UC Santa Barbara

Himalayan Linguistics

Title

A descriptive grammar of Denjongke [HL Archive 10]

Permalink

https://escholarship.org/uc/item/6xs3r33s

Journal Himalayan Linguistics, 0(0)

Author Yliniemi, Juha Sakari

Publication Date 2021-07-31

DOI 10.5070/H920146466

Copyright Information

Copyright 2021 by the author(s). This work is made available under the terms of a Creative Commons Attribution-NonCommercial-NoDerivatives License, available at https://creativecommons.org/licenses/by-nc-nd/4.0/

Peer reviewed



A free refereed web journal and archive devoted to the study of the languages of the Himalayas

A Descriptive Grammar of Denjongke (Sikkimese Bhutia)

Juha Yliniemi

SIL International

Himalayan Linguistics Archive 10

Himalayan Linguistics Archive No. 10 i-xxx, 1-687 ASSN 1544-7502 © 2021. All rights reserved

Abstract

This work is a descriptive grammar of Denjongke, or Sikkimese Bhutia (also known as Lhoke or Sikkimese) (ISO 639-3 sip), an underdescribed and endangered Tibeto-Burman, Tibetic language spoken in the Indian state of Sikkim. The study is based on original fieldwork conducted over more than six years. The theoretical framework is functionalist-typological and may further be characterized as an application of Basic Linguistic Theory, which relies on the power of prose, instead of formalisms, to describe linguistic phenomena. Traditional grammatical terms are complemented by recourse to up-to-date typological information. The discussion is data-oriented and aims to describe Denjongke on its own terms, making a distinction between language-internal descriptive categories and cross-linguistic comparative concepts.

Denjongke has 43 consonants and eight vowels (if long vowels are not counted separately). Nasalization and length are contrastive in vowels. Words are phonologically divided into high register, which is associated with high pitch and modal/stiff voice, and low register, which is associated with low pitch and breathy voice. The register of a word is partly unpredictable from the initial consonant, so Denjongke is a tone language, although tone does not bear as great a functional load as in many well-known tone languages such as Cantonese or Vietnamese.

The present analysis establishes four major word classes (nouns, verbs, adjectives and adverbs) and eleven minor word classes. Many often-used nouns and verbs can be divided into ordinary and honorific register so that the same ohject or action may be referred to by two different lexical items. Similarly to many other Tibeto-Burman languages, Denjongke has phonologically related pairs of controllable vs. non-controllable verbs. The class of adjectives, although deriving from stative verbs, is morphologically distinguished from other word classes by a number of adjectivizing derivative suffixes. Numerals follow both decimal and vigesimal systems. Unlike many Tibetic languages, Denjongke does not make a clusivity distinction in first person plural pronouns. There are five case-marking enclitics some of which may be stacked. Within demonstratives, an interesting feature is the existence of an emphatically demonstrative proximal form (*dodi*) in addition to the ordinary proximal (*di*) and distal (*odi*). Proximal *di* has grammaticalized into a non-referential marker =*di* which may attach to proper names and other demonstratives. Among the twenty-one clitics, the behaviour of the attention marker =*co* provides interesting insight into phenomena surrounding the concepts of contrastive focus and mirativity.

Because the marking of A argument and P argument is to a considerable degree conditioned by pragmatics and lexical choices, categorizing Denjongke as either nominative-accusative or absolutive-ergative language is not feasible. Intransitive clauses may be agentive marked for emphatic purposes, while the marking of A argument in transitive clauses shows signs of both syntactic control (some verbs require agentive marking in the past tense) and pragmatic control (A arguments with and without agentive marking are offered in elicitation). The marking of P argument is sensitive to animacy, identifiability/specificity and affectedness.

Denjongke has a particularly wide array of copula forms, which mark three evidential values: personal, sensorial and neutral. The present study shows that the semantically oriented category "personal" differs from the more syntactically-oriented Lhasa Tibetan category "egophoric". It also suggests how egophoricity may have developed in Tibetic languages. Among copulas, an interesting detail is that the sensorial du^2 , which typically functions as an existential, can be used as an equative if the proposition describes something that held in the past.

Relative clauses are a subclass of constituent-modifying clauses, which modify a clausal constituent by a nominalized and genetivized clause. Probably as an influence from Indo-Aryan languages, Denjongke also uses correlative clauses, in which a clause with a question word is

linked to another clause with a resumptive demonstrative, which is a coreferential with the question word. Due to Denjongke being a clause-chaining language in which one sentence typically has only one finite verb, the concept of coordination proves elusive on clausal level. Denjongke has a wide variety of adverbial clauses, which are expressed through various constructions, including ten converbs. Simultaneity (expressed with seven constructions) and causality (expressed with eight constructions) are the functions which show the greatest constructional variety

Two rarely described categories of vocabulary which are addressed in this grammar are ideophones and terms adults use when talking to children. Denjongke ideophones are a semantically, phonologically and morphologically distinct set of words which provide vivid descriptions of sensory experiences. Nonnormative ideophones associate a sequence of phonemes (C1aC2.C1oC2) with nonnormativity (e.g. walking as/like a drunkard instead of walking normally). An investigation into the terminology used when talking to small children demonstrates that adults show empathy to children with respect to children's enunciatory ability and life-experiences.

iii

बे'झून'ग्री' बे'र्बेल' वृत्त'क्षर' बे' ऑन्'र्ये'		નેવ વને વ વાસન છે જેવા સુ વન લગ્ન દો બેન વે
ศ⁄ж [kʰo]/[mò]	>	رج·/عِ: [kʰu]/[mù]
ਬੁੱਕਾਧੋਂ [,] [bompo]	>	ਬੁੱਕਾਦਾ [bompu]
٦٠؏٠ [t`ata]	>	ڗۥٚۼۣٚ؞ [tʿato]
र्ळ्यायान्ता [tsʰuːna]	>	र्ह्तयाबा दे. [tshu:ne] / र्ह्तयाबा र्दे [tshu:no]
गेष'र्य' [ke:po]	>	गोषाय [ke:p] / गोषा दी [ke:po]
གནང་བོ་ སྒང་ [náŋbo bɛʔ]	>	गवत्तवः भ्रतः [na:m be?]/ गवतःरां भ्रतः [na:bo be?]
aहेंग'र्ळर'र्ने'यया [dzɛktsʰarbolɛ]	>	ત≓્વાૠઁતુ.વાય. [qsktsponts]
ર્ફ્રોંદ્ર ર્ಹર ર્વે બન્ન [mjòŋtsʰarbolɛ]	>	ڲٚڷڗۥؘۿٚؗؗۻٚ؞ؚڡٳ؆ [mjồ̀:tsʰoːlɛ]
वि'र्ळ्न'र्ने'लष' [cítsʰarbolɛ]	>	ર્ન ર્સુત્ર ભષ્પ [cítsublɛ]
ฐ์พ [.] พัศ [.] รัฐรา [pʿjasõːbodãː]	>	ฮ์สาสักสากกา [pʿjasomdãː]
ấଷເຮົາ໗? [tʰoːtɛɛka]	>	ইন্স'ত? [tʰoːtɛa]
र्जुषा प्रतिव : तत्या गो : [p jazenduke]	>	र्जुष्ग प्तवित्र गो [pʻjazyŋgɛ] (Tashiding)

यात्र ब्रेन्, टेन्रुप, क्र्या, झैंट, ग्री. जटी- 1822 का पट, 12. संट. क्रुंच, ज्यू, जीह्य, वर्ष क्रिंग, व्राक्षेय, व्या, ज्यू, जुंथ, व्या, ब्रेंग, ब्री. वर्ष, ज्यू, जुंथ, जुंथ, जुंथ, ब्री. वर्ष, क्रुंच, भ्री. जुंध, ज जुंध, जुं

મત્ત ચુયા રુ ચેનુ આપતુ તેન તેને તે આ સુંત્ર નું સુંતે અને સુંત્ર તે સુંત્ર સુંતર તે આ સુંત્ર સું સું સું સુંત્ર સું સું સુંત્ર સુંત્ર સુંત્ર સું સું સુંત્ર સુંત્ર સુંત્ર સુંત્ર સુંત્ર સુંત્ર સુંત્ર સુંત્ર સુંત્ર સું સુંત્ર સુંત્ય સુંત્ર સુંત્ય સુંત્ય સુંત્ય સુંત્ સુંત્ર સુંત્ર સુંત્ય સુંત્ય સુંત્ય સુંત્ય સુંત્ય સું સુંત્ર સું સુંત્ય સુંત્ય સુંત્ય સુંત્ય સુંત્ય સુંત્ય સુંત્ય સુંત્ય સું સુંત્ય સુંત્ય સુંત્ય સુંત્ય સુંત્ય સું સુંત્ય સ

સુ.ર. જીટ. યુટે. છે. છે. (Juha Yliniemi)

नेया वन्तिः मेंत्र मेंग विश्वसाविया ईश्वार्या मृत्ता विव्याग्नीः श्रुपावच्च चित्रवे श्रूता प्रत्ये र्स्यायाय्या र्झ्या क्षेत्र मुत्रा विश्वसाव्य विद्या हैं स्वयाया विद्या स्वयाया क्या में स्वयाया स्वयाया स्वयाया स्वयाया क्या स्वर्या योग्रेया विश्वसाव्य विद्याया प्रत्याया विद्यायाया क्या स्वयाया स्वयाया स्वयाया स्वयाया स्वयाया क्याया क

તેન. તરી. ક્રે. તદ્યાસી શૈર-ત્રી. છા. વગ્યા રૂટે છે. તેને તેને સે ક્રે. અંગ આવવે. ક્રે. દ્વ. ક્રે. તરી. ખૂન. સૈંગોગ. જૂને. બુને. જેવો. તર્જવો પૂર્ણ, દ્વ. રુ. છુ, બુને. જૂની તે. બંન. ક્રે. તરી. રુ. જૂની

Acknowledgements

It is without question that completing a work as extensive as this grammar would not have been possible without help from a great number of people from the language community, academia and my circle of friends and relatives. I would first like to thank my very first consultants Khichung Bhutia (Tadungchen), Sonam Gyatso Lachungpa (Lachung), Rabden Bhutia (Tashiding) and Tobjor Bhutia (Ralang) with whom I started learning Denjongke phonology in 2004. With Rabden I also studied for a week in Warsaw in 2009. Tobjor, on the other hand, was my first consultant upon my resuming Denjongke studies in 2011 and has continued to work with me until the present day. I would also like to thank Padma Shri Norden Tshering Bhutia and Shri Pema Rinzing Takchungdarpo, the key figures in developing literary Bhutia, for endorsing this project. Shri Pema Rinzing also provided me with language recordings and gave a lot of helpful feedback on the spelling of example sentences.

My first stint in a village took place in 2012 in Tashiding, West Sikkim. I owe a great debt of gratitude to many wonderful people there with whom I continued to work over the years. I would especially like to thank Phurba Tshering and his family, Dauge and his family (especially Samten and Namdol Gyatso), Gyaltshen and his family, Pempa Doma, Dawa Namgyal, Jampal Dorje and his family, Ugyen Tshering and family, Ugyen Thrinley, Sonam Gongphel, Rinzing Chodra, RS Sir and family, Neydup Sir and Namdol Sir (President of Bhutia Literary Association).

My second village stint (in 2013) took place in Upper Martam (East Sikkim), where I was helped by a great number of people. On a practical level, thanks are due to Karma BB Bhutia for providing me and my family accommodation. I'm grateful to Khempu Karma Tshering and retired Bhutia teachers Tashi Lopen, Dorje Lopen and Rinzing Lopen, for teaching me their language. In Martam I also met Karma Namgyal, who became a most faithful consultant until the present day. Thanks to him for securing long-awaited feedback from Sikkim at the last minute before an earlier version of this grammar was submitted as a PhD thesis.

My heartfelt thanks are also due to Yeshe Rinzing Bhutia (Bhutia language Text Book Officer at the time of my research) for hours of recording and discussion in Ranka and Gangtok, and also for providing feedback on the spelling of example sentences in this grammar. Moreover, I was priviledged to work extensively with Karma Lobsang Bhutia (Bhutia teacher at West Point Senior Secondary School), who also arranged a wonderful research trip to Barapathing and whose book was used as data in the present work. I am greatly indebted to KT Sir of Wok for extravagant hospitality and help in my studies of Lhoke until the present day.

I would also like to thank Captain Yongda for an unforgettable interview session in Pemayantse, Sonam Gyatso Dokhangba for recording sessions and help in spell-checking, Rinzing Chewang Kazi of Lachen Post Office for great fellowship and recording sessions, and Chewang Genje and Wanchuk for helping me make recordings of Lachung variety of Lhoke. At Ranka, I was priviledged to learn Bhutia from many people among whom were Kempu Tobden, Tashi Tshering Bhutia and Ranka Senior Secondary School teachers Sonam Sir, Pema Sir and Sange Sir. My family felt priviledged to have such generous landlords as Pema Dadul, Pasang Lhamu and family, who also helped as occasional language consultants.

I am grateful to Bhaichung Tshering Bhutia, HoD of the Department of Bhutia, Sikkim University, for writing the works which were used as the main written data for this study and also for making his own and other books available to me. I thank Dr. Kunzang Namgyal both for checking the example sentences for some 200 pages of a manuscript of this work and for providing insightful comments that greatly improved the analysis. Thanks also to him for patiently answering my questions via Whatsapp! I am also thankful for Ms Sonam Doma Bhutia, who typed several written sources on computer and thus enabled computer-analysis of data.

In the academia, I would like to thank my first supervisor Professor Fred Karlsson who made contact with Sikkim University, thereby initiating my research visa process. I am also grateful to my main supervisor Professor Matti Miestamo for ever helpful feedback on my writing as well as numerous hours of discussion on linguistic topics related to my PhD thesis on Denjongke grammar. I would like to extend thanks to Sikkim University in general and particularly to my advisor Dr. Samar Sinha for fluent cooperation and stimulating conversations on linguistics and other matters. Thanks are also due to Lotta Aunio for providing useful comments on phonology; Bertil Tikkanen for providing feedback on parts of the manuscript and Kaius Sinnemäki for commenting on my writing and for many enlightening discussions on linguistics and more important matters. My colleagues Erik Andvik, Brad and Wendy Chamberlain, Anette Helgestadt-Tombleson and Steve and Jean Watters helped keep my linguistic candle (as well as other candles) burning, hopefully at just the one end!

Many scholars gave feedback on article drafts concerning topics which were included in this grammar. They are, in alphabetical order, Alexandra Aikhenvald, Erik Andvik, Ellen Bartee, Lon Diehl, Scott DeLancey, Jonathan Evans, Nathan W. Hill, Hella Olbertz, Kaius Sinnemäki. I also thank external examiners Nicolas Tournadre and Carol Genetti both for encouragement and constructive criticisms which improved the work.

Thanks to proof-reading by Hazel Sorsamo, many sentences in the first chapter are now clearer. Don Killian and Chris Wilde helped me to access relevant articles and books. Chris Wilde also provided crucial help in map-making. Stephen Kelly and Esteban Witzemann provided invaluable computer support, which saved a lot of time. Siiri Yliniemi helped me check the cross-references within the grammar. Many thanks are due to You-Jing Lin for hours of laborious editing, which helped the present work become publishable through *Himalayan Linguistics*.

I am grateful to The Finnish Cultural Foundation (Suomen Kulttuurirahasto) for a one-year research grant for this project. I also thank friends in Sodankylä, Rovaniemi, Tornio, Oulu, Kempele, Laukaa, Jyväskylä, Riihimäki and Helsinki for all types of support over the years. Thanks is also due to Richard Brewis for encouragement and support right from the outset.

This grammar was very much a family effort and as such would most likely have been impossible without crucial help from (in chronological order) Ruut Uusitalo, Heidi Örtengren, Jenna Honkanen, Hanna Kuparinen, Inkeri Laurila, Elisa Nykänen, Johanna Lusenius and Jenny Kandén. I'm grateful to Tuula and the late Viljo Paananen for their help on site in Sikkim. My mother Anne Ludwig is to be admired for her outwardly stoic acquiescence at being robbed of her only grandchildren for considerable periods of time.

The success of this grammar-writing process was certainly helped by an outstanding home team made up of world-class fellow-adventurers Siiri, Sara, Lyydia and Joose, who kept on keeping our spirits up and taking advantage of things I struggled to enjoy (e.g. leeches, leaking roofs, flooded roof-terraces and power-cuts). The role of my wife Mari in this project and my life in general could only be expressed in poetry too intimate and lengthy to be printed here. Therefore I condense my gratitude to one word: she was, and is, "special," a British understatement for which I claim all its most wonderful senses.

I take full responsibility for all the remaining errors. I would also like to thank, in advance, those who care enough to read this grammar and point out its mistakes and inadequacies. Blessed are the true fault-finders, for they shall be called midwives of truth.

Lastly, thanks to my Creator and Redeemer, who always knew exactly what on earth was going on in Denjongke, for letting me enjoy the process of taking a brief glance at it!

Brief Table of contents

A	bstract त्र'नईेट्र'	(Foreword in Denjongke)	i iii
A	cknow	ledgements	v
В	rief Ta	ble of contents	vii
F	ull Tab	le of Contents	xi
L	ist of '	Tables	XXV
L	ist of F	igures	xxviii
L	ist of N	ſaps	xxix
A	bbrevi	ations	XXX
1	Intr	oduction	1
	1.1	The language and the people	1
	1.2	Research approach and data	
2	Pho	nology	
	2.1	Introduction	
	2.2	Consonants	
	2.3	Vowels	
		Syllable structure	
	2.5	Segmental phonological processes	
	2.6	Register, pitch and tone	
	2.7	Stress or tone	65
	2.8	Morphophonology	
	2.9	Phonemic script	
	2.10	Summary remarks	71
3	Wo	rd classes, suffixes and clitics	72
	31	Words affixes and clitics	77
	3.1	Nouns	
	3.2	Verbs and verbal affixes	
	3.4	Adjectives	104
	3.5	Adverbs	112
	3.6	Minor word classes	
	3.7	Clitics	
	3.8	Summary remarks	147
4	Phr	asal constituents	
		Noue shares	140
	4.1	Noun phrase	
	4.Z	A diactive and advert phrases	
	4.3 1 1	Aujective and advero phrases	
	4.4 15	Summers remerks	
	4.)	Summary remarks	

5 Ba	sic clause structure	
5.1	Introduction to concepts and terminology	
5.2	Intransitive clauses	
5.3	Transitive clauses	
5.4	Clauses with verbs of being and becoming	
5.5	Valency modification	
5.6	Adverbial modification	
5.7	Summary remarks	
		050
6 De	eixis and reference	
6.1	Personal pronouns	
6.2	Reflexives and reciprocals	
6.3	Indefinite reference	
6.4	Demonstratives	
6.5	Summary remarks	
7 Co	spulas and evidentiality in copulas	272
7.1	Definition of evidentiality	
7.2	Simple copulas	
7.3	Complex copulas	
7.4	Simple copulas compared with some other Tibetic languages	
7.5	Summary remarks	
9 Та	non association dentity	215
8 Ie	nse, aspect and modality	
8.1	Past, completive and perfect forms	
8.2	Present habitual and future forms	
8.3	Imperfective, progressive and continuous forms	
8.4	Tense, aspect and modality with the construction VERB-INF EX	
8.5	Modality (probability, possibility and necessity)	
8.6	Summary remarks	
9 Ev	identiality marked by auxiliaries and clitics	
0.1	Evidentiality montrod by available -	205
9.1	Evidentiality marked by auxiliaries	
9.2	Evidentiality marked by clitics	
9.3	Summary remarks	
10	Negation	
10.1	Clausal negation	410
10.1	Non-clausal negation	
10.2	Other aspects of negation	лэсан таранан таранан таранан тарактан тарактан тарактан тарактан тарактан тарактан тарактан тарактан тарактан Лэс тарактан
10.3	Summary remarks	
10.4	Summary Temarks	
11	Non-declarative clauses	

11.1	Interrogatives	
11.2	Exclamatives	
11.3	Imperative	
11.4	Hortative - <i>kε/gε</i>	
11.5	Optative with <i>tcu?</i>	
11.6	Summary remarks	
12 0	Connecting finite clauses	
12.1	Introduction	
12.2	Monosyndetic connectors	
12.3	Bisyndetic connectors	
12.4	Summary remarks	
13 (Constituent-modifying clauses	489
15		
13.1	Introduction	
13.2	Relative clauses	
13.3	Correlative clauses	
13.4	Noun complement clauses	
13.5	Postposition complement clauses	
13.6	Summary remarks	
14 (Complement clauses	515
14.1	Non-finite complement clauses	
14.2	Finite (clause-like) complement clauses	
14.3	Summary remarks	
15	Adverbial clauses	
15 1	Introduction to forms	528
15.1	Nonfinal _ti/di	
15.2	Temporal clauses	534
15.5	Causal clauses	548
15.1	Purnosive clauses	556
15.6	Conditional clauses	560
15.0	Concessive clauses	564
15.7	Clauses of circumstance and manner	566
15.0	Additive clauses	577
15.10	Substitutive clauses with ts^hanlo	579
15.10	Comparative clauses with $=l_{\epsilon}(l_{ako}=\text{FMPH})$	579
15.12	Various functions of the terminative converts $-s\tilde{a}$ and $-son z\tilde{a}$	580
15.13	Summary remarks	
1.6		
16 I	Discourse phenomena	
16.1	Emphatic clitics	
16.2	Clause-final cliffics	
16.3	Assertive and exclamative tags	

16.4	Discourse particles $t'a$ and $t\varepsilon$	
16.5	Recapitulation	
16.6	Right-dislocation	
16.7	A note on intonation and discourse	
16.8	Summary remarks	

17 I	Notes on lexicon	617
17.1	Ideophones	
17.2	Kinship terminology	
17.3	Names	
17.4	Colours	
17.5	Vocabulary used with small children	
17.6	Summary remarks	

Appendix 1: Text excerpts	649
Appendix 2: Differences between spoken and written language	670
Appendix 3: Vowel plot measurements	673
Appendix 4: Letter of informed consent	675
References	677

Full Table of Contents

Abstract	i
జైశానకైన్ (Foreword in Denjongke)	iii
Acknowledgements	v
Brief Table of contents	vii
Full Table of Contents	xi
List of Tables	xxv
List of Figures	xxviii
List of Maps	xxix
Abbreviations	xxix
1 Introduction	1
1.1 The language and the people	1
1.1.1 Name of the language	1
1.1.2 Number of speakers	6
1.1.3 Genetic affiliation	7
1.1.4 Previous research	9
1.1.5 Origins of the people	
1.1.6 From spoken to written language	11
1.1.7 Religion	11
1.1.8 The honorific system and social relations	
1.1.9 Language contact and multilinguality	
1.1.10 Language endangerment	
1.1.11 Central linguistic features	14
1.2 Research approach and data	16
1.2.1 Dialects and the language described here	16
1.2.2 Theoretical background	
1.2.3 Data	
1.2.3.1 Oral data and consultants	
1.2.3.2 Literary data	
1.2.4 Methodology and software	
1.2.5 Equipment	
1.2.6 Linguistic examples	
1.2.7 Written Denjongke	
1.2.7.1 Introduction to written Denjongke	
1.2.7.2 Word breaks	
1.2.7.3 Syllable merging	

	1	.2.7.4	4 Phonological writing	29
	1	.2.7.5	5 Differences between spoken and written language	30
	1.2	.8	Transliteration	30
	1.2	.9	Organization of the grammar	30
2	Pho	onolo	gy	32
	2.1	Intro	oduction	32
	2.2	Con	sonants	32
	2.2	.1	Phonetic descriptions and contrastive sets for consonant phonemes	33
	2	.2.1.	Plosives and affricates	33
	2	.2.1.2	2 Fricatives and central approximants	39
	2	.2.1.3	3 Nasals	41
	2	.2.1.4	4 Liquids	42
	2.2	.2	The lightly aspirated "breathy" consonants	43
	2.3	Vov	vels	46
	2.3	.1	Phonetic descriptions and contrastive sets for vowel qualities	47
	2.3	.2	Length	50
	2.3	.3	Diphthongs	52
	2.3	.4	Nasalization	53
	2.4	Syll	able structure	54
	2.5	Seg	mental phonological processes	55
	2.5	.1	Vowel assimilation	55
	2.5	.2	Elision	56
	2.5	.3	Consonant lenition	56
	2.6	Reg	ister, pitch and tone	58
	2.6	.1	Monosyllables	59
	2	.6.1.	1 Register-internal pitch differences	60
	2	.6.1.2	2 Final glottal and pitch	62
	2.6	.2	Disyllables	63
	2.7	Stre	ss or tone	65
	2.0		Markal and firms	00
	2.8	.1 ว	Verbal sumxes	00
	2.8	.2	Regarding prefixes	08
	2.8	.5	A continue coop	09
	2.8	.4 DL -	Agenuve case	09
	2.9 2.10	Sum	mary remarks	70
3	Wo	rd cl	asses, suffixes and clitics	72
-	3.1	Woi	ds, affixes and clitics	72
			· · · · · · · · · · · · · · · · · · ·	

3.2 No	uns	73
3.2.1	Introduction to nouns	73
3.2.2	Ordinary and honorific nouns	74
3.2.3	Deriving nouns from verbs	76
3.2.4	Nominal suffixes	77
3.2.4	.1 Nominal suffix <i>-po/bo</i>	77
3.2.4	.2 Nominal suffix $-m(u)$	79
3.2.4	.3 Nominalizer $-k^{h}\tilde{\epsilon}$:	80
3.2.4	.4 Nominalizer - <i>sa</i>	
3.2.4	.5 Nominalizer - <i>tã</i> :	
3.2.5	Compounding	
3.3 Ve	rbs and verbal affixes	
3.3.1	Defining criteria for verbs	
3.3.2	Stative and dynamic verbs	
3.3.3	Controllable and non-controllable verbs	
3.3.4	Ordinary, honorific and humilific verbs	
3.3.5	Reduplication	89
3.3.6	Verbal affixes	
3.3.6	.1 Infinitives -εε?, -po/bo and -ni	91
3.3.6	5.2 Nominalizer $-k^{h}\tilde{\epsilon}$:	
3.3.6	5.3 Completive - <i>ts^ha(:</i>)	
3.3.6	.4 Imperfective - <i>to/do</i>	
3.3.6	5.5 Progressive $-t\varepsilon\tilde{\varepsilon}:/z\tilde{\varepsilon}:/zin$ (also $-t\varepsilon ou/zou$)	
3.3.6	.6 Imminent future marker - <i>rap</i>	95
3.3.6	.7 Past $-t \epsilon \epsilon / z \epsilon$	95
3.3.6	8.8 Probabilitative -to	95
3.3.6	5.9 Imperative suffixes $-tc^{h}i$, $-da$ and $-na$	96
3.3.6	5.10 Hortative $-k\varepsilon/g\varepsilon$	
3.3.6	.11 Interrogative suffixes -ka/ga and -kam/gam	97
3.3.6	5.12 Exclamative $-l\phi(2)$	
3.3.6	Nonfinal converb - <i>ti/di</i>	
3.3.6	.14 Circumstantial-purposive converb - <i>pa/ba</i>	
3.3.6	6.15 Conditional converb - $(patee)ne/(batee)ne$	
3.3.6	.16 Concessive converb - <i>ruŋ</i>	
3.3.6	.17 Terminative converbs <i>-sã</i> : and <i>-sonzã</i> :	
3.3.6	5.18 Simultaneous converb - <i>sondã:/somdã:/sumdã:/tsubda:</i>	
3.3.6	5.19 Simultaneity markers - <i>kap</i> , $-d\tilde{y}$: and $-r\varepsilon\eta k^ha$	

3.3.7	Copulas	
3.4 Adj	ectives	
3.4.1	Defining criteria for adjectives	
3.4.2	Adjective-forming suffixes	106
3.4.3	Adjective-modifying suffixes	
3.5 Adv	erbs	
3.5.1	Introduction to adverbs	
3.5.2	Adverb derivation and types of adverbs	
3.5.2.	Manner adverbs	
3.5.2.2	2 Locative adverbs	
3.5.2.3	3 Temporal adverbs	116
3.5.2.4	4 Quantifying adverbs	117
3.5.2.	5 Other adverbs	119
3.5.2.0	5 Directional and approximative suffix -tcika '-abouts, around'	119
3.6 Min	or word classes	
3.6.1	Personal pronouns	
3.6.2	Reflexive pronouns	
3.6.3	Reciprocal pronouns	
3.6.4	Indefinite pronouns	
3.6.5	Demonstratives	
3.6.6	Question words	
3.6.7	Numerals	
3.6.8	Postpositions	
3.6.9	Discourse connectives	
3.6.10	Interjections	
3.6.11	Discourse particles	
3.7 Clit	ics	
3.7.1	Case clitics	
3.7.1.	I Introduction to case clitics	
3.7.1.2	Postpositions or cases: $=s\tilde{a}$: 'until' and $=tsa(:)$ 'at, by, with'	141
3.7.1.3	3 Case-stacking	
3.7.2	Emphatic clitics	
3.7.3	Clausal clitics	144
3.7.4	Other clitics	145
3.7.4.	Plural clitic = <i>tsu</i>	
3.7.4.2	2 Indefiniteness clitic =tei?	145
3.7.4.3	B Honorific clitics $= la(:)$ and $= tc^{h}o$:	146

3.7.4.4 Urgetive clitic $=m\phi^2$	
3.8 Summary remarks	
4 Phrasal constituents	
4.1 Noun phrase	
4.1.1 Introduction to noun phrase	
4.1.2 Prenominal modifiers	
4.1.2.1 Demonstrative	
4.1.2.2 Genitive attributes	
4.1.2.3 Relative clause with $-k^{h}\tilde{\epsilon}$	
4.1.2.4 Juxtaposition	
4.1.2.4.1 Appositional juxtaposition	
4.1.2.4.2 Non-appositional juxtaposition	
4.1.3 Postnominal modifiers	
4.1.3.1 Adjectival modifiers	
4.1.3.2 Numeral modifiers	
4.1.3.3 Quantifying modifiers	
4.1.3.4 Demonstrative modifiers	
4.1.3.5 Post-head relative clauses as modifiers	
4.1.4 Noun-phrase-final clitics	
4.1.5 Plural marker $=$ <i>tsu</i>	
4.1.6 Definiteness and indefiniteness	
4.1.7 Coordination of noun phrases	
4.1.8 Reduplication of noun phrase	
4.2 The verb complex	
4.2.1 Structure of the verb complex	
4.2.2 Phrasal verbs	
4.2.3 Verb serialization	
4.2.3.1 Translocative secondary verb gju (past $s\tilde{o}$)	
4.2.3.2 Benefactive secondary verb <i>p</i> ' <i>in</i> 'give'	
4.2.3.3 Beneficiary secondary verb <i>t</i> ^h op 'find, receive'	
4.2.3.4 Secondary verb $t\tilde{a}$: 'send'	
4.2.4 Summarizing examples of the structure of the verb cor	nplex 183
4.3 Adjective and adverb phrases	
4.3.1 Quantifying adverb as modifier	
4.3.2 Nominalized clause as modifier	
4.3.3 Phrasal adjectives	
4.3.4 Reduplication	

	4.	3.5	Adverb phrase	192
	4.4	Nun	neral phrase	192
	4.5	Sum	mary remarks	195
5	B	asic cla	ause structure	196
	5.1	Intro	oduction to concepts and terminology	196
	5.2	Intra	ansitive clauses	199
	5.5	11ai 2 1	A roumont A	201
	5. 5	2.1 2.2	Argument R	203
	5.	3.2 3.3	Ditransitive clauses (T and P argument)	207
	5.1	J.J Clar	uses with verbs of being and becoming	212
	5.4	/ 1	Equative along	215
	5.	4.1 1 2	Existential and locative clauses	215
	5.	4.2 1 3	Predicative possession	210
	5.	ч.5 ДД	Predicate adjectives	210
	5.	4.5	Clauses of becoming and happening	220
	5.5	Vale	ency modification	223
	5	5.1	Argument suppression	223
	5.	5.2	Causative	224
	5.6	Adv	rerbial modification	225
	5.	6.1	Case-marked nouns as adverbials	226
		5.6.1.1	1 Dative-locative $= lo$	226
		5.6.1.2	2 Locative = na	229
		5.6.1.3	3 Ablative $=l\varepsilon$	229
		5.6.	1.3.1 Spatio-temporal uses	230
		5.6.	1.3.2 Comparative uses	231
	5.	6.2	Postposition phrases as adverbials	231
	5.	6.3	Adverbs as adverbials	242
		5.6.3.1	Adverbs of manner	242
		5.6.3.2	2 Locative adverbs	243
		5.6.3.3	3 Temporal adverbs	244
		5.6.3.4	Verb-modifying quantitative adverbs	246
		5.6.3.5	5 Other adverbs	249
	5.7	Sum	mary remarks	250
6	D	eixis a	nd reference	252
	6.1	Pers	onal pronouns	252
	6.2	Refl	exives and reciprocals	256
	0.3	mae		238

	6.3	3.1	Indefinite pronouns	258
	6.3	3.2	Indefinite reference with question words	261
	6.4	Den	nonstratives	265
	6.5	Sum	mary remarks	270
7	Co	pulas	and evidentiality in copulas	272
	7.1	Defi	inition of evidentiality	272
	7.2 7 0) 1	Personal copulas	275
	<i>1.2</i>	2.1 7 2 1 °	Personal copulas	275
	,	7.2.1. 7.2.1.	2 Personal existential $i\dot{a}^2$	275
	7 0))))	Sensorial conula du^2	270
	, 1.2	 	Sensorial copula uu	201
	,	7.2.2. 7.2.2	2 Past equative uses of du^2	281
	,	7 7 7 ?	$\frac{1}{2} \text{Intensifier } kc$	205
	7 0	7.2.2 7 3	Equative neutral bs^2 in comparison with other copulas	280
	7.2 7.2	2.5 2 A	Apparentive equative de/re	207
	7.2 7.2	2.4	Other forms used as copulas	292
	, 1.2		The work \hat{x} (some ' or evistor ticl	202
		7.2.3.	The verb of come as existential	293
	70	1.2.5.2	2 Reportative $=lo$ as equative substitute	294 204
	1.3	Con		296
	7.3	3.1	Combinatory copulas <i>imbeP</i> and <i>induP</i>	296
	7.3	3.2 7.2 a î	Nominalized copula constructions	298 200
	,	1.3.2.	Evidentially neutral constructions (ending in $b\epsilon P$)	299
		7.3.	2.1.1 Equative constructions	299
		7.3.	2.1.2 Existential constructions	301
	,	7.3.2.2	2 Personal constructions (ending in \tilde{t})	305
		7.3.	2.2.1 Equative constructions	305
		7.3.	2.2.2 Existential constructions	308
	7.4 7.5	Sim Sun	ple copulas compared with some other Tibetic languages	310313
8	Те	nse, a	spect and modality	315
	8.1	Past	, completive and perfect forms	315
	8.1	1.1	Past forms	315
	8.1	1.2	Completive	320
	8.1	1.3	Secondary verb $mj\tilde{o}$: 'finish, experience'	322
	8.1	l.4	Perfect	323
	8.1	1.5	Resultative	326

8.1.6	Sensorial resultative/past	
8.1.7	Resultative secondary verb zak	
8.1.8	Iterative past	
8.1.8.1	Iterative with -kjā:	
8.1.8.2	Iterative with <i>-tcim</i>	
8.1.9	Note on the Sandberg's (1895) past forms	
8.2 Prese	ent habitual and future forms	
8.2.1	Steady state present	
8.2.2	Simple present	
8.2.3	Present habitual I	
8.2.4	Present habitual II	
8.2.5	Nonpast	
8.2.6	Future with secondary verb $\dot{\vec{o}}$: 'come'	
8.2.7	Imminent future	
8.2.8	Tense, aspect and modality with the infinitive - <i>ni</i>	
8.2.8.1	Uncertain future	
8.2.8.2	Inability	
8.2.8.3	Unrealized planned activity	
8.2.8.4	Future-oriented questions	
8.2.8.5	Request and suggestion	
8.2.8.6	Future conditional	
8.2.8.7	Other uses of - <i>pi</i>	
8.3 Impe	erfective, progressive and continuous forms	
8.3.1	Imperfective	
8.3.2	Continuous	
8.3.3	Progressive VERB- $t_{\varepsilon}\tilde{\varepsilon}:/z\tilde{\varepsilon}:/zin$ EX and VERB- $t_{\varepsilon}u\eta g\varepsilon/zu\eta g\varepsilon$	
8.3.3.1	Progressive VERB- <i>tcɛ̃:/zɛ̃:/zin</i> EX	
8.3.3.2	Alterphoric progressive VERB- <i>tsuŋgɛ/zuŋgɛ</i>	
8.3.4	Durative secondary verb bak	
8.4 Tens8.5 Mod	e, aspect and modality with the construction VERB-INF EXality (probability, possibility and necessity)	361 363
8.5.1	Probabilitative -to	
8.5.2	Apparentive <i>da</i>	
8.5.3	Approximatives $da:gju$ and $t^{h} \varepsilon:gju$	
8.5.4	Permissives <i>tc^ho?</i> and <i>tup</i>	
8.5.5	Temporal evaluative <i>rɛn</i> 'be time to'	
8.5.6	Moral and practical evaluative (mi-)le? 'be (not) good'	

	8.5.7	General abilitatives <i>ts^hu?</i> 'be able to' and <i>cè</i> .' 'know'	
	8.5.8	Mental abilitative <i>nùm</i> 'dare, have courage to'	
	8.5.9	Physical abilitative $k^h \phi$? 'to have the strength to'	
	8.5.10	Temporal abilitative $l\hat{o}$: 'have time to'	
	8.5.11	Deontic modal go? 'be needed, must'	
	8.5.12	Possibility with <i>si?</i> 'be possible'	
	8.5.13	Tentativity with <i>ta</i> 'look'	
	8.5.14	Pretensive - <i>dzin t</i> ^h ap	
8	8.6 Sum	mary remarks	
9	Evidenti	ality marked by auxiliaries and clitics	
Ģ	9.1 Evic	lentiality marked by auxiliaries	
	9.1.1	Equatives \tilde{t} and $b\varepsilon$? as auxiliaries	
	9.1.1.1	Personal forms with 1st person actor	
	9.1.1.2	2 Personal forms with non-1st person actor	
	9.1.1.3	3 Neutral forms with non-1st person actor	
	9.1.1.4	4 Neutral forms with 1st person actor:	
	9.1.1.5	5 Evidentiality with complex equative auxiliaries	
	9.1.2	Existentials <i>j</i> @? and <i>du</i> ? as auxiliaries	
	9.1.2.1	Personal forms with 1st person actor	393
	9.1.2.2	2 Personal forms with non-1st person actor	
	9.1.2.3	3 Sensorial and neutral forms with non-1st person actor	
	9.1.2.4	4 Sensorial and neutral forms with 1st person actor	
	9.1.2.5	5 Evidentiality with complex existential auxiliaries	
	9.1.3	Evidentiality with the completive <i>-ts^ha(:)</i>	
Ç	9.2 Evic	lentiality marked by clitics	
	9.2.1	Reportative =lo	
	9.2.1.1	Declarative use	
	9.2.1.2	2 Interrogative use	
	9.2.1.3	3 Imperative use	
	9.2.2	Quotative $=s(\varepsilon)$	
Ç	9.3 Sum	mary remarks	
10	Negat	ion	
1	10.1 Clau	isal negation	
	10.1.1	Standard negation	
	10.1.2	Negation in copular clauses	
	10.1.3	Negation in non-declarative clauses	
	10.1.3	.1 Negation in interrogative, imperative, hortative and optative	

10.1.3	8.2 Negation of interrogative copulas	420
10.1.4	Negation in subordinate clauses	420
10.2 Non	n-clausal negation	421
10.2.1	Negated replies	421
10.2.2	Negative indefinites and quantifiers	423
10.2.3	Negation of adjectives	424
10.2.4	Privatives	425
10.3 Oth	er aspects of negation	426
10.3.1	A note on negation in complex clauses	426
10.3.2	Negated restrictive -ma 'more than (+NEG.EX), only'	426
10.3.3	Variety marking idiom NEG-VERB gu-VERB	428
10.4 Sum	nmary remarks	428
11 Non-d	leclarative clauses	429
11.1 Inte	rrogatives	429
11.1.1	Polar questions	431
11.1.1	.1 Polar questions with rising intonation and <i>-po</i> -infinitive	432
11.1.1	.2 Polar questions with -ka/ga and pá	433
11.1.1	.3 Simple polar questions with the attenuated markers -kam/gam and pán	ı. 445
11.1.1	.4 Polar questions with the interrogative copula <i>bo</i>	447
11.1.1	.5 Polar interrogatives with preverbal <i>á</i> (Lachung)	447
11.1.1	.6 Summary on polar questions	448
11.1.2	Question words and content questions	449
11.1.2	2.1 Question words	449
11.1.2	2.2 Content questions without final question markers	451
11.1.2	2.3 Content questions with <i>-kam/gam</i> and <i>nám</i>	454
11.1.2	Content questions with <i>bo</i>	455
11.1.3	Alternative questions	456
11.1.3	Alternative questions with <i>-kam/gam</i> and <i>pám</i>	456
11.1.3	Alternative questions with <i>bo</i>	458
11.1.3	Alternative questions with <i>-ka/ga</i>	459
11.1.3	Alternative questions with $-l\phi^2$	460
11.1.4	Tag questions	460
11.1.5	Questions with the reportative =lo	462
11.1.6	Exclamative questions with (h)o:	463
11.2 Exc	lamatives	464
11.2.1	Exclamation with - <i>l</i> ø?	464
11.2.2	Exclamative use of the interrogated copula $b\varepsilon$ -ka	465

11.2.3 Interjections	
11.3 Imperative	
11.3.1 Verb root as imperative	
11.3.2 Imperative suffixes <i>-tc^hi</i> , <i>-da</i> , <i>-na</i>	
11.3.3 Urgentive $=m\phi^2$	
11.4 Hortative - <i>kε/gε</i>	
11.5 Optative with <i>tcu</i> ?	
11.6 Summary remarks	
12 Connecting finite clauses	
12.1 Introduction	
12.2 Nonosyndetic connectors	
12.4 Summary remarks	
13 Constituent-modifying clauses	
13.1 Introduction	
13.2 Relative clauses	
13.2.1 Relativization by $-k^{h}\tilde{\epsilon}$	
13.2.1.1 Pre-head RCs	
13.2.1.2 Post-head RCs	
13.2.1.3 Headless RCs	
13.2.1.4 Internally-headed RCs	
13.2.2 Relativization by <i>-po/bo</i>	
13.2.2.1 Pre-head RCs	
13.2.2.2 Post-head RCs	500
13.2.3 Spatial nominalizer - <i>sa</i>	
13.2.3.1 Pre-head RCs	
13.2.3.2 Headless RCs	505
13.2.4 Quantifying nominalizer $-ts^{h}\varepsilon^{2}$	
13.3 Correlative clauses	
13.4 Noun complement clauses	507
13.4.1 Nominalization with -po/bo	
13.4.2 Nominalization with - <i>ce</i> ?	
13.4.3 Finite clause with a complementizer	
13.4.4 Genetive-marked finite clause	
13.5 Postposition complement clauses	
13.5.1 Genitivized -po-infinitive	
13.5.2 Bare - $\varepsilon \varepsilon$ -infinitive	
13.5.3 Sentence-like complement	
13.6 Summary remarks	

14 Complement clauses	515
14.1 Non-finite complement clauses	515
14.1.1 Complement clauses with <i>-po</i> -infinitive	515
14.1.2 Complement clauses with - <i>ce</i> -infinitive	520
14.1.3 Complement clauses with progressive $-tc\tilde{\varepsilon}:/z\tilde{\varepsilon}:/zin$	521
14.1.4 Postposition clause with <i>kor</i> 'about' as a clausal complement	521
14.2 Finite (clause-like) complement clauses	521
14.2.1 Finite complement clauses without complementizer	521
14.2.1.1 Declarative complement	522
14.2.1.2 Interrogative complement	522
14.2.2 Finite complement clauses with complementizer	523
14.2.2.1 Complementizers $=s\varepsilon$ and $lap(ti)$	523
14.2.2.2 Complementizer <i>ki</i> (loan from Nepali)	526
14.2.3 Complement clauses with a resumptive demonstrative	526
14.3 Summary remarks	526
15 Adverbial clauses	528
15.1 Introduction to forms	528
15.2 Nonfinal converb marker <i>-ti/di</i>	530
15.2.1 Antonion clauses	554
15.3.1 Anterior clauses	334 524
15.3.1.2 Anteriority with the postpositions gights 'after' and d_{70} (a) 'after'	534
15.3.1.2 Anteriority with the postpositions $g/able$ after and $d_{2}e_{-}(lo)$ after	550
15.3.2 Bostarior clauses with postposition <i>nonla</i> 'before'	537
15.3.2 Fosterior clauses with postposition <i>fience</i> before	550 540
15.3.3 Simultaneous no $d\tilde{a}$	540 540
15.3.3.2 Simultaneous converb markers <u>soudā</u> :/soundā:/tsubdā:	540 542
15.3.3.3 Simultaneity with $q\bar{q}$, 'time'	543
15.3.3.4 Simultaneity with kap 'time'	544
15.3.3.5 Simultaneous converb marker $-dv^{-}$	546
15.3.3.6 Simultaneous converb marker $rsnk^ha$	547
15.3.3.7 Simultaneity with $n'a mu$ 'in between'	548
15.4 Causal clauses	548
15.4.1 Causality with connector k'amiasens 'because'	5/19
15.4.2 Causality with $\sigma_i um ts^h \tilde{c}$ 'reason'	5 4 9 550
15.4.3 Causality with ken 'cause condition'	550
20.1.2 Currently marker current, condition	

15.4.5	Causality with ablative $=l\varepsilon$	552
15.4.6	Causality with agentive	553
15.4.7	Causality through nonfinal converb p 'jati(ki) 'doing'	555
15.4.8	Causal uses of the circumstantial-purposive converb	556
15.5 Pu	rposive clauses	556
15.5.1	Purpose with circumstantial-purposive converb marker -pa/ba	556
15.5.2	Purpose with the postposition <i>t'ønlɛ/t'ønlo</i>	557
15.5.3	Purpose with nonfinal converb làpti 'saying'	558
15.6 Co	nditional clauses	
15.7 Co	ncessive clauses	
15.8 Cla	auses of circumstance and manner	
15.8.1	Circumstantial-purposive converb marker - <i>pa/ba</i>	
15.8.2	Circumstantial construction with postposition <i>nåŋca/nåŋlo</i> 'inside'	
15.8.3	Circumstantial/manner use of progressive $-t\epsilon\tilde{\epsilon}:/z\tilde{\epsilon}:/zin$	
15.8.4	Circumstantial clauses with $k^h a = lo$	
15.8.5	Comparative manner	
15.8.	5.1 Comparative manner with $d\varepsilon m$	
15.8.	5.2 Comparative manner with <i>nàŋtar(gi)/nàŋzin</i>	
15.8.	5.3 Comparative manner with $t' on zin(gi)$	575
15.8.6	Genitivized -po-infinitive	
15.9 Ad	ditive clauses	
15.9.1	Additive with <i>mits^he?</i>	577
15.9.2	Additive with <i>tɛŋlo</i>	577
15.9.3	Additive with $k^h a = lo$	
15.9.4	Negated additive with <i>mèmbo</i>	
15.10 Su	bstitutive clauses with <i>ts^haplo</i>	579
15.11 Co	mparative clauses with $=l\varepsilon$ (<i>lako</i> =EMPH)	
15.12 Va 15.13 Su	mmary remarks	
16 Disc	purse phenomena	
16.1 En	phatic clitics	
16.1.1	Anaphoric emphatic $=r\tilde{a}$	585
16.1.2	Contrastive emphatic $=to$	590
16.1.2	Demonstrative-emphatic $= di$	593
16.1.4	Topicalizer-emphatic $=ni/n\epsilon$	598
16.1.5	Additive emphatic $=i\tilde{a}$:	
16.2 Cl:	ause-final clitics	
16.2.1	Honorific $=la$	602
10.2.1		

16.2.2	Attention marker = co	603
16.2.3	Non-commitment marker =ki/gi	605
16.3 As	ssertive and exclamative tags	606
16.3.1	Assertive tag <i>pá</i>	606
16.3.2	Exclamative tag (h)o:	610
16.4 Di	scourse particles $t'a$ and $t\varepsilon$	612
16.5 Re	ecapitulation	
16.6 Ri 16.7 A	ght-dislocation	
16.8 St	mmary remarks	
17 Note	es on lexicon	617
17.1 Id	eophones	617
17.1.1	Introduction	617
17.1.2	Ideophones expressing nonnormativity	619
17.1.3	Fully reduplicated ideophones	
17.1.4	Near-reduplicated ideophones with vowel change	
17.1.5	Rhyming ideophones	
17.1.6	Onomatopoeic ideophones	
17.1.7	Ideophonic suffixes	
17.2 Ki	nship terminology	636
17.2.1	Consanguineal relatives and their spouses	636
17.2.2	One's spouses relatives	
17.3 Na	ames	641
17.4 Co	blours	
17.4.1	Clear colours	
17.4.2	Pale colours	
17.4.3	Dark colours	
17.4.4	Other colour terms	
17.5 Vo	ocabulary used with small children	
17.6 St	mmary remarks	
Appendix	: Text excerpts	
Proverbs Riddles		
Folkstor	7	
Excerpt	from the Novel Richhi	
Excerpt	rom discussion	
Appendix 2	2: Differences between spoken and written language	670
Appendix 3	3: Vowel plot measurements	
Appendix 4	E Letter of informed consent	

References

List of Tables

Table 1.1. Summary of language names	6
Table 1.2. How to refer to the people who speak Denjongke	6
Table 1.3. UNESCO's Language Vitality and Endangerment framework	. 13
Table 1.4. Some linguistic differences between Tashiding and Lachung	. 17
Table 1.5. Consultants and types of data	. 19
Table 1.6. The Wylie system	. 30
Table 2.1. Consonant phonemes in Denjongke	. 33
Table 2.2. Denjongke vowels	. 46
Table 2.3. Three-way contrast of $\frac{2}{\sqrt{vs}}$ vs. $\frac{1}{ik}$ shown by differing F1 values	51
Table 2.4. Syllable structure	. 55
Table 2.5. Reduction of nominalizer - <i>po/bo</i> to - <i>m</i>	. 57
Table 2.6. Features of high and low register	. 58
Table 2.7. Voicing alternation in verbal suffixes	. 66
Table 2.8. Verb suffixes with verb roots ending in $/\eta$ /	. 67
Table 2.9. Consonant elision in verbal suffixes with initial -p	. 68
Table 2.10. Consonant elision in verbal suffixes with initial -k	. 68
Table 2.11. Negation of low vs. high register verbs	. 68
Table 3.1. Ordinary and honorific nouns with no formal resemblance	. 74
Table 3.2. Honorific nouns formed by compounding	. 75
Table 3.3. Honorific nouns formed by replacing a syllable	. 76
Table 3.4. Honorific nouns formed by prefixing and final syllable elision	. 76
Table 3.5. Nominalizing markers	.77
Table 3.6 Uses of the suffix -po/bo	.78
Table 3.7. Some masculine-feminine noun pairs	. 80
Table 3.8. Non-controllable and controllable verb pairs	. 85
Table 3.9. Ordinary-honorific-humilific triads of verbs	. 87
Table 3.10. Some ordinary-honorific pairs of verbs	. 87
Table 3.11. Verbal suffixes	.91
Table 3.12. Simple copulas 1	102
Table 3.13. Direct and attenuated questions with copulas	103
Table 3.14. Adverbs of manner related to sleeping	115
Table 3.15. Other adverbs of manner	115
Table 3.16. Some locative adverbs	116
Table 3.17. Noun-like temporal adverbs referring to times of day	116
Table 3.18. Days and years	117
Table 3.19. Other temporal adverbs 1	117
Table 3.20. Indefinite temporal adverbs	117
Table 3.21. Verb-modifying quantitative adverbs	118
Table 3.22. Adjective and adverb-modifying adverbs 1	118
Table 3.23. Numeral-modifying adverbs	119
Table 3.24. Epistemic adverbs	119
Table 3.25. Personal pronouns 1	120
Table 3.26. Reflexive pronouns 1	121
Table 3.27. Reciprocal pronouns	121
Table 3.28. Indefinite pronouns 1	122
Table 3.29. Demonstrative roots 1	122

Table 3.30. Derived demonstratives	123
Table 3.31. Proadverbs of manner and proadjectives	123
Table 3.32. Question words	124
Table 3.33. Numbers 1-20	125
Table 3.34. Two forms for referring to full tens	125
Table 3.35. Decimal system numbers 21-60	126
Table 3.36. Decimal system numbers 61-100	126
Table 3.37. Examples of numbers from 100 onwards	127
Table 3.38. Other numeral-related vocabulary	127
Table 3.39. Examples of the vigesimal number system	128
Table 3.40 Nominal roots used in postpositions listed in Table 3.41	131
Table 3.41. Postpositions	132
Table 3.42. Clause/sentence connectors	137
Table 3.43. Some interjections	138
Table 3.44. Case clitics	139
Table 3.45. Emphatic clitics	144
Table 3.46. Clausal clitics	144
Table 4.1. Some titles	154
Table 4.2. Quantifying pronouns	159
Table 4.3. Some phrasal verbs formed with the verbalizers <i>p</i> ' <i>ja</i> , <i>kjap</i> , <i>tã</i> : and <i>tap</i>	171
Table 4.4. Some other phrasal verbs	171
Table 4.5. Serial verbs from converbs	174
Table 4.6. Secondary verbs	177
Table 4.7. Adjective and adverb-modifying quantifying adverbs	186
Table 4.8. Pre-numeral modifiers	192
Table 4.9. Post-numeral modifiers	193
Table 5.1. Summary of argument marking options	198
Table 5.2. Marking of A argument in the past tense of some verbs (consultant KN)	205
Table 5.3. Postpositions	232
Table 5.4. Verb-modifying quantitative adverbs	246
Table 6.1. Personal pronouns	252
Table 6.2. Reciprocal pronouns	257
Table 6.3. Indefinite pronouns	259
Table 6.4. Affirmative indefinite reference with question words	262
Table 6.5. Negated indefinite reference with question words	262
Table 6.6. Demonstrative roots	266
Table 6.7. Derived demonstratives	267
Table 6.8. Proadverbs of manner and proadjectives	270
Table 7.1.Basic copulas	273
Table 7.2. Nominalized copulas	299
Table 7.3. Dzongkha copulas (adapted from van Driem 1998)	310
Table 8.1. Past constructions	315
Table 8.2. Present habitual and future constructions	333
Table 8.3. Imperfective, progressive and continuous constructions	349
Table 8.4. Modality markers	364
Table 9.1. Constructions with equatives as auxiliaries	386
Table 9.2. Constructions with existentials as auxiliaries	392
Table 9.3. Constructions with the completive -tsha(:)	399
Table 10.1. Clausal negation formatives	410
Table 10.2. Negation of declarative final forms	412

Table 10.3. (A)symmetry in finite negated constructions	414
Table 10.4. Nominalized copulas	415
Table 10.5. Negation of non-declarative non-copular clauses	416
Table 10.6. Negation of interrogatives	.417
Table 10.7. Negation of interrogated equative copulas	420
Table 10.8. Negation of interrogated existential copulas	420
Table 10.9. Negation of nonfinal forms	421
Table 10.10. Adjectives negated by a prefix	424
Table 10.11. Adjectives negated by a negated copula	424
Table 10.12. Adjectives meaning 'different (kinds of)'	425
Table 11.1. Interrogative morphemes	430
Table 11.2. Negating copulas with -ka/ga	.431
Table 11.3. Some interjections	466
Table 12.1. Monosyndetic clause connectors	479
Table 12.2. Bisyndetic clause connectors	. 487
Table 13.1. Constituent-modifying clauses	. 489
Table 13.2. Temporal and aspectual values in pre-head RCs	492
Table 13.3. Some postpositions which take complement clauses	511
Table 14.1. Elements forming complement clauses	515
Table 14.2. Verb types receiving a complement clause with -po/bo	516
Table 15.1. Converbal endings used in adverbial clauses	528
Table 15.2. Postpositions heading adverbial clauses	528
Table 15.3. Nouns heading adverbial clauses	529
Table 15.4. Other constructions used in adverbial clauses	529
Table 15.5. Simultaneous dependent verbal constructions	540
Table 16.1. Emphatic clitics	585
Table 16.2. Clausal clitics	602
Table 17.1. Ideophones expressing nonnormativity	620
Table 17.2. Fully reduplicated ideophones	627
Table 17.3. Complex ideophones	628
Table 17.4. Near-reduplicated ideophones	630
Table 17.5. Rhyming ideophones	632
Table 17.6. Ego's grandparents' generation and beyond	637
Table 17.7. Matrilineal parents' generation	637
Table 17.8. Patrilineal parents' generation	637
Table 17.9. Ego's own generation	638
Table 17.10. Generations younger than ego	639
Table 17.11. Terms for addressing strangers of various ages	640
Table 17.12. Spouses relatives	640
Table 17.13. Names associated with days of the week	641
Table 17.14. Names according to gender	642
Table 17.15. Basic colour terms	642
Table 17.16. Some compound colour terms	643
Table 17.17. Colours terms with ideophonic suffixes	643
Table 17.18. Pale colours with -lop (Tashiding)	644
Table 17.19. Pale colours with -sa loksy: (Martam)	644
Table 17.20. Dark colours with -na? loksy: (Martam)	645
Table 17.21. Other coulour terms from Tashiding (consultant DB)	645
Table 17.22. Child talk vocabulary	646

List of Figures

Figure 1.1. Shafer's (1955) classification of Bodic languages	7
Figure 1.2. Bradley's (2002) grouping of Tibeto-Burman languages	8
Figure 1.3. Affiliation of Denjongke in Glottolog	9
Figure 2.1. Wave forms from $k' \varepsilon : t \varepsilon \tilde{\varepsilon}$: (top) and $k' \varepsilon : t \varepsilon^h \tilde{\varepsilon}$: (bottom)	34
Figure 2.2. Prenasalization and prevoicing in the initial in /goko/ 'garlic'	34
Figure 2.3. Initial consonant duration in /khap/ 'needle', /k'ar/ 'what?' and /ka/ 'who?'	44
Figure 2.4. Consonant duration in /phu:/ 'fly', /p'u/ 'boy' and /pu:/ 'pack' (RB)	44
Figure 2.5. Two pronunciations (in context) of the word /k'ɛː/ [k'ɛː] 'line, order' (SG)	45
Figure 2.6. Wave forms and pitch of /tsa/ 'grass' and /ts'a/ 'make-up' (TB)	45
Figure 2.7. Vowel plot from consultant TB (Ralang)	47
Figure 2.8. Pitch traces of /má/ 'wound' and /mà/ 'mother' in isolation and in context (KN	√) 59
Figure 2.9. Pitch traces of /ká/ 'order' and /k'a/ 'what, where' in isolation and in context ((KN)
-	59
Figure 2.10. Pitch traces of /kom/ 'dry (verb)' and /gom/ 'door' in isolation and in context	t
(KN)	60
Figure 2.11. Sequentially uttered /ná/ 'five', /nà/ 'I' and /nâ/ 'drum' (RB)	60
Figure 2.12. Contrast between /ta/ 'horse' and /ta?/ 'tiger' when pronounced in isolation	61
Figure 2.13. Contextual tonal contrast between /ta/ 'horse' and /ta?/ 'tiger'	61
Figure 2.14. Pitch contrast with voiced stop initials /zí:/ 'look' and /zi:/ 'split'	62
Figure 2.15. Contrast of glottal and non-glottal ending in /nà?/ 'speech' and /nà/ 'I'	63
Figure 2.16. Pitch contrast in /pømpu/ 'leader' and /p'ømpu/ 'Bön practitioner'	64
Figure 2.17. Pitch contrast in /námtco?/ 'ear' and /màmtchu/ 'lower lip'	64
Figure 2.18. Pitch in postpositions $t \in \eta k^h a$ and $n a \eta \in a$	65
Figure 2.19. <i>nà</i> vs. <i>nà</i> : in context (consultant KN)	70
Figure 3.1. Wave forms from nonsensical p'ja-tci and imperative p'ja-tchi 'do!' (KUN)	96
Figure 4.1. Structure of the noun phrase	. 149
Figure 4.2. Structure of the verb complex in declarative mood	. 170
Figure 4.3. Structure of the verb complex in declarative mood	. 183
Figure 7.1. Hypothesis of grammaticalization of WT yin	. 313
Figure 9.1. The pitch in reportative $=lo$ (9.69) and tag question lo (9.70) contrasted (TB).	. 405
Figure 9.2. Pitch trace of original (9.71) and TB's and KL's repetitions of it respectively .	. 406
Figure 11.1. Intonation in polar question (11.2)	
Figure 11.2. Intonation in polar question (11.3)	. 432
Figure 11.3. Intonation in polar question (11.13) with -ka/ga	. 435
Figure 11.4. Intonation in polar question (11.14) with -ka/ga	. 435
Figure 11.5. Intonation in declarative (11.15), cf. Figure 11.3	. 436
Figure 11.6. Intonation in polar question (11.16) with <i>pá</i>	. 436
Figure 11.7. Intonation in polar question (11.17) with <i>pá</i>	. 436
Figure 11.8. Intonation in content question (with a wh-word) (11.87)	. 453
Figure 11.9. Intonation in content question (11.94) with pám	. 455
Figure 11.10. Intonation in alternative question (11.108) with bo	. 459
Figure 11.11. Intonation with tag question <i>iŋga</i> in (11.115)	. 461
Figure 11.12. Intonation with tag question lo in (11.120)	. 462
Figure 11.13. Rising intonation with lo in polar question (11.121)	. 462
Figure 11.14. Low intonation with <i>lo</i> in content question (11.122)	. 463
Figure 11.15. Rising intonation with <i>inam</i> in imperative (11.157)	. 472
Figure 16.1. Rising intonation on <i>pá</i> in (16.93)	. 609
Figure 16.2. Rising intonation on <i>pá</i> in (16.94)	. 609
Figure 16.3. Falling intonation on <i>pá</i> in (16.95)	. 609

List of Maps

Map 1.1. Sikkim within India	2
Map 1.2. Sikkim (India), Chumbi (China) and Ha (Bhutan)	3
Map 1.3. Consultants' places of origin within Sikkim	. 23

Abbreviations					
1	first person	IN	intensifier		
2	second person	INDF	indefinite		
3sgm	third person singular masculine	INF	infinitivizer		
3SGF	third person singular feminine	L	low-level		
ABL	ablative	LNK	linker		
ADJZR	adjectivizer	LOC	locative		
ADVZR	adverbializer	М	mid-level		

AEMPH	anaphoric emphatic	NC	non-commitment
AGT	agentive	NE	neutral
AO	addressee-oriented	NF	nonfinal
APH	alterphoric	NEG	negative
APPR	approximative	NMLZ	nominalizer
ASR	assertive	NN	nonnormative
ASSOC	associative	NPST	nonpast
AT	attention marker	NUM	numeral
ATTQ	attenuated question	ORD	ordinal number
СЕМРН	contrastive emphatic	PER	personal
CHT	child talk	PFV	perfective
CIRC	circumstantial	PL	plural
CMPL	completive	PN	personal name
CNG	connegative	PRF	perfect
CNTJ	conjunction	PQ	polar question
COL	collectivizer	PRET	pretensive
COMP	complementizer	PROB	probabilitative
COND	conditional	PROG	progressive
CONJ	conjunction	PST	past
СОР	copula	PUR	purposive
DAT	dative-locative	Q	question
DEMPH	demonstrative-emphatic	QUA	quantifier
DIM	diminuative	QUO	quotative
EGO	ego(phoric)	REFL	reflexive pronoun
EMPH	emphatic	REP	reportative
EQU	equative	SEN	sensorial
EX	existential	SG	singular
EXPER	experiental	SIM	simultaneous
EXCLAM	exclamative	SPAT	spatial
FRN	friendly	SUG	suggestive
FUT	future	SUP	superlative
GEN	genitive	TAG	tag
HON	honorific	TERM	terminative
HON	honorific	TOP	topicalizer-emphatic
HORT	hortative	TPN	toponym
IDEO	ideophone	URG	urgentive
IMF	imminent future	VBLZ	verbalizer
IMP	imperative	WD	written Denjongke
IPFV	imperfective	WT	Written Tibetan

1 Introduction

This is a grammar of Denjongke, also known as Lhoke, Sikkim(ese) Bhutia and simply Sikkimese, a Tibetic language spoken in the Indian state of Sikkim. This introductory chapter first provides background information on the language (Denjongke) and the people who speak it (Denjongpos¹), see §1.1. Subsequently, methodology and data are described in §1.2.

1.1 The language and the people

This section introduces the language and the people who speak it. Topics covered are the language names (\$1.1.1), number of speakers (\$1.1.2), genetic affiliation (\$1.1.3), previous research (\$1.1.4), origins of the people (\$1.1.5), the history of written Denjongke (\$1.1.6), religion of Denjongpos (\$1.1.7), the honorific system and social relations (\$1.1.8), language contact and multilinguality (\$1.1.9) and language endangerment (\$1.1.10). The last section provides an overview of some central linguistic features of Denjongke (\$1.1.1).

1.1.1 Name of the language

The language which is the subject of this grammar has four main names none of which are without problems: Denjongke, Lhoke, Bhutia and Sikkimese.² The ISO 639-3 code for the language is 'sip', while the glottocode (see glottolog.org) is 'sikk1242'. The name "Denjongke" /dɛndzoŋkɛ?/³ (AENNY e^{-NYNY} . 'bras-ljongs-skad 'the language of Sikkim') is chosen as the main title because it is probably the least problematic. The word dendzã. which literally means 'rice-valley' or 'fruit(ful)-valley', has become the term that refers to Sikkim in both Denjongke and Central Tibetan. The last part of the language name, ke?, refers to 'sound' or 'language'. The first describer of the language, Sandberg (1888), reasons as follows: "as the Bhutias both in numbers and in power are the predominant people of the land, we may, we think, not unreasonably speak of the Bhutia tongue as the Dé-jong Ké or vernacular of Sikkim." In the revised edition of the grammar (Sandberg 1895), the language name is modified to the phonetically more representative "Dén-jong Ké." Sandberg's rationale for a geopolitical choice for the language name was the numerical and political strength of the Sikkimese Bhutias, or Denjongke speakers, in Sikkim, i.e. Denjong. However, now that the times of the Bhutia ruling dynasty are over and the Denjongpos/Bhutias/Lhopos are a minority in numbers, the original rationale for language name choice is no longer valid.

¹ This word also occurs as Denjongpa. Some of my consultants commented that Denjongpa is the Tibetan and Denjongpo the Denjongke pronunciation. Henceforth, the spelling Denjongpo is used.

 $^{^{2}}$ See also Mullard (2011: 21, 37), who addresses the difficulty of choosing an appropriate term for referring to the people who speak the language.

Map 1.1. Sikkim within India



Another challenge with the term "Denjongke" is that in being a geopolitical term based on present international borders ('language spoken in Sikkim'), it overlooks historical and linguistic unity of Sikkim and its adjacent areas Chumbi (China) and Ha (Bhutan).⁴ According to their traditional lore, the Lhopos started to spread to Sikkim and Ha through Chumbi valley (Balikci 2008: 68-70). The Tibetic variety spoken in Sikkim is indeed to a great degree intelligible with the Tibetic varieties of the Chumbi and Ha.⁵ However, as the data presented in this grammar were gathered in Sikkim, the geopolitically based term "Denjongke," 'language of Sikkim', is descriptive of the data.

Another name for the language is "Lhoke" (*Single three the stand* south(ern) language"). This term also has both advantages and disadvantages. The first of the two advantages over the language name "Denjongke" is that "Lhoke" is the typically used endonym for the language when speaking the language. The second advantage is that the term "Lhoke" has potential to refer to speakers outside the geopolitical area of Sikkim and is therefore more fit than Denjongke to refer to the linguistic and historical unity of Sikkim to adjacent regions in the east. The name "Lhoke" suggests a (Tibetic) language spoken south of Tibet. Tsichudarpa's (2018: 47) enigmatic statement that "there are thousands that speak this language outside India too" may refer to the fact the Tibetic variety spoken in the Chumbi valley in China is so close to Denjongke as to be considered the same language. Walsh's (1905) vocabulary, ⁶ along with his notes on verb forms ⁷ and historical

⁴ The language spoken in Chumbi is in Lachung (North Sikkim) called *tc 'umbø: ke?* (artshire' gyung-ba'i skad).

⁵ A group of people in Ha claimed to understand most or all they heard, when I played them a recording of speech from Lachung (North Sikkim).

⁶ Tromowa words are most of the time identical with "Sikhimese" (=Denjongke) and both contrast with Tibetan.

⁷ All the tense/aspect/modality related verbal constructions in Walsh (1905: iv-v) with the exception of the form "lapbo-she," which I do not recognize as resembling any Denjongke form, are identical with the present description of Denjongke presented especially in §8.

phonology,⁸ indeed give preliminary evidence that the Tibetic variety spoken in the Chumbi valley could be linguistically considered to be the same language or part of the same dialectal continuum as Denjongke.



Map 1.2. Sikkim (India), Chumbi (China) and Ha (Bhutan)

The historical connection of Sikkim with Chumbi and Ha is illustrated by an interesting anthropological insight: all the lhopo descent groups around Sikkim worship the same mountain deity, Masang Khungdü (WD⁹ arvarar gravarar gravarar ma-sangs khyung-'dus/bdud), who is said to reside near the point where the borders of Sikkim, Chumbi and Bhutan join (Balikci 2008: 73). Other Tibetic groups that have later come to Sikkim from Tibet or Nepal do not share this ritual characteristic. The ritual and linguistic similarity between Tibetic groups in Sikkim, Chumbi and Ha lead Balikci (2008: 73) to hypothesize that earlier in history these areas "were perhaps once populated by a somewhat homogenised Lhopo population." Balikci (2008: 73) further offers the interesting observation that during the Tibetan refugee crisis of 1959, the Chogyal (religious king) welcomed asylum seekers from Chumbi into Sikkim as "Sikkimese" whereas other refugees were assigned the outsider status "Tibetans." Because of the historical and linguistic unity of Sikkim

⁸ Walsh (1905: vi) lists some ways in which Tromowa pronunciation differs from Central Tibetan. Three are listed here. The first is the neutralization of the difference \tilde{a} : vs \tilde{o} : so that only \tilde{o} : is used. The same feature also occurs in Denjongke spoken in North and East Sikkim (but not in West Sikkim, where there still is a contrast between \tilde{a} : and \tilde{o} :). The second difference is the pronunciation of WT glide *y* as separate from the bilabial plosive (e.g. *pja*) whereas Central Tibetan exemplifies a merged pronunciation (e.g. *tea*). Again, Denjongke follows the Tromowa pattern. The third difference in pronunciation that distinguishes Tromowa, and Denjongke, from Central Tibetan is the tendency to replace the Central Tibetan glide *r* with *y*, i.e. *kja* 'hair' (Tromowa, Denjongke) vs. *ta* 'hair' (Central Tibetan, *r* causes retroflexivization).

⁹ Throughout the grammar, WD refers to written Denjongke and WT to Written Tibetan.

with Chumbi and Ha, the term "Lhoke" seems a more lucrative candidate for a general language name than "Denjongke."

The geographically larger referential scope of the term "Lhoke," however, is also disadvantageous because the same or similar names are used for geographically close related languages. Grierson (1909: 129) calls "Bhōtiā of Bhutan" (i.e. Dzongkha) by the term "Lhoke." Following Grierson's tradition, Tikkanen's (1991: 10) Hindi grammar (in Finnish) published a map (by Bertil Tikkanen and Virpi Hämeen-Anttila) depicting the language situation in South Asia, in which the language name "Lhoke" is written within Bhutan. In the same vein, the 13th edition of the Ethnologue (Grimes 1996) lists "Lhoke" as an alternative name for Dzongkha. Moreover, Genetti (1986: 387) lists "Lhoke" as a Western Bodic language, separate from the South Bodish language "Danjongka" (most likely referring to the same language as Denjongke here). Walsh also lists (1905: 4) "Lho-yü" as the "Sikhimese" and "Tromowo"¹⁰ word for Bhutan, in constrast to the Tibetan word and 'brug-yul 'Bhutan'. Due to the association of the term "Lhoke" with Dzongkha, it was recommended by Khenpo Lha Tshering (2016), the principal of Higher Institute of Nyingmapa Studies in Gangtok, that the term "Denjongke" be used rather than "Lhoke." Cognates of the term "Lhoke" are also used for other Tibetic languages: "Lhoket" is an alternative name for "Lhomi" (Nepal, Vesalainen [2016: 2]), and "Lhoke" and "Loket" are given as alternative names for "Lhowa" (Nepal, the Ethnologue [Simons & Fennig 2017]).¹¹

Within Sikkim, the typically used exonym for Denjongke, when speaking either English or Nepali, is "Bhutia/Bhotia" (Nepali $\mathfrak{AllCAll} b^h utia/b^h otia$).¹² This term is problematic because it represents an overly simplified categorization by outsiders who group all the Tibetan-related peoples and languages basically as one, those coming from *bhot* 'Tibet' (Nepali)¹³. When more precision is needed, the term is amplified by a geographical location, e.g. "Bhōtiā of Bhutan" (Grierson 1909: 129), "Humla Bhutia" (Wilde 2001) or "Sikkim Bhutia" (Ethnologue, Simons & Fennig 2017). Because the term "Bhutia" is known to other language communities, it has become the preferred language name when speakers seek recognition for their language from outside their own community. Currently, there is a political motivation for uniting all Tibetan-related peoples and languages under the common name "Bhoti" to gain more influence within India. Some Denjongke speakers want to make a distinction between "Bhutia," which refers to Denjongke speakers and their languages, and "Bhotiya," which refers to Tibetan-related peoples of the southern Himalayas more generally,¹⁴ but as far as I understand, this distinction has not received overall acceptance, especially within other linguistically related groups.¹⁵

¹⁰ Tibetic variety spoken in the Chumbi valley (of China) situated between Sikkim and Bhutan.

¹¹ I once heard a Lhomi speaker refer to his language as [loke?].

¹² The Census of India 2011 reports that there are 229,954 speakers of "Bhotia" in India and lists speakers in all the Indian states except Jharkhand, Odisha and Telangana. States with more than a thousand "Bhotia" speakers are Jammu and Kashmir (107451), Arunachal Pradesh (62458), Sikkim (41889), Uttarkand (9287), West Bengal (4293) and Himachal Pradesh (2012). Because Sherpas and Tibetans are listed separately, it may be presumed that the number of "Bhotia" speakers in Sikkim (41889) refers to Denjongke speakers. The number assigned to West Bengal (4293) may refer to Denjongke speakers living in and around Darjeeling and Kalimpong.

¹³ This Nepali word most likely derives from Written Tibetan 55. bod 'Tibet'.

¹⁴ This view is expressed in the Wikipedia article on "Bhutia" (13 Feb 2018)

¹⁵ When I explained, in Nepali, to one lady that I was carrying out research on the "Bhutia" language, she retorted, "Which Bhutia? There are many Bhutias." It turned out that the lady was a Sherpa, a member of a smaller "Bhutia" group, who may be wary of others "hijacking" their identity term.
The term "Sikkimese," which is the main language name for Denjongke in Walsh (1905) and the Ethnologue (Simons & Fennig 2017), is essentially an English translation of Sandberg's (1888) original term "Dé-jong Ké" ('the language of Sikkim'). The term was in active use a few decades ago but has since then become politically incorrect, because of demographic and political realities (personal communication, Tsewang Topden, Ambassador of India). The prevalence of the Anglocentric term "Sikkimese" as a language name a few decades ago probably reflected the general cultural atmosphere where "the elite looked up towards and emulated the cultures of both Tibet and the British at the cost of losing their own" (Balikci 2008: 11). The term "Lhoke" 'south(ern) language', on the other hand, may be seen to represent a Tibeto-centric vision which underspecifies the language as being spoken "somewhere" south of Tibet. The term "Denjongke" has essentially the same meaning as the Anglo-centric "Sikkimese" but escapes colonial overtones and is linguistically opaque enough not to cause offense among the other ethno-linguistic groups.

In addition to the four terms mentioned above, the highlanders of Lachung call their own language $j\dot{a}:k\epsilon^2$ (WD (WT, WT)) 'up(per) language' and the Denjongke varieties spoken in lower altitudes $m\dot{a}:k\epsilon^2$ (WD (WT, WT)) 'low(er) language'.

To summarize, the main language name chosen for this grammar is Denjongke. The main reasons for this choice are that the term "Denjongke" is distinctive enough (contra "Lhoke" and "Bhutia"), more endonymic than "Bhutia" and "Sikkimese," politically correct (contra Sikkimese), and in harmony with an ongoing research tradition beginning with Sandberg (1888, 1895) and Grierson (1909). Moreover, it is reflected in later references such as Genetti's (1986: 387) "Danjongka"¹⁶, Bradley's "Danjong" (Bradley 1997: 6), and is also consistent with my own earlier work (Yliniemi 2016a, 2017).

The two counter-arguments for using the language name "Denjongke" are that it is not the most typical endonym for the language (which is "Lhoke") and that the geopolitically oriented term "Denjongke" ('language of Sikkim') downplays the linguistic similarity of the Tibetic varieties in Chumbi (China) and Ha (Bhutan). In response to the prior argument, it can be said that although "Lhoke" is the typical endonym, the term "Denjongke" is much more distinctive and understandable when considering the whole Tibetosphere. In response to the second argument it should be noted that the present description is limited to the Tibetic variety in Sikkim and does not claim to represent varieties east of Sikkim, although they do seem to form a dialect continuum with Denjongke.

The different language names and the terms used for people who speak Denjongke are summarized in Table 1.1 and Table 1.2 respectively.

¹⁶ My MA thesis (Yliniemi 2005) uses the mistaken form Denjongka (cf. Genetti 1986: 387) for the language. This mistake, as far as I remember, derives from the alternative language names listed in the Ethnologue (Simons & Fennig 2017).

Name of lan	iguage		Brief description
Denjongke	[dɛndzoŋkɛ?]	'language of Sikkim'	Pan-Tibetan endonym geopolitically
			limited to Sikkim (used when speaking
			Tibetan and Denjongke)
Lhoke	[loke?]	'south(ern) language'	Somewhat vague Tibeto-centric
			endonym with potential to cover
			language varieties spoken in a larger
			area than present Sikkim (used when
			speaking Denjongke)
Bhutia	[bʰut̪ia/bʰot̪ia]	'person or language	Term used with non-Tibetic language
		related to Tibet'	communities in Sikkim (used when
			speaking Nepali and English)
Sikkimese		'language of Sikkim'	Anglo-centric somewhat obsolete and
			politically incorrect endonym (used to
			be used when speaking English)

Table 1.1. Summary of language names

Table 1.2. How to refer to the people who speak Denjongke

Language name	Speaker name
Denjongke	Denjongpo/Denjongpa (lit. 'Sikkim-dweller')
Lhoke	Lhopo ('southener')
Bhutia	Bhutia ('person/people of Tibetan origin')
Sikkimese	Sikkimese (obsolete as a reference to Denjongke/Denjongpos only)

1.1.2 Number of speakers

In personal communication some Denjongke language teachers have estimated the current number of speakers at 25–30,000. The Ethnologue (Simons & Fennig 2017) lists 70,300 speakers (as of 2001), which is an over-estimated number, unless the number proposes to include speakers of Tromowa in the Chumbi valley (China).

Establishing the number of Denjongke speakers is complicated by at least four factors. The first is that competence in Denjongke and ethnic identity associated with the language coincide to a diminishing degree, i.e. increasing numbers of Denjongpos, especially among children and young people, do not speak Denjongke. The State Socio-Economic Census of 2006 (as cited in Tsichudarpo 2018: 46) lists 49,837 ethnic Bhutias (presumably meaning "Denjongpos"), comprising 8,57% of the population of Sikkim. However, the speakers of the language are much fewer than those who identify with the group ethnically.

The second factor that complicates the counting of speakers is that many ethnic Lepchas, who have lived in close contact to Denjongke speakers for several centuries, also speak Denjongke as either first or second language. The third factor is that it is difficult to determine how far to cast the net in search of Denjongke speakers. Varieties of Tibetic, which are intelligible to at least some Denjongpos, are spoken outside of Sikkim, both within India and outside of India. Within India, the language is said to be spoken in Darjeeling and Kalimpong (see Map 1.2), especially in Bhutia Busty (Darjeeling) and Pedong (near Kalimpong). As pointed out in the previous section, outside of India the Tibetic varieties in Chumbi valley (Groma/Tromowa, ISO 639-3 'gro') and Ha (considered a dialect of Dzongkha) are to some degree intelligible especially with Denjongke spoken in the northern village of Lachung. For Tromowa, the Ethonologue lists 26,800 speakers,

of whom 12,800 are reported to live in the Chumbi valley. The rest are presumably assigned to India and Bhutan.

The fourth reason for difficulty in counting the number of speakers is that in a language endangerment situation (such as the one exemplified by Denjongke) ethnic Denjongpos' language competence occurs on a continuum from almost no knowledge to great fluency. It is difficult to define how much of a language a person needs to speak/understand to be considered a speaker of a language. Turin (2011) reports an interesting statistic from the first modern linguistic survey of Sikkim, conducted among secondary school students, stating that while 10% of the interviewees reported Bhutia as their mother tongue, only 7% claimed an ability to speak the language.¹⁷

1.1.3 Genetic affiliation

In Shafer's (1955) classic classification of Sino-Tibetan languages, Denjongke is part of the South Bodish branch within the larger Bodic division, which together with Sinitic, Daic, Burmic, Baric and Karenic divisions form the Sino-Tibetan language family. Shafer's subgroupings under the Bodic division are presented in Figure 1.3 (emphasis added).

Figure 1.1. Shafer's (1955) classification of Bodic languages

Bodic division Bodish section Bodish branch Central Bodish unit Lhoskad (Lhoke), Lhasa, Sikkim... South Bodish unit Groma (Tromowa): upper, lower Sikkimese **Dand**źongka East Bodish unit Tshangla branch Rgyarong branch Gurung branch West Himalayish section West Central Himalayish section East Himalayish section Newarish Digarish Midźuish Hrusish Dhimalish Miśingish Dzorgaish

Shafer's (1955) language names present some difficulty because four terms (those in bold above) could potentially refer to the subject of the present study, two of them listed as Central Bodish and two as South Bodish languages. However, considering Shafer's (1955: 101) criteria of

¹⁷ A similar result was reported for Lepchas and Limbus, which were both more often referred to as the mother tongue (6% and 4% respectively) than as a language of spoken competence (5% and 3% respectively).

distinguishing Central Bodish from South Bodish ("[c]haracterisized by the shift of [WT] -r- to y-") it becomes clear that the language described in this grammar is identified as Sikkimese and/or Dandźongka of the South Bodish unit in Shafer's work. The terms "Lhoskad (Lhoke)" and "Sikkim" under Central Tibetan remain somewhat mysterious.¹⁸ The term "Dandźongka" may refer to the same language as Sikkimese, or it may represent another Tibetic variety, possibly Dzongkha.

According to Bradley's (2002: 75) grouping, Denjongke is a Tibeto-Burman, Western Tibetan/Bodic language, see Figure 1.2. Bradley's Western Tibetan corresponds, in his own words, "mainly to Shafer's Bodic group" (2002: 75).

Figure 1.2. Bradley's (2002) grouping of Tibeto-Burman languages



The Ethnologue (Simons & Fennig 2017) lists Denjongke (using the name "Sikkimese"), along with Dzongkha (Bhutan) and seven smaller languages spoken in Bhutan and Chumbi valley (China), as a Western Tibeto-Burman, Bodish, Central Bodish, Southern language. Denjongke may also be referred to as a Tibetic language. The term "Tibetic" refers to languages derived from Old Tibetan (Tournadre 2008: 283; Tournadre 2014). "Tibetic" differs from "Bodish" in exluding "Tamangic and any other Bodish languages not derived from Old Tibetan" (Tournadre 2008: 283).

The affiliation of the language in Glottolog (glottolog.org) is given in Figure 1.3. The numbers refer to the number of languages listed for each grouping.

¹⁸ One or both of these terms may refer to languages spoken by Tibetan-related peoples who have come to Sikkim rather late. For instance, Mullard (2011: 37) states that "there seems to have been a substantial movement of Tibetans from Eastern Tibet during the 1920s. These migrants settled in regions close to Sikkim-Bhutan border."

Figure 1.3. Affiliation of Denjongke in Glottolog Sino-Tibetan (488), Bodic (82), Bodish (54), Old-Modern Tibetan (43), Tibetic (42), Southern Tibetic (9), Dzongkhic (6), Sikkimese

The other five Dzongkhic languages referred to in Figure 1.3 are Chocangacakha (see Tournadre & Rigzin 2015), Groma (see Walsh 1905) and what Glottolog terms "nuclear Dzongkhic" languages Dzongkha (see van Driem 1998 and Watters 2018), Layakha and Lunakha.

1.1.4 Previous research

A sketch of Denjongke grammar was written by Sandberg in 1888 and an expanded second edition of 144 pages in 1895. Sandberg's grammar has a three-and-half page introduction to pronunciation and then continues with 40 pages of grammar. The grammar part is followed by 38 pages of example sentences that the author thought might prove helpful for language learners. The example sentences are divided into topical headings such as "horses and guns," "engaging coolies," "the weather," "shooting in the hills" and "talk on religion." The appendices consist of a list of animals names, toponyms and a collections of Lepcha¹⁹ words and phrases.

Grierson's (1909:119-128) survey entry on "Dänjong-kä" provides some distilled phonological, etymological and grammatical information derived from Sandberg's description. A new contribution to Sandberg's (1895) collection of isolated sentences is a coherent text, the Prodigal Son story (obtained from David Macdonald), which is given in Tibetan script, transliterated Roman script and a glossed phonological Roman script.

Walsh (1905) provides a 34-page vocabulary list of the Tromowa languge (WT ﷺ) gro-moba), a Tibetic variety spoken in the Chumbi valley. He includes comparative examples of equivalent words from both "Sikkimese" (Denjongke) and Central Tibetan. The Sikkimese words are mostly identical with Tromowa words, testifying to the very close relationship between the Tibetic varieties spoken in Sikkim and Chumbi.

Shafer (1974) presents some lexical data and historical-comparative observations on Denjongke phonology. My MA thesis, Yliniemi (2005), was the first treatise of Denjongke phonology. Yliniemi (2016) decribes the functions of Denjongke attention marker =co and also provides a glossed text sample from Bhaichung Tsichudarpo's novel Richhi (an edited version of which can be found in Appendix 1). Yliniemi (2017) describes Denjongke copulas and evidentiality (chapter 7 of the present work is an updated version of Yliniemi 2017).

For anthropological information on Denjongpos/Lhopos, refer to Balikci (2002) and especially Balikci (2008), which is an in-depth study of rituals in a Denjongke speaking community in North Sikkim. Yliniemi (2018) presents a micro-anthropological study of the every-day life of one Denjongpo/Lhopo family.

¹⁹ Lepcha is a Tibeto-Burman languages spoken by a people who are considered the earliest inhabitants of Sikkim. For a Lepcha grammar, see Plaisier (2006).

1.1.5 Origins of the people

In terms of modern historical research, the origins of the Sikkimese Denjongpos/Lhopos are not well established (Balikci 2008: 65, Mullard 2011: 36). It is likely that Tibetic peoples from adjacent areas came to Sikkim in several migrations from perhaps as early as the ninth century up until the 20th century (Mullard 2011: 36, 77). According to Mullard (2011: 73) there are approximately thirty-four clan names among the Denjongpos, and some of them may represent places of origin. For instance, the clan name anti-instance, the clan name anti-instance, the clan name anti-instance, the clan name anti-instance.

The twelve tribes who claim descent from Gye Bumsa view other Lhopo tribes as less prestigious late-comers (Balikci 2008: 71). The other tribes, in defence of their status, have claimed that they came to Sikkim before Gye Bumsa's time (Balikci 2008: 73-74). In favour of this claim, Mullard (2011: 77) hypothesizes that Tibetan migration to Sikkim likely began at the same time as migration to Bhutan, that is, in the ninth century. Thus, the official historical narrative may describe the origin of the tribes which trace their roots to Gye Bumsa but does not tell the whole story of Tibeto-Sikkimese migrations.

Denjongpos have historically had a close relationship with the Lepchas. This relationship is said to originate from Gye Bumsa's time. After receiving a male heir, Gye Bumsa returned grateful to Sikkim and made an alliance with the Lepcha bongthing Thekongtek. In the accompanying ritual the territorial deities of Sikkim, the most formidable of these the mountain Kangchendzönga, were invoked as witnesses of the covenant. This alliance is still annually commemorated in the Pang Lhabsol ritual (קקק, graping dpang lha-gsol 'offering to the witness gods').

A third group of early inhabitants of Sikkim are the Limbu. A legal charter dated 1663 records an agreement between the three groups Bhutia, Lepcha and Limbu to unite under the first Bhutia king Phuntsog Namgyal (WT *phun-tshogs rnam-rgyal*) who reigned 1642-1670²¹ (Mullard 2011: 5). In Denjongke and Tibetan, this agreement is known as $\frac{2}{3}$, $\frac{2}{3}$,

²⁰ Male ritual specialist of the Lecphas (Balikci 2008: 378).

²¹ According to the traditional account. Mullard (2011) argues for the years c. 1646 - c. 1670.

1.1.6 From spoken to written language

Until Sikkim became part of India in 1975, Denjongke was exclusively an oral language while Classical Tibetan was used for writing (Dewan 2012: 171, 418). Under Indian rule, Denjongke, along with ten other minority languages of Sikkim, was gradually introduced as an elective subject in schools. In many localities such as Tingchim (North) and Lingdum it was not until the late 1980s that vernacular language classes in Lhoke were introduced in schools (Balikci 2008: 327). For this purpose a literary form of the language was needed. Through the efforts of *Norden Tshering Bhutia* مَتَرَجَعَ عَتَرَجَعَ عَتَرَجَ عَتَرَجَعَ عَتَرَجَ عَتَرَجَعَ عَتَرَجَعَ عَتَرَجَعَ عَتَرَجَعَ عَتَرَجَعَ عَتَرَجَ عَتَرَجَعَ عَتَرَجَ عَتَرَجَ عَتَرَجَعَ عَتَرَجَ عَتَرَجَعَ عَتَرَجَ عَتَرَجَ عَتَرَجَ عَتَرَجَ عَتَرَج

More recently, other types of literature have appeared. In 1996, *Bhaichung Tsichudarpo* المنتخرية المنتخذية المنتخذين المنتخذية المنتخذين المنتذين المنتخذين المنتذين المنتذين المنتذين المن المنتذين المنتخذين المنتخذين المنتخذين المنتخذين المنتذين المنتذين المنتذين المن المنتذين المنتذين المنتذين المنتذين المنتذين المن المنتذين المنتذين المنتذين المنتذين المنتذين المنتذين المن المنتذين المنتذين المنتذين المن المنتذين المنتذين المن

Two newspapers are published in Denjongke. The Department of Information and Public Relations, Government of Sikkim, publishes a weekly paper called Sikkim Herald. The first issue appeared already in 1956 in English and in 1962 the paper began to be published also in Nepali and Tibetan. Denjongke language editions became possible after the 1975 merger with India, when Denjongke became a written language. Nowadays, Sikkim Herald publishes news translated from English in all the eleven official languages of Sikkim. The second Denjongke newspaper is $\neg \forall a$.

Currently, the Bhutia Language Website Development Committee (ﷺ شَامَةُ اللَّذِي اللَّهُ اللللَّهُ اللَّهُ الللَّهُ اللَّهُ اللَّ

For a technical introduction to how Denjongke writing differs from Classical Tibetan and how it is applied in this grammar, refer to §1.2.7 below.

1.1.7 Religion

According to the State Socio-Economic Census 2006 (as cited in Tsichudarpo 2018:46), the great majority (97,79%) of Denjongke speakers are Tibetan Buddhist. Other religions having adherents among the community are Christianity (1,19%), Hinduism (1 %) and Islam (0,02%). The main sects of Tibetan Buddhism followed in Sikkim are Nyingmapa and Kagyupa, which are distinct from the Gelukpa sect led by the Dalai Lama.

Sikkim is known in Denjongke and Tibetan as *bey:* (WD $\mathbb{R}^{\mathbb{N}} \mathbb{Q}^{\mathbb{N}}$ *sbas-yul*) 'hidden land', one of the sacred valleys which the 8th century Indian tantric Buddhist master Padmasambhava, also known as Guru Rimpoche, is said to have specifically blessed as gateway places where physical

and spiritual world overlap.²² Moreover, Sikkim is viewed by many Denjongpos as a paradise on earth whose fate foretells the fate of the whole world (Yongda 2016). For a more detailed description of religion among the Denjongpos, refer especially to Balikci (2008) but also to Yliniemi (2018).

1.1.8 The honorific system and social relations

Similarly to many other Tibetic languages, Denjongke uses an honorific system to give linguistic expression to social hierarchy. A central feature of the honorific system is the division of many nouns, personal pronouns and verbs into two groups, the "ordinary" words known as ceyke? and the simple or $p^{h}e(l)ke?$ and the superiors of the simple of $p^{h}e(l)ke?$ and the honorific words known as ceyke? and social superiors by using the honorific forms with them. The inhabitants of Lachen and Lachung in the north, however, are known for their more direct way of speech in which honorifics are used less frequently. Therefore the speech of Lachenpas and Lachungpas seems offensive to many more southern speakers. The speech of the southeners, on the other hand, may seem too slow and wordy in the ears of the Lachenpas and Lachungpas. A consultant from Lachen commented that the speech of the southern Denjongke speakers makes him feel drowsy. As an example of the difference between the speech in Lachen/Lachung and the speech in West Sikkim, consider the trisyllabic question (1.1), which one might hear in Lachung, and the eight-syllable equivalent (1.2), which one might hear in Tashiding.

(1.1) قَحْرَ مِعْ? *tc^hø? k'a: gju*?
2SG.L where go
'Where are you going?'

(1.2) هُجَ بِهِمَ مَّاجَ جَعَ مَّ مَعَ جَنَ مَعَ جَنَ (مَن)? *lɛŋɛɛ? k'ana tɛ'øm-bo nấُ:-do (bo)*? PRN.HON where go.HON-2INF do.HON-IPFV (EQU.NE.Q) 'Where are you going?'

The ability to use the honorific forms is generally considered a sign of skillful language use. Many young speakers who are unable to use the honorifics correctly are ashamed to speak the language. The honorific words are often identical with Classical Tibetan and with the honorific varieties of other Tibetic languages. Hence, one of my consultants, who is educated in Classical Tibetan, said that he is able to understand the Dzongkha in televised sessions of the Bhutan Parliament but talking to a Dzongkha-speaking farmer would be more difficult.

1.1.9 Language contact and multilinguality

Most Denjongke speakers are to some degree bilingual in Nepali, although some elderly people in some rural communities may be monolingual Denjongke speakers. Domains of language use

 $^{^{22}}$ A Tibetan monk's attempt to open the passage to the spiritual realm through a gateway location in Sikkim is recorded in Shor (2017).

²³ A few verbs have an additional humilific form, which signifies that the speaker acknowledges the addressee's equal or higher social status, see §3.3.4.

among Denjongke speakers are being lost to Nepali, English and Hindi. Nepali is used in the dayto-day life in the market and in offices, where the Denjongpos communicate with members of other communities and also with members of their own community who do not speak Denjongke. English is the official medium of instruction in all schools. Although school books are in English, oral instruction, because of the teachers' weak command of spoken English, may be given in Nepali. Hindi dominates the domain of entertainment (TV, music) and is considered important for career prospects in other places in India. The main domain for using Denjongke is the home, but even there the language in most cases is not being successfully transmitted to children because of the educational choices outlined in the next section. Historically, Denjongke elites have looked up to Tibet for cultural and linguistic influences (Balikci 2011: 11-12). The perceived prestige of Tibetan is well illustrated by the words of one of my elderly consultants, according to whom the main reason for learning Denjongke was that, as a "gateway language," it would later facilitate the learning of Tibetan, the source of religious heritage.

1.1.10 Language endangerment

Denjongke is rapidly losing speakers among children. For this reason, the language has been characterized as "severely endangered" (Turin 2014: 384) and "moribund" (van Driem 2007: 312). In terms of EGIDS²⁴ scale, the status of Denjongke is on level 6b or 7, depending on the community. Level 6b, termed "threatened," is characterized in the following way: "The language is used for face-to-face communication within all generations, but it is losing users." In the country-side, especially near big monasteries, the language is still to some degree being transmitted to the children. Language status level 7, termed "shifting," has the following description: "The childbearing generation can use the language among themselves, but it is not being transmitted to children." This may be the situation with most Denjongke speakers.

Within UNESCO's Language Vitality and Endangerment framework, see Table 1.2, the language may be described, depending on the community, as "vulnerable," "definitely endangered" or "severely endangered." The description "vulnerable" is applicable to some rural settings (such as the monastery hill in Tashiding), while "severely endangered" characterizes the situation of many urban Lhopos (for instance in Gangtok). In my estimation, the majority of children do not currently learn to speak the language, thus "definitely endangered" is an apt classification.

safe	language is spoken by all generations; intergenerational transmission is uninterrupted
vulnerable	most children speak the language, but it may be restricted to certain
	domains (e.g., home)
definitely endangered	children no longer learn the language as mother tongue in the home
severely endangered	language is spoken by grandparents and older generations; while the parent generation may understand it, they do not speak it to children or among themselves
critically endangered	the youngest speakers are grandparents and older, and they speak the language partially and infrequently
extinct	there are no speakers left

T 11 12 INFOCO	у т	\mathbf{x}_{1} , \mathbf{x}_{1} , \mathbf{x}_{1} , \mathbf{x}_{1} , \mathbf{x}_{1}	F 1	C 1
LADIE I A LINENCO	s Language	Vitality and	Endangerment	Tramework
1 abic 1.5. OI LDCC	5 Language	v manny and	Lindangerment	mannework

²⁴ https://www.ethnologue.com/about/language-status (consulted 13 Feb 2018)

With the lack of language competence among younger Denjongpos, the concept of "mother tongue" is being interpreted in the sense of historical and ethnic belonging. The number of Denjongpo who claim Denjongke/Lhoke/Bhutia as their mother tongue exceeds the number of those who claim the ability to speak the language (Turin 2011: 136, Turin 2014: 384).

Some of the reasons for language endangerment, which I have heard from the speakers themselves, are children's schooling outside the language community, fear of losing face, and the lack of economic value in knowing Denjongke. First, private schools, which are preferred to local government schools, are typically situated far away from the pupils' homes. Therefore the pupils have to live outside their homes and lose connection with the language community. Furthermore, unlike government schools, private schools are not obliged to offer Denjongke or the other minority languages of Sikkim as subjects. The number of Denjongke speakers has declined since the 1960, whereas the number of Lepcha speakers has remained fairly stable (Turin 2014: 385). Turin (2014: 385) suggests that the difference may be due to the greater wealth, more education and urbanisation of the Lhopos as compared to the more rural Lepchas.

Second, the fear of losing face is a serious factor in language loss, causing younger speakers to reply in Nepali to their parents when asked a question in Denjongke. One particular linguistic factor, which exacerbates the situation, is the difficulty of the honorific system, where, in order to establish oneself as an esteemed speaker, one has to master two different sets of vocabulary (see §3.2.2 and §3.3.4). Lack of training in the use of the honorific forms, which should be used with one's superiors, along with an occasional rebuke for not showing respect linguistically, has resulted in some younger speakers avoiding using Denjongke at all.

The third factor contributing to language endangerment is that the speakers do not see any economic value in knowing Denjongke. For this reason, even the children of Denjongke language teachers, do not choose Denjongke as an elective subject at school but rather go for languages of more economic prospects, Nepali, English and Hindi.

Although the language is threatened, recent years have seen a thrust toward revitalization among some members of the community. Language and culture oriented Whatsapp-groups have been established. Dictionaries, vocabularies and poem collections of various sizes have been published. Audio and video recordings of traditional songs have been produced. In 2017, the first Bhutia film *byakay* (WD Spar *b'ya-skyel* /p'jake:/ 'chicken-bringing'²⁵) was released. Currently, the Bhutia Literary Association (BLA) is preparing website, which contains cultural and linguistic information about the Denjongpos and Denjongke.

1.1.11 Central linguistic features

This sections provides an overview of the central linguistic features of Denjongke. In the present analysis, Denjongke has 43 consonants and eight vowels (or 13 if lengthened vowels are counted

²⁵ The name refers to the customs of bringing chicken meat to the family of a new-born baby.

separately). Both length and nasalization are contrastive in vowels. Denjongke words are divided into high and low register based on pitch and voice quality. The register of a word is predictable with some initial consonants. The unpredictability of register with other initials and lack of clear voicing difference (breathy vs. modal) leave pitch to be the main contrastive feature in some minimal pairs. Therefore, Denjongke may be termed a tone language, although tone does not bear as great a functional load as in some more well-known tone languages (e.g. Thai). Denjongke syllable structure (C)(G)V(C/V) is more simplified than in phonologically more "archaic" (i.e. more like Written Tibetan) Tibetic languages such as Ladakhi, Balti and Amdo.

The distinction between ordinary and honorific language is seen on many levels. Denjongke makes a distinction between ordinary and honorific nouns. The honorific forms may be derived from ordinary forms but may also be morphologically unrelated. Many verbs also have morphologically unrelated ordinary and honorific equivalents, and some verbs have, moreover, humilific forms. Even those verbs which do not have honorific counterparts can be formed into honorific constructions through a specific nominalized construction. The language also has honorific clitics which attach to the end of the sentence or to a noun phrase which refers to people.

The present study presents 45 phonologically related pairs of controllable vs. non-controllable verbs. Denjongke has a lexical class of adjectives most of which are derived from stative verbs through various adjectivizing morphemes. Similarly to Dzongkha (van Driem 1998), Denjongke does not make a distinction between inclusive and exclusive first person plural pronouns, unlike many other Tibetic languages (see Hill [2010], Bielmeier [1985: 76], Ebihara [undated], Bartee [2007:108], Haller [2000: 50] and Vesalainen [2016: 21]). For numerals, both decimal and vigesimal systems are in use. Denjongke has a morphologically, phonologically and semantically distinct class of adjectives/adverbs termed ideophones, which are used for vivid representation of ideas.

The prominent word order in Denjongke is verb-final APV (or SOV), although right dislocated elements occasionally occur after the verb. Denjongke argument marking, which cannot be characterized as either nominative-accusative or ergative-absolutive, shows a pragmatically conditioned agentive-marking pattern. With some transitive verbs, agentive marking of an A argument is obligatory in past tense whereas other transitive verbs are exempt from such a requirement. The marking of argument P is sensitive to animacy, identifiability/specificity and affectedness of the referent. Denjongke does not have a separate passive construction but functional passives can be formed by suppressing the A argument. Grammatical and spatial relations are marked by five case-marking clitics, some of which can be stacked for double or even triple case marking.

Denjongke has a rich array of simple and complex copulas which mark the basic evidential distinctions of personal, sensorial and neutral. The copulas also mark evidentiality as auxiliaries in periphrastic constructions with various tense, aspect and modal values. The use of personal evidentials is more semantic and less restricted by the requirement for the 1st person to be syntactically present than is recorded to be the case with the egophoric category in Standard Tibetan (Tournadre & Dorje 2003). The present analysis lists 24 secondary verbs which add semantic nuance to the primary verb. The semantic effect of secondary verbs can in most cases, but not always, be described in terms of tense, aspect and mood. Secondary verbs, affixes and nominalized verbs accompanied by copular auxiliaries form a tense-aspect system of nine past-oriented constructions, seven present habitual and future constructions and five progressive/imperfective

type of constructions. Modality is expressed by 11 secondary verbs but also four other constructions.

Interrogation, which occurs in direct and attenuated forms, is accomplished by interrogative copulas and interrogative suffixes, which form a rather complicated system. Denjongke is a clause-chaining language where one sentence has only one finite verb. Genitivization of nominalized clauses is used as a strategy for forming relative clauses, noun complement clauses and postposition complement clauses.

1.2 Research approach and data

This section describes the research approach and data of this grammar. I begin by outlining which language varieties are covered by the present research (§1.2.1). The theoretical background is then addressed in §1.2.2 and the types of spoken and written data in §1.2.3. This is followed by a discussion on methodology and software (§1.2.4) and technical descriptions of equipment used in recording (§1.2.5). Section §1.2.6 introduces conventions used in linguistic examples. The following section §1.2.7 gives details of choices made in using the Denjongke script. Conventions of transliteration are introduced in §1.2.8. The last section §1.2.9 describes how the data and discussion are organized in the remaining chapters.

1.2.1 Dialects and the language described here

As far as I understand, Denjongke varieties spoken all over Sikkim are mutually fairly easily understandable, although some variance occurs in phonology and lexicon. As an example of difference in the phoneme inventory, the consultant from Ralang (West Sikkim) merges /z/ and /dz/ into one phoneme /dz/. As another example of a merger, /ã:/ and /õ:/, which are distinguished in West Sikkim merge into /õ:/ in the East and the North. These features signal the geographic area the speaker comes from but are not, to my knowledge, functionally of much import, because the resulting homonymy is rare. A conspicuous morphological point of variance is the conditional marker, which is /no/ or /nu/ in East and North Sikkim but /nɛ/ or /ni/ in South and West Sikkim.²⁶

The greatest dialectal difference seems to occur between the language spoken in the northern villages of Lachen and Lachung and those varieties spoken in more southern locations. As noted above, Lachenpas and Lachungpas are less likely to use honorifics than the speakers elsewhere. Table 1.1 presents some illustrative differences mainly in the phonology but also in the lexicon and grammar of the language varieties spoken in Lachung and Tashiding.

²⁶ Consultant KUN told me that because of the variation in pronouncing the conditional form, the standardizers of the writing system decided to write the conditional, similarly to Classical Tibetan, as /na/ (\underline{a} , *na*), which, as far as I understand, does not reflect the spoken language of speakers from any location. Nowadays the form /na/ can occasionally be heard in the spoken language of some literate speakers.

Lexeme	Tashiding	Lachung
'after'	gjablɛ	$\epsilon \dot{y}$: $l\epsilon$ (the same as in Dzongkha)
'flower'	mìnto?	<i>mè:to?</i> (Central Tibetan pronunciation)
'tradition'	lùksø:	lòkso:/lòːso:
'coral'	p`juru	tcuru (Central Tibetan pronunciation)
Other		
morpheme		
plural	=tsu	<i>=tso</i> (Central Tibetan pronunciation)
pre-verbal	is not used	is used
negation particle		
<i>á</i> (see §11.1.1.5)		

Table 1.4. Some linguistic differences between Tashiding and Lachung

When visiting the Ha region of Bhutan, which is the closest part of Bhutan to Sikkim, I played to some locals audio samples of Denjongke spoken in Tashiding (West Sikkim) and Lachung (North Sikkim). The group of hearers commented that they could understand most or everything of the Lachung consultants' speech, whereas understanding the Tashiding consultant was more difficult. This linguistic fact provides evidence that supports the local belief in Lachung that the people there have originally come from the Ha region.

This grammar is based on data from a large number of consultants from various geographic locations, see Table 1.2. Therefore the present description is richer in terms of geographic variety than a description based on a very limited geographical location such as one village. Including this broad geographical area was also made necessary for practical reasons: it was not possible to stay in one Denjongke speaking village for extended periods of time and it proved more feasible to work less intensively with several consultants than more intensively with one or two consultants. This geographically/dialectally eclectic approach did not seem problematic since the basic syntactic constructions are fairly unified across various localities, although some constructions may be more frequent in one location than in another. On the positive side, the current approach provides a fuller picture of the use of Denjongke because variation is noted and, to the degree of my awareness, linked with certain geographic locations. Nevertheless, this grammar focuses not on $j\hat{a}:k\epsilon^2 = 4\pi \sin^2 f^2$ 'language of the highlands', which is the description of some Lachenpas and Lachungpas of their own language, but $m\hat{a}:k\epsilon^2 = 4\pi \sin^2 f^2$ 'language of the lowlands', a word used by some higlanders of those living at lower altitudes and in more southern locations.²⁷

The greatest contributor to the written Denjongke data used as data in this grammar is Bhaichung Tsichudarpo, whose works make up approximately 90% of the digitized written data (134 pages and 56,474 words according to MS-Word). As a result, Tsichudarpo's place of origin, Yangang (South Sikkim, see Map 1.3), is well represented in the data. It is, however, worth noting that none of my consultants from whom spoken data was collected hails from there.

Of the Denjongke speaking village locations, my research has most been affected by Tashiding (West Sikkim), where I stayed, on several occasions from 2012, for a total of more than ten weeks. Another field location was Upper Martam, where I stayed six weeks in 2013 and visited several times after that. From June 2013 to August 2016 and again from November 2017 to July 2018, I

²⁷ I record, however, some examples from speakers of Lachen and particularly Lachung.

stayed mainly in Ranka, East Sikkim, from where I could visit and work with several Denjongke teachers living in and around Gangtok.

1.2.2 Theoretical background

This grammar is based on the functionalist-typological approach embodied in such works as the edited Shopen (2007) trilogy and informed by fieldwork guides such as Newman & Ratliff (2001). The work may also be characterized as an application of Dixon's (1997:128) "Basic Linguistic Theory," aiming to combine maximal intelligibility, which results from basing the description on traditional grammatical terms, and nuanced analysis, which is enabled by recourse to up-to-date typological information. In this way, I hope the grammar will be more widely understood than when basing it on a more marginal framework (Noonan 2006).

The approach here relies on the power of prose, instead of formalisms, to describe linguistic phenomena. While using traditional grammatical terms in describing phenomena in Denjongke, I have tried to keep in mind Haspelmath's (2010a) warning that analysis should not be limited even by the conceptual framework of such eclectic approaches as Basic Linguistic Theory. The aim has been to describe Denjongke on its own terms, making a distinction between language-internal descriptive categories and cross-linguistic comparative concepts (Haspelmath 2010b). Moreover, this grammar is data-oriented in that I aim to give a lot of examples, which not only illuminate the points under discussion but may also provide insight into other topics that have not been covered or envisioned by the present author.

1.2.3 Data

The data for this grammar come from both oral (§1.2.3.1) and written sources (§1.2.3.2).

1.2.3.1 Oral data and consultants

The data were gathered initially during six weeks in 2004 (mainly for my MA thesis) and later extensively between 2012-2018, during which I spent more time in the language area (Sikkim) or its vicinity (Kalimpong, West Bengal) than outside of it. Before 2012 I had achieved some oral competence and literacy in Nepali and so was able to use Nepali as a contact language in my initial approach to Denjongke speakers. Gradually, I was able to achieve some competence in speaking, listening, reading and writing Denjongke and so shifted to a monolingual method where all sessions were conducted in Denjongke. Developing speaking ability in the language enabled me to use participant observation as a means of data gathering and I was also able to use recordings of the elicitation sessions as data.²⁸ Developing my own intuitions about the language also helped make the analysis more reliable. Although many example sentences in the present work are marked as elicited, the elicited examples are almost never translations from another language but responses to prompts in Denjongke or, with two consultants, in English. The common languages between me and the consultants were mainly Denjongke and Nepali. Only two of my occasional consultants (PT and KT) were competent in English. An illustrative collection of Denjongke texts is provided in Appendix 1. I am planning to make later available some material from Appendix 1 and also other analyzed texts through archiving (preliminarily in FINCLARIN's language bank²⁹). The following types of spoken data have been used in the analysis and examples:

²⁸ For a defense of the monolingual method, see Everett (2001).

²⁹ https://kitwiki.csc.fi/twiki/bin/view/FinCLARIN/KielipankkiFrontpage

1) Video recorded monologues, interviews and songs

2) Audio-recorded monologues, interviews and songs

3) Audio-recorded conversation of several people recorded by a consultant (KN) in the author's absence

4) Audio-recorded conversation of several people recorded by the author

5) Audio-recorded elicitation, language learning and conversation sessions with the author present and occasionally absent for a time

6) Words, minimal pairs and sentences elicited for phonological analysis

7) Existing Denjongke song albums

8) About 1200 pages of field notes, which include notes from recorded and non-recorded elicitation, overheard language, notes from audio and video-recordings which have not been transcribed, examples from written sources which have not been digitized, initial transcription and interlinearization.

9) One example, (9.58b), is taken from Kunzang Rapten Denjongpo's 2021 short film and gradient gnyid-lam /pílam/ 'dream' (1:12), see https://www.youtube.com/watch?v=SBp2FAZ94AY

The monologues mentioned in 1) and 2) above include such categories as folk-story, travel story, remembering old times, description of a cultural tradition, joke, riddle, proverb with or without explanation, speech in a formal setting, pedagogical speech, and the pear story.³⁰

I was fortunate to work with many consultants. Almost all the consultant signed an informed consent form, see Appendix 3. With some consultants, an informed consent paper was not available at the time. In these cases, I believe no harm is done to their person, because the consultants' names are not identified and data received from them is not archived. Although most consultants indicated that they may be identified by their name, I decided to introduce the consultants by a combination of two or three capital letters, because being identified by name may in some cases have unforeseen consequences, especially if the topic under discussion is in one sense or another sensitive. The consultants from whom spoken data used in this grammar have been obtained are presented in Table 1.2, along with the types of data gained (m.=male, f.=female). The places of origin of the consultants are shown in Map 1.3 under Table 1.2. Table 1.2 does not present all my data but only those recordings from which example sentences were taken. The recordings referred to in Table 1.2 are audio-recordings, unless separately marked as "video."

Code	Gender,	Location	Type of data
	age		
BT	m. 50+	Tashiding	Grammar exposition; formal exposition (monologue) on
			the correspondence of English and Denjongke verbal forms
			(13:32)
DB	m. 60+	Tashiding	1) Life story (4:15)
			2) Trip story; story of a one-day tour in West Sikkim taken
			the previous day (14:43 min)
GB	m. 40+	Tashiding	Unrecorded elicitation

Table 1.5. Consultants and types of data

³⁰ The pear story is a story prompted by showing a consultant the Pear Film, a six-minute film produced at the University of California at Berkeley in 1975 for studying narrative crosslinguistically, see Chafe (1980). The Pear Film is available at: http://pearstories.org/.

JD	m. 15	Tashiding	Life story (00:47)		
JDF	m. 40+	Tashiding	Axe story; famous pedagogic story of a farmer who lost an		
			axe and regained it with the help of a water-god (3:52)		
JDG	m. 70+	Tashiding	Field notes of unrecorded elicitation		
NAB	m. 30+	Tashiding	BLA 7, recorded formal talk in a meeting (28:54)		
NB	m. 30+	Tashiding	Recordings of elicitation sessions		
PAD	m. ?	Tashiding	1) Bet story; a story sent in a whatsapp group about a bet		
			between a crafty peddler and an intelligent farmer (17:10)		
			2) Tashiding story; story of one clan living in Tashiding		
			(3:00)		
PED	f. 30+	Tashiding	Life story (16:21)		
PT	m. 30+	Tashiding	Kitchen discussion, free discussion between up to six		
-			people (three recordings altogether 1:44:09)		
PTA	f. 60+	Tashiding	Kitchen discussion, see PT		
PTM	f. 60+	Tashiding	Kitchen discussion, see PT		
PTW	f. 30+	Tashiding	Kitchen discussion, see PT		
RB	m. 20+	Tashiding	1) Pear story (1:50)		
			2) Butcher story (2:04)		
RBM	f. 60+	Tashiding	1) Roof discussion; free discussion between four relatives		
			on the roof (4:34)		
			2) Story of my son (1:27)		
RS	m 60+	Tashiding	1) Driver joke (2:38)		
			2) Pupil joke (3:00)		
			3) Language situation; monologue on the language		
			situation from Denjongpo perspective (14:12)		
			4) Bee story; a story of a competition between a bumble		
			bee and balsam flower $(3:24)$		
			5) Song intro; recording of a song with introduction (4:38)		
			6) (in)auspicious days; an exposition on astrology (15:07)		
			7) Intro to duetto; recording of a song with introduction (4.29)		
			(4.50) 8) Animal song intro: recording of a song with introduction		
			(8.32)		
			(0.52) 9) On songe: introduction to types of songe in general		
			(2.00)		
SM	m 50+	Tashiding	Kitchen discussion see under PT		
SN	f_{50+}	Tashiding	Kitchen discussion, see under PT		
UT	m_{30+}	Tashiding	1) Recorded elicitation session		
01		1 wonnening	2) Proverb: recorded proverbs with explanations		
			3) Riddle: recorded riddles with explanations		
UTR	m. 20+	Tashiding	Plains story: personal story of living in the plains of India		
			(6:57)		
UU	m. 40+	Tashiding	Deer story; folkstory of a deer, given fully in the appendix		
			(1:36)		
YB	m. 20+	Tashiding	Restaurant discussion; recorded discussion of several		
			people in a restaurant (including the author)		

TB	m. 40+	Ralang	1) Story of two bulls (2:33)
		U U	2) Story of the Buddha (1:43)
			3) Phone call; telephone conversation with a brother who
			is in Delhi (3:42)
			4) Funeral customs; a monologue exposition on funeral
			customs in Ralang (1:28)
			5) Life story $(2:40)$
			6) Life in gumpa (monastery) (1:55)
			7) Comment on a video: recorded descriptions on what
			happened on videos
			8) Phone call: unrecorded phone call, field notes
BB	m. 30+	Barapathing	BB discussion, see KL
BBP	m 70+	Barapathing	BB discussion, see KL
RP	m_{30+}	Barapathing	BB discussion see KI
KI	m 10^{+}	Barapathing	1) BLA 12 discussion with some younger speakers mostly
IXL.	III. 4 0+	Darapating	monologue (15:12)
			2) Discussion with DR a free discussion between two
			2) Discussion with DR, a nee discussion between two
			3) BB discussion: a group discussion of several people in
			Barapathing
			1) Phone call (within a longer eligitation recording)
V T	m 601	Dormoolz	1) Animal story falkstory on how a marten killed an
ΓI	III. $00+$	Defineok	1) Annual story, lockstory on now a marten kined an
			2) Discussion with TD a free discussion between two
			2) Discussion with TB, a free discussion between two
			people (several recordings of discussion)
			3) Intro to an ode; recording of an ode with an introduction
			(3:06)
VD	40	T7 ·	4) Recorded phone call (5:49)
YR	m. 40+	Kewsing	1) Canteen video; pedagogical exposition prompted by
			questions (22:24)
			2) Boys' and girls' clothing; an exposition prompted by a
			question (3:26)
CY	m. 70+	Pemayantse	1) Structured video interview conducted by KN (1:20:17)
			2) Monologue before interview (12:58)
SGD	m. 50+	Barphung	1) Monologue exposition of wedding customs (28:19)
			2) Cave story; story of Sikkimese caves (5:21)
AB	m. ?	Martam	Kitchen discussion, see KN
KN	m. 20+	Martam	1) Kitchen discussion, a discussion by KN's family,
			recorded by KN
			2) Phone call (field notes 5, 100)
			3) Photo discussion; consultant's recorded responses to
			photos
			4) Phone call 2 (1:15)
			5) Phone call 3 (field notes)
KNA	m. 70+	Martam	Kitchen discussion, see KN
KNM	f. 60+	Martam	Kitchen discussion, see KN

KNU	m. ?	Martam	Kitchen discussion, see KN				
LT	m. 30+	Martam	Kitchen discussion, see KN				
DR	m. 70+	Phodong	Discussion with KL, a free discussion between two people (33:45)				
KUN	m. 30+	Lachung	Recorded elicitation session (1:43:18)				
LA	f. 60+	Lachung	 Intro to Lachung; a monologue which introduces life in the northern village of Lachung (5:07) Birth in Lachung; an exposition on customs relating to birth of a baby (2:12) Funerals; monologue on funeral customs in Lachung 				
			(4:15)				
PD	m. 40+	Lingdum	 Intro video; an introduction to a house (1:02) Living room video; an introduction to a room in a house (00:55) Outside video: an introduction to some items around the 				
			house, prompted by questions (6:35)				
			4) Altar room video (4:27)				
			5) Goat shed video (1:20)				
			6) Surroundings video; an introduction to the surroundings of a house (1:37)				
			7) Storeroom video (2.29)				
			 8) Spatial topography interview; based on pictures (1:07:04) 				
			9) Interview; structured bilingual (Nepali, Denjongke) interview on everyday life of the consultant's family (1:42:22)				
SS	m. 50+	Lingdum	Proverb explanation; recording of a proverb with explanation				
PL	f. 30+	Lingdok	Interview on farming conducted by the present author (36:11)				
KLT	m. 50+	near Tashiding	Bumchu video; an exposition of the origin of the Bumchu festival in Tashiding produced by Namgyal Institute of Tibetology				
RL	m. 40+	Lachen	Several recorded interviews with the author				
PB	m. 20+	Gyalshing	Discussion with TB (16:21)				
LT	m. 30+	Rinchenpong	Unrecorded elicitation				
RD	m. 50+	Phodong	BLA 9; recorded formal talk in a meeting (9:01)				
DL	m. ?	?	About food; post in a Whatsapp group (2:25)				
NT	m. 70+	Tathongchen	BLA 6, recorded formal talk in a meeting (10:20)				

Overheard clauses are marked "oh" and assigned a place of hearing. If an example sentence originally occurred in a message to a Whatsapp group, it is separately mentioned. Functionally, language material from a large Whatsapp group where all participants do not know each other could be classified as "overheard." For some simple examples no source is marked.

The aim of the research has been to treat the consultants in conformity to the three basic ethical principle described by TENK (Tutkimuseettinen neuvottelukunta, Finnish National Board on

Research Integrity), i.e. respecting the autonomy of the research subjects, avoiding harm and protecting privacy.

Map 1.3. Consultants' places of origin within Sikkim



1.2.3.2 Literary data

Several written works are used as data. In the written data, works of Bhaichung Tsichudarpo take precedence. His works, most of which are designed as audio-plays, have a lot of dialogue and use of colloquial expression. His texts represent the actual spoken language and are said to be intelligible for ordinary villagers.

Works by Bhaichung Tsichudarpo (WD ธูลิ ซูร ซัญ จรูร รัง bha'i-cung tshes-bcu-dar-po):

1) A novel called Ref. richhi /ritchi/ 'hope', 173 pages, see Tsichudarpo (1996).³¹

2) A play called an if mam-rtog /námto?/ 'superstition', 42 pages, see Tsichudarpo (1997).

3) A play called sign 'nga'i 'gan /nè: gɛn/ 'my responsibility', 27 pages, within Tsichudarpo (2008).

4) A play called aga gar mthun-sgril /thyndi:/ 'unity', 19 pages, within Tsichudarpo (2008).

³¹ This is the first and, thus far, the only Denjongke novel.

6) A collection of folk-stories and moral teachings $\frac{1}{5}$ and $\frac{1}$

Work by Sonam Gyatso Dokhangba (WD مَعْمَ مَعَمَ عَلَى اللهُ عَلَى اللهُ عَلَى اللهُ مُواللهُ اللهُ مُحَاللهُ اللهُ اللهُ اللهُ اللهُ اللهُ مُحَاللهُ مُواللهُ اللهُ مُحَاللهُ الللهُ اللهُ الل

7) A description of Denjongpo marriage customs artight and the sharphung ling-dam 'grolis /barphung lingdam doli?/ 'the custom of Barfung-Lingdam' (English name: Sikkimese marriage custom and rites), 143 pages, see Dokhangba (2001). In example clauses this work is referred to simply as *sbar-phung*.

Work by Pema Rinzing Takchungdarpo (WD रान् अ' रेग् तर्हेन झुग् छून न्र रॉ' pad-ma rig-'dzin stag-cungdar-po):

8) Class 12 textbook तज्ञअ ब्रेन्ज गहान गहान 'bras-ljongs gsung-gtam 'Stories (from) Sikkim', 73 pages, see Takchungdarpo (1987).

A "compilation" by Tshering Thendup Bhutia (WD ঊ দিন' দ্বায়ুন' এর্ষা র্ট্রন্মের্ঘ tshe-ring don-grub 'bras-ljongs-po) and Thupten Palzang Bhutia (WD দ্রন'নঙ্গুব্ দেশ'নরন' এর্ষা র্ট্রন্মের্ঘ thub-bstan dpalbzang 'bras-ljongs-po):

9) A discussion on language situation in the world مَعْرَبُ عَبَّلَ عَبَّلَ عَبَقَ عَبَقَ عَبَقَ عَبَقَ عَلَيْ عَامَ عَلَيْ مَا يَعْلَيْ عَلَيْ عَلَ

A compilation³² of Tashi Denjongpo (WD স্মাণ্ট্ৰমা নহামান্ট্ৰমোহ্য bkra-shis 'bras-ljongs-po), Pema Rinzing Takchungdarpo (WD স্বাজ মিশ্বেইর স্থ্রশাস্তদ্বেমার্ঘ pad-ma rig-'dzin stag-cung-dar-po) and Bhaichung Tsichudarpo (WD হ্লবিস্তদ স্থ্রাস্তদ্বেমার্ঘ bha'i-cung tshes-bcu-dar-po):

10) Class 7 textbook ਡ਼੍ਰਾਘੇਗ੍ਰਾ ਡ਼੍ਰੋਕਾਜ਼੍ਰੇਕਾ ਕਸ਼ੂਰਾਜ਼ *lho-yig slob-deb bdun-po*, 71 pages, see Denzongpo et al (2011).

11) Class 8 textbook ਣ੍ਰਿੰਘੇਗ੍ਰ' ਸ਼ੁੱਤਾਂ ਤ੍ਰੇਤ' ਤਸ਼ੂਨ ਤੋਂ *lho-yig slob-deb brgyad-po*, 63 pages, see Denzongpo et al (2009).

13) Annual magazine of the newspaper নৃ'ল্লন' নহাম' (*da-lto'i 'bras-ljongs* 'Sikkim today') from year 2003.

Of the above works, the following were typed on the computer in order to facilitate computerized searches: all the items 1-5 mentioned under Bhaicung Tsichudarpo, the first story in 6 (28 pages) and a dialogue from 7 (6 pages).

³² I do not know whether the compilers resort to already existing sources in Denjongke, translate from existing Tibetan materials or compose themselves.

1.2.4 Methodology and software

In phonology, the initial analysis was based on a collection of about 1000 words, all of which were recorded separately with two or three repetitions. Some of these words were also recorded in sentence frames to enable research on tone/pitch. Minimal or analogous pairs of words were used to establish distinctive sounds (phonemes). Words were also acoustically analyzed in Praat software to improve understanding on such issues as prenasalization, voicing, aspiration, breathiness, length, tone/pitch and intonation. Diagrams from Praat are presented in relevant parts of this grammar.

At an initial stage of research, i.e. during my MA thesis writing, I used the Toolbox software for storing data. Later, during my PhD research, I shifted to FieldWorks Language Explorer software, which among other things enables lexicon building, text collection, interlinearization and concordance searches of the data, the last of which proved particularly helpful. I first collected elicited data, partly for language learning, and then moved to collecting natural data from various genres of speech. The first recordings were short stories. Later, other types of spoken data were added. The main task was transcribing and translating texts with native speakers. After having acquired some competence in the language, I was myself able to do the initial transcription, which was then checked and supplemented with native speakers. In addition to working with transcribed texts, I listened to untranscribed recordings to spot various constructions and morphemes.

Reading and identifying grammatical constructions in the written sources was first done with hard copies of books. Later, after some literary texts had been typed and stored in an MS Word-file, I was able to do searches for grammatical constructions within the file.

1.2.5 Equipment

The audio recordings of 2004 were done on a minidisc recorder using an external microphone. The recordings in 2012-2018 were captured as WAV-files (either 44.1Hz/24bit or 44.1Hz/16bit) on Olympus LS-10 and LS-11 solid state recorders using the recorder's own microphone. Video files were captured on Canon EOS 700D camera, iPad (3. gen) and Canon Legria-video camera (non-HD). Audio recordings of the video-sessions were made on Olympus LS-11 solid state recorder.

1.2.6 Linguistic examples

Linguistic examples are numbered so that the number before the full stop reveals the chapter and the digits after the full stop show the example number within the chapter, e.g. (4.33) refers to the thirty-third example in chapter four. The same example may occur in different parts of the grammar, illustrating different grammatical points.

Examples consist of four lines, the first line presenting the Denjongke script, the second line the phonological script written in IPA, and the third line morpheme-glosses. Morpheme glosses follow Leipzig glossing rules, supplemented with other glosses not found in the Leipzig rules. All the abbreviations are listed above. The fourth line in examples offers a fairly literal translation into English, aiming to reflect the Denjongke constructions used. English words³³ within Denjongke speech are on the first line written in Denjongke/Tibetan script followed by the same word written in Roman script in brackets, see (1.3).

³³Generally, English words are considered instances of code-mixing, although the most frequently used English words (while talking Denjongke) may approach the status of loan words.

Nepali code-mixed words are written on the first line in Denjongke script with a following (Nep.), which indicates that the previous word represents a Denjongke writing of a Nepali word, see (1.4), where the Nepali word p_Ani 'also' is adapted to Denjongke pronunciation as pun(i).³⁴ Morphemes addressed in the discussion are typically given in bold, see *pun* in (1.4).

(1.4) $\operatorname{gr}\mathfrak{F} \mathfrak{Tap}(x, x, y; \mathfrak{F}) (\operatorname{Nep.}) \mathfrak{F}(x, y, y; \eta)$ $\operatorname{atsi} ro: ram pun p' ja - \varepsilon \varepsilon = lo = ki^{35}.$ a.bit help also(Nep.) do-INF=REP=NC '(He) is also going to help us, I hear.' (TB discussion with KT)

As shown in (1.3) and (1.4), code-mixed words are also indicated on the morpheme-gloss line by brackets after the gloss, e.g. in (1.3) *juniva:siti* is glossed as 'university(Eng.)'. Within the English translation the following items are given in brackets: 1) Elements which are not explicitly expressed by Denjongke but are required by English grammar, see (it) in (1.5), 2) elements which are suggested by Denjongke but not as explicitly stated as in English, see (I saw) in (1.5), 3) elements that help the reader understand the context and meaning of the clause better, see (the price of) in (1.6).

- (1.6) $f_{\nabla} \circ a_{\nabla} \circ \tilde{f}_{\nabla} \circ \tilde{g}_{\nabla} \circ \tilde{g}_{\nabla} \circ \tilde{g}_{\nabla} \circ \tilde{f}_{\nabla} \circ \tilde{g}_{\nabla} \circ \tilde{f}_{\nabla} \circ \tilde{g}_{\nabla} \circ \tilde{f}_{\nabla} \circ \tilde{g}_{\nabla} \circ \tilde{f}_{\nabla} \circ \tilde{f}_{\nabla}$

Three dots (...) in the middle of an example sentence mean that some words have been elided. Incomplete sentences are marked in the examples in three ways: 1) in Denjongke script by ending with the syllable-break marker \cdot (rather than the equivalent of full stop 1), 2) in the phonemic script

³⁴ Similar to English code-mixing, the most frequently used Nepali words may have attained the status of a loan word. In the case of Nepali *pAni* 'also', lexicalization is suggested by adapted pronunciation, *pun*. Typically speakers, however, are aware that *pun* originates in Nepali and that the semantic equivalent in "proper" Denjongke is $=j\tilde{a}$. ³⁵=*ki* is a loan from Nepali, see §16.2.3.

³⁶ The word order in this spoken example is unconventional. The verb complex (here $s\dot{a}$ - $ts^{h}a \ du$ - $k\varepsilon$) typically occurs finally.

by ending with no punctuation (in opposition to marking the end of the sentence by a full stop), and 3) in the free translation by ending with three dots, in opposition to a full stop.

1.2.7 Written Denjongke

This section introduces the special features of written Denjongke in general (§1.2.7.1) and also describes the particular decisions made to represent spoken language in Denjongke writing (§1.2.7.2-4). The remainder of this grammar uses the following abbreviations: WT (Written Tibetan), WD (written Denjongke) and D (Denjongke).

1.2.7.1 Introduction to written Denjongke

The most important WD innovation to the Tibetan writing system is the application of the *tsha-lag* $\mathfrak{s}^{r} \mathfrak{A} \mathfrak{P}^{37}$ (as \mathfrak{s} in \mathfrak{g}) to letters with which it cannot occur in Classical Tibetan. One reason for this innovation was that some of the historical labial-palatal sequences, for instance /pj/ and /mj/, which in Lhasa Tibetan have merged into /tc/ and /n/ respectively, are in Denjongke pronounced as sequences /pj/ /mj/ and written \mathfrak{s} and \mathfrak{s} respectively. Another reason was the need to introduce spellings for frequently heard foreign loan words such as proper names which have such consonantal sequences that appear the WT but have since developed into retroflexes in the inherited lexicon of Denjongke. For instance /kr/, as in Khrishna, and /pr/, as in Pradhan, are now written as \mathfrak{T} and \mathfrak{T} respectively because \mathfrak{T} and \mathfrak{T} without a *tsha-lag* are pronounced as a retroflex /t/. Another innovation in WD is word-breaks (see any example clause). The non-standardized character of the written language is also seen in that some words are spelled in various ways by different authors, and sometimes by the same author, e.g. $\mathfrak{A}\mathfrak{P}\mathfrak{A}$ and $\mathfrak{A}\mathfrak{A}$ for $l\mathfrak{E}m$ 'good'.³⁸

It is important to acknowledge that the WD form given in the example sentences does not claim any degree of standardization. When examples are taken from written sources the spelling follows the original, although the same author may in another passage write the same word in a different way. Some obvious mistakes, however, are corrected (e.g. forgetting a vowel marker from above a consonant).

Another important issue to acknowledge is that when spoken language is represented in WD, the result may seem faulty to those familiar with literary language. This is so because spoken language is produced more spontaneously than written language. The spoken examples may include false starts and have repetition which may seem unnecessary for those who see a written text. The reason for writing the spoken examples in the Denjongke script is to make them more accessible to those who know the Tibetan/Denjongke script but are not familiar with the phonemic script based on the International Phonetic Alphabet. WD is not standardized, especially with respect to word breaks. Therefore whatever principles are used in the present work, they are likely to break some precedent in Denjongke literature. The following sections outline the principles used in representing spoken Denjongke in the Denjongke script in this grammar. The discussion is divided into the topics of word-breaks (§1.2.7.2), syllable merging (§1.2.7.3) and phonological writing (§1.2.7.4)

³⁷ Also known as झुँ हुंगुरुगुरुग कॅ'ल्पग' sgra-rtags tsha-lag.

³⁸ According to an emerging standard, the correct spelling at present is Amar, which retains the etymological connection to the related form Amar' (be) good'.

1.2.7.2 Word breaks

Monosyllabic clause-medial clitics $u_{\Box} = j\tilde{a}$: 'too, even', $z_{\Box} = r\tilde{a}$: (anaphoric emphatic), $\tilde{\beta} = to$ (contrastive emphatic) and clause-final clitics $\tilde{\beta} = co$ (attention marker), $u_{\Box} = la(:)$ (honorific), $\tilde{\alpha} = s\varepsilon/si$ (quotative) and $\tilde{\alpha} = lo$ (reportative) are written together with the word they follow, e.g. $\Box u_{\Box} = j\tilde{a}$: 'I too'. An exception is the demonstrative-emphatic clitic $a\tilde{\beta} = di$, which is, following the novel Richhi, written separately. The reason why di/=di is written separately by Tsichudarpo and other authors is probably that the demonstrative uses, which would naturally occur with a wordbreak, are not always easy to tell apart from emphatic cliticized uses. Cliticized monosyllabic postpositions are written together with the previous word, while disyllabic longer versions are written separately, $\tilde{\beta}_{a} = s\tilde{a}$: 'until home/house', $\tilde{\beta}_{a} = z \tilde{\beta} + khim s\tilde{a}$: 'until home/house', $\tilde{\beta}_{a} = z \tilde{\beta} + khim s\tilde{a}$: 'at home/house', $\tilde{\beta}_{a} = s\tilde{\delta} + khim s\tilde{\delta} + khim s\tilde{\delta}$: 'at home/house', $\tilde{\beta}_{a} = s\tilde{\delta} + khim s\tilde{\delta} + khim s\tilde{\delta}$.

Similarly to Tsichudarpo, double case marking, or case stacking, is written together, e.g. locative following a genitive $\neg a \cdot k^h a = i = na$ 'in the mouth' (occurs also as simply locative $\neg a \cdot k^h a = na$). Serialized verbs are written separately, e.g. $\neg \neg a \cdot k = bak \circ n - di$ (also $ba \cdot on - di$) 'is bringing (lit. carry come-NF)'. Interrogative copulas $\neg na$ and $\neg a \cdot na$, in analogy to copulas and copular auxiliaries are written separately, whereas interrogative suffixes $\neg na/a$ and $\neg a \cdot na/a$ are written together with the previous word.

- - c) גאלי פריד געויד לעויד אויד אויד אויד געויד געוי

The third, practical reason for writing the auxiliaries separately is that this practice enables smoother reading of glosses. If complex auxiliaries were to be written together with verbal suffixes, the resulting verbal units could add up to six or more syllables, causing great difficulty in reading glosses.

1.2.7.3 Syllable merging

In spoken language, the nominalized verb VERB-*po* may be abbreviated to VERB-*b* (with verbs ending in glottal stop or *p* and followed by the equative $b\varepsilon^2$) or to VERB-*m* (with verbs ending in a the velar nasal and followed by the equative $b\varepsilon^2$ or \tilde{t}). The abbreviated form is represented in an experimental way in the Denjongke script by omitting vowel marking from the infinitive marker -*po* $\tilde{\tau}$ and representing the consonant value (either -*b* or -*m*) in the previous syllable (changing τ ' into τ ' because τ ' does not occur syllable-finally), e.g. $\tau \tau = s \varepsilon - po \delta \varepsilon^2$ 'killed' vs. $\tau \tau = s \varepsilon - b \delta \varepsilon^2$ vs. $\tau = s \varepsilon - b \delta \varepsilon^2$ 'killed' vs. $\tau = s \varepsilon - b \delta \varepsilon^2$ 'killed'; $\tau = s \varepsilon - b \delta \varepsilon^2 + \delta \varepsilon^2 + \delta \varepsilon^2 + \delta \varepsilon^2 + \delta \varepsilon - \delta \varepsilon^2 + \delta \varepsilon^2 +$

1.2.7.4 Phonological writing

The conventions of writing Denjongke in this grammar are at times innovative in that I introduce phonological spellings of words and constructions which either do not occur at all in written Denjongke or are written in a more (historically) conservative way, which does not explicitly reveal pronunciation. The reason for using more phonological ways of writing written Denjongke is to give those readers who know Tibetan characters but not the phonetic alphabet better access to spoken pronunciations. An example of an innovative spelling is given in (1.8), where (a) illustrates a written clause from the novel Richhi along with a reading-style pronunciation of written Denjongke in Roman based script. Example (1.8b), on the other hand, presents a corresponding colloquial pronunciation along with the innovative phonemic Denjongke spelling.

(1.8) a) ผู้สาสา กังๆ เฮ้สาร์กั เพิสาสุล?

 $k^{h}im = na$ $l\partial k$ - $ts^{h}a$:-bo *in-nam*? house=LOC return-CMPL-2INF EQU.PER-ATTQ 'I wonder whether (he) has returned home.' (Richhi 24)

b) أَيْهَا عَرَ مَرْسَا عَلَى مَعْمَا عَلَى اللَّهُ مُعْمَا عَلَى اللَّهُ مَعْمَا عَلَى اللَّهُ مُعْمَا عَلَى اللَّهُ مُعْمَا عَلَى اللَّهُ مَعْمَا عَلَى اللَّهُ مُعْمَا عَلَى الْعُمَا عَلَى الْعُلَى الْعُلَى الْعُلَى الْعُلَى اللَّهُ مُعْمَا عَلَى الْحُمَا عَلَى الْعُلَى الْحُلْقُلُولَ عَلَى الْعُلَى الْحُلَى عَلَى الْعُلَى الْحُلَى الْعُلَى الْحُلَى الْحُلَى الْحُلُكُلَى الْحُلَى الْحُلُكُلَى الْعُلَى الْحُلُكُلَى الْحُلَى الْحُلُكُلَى الْحُلَى الْحُلُكُلَى الْحُلَى الْحُلُكُلَى الْحُلَى الْحُلُكُلَى الْحُلُكُلَى الْحُلُكُلَى الْحُلَى الْحُلْعُلَى الْحُلْحُلُكُلُكُلَى الْحُلَى الْحُلْحُلَى الْحُلْحُلَى الْحُلْحُلَى الْحُلْحُلَى الْحُلْحُلَى الْحُلْحُلَى الْحُلْحُلُكُلَى الْحُلْحُلَى الْحُلْحُلَى الْحُلْحُلَى الْحُلْحُلَى الْحُلُكُلَى الْحُلْحُلُكُلَى الْحُلْحُلَى الْحُلْحُلُكُلَى الْحُلُكُ الْحُلْحُلُكُلُكُ الْحُلُكُلُكُلُكُلُكُلُ حُلْحُ عَلَى الْحُلْحُلُلُكُلُكُلُ حُلْحُلُكُ الْحُلُكُ عَلَى الْحُلْحُلُكُ عَلَى الْحُلْحُلُكُلُكُ عَلَى الْحُلْحُلُكُلُ عَلَى الْحُلُحُلُكُلُكُلُحُلُكُ الْحُلُحُلُكُلُ الْحُلُحُلُ عَلَى الْحُلْحُلَى الْحُلْحُلْحُلَى حُلَى الْحُلْحُلَيْ عُ

The spelling innovations in (1.8b) are $\mathfrak{F}_{\mathfrak{T}}$ instead of $\mathfrak{F}_{\mathfrak{T}}$ to represent colloquial pronunciation $-ts^{h}ou$ and $\mathfrak{F}_{\mathfrak{T}}$ instead of $\mathfrak{F}_{\mathfrak{T}}$ to represent the colloquial pronunciation $p\acute{a}m$.

Another phonologically based innovation is to write the final syllable of many nouns as it is heard pronounced in Denjongke (-pu/bu) and not as it occurs in Written Tibetan (-po/bo), e.g. the

word *lómpu* 'minister' is here written as $\arg_{i} \operatorname{sign} blon-pu$ rather than as $\arg_{i} \operatorname{sign} \operatorname{sign} blon-po.^{40}$ The latter spelling is often followed in WD, although some authors are open to the more phonological spelling. Furthermore, one WD form used here which I have not come accross in literature is the double genitive, which is prevalent in spoken Denjongke, e.g. the spoken form $\eta atci=gi$ [we.GEN=GEN] is represented as $\neg_{i} \operatorname{sign} nga-ca'i-ki$, although in written language the simple genitival expression $\neg_{i} \operatorname{sign} nga-ca'i$ would be used instead.

1.2.7.5 Differences between spoken and written language

Generally, written language exemplified by such works as the novel Richhi corresponds to careful spoken language. There are, however, some phonological, morphosyntactic, pragmatics-related and other differences between spoken and written language. These differences are discussed in relevant sections throughout the grammar and summarized in Appendix 2. Unsurprisingly, spoken language is associated with phonological and morphosyntactic reduction. However, in some constructions, spoken language shows morphosyntactic expansion (i.e. more form) and flexibility compared to written language. Certain discourse-oriented morphemes and phenomena are more frequent in spoken language. Finally, spoken language is heavily influenced by Nepali and English, whereas written language looks to Literary Tibetan for guidance. For a fuller description of the differences, see Appendix 2.

1.2.8 Transliteration

			~	-			
Ł	ka	E.	kha	5	ga	Ц	nga
Ø	ca	в	cha	Щ	ja	9	nya
5	ta	Ø	tha	5	da	б	na
7	ра	∇	pha	5	ba	ਲ	ma
ભ્ર	tsa	ક્ષ	tsha	Л	dza	Ħ	wa
Ø	zha	Π7	za	מ	'a	ฮ	ya
۲	ra	ਕ	la	٥	sha	N	sa
5	ha	জ	а				

Table 1.6. The Wylie system

1.2.9 Organization of the grammar

Each grammar-writer must make a decision between formal and functional orientation, which are in tension (e.g. Payne 2014). Formally oriented traditional grammars take linguistic forms as the

⁴⁰ However, if the word occurs in a written source, I follow the original spelling.

⁴¹*Tsha-lag* also occurs with *f* resulting in *f sh ra*.

starting point for organization. The various functions of a certain form are then listed under formal headings. In a functionally oriented grammar, such as the current one, the general direction of description is from function to form. This means that the different functions of the same form or construction may be described under two or more functional heading. Cross-references throughout the grammar will point the reader to other sections where the other uses of a certain morpheme or construction are discussed.

The remainder of the grammar is organized as follows. Chapter 2 deals with segmental and suprasegmental phonology. Morphology, along with etymological information, is presented in §3, which describes word classes, suffixes and clitics. Parts of §3 may seem heavy for readers who are not interested in Tibetic etymologies. Those readers may want to follow the references provided in §3 to functionally-oriented accounts provided in later chapters. Chapter 4 moves the discussion to phrase-level syntax and §5 describes basic clausal syntax. The remaining chapters are organized under functional headings, thus the main direction of presentation is from function to form. Chapter 6 addresses deixis and reference by describing the use of various pronouns and indefinite expressions. The description of evidentiality begins in §7, which discusses copulas, the hotbed of evidentiality in Denjongke. This is followed by a treatment of tense, aspect and modality marking in §8. Chapter 9 reintroduces the topic of evidentiality by showing how it is marked in non-copular clauses by auxiliaries and other markers. The reason why §8 on tense, aspect and mood is placed between the two chapters on evidentiality (§7 and §9) is that it is easier to discuss evidentiality in periphrastic constructions after those constructions have been introduced. While negated constructions are illustrated throughout the grammar, §10 provides a summary of negated constructions with special reference to "symmetry" vs. "asymmetry" (Miestamo 2005). Nondeclarative clauses (interrogative, exclamative, imperative, hortative and optative) are the topic of \$11.

Chapters 12-15 discuss clause-combining. Chapter 12 shows how finite clauses are combined and §13-15 address subordinate/dependent clauses. Dependent clauses are divided into constituentmodifying clauses (consisting of relative clauses, noun complement clauses and postposition complement clauses) (§13), complement clauses (§14) and adverbial clauses (§15). Chapter 16 on discourse phenomena describes the use of discourse-oriented clitics, tags and particles, and also addresses other discourse-related topics. Finally, §17 provides notes on vocabulary from some typologically and culturally interesting semantic domains.

The end of each chapter has a section called "Summary remarks," which revisits the central features that have been discussed in that chapter, particularly typologically interesting ones. The appendices provide texts from various genres (Appendix 1), a summary of differences between spoken and written language (Appendix 2), results from vowel plot measurements (Appendix 3) and the informed consent letter used with the consultants (Appendix 4).

2 Phonology

This chapter describes Denjongke phonology. The discussion begins with a short summary statement (§2.1). After that, separate sections are dedicated to consonants (§2.2), vowels (§2.3), syllable (§2.4), some phonological processes (§2.5) and tone/pitch/register (§2.6). Section (§2.7) discusses the relationship of stress and tone, while (§2.8) addresses some morphophonological phenomena. The last section (§2.9) comments on the phonological script used in the present work. Some notes on intonation will be presented in relevant sections in other chapters.

2.1 Introduction

In the present analysis, Denjongke has 43 consonants and eight vowels (or 13 if lengthened vowels are counted separately). Both length and nasalization are contrastive in vowels. Denjongke words are divided into high and low register based on pitch and voice quality. When a word has an initial obstruent, its register can be predicted based on the initial phoneme. With sibilants and sonorants (nasals and liquids), however, register is unpredictable. The lack of clear voicing difference (breathy vs. modal) and unpredictability of register with some initials leave pitch to be the main contrastive feature in some minimal pairs. Therefore, Denjongke may be termed a tone language, although tone does not bear as great a functional load as in some more well-known tone languages (e.g. Thai).

Syllable onset clusters in Denjongke are more simplified than in phonologically more "archaic" (i.e. more Written-Tibetan-like) Tibetic languages such as Ladakhi, Balti and Amdo. Denjongke syllable structure is (C)(G)V(C/V). Possibly the most controversial part of the present study is the analysis of lightly aspirated, breathy obstruents as separate phonemes rather than as low-register realizations of voiceless obstruents, a decision which increases the number of consonant phonemes (plosives and affricates) by six. The breathy obstruents are treated separately in §2.2.2.

Throughout this grammar the near-open unrounded central vowel is for typographical reasons written as /a/ when reference is made to the phoneme. The phonetic symbol [v] is only used in the phonetic descriptions in this chapter, i.e. /ápo/ [?épo] wrš 'father'.

2.2 Consonants

The consonant phonemes of Denjongke are presented in Table 2.1 below.

		Bilabial	Dento-alveolar	Post-alv.	Alvpal.	Velar	Glottal
	Voiceless unaspirated	р	t	t		k	?
Dission	Voiceless aspirated	p ^h	t ^h	t ^h		k ^h	
Plosive	Voiced	b	d	d		g	
	"breathy"	p'	t'	ť		kʻ	
	Voiceless unaspirated		ts		tç		
A ffmianta	Voiceless aspirated		ts ^h		t¢ ^h		
Alfricate	Voiced		dz		dz		
	"breathy"		tsʻ		tçʻ		
Emicrotive	Voiceless		S		ç		h
Flicative	Voiced		Z		Z		
Nacal	Voiced	m	n		n	ŋ	
INASAI	Voiceless	ņ	n		ņ	ņ	
Latanal	Voiced		1				
Lateral	Voiceless		ļ				
Photic	Voiced		r				
Knotic	Voiceless		ŗ				
Central appro	oximant				j		

Table 2.1. Consonant phonemes in Denjongke

All consonants, except the glottal /?/, occur word-initially⁴². In word-medial position, the importance of aspiration is reduced (i.e. it tends to appear only in emphatically careful speech) and the lightly aspirated ("breathy") series, voiceless liquids and voiceless nasals do not occur at all. The following consonants occur as syllable coda: /p/, /k/, /?/, /m/, /n/, /ŋ/, /r/ and marginally /l/.

2.2.1 Phonetic descriptions and contrastive sets for consonant phonemes

2.2.1.1 Plosives and affricates

All Denjongke phonemes are pronounced with egressive lung air. Plosives and affricates have a four-way contrast in voicing/aspiration: 1) voiced, 2) voiceless unaspirated, 3) voiceless lightly but inconsistently aspirated and followed by breathy voice ("breathy consonant"), 4) voiceless heavily aspirated.⁴³ The four-way contrast occurs only in word-initial position. Word-medially there is a three-way contrast: voiceless aspirated vs. voiceless unaspirated vs. voiced.⁴⁴ The prominence of aspiration, however, is diminished word-medially. Many words which alone have an aspirated initial are as second member of a compound pronounced as unaspirated or with reduced aspiration. Nevertheless, one can still hear word-medial aspirates, especially in words pronounced in isolation.

⁴² The glottal stop, however, occurs word-initially phonetically.

⁴³ Similarly Watters (2002) on five other Tibetic languages spoken in the Southern Himalayas.

⁴⁴ The slightly aspirated breathy consonants become voiced word-medially, e.g. $t'u^2 = \frac{1}{2}$ 'six', /teudup/ [teurup] $= \frac{1}{2}$ 'sixteen' (lit. 'ten-six').

There may be dialectal variation in the realization of word-medial aspiration. For instance, the word $/k^{h}at^{h}u?/_{[\pi'\in\pi']}$ 'direct' was pronounced $[k^{h}\acute{e}t^{h}u?]$ by consultant KN from Martam (East Sikkim) and $[k^{h}\acute{e}tu?]$ by TB from Ralang (South Sikkim). Figure 2.1 illustrates aspiration difference in word-medial affricates by providing wave forms from KN's pronunciation of $k'\epsilon:t\epsilon\tilde{\epsilon}:$ and $k'\epsilon:t\epsilon^{h}\tilde{\epsilon}:$, both meaning 'important' but using different adjectivizing suffixes.



Figure 2.1. Wave forms from $k' \varepsilon t \varepsilon \tilde{\varepsilon}$: (top) and $k' \varepsilon t \varepsilon^h \tilde{\varepsilon}$: (bottom)

In word-final position, only the voiceless unaspirated /p/, /k/ and /2/ occur, /p/ mostly realized as unreleased [p[¬]] and the velar alternating with the glottal stop [k]~[2]. The glottal stop, in addition to being an allophone of word-final /k/, also contrasts with non-glottal endings in other environments.

Voiced stops fricativize word-medially with some speakers, e.g. $/k^{h}\epsilon di/[k^{h}\epsilon \delta i] = \sqrt{mkhal-ril} kidney', /p`õ:bu/[p`õ:bu] <math>\forall \neg \neg bong-bu$ 'donkey'. Another phonetically interesting feature is that when pronounced in isolation, voiced stops may be either pre-nasalized or "prevoiced." These two options are illustrated in Figure 2.2, which have the same word /goko/ $\forall \neg \neg \gamma$ sgog-ko 'garlic' as pronounced by RB (Tashiding) and TB (Ralang).



Figure 2.2. Prenasalization and prevoicing in the initial in /goko/ 'garlic'

As can be seen in Figure 2.2, the prenasalized onset of [ngòkɔ] is voiced throughout, but the second word, written here as [gkòkɔ], has a period of weak voicing (shown by the wave form and

pitch) followed by a voiceless release. Watters (2002: 4) reports similar "prevoiced" stops in Sherpa (Solu Khumbu) and other Tibetic languages.

Plosives and affricates are now presented according to the place of articulation beginning from bilabial and moving backwards in the articulatory tract. Description of phonemes and allophones is followed by minimal/analogous sets, which illustrate that the sounds in question differentiate meaning. Plosives and affricates do not usually occur as geminates. The exceptions are mentioned in the discussion below.

Bilabial plosives

- /p/ [p] voiceless unaspirated bilabial plosive; word-initially, word-medially and wordfinally utterance-medially: /pa:/ [pé:] المجة 'kindle', /ápo/ [?épo] الماتية' 'father', /hap sék^hẽ:/ [hép sẹk^hệ:] المجة 'ki אַרָאָרָאָר' (the word) called [hep] ('to bark')'
 - [p[¬]] voiceless unreleased bilabial plosive; utterance-finally: /hap/ [hép[¬]] المجم⁻ 'bark (v.)', /íp/ [íp[¬]] المج⁻ 'hide (intr.)'
- /pʰ/ [pʰ] voiceless aspirated bilabial plosive; word-initially and word-medially: /pʰjak/ [pʰják]~[pʰjá?] مِخِرْمٍ 'sweep', /kʰimpʰja?/ [kʰímpʰjæ?] المُعْدَمَةُ (broom'
- /b/ [b] voiced bilabial plosive; word-initially, word-medially following nasal, and also inter-vocalically in variation with [β]: /bu/ [bù] ^A/_A, 'middle', /ámbi?/ [?émbi?] ^{MATAT} 'mango', /p'õ:bu/ [p'ồ̃:βu]~[p'ồ̃:bu] ^A/_A, 'donkey'
 - [β] voiced bilabial fricative; inter-vocalically preceding vowels other than /a/, in variation with /b/: /p'õ:bu/ [p'ồ̃:βu]~[p'Õ̃:bu] $\exists \neg \neg \exists$ 'donkey', /rỳbɛ/ [Jỳβɛ] $\exists \neg \neg \exists \neg \neg \neg$ 'tortoise'
 - [w] voiced labio-velar approximant; inter-vocalically preceding /a/: /t[°]iba/ [t[°]iwe] ⁵/₂^{-π}
 [°](question', /nába/ [néwe] ₅³/₂^{-π}
 [°](hell⁴⁵; the nominalizer /-po~bo/ is intervocalically variously realized as [-bo~βo~wo]
- /p[°]/ [p[°]] voiceless lightly but inconsistently aspirated bilabial plosive followed by breathy voice; word-initially: /p[°]a/ [p[°]ġ] קי 'cow', /p[°]o:/ [p[°]ġ:] לאי 'call'

/pu/	Ę.	'skin hair'	/pjak/	নর্হ্যগ	'peel'	/pa:/ 🛐 🏹 '	'kindle'
$/p^{h}u/$	સુ.)	'blow'	/pʰjak/	দের্দ্রিয়া.	'sweep'	/pʰaː/ ཕར	'expand'
/bu/	Æ.	'middle'	/bjak/	শ্র্র্যা'	'come close'	/ba/ $\Xi(v)$	'hide'
/p`u/	£.	'boy'	/p`ja?/	র্হুন্ম.	'rock'	/p`aː/ བར་	'interval'

The only geminate within bilabial plosives in my data is /bb/, which occurs when the combination of the infinitive marker *-po/bo* and the equative $b\varepsilon^2$ (*-po b* ε^2) merges into *-bb* ε^2 , e.g. $j\partial -po b\varepsilon^2$ (WD $\Im_{\neg} \Im' \Im_{\neg} yod$ -*po sbad*) > $j\dot{\varepsilon}bb\varepsilon^2$ (WD $\Im_{\neg} \Im_{\neg} \Im_{\neg} yod$ -*b* ε^2).

⁴⁵ Although [w] can be shown to be an allophone of /b/, the transcription in example clauses follows a more phonetic writing *t'iwa* 'question' and *náwa* 'hell' because the phonetic writing was deemed to better result in correct pronunciation.

Dento-alveolar plosives and affricates

The dento-alveolar plosives and affricates are pronounced with the tongue touching the back of the upper teeth and the alveolar ridge (unlike /s/, /z/, /r/ and /r/, which are alveolar).

- /t/ [t] voiceless unaspirated dento-alveolar laminal plosive; word-initially and wordmedially: /ta/ [té] جَنْ 'horse', /pito?/ [píto?] جَنْ جَسَ' 'hip'
- /t^h/ $[t^h]$ voiceless aspirated dento-alveolar laminal plosive; word-initially and wordmedially: /t^ho/ $[t^h5] \equiv$ 'list', /t^huŋt^huŋ/ $[t^h \acute{u}nt^h unt] = t^h (t^h)$ ' (the second second
 - [θ] voiceless dental fricative; word-medially, with some speakers: /t^ha:t^ha?/ [t^hé:θe?] s¹⁷⁴</sup>, 'back of neck'
- /t'/ [t] voiceless lightly but inconsistently aspirated dento-alveolar plosive followed by breathy voice; word-initially: /t'a/ [t'p] 5 'now', /t'om/ [t'pm] 5 ' bear'

/ta/	સં	'look'	/tɛk/ স্থ্রিযামা	'place up, keep
$/t^{h}a/$	শ্বদ.	'end'	/tʰak/ র্রন্য	'grind'
/da/	बद्दर	'arrow'	/dɛk/ हेगु	'place inside'
/t`a/	5.	'now'	/t`ɛk/ देयां	'chase away'

- /ts/ [ts] voiceless unaspirated dento-alveolar laminal affricate; word-initially and wordmedially: /tsi/ [tsí] हूूूूूूूू 'astrology', /nɛ̈́:tso/ [n̥ɛ̀:ts̪sɔ] तू 😤 'parrot'
- /ts^h/ [ts^h] voiceless aspirated dento-alveolar laminal affricate; word-initially and wordmedially: /ts^ho/ [ts^h5] as 'lake', /pè:ts^ho/ [pè:ts^h5] caras' rest'
- /dz/ [d̪z] voiced dento-alveolar laminal affricate; word-initially and word-medially /dziŋ/ [d̪ziŋ] هَجَتَ 'fight', /dzabridzobri/ [d̪zibrɪd̪zobrɪ] هَجَتَ 'uneven (of landscape)', /k'adzø?/ [k'iˈbd̪zø?] ག་ཆོད་ 'how many'
- /ts^{*}/ [ts^{*}] voiceless lightly but inconsistently aspirated dento-alveolar laminal affricate followed by breathy voice; word-initially: /ts^{*}a/ [ts^{*}b] \in 'make-up'

 $\label{eq:linear_state} \begin{array}{ll} \medskip {\sc s} \medskip$

Postalveolar (apical) plosives

These plosives may alternatively be termed "retroflex," although the tongue is not curled strongly backwards as in, for instance, some Dravidian languages.

- /t/ [t] voiceless unaspirated postalveolar apical plosive; word-initially and wordmedially: /takta?/ [t̪śkˈt̪ɐʔ] শ্র্বান্ত্র্বাঝ 'hard', /tsʰat̥aʔ/ [t̪sʰét̪ɐʔ] র্ক্র্বাঝ 'hot'
- $/t^h / [t^h] \quad \text{voiceless aspirated postalveolar apical plosive; word-initially and word-medially:} \\ /t^h om / [t^h \acute{o}m] \check{g}_{\breve{M}'} `town', /s\acute{a}t^ha / [s\acute{e}t^he] _{\breve{M}'\breve{B}'} `map'$
- /d/ [d] voiced postalveolar apical plosive; word-initially and word-medially: /dik/ [dìk]

- [r] voiced alveolar flap; intervocalically in /teudup/ [teúrup] $\neg_{\overline{S}',\overline{S}'}$ 'sixteen', and also in /da be?/ $\alpha_{\overline{S}'} \cong_{\overline{S}'}$ abbreviated to /de:/ [dè:]~[rè:] $\alpha_{\overline{S},\overline{S}'}$ 'to be similar'.
- $/t^{\prime}$ [t°] voiceless lightly but inconsistently aspirated postalveolar apical plosive followed by breathy voice; word-initially: $/t^{\circ}u^{2}/[t^{\circ}u^{2}]_{5}$ 'six', $/t^{\circ}o$:pa/[$t^{\circ}o$:pa] ξ° 'morning'

/tak/ गुणु' '(be) hard'

/thap/ act' 'act'

/dak/ join' 'join'

/t'ak/ 51 'get well'

Alveolo-palatal affricates

- /tɛ/ [tɛ] voiceless unaspirated alveolo-palatal affricate; word-initially and word-medially: /tɛɛ/ [tɛɛ́] ː 'tongue', /sòtɛ̯a?/ [sɔ̀tɛɐ?] ː sག' 'nettle'
- /tc^h/ [tc^h] voiceless aspirated alveolo-palatal affricate; word-initially and word-medially: /tc^ha?/ [tc^hɐ̯?] gq 'hand (h.)', /ts^hatc^hu/ [ts^hétc^hu] sres 'hot spring'
- /dz/ [dz] voiced alveolo-palatal affricate; word-initially and word-medially: /dzip/ [dzip]
- /tɛ'/ [tɛ'] voiceless lightly but inconsistently aspirated alveolo-palatal affrecate followed by breathy voice: word-initially /tɛ'a/ [tɛ'ɐ̯] = 'tea', /tɛ'um/ [tɛ'u̯m] = a' 'lady, madam'
 - [z] voiced alveolo-palatal fricative; inter-vocalically: /sǿ:tçʿa/ [sǿ:zɐ] المَرْمَاتِ: 'tea (hon.)'

/tcak/	নন্তব্য'	'break (tr.)'
/tehak/	ळग्रा '	'break (int.)'
/dza?/	শ্রুযাম'	'tongue (h)'
/tɛ`ako/	ह्या र्गे	'thief, robber'

Velar stops

The sequence [velar stop] + /j/ does not occur preceding front vowels /i/, /e/, / ϵ / /y/ and / ϕ /. The front vowels cause palatalization on the velar plosives and, consequently, the potential sequence [velar plosive] + [palatal approximant] + [unrounded front vowel] is indistinguishable from the sequence [velar plosive] + [unrounded front vowel].

- - [k]~[?] utterance-finally: /zik/ [zìk]~[zì?] ग्विंग 'leopard'
 - [c] voiceless unaspirated palatal plosive; word-initially and word-medially when followed by /j/ or /i/: /kjokju?/ [cjścju?] juj juj juj 'crooked', /kju?/ [cjú?]~[cjú?]~[cjú?]] juj 'vomit', /ki/ [cí] jj 'be born'

- [c^h] voiceless aspirated palatal plosive; word-initially and word-medially when followed by /j/ or /i/: /k^hja?/ [c^hjæ?] [g^q] 'blood', /p`ik^hjap/ [p`ìc^hjæp'] `a'gq' 'rice-sifting plate', /k^hi/ [c^hí] ^{*}g' 'dog'
- /g/ [g] voiced velar plosive; word-initially and word-medially in other contexts than inter-vocalically or when followed by /j/ or /i/: /ga/ [gè] جَالَ 'ginger', /l̥ɛŋɡɛ?/ [l̥ɛ́ŋɡɛ?] جَالَ 'you (h.)'
 - [J] voiced palatal plosive; word-initially and medially when followed by /j/ or /i/: /gjuk/ [jjùk] قِرَامِ، 'run'⁴⁶, /p^ho:gja?/ [p^hó:jjɛ?] خَرَقِرَامِ، 'husband', /gim/ [jìm] مِقْهَم، 'gaze'
 - [ɣ] voiced velar fricative; word-medially inter-vocalically: /rigõ:/ [μἰγõ:] جُنَقِتِ 'rabbit', /tɛʰigɛ:/ [tɛʰíɣɛ:] كَتَقَرَّسِ 'foreign'
- /k^c/ [k^c] voiceless lightly but inconsistently aspirated velar plosive followed by breathy voice; word-initially: /k^can/ [k^càn] الم: 'what', /k^cola?/ [k^càlɐ?] (الم: 'what', /k^cola?/ [k^càlɐ?] (الم: 'what', /k^cola?/ [k^càlɐ?]) (الم: 'what', /k^cola?/ [k^càlɐ?])
 - [c^c] voiceless lightly but inconsistently aspirated palatal plosive followed by breathy voice; word-initially when followed by /j/ or /i/: /k^cjõ:/ [c^cjồ:] 颈穴 'village', /k^ci/ [c^c] 资 '(big) knife'

/kɛː/ क्रेन	'bring'	/ka/ גאוויב.	'order'	/kom/	ર્ક્સેચ.	'thirst'
/kʰɛː/ ལ	'tax'	/kʰa/ ˌʌ·	'mouth'	/khom/	শিক	'dry (intr.)'
/ge:/ An	'cross'	/ga/ झु.	'ginger'/	/gom/	ই্র্মিঝ'	'door'
/k`e:/ আঁষ্ণ	'separate'	/k`a:/ ག	'what, where'	/k`om/	র্যারু'	'gain experience'

The velar stop does not typically occur as a geminate although my data has two exceptions, $t^h ukky$ (CY)/ $t^h ukk\epsilon$ (PT) 'fixed, not moving' and $t\epsilon^h ukk\epsilon$? 'Nepali language' (CY)⁴⁷.

The glottal stop

The glottal stop /?/ is phonemic only in the word-final position, in which it contrasts with non-glottal vowel endings and final /k/ $[k^{7}]$ ~[?].⁴⁸

/k ^h ε?/ /k ^h εk/ /k ^h ε/	[kʰɛ́?] [kʰɛ́kʰ]~ [kʰɛ́]	·[kʰɛ́?]	छन् छग वि	'difference 'freeze' 'profit'	e'		
/tsho?/	[t̪sʰóʔ]	পঞ্জ.	ʻoffei	ring'	/kʰaʔ/[kʰɐ́ʔ]	[यय]	'liquid, soup'
/tsho/	[t̪sʰɔ́]	প্রুযাধ্য.	ʻlake		/kʰa/ [kʰɐ́]	[य	'mouth'

⁴⁶ Sandberg (1895: 20) comments that "[t]o say 'gy' rightly, personally I have found it almost advisable, strange though it may seem, to pronounce it as dy. Thus gyuk-she 'to run' is almost dyuk-she."

⁴⁷ Other consultants pronounced this word without gemination as $te^h uke$?.

⁴⁸ Historically, the contrast between /?/ [?] and /k/ [k]~[?] derives from differing WT finals, WT -g > /k/ [k]~[?], WT -d/?/ [?], WT -s > /?/ [?] in some words, elided in others.

Word-initially, the glottal stop phonetically contrasts with the high register [h] and the low register [h].⁴⁹ Because the initial glottal only occurs in the high register, it is here considered a phonetic feature of initial vowels, e.g. /ám/ [?ám] 'mother'.⁵⁰

/ám/	[?ém]	জেন্স'	'mother'
/háp/	[hép]]	ন্দ্র-	'to bark'
/à:m/	[hɐ̯̀ːm]	ત્રુંચ.	'jackal'

The phonemic status of /?/, however, is not clear-cut even word-finally, because the realization of final glottals in continuous speech overlaps with vowel length, which also occurs independently of glottal stop (see §2.3.2).

When occurring utterance-finally (e.g. when pronounced in isolation), words ending in a glottal stop such as $/dz\epsilon^{2/}$ 'gunpowder, bullet', $/dz\epsilon^{2/}$ 'leprosy' and $/zi^{2/}$ 'leopard' have various degrees of length, but in continuous speech they are most of the time realized with a long vowel without the glottal.⁵¹ A glottal stop coda works analogously to long vowel codas in that it allows, unlike short syllables, a three-way contrast */i*^{2/}, */e*^{2/}, */* $\epsilon^{2/}$.

Within back vowels, the presence of a glottal stop, similarly to vowel length, raises vowel quality, e.g. /lò/ [lò] $\stackrel{\text{def}}{=}$ 'year' vs. /ló?/ [ló?] $\stackrel{\text{def}}{=}$ 'light' (cf. /ko/ [kó] $\stackrel{\text{def}}{=}$ 'dig' vs. /ko:/ [kó:] $\stackrel{\text{def}}{=}$ 'throw') and /lú/ [lú] $\stackrel{\text{def}}{=}$ 'song' vs. /lú?/ [lú?] $\stackrel{\text{def}}{=}$ 'sheep' (cf. /ku/ [kó] $\stackrel{\text{def}}{=}$ 'body' vs. /ku:/ [kú:] $\stackrel{\text{def}}{=}$ 'laddle for pouring rice flour dough'). The same phenomenon is also seen within different pronunciations of words with -k/? variation at the coda: one consultant pronounced /gok/ $\stackrel{\text{def}}{=}$ 'to crawl' both as [gòk] (F1 550 Hz) and as [gò?] (F1 400 Hz).

A phonetic glottal stop often occurs accompanying an utterance-final nasalized vowel, e,g. /sáŋ/ [sɛ́:?] $\nabla \nabla \nabla \nabla$ 'incense'. At least in the speech of consultant TB, the glottal distinction between /tá?/ $\nabla \nabla \nabla$ 'tiger' and /ta/ 'horse' ∇ is neutralized when a case marker is added, i.e. /ta(?)=lo/ [té=lo] $\nabla \nabla$ 'to the tiger' and /ta/ [té=lo] $\nabla \nabla$ 'to the horse' become indistinguishable.

2.2.1.2 Fricatives and central approximants

Denjongke has five fricatives /s, z, ε , z, h/ and one central approximant /j/. The voiceless fricatives /s, ε / and the central approximant /j/ occur in both high and low register and thus give evidence for tonal contrasts in Denjongke (see §2.6). Voiced sibilants occur only in the low register. In high register /h/ contrasts with initial vowels, which have intrinsic phonetic initial [?]. Low register initial vowels, on the other hand, have an intrinsic initial [ħ], which does not contrast with other

 $^{^{49}}$ In WT/WD, [?], [h] and [h] correspond to $\mathfrak{R}, \mathfrak{H}$ and a respectively.

⁵⁰ Because Lhasa Tibetan similarly has /fi/ preceding low register vocalic onsets and /?/ preceding high register vocalic onsets, Kjellin (1976: 319) comments that "[e]very syllable must begin with a consonant."

⁵¹ Similar observation on the interrelatedness of the glottal stop and length in Tibetan spoken in Nangchen has been made by Causemann (1989: 29).

⁵² Some of my consultants, such as TB from Ralang, systematically pronounce /z/ as /dz/, thus having one phoneme less than the others. I am uncertain whether the lack of /dz/ vs. /z/ distinction is compensated elsewhere in phonology.

laryngeal or glottal initials. Consequently, word-initial [?] and [ĥ] are here considered phonetic markers of high and low register initial vowels respectively, whereas /h/ is considered a phoneme which occurs only in the high register.

- /s/ [s] voiceless alveolar grooved fricative; word-initially and word-medially: /só/ [só]
- /z/ [z] voiced alveolar grooved fricative; word-initially and word-medially: /zo/ [zɔ] ସୁଇଁ 'make', /kuzu?/ [kúzu?] ଲୁଂସ୍ଟ୍ରସ୍ବ୍ 'body (hon)'
- /ɛ/ [ɛ] voiceless alveolo-palatal grooved fricative; word-initially and word-medially: /ɛó/ [ɛɔ́] ភ្ម៍ 'dice', /ɛò/[ɛɔ̯̀] สั 'curd', /sǿ:ɛip/ [sǿ:ɛip']~[sǿ:zip'] ୩୪୩୩ ନିସଂ 'snacks'
- /z/
 [z] voiceless alveolo-palatal grooved fricative; word-initially and word-medially:

 /zak/ [zɐ̯k] جامع: 'put', /ázã:/ [ʔɛ́:zɛ̃:] هرمج: 'maternal uncle'
 - [j] voiced palatal approximant; at least word-initially, in fast speech of some speakers: /zak/ [jèk] קמן. 'put'
- /h/ [h] voiceless laryngeal fricative; word-initially: /hap/ [hép] 557 'bark', /hup/ [húp]
 - [ĥ] voiced laryngeal fricative; inter-vocalically: /màhi/ [mɐ̀hiֵ] אויל: 'buffalo', /pahip/ [pɐ́hi̯p] אויל: 'container for millet beer'
- /j/ [j] voiced palatal approximant; preceding non-front vowels: /já?/ [jý?] ग्यापा 'yak', /jòu/ [jòu] थॅादु: 'up'
 - [h] voiced laryngeal fricative; preceding front vowels: /jìgi/ [hìgɪ] شَنَمَ 'letter', /jŵ?/~/jè?/ [jŵ?]~[jè?]~[hè?] شرح 'exist'

/s/ vs. /z/ vs. /c/ vs. /z/

/zo/ /cò/ /zo/	ন্দ্র র্নি দ্র্বি	'make' 'dice, gambling' 'milk (v.)'	/za:/ /cá/ /(d)za/	যারন প'	'day, planet' 'meat' 'rainbow'
/s/ vs	5. /z/		/c/ vs. /z		

			· · ·	
/sík/ याहीया	'shiver'	/çé:/	विह्य	'know'
/zik/ यहियां	'leopard'	/ze:/	ননিম'	'have, eat (hon.)'

/j/ vs. /h/

/jákca/ गप्पण न' 'yak meat' /hakca/ ठृगान' 'quality of bad(ly cooked) rice'

/j/ vs. /h/ vs. /Ѷ/ vs. /r/ vs. /z/ vs. /dz/

/jầː/	[jềː]	とう.	'again'
/hãː/	[hế̃:]	55.	'squander'
/à̀ː/	[ĥề̃:]	קקי	'lie'
/rã:/	[rề̃ː]	<u> २</u> ८'	'self'
/zaŋ/	[zềː]	নিৰ্বন্থা	'get up (hon.)'
/dzaŋ/	[dzề̃:]	35 ⁻ .	'study'
In the word-initial position, /h/ contrasts phonetically with high register prevocalic [?] and the low-register pre-vocalic [ĥ] (see contrastive set under glottal stop).

One consultant (TB, Ralang) pronounced the word $\langle \delta t \epsilon \rangle$ [fidte] $\exists a_{\forall} \vdots \vdots$ 'down(hill)' as [wete], giving some evidence for the phoneme /w/ is his speech. As this is the only evidence for /w/, the labio-velar in [wete] is here considered an allophonic effect of vowel rounding in $\langle \delta t \epsilon \rangle$.

2.2.1.3 Nasals

Denjongke has eight nasals, the voiced /m/, /n/, /ŋ/, /ŋ/ and the voiceless or preaspirated /m/ /n/ /ŋ/ /ŋ/. The voiceless set is more precisely defined as voiced preceded by voicelessness, [mm] [nn] [nŋ] and [ŋŋ]. Some speakers do not pronounce the voiceless nasals, instead uttering them identically with high register voiced nasals. The voiced nasals occur syllable and word-initially, medially and finally, except for /n/, which does not occur syllable or word-finally. The voiceless nasals occur only word-initially. There is no contrast between the velar and palatal nasals preceding front vowels; only the palata nasal occurs preceding front vowels. Word-final alveolar /n/ and velar /n/ alternate with a long, nasalized vowel, e.g. /sǿn/ [sǿn]~[sǿī] སོན 'seed', /taŋ/ [ṯáŋ]~[ṯáː] གདང 'send'⁵³. In continuous speech, nasals assimilate to the place of articulation of the following consonant, e,g. /ín-bo/⁵⁴ [ím-bo] མོན་ 'EQU-2INF'. When speaking fast, some speakers pronounce /n/ and /n/ as [j] and [j] respectively.

- /m/ [m] voiced bilabial nasal stop; word-initially, medially and finally: /mí/ [mí] à 'human', /dimi?/ [dìmi?] à àur 'key', /ám/ [?ém] war 'mother'
- /n/ [n] voiced alveolo-palatal nasal stop; word-initially and medially: /nà/ [nè] 5. 'fish', /là:na/ [lè:ne] and car, 'upper arm'
- /ŋ/ [ŋ] voiced dental nasal stop; word-initially, medially and finally: /ŋà/ [ŋè]

 'I', /l̥aŋa/ [l̪éŋɐ]

 [l̪éŋɐ]

 [sy-- 'pan', /tã:/ or /taŋ/ [t̪éŋ]~[t̪ế]

 ¬¬- 'send'
- /m/ [m] voiceless bilabial nasal stop: word-initially: /m̥a/ [m̥ɐ́] אָמָר 'down, low(er)', /m̥ɛːc̥am/ [m̥ɛ́ːc̥ɐm]⁵⁵ אָק' אָק' אָסָאָר 'young woman'
- /n/ [n] voiceless alveolo-palatal nasal stop; word-initially: /nou/ [nou/ [nou] ฐ ; station as al mucus'
- /ŋ/ [ŋ]⁵⁷ voiceless velar nasal stop; word-initially preceding non-front vowels:/ŋa?/ [ŋé?] ²²⁴/³/³</sup> 'invocation', /ŋɔmpu/~/ŋømpu/ [ŋómpu]~[hømpu] ²⁴/₅, 'blue-green'
 - [n] voiceless palatal nasal stop; word-initially preceding front vowels: /nim/ [ním] אָקאי 'sister-in-law', /ny:na?/ [ný:nɐ?] הַמיאַקין' 'sweat'

 $^{^{53}}$ Final velar is most of the time pronounced as a lengthened nasalized vowel, although I have also heard realizations as [ŋ], especially in Tashiding, West Sikkim.

⁵⁴ In example sentences in other chapters, however, nasals preceding plosives are written phonetically (i.e. *im-bo*) to ensure a smoother reading experience.

⁵⁵ Some pronounce the word [mpicem].

⁵⁶ The latter spelling suggests a connection with the word \mathbb{R}° 'nose'.

⁵⁷ Some speakers pronounce $/\eta/$ as [h], as in $/\eta$ aru/ [η aru] ~[haru] 'morning'.

/m/ vs. /n/ vs. /ŋ/ vs. /ŋ/	/m/ vs. /m̥/	/n/ vs. /n̥/
/mà/ क' 'mother' /nà:/ क' 'here' /nà/ ज़' 'fish' /ŋà/ रू' 'I'	/mà/ অ' 'mother' /má/ ক্র' 'wound' /m̥a/ নৃঅন্ 'down, low(er)'	/nà:/ ན 'here' /ná/ ན 'ear' /ŋa/ ฐ 'nose
/ ɲ/ vs. /ŋ/ /ɲìm/ ਰੋਕਾ 'sun, day' ⁵⁸ /ɲim/ ਸ਼੍ਰੇਨ੍ਕਾ 'sister-in-law'	/ŋ/ vs. /ŋ/ /ŋàk/ [ŋà?] দ্ব্য 'speech' /ŋa?/ [ŋá?] স্থ্য্ব্য 'invocation	n'

The contrast between the voiceless alveolo-palatal and velar stops is very marginal. The only
contrastive pair in my data, presented below, is based on alternate pronunciations of the word
/ŋompu/~/nømpu/ 'blue-green'. While some people pronounce the word with a velar nasal
[ŋ́śmpu], others use the alveolo-palatal [nømpu].

/ŋ/ vs. /ŋ/ $/\check{p}$ ou/ [\check{p} du] \check{g}_{G} , \check{g}_{G} 'nasal mucus' /ŋ̊ompu/~/nømpu/ [ŋómpu]~[nømpu]~[hømpu] 🖉 ག་ཕུ་ 'blue-green'

The nasals /m/, /n/ and /n/ occur as geminates both morpheme-internally (2.1) and across morpheme boundary (2.2).⁵⁹

(2.1)	mè nn i	మెశ్ర'శే'	'perhaps'
	lè mm o	మేదాను'మే'	'good' (RS) ⁶⁰
	du ŋŋ al	జైర్పారాజ్యం'	'suffering'
(2.2)	là m-m ɛʔ ɲé n=n a kʰɛ ŋ-ŋ a/kʰɛŋ-ga	^{ন্য} অ:স্ক্লি ন্ যান্তব'ব' অট্টিব'যা'	<pre>'below the road (lit. road-below)' 'at the wedding (lit. wedding=LOC)' 'do (you) know (lit. know-PQ)'</pre>

2.2.1.4 Liquids

Denjongke has two lateral approximants, the voiced /l/ and the voiceless (or voiced preceded by a period of voicelessness) /l/. The voicing distinction holds only word-initially. All laterals are voiced word-medially. Word-finally /l/ only occurs in spelling-style and reading-style pronunciation (see Sprigg 1991) of words which end in /l/ in written form. In ordinary spoken language, however, the written final -l of WT/WD is realized as vowel fronting and lengthening, e.g. WT/WD upon gsal 'clarify' may be pronounced as [sél] when reading but is pronounced as [sé:] in ordinary conversation.

⁵⁸ A literary distinction can be made between and 'nyin' day' and an 'nyim' sun'.

⁵⁹ Disyllabic words are challenging for morphemic analysis, because it is not always clear whether the speakers are aware of the constitutive parts of the word.

⁶⁰ This seems a rare pronunciation of the word and may be connected particularly to the astrological context of good/auspicious and bad/inauspicious stellar positions, the context in which the word was spoken.

- $\label{eq:linear} \begin{array}{ll} /l/ & [l] & \mbox{voiced dento-alveolar lateral approximant; syllable and word-initially and medially: /lú/ [lú] <math>\mbox{\underline{n}}$ ' song', /ts^hilu/ [ts^hílu] $\mbox{$\underline{s}$}$ ' fat'
- /l/ [l] voiceless dento-alveolar lateral approximant; word-initially: /la/ [lˈɛ́] عِ: 'deity', /lok/ [lĺók] عَرْمَا' 'lift up'

There are two rhotics, a voiced and a voiceless one. Voicing distinction holds only in wordinitial position. In the following list, the phones listed under /r/ depict tendencies, not fixed rules, i.e. /r/ may occasionally be word-initially realized as [r] as well as [1], and word-finally as [r] as well as [r].

- /r/ [1] voiced alveolar central fricative; utterance-initially: /ra/ $[1\hat{p}\hat{e}] = \hat{v}^{\circ}$ (goat', /ri/ $[1\hat{p}\hat{e}] = \hat{v}^{\circ}$) while \hat{v}°
 - [1] voiced alveolar central approximant; word-medially preceding /l/: /korlɛ/ [kɔ́ɹlɛ] المجترعاتين 'about'
 - [r] voiced alveolar flap; word-medially: /ára?/ [?ére?] هرجم 'liquor', /dumra/ [dumre] عرض 'garden'
 - [r] voiced alveolar trill; word-finally: /k'ur/ [k'ur] المجر 'tent', /kor/ [kór] المجر 'theme'
- /r/ [J]~[r] voiceless alveolar fricative, flap or trill, depending on the speaker; wordinitially and possibly word-medially (I have only one example): /rɛ/ [rɛ́/ [rɛ́]]]]]]]]])])])])]]

/l/ vs. /l/

/là/	ন্দ 'pass'	/lò/	র্শ 'year'	/lòk/ र्वेषा [.]	'return'
/lá/	ੜ' 'life force'	/ló/	ੜ੍ਹੋ [:] 'mind'	/lók/ र्ह्यया	'cause to return'
/ļa/	झ [.] 'deity'	/ ļ o/	ਡ਼੍ਰੋ [:] 'south'	/lok/ ﷺ.	'lift'

/lùk/ सुष्	'drop (intr.); sheep'
/lúk/ ह्यूयां	'pour'
/luk/ ga	'take apart'

/r/ vs. /r/

/re:/	[ːś1]	<u> ন</u> ্মণ	'be torn'	/ram/ म्हा	'be broken'
/rɛ/	[ːśː]	5971	'tear'	/ram/ 5ुवा [.]	'break (trans.)'

2.2.2 The lightly aspirated "breathy" consonants

A major challenge in interpretation was how to treat the series of plosives and affricates that were above described as lightly but inconsistently aspirated and followed by breathy voice,⁶² hence the IPA symbol [^c] for "light aspiration" for marking them. Eberhardt & Mehnert (1978: 129-130) have shown a difference in the degree of aspiration between the aspirated ("strong aspiration") and the historically devoiced plosives and affricates ("less intense aspiration") in three varieties of Tibetan

⁶¹ Utterance-medially, as in the intervocalic position in the sentence $\dot{o}dilo \ lap \ goee \ be2$ 'It is to be called _', /ra/ and /ri/ are likely realized with a flap [r].

⁶² These "devoiced consonants" were historically voiced but have since lost voicing and given rise to tonal constrasts in Tibetic languages. These consonants correspond to the WT (Written Tibetan) characters in the following way: /k^{*}/ > η , /te^{*}/ > ξ , /t^{*}/ > ξ .

(Lhasa, Bathang and Derge)⁶³. Similarly, Watters (2002) describes the "devoiced series" in Dzongkha, Lhomi, Sherpa (Solu Khumbu), Dolpo Tibetan and Mugom Tibetan as "voiceless sometimes with slight aspiration and followed usually by breathy voice," contrasting with "voiceless without aspiration" and "voiceless with heavy aspiration (followed by modal voice)." Watters (2003) suggests that the consonants with slight aspiration have the feature [+spread], referring to the spread glottis causing the slight but inconsistent aspiration and often breathy vowel quality on the following vowel.

The difference in aspiration between $/k^{h}$, $/k^{c}/$ and /k/ in Denjongke is shown in Figure 2.3, where the duration of aspiration is 0,8 seconds ($/k^{h}ap/\sqrt{n}$, 'needle'), 0,6 ($/k^{c}ar/\sqrt{n}$, 'what?') seconds and 0,2 seconds ($/ka/\sqrt{n}$, 'who?') respectively.



Figure 2.3. Initial consonant duration in /khap/ 'needle', /k'ar/ 'what?' and /ka/ 'who?'

Similar difference in aspiration for the bilabial set $/p^h/$, $/p^r/$, /p/ in $/p^hu$:/ respectively', $/p^ru/respectively'$ and /pu:/respectively' is given in Figure 2.4, where there are again clear differences in the duration of aspiration.



Figure 2.4. Consonant duration in /p^hu:/ 'fly', /p^su/ 'boy' and /pu:/ 'pack' (RB)

The aspiration in the breathy set, however, is not consistent, as shown in Figure 2.5, where the same word $/k \epsilon / [k \epsilon] gar$ time, order has a considerable difference in aspiration between the two different pronunciations in the same story. The latter pronunciation is probably indistinguishable from an unaspirated plosive.

⁶³ In the Lhasa language variety, the difference in the degree of aspiration between low and high register words was smaller than in the other varieties, perhaps giving justification to present analyses of Lhasa Tibetan where no aspiration differences are reported as significant.



Figure 2.5. Two pronunciations (in context) of the word /k'ɛː/ [k'ɛː] 'line, order' (SG)

In Figure 2.5, the aspiration in the second pronunciation of $/k^{\epsilon}$: [k' ϵ :] overlaps with the unaspirated /k/. The aspiration in the breathy series also occasionally overlaps with the aspirated /k^h/. With one speaker, for instance, in three consecutive pronunciations of /k' ε :/ [k' $\dot{\varepsilon}$:] 'line, order', one instance had more aspiration than the other two, overlapping with $/k^{h}\epsilon$: $[k^{h}\epsilon]$ and the other two in the other two in the second duration of aspiration.

Figure 2.6 presents the wave forms of the affricate minimal pair /tsa/ & 'grass' vs. /ts'a/ 'makeup', first pronounced in isolation and then in the frame $di _ b\epsilon^2$ 'this is '.



Figure 2.6. Wave forms and pitch of /tsa/ 'grass' and /ts'a/ 'make-up' (TB)

As shown in Figure 2.6, in isolation /ts'a/ 'make-up' is pronounced longer than /tsa/ 'grass', whereas in context the length of the two words is probably indistinguishable. The breathiness on ts'a is more clearly audible when the word is pronounced in isolation. Because length and breathiness are less distinctive in the sentence frame, pitch difference becomes more central. The pitch traces in the frame clauses in Figure 2.6 indicate that ts'a is pronounced in a considerably lower pitch than *tsa*.

The historically devoiced consonants are followed by low pitch and breathy voice. Therefore I refer to them as "breathy consonants."⁶⁴ Precedents within Tibetic languages for analyzing breathiness as a consonantal feature are Causemann (1989: 31) and Watters (2003). Alternatively, breathiness could be analyzed as a vocalic or a suprasegmental feature. The benefits or analyzing

⁶⁴ Beyer (1992: 27) also refers to the "breathiness" of Lhasa Tibetan reflexes of Old Tibetan voiced initials.

breathiness as a consonantal feature are doing justice to the varying degrees of aspiration in consonants and making breathiness predictable on the basis of the consonant.⁶⁵

The high and low register difference applies also to the sibilants, voiced nasals and voiced liquids.⁶⁶ Because there is no evidence for any consonantal phonetic difference between the high and low register sibilants, only one sibilant phoneme is posited for each place of articulation (similarly Watters 2002: 12). Similarly to sibilants, there is no phonetic difference between high and low register consonants for nasals and liquids, and therefore only one phoneme that corresponds to both registers is posited for each manner and place of articulation (excluding the voiced vs. voiceless distinction, which is represented).

2.3 Vowels

This section begins with an overview of Denjongke vowel phonemes. Then, each of the vowels is described in more detail and minimal pairs presented (\$2.3.1). That is followed by a discussion on vowel length (\$2.3.2), diphthongs (\$2.3.3) and nasalization (\$2.3.4).

Denjongke vowel phonemes with length-values are presented in Table 2.2 below.

	Front		Middle		Back				
Short	Long		Short	Long	Short	Long			
i	i	y:			u	u:			
	e: ⁶⁷	ØĽ			0	0:			
3	23								
			а	a:					

Table 2.2. Denjongke vowels

Figure 2.7 below gives consultant TB's vowel plot based on the average value of manual F1 and F2 measurements from four to six different words per vowel value (except *u*: had only two example words). The words along with the measuring results are given in Appendix 3. Short and long vowels for /y/ and $/\phi/$ are not given separately, because length is not as clearly contrastive in these two vowels as in the other vowels, see §2.3.2 on vowel length.

⁶⁵ Analyzing breathiness as a primarily vocalic feature would also cause the historically distinct identity between the voiceless and devoiced consonants, which is reflected in Denjongke writing, to be lost, i.e. both $\overline{\eta}$ and $\overline{\eta}$ would be seen as instances of /k/. That would be pedagogically disadvantageous, because for literate Denjongpos breathiness is already associated with certain consonant characters.

⁶⁶ Voiceless/preaspirated nasals and liquids are always high register. Among sibilants, WT \approx and α are realized as high register /s/ and /c/ respectively (high pitch, modal voice), and WT \equiv and α as low register /s/ and /c/ respectively (low pitch, some breathiness).

⁶⁷ The long realization [e:] is in complementary distribution with a short variant which ends in a glottal [e?] (e.g. *dze?* $a \in \$ 'leprosy'), i.e. there is no contrast between [e:?] and [e?]. As the glottal stop in [e?] is in sentential context often elided and causes vowel lengthening instead, [e:] and [e?] are phonologically interrelated and thus the short variant is not here represented as a separate phoneme.



Figure 2.7. Vowel plot from consultant TB (Ralang)

The most conspicuous features of the vowel plot in Figure 2.7 are 1) the proximity of /i/ and /e:/, 2) the proximity of /u/ and /o:/, and 3) the relatively big F1 difference between /o/ and /o:/. Evidence for considering /o:/ as a lengthened variant of /o/ rather than the two being unrelated vowels /o:/ and /ɔ/ respectively is provided by the variant pronunciations of the word /gok/ [gòk]~[gò?] 'crawl'. Because vowels followed by a glottal are pronounced analogously in quality to long vowels, the variation in the pronunciations of /gok/ [gòk]~[gò?] \breve{s}_{1} 'crawl' suggests that /o/ [ɔ] and /o:/ [o:] should be considered, analogously to /ok/ [ɔk]~[o?], phonologically related so that /o:/ [o:] is the lengthened variant of /o/ [ɔ]. Lengthening, however, is accompanied by a considerable change in vowel height.

2.3.1 Phonetic descriptions and contrastive sets for vowel qualities

The following list decribes the various vowel values and their allophones in Denjongke. Note that /e:/, /y:/ and /ø:/ are marked as intrinsically long vowels, which have short allophones in specific contexts.⁶⁸ With all the vowels /e:/, /y:/ and /ø:/ such a context is a following glottal stop (for the interrelationship of length and final glottal, see §2.3.2). Moreover, /y:/ and /ø:/ are realized as short allophones when they precede the nasal /n/. However, if the nasal is word-final, the pronunciation varies between $[yn]\sim[\tilde{y}:]$ and $[øn]\sim[\tilde{o}:]$, e.g. /dyn/ [dŷn] \sim [dŷ:] \neg_{5} ¬, 'seven', /lópøn/ [lópøn] \sim [lópøi] \approx [lópøi] \approx [lópøi] \approx [lópøi] \approx [lópøi] \approx [lópøi] \approx

⁶⁸ The reason why /y:/ and /ø:/ are intrinsically long is that they have historically arisen from sequences where /u/ (in the case of present /y:/) and /o/ (in the case of present /ø/) have been followed by one of the consonants *l*, *d*, *s* or *n*. Final *l* has resulted in vowel fronting and lengthening, e.g. *ky*: (WT $\frac{1}{2}$, *skul*) 'cause to move' and *k^hø*: (WT $\frac{1}{2}$, *khol*) 'boil (intr.)'. Final *n* has resulted in vowel fronting followed by a nasal stop or nasalization, e.g. *pyn/pỹ*: (WT $\frac{1}{2}$, *spun*) 'brother' and *lópøn/lópø*: (WT $\frac{1}{2}$, *slob-dpon*) 'teacher'. Final *d* has resulted in vowel fronting and a final glottal stop, e.g. *ly*? (WT $\frac{1}{2}$, *lud*) 'fertilizer' and *jò*? (WT $\frac{1}{2}$, *yod*) 'exist'. The resulting vowel can be pronounced with various degrees of length and in utterance-medial position the glottal is typically omitted and the vowel sound lengthened, e.g. [lý?]~[lý:?]~[lý:]' fertilizer' and [j*ò*?]~[j*ò*:?]~(j*ò*:?] 'exist'. The reflexes of historical *u* and *o* followed by *s* are more irregular. Typically the final sibilant has not caused vowel fronting (e.g. *lù*? [WT $\frac{1}{2}$, *lus*] 'remain' and *t'o*? [WT $\frac{1}{2}$, *lus*] 'load', *te^ho*? [WT $\frac{1}{2}$, *lus*] 'body' and the Tibetan-influenced alternative spelling of *te^ho*?, *te^hø*? [WT $\frac{1}{2}$, *chos*] 'teaching'). Similarly to reflexes of final *d*, the vowel sound preceding the glottal stop occurs in various lengths and may be dropped altogether in utterance-medial context where the vowel occurs long.

- /i/ [I] short near-close front unrounded vowel: $/k^{h}i/[k^{h}i] \stackrel{\circ}{\otimes} 'dog', /k^{h}im/[k^{h}im] \stackrel{\circ}{\otimes} ^{N'}$ 'house', $/si/[si] \stackrel{\circ}{\exists} 'receive'$
- /i:/ [i:] long close front unrounded vowel: /ki:/ [kí:] גַזָּאָמִי 'wrap', /sí:/ [sí:] גַאָּמִי 'cool'
- /e:/ [e:] long close-mid front unrounded vowel: /sé:/ [sé:] गुलेम: 'gold', /sé:p/ [sé:p] लेमन: 'yellow', /ge:/ [gè:] এট্রন' 'fall'
- ϵ [ϵ](~[ϵ]) short open-mid front unrounded vowel: /k ϵ / [k ϵ] $\frac{1}{2}$ 'neck', /s ϵ m/ [s ϵ m] $\frac{1}{2}$ (s ϵ m] $\frac{1}{2}$ (s ϵ m) (s ϵ m)
- / ϵ :/ [ϵ :](~[ϵ]) long open-mid front unrounded vowel: / $s\dot{\epsilon}$:/ [$s\dot{\epsilon}$:] $q_{NA'}$ 'clear', / $g\epsilon$:p/ [$g\dot{\epsilon}$:p] $g_{A'A'}$ 'king'
- /a/ [v] short near-open central unrounded vowel; when not preceding palatal or alveolopalatal consonants: /k^ha/ [k^hté] r_{v} 'mouth', /ra/ [rà] τ_{v} 'goat'
 - [æ](~[ɐ̯]) fronted near-open front unrounded vowel; following palatal or alveolo-palatal consonants: /p'ja/ [p'jæ] ਤ 'chicken', /tç'a/ [tç'à] ਵ 'tea', /pà/ [pɐ] ਤ 'fish'
- /a:/ [v:] long near-open central unrounded vowel; when not preceeding palatal or alveolopalatal consonants: /p`a:/ [p`ġ:] བུར; 'interval', /pʰa:m/ [pʰɐ́:m] ཕྱམ; (?) 'hug' (TB)
 - [æ:](~[v:]) long near-open front unrounded vowel; following palatal or alveolopalatal consonants: /ná:ciŋ/ [né:ciŋ] གའའ་ཐོང་ 'yoke', /tɕʰa:lɛʔ/ [tɕʰé:lɛʔ] ཕྱག་ལས་ 'work (hon.)'; however, nasalization seems to counteract the fronting, /ázã:/ [vɛṽɛː] ལ་ལང་ 'maternal uncle'
- /u/ [v] short near-close back rounded vowel; when not following palatal or alveolopalatal consonants: /ku/ [kú] [y] 'body', /p'um/ [p'ùm] [y] 'iri' 'girl'
 - [<code>u</code>] short close central rounded vowel; following palatal and alveolo-palatal consonants: /gjuk/ [<code>jjuk</code>] <code>Jqr</code> 'run', /tcu/ [tcú] <code>¬\varsets' 'ten'</code>
- /u:/ [u:] long close back rounded vowel; when not following palatal or alveolo-palatal consonants: /ku:/ [kú:] (JG: 'laddle for pouring rice flour dough', /p^hu:/ [p^hú:] (JT: 'fly')
 - [u:] long close central rounded vowel; following palatal and alveolo-palatal consonants: /dzu:/ [dzu:/ [dzu
- /o/ [ɔ] short open-mid back rounded vowel; elsewhere: /t`om/ [tַ`om] རྒྱ་ 'bear', /ko/ [kɔ́] རྒྱ་ 'dig'
- /o:/ [o:] long close-mid back rounded vowel: /t'o:m/ [t'ò:m] جَمِع 'trousers', /ko:/ [kó:] المُحْمَد 'throw (away)'
- $/y/ [y:] long close front rounded vowel; when not followed by /n/ or /?/: /ky:/ [ký:] = drive', /p^hy:/ [p^hý:] = drive', /p^hy:/ [p^hý:] = drive', /p^hy:/ [p^hý:] = drive', /p^hy:/ [p^hy:/ [p^hy:/$
 - [y] short close front rounded vowel; when followed by /n/ or /?⁶⁹: /dyn/ [dŷn]~[dŷ:] المرتبع: 'seven', /pynlo/ [pýnlo] المجتم: 'younger brother', /lŷ?/ [lŷ?]~[lŷ?] المجتم: 'fertilizer'
- /ø/ [ø:] long close-mid front rounded vowel; when not followed by /n/ or /?/: /kø:/ [kø:] المجاهر 'boil (tr.)', /rø:m/ [rø:m] المجاهر المحافظ 'cymbal'
 - [ø] short close-mid front rounded vowel; when followed by /n/ or /?/: /k`øn/ [k``øn]~[k``®:] سَرَح: 'wear', /pønpu/ [pǿmpu] حِنْمَ عَنْ 'chief', /tɕʰø?/ [tɕʰǿ?]~[tɕʰǿ(:)] الحَمَ 'you'

⁶⁹ There is also a marginal short /y/ in disyllabic compounds. If the first part of the compound is pronounced in isolation, it ends in a glottal. In the compound the glottal, however, may be elided, e.g. te^hymi 'lamp for offering' (from $te^h\phi$? 'offering' and mi 'fire', the vowel quality ϕ assimilates into y in the compound).

Denjongke unrounded front vowels proved complicated to analyze, a problem also faced in many other Tibetic languages.⁷⁰ Because of the initial difficulties, I carried out an acoustic study on F1 values (corresponding to vowel height) in monosyllabic words containing front unrounded vowels with five speakers from different locations. A detailed account of the study is found in Yliniemi (2014). Here I only summarize the main findings:

- There is a two way contrast /i/ vs. /ε/ with short vowels but a three way contrast /i:/ vs. /e:/ vs. /ε:/ with long vowels.
- 2) There is some overlap in the F1 value of long /e:/and short /i/.
- 3) The short vowel i tends to be realized as lower [1] than the long vowel i. i. i.
- 4) Denjongke has both short /ε/ and long /ε:/ with roughly the same F1 values, contra Dzongkha (van Driem 1992: 67) and Dege Tibetan (Häsler 1999: 24), in which open-mid /ε/ is reported to occur only as a long vowel [ε:] and the short vowel contrasting with /i/ is the higher /e/).

The back rounded vowels /u/ and /o/ are realized as closer when long ([u:] [o:]), and more open when short ([υ] [\mathfrak{s}]). A final glottal heightens vowel value similarly to length, e.g. /ló/ [$\mathfrak{l}\mathfrak{s}$] \mathfrak{s} 'mind' vs. /ló?/ [$\mathfrak{l}\mathfrak{s}$?] \mathfrak{s} 'light', /gok/ [$\mathfrak{g}\mathfrak{s}$ k]~[$\mathfrak{g}\mathfrak{s}$?] \mathfrak{s} 'crawl' (see also the section on glottal stop in §2.2.1.1 above). The following minimal sets give evidence for vowel quality differences between /i/, / ε /, /a/, /o/, /u/, /y/ and /ø/.

/kʰi/ ౄ̀·	'dog'	/ki:/ দুশ্রুম্বাম্বা	'wrap'
/khe/ r	'profit'	/ke:/ mar	'bring'
/kʰa/ /א׳	'mouth'	/ka:/ ཀྱ̃ (?)	'split'
$/k^{\rm h}o/$ at \tilde{k}	'need'	/ko:/ mar	'throw'
/kʰu/ ṟṟ	'he'	/kuː/ ཀྲོཕ.	'laddle for pouring rice flour dough'
/khø:/ ภูลัภา	'boil (intr.)'	/køː/ ลีลา	'boil (tr.)'
$/k^hy$:/(=/ k^hu	ui/) હ્યુત્રે· 'his'	/ky:/ An	'drive'

Within long vowels, an additional vowel /e:/ is introduced between $\frac{\varepsilon}{\epsilon}$ and $\frac{i:}{\epsilon}$.

/sí:/	দ্রমন্ম 'cool'	/gi:/	দ্বদ্বীস.	'go around'	/sìːp/	রিন্দান' 'dew'
/séː/	गर्भेन: 'gold'	/geː/	ત્ર્યોુભ.	'fall'	/sé:m/	ग्रोन्स 'bamboo slat wall'
/sέː/	ग्राला 'clear'	/ge:/	ক্রুন্থ'	'win'	/sɛ́:m/	জ্যমন্য 'daughter (hon.)'

Especially in varieties of Denjongke spoken in East and North Sikkim, /a/ followed by the velar nasal /ŋ/ is pronounced as [p]~[ɔ], hence /taŋ/ $\neg_{\overline{1}\overline{7}\overline{5}}$ 'send' is typically pronounced [téŋ]~[tế:] in Tashiding (West Sikkim) but [tố:]~[tố:] in East and North Sikkim. Therefore, in eastern and northern varieties of Denjongke and opposition between /a/ and /o/ seems to neutralize before /ŋ/.

⁷⁰ Watters (2002: 16), having carried out a phonetic study of five Tibetic languages, calls /e/ and / ϵ / "problematic" and continues that "it isn't always clear whether the vowel is /e/ or / ϵ / in short vowels, and as such whether or not / ϵ / occurs only in long vowels where it is clearly heard as such." Van Driem (1998: 66), on the other hand, posits a short /e/, a long /e:/ and an always long / ϵ :/ for Dzongkha, but comments that the difference between /e/ and /e:/ is actually "more often one of timbre [=quality] than of length." Furthermore, Tournadre & Dorje (2003: 35) describes /e/ and / \ddot{a} / (same as / ϵ /) as separate phonemes, but then comments on /e/ that when "followed by a consonant (closed syllable), it is pronounced like / \ddot{a} ."

Particularly younger speakers are losing or have lost rounding in front vowels /ø/ and /y/, which are pronounced as [e]~[ϵ] and [i] respectively.⁷¹ This development is probably affected by the lack of /ø/ and /y/ in Nepali and English. Interestingly, even speakers who clearly use /ø/ and /y/, tend to unround /ø/ when the infinitive marker -*po/bo* follows, e.g. /t^høn/ [t^hő:]~[t^hón] ž_A⁻ 'happen' > [t^hém-bɔ] ž_A⁻ ' τ ', /dø?/ [då?] ž_A⁻ 'sit' > [dɛ̀(:)pɔ] ž_A⁻ ' τ '.

2.3.2 Length

Vowel length in Denjongke is a complex phenomenon related to other features like vowel quality and the presence/absence of the glottal stop. Historically, vowel length is derived from elided WT final consonants. In the careful, comparative pronunciations for the recording, long vowels were often pronounced as markedly long.⁷² In the following minimal pairs, showing length contrast for each of the vowels, it is seen that the lengthened /i:/, /u:/ and /o:/ are higher in quality than their short counterparts /i/, /u/ and /o/.⁷³ The situation with the unrounded front vowels /e/ and / ϵ /, as described below, is more complicated.

/i/

/sí/	[sí]	ধ্য	'trouble, envy'	/ki/	झे.	'be born' ⁷⁴
/sí:/	[síː]	বর্মনা	'(feel) cool'	/ki:/	ন্ সূর্যামা	'wrap'

/e:/ occurs only as a long vowel, contrasting with /i:/ and / ϵ :/.

/ɛ/ /kɛ/	۲. ۲.	'neck'	/k ^h ɛta?/ ผิ·รงง	'cheap'	/gɛp/ ᡵᡪ· '1	bag'
/kɛː/	AT AT	'bring'	/khe:ta?/ ষ্বায়মা:5্য	skilful'	/gɛːp/ གྱལག་ ']	king'
/a/ /ka/ /kaː/	ساً: (?) المالية:	'order' 'split'	/p`a/ ¬· 'co /p`a:/ ¬¬· 'int	w' erval'	/pʰam/ ಸನು /pʰaːm/ ಸೃನು	'parents' 'hug' (TB)
/u/ /k ^h u/ /k ^h u:/	[kʰớ] [kʰúː]	ଞ୍ଜ 'he' ଞ୍ର 'bread'	/pʰu/ [pʰú] તસ. /pʰuː/ [pʰúː] તસ.	'blow'	/zu/ [zờ] ។ /zu:(?)/ ⁷⁵ [zùː] ។	ig [,] 'melt' igयाब' 'to sit (hon.)'
/o/ /t`om /t`o:n	/ [tˁ: n/ [tˁ	àm] جॅه [,] 'bear àːm] جٚمه، 'trous	, , sers' /	/tsʰo/ [tsʰɔ́~tsʰứ́ /tsʰoː/[tsʰóː]	ć] مخ 'lak ಹ:ব: 'aun	e' it's husband'

⁷¹ The generational difference is clearly illustrated on a song recording where a father ends a line in the long syllable [mø:] while his two children sing a resounding [me:].

⁷² This is in line with Hildebrandt's (2005:24) observation that Manange words in isolation had longer vowels values than when pronounced medially in a context.

⁷³ In Dinri Tibetan (Herrmann 1989: 21) and Drokpa Tibetan (Kretscmar 1986: 23) the quality opposition for long and short vowels applies to all vowels.

⁷⁴ Vowel length in this word was somewhat inconsistent between different speakers.

⁷⁵ Some speakers have glottal ending, others do not.

/ko/ [kó] ¥ 'dig' /ko:/ [kó:] * throw'

/y/ and /ø/

The front rounded vowels /y/ and /ø/ are basically long but may be realized as short when ending in a glottal stop (which varies in pronunciation with length). Short realizations as [y] and [ø] are also possible when the vowel is followed by /n/, either word-finally or syllable-finally. Wordfinally, the combination ends in a short vowel followed by a nasal stop or a lengthened nasalized vowel, e.g. /sǿn/ [sǿn]~[sǿ:] 'seed'. Syllable-finally (but word-medially) /n/ assimilates to the following plosive and the vowel is realized as short, e.g. /pøn-pu/ [pǿmpu] $\neg ζ$ t̃a ເງ.' chief'.⁷⁶ Although no minimal pairs have been found for /y/ vs. /y:/ or /ø/ vs. /ø:/, the following two word pairs illustrate the occurrence of short and long realizations.

/çờn/	र्वेव '	'ride'	/t ^h yn/	মদ্বুর'	'fit, suit'
/çờːm/	র্নিশব্য	'cockroach'	/t`y:p/	হ্রিম্ব	'small bell'

The glottal stop affects length. Utterance-finally the sequence /V?/ is usually realized with various vowel lengths that end in a glottal stop. In sentence-medial context, however, /V?/ is usually realised as [V:] without the glottal (similarly Häsler 1999: 24 for Dege Tibetan). For instance, when pronounced in isolation, words ending in a glottal stop such as $/dz\epsilon?/ \neq 1$ (gunpowder', $/dze?/ \neq 1$) 'leprosy' and /zi?/ = 1 'leopard' have various degrees of length, but in continuous speech they are most of the time realized with a long vowel without the glottal.⁷⁷

A syllable coda with a glottal stop works analogously to a coda with a long vowel in that it allows, unlike short syllables, a three-way contrast /i?/, /e?/, / ϵ ?/. This is shown in Table 2.3, which shows the F1 values of / ϵ ?/, /e?/, /ik/ [i?]~[ik] and /i:/ (the value of /i:/ is given for comparison) taken from an acoustic study with five consultants (RB, TB, PT, NB and TL). WD and WT refer to Written Denjongke and Written Tibetan respectively.

Lexeme	Gloss	WD	WT	RB	ТВ	РТ	TL
42. /dze?/	bullet	rdzas	rdzas	405-	550-	580-	565
				440	560	600	
43. /dze?/	leprosy	mdze	mdze		350-	370-	400-
					410	400	430
44.	leopard	gzig	gzig	330-	260-	380	
RB /zi?/,				345	350		
PT /zik/							
TB /dzi?/							
47. /ziː/	to split	78	?			300-	260-
						330	310

Table 2.3. Three-way contrast of $\frac{\epsilon^2}{vs}$. $\frac{ik}{ir}=\frac{1}{ik}$ shown by differing F1 values

⁷⁶ See footnote 68 above for historical origin of /y/ and $/\phi/$.

⁷⁷ Similarly, Causemann (1989: 29) notes that in Nangchen Tibetan glottal endings lengthen the vowel. Mazaudon and Michailovsky (1988: 123), on the other hand, point out that the WT rhymes -d, -g -s, which in Denjongke are often realized as a glottal stop, are in Dzongkha realized as length (and level pitch).

⁷⁸ NB commented: "We don't yet have a written form for this word."

When a verb-final glottal is followed by verbal suffixes, the vowel may be realized as either short or long, depending on other verb forms with which a contrast needs to be established. If a contrast needs to established with a long (glottal-less) vowel, the glottal stop is dropped without lengthening the vowel (WT below stands for Written Tibetan):

WT shad > $/c\epsilon^2/$, $/c\epsilon^2/comb-INF'$	WT $brjed > /dz\epsilon?/, /dz\epsilon-c\epsilon?/$ 'forget-INF'
WT $bshal > /c\epsilon!/, /c\epsilon!-c\epsilon?/$ 'rinse-INF'	WT $m jal > /dz\epsilon!/, /dz\epsilon!-c\epsilon?/$ 'meet-INF'

If a contrast needs to established with a short (glottal-less) vowel, the glottal stop is realized as a lengthed vowel:

WT btag > /ta?/, /ta:- $c\epsilon$?/ 'append-INF' WT lta > /ta/, /ta- $c\epsilon$?/ 'watch-INF'

Finally, vowel length in short monosyllabic words appears to be quite flexible, leaving a lot of room for context to arbitrate phonemic length. For instance, when hearing a recording of the comparative word pair /ta: $\epsilon\epsilon$?/ [té: $\epsilon\epsilon$?] recorded r

2.3.3 Diphthongs

Diphthong is here defined as a combination of two vowel sounds within one syllable. The two vowel sounds within a diphthong are further considered to consist of two vowel phonemes. The most frequent second vowel of a diphthong is /u/, e.g. /jɛ̀u/ (WD גָּבָּוֹל *dbye-bo*) 'differentiation'.⁷⁹ The following diphthongs have been found to occur in noun bases: /ou/, /ɛu/, /iu/, /ai/ and /oi/. Of these /oi/ and /ai/ are relatively rare within word stems, occurring mainly in the words /ói/ [ói]~[ới] šià' 'hey!' and /ái/ sià' 'older sister', /m̪aicam/~/m̪ɛ:cam/ צֵּקֹיָקְאָסִאָּ

Word stem-internal diphthongs

/ou/~/au/ ⁸⁰	/lóu/~/láu/ ਗ਼ੑਜ਼੶ 'lungs', /rou/~/rau/ ར་ན་ 'fence', /jòu/ གོན་ 'up', /mòu/ སོན་ 'down'
/ɛu/	/theu/ ag. 'dust', /p'jeu/ ag. 'hurry', /jeu/ ag. 'differentiation', /kjeu/ mg. 'rice
	measuring vessel'
/iu/	/lìu/ هَج: 'phase', /biu/ هَجْج: 'snake', /piu/ هَجْج: 'monkey', /diu/ مَجْج: 'bullet'
/ai/	/ái/ आद? 'older sister', /maicam/~/mɛːcam/ हात्र: ga.g.a. 'bachelorette'
/oi/	/ói/ [ói]~[ǿi] &ar 'hey!'

⁷⁹ Final /u/ in a diphthong is traditionally typically written as $\approx bo$, a form which functions both as a historical nominal suffix and a productive infinitivizer/nominalizer of verbs. Nowadays, some writers are experimenting with a more phonetic spelling $\underline{G}'u$.

⁸⁰ The off-glide is pronounced at various degrees of fronting and roundedness between [au] and [ou]. There is no contrast between /au/ and /ou/.

The following four words provide evidence for distinctions /ɛu/ vs. /ou/ and /iu/ vs. /ɛu/:

/eu/ v	vs. /ou	/	/iu/ vs. /ɛu/			
mεu	ষ্ট্র'র্ন	'mole (in skin)'	tiu	ह्वेतुः, ह्वे'र्च	'navel'	
mòu	ર્વેતુ'	'down'	$t^{h}\epsilon u$	ਬੇਨੂਾ, ਬੇ ਸੱਾ	'dust'	

Genetival diphthongs

/ɛi/	/kɛ-i/ क्वेते 'neck's', /pɛ-i/ त्रोते 'example's'
/ai/	/sa-i/ જાવે 'soil's', /ra-i/ ત્રવે 'goat's'
/oi/	/só-i/ कॅंदि· 'tooth's', /go-i/ अर्गेंदि· 'head's'
/ui/	/mù-i/ દ્યુંતે [.] 'her', /kʰu-i/ _{ઙિ} ંતે [.] 'his'

The diphthong $/\epsilon i/$ also occurs in the interjections $\dot{a}dz\epsilon i \, \varpi \in (\text{surprise})$ and $k\epsilon i \, \varpi$ (honorific address), and /a i/ occurs in the interjection aijo: $\mathfrak{F} \mathfrak{G} \mathfrak{G}$ (discomfort), underlining the phonologically distinct character of interjections.

2.3.4 Nasalization

Distinguishing between nasalization and nasal stops is challenging in Denjongke. All monophthong vowels except /e:/⁸¹ occur as nasalized (and lengthened). Nasalization derives from a historical syllable-final nasal /n/ or /ŋ/. Often a nasalized vowel has an alternative pronunciation with a final /n/ or /ŋ/, although in some cases, where only a nasalized pronunciation exists in spoken language, it is impossible to determine the underlying nasal without reference to the written language. Pronunciations with a nasal stop are probably more usual with literary speakers in careful speech, whereas elsewhere nasalized vowels are used. In some speech varieties especially in North and East Sikkim the difference between /aŋ/~/ã:/ and /oŋ/~/õ:/ appears to disappear, because both are pronounced [$\tilde{3}$:]~[$\tilde{0}$:]. The distinction is, however, held in West Sikkim, for instance in Tashiding.

/p`in/	[p`ïː]~[p`ïn]	ਉੱਕ.	'give'
/séŋ/	[sɛ̃́:]~[sɛ́ŋ]	RIE.	'elevate'
/tã:/	[t̪ɐ̃̃ː]~[t̪ɐ́ŋ]	বদন:	'send'
/k`uŋ/	[kʿײֲ̃ː]~[k'ײ̀ฏ]	याूर्ट.	'roof'
/ốː/	[ấː]	595.	'empowerment'
/pyn/	[pỹ̃:]~[pýn]	्ध्रव.	'brother'
/k`øn/	[kʿʾĝː]~[kʿʾġn]	ਸ਼ੁੱਕ.	'wear (clothes)'

 $^{^{81}}$ That is, I do not have evidence for a contrast between / $\tilde{\epsilon}:$ / and / $\tilde{e}:$ /.

this tendency. With /i/ and / ϵ /, the final velar nasal contrasts both historically and syncronically with the dental nasal /n/, e.g. *p* 'in \Im_{a} 'give', *sén* psg 'hear (hon.)'. Such contrast, however, does not occur with the back vowels, because historically a final /n/ has caused vowel fronting /a/ > / ϵ /, /o/ > / ϕ /, /u/ > /y/, e.g. WT psg 'gan > Denjongke gen 'responsibility'.⁸² Thus, within back vowels nasalization points necessarily towards a historical velar nasal which does not contrast synchronously with a dental nasal (contrast with bilabial *m* is unproblematic because *m* is always pronounced as a nasal stop), whereas with front vowels, nasalization is ambiguous between /ŋ/ and /n/.

The phonemic transcription used in this grammar marks nasalization rather than a nasal stop on back vowels /a/ and /o/ where it actually occurs in pronunciation. This practice allows the writing to correspond more clearly to pronunciation. It also does not cause any phonological (or even historical) information to disappear, because with back vowels nasalization is unequivocally tracable to a velar nasal. Following the back vowel /u/, however, a velar nasal is often heard instead of a nasal vowel. The reason for this may be that the physical effort involved in producing nasalization causes the back of the tongue to raise towards the velum/pharynx. Since the back of the tongue is already close to the velum in pronouncing /u/, the act of nasalization easily produces a velar nasal. A final velar nasal following /u/ is marked, because it is often heard as such in pronunciation, e.g. $t^h u\eta$ eggs 'drink'.

Although word-finally a historical velar nasal following /a/ and /o/ is pronounced as a nasalized vowel, at the end of the first syllable of disyllabic words the velar nasal is often pronounced and thus also represented in the phonemic script, e.g. $t'anpu/t'\tilde{a}:pu = \sum_{i=1}^{n} t'anpu/t'\tilde{a}:pu = t'anpu/t'anpu/t'\tilde{a}:pu = t'anpu/t'anpu/t'anpu/t'anpu/$

2.4 Syllable structure

The syllable in Denjongke is of the form (C) (G) V (C/V).⁸³ The mandatory vowel may be preceded by any of the consonant phonemes listed in Table 2.1. The glide is almost always /j/, but there is also a marginal glide /r/, pronounced [r], which occurs rarely and not in all Denjongke varieties⁸⁴. The glide /j/ may follow the velar stops /k/, /k^h/, /g/ and /k^c/, the bilabial stops /p/, /p^h/, /b/ and /p^c/, and the bilabial nasal /m/. All vowels may fill the mandatory vowel position, either as short or long. The second vowel position in diphthongs is reserved for close vowels /u/ and /i/.The last consonant may be a plosive /p/, /k/, /?/, a nasal /m/, /n/, /ŋ/⁸⁵ or the rhotic /r/. The lateral /l/ occurs in syllablefinal position in reading and spelling style pronunciations when the Written Tibetan (WT) or Written Denjongke (WD) has a final /l/, as in WT/WD *gsal* /sé:/ [sé:] 'clarify', reading-style [sal]. The syllable structure is summarized in Table 2.4 and examples of the various syllable patterns are given below. Diphthongs are here analyzed as VV and long vowels as V. The reason for the differing analyses is the difference in distribution: dipthongs do not occur in closed syllables.

⁸² An important exception to the fronting rule is $k'an \sqrt{n}$ 'what', which is not pronounced k'en despite the final nasal.

 $^{^{83}}$ C = consonant, G = glide, V = vowel.

⁸⁴ However clusters such as /pr/ and /kr/ do occur in loan words (e.g. Nepali names) and ideophones (see §17.1).

⁸⁵ Final -n and $-\eta$ are often realized as nasalization and vowel lengthening.

Table 2.4. Syllable structure

(C)	*	(G)	V	(C)
All	consonants	/j/ and marginally	All vowels	/p/, /k/, /ʔ/, /m/,
excep	ot /?/	/r/ in some speech		/n/,/ŋ/, /r/ and
_		varieties		marginally /l/
				(V)
				/i/, /u/

Open syllables

CGVV	/kjɛu/ŋg· 'rice measuring vessel' (KN), /p 'jɛu/ ʒg· 'hurry'
CGV	/kja/ x 'hair', /p'ja/ 51 'do', /bja/ 25' 'rice', /p'jo/ 31 'escape'
CVV	/jou/ xg. 'up', /khau/~/khou/ rst 'snow', /theu/ ag 'dust'
CV	/là/ ar 'mountain pass', /só/ šr 'tooth', /khu:/ Bg 'bread'
V	/ỳː/ ལ̯ལ 'country', /ǿː/ [ɦøː] འ̀འֵ (?) 'eagle'
VV	/ái/ wig. 'older sister', /ói/~/ới/ wig. 'hey!'

Closed syllables

CGVC	/gjap/ _Ð ¬'	'back', /mjõː/ མོང·	'finish', /prɛk/	[p.i.ék] ⁸⁶ द्येंग	r 'cut (grass)' ((RB),
------	------------------------	---------------------	------------------	-------------------------------	-------------------	-------

CVC /k'ur/ אָד. 'tent', /sák/ האזקאי 'accumulate', /t'oːm/ בָּאַשּ' 'trousers'

VC /íp/ khur 'hide', /ám/ khar 'mother', /à:m/ gar 'jackal'

2.5 Segmental phonological processes

2.5.1 Vowel assimilation⁸⁷

In Denjongke, vowel height, roundedness and frontness are affected by other vowels. This section provides only preliminary notes, which should be followed by a more detailed study. In the compound in example (2.3) below, assimilation is bi-directional in that the second vowel /i/ causes heightening of the first one from $/\phi/$ to /y/ and the first vowel $/\phi/$ causes the rounding of the second one from /i/ to /y/.

(2.3) $/tc^h \phi ?/ = tc^h y mi/ = 10^{-1} /tc^h y mi/ = 10^{-1} /t$

In (2.3), the height assimilation $|\phi| > |y|$ is more stable and may hence be considered to have become lexicalized so that speakers are not necessarily aware that the first syllable of /tc^hymi/ derives from /tc^h ϕ ?/ 'offering'. The rounding assimilation [y]~[i] in (2.3) is more subject to variation between different pronunciations of the word and may hence be considered allophonic.

In (2.3), assimilation works across syllable boundary. The next examples illustrate assimilation within the syllable. The most frequent type of assimilation within the syllable is the fronting of the previous vowel by syllable-final /i/, see (2.4), or the fronting of the following vowel by /j/ in the syllable onset, see (2.5).

⁸⁶ A more typical pronunciation is *pjɛk*.

⁸⁷ Vowel assimilation/harmony in Tibetic languges has been addressed by several researchers (Sprigg 1961, 1980, Miller 1966, Haller 2012). Hari (1979: 28) comments on Lhasa Tibetan that vowel analysis is complicated by "extensive and intricate processes of vowel height approximation in polysyllabic words."

⁸⁸ WT/WD mchod-me

(2.4) a) /mù=i/ [mùi]~[mỳi]~[mỳ:] المراكة 'her'

- b) /oi/ [ói]~[ới] र्लेवे 'hey!'
- c) /maicam/~/me:cam/ >ar 'grav 'bachelorette, young woman'
- (2.5) /gjompo/ [gjòmpo]~[gjòmpo]~[gòmpo] جَنْرَجَ بَنْ 'monastery'

In example (2.4a), the genitive marker =i is attached to a base ending in *u*. The first vowel /u/ determines rounding whereas the latter vowel /i/ determines frontness for the resulting long vowel [y:]. The variants in (2.4c) represent pronunciations in different localities, suggesting that /m̥ɛ:cam/ is a stabilized fronted pronunciation of /m̥aicam/. As shown by (2.3-5), the front vowel /i/ is central factor in vowel assimilation.

2.5.2 Elision

Both vowels and consonants in frequently used constructrions may be elided. In fast speech, final vowels are often elided, e.g. the reportative marker $=s\epsilon/si$ frequently becomes =s and the attention marker $=\epsilon o$ is realized as $=\epsilon$. The long vowel in /ma- ϵe : t ϵ / is dropped to form the idiom /ma $\epsilon t\epsilon$ /' I don't know, who knows'.

In fast speech, also medial consonants are often elided in frequently-used words, see (2.6-10).

(2.6) A5' 35' be? da > de:/re: similar EQU.NE 'be similar' (2.7) *ซึ*เลริ ódi > *ói* 'that' (2.8) ฉรั ฉรั dodi > doi 'right this (one)' (2.9) 5'75 t'arin $> t' \varepsilon i \eta$ 'today' (2.10)गामीषा अन्न र्च? làp-o > kai làpo ka-ki who=AGT say-2INF 'Who said (so)?.' (Tashiding 5 p. 75)

2.5.3 Consonant lenition

Lenition is a process in which a consonant becomes more sonorous, or more vowel-like. Infinitive marker *-po/bo*, for instance, often goes in fast speech through vowel elision and consonant lenition,

becoming -m. Table 2.5 illustrates this lenition process in the periphrastic past construction where the nominalizer is followed by an equative copula (for periphrastic past, see §8.1.1).

1 4010 2.5. 100	duction of m	omn	p = p = p = p = p = p = p = p = p = p =		
Full form			Abbr. form	Source	
t ^h õː-po í̇́ː	अर्वे८ र्थे क्षेत्र	>	[tʰõːm í̃ː]	(PL)	
nấ̃:-bo bɛ?	ਧਕਿਸਟਾਬੋਂ ਐਕਾ	>	[nɐ̃́ːm bɛ?]	(CY)	
òm-bo bε?	र्देन हे. इन	>	[òm bɛ?]	(NB)	
số:-bo bɛ?	র্ঝন:র্ন শ্বন্	>	[sóm bɛ?]	(DB)	
sõ:-bo bɛ?	र्षत र्य श्रु	>	[sóm bɛ?]	(DB)	_

Table 2.5. Reduction of nominalizer -po/bo to -m

Other examples where the nominalizer is reduced to -m are (2.11) and (2.12):

- (2.11) $\operatorname{qgc} \operatorname{sc} \operatorname{sc}$ $n\tilde{a}$:-bo-d\tilde{a}: > $n\tilde{a}$:-m-d\tilde{a}: do.HON-2INF-CONJ 'when doing' (CY)
- (2.12) $\eta_{\overline{A}} \subset \eta_{\overline{A}} \subset \overline{A}$ $n\tilde{a}:-n\tilde{a}:-bo > n\tilde{a}:-na-m$ do.HON-do.HON-2INF 'done' (CY)

In intervocalic position /tsh/ is often simplified to /s/.

(2.13) ג׳ גַבָּאַישֿאַ

ŋà dzø:-ts^ha: > ŋà dzø:-sa
1SG make.mistake-CMPL
'I (have) made a mistake.' (JD minimal pair recording)

- (2.15) র্চ্চ'ন্ড্র্ব্য

ts^halum > *sálum* (when preceded by a word ending in a vowel). 'orange'

Voiced plosives may become nasals when followed by a nasalized vowel, e.g. $/g/ > /\eta/(2.16)$ and /d/ > /n/(2.17).

(2.16) $\widetilde{\mathfrak{A}}' \widetilde{\mathfrak{A}} \widetilde{\mathfrak{A}}' \widetilde{\mathfrak{A}} \widetilde{\mathfrak{A}}' \widetilde{\mathfrak{A}} \widetilde{\mathfrak{A}}'$ $\acute{odi:} g\widetilde{a}: > \acute{odi} \eta \widetilde{a}:$ that.GEN time 'at that time' (2.17) a) 듯듯` t'ã:/dã:/dõ: > nà/nò (at least in Martam, East Sikkim) 'and'
b) 뭐지 듯데데' 흔히 가 있어지 않는 가 있어지 않는 가 있어요.
b) 뭐지 듯데데' 흔히 가 있어요.
b) 귀지 듯데데' 흔히 가 있어요.
b) 귀지 듯데데' 흔히 가 있어요.
c) 귀지 드 더 나 있는 가 있어요.
c) 귀지 드 너 나 있는 가 있어요.
c) 귀지 드 너 나 있는 가 있어요.
c) 귀지 한 가 있는 것 같아 한 가 있어요.
c) 귀지 한 가 있는 것 같아 한 가 있는 것 않아요.
c) 귀지 한 가 있는 것 않아 요.
c) 귀지 한 가 있는 것 않아 요.
c) 귀지 한 가 있는 것 않아 요.
c) 귀지 한 가 있는 것 않아

Word-medial aspiration is weakened or disappears, especially in sentential context.

(2.18) 积'风花、 láŋk^hor > laŋk^hor~láŋkor 'taxi'

For regular morphophonemic alternation refer to §2.8 below.

2.6 Register, pitch and tone

Denjongke words occur in two registers, high and low. The two registers consist of a bundle of features listed in Table 2.6.

Table 2.6.	Features	of high	and low	register
		()		()

High register	Low register
-Modal, stiff or creaky voice on vowel	-Breathy or modal voice on vowel
-High pitch	-Low pitch
-Voiceless and (strongly) aspirated consonants	-Breathy and voiced consonants
-After initials /p/, /p ^h /, /t/, /t ^h /, /k/, /k ^h /, /t ^c /, /t ^c ^h /,	-After initials /b/, /p [°] /, /d/, /t [°] /, /g/, /k [°] /,
/ts/, /tsʰ/, /m̥/ /n̥/ /n̥/ /ŋ̥/, /l̥/, /ŋ̥/, /h/	$/dz/, /tc^{\circ}/, /dz/, /ts^{\circ}/, /z/, /z/, /r/$

As seen in Table 2.6, register is often predictable from the initial phoneme of the word. However, for the following initials register is not predictable (i.e. these initials occur both in high and low register): /V/, /s/, /e/, /m/, /n/, /n/, /n/, /n/, /n/, /n/, /n/ and /j/. Because breathiness is not always very prominent with these initials, pitch plays an integral role in disambiguation.⁸⁹ Therefore Denjongke may be termed a tone language, although contrastive pitch does not bear as big a functional load in Denjongke as in well-known tone languages such as Mandarin Chinese and Vietnamese. High tone is phonetically a fall from high to low pitch (in long vowels pitch is not as sharply falling) and low tone is realized as a slight rise from low to higher pitch (although pitch following voiced stops is not as clearly rising).

⁸⁹ This is unlike Lhomi, in which voice quality (modal vs. breathy) is more salient than pitch with sonorant initials (Watters 2003: 253).

In disyllabic words, the difference between high and low tone is the relative height of the first syllable in relation to the second syllable. Both high and low register disyllabic words have the syllable pitch pattern high-low, but in low tone words the first syllable starts lower than in high tone words and has a greater pitch difference to the second syllable than in high tone words. The following sections give evidence for these claims, §2.6.1 for monosyllables and §2.6.2 for dissyllabic words.

2.6.1 Monosyllables

After initial tonal analysis with several speakers, I conducted a more systematic analysis of 84 monosyllabic nouns with KN from Upper Martam (East Sikkim). The words were first pronounced in isolation and then in a carrier sentence *ódilo* _ *làp go:cɛ bɛ?* 'This has to be said...'. The following minimal pairs illustrate the contrastive pitch within sonorant initials (/má/ a_i 'wound' vs. /mà/ a_i 'mother'), plosive initials in open syllable (/ka/ a_i or 'order' vs. /k'a/ a_i 'what, where') and closed syllable (/kom/ a_i 'thirst' vs. /gom/ a_i 'door'). Word duration in Figures 2.8-10 is marked with vertical dotted lines.





The defining pitch pattern in Figure 2.8, high falling in /má/ 'wound' and low rising in /mà/ 'mother', is seen on the latter part of the word on the vowel. The rapid rise at the onset of /má/ signifies a short pause and glottalization. In /mà/, the nasal has a falling pitch and the vowel a rising pitch both in isolation and sentential context.

The word pairs in Figures 2.9 and 2.10 illustrate that pitch in predictably high/low register words is similar to /má/ and /mà/ in Figure 2.8.







Figure 2.10. Pitch traces of /kom/ 'dry (verb)' and /gom/ 'door' in isolation and in context (KN)

Figures 2.9-10, which show that the high register words have a falling pitch and the low register words a rising pitch, are representative of all the words recorded in the tonal study. All the high register words (both open and closed syllables) had a high falling pitch, although with long vowels the pitch was more level than with short ones. Low register words had a low rising pitch, although the pitch following voiced stops was not as clearly rising.

2.6.1.1 Register-internal pitch differences

In the four-tone systems suggested for some Tibetic (e.g. Vesalainen & Vesalainen 1976, Hari 1979) and other Himalayan languages (Watters 2002), register-internal tonal contrasts are reported both within the high and the low register. In Denjongke, however, it is difficult to find evidence for register internal tonal contrasts. Nevertheless, there are some word pairs that give some initial evidence for marginal register-internal pitch contrasts both in high-register and low register. I first give some evidence for limited tonal contrasts in the high register and then in the low register.

Consultants from Ralang (TB) and Upper Martam (KN) did not have a pitch difference between the words /ŋá/ (WT g· *rnga*) 'five' and /ŋá/ (WT g· *rnga*) 'drum'. A consultant from Tashiding (RB), however, pronounced a pitch difference in /ŋá/ 'five' (high level) and /ŋá/ 'drum' (high falling). This is illustrated in Figure 2.11, presenting isolated pronunciations of /ŋá/ 'five', /ŋà/ 'I' and /ŋâ/ 'drum'. More research is needed in order to determine whether Figure 2.11 shows list intonation, whether other triplets give evidence for a three way pitch contrast and whether the threeway contrast occurs in all vowels.



Figure 2.11. Sequentially uttered /ŋá/ 'five', /ŋà/ 'I' and /ŋâ/ 'drum' (RB)

For the same speaker from Tashiding (RB), the words $/ta/\frac{1}{5}$ 'horse' and $/ta?/\frac{1}{5}$ 'tiger' were segmentally contrastive when the words were pronounced in isolation, [ta] 'horse', [tá:?] 'tiger'.

This can be seen in Figure 2.12 where /ta?/ is pronounced longer than /ta/ (the words are pronounced three times on the recording).



Figure 2.12. Contrast between /ta/ 'horse' and /ta?/ 'tiger' when pronounced in isolation

In sentential context, however, the glottal stop of /ta?/ was elided and the two words /ta/ and /ta?/ were pronounced with equal or almost equal length. The most obvious contrast became pitch, which was falling in /ta?/ and level in /ta/. The pitch contrast is shown in Figure 2.13. The carrier sentence is $\eta \dot{a}_{\perp}$ tato $\tilde{t}'(r_{\perp} = g(\tilde{r})$ ($g(r_{\perp})$) (I'm looking at _').



Figure 2.13. Contextual tonal contrast between /ta/ 'horse' and /ta?/ 'tiger'

It was shown above that one speaker from Tashiding (West Sikkim) may have a tonal contrast in the high register between high level and high falling tone. This contrast, however, has not been attested with other speakers. Several speakers, on the other hand, have provided some evidence for a tonal contrast in the low register. This is illustrated in Figure 2.14, featuring a pitch difference for segmentally identical words /zí:/ $\eta \exists \eta \forall \cdot$ 'look' and /zi:/ $\eta \exists \forall \cdot$ 'split'. Both words are pronounced in an honorific imperative construction *cin VERB-po nã*: resulting in meanings 'Please look at the tree' and 'Please split the tree' respectively.



Figure 2.14. Pitch contrast with voiced stop initials /zí:/ 'look' and /zì:/ 'split'

Figure 2.14 shows that /zí/ 'look' is pronounced with higher and basically level pitch whereas /zì:/ 'split', analogously to typical low register words, is pronounced with a lower lightly rising pitch. Thus, this word pair appears to exemplify a high vs. low register split within the low register. There is, however, also a segmental difference. The infinitive marker -*po/bo* is a voiceless [p] with /zí:/ 'look' and a voiced [β] with /zì:/ 'split'.

2.6.1.2 Final glottal and pitch

When pronounced in isolation, words ending in a glottal stop may be pronounced with various lengths, e.g. $/k^{h}a?/[k^{h}v?]\sim[k^{h}v?] [var]$ 'soup'. In sentential context in the tonal study, the glottal was usually elided and the pitch was falling in both $/k^{h}a/$ 'mouth' and $/k^{h}a?/$ 'soup'. The contrast between $/k^{h}a?/$ and $/k^{h}a/$ became one of length and perhaps also some glottal quality, or creakiness, on the vowel in $/k^{h}a?/$. With back vowels, there is the additional contrast of vowel quality. The vowel in the open syllable is lower than in the closed, e.g. /lo/[lo?] [solation] [lo?] (context) and $/k^{h}a?/$.

In comparative recordings with KN (Martam), a high register word with a glottal ending (e.g. $/\eta a?/$ solution) had a short vowel followed by a glottal stop when pronounced in isolation. In context, however, the words were pronounced with a falling pitch and a long vowel with the glottal elided. Thus, in sentential context, the difference between words with glottal ending and non-glottal ending was length, the glottal stop being pronounced as additional length.

Low register glottal ending, however, was pronounced even in context, as can be seen in Figure 2.15, where a glottal ending is contrasted with a non-glottal ending. The frame sentence is *ódilo_làp go:cɛ bɛ?* (क्रॅ'acct'ă'_ava' cățăv: act call this _'.



Figure 2.15. Contrast of glottal and non-glottal ending in $/\eta a^2/$ speech' and $/\eta a^2/I'$

The glottal stop of $/\eta a^2/(ngag)$ in Figure 2.15 is seen as a sudden fall in the pitch at the end of the pronunciation of the word. The effect of the glottal on the pitch trace can be clearly seen by comparing $/\eta a^2/$ speech' to the non-glottal ending in $/\eta a^2/(nga)$ 'I'.

The next section extends the discussion on tone to disyllabic nouns.

2.6.2 Disyllables

In an acoustic study of 29 disyllabic nouns with KN both high and low-register/tone⁹⁰ disyllabic nouns were pronounced in a low-high pitch pattern, although sometimes the starting point in the pitch of the falling first syllable was higher than the second syllable. When pronounced in isolation, the first syllable in high-tone disyllabic nouns was sometimes pronounced on equal pitch level with the second syllable. In context, however, the first syllable usually became lower in pitch than the second. The same pattern was observed also with other speakers: equal pitch levels or high(er)low(er) pattern in disyllables were in clausal context changed to low(er)-high(er) pattern.

No consistent difference in the absolute pitch height of the second syllable of high vs. low-tone words was perceived. In some cases, the second syllable of a low-tone word was higher than the second syllable of a high-tone word. Consequently, the contrastive factors between high and lowtone words were 1) pitch height in the first syllable and 2) pitch difference between the first and second syllable. A low-tone (disyllabic) word had a lower starting pitch than a high-tone word, and low-tone words had at least 10 Hz greater pitch difference between the first and second syllable than high-tone words.

The difference of high and low-tone disyllabic words is illustrated in Figure 2.16 with the words /pømpu/ [pømpu] 5र्द्य र्रें 'leader' and /p'ømpu/ [p'gmpu] र्द्य र्रें 'Bön practitioner'.91 The words are first pronounced two times in isolation and then in the same carrier sentence as in Figure 2.16 (location of [pǿmpu]/[p'ŵmpu] in the sentence is marked by the dotted line).

⁹⁰ Strictly speaking, the word "register" should be used for those words in which the pitch is predictable from the first consonant (e.g. the slightly aspirated "breathy" series) and "tone" for those words in which pitch is unpredictable from the first consonant. For brevity, however, I use "tone" in this section when "register/tone" is meant.

⁹¹ Whereas the historical nominal suffix -po (often -pu in nominals), by which many nouns have been formed, is typically in nouns pronounced with higher pitch than the previous syllable, the synchronically operational infinitive marker -po/bo following verbs is typically pronounced with lower pitch than the preceding verb stem.



Figure 2.16. Pitch contrast in /pømpu/ 'leader' and /p'ømpu/ 'Bön practitioner'

As can be seen in Figure 2.16, the pitch difference of the first and second syllable in the first utterance of /p'ompu/ is much greater than the same difference in /pompu/. The second syllables are roughly on the same pitch level. The main factor in disambiguating /pompu/ and /p'ompu/ in context seems to be the level of the first syllable in comparison to the previous word. It is noteworthy that the characteristically rising pitch of low-tone monosyllables is changed for a level/falling pitch in the first syllable of disyllabic words.

Figure 2.17 illustrates the pitch difference in high and low-tone words with sonorant initials (/námtco?/ क्रबरक्र 'ear' and /màmtc^hu/ क्रबरक्र 'lower lip').



Figure 2.17. Pitch contrast in /námtco?/ 'ear' and /màmtchu/ 'lower lip'

When pronounced in isolation, the low-tone word /màmte^hu/ has a rising pitch on the vowel of the first syllable. In context, however, the first syllable becomes falling, presumably because the syllable, which is voiced throughout, has to reach a lower level of pitch compared to equivalent high-tone word (cf. /námtco?/, which continues in a pitch level similar to the previous word).

Tonal differences are retained in disyllabic postpositions, as shown in Figure 2.18, which produces the pitch traces from the clauses in (2.19). In Figure 2.18, the high tone postposition $tegk^ha$

(2.19)a) तर्ने हिंबा झेनावर थेन di k^him t^henk^ha jò?. this house above EX.PER 'It is on/above the house.' (KUN e) b) เวรา เมิม สราย พัรเ di k^him j*à*?. nàŋca this house inside EX.PER 'It is inside the house.' (KUN e) Figure 2.18. Pitch in postpositions $tenk^ha$ and nanea



Note that with the postpositions in Figure 2.18 the high tone is realized as a high-low sequence and the low tone as low-high sequence. This somewhat contrasts with what was above reported on high and low tone in disyllabic nouns. With nouns, the most important correlate of tone was shown to be the degree of rise from low to high pitch (greater rise with low tone words and lesser rise with high tone words).

In summary of tone and register, Denjongke words can be divided into high and low register. High register is associated with high(er) pitch and modal or stiff voice. Low register is associated with low(er) pitch and breathy phonation type. Register is predictable from the initial consonant of the word except when the initial is a sibilant, voiced nasal, voiced lateral or a vowel. Breathiness appears to be more difficult to perceive in sibilants, nasals and laterals than in plosives and affricates. Pitch seems to be the decisive factor in disambiguation. Therefore, Denjongke may be called a tonal language. Pitch is only marginally contrastive within the same register. In disyllabic words, the difference between high and low tone is realized as a difference in pitch rise between the first syllable and the second syllable (low tone words have a lower starting point and a greater rise in pitch than high tone words).

2.7 Stress or tone

I have not found stress to be a useful category for describing Denjongke. Caplow's (2016) study on disyllabic words in Balti, a toneless varity of Tibetic preserving archaic phonological features, shows that non-verbs (nouns, adjectives, numerals) stress the second syllable and verbs stress the first syllable. Moreover, Caplow (2016: 47) suggests that the transphonologization of stress, whose

primary correlate is fundamental frequency⁹², into tone may explain how toneless Proto-Tibetan developed tones. Caplow's observation bears resemblance to Denjongke in that Denjongke disyllabic nouns, when pronounced in clausal context, have higher pitch on the second syllable. There are, however, two reasons for not considering the pitch difference in disyllabic nouns in Denjongke as constituting stress. The first is that, Denjongke is sensitive to *how much* the pitch rises from the first syllable to the second one in disyllabic words (see §2.6.2), a distinction that is better described in terms of tone than stress. Second, the correlates of stress (pitch, intensity, vowel duration, vowel quality) are inconsistently pronounced in disyllabic words, especially when comparing words pronounced in context with those pronounced in isolation.

Similarly to stress in Balti (Caplow 2016), tone in verbs occurs on the first syllable, i.e. the monosyllabic verb root, whereas the tonally neutral verbal suffix attached to the verb may receive various pitches, depending on the intonational contexts. For instance, the nonfinal marker -di/ti may occur at either higher or lower pitch than the preceding verb root. One reason for heightened pitch on the verb suffix is anticipation of continuation, see §16.7.

2.8 Morphophonology

This section on morphophonology addresses variation of verbal suffixes (\$2.8.1), variation of negator prefix *ma*(*n*)- (\$2.8.2), reduction of the genitivized infinitive marker (\$2.8.3) and variation in agentive marking (\$2.8.4).

2.8.1 Verbal suffixes

Some verbal suffixes have two or more forms, depending on which sound the verb root ends in. These forms are summarized in Table 2.7. The form of the suffix following verb roots ending in /ŋ/ (usually pronounced as a lengthened nasalized vowel) cannot be phonologically predicted. Similarly, the form of the nonfinal suffix -ti/di is unpredictable with roots ending in a vowel. The verbs in Table 2.7 have the followings meanings: $kjap \frac{1}{2} \nabla \nabla^2$ 'strike, do', $p^h jak \operatorname{aggm}$ 'sweep', $\epsilon \epsilon' 2 \operatorname{aggm}$ ' comb', $p' ja \operatorname{ggm}$ ' do, $zo \operatorname{aggm}$ ' make', $p^h u \operatorname{ggm}$ ' blow', $tea: \operatorname{aggm}$ ' come (hum.)', $ts^h o: \operatorname{ggm}$ ' feel', $p^h y: \operatorname{ggm}$ ' offer, $len \operatorname{aggm}$ ' take', bom gas' grow', $mj\tilde{o}: \operatorname{ggm}$ ' finish, $l\tilde{o}: \operatorname{aggm}$ ' stand, $l\tilde{a}: \operatorname{aggm}$ ' be enough', $t\tilde{a}: \operatorname{aggm}$ ' send', $n\tilde{a}: \operatorname{aggm}$ ' give (hon.)', $t^h \tilde{o}: \operatorname{aggm}$ ' see'.

Suffix	Preceding context	Form	Examples
Infinitivizer	voiceless	-0	kjap-o, pʰjak-o, ɕɛ́-po (ɕɛ́?)
-po/-bo	short vowel	- <i>U</i>	р'ја-и, zo-и, р ^њ и:
	long vowel	-bo/βo/wo	$t \in a$:-bo, $ts^h o$:-bo, $p^h y$:-bo
	-n, -m	-bo	lèm-bo, bom-bo
	-ŋ	-po	mjồː-po, lồː-po
		-bo	tãː-bo, nấː-bo
Past/perfective	voiceless	-tce	kjap-t $\varepsilon\varepsilon$, p^h jak-t $\varepsilon\varepsilon$, ε é-t $\varepsilon\varepsilon$ ($\varepsilon\varepsilon$?)
-tee/ze	voiced	-ZE	p'ja-zɛ, zo-zɛ, lèn-zɛ, bom-zɛ
	-ŋ	-tee	$t^{h}\tilde{o}$:-t $\varepsilon\varepsilon$, $l\tilde{o}$:-t $\varepsilon\varepsilon$
		-ZE	tã:-zɛ, nấ:-zɛ

Table 2.7. Voicing alternation in verbal suffixes

⁹² In Balti, vowel duration was found to be a "weaker and inconsistent" cue for stress while intensity was deemed "not a factor" (Caplow 2016: 47).

	voiceless	-tɛɛ̃:	kjap-t $\varepsilon \tilde{\varepsilon}$:, p^h jak-t $\varepsilon \tilde{\varepsilon}$:, $\varepsilon \tilde{\varepsilon}$ -t $\varepsilon \tilde{\varepsilon}$: ($\varepsilon \varepsilon$?)
Progressive	voiced	-zẽː/zin	p'ja-zẽ:, zo-zẽ:, lèn-zẽ:, bom-zẽ:
-tɛɛ̃ː/zɛ̃ː/zin	-ŋ	-tɛɛ̃ː	$t^h \tilde{o}$:- $t \epsilon \tilde{\epsilon}$:, $l \tilde{o}$:- $t \epsilon \tilde{\epsilon}$:
		-zẽ:	$t\tilde{a}$:- $z\tilde{\epsilon}$:, $n\tilde{a}$:- $z\tilde{\epsilon}$:
Imperfective	voiceless	-to	kjap-to, p ^h jak-to, εέ-to (εε?)
-to/do	voiced	-do	p'ja-do, zo-do, lèn-do, bom-do
	-ŋ	-to	mjồː-to, lầː-to
		-do	tãː-do, nấː-do
Nonfinal	voiceless	-ti	kjap-ti, pʰjak-ti, ɕέ-ti (ɕε?)
-ti/di	voiced C	-di	lèn-di, bom-di
	(except - <i>ŋ</i>)		
	vowel	-ti	p'ja-ti, zo-ti
		-di	$p^h y$:-di, sà-di
	-ŋ	-ti	mjồː-ti, lồː-ti
	0	-di	tãː-di, lầː-di

As suggested by the word $\epsilon \epsilon^2$ 'comb' in Table 2.7, root-final glottal stop, although it disappears when a suffix is added, is treated as a voiceless ending. Other examples in addition to $\epsilon \epsilon^2$ are ta^2 $\sqrt{2}\sqrt{3}\sqrt{3}$ 'append' > ta:-po and $\epsilon^2 \approx \frac{1}{2}\sqrt{2}\sqrt{2}$ 'push' > $\epsilon - t\epsilon \epsilon n$. As further seen in Table 2.7, verb roots ending in /ŋ/ may have either a voiceless or a voiced onset in the suffix. The correct form has to be learnt by heart. Table 2.8 presents the correct suffix forms for some common verbs ending in - η .

Table 2.8. Verb suffixes with verb roots ending in $/\eta/$

Voiceless (-po, -tee, tee:, -to)	Voiced (-bo, $-z\varepsilon$, $-z\tilde{\varepsilon}$, do)	
$t^h \widetilde{O}$ ਂ ਗਰੱਨਾ 'see'	tã: ヮヮヽ 'send'	thun aga 'drink'
lầ: ara 'be enough'	nấ: གནང [.] 'give (hon.)'	<i>tsõ:</i> দর্জন 'sell'
$l\dot{\tilde{o}}$: बॅद्र 'stand (up)'	số: శాధా 'go.PFV (suppl.)'	súŋ المجامع: 'say (hon.)'
<i>mjồ</i> : ಫ್ರೆ⊏ 'finish'	$\dot{\tilde{o}}$: \check{a} 5. 'come'	ťãː הַהַאַי 'carry (hon.)'

The nominalizer -po occurs in four forms, -o, -u, -po, -bo (phonetically there is a fifth form - $\beta o/wo$, which is the realization of /b/ after long vowels). When preceded by root final /p/ or /k/, the bilabial stop is elided from the suffix, e.g. $kjap \frac{1}{2} \nabla \nabla^{*}$ 'strike' > kjap-o, $p^{h}jak \exp' > p^{h}jak$ -o. Although root final /k/ is utterance-finally typically realized as a glottal stop [?], there is a difference between roots having final /k/ [k]~[?] and those having a final /?/ [?]. Whereas roots ending in /k/ retain the velar stop and delete the bilabial stop from the suffix, roots ending in /?/ have a rather long vowel followed by the full infinitive marker -po, e.g. $t'ak \nabla''$ 'get well' > t'ak-o, $ta^2 \nabla \nabla''$ 'adorn' > ta:-po. When preceded by a root ending in a short (non-nasalized) vowel, the suffix becomes -u, e.g. $p'ja \nabla''$ 'do' > p'ja-u, $zo \nabla''$ 'make' > zo-u. In the case of final /o/, the vowel may also be just lengthened, e.g. $zo \nabla''$ 'make' > zo. If the vowel is long (usually because of a historical ending in /r/, /l/ or /s/), the nominalizer becomes -bo [bo]~[\betao]~[wo], as in $tea: \nabla \nabla \nabla'$ 'come (hum.)' > tea:-bo [teá: β o], ts^{h_O} : $\mathfrak{F}_{\mathfrak{T}}$ 'feel' > ts^{h_O} :-bo 'getting big', $len \otimes \nabla' > lem-bo$ 'taking'. Similarly to other suffixes, final /n/ may obtain either voiceless or voiced suffix, e.g. mjo:-po $\mathfrak{F}_{\mathfrak{T}}$ 'finished, finishing' vs. $t\tilde{a}$:-bo $\nabla \nabla \nabla \nabla''$ 'sent, sending'.

Suffix	Preceding context	Form	Example
Infinitivizer -po/bo/u	-p, -k	-0	kjap-o, pʰjak-o
Purpose/manner -pa	-p, -k	- <i>a</i>	kjap-a, p ^h jak-a
Conditional -patcene	-p, -k	-atsene	kjap-atcene, p ^h jak-atcene

Table 2.9. Consonant elision in verbal suffixes with initial -p

In verbal suffixes with initial k-, the initial plosive is elided when the preceding verb root ends in -k, making the interrogative and purposive/circumstantial forms of $p^h jak$ 'sweep' homophonous, $p^h jak-a$, see Table 2.10 (and §15.5.1 for purposive/circumstantial marker).

Table 2.10. Consonant elision in verbal suffixes with intial -k

Suffix	Preceding context	Form	Example
Polar interrogative -ka/ga	- <i>k</i>	-a	p ^h jak-a
Attenuated interrogative -kam/gam	- <i>k</i>	-am	p ^h jak-am

2.8.2 Negating prefixes

The negating prefixes are *ma*- (perfective) and *mi*- (imperfective). With verb roots in the low register, a nasal occurs between the negator and verb root, e.g. go_{75} $\sqrt[5]{5}$ (need' > *min-go* 2 $\sqrt[5]{5}$) $\sqrt[5]{5}$ (need not', sa = 1 'eat' > *man-za* $\sqrt[5]{5}$ (did not eat, don't eat'. Table 2.11 contrasts negation in low-register and high-register words.

Table 2.11. Negation of low vs. high register verbs

Low register		High register		
Affirmative	Negated	Affirmative	Negated	
p'in fja. 'give'	mam-bin, mim-bin	pik दोग' 'pull out'	ma-pik, mi-pik	
$t' \varepsilon$: $\neg \alpha$ 'have time to'	man-dɛː, min-dɛː	tek झेगल' 'place up'	ma-tɛk, mi-tɛk	
<i>t'õ:</i> (hon.)'	man-dõː, min-dõː	t ^h uŋ ¬' 'shine (hon.)'	ma-t ^h uŋ, mi-t ^h uŋ	
k'o र्ग 'understand'	maŋ-go, miŋ-go	kʰøː řɨঝ· 'boil (intr.)'	ma-kʰøː, mi-kʰøː	
$s\dot{a}$ = 'eat'	man-za, min-za	só गर्रें 'tend'	ma-só, mi-só	
bak הקקה' carry'	mam-bak, mim-bak	<i>pã</i> .' Ŋ⊏' 'forsake'	та-раŋ, ті-раŋ	
dò? ۆخ: 'sit'	man-dø?, min-dø?	tøn यहेंब. 'take out'	ma-tøn, mi-tøn	
go? न्यॅूबा 'need'	maŋ-go?, miŋ-go?	kok कॅल. 'dig'	ma-kok, mi-kok	
zim ग्रहेअ [.] 'sleep (hon.)	man-zim, min-zim	sák नजग्राजा 'accumulate'	ma-sak, mi-sák	
zak न्तवग 'put'	man-zak, min-zak	cá: جم 'shine'	ma-ɛaː, mi-ɛáː	
dze: क्रह्रया' 'meet (hon.)'	man-dzɛː, min-dzɛː	tea: אסא' 'come (hum.)'	ma-tɛaː, mi-tɛaː	

The frequent verb *p'ja* has, in addition to the regular *mam-bja/mim-bja*, a special, reduced negated form *ma-jà/mi-jà*.

2.8.3 Reduction of the genetivized infinitivizer

The genitivized infinitivizer $-b\phi$: [w ϕ :] is by some speakers, and especially in fast speech, reduced to [i], see (2.20-21).

(2.20) a) $\operatorname{gen} \widehat{a} := di$ $\widehat{cu} = w \overline{o}: \qquad g \widetilde{a} := di$ $\operatorname{say} = 2 \operatorname{INF.GEN} \qquad \operatorname{time} = \operatorname{DEMPH}$ 'when saying' (KLT) b) $\operatorname{gen}^{2,93} \operatorname{spn} \operatorname{end}^{2}$ $\widehat{cu} = i \qquad g \widetilde{a} := di$ $\operatorname{say} = 2 \operatorname{INF.GEN} \qquad \operatorname{time} = \operatorname{DEMPH}$

'when saying' (KLT)

(2.21) a) සු·ති? සුද

ta-wø: gã: look-2INF.GEN time 'when looking' (SD)

b) ਕੋਟੇ ਸੋਟ *ta-i gã:* look-2INF.GEN time 'when looking' (SD)

2.8.4 Agentive case

With the personal pronouns $\eta \dot{a} = (1', k^h u_{\mathbb{N}})$ the' and $m \dot{u} \in (1')$ she' the agentive case can be marked by vowel lengthening and raise of tone from low to high.⁹⁴ With $k^h u$, which is already high tone, the modification reduces to vowel lengthening, $k^h u : \mathbb{N}^n$ 'I.AGT'. With $\eta \dot{a}$ and $m \dot{u}$, the tone changes from low to high along with vowel lengthening, $\eta \dot{a} := \mathbb{N}^n$ 'I.AGT', $m \dot{u} := \mathbb{N}^n$ 'she.AGT'. Figure 2.19 illustrates the tonal difference between $\eta \dot{a}$ and $\eta \dot{a}$. The clause with $\eta \dot{a}$, which is actually infelicitous, was produced just for comparison. Vowel length is not clearly visible in Figure 2.19, because the pitch traces also record the voicing in /ŋ/. Manual measurements of vowel lengths in $\eta \dot{a}$ and $\eta \dot{a}$: yielded 0,12 seconds and 0,22 seconds respectively, showing a clear difference in length. The agentivization of $m \dot{u}$ 'she' functions analogously to $\eta \dot{a}$.

⁹³ This written form, along with the analogous form in (2.21b), is highly experimental and used here only to provide a written representation of reduced pronunciation in spoken language.

⁹⁴ An alternative strategy is to use the agentive marker =ki/gi, e.g. $\eta \dot{a}(z)=gi$, $k^{h}u(z)=gi$, $m\dot{u}(z)=gi$.

Figure 2.19. *ŋà* vs. *ŋá*: in context (consultant KN)



2.9 Phonemic script

The phonetic/phonemic symbols are from the IPA except for /g/ representing IPA /q/ and /a/ representing IPA / ν /. When italicized in example sentences, /a/ becomes /a/. Two major issues related to the phonological script were how to treat nasals/nasalization and the glottal stop. For reasons given in §2.3.4, final nasal stops and nasalized vowels are in the examples written as they are actually pronounced. That is, the historical velar nasal following back vowels /a/ and /o/ is phonemically written as nasalization (i.e. $t\tilde{a}$: 555; 'send', $t^h\tilde{o}$: $a\tilde{a}$: 'see') while with other vowels the same historical velar tends to be retained in pronunciation (i.e. $\epsilon i \eta = t$, $\epsilon i \eta = t$, GET 'drink'). Similarly, lexeme-internal nasals and nasalized vowels are represented as actually pronounced. This means that verb root preceding a suffix may be written in various ways, depending on the suffix with which the root assimiates, i.e. the equative copula \hat{i} occurs in various forms: *im-bo* (with infinitivizer -*po/bo*), *in-do* (with imperfective marker -*to/do*) and *in-k^h* $\tilde{\varepsilon}$: (with nominalizer $-k^{h}\tilde{\epsilon}$.) A nasal which assimilates to a following retroflex is written as n (as in *in-to*) rather than η , because the retroflex nasal does not occur as an independent phoneme. Nasal assimilation, however, is not represented over word boundaries which are marked by spaces, i.e. the complex copula $i\eta \cdot k^{h} \epsilon n \ b \epsilon 2$ is written as such although its pronunciation is typically [in-k^h \epsilon m $b\epsilon$?] (for reasons why the final auxiliary is written separately, see §1.2.7.2).

The second problematic issue in the phonemic script is the status of the glottal stop. The glottal is phonemic word-finally and it functions in an intricate relationship with pitch, vowel length and vowel quality, as discussed in §2.6.1.2. In brief, word-final glottal stops are marked in the phonemic scripts although they are realized only when followed by a pause. When a word-final glottal occurs in another context than preceding a pause, the glottal is typically realized as lengthening of the vowel and, at least for some speakers, a fall in pitch. The phonetic realization of underlying word-final glottals stops requires more research, and a fruitful starting point at this point is to mark them for those words in which the glottal occurs when the word is pronounced in isolation (and so also followed by a pause). However, stem-final glottal stops are not written if the stem is followed by other morphemes and, thus, the glottal is not word-final, e.g. ta:=lo 'tiger=DAT' (ta2 'tiger'), $c \epsilon - c \epsilon (2)$ is not

written when an auxiliary follows. Although the auxiliary is written separately it is phonologically part of the same utterance with $-\epsilon\epsilon(2)$ and hardly ever divided from the infinitive by a pause.

The phonemic transcription below attempts to follow spoken pronunciation, not reading or spelling style pronunciation (for discussion on the differences see Sprigg 1991), e.g. the progressive marker $\neg \partial \neg$ *bzhin* is transcribed in literary examples, following spoken pronunciation, as $tc\tilde{\epsilon}:/z\tilde{\epsilon}:$ despite *zin* being the reading-style pronunciation.

2.10 Summary remarks

This chapter showed that Denjongke has 43 consonants all of which, with the exception of /?/ (which is phonetic word-initially), occur word-initially and 8 of which occur word-finally. One of the distinguishing features of the present phonological analysis was seen to be that plosives and affricates have a four-way contrast in voicing/aspiration. A detailed treatment was given to the perhaps controversial category of lightly and inconsistently aspirated ("breathy") consonants (§2.2.2). It was also shown that the phonemic category of voicing in voiced plosives and affricates can be phonetically produced as either prevoicing (i.e. [gka]) or prenasalization (i.e. [ŋga]). Moreover, Denjongke was seen to have quite a rich variety of preaspirated consonants, four nasals and two liquids.

This chapter also showed that Denjongke has eight vowels (if long vowels are not counted separately). It was shown that nasalization and length are contrastive in vowels. The relationship of front unrounded vowels proved particularly tricky for analysis, but a three-way distinction in long vowels between /i:/, /e:/ and / ϵ :/ was firmly established. Final glottal, which is pronounced utterance-finally, is realized as length utterance-medially.

The section on register, pitch and tone showed that Denjongke words are divided into high register and low register. High register was seen to be associated with stiff voice quality and high pitch, low register with breathy voice and low pitch. The conclusion was that because pitch is only partly but not always predictable from the initial consonant, Denjongke can be called a tone language. I also presented some initial evidence that there may be pitch-contrasts within the low register. A study of pitch phenomena in disyllabic words showed that both high-register and low-register disyllabic nouns (pronounced in a sentence-frame) have a low-high pitch pattern. The difference was shown to be that low-register words start lower and have a greater pitch difference between the first and the second syllable, whereas high-register words start higher and have a smaller frequency difference in pitch rise.

I also described some segmental phonological processes (vowel assimilation, phoneme elision and consonant lenition) and showed that within morphophonology, some allomorphs show a partly unpredictable voicing pattern which has to be learned on a case-by-case basis (e.g. nonfinal *-ti* vs. *-di* in §2.8.1).

3 Word classes, suffixes and clitics

This chapter provides an overview of Denjongke word classes, affixes and clitics. Whereas the discussion in this chapter focuses on morphology, the functions of the different formatives are described and discussed in more detail in later chapters. I begin with a general discussion on the terms "word," "affix" and "clitic" (§3.1). That is followed by an introduction to word classes and the types of subclasses that can be identified within word classes.

Denjongke has four major (or open) word classes, nouns (§3.2), verbs (§3.3), adjectives (§3.4) and adverbs (§3.5). Major word classes differ from minor word classes in having more lexemes and in being more open to adding new lexemes (hence the term "open word class"). Moreover, major word classes typically consist of content words, which are less frequent and have a more specific meaning than the members of minor/closed word classes, which may also be called function words (Haspelmath 2001: 16539).

Minor word classes (see §3.6), consisting of function words, have fewer lexemes than open word classes and are less open to new words (hence the alternative term "closed word classes"). Denjongke minor word classes are personal pronouns (§3.6.1), reflexive pronouns (§3.6.2), reciprocal pronouns (§3.6.3), indefinite pronouns (§3.6.4), demonstratives (§3.6.5), question words (§3.6.6), numerals (§3.6.7), postpositions (§3.6.8), connectives (§3.6.9), interjections (§3.6.10) and discourse particles (§3.6.11). Affixes are described with the word class they attach to. Clitics, because of their transcategorial nature (see §3.1), are treated under a separate heading in §3.7. Onomatopoeic words are treated as a subclass of ideophones, which are syntactically adjectives or adverbs but have distinctive phonological, morphological and semantic features, see §17.1.

3.1 Words, affixes and clitics

This section briefly discusses the definition of word, affix and clitic in Denjongke. Word is crosslinguistically a challenging concept which may be defined using grammatical, phonological and/or orthographical criteria (Aikhenvald 2007: 1-2, Dixon 2010b: 3-19). Phonologically word in Denjongke may be defined, following Payne's (2006: 20) working definition, as "the smallest structural unit that can occur between pauses." This implies that words (unlike clitics) are not phonologically bound to other morphemes and may hence be used independently, for instance, as short answers to content questions.

Grammatical criteria and phonological criteria for wordhood, however, do not always coincide. For instance, some Denjongke postpositions have two forms, a disyllabic, phonologically independent form, which may occur as an answer to a content question (e.g. $s\hat{a}:t\varepsilon$ 'until, straight on' [consultant KT]), and a monosyllabic cliticised form, which typically cannot occur independently as an answer to a question (e.g. $=s\hat{a}:$ 'until').⁹⁵ A fact suggesting that $s\hat{a}:t\varepsilon$ 'until' is a word and $=s\hat{a}:$ 'until' a clitic is that the word $s\hat{a}:t\varepsilon$, unlike its cliticized form, has, at least in some varieties of Denjongke, the (secondary) meaning 'straight on' (as in an answer to the question 'Where shall we go from this crossroads?'). In the novel Richhi, $s\hat{a}:t\varepsilon$ (WD $=s\hat{s}: zang-ste$) is separated in writing from the previous word by a space whereas the clitic $=s\hat{a}:$ (WD $=s\hat{s}: zang$) is attached to the previous word. Grammatically, however, both $s\hat{a}:t\varepsilon$ 'until, straight on' and $=s\hat{a}:$ 'until' function essentially identically in phrases such as $t^hor\hat{a}: s\hat{a}:t\varepsilon/t^hor\hat{a}:=s\hat{a}:$ 'until tomorrow',

⁹⁵ These morphemes are somewhat analogous to *not* and *n't* in English (see Dixon & Aikhenvald's 2003: 27).

suggesting that the phonologically reduced form $=s\tilde{a}$: is as much a grammatical word as the fuller form $s\dot{\tilde{a}}$: $t\epsilon$. Another form which occurs both as a monosyllabic clitic and a disyllabic word is the relator noun =tsa:/ $tsak^{h}a$ 'at (the root of)'.

Affixes and clitics, in contrast to phonological words, depend phonologically on the word they are attached to. Syntactically, affixes and clitics "cannot govern or be governed by other words, cannot require or undergo agreement, and cannot head phrases" (Bickel & Nichols 2007: 172). The main criteria used here to distinguish clitics from affixes, following Bickel & Nichols (2007: 174-175), is transcategoriality: clitics are freer than affixes to occur with more than one type of part of speech or phrase. For instance, verbal affixes only occur attached to a verb stem. Case clitics, on the other hand, occur both with noun phrases and attached to the verb complex (see Tournadre 2010 for transcategoriality of Classical Tibetan cases). The plural marker =*tsu* occurs at the end of the NP, where it may attach to both nouns and noun modifiers. For that reason =*tsu* is here considered a clitic. Denjongke clitics can be divided into the following categories: case clitics (§3.7.1), emphatic clitics (§3.7.2), clausal clitics (§3.7.3) and other clitics (§3.7.4).

3.2 Nouns

This section introduces the characteristics and distinctive features of nouns in general (\$3.2.1) and then goes on to describe ordinary and honorific nouns (\$3.2.2), the principles of deriving nominals from verbs (\$3.2.3), nominal suffixes (\$3.2.4) and compounding (\$3.2.5).

3.2.1 Introduction to nouns

Nouns (from Latin nomen 'name') prototypically refer to physical entities such as objects, living creatures and places but are by extension also used for abstract entities (e.g. *bjakha* restraction (from verbs) and adjectives by their ability to be possessed and modified by numerals and adjectives. Moreover, unlike verbs, nouns allow the plural marker =*tsu*, case clitics, demonstratives and various emphatic clitics to be attached to the base form. Some of these morphemes may also be attached to the verb, but only after some additional verbal morphology such as the nominalizer in (3.2). In (3.1) the plural marker attaches to noun base but (3.2) illustrates that additional verbal morphology (here nominalizer $-k^h \tilde{\epsilon}$) has to be added to the verb before attaching the plural marker.

(3.1) ह्यते मी.⁹⁶ चन्या मा.र्झ. sá=i=gi

 $s\dot{a}=i=gi$ daku=tsuground=GEN=GEN owner=PL 'Land-owners (lit. owner's of land)' (BP, BB discussion)

⁹⁶ I have not seen double genitive, which is prevalent in spoken Denjongke, been represented in written Denjongke. The innovative writing used here and elsewhere in this grammar is a written representation of spoken language and may seem contrary to good literary style.

Examples (3.3) and (3.4) show that the same is true with case clitics: the dative-locative =lo may be directly attached to a noun base (3.3) but does not typically attach to a verb root without some additional marking (here progressive), see (3.4)⁹⁷.

mi=lo' $c \epsilon nt^h \tilde{o}$: $ma-s \dot{a}$:=s. human=DAT contempt NEG-heap=QUO 'Do not look down on people (it is said).' (UU, Deer story 1)

(3.4) $ac_{7} c_{7} c_{$

Most Denjongke nouns are mono- or disyllabic. Nouns containing more than two syllables are likely some type of compounds, such as *bjam-ka:-riŋ* 'mosquito' (lit. 'fly-foot-long'), although exceptions exist, e.g. *kaŋkara* 'crab'.⁹⁸

3.2.2 Ordinary and honorific nouns

Typically of Tibetic languages, Denjongke has an honorific system in nouns, which means that two different lexical forms are used for the same referent, one in honorific register and the other in ordinary register. Not all nouns have an honorific form, and not all speakers know all the honorific forms, although using them is considered a sign of linguistic acumen. The honorific forms are often the same as or similar to honorific forms in Lhasa Tibetan. Honorific nouns are formed in several different ways, which are described here. Table 3.1 presents examples in which the ordinary and honorific forms bear no formal resemblance.

Ordinary register		Honorific	register	Gloss
ठायें।	go	<u>5</u> .	ú	head
ন্যে.	kha	Ga1.	sè:	mouth
લે	tee	শ্রিয়ার্থ.	dza?	tongue
झ्र.ग्रेग.	nagu?	শৃদন্ধ'	са́:	nose
শ্দ≈, শ্দ∽র্থ	kaːm, kãːpo	ৰ্বম্থ	сàр	foot
ञेया'€्या'	mì:do?	झेव.	tsen	eye
ক্র্ম'র্স্র্যা'	námt£0?	श्रुव् (र्द्धण)	pén(tco?)	ear
λ.	só	केंग्रेंग	ts ^h em	tooth
व्यया गें।	làko	শ্ৰন্য.	t¢ ^h a?	hand
रू,	t\$^hu	æ7.	t¢ ^h ap	water

Table 3.1. Ordinary and honorific nouns with no formal resemblance

⁹⁷ For an exception to this rule, see example (15.106c).

⁹⁸ Unanalyzable nouns having more than two syllables in Donwang Tibetan are typically names of small animals (Bartee 2007: 91). I am thankful for Bertil Tikkanen for pointing out that the word *kaŋkara* is an Indo-Aryan Ioan. In Nepali spoken in Sikkim the word occurs as *gaŋŋata* 'crab'. Oriya (*kaŋkara* 'crab') and Tirahi (*kaŋgara* 'spider') also have pronunciations almost identical to Denjongke (Turner 1962-1966).

ਬ੍ਰਿਕਾ	k ^h im	यचित्रायम्.	zimkhã:	house
झेवा.	<i>lam</i>	ৰ্বনথ তথা	sàptsa?	shoe
ই বিশ্বমান্য	to:pa?	য়৾৾য়৾ঀ৽৸৸য়ৣয়৽	sǿːţym	vegetables (with rice)
র্বান্থ'	k'o	ব'ন্রন'	nàza	clothes
એંદ્ર.	mìŋ	মৰ্চ্ব	ts ^h en	name
জার্যা	ápo	と す.	jàːp	father
জ.অ.	áma	લીથા.	jùm	mother
£.	р'и	শ্রন্থ	sé?	son
न्विभ.	p`um	শ্রদ্রাধ্যম.	sé:m	daughter

In other cases, the ordinary and honorific forms resemble each other. First, the honorific form may be a compound where a monosyllabic honorific word, either a noun referring to a body part or a relevant verb, forms a compound with the ordinary form, see Table 3.2. Typically the honorific noun is preposed to the ordinary form, but postposing the honorific word to the ordinary word is also possible (see $k\varepsilon$:dza? m_{i} : "anguage" in Table 3.2). Simple compounding is a common strategy when the ordinary noun is monosyllabic and thus the resulting form does not exceed two syllables.

Table 3.2. Honorific nouns formed by compounding

Independent use	Honorific with the affix	Ordinary register	Gloss
ন্দেম <i>càp</i> 'foot (hon.)'	ଜ୍ମଷ୍ୟଙ୍କୁଷ୍ୟ <i>càplam</i>	झुम्र' <i>lam</i>	shoe
	ลุกพากลังา càpkor	ראָ <i>kor</i>	tour
\mathbf{F} ku 'body (hon.)'	भुं पाञ्चगर्या kuzu?	যান্ত্রবাধ্য <i>zu?</i>	body
	ay'sar' kuõ:	קקקי <i>ố</i> :	power
डाया' <i>tc^ha?</i> 'hand (hon.)'	छग्-नृहत्य tehani:	JEAN NÝ.	money
	द्यगाल्पसः teha:le?	पॉर्पाय: jó?, avar lè?	work, deed
ন্থ্ৰন্থ dza? 'tongue (hon.)'	ন্নন্-'শ্রুবাঝ' kɛːdza?99	strict kε?	language
ह्याय thu? 'mind (hon.)'	मुगाया तयात्र thugen	त्रगत. gen	responsibility
ar cè: 'face (hon.)'	פתייתובה: cè:dõ:	מובר: dõ:	face
যার্মন্য sø: 'offer (hon.)'	มสัญ. E. Sø:za	E' te'a	tea
মন্ন ze: 'eat, have (hon.)'	त्रावेष:घुत्रव: ze:thum	ष्ट्रन्त्र t ^h um	spoon
المالية: súŋ 'say (hon.)'	anger: súnke?	بار ke?	voice

In other instances, where the ordinary noun is disyllabic, the honorific word replaces the first syllable of the ordinary form, see Table 4.3.

⁹⁹ This meaning is from consultant CY. Consultant KUN, on the other hand, gave this word the meaning 'accent, distinct way of pronouncing a language'. KUN did not have an honorific equivalent for $k\epsilon^2$ 'language'.

Ordinary	y	Honorifi	C	Gloss	
pyntc ^h e?	झुव अळेट्र.	kut¢^e?	ৠ.ૹૹ૾ઽ.	sibling, relative	
ára?	জ্ঞান্যা	ze:ra?	ননিম'ন্যা'	alcohol	
pa(h)ip	ন'র্দৃন'	sǿ:(h)ip	শর্মিঅ'র্নৃন্ন'	beer container (of bamboo)	
ákar	জে'শাম'	ze:kar	র্বারীকাশ্যাম'	chilli	
døsa	र्श्वन् स	zu:sa	ন্ত্রিযাম'ম'	residence	
zuːɲɛ̃ː	याज्यायाया मुत्रवा	kupẽ:	भु'ग्रहेव'	image	
mìkce:	र्व्या नेपा	tcence:	ਝੂਰ-ਸੇਆ	spectacles	

Table 3.3. Honorific nouns formed by replacing a syllable

The honorific prefix may also be prefixed to the ordinary form with the result that the final syllable of the ordinary form is dropped because there is a strong preference for disyllabicity in nouns, see Table 3.4, where the common syllable between the ordinary and honorific forms is given in bold. Note that there may be a phonological difference in how the historically same syllable is realized as the first syllable of a word and as the second syllable of the word (e.g. WD for smon becomes $m\epsilon$ - in mélam 'prayer' and $-m\delta$: in thum\delta: 'prayer [hon.]').

Table 3.4. Honorific nouns formed by prefixing and final syllable elision

Ordinar	у	Honorifi	c	Gloss
mé lam	ষ্ট্র্বি'শেষ'	thu mõ :	শ্রিন্যম:ষ্ট্রব:	prayer, wish
càm bu	ਕ੍ਰੋਕਾ.ਦੁ.	ú za	75.°g.	hat
có ku	र्वेगारगु.	tc ^h a co?	छगा र्वेगा	paper
ka jø	קזודיעאי	sè∶ kar	ৰম্প'দশান	cup
tsam po	रूवा र्ये'	sǿ: tsam	यर्षेल. क्र	tsampa
dø :po	वर्नेन्'र्थे'	t ^h u dø?	ष्ठ्रगुरु: २२ॅन	wish, desire
p ɛtɕʰa	न्दो'क'	$t \epsilon^h a p \epsilon$	छग्।'न्ये'	book
nú ku	\$J'T]'	tɕʰa ɲu?	स्रुग] श्रुग]	pen
ka jø:	ᠵ᠋ᡢᠯᢩ᠆᠋᠄ᡫᠮᢆᠬ	<i>ω</i> ὲ∶ ka(r)	वयानगार.	cup, bowl
qo mo?	ર્ફે.સં.	ze:do	শ্বিশ:শ্র্র	bamboo straw

3.2.3 Deriving nouns from verbs

In her analysis of Tibeto-Burman languages of the Himalayas, Genetti (2011: 164) points out that nominalization may occur both on the morphological level (producing lexical nouns) and on the syntactic level (allowing a grammatical clause to be treated as a noun phrase). Denjongke has several productive morphemes that allow verbs to be treated as nouns or clauses to be treated as noun phrases, see Table 3.5.
Infiniti	Infinitive		with <i>p^hy:</i> 'offer (hon.)'
-687	'to x'	р ^ь у:-се?	'to offer'
-ni	'to x'	p ^h y:-ni	'to offer' ¹⁰⁰
-po/bo	'(the act of) x-ing'	p ^h y:-bo	'(the act of) offering'
Nomin	alizer		
$-k^{h}\widetilde{e}$:	'the one x-ing'	p^hy :- $k^h \widetilde{e}$:	'the one offering'
-sa	'the place of x-ing'	p ^h y:-sa	'place of offering'
-tã:	'the way of x-ing'	p ^h y:-tã:	'way of offering'

Table 3.5. Nominalizing markers

All the markers in Table 3.5 are highly productive in that they can be added to almost any verb. Semantically, the forms ending in $-c\varepsilon^2$, -ni and -po/bo are verbal nouns which refer to the action denoted by the verb, whereas the markers $-k^h\tilde{\varepsilon}$:, -sa and $-t\tilde{a}$: derive noun-like words referring to person, place or way of doing respectively. Because of this difference in semantics, it is useful to make a conceptual and terminological distinction between infinitive markers ($-c\varepsilon^2$, -ni and -po/bo) and other nominalizers ($-k^h\tilde{\varepsilon}$:, -sa and $-t\tilde{a}$:), although strictly speaking also the infinitives are nominalized forms in that they allow verbs and clauses to function like nouns or noun phrases. A syntactic distinction between the infinitive markers and the nominalizers -sa and $-t\tilde{a}$: is that the former participate in tense, aspect and evidentiality-marking auxiliary constructions (e.g. p^hy : $-c\varepsilon$ $b\varepsilon^2$ [offer-INF EQU.NE] 'will offer, offers') whereas the latter do not.¹⁰¹ Furthermore, infinitives are the forms used in complement clauses (see §14), which is the basic function of infinitives/masdars cross-linguistically (Shagal 2017: 5).

All the markers in Table 3.5 can nominalize a clause. To accommodate uses as clausal nominalizers, I refer to $-k^{h}\tilde{\epsilon}$, -sa and $-t\tilde{a}$: as "nominalizers" rather than as "nominal suffixes." The general term "nominalizer" subsumes also infinitives, which are here considered a special class of nominalized forms (i.e. those nominalized forms which nominalize action itself). The three infinitive forms are discussed under verbal suffixes in §3.3.6. The next section describes the nominalizers $-k^{h}\tilde{\epsilon}$:, -sa and $-t\tilde{a}$: and other nominal suffixes.

3.2.4 Nominal suffixes

The nominal suffixes described in this section can be divided into simple nominal suffixes and nominalizers. Simple nominal suffixes *-po* (§3.2.4.1) and *-m(u)* (§3.2.4.2) attach to a noun and derive another noun. They do not participate in clausal nominalization. Nominalizers $-k^{h}\tilde{\epsilon}$: (§3.2.4.3), *-sa* (§3.2.4.4) and $t\tilde{a}$: (§3.2.4.5), on the other hand, not only derive nouns from verbs but also nominalize clauses.

3.2.4.1 Nominal suffix -po/bo

The suffix -po/bo is related to the nominal suffix -pa (WT $\leq pa$) 'person having to do with' (Beyer 1992: 120) and the nominalizer -pa/ba in Classical Tibetan (Beyer 1992: 299). The uses of -po/bo are varied and complex in Denjongke, as shown by the summary of uses in Table 3.5.

 $^{^{100}}$ -*pi* is close in meaning to -*c*e? but less productive. For more information, see §3.3.6.1 and §8.2.8.

¹⁰¹ The nominalizer $-k^{h}\tilde{\epsilon}$, on the other hand, participates in the present habitual construction ($p^{h}y$:- $k^{h}\epsilon n \ b\epsilon$? [offer -NMLZ EQU.NE] 'offers'). However, because of its otherwise noun-like semantics, it is not called an infinitive.

10010 010 00	•••••••••	$P \circ i \circ \circ$				
	Unproductive	Productive				
Form	- <i>po</i>	-po/bo				
Attaches to	noun,	verb numeral			noun	
	verb (rare)				pers. name	place name
Glossing	not glossed	II infinitive	collective	ordinal	associative	associative

Table 3.6 Uses of the suffix -po/bo

As suggested by Table 3.6, a distinction should be made between historical, unproductive uses (-po) and synchronic, productive uses (-po/bo).¹⁰² Whereas the unproductive uses have generally resisted voicing assimilation (e.g. *lùŋ-po/lùmpo* Sec 'locality, place'), the productive uses are more prone to voicing assimilation (e.g. *dm-bo* 'the act of coming', see also Table 2.7). As shown by the last row in Table 3.6, in this grammar those uses of *-po* which are considered lexicalized and unproductive (i.e. when *-po* is seen attached to other nouns than personal names and place names) are not glossed at all. The productive uses, on the other hand, are glossed according to the specific uses, because this practice is more informative than using the same gloss (for instance "nominalizer") for all the uses.

Historically, -po has been used to form lexical nouns from verbs:

tsøpo/tsɛpo 'debate (noun)' from WT ₹5⁻ *rtsod* 'debate (verb)'

More often, however, *-po* has derived nouns from other nouns:

sàmpo 'bridge'	from WT and sam 'line, continuity'
lùmpo 'locality, place'	from WT est 'ung 'valley, land'.

Synchronically, -po/bo may still attach as a derivational marker to nouns, but the productive uses seem to be limited to personal names and place names, which are associative in meaning. With noun referring to a person, the associative meaning is 'those associated with person x' and with a location the meaning is 'person who is from location x'. For an associative meaning where -po/bo attaches to a personal noun, consider (3.5).¹⁰³

(3.5) ਲ੍ਹਾਜ਼ੁਰਾ ਛੋਂ ਸੋਸ ਸੱੱਧ ਸ ਸੋਸ ਵੇ ਕਾ ਨੇ ਰੋਕਾ ਲੱਗ ਸੇ ਕੇ ਸਾ ਨੇ ਕਿ ਸਾ ਕਿ ਸਾ ਕੇ ਸਾ ਕੇ ਸਾ ਕੇ ਸਾ ਕੇ ਸਾ ਕੇ ਸਾ ਕੇ ਸ

úgjen ts^heriŋ-bo t'ariŋ p^hina hoţel oupeniŋ p'ja-wa
PN PN-ASSOC today over.there hotel(Eng.) opening(Eng.) do-PUR số:?
go.PFV
'Did Ugyen Tshering and his family go to open the hotel today?' (interrogation by rising intonation) (PT kitchen discussion)

For examples of associative meaning with nouns referring to places, consider the following words:

¹⁰² For similar argumentation for -pa vs. -pa/ba in Purik, see Zemp (2018: 110).

¹⁰³ For the associative use of the largely similar morpheme -pa in the Tibetic language Purik, see Zemp (2018: 112).

dɛndzoŋ-po	'Sikkim-dweller; person	from dendzõ: (กรุงาลักง 'bras-ljongs 'Sikkim')
	of Sikkimese Bhutia ethnicity	y'
gjagar-bo	'Indian'	from gjagar 'India' (WD གྱ་གར་ rgya-gar)
p ^h iliŋ-po/	'foreigner'	from p ^h iliŋ 'out(side)' (WD 資訊云 ph'yi-gling)
t& ^h iliŋ-po ¹⁰⁴		from tchilin 'out(side)' (WD g) are phyi-gling)

Occasionally, the more typically Central Tibetan ending -pa/ba is heard instead of -po/bo, e.g. *làtɛuŋba/làtɛuŋbo* 'person from Lachung', $p^hiliŋbo/p^hiliŋba$ 'foreigner'.

Some more lexicalized forms may also be characterized as associative:

nàpo	'fisherman'	from WT 3 nya 'fish'
sìŋpo	'farmer'	from WT and 'field'
nè:po	'patient'	from WT and 'illness'
nàŋpo	'insider, Buddhist'	from WT and 'inside(s)'
tɛʰipo	'outsider, non-Buddhist'	from WT gr phyi 'outisde'

Occasionally the suffix *-po/bo* also attaches to a verb to mark the agent, e.g. $k^h u$: *zo:-bo* [bread make-*po*] 'bread maker, baker', *tea? dum-bo* [iron hit-*po*] 'iron-hitter, blacksmith'. This use of *-po/bo* overlaps the semantic domain of the nominalizer $-k^h \tilde{\varepsilon}$: (§3.2.4.3), which is the typical morpheme for referring to the doer of an action. My hypothesis is that when referring to the agent of an action, *-po/bo* is more lexicalized and refers to stable identity whereas $-k^h \tilde{\varepsilon}$: is more likely used on an ad hoc basis and refers to the doer of an action in a specific situation.

As shown in Table 3.6, the suffix -po may also attach to numerals to form a collectivized nominal with the meaning 'a group consisting of x (number) instances of y (noun)' (see §3.6.7 for collective uses of -po/bo and the similarly functioning collectivizer -ga).

3.2.4.2 Nominal suffix -m(u)

The full form of the nominal suffix -m(u) is homophonous with the feminine third person pronoun and mainly occurs in words that have female referents, see Table 3.7, although it also occurs in some words with no clear feminine connection, e.g. WD $\frac{1}{2} \sqrt{n} rol - mo > r \sigma m$ 'cymbal'.

¹⁰⁴ The variant initials $t\varepsilon^{hi}$ and p^{hi} derive from two different reflexes of WT $\frac{1}{2}$ phyi 'out(side)', the first corresponding to the typical Central Tibetan pronunciation and the latter (p^{hi}) to the typical Sikkimese reflex of $\frac{1}{2}$. Yet another alternative pronunciation for the word is $t\varepsilon^{hi}rig$ -po. The meaning 'foreigner' may also be expressed by the word $t\varepsilon^{hi}g\varepsilon$ -po $(\frac{1}{2}g^{-1}\varepsilon' phyi-rgyal-po)$.

Masculine			Feminine			
Noun	WD	Meaning	Noun	WD	Meaning	
p'u	-5 [.] ри	'boy'	p'um	ਜੁਕ਼ਾ <i>bum</i>	'girl'	
<i>ļa</i>	झ lha	'god'	<i>lamu</i>	झुर्खें lha-mo	'goddes'	
gɛːp, gɛːpo, gɛːpu	ক্রুণ্ণন' rgyalb ক্রুণ্ণর্না rgyal-po	'king'	gɛːm(u)	ਜ਼੍ਰੁਧਕਾ rgyalm ਜ਼ੁਧਾਕਾਂ rgyal-mo	'queen'	
pʻjapu	ई'र्बे' b'ya-pho	'rooster, cockerel'	p'jam	ৰ্ভ্ৰঝ [.] b'yam	'femal hen'	bird,
jóku	यविंग्या'गु' gyog-ku	'servant'	jóːm	यॉर्थेयास्र gyogm	'servant (fem.)'	
gopøn	दर्षे'न्देंग्न' 'go-dpon	'director, principal (m.)	gopøm	दर्षा-दर्भवम्र 'go-dponm	'director, principal (fem.)'	
lópøn	र्भ्रेन'र्न्सन' slo-dpon	'teacher (m.)'	lópøm	र्झन'दर्भेब्रब' slo-dponm	'female teacher'	
t ^h apøn	নদ্রন'ন্ ^হ র্যন' 'khrab-dpon	'actor'	t ^h apøm	নন্দ্রন'ন্র্ম্বর্ম' 'khrab-dponm	'actress'	

 Table 3.7. Some masculine-feminine noun pairs

In Table 3.7, the last three words differ from other feminine words in that rather than replacing -po/pu with -mo/mu (as with $g\varepsilon:m[u]$) or adding -mo/mu to the masculine stem (as with lamu), it is the form $-p\phi n$ (WT $\neg \forall \exists \forall door master'$) which is modified by changing the final nasal.

Many Denjongke nouns are cognate with Written Tibetan nouns ending in -po/pa and -mo/ma. The realizations of these nominal suffixes, however, have three notable features each of which distinguish Denjongke from Lhasa Tibetan. First, the final vowel in the suffix is usually dropped if the root to which the suffix is added ends in a vowel, resulting in monosyllabic words ending in /p/ and /m/, e.g. WT add the suffix is added ends in a vowel, resulting in monosyllabic words ending in /p/ and /m/, e.g. WT add the suffix is retained in Lhasa Tibetan, usually results in -po in Denjongke, e.g. WT add to retain the suffix is retained in Lhasa Tibetan, usually results in -po in Denjongke, e.g. WT add to retain the suffix is gompo 'monastery'. Third, in Denjongke /p/ in the nominalizer is elided when preceded by a velar stop, WT add to retain the suffix lag-pa > D add to retain the stop.

A less frequent nominal suffix is $-k^h a$, which also functions as an adverbializer (see §3.5.2.2). It has derived some nouns from verbs, e.g. $dz\varepsilon$: 'meet (hon.)' > $dz\varepsilon$: $k^h a$ $dz\varepsilon$: 'meeting'. However, $-k^h a$ does not seem synchronically as productive a marker as -po, -m(u), $-k^h \varepsilon$:, -sa and $-t\widetilde{a}$.

3.2.4.3 Nominalizer $-k^{h}\tilde{\epsilon}$

The nominalizer $-k^{h}\tilde{\epsilon}$: can be added to any verb which allows an actor. The combination refers to the person who does the action:

 $p'ja-k^{h}\tilde{\epsilon}$: לאישועק 'doer' $ton-k^{h}\tilde{\epsilon}$: אָשָׁק־שועק' 'the one who shows' $t\tilde{a}$:- $k^{h}\tilde{\epsilon}$: הק-הישועק' 'sender'

The form $-k^{h}\tilde{\epsilon}$ derives from Classical Tibetan Arg *mkhan* 'skilled in' (Beyer 1992: 120). In Classical Tibetan, *-mkhan* may attach to nouns (e.g. *lam* 'road', *lam-mkhan* 'guide'), but in

Denjongke it is postposed to verbs. Although in derivational nominalization, which is a lexical/morphological process, the nominalizer $-k^{h}\tilde{\epsilon}$: expresses the meaning 'the one who does action x', in clausal nominalization $-k^{h}\tilde{\epsilon}$: may express the meaning 'the one which is x-ed'. The use of $-k^{h}\tilde{\epsilon}$: in clausal nominalization is described in §13.2.1 (relative clauses).

3.2.4.4 Nominalizer -sa

The spatial nominalizer -*sa* is quite productive in turning verbs into nouns meaning 'the place of/for x-ing'. The form is homophonous with the noun *sá* 'ground, earth, soil' (WD $\propto sa$), which also occurs in Classical Tibetan and many other Tibetic languages.

zak-sa	শৰ্মা-শ	'place to put something, storage'
dø-sa	<u>َ</u> هٓح ٰ אַ	'place to stay, dwelling'
zu:-sa	ন্ব্ব্ব্ব্ব্য্থ্য জ	'place to stay, dwelling (hon.)'
dzim-sa	শ্র্রিঝঝ'ঝ'	'place to sleep, bedroom (hon.)'
ki-sa	સ્નુંચ'ચ	'place of birth'.

The use of -sa as a clausal nominalizer in relative clauses is described in §13.2.3.

In addition to nominalizing uses where *-sa* attaches to verbs to form nouns or noun phrases, *-sa* may attach to some nouns to form a compound:

3.2.4.5 Nominalizer -tã:

The nominalizer $-t\tilde{a}$, deriving from WT stang 'manner, mode', turns verbs into nouns with the meaning 'the manner of x-ing'.

né:-tã:	শবৃথ্যস্থদথ্য	'situation, condition (lit. dwell-manner)'
k ^h a-lap-tã:	বি'শব'শ্বন্থন্থ	'manner of speaking (lit. mouth-speak-manner)'
p'ja-tã:	র্হারা-শ্রদর্ম	'manner of doing (lit. do-manner)'
t€ ^h a∶-tã:	ক্র্যার্থ:স্ট্র দ্র শ	'composition (lit. be.established-manner)'

3.2.5 Compounding

This section briefly outlines the ways compound nouns are formed in Denjongke. Compound formation processes are very similar to those already described in detail in related languages such as Standard Tibetan (Tournadre & Dorje 2003: 255-257). Compound nouns can be of the forms NOUN + NOUN, NOUN + VERB and VERB + NOUN.

NOUN + NOUN

The semantic relationship of the two nouns that form a compound can be various. For instance, the nouns may be (close to) synonyms, as in (3.6). The use of two similar nouns to form a new noun reveals the preference in Denjongke for disyllabicity in nouns. This preference may be motivated by the need to disambiguate nominal lexemes from potentially homonymous monosyllabic verbal lexemes.

(3.6) lùk-sø: এল্ব্যাইনে 'tradition (lit. tradition-custom)'

The components of a compound may also be opposites of one type or another:

(3.7) $p^{h}a$ -ma, $p^{h}am \leq x$, $z \equiv x$ 'parents (lit. father-mother)'

Very often, the first part of the compound functions as a specifying attribute to the second part, see (3.8).

(3.8)	a) mìk-tc ^h u	र्व्यया'कु'	'tear (lit. eye-water)'
	b) <i>tc^hu-mi?</i>	कु'र्थया'	'spring (lit. water-eye)'
	с) <i>bja-n</i> ɛ?	ਕੁਤੂਂ ਕ੍ਰਤਾ	'illness of the rainy season (lit. summer-illness or rice-illness)'
	d) ádzo-ta?	अ' <u>हे</u> झूया'	'tiger (lit. grandfather-tiger)'
	e) <i>álu-kiu</i>	ષ્ઠા ભુ ગોતુ.	'potato (lit. alu-root [alu is potato in Nepali])'

NOUN + VERB

Second, compounds may be of the form NOUN + VERB, see (3.9).

(3.9)	ám-tɛuŋ	ଷଷ'ନ୍ତ୍⊂'	'mother's younger sister, father's younger brother's wife
			(lit. mother-be.small)'
	ám-bom	ભ્રશ્ન સુંચ	'mother's elder sister, father's elder brother's wife
			(lit. mother-be.big)'
	k ^h im-p ^h ja?	ট্রিঝ:র্দ্রুবা.	'broom (lit. house-sweep)'

VERB + NOUN

Third, a verb may precede the noun in a VERB + NOUN structure, as shown in (3.10).

(3.10) $t^{h}u\eta - tc^{h}u q_{B} = \frac{1}{2}$ 'drinking water (lit. drink-water)'

3.3 Verbs and verbal affixes

This section first introduces defining criteria for verbs ($\S3.3.1$). The introduction is followed by three subsections on verb classification, i.e. stative and dynamic verbs (\$3.3.2), controllable and non-controllable verbs (\$3.3.3), and ordinary, honorific and humilific verbs (\$3.3.4). The next two sections describe the morphological processes of reduplication (\$3.3.5) and verbal affixation (\$3.3.6). The last part provides an introduction to a special case of verbs, the copulas (\$3.3.7).

3.3.1 Defining criteria for verbs

Verbs in Denjongke are words that describe events (e.g. 'to hit'), processes (e.g. 'to walk'), states (e.g. 'to love', 'to be long') and being (e.g. 'is', 'there is'). Basic Denjongke word order is APV (or SOV), and syntactically Denjongke verbs are distinguished from other word classes by their ability to act as the head of the predicate/verb complex, which occurs at the end of the clause. The three major divisions in verbs are stative vs. dynamic verbs, controllable vs. non-controllable verbs and honorific vs. ordinary verbs, see §3.3.2-4.

In his grammar of the Tibeto-Burman language Lahu, Matisoff (1973: 193) uses a criterion for verbhood which, he says, at the time also worked for all studied Tibeto-Burman languages, namely the ability to follow directly the "negative adverb $m\hat{a}$." This criterion also works quite nicely in

Denjongke: it is almost exlusively verbs that can be preceded by the perfective negator *ma*- and imperfective negator *mi*-. The only problem with this definition is that there are a few adjectives derived from stative verbs that may also take the negating prefix, e.g. *ma-tsã:m* 'dirty' (cf. *tsã:po* 'clean') derived from the verb *tsã:* 'be clean' (WD rsc: gtsang).¹⁰⁵ However, if these adjectives, and I have not found many, are seen as essentially nominalized verbs expressing a property concept, the definition holds.

Another potential problem with applying Matisoff's criterion to Denjongke is that the negated forms of copulas, which are here analyzed as verbs, do not use the prefixed *ma*- but have separate negated forms instead, i.e. personal negated equative $m\tilde{\epsilon}$: (cf. positive \tilde{i} :), neutral negated equative $m\tilde{\epsilon}mb\epsilon$ (cf. positive $b\epsilon$?), personal negated existential mere? (cf. positive je?) and sensorial negated existential mere? (cf. positive je?) and sensorial negated existential mere (cf. positive du?). All of these negated copulas can, however, be easily seen as derived from positive constructions supplemented by the negators *ma*- and *mi*-.

Verbs are typically monosyllabic, a fact that distinguishes verbs from adjectives (if the monosyllabic property concept words are considered stative verbs, see §3.4.1), which tend to have two or more syllables (although some exceptions exist), but not from nouns, many of which are monosyllabic. In their base forms, many verbs and monosyllabic nouns that have unrelated meanings are homonymous, e.g. $p'ja \leq n'$ (do (verb)' and $p'ja \leq b'$ (bird (noun)', $ta \in b'$ (look (verb)' and ta = b' (noun)', $ga \leq n' < b'$ (look (verb)' and ga = b' (saddle'/ga = b' (look (verb)' and ta = b' (noun). Therefore, it is only when used in a sentence, in a certain syntactic position and with additional verbal morphology, that some verbs are distinguished from nouns. The presence of two or more syllables in a verb suggest that the verb in question is a phrasal verb (e.g. lo $t\epsilon 2 \leq n' < b' (trust', consisting of lo' (trust', see §4.2.2))$ or a serial verb construction (e.g. bak $\delta : can < b' (can < b' ($

Unlike Written Tibetan and Lhasa Tibetan (see Denwood 1999: 105-108), Denjongke verbs do not have differing stems based on tense, aspect and mood (TAM) values. Verbs are uniform across different TAM values. Exceptions are $gju \, \operatorname{com}^{\circ}$ 'go, walk' with the perfective (past and imperative) form $s\delta$: $\operatorname{com}^{\circ}$ ' $\operatorname{com}^{\circ}$ 'come' with the imperative form $s\delta$? $\operatorname{com}^{\circ}$ ' $\operatorname{com}^{\circ}$ 'come' with the imperative form $s\delta$? $\operatorname{com}^{\circ}$ ' $\operatorname{com}^{\circ}$ 'and ' $\operatorname{com}^{\circ}$ ' $\operatorname{com}^{\circ}$ ' 'and ' $\operatorname{com}^{\circ}$ ' ' $\operatorname{com}^{\circ}$ ' ' $\operatorname{com}^{\circ}$ ' ' $\operatorname{com}^{\circ}$ ' ' $\operatorname{com}^{\circ}$ ' ' $\operatorname{com}^{\circ}$ ' ' $\operatorname{com}^{\circ}$ ' ' \operatorname

Lastly, the verbhood of a word is revealed by the ability to receive exclusively verbal suffixes, which are listed in §3.3.6.

3.3.2 Stative and dynamic verbs

Denjongke verbs may be divided into stative and dynamic. Stative verbs express time-stable qualities (e.g. $ga rac{1}{3}$, 'love, like'), adjective-like property concepts (e.g. rin = r

¹⁰⁵ The usual nominalizing suffix *-po/bo*, as seen in the positive form $ts\tilde{a}:po$ 'clean', is often reduced in spoken language to *-m* when preceded by a nasal vowel/nasal, e.g. $s\tilde{o}:-bo b\epsilon$? [go.PFV-NMLZ EQU]> $som b\epsilon$? (someone) went'. ¹⁰⁶ In Tashiding, West-Sikkim, the invariable verb $j\tilde{a}:$ 'go' is often used instead of gju and $s\tilde{o}:$.

that hold in the present, as in (3.12-14). Note that in (3.14) $j\hat{\epsilon}bb\epsilon^2$ is a reduced pronunction of $j\hat{\phi}$ po $b\epsilon^2$ [EX-2INF EQU.NE].

- (3.11) $\vec{r} \sim \vec{r} \sim \vec{r} = \vec{r} \cdot \vec{r} \cdot$
- (3.12) אָ ה'אָ' קַשְרָא' אָשָן $k^h u \quad \eta a = lo \quad ga - u \quad \tilde{i}$. 3SGM 1SG=DAT like-2INF EQU.PER 'He likes me.' (KT e)
- (3.13) $a\hat{f}' \equiv a \eta' \tilde{\eta}' a\hat{f}' \equiv \hat{f}' \equiv \hat{f}' \equiv \hat{f}' \equiv \hat{f}' \equiv \hat{f}' = \hat{f}' =$
- (3.14) $\exists_{2^{A_{1}}} \tilde{\eta} \cdot \check{s}_{2^{A_{1}}} \tilde{\eta} \cdot$

Stative and eventive verbs also differ in their tendency to occur with certain verbal suffixes. For instance, I did not find naturally occurring examples of stative verbs occurring with the perfect marker $-ts^ha$. In elicitation, however, it became clear that the completive suffix can occur with stative verbs, see (3.15), where the usually stative verbs ga 'like' and $c\dot{e}$: 'know' refer to events.

(3.15) a) $\underset{ja}{\text{even}} \overset{\text{even}}{=} \int \frac{1}{2} \frac{1$

The bare roots of stative verbs without additional marking may be used as predicates describing steady states:

(3.16) $\neg \widetilde{a} \neg \widetilde{a} \neg \widetilde{a} \neg \gamma \alpha \beta$ $\eta a t \varepsilon^h \phi = lo ga.$ 1SG 2SG.L=DAT like 'I like you.' (KN e)

3.3.3 Controllable and non-controllable verbs

Similarly to other Tibetic languages, and also other Tibeto-Burman languages (e.g. Sun 1999, Ding 2014: 118,), many Denjongke verbs form phonetically similar pairs in which one of the verbs describes a non-controllable (or non-volitional) action that happens by itself (e.g. $te^{h}a^{2} \equiv q^{-1}$ 'break [intr.]') and the other verb describes an equivalent controllable (or volitional) action as caused by someone ($tea^{2} \equiv q^{-1}$ 'break [tr.]'). In other Tibetic languages this distinction has been referred to as controllable vs. non-controllable (Shigatse Tibetan and Themchen Tibetan, Haller 2000: 175-176; Dege (Sde.dge) Tibetan, Häsler 1999: 134), transitive vs. intransitive (Donwang Tibetan, Bartee 2007: 122) and causative vs. resultative (Standard Tibetan, Tournadre & Dorje 2003: 352). The terms "transitive" and "intransitive" are infelicitous for Denjongke, because both groups include both transitive and intransitive members, e.g. dzy: aggq 'insert (controllable)' and $ts^{h}y^{2} \notin gq$ ' enter, end up (non-controllable)' are both intransitive¹⁰⁷, and $tsuk \equiv qgq$ 'insert (controllable)' and $suk \equiv qq$ ' insert (non-controllable)' both occur in transitive clauses. Example (5.7) illustrates the transitive use of non-controllable suk 'insert, pierce'.

(3.17) $rac{d}{d}rac{d}rac{d}rac{d}{d}rac{d}rac{d}{d}rac{d}{d}rac{d}{d}rac{d}{d}rac{d}{d}rac{d}rac{d}rac{d}rac{d}{d}rac$

The same Denjongke verbs are also problematic when using terminology from Tournadre & Dorje (2003: 352), who define causative verbs as "both transitive and volitional" and resultative verbs as "usually both intransitive and non-volitional," making a (syntactic) transitivity division between the two verb classes. Therefore I have here adopted the terms "controllable" vs. "non-controllable" verbs. An alternative choice of terminology could have been volitional vs. non-volitional verbs.

Tournadre & Dorje (2003: 352) comment that Classical Tibetan has more than 200 such verb pairs and add that in Central Tibetan the number has been reduced to "a few dozen." Bartee (2007: 122-123) found thirteen such pairs in Dongwang Tibetan. Table 3.8 lists 45 such pairs in Denjongke. The disability to occur in the imperative may be used as a test for non-controllable verbs (Häsler 1999: 134). The words in Table 3.8 are listed according to the place of articulation of the first consonant, beginning from labial and proceeding through alveolar and palatal to velar.

Non-c	ontrolla	ble	Contro	llable	
p'ap	নন.	'come down, descend'	p ^h ap	শ্বন.	'take down, cause to come down'
р'є?	শ্রন্থ'	'explode (intr.)'	gɛ?	শ্রুর্থ,	'cause to explode'
p'ik	নিশ্ব'	'come off'	pik	ন্যব্য'	'take off, remove'
p'ok	র্বন্য'	'be dislocated, removed'	pok	শ্বন্	'dislocate, remove'
p'y?	55.	'be detached'	$p^{h}y^{2}$	КĽ.	'detach'
p'jaŋ	<u> </u>	'hang (intr.)'	pjaŋ	<u> </u>	'hang (tr.)'
ba:	নন্দ্র	'catch fire, burn'	pa:	<u>श्</u> रू.	'set on fire, burn'

Table 3.8. Non-controllable and controllable verb pairs

¹⁰⁷ if intransitivity is defined as the lack of an affected patient-like argument.

bõ:	新 「 ・	'get wet'	bã:	塑丁.	'make wet'
bja:	ଏଥିଣ.	'get stuck'	pja:	দর্ঘ	'stick'
t ^h O.	দুর্ন.	'be scattered'	to:	শ্র্র্টিশ'	'scatter'
t ^h øn	ਬੱਕ੍'	come out	tøn	हेंन.	take out
ťø:	Ť́́́т	become a hole	tø:	नर्हेवा.	make a hole
dy:	લનુત્ર.	'become soft, tame'	ty:	ন্থ্রুহ্য'	'tame, soften'
dzak	≝বাঝ.	'drip'	tsak	₹ग.	'make drip'
ts ^h ik	র্ক্রযাথ্য	'burn (intr.)'	sék	द्येया.	'burn (tr.)'
ts ^h o:	র্দ্ধথ্য	'be cooked'	tso:	নর্স্ <u>ড</u> ি	'cook'
ts ^h y?	هْم:	'enter (non-vol.), end up'	dzy:	AÉU.	'enter'
sùk	नुग.	'be inserted'	tsuk	নর্ন্তৃযা'	'insert'
sớ:	র্হ্রাব্য'	'survive'	só	यार्दे.	'rescue'
lùk	सुया.	'slip out from inside'	ļuk	क्षेत्रा.	'take out from inside'
lồ:	άς.	'stand'	ļõ:	۲	'raise up'
ram	শপ্রধ্য.	'be destroyed'	ram	হ্যঝ'	'destroy'
re:	শ্রুম.	'be torn'	re:	5 ^{A1.}	'tear'
t ^h uk	নেদ্রিযাঝ.	be mixed	ţuk	5्रगुग	mix
f'ip	দ্রীন.	'become polluted'	dip	क्षेत्र'	'pollute'
ťõ:	র্শ্রদক্ষ	'die'	ţõ:	নগ্র্মিদঝ'	'kill (hon.)'
t'up	ন্যুন'	'be fulfilled'	dup	দের্ঘ্রুন:	'achieve'
dok	ন্দ্র্যু.	'become frightened'	tok	ন্র্য্যুন্য'	'frighten'
sàk	ৰেশ্ৰম্য.	'lose taste (of beer)'	cák	निया.	'put taste (to beer)' thicken
cìk	নিয	'be destroyed'	сík	न्दर्मिया.	'destroy'
сù	q.	'melt (intr.)'	ZU	ମ୍ମ୍.	'melt (tr.)'
сỳ:	নেশ	'be moved'	бý:	स्रू.	'move'
tc ^h ak	ळग्	'be(come) broken'	tsak	নন্তব্য'	'break'
tc^he?	ھح'	'be cut off'	tce?	নন্তন্'	'cut off (e.g. rope, road)'
tc'ã:	<u>5</u> 5.	'become trained'	dzã:	Sgr.	'train'
nè:	Fa.	'sleep'	jne:	જ્યુતા.	'put to sleep'
jềŋ	બે⊏જા	'be pacified, entertained'	jếŋ	षाप्पेम्.	'pacify, entertain'
k ^h i:	ଏଥ୍ରିଏ.	'be encircled'	ki	519.	'encircle'
k ^h om	শিষ্ণ	'become dry'	kam	ন্ন্নব্য'	'dry'
k ^h uk	ষ্ট্রেম].	'be drawn, invite oneself'	kuk	रगुया.	'call, invite'
k'ã:	קר.	'be filled'	kã:	翌 万文7.	'fill'
k'õ:	र्वेष्ट्र'	'be bent'	kõ:	مَآج′	'bend'
k'uk	য্য্য্য	'be bowed, be bent'	kuk	रगुंग'	'bend'
			guk	নেযাযা.	'bow, bend oneself' ¹⁰⁸
k ^h ø:	দিন্দ	'boil (intr.)'	kø:	MA.	'boil (tr.)'
gak	নেযাযা.	'stop (intr.)'	kak	त्त्रंग[य].	'stop (tr.)'

Because the term "volitionality" has been mentioned in the discussion above, a note on the difference between Denjongke and Tournadre & Dorje's (2003) "Standard Tibetan" is in order. In

¹⁰⁸ Here the reflexive form guk 'bow, bend oneself' forms a triplet of phonologically and semantically similar verbs along with k'uk 'be bowed, be bent' and kuk 'bend'

Denjongke the verbal morphology after the volitional verb $ta \in 100$ 'look' in ta-u \tilde{t} . 'I looked' is identical with the non-volitional $t^{h}\tilde{o}$: $\exists i \in 1000$ 'see' in $t^{h}\tilde{o}$: bo \tilde{t} : 'I saw' (-bo and -u are allomorphs), whereas the equivalent expressions in Standard Tibetan have differing auxiliaries, voluntary -payin and non-voluntary -cung (Tournadre & Dorje 2003: 141). Hence, unlike on Standard Tibetan, volitionality is not coded in the Denjongke verbal endings.

3.3.4 Ordinary, honorific and humilific verbs

Similarly to Standard Tibetan (Tournadre & Dorje 2003: 447) and many other Tibetic languages, Denjongke uses different verbs to refer to the same situation on different levels of deference. A few actions may be described by three verbs on three different levels related to deference: ordinary, honorific and humilific. Usually, an ordinary verb is used with friends and one's social inferiors. Honorific and humilific verbs are used when talking to and referring to one's elders and social superiors. Using honorific verbs shows deference to the addressee and/or the referent of the clause, and the use of humilific verbs implies the speaker's humility. Humilific forms are rare, only a handful of verbs form triads of ordinary, honorific and humilific forms, see Table 3.9. As seen in Table 3.9, the honorific ze: and humilific $e\hat{u}$ have a wide range of meanings corresponding to several more specific ordinary level verbs.

Ordi	inary	Hono	rific	Hun	nilific	Gloss
sà	T	ze:	শন্ধিম'	сù	લુ.	'eat'
t ^h uŋ	तम्न	ze:	নাৰ্ন্ম'	сù	લુ.	'drink'
lèn	लेवु:	ze:	ଅନ୍ୟି:	сù	લુ.	'receive, accept'
làp	27.	súŋ	यसुद	сù	<u>g</u> .	'say'
p'in	ਉੱਕ.	nấ:	শ্বন্দ'	$p^h y$:	র্মনা.	'give'
ò:	й́г.	tc 'øn	ইন্	tɛa:	বন্তন	'come'

Table 3.9. Ordinary-honorific-humilific triads of verbs

Ordinary-honorific pairs are more frequent than triads presented in Table 3.9, see Table 3.10. Many verbs lack honorific or humilific equivalents. These verbs may be formed into honorific periphrastic constructions VERB-2INF $n\tilde{a}$: with the help of the verb $n\tilde{a}$: $\eta \eta \varsigma$: 'grant, give (hon.)', e.g. $kjap_{\pi} \eta \eta \varsigma$: 'grant, give (hon.)', e.g. $kjap_{\pi} \eta \eta \varsigma$: 'please do, please strike'.

Table 3.10. Some ordinary-honorific pairs of verbs

Ordin	ary	Honori	fic	Gloss
p'in	ईव.	nấ:	শ্বন্দ'	'give'
p'ja	র্হাব্য.	nấ:	শ্বন্দ'	'do'
bak	নন্দ্রা.	nám	বস্থুমন্ম.	'carry'
ta	ar.	zi:	য্রিযাম'	'look'
t ^h uŋ	ನರ್ಷ.	ze:	নাৰ্ন্ম'	'drink'
sà	¤ .	ze:	শ্বন্ধি'	'eat'
nà	ਰ.	núŋ		'become ill'
lồ:	AT.	zã:	নন্দ্ৰম'	'stand'
сé:	नेष.	k ^h ɛn	यामुन.	'know'
сí	م	ťõ:	मॅान्य.	'die'
nè:	નુતા.	zim	শ্রিঝঝ'	'sleep'
pèn	नन्.	sén	শশ্বন	'listen'

ki	ક્ ર્યુચ્પ	(ku)t ^h uŋ	ন্দ্র্দুম.	'be born'
k ^h ju:	র্হিদ্র.	sí:	র্বমিন্দ্র'	'wash'
ga	न्यात.	ge:	न्त्रीयः	'rejoice'
gju	त्मु.	tc 'øn	र्चेन.	ʻgoʻ
ŋù	5	сúт	নন্দুম'	'weep'

In addition to the forms given in Table 3.10, the verbs for eating and drinking also have the hyperhonorific form $te^{h}\phi^{2}$ eat, drink (hyper-honorific)', which may be used, for instance, in the presence of high lamas and royalty.

(3.18) র'র' র' ঝর' র'র' ৸শ

sà-nɛ sà, man-za-nɛ pa?
eat-COND eat NEG-eat-COND eat.AO.HUM
'If you are about to eat, eat. If you are not about to eat, eat (under duress).' (Richhi 65)

¹⁰⁹ Nepali, in which most Denjongke speakers have at least some competence, functions analogically. When requesting someone to eat, using the word *linu* 'take' is considered more polite than khanu '*eat*', e.g. *linuhos* 'Please have (some)' vs. *khanuhos* 'Please eat'.

- (3.19) $\widehat{\neg} \neg \neg \widehat{\neg} \cdot \widetilde{a} \neg \widehat{\neg} \cdot \widetilde{a} \neg \widehat{\neg} \cdot \widehat{a} \neg \widehat{\neg} \cdot \widehat{a} \neg \widehat{g} \widehat{\epsilon}$ $t' \varepsilon p = di$ $k' \widetilde{o} : t \varepsilon h \widetilde{\varepsilon} : ma - j a - g \varepsilon$ l a p - o $\widetilde{t} :.$ book=DEMPH expensive NEG-do-HOR say-2INF EQU.PER
 'I said (to him): 'Let's not make the book expensive.'' (KL BLA 12)
- ສີ້या' गठिया'गढ़िय'ठेया' गर्छे'ह्ये' क्वॅत्र'पदे' क्षु'यत्त्र'यर' त' त'ठया' तठर'ष्ट्रे'गी' खु'र्ये' क्षेत्रा (3.20)ts^hik tci-pí:=tci? tsotci lémpø: kumdynk^ha nàtea? ťa word one-two=INDF chief minister.GEN in.the.presence now 1PL ĩ. tcaː-diki **cù** ː say.HUM. 2INF come.HUM-NF EQU.PER 'Coming to the presence of the Chief Minister we said a few words.' (NAB BLA 7)
- (3.21) \check{p} \check{q} \check{d} \check{d}

3.3.5 Reduplication

Denjongke uses the morphological process of verb root reduplication to mark completion/resultativity, as shown by the perfect construction in (3.22), or iteration/continuity, as shown by the nominalized complement construction in (3.23):

- (3.22) $\alpha \hat{f}^{*} \tilde{\mathfrak{H}}^{ar} \hat{\mathfrak{H}}^{*} \tilde{\mathfrak{H}}^{r} \tilde{\mathfrak{H}}^{$
- (3.23) بِحْمَّ يَعْمَ بَنْ عَلَى اللَّهُ اللَّهُ عَلَى اللَّهُ (He) saw green flies hovering (and hovering over the corpse). (KT animal story)

Reduplication by itself may function as a nominalized construction, as shown by (3.24), where a postposition is postposed to a reduplicated verb stem, and by (3.25), where the reduplicated verb marks a complement clause functioning as an argument of the verb $n \dot{\epsilon} n$ 'listen'.¹¹⁰

(3.24) $\operatorname{cam}^{\operatorname{am}} \mathfrak{F}^{\operatorname{am}} \mathfrak{F}^{\operatorname$

¹¹⁰ With nouns, reduplication marks iteration, see §4.1.8.

(3.25) الإنتهام، حريق (Nep.) عَمَامَة، حَرَّ اللَّذَي اللَّهُ عَمَامَةً، اللَّهُ عَمَامَةً، حَمَّ اللَّهُ عَمَامَةً، اللَّهُ عَمامَةً، اللَّعْمَامَةُ عَمامَةً، اللَّهُ عَمامَةً، اللَّهُ عَمامَةً، اللَّهُ عَمامَةً، المَامَةُ عَمامَةً، المَامَةُ عَمامَةً، المَامَةُ عَمامَةً، المَامَةُ عَمامَةً، اللَّهُ عَمامَةً، اللَّهُ عَمامَةً، اللَّهُ عَمامَةً، اللَّهُ عَمامَةً، المَامَةُ عَمامَةً، المَامَةُ عَمامَةً، المَامَةُ عَمامَةً، عَمامَةً، المَامَةُ عَمامَةً، اللَّهُ عَمامَةً، اللَّهُ عَمامَةً، اللَّالَةُ عَمامَةً، المَامَةُ عَمامَةًا عَمامَةً، ومَا عَمامَةُ عَمامَةً، عَمامَةُ عَمامَةً، ومعامَّةُ عَمامَةًا عَمامَةًا عَمامَةًا عَمامَةًا عَمامَةًا عَمامَةً مَامَةًا عَمامَةًا مَامَةًا عَمامَةًا مُعَامَةًا مُعَامَةً، ومع مَامَةًا عَمامَةُ مَامَةًا مُعَمَامَةً، ومع مَامَةُ عَمامَةًا عَمامَةًا عَمامَةًا مَا عَمامَةً مَامَةًا عَمامَةً عَمامَةًا عَمامَةً مَامَةًا عَمامَةً مَامَةً مَامَةًا عَمامَةً مَامَةًا مَامَةً مَامَةً مَامَةًا مُعَامَةً مَامَةً م المَامَامُ اللَّالِي اللَّالِي اللَّالِي عَلَيْ عَلَيْمَامَةًا عَلَيْ عَمامَةًا عَلَيْكَامَةُ عَمامَةً عَلَيْ عَ المَامَةُ عَمامَةُ عَمامَةً عَلَيْ عَلَيْ عَلَيْ عَلَيْ عَلَيْ عَلَيْ عَلَيْ عَلَي عَلَيْ عَلَيْ عَلَيْ عَلَي ع

The various constructions using reduplication, along with non-reduplicating constructions, are discussed more in detail in §8.

3.3.6 Verbal affixes

Verbal affixes in Denjongke consist of the negator prefixes *ma*- (perfective) and *mi*- (imperfective) and several inflectional suffixes listed in Table 3.11, where the affixes are grouped according to their distribution in the verb complex.¹¹¹ The first group of suffixes in Table 3.11 consists of forms that participate in finite constructions which can end a sentence (hence the full stop in the left-most column). The left-most column shows in which type of construction the suffix typically occurs, signalling the placing of the suffix by _, e.g. VERB-_ EQU refers to a syntagm such as *kjap-ce be?* [do-INF EQU.NE] 'does, will do'. Some suffixes, such as -ce? and $-tce\tilde{e}:/z\tilde{e}:$ respectively, form finite constructions with an equative (EQU) or an existential auxiliary (EX). Others, such as ts^ha :, can finish a finite sentence with or without an equative auxiliary, while still others, such as -tce, occur exclusively without an auxiliary.

The second group of suffixes consists of exclusively nonfinal forms, which cannot end a sentence by themselves (hence the three dots ... in the left-most column). These suffixes attach directly to the verb root (marked V) and are not followed by any other suffixes listed in Table 3.11.¹¹² Note that secondary verbs¹¹³, which may also mark aspectual and modal information (see §4.2.3), are not listed in Table 3.11. The column on the right refers to sections where the use of the suffix is illustrated in a specific construction.

¹¹¹ The copulas are the least verb-like with respect to suffixes. The copulas do not typically occur with the perfect, past, progressive or nonfinal converb suffixes, and non-personal copulas $b\epsilon^2$ and du^2 only occur with the polar question suffix.

¹¹² However, emphatic clitics such as $=r\tilde{a}$:, =di and $=j\tilde{a}$: (see §16.1) may follow at least some of these forms. Moreover, the dative-locative =lo and ablative $=l\varepsilon$ may follow the circumstantial-purposive converb -pa/ba and progressive $-t\varepsilon\tilde{\varepsilon}:/z\tilde{\varepsilon}$: when it functions as an adverbial clause marker.

¹¹³ The word "secondary" refers to the secondary semantic effect that the secondary verbs have in a verb complex in comparison to the primary verbs, see §4.2.3 for details.

Suffixes that participate in finite constructions ¹¹⁴					
VEQU.	-ce(?)	infinitive I	§8.2.5, §8.4, §13.4.2, §13.5.2,		
			§14.1.2		
	-po	infinitive II	§8.1.1, §8.1.4, §8.1.8, §13.2.2		
	-ni	infinitive III	§8.2.8		
	$-k^h \tilde{e}$:	nominalizer	§8.2.3		
V(EQU).	-ts ^h a:	completive	§8.1.2, §9.1.3		
	-to/do	imperfective	§8.3.1		
VEX.	-t $arepsilonarepsilon :/ arepsilon arepsilon :/ arepsilon arepsilon :/ arepsilon arepsilon :/ arepsilon :: arepsilon :/ arepsilon $	progressive	§8.3.3		
VEX/EQU.	-rap	imminent future	§8.2.7		
V	-tGE/ZE	past	§8.1.1		
	-to	probabilitative	§8.5.1		
	-tc ^h i	friendly imperative	§11.3.2		
	-da	friendly imperative	§11.3.2		
	-na	suggestive	§11.3.2		
	-kɛ/gɛ	hortative	§11.4		
	-ka/ga	polar interrogative	§11.1.1.2		
	-kam/gam	attenuated interrogative	§11.1.1.3		
	-lø?	exclamative	§11.1.3.4, §11.2.1		
Exclusively	nonfinal suffixes				
V	-ti/di	nonfinal converb	§15.2		
	-pa/ba ¹¹⁶	circumstantial/purposive	§15.5.1, §15.8.1		
			e15.c		
	-(patse)ne	conditional converb	§15.0		
	-ruŋ	concessive converb	§15.7		
	-sal, -sonzal	terminative converb	§15.12		
	-sondă:/sumdă:	simultaneous converb	§15.3.3.2		
	-kap	simultaneous converb	§15.3.3.4		
	$-d\tilde{y}$:	simultaneous converb	§15.3.3.5		
	-rɛŋkʰa	simultaneous converb	§15.3.3.6		

While the function of the verb suffixes in Table 3.11 is illustrated in later chapters (see references within the table), the following subsection provides etymological and comparative morphological information of the forms, along with introductory examples. The verb suffixes are discussed in the same order they occur in Table 3.11.

3.3.6.1 Infinitives -ce?, -po/bo and -pi

The present analysis posits three infinitive forms in Denjongke. As was already pointed out in §3.2.3, the term "infinitive" is applied to nominalizing suffixes which refer to the verbal action

¹¹⁴ Some of these forms also participate in non-finite constructions, e.g. the infinitive $-\epsilon\epsilon^2$ may form complement clauses, see §14.1.2, and the progressive $-t\epsilon\tilde{\epsilon}:/z\tilde{\epsilon}:$ adverbial clauses of manner, see §15.8.3.

¹¹⁵ The form *zin* is the reading-style pronunciation used by literate speakers, $te\tilde{\epsilon}$: and $z\tilde{\epsilon}$: are spoken variants. In Martam (East Sikkim) *teou* is used instead of $te\tilde{\epsilon}$:/ $z\tilde{\epsilon}$:.

¹¹⁶ This form has developed some finite-looking uses, see §15.8.1.

itself. The nominalizing suffixes marking more noun-like concepts, i.e. person $(-k^{h}\tilde{\epsilon})$, place (-sa) or manner of doing $(-t\tilde{a})$, are called by the general term nominalizer. Of the infinitive markers, $-\epsilon\epsilon^{2}$ and -po are more central and productive, whereas -ni is more marginal and semantically close to $-\epsilon\epsilon^{2}$.¹¹⁷

The difference between the two main infinitive markers $-\varepsilon\epsilon^2$ and -po/bo may be described in terms of spatiotemporal boundedness. Whereas verbal action marked by -po/bo can be characterized as spatiotemporally bounded, $-\varepsilon\epsilon^2$ marks unbounded action. Because the infinitive marked by $-\varepsilon\epsilon^2$ is unbounded, it is used as a copula subject which refers to action in general. Unboundedness or open-endedness of the infinitive marked by $-\varepsilon\epsilon^2$ is reflected in the fact that when followed by an auxiliary copula the construction with $-\varepsilon\epsilon^2$ results in a future meaning (e.g. $\tilde{o}-\varepsilon\epsilon$ \tilde{t} : [come-INF EQU.PER] 'is coming/will come'). The bounded infinitive marked by -po/bo, on the other hand, typically refers to a specific action. It occurs in constructions with an auxiliary and in complement clauses. When followed by an auxiliary copula, the construction obtains a past meaning (∂m -bo \tilde{t} : [come-2INF EQU.PER] 'came'). The form V- $\varepsilon\epsilon^2$ is preferable to V-po/bo as a dictionary form because many nouns end in -po. Therefore it would be difficult to distinguish verb forms ending in -po from nouns by formal criteria. The form $-\varepsilon\epsilon^2$, in contrast, is a distinctly verbal marker.

While the infinitive marker -po/bo has an etymon in the Classical Tibetan nominalizer -pa/ba (Beyer 1992: 299), the origin of the form $-\varepsilon\epsilon^{2}$ (WD $\operatorname{sp}^{-} shad$) is unknown. The infinitive marker $-\varepsilon\epsilon^{2}$ is used in the finite nonpast construction (§8.2.5) and several non-finite constructions, i.e. noun complement clauses (§13.4.2), postposition complement clauses (§13.5.2), complement clauses (§14.1.2) and in the construction VERB-INF EX (§8.4). In addition, the infinitive $-\varepsilon\epsilon^{2}$ also occurs in short questions such as $k'an p'ja-\varepsilon\epsilon^{2}$ [what do-INF] 'what to do?', $k'ana gju-\varepsilon\epsilon^{2}$? [where go-INF] 'where to go?'.

The infinitive marker -po/bo occurs in a host of both finite and non-finite constructions. In finite constructions, it occurs in the periphrastic past (§8.1.1), perfect (§8.1.4) and iterative past constructions (§8.1.8). It is the default marker for clausal nominalization of action and is used in relative clauses (§13.2.2), correlative clauses (§13.3), noun complement clauses (§13.4.1), postposition complement clauses (§13.5.1), complement clauses (§14.1.1) and in various adverbial clause constructions (§15). The uses of $-c\epsilon^2$ and -po/bo are illustrated in (3.26), which exemplify the present habitual and periphrastic past constructions respectively. The first infinitive $-c\epsilon^2$ is glossed as simply INF, the second infinitive -po/bo as 2INF and the third infinitive -pi as 3INF.

¹¹⁷ Similarly to the analysis here, Zemp (2018: 435) applies the term "infinitive" to verb forms marked by *-pa*, which has similar functions to Denjongke *-po*. Yukawa (2017: 210ff), on the other hand, posits five infinitives marked by $\neg r$, \overline{g} , \overline{gyi} , \overline{gyi} , \overline{gyi} , rgyu, \overline{gv} *rtsis* and $\alpha \overline{z} \overline{\gamma}$ 'dod in Lhasa Tibetan.

b)
 שׁ הַכְּהָשׁ הַכָּהַ בָּשִּ בַּשִׁ בַּשְׁ בַשְׁ בַשְׁ בַשְׁ בַּשְׁ בַשְׁ בַשְׁבָשָׁ בַשְׁ בַשְׁבָי בַשְׁ בַשְׁבַשָּׁב בַשְׁבָשָׁב בַשְׁבַשָּׁב בַשְׁבַשָּׁב בַשְׁבַשָּׁב בַשְׁבַשָּׁב בַשְׁבַש בַשְׁבַש בַשְׁבָש בַשְׁבַי בַשְׁב בַעוּ בַעוּב בַעוּב בַשְׁב בַשְׁב בַשְׁב בַשְׁב בַעוּי בַשְׁב בַשְׁב בַשְׁב בַשְׁב בַעוּב בַשְׁב בַעוּי בַשְׁב בַשְׁי בַשְׁב בַשְׁב בַשְׁב בַשְׁב בַשְׁב בעוּב בעוּב בעוּב בעוּב בעוּב בעוּשַעוּי בעוּב בעוּשַעוּ בעוּב בעוּב בעוּב בעוּשַ בעוּשַי בעוּב בעוּב בעוּשַי בעוּשַי בעוּב בעוּשַי בעוּב בעוּשַי בעוּשַי בעוּב בעוּי בעוּב בעוּיש בעוּיי בעוּייי בעוּיי בעוּיי בעוּיי בעוּיי בעוּייי בעוּיי בעוּיי בעוּיי בעוּיי בעוּייי בעוּיי בעוּיי בעוּיי בעוּייי בעוּייי בעוּישַייי בעוּייי בעוּייי בעוּישַיי בעוּיי ב

The uses of infinitive III (marked with *-ni*, written \widehat{g} *nye*) somewhat overlap with those of infinitive I (marked with *-ce[2]*). Sandberg (1895: 40) reports two infinitive forms *-she* (*-ce2*) and *-nyi* (*-ni*) for Denjongke and comments that the former is used in Denjongke spoken in Sikkim and the Tibetan variety spoken in the Tsang region of Tibet¹¹⁸, whereas the latter is used in Denjongke spoken in the Darjeeling district. Sandberg (1895: 40) gives $t^h u \eta$ -*ni* \widehat{t} : as an alternative for $t^h u \eta$ -*ce* \widehat{t} : '(I) shall drink'. In my data, which is from Sikkim, *-ni* does not occur as a regular alternative to *-ce(2)* but it does occur in quite a few fixed expressions, sometimes overlapping with *-ce(2)*. For an example of a specific construction where *-ni* is used, consider (3.27).

(3.27) 天 只要下资: 受知 花 說到
 tc 'a t^huŋ-**ni** p'ja-u 花:
 tea drink-INF do-2INF EQU.PER
 'I was about to drink tea./I attempted to drink tea/I would like to drink tea.' (TB e)

For a more detailed discussion on the various uses of *-µi*, refer to §8.2.8.

3.3.6.2 Nominalizer $-k^{h}\tilde{\epsilon}$:

The nominalizer $-k^{h}\tilde{\varepsilon}$ is related to Classical Tibetan $\overline{s_{NS}}$ mkkan, which functions both as a noun suffix meaning 'skilled in' (Beyer 1992: 120) and a nominalizer meaning 'person involved in proposition' (Beyer 1992: 301). Cognates of $-k^{h}\tilde{\varepsilon}$ are found in both Tibetic (e.g. Lhomi, see Vesalainen 2016: 224; Dege Tibetan, see Häsler 1999: 240) and non-Tibetic Himalayan languages (e.g. Tshangla, see Andvik 2010: 238). In Denjongke, the basic function of the nominalizer $-k^{h}\tilde{\varepsilon}$ is to derive from a verb a nominal that refers to the person who is the agent of the verb (quite similarly to English *-er* in sow > sow-er), see §3.2.4.3. However, $-k^{h}\tilde{\varepsilon}$ also occurs as part of the present habitual auxiliary construction:

(3.28) $\exists x^{n} e^{ix^{n}} e^{ix^{n}} = \forall e^{ix^{n}} e$

For more examples of $-k^{h}\tilde{\varepsilon}$: in the present habitual use, see §8.2.3

3.3.6.3 Completive -*tsha(:*)

The completive form $-ts^ha(:)$ (written $\mathfrak{s}_{\overline{x}}$ tshar) derives from the Classical Tibetan verb $\mathfrak{s}_{\overline{x}}$ tshar 'complete' and denotes a completed action. Cognates of $-ts^ha$: are used in a sense similar to Denjongke in other Tibetic languages, such as Lhomi (Vesalainen 2016: 222), Dzongkha (Watters

¹¹⁸ Sandberg (1895: 12) reports Sikkimese Bhutias (=Denjongpos/Lhopos) to have originally come from the Tsang region in Tibet.

2018: 258) and Lhasa Tibetan (Denwood 1999: 174). The completive attaches directly to the verb root and can finish a sentence, see (3.29).

(3.29) ה. مالَنَّسْمَا عَلَى اللَّهُ اللَّهُ عَلَى الْحَامَ عَلَى اللَّهُ عَلَى اللَّالَ عَلَى الْحَامَةُ عَلَى الْحَامِ عَلَى الْحَامَةُ عَلَى الْحَامَةُ عَلَى الْحَامَةُ عَلَى الْحَامَةُ عَلَى الْحَامِ عَلَى الْحَامِ عَلَى الْحَامِ عَلَى الْحَامَ عَلَى الْحَامَ عَلَى الْحَالَةُ عَلَى اللَّهُ عَلَى عَلَى الْحَامَ عَلَى الَ

The completive $-ts^ha(:)$ still retains some regular verbal qualities such as the ability to be nominalized with -po/bo (resulting in the form $-ts^hou$ in some constructions), although it does not function as a fully productive verb. For a more detailed treatment, see §8.1.2 (completive construction), and §9.1.3 (evidentiality).

3.3.6.4 Imperfective -to/do

The imperfective marker -to/do (written $\tilde{\gamma}/\tilde{\gamma}$ to/do) attaches directly to the verb root. A cognate progressive form -do is found in Dzongkha (Watters 2018: 464). The imperfective form of a verb can end a sentence either by itself or in conjunction with an equative copula (VERB-to $\tilde{t}:/b\epsilon$?). The range of uses of the imperfective covers past habitual and progressive, present habitual and progressive, and immediate future. For a present habitual example, see (3.30).

(3.30) جَعَرَ حَرَّ مَعْرَ عَلَى اللَّهُ عَلَى اللَّالَ عَلَى اللَّهُ عَلَى اللَّهُ عَلَى اللَّهُ عَلَى اللَّالِي عَلَى اللَّ

For the main discussion on -to/do, refer to §8.3.1.

3.3.6.5 Progressive -tcɛ̃:/zɛ̃:/zin (also -tcou/zou)

This form derives from WD/WT schat bzhin, which according to Jäschke (1881: 483) has the meanings 'face, countenance', 'agreeably, in conformity, according to' and 'like, as'. Jäschke (1881: 483) also notes that schat bzhin may mark present participle and schat bzhin-du gerund. The reading-style pronunciation of the form in Denjongke is zin, a form which also occurs in the spoken language of literate speakers. The forms $-tc\tilde{e}:/z\tilde{e}:$ are allomorphs, whereas -tcou is the variant used in the village of Martam (and possibly also in other villages). The progressive suffix is used in conjunction with existential copulas in periphrastic finite progressive constructions, as shown in (3.31).

(3.31) $\tilde{\alpha}$ ς ς $\tilde{\gamma}$ $\tilde{\gamma}$

'She/he is reading a book now (I see).' (KN e)

Phonetically, $-t\varepsilon\tilde{\epsilon}:/z\tilde{\epsilon}:$ is realised as $-t\varepsilon\epsilon n/z\epsilon n$ when followed by a dental (i.e. *kjap-t\varepsilon\epsilonn du?*). For more examples, consider §8.3.3 (progressive construction), and §15.8.3 (adverbial clause). The

progressive form differs from English *-ing* in that the Denjongke form, unlike the English form, does not occur as a nominal modifier (e.g. dying man).

3.3.6.6 Imminent future marker *-rap*

The imminent future marker *-rap* is cognate with WT spaw grabs 'preparation, arrangements', which is used quite similarly to Denjongke in Jäschke's (1881) dictionary entry. It co-occurs with either an equative or an existential copula and marks something that, in the speaker's opinion, is just about to happen in the immediate future.

(3.32) B' AB' AT $arg k^{h}u$ gju-**rap** be?.

3SGM go-IMF EQU.NE 'He's about to go.' (KN e)

For a more detailed treatment, refer to §8.2.7.

3.3.6.7 Past -tce/ze

The past marker $-t\varepsilon \epsilon/z\epsilon$ (written $\overline{s}/\overline{a}$ ce/zhe) likely derives from WT \overline{s} by as 'do'. Sandberg (1895: 42) reports the forms "zhe and che" as past forms of the verb 'do':

(3.33) (Sandberg 1895: 42) pya-she 'to do' zhe or che 'did' zhe song 'has done'

A similar "witnessed past" form $\hat{s} ci$ (with the allomorph $\hat{a} yi$) occurs in Dzongkha (van Driem 1998: 267). The past suffix $-t\epsilon \epsilon/z\epsilon$ is a final marker which cannot be followed by other verbal markers (except the interrogative) or auxiliaries. In my data, $-t\epsilon \epsilon/z\epsilon$ occurs only as a past marker, not as a past tense form of the verb p'ja 'do', see (3.34).

(3.34) حمر ماكلام على المرابع المرابع (3.34) *nà: jó? p'ja-ze*. 1SG.AGT work do-PST 'I worked.' (KN e)

For a more detailed treatment of the past marker, refer to §8.1.1.

3.3.6.8 Probabilitative -to

The probabilitative *-to* derives from Classical Tibetan ch 'gro 'go', reflexes of which are used as a marker similar to Denjongke probabilitative in many Tibetic languages, for instance Standard Tibetan (Tournadre & Dorje 2003: 236), Lhomi (Vesalainen 2016: 203) and Kyirong Tibetan (Huber 2002: 188). Written forms of the probabilitative that I have seen used in written Denjongke are ch gro, ch pro (Richhi) and ch kro, the last of which is used in the present work to represent

spoken examples.¹¹⁹ The probabilitative marker attaches directly to the verb root, constructing a final verb form which signals that the speaker considers it possible or probable that the proposition is true, i.e. the meaning ranges from 'maybe' to 'probably'.

(3.35) $\widetilde{\mathfrak{s}}_{5}$, 5, $\widetilde{\mathfrak{s}}_{5}$, $\widetilde{\mathfrak{s}}_{7}$, $\widetilde{\mathfrak{s}}_{7}$, $\widetilde{\mathfrak{s}}_{6}$, $\widetilde{\mathfrak{s}}_{7}$, $\widetilde{\mathfrak{s}}_{6}$, $\widetilde{\mathfrak{s}}_{7}$, $\widetilde{\mathfrak{s}}$

For more examples, refer to §8.5.1.

3.3.6.9 Imperative suffixes -tchi, -da and -na

The friendly imperative forms $-tc^{hi}$ (WD $\cong chi$, although the phonetically less accurate form $\boxtimes_{\neg} cig$ from Central Tibetan is used by many authors) and -da (WD $\neg da$), and the suggestive -na (WD $\neg a$) attach to the verb root. The two first ones have cognates WT $\exists_{\neg \neg} shig$ (phon. -ci) and WT $\neg \neg dang$ (phon. -ta) in Standard Tibetan (Tournadre & Dorje 2003: 244). Denwood (1999: 168) calls these two markers "friendly imperatives," a term which is also adopted here because the use is similar.

- (3.36) a) $\operatorname{sgr}(\operatorname{soup}) \operatorname{rade}(\check{z}' \operatorname{rade})$ su:p ze:-po $n\check{a}:-te^{h}i.$ $\operatorname{soup}(\operatorname{Eng.})$ have-2INF do.HON-IMP.FRN 'Please, have (some) soup.' (KT discussion with TB)

Several consultants have assured me that one of the friendly imperative markers is indeed $-tc^{hi}$ and not -tci. Figure 3.1. provides some evidence, although the experiment was somewhat artificial. Consultant KUN was asked to contrast the nonsensical p'ja-tci with the imperative form p'ja- tc^{hi} . The wave forms of the two forms are given in Figure 3.1.





¹¹⁹ The form $\underline{\tilde{\mathfrak{T}}}$ kro is preferable to $\underline{\tilde{\mathfrak{T}}}$ gro, because the latter would word-medially typically result in voiced pronunciation. The form $\underline{\tilde{\mathfrak{T}}}$ kro is also preferable to $\underline{\tilde{\mathfrak{T}}}$ pro, because the former retains the velar place of articulation suggested by the etymon $\alpha \underline{\tilde{\mathfrak{T}}}$ gro.

As seen in Figure 3.1, the prevocalic affrication is more prominent in p'ja- tc^hi than in p'ja-tci. I suspect that sometimes, although not in Figure 3.1, the difference to -tci is rather voicelessness, tcj. At present, the form is written $-tc^hi$.

The suggestive -na is probably an old conditional form, which, although typically nowadays pronounced as -ne/ni (West Sikkim) or -no/nu (East and North Sikkim), is still often written as WD $a_i \cdot na$ and also pronounced -na by some literate speakers as na. Denwood (1999: 168) calls an analogous form in Lhasa Tibetan a "suggestive particle," translated as "what if," thus suggesting a relationship with the conditional form. Interestingly, the form -na is also used for making a request more polite in Nepali, a language in which most Denjongke speakers are at least to some degree bilingual. I follow Denwood (1999: 168) in naming -na a suggestive suffix, which softens the tone of the imperative. The term "honorific imperative," which I first considered as a term, is not appropriate for -na, because the presence of honorifics requires the use of other honorifics in the same clause (i.e. the use of an honorific verb would in good style require also using an honorific noun), but -na may attach to both honorific and ordinary verb stems, see the use with an ordinary verb in (3.37).

(3.37) هرسر شراع کې مین م màla? lòkti gju-**na** quick again go-SUG 'What if you go quickly back...' (Nga'i 'gan 9-10)

For a more detailed description of imperatives, see §11.3.2.

3.3.6.10 Hortative -*k*ε/gε

The hortative form $-k\epsilon/g\epsilon$, which is used for first person singular ('let me') and plural ('let us') exhortation, is cognate with a similarly functioning morpheme in neighbouring Dzongkha (van Driem 1998: 235).

(3.38) rightarrow range and range

For a more detailed description and more examples, refer to §11.4.

3.3.6.11 Interrogative suffixes -ka/ga and -kam/gam

While the origin of the polar interrogative suffix -ka/ga is not known to me, the attenuated form -kam/gam seems related to the Classical Tibetan polar question marker -am (Beyer 1992: 357). The polar interrogative marker -ka/ga and the attenuated interrogative -kam/gam can attach either to the verb root or one of the suffixes -to/do (imperfective), $-tc\epsilon/z\epsilon$ (past) or $-c\epsilon$ (2) (infinitive).

this road=DEMPH TPN arrive-ATTQ 'Does this road arrive in Siliguri, I wonder?' (KN e)

For the full discussion, see §11.1.

3.3.6.12 Exclamative -*lø(?*)

The marker $-l\phi^2$ can attach to (monosyllabic) stative verbs. It is the reflex of Written Tibetan $\breve{x}_{N'}$ los, for which Jäschke (1881: 554) gives the meaning 'in truth, indeed" and provides the following example: *mgon-skyabs rang-los yin* 'He is indeed the helper.' In Standard Tibetan $-l\phi^2$ (WD $\breve{x}_{N'}$ los) forms a how-question attached to monosyllabic adjectives/verbs (Tournadre & Dorje 2003: 230), but in Denjongke the form denotes an exclamative rhetorical question, see (3.40).

(3.40) \P_{B}^{B} \P_{B}^{B} \P_{B}^{B} $\stackrel{\text{A}}{\text{B}}$ $\stackrel{\text{A}}{\text{B}}$

For more information on -lø?, see §11.1.3.4 (alternative questions) and §11.2.1 (exclamatives).

3.3.6.13 Nonfinal converb -ti/di

The nonfinal converb -ti/di (written $\frac{1}{2}$, $\frac{1}{2}$, sti/sdi) is cognate with the form $\frac{1}{2}$, $\frac{1}{2}$, ste/te/de (pronounced -di) in Dzongkha (van Driem 1998: 296) and bears functional similarity to Lhasa Tibetan subordination marker -ni ($\frac{1}{2}$, nas) (Denwood 1999: 221). In the most basic sense, the nonfinal converb shows that the sentence is not finished (hence the term "nonfinal"). It typically implies anteriority but the temporal relationship of the actions denoted by the nonfinal and final verbs is determined by the verb and other contextual factors. For an example, refer to (3.41).

(3.41) $\vec{\beta} \cdot a\vec{\beta} \vec{\alpha} \cdot \vec{\alpha} \vec{n} \cdot \vec{\beta} \cdot \vec{n} \cdot \vec{n} \cdot \vec{n} \cdot \vec{\beta} \cdot \vec{n} \cdot$

For more information on the nonfinal marker, refer to §15.2.

¹²⁰ This word is used specifically for the cleaning done for the annual New Year's festival Losung.

¹²¹ The voicing/voicelessness of -ti/di following a velar nasal (usually realized as a nasalized vowel) has to be learnt on a case by case basis, for instance $t^hu\eta$ -di 'drinking', $s\delta$:-di 'going', ∂n -di 'coming', riŋdi 'becoming long', but mjoŋti/mjõ:-ti 'finishing', $te^h\delta$:-ti 'rushing' (see also §2.8.1).

3.3.6.14 Circumstantial-purposive converb -pa/ba

The circumstantial-purposive converb *-pa/ba* marks an attendant circumstance to the verb in the main clause, see (3.42).¹²² Mainly with directional verbs (e.g. 'come', 'go'), but also in other specific contexts, the form is interpreted as expressing purpose, see (3.43). Purposive uses are typically affirmative, whereas circumstantial uses are typically negated, although circumstantial uses also occur in the affirmative. In the glossing, circumstantial (negated) uses are marked CIRC and purposive (affirmed) uses PUR. The written Denjongke form of *-pa/ba* is $r_1(r_1)/r_2(r_2) pa(r)/ba(r)$, which is a reflex of the Classical/Written Tibetan nominalizer *-pa* followed by the (optional) locative case marker *-r*.

- (3.43) הישמי מוֹקאינוֹי בְּלִישׁמְאִי פִּיַיִדִּי השבּינוֹי שִּלְישׁמְאַי פָיַרִדִי השבּינוֹי שּׁק *nàtea? ní:-po bhaila ta-wa tea:-bo í*:. 1PL two-COLPN watch-PUR come.HUM-2INF EQU.PER 'The two of us came to see Bhaila.' (Richhi 11)

The circumstantial form may be followed by dative-locative =lo, repeating a strategy which was already used historically (-*r*) but had become inconspicuous. Currently, as -*r* is not recognized as a locative anymore, a new locative marker =lo may be added. For more examples on the adverbial use, see §15.5.1 (purposive) and §15.8.1 (circumstantial). The circumstantial -pa/ba also occurs as a final marker, the use of which I do not fully understand, see the latter part of §15.8.1.

3.3.6.15 Conditional converb -(patce)ne/(batce)ne

¹²² The North Western Tibetic language Purik (Jammu and Kashmir) also uses -pa for purposive clauses (Zemp 2018: 441). In Purik, however, -pa (called "infinitive") has a wider range of uses than the Denjongke purposive marker, covering some of the uses similar to Denjongke infinitive -po/bo.

¹²³ The form used in the novel Richhi is $\overline{\eta}$ *na*. Sandberg (1895: 56) reports the spoken *-ne* and *-nu* and literary *-na*. The conditional form *-na* is used in many Tibetic languages, such as Dege Tibetan (Häsler 1999: 250) and Lhomi (Vesalainen 2016: 250).

(patee)ne/(batee)ne may be accompanied by the non-mandatory clause-initial k'e:si? 'if'. For an example of the conditional, consider (3.44).

(3.44) দান্তবা বা নী শামাৰ্য দেই মন্তব্য ন্তব্ব *nàtea? nà: mí ke:po dzom-bateene* 1PL here human a.lot gather-COND 'If we gather here as many people...' (NAB BLA 7)

For the main discussion on conditional clauses, see §15.6.

3.3.6.16 Concessive converb -ruy

The concessive converb *-run* (written 35^{-rung}) attaches directly to the verb stem to form subordinate clauses with the meaning 'although, even if', see (3.45).

(3.45) حَقَّتْ مَعْنَ مَجْتَمَ مَجْتَمَ مَجْتَمَ مَجْتَمَ مَجْتَمَ مَجْتَمَ مَجْتَمَ مَجْتَمَ مَجْتَمَ مَعْنَ مَجْتَمَ مَعْنَ مَجْتَمَ مَعْنَ مَجْتَمَ مُعْنَ مَجْتَمَ مُعْنَ مَجْتَم مُعْنَ مُعْن مُعْعْ مُعْن مُعْن مُعْن مُعْن مُعْن مُع

Neighbouring language Dzongkha has a cognate concessive form -ru (van Driem 1998: 308). A more detailed treatment of Denjongke concessive clauses follows in §15.7.

3.3.6.17 Terminative converbs -sã: and -sonzã:

The terminative converbs $-s\tilde{a}$: and $-sonz\tilde{a}$: both probably derive from the terminative postposition $=s\tilde{a}$: (WD $\equiv r : zang$).¹²⁴ The first part of $-sonz\tilde{a}$: is probably historically the secondary verb WT song $\approx r : go'$ (the sibilant in $=s\tilde{a}$: becomes voiced after $s\tilde{a}$:). The terminative converb obtains a variety of meanings ranging from terminative to simultaneous to causal, the last of which is illustrated in (3.46). For more examples, see §15.12.

(3.46) هَٰ مَحْجَمَّ مَوْمَا مَّرْجَا مَّحْ حَجَّ *odem lùksø: jð:-sonzã:* like.that custom EX-TERM 'Since there is such a custom...' (SGD wedding customs)

In my data, only the short form $-s\tilde{a}$: (WD = zang) occurs in writing, whereas the spoken language uses both forms $-s\tilde{a}$: and $-sonz\tilde{a}$:.

3.3.6.18 Simultaneous converb -sondã:/somdã:/sumdã:/tsubda:

The simultaneous converbal endings *-sondã:/somdã:/sumdã:/tsubda:* do not occur in written language. My current hypothesis is that the forms derive from the nominalized/infinitivized form of the secondary verb $s\tilde{\delta}$: 'go (past)' followed by the conjunct $t'\tilde{a}$: 'and', $s\tilde{\delta}$:-bo $t'\tilde{a}$: [go.PFV-2INF and], a type of simultaneous construction that occurs in both written and spoken Denjongke, see §15.3.3.1. The two forms *-somdã:~sumdã*: arise from the reduction of the nominalizer *-po/bo* to *-m*,

¹²⁴ An alternative origin is WT $\leq r$ 'because', which may be reflected in the causal uses of the terminative construction.

a process which occurs elsewhere in fast speech, e.g. $t^{h}\tilde{o}:-bo\ b\varepsilon^{2} > t^{h}\tilde{o}:-m\ b\varepsilon^{2}$ '(he) saw'. The form *son-dã*: could then be a further assimilation of the nasal. Although individual speakers may favour one of the forms *-sondã*:/*somdã*:/*sumdã*:, the data bear some evidence that these three forms may fall within the enunciatory potential of one person.

The forms -sondã:/somdã:/sumdã: are to be contrasted with the form -tsubda:, which only occurred in the speech of one elderly speaker from Pemayangtse (West Sikkim). This form derives less likely from the secondary verb số:. A possible origin of the form is the nominalized completive marker *- ts^ha :- $bo=d\tilde{a}$: > * $ts^hou=d\tilde{a}$: > $tsub-d\tilde{a}$:. These observations are as yet hypotheses. The forms are in the present work presented as unified converbal suffixes and written with the experimental Denjongke spellings tsis - song-dang for -sondã:, tsis - song-dang for both -som-dã: and sum-dã:, and tsis - tsub-dang for -tsubdã:. These forms code action that at least partly temporally overlaps with the action denoted by the following verb, see (3.47).

(3.47) קאי קדיאָראיקרי אָז פּרָ אָזאַראין אָזאַראיקרי אָז פּרן *אות בּמ:-sumdã: ódɛ: בּוֹ-kʰɛn bɛ?.* sun shine-SIM like.that die-NMLZ EQU.NE 'When the sun shines, (they) die like that.' (KT discussion with TB)

For more examples and discussion, see §15.3.3.2.

3.3.6.19 Simultaneity markers -kap, -dỹ: and -rɛŋkʰa

The simultaneity markers -kap (written $\frac{1}{2}\sqrt{N}$ skabs), $-d\tilde{y}$: (written \sqrt{N} dus) and $-renk^ha^{125}$ (written \sqrt{N} and $renk^ha^{125}$ (written \sqrt{N} dus) and \sqrt{N} dus) and $renk^ha^{125}$ (written \sqrt{N} dus) and $\sqrt{$

- (3.48) المجرة مع المحترفة المحتمة المحتمة الحترفة المحتمة المحترفة المحترفة المحترفة المحت
- (3.49) مَقْ جَـَنْجَ مَ مَحَجَمَّ go t'anpo nà: òn-**dỹ:** start first here some-SIM 'When (I) at first came here...' (KT life story)

¹²⁵ Some speakers pronounce the final vowel long, $-r\epsilon\eta k^h a$.

All these suffixes express the meaning 'when, while'. A more detailed functional treatment of these forms is found in §15.3.3.4-6.

3.3.7 Copulas

Copulas are a subclass of verbs that have little independent meaning apart from linking two arguments. In other words, copulas "have relational rather than referential meaning" (Dixon 2010: 159). In Denjongke, copulas can be identified as those verbs which can link a nominal argument to an adjectival argument. This definition includes both equative copulas and existential copulas. Equative copulas can link together two non-marked noun phrases or a non-marked noun phrase to an adjective phrase.¹²⁶ Existential copulas, in addition to linking an non-marked noun phrase to an adjective phrase, can occur with one non-marked argument (pure existential use) or link a non-marked noun phrase to a(n optionally) case-marked noun phrase (locative and possessive uses).

Morphohologically copulas differ from other verbs in that through frequent use interrogative and negating elements have merged into separate forms which do not occur with other verbs, e.g. $m\tilde{\epsilon}$: (< *ma- \tilde{i}), $m\epsilon n$ -a (< * $m\epsilon n$ -na, the interrogative -na does not occur with other verbs). Denjongke copulas are not totally devoid of referential meaning (i.e. other meaning than mere linking function), because they encode evidential distinctions. Simple copulas are summarized in Table 3.12, which does not include interrogative forms of $j\partial 2/me^2$ and and $du^2/mindu^2$ which are formed regularly by the polar quation suffix -ka/ga.

Table .							
		Personal		Sensorial	Neutral		
					Ordinary		Apparentive
		decl.	interr.		decl.	interr.	
EQ	PRS	<i>ĩ̃:/mɛ̃̃:</i> ผิส [.] /ॺส.	pá/mèna न ^{:/} ब्रब्		<i>bɛ?/mɛ̀mbɛ?</i> ఇర్ ^{:/al} a៊:ఇర్:	bo/mèmbo 127	de:/re: 255 ^{,128}
	PST				$(also = p\epsilon 2)$	র্ন'/মন্ব'র্ন'	
EX (pos./	neg.)	jø?/mè? ਘੋਨ੍'/ਕੇਨ		du?/mìndu? २५ुग ^{,(} ब्रेब:२५ुग]	(j <i>èbbe?/mèbbe?</i>) (ଝ୍ୟୁମ୍ଦ୍ୟୁମ୍/ଛ୍ସିମ୍ଦ୍ୟୁମ୍)		

Table 3.12. Simple copulas

The use of the copulas is described in §7. Comments in this section are limited to phonology, morphology and etymology. While the personal equative \hat{i} and the existentials du^2 and $j\partial^2$ have clear Classical Tibetan etymons $\log yin$, $\log yin$, $\log yin$ and $\log yod$ respectively, the origin of evidentially neutral equative $b\varepsilon^{2^{129}}$ is unclear. Semantically $b\varepsilon^2$ is somewhat similar to Lhasa Tibetan $\exists_{\nabla} red$. Morphologically it resembles the Shigatse evidentially neutral copula $pi\dot{e}$ (Haller 2000: 186), the Lhomi copula bet^{130} (neg. *mem-pet*, Vesalainen 2016) and the last syllable of the Kyirong Tibetan

¹²⁶ "Non-marked" noun refers to a noun stem with no overt case marking.

¹²⁷ This form is homophonic and homographic (in WD) with the nominalized form $m \hat{e}m$ -bo used in such expression as $m \hat{e}m$ -bo $b \epsilon ?$ 'is/was not'.

¹²⁸ This is an innovative WD form.

¹²⁹ I have heard some older speakers in Barapathing, East Sikkim, use the form $m\epsilon^2$ instead of $b\epsilon^2$. Similar ambivalence is seen in Grierson (1909: 121), who lists " $b\ddot{a}$, $p\ddot{a}$ and $m\ddot{a}$ " as copula options (in addition to "*in* or *yin*"). The story of the prodigal son accompanying Grierson's description (gotten through David MacDonald) has the written form $m\ddot{a}$ " smad and the pronunciation given as " $m\ddot{a}$ " (Grierson 1909: 123, 125).

¹³⁰ Word-final /t/ is realized as a glottal stop in Lhomi copulas (Olavi Vesalainen, personal communication).

(Lende) copula *jimbe*:, which codes recently acquired generally valid facts (Huber 2000: 157). Moreover, $[b\epsilon(2)]$ is found instead of [re2] / red / in some Tsang Tibetan varieties (Tournadre & Jiatso 2001: 82). According to Bielmeier (2000: 121), the Shigatse $pi \not e$ and Lhomi *bet* derive from Written Tibetan g_{5} byed 'make'. The same may be true of Denjongke $b\epsilon^2$. The neutral existential forms $j \partial b b \epsilon^2$ and $m \dot{\epsilon} b b \epsilon^2$ are phonological reductions of the fuller nominalized forms $j \sigma$:-po $b\epsilon^2$ and $m \dot{\epsilon} b \epsilon^2$.

The interrogative copula $p\dot{a}$ very likely derives from a historical interrogated personal copula **in-na*, which has productive cognates at least in Dzongkha *in-na* (van Driem 1998: 367), Shigatse Tibetan $j\dot{i}$ -*na* (Haller 2000: 75) and Standard Tibetan *jìn-na* (Tournadre & Dorje 2003: 223). The historical affirmative interrogative **in-na* has through frequent use been synchronically reduced to $p\dot{a}$, while its negative, still productive counterpart $m\dot{e}n$ -(*n*)*a* leaves the copular origin more transparent ($m\dot{e}n$ is the negation of personal equative \dot{i} :). It is not at all clear whether $p\dot{a}$ retains high pitch/register which would be expected on the basis of the source form *in-na* $\log_{\sqrt{\gamma}}$. As long as a detailed phonological study on its behaviour is unavailable, I find it instructive to mark the high pitch in order to retain the connection to the source form and thus keep open to discussion the possible phonological/phonetic effects caused by the source. The apparentive equative dez:/rz;, which I have not seen used in Denjongke writing, is a reduction of the fuller form $da \ bz$? 'be similar', which is also in use.

The interrogated forms given in Table 3.12 have corresponding attenuated forms given in Table 3.13.

1 4010 01101						
Marker	Polarity	Direct polar quest	ions	Att. polar and content questions		
type						
Personal	Affirm.	pá, íŋ-ga	<i>ज़ॱ/</i> ऄॱॺॱ, ऄॺॱॻॱ	pám	ਝ੍ਹਕਾ/ਯੋਰਕਾ/ਯੋਰਾਰਕਾ	
equative				íŋ-gam	জিব 'যাঝ'	
	Neg.	mèn-a, mèŋ-ga	ठाव'व', ठाव'ग'	mèn-am, mèŋ-gam	ষর'রম', মর'শম'	
Neutral	Affirm.	bo, bε-ka	র্বা, স্থ্রদ্রাশা	bɛ-kam	ञ्चन्'गव'	
equative	Neg.	mèmbo, mèmbe-ka	यवर्ग्त, यवःश्चन्गा	mèmbe-kam (?) ¹³¹	षव् :भ्रद् :गाव	

Table 3.13. Direct and attenuated questions with copulas

It is highly likely that WD क्षेत्र ज्ञ *in-nam*, which occurs nine times in the novel *Richhi*, reflects the spoken pronunciation [nám], although [i-nam] is also heard in spoken language, especially as a tag appended to imperatives. Consequently, in examples taken from *Richhi* I have written WD क्षेत्र ज्ञ *in-nam* as *nám* in the phonemic transcription.

The sensorial existential du^2 may occur together with the intensifier $-k\varepsilon$, i.e. $du-k\varepsilon$. As discussed in §7.2.2.3, the use of $-k\varepsilon$ seems to add assertiveness and certainty to the statement based on sensorial experience, although the exact semantics of $-k\varepsilon$ are difficult to unravel. The only other verbal root to which $-k\varepsilon$ can be attached is, to my knowledge, the verb and v is a single of the altight, suit', which forms any igrigs-ke /dike/ 'it's alright'.¹³² Other constructions where the intensifier possibly occurs are the progressive construction $-t\varepsilon ung\varepsilon/zung\varepsilon$ (see §8.3.3) and the completive

¹³¹ I do not currently have examples of this negated form but its existence can be hypothesized on the basis of the positive form $b\varepsilon$ -kam.

¹³² This form may be influenced by the frequent Hindi expression $t^{hik} h\epsilon$: 'it's alright'.

construction $-ts^h a - k\varepsilon$ (see §9.1.3), in both of which the last element may be $-k\varepsilon$ retained from reduced $-du - k\varepsilon$. The intensifier $-k\varepsilon$, which occurs with du^2 should not be confused with the homophonous hortative marker $-k\varepsilon/g\varepsilon$, see §11.4.

3.4 Adjectives

This section discusses the defining criteria of adjectives (§3.4.1) and then describes adjective forming suffixes (§3.4.2) and adjective-modifying suffixes (§3.4.3).

3.4.1 Defining criteria for adjectives

In some languages, there is no separate class of adjectives, which would be distinguishable from verbs and nouns. In Denjongke, however, there are some morphosyntactic criteria for positing a separate class of adjectives. Adjectives are distinguishable from verbs by their ability to appear as copula complements and from nouns by their ability to act more freely as noun modifiers (but see §4.1.2.4 for examples of bare nouns as modifiers of other nouns). Most adjectives are di- or trisyllabic, and the monosyllabic ones derive historically from disyllabic constructions in which the last syllable has been reduced and incorporated into the first one, e.g. $l \epsilon m$ 'good' < WT *legs-mo/legs-po*, s d c p(u) 'new' < WT *gsar.po*.

Because the suffix -*po* has been historically used to form both adjectives and nouns, for instance t'apo 'monk' and *zaŋpo* 'good', adjectives ending in -*po/bo* cannot be distinguished from nouns by the morphology of their citation forms. Synchronically, however, many adjectives are being formed by adjective suffixes such as -*ta*?, -*tc*^h*itc*^h*i*</sup> and -*tom*, which attach to stative verbs, and -*tc*^h*ita*?, which attaches to nouns, see (3.51).

(3.51)	tɛãː-ţaʔ	ন্তন:ন্র্যাঝা	'beautiful'	(from $t \in \tilde{a}$.' $\mathfrak{s} \subset \mathfrak{s}$ ' be beautiful')
	tɛãː-tɛʰitɛʰi	<u>रू</u>	'beautiful'	(from <i>tcã:</i> $\overline{s}_{\overline{s}}$ 'be beautiful')
	dzam-tom	ন্দ্র স্রি	'easy'	(from dzam aga' 'be easy')
	kʰõː[o-tɕʰi[a?	য়িন্দ'র্দ্রি'ক্ট'র্ব্রেশাঝ'	'quick to anger'	(from $k^h \tilde{o} to \check{h} \neg \check{h}$ 'anger')
	gja-tɕʰiʈaʔ	ক্রু'ক্ল'ন্ব্যাম'	'vast'	(from $gja_{\overline{\mathfrak{Y}}}$ 'extent')

More adjective-forming suffixes are described below. Adjectives in general are derived through adjectival suffixes from stative verbs and nouns.

Apart from suffixes, another morphological cue for distinguishing adjectives from nouns is reduplication, which is frequent with adjectives but not with nouns (e.g. *dumdum* 'short', $te^{hu}\eta te^{hu}\eta$ 'small'). Furthermore, gradient adjectives may be distinguished from nouns (and verbs) by the ability to take the superlative suffix - eo^2 , e.g. *zaypo* 'good' > $z\tilde{a}$:- eo^2 'best', $k'e:te^{hi}ta^2$ 'important' > $k'e:te^{hi}-eo^2$ 'the most important'. Adjectives do not have a separate comparative form (for comparison of adjectives, see §5.6.1.3.2).

Although adjectives can be distinguished from verbs morphosyntactically, there is a close relationship between some verbs and adjectives. Many adjectives are derived from monosyllabic property concept verbs, and many of these verbs are still used to express the same properties as the adjectives, e.g. $l\hat{\epsilon}$: 'be good' > $l\hat{\epsilon}m$ 'good', dzam 'be easy' > dzampu, dzamtom, dzamta? 'easy', $r\hat{i}$: 'be long(er)' > riyku, riyta? 'long'. Some of the adjectives derived from verbs also take the verbal negator prefix in adjectival negation, e.g. $t\epsilon \tilde{a}$: 'be beautiful' > $t\epsilon \tilde{a}$:-ta? 'beautiful' > ma- $t\epsilon \tilde{a}$:m 'ugly, not beautiful'.

Examples (3.52-55) illustrate the same root used a) as an adjective and b) as a verb. In (3.53b), the verbal strategy is the preferred one in forming an alternative question.

(3.52)	a) ག་འདོ་ འྲུམ་ཕྱུད? kʿadi dzampu bɛ?? which easy EQU.NE 'Which (one) is easier?' (KT e)
	b) קימה מבסיקסי? <i>k'adi dzam-gam?</i> which be.easy-ATTQ 'Which (one) is easier, I wonder?' (KT e)
(3.53)	 a) הי מקיקי מלי קלי קלקי קקיקי מלמי קדיקיין מרשי מלמי קדיקיין מרשי מלקיקי מלמי קדיקיקי מלמי קדיקיין מרשי מרשי מלמי קדיקיין מרשי מרשי מרשי מרשי מרשי מרשי מרשי מרשי
	b) ארך די שרא יאא ישרא יאין ישא? di t'a den-gam min-den-gam? this now be.true-ATTQ NEG-be.true-ATTQ 'Now is that true or not true?' (DR discussion with KL)
(3.54)	a) a_{1}^{2} , a_{2}^{2} , a_{1}^{2} , a_{2}^{2} , a_{3}^{2} , a_{1}^{2} , a_{1}^{2} , a_{1}^{2} , a_{1}^{2} , a_{1}^{2} , a_{2}^{2} , a_{1}^{2} , a_{1}^{2} , a_{2}^{2} , a_{1}^{2} ,
	b) $\in \neg $
(3.55)	a) װֻّרֵי אָרֵי אָרָי יוּצּׁיוִי אָרֵי פִיאָד רֵי אָזָי אָאָאן k'jõ: tc''uŋtc''uŋ =tci? nàŋca=lo ŋà ki: bom. village small=INDF inside=DAT 1SG be.born become.big 'I was born and grew up in a small village.' (KT life story)
	b) ה׳ קשַקַּקִי אָה׳ אָרָ׳ הַרְ׳ אָקַקִּמִי אָדָי <i>אָם zu: teʰuŋ-cɛ=di kjap-ti</i> 1sg body be.small-INF=DEMPH do-NF 'because my body was small' (CY interview)

Adjectival uses of the monosyllabic, verb-like property concept words are rare but do exist, especially in idioms such as (3.56), where the interpretation of each monosyllabic adjective is aided by the presence of the other $(m\dot{a}\eta pu > m\dot{a}\eta/m\ddot{a}$: '(be) many', rigku > rig '(be) long').

(3.56) مَنْ عَالَ بَوْلَ حَلَّ مَرْعَ اللَّهُ عَلَى اللَّالَةُ عَلَى اللَّهُ عَلَى اللَّ الْعَلَى اللَّهُ عَلَى اللَّ

The short forms may also occur in contexts where a longer form would usually be expected, see (3.57) employing $c \epsilon m$ 'stupid', although the longer form $c \epsilon m po$ is used elsewhere in the same story in identical position.

(3.57) דָאי פאאי פר. *t'om cèm=di* bear stupid=DEMPH 'the stupid bear' (KT animal story)

In some uses, it is not clear, whether the monosyllabic form is a verb or an adjective, see (3.58) where the form has an unmistakably verbal ending and (3.59), which occurs in a syntactic position where both verbs and adjectives may occur.

(3.58) **BAND BAND**

tc^ha:p **b**om-ts^ha:. rain grow-CMPL 'Rain has increased.' (oh, Tashiding)

(3.59) ธรรา ซู้สา (รุรา)

tc^ha:p **bom** du-kε. rain big/grow EX.SEN-IN 'Rain has increased.' (oh, Tashiding)

After this introduction, the following two sections describe adjective forming suffixes and adjective-modifying suffixes.

3.4.2 Adjective-forming suffixes

Denjongke adjectives are formed from stative verb or noun roots. The most frequent adjectiveforming suffix is **-***ta***?**, which does not have the "excessive" meaning that its cognate has in Standard Tibetan (Tournadre & Dorje 2003: 229). It typically attaches to stative verbs but occasionally also to nouns (e.g. *nenta***?** 'pleasant to hear' from sps 'ear [hon.]'). For examples, consider (3.60):

(3.60)			
ma-ta?	নঝনে:শ্রাবাঝা, নঝনে:ন্র্বাঝা	'low'	(from ma नुवाद' 'be low')
tuk-ta?	स्रुया-द्रयाय.	'thick'	(from tuk (from tuk ')
t ^h o-ta?	र्झे न्यूयाय	'high'	(from $t^h o$ \breve{s} 'be high')
t`ok-ţa?	र्देया झूंयाखा, देवा द्वाखा	'narrow'	(from t'ok र्न्ग' 'be narrow')
dɛn-ţa?	यन्तेव'न्याय'	'true'	(from den المرتجم: 'be true')
ts ^h a-ta?	र्ळ'झ्रीयांस', र्ळ'ट्रयांस'	'hot'	(from <i>ts^ha s</i> ['] 'be hot')
ts ^h ik-ta?	तर्ळेया'झुयाबा, तर्ळेया'न्याबा	'hot (of weather)'	(from ts ^h ik तळेंग 'burn')
sấː-ţa?	याह्य र ग्रीयाह्य, याह्य र याह्य याह्य	'secret'	(from sấ: קאק: 'keep secret')
sé:-ta?	ปลาร์มีปลา. ปลาร์ปลา.	'clear'	(from sé: approv 'be clear')
zã:-ta?	ন্রন:ন্যাঝ	'good (of health)'	(from $z\tilde{a}$. $\neg \exists \neg$ ' 'be good/healthy')
lèː-ţa?	लेगस.न्यांस.	'good'	(from le: लेगाय' 'be good')
riŋ-ţa?	रेट'र्याय'	'long'	(from rin $\exists \exists$ ''''''''''''''''''''''''''''''''''
cìm-ta?	विम्र'र्याया	'delicious'	(from <i>cim</i> दिय [,] 'be delicious')
tɛãː-ţaʔ	তন স্নাযা, তন দ্র্যাঝ ^{.133}	'beautiful'	(from $tc\tilde{a}$: sr 'be beautiful')
pám-ta?	ঝন্তরা:র্যামা	'equal'	(from nom अनुस' 'be equal')
pen-ta?	ञ्चेत्र द्वगर्थ.	'pleasant to hear'	(from nen हुन 'ear (hon.)')
jầː-ta?	พ่นสามาลา, พนสามาลา	'wide'	$(\text{from } j\hat{a}: u_{ray} \text{ 'be wide'})$
jàmtsʰɛn-ʈaʔ	অ'অর্চ্চর'র্যাম'	'amazing'	(from jàmtshen urated' 'be amazed')
kjo-ţa?	र्झे द्वार्य	'crooked'	(from kjo 新 'be sad')
k`jãː-ţa?	য্রান: র্যাঝ	'cold (of weather)'	(from $k'j\tilde{a}$. $\mathfrak{g}_{\pi'}$, 'be cold')
k`juŋ-ţa?	गुट्-'र्याय'	'clever'	(from <i>k'juŋ</i> الج: 'be clever')
ga-ta?	र्यात: र्याय:	ʻglad'	(from ga جرم ، 'rejoice, like')
gjaː-ţaʔ	क्तुयांस्र: द्यांस	'fat'	(from gja: المَالَمَانَ 'be fat')

(2, co)

The suffix -p(o), which has been historically used in noun formation is also a historical adjectivizer. All the basic colour terms and some frequent adjectives have been derived by -po, as shown in (3.61). While *bompu* $\max_{i \in \mathbb{N}}$ 'big' and *rinku* $\max_{i \in \mathbb{N}}$ 'long' derive from stative verbs, I have not seen short forms of the colour terms such as $m\dot{a}$: $\max_{i \in \mathbb{N}}$ used as verbs. The short colour terms, however, occur in compounds such as t'o:-na? is an 'load-black' (time of economic oppression of peasants in Sikkimese history).

(3.61)			
<i>bom-pu</i> ¹³⁴	ર્ક્ષેન્ન'નુ'	'big'	(from $bom \check{\mathfrak{A}}^{w}$ 'be big')
riŋ-ku	र्नू म्'गा	'long'	(from <i>rin</i> रेन 'be long')
sìm-pu	विमार्ग	'delicious'	(from <i>cim</i> त्रेज्ञ [.] 'be delicious')
k ^h ak-u	[यया.यी.	'bitter'	(from k^hak [NAT]. 'be bitter')
máːp(u)	न्वर्म् न्वर्स्ः	'red'	
sé:p(u)	শ্বন্থা/ শ্বন্থা	'yellow'	

¹³³ As shown by the different spellings for this word and others in (3.60), there is some variation in writing the adjectivizing suffix. For instance, the first variant here occurs in a dictionary (Lama 2013) and the second one in the novel Richhi. It seems safe to assume that the form some can be generally used for writing the adjectivizing suffix *-ta*?. ¹³⁴ Sandberg (1895: 32-33) lists *tehempo* and *tehe* as 'great' and *bompo* as 'thick (also 'loud')'. Walsh (1905: 4) glosses *bompu* as 'big'.

nàku	वया'गा'	'black'
потри/ŋотри	र्श्वेव र्धे	'blue/green'
ka:p(u)	<i>ব্যা</i> ম্ব ^{./} ব্যামার্য	'white'

All the words in (3.61) deriving from verbs, can also be adjectivized by -ta?. A difference between the present description of Denjongke and that of Sandberg (1895) is that the adjectives in Sandberg's description systematically take the ending -po (and -mo is some cases), whereas -ta? is not mentioned at all as an adjective suffix, e.g. Sandberg's (1895: 71) ts^hapo 'hot' (here $ts^hata?$) and Sandberg's (1895: 69) k^hakpo 'difficult' (here k'akta?).

The suffix -m(o), which has been used for deriving feminine nouns (see §3.2.4.2), is also used as an adjective-forming suffix in a few adjectives.

(3.62)			
ťøː-m	<u> </u> б́Га ^и	'warm'	(from $t' \phi^2 \leq 5$ 'heat')
k ^h øː-m	ট্রিন্সম	'cold (of liquid)'	(from $k^{h} \emptyset$. $\mathfrak{F}_{\mathcal{A}}$ 'to be cold')
ky:-mo	સ્યુ ^{ર્} ય ર્સે.	'common, ordinary'	(from WT styus 'altogether')
lè-m	वोस'/वोषास'	'good'	(from <i>lɛ̃?</i> लेगूम' 'be good')
mà-lɛ-m/mà-lɛ-p	ঝ'ন্দিঝ'/ঝ'ন্দিবাঝ' ঝ'ন্দিবাঝ'র্ন্য'	'not good, bad'	

The derivational suffixes $-tc^{h}ifa?$ (meaning 'great, big', written more traditionally $\widehat{\mathbb{B}}_{\mathbb{Z}}^{\mathbb{N}}$ chedrags and more phonologically $\widehat{\mathbb{B}}_{\mathbb{Z}}^{\mathbb{N}}$ chi-drags), $-tc^{h}\varepsilon$ (meaning 'great, big') and $-tc^{h}u\eta$ (meaning 'small') turn nouns into adjectives:

(3.63)

tsa-tɕʰit̥aʔ	₹'ळे'॒ऽ्याय	'invaluable'	(from <i>tsa</i> ₹ 'root, nerve')
sém-t¢ ^h uŋ	ষ৾য়য়''&ৢ⊏'	'humble'	(from <i>sém</i> रोग्रय' 'mind')
dzik-tɕʰiţa?	ॾॖऀऀऀ॑॔ॖॕॱॾऀॱड़॒ऺॺॺ	'heavy'	(from WT En ljid 'weight, heaviness')
cúk-tc ^h ẽ:	नुगषाः केवः	'strong'	(from cúk grav 'strength')
pàm-t& ^h ɛ̃:	नुव्यया केव्	'proud'	(from <i>nàm مجمع</i> 'arrogance')
pàm-t& ^h uŋ	नेथाया.कीट.	'humble'	(from <i>nàm جمع</i> ر 'arrogance')
kʰõːto-tɕʰita?	র্বিন্দ:র্রি:ক্ল:নুবাৃঝ:	'quick-tempered'	(from $k^{h}\tilde{o}$ to $\tilde{h} \subset \tilde{h}$ 'anger')
k'ɛː-tɕʰɛ̃ː	यात्थ'ळेतु'	'important'	(from k'ɛː གལ་ 'importance')
gja-tɛʰiʈa?	क्तु:क्र:ट्याय	'vast'	(from gja 🛒 'extent')

The reduplicated suffix $-te^{hite^{hi}}$ (deriving from WT \hat{a} che 'great, big', written $\hat{a} \cdot \hat{a} \cdot or \hat{a} \cdot \hat{a}$) forms adjectives from stative verbs:

(3.64)			
pe:-tc ^h itc ^h i	न्दोषाक्वेके	'charming (of child)'	(?)
tɛãː-tɕʰitɕʰi	જ્રમ્ 'સે'સે'	'beautiful'	(from tcã: sr' 'be beautiful')
t& ^h uŋ-t& ^h it& ^h i	ਛ੍ਹ⊏'ಹੇ'ಹੇ'	'small'	(from te ^h uŋ ᠳᠲ' 'be small')
dzam-tɛʰitɛʰi	પદલાક્ષ.થ.	'easy'	(from dzam agar 'be easy')
pen-te ^h ite ^h i	ਝ੍ਹਰ 'ਛੇ'ਛੇ'	'pleasant to hear'	(from nen हुन 'ear (hon.)')
ga-te ^h ite ^h i	न्यातःक्रेंकें	ʻglad'	(from ga קקק' 'rejoice, like')

Specific intensifying suffixes not occuring with other adjectives may be used with the frequent adjectives *rinku* and *bompu*. The suffix $-k^h jam$ (WD rist khyam) is used in Tashiding (West Sikkim) and *-pam/kam* (WD rist max/max) in Martam (East Sikkim) to form adjectives from stative verbs. The consultants thought that adjectives formed with $-k^h jam$ or -pam/kam, when compared with the ordinary forms with *-po*, implied a greater degree, see (3.65).

(3.65)	a)	Tashiding bom-k ^h jam riŋ-k ^h jam	র্শ্বরু'ট্রেপ্র' হুল'ট্রিপ্র'	'big' (bigger than <i>bom-pu</i> ধ্র্রান্ড্র') 'long' (longer than <i>riŋ-ku</i> २८-স')
	b)	Martam bompam riŋkam	র্শ্বিরাদ্যরা দিন্দাযা	'big' (bigger than <i>bom-pu</i> স্থ্র্র্র্স্ স্:) 'long' (longer than <i>riŋ-ku</i> २८ गु.)

The derivative suffix $-tcen/tc\tilde{e}$: (WT/WD \overline{eq} can) has the meaning 'having, bearing'. It attaches to nouns to form adjectives, see (3.66), but has historically also formed nouns, see (3.67).

(3.66) k'ɛː-tɕɛ̃: p'uzi-tɕɛ̃: rik(o)-tɕɛ̃:	याव्यःच्ठत्रः सुःयाविष्यःच्ठत्रः न्देयाःर्गे:च्ठत्	'important' 'child-having' 'intelligent'	(from k'ɛː གལ་ 'importance') (from p'uzi नुःगविषः 'children') (from riko देषार्गे 'intellect')
(3.67) sím-tcẽ:	ম৾ঝম'ডব'	'animal'	(from <i>sém</i> ন্ট্র্যন্থ 'mind')

Other adjective-forming suffixes are *-tom* and *-ba/wa* (which looks like a circumstantial converb, see §15.8.1).

(3.68) <i>ts^ha-tom</i>	र्क में ज़	'hot'	(from $ts^h a \notin be hot'$)
dzam-tom	<u> २</u> ह्ब'र्नेब'	'easy'	(from dzam तहता' 'be easy, soft')
dzø:-wa	र्म्स्थ'न'	'mistaken, false'	(from dzø: cerr')
nòr-wa, nòr-wo	ਰ੍ਰੇਂਨਾਧਾ, ਰ੍ਰੇਂਨਾਧੇਂ	'mistaken, false'	(from nòr र्ज्नेन: 'err')

Adjectives may also be formed by adding a reduplicated suffix to a stative verb. Consultant KN commented that the reduplicated derivative suffixes add intensity to the adjective beyond what is implied by a non-reduplicated suffix, e.g. *dzamtõ:to* 'easy' is even easier than the alternatives *dzampu* or *dzamta?* 'easy'.

(3.69)

pe:-tokto?	<u>न्द्येख</u> र्फेयार्फेया	'charming'	(origin unknown)
dum-bɛbɛ?	תקמיקריקר	'short'	(from <i>dum</i> त्युत्र, 'be short')
ts ^h a-tokto?	र्ळ र्जेया र्नेया	'hot'	(from <i>ts^ha</i> \mathfrak{s}^{\cdot} 'heat, hot')
síː-tõːto	বর্মিন্য'র্ন্ন='র্ন্ন'	'refreshingly cold'	(from si. ज्ञीया 'feel cool')
nàk-susu?	न्याः स्र्याः स्र्याः	'dark'	(from <i>nàk</i> ज्या [.] 'black')
t'o-tiptip ¹³⁵	র্ন্র নির্মান দ্বানা	'warm'	(from $t' \phi ? \stackrel{'}{\leq} 5'$ 'heat (v.)')
dzam-tõ:to	<u> </u>	'easy'	(from dzam agai 'be easy')
k ^h ø:-takta?	ট্রিন্ম'দশ্য'দশ্য'	'cold'	(from $k^h \emptyset$. $\tilde{\mathbb{B}}^{\mathbb{A}^r}$ 'be cold')
k ^h ø:-si:si:	นี้ณาสีณาสีณ ^{.136}	'chilly'	(from $k^h \emptyset$. $\tilde{\mathbb{S}}^{n'}$ 'be cold')

- (3.70) รับนิก ตู้ลบสิลบสิลบสิลบสิลบสา t'o:pø: k^hø:si:si: lúŋma morning.GEN chilly wind 'morning's chilly wind' (Richhi 1)

Colour words excel in reduplicated suffixes that are rare in other words, see §17.4.

Other adjectival endings, which are of unkonwn origin, are *-nam*, *-su*?, *-p^h\varepsilon m/p^h ym* and *-ka*, see (3.72).

(3.72)			
gjaː-nam	শ্র্র্যাম'বৃষ'	'fat'	(gja: कुष्पूष 'be fat')
nàk-su?	नया-सुया-	'dark'	(nàk ज्य 'black')
tuk-p ^h ɛm/tuk-p ^h ym	ञ्चया'म्वेस'	'thick'	(tuk हुन' 'be thick')
pjaŋ-ka	र्द्ध्यूत्त.यो.	'poor'	(WT gr sprang 'poor')

While $-ta^2$ appears to be the most productive adjectival suffix, often two or more alternative adjectival endings may be attached to the same stative verb root, as shown in (3.73).

¹³⁵ also *t'ø:tiptip*

¹³⁶ also ฐัญารสิญารสิญา

Reduplication is a frequent strategy for forming adjectives from stative verbs (e.g. *dumdum* 'short' from *dum* $_{abs}$ ' be short'). These forms also occur as non-reduplicated adjectives with a suffix (e.g. *dumta?* $_{abs}$ ' $_{abs}$ 'short'). Reduplication and near reduplication are here considered ideophonic features, which are essential in forming ideophones, see §17.1. Ideophones are often ambiguous with reference to their status as adjectives or adverbs. The words listed in (3.74) are used mainly adjectivally. Other reduplicated words, which seem more ambiguous with reference to adjective vs. adverb distinction are introduced under ideophones in §17.1.

(3.74)

dop-dop	ষ্ট্র্য:র্ষ্ট্র্য	'slow (of animate being)'
dum-dum	તનુરા તનુરા	'short (vertical)'
t ^h uŋ-t ^h uŋ	<u> </u>	'short (horizontal)'
sáp-sap	শ্রব:শ্রব:	'thin'
sóp-sop	র্মন'র্মন'	'soft'
rip-rip	হ্য্যিন'হ্য্যিন'	'dim'
tc ^h uŋ-tc ^h uŋ	æु⊂ॱæु⊂'	'small'
pòp-pop	র্নৃন'র্নৃন'	'soft'
jầː-jãː	ちん.	'light (opp. of heavy)'

3.4.3 Adjective-modifying suffixes

Adjectives can be modified by the diminuative and superlative suffixes, which attach to the adjective root. The diminuative suffix (WD $\operatorname{grav}/\operatorname{grav}$ or $\operatorname{grav}/\operatorname{grav}$), which replaces the last syllable of the adjective, lessens the quality or quantity expressed by the adjective.

(3.75)						
bom-pu	ર્સેઅ'નુ'	'big'	>	bom-sy:	ર્ક્ષેત્ર.શ્રેના	'quite big'
ma-ta?	न्रवतःन्यूयाः	'low'	>	ma-sy:	न्ह्यत.श्रिज्ञ.	'quite low'
t ^h o-ta?	अंर्वे द्वग्र	'high'	>	t ^h o-sy:	અંધેં સુવ્ય	'quite high'
tuk-p ^h ɛm	्रश्वया'म्वेव्य'	'thick'	>	tuk-sy:	ह्यया-सुत्य	'quite thick'
t& ^h uŋ-t& ^h uŋ	æुं⊤ æुं⊏'æुं⊏'	'small'	>	t& ^h uŋ-sy:	જીં⊏.ધોના.	'quite small'
рор-рор	র্নৃন'র্নৃন'	'soft'	>	nòp-sy:	ର୍ଶିସଂକ୍ଷାଦ୍ୟ	'quite soft'
khõ:totchi-ta?	র্মিন্দ ব্রি'ক্ট'ন্র্যাম	'quick to anger'	>	khõ:totchi-s	y. ૉવઽ ૉૉ જે જે જોવા	'quite quick to anger'

The phonological realization of the diminuative in (3.75) comes from consultant KN. An alternative pronunciation - ϵy : is suggested by the word $\operatorname{deg}(\overline{w}, \overline{g})$ tshap-che-shus / $ts^hapte\varepsilon ey$:/ 'quite

serious (of medical condition)' in the novel Richhi. When shown this word in Richhi, KN responded by commenting that there is probably an error in the text, the right form being der(3, N) tshap-che-sus /tshaptcesy:/. However, the form -ey: represents more likely a variant pronunciation than an error, as shown by (3.76) below. The alternative form -sy: is illustrated in (3.77). The same formative appears to be used as reduplicated in the quantifier ey':ey: 'a bit' (WD der(N) shus-shus or der(2, N) shull. It also occurs in the negated perfect construction tsi-ey: mè? [play-trace EX.PER] 'has not ever played', see §8.1.4.

The superlative marker $-co^2$ (WD ξ_{n} shos) intensifies the degree of the adjective. It attaches to the root from which the adjective is derived, e.g. $tc^hu\eta tc^hu\eta$ 'small' > $tc^hu\eta co^2$ 'the smallest', tc^ham - ta^2 'agreeable' > tc^ham - co^2 'the most agreeable'.

קאי קדיאי הקאי קאי פריאאי הק' מיקי אקן ה'ווד ה'amo=di: (3.78)p'amo=di: tchun-co=lo p'um p'um de: middle.one=DEMPH.AGT daughter small-SUP=DAT daughter like.this làp-o hE?. say-2INF EOU.NE 'The middle-born daughter said like this to the youngest daughter:...' (rna-gsung 2)

Denjongke does not have a separate comparative form of adjective. Comparison is accomplished with the help of the ablative case, see §5.6.1.3.2.

3.5 Adverbs

This section first provides an introduction to adverbs (§3.5.1) and then introduces the various types of adverbs and their derivation (§3.5.2).

3.5.1 Introduction to adverbs

Adverbs are here defined as a somewhat heterogeneous group of words that modify other constituents than nouns (see Schachter & Shopen 2007: 20). That is, adverbs modify verbs, adjectives, other adverbs and whole clauses. This definition of an adverb is mainly syntactic. Several time words such as $t'o:pa \leq v_1$ '(in the) morning' and $p^hiru2 \geq v_2$ '(in the) evening' are interpreted as adverbs, because their citation forms occur as adverbials without case marking.
- (3.79) $\vec{r} \cdot \vec{r} \cdot$
- (3.80) ثَلَّاتِي العَلَيْمَ العَلَيْقِ اللَّهِ اللَّهِ اللَّهُ عَلَيْتُ اللَّهُ (Richi 4) (Richhi 4) (Richhi 4)

Temporal and locative adverbs, however, also have the nominal feature of occurring as genitive modifiers:

- (3.81) छेंद्र गो जवावा *p^hiru=gi cè*:*la*? evening.GEN meal.HON 'evening's meal' (Richhi 62)
- (3.82) क्रें'त्र'में' कु

óna=gi tc^hu there=GEN water 'the water (of) there' (UTR plains story)

Temporal and locative adverbs also take ablative case to express spatial or temporal starting point:

(3.83) בָּיִבְיִמְשָּׁי שָּאַ עוּד: שָּׁרְ־שׁרִשָּׁי שָּרָן *f'o:pa=le* sàm k'are mè:-k^hen be?. morning.ABL food.HON any NEG.EX-NMLZ EQU-NE. 'Since morning there has not been any food.' (DB trip story)

Some locational and temporal adverbs may receive optional dative-locative marking, which is also a noun-like quality, e.g. t'o:pa '(in the) morning' > t'o:pa=lo 'in the morning'.

There is not always a clear distinction between adjectives and adverbs in that adjectives may be used adverbially without modification, as shown by the adverbial (3.84a) and adjectival use (3.84b) of $mala_{avav}$ 'quick(ly)'. For similar uses of $n \epsilon mu$ and equivalent the adverbial in (3.85) and adjective in (3.86).

(3.84) a) ها تعامی (معلق) màla? gju?. quickly go 'Go quickly!' (KN e)

	b) हिंदेगी चर्छ्य (bike)	તર્ને લેન ક્ષે સાલવા ધેની		
	k ^h u=i=gi	baik=di	lèpti màla?	jờ?.
	3SGM=GEN=GEN 'His (motor)bike	bike(Eng.)=DEMP is very fast.' (NB	H very fast e)	EX.PER
(3.85)	बजे.खु.२८ भ्रुव.पॉ प्रञ्जव.पह né:mu=rã: mémp really=AEMPH docto 'Doctor Tenzing rea	ৰ' ক্তু'ৰ্কচ' শুনিম'ৰ্ম' স্নিবম po tendzi: te ^h ute or PN clock Illy arrives at two o	hø? ní:=lo .time two=DAT o'clock.' (Richh	<i>lɛp.</i> arrive i 31)
(3.86)	ञ्चतः क्रुषः ग्रीषः पासुनः र्थेः पात्रनः द <i>lenge:=ki sú</i> PRN.HON=AGT sa 'What you said is tri	म्राप्त्व [.] अवे.सु. झन् <i>m-bo nấ</i> y.HON-2INF do ue.' (TB e)	:- <i>k^hẽ: nế:1</i> .HON-NMLZ true	<i>mu bε?</i> . e/real EQU.NE

Morphological cues for adverbhood are discussed in the next section on adverb derivation (§3.5.2). For ideophones, a special category of adjectives and adverbs, refer to §17.1. For the use of adverbs in clausal context, see §5.6.3.

3.5.2 Adverb derivation and types of adverbs

The following paragraphs introduce manner (\$3.5.2.1), locational (\$3.5.2.2), temporal (\$3.5.2.3), quantifying (\$3.5.2.4) and other adverbs (\$3.5.2.5). The last section describes the approximative and directional adverbial suffix *-tcika* '-abouts, around; towards' (\$3.5.2.6).

3.5.2.1 Manner adverbs

Manner adverbs are typically formed by the adverbializer -p'ja(ti) = p'ja(ti), which attaches to adjectives, see (3.87). The adverbializer -p'ja(ti) derives from the nonfinal converb form of the verb p'ja 'do', p'ja-ti, where the converbal ending may be dropped. Similar use of the verb 'do' as an adverbializer is reported for Lhasa Tibetan gw' byas /tee:/ (Denwood 1999: 186) and Dzongkha gsg' bad /be~ba/ (van Driem 1998: 317). Both the short form p'ja and the long form p'ja-ti are in use, as seen in (3.87) and (3.88). The short from -p'ja is written as a suffix, whereas the converbal form p'ja-ti is written separately, reflecting its less grammaticalized status.

(3.87)	Adjective			A	dverb			
	k`aly?	বা'শ্রিম'	'slow'	k	ʻaly-pʻja, k'aly pʻja	ı-ti		'slowly'
	lèm	ন্দ্র্যারু.	'good'	lè	m-p'ja, lèm p'ja-ti			'well'
	dzamts ^h its ^h i	ุ่ เสียง เชิง เชิง	'soft'	d	zamtɛʰitɕʰi-pˈja, dz	amt¢ ^h it¢ ^l	hi pʻja-ti	'softly'
(3.88)	લને ર્નુજ ગાલ્લા ગાલ <i>dદ:-p'ja like.that-ADVZ</i> '(They) speak	אי לאיאי פא <i>k'aly.</i> R slow like that,	≡ત્રા'ક્રે'ક્રે' ક્રીં ? <i>k 'aly?</i> slow slowly,	^{N'} ಕ್ರಾನನ'	ز قانطر العالي <i>dzamtc^hitc^hi-p'ja</i> soft-ADVZR ou know.' (RL)	<i>kjap</i> speak	<i>bε=εο=</i> EQU.NE=	<i>la.</i> =AT=HON

Manner adverbs may also be formed from adjectives by reduplication, e.g. $k'aly? k'aly? \P' \P'$ $\P \P'$ 'slowly', which is an alternative to k'aly-p'ja, although reduplication can also co-occur with the adverbializer -p'ja(-ti), see (3.88). Reduplication is considered an ideophonic feature, see §17.1.

There are also nonderived adverbs of manner which are not marked by p'ja(ti). Tables 3.15 and 3.16 list adverbs of manner related to sleeping and other adverbs of manner respectively.

Table 3	Table 3.14. Adverbs of manner related to sleeping				
k'õ:ke	र्वेष्ट:भ्रुव्य:	'(sleeping) on one's back'			
k ^h abup	দ্র-রূব-	'(sleeping) on one's tummy'			
sù:te	नुर हे'	'(sleeping) on one's side'			

pámtei(lo)	ঝন্তঝ'ন্তব্য'(র্ন্য')	'together'
t ^h alamki	ાર્ બાર્ચ્સ શેખ.	'clearly'
hatok ^h a, hatolo	न्दर्भवायर, न्दर्भवार्थ	'suddenly'

The adverb *pámtei(lo)* angar Eqr (ar) 'together', is closely related to the postposition *pámpu/pámtei?* angar Eqr/angar Eqr 'with'. In my data, the adverb typically occurs with the dative-locative marker added to the postpositional form but in one instance a form identical with the postposition is used as an adverb.

For examples on the uses of the adverbs of manner, refer to §5.6.3.1.

3.5.2.2 Locative adverbs

Many locative adverbs are formed from nouns by the suffix $-k^ha$. The form probably derives from WT $rac{}^{}$ *kha* 'mouth, face, (front) side' (Jäschke 1881: 34), which, as suggested by the written Denjongke form $rac{}^{}$ *khar* in the novel Richhi, is supplemented by the historical locative marker *-r*. The adverbializer $-k^ha$ is not as productive in forming locative adverbs as -p'ja(ti) is in forming manner adverbs.

gjap	<u>. r</u> đ	'back'	>	gjap-k ^h a	'in the back'
sá	<u>N</u> .	'soil'	>	sá-k ^h a	'on the floor, on the ground'
làm	নাধ্য'	'road'	>	làm-k ^h a	'on the road/way'

As an indication that $-k^h a$ is a derivational suffix rather than an inflectional marker like case, $-k^h a$ may be supplemented with the dative-locative marker, e.g. $lamk^h a = lo$, $sak^h a = lo$. The form $-k^h a$ also converts some verbs into nouns and thus functions as nominalizer, e.g. $dze: aze: k^h a$ 'meeting, place/occasion to meet'. Sometimes, $-k^h a$ appended to a noun does not change the meaning, e.g. $k'j\tilde{o}: zz$ 'village' > k'joy- $k^h a$ 'village'.

Some other locative adverbs are listed in Table 3.16.

Table 3.16. Some locative adverbs

10010 5.10.1		
t ^h aːpi	ষশ'ন্ট'	'close, near'
t ^h aːriŋ	वग'रेट'	'far away'
làmta?	অঝ'দগ্য'	'above the road'
làmmɛ?	નાથા.શૈંટ.	'below the road'
jà: ¹³⁷ , jàtɛ	બત્ર', બ'જ્રે'	'up(wards)'
ò?, òtε	र्तेया', र्तेया'झे	· 'down(wards)'

In addition to forms in Table 3.16, pro-adverbial demonstratives (e.g. $n\dot{a}$: \vec{a} : 'here', $ts^h u:k^h a \vec{a}$ ' $\vec{a} \vec{a}$ ' $\vec{a} \vec{a}$ ' (nee'), $ts^h u:k^h a \vec{a}$ ' $\vec{a} \vec{a}$ ' (nee'), $ts^h u:k^h a \vec{a}$ ' $\vec{a} \vec{a}$ ' (nee'), $ts^h u:k^h a \vec{a}$ ' \vec{a} ' (outside'), $p^h i lo 2$ ' \vec{a} ' \vec{a} ' (outside'), bu:na \vec{a} ' \vec{a} ' (in the middle') are used as locative adverbs. For the uses of locative adverbs in clausal context, see §5.6.3.2.

3.5.2.3 Temporal adverbs

Temporal adverbs are here divided into those referring to times of day (Table 3.17), those referring to days and years (Table 3.18) and other temporal adverbs (Table 3.19). Words from the first two categories also function as nouns, although their most frequent use is adverbial.

1 4010 5.17.	Noull like tel	ipolar adveros referring to times of day
ŋaru, haru	প্র'র', দৃ'র'	'(in the) morning, tomorrow morning'
t'o:pa	র্নু:ন্য:	'(in the) morning'
nìта	ને.જ.	'(in) day-time'
nìŋguŋ	वेत्र गु∽	'(at) noon'
nìmp ^h i?	नेवा होंन्।	'(at) mid-day'
p ^h i:ts ^h am	ર્સુ સર્જ્યયા	'(at) dusk'
p ^h i(:)ru?	ફ્રી.ર.	'(in the) evening, night (after dark)'
nùpp ^h i?	તૃ્વ'ર્સેન્'	'(at) midnight'

Table 3.17. Noun-like temporal adverbs referring to times of day

The adverbs referring to times of the day may be followed by the spatiotemporal markers $-k^ha$ and/or =lo, e.g. $\eta aru(lo)$, $\eta aru(khalo)$ 'in the morning', $p^hi:ts^hamlo$, $p^hi:ts^hamk^ha(lo)$ 'at dusk'. Moreover, the form ηalo grave, which does not occur as a noun, can express 'in the morning'.

¹³⁷ The shorter forms $j\dot{a}$: and \dot{o} ? are used especially with verbs of motion, e.g. $j\dot{a}$: \dot{o} ? $gju-k^{h}\tilde{\varepsilon}$: $\Box x^{-}$ $\tilde{\kappa} = 0$ [up down go-NMLZ] 'those who go up and down' (Richhi 158) and $j\dot{a}$: also with $l\dot{o}$: 'rise', e.g. $j\dot{a}$: $l\tilde{o}$: $\Box x^{-}$ $\tilde{\kappa} = x^{-}$ 'rise up, stand up'.

gu:nup	न्गु'कुंग'	four days ago	guːniŋ	न्गुःवेन्ः	'3rd year before this one'
ze:nup	यविषातुनः	three days ago	<i>zo:niŋ</i> ¹³⁸	गर्वि वेट	'the year before last year'
k ^h anup	षि'तुन	the day before yesterday			
dã:	אקר.	yesterday	nàpiŋ ¹³⁹	व'वेन	'last year'
t'ariŋ	न्'र्रूट'	today	t'utci?	र्रु:चेग'	'this year'
t ^h orã:	র্ষ্র'মন্দ্র্য'	tomorrow	sấːpø?	শ্বন্দ্র্র্	'next year'
náŋtsi	য়ঀ৸য়৾৾৾য়৾	the day after tomorrow			
ze:tsi	यविषाः कें	in 3 days from now	ze:pø?	শ্বন্ধার্থান্য	'2nd year after this one'
gu:tsi	न्गु'कें'	in 4 days from now	gu:pø?	ন্ন্যু:ইন্	'3rd year after this one'

Table 3.18. Days and years

Table 3.19. Other temporal adverbs

k ^h atsã:	[4]、タト	'some days/time ago'
t'itsi	<u> </u>	'recently'
ť asã:	5.25	'this morning'
làmsã:	ਅਕਾੜਸ:	'suddenly'
t'ato	୵ୖୖୢୄ	'now'
t`aruŋ	न्'रुन्'	'again, yet, still'
te:ra	<u>ह</u> ें'रू'	'again'

In addition to the above-mentioned temporal adverbs, the postpositions gjable $\mathfrak{g}_{\mathfrak{T}'\mathfrak{M}}$ 'after(wards)' and $p\acute{e}nle/hale/hema$ $\mathfrak{g}_{\mathfrak{T}'\mathfrak{M}}'/\mathfrak{H$

The indefinite temporal adverbs, which use reduplication, express an unspecific temporal reference point (cf. indefinite pronouns, which express indefinite person reference, see §6.3.1):

Table 3.20. Indefinite temporal adverbs

p'aːp'aː(na)	नम'नम'(a')	'sometimes, now and then'
kapkap(na)	শ্লনম'শ্লনম'(বৃ')	'sometimes, now and then'
rɛgaː(rɛgaː)	रे.तयात.(रर	'sometimes(sometimes)'

For indefinite adverbial expressions corresponding to English *whenever*, *wherever*, *however* and *for whatever reason ("whyever")* refer to §6.3.2. Temporal adverbs are exemplified in §5.6.3.3.

3.5.2.4 Quantifying adverbs

Because adverbs were above defined as words which modify other words than nouns, quantifying adverbs can be defined as words which quantitatively modify other words than nouns. Quantitative adverbs can be divided into verb-modifying (Table 3.21), adjective/adverb-modifying (Table 3.22)

¹³⁸ also *zø:niŋ*

¹³⁹ also nàni? ज्र'हेन्'

and numeral-modifying adverbs (Table 3.23). Two adverbs, $l \hat{\epsilon} p(ti)^{140} \operatorname{eq}(\underline{s})$ 'very (much)' and $dtsi(m) \operatorname{eq}(\underline{s})$ 'a bit', occur as both verb and adjective modifiers.

$\frac{1}{l\hat{\varepsilon}p(ti)}$	<u>০ mounying q</u> ন্দিব'(ষ্ট্রু')	'very much'
kεːp, kεːpo ¹⁴¹	गोबाया, गोबार्या	'much, a lot'
màŋpu, màŋpo ^{14.}	² અંદ્ર સું, અંદ્ર સેં	'much, a lot'
ts ^h ɛdɛ̃ː	र्बन् स्व	'considerably'
màŋts ^h ø?	مح فر	'to great degree, more (than)'
pùŋts ^h ø?	ورجر بي الم	'little, less (than)'
сў:су:	রিদ্রা.রিদ্রা.	'a bit'
eý:tey:	्मुष्य:च्चेया:	'a bit'
cýry	. सुह्य में	'a bit'
átsi(m)	জ'র্ন্থ(ম')	'a bit'
átem	ष्णः झेव्रा (?)	'a bit' (rare)
րնդրսդ	લુદ્ર લુદ્ર	'little, few'
tse:/dze:	ਡੇ./ਪੁਵੇ.	'at all' (+negation)
tsa(:)lɛ	3 and	'at all' (+negation)

Table 3.21. Verb-modifying quantitative adverbs

Table 3.22. Adjective and adverb-modifying adverbs

lèp(ti)	ભેવ (જ્વે')	'very much'
pemissiki	<u>ન્</u> યે સે જેનુ છે	'extraordinarily' (in Martam: pesimipø:/pemisipo)
k'ã:mentse?	गंद्र :अत्र : र्वत्	'limitless'
átsi(m)	জ'র্ন্ড(ম')	'a bit'

¹⁴⁰ This form is often pronounced with markedly high intonation.

¹⁴¹ $k\epsilon:p$ is a frequent and versatile quantifying morpheme which can modify a verb/clause or a noun. It also occurs as the second (quantifying) argument of a copula and independently as an indefinite pronominal 'many'.

¹⁴² m angu/m angu/m

Pre-nu	ımeral		
	halam	হ'শঝ'	'about, approximately'
	t'yːmɛnɛ	नुषाःग्वनुः	'about, approximately; almost'
	dɛːtɕi?	तने.रूग.	'about, this much' (lit. 'like.this-one')
Post-n	numeral		
c	ļakts ¹ ø?	झ्या र्केट्-	'over, more than' (lit. 'more.than-limit')
	kort£i?	র্ন্নুন:ন্তব্য'	'about' (literally 'around-one')

Table 3.23. Numeral-modifying adverbs

For examples of verb-modifying quantitative adverbs, see §5.6.3.4. Adjective/adverb-modifying adverbs and numeral-modifying adverbs are illustrated in §4.3.1 and §4.4 respectively.

3.5.2.5 Other adverbs

Other adverbs include the epistemic adverbs (Table 3.24) and the restrictive evaluative adverb teiku/teuku (η) $\exists \eta$: η : 'only'.

Table 3.24. Epistemic adverbs

néːmu(ra)	ಷನ್ರ`ಸ್ರ'(ಸ')	'really'
mèntsene/mèntseno	য়ৢৢৢৢৢৢৢৢয়৾৾৻৵য়য়ৢৢয়৾৾য়৾৾য়৾	'perhaps, maybe'
mènne/mènni ¹⁴³	वाव वे.	'perhaps, maybe'

Epistemic adverbs and the evaluative *tciku* 'only' are, together with other verb/clause-modifying adverbs are exemplified in §5.6.3.5.

3.5.2.6 Directional and approximative suffix -tcika '-abouts, around'

Adverbs of time and place may be followed by the suffix $-t\epsilon ika \exp[\pi/(8\pi)]$ which marks directionality 'in the direction of, towards' and/or approximativeness '-abouts, around', thus functioning similarly to the Nepali affix *-tira*. The directional meaning is illustrated in (3.89) and (3.90).

- (3.89) مَا مَحَ مَعْمَا بِيَّلَ مَعْمَا مُعْمَا مُحْمَا مُعْمَا مُحْمَا مُحْمَا مُعْمَا مُحْمَا مُحْمَا مُحْمَا مُعْمَا مُحْمَا مُعْمَا مُعْمَا مُعْمَا مُحْمَا مُحْمَا مُحْمَا مُحْمَا مُحْمَا مُحْمَا مُحْمَا مُحْمَا مُعْمَا مُحْمَا مُحْمَا مُعْمَا مُعْمَا مُعْمَا مُعْمَا مُعْمَا مُعْمَا مُعْمَا مُعْمَا مُحْمَا مُعْمَا مُعْمَا مُ
- (3.90) $\int \tilde{\eta} \vec{a} \cdot \tilde{z} \cdot \omega \vec{a} \cdot \vec{a}$

The approximative meaning (glossed APPR) is illustrated in (3.91) and (3.92).

¹⁴³ The form *menni* is from consultant KN (Martam).

(3.91)	p ^h iːtsʰam-tɕika	દ્વું અર્ಹઅષ્ય રહેવા ગા	'at about dusk, around dusk',
	k`ana-t£ika	या'व' रुया'गा'	'approximately where, whereabouts'.
	t`ãːpu-tɕika	न्द्र:रुग्रेच्या:गा	'once long ago' (lit. 'long.ago- <i>tcika</i> ')
	t`atar-t£ika	८'झर छेगांगां	'at around this time, '?nowabouts' (lit. 'now-tcika')

(3.92) (3.92)

In addition to the markers above, the quantifying nominalizer $-ts^{h} \varepsilon^{2} = ts^{-1} \varepsilon^{2}$ may attach to verbs to express 'as much as is x-ed', see §13.2.4.

3.6 Minor word classes

The minor word classes are personal pronouns (§3.6.1), reflexive pronouns (§3.6.2), reciprocal pronouns (§3.6.3), indefinite pronouns (§3.6.4), demonstratives (§3.6.5), question words (§3.6.6), numerals (§3.6.7), postpositions (§3.6.8), connectives (§3.6.9), interjections (§3.6.10) and discourse particles (§3.6.11).

3.6.1 Personal pronouns

Denjongke personal pronouns are summarized in Table 3.25. The 2PL form $te^h \phi = tsu$ \mathfrak{F}_{γ} is given in brackets because it is a marginal form not accepted by all speakers.

				D1 1	
			Singular	Plural	
1p			ŋà 二	nàtea? ८'ख्य	
2p	low-level		$tc^h \phi$? \breve{a}_{5} .	(<i>tc^hø:=tsu</i>	<i>k`utca?</i> শ্ য 'তথ্য'
	mid-level		<i>rã:</i> ᠵᠵ	rã:=tsu रूट:र्डु:	
	honorific		lenge? झुत्र मुख	k'utca lenge? गु'ठग' झुव'हु	দুরা'
				<i>ॄृृृɛŋgɛ:(=tsu)</i> झुत्र'मुख'(र्डु')	
3p	ordinary	masc.	$k^h u$ \mathbb{R}^n	$k^h ilde{o}$: The second se	
		fem.	mù/mò ह्र		
	honorific		$k^h ilde{o}$: $ ilde{o}$:	khõ: lenge:(=tsu) ॉवन: झुर	এক্টিন্দ্র.
			khõ: lenge? ॉवन झुब कुब		

Table 3.25. Personal pronouns

The first person pronoun is $\eta \dot{a} = from$ which the plural form $\eta \dot{a}tea? = from$ is formed by adding the Classical Tibetan plural marker from cag. According to Beyer (1992: 230), in Classical Tibetan from cag occurs "only after personal determiners," an observation that also holds for Denjongke.¹⁴⁴ Unlike many Tibetic languages such as Old Tibetan (Hill 2010), Balti (Bielmeier 1985: 76), Amdo (Ebihara undated), Dongwang (Bartee 2007:108), Shigatse (Haller 2000: 50) and Lhomi

¹⁴⁴ However, Sandberg (1895: 23) reports *-tea?* as a plural marker that can be used, unlike in my data, with at least some common nouns, see §3.7.4.1.

(Vesalainen 2016: 21), which have an inclusive vs. exclusive distinction in first person plural pronouns, Denjongke pronouns do not make a clusivity distinction.¹⁴⁵ The honorific personal pronoun lenge2 and responses in that I am not aware of it being used as a personal pronoun in other Tibetic languages.

The use of the personal pronouns is discussed in §6.1.

3.6.2 Reflexive pronouns

Denjongke has three reflexive pronouns based on $r\tilde{a}$: $\pi\pi$ 'self':

Table 3.26. Reflexive pronouns

Form	•	Notes on use	Gloss
=rã:/=ra	<u> ۲</u> ۲'	attaches to personal pronouns	'-self'
rãːmẽː/rõːmẽː	रूट'खतु'	used independently	'oneself'
rãːrãː soːsoː	নদানদা ঈার্মা	distributive use, typically co-occurs with a	'each oneself'
		noun, personal pronoun or indefinite	
		pronoun	

The same form $r\tilde{a}$: is also used as the mid-level second person singular pronoun, see §3.6.1. The reflexive $=r\tilde{a}$:/=ra has also grammaticalized into an anaphoric emphatic clitic, see §16.1.1. The uses of the reflexive pronouns are illustrated in §6.2.

3.6.3 Reciprocal pronouns

The three reciprocal pronominals occurring in my data are listed in Table 3.27.

Table 3.27. Reciprocal pronouns

Form			Gloss
tci:=ki, tci:(=lo)	যন্তিযা	'শীঝ' শৃন্ঠিশা'(র্নি')	'one to another' (lit. 'one to one')
tci:=ki zen(=lo)	ম্বন্ঠিমা	'শীম' শাৰ্বি'(র্মি')	'one to another' (used in Richhi instead of the first form)
p ^h entsỹ: ¹⁴⁶	শ্বৰ'স্ক্	ईव	'each other'

The forms from which the reciprocals are formed are the numeral tei? $\eta \in \eta'$ one' and the demonstrative $zen \eta \in \eta'$ other' (demonstrativity of zen is defined in opposition to something else determined by the context). The form $p^{hentshy}$: $\forall \eta \in \eta' \in \eta'$ also occurs in Written Tibetan with the meaning 'mutual, reciprocal, hither thither, each other'. The reciprocal pronouns are further illustrated in §6.2.

3.6.4 Indefinite pronouns

Indefinite pronouns are words that can replace a noun phrase (hence the term "pronoun") and refer to people, objects or places without exactly specifying the referent (hence the characterization indefinite). All indefinite pronouns listed in Table 3.28, except for the last two, can be further characterized as quantifying pronouns. In addition to independent uses, the indefinite pronouns are used as noun modifiers. The specific numeral $k'\tilde{a}:pu = \sqrt{r_s}$ one full measure of may follow $t^hamtee?$

¹⁴⁵ Neither does Dzongkha, another southern Tibetic language (van Driem 1998).

¹⁴⁶ The demonstrative expression $p^{h}a$: $ts^{h}u$: $q_{\overline{x}}$, $\overline{q}_{\overline{x}}$, 'thither hither' may be used in a similar sense.

Table 3.28. Indefinite pronouns

t ^h amtce? (k'ãːpu)	রঝঝান্তন্র (বাদান্থা)	'all, everyone'
kʰɛːlɛ (kˈãːpu)	यिंग्ये' (याद्र'सुं')	'all, everyone'
tɛʰaːlɛ (kˈãːpu)	ক্র.ল. <u>খ</u>	'all, everyone'
dzaŋki	<u>بر</u> - بیل 147	'all, everyone' (Lachung)
màŋtɕʰiɛo?	અમ્ રે રેવ્ય	'most' (includes the adjectival superlative ending -co?)
màŋtɕʰiţa?	ञन्दः क्वे:न्द्रयाह्य:	'most' (includes the adjectival ending -ta?)
khace?	M. 44.	'some(one)'
làri?	artar	'some(one)'
rere	<u> र</u> े. <u>र</u> े.	'each one'
ripi (ripi)	रे'गवि़्रा' (रे'गवि्रा')	'a few (people)' (lit. one-two)
ka:kut£i?	गा'गु'र्च्चिया'	'a few'
tci:pi:	যান্তিযা'যান্ত্রিম'	'a few' (lit. one-two)
làla…làla	ସଂସ' ସଂସ'	'someothers'
<i>ri, -ri</i> (also <i>rε</i>)	<u>र</u> ें	'one, each'

The independent uses of the indefinite pronouns are described in §6.3.1, whereas uses as noun modifiers are exemplified in §4.1.3.3.

3.6.5 Demonstratives

Demonstratives are deictic words which define a person, object or location in terms of its spatial relationship to the speaker. Demonstratives may be pronouns, pro-adjectives or pro-adverbs. With $z \in n$ and $z \in n$ and

Table 3.29.	Demonstrative roots
-------------	---------------------

14010 0.2	· Demons	
di	مثر:	proximal, 'this'
do-	٩Ť.	emphatic proximal, 'this right here'
<i>ó</i> -	હેંગ	distal, 'that'
nà:	व.	'here'
p ^h ou, p ^h i-	र्चतुः, चि	'over there'
jòu, jì-	ર્વેતુ, ધે	'up (there)'
mòu, mì-	ર્વેતુ', ત્રે'	'down (there)'
p ^h a(:)	শ্(ম)'	'over there, thither, on the other side'
ts ^h u(:)	థ్ర్(న)'	'here, hither'
zen	শাৰব'	'other'

Table 3.30 presents other demonstratives that derive from the roots of Table 3.29. The list of locative expressions in Table 3.30 is not exhaustive but only presents the forms which I have come across in my present data. Note the intensifying reduplication in locative forms. The first syllable of the reduplicated expressions is typically accompanied by a higher pitch.

¹⁴⁷ This written form given by consultant KUN is surprising in that it suggests pronunciation as *ts'aŋki* rather than *dzaŋki*.

Root		Derived object	Derived location
do-	emphatic proximal	dodi 'this right here'	dodikha, dona 'right here'
di	proximal	di 'this'	<i>dikha</i> 'here' (cf. <i>nà</i> : 'here')
<i>ó</i> -	distal	ódi 'that'	<i>óna, ók^ha</i> 'there'
nà:	'here'	$n\dot{a}=di$ 'the one here'	$nak^{h}a(=lo)$ 'here'
<i>р</i> ^{<i>h</i>} ои,	'over there'	p ^h ou=di, p ^h idi	p ^h ou=na, p ^h ouk ^h a, p ^h ok ^h a, p ^h ina, p ^h ik ^h a
$p^{h}i$ -		'that over there'	'over there'
			p ^h o:p ^h ina, p ^h o:p ^h ou=na, p ^h o:p ^h ouk ^h a
			'way over there
jòu,	'up (there)' <i>jòdi, jìdi</i>		<i>jìna, jòuk^ha</i> 'up (there)'
jì-		'that up there'	<i>jòːjina, jòina</i> 'way up (there)'
тòи,	'down (there)'	mòdi, mìdi	$m \partial u = na$, $m \partial u k^h a = lo$, $m \partial u = lo$, $m \partial na$
mì-		'that down there'	'down (there)'
			$m \hat{o}:mouk^h a = lo$ 'way down (there)'
p ^h a(:)	'further, on the	$p^{h}a = di$ 'the one	$p^{h}a:k^{h}a$ 'on the other side, further'
	other side'	thither/further'	
ts ^h u(:)	'closer, on this	$ts^h u = di$ 'the one	$ts^h u: k^h a$ 'on this side, nearer'
	side'	hither/closer'	
zen	'other'	$z \in n = di$ 'the other'	$z \in nk^h a (=lo)$ 'in another place'

 Table 3.30. Derived demonstratives

It is a moot point whether some of the forms in Table 3.30 should be considered lexemes or combinations of a root lexeme with a following demonstrative-emphatic clitic =di or a case clitic =na/=lo. Forms such as *jidi* 'that up there' and $p^{h}ina$ 'over there' are more lexeme-like than the semantically equivalent forms $j\partial u=di$ and $p^{h}ou=na$ because *ji*- and $p^{h}i$ - do not occur independently whereas $j\partial u$ and $p^{h}ou$ do. The forms $p^{h}idi$ and $p^{h}ina$ may be considered phonologically reduced, lexicalized variants of $p^{h}ou=di$ and $p^{h}ou=na$.

Denjongke also has forms with the double function of proadverb of manner and proadjective, see Table 3.31.

Table 3.31. Proadverbs of manner and proadjectives

dodem	पर्ने पर्नेग	'like this right here'
dɛː, dɛm, dɛp	מלן, מלמי, מלחי	proximal, cataphoric, 'like this', 'like that', 'such'
ódɛː, ódɛm, ódɛp	क्षेंप्तने, क्षेंपनेवा, क्षेंपनेन	distal, anaphoric, 'like this', 'like that', 'such'

In addition to the derivations presented above, the distal \dot{o} - can combine with $-ts^h \partial 2$ 'limit, as much as' to form the quantitative \dot{o} - $ts^h \partial 2/\dot{o}$ - $dz \partial 2$ $\breve{e} = \dot{c} \cdot \dot{c}$ 'that much/many' (cf. $k'adz \partial 2 = (\ddot{e} - \dot{c})^2 + (\dot{e} - \dot{c})^2$

Demonstrative forms are illustrated and further discussed in §6.4.

3.6.6 Question words

Denjongke question words are listed in Table 3.32.

k'an, k'ar	শ্ববৃ', শ্বস্	what
k'adi	ما.ريل.	which
k`ana, k`akʰa	या'त, या'य(र्र)'	where
ka	या.	who
nàm	ववा.	when
k'ambja ¹⁴⁸	শ্বন-র্ভূম্ম	why
k'adzø?, k'ats ^h ø?	या'र्केंट्र'	how many
k'atem	या'झेव्र'	what kind
k'ate	या झे	how
k'a: ¹⁴⁹	শ্বা	what, where, why (general interrogative, contextually
		interpreted)

Table 3.32. Question words

For examples on question words, consider §11.1.2. Question words are used as the basis for forming indefinite expressions with meanings such as 'whatever', 'wherever' and 'for whatever reason', see §6.3.2.

3.6.7 Numerals

Numerals are words that express exact numbers. Numerals can occur both as independent verbal arguments (3.93) or as noun modifiers (3.94). Note that in (3.93) the indefinite marker =tci? expresses the approximative meaning 'some'.

- (3.93) $\underbrace{\Re}_{\nabla} \underbrace{\Re}_{\nabla} \widehat{\Re}_{\nabla} \underbrace{\Re}_{\nabla} \widehat{\Re}_{\nabla} \underbrace{\Re}_{\nabla} \widehat{\Re}_{\nabla} \underbrace{\Re}_{\nabla} \widehat{\Re$
- (3.94) دَ هَمَّا مَرْجَحَ عَلَمَ عَلَى اللَّهُ (RS duetto) (RS duetto) المَعْنَا عَلَى اللَّهُ عَلَى الْحَلَى الْحَ الْحَلَى الْحَلَ

¹⁴⁸ Derives from *k'an* 'what' and *p'ja(-ti)* 'do(-NF)', also occurs as *k'amja, and k'am* ε and in the fuller converbal construction *kan p'ja-ti* [what do+NF].

¹⁴⁹ Direction from location can be questioned any 'from where'.

1 4010		0120			
1	tci?	শ্বন্থিযা'	11	teuktei?	यहुःग्विंग
2	ní:	যবিশ্ব	12	teuːpiː	নন্তু:শবিষ্ণ
3	súm	শশ্রুষ'	13	teuksum	ଅନ୍ତି.ସାହ୍ୟିକ.
4	zi	ଅନ୍ତି.	14	tsipzi	ત્રહુ'નંલે'
5	ŋá	સં	15	tɕɛːŋa	নষ্ঠ'শ্ৰ'
6	t`u?/t`uk	र्जुया'	16	teu:rup	पछ <u>ुः न</u> ुग
7	dyn	ननुबुः	17	tsopdyn	नर्हें ननुतुः
8	gɛ?	নক্তুই.	18	tcopge?	৸৾৾ঌ৾৾৾৽৸য়ৣঢ়
9	gu	ন্ন্যু	19	t£u:gu	न्द्रु:न्गु.
10	tɛu(tʰamba)	বন্তু (প্রঝ'ন্য')	20	pícu(tʰamba)	વે.સ.(દ્યય.ત.)

Table 3.33. Numbers 1-20

In counting numbers between 20 and 99, one morpheme (word) is used for full tens but another morpheme (clitic) is used to refer to the same full tens in the following nine digits, e.g. *picu* $3 \cdot 3$ 'twenty' but *tsa-tci?* $3 \cdot 3$ 'twenty-one' and *tsa-gu* $3 \cdot 5$ 'twenty-nine', where the dependent form *tsa-* $3 \cdot 3$ represents twenty in numbers 21-29. As shown in Table 3.34, many of the dependent forms marking full tens bear phonetic similarity to the corresponding numbers between two and nine (and 50 between 51 and 59 is segmentally identical with number five).

Table 3.34. Two forms for referring to full tens

			0		
20	nícu	<u> </u> હે.સે.	21-29	tsa-	₹
30	súmteu	गह्युव्य'नळु'	31-39	so-	र्वे
40	ziptsu	ସାିଶ୍ୟୁ ସଞ୍ଚ	41-49	sè-	e.
50	ŋáptsu	ମ୍ଚ.ଅହି.	51-59	ŋà-	5.
60	t'ukteu	<u>न</u> ुग'नळु'	61-69	re-	रे.
70	dynteu	ननुन-नन्हु	71-79	t'øn-	र्नेन'
80	getsu	নক্র্-নেন্তু:	81-89	kʻja-	मु.
90	guptsu	<i>ન્</i> ગુ'નજુ'	91-99	k'o-	र्वे.

In counting (as in 19, 20, 21) twenty-one is expressed as $tsa-tei? = 10^{-10}$, but when referring to one number in a sentence, for instance when telling one's age, a more complex form is typically used.¹⁵⁰ The complex form combines the two morphemes for tens, e.g. *picu tsa-tei?* $3 = 100^{-10}$ (lit. 'twenty twenty-one). The full forms of numbers 21-100 are given in Table 3.35 and 3.36.

¹⁵⁰ Consultant RL from Lachen counted *pi: tsa-tci?* '21', *pi: tsa-pi:* '22'.

21	pícu tsa-tci?	ने.स. ≆.गठ्या.	41	ziptsu sè-tsi?	ସଜ୍ୱି'ସ୍ୱଞ୍ଚୁ' ଜ୍ୱି'ସ୍ବୃତ୍ସିସ୍'
22	nícu tsa-ni:	वि.स. ஆ.गुथुर	42	ziptsu sè-ní:	ସଜ୍ୱି'ସଞ୍ଚୁ' ଜ୍ୱି'ସ୍ଟର୍ଟିଷ'
23	pícu tsa-sum	છે.ને. ≨.તાજીજા.	43	zipteu eè-súm	ସଜ୍ୱି'ସ୍ୱଞ୍ଗୁ' ଜ୍ୱି'ସ୍ୟୁଷ୍ଟ
24	pícu tsa-zi	છે.સે. ≨.ટાલુ.	44	ziptsu sè-zi	ସଜ୍ୱି'ସ୍ୱଞ୍ଗ' ଜ୍ୱି'ସଜ୍ୱି'
25	pícu tsa-ŋa	હે.સે. £.इ.	45	ziptsu sè-ŋá	ସଜ୍ୱି ସଞ୍ଚୁ ଜି:ଝ୍ର'
26	pícu tsa-ť u?	वि.स. ≨.रीग.	46	ziptsu sè-t'u?	ସଜ୍ୱି'ସଞ୍ଚୁ' ଜ୍ୱି'ଲ୍ରୁଣ୍'
27	pícu tsa-dyn	ંગે.નુ. જ્ઞ.નટ્વ.	47	ziptsu sè-dyn	ਧੱਕੇ ਸਨ੍ਹਾ ਕੇ ਸਨੂਰ
28	pícu tsa-ge?	৾ঀ৾৾৾ঀৢ৾৾৾৾৾ৼৼ৾৽য়ৢ৾৾৾৾৾	48	ziptsu sè-ge?	ସଜ୍ୱି'ସଞ୍ଚୁ' ଜ୍ୱି'ସକ୍ତୁମ୍'
29	pícu tsa-gu	કે.સ. ક્ર.રંગે.	49	ziptsu sè-gu	ସଜି'ସ୍ତୁ' ଜି'ମ୍ୟୁ'
30	súmteu	યાસુસ'નર્સુ'	50	ŋápteu	ମ୍ଚ
31	súmteu só-tei?	শব্জুম'নস্তু' র্ম'শ্বতিশ'	51	nápteu nà-tei?	শ্রু'নন্তু' দ'যন্তিযা'
32	súmteu só-pi:	শধ্যম'নস্তু' র্ম'শনিম'	52	nápteu nà-ni:	শ্রু'নন্তু' দ'শবিষ্ণ
33	súmteu só-sum	યાસુઅ'નર્સ્ટુ' સેં'યાસુઅ'	53	ŋáptɛu ŋà-sum	শ্র'নন্তু' দ'শধ্যুঝ'
34	súmteu so-zi	୩ଣ୍ଠୁୟ'ସଞ୍ଚୁ' ର୍ଶି'ସର୍ବି'	54	nápteu nà-zi	ഷ୍ଟମଷ୍ଡ ୮୮୩ଣ୍ଟି
35	súmteu só-ŋa	યાસુસ્ર'ગરુ: ર્સે'સ્ટ'	55	nápteu nà-na	ଞ୍ଜ୍ୟୁମ୍ବର୍ଷ୍ଣ ମାହୁ
36	súmteu só-ť u?	યાસુઅ'નરહુ' સેં'દુ્યા	56	nápteu nà-t`u?	ञ्च'नळु' म'मुग'
37	súmteu só-dyn	ષષ્ઠ્રચ'નજુ' ર્ષે'નનુવ'	57	ŋáptɛu ŋà-dyn	ञ्च'नरु' म'नमुतु'
38	súmteu só-ge?	য়য়ৢয়৾৾য়ড়ৢ৾৾৾য়৾য়য়ৢঢ়৾	58	nápteu nà-ge?	ञ्च [.] यहुः ८.नमु <u>र</u> ि
39	súmteu só-gu	यसिकायळे. यू.च्येस.	59	nápteu nà-gu	ञ्च'नहु' म'म्गु'
40	zipteu	ମଜ୍ୱି ମଞ୍ଚ	60	t'ukteu	<u>न</u> ुग'नरु'

Table 3.35. Decimal system numbers 21-60

Table 3.36. Decimal system numbers 61-100

61	t'ukteu re:-tei?	नुया'चळु' रे'यचिया'	81	getsu k'ja-tsi?	নদ্রুদ্র'নস্তু' ग্র্যু'শৃষ্ঠিশা'
62	ť ukteu rez-niz	जुया नरहु ने यकिषा	82	getcu k'ja-ni:	নদ্রুদ্র'নস্তু' ग্র্যু'শবিষ্ণ'
63	ť ukteu re:-sum	नुग'मन्हु' ने'गह्युत्र'	83	geteu k'ja-sum	പ्यमेटप्रह. गी.गश्रंश.
64	t'ukteu re:-zi	नुग'मळु' रे'मवि'	84	getsu k'ja-zi	ମକ୍ରୁମ୍'ମଷ୍ଡ' ଶ୍ର'ମଜି'
65	ť ukteu re:-ŋa	नुग'म्रहु' रे'झु'	85	getsu k'ja-ŋa	নক্র্র-'নন্ত্' ग্র'শ্র'
66	t'ukteu re:-t'u?	नुग'मळु' रे'नुग'	86	getsu k'ja-t'u?	৸য়ৢৢৢৢৢৢৢৢৢৢ৸ৢ
67	ť ukteu re:-dyn	डुग'नरु' रे'न5ुब'	87	geteu k'ja-dyn	નજીનું. મહે. મો. નરે થે.
68	t'ukteu re:-ge?	કુંગ'નરું' રે'નક્રુન્'	88	getsu k'ja-ge?	৸য়ৢৢৢৢৢৢৢৢৢৢ৸য়ৢৢৢৢৢৢৢৢৢৢৢৢৢৢৢৢৢ
69	ť ukteu re:-gu	नुग'मळु' रे'न्गु'	89	geteu k'ja-gu	৸য়ৢৢৢৢৢৢৢৢৢৢৢ৸য়ৢৢৢৢৢৢৢৢ
70	dyntsu	নদ্ব-নস্তু-	90	gupteu	<u> ન્</u> યુ'નજુ'
71	dynteu t'øn-tei?	यमुत्र यन्तुः मॅत्र यन्त्रियाः	91	gupteu k'o-tei?	न्यु'चळु' येॅायच्यि'
72	dynteu t'øn-pí:	यमुत्र यहुः मॅत्र यहिषा	92	gupteu k'o-ní:	न्यु'मङ् यें यक्रिय
73	dynteu t'øn-súm	नमुत्र नरुः मेंत्र गसुम्र	93	gupteu k'o-sum	ન્યુ'નજુ' યેં યજ્ઞુસ'
74	dynteu t'øn-zi	यमुब्र'यहु' मॅब्र'यबे'	94	gupteu k'o-zi	ମ୍ଣ୍ୟୁ'ମ୍ବଞ୍ଗ ସିଁ'ମ୍ବି'
75	dynteu t'øn-ŋá	नमुनु नर्छु मेंनु झु	95	gupteu k'o-ŋa	ન્યુ'નજુ' યેં'સ્ટ'
76	dyntɛu t'øn-t'u?	नमुनु नर्छु मेंनु मुग	96	gupteu k'o-t'u?	ન્ યુ'નજુ' થેં', <u>ન</u> ુય
77	dyntsu t`øn-dyn	ननुवानङुः र्नेवाननुवा	97	gupteu k'o-dyn	দ্যানন্তু যাঁন্দ্র
78	dynteu t'øn-ge?	৸ঢ়ৢঀ৾৾৾ঀড়ৢ৾৾৽৾ঀ৾ঀ৾৸য়ৢৢ৾৾ঀ	98	gupteu k'o-ge?	ઽૣਗ਼ૢૻઽૢૢૢૢૢૢૢૢૢૢૢૢૢૢૢઌૢૢૢૢૢૢઌૢ
79	dynteu t'øn-gu	नमुनु नर्डु मेनु मुगु	99	gupteu k'o-gu	न्यु'नङ् यें'न्यु'
80	geteu	৸য়ৢঢ়ৢ৾৾৾৾ঀড়ৢ	100	tsik gja, gja tsi?,	যহিঁযা' নদ্র্যু', নদ্র্যু' যহিঁযা',
				gja(t ^h amba)	নক্রুন্ ' রম'শ'

Examples of numbers from one hundred onwards are given in Table 3.37, which introduces the words $t\tilde{o}:(t^{h}a?) \cong (t^{h}a?)$ 'thousand', $t^{h}i \cong$ 'ten thousand', bum and 'hundred thousand' and sája www 'million'.

1 4010 5.57.1	2xumples of numbers from 100 onwa	105
100	teik gja, gja tei?, gja(tʰamba)	শৃহ্বিশ' নদ্রু', নদ্রু' শৃহ্বিশ', নদ্রুস্'প্রঝ'ম'
180	gja t'ã: gɛtɕu	79. 75. 79. 78.
200	ní: gja, gja ní:	ग्विंश नमु, नमु, गविंश
1000	teik tõ:, tõ:tʰaʔ teiʔ	य[ठेया' झॅंट', झॅंट'श्चया' यठिया'
2955	pí: tõ: gubgja ŋáptɛu ŋàgu	યવિશ્વા ક્ષેંદ્ર દેશી.તથી. સંત્યે દ. ટેલી.
10,000	teik tʰi, tʰi tei?, tõːtʰa? teu(tʰamba)	শৃষ্ঠিনা' দ্রি', দ্রি' শৃষ্ঠিনা', স্ট্রিন্-শ্রেশা' অন্তু'(প্রঝান্য')
100,000	bum tɕi?	तसुम्रा गर्छिया
160,000	bum tɕi? t`ãː tõːţa? kʰɛː súm ¹⁵¹	तत्रुमः चर्डियाः दत्तः झ्रेन्टःस्रवाः ायव्यः वश्त्रित्रः
1,000,000	sája tei?	षाया मर्चिमा

Table 3.37. Examples of numbers from 100 onwards

For other numerals, consider Table 3.38.

tõ: k'ã:pu	र्झेन्टः गन्दःसुः	'all'	lopo tõ: k'ã:pu الجَبْتَة الجَجْبَة (ilt. full the Lhopos' (lit. full
			thousand Lhopos)
p ^h i?	छेन्.	'half'	súm t ʿāː pʰiʔ གསུམ་ དང་ ཕྱོད་ 'three and half'
te ^h a	æ	'pair'	lấ: tɛʰa tɛiʔ म्नूह: क पहिंग: 'a pair of bulls'
k'ã:(pu)	बन्द.सु.	'(one) full'	tchum tsaŋku k'ã: कुबा रूट्गा गृत [rice container full] 'one
			container of rice'
			para k'ãː(pu) दाऱर यादर(दुर) 'fistful'
do/t'o	ŕ	'two'	botor do ži fr z' z' 'two bottles'

Table 3.38. Other numeral-related vocabulary

The vigesimal system illustrated in Table 3.39 is build around the number twenty, for which the word $k^{h} \varepsilon$ rear 'twenty, score' is used. The numbers before the next full score are expressed by adding a suitable number to the full score, e.g. fifty-five is 'two scores and fifteen'. The conjunct $t'\tilde{a}$ is used in conjoining the complex numerals.

¹⁵¹ For $k^h \varepsilon$, refer to the vigesimal system explained below.

20	k ^h ɛː tɕiʔ	ायला यार्ड्या	score-one
21	kʰɛː tɕiʔ (tʿãː) tɕiʔ	বিদ্রা যন্তিযা (८८८) যন্তিযা	score-one (and) one
25	kʰɛː tɕiʔ (tʿãː) ŋá	العاما ماتحما ((٦٦) 2	score-one (and) five
30	$k^h \varepsilon$: $t \varepsilon i$? ($t \cdot \tilde{a}$:) $t \varepsilon u(t^h amba)$	বিশ গঠিয়া (১৯.) এপ্ত (প্রমান)	score-one (and) ten
39	kʰɛː tɕi? tɕu tʿãː gu	ायन्भः यन्डियाः चन्छः मृमः मृत्युः	score-one ten and nine
40	k ^h ɛː ɲíː	म्भा यानुषा	score-two
41	kʰɛː ɲíː (tʿãː) tɕi?	المحمد ما وكما. (كركر) ما حكما.	score-two (and) one
55	kʰɛː ɲíː t ʿãː tɕɛːŋa	मिया यहिषा नूमा महुन्धा	score-two and fifteen
59	kʰɛː ɲíː tɕu tʿãː gu	ायत्य यानिषा यन्तुः नूमः नृगुः	score-two ten and nine
60	k ^h ɛː súm	यित्रा. यसिक.	score-three
80	k ^h ɛː zi	म्प्रिय रावि	score-four
100	teik gja, gja tei?,	यचिया' नक्तु', नक्तु' यचिया',	(one) hundred
	gja(tʰamba)	ন্দ্রু'দৃশ্ব'ন'	

Table 3.39. Examples of the vigesimal number system

I have not carried out a detailed study on the division of labour of the decimal and vigesimal systems, but I here offer some initial observations. In my data, the vigesimal system is used at least when talking about prices of items (3.95), age of people (3.96) and number of people (3.97).

- (3.95) $r \in \Re r : \Re q : q \in q : \Im q = \Im q : \Re q : \pi : t \in i ? p : in-c \tilde{i}:.$ 1SG thousand score one give-NPST.PER 'I will give twenty thousand (rupees).' (TB bulls story)
- (3.96) आहं मुषार्यें यें विया यवी केंट्र

 $\acute{a}dzo$ $g\varepsilon:po$ $l\grave{o}$ $k^h\varepsilon:$ zi- $ts^h\emptyset$? grandfather old.man year score four-about 'old grandfather of some eighty years' (Richhi 78)

(3.97) a) दुवापार पर्धेक्षा है। पवा पति नर हा छन्

 $k^{h}\varepsilon$: $g\tilde{a}$: dom-ti $k^{h}\varepsilon$: zi t' \tilde{a} : $\eta\dot{a}$ b ε ?. family gather-NF score four and five EQU.NE 'There are altogether 85 families.' (Richhi 48)

b) ଝ୍ରଂଗ୍ରଞ୍ଗ ୩ନ୍ତିଷଂଶ୍ୱ ୫ଂଗ୍ରଞ୍ଚଟ ସମ୍ପି ବିଶ୍ୱ ଶ୍ୱିଂ ସ୍ପିଷାର୍ଥି) ଶ୍ରିସଂମଧ୍ୟିର ଭିମ୍ବା

 $\eta \dot{a} bg ja$ $\eta \dot{i} cu$ $tsa:-g\epsilon?$ $d\epsilon: tci?$ lori=ki $l \dot{o} p \tilde{\epsilon}:$ five.hundredtwentytwenty-eightthus.muchLhopo.people=GENteacher $j \dot{o} ?.$ EX.PER'There are as many as 528 (language) teachers of the Lhopos.' (RD BLA9)

The decimal system is used for pointing out the year when something took place (3.98), the number of years since something happened (3.99) and dates (3.100)

- धुः र्ये गर्छ्या क्रेंट द्यापक्य पर्व पहर देव पक्ति वट शर्य (3.98)gubgja dynteu t'øn-ge? tchilo¹⁵² tcik-tõ: nànca=lo one-thousand nine.hundred seventy seventy-eight inside=DAT vear 'In 1978' (KT life story) (3.99) dak-ti døː-ti t'a lò t'a nà za:ts^hã: nícu now 1SG married.couple bind-NF stay-NF now year twenty $l\tilde{a}$:- $ts^{h}a$:=s. tsapd*ṽ:*-ts^hø? twenty.seven-about reach-CMPL=QUO 'Now it's about twenty-seven years since I established the tie of matrimony (I tell).' (KT life story)
- (3.100) क्रेंबा यानेवा मु दर्न के लोगाया

 $ts^{h}e:$ picu = di $mi-l\hat{e}?$. date twenty=DEMPH NEG-be.good 'The twentieth day is not good.' (RS astrology)

In addition to the *picu* $\widehat{\mathfrak{g}}$, $k^h \varepsilon$: rear and *tsa*- \mathfrak{F} , all meanings 'twenty', a fourth numeral $p \hat{\varepsilon} r \widehat{\mathfrak{g}} \tau$ is used for referring to twenty with respect to to dates of the month, i.e. the form is used for the 21-29th days of the month, e.g. $p \hat{\varepsilon} r = g u \widehat{\mathfrak{g}} \tau \widehat{\mathfrak{g}} \overline{\mathfrak{g}}$ 'the twenty ninth day of the month'.¹⁵³

(3.101) A: $\vec{\beta}$: $\exists av \vec{\beta} q \neg \vec{\alpha} \vec{\beta} \neg \vec{\alpha} \vec{\gamma}$, $\vec{\alpha} \vec{\gamma}$, $\vec{\alpha} \vec{\gamma}$, $\vec{\gamma} \vec{\eta} \vec{\alpha} \vec{\gamma}$, $\vec{\gamma} \vec{\eta} \vec{\gamma}$? $te \ r \hat{u} mte? \ gj \ mp \ gi = gi = di$ $n \ mo$,

te rùmte? gjømpø:=gi=di nàm mo, gjømpo nínku=gi? so TPN monastery.GEN=GEN=DEMPH when EQU.NE.Q monastery old.=GEN 'So when is the Rumtek monastery's (ceremony), the old monastery's?'

B: a_{7}^{2} $g_{7}^{2} = ra$ $b\epsilon^{2}$, dou $t\epsilon^{2} = gi$. this twenties.of.a.month=AEMPH EQU.NE month ten-ORD.GEN=GEN 'It's in the twenties (of the month), of the tenth month.' (KN kitchen discussion)

Numerals are typically used as noun modifiers (see §4.1.3.2), although they may also occur independently (see [4.166] in §4.4).

The suffix $-t^hamba$ can be added not only to round tens and hundreds, as described for Standard Tibetan by Tournadre & Dorje (2003), but also to small non-round numbers, see (3.102). The suffix

¹⁵² Lit. 'outsider years', referring to the western calender years. A more local way of referring to years is through animals associated with each year in the Tibetan astrological twelve-year cycle (e.g. year of the horse).

¹⁵³ The word *ts^he:* 'date of the month' is used for referring to the dates of the month up to the twentieth day, *ts^he: picu* 'day twenty'. After that, the specific numeral *p* \dot{r} 'twenties (of a month)' is used for referring to days 21-29, i.e. *p* \dot{r} *gu* 'the twenty ninth of the month'. The thirtieth day of the month is *námgã:*, which literally means 'black moon'.

 $-t^hamba$ has little other meaning than highlighting the presence of a numeral, hence the gloss NUM (numeral). It is particularly common with number ten, see (3.103).

(3.102) דָר הַמִּיֹחִן אָ הַיִּשְיַשְׁמִישִּאַי אָשָאַי אָשָאַי אָדָן *tɛ nè:=gi nà: p'apuzi súm-thamba jò?*. then ISG.GEN=GEN here children three-NUM EX.PER 'Then I have three children here.' (KT life story)

In addition to marking ordinal numbers, -po/bo can form collective nouns from numerals, see (3.104) and (3.105), where the noun preceding the numeral reveals what the collective entity consists of. The nominalized numeral signifies that the enumerated elements form one entity. This type of derivative use of -po/bo with numerals is here termed collectivizer (glossed COL).

(3.104) দ'ন্ডবা বারিমার্যা হ্লাবি'এবামা স্ল'বন বন্তনার্য জিন্। nàtea? **ní:-po** bhaila ta-wa tea:-bo í:. 1PL two-COL PN look-PUR come.HUM-2INF EQU.PER 'The two of us came to see Bhaila.' (Richhi 11)

(3.105) क्षें वर्दे हे रुग हार्य वर्दे

ódi tiru? **ŋá-po**=*di* that rupee five-COL=DEMPH 'those five rupees' (RS pupil joke)

Numerals referring to 2-9 people may be marked by another collectivizer, the suffix -ga.

(3.106)	*nè:	p'um	tci-ga	jờ?.	
	рè:	p`um	níŋ-ga	jờ?.	'I have two daughters.'
	nè:	p'um	súm-ga	jờ?.	'I have three daughters.'
	nè:	p'um	zi-ga	jờ?.	'I have four daughters.'
	nè:	p'um	gu-ga	jờ?.	'I have nine daughters.' (KN e)
	*pè:	p'um	tcu-ga	jờ?.	
	1SG.GEN	girl	NUM-COL	EX.PER	

The suffix $-ts^h @? \notin_{\nabla}$ 'about, some, measure of' (from WT \notin_{∇} *tshod* 'measure', which is in meaning very similar to WT \notin_{∇} *tshad* 'measure') attaches to numerals and marks an inexact amount. For examples, refer to (3.99) above and §4.4 below. The restrictive suffix *-ma*, which attaches to numerals and quantifying words and which is accompanied by a negated existential, is described in §10.3.2.

3.6.8 Postpositions

In Denjongke, words marking various spatio-temporal and abstract relations such as 'on', 'after' and 'according to' are here termed postpositions, which underlines the syntactic fact that the word expressing the relation to the noun occurs after the nominal. Some of the postpositions bear more noun-like characteristics and can be described as relator nouns (RN), whereas for other postpositions such a characterization is less fitting. Thus, RNs are here considered a subcategory of postpositions. RNs are historical nouns which through frequent use in relational contexts have shed some of their noun-like characteristics.¹⁵⁴ According to Aissen (1987:11) "[t]he term relational noun comes from Mayan grammatical theory and refers to a set of obligatorily possessed noun stems which denote grammatical or thematic relations. The nominal which actually bears the relation functions as genitive of the relational noun." Those postpositions in Denjongke which most resemble Aissen's (1987: 11) definition of relational nouns (=relator nouns), however, show further grammaticalization from Aissen definition in that Denjongke RNs are not obligatorily possessed, i.e. the nominal which bears the relation function is not always genitive marked.

Table 3.40 lists nominal roots from which postpositions derive, while postpositions are listed in Table 3.41. The suffixes which have been added to the root forms in Table 3.41 are *-tar* (WT/WD $rac{} a r$) 'according to', *-zin* (WT/WD $rac{} a r$) 'likeness, similar to', and the locative markers *-k^ha* (locative suffix), *=lo* (dative-locative case), *=le* (ablative case) and *=na* (locative case). Although the postpositions in Table 3.41 are analyzed into their constitutive parts, the combinations have through frequent use grammaticalized towards lexicalization.

Noun ro	ot	Gloss
nàŋ	वर्ट.	'inside(s)'
t'øn	र्नेव'	'meaning, purpose'
ts ^h ap	র্জন'	'replacement'
tsa	₹	'root, base'
t ^h o?	র্ষনা.	'roof'
go	অর্য্য	'head'
kor	ب ېخ.	'neighborhood'
no:	Ϋ́Ν.	'side, direction'
gjap	ক্রুন'	'back'
dze:	È.™.	'trace, trail'
dyn	षम्तुत्र.	'front'
p'a:	নন'	'(space in) between'
nén	र्धेव.	'earlier (time)'
teŋ	R-	'top'
ò?	র্বৈশ্ব'	'low(er) place'
bu(:)	AT.	'middle'
bolo?	۲٩٠ مَتْ سَمَّ ¹⁵⁵	'close'

Table 3.40 Nominal roots used in postpositions listed in Table 3.41

¹⁵⁵ In Dzongkha, the cognate is written জ্রান্বাব্য.

¹⁵⁴ For a general discussion on relator nouns, see Starosta (1985). For particular languages, see DeLancey (1997b) for Classical and Lhasa Tibetan, Huber (2002: 112) for Kyirong Tibetan, Watters (2002: 137) for Kham (Tibeto-Burman, Nepal), Daudey (2014: 160-161) for Wangdu Pumi (Tibeto-Burman, China) and Watters (2018: 198ff) for Dzongkha (Tibetic, Bhutan).

Postposition		Gloss	
nàŋ-tar, nàŋ-zin	वट्रा झ्रे. वट्र पविवर	'according to, similar to'	
t'øn-zin(gi)	र्देव'पविव'(ग्रीय')	'according to,	
		in accordance with, in view of'	
(t'ãː) pámtɛi?,	(८८) अनुसः छेया,	'with'	
(ť či) pámpu	(८८) बालुखासु		
$s\dot{\tilde{a}}$: $t\varepsilon$, $=s\tilde{a}$:	<u> </u>	'until'	
mèmbo ¹⁵⁶	মবৃ'র্ন'	'except'	
<i>mèntã:</i> (rare)	য়ঀ৾৾৾য়ঢ়৾৾৾	'except'	
$ts^hap=lo, ts^hama(=lo)$	র্চন'র্নি', র্চ্রন'রে'(র্নি')	'instead of'	
$tsa-k^ha, tsa:=lo, =tsa:$	₹'ष(म)', इ'й', इ(म)'	'at, by, with'	
t 'øn=lo, t 'øn=l ε ,	र्देवार्थें, देवायबा,	'for (the purpose of)'	
t'ønda=lo, t'ønda=le	र्नेव न्या यें, नेव न्या अषा		
$t^h o:= l\varepsilon$	র্ষনা নামা	'through, via, on the basis of, by'	
$(=l\varepsilon)$ gjy:ti	(ગય.) ર્શુટ્ટ કે.	'through, via, by'	
$go:=l\varepsilon$	ন্য্র্যা এক্ষা	'from, through'	
$kor=lo, kor=l\varepsilon$	ૻ૾ૣૣૼૣૻઌૼૻૻ, ૻ૾ૣૢૼૼૣૻઌૹૻ	'about'	
ηο:=lo/hõ:=lo, ηο:tε/hõ:tε	ર્દેષાર્યેં, દેંષાસ્ટ્રેં	'toward, in the direction of'	
	Karaa.	'from the direction/side of, through'	
gjab=lo, gjab=lɛ	য়ৢৢৢৢৢৢৢৢৢৢৢৢৢৢৢৢৢৢৢৢৢৢ	'behind, after'	
$dz \varepsilon = lo^{157}$	हेरा'र्ये'	'after'	
$(kum)dyn-k^ha, (kum)d\tilde{y}:=lo,$	(भ्रु')अनुव'ाप(र)', (भ्रु')अनुव'र्थे',	'in front of,	
kumdỹ:	भु'व्यन्त्व.	in the presence of'	
p'a = na	न्नर'व'	'between'	
pén=lo, pén=lɛ, pǿma,	ર્થેવ ખેં', ર્થેવ બષા, ર્થેવ યા,	'before'	
hen=le, hema	हन्यया, हन्या		
$t \in \eta - k^h a, t \in \eta = lo$	ह्येन्ग्य(म), ह्येन्ग्ये	'above, on'	
$=gu, gu=lo^{158}$	व्यगु'(त्रॅ') ^{,159}	'above, on'	
$\dot{o} = lo, \dot{o} = l\varepsilon$	र्तेग'र्भे', तेंग'यर्भ'	'below'	
bu:=na, bu:=lo	श्च न, श्च में	'in the middle of'	
$bolo(-k^ha), bolo=lo$	तर्रे र्येषा (षर), तर्रे र्येषा ये	'next to'	
$n a \eta \epsilon a (= lo), n a \eta = lo,$	वृत्तः वृः (याँ), वृत्तः याँ,	'inside'	
nàŋ=na	वन्त.		
p ^h ilo?	र्धु र्त्यम	'outside'	
paŋk ^h a	ふり、(方)、	'outside'	

Table 3.41. Postpositions

¹⁵⁶ Occurs in a negated or interrogated clause. ¹⁵⁷ This word from Tibetan is mainly used by literate people with monastic training. ¹⁵⁸ This form occurs only in data from Martam (East Sikkim). The innovative written form and used here suggests an origin with and 'head' and a vowel change from o to u. A cognate form =gu is used in Dzongkha as a "contact locative" (Watters 2018: 200). ¹⁵⁹ Experimental spelling reflecting the hypothesized origin go 'head' (WD and 'mgo).

The ability of the complement noun to occur in the genitive can be used as a delineating criterion for distinguishing RN (with which the noun complement may occur either with or without genitive marking) from other postpositions (with which the noun complement cannot occur in the genitive). Following this criterion, the seven first items in Table 3.40 (*nàŋtar, nàŋzin, t'ønzin(gi), pámtɛi?, pámpu, sã:tɛ, =sã:*) are postpositions which are not relator nouns whereas the rest are postpositions which are relator nouns. Note that most of the relator nouns have several forms, depending on which locational suffix or case clitic is used in their formation (e.g. *nàŋɛa/nàŋlo* 'inside', *nàŋlɛ* 'from inside').

Example (3.107) exemplifies a postposition which is an RN, because it may occur with a genitive-marked nominal complement (3.107a). The construction in (3.107b) can be considered a more grammaticalized use because there is no genitive marking. Note that the RN is in glossing treated like the other (non-RN-like) postpositions and not like a noun, hence the gloss 'after' rather than [back=ABL].

(3.107) a) র্ন্য ব্যন্তিবা'নী' ক্রুন্স'এম্বা'

lò tci:=ki gjablɛ year one=GEN after 'one year later' (SGD wedding customs)

b) क्रें ग्रेंचा कुन्य का *jim tei? gjable* day one after 'after one day' (KT animal story)

Example (3.108), on the other hand, illustrates a non-RN postposition. With these postpositions the complement noun is not genitivized.

(3.108) र्रेगुर्लार्ड्र अनुसासु

ro:=tsu pàmpu friend=PL with 'with friends' (TB phone call)

The remainder of this section provides a more detailed description on the characteristics of relator nouns. The latter part of the discussion points out the morpho-phonolological factor which conditions whether the complement noun of a relator noun (RN) is genitive marked or not.

Rarely, and exclusively in the spoken language, RNs occur in relational context in their bare monosyllabic root form:

(3.109) দ্রিমা বাদ্ধারা ক্রানা

nìm súm gjap day three back 'After three days...' (TB funeral customs) (3.110) $\neg \check{m} \check{a} \check{\gamma} \check{\gamma} \check{\gamma}$ $t'a \quad \acute{o}(d)i \ ten$ now that top 'Now above (=in addition to) that...' (SGD wedding customs)

Typically RNs occur with additional spatial (dative-locative case =lo, ablative case = $l\varepsilon$, locative suffix - $k^h a$) or genitive marking (=ki/gi).

 $t\varepsilon$ ódi $t\varepsilon y k^h a = lo$ the that above=DAT 'then above (in additional to) that...' (CY interview)

(3.112) ଦିଁ ଅଂସ୍ଟ୍ର(ମ୍ୱି) ଶ୍ରୁସଂୟଷଂ

lò ŋáptcu(=*gi*) *gjablɛ* year fifty(=GEN) after 'after fifty years' (KN e)

(3.113) तन्त्रस'र्ब्येन्स' वृत्त'यी' यानुस'क्रुस'र्दु'

dendzõ: nàŋ=gi tamgy:=tsu Sikkim inside=GEN legend=PL 'legends within Sikkim' (SGD cave story)

In (3.112), the grammaticalization of the ablative form of gjap 'back' towards being a postposition (cum adverb) gjable is signified by the lack of clitic-marking = and the postpositional gloss 'after'. In (3.111), the dative-locative in $tegk^ha=lo$ is separately marked, because =lo is an optional emphatic locative element, $tegk^ha$ itself already meaning 'above'.

RNs have four features that distinguish them from typical nouns. One feature of grammaticalization towards being a postposition is semantic bleaching, where the meaning of a noun becomes less literal and more abstract (see DeLancey 1997b: 56). For instance, the dativelocative and ablative forms of the word gjap 'back', $g_{jab} = lo/g_{jab} = l\varepsilon$, have through frequent use obtained the abstract meaning 'after, behind', whereas the literal meaning 'in the back (of a human or an animal)' has become to be marked with the less productive locative suffix $-k^h a$, $g_{japk^h a}$. Another non-noun-like feature of RNs is the ability to have a non-case-marked complement/modifier, whereas noun modifiers are typically genitive marked, e.g. $\epsilon \dot{a}: lo(=ki) t \epsilon \eta k^h a$ [bamboo.slit.wall(=GEN) on] 'on the bamboo wall'. A third more postpositional than nominal feature of RNs is the ability to occur with a derivational suffix which does not occur with other nouns. For instance, the RN *này* 'inside' occurs in colloquial language with the formative $-\epsilon a^{160}$ (nànea 'inside'), which does not occur with other nouns and whose origin is unclear (written language prefers the more typical nominal ending =lo, as in $na\eta = lo$ 'inside'). A fourth feature of RNs is that their independent uses in non-relational contexts are rare. For instance, $t \epsilon \eta / t \tilde{\epsilon}$: 'top, upper surface' rarely occurs outside the relational constructions with additional marking $(t \in \eta k^h a / t \in \eta = lo$ 'above', $t \in \eta = l \in$ 'from above', $t \in \eta = gi$ 'of above').

¹⁶⁰ Written and *shar* in the novel Richhi.

RNs also have noun like features. Locative RNs inflect for the dative-locative, ablative and genitive case, e.g. nayea/naylo 'inside' also occurs in the ablative form nay=le 'from inside' and genitive nay=gi 'of the inside'. Another noun-like feature is the ability to have genitive-marked nominal modifiers/complements, e.g. ea:lo=ki tey=lo 'on the bamboo slit wall'. RNs also have adverbial characteristics. The locative adverbial suffix $-k^ha$ is used in some RNs in variation with the dative-locative case marker =lo, e.g. $teyk^ha$, tey=lo 'on, above', $dyyk^ha$, dyn=lo 'in front of'. Most of the RNs also have independent uses as adverbs, e.g. penle '(in) earlier (times)', gjable 'afterwards', nayea 'inside'.

A study of the most frequent RNs in the novel Richhi, ten = c, 'above' (including forms $tenk^ha$, ten = lo, ten = gi, altogether 81 instances) and nan = c, 'inside' (including nan = lo, nan = le, nan = gi, altogether 74 instances) yields the following result concerning genitive marking: those complements/modifiers which allow short genitive marking through vowel modification (glide or fronting) and, consequently, do not add an extra syllable are genitive marked, whereas those complements/modifiers for which genitive marking would require adding a syllable (=ki/gi) are not genitive marked. In other words, typically only the short genitive is marked. Constructions with the short genitive are illustrated in (3.114-116).

(3.114) LA BL'A

pè: $t \in \eta = lo$ 1SG.GEN top=DAT 'on me' (Richhi 63)

(3.115) xx-362. 35-1418

 $k^{h}\tilde{o}:=tsu=i$ $n \partial \eta = l\varepsilon$ 3PL=PL=GEN inside=ABL 'from among them' (Richhi 45)

(3.116) क्वॅनःगुतेः वृत्तःगेः कुःमि

 $l \dot{b} d \phi$: $n \dot{a} \eta = g i g j a t^h i$ school.GEN inside=GEN chair 'chair from inside the school' (Richhi 88)

For the non-genitive-marked constructions in Richhi, consider (3.117-119).

(3.117) न्या र्वेया हेन्ट र्वे

 $\epsilon \dot{a}: lo?$ $t \epsilon \eta = lo$ bamboo.wall top=DAT 'on the bamboo wall' (Richhi 98)

(3.118) अर्कें'तर्षेत्र' वृद्त'र्थे'

ts^hokor nàŋ=lo pond inside=DAT 'in the pond' (Richhi 32) (3.119) $\check{a}(\hat{a}, \hat{b})$ $\check{a}(\hat{a}, \hat{b})$ $\check{a}(\hat{a}, \hat{b})$ $\check{a}(\hat{a}, \hat{b})$ $m\dot{u}=i$ $k^{h}im$ $n\dot{a}\eta=gi$ $l\dot{o}gju$? 3SGF=GEN house inside=GEN story 'news from her home' (Richhi 95)

Two exceptions were found in which the long genitive form was used: $k^{h}\epsilon p = ki \ t\epsilon\eta k^{h}a \ and \ be not the long genitive form was omitted, <math>k\epsilon \ t\epsilon\eta = lo \ a\eta \ a\eta \ a\eta \ be not \ barbor \ bar$

The observation made on the basis of the novel Richhi is confirmed by my spoken data. Out of the 714 instances of modifiers/complements for the forms nangea, nanglo, nangle and nang=gi only 9 are marked with the longer genitive form -ki/gi. The other instances where the genitive would obtain the longer form are not marked for genitive. On the other hand, those vowel-final complements/modifiers which can occur with short genitive marking are either genitive marked or non-marked. For instance, the noun *lóbda* 'school', which can occur in short genitive, occurs in my corpus both as non-marked (*lóbda nangea*) and as genitive (*lóbdø: nangea*).

The general principle applies that RN complements may be genitive marked unless marking the genitive increases morphophonological complexity by adding an extra syllable. For an illustration, consider (3.120) and (3.121), both used in the same story by the same speaker. The word meaning 'foot' has two forms, $k\tilde{a}:po$ π_{\Box} , and $ka:m_{\Xi}$. With the first one, the (short) genitive is formed by vowel modification, $k\tilde{a}:po$:, whereas the latter uses the longer form =ki/gi. The word which allows short genitive marking is genitive marked (3.120), but the other word, which would occur with the longer genitive, remains non-marked (3.121).

(3.120) ਤੋਂ ਨੂਸ ਸ਼ਾ ਦ ਦ ਦ

p'itcuŋ¹⁶¹ kã:pø: nàŋɛa bird foot.GEN inside 'in the bird's foot' (PD bet story)

(3.121) र्चे रुद्र गे मूट्या क्रेट विर

p'itcuŋ=gi ka:m tɛŋkʰa bird=GEN foot above 'in (lit. on) the bird's foot' (PD bet story)

If the RN complement has an indefiniteness marker, no genitive marking is typically used.

(3.122) ८ सि. रे.मु. बाह्या. केंटावाया रे. टेर्गटाहीया.

t'a $k^h u$ *rubi=tei?* $t \in \eta k^h a = lo pj\tilde{a}$:-*diki* now 3SGM climber=INDF on=DAT hang-NF 'Now, hanging in a climber plant...' (KTL animal story)

The use of postpositions and postposition phrases as adverbials are further exemplified in §5.6.2.

¹⁶¹ Interestingly, the complement of the RN ($k\tilde{a}:p\sigma$) is genitive marked whereas the nominal modifier of $k\tilde{a}:p\sigma$ 'foot', *p*'*itcuŋ* 'bird' remains non-marked. In example (4.22), however, the modifier if genitive-marked. This speaker may have a tendency to avoid two contiguous non-marked modifiers/complements.

3.6.9 Discourse connectives

This class of words is termed discourse connectives (similarly Coupe 2007 for Monsen Ao) rather than conjunctions, because the word conjunction suggests the occurrence of coordination, in which two clauses are combined into one sentence. In Denjongke, however, it is not at all clear that the connected clauses would form one sentence, see §12. The term "discourse connective" applies here both to clause-combining and sentence-combining cohesion-adding words.

t'izã:	र्ने'चर्त्र'	'but'
โะruŋ	ક્ષેવ'રુદ્ર'	'but, however', lit. 'although it is (so)' [EQU-CONC]
ínajã:	ક્ષેવ'વ'ਘ∽'	'but, however', lit. 'even if it is (so)' [EQU-COND-still]
índã:jã:	क्षेत्र'नूम्'ਘम्'	'but, however' lit. 'it is (so) and yet' [EQU-and-still]
ĩ:jã:	ਲੇਕ੍ਰਾਘ~ਾ	'but, however', lit. 'it is (so) yet' [EQU-still]
zenne	यालव 'वे'	'otherwise' lit. [other-COND] or [other-TOP]
zẽ:mene	यालव खव वे'	'otherwise', lit. 'if there is nothing else' [other-NEG.EQU-COND]
jầ	ど て、	'and, (then) again, still'
jầ:ne	धन्दाने.	'or'
jã:mene	धम्म अत्र ते	'or in other case, otherwise', lit. 'and if it is not (so)' [and-
		NEG.EQU-COND]
mi-tshe?	ঈ'র্চ্ব'	'not only (but also); moreover; in addition ', lit. 'not-stop(ping)'
k'ambjasene ¹⁶²	गव-धुग्र-गे-वे-	'because, this is for the reason that', lit. 'if told why' [why-say-
		COND]
dilɛ	तर्ने लाहा	'then' (lit. 'this=ABL', probably already lexicalized)
t'ene	ने'क्श'	'then, in that case' (often collocates with $t\varepsilon$ 'well, then', as in $t\varepsilon$
		t ' $\epsilon n \epsilon$ 'well then')

Table 3.42. Clause/sentence connectors

All the discourse connectors in Table 3.42 connect finite clauses and therefore, they are described, with two exceptions, in §12, which describes how finite clauses are connected. The first exception is *mi-tshe2* 'moreover, not only', which occurs both as a looser connector of finite clauses and an additive adverbial clause marker (see §12 and §15.9.1 respectively). The second exception is *k'amjasene* 'because', which is functionally similar to formally-subordinated causal adverbial clauses and is therefore described along with the functionally similar clauses in §15.4.

3.6.10 Interjections

Interjections are words that comprise an utterance in themselves. They are often phonologically distinct (Schachter & Shopen 2007: 57) and usually express the speaker's spontaneous emotions and reactions to something they have experienced or heard. Interjections usually occur at the beginning of a clause and are often followed by a pause. Phonologically distinctive characteristics of interjections are emphatically long vowels, the use of diphthongs /ɛi/ and /ai/, which do not otherwise occur in word roots (but do occur in the genitive forms, e.g. kei $\frac{1}{2}$ of neck', sai $\frac{1}{2}$ of

¹⁶² The first part 'why' in this word may be pronounced k'amja, k'ambja or k'amp'ja, depending on the level of phonological reduction. The last pronunciation k'amp'ja suggests a succession of words rather than a single word, because p' typically only occurs word-initially. The word of speaking $s\dot{\epsilon}$ (which may be pronounced si) can be replaced by lap 'say' or ϵu 'say (hum.)', e.g. $k'amjalapn\epsilon$, $k'amjacun\epsilon$. The last syllable, which is a conditional marker, may also take the forms *-no* and *-na*, the latter of which is probably affected by Tibetan spelling, e.g. k'amjaseno (eastern and northern pronunciation), k'amjasena (literary pronunciation).

ground') and the otherwise non-occurring final consonant $/p^h/[f]$ in if_{MS} (discomfort). Table 3.43 lists some Denjongke interjections. The order of items follows loosely the order of positive-neutral-negative. The interjections of surprise are neutral in that they may involve either positive or negative emotions.

Form		Gloss	Function	
áme:	લ્ય'એ'	'wow'	expressing enthrallment, rapture	
làso	এবাঝ'র্মি'	'okay'	approval	
ák(ʰ)aː	জ'ম'	'gee, gosh'	surprise, amazement	
ádzei	<i>ਕ</i> ਾਵੈਂ'	'gee, gosh'	surprise, amazement, quite similar to <i>ák^ha</i> :	
ójε, ói	લેં' બે', લેંતે'	'oh, hey'	used for getting someone's attention	
έː, jàː	જો', પ્પ'	ʻoh'	expresses engagement or surprise when listening,	
			keeps the conversation going	
á:	GN.	'hey',	1) informal address to get someone's attention	
		'yes',	2) non-honorific response to being called, addressed to social	
		'what?'	inferior (corresponds to honorific <i>la:</i> below)	
là:	ন্যব্যম্ব	'yes'	1) polite response to being called	
		'excuse	2) expressing that the speaker did not hear or understand	
		me?'	what was said ¹⁶³	
kei	ਸ਼੍ਰੋ', ਸ਼੍ਰੈ'	'O (voc.)'	honorific address (e.g. 'O lama')	
tc ^h a	æ .	'alas'	disappointment (e.g. after a bad shot in a game of carrom or	
			kerembot), loan from Nepali	
t€ ^h E:	ਲੇ.	'ah, oh no'	disapproval, discomfort (e.g. when someone does not answer	
			phone), the response to being tickled	
úf	জীন্ব.	'phew'	expression of pain or discomfort	
ádzi:	क्ष'हें'	'oh no'	1) (unpleasant) surprise, 2) fear	
áijo:	ક્ષે'ર્ધે'	'ouch'	pain or fear of pain	
á(i)ja:	જ્ઞે'બ્રૂ', જ્ઞ'બ'			
óho:	ði F.	'oh no',	sadness, response to bad news (KT)	
		'gosh'	(neutral) surprise at information (KUN)	

Table 3.43. Some interjections

For examples of the interjections, consider §11.2.3.

3.6.11 Discourse particles

Two monosyllabic discourse particles, t'a 'now' (WD (\overline{f}, da)) and $t\varepsilon$ (WD (\overline{f}, te)) 'then, so', are frequently used in spoken Denjongke. The two particles may co-occur, occur independently and occur more than once in a clause. For an illustrative example, consider (3.123).

(3.123) हे. नुर्हें हे. नु वरहन्य हे नु की

 $t\varepsilon$ t'ato $t\varepsilon$ t'a làtc^huŋ=la $t\varepsilon$ t'a lí=di so now so now TPN=DAT so now apple=DEMPH 'So now in Lachung apples...' (LA intro to Lachung)

¹⁶³ Likely to be frequently heard by a language learner.

As seen in (3.123), t'a and $t\varepsilon$ can be used as fillers, when the speaker is not yet sure what to say. For a more detailed description of t'a and $t\varepsilon$, refer to §16.4

3.7 Clitics

Denjongke clitics are here divided into case clitics (§3.7.1), emphatic clitics (§3.7.2), clausal clitics (§3.7.3) and other clitics (§3.7.4).

3.7.1 Case clitics

This section first introduces the case clitics 164 (§3.7.1.1), then discusses monosyllabic postpositions, which resemble case-markers (§3.7.1.2), and lastly addresses case-stacking (§3.7.1.3).

3.7.1.1 Introduction to case clitics

Case clitics are here termed clitics because of their transcategoriality, i.e. the ability to occur with more than one word class.¹⁶⁵ Within the noun phrase, case clitics attach to the last word, which may be a noun, adjective, demonstrative, numeral or a quantifier. Some cases also occur attached to adverbs and a few verbal suffixes (see example [3.4] above). Cases may be divided into grammatical cases (G in Table 3.44) and spatial cases (S in Table 3.44). Grammatical cases express syntactic relations (agent, patient, recipient, possessor in possessive constructions), whereas spatial cases express location and, by extension, time. The dative-locative case inhabits both categories, as suggested by its name and shown in Table 3.44.

Case name	Form	Gram./Spat.	Origin	Functions described in
agentive	= ki/gihigh pitch ¹⁶⁶	G	WT ग्रे [.] - <i>kyis</i> , जेल - <i>gis</i> , ग्रेल -gyis, ब - <i>s</i> , ùब - <i>yis</i>	§5.1-3, §15.4.6
genitive	=ki/gi/i	G	WT ग़ु' -kyi, ग़ु' -gyi, गे' -gi, ते' -'i, थे' -yi	\$3.6.8, \$5.4.3, \$15.8.6
dative- locative	=lo	G and S	WT at la (dative) (?)	\$5.3.2, \$5.4.3 \$5.6.1.1
ablative	=lɛ	S	WT any las (elative)	\$5.6.1.3, \$15.3.1.1, \$15.4.3
locative	=na	S	WT ज्' na (locative)	§5.6.1.2

Table 3.44. Case clitics

¹⁶⁴ The word "case" derives from Latin *casus* 'fall(ing)', a loan from Greek *ptõsis* 'fall(ing)' (Haspelmath 2009: 506). Thus, the word suggests a falling away from the standard citation form (Blake 1994: 19).

¹⁶⁵ For transcategoriality of Classical Tibetan cases, see Tournadre (2010). The application of the term "case" to clitics differs from the Sanskrit, Greek and Latin based tradition where case is a word-level morphological category. In Denjongke, case functions on the phrasal level.

¹⁶⁶ In agentive personal pronouns $\eta \dot{a}$: (vs. $\eta \dot{a}$) and $m \dot{u}$: (vs. $m \dot{u}$).

It seems that whereas Classical Tibetan is, in Stassen's (2000) terminology, a WITH-language, which does not make a clear distinction between comitative and coordinative uses, Denjongke has developed towards being an AND-language, which tends to distinguish coordinative (marked with $t'\tilde{a}$) and comitative constructions (marked with *pámpu*).

The main functions of the grammatical and spatial case clitics are summarized here with references to relevant sections in later chapters. The **agentive** marker =ki/gi marks the animate (agentive) or inanimate (instrumental) causer of verbal action. Its use is partly syntactically and partly semantico-pragmatically governed. With nominals ending in a vowel, the agentive case may, alternatively, be marked by lengthening of the vowel and by high tone, e.g. ηa 'I', $\eta a := gi$ 'I=AGT', $\eta a := gi$ 'I=AGT', n arks' 'I.AGT'; n arks' horbu', n arks' = gi 'Norbu=AGT', n arks' 'Norbu.AGT'. The use of the agentive in clausal argument marking is discussed in §5.2 (intransitive clauses) and §5.3 (transitive clauses).¹⁶⁹

The **genitive** marker has two forms, longer =ki/gi, which is homophonous with the agentive case, and shorter =i. The shorter form can only be attached to stems ending in a short vowel. The genitive =i following final -a or -o in disyllabic words merges into -o: (or e:). However, the longer form may also be used with vocalically ending nouns, especially in spoken language. The main function of the genitive case is to mark a nominal or a clause as a modifier of a nominal, see §4.1.2.2. Although there is considerable variation in the voicing of the velar plosive in -ki/gi in both the agentive and genitive, in the examples of this grammar =gi follows voiced consonants and

¹⁶⁷ In sentential context typically pronounced $d\tilde{a}$:.

¹⁶⁸ The term "associative" is in the present work used differently. It refers to the use of suffix *-po* when it occurs with proper names or place names with the meaning 'x and his/her associates' or 'person from x' respectively.

¹⁶⁹ At present it is unclear, whether agentive marking by lengthing the vowel in disyllabic words such as *nòrbu* is only part of the reading-style pronunciation, and hence used only by literate speakers, or whether it is also used by non-literate speakers. The reason why the reading-style pronunciation is motivated to lengthen the vowel is that the agentive case in words ending in a vowel may in writing be marked, in addition to full -ki/gi, by final letter $-s \propto v$, which prompts a response from the reader to distinguish the form from the nominative. Another way to pronounce the final $-s \propto v$ in reading is to produce, in harmony with Tibetan spelling rules, a fronted vowel, e.g. /u/ > /y:/, /o > /ø:/, although in Denjongke cognates Written Tibetan final $-s \propto v$ do not, most of the time, cause vowel fronting, e.g. WT $q \propto lus > /lu:/$ 'remain'.

vowels, whereas =ki follows voiceless consonants, including an underlying glottal stop (e.g. $tc^h \phi$? $\mathfrak{s}_{\overline{\uparrow}}$ 'you' > $tc^h \phi$:= $ki \mathfrak{s}_{\overline{\uparrow}} \mathfrak{s}_{\overline{\uparrow}}$ [you=GEN]).

The **dative-locative** case can mark nominal P(atient) and R(ecipient) arguments in a clause. It can also mark a nominal adverbial (attached to the noun phrase, §5.6.1.1), an adverbial clause (attached to a verb suffix, §15.5.1, §15.8.3) or a possessor in a possessive clause (§5.4.3). It also functions as an additional marker in locative postpositions (see, for instance, example [3.111]).

The **locative** case marker =na is almost homophonous with the locative demonstrative $n\dot{a}$: 'here'. Its basic function is to mark location within three-dimensional space, whereas the basic function of =lo is to locate an object two-dimensionally. The locative case marks nominal locative adverbials (e.g. $k^{h}im = na$ gava. 'in[side] the house', see §5.6.1.2). Unlike the dative-locative marker, =na may be attached to the short genitive form of a noun, a feature well understandable if =naderives from the relator noun $n\dot{a}\eta$ 'inside' (for relator nouns, see §3.6.8).

The **ablative** marker $=l\varepsilon$ encodes movement from a spatial source (e.g. from the house) or temporal source (e.g. from yesterday). Similarly to =lo, with which it sometimes overlaps, $=l\varepsilon$ occurs in both nominal (§5.6.1.3) and verbal adverbial uses (see §15.3.1.1). The ablative is also used for comparison, see §5.6.1.3.2 and §15.11.

3.7.1.2 Postpositions or cases: $=s\tilde{a}$: 'until' and =tsa(z) 'at, by, with'

The forms $=s\tilde{a}$: $\exists \forall \vdots \forall i$ until' and $=tsa(:) \notin (\forall; i)$ 'at, by, with' are problematic for analysis in that they resemble cases. In distribution $=s\tilde{a}$: resembles cases in being able to occur with nouns, adverbs and verbs, see (3.125-127) respectively.

(3.125) दे'त' तत्र गुव्र वर्ष क्वे का र्य

 p^{h} ina **ba:** $pa=s\tilde{a}:$ lep-o mè?. over.there TPN=until arrive-2INF NEG.EX.PER 'I haven't been able to arrive over there in Bermeok.' (KT, discussion with TB)

- (3.126) ماجم، هَرَجُرَ مَا تَعْجَ عَجَ مَجُعَ جَمَامَ، مَرْجَ مَعْجَمَامِهِ، *tam ódi k'adzo=sã: dɛnta? jò:=kam=la?* speech that how.much=until true EX.PER=ATTQ=HON 'I wonder to what degree that claim is true?' (CY interview)
- (3.127) جَرَحَتَ جَرَحَتَ هُمَّا هَا مَدْ عَامَ مَعْ مُعْتَ هُمَا مَعْ مُعْتَ عَرْجَ عَلَيْ اللَّهُ عَلَيْ الْمُعْ عَلَيْ اللَّهُ عَلَيْ الْحَلَيْ عَلَيْ اللَّهُ عَلَيْ اللَّهُ عَلَيْ الْحَلَيْ عَلَيْ الْحَلَيْ عَلَيْ الْحَلَيْ الْحَلَيْ عَلَيْ الْحَلَيْ عَلَيْ الْحَلَيْ عَلَيْ الْحَلَيْ عَلَيْ الْحَلَيْ عَلَيْ الْحَلَيْ عَلَيْنَا الْحَلَيْ الْحَلَيْ عَلَيْ الْحَلَيْ الْحَلَيْ عَلَيْ الْحَلَيْ عَلَيْ الْحَلْ الْحَلَيْ عَلَيْ الْحَلَيْ عَلَيْ عَلَيْ الْحَلَيْ عَلَيْ الْحَلَيْ الْحَلَيْ الْحَلَيْ عَلَيْ الْحَلَيْ الْحَلَيْ الْحَلَيْ الْحَلَيْ الْحَلَيْ الْحَلَيْ عَلَيْ الْحَلَيْ عَلَيْ الْحَلَيْ الْحَلَيْ عَلَيْ الْحَ الْحَلَيْ الْحَلَيْ الْحَلَيْ الْحَلَيْ الْحَلَيْ الْحَلَيْ عَلَيْ الْحَلَيْ عَلَيْ الْحَلَيْ الْحَلَيْ الْحَلَيْ الْحَالَةُ عَلَيْ الْحَلَيْ الْحَلَيْ الْحَلَيْ الْحَلَيْ الْحَلَيْ الْحَلَيْ الْحَلَيْ الْحَلَيْ الْحَلَيْلَةُ الْحَلَيْ الْ

The clitic $=s\tilde{a}$, however, is distinguished from case-markers by the ability to be formed into a full-blown disyllabic postposition, see (3.128), something which does not happen with case-markers.

¹⁷⁰ The way of writing Bringar sbar-gnya' 'burnt-neck' derives from Sandberg (1895: 124), who relies on Dr. Waddell.

(3.128) ८ दें बटा छे र व केंप्राया वटा वार्या र पार्थेया मुनका पविव थेंटा

t'ato $s\hat{a}$: $t\epsilon$ $n\dot{a}$ $n\dot{a}$: $l\dot{a}pta$ $n\dot{a}\eta ca=lo=r\tilde{a}$: $j\dot{o}$? $kjap-z\tilde{\epsilon}$: $j\dot{o}$?. now until 1SG here school inside=DAT=AEMPH job do-PROG EX.PER 'Until now I have been working here at the school.' (KT life story)

Because of the presence of two synonymic variant forms $=s\tilde{a}$: and $s\tilde{a}$: $t\varepsilon$, $=s\tilde{a}$: is here considered a cliticized form of the postposition $s\tilde{a}$: $t\varepsilon$. There is, however, some fluidity in this criteria, because the locative case marker =na also has a rather synonymic corresponding postposition $n\tilde{a}\eta ca/n\tilde{a}\eta lo$, and thus =na could be argued, in analogy with $=s\tilde{a}$: and $s\tilde{a}$: $t\varepsilon$, to be a cliticized postposition $n\tilde{a}\eta ca/n\tilde{a}\eta lo$. However, because =na is more frequently used in core argument marking and shows more phonological reduction from its corresponding postposition than $=s\tilde{a}$:, =na is here analyzed as a case marker (along with a separate postposition $n\tilde{a}\eta ca$) and $=s\tilde{a}/s\tilde{a}$: $t\varepsilon$ as postposition with a longer and shorter form.

The monosyllabic form =tsa(:) 'by, with', which also resembles a case marker, is here analyzed as a postposition. A formal argument for this analysis is that the derivational locative suffix $-k^ha$ may be suffixed to (=)tsa(:). The suffix $-k^ha$ does not occur with cases but it does occur with relator nouns, e.g. $teyk^ha$ 'on, above', $dyyk^ha$ 'in front of', $tsak^ha$ 'at (the root of)'. Example (3.129) illustrates the use of tsa without $-k^ha$ (a) and with it (b).

- - b) هَمْ مَرْيَطْ (school) حَرْلَمْ حَرْهَمْ؟ iskul tsak^ha=ra óna? school(Eng.) by=AEMPH there 'There at the school?' (KT, discussion with TB)

A further phonological argument for a distinguishing $=s\tilde{a}$: and =tsa(:) from cases is that the vowels in $=s\tilde{a}$: and =tsa(:) tend to be longer than the short vowels in case markers.¹⁷¹

3.7.1.3 Case-stacking

¹⁷¹ An exception is the special case of agentive, where agentivity is shown by vowel lengthening, see §15.4.6.

¹⁷² An analogy can be found in English postpositions combining two elements, e.g. *onto*, *upon*, *within*. For case-stacking in other languages, see for instance Sadler & Nordlinger (2006).

form, e.g. súm-pø:=na anglar that i [three-ORD.GEN=LOC] 'in the third one'. Double genitive marking is a feature of the spoken language, whereas in writing single marking is used.

The ablative case $=l\varepsilon$ and dative-locative =lo may attach to the locative marker, as shown in (3.130) and (3.131) respectively.

 $t^{hinl \varepsilon:=ki}$ $k^{him=na=l\varepsilon}$ $t'ato^{173}=s\tilde{a}$: $ka=j\tilde{a}$: $ma-l\varepsilon p-\varepsilon \varepsilon=di$: PN=GEN house=LOC=ABL now=until who=even NEG-arrive-INF=DEMPH.AGT 'Because no one has so far arrived from Thrinley's house...' (Richhi 43)

(3.131) Jan Bararár

p'u=i $k^{h}im=na=lo$ boy=GEN house=LOC=DAT 'In(side) the boy's house' (SGD wedding customs)

Example (3.132) has the ablative appended to the genitive:

(3.132) दरिंगी दिवा झुरु कुरु ग्रे जिया केंद्र ता केंद्र

 $n\dot{e}:=gi$ $k^{h}im$ lenge:=ki=le ninku $\tilde{i}:$. 1SG.GEN=GEN house PRN.HON=GEN=ABL old EQU.PER 'My house is older than yours.' (TB e)

A combination with an initial short genitive form may amount to three stacked cases, as illustrated by an example from the novel Richhi:

(3.133) र्ज्ञेन मुनि न में

lóbdø:=na=lo school.GEN=LOC=DAT 'in the school' (Richhi 31)

An alternative interpretation would be to treat *nalo* and *nale* as reduced postpositions meaning 'within' and 'from within' respectively, co-existing with the postpositions *nàŋɛa/nàŋlo/nàŋlɛ* '(from) inside'. However, the fact that the postpositional forms *nàŋɛa* and *nàŋlo* are more likely than =na=lo and $=na=l\varepsilon$ to be used alone as simple utterances favours the interpretation of =na=lo and $=na=l\varepsilon$ as instances of case marking. Although the examples of locative case stacking illustrate the gradience of linguistic categories, they cannot question the presence of the phenomenon of case-stacking in Denjongke. The construction $=ki=l\varepsilon$ with the grammatical genitive case in (3.132) is a clear example of case-stacking which cannot be interpreted as a postposition.

3.7.2 Emphatic clitics

Whereas case clitics express various relationships between verbal arguments, the use of emphatic clitics (see Table 3.45) directs the addressee's attention in various ways to a certain constituent in the clause and sometimes the whole clause. Emphatic clitics occur after case clitics.

¹⁷³ The spoken form *t'ato* corresponds to $\neg \exists da-lta$ 'now' used in Richhi. The form used in Richhi is influenced by Written Tibetan.

Form	-	Label	Function described in
=rã:/=ra	<u> २८</u> ./२.	anaphoric emphatic	§16.1.1
=to	র্ট	contrastive emphatic	§16.1.2
=di	حمح.	demonstrative-emphatic	§16.1.3
=ni/nɛ	वे [.]	topicalizer-emphatic	§16.1.4
=jã:	೮ಗ.	additive emphatic	§16.1.5

Table 3.45. Emphatic clitics

The anaphoric emphatic $=r\tilde{a}$: derives from Classical Tibetan $\pi rang$ '-self, same' (Beyer (1992: 218), the meaning of which also helps understand the use of the Denjongke clitic. The contrastive emphatic =to \tilde{r} is most likely a loan of the Nepali contrastive emphatic ta π and/or the Hindi contrastive emphatic to π .¹⁷⁴ The demonstrative-emphatic =di $\alpha rac{r}$ is a grammaticalized form of the proximal demonstrative di 'this'. The topicalizer-emphatic $=n\epsilon/ni$ \bar{a} ' derives from Classical Tibetan "topicalizer" \bar{a} ni (Beyer 1992: 275). Its cognates in other Tibetic languages have been termed "topic marker" (Lhasa Tibetan, Denwood [1999: 103]), "topic particle" (Kyirong Tibetan, Huber [2002: 108]) and "focus marker" (Lamjung Yolmo, Gawne [2013: 487]). The additive emphatic $=j\tilde{a}$: ωrac derives from Classical Tibetan, where it is used in a similar way (Jäschke 1881: 505). For full functional description of the emphatic clitics, see §16.1.

3.7.3 Clausal clitics

Clausal clitics (see Table 3.46), which have scope over the whole preceding clause, attach to the end of the verb complex.

Tuble 5.16. Chuba chucs					
Form		Label	Function described in		
=lo	র্না	reportative	§9.2.1		
$=s(\varepsilon)$	र्बे'	quotative	§9.2.2		
=la	অধ্যম.	honorific (also attaches to nouns)	§16.2.1 (also §3.7.4.3, §11.2.3)		
=60	Ĩ.	attention marker (extended use with nouns)	§16.2.2		
=ki/gi	गो	non-commitment marker	§16.2.3		

Table 3.46. Clausal clitics

The forms =lo, $=s(\varepsilon)$ and =la have recognisable cognates in other Tibetic languages. For instance, the reportative =lo (WT $\approx lo$), deriving from Classical Tibetan (Jäschke 1881: 551-552), has cognates at least in Dzongkha (van Driem 1998: 405-406), Lamjung Yolmo (Gawne 2015), Lhomi (Vesalainen 2016:189) and Kyirong Tibetan (Huber 2002: 107). The quotative WT $\exists x cer$ and honorific WT arg s are also widely attested in Tibetic languages. In addition to being a clausal clitic, the honorific =la also attaches to nouns (§3.7.4.3) and functions as an interjection (§11.2.3).

To my knowledge, anything resembling in form and function the attention marker $= \varepsilon o \notin f$ has not been reported in Tibetic languages. However, at least three Bhutanese, Eastern Bodish languages

¹⁷⁴ Huber (2002: 111) analyzes similarly functioning Kyirong Tibetan form *-ta* as a reflex of WT \neg *da* 'now', for which Jäschke's (1881) dictionary describes a colloquial emphatic use. In Denjongke, however, the reflex of WT \neg *da* 'now', *t'a*, functions as an independent discourse particle which can occur clause-initially, a context in which Kyirong *-ta* does not occur.

have a probable cognate of =co. First, Kurtöp is reported to have an "emphatic particle" *sho* which occurs following both verbs and nominals (Hyslop 2011a: 500-502). Two of the example sentences given by Hyslop (2011a: 501), both of them postposed to a verb, are followed by exclamation markers, which are reported to be used with surprising information (p. 679-680), a use also covered by Denjongke =co. Second, Hyslop and Tshering (2010) have also found a similarly functioning "sentence final particle" co in Dakpa (a.k.a. Tawang Monpa). Third, Andvik's (2010: 441) grammar of Tshangla describes a "marked topic particle" *sho*, which covers some of the same semantic field as Denjongke =co. Nathan W. Hill (personal communication) suspects that =co is derived from Classical Tibetan for shog, the imperative form for the verb 'come', which is also used in hortatives.

The non-commitment marker =ki/gi is a loan from Nepali ki क, which has been mainly used for marking polar questions and 'or' in alternative questions (see, for instance, Turner 1962-1966) but which has lately developed a declarative use expressing the speaker's non-commitment to and uncertainty about a statement.

The function of clausal clitics is described in $9.2 (=lo \text{ and } =s\varepsilon)$ and 16.2 (=la, =co and =ki/gi).

3.7.4 Other clitics

There are five further morphemes which, because of their transcategorial nature, are analyzed as clitics. These clitics are the plural $=tsu \notin (\$3.7.4.1)$, the indefiniteness clitic =tci? (a) \Re (\$3.7.4.2), the honorific clitics =la(:) and $=tc^{h}o:$ and $=tc^{$

3.7.4.1 Plural clitic =tsu

3.7.4.2 Indefiniteness clitic =tci?

The indefiniteness clitic $=t\epsilon i ?$ (η) \Im derives from the numeral $t\epsilon i ? \eta \Im$ 'one'. While in some contexts it is phonologically, syntactically and even semantically difficult to distinguish between the numeral and the indefinite uses, the vowel in the clitic is typically pronounced shorter than in the numeral. For one example, consider (3.134), where the clitic is attached to the numeral.

For more examples, refer to §4.1.6.

3.7.4.3 Honorific clitics =la(:) and $=tc^{h}o$:

(3.135) a) สูสาญญาณา (นาสูนา นาชณิ)

tcam=la:, $k^{h}anup$ $\eta \dot{a}tci$ madam=HONthe.day.before.yesterday1PL.GEN'Madam, the day before yesterday our...' (Richhi 8)

'The father came back home yesterday and is (there/at home).' (Richhi 55)

jò?.

EX.PER

The less frequent clitic $=te^{h}o$: can be translated as 'most excellent, honorable' and is used of people of considerable social status. The morpheme $te^{h}o$: is considered a clitic because, similarly to plural =tsu, it attaches to the last word of the noun phrase, whether the last word is the head noun or a modifier.

(3.136) a) कॅर्बा मुना यहा र्झेनबा स्वा मुना वर्केवा

 $tc^{h} \phi g \epsilon$: $t^{h} utop$ $n \dot{a} m g \epsilon l = tc^{h} o$: king PN PN-most.excellent 'most excellent king (chogyal) Thutop Namgyal' (CY interview)

3.7.4.4 Urgetive clitic =*mø*?

The urgentive marker $=m\phi^2 \, \breve{x}_{\neg}$, which attaches to verbs, is analyzed as a clitic rather than an affix because it may attach, in addition to the verb root, also to other markers such as imperative, hortative and optative, see §11.3.3. The urgentive form most likely derives from WT \breve{x}_{\neg} mod, an emphatic verb of being 'to be indeed'. Sandberg (1895: 57) reports $m\dot{\partial}$: as an independent verb, but in my data such uses are not found.

(3.137) $\hat{\mathbf{w}}$! $\hat{\mathbf{y}}$. $\hat{\mathbf{y}}$

For further examples on $=m\phi$? see, §11.3.3.

3.8 Summary remarks

This chapter, the only one focusing on morphology and etymology, introduced Denjongke word classes, affixes and clitics. The main criteria used in distinguishing clitics from affixes was transcategoriality, i.e. the ability to attach to words of more than one word class. It was shown that Denjongke has four major word classes (nouns, verbs, adjectives and adverbs) and eleven minor word classes. Many nouns and verbs can be divided into ordinary and honorific registers. The discussion on verbs listed 45 phonologically related pairs of controllable vs. non-controllable verbs. I also provided etymological information on verbal suffixes and gave introductory examples of each form in clausal context.

Adjectives were shown to be a word class which, although deriving from stative verbs, is morphologically distinguished from other word classes by a number of adjectivizing suffixes. Adverbs were seen to be a word class which, although partly overlapping with adjectives, are also distinguished from them morphologically and syntactically. Time words such as *t'o:pa* '(in the) morning', which have both nominal and adverbial characteristics, were on distributional grounds analyzed as temporal adverbs. Numerals were seen to follow both decimal and vigesimal systems. It was shown that postpositions divide into more noun-like postpositions, which were called relator nouns, and less noun-like postpositions. That is, relator nouns were treated as a subclass of postpositions.

The last major section introduced 21 clitics (all monosyllabic), which were divided into five case clitics, two cliticized postpositions, four emphatic clitics, five clausal clitics and five other clitics (plural, indefinite, urgentive and two honorific clitics). It was shown that cases divide into grammatical and spatial cases, with the dative-locative occupying both categories. A typologically interesting feature was seen to be case-stacking of up to three case markers.

4 Phrasal constituents

This chapter moves the discussion from morphology and etymology to syntax by describing the constituents in nouns phrases (§4.1), the verb complex (§4.2), adjective and adverb phrases (§4.3) and numeral phrases (§4.4).

4.1 Noun phrase

This section first provides an introduction to the structure of the noun phrase (\$4.1.1) and then describes prenominal (\$4.1.2) and postnominal modifiers (\$4.1.3). The section after that describes the ordering of clitics at the end of the noun phrase (\$4.1.4). This is followed by sections on the plural marker (\$4.1.5), (in)definiteness (\$4.1.6), coordination (\$4.1.7) and reduplication (\$4.1.8).

4.1.1 Introduction to noun phrase

The noun phrase may consists of 1) a noun head with its preceding and/or following modifiers (4.1), 2) a proform (pronoun, demonstrative or question word) (4.2), or 3) a nominalized clause (4.3).¹⁷⁵ In the following discussion, square brackets are used for marking noun phrases, if the noun phrases are given in clausal context or if noun phrases are embedded within each other in a complex way. If the whole example consists of simply one noun phrase, there are no square brackets. When the noun phrase contains more than one word, the head noun and its equivalent in the English translation are underlined.

Noun with preceding and following modifiers

hindu-i <u>p'um</u> nén kjap-kjap-o=loHindu=GEN girl wedding do-RDP-2INF =DAT 'to a <u>girl</u> of Hindus' who has been married...'/ 'to a married Hindu <u>girl</u>...' (sbar-phung 88)

 b) هَٰ حَرَّ جَعَ عَامَهُ حَرَّ حَمَّ *ódi* <u>p'um</u> ní: dɛ:tɛi? that girl two that.much 'those (as much as) two <u>girls</u>' (SGD wedding customs)

Pro-form

(4.2) a) BY $\widehat{\mathbb{G}}^{k} \widehat{\mathbb{G}}^{k} \widehat{\mathbb{G}}^{k} \widehat{\mathbb{G}}^{k}$ $[k^{h}u] p^{h}i:-p(o) b\varepsilon^{2}$. 3SGM be.late-2INF EQU.NE '[He] was late.' (RS pupil joke)

¹⁷⁵ The nominalized clause in (4.3a) is analyzed as a complement clause (see §14.1.1) and (4.3b) as a headless relative clause (see §13.2.1.3)
b) אָז'יִבְלְ'מִי בְלָ' שְּק' מִיבִיאָ? [ódi=lo=di] [k'an] làp-to? that=DAT=DEMPH what say-IPFV '[What] is [that] called?' (PD intro video)

Nominalized clause

(4.3) a) ग्वत्रकार्झर लेगाय नुरू हैं - حصر क्रॅर र्ह्याकार्ट (مَعْتَمَ مَعْتَمَ عَلَيْهُ اللَّهُ اللَّهُ الْعَلَى الْعَلَى اللَّهُ الْحَافَةُ اللَّهُ الْعَلَى الْعَلَى الْعَلَى الْعَلَى الْعَلَى الْعَلَى الْعَلَى اللَّهُ الْعَلَى اللَّهُ الْعَلَى اللَّهُ الْعَلَى اللَّهُ اللَّهُ الْعَلَى اللَّهُ (1.3) a) ग्वत्रकार्झर लेगाय कि able able in the same and the same and

The ensuing discussion focuses on the first option, the type and order of preceding and following modifiers in the noun phrase. The structure of Denjongke noun phrase is summarized in Figure 4.1, which summarizes the order of various constituents. The abbreviation =EMPH covers both =CEMPH and =AEMPH. The notion quantifier (QUA) includes quantifying pronouns and some versatile quantifying adverbs (see §4.1.3.3).

Figure 4.1. Structure of the noun phrase





 $(GEN.ATTR) = \begin{cases} (NP=GEN) \\ (ADV=GEN) \\ (PP=GEN) \\ (RC \text{ with } -po \text{ or } -sa \\ (NCC) \end{cases} \}$

(Abbreaviations in Figures 4.1 and 4.2 (from left to right): DEM demonstrative, NP=GEN noun phrase in genitive, PP=GEN postpositional phrase in genitive, RC relative clause, NCC noun complement clause, =DEMPH demonstrativeemphatic, INDF=indefiniteness marker, ADJ adjective, NUM numeral, QUA quantifier, EMPH emphatic)

The order of noun modifiers is demonstrative + noun + adjective + numeral + quantifier (+ emphatic), see (4.4). Numerals and quantifiers are usually exclusive of each other, but certain

¹⁷⁶ The additive emphatic 'too, even, yet'

quantifiers, such as $t^hamtee?$ 'all' in (4.4) may co-occur with numerals, although the combination is not very frequent.

(4.4) ซัง เสริ. เฏิม. สมมารา. สมมารา. สรา.
 ódi khim sá:p súm thamtee=di

that house new three all=DEMPH 'all those three new <u>houses</u>' (KN e)

The following subsections describe prenominal (§4.1.2) and postnominal modifiers (§4.1.3)

4.1.2 Prenominal modifiers

A prenominal modifier of a noun phrase can be a demonstrative (§4.1.2.1), genitive attribute (§4.1.2.2), a relative clause with $-k^{h}\tilde{\epsilon}$: (§4.1.2.3) or a juxtaposed nominal (§4.1.2.4).

4.1.2.1 Demonstrative

Distal $\dot{o}di$, proximal di and, less frequently, the emphatic proximal dodi, which can all occur as independent pronouns, do also occur either as prenominal modifiers or postnominal modifiers (for postnominal uses, see §4.1.3.4). Examples (4.5a) and (4.5.b) illustrate prenominal uses of $\dot{o}di$ and di respectively.

t'izã: nà: $d \in ndz \tilde{o}:=na$ $\partial \eta \cdot k^h \in n=tsu=lo$ [$\delta di \ \underline{c} \delta ku$] $j \in bb \in =la$. but here Sikkim=LOC come-NMLZ=PL=DAT that paper EX.NE=HON 'But those who came here to Sikkim had [that <u>document</u>].' (CY interview)

In addition, the demonstrative pro-adverb *nà*: 'here' may modify a noun.

(4.6) गहा, केंद्र व' कें' केंगे धर राज्य केंग केंद्र केंग

karma, $te^{h}\phi^{2}$ [*nà:* $\underline{\phi ni}=lo$] $j\dot{a}:=tsa:=s\tilde{a}:$ $k\varepsilon:$ p'in lo.PN 2SG.L here small.child=DAT up=by=until bring give TAG.Q 'Karma, you take [the <u>child</u> here] all the way up, okay.' (Richhi 40)

Other pro-adverbial demonstratives, when used as noun modifiers, have to be genitive marked, see §4.1.2.2 below.

When co-occurring with a genitive attribute, the reference of the demonstrative is contextually determined. In (4.7a), the demonstrative modifies the noun which functions as the genitive attribute, whereas in (4.7b) the demonstrative modifies the head noun, not the genitive attribute. The modified word is underlined.

(4.7) a) क्षें प्रदे ज्ञूद यो क्रें प्रदे हुं

[*ódi* <u>gan</u>=gi] *lømpu* di=tsu that time=GEN minister this=PL 'ministers of that time' (CY interview)

b) र्खेंग्दर्न. अग्रेया में यो. यिष. पर्ट.

 \acute{odi} tea:co:=ki <u>k^him</u>=di

that corrugated.iron=GEN house=DEMPH

'that house of corrugated iron' (not: 'house of that corrugated iron') (KL, BB discussion)

4.1.2.2 Genitive attributes

The genitive-marked noun modifier may be a noun phrase (4.8), adverb (4.9), postposition phrase (4.10-11), relative clause (4.12) or a noun complement clause (4.13). In the following illustrative examples, genitive attributes are marked with square brackets.

Noun phrase

(4.8) a) 5à. m.t.

[*nè:*] *ápo* 1SG.GEN father 'my father' (DB life story)

b) ୮.୭ଘି.୩. ସମ୍ବନ୍ଧ ହିଁ ୮୬.୩.୩. ଅ. ଅ. ଅ.

[[*ŋàtɕi=gi*] *dɛndzoŋ=gi*] *míri?* 1PL.GEN=GEN Sikkim=GEN people 'people of our Sikkim' (NAB BLA 7)

c) ८ र छते में तर्र ते कुष [ŋàtɛi=gi=di] lògju? 1PL.GEN=GEN=DEMPH story 'this story of ours' (YR canteen video)

Note that in (4.8c) the demonstrative-emphatic =di intervenes between the noun and its genitive modifier and that the genitive is double marked.

Adverb phrase (independent uses of postpositions are inluded within adverbs here)

(4.9) a) क्रेंग्ज्'में छु

 $[\acute{o}na=gi]$ tc^hu there=GEN water 'the water (of) there' (UTR plains story)

b) דְיאָרִישִׁי מָקָי מָלָישִ [t'ariŋ=gi] tsho:du:=di=na today=GEN meeting=DEMPH=LOC 'in today's meeting' (RD BLA 9) c) र्धेव र्येगी र्स्वेय मु

[*nénlo=gi*] *lóbda* before=GEN school 'the school before' / 'the school of days gone' (Richhi 44)

d) देंगायी के

[$\partial = gi$] mí under=GEN human 'people (who live) down' (LA birth in Lachung)

e) مَتْرَنْمْسَانَا بِجَارَ عَلَمْ مَنْمَانَ بَجَارَ عَلَمَ مَنْمَانَ بَجَارَ حَسَانَا بِجَارَ عَلَمَانَ اللَّهِ اللَّهُ عَلَيْنَا اللَّهُ اللَّهُ عَلَيْنَا الْحَالَيْنَا الْحَلْنَا الْحَالَيْنَا الْحَلْقَانَ الْحَالَيْنَا الْحَلَيْنَا الْحَلَيْنَ الْحَلَيْنَ الْحَالَيْنَا الْحَلَيْنَا الْحَلَيْنَا الْحَلَيْنَا الْحَالَةُ اللَّالِي الْحَالَةُ الْحَالَيْنَا الْحَالَيْ الْحَالَيْنَا الْحَالَةُ اللَّهُ عَلَيْنَا الْحَالَيْ الْحَالَيْنَ الْحَالَيْنَا الْحَلَيْنَا الْحَالَيْ الْحَالَيْ الْحَالَيْ الْحَالَيْ الْحَالَيْنَا الْحَالَيْ الْحَالَيْ الْحَالَيْنَا الْحَالَيْ الْحَالَيْنَا الْحَالَيْ الْحَالَيْ الْحَالَةُ الْحَالَةُ الْحَالَةُ الْحَالَةُ الْحَالَةُ الْحَالَى الْحَالَيْنَا الْحَالَيْ الْحَالَةُ الْحَالَيْ الْحَالَيْنَا الْحَالَيْنَا الْحَالَيْنَا الْحَالَةُ الْحَالَيْنَا الْحَالَةُ عَلَيْنَا الْحَالَيْنَا الْحَالَةُ الْحَالَةُ الْحَالَةُ الْحَالَةُ الْحَالَيْنَا الْحَالَيْنَا الْحَالَةُ الْحَالَيْحَالَةُ الْحَالَيْكَ الْحَالَةُ الْحَالَيْ الْحَالَ لَكَ الْحَالَيْكَ الْحَالَةُ الْحَالَةُ الْحَالَةُ الْحَالَيْ الْحَالَةُ الْحَالَةُ الْحَالَةُ الْحَالَيْ لَكَا الْحَالَ الْحَالَيْ الْحَالَيْكَ الْحَالَيْ الْحَالَيْ الْحَالَةُ الْحَالَةُ الْحَالَةُ الْحَالَةُ الْحَالَةُ الْحَالَيْ الْحَا

In postposition phrases, the genitive marker =ki/gi replaces the last suffix/syllable of the noncase-marked postposition, as shown in (4.10), where the postpositions modify the noun k^{him} (WD gave *khyim*) 'house'.

(4.10)	bolok ^h a	तर्चे लेंगा वर	'next to' >	k ^h im bolo=gi	เปิลา ฉรังพัญญา
	tɛŋkʰa/tɛŋ=lo	ह्रेट वर /हेट कें	'above' >	k ^h im tɛŋ=gi	छिव्रा झे⊏ गी
	∂:lo/∂:lɛ	র্ববার্ন্মে/র্ববাঞ্ম	'under' >	khim ò:=gi	ष्ठित्रः देयायोः
	nàŋɕa/nàŋlo	קביף(ב)י/קביתי	'inside' >	k ^h im naŋ=gi	ਤ੍ਰਿਕਾ ਕ੍ਰ–ਾਧੀ

Postpositional phrases as genitival attributes frequently include other embdedded genitive attributes, as shown by the examples of gradually increasing complexity in (4.11).

Postposition phrase a) तज्ञसार्खेट्रसा वट्टायी यानवा कुषार्छु (4.11) $[dendz \tilde{o}: n a \eta = gi]$ tamgy:=tsu Sikkim inside=GEN legend=PL 'legends within Sikkim' (SGD cave story) b) พุสเฉิ ฉรังสังพา¹⁷⁷ สุจาญญา [karmø:] bolo=gi] kupkja? PN.GEN next.to=GEN seat 'the seat next to Karma' (Richhi 125) c) र्चेर्स्डित अर्वेति छेट यो हा ten=gi] má $[[p'ots \emptyset] go=i]$ child.GEN head=GEN top=GEN wound 'wound on the child's head' (Richhi 3)

¹⁷⁷ Both the written forms בְאָיאָי 'bo-lo and בְאָיאָש 'bo-log occur in Richhi.

Relative clauses and noun complement clauses are discussed in detail in §13. These attribute types are here illustrated just by one example each:

Relative clause

(4.12) গ্রান্ড

गु'रु' रेव'र्रे'के' ८८.सु' वर्ळव्यवा रावुगवा रावुगवा रादी गवा

 $[guru rimpute^{h}e t'\tilde{a}:pu$ $ts^{h}am$ zu:-zu-bø:] $n\dot{e}:$ guru Rimpoche long.agoseclusionsit.HON-RDP-2INF.GENsite'a site [where Guru Rimpoche used to sit in solitary meditation]' (SGD cave story)

Noun complement clause

(4.13) مَرْآَبُ مَرْ مَرْهُمْ بَعْرَامَةُ مَرْ عَلَيْهُمْ الْحَامَةُ مَرْ عَلَيْهُمْ الْحَامَةُ مَرْ عَلَيْهُمْ الْحَامَةُ مَنْ الْحَامَةُ مَا الْحَامَةُ مَنْ الْحَامَةُ مَنْ الْحَامَةُ مَنْ الْحَامَةُ مَنْ الْحَامَةُ مَنْ الْحَامَةُ مَا الْحَامَةُ مَا الْحَامَةُ مَالْحَامَةُ مَا الْحَامَةُ مَا الْحَامَةُ مَا الْحَامَةُ مَالْحَامَةُ مَا الْحَامَةُ مَا الْحَامَةُ مَالْحَامَةُ مَا الْحَامَةُ مَا الْحَامَةُ مَالْحَامَةُ مَا الْحَامَةُ مَا الْحَامَةُ مَا الْحَامَةُ مَا الْحَامَةُ مَالْحَامَةُ الْحَامَةُ مَا الْحَامَةُ مَالْحَالْحَامَةُ مَا الْحَامَةُ مَا الْحَامَةُ مَالْحَالْحَامَةُ مَا الْحَامَةُ مَا الْحَامَةُ مَالْحَامَةُ مَا الْحَامَةُ الْحَامَةُ مَا الْحَامَةُ مَا الْحَامَةُ مَالْحَامَةُ الْحَامَةُ مَا الْحَامَةُ الْحَامَةُ مَا الْحَامَةُ مَا الْحَامَةُ مَا الْحَامَةُ الْحَامَةُ الْحَامَةُ مَالْحَامَا الْحَامَةُ مَا الْحَامَا الْحَامَةُ مَا الْحَامَةُ

In spoken language, genitive marking is sometimes dropped from noun modifiers. Example (4.14) provides two phrases from the same story referring to the same event. In a) the modifier p'itcug 'bird', modifying the noun ka:m/ka:po 'foot', occurs with genitive marking but in b) without.

b) לא פרי קריגולי קריקי
 [p'itcuŋ] kã:pø: nàŋca
 bird foot.GEN inside
 'in the bird's foot' (PAD bet story)

Leaving out genitive marking is particularly frequent with toponymic modifiers, see §4.1.2.4.2.

4.1.2.3 Relative clause with $-k^{h}\tilde{e}$:

Noun-modifying relative clauses with the nominalizer $-k^{h}\tilde{\epsilon}^{\cdot}$, which do not require (nor allow) genitive marking, are described in §13.2.1. For an introductory example, consider (4.15).

4.1.2.4 Juxtaposition

A noun may also be modified by a noun phrase which is placed in juxtaposition to the noun. In these cases, it is sometimes difficult to determine which noun phrase modifies which one. Constructions with juxtaposition are used to refer especially to people and places. Therefore a juxtapositional construction typically contains a personal name or a toponym, which is juxtaposed

to another noun phrase. Juxtaposition may be either appositional or non-appositional. In appositional juxtaposition, a noun phrase which modifies a noun has the same referent as the noun it modifies (Haspelmath 1993: 256). In non-appositional juxtaposition, the juxtaposed noun is not co-referential with the modified noun.

4.1.2.4.1 Appositional juxtaposition

An apposition to a personal name may be a title (4.16-17) or a kinship term (4.18-19). The kinship term typically occurs before the name, although a reverse order is also possible. Some frequent titles are listed in Table 4.1.

Table 4.1. Sc	ome titles	
києар	ଞ୍ଚ୍ୟୁ'ଜ୍ୟୁ'	'mister'
tcam, l̥atcam	સુરા, સુસુરા	'mrs'
lópøn	শ্ধ্র্যন'ন্র্যুব'	'teacher (m.)'
lópøm	สัสาราัสม.	'teacher (fem.)'
te ^h o:p	র্ক্রম্বা / র্ক্রিঝার্ন্যা	'monk'
k ^h empo	ঝাম্মব্ 'র্যা'	'doctor of Buddhist studies'

- (4.16) র্জরা মর্লন্থন নগ্য স্থ্রন *ómdze? sámdup* second.lama PN 'omdze Samdrup' (LT kitchen discussion)
- (4.17) দ্যা'পিম' ঈ্লি'ন্র্য্র' *taci lópøn* PN teacher 'teacher Tashi' (oh, Martam)
- (4.18) هري چرک ج *áku bhaitcuŋ* father's.younger.brother PN 'Uncle Bhaicung' (KN kitchen discussion)
- (4.19) अर्हे अतु हेर्गेट खप्ट

ádzo mènt^hekoŋsalã: grandfather PN 'grandfather Mon Thekongsalang' (CY interview)

A person may also be referred to by two appositional titles:

(4.20) هترتا¹⁷⁸ المعرية (captain) *jà:p kæpten* nobleman captain(Eng) 'Honorable Captain' (KN CY interview)

In the following two examples, two appositional nouns refer to the same location. In (4.21), the latter noun clarifies the referent of the toponym by a common noun. In (4.22), the second word specifies a referent from a group of possible referents suggested by the first word ($b\epsilon jyl$).

- (4.21) $\operatorname{Her} \widetilde{\mathcal{F}}^{\operatorname{M}}$ Kar $g\widetilde{a}:to?$ t^{hom} TPN town 'Gangtok town' / 'the town of Gangtok' (YR canteen video)
- (4.22) भ्रुषाध्याः तन्त्रषासाः हिन्ताः

bejyl dɛmadzõ: hidden.land fruit.valley/Sikkim 'hidden land of Sikkim' (CY interview)

An appositional construction may also consist of a noun phrase and a personal pronoun (4.23) or two personal pronouns (4.24):

(4.24) المَّامَّةُ المَّامَّةُ المَّامَّةُ المَّامَةُ المَّامَةُ المَّامَةُ المَّامَةُ المَّامَةُ المَّامَةُ المُحْدَامَةُ المَّامَةُ المُحْدَامَةُ المَّامَةُ المُحْدَامَةُ المَّامَةُ المُحْدَامَةُ المَّامَةُ المُحْدَمَةُ المَامَةُ المُحْدَمَةُ المَامَةُ المُحْدَمَةُ المَحْدَمَةُ المَحْدَمَةُ المَحْدَمَةُ المَحْدَمَةُ المَحْدَمَةُ المَحْدَمَةُ المَحْدَمَةُ المَحْدَمَةُ المَحْدَمَةُ المحْدَمَةُ المُحْدَمَةُ المُحْمَاحُيْنُ المُحْدَمَةُ مُحْدَمَةُ مُحْدَمَةُ المُحْمَعُ مُحْدَمَةُ المُحْدَمَةُ المُحْدَمَةُ المُحْدَمَةُ المُحْمَدُ مُحْدَمَةُ المُحْدَمَةُ مُحْدَمَةُ مُحْدَمَةُ مُحْدَمَةُ مُحْدَمَةُ مُحْدَمُ مُ

In some instances, appositions lexicalize. For instance, the form $\dot{a}dzota? \ll \dot{\Xi} \approx 1$ 'tiger' (TB) deriving from $\dot{a}dzo ta?$ 'grandpa tiger', is used by some speakers as the main word for referring to the feline.¹⁷⁹

4.1.2.4.2 Non-appositional juxtaposition

People may be identified with reference to their place of origin with a non-appositional construction where a toponym is combined with a kinship term or a title, see (4.25). Although these syntagms resemble compounds, they are distinct from prototypical nominal compounds in that the prototypical compound noun combines two monosyllabic words (e.g. $mik-te^{hu}$ 'eye-water, tear'), whereas the non-appositional juxtaposition combines longer syntagms, which are much longer than

¹⁷⁸ This word is probably a phonologically reduced version of WT $u_{i} = rabs$ 'nobleman', which has become homophonous with $j\dot{a}:p$ 'father (hon.)' (WD $u_{\nabla i} yab$).

¹⁷⁹ A motivation for this grammaticalization may be that $ta 2 \frac{1}{5}$ is almost homophonous with $ta_{\frac{1}{5}}$ 'horse'.

what would be expected of a word in Denjongke, into a single concept (e.g. 4.25a). Prosodic phenomena involved in non-appositional juxtaposition, however, remain a subject for further study.

(4.25) a) 되ってあい歌 (新元) (4.25)

p^haregõ: ádzo TPN grandfather 'Paramgang grandfather' / 'Grandfather from/of Paramgang' (KNA kitchen discussion)

- b) هُتَ مَعْنَا عَلَيْهُمْ اللَّهُ عَلَيْهُمُ اللَّهُ عَلَيْهُمْ اللَّهُ عَلَيْهُمُ اللَّهُ عَلَيْهُ اللَّهُ عَلَيْهُ اللَّالَ اللَّهُ عَلَيْهُ اللَّالَةُ اللَّالَ اللَّالَ اللَّالَ الْحُلْقُلُ عَلَيْلَةُ عَلَيْ الْحُلْحُامُ اللَّهُ عَلَيْ الْحُلْقُلُ عَلَيْنُ الْحُلُكُ اللَّهُ عَلَيْهُمُ اللَّالِ اللَّهُ عَلَيْلُ اللَّالِ اللَّعْلَيْلُ الْحُلْحُلُكُمُ اللَّهُ عَلَيْلُولُ اللَّالَةُ اللَّالَةُ اللَّالِي حَلَيْ اللَّالِي الْحُلْحُلْكُمُ اللَّالَةُ اللَّالِي عَلَيْ الْحُلُكُمُ اللَّالِ اللَّالِي عَلَيْ الْحُلُكُمُ اللَّالِي عَلَيْ الللَّالِ اللَّالَةُ عَلَيْ الْحُلُكُمُ اللَّكُمُ اللَّالِي عَلَيْ الْحُلْحُالِي الْحُلْحُالِي الللَّا عَلَيْ الْحُلُكُمُ اللَّالَةُ عَلَيْ الْحُلُكُ اللللْحُلُكُمُ الْحُلُكُمُ الْحُلُكُمُ اللللْحُلُحُلُي الْحُلْحُلُكُمُ اللَّالِي عَلَيْ الْحُلُولُ اللَّالِي عَلَيْ اللْحُلْحُلُكُمُ اللْحُلُكُمُ الْحُلُكُلُحُالِي الْحُلُحُلْحُلُكُمُ اللْحُلُحُلُحُلُكُلُحُلُكُمُ اللْحُلُحُلُكُ الْحُلُكُمُ اللللَّا عَلَيْ الْحُلُكُ اللللَّا عَلَيْلُ الْحُلُكُمُ الْحُلُحُلْحُلُكُمُ الْحُلُحُلْحُلُكُمُ الْحُلُولُ الْحُلُكُلُحُلُكُمُ الْحُلُحُلُ عَلَيْ الْحُلُحُلُ عَلَيْلُ الْحُلُكُ الْحُلُحُلُكُمُ الْحُلُحُلُكُ الْحُلُكُمُ الْحُلُكُ حُلُكُمُ الْحُلُكُلُحُلُلُ عَلَيْ الْحُلُحُلُكُ عَلَيْ الْحُلُكُ الْحُلُولُ الْحُلُكُ الْحُعَلَي حُلَيْلُ حُلَيْ حُلُلْحُلُكُ الْحُلْحُ ع
- c) จริสาขุรสา พจา/พรจา ซิลีสายูา

dentam jà:p/jà:p bompu TPN father/nobleman big 'the great father/nobleman of Dentam' (CY interview)

d) ८.२५.भी. पर्यथा क्रिंटका. भेजास.

yàtci=gi dɛndzõ: gɛ:pu 1PL.GEN=GEN Sikkim king 'our king of Sikkim' (CY interview)

Some adjectives (4.26a) and adverbs (4.26b) precede a noun without genitive marking:

(4.26) a) 읽고 (위기 (4.26)

*tc^hiliŋ kε*2¹⁸¹ foreign language 'English language' (DR discussion with KL)

b) झण्ठाभेत्तः आयात्रज्ञ *t^ha:-riŋ sánɛ?* 'distance-long region' 'far-away region' (Richhi 1)

Some nouns may also modify other nouns without genitivization. Each of the words used in (4.27) may also occur independently as a noun phrase head.

¹⁸⁰ The Denjongke script here probably reflects the Classical Tibetan spelling, the Denjongke spelling met elsewhere being across *zhing-bshags*, see (4.8) above. The phonological script reflects the Denjongke spelling and pronunciation.

¹⁸¹ For historical reasons, this combination, which literally means 'foreigners' language', has acquired the more specific meaning 'English language'. There is also a more specific word referring to English, *indzi ke?* (WD 方式 完 新方' *dbyin-ji skad*).

(4.27) a) ইণ্ট্রব্যুষ্ণ ঈণ্টব্য p^hogja? mi=tci? husband human=INDF 'a male person' (nga'i 'gan 8)

b) भें' मु'सु'

mí gapu human elder 'an elder man' (KN kitchen discussion)

b) त्रे' र्से'कुगुरा' का'र्ये'

mí p^hogja gapu human husband elder 'an elderly male person' (rnam-rtog 28)

4.1.3 Postnominal modifiers

Postnominal word-level modifiers (for clitics, see §4.1.4) occur in the following order: (ADJ) (NUM) (QUA) (DEM). Postnominal modifiers other than the genitive-marked relative clause bear no morphological cues revealing their modifying relationship to the head noun.

4.1.3.1 Adjectival modifiers

A noun-modifying adjective phrase typically occurs postposed to the head noun.

(4.28) $\operatorname{sgc} \operatorname{sgc} \operatorname{sgc} \operatorname{sgc}$ $g\tilde{a}$: $te^{h}u\eta te^{h}u\eta$ hill small 'a small hill' (TB e)

(4.29) गहेतु र्झेयरसु

nén bompu wedding big 'big wedding' (DB life story)

The adjective itself may be preceded by a modifier:

(4.30) **अ**'रे' लेप' क्वेंगरा'

ári [*l*èp bompu] paddy.field very.much big 'a very big paddy field' (TB bull story)

(4.31) gqq_N : $\widehat{\pi}q_N$: $q_{\Gamma'}$: aq_N : aq_N : fq_N : aq_N : fq_N : f_{μ} (k' \tilde{a} :mentse? k^heta?] understanding extremely sophisticated 'extremely sophisticated understanding' (CY interview)

When co-occurring with a numeral modifier, the adjective (phrase) may occur preceding the noun, as shown in (4.32a). However, another consultant wanted to correct the order in (4.32a), after seeing it in writing, to (4.32b).

(4.32) a) क्रॅंग्स्ट्रेंग् कुट्र कुट्र हुंग प्रति म्रि मा

 \acute{odi} te^huŋte^huŋ k^hi zi-t^hamba that small dog four-NUM 'those four small dogs' (TB e)

b) אָ אָדי אָד' אָל' אָאיזי אָל' אָל *k^hi te^huŋte^huŋ zi-t^hamba di=tsu* dog small four-NUM this=PL 'those/these four small dogs' (PR e)

4.1.3.2 Numeral modifiers

Numerals and quantifying pronouns can function as quantitative modifiers of nouns. Typically these two modifier types are exclusive of each other (4.33), but some quantifiers such as $t^hamtee2$ 'all' can occur with numerals (4.34).

(4.33) a) দ্রুঝা বাধ্যুঝা

p'um súm girl three 'three girls' (KN e)

b) सुम्रा गोषार्चें

p'um kɛ:p(o) girl many 'many girls' (KN e)

(4.34) i = 0

For two additional examples of modifying numerals, consider (4.35) and (4.36).

(4.35) अर्हे मुखार्चे कें विया पवि केंट्र

 $\dot{a}dzo$ $g\varepsilon:po$ [$l\partial$ $k^h\varepsilon:$ zi- $ts^h\theta$?] grandfather old.man year score four-about 'old grandfather of some eighty years' (Richhi 78)

(4.36) ה. هما. ماها عليه تحريجة عليهما بنا *nà* [*cà? súm*] *dø:-ruŋ dikɛ.* 1SG night three stay-CONC be.alright 'Even if I stay three nights, it's alright.' (RS duetto) When co-occurring with an adjectival modifier, the numeral may precede the noun in spoken language, as shown by spoken example (4.37a). According to consultant KUN, however, written language prefers the orderings given in (4.37b) and (4.37c).¹⁸²

(4.37) a) क्षें वर्दने गुरुवा नवा पा देवा क्वें कार्यु

ódi súm-thamba t'ɛp bompu that three-NUM book big 'those three big books' (TB e)

b) देनः र्श्वेमसुः गुसुमानमामा सिंग्दर्नः

t'ɛp bompu súm-tʰamba ódi book big three-NUM that 'those three big books' (KUN e)

c) สังสรา ราวา ซัลงสง ขุญลง สลง นา สรา

ódi t'ɛp bompu súm-t^hamba=di that book big three-NUM=DEMPH 'those three big books' (KUN e)

4.1.3.3 Quantifying modifiers

Quantifying modifiers include quantifying pronouns (a subclass of indefinite pronouns, see §3.6.4 and §6.3.1), versatile quantifying adverbs $k \in po$ 'much, a lot', màypo 'much, a lot' and pùypuy 'little, few' (see §3.5.2.4) and the specific numeral $k'\tilde{a}:pu$ 'one full (vessel)' (and the variant k'o:m from Lachung). Quantifying pronouns are listed in Table 4.2 and exemplified after the table. They also occur independently as pronouns, see §6.3.1.

Table 4.2. Quan	infying pronouns	
t ^h amtse?	রমন্য তদ	'all, totally'.
kʰɛːlɛ	मि'ले'	'all, totally'
$t\epsilon^h a:(l\epsilon)$	ଇ.(ଏୟ.)	'all, totally'
dzaŋki	<u>इं</u> ट्र'गो'	'all' (Lachung)
rere	रे.रे.	'each (one)'
màŋtɛiɛ0? ¹⁸³	ag`a૾૽ર્વેષ', a⊏`aે`વેષ'	'most'
khace?	A. 44.	'some'
làri?	a.t.	'some'
làla(làla)	ୟ'ୟ'(ୟ'ୟ')	'some(others)'
ripi (ripi)	रे'गविष' (रे'गविष')	'a few (of people)'
ka:kut¢i?	गा'गा'र्च्च्या'	'a few'
tei:ní:	यचिया यानेत्रा	'a few'

Table 4.2. Quantifying pronouns

¹⁸² Yet another ordering suggested by one consultant was *súm-thamba ódi t'ep bompu*.

¹⁸³ This form uses the superlative suffix -co2 and hence formally resembles adjectives. It can, however, be used as an independent pronoun 'most'. On this distributional basis, the word is here listed as a pronoun.

(4.38) 5' à' ANNIST' 5' NAN (4.38)

t'a [*mí t*^{*h*}*amtce?*] *halede:*. now human all be.surprised 'Now all the people were amazed.' (SGD cave story)

- (4.39) $\int \dot{x} \, \bar{x} \, \bar{\alpha}^{\dagger} \, \bar{x} \, \bar{x} \, \bar{\alpha}^{\dagger} \, \bar{\alpha}^$
- (4.40) rightarrow single a transformation <math>rightarrow single a gatsi martam=gi [mi tsha:le] lèm jò?. 1PL.GEN TPN=GEN human all good EX.PER 'All people of our Martam are good.' (KN e)

The use of the quantifying pronoun *dzaŋki* is in my data limited to a few elicited examples from a speaker from Lachung:

- (4.41) $a\hat{\gamma} \in \bar{\gamma}\hat{\eta}^{*} a\nabla \eta^{*} \tilde{\eta}\eta^{*}\tilde{\eta}$ $[\underline{di} \quad dza\eta ki] \quad ba? \quad c\dot{o}:=lo.$ this all carry come=REP 'Bring this all, it was said.' (KUN e)
- (4.42) $\hat{a} \cdot \hat{x} \cdot \hat{x} \cdot \hat{\eta} \cdot \hat{\eta} \cdot \hat{\eta} \cdot \hat{\eta} \cdot \hat{\eta} \cdot \hat{y} \cdot$
- - b) היפּחִי פֶּּקִיהָּלָיחִי הַאַילֹהִי וְמִיּחְאִי אֹק׳ היהָהיזַה׳
 *חָמׁנְהַמִי פָּ*אָי פָּּקיהָילָיחִי הַאַדּיבּוּ ו<u>ווי</u> לאַז היהָהיזַה׳
 חָמׁנָהַמַ וְבָּחָקָבּ:=tsu=gi [<u>t'ytshø?</u> khace?] lèn tã:-ruŋ
 1PL PRN.HON=PL=GEN time some take send-CONC
 'Even if we take some of your time...' (BLA Namdol)

(4.45) هَ[·] مِن² جَمَ[·] <u>mí</u> **làri?** human some 'some people' (CY interview)

- (4.46) אַיאָליאָי מוּאָשָאָד מימי מימיאוֹאָי מָלָי p'umy:=gi [<u>néntsʰɛ̃:</u> **làla**=gi=di] girl.GEN=GEN relative some=AGT=DEMPH 'Some(one) of the girl's relatives (says):' (SGD wedding customs)

The pronoun *ripi (ripi)* occurs in my data only in independent use, see §6.3.1, but presumably it may also modify a noun.

The versatile quantifying adverbs $k\epsilon:po$ 'a lot, many', manpu/manpo 'a lot, many' and pannun 'little, few', which occur as verb modifiers (see §5.6.3.4), can also be used for noun modification:

- (4.48) $\alpha \in \mathbb{R}^{n}$ $\widehat{\mathbb{R}}^{n}$ $\widehat{\mathbb{R}^{n}$ $\widehat{\mathbb{R}}^{n}$ $\widehat{\mathbb{R}}^{n}$ $\widehat{\mathbb{R}}^{n}$ $\widehat{\mathbb{R}}^{n}$ $\widehat{\mathbb{R}}^{n}$ $\widehat{\mathbb{R}^{n}$ $\widehat{\mathbb{R}}^{n}$ $\widehat{\mathbb{R}^{n}$ $\widehat{\mathbb{R}^{n}}$ $\widehat{\mathbb{R}^{n}$ $\widehat{\mathbb{R}^{n}$ $\widehat{\mathbb{R}}^{n}$
- (4.49) ך: רְיָשׁרָאַי חְאִידִי שָׁקִידִי שָּקן t'a t'atei? [<u>tam</u> **kɛ:po**] t^høm-bo bɛ?. now recently speech many happen-2INF EQU.NE 'Just now, there was a lot of speaking.' (KL BLA 12)

Note, however, that in (4.49) $k\epsilon:po$ could be analyzed two ways, either as modifier of tam ('[many words] happened') or as a complement of the verb $t^{h} on$ 'happen, become' ('[words] became [many]').

(4.50) نَا ثَلْتُ عَلَى اللَّهُ اللَّهُ عَلَى اللَّهُ اللَّهُ اللَّهُ اللَّهُ اللَّهُ المَّاتِ اللَّهُ اللَّ اللَّهُ اللَّ اللَّهُ اللَّالَ اللَّهُ اللَّهُ اللَّهُ اللَّهُ اللَّهُ اللَّالَةُ اللَّا اللَّهُ اللَّهُ اللَّهُ اللَّهُ اللَّهُ اللَّهُ اللَّهُ الللَّا اللَّهُ اللَّهُ الللَّا اللَّالَةُ اللَّالَةُ اللَّالَةُ اللَّالَةُ اللَّالَةُ اللَّ اللَّ اللَّالَةُ اللَّالَةُ اللَّالَةُ اللَّالَةُ اللَّ اللَّالَ اللَّالَ لَلْحُالَةُ اللَّالَةُ اللَّالَةُ اللَّالَةُ اللَّالَةُ اللَّالَةُ الللَّالَةُ اللَّالَةُ اللَّالَةُ اللَّ اللَّالَةُ اللَّالَ اللَّالَ اللَّالَةُ اللَّالَةُ اللَّالَةُ اللَّالَةُ اللَّالَةُ الْحُلَيْ لَاللَّا الللَّا الللَّا اللَّا ا

¹⁸⁴ 50 paisa is equivalent to 8 *ána*. One *ána* is thus 6,25 paisa (1 paisa is a hundredth of a rupee).

The specific numeral and $k'\tilde{a}:pu/k'\tilde{o}:pu = 1$ for a consultant from Lachung (the first instance in [4.52]). While $k'\tilde{a}:pu/k'\tilde{o}:pu$ is a frequent and geographically widely used morpheme, the related form k'o:m = 1 form $k'\tilde{o}:m = 1$ form $k'\tilde{a}:pu/k'\tilde{o}:pu$ occurs only once in my data from a consultant from Lachung (the first instance in [4.52]). While $k'\tilde{a}:pu/k'\tilde{o}:pu$ occurs as an emphatic modifier of the words meaning 'all', i.e. $t^{h}amtee^{2}$ ($k'\tilde{a}:pu$), $k^{h}e:le$ ($k'\tilde{a}:pu$), $te^{h}a:le$ ($k'\tilde{a}:pu$) 'entirely all', in the sole example k'o:m modifies an ordinary noun. The head word of $k'\tilde{a}:pu$ may either occur independently as a quantitative pronoun (the second instance in [4.52]) or as a noun modifier (4.53).

- (4.52)[néntsẽ: k'oːm], átcu, ái, ádzo, $p^{h}amin$, relatives full elder.brother elder.sister grandfather father's.relative [kʰɛːlɛ $k'\tilde{o}:pu=gi$] ta:- $\varepsilon\varepsilon$ k'an-run j*à-pats*e p'ia-run bε?. what-CONC EX-COND do-CONC entirely full=AGT append-INF EQU.NE 'All relatives, elder brother, elder sister, grandfather, father's side's relatives whatever (relative) is there, they all offer (a ceremonial scarf).' (LA intro to Lachung)

4.1.3.4 Demonstrative modifiers

Distal *ódi*, proximal *di* and emphatic proximal *dodi* were above shown to occur as prenominal modifiers. They also occur as postnominal modifiers:

- (4.55) $\mathfrak{g}^{\mathsf{r}} \mathfrak{g}^{\mathsf{r}} \mathfrak{g}} \mathfrak{g$
- (4.56) জ'ন্ ক্রান্স জ'নেন্ট' <u>ána</u> gɛ:m ódi old.lady old.woman that 'that old lady' (rna-gsung 6)

Other postnominal demonstrative modifiers are $p^{h}ou=di/p^{h}idi$ $\arg (a_{1}^{2},a_{2}^{2},a_{3}^{2},a_{$

(4.58) वृत्तः क्षेप्तत्ते का जुः

 $cin p^{h}idi = lo$ árutree that.over.there=DAT peach 'that tree over there (has) peache(s)' (PD surroundings video)

4.1.3.5 Post-head relative clauses as modifiers

Noun phrases can have a modifying post-head relative clause. Example (4.59) illustrates such a use.

(4.59) الج المَح المَح

<u> $n\dot{a}mu$ </u> $\dot{o}di$ [$k'jank^ha$ (kjap) $ma-ts^hu-po=di$] camel that counting (do) NEG-be.able.to-2INF =DEMPH 'those camels, which could not be counted' (PAD bet story)

A fuller treatment of post-head relative clauses can be found in §13.2.1.2 and §13.2.2.2.

4.1.4 Noun-phrase-final clitics

Clitics attach to the noun phrase after word-level modifiers. The order of clitics is (=PL) (=CASE) (=AEMPH/CEMPH) (=DEMPH) (=TOP) (= $j\tilde{a}$: 'even, too, yet'). Examples (4.60-67) provide evidence for this ordering. Typically only one of the clitics =AEMPH/CEMPH, =DEMPH, or =TOP occurs in a noun phrase, but some combinations are possible, see (4.64-65). For the complexity of analyzing =DEMPH in relation to the proximal demonstrative *di*, see §16.1.3. The plural clitic =*tsu* is treated separately in §4.1.5.

NOUN=PL=CASE= $j\tilde{a}$:

(4.60) तुवार्ङ्जा पट वयात् ह क्रुया चेंत्र

 $n \hat{u}m = tsu = lo = j\tilde{a}$: younger.sister.of.a.woman=PL=DAT=too bed=LOC tea bring give '(She) also brings tea to bed for (her) younger sisters.' (Richhi 5)

NOUN=PL=CASE=DEMPH

t'indy:=gi p'jaby:=tsu=lo=di present=GEN child=PL=DAT=DEMPH 'For children of the present day...' (YR canteen video)

NOUN=CASE=AEMPH

(4.62) ८ ८ ८ २ ४ में राष्ट्र में राष्ट्र के राष्ट्र

ŋà t'ato maisur=lo=rã: jò?.
1SG now TPN=DAT=AEMPH EX.PER
'I'm now in Mysore (itself).' (Richhi 56)

NOUN=CASE=TOP

(4.63) אָק־יאָז' מיק־יאָז' מיק־יאָז' (Nep.) אָק־ימאז'קֿ' מֿאַז'אָד (Nep.) *zuŋlu làp-øtcɛnɛ t'a nɛpali kɛ:=lɛ=nɛ loggit.* folksong say-COND now Nepali language=ABL=TOP folksong(Nep.) 'If (we) say [zuŋlu], now (that) is in Nepali language [loggit].' (RS)

It is quite rare for two emphatic clitics to occur contiguously but at least the combinations =AEMPH=DEMPH (4.64) and DEMPH=TOP (4.65) are possible:

NOUN=AEMPH=DEMPH

(4.64) **क्रॅ**'वर्ने'रूट' वर्ने'

ódi=rã:=di that=AEMPH=DEMPH 'that indeed' (YR canteen video)

NOUN=DEMPH=TOP

(4.65) रोग्रया पर्ने वे यानवा योन पर रहें पर्युर

 $s \epsilon m = di = n \epsilon$ $t \epsilon nme?$ $p^h a$: $t s^h u$: $p^h u$:mind=DEMPH=TOPimpermanencethitherhitherfly'The mind flies here and there restlessly...' (Richhi 86)

The clitic = $j\tilde{a}$: 'even, too, yet' is frequently postposed to =DEMPH, see (4.66).

NOUN=DEMPH= $j\tilde{a}$:

(4.66) พ⁻นั่ง คราพรา

ápo=di=jã: father=DEMPH=too 'also the father'

In my data, $j\tilde{a}$: also occurs attached to $=r\tilde{a}$:, see (4.67). However, in all the instances $=r\tilde{a}$: functions as a simple reflexive (see §6.2) rather than in the more grammaticalized function of =AEMPH (see §16.1.1).

NOUN=REFL= $j\tilde{a}$:

(4.67) מישאל הפיצר אבישרי אמי איאר אלי

 $m \dot{a}t \varepsilon^{h} \tilde{\varepsilon}: zi - po$ $k^{h} \tilde{o}:=r \tilde{a}:=j \tilde{a}:$ $s \dot{a}m \quad s \dot{a} - \varepsilon \varepsilon \quad \tilde{i}:$. cook four-COL 3PL=REFL=too food eat-INF EQU.PER 'The four cooks themselves also eat food.' (Richhi 90) In addition to the emphatic clitics, the attention marker =co, which is a clausal clitic, can also attach to a noun phrase to mark a topic-switching question, see (4.68). Even in its phrasal use, =co retains its clausal feature of being followed by a pause. For more on =co, see §16.2.2.

(4.68) 「 報句 花句· 貴子 マモデャ ダヤデデ ෯句 方· 정· 教子 중· 教子 ?
 ŋà mémpø: dzã:dar p 'ja-do î:. tɛ p 'usim=tsu=co? 1SG physician.GEN training do-IPFV EQU.PER then younger.sister=PL=AT
 'I'm doing physician's training. And what about the sisters then?' (Richhi 127)

Whereas the emphatic clitics are discussed more in detail in §16.1, the plural marker is described here.

4.1.5 Plural marker =*tsu*

The plural marker =tsu is an enclitic, which attaches to the last word of the noun phrase, which may either be a noun, as in (4.69), or a noun modifier, as in (4.70) and (4.71).

- (4.70) ই শ্র্র্যান্তার্ন্ত do bompu=tsu stone big=PL 'big stones'
- (4.71) 美· 對和识 여子爱 do bompu di=**tsu** stone big this=PL 'these big stones'

The plural marker is frequently elided when plurality is otherwise obvious from the context. In (4.72) and (4.73), the plural marker is dropped with a numeral and a quantifier respectively, but (4.74) retains the plural marker despite the quantifier.

(4.72) র্নার্স্টা বাদ্যারা

p'otso súm child three 'three children'

(4.73) *ई* ग्रेयन

do $k \epsilon p$ stone a.lot 'a lot of stones' (4.74) विद्यंगी द्वियादा देवार्दु गोषार्थ वर्त्तगांगी

 $k^{h}on=gi$ $k^{h}im=na$ $t'\varepsilon p=tsu$ $k\varepsilon:po du-k\varepsilon.$ 3SG.HON=GEN house=LOC book=PL a.lot EX.SEN-IN 'There are a lot of books in his house.' (YR e)

In a list of items, a final plural marker has the meaning 'and so on; and such things; et cetera' (cf. WT *la-sog-pa* 'et cetera'), see (4.75).

(4.75) याबेसार्रे इसार्ये र्डु यावगाया

ze:ro tsampo=tsu zak-sa rice.snack flour=PL put-SPAT.NMLZ 'a place to put zero (traditional snack made by frying strips of thin rice dough on a pan), flours and such things' (PD, living room video)

The plural marker =*tsu* can also attach to an infinitive form of a verb:

(4.76) अष ५६'गीष पत र्डाष न, खुरा छे ते राख्या रेत राखे थे राष्ट्र छ्या ५ क्यु छ्या ५ क्या न, छुं *iy:=di=gi k'an p'ja-ce? cù-watee nàtea? rimpute^he=lo*body=DEMPH=AGT what do-INF say.HUM-COND 1PL Rimpoche=DAT *gjømpa gju te^ha:ts^he:-ce=tsu*monastery go prostrate-INF=PL
'When it comes to what to do with the body, (we should practise) going to monastery to Rimpoche and doing prostrations and such things.' (YR canteen video)

Mass nouns such as $tc^h u$ 'water' may receive plural marking.

(4.77) $a\hat{\zeta} = a \bar{\zeta} \cdot a \bar{$

The plural marking in (4.77) could suggest spatial limitation, iterativity (i.e. the many different occasions when water is poured into the cauldron) or a meaning similar to that in (4.75) 'and such things'.

The plural =tsu may also attach to personal names to refer, similarly to the suffix -po/bo in §4.2.1, to the person and those associated with him/her:¹⁸⁵

(4.78) দ্রুন'অর্চ্জর্ ব্র্র্ gjalts^hen=**tsu** PN=PL 'Gyalchen and those associated with him' (KN kitchen discussion)

¹⁸⁵ In Nepali, the plural marker *-haru* is used similarly.

4.1.6 Definiteness and indefiniteness

Definiteness and indefiniteness may be expressed, respectively, by the demonstrative-emphatic =di and the indefiniteness marker $=t\epsilon i(2)^{186}$, which is homophonous with $t\epsilon i2$ one', although the vowel quality in the numeral tends to be pronounced longer. The semantic field covered by =di extends beyond typical definitions of definiteness because =di can also mark noun phrases which are by definition definite, e.g. personal names and personal pronouns. Moreover, the lack of the demonstrative-emphatic =di does not equate lack of definiteness, i.e. nouns that are semantically definite are not necessarily marked by =di. For a description of the uses of =di, see §16.1.3.

The fact that the indefinite marker is distinct from numeral $t\epsilon i 2$ v is shown by the following examples, where $=t\epsilon i(2)$ follows the quantifier $k\epsilon : po$ 'many, a lot' (4.79), the numeral $t\epsilon i 2$ (4.80), other numbers (4.81) and the plural marker (4.82).

- (4.79) $\alpha \in [\alpha, \beta] \subset [\alpha]$ $(\alpha, \gamma) \in [\alpha]$ $(\alpha, \gamma) \in [\alpha]$ dzamlin=gi $\dot{y}: k\varepsilon:po=tci=na$ world=GEN place many=INDF=LOC 'in many places in the world' ('dzam-gling skad-yig 62)
- (4.80) אראי אַזאַי און און איז אַז' און און איז אַז' און און איז אַז'ן און איז אַז'ן און איז'ן און איז'ן און איז'ן איזאין איז'ן איז'אין איז'אין איזאין איז'ן איז'ן איזאין איז'אין איזאיזאין איז'אין איזאיזאין איז'אין איזיאין איזאאין איזאין איזיאין איזאיאאיאאיזאין איזאיאין איזאין איאין איזאיאיאין איאין איאיאין אי
- (4.81) $\underbrace{\Re}_{\nabla} \operatorname{Rav} \underset{\cong}{\mathbb{R}} \underbrace{\Im}_{\nabla} \operatorname{Rav} \underset{\cong}{\mathbb{R}} \operatorname{R$
- (4.82) $\omega \in \tilde{\alpha} : \tilde{\alpha} : \tilde{\alpha} : \tilde{\omega} : \tilde{\omega$

The indefiniteness marker may be followed by case marking, as in (4.79) and the first instance of $=t\epsilon i i$ in (4.83). In the second instance in (4.83), $=t\epsilon i i$ obtains the meaning 'some'.

In indefinite expressions such as 'whoever' and 'whatever', which are formed from an interrogative word and a concessive form of the equative \tilde{t} , the indefiniteness marker may occur either attached to the interrogative word (4.84) or the concessive suffix (4.85).

¹⁸⁶ The glottal is represented in the phonemic transcription word-finally (when a potential pause may occur) but not if the indefinite marker is followed by case marking.

- (4.84) $\eta' \hat{\prec}' \eta \hat{\otimes} \eta' \hat{\otimes} \eta' \hat{\otimes} \eta' \hat{\prec} \hat{\prec}'$ $k' ar \varepsilon = tci ? \tilde{i}: ru\eta$ what=INDF EQU-CONC 'whatever' (KT discussion with TB)
- (4.85) מוק׳ אלא ישר׳ מושליי אלא ישר׳ שושיי אלא ישר׳ מושיי אלא ישר׳ שושיי אלא ישרי אלא ישרי אלא ישרי אלא ישרי אלג ווייע אומד בעט-CONC=INDF happen-COND 'whatever happens' (PED life story)

4.1.7 Coordination of noun phrases

Coordination of noun phrases may be "asyndetic" or "monosyndetic" (see Haspelmath 2007: 6). In an asyndetic construction, there is no overt connector but coordination is accomplished through juxtaposition. Mere juxtaposition is especially used with elements which frequently occur together:

 $k^{h}im = na$ ápo ám $k^{h}jo:m$ dik-ti house=LOC father mother gift arrange-NF 'At home, the father (and) mother arrange a gift (and)...' (SD wedding customs)

Monosyndetic coordination uses the connectors $t'\tilde{a}$: 'and' for inclusive coordination and $j\tilde{a}:n\varepsilon$ 'or' for presenting options:

(4.87) क्रेंबा क्षेंच, नगम कें नम भ्रम अ

 $tc^{h} \phi d\tilde{\epsilon}$; $\dot{\phi} \eta m u$ $t'\tilde{a}$: karma PN PN and PN 'Choden, Wangmo and Karma' (Richhi 2)

(4.88) ८'र्थे, रूट'र्थे' थट के' विट थे'

 $\eta \dot{a} = lo, \quad r \tilde{a} := lo \quad j \tilde{a} : n \varepsilon \ k^{h} \tilde{o} := lo$ $1 \text{SG} = \text{DAT} \quad 2 \text{SG.M} = \text{DAT} \text{ or } \quad 3 \text{S.HON} = \text{DAT}$ 'To me, you or him' (KT discussion with TB)

The connector $j\tilde{a}.n\varepsilon$ 'or' is also frequently used for introducing alternative dictions and therefore in meaning resembles 'in other words'. In (4.89) the speaker gives an interpretation of an old poetic expression (mother's body's curd is the milk from her breasts) and in (4.90) the speaker provides the Denjongke equivalent of an English word.

(4.89) क्षेंग्रा यादावें महायी युषायी कें

óm $j\tilde{a}:n\varepsilon$ $ra\eta=gi$ $l\tilde{y}:=ki$ $\varepsilon \delta$ milk or own=GEN body=GEN curd 'milk, or (her) own body's curd' (SGD wedding customs) (4.90) קדי לאדי (Buddhist) שריאי אריאי מריאי שלי איז) אריאי איז (4.90)

buddis $j\tilde{a}:n\varepsilon$ $nan-p\phi:$ $tc^h\phi:=ki$ nance=loBuddhist(Eng.)orinside-NMLZ.GEN religious.teaching=GENinside=DAT'within Buddhist, or insiders' teaching...' (KTL life story)inside=DAT

4.1.8 Reduplication of noun phrase

Reduplication of the noun phrase (4.91) or in the noun phrase (4.92) marks iteration or distributive function:

(4.91) *कें*' कें' कॅर' रे'

 $l\hat{o}$ $l\hat{o}$ ts^ha : $r\varepsilon$ year year turn one 'once every year' (KN e)

(4.92) हि'रुग' पक्तु'रे' पक्तु'रे'

tiru? gja-ri gja-ri rupee hundred-one hundred-one 'one hundred and one hundred rupees (=a hundred rupees each)' (DB day trip)

4.2 The verb complex

This section first introduces the structure of the verb complex (§4.2.1) and then describes complex verbs, namely phrasal verbs (§4.2.2) and serial verbs (§4.2.3). The last section gives examples of various types of combinations of elements in the verb complex (§4.2.4). In the ensuing discussion, "complex verb" refers to verbs consisting of more elements than one verb root (either phrasal verbs or serial verbs) and "verb complex" refers to the verb and all verbal suffixes and auxiliaries accompanying a verb in a clause.

4.2.1 Structure of the verb complex

The structure of Denjongke verb complex is presented in Figure 4.2, using the following abbreviations:

Ν	=	Noun (in a phrasal verb)
NEG	=	Negator prefix
V1	=	Primary verb, may consist of several verb roots forming the semantic core of
		the serial verb construction, as in (4.99)
RDP	=	reduplication of the verb stem, occurs in some constructions
V2	=	Secondary verb, which does not belong to the semantic core of the primary verb
		but which brings nuance to the primary verbal meaning
TAM	=	Tense/aspect/modality marking suffix (see §8)
AUX	=	Equative auxiliary (\tilde{t} , $b\epsilon$?) or existential auxiliary ($j\dot{a}$?, du ?), also in complex,
		nominalized forms as in (4.135), may occur cliticized as in (4.132)
NMLZ	=	Nominalizer - $k^h \tilde{\epsilon}$:
INF	=	Infinitive marker <i>-po/bo</i> or <i>-cc</i> ?
CFC	=	Clause final clitic(s) = co , = la , = lo , = $s\varepsilon$, = ki (see §3.7.3)

Figure 4.2. Structure of the verb complex in declarative mood

(N) (NEG)	V1	(-RDP)	(NEG)	(V2) [((=)AUX) -TAM -NMLZ -INF	(AUX) AUX AUX	(=CFC)	

In Figure 4.2, the elements in brackets are optional and items without brackets obligatory. Thus the only obligatory element is the primary verb. The four sequences of items within the curly brackets present options, i.e. a verb root may be followed by $((=)AUX)^{187}$, -TAM (AUX), -NMLZ AUX or -INF AUX. Note that the elements within the curly brackets are attached either to the primary verbal expression V1, if there is no nuance-introducing secondary verb, or to the secondary verb V2, if there is one. Various combinations of the items in Table 4.2 are illustrated in the last section §4.2.4, after the complex verbs have been first introduced.

4.2.2 Phrasal verbs

Complex verbal expressions, or briefly complex verbs, are here divided into two categories. The first is phrasal verbs (discussed in this section), in which a noun forms a verbal expression usually together with a semantically bleached verbalizer but also with other verbs. The second category is serial verbs (see §4.2.3.), in which two to four verb roots occur adjacently without intervening morphology.

Denjongke, like many other Tibetic languages, has plenty of complex predicates where a verb is accompanied by another preceding element, usually a noun but sometimes also an adjective or a verb. Denwood (1999: 109) calls these complex predicates "phrasal verbs." Tournadre & Dorje (2003: 204) use the term "compound verbs." Bartee (2007: 143) makes a distinction between "compound verbs" and "phrasal verbs," the latter being formed with verbalizers. Because the term "compound verb" may invoke the idea of two verb roots joined together (termed here "serialization"), I apply the term "phrasal verbs" to all combinations of a noun with a verb.

The main verbal element in phrasal verbs is often a semantically rather vacuous verb, which may be called a "light verb" (Jespersen 1965: 117) or "verbalizer" (Denwood 1999: 109, Bartee 2007: 143, Tournadre & Dorje 2003: 403). The most common verbalizers in Denjogke are $p'ja \leq 100$ 'do', kjap = 100 (do', kjap = 100 (do', tap = 100), 'send, do', tap = 100 (strike', $t\tilde{a}$: tap = 100 (strike', $t\tilde{a}$: tap = 100) (strike', $t\tilde{a}$: tap = 100 (strike'), $t\tilde{a}$: tap = 100 (strike'), t

¹⁸⁷ Brackets around the clitic marker in ((=)AUX) refer to the fact that the auxiliary may occur as cliticized or as a separate word (e.g. equative copula = $p\epsilon^2$ vs. $b\epsilon^2$)

p'ɛtɕø? p'ja	নিন্'র্ষ্ট্রন' র্রুম'	'use' (lit. 'use do')
jó? p'ja	শর্মিশ' র্ভ্রন্ধ'	'work' (lit. 'work do')
à: kjap	पट. भीयका.	'lie' (lit. 'lie do/strike')
daku kjap	নদ্বা'শ্য' ক্রদেঝ'	'govern' (lit. 'owner do/strike')
nóːsam tãː	धर्बे नषधा यहरा	'think' (lit. 'thought send')
jàrgɛ tãː	धर क्रेश यहर	'develop, progress' (lit. 'progress send')
mélam tap	ষ্ক্র্র্ব'এম' বদ্ব'	'pray' (lit. 'prayer sow')
só tap	র্ষা বদুবা	'bite' (lit. 'tooth strike').

Table 4.3. Some phrasal verbs formed with the verbalizers *p'ja*, *kjap*, *tã*: and *tap*

Table 4.4. exemplifies some other phrasal verbs, in which a noun and a verb have a strong collocation but in which the verbal element is rarer and not semantically bleached, and therefore not a typical "verbalizer."

Table 4.4. Some other phrasal verbs

sém ga:	শ্রপ্রধ্য প্রদান.	'rejoice' (lit. 'mind rejoice')
k'jøpo to:	র্গ্রন্'র্থ' স্থ্রিগব্য	'be hungry' (lit. 'stomach be.hungry')'
k ^h a kom	या र्झ्रेका	'be thirsty' (lit. 'mouth dry')
go tsuk	নর্যাঁ: দার্ন্তুবাঝ'	'begin' (lit. 'start plant'),'
ŋò cé:	हे. चेहा.	'know (a person)' (lit. 'face know')
mèːp zo	ਕੇਨ੍ਨਾ ਨੜੇਂ	'destroy' (lit. 'not-existing make')

Phrasal verbs function as units is terms of their prosodic unity, but there is also some syntactic separateness in the components of compound verbs in that intervening elements may occur between them. Example (4.93b) illustrates an intervening negator ma. The adverbial modifier $l\epsilon p$ 'very (much)' may occur preceding the compound (4.94a) or within it (4.94b).

(4.93) a) สับ จุธุรุเ

só tap tooth strike 'Bite!'

b) র্ষা ঝাবদুবা

só ma-tap tooth NEG-strike 'Don't bite!'

(4.94) a) अन्न में के राज क

 $d\tilde{a}$: $\eta \dot{a}$ $l \dot{\epsilon} p$ $s \dot{\epsilon} m$ g a - u \dot{t} :. yesterday 1SG very.much mind rejoice-2INF EQU.PER 'Yesterday I was very happy.' (KN e) b) אקה׳ ה׳ ה׳ קזוְה׳ ה׳ הַזוֹן
 dã: ŋà sém lèp ga-u í̇́:.
 yesterday 1SG mind very.much rejoice-2INF EQU.PER
 'Yesterday I was very happy.' (KN e)

Typically only one of the verbalizers may occur in a certain compound, but some constructions allow the use of two different verbalizers with no noticeable difference in meaning.

 (4.95)
 a) jó? p'ja/kjap
 ขุณัญ รู้สุง/ภูรุจง' 'work'

 b) jàrgɛ tã:/p'ja
 พระสูง ราธุร:/ รู้จง' 'develop (tr.)'

 c) cúk kjap/p'ja
 ศูญญา รูรุจง'/รู้จง' 'exert force'

Some bivalent compounds may be reduced to monovalent ones by using the verb $t^h @n$ 'become':

(4.96) a) jàrgɛ tã: المجتوم: 'develop (tr.)'
 b) jàrgɛ tʰøn المجتوم: 'develop (intr.)'

The verbalizer p'ja is commonly used in borrowing verbal expressions from other languages such as English.¹⁸⁸ The following three expressions were used in a public speech (English ortography in square brackets).

(4.97)	a) [changes] <i>p'ja</i>	'make changes'
	b) [follow] <i>p'ja</i>	'follow'
	c) [message pass] p'ja	'pass on a message' (NAB)

Although most phrasal verbs are transparent in that the speakers are aware of the individual meanings of the composite parts of the verb, some phrasal verbs, such as *hako* (or *ha k'o*) 'know, understand' are non-transparent in that speakers are not aware of the meaning of *ha*, although *k'o* has the independent meaning 'understand'.

4.2.3 Verb serialization

In verb serialization, two to four verb roots, which could occur independently¹⁸⁹, occur adjacent to each other without any intervening morphology, see (4.98-101).

¹⁸⁸ An analogous process takes place in Nepali with the verbalizer *garnu* 'do to'.

¹⁸⁹ A verb which can occur independently as an ordinary verb participates, when occurring with other verbs, in "serialization." Further stages of grammaticalization in which the verb no longer functions independently as an ordinary verb may be called "auxiliarization" and "morphologization" (DeLancey 1991: 2). Verbs that have both independent and auxiliary uses, as the secondary verbs here do, can be called "versatile verbs" (Matisoff 1969, 1973, Delancey 1991).

- (4.99) التركَّ ثَمَّا عَيْمَ حَدَيْقَ عَلَى اللَّهُ عَلَى الْحَامَةُ عَلَى اللَّهُ عَلَى اللَّهُ عَلَى اللَّهُ عَلَى اللَّهُ عَلَى اللَّهُ عَلَيْ اللَّهُ عَلَى اللَّا عَلَى الْحَالَى الْحَالَى الْحَالَةُ عَلَى اللَّا عَلَى اللَّهُ عَلَى الْحَالَى الْحَالَةُ عَلَى الْحَالِي عَلَى اللَّا عَلَى الْحَالَةُ عَلَى الْحَالِ عَلَى الْحَالَى اللَّالَ عَلَى اللَّالَةُ عَلَى الْحَالَةُ عَلَى اللَّالَ عَلَى اللَّهُ عَلَى الْحَالَةُ عَلَى اللَّهُ عَلَى اللَّالَ عَلَى الْحَال اللَّا عَلَيْ اللَّا عَلَيْ عَلَى اللَّالَةُ عَلَى اللَّا عَلَى اللَّا عَلَى اللَّا عَلَى اللَّالَةُ عَلَى اللَّ المَا عَلَيْ عَ
- (4.100) $\operatorname{cd}_{\mathcal{A}}$ $\operatorname{cd}_{\mathcal{A}$
- (4.101) \check{a} \check{a} \check{c} \check{a} \check{c} \check{a} \check{a}

When asked how (4.101) would be negated, consultant KN offered the form in (4.102), which retains only one (the most central) verb from the affirmative construction. The number of verbs in a negated serial verb construction in my data is never more than two.

(4.102) (אַ הַרְיָדִי אַ שִרָאַ אַרָן *odera* **mi-t**^ha:-k^hen be?. like.that NEG-be.released-NMLZ EQU.PER 'Like that, (they) are not released.' (KN e)

These serial verb constructions (SVC) have most likely developed from nonfinal converbal constructions (see §15.2) where the converbal ending has been dropped (see DeLancey 1991: 4). Table 4.5 lists the most common combinations of two verbs which occur in both converbal and serialized constructions. The latter verb in these combinations is, with the exception of $mj\tilde{o}$: 'finish' a verb of motion.

¹⁹⁰ According to consultant KUN, an agentive form tci=ki would have been expected in this clause.

Converbal	Serialized	Meaning
k^h ik-ti $\hat{ec{O}}$ ్ ప్రేశ్రాశ్రీ దేవా	$k^{h}ik\widetilde{\!\!\mathcal{O}}$: By $\widetilde{\!\!\mathcal{O}}$	bring, lead (a human or a large animal) (lit.
		lead come)
kʰik-ti gju שַר אין	k ^h ik gju খ্রিয়' নেয়ু'	take, lead away (a human or a large animal)
	(past k ^h ik số: छिंग' कॅर्न्र')	(lit. lead come)
bak-ti ö́: འདག་སྲི་ འོང་	bak o: 2547 Az	bring (a thing) (lit. carry come)
bak-ti gju ଘଟ୍ସମାଂଞ୍ଚି' ଘଣ୍ଡୁ'	bak gju רְאָרָא Tag'	take away (a thing) (lit. carry go)
	(past <i>bak sốঁ:</i> রন্যা ⁻ র্মন ⁻)	
<i>lòk-ti ồ̃: حَ</i> مَّر بَجَ، حَر	lòk ổː མོག་ མོང་	return, come back (lit. return come)
lòk-ti gju མོག་ཐི་ འགྱུ་	lòk gju র্ন্যা নয়ু	go back (lit. return go)
	(past <i>lòk số</i> : कॅप्प र्स्न)	
lồ้:-ti gju ลัธาริ่า ลมูา	lồ้: gju ลั⊏ ลุฏู	go away (lit. stand go)
	$(\text{past } l \hat{o} : s \hat{o} : \tilde{a} = \tilde{a} =)^{191}$	
VERB-ti mjồ: -ह्रे' र्ज्जूद	VERB mjồ: र्ज्जु≒	finish doing something (lit. VERB finish)

Table 4.5. Serial verbs from converbs

In negation, the negator occurs in between the verbs, e.g. $l\partial k mi \cdot \partial z$ 'does not return', *bak maso*' 'did not take (away)'. This feature violates Haspelmath's (2016) criterion 6 for serial verb constructions, which states that the negator element in SVCs occurs "preceding the first verb or following the last verb."

Examples (4.103) and (4.104) further illustrate the irregularity of nonfinal marking in a series of three verbs. Both examples, occurring in the same story, use the same three verbs but place the nonfinal marker in different places. Note that both examples resemble the SVC in (4.100), which has no nonfinal marking.

 $r\tilde{a}:d\tilde{o}:$ t^hu-ti bak $\tilde{o}m$ -bo b ϵ ?. basket pick-NF carry come-2INF EQU.NE '(He) took and brought the box.' (Class 8 textbook 14)

> basket=DEMPH what do-NF pick carry-NF come-IPFV 'Why are (you) taking and bringing the box?' (Class 8 textbook 15)

In serial verb constructions (SVC), the verbs may have three types of relationship to the other verbs in the construction. First, a component verb may be part of the semantic core meaning of the SVC. This is the case with the SVCs in Table 4.6, with the exception of $mj\tilde{o}$: 'finish', which marks completive aspectual information. For instance, the act of bringing (*bak* \tilde{o} :) essentially consists of

¹⁹¹ Consultant KN (Martam, East Sikkim) comments that in Tashiding (West Sikkim) $j\hat{a}$: gju (lit. go go) is used for 'go away' instead of $l\hat{o}$: gju.

Second, the SVC may consist of a combination of a verb that provides the semantic core of the expression and a semantically bleached verb, verbalizer, which provides not much more information than that the expression is a verb. For examples, refer to (4.105-108), where (a) exemplifies the verb without a verbalizer and (b) a use with verbalizer.¹⁹³

(4.105)	a) המיחן אָה מקרימי הה שׁק־שֿמידָיחן? nè:=gi duŋ=di=lo rã: jìteʰɛ-to-ka? 1SG.GEN=GEN story=DEMPH=DAT 2SG.M believe-IPFV-PQ 'Do you believe my story?' (PAD bet story)
	b) װֶג: אָרָי גרֹי אֹים שֶׁרָ שָּׁת בָזוֹאי אַנים בָזוֹ אַים אַרן גאיץ: duŋ=di=lo lɛ̀p jìtɛʰɛ p'ja go:-kʰɛn bɛ?. he.GEN story=DEMPH=DAT very.much believe do be.needed-NMLZ EQU.NE '(I) have to very much believe his story.' (PAD bet story)
(4.106)	a) قَامَّا عَلَمَا عَلَمَا عَلَمَا عَلَمَا عَلَمَ عَلَي عَلَمَ عَلَي عَلَمَ عَلَي عَلَ clothes excellent excellent bring-IPFV EQU.NE '(They) bring fine, fine clothes.' (LA, birth in Lachung)
	b) तरीय क्रिय ग्रंभ रक्ष वर्ग <i>dep ke: p'ja-renk^ha</i> like.this bring do-SIM 'when (she) brought (it) like this' (RB pear story)
(4.107)	a) בו או
	b) \tilde{w} 'ਰ' 'ਚ'' ''' ''' ''' ''''''''''''''''''

'There (she) looked here and there.' (UTR plains story)

¹⁹² Note that according to Haspelmath (2016) synonymic verb combinations are not Serial Verb Constructions.

¹⁹³ Already Sandberg (1894: 76-77), similarly to Tournadre & Dorje (2003: 204) later, noted that such compound forms in Central Tibetan were preferred to the more simple forms in colloquial Tibetan. A functional factor for this preference may be that by using the longer forms the speaker gives more processing time to the addressee and avoids homonymy, which is prevalent especially in monosyllables.

t'ato t^hamtɛɛ? k^hõ: lópța **gju**-do. now all 3PL school go-IPFV 'Now they all go to school.' (PED life story)

b) राशनात् तमु मुनया तनगान्ने मी तेन हेः

ralã:=na gju kjap bak-tiki òn-di TPN=LOC go strike carry-NF come-NF 'keeping on walking and coming to Ralang' (PAD Tashiding story)

In (4.108b) the serial construction *gju kjap* 'go strike' may have a stronger association with actual walking than mere *gju* which typically refers to simply going. Moreover, in (4.108b), *gju kjap* is followed by a third serial verb, the secondary verb *bak*, which emphasizes durativity of action, see §8.3.4.

Third, in some cases the last member of the SVC is neither a verbalizer nor does it participate in forming the core lexical meaning of SVC but rather adds tense, aspect and modality related information or other semantic nuance to the SVC in question. These verbs are here termed secondary verbs, referring to the secondary nature of their semantic effect. Secondary verbs are summarized in Table 4.6, where the left-most column divides the verbs according to the construction in which the secondary verb occurs. The notion VERB refers to the primary verb in the construction, while the underscore _ refers to the position of the secondary verb. The full stop in "VERB_." reveals that the sentence may end in the secondary verb root. The asterisk * marks that the secondary verb collocates strongly with the construction in question (but is not completely limited to the construction).

The majority of the secondary verbs inflect like ordinary verbs because they also function as independent non-serialized verbs, i.e. ordinary verbs. The right-most column divides the verbs into clear tense-aspect markers (TA), clear modality markers (M), causative marker (C) and others (O). Delineation between the categories modality (M) and other (O) is by no means simple. The category modal is assigned to those markers which are in linguistics typically treated under the category modality (ability, obligation, permission, possibility).

Constr.	Verb			Function	
VERB	ts ^h a:	<u> </u>	'complete' ¹⁹⁴	completive	TA
	ò:	йт.	'come'	future	TA
VERB_EX	dø:	Ĩή.	'sit, stay'	continuous	TA
	zak	নাৰ্বনা.	'put'	resultative	TA
	lồ:	र्भत.	'have time to'	temporal abilitative	М
VERB_ ti*	bak	ন্দ্রন্যা.	'carry'	durative	TA
VERB_ $\dot{\tilde{o}}$.*	sí?	وليتر	'be possible'	possibility	М
VERB mi_*					
VERB mi_*	lè?	ন্টবাব্য'	'be good'	moral or practical evaluative	М
VERB(_to)	ren	त्रवु'	'be time to'	temporal evaluative	Μ
Like	mjõ	الجام	'finish; experience'	finished or experienced action	TA
ordinary	ts ^h u?	র্ন্ত্র্যাম্য'	'be able'	abilitative	Μ
verb	сé:	<i>ने</i> ह्य.	'know'		М
	nùm	ਗ੍ਰਕਾ	'dare'	mental abilitative	М
	k ^h ø?	র্জিন্'	'have the strength to'	physical abilitative	М
	g0? ¹⁹⁵	न्यॅायः	'need to'	obligation/deontic	М
	tc ^h o?	र्केया'	'be alright'	objective permissive	М
	tup	বদ্রুব'	'deem fitting'	subjective permissive	М
	ta	ar Ar	'look'	tentative or experimental action	М
	tsuk	নস্তৃযা	'cause'	causative	С
	gju	ন্মূ;	ʻgo'	translocativity or disappearing,	0
				gradual change towards a state	
	p'in	ਤੇੱਕਾ	'give'	benefactive	0
	t ^h op	র্ষন'	'find, receive'	beneficiary	0
	tã:	বদুদ'	'send'	literal sending; semantically	0
				bleached verbalizer; non-honorific	
				imperative marker	
	nấ:	गव्र-	'grant'	honorific imperative marker	0

Table 4.6. Secondary verbs

As shown by Table 4.6, all tense-aspect markers, with the exception of *bak* and $mj\tilde{o}$, which inflect like an ordinary verb, are unlike ordinary verbs, either being sentence-final markers (ts^ha , \tilde{o}) or being followed by an existential auxiliary ($d\omega$, *zak*, $l\tilde{o}$). Modality markers occur either in specific constructions ($l\tilde{o}$, si) or behave like ordinary verbs (e.g. ts^hu , se, go, te^ho , tup). Verbs in the category "others" behave like ordinary verbs.

While the use of tense-aspect marking and modality-marking secondary verbs are described in §8, this section illustrates the use of the secondary verbs categorized as "other" in Table 4.7. It should be noted that at least some of the verbs in Table 4.6 occur, in addition to SVCs, also in

¹⁹⁴ ts^ha : is the only grammatical secondary verb which does not occur as a primary verb (having been repleced by $mj\partial \eta$ 'finish'). It, however, has some characteristics of a verb, such as the ability to occur in nominalized constructions, see §8.1.2.

¹⁹⁵ Occasionally pronounced *ko*? when preceded by a voiceless consonant.

converbal constructions, e.g. *sà-ti mjo*² [eat-NF finish] 'finish eating', *go-ti p'in* [divide-NF give] 'divide and give' (or possibly 'divide for someone's benefit').¹⁹⁶

4.2.3.1 Translocative secondary verb *gju* (past *ső*?)

In addition to the uses where $gju \, \operatorname{qgv}$ 'go' (past $s \delta : \operatorname{scv}$) belongs to the semantic core of the verbal expression (whether as lone primary verb or a part of a SVC), it is used as a secondary verb which emphasizes translocativity (going away from the speaker) which is already expressed by the primary verb, or gradual change of state. By extension, translocativity may refer to disappearance. In example (4.109) the secondary verb $s \delta : \operatorname{occurs}$ with TAM-marking whereas in (4.110) TAM-marking is dropped.

(4.110) জাঁব এমা দ্র' এর্চ্রন মন্য

ónale $k^h u$ **bjã**: **số**:. then 3SGM disappear go.PFV 'Then he disappeared.' (KTL animal story)

Whereas in my data the secondary verb $s\delta$: is associated with literal going or somewhat metaphorical going, disappearing, Sandberg (1895: 42) reports $s\delta$: as a more grammaticalized past marker which does not have connection to literal going but may be used in phrases such as (produced in Sandberg's ortography) *zhe song* 'has done', *shi song* 'died', *t'ong song zhe* 'has seen' and (4.111) (Denjongke script, glossing and emphasis mine, Roman script from Sandberg).

(4.111) Sandberg (1895: 42) $\tilde{\Lambda}' \ \hat{u}' \hat{q}' \ q \hat{\varsigma} q^{*} \hat{\mathfrak{G}}'^{197} \ \tilde{\Lambda} \tilde{\varsigma}' \ \alpha \varsigma q \eta$ *Kho yige chi p'i song du'.* he letter INDF write go.PFV EX.SEN 'He has written a letter.'

Although example (4.110) above resembles in form Sandberg's *zhe song* 'has done', the forms differ semantically in that (4.110) involves actual going out of sight whereas Sandberg's *zhe song* 'has done' does not. Formulations such as the one exemplified in (4.111), which do not involve actual going, do not occur in my data nor are they accepted by my consultants. Borderline cases

¹⁹⁶ It is worth noting that *p'in* 'give', *zak* 'put' and *mjo*.' 'finish' in compounds are used quite analogously with the equivalent Nepali verbs *dinu* 'give', *haalnu/raaknu* 'put' and *saknu* 'finish', although in Nepali the second verb is not attached to the other root directly but after an intervening connector vowel *-i*-, e.g. *bhan-i-di-nu* [say-LNK-give-INF] 'to say (for me)'.

¹⁹⁷ The more frequent form of the word is $t'i \ge 1$. The pronunciation p'i recorded by Sandberg is here written as ≥ 1 , because the form reflects the change of the WT glide *r* into *y*.

are (4.112) and (4.113), involving the verbs $t'\tilde{o}$: 'die (hon.)' and ϵi 'die', which can be conceived as expressing a type of going or disappearing.

(4.112) $\check{a}_{1}' [aa_{1}' a_{N}] \check{a}_{1}' \check{a}_{1}' \check{c}_{1}' \check{a}_{1}' \check{c}_{1}' \check{b}_{1}' \check{a}_{1}' \check{a}_{1}'$

(4.113) \mathbb{R}^{t} $\hat{\mathfrak{R}}^{t}$ $\hat{\mathfrak{R}}^{t}$ \mathfrak{R}^{t} \mathfrak{R}^{t} \mathfrak{R}^{t} \mathfrak{S}^{t} $\mathfrak{S}^{$

The borderline status of $\epsilon i s \delta$: is seen in the fact that upon asking about the felicity of the Sandbergian form $\epsilon i s \delta$:- $z\epsilon$ [die go-PST], two of my consultants had varying responses. Whereas consultant PT (Tashiding) acknowledged the semantics suggested by Sandberg, see (4.114a), consultant KN (Martam) gave the clause a purposive reading, see (4.114b)

(4.114) a) $\widehat{\mathfrak{H}}^{*} \widetilde{\mathfrak{h}} \subset \widehat{\mathfrak{h}}^{*}$ $\mathfrak{s} \widetilde{\mathfrak{o}}^{*} - \mathfrak{z} \varepsilon.$ die go-PST '(He) died./(He) went and died.' (PT e) b) $\widehat{\mathfrak{H}}^{*} \widetilde{\mathfrak{h}} \subset \widehat{\mathfrak{h}}^{*}$ $\mathfrak{s} \widetilde{\mathfrak{o}}^{*} - \mathfrak{z} \varepsilon.$

die go-PST '(He) went (there) to die.' (KN e)

The verb gju (past $s\delta$?) often forms a SVC with a preceding $l\delta$? 'rise' with the resulting meaning 'go away'. This serial construction can also function as a complex secondary verbal construction which marks gradual change toward a state, see (4.115) and (4.116).

nø:po ín-nε=to ádzo=gi dem=sã: làklok mischief grandfather=AGT like.that=until IDEO.NN EQU-COND=CEMPH *ได้: gju-se i*. nấ:-bo=lɛ f'ak do.HON-2INF = ABL be.cured rise go-INF EQU.PER 'If (this) is (about spiritual) mischief, after the grandfather has done so many (unintelligible) incantations, (the disease) will go and get healed.' (rnam-rtog 17)

In (4.116), an elderly person has just commented on the wrinkles on his hands, comparing his skin to the smooth skin of younger people. Another person comments:

(4.116) เหาะา สัาทิสารา สัาทิสารา ขารูลา ฉลัสา สัรา ฉุญาสุกสา ธุร

lố: k'ajem t^høn ápa *l* \hat{o} =gi=ra l*ò=gi=ra* father what.is.it become rise year=AGT=AEMPH year=AGT=AEMPH gju-k^hen $b\epsilon$?. go-NMLZ EQU.NE 'Father, it's year by year that (they) go on becoming what's it (=wrinkled).' (KNU kitchen discussion)

Some western varieties of Denjongke use $j\dot{a}$: gju instead of $l\ddot{o}$: gju for 'go away'. Both of the components $j\dot{a}$: and gju mean 'go' and can be used independently. The serialization $j\dot{a}$: gju may be used similarly to $l\ddot{o}$: gju as a secondary verbal construction expressing movement towards a state:

(4.117) مَنْ اللَّهُ اللَّهُ اللَّهُ اللَّهُ عَلَى اللَّالِي الْحَلَى اللَّهُ عَلَى اللَّ

4.2.3.2 Benefactive secondary verb *p*'in 'give'

Although p in \exists_{\forall} give' is used in SVCs to refer to literal giving, see (4.118), it is also used as a benefactive secondary verb which expresses that an action is done for the benefit of some recipient, see (4.119).

- (4.118) ພໍ່ຫາສາຊ ຫຼືສາ^{.198} ພໍ່າຫ້ ຫລືສາ ຂ້ຳ ຕ້ຳລ້ຳ ພໍ່ຫ້ ຫຼືສາ ຊອງອີສ $jik^{h} \varepsilon \eta = gi$ jigi $s\dot{\varepsilon}:-di$ $k^{h} \tilde{o}:=lo$ jigi ni: $t^{h}u$ p'in. postman=AGT letter choose-NF 3PL=DAT letter two pick give 'The postman, selecting letters, picks two letters and gives (them) to him.' (Richhi 149)
- (4.119) শারা শৈদ স্থ্র জ্বারা দ্রীরা দ্রীরা দ্রীরা

karma $l\tilde{\delta}$:-*ti* gom $p^{h}i$: p'in. PN rise-NF door open give 'Karma rises and opens the door (for the one who is knocking).' (Richhi 135)

For further examples of benefactive use, consider (4.120-122), the last of which seems ambiguous between literal and benefactive reading.

(4.120) المراجعة مراجعة المراجعة المراجعة مراجعة المراجعة المراجعة مراجعة المراجعة المراجعة مراجعة مراجعة مراجعة مراجعة مراجعة مراجعة مراجعة م مراجعة المراجعة مراجعة م مراجعة المراجعة المراجعة المراجعة مراجعة المراجعة المراجعة مراجعة المراجعة مراجعة مراجعة مراجعة مراجعة مراجعة مراجعة مراجعة مليعة مراجعة مماحية مراجعة مراجعة مراجعة مراجع

> lala=lo k^{him} **zo** p'in-do $b\varepsilon^{2}$. some=DAT house make give-IPFV EQU.NE 'For some houses are being built (by the government).' (RBM discussion on roof)

¹⁹⁸ This form is surprising because here nominalizer $a_{N = a_{N}} - k^{h} \varepsilon n$ attaches to a noun instead of a verb. The author may have intended to write \hat{u}_{M} $\hat{g}_{M' = a_{N = a_{N}}} ji 2 k \varepsilon - k^{h} \tilde{\varepsilon}$ 'letter-bringer, postman' (this is how another informant wanted to correct the expression).

- (4.121) הَ تَحْمَّا تَّالَى اللَّهِ اللَّهِ عَلَى اللَّهُ اللَّهُ عَلَى اللَّهُ اللَّهُ عَلَى اللَّهُ اللَّهُ عَلَى اللَّعْنَى اللَّهُ عَلَى اللَّالَ عَلَى اللَّعْلَى اللَّهُ عَلَى اللَّهُ عَلَى اللَّهُ عَلَى اللَّالَةُ عَلَى اللَّالَةُ عَلَى اللَّهُ عَلَى اللَّالَةُ عَلَى اللَّهُ عَلَى اللَّهُ عَلَى اللَّهُ عَلَى اللَّهُ عَلَى اللَّهُ عَلَى اللَّهُ عَلَى اللَّ (Indef and the second strike are speaking their language, Nepali language (for their benefit).' (RS language situation)
- (4.122) हे' आ ले गोष'र' झुग र्जुब हे' र र र गा में , नुष' नुष' र जा हे' तरा में र जेंद' दे हो र हे जा र र जा में स् *ázi=gi=ra* duk p'ja-ti ŋàtɛa=lo tε сý:су: t'a sàm=tsu then elder.sister=AGT=AEMPH pain do-NF 1PL=DAT a.bit now food=PL ba? Ő: p'in-do bɛ?. carry come give-IPFV EQU.NE 'It was the elder sister who used to toil and bring a bit food (for us).' (PED life story)

Although the secondary verb *p* '*in* typically expresses benefactive semantics, it may also be used to emphasize mere recipiency in a context where the action is not (obviously and immediately) beneficial for the recipient:

 $\eta \dot{a}$: $k^h u = lo$ tipp'im-bo \tilde{t} :1SG.AGT3SGM=DAThitgive-2INFEQU.PER'I hit him.'(KN e)

4.2.3.3 Beneficiary secondary verb *t*^hop 'find, receive'

Whereas with the secondary verb p'in the actor¹⁹⁹ in the clause is the benefactor, by using the secondary verb $t^{h}op \not\equiv_{\nabla}$ 'find, receive' the speaker takes the opposite viewpoint and sees the actor in the clause as a *beneficiary* who receives the opportunity denoted by the preceding verb. An apt English translation is 'get to do'. For examples, consider the affirmative uses in (4.124) and negated uses in (4.125).

(4.124) a) हेंद्रार्थे र्जुवायत्र वर्दा र्जुवा हे गुवया हेंदा छत्।

 $ts\phi po$ $p'ja-c\varepsilon=di=p'ja$ $t\varepsilon$ $n\dot{\varepsilon}:kor$ kjap $t^{h}op$ $b\varepsilon$?. debate do-INF=DEMPH=ADVZR so sightseeing do receive EQU.NE 'Because of (participating) the debate (he) got (the opportunity) to do sightseeing.' (KNA kitchen discussion)

b) דֿמַשְׁמִיּאָל קָרָי יוּשָׂי אַקּאישׁי אַשָּׁרָי צָּקִייזאָ דַקּערָאָל קַשְּרָאָל קַשְרָאָשָּׁרָ ro:=tsu t'ã: karma námtei? phe? thop-ø: ga:tsho: friend=PL and PN with meet receive-2INF.GEN gladness 'the gladness of getting a chance to meet friends and Karma'(Richhi 67)

¹⁹⁹ "Actor" here and elsewhere is used as a semantic term that refers to the person/entity initiating or causing action in either an intransitive clause or a transitive clause. I prefer "actor" to "agent" because the latter term may be associated with A argument and agentive case.

(4.125) a) द्वि आंक्षे केंबा झेंगा वार्वता

ázi $tc^{h}o?$ nè: dok ma-t^hop. elder.sister doctrine read NEG-receive 1SG.GEN 'My sister didn't get (a chance) to study.' (PED life story) *mìnto=di* dzip $mi-t^{h}op=s$, $tiz\tilde{a}$: $tc^h \phi$:=kipè:=gi dzip 2SG.L=AGT 1SG.GEN=GEN flower=DEMPH suck NEG-receive=QUO but suck mi- $ts^h u = s$ làp. NEG-be.able.to=QUO say (But [s]he) said, you will not get to suck (the nectar from) my flower, (you) cannot suck.' (RS bee story)

The use of the secondary verb $t^{h}op$ is close in meaning to abilitative $ts^{h}u$? 'be able to'. Thus, $t^{h}op$ could be alternatively be said to express "availability" as a modal category (for modal markers, see §8.5)

4.2.3.4 Secondary verb tã: 'send'

The secondary verb $t\tilde{a}$: $\neg \neg \neg \neg$ 'send' can be used in a context which involves quite literal sending (away), see (4.126), or in a more abstract verbalizing use with little additional meaning to the primary verb, see (4.127).

- (4.126) a) هَلَامَ عَلَى اللَّهُ عَلَى اللَّالَ عَلَى اللَّهُ عَلَى اللَّهُ عَلَى الْحَلَى اللَّهُ عَلَى اللَّهُ عَلَى اللَّهُ عَلَى اللَّهُ عَلَى اللَّهُ عَلَى اللَّهُ عَلَى الْحَلَى اللَّهُ عَلَى اللَّ عَلَى اللَّهُ عَلَى الْحَلَى المَعْلَى اللَّهُ عَلَى الْحَلَى اللَّهُ عَلَى اللَّ عَلَى اللَّهُ عَلَى اللَّهُ عَلَى اللَّهُ عَلَى الْحَلَى الْحَلَى الْحَلَى اللَّ عَلَى الْحَلَى اللَّالَةُ عَلَى الْحَلَى الْحَلَى الْحَلَى الْحَلَى اللَّالَ عَلَى الللَّهُ عَلَى اللَّهُ عَلَى اللَّهُ عَلَى اللَّهُ
 - b) $\tilde{\mathfrak{A}}^{r}$ \mathfrak{g}^{r} $\mathfrak{$
- (4.127) a) ८. ज्या क्षेत्र कुषार्द्ध गी रुषा केंट्र वि में का सेन्द्र सेन्द्र र

 $y \dot{a}t \varepsilon a$? $l \varepsilon \eta g \varepsilon$:=tsu=gi $t'yts^h \phi$? $k^h a \varepsilon \varepsilon$? $l \dot{\varepsilon} n$ $t \tilde{a}$:- $ru\eta$ 1PL PRN.HON=PL=GEN time some take SEND-CONC 'Even if we take some of your time...' (NAB BLA 7)

c) אי המקיטי מהי מהי הקריהי אהן be?. số: làp tã:-m сá sé-pa meat kill-PUR go.PFV say send-2INF EQU.NE '(He) went to kill meat, (she) said.' (KT animal story) d) ગ્નન જ્વા હત્ય છે. સંચાય ĩ. ke:da $t c^h u \eta t c^h u \eta p' j a t \tilde{a}$:- $c \epsilon$ sound small do send-INF EQU.PER '(It) sends out a small sound.' (Richhi 4) e) ଖିଁ '''' ସୁଁଶ୍ୱ 'ହି' କରି' ଭିଳ' କର୍ନିଶ୍ୱ 'ହି' କ୍ରି'ସା ଏକ୍ଟର' ଏକ୍ଷା ଅନ୍ୟ ଏକ୍ଟର୍ କ୍ରାର୍ଯ୍ୟ କର୍ନ୍ୟା nấ:-nɛ ónalε tc'øn-di nè: mìn tøn-di ť'i-u then come.HON-NF 1SG.GEN name show-NF ask-2INF do.HON-COND làmsã: làp p'in tã. immediately say give send 'Then, if (you) go, show my name and ask, (they will) immediately tell (?and send you in the right direction).' (Richhi 102)

Example (4.127e) is somewhat ambiguous between literal sending and a more grammaticalized use. Consultant KN commented that no literal sending is implied but in the context literal sending in the right direction (by instructing) would seem a natural reading.

The secondary verb $t\tilde{a}$ is also used as an additional marker in imperatives, see (4.128).

(4.128) ซัเซ็ก การ ลาสาร รากกา ótsõ: t'ã: benda t'a

ótső: t'ã: bɛnda t'a: tã:. onion and tomato cut send 'Cut the onions and tomatoes.' (PT kitchen discussion)

The imperative use of $t\tilde{a}$: and its honorific (imperative) counterpart $n\tilde{a}$: 'grant' are described in §11.3.

4.2.4 Summarizing examples of the structure of the verb complex

This section provides summarizing examples of various combinations that the elements in the verb complex may take. The structure of the verb complex is, for the readers' convenience, repeated in Figure 4.3 below.

T' 10	C , , ,	C .1	1	1 .	1 1	, •	1
$H_1 \sigma_{11} r_P \Delta \Lambda$	Structure	of the	verh com	nley 11	n dect	arative	mood
I Iguite T.J.	Suuciaic	or the		присл п	i ucci	aranve	moou

$(IN) (INEG) \vee I (-KDP) (INEG) (V2) \begin{bmatrix} ((=)AUX) & (=CPC) \\ -TAM & (AUX) \\ -NMLZ & AUX \\ -INF & AUX \end{bmatrix} = (=CPC)$	(N) (NEG)	V1	(-RDP)	(NEG)	(V2)	((=)AUX) -TAM -NMLZ -INF	(AUX) AUX AUX	(=CFC)	
---	-----------	----	--------	-------	------	-----------------------------------	---------------------	--------	--

Negation is more complicated than Figure 4.3 suggests, because in addition to using a negator prefix, negation can also be accomplished by a negated auxiliary copula, see (4.139) below.

Negated serialized verbs cannot have more than two verbs in sequence, with the negator in between, see (4.130). The minimal verb complex consisting only of the primary verb, is presented in (4.129), exemplifying a steady-state present construction (see §8.2.1).

(4.129) LAN ANI V1 cé:. ηά: 1SG.AGT know 'I know (it).'

For other combinations of elements, consider (4.130-139).

(4.130) दते आ ले केंबा ह्यांग आ हेंना

nè:

V1 NEG-V2

dok ma-thop.

elder.sister doctrine read NEG-receive 1SG.GEN 'My sister didn't get (a chance) to study.' (PED life story)

tcho?

(4.131) त्रां कुषा वा याषाया द्याषा र्जुषा क्षे क्षे तदे याष्ट्र याष्ट्र याष्ट्र व्या दर्या

ázi

V1 V2 AUX lògju? k^ha sé:ta? p'ja-ti óde: súŋ za: du?. mouth clear story do-NF like.that say.HON put EX.SEN 'It has been so said in clear words.' (KLT Bumchu video)

(4.132) בישה אך תלימי שדישמי הקבי קממימקן

Ν V1 V2=AUX nàtci ke:=di=lo jàrge tã: go=pe?. 1PL.GEN language=DEMPH=DAT progress send be.needed=EQU.NE 'Our language needs to be developed.' (KL BLA 12)

(4.133) במי מושמיו הר בלי באי אורישדיו

V1 V2-TAM jó?=di p'ja mjõz-tshaz. ηά: **1SG.AGT** work=DEMPH do finish-CMPL 'I finished doing the work.' (KN e)

(4.134) र्रेज़ान्द्रांतु आधुत्र तेट्रांगु भगवा

V1 V2-TAM=CFC $k^h \tilde{\epsilon}$: ồ:-to=la. lópøn teacher know.HON come-PROB=HON '(You) perhaps know (that) teacher.' (KT discussion with TB)
(4.135) ସଦ୍ୟ ସମୁଷ୍ଠ ଅନୁସ୍ୟାନ୍ତି ଭିକ୍ ଆସ୍କୁ ଅନୁ V1-TAM AUX \tilde{a} : $d\varepsilon m = t \varepsilon i$? kjap-to íŋ-k^hɛn be?. lie like=INDF do-IPFV EQU-NMLZ EQU.NE 'he tells such a lie' (PAD bet story) (4.136) केंद्र' गत्र' दर्जुट्र' भया है' है' क्ष'र्ये गें। V2-INF =CFC/AUX²⁰⁰ V1 tc^hø? kan bjõ: làp-ti ťï ta-u=lo. what disappear say-NF ask look-2INF = REP 2sg.l 'What of yours disappeared he said (so the story goes).' (JDF axe story) (4.137) र के क्वें केंबर केंबर में पर्या गी V1-RDP-INF AUX du-ke. rodzou ten-tem-bo horns show-RDP-2INF EX.SEN-IN '(Its) horns are out showing (as I see in the picture).' (KN e) (4.138) בישה ארן מלי שדיאמי אימצא יארן N NEG-V1-INF AUX ŋàtci ke:=di jàrge? mi-t^høn-ce bE?. 1PL.GEN language=DEMPH development NEG-happen-INF EQU.NE 'Our language will not develop.' (KL BLA 12) V1-INF AUX mù loke kjap-ce? mè?. tε

well 3SGF Lhoke speak-INF NEG.EX.PER 'Well, she cannot speak Lhoke.' (SN kitchen discussion)

Example (4.140) is challenging for the current analysis. The verb complex includes the element $j\dot{a}$: gju which was above in §4.2.3.1 introduced as a complex translocative secondary verb but it also includes the durative secondary verb *bak/ba2/ba*: 'carry' (see §8.3.4). Therefore it seems like in (4.140) there are, according to the present analysis, two secondary verbs. The first secondary verb (*bak*) gives a durative nuance to the primary verb t^ha : 'be released, escape' and that complex construction becomes the primary verb for the secondary verbal construction $j\dot{a}$: gju, which further gives translocative nuance to the whole expression.

²⁰⁰ Here the reportative is used in place of an equative copula auxiliary. For more examples of the reportative substituting equative copulas, see §7.2.5.2.

(4.140) สังคราร สรง ครา แลง เลย สรา

1	[V1	V2]		V2	AUX
$\delta d\varepsilon = ra$	tha:	ba?	jàː	gju	bE?.
like.that=AEMPH	be.released	l carry	go	go	EQU.NE
'Like that (they)	keep on goi	ing fre	e.' (C	Y inte	rview)

4.3 Adjective and adverb phrases

The terms "adjective phrase" and "adverb phrase" refer here to a combination of an adjective/adverb and its modifier. The adjective phrase occurs either independently in an attributive/predicative clause or as a modifier following a noun. The adjective phrase consists of an adjective and a possible modifier. The adjective modifier may be an adverb (§4.3.1) or a nominalized clause (§4.3.2). In phrasal adjectives, which are a special case of adjective phrases, the adjective has a pre-modifying noun (§4.3.3). After addressing briefly reduplication (§4.3.4), this section finished with a couple of examples of adverb phrase, which is very similar in function to adjective phrases (§4.3.5).

4.3.1 Quantifying adverb as modifier

The adjective and adverb-modifying adverbs occurring in my data are listed in Table 4.7.

1 auto 4.7	. Aujective a	
lèp(ti)	ચેત્ર જ્વે.	'very much'
pemissiki	ન્વે શ્રે શ્વેન્ શે	'extraordinarily' (Martam: pesimipø:/pemisipo)
k'ã:mentsel	वित्राख्य किंदि.	'limitlessly, extremely'
átsi(m)	ષ્ઠા.છ્ર(થ).	'a bit'

Table 4.7. Ad	iective and	adverb-m	odifving	quantifying	adverbs
10010 11/110	jeen te ana	""""""""""""""""""""""""""""""""""""""		quantur , mg	aaroroo
	,		2 0		

Example (4.141) shows that the adverbs in Table 4.7 may be used both with adjectives and adverbs of manner, i.e. with or without the adverbializer p'ja(-ti).

(4.141) جِنَابَهُ عَاجَابًا اللَّهِ اللَّهُ عَلَى اللَّهُ عَلَى اللَّهُ عَلَى اللَّهُ عَلَى اللَّهُ الْحَالَ الْحَالَ اللَّهُ عَلَى الْحَالَ الْحَاطَ الْحَالَ الْحَالَ الْحَالُ الْحَالَ حَالَ حَالَ حَالَ الْحَالَ الْحَالَ الْحَالَ الْحَالَ الْحَالَ الْحَالَ حَالَ حَالَ حَالَ الْحَالِ حَالَ الْحَالَ الْحَالَ الْحَالَ حَالَ حَالَ حَالَ الْحَالَ الْحَالَ حَالَ وَحَالَ حَالَ عَلَى حَالَ الْحَالُ حَالَ حَال وَحَالَ حَالَ حَالُحَالُ حَالَ عَاجَاحَا حَالَ حَالَ حَالَ حَالَ حَالَ حَالَ حَالَ

The use of the modifiers is more frequent in adjective attribute clauses, where the adjective occurs independently (4.142), than it is when the adjective modifies a noun (4.143).

Independent uses (4.142) a) $\operatorname{art}^{\mathbb{R}} \operatorname{alg} \operatorname{sgarg} \operatorname{alg} \operatorname{alg} \operatorname{sgarg} \operatorname{alg} \operatorname{sgarg} \operatorname{alg} \operatorname{sgarg} \operatorname{sgarg} \operatorname{alg} \operatorname{sgarg} \operatorname{sgar$

b) हु दूर द केरा हु तकका में का की का ĩ. k^hu ťã: nà lèpti tcham-co? and 1SG very.much agreeable-SUP EQU.PER 3SGM 'He and I are in the very best of terms.' (KN e) c) केंग दने लेग लेगाय थेंन मन दी का होगी lèm jò-ce=di $ts^{h}ik=di$ lèp p'ja-tiki word=DEMPH very.much good EX-INF=DEMPH do-NF 'Because the words are very good...' (RS song intro) d) नूम्स्य महाम करी हुं के मार माय के माय t'ã:py: sún di=tsu **lèp=ra** k'exterita? î. long.ago.GEN story this=PL very.much=AEMPH important EQU.PER 'These ancient stories are very important.' (KT animal story) lèpti=ra súŋlu? ódi: dã: penta? du?. song.HON that.GEN melody very.much=AEMPH pleasant.to.hear EX.SEN 'The melody of that song is very pleasant.' (Richhi 90) f) तर्ने गन जव कर्न लेगवा भ्रम k'ã:mentse? lèm be?. di good EQU.NE this extremely 'This is extremely good.' (KL discussion with DR) g) ८.८२ मी. ज.४८८ के. जावव. पट्ट. ट्रांके. हुर. में जावा. ह्याय. ट्याया làt c^h uŋ s \acute{c} - k^h $\epsilon n=di$ pemisiki lèm, dzikta?. nàtci=gi say-NMLZ=DEMPH extraordinarily 1PL.GEN=GEN TPN good excellent 'Our what is called Lachung (is) extraordinarily good, excellent.' (LA intro to Lachung) Noun-modifying uses (4.143) a) केंगा लेग मन मन दु ... ts^hik **l**èp màndou word very.much different 'very different words' (DR discussion with KL) b) ह्याया रेया. याट. अव. क्र्ट. आवया टेयाया. t^huri? k'ã:mentse? k^hE:fa?

understanding extremely learned 'exteremely learned way of thinking' (CY introduction) c) वाका प्रयोग प्रामुमें आ हैं क्रें क्रें क्रिंग सुवा गहेवा làmsika:tor átsi bom-sy:=tɛi? type.of.dough.effigy a.bit big-DIM=INDF 'a rather big lamsika-torma' (KN apo, discussion in kitchen)

d) קיין מישי מיצי מיקיומי
 kale? átsi tshata?
 difficulty a.bit hot
 'rather difficult' (TB discussion with KT)

Note that the modifier $l \hat{e} p(ti)$ 'very much' can modify the superlative form, indicating that the superlative does not necessarily refer to the last instance of a quality before the end of the scale.

(4.144) $\mathbb{R}^{n} \subseteq \mathbb{R}^{n} \subseteq \mathbb{R}^{n} \stackrel{\text{rest}}{\cong} \mathbb{R}^{n} \stackrel{\text{$

Moreover, using the superlative does not necessitate definiteness²⁰¹, as shown by the cooccurrence of the superlative with an indefinite marker:

(4.145) ה־פּה׳ ה׳ הַצְּׁקִי מְשָּׁקִ׳ מֹק־צִׁי הַרְ׳ p'a dzik-co=tci? $j\dot{p}$ -po=di 1PL.GEN cow excellent-SUP=INDF EX-2INF=DEMPH 'our cow which is a most excellent one' (rna-gsung 2)

In spoken language, the adjective modifier may occur preposed to the noun that the adjective phrase modifies, as shown in (4.146) and (4.147).

(4.146)	רטר פאי אר ער איז						
	tce:zip	pʻja-ce=di	lèp	kale?	tsʰaţa?	bɛ?.	
	research	n do-INF=DEMPH	very.much	difficulty	hot	EQU.NE	
	'Doing	research is very di	ifficult.' (YF	R canteen vi	deo)		

(4.147) ך แลา ซี แลา ซี ส เพ ซิ ซิลา เล ส์ เลลา เลรา มีสาลรา นักามเลรา มา

t'a	ỳ∶tso	ỳ:tso=na	átsi=t¢i?	doly?	daminda	j <i>à:-k</i> ʰɛn	bɛ?.
now	district	district=LOC	a.bit=INDF	tradition	various	EX-NMLZ	EQU.NE
'Now	, in diffe	rent districts, th	he traditions	are a bit d	lifferent.' (SGD wedding	customs)

In an adjective attribute clause with a copula or a change-of-state verb, the adjective modifier may occur before the copula subject which is modified by the adjective phrase:

²⁰¹ Cf. English definite *the most wanted man* vs. indefinite *a most wanted man*.

(4.148) a) ८ राज्वी का ते रे जुवाका (materials) क्षाया के ते पर राज्या लुट ट्राका के जुवाका क nàtci mətiəriəls nàtea? **nùnta?** ta-watcene=di lèp 1PL.GEN materials(Eng.) look-COND=DEMPH very.much 1PL few $\tilde{i} = la$ EQU.PER=HON 'If (we) look at our (human?) materials, we are very few.' (YR canteen video) b) यात्र अत्र कत् रत्यो में रहेयाय द्वाय क्षत् याय वा k'ã:mentse? ran=gi=to dzikta? $b\epsilon$? làp=lo. extremely 2SG.M=GEN=CEMPH excellent EQU.NE say=REP '(Story of) yours is extremely great, he said (so the story goes).' (PAD bet story) c) यान अत्र केन न र रुया सेमसा न यात न यात हुन ले। k'ã:mentse? nàtea? sém gata? tc'uŋ-ze. mind joyful become-PST extremely 1pl

'We became extremely joyful.' (CY interview)

In (4.148c), the modifier of *gata?* 'joyful' precedes both the experiencer ($\eta \dot{a}tea$?) and the nominal element of the phrasal adverb ($s\dot{e}m$).

More frequently than as an adjective modifier $l \hat{\epsilon} p$ occurs as an independent verb-modifying adverb:

(4.149) a) בּחַשִי אֹסי דְּבִי מֹן t^ha: lɛ́p riŋ-zɛ. distance very.much be.long-PST 'The distance became very long.' (KN e)
b) هٚימָלָ אָר מָלָ דָ' אֹסִי חָמָּרָיזַ אָאָמי קָר אֹמן

ódi $g\tilde{a}$:=di te **lèp** népo ke:- $\epsilon\tilde{i}$: that time=DEMPH then very.much harm bring-NPST.PER 'That time it will wreak a lot of havoc.' (PL interview)

Because the verb modifying uses are more frequent and adjective-modifying uses occur with a copula, it may be surmised that $l\hat{e}p$ is basically a verb-modifier but in the presence of a semantically vacuous verb, copula, the modifying potential can be directed to a predicate adjective. The basic nature of $l\hat{e}p$ as a verb-modifier also explains why other elements may occur between $l\hat{e}p$ and the adjective it modifies.

4.3.2 Nominalized clause as modifier

A nominalized clause as modifier may occur either before the adjective (4.150) or after it (4.151). In my data, the pre-adjectival modifying clause may be marked by $-k^{h}\tilde{\varepsilon}$, as shown in (4.150a), or -po/bo, as shown in (4.150b-c) and (4.151). The clause marked with -po/bo may be either genitivized as in (4.150b) or not genetivized as in (4.150c) and (4.151).

(4.150) a) तन्त्रया के र्स्त्याया आवत र्श्वे अन्तु

 $[bak mi-ts^hu-k^h\tilde{\epsilon}:]$ $bompu^{202}$ carry NEG-be.able.to-NMLZbig'(so) big (it) cannot be carried' (KN e)

b) तनमा से र्खमाय ने सेंसरा

[*bak mi-ts^hu-po*:] bompu carry NEG-be.able.to-2INF.GEN big '(so) big (it) cannot be carried' (KN e)

c) বন্যণ রীস্ক্র্যামার্থ স্থ্রিস্কার্য স্থ্রস্কান্থ [bak mi-ts^hu-**po**] bompu carry NEG-be.able.to-2INF big '(so) big (it) cannot be carried' (KN e)

(4.151) ผู้สาวกรี่นิ่า สุราวัน สุขาญขาญขา ลิเซ์ ทาเติสุลา รับอรัสรา ฮูลา ลิเซ์ขุลารับ พีรารับ พูรา $k^{h}im=di$ $n a \eta = lo...$ n a ksusu? [mi=tsu] ka pám nòt^hø? house=DEMPH.GEN inside=DAT dark human=PL who EQU.ATTQ recognition p'ja mi-ts^hu-**po**] j*à-po bε*?. NEG-be.able.to-2INF EX-2INF do EOU.NE 'Inside the house... it was (so) dark [that it was not possible to recognize who people were].' (rna-gsung 7)

The pro-adverb $d\epsilon m$ 'like (it)' may be used along with nominalization to form a comparative modifier. Note that the verb is reduplicated.

(4.152) تَا سَمَا سَحَتْ تَهُمَا حَجْ عَلَيْ تَعْمَا تَعْمَا تَحْتَ تَعْمَا تَعْمَا تَحْتَ عَلَيْ مَا عَلَيْ مَا يَعْلَيْ عَلَيْ عَل

4.3.3 Phrasal adjectives

In analogy to phrasal verbs (§4.2.2), an adjective with a preposed noun may be termed a phrasal adjective, see (4.153-157).

²⁰² Consultant KUN would prefer the negator ma- in this construction.

(4.153)	riko lèm	र्त्तेयार्गे. व्येयाव्य	'intelligent (intellect good)'
	riko ts ^h aţa?	र्त्तेयांगां' कॅ'र्ट्रयाख'	'intelligent (intellect hot)'
	sém ť `ãːpu	र्षेम्रेयां यूट्रायें	'honest (mind straight)'
	tiŋ riŋku	यान्निन्तः सेन्तःगुः	'deep (depth long)'
	pu sópsop	শ্বৃ'র্য্রন'র্য্রন'	'fluffy with hair (hair fluffy)'
	ņum riːriː	ु भ्रुम्ना मेला मेला	'smooth (of skin) (oil round)'

(4.154) $\widetilde{\mathfrak{s}}_{\Gamma}$: $\widehat{\mathfrak{c}}_{\eta}$: $\widehat{\mathfrak{$

(4.155) अर्कें योनेन मेन गा

ts^ho [*tiŋ riŋku*] lake depth long '[deep] lake' (KN e)

(4.156) (אַזיָ הָלָ אָי אָז'ה אוֹדָרִילִי אָדן lako=di [pu sópsop] t^hõ:-po bɛ?. hand=DEMPH hair fluffy see-2INF EQU.NE '(She) saw (that) the hand (was) [fluffy with hair]' (Dras-ljongs gsung-gtam 40)

(4.157) दते आ अते झुवा झुवा रेनारेना धेंदा

nè: $\acute{amo:}$ $tc^ha?$ [*num rì:ri:*] jo?. 1SG.GEN mother.GEN hand.HON oil round EX.EQU 'My mother's hand is [smooth].' (Dras-ljongs gsung-gtam 40)

Ad hoc adjectivals can be formed from nouns by adding to a noun a nominalized existential:

 $mi s \epsilon m - \epsilon u^2 m e^{-po}$ [man mind-strength NEG.EX-2INF]'cowardly man' $mi n e^{-k} \epsilon^{-k}$ [man illness NEG.EX-NMLZ]'man with no illness, healthy man' $ts^{ho} tin m e^{-po/m e^{-k} \epsilon^{-k}}$ [lake depth NEG.EX-2INF/NMLZ]'shallow (=not deep) lake'

Formally these property concept expressions are post-head relative clauses, see §13.2.2.2. The negation of adjectives is described in §10.2.3 under the chapter on negation.

4.3.4 Reduplication

Reduplication of the full adjective in an adjective phrase denotes intensification of the quality:

t'ã:py:=ki t'aca? [*níŋku níŋku*] *di=tsu* long.ago.GEN=GEN monastery.building old old this=PL 'These [old, old] monastery buildings of long ago' (DB trip story)

4.3.5 Adverb phrase

Adjectives and adverbs are partly overlapping categories and adverbs are often derived from adjectives. Therefore it is no surprise that the same quantifying modifiers which are used in adjective phrases (see 4.2) can also be used with adverbs to form complex adverb phrases.

 (4.159) مَامَاتَهُ عَلَّا حَدَ مَاحَلًا
 [*lɛ̀pti tʰa:riŋ*] mė?. very far.away NEG.EX.PER 'It's not [very far away].' (Richhi 70)
 (4.160) مَحَاتَهُ مَاتَ مَحَاتَ هَاتَهُ مَاتَهُ مَاتَّهُ مَاتَّهُ مَاتَّهُ مَاتَّهُ مَاتَّهُ مَاتَّاتُهُ مَاتَّاتُهُ مَاتَّةُ مَاتَّهُ مَاتَّهُ مَاتَّاتُهُ مَاتَّاتُهُ مَاتَّةً مَاتَتُهُ مَاتَتُهُ مَاتَتُهُ مُنْتُنَا مُنْتُكُمُ مُنْتُكُمُ مُنْتُنَا مُنْتُكُمُ مُنْتُنَاتُ مَاتَتُهُ مُنْتُكُمُ مُنْتُكُمُ مُنْتُنَا مُنْتُنَا مُنْتُنَاتُ مُنْتُنَا مُنْتُنَا مُنْتُكُمُ مُنْتُنَا مُنْتُنَا مُنْتُنَا مُنْتُنَاتُ مُنْتُ مُنْتُكُمُ مُنْتُ مُنْتُ مُنْتُنَا مُنْتُنَاتُ مُنْتُنَا مُنْتُعُاتُ مُنْتُكَمَاتُ مُنْتُنَاتُ مُنْتُعُاتُهُ مُنْتُنَاتُ مُنْتُنَاتُ مُنْتُنَاتُ مُنْتُنَاتُ مُنْتُكُمُ مُنْتُنَاتُ مُنْتُنَاتُهُ مُنْتُنَاتُ مُنْتُنَاتُ مُنْتُنَاتُ مُنْتُنَاتُ مُنْتُنَاتُ مُنْتُنَاتُ مُنْتُنَاتُ مُنْتُنَاتُ مُنْتُنَاتُ مُنْتُعُاتُ مُنْتُكُمُ مُنْتُ مُعْتَنَاتُ مُنْتُكُمُ مُنْتُ مُنْتُكُمُ مُنَاتُ مُعْتَنَاتُ مُعْتَنَا مُنْتُ مُنَاتُ مُنْتُكُمُ مُنْتُ مُعْتَنَاتُ مُنْتُ مُعْتَنَاتُ مُنْتُعُاتُ مُنْتُعُاتُ مُنْتُكُمُ مُنْتُعُاتُ مُنْتُكُمُ مُنْتُعُاتُ مُنْتُعُاتُ مُنْتُنَا مُنْتُكُمُ مُنْتُ مُنْتُعُاتُ مُنْتُعُاتُ مُنْتُعُاتُ مُنْتُعُاتُ مُنْتُ مُنْتُعُ مُنْتُعُاتُ مُنْتُ مُنْتُعُنْتُ مُنْتُعُنْتُ مُنْتُعُنَاتُ مُنْتُ مُنْتُ مُنْتُعُنْتُ مُنْتُنُ مُنْتُ مُنْتُ مُنْتُ مُنْتُ مُنْتُ مُنْتُ مُنْتُعُتُ مُنْتُ مُ مُنْتُ مُنْتُ مُنْتُ مُنْتُ مُ مُنْتُ م

 $k^{h}o\eta = gi = di$ [*átsi lɛ̀m-p'ja*] $kjap- \varepsilon \varepsilon = la$. 3PL=AGT=DEMPH a.bit good-ADVZR strike-INF=HON 'They speak (the language) [rather well].' (RL interview)

4.4 Numeral phrase

The term "numeral phrase" refers to a combination of a numeral and a quantifier which modifies it. Quantifiers may be preposed (Table 4.8) or postposed (Table 4.9) to numerals they modify.

Table 4.8.	Pre-num	eral modifiers
t'yːmɛnɛ	नुषाःखवावेः	'about, approximately; almost'
halam	ર્ઝ'ભય'	'about, approximately'

The numeral-modifying quantifiers t'y:mene 'about, approximately; almost' and *halam* 'about, approximately' are both preposed to their head word. The quantifier t'y:mene, analogously to Nepali $dz^{h}andei$, covers both the meanings 'about' (4.161a-b) and 'almost' (4.161.c). Note the reduced form t'y:min in (4.161b). A nominal may intervene between the numeral and its modifier (4.161a).

- - b) ਤੁੱ ਨੂਲਾਗਰ ਰੋ ਸੁਲ੍ਹਾੜਾ ਸੁਲ੍ਹਾਨੂ ਸਾਲ ਨੂੰ ਨੂੰ ਸਾਲ ਨੂੰ ਨੂੰ ਸਾਲ ਸ਼ੁੱਸੀ p'ja t'y:min t<u>eena</u> teu:dup dø:tei? lo: to t'a sà-tiki. hen about fifteen sixteen that much reach-IPFV now eat-NF '(The number of) chicken (they) have eaten reaches now approximately some fifteen, sixteen.' (PL interview)

c) त्रायवावे र्ययार्गे श्वन

t'y:mene p^hok-o $b\epsilon$?.almosthit-2INFEQU.NE'(It) almost hit (us).' (KUN e)

In the novel Richhi, both of the two instances of the pre-numeral modifier *halam* are followed by the post-numeral suffix $-ts^h \emptyset$ (see Table 4.9):

(4.162) مَجْ اِلْعَبَّ عَلَى اَلْحَانَ عَلَى اَلَّهُ اَلَّهُ الْحَانَ عَلَى الْحَانَ الْحَانِ الْحَانَ لَكَلُ الْحَانَ الْ

The post-numeral modifiers are listed in Table 4.9 and illustrated in the same order below the table. The approximative suffix $-ts^h \partial r$ is described here because it is functionally similar to the word-level numeral modifiers.

Table 4.9.	Post-numeral	moc	lifiers

T 11 40 D

dɛːtɕi?	तन्दे:रुग	'about' (lit. 'like.that-one')
lakts™ø?	क्षेगा र्केंट्र.	'over, more than'
màŋts ^h ø?	ਕਟਾਲੋਂਟ.	'over, more than'
kort£i?	र्झेन्र:ठेयां	'around, about' (literally 'neighborhood-one')
teiku, teuku	(या)ठिया गा	'only'
mèmbo (+NEG.VERB)	ঝব্'র্ন	'no more than, only'
-ts ^h ø?	هرا.	'about, some, amount of'

(4.163) गहिन' (Nep.) गहिन' (Nep.) गहिन' (Nep.) गहिन' तर्ने खेग' झुग' न्वॅाक' या

ghanţa tci? ghanţa ní: dɛ:tci? gu: go:=lo. hour(Nep.) one hour(Nep.) two like.that wait be.needed=REP '(We) need to wait for some one or two hours, I hear.' (DB day trip)

The modifier $lakts^h @ 2(4.164a)$ derives from the verb lak 'be more than', which can be used also independently (4.164b).

(4.164) a) र्ख्वनायाः संगत्निते व गोधेणा मुनया छे। या गासुमा क्षार्म, र्यान्सन, र्यन

lóbda ódi:=na jó? kjap-ti lò súm laktshø? số:-tsha:.
school that.GEN=LOC work do-NF year three more.than go.PFV-CMPL
'More than three years have gone (her) working at that school.'/ 'More than three years have gone (since she started) working at that school.' (Richhi 11)

(4.165) র্না ব্যতিবা (এমা) মা স্টের্

 $l \delta$ $t \epsilon i : (=l \epsilon)$ **måŋts^hø?** year one(=ABL) more 'more than one year' (KN e) (4.166) a) নন্তু স্নাঁম তিশা kortci? tcu about ten 'about ten' (KN e) b) ব্র্যু শ্ল্রিম উব্য kortci? gja hundred about 'about a hundred' (KN e) c) नरु'गुरुग' क्रॅंग' रेंग' *tcuktci? kortci? eleven about (4.167) हुः हुः हुगः ईगःगुः p'u ná t'u? **tsiku** boy five six only

'only five, six boys' (PL interview)

In addition to being a formative in such words as *lakts^hø?* and *màŋts^hø?* ('over, more than'), the monosyllabic approximative suffix $-ts^h a/2$ \Im_{∇} 'about, some, as much as', attaches to numerals.

(4.168) a) بن مرتم المعالية المحالية المعالية ا

b) הَ تَحْمَ² سَمَ² عَلَى مَعْمَ² عَلَى مَعْمَ² مَعْمَ² عَلَى مَعْمَ² عَلَى مَعْمَ² عَلَى مَعْمَ² عَلَى مَعْمَ² مَعْم

The marker $m \dot{\epsilon} m bo$ (nominalized negated existential copula) is followed by a negated verb. When postposed to numerals, the construction has the meaning 'no more than, only':

(4.169) الله: ترمين تركم: تر

The formative $m \hat{e} m bo$ also functions as a marker of negated additive adverbial clauses, see §15.9.4.

The specific numeral $k'\tilde{a}:pu/k'\tilde{o}:pu$ 'one full measure of' occurs as a numeral modifier in the idiom $t\tilde{o}: k'\tilde{a}:pu \notin \mathbb{R}^{n}$ 'all' (lit. thousand-one.full.measure), illustrated in (4.170).

(4.170) مجمَّتَهُ الله المَّاتَى المَاتَحَةُ المَّاتَ المَّاتَةُ المَّاتَةُ المَّاتَةُ المَّاتَةُ المُحْتَةُ المُحْتَةُ المُحْتَةُ المُحْتَةُ المُحْتَةُ المُحْتَةُ المُحْتَةُ المُحْتَةُ (NAB BLA 7) (4.170) مجمعة المحتاطة (1.170) مجمعة المحتاطة المحتاطة

4.5 Summary remarks

This chapter described how words can be combined into phrases or, in the case of verbs, into a "complex." It was shown that nouns take both prenominal modifiers, which are typically genitive marked, and postnominal modifiers, which are typically not genitive marked. In clausal context, nouns frequently take up to three clitics, including plural, case and emphatic clitics. The numeral *tci2* 'one' has grammaticalized into an indefiniteness marker.

The section on the verb complex showed that there are two types of complex verbs, phrasal verbs, which combine a nominal element to a verb, and serial verbs, which juxtapose two to four verb roots. Verbs in a serial verb construction were seen to have three types of semantic relationship. In the first, the verbs together form the semantic core of the construction (e.g. *bak* δ . 'bring', lit. 'carry-come'). In the second (and rare) case, the combined verbs are synonymic (*t'ɛk da:* 'chase away' lit. 'chase-chase'). In the third relationship, the last verb does not belong to the semantic core of the construction but brings semantic nuance to it. Verbs which bring semantic nuance to the primary verb were termed secondary verbs, which often express tense, aspect and modality-related values.

Quantifying adjective modifiers were shown to be loosely connected with the adjective they modify in that other elements may come in between. It was also shown that adjectives, analogously to phrasal verbs, can occur as phrasal adjectives with nominal premodifiers. The last section illustrated the use of nine numeral modifiers.

5 Basic clause structure

This chapter discusses basic clause structure in Denjongke, focusing on the number and type of NP arguments that verbs may take. The treatment begins with an introduction to concepts and terminology (§5.1). Then the description is divided into intransitive clauses (§5.2), transitive clauses (§5.3), clauses with verbs of being and becoming (§5.4), valency modification (§5.5) and the use of adverbials (§5.6).

5.1 Introduction to concepts and terminology

The terms "transitive" and "intransitive" are here used syntactically ²⁰³ as referring to the potential ²⁰⁴ number of core arguments of a verb. Transitive clauses have two potential core arguments, whereas intransitive clauses have only one argument. Transitive clauses are further divided into monotransitive clauses and ditransitive clauses, the latter having an additional argument expressing the recipient of the action. The discussion on argument structure uses the typologically motivated terms S, A, P, T and R. The term "S" refers to the sole core argument of an intransitive clause, such as (5.1). In this section (§5.1), argument S is given in bold, while A is underlined. Other arguments are not marked typographically.

(5.1) हुः कुगु न्द्रित दन् गुगो

k^hu gjuk-tεεn du-kε. 3SGM run-PROG EX.SEN-IN 'He's running.' (TB e)

In addition to one core argument, an intransitive clause may have some additional, peripheral arguments such as adverbials expressing time, location and manner. Example (5.2) has one core argument S (ηa) and two peripheral arguments, the temporal adverbial *ódi gjaple* 'after that' and the directional $g\tilde{a}$:to?(=lo) 'to Gangtok'.

(5.2) ซังลร์ สูวาจจ รา สุราร์ขา(กัง) ลีราร์ พิสุ

ódi gjaple $\eta \dot{a}$ gã:to?(=lo) *òm-bo* \dot{i} : that after 1SG TPN(=DAT) come-2INF EQU.PER 'After that I came to Gangtok.' (TB life story, KN e)

In a transitive clause, argument A is the most important argument for the success of the verbal action (and usually the most agent like argument, hence A), and P is the other argument in a transitive clause (and usually the most patient like argument, hence P), see (5.3). Elements in brackets are pragmatically conditioned instances of differential marking of A and P, which is described more in detail below. The differential marking of the P argument was already hinted at by Sandberg (1895: 22) who reports both k^{him} (zero-marked) and $k^{him}=lo$ (dative-locative-marked) as "accusative" forms of the word for 'house'.

²⁰³ In contrast to the more semantically oriented use of the terms in Hopper & Thompson (1980).

²⁰⁴ The word "potential" is used because of the prevalence of zero anaphora, or contextual elision of core arguments, which is described below.

(5.3) สูญาลส์สา (ฏิลา) ทุสา (กับ) จลรารับ พรา

 $gjalts^{h}en(=gi)$ karma(=lo) $s\acute{e}$ -po be?. PN(=AGT) PN(=DAT) kill-2INF EQU.NE 'Gyaltshen killed Karma.' (KN e)

Ditransitive clauses, such as those formed around verbs of giving, have, in addition to A, a recipient-like argument R and a theme-like argument T (Haspelmath 2005), see (5.4), where lala=lo 'to some' is R and man 'privilege' T.

(5.4) ज्वू ग्यू भाषा भाषा के का (Nep.) होता ही सुन

 $\underline{zun=gi}$ lala=lo man p'in-do $b\epsilon^2$. government=AGT some=DAT privilege(Nep.) give-IPFV EQU.NE 'The government is giving privileges to some.' (RBM discussion on the roof)

The dominant word order in declarative clauses is SV in intransitive clauses and APV in transitive clauses. For a note on right-dislocation, in which arguments may occur after the verb, refer to §16.6. Ditransitive clauses have the order ATRV or ARTV, depending on whether T or R is more focal and thus comes first (see §5.3.3). Other arguments than the above-mentioned S, A, P, R and T are peripheral. Typically of clausal PV order, Denjongke genitives precede the noun they modify and postpositions are used rather than prepositions (conforming to Greenberg's [1966] correlations).

In nominative-accusative (NA) languages, S and A are aligned in having the same form contrasting with O. Ergative-absolutive (EA) languages, on the other hand, align S and O, which contrast in form with A. Denjongke is clearly neither a NA language nor a EA language, because all of the three arguments S, A and P evince split patterns with two ways of marking depending on the context. In intransitive clauses, S argument is usually zero marked, but sometimes the same intransitive verb (which though has to be a controllable verb) may occur either as zero or agentive marked, depending on the context. In transitive clauses, A argument may be agentive or zero marked with the same verb, depending on the context. Lastly, P argument marking is also split between zero-marking and dative-locative marking. Factors that have been identified as conditioning this differential marking are animacy, specificity/identifiability and affectedness of the P argument, which are all exemplified in the discussion below. Because Denjongke does not have a clear EA alignment but allows considerable pragmatic conditioning of argument marking, I use the term "agentive" rather than "ergative" for the A argument.²⁰⁵ For the same reason, nonmarked S and P arguments are called "zero marked" rather than "absolutive." The options for clausal argument marking are summarized in Table 5.1, which, however, does not include information on obligatory agentive marking with certain past/perfective verb forms (see §5.3.2).

²⁰⁵ The Denjongke agentive though may be seen as an ergative-in-making, see Coupe (2017).

I dole officie	ammary or argam	ent marking opti	ons
Argument	Zero-marking	Agentive	Dative-locative
S	Х	(X)	
А	Х	Х	(X)
Р	Х		Х
R			Х
Т	Х		

Table 5.1. Summary of argument marking options

Instances marked with brackets in Table 5.1 occur infrequently. The agentive S arguments occur only in specific cases to emphasize identity or agentivity (see §5.2). The dative-locative A argument occurs with a couple of transitive verbs (see §5.3.1).

The reason for the splits in argument marking in Denjongke is that the argument marking in the language is to some degree directly controlled by semantics/pragmatics and not by syntacticallyoriented grammatical relations, which are abstractions based on prototypical situations. Dixon (1994: 1) makes a distinction between languages that map semantics directly onto morphosyntax and languages that use an intermediate layer of grammatical relations where arguments are marked according a prototypical situation (e.g. the arguments of the English verb *hit* are marked the same way irrespective of whether the act was intentional or not). Dixon further notes that the division into NA and EA works only for the latter, syntactically oriented languages, not to semantically oriented languages. As a system where semantics/pragmatics play an important role in argument marking, Denjongke argument marking cannot, thus, be naturally termed either NA or EA.

According to LaPolla (1995: 189-190) argument marking within Tibeto-Burman languages may be divided into non-paradigmatic and paradigmatic, with a grammaticalization continuum from the former to the latter. In non-paradigmatic languages, overt marking of arguments is motivated by "disambiguation of two potential agents" (LaPolla 1995: 189). Paradigmatic languages, on the other hand, "have relatively stable paradigmatic ergative system" (LaPolla 1995: 216) with "semantic and pragmatic functions beyond simple disambiguation" (LaPolla 1995: 190). As examples of non-paradigmatic languages LaPolla (1995: 214) gives Dulong, Namuzi, Hani, Naxi, Achang, Nusu and Deng languages. Examples of paradigmatic languages are "Chepang, Newari, Kham, Sunwar, and most Tibetan dialects" (LaPolla 1995: 216). Even if it were true that most Tibetic languages have relatively stable paradigmatic ergative systems, Denjongke seems not to be one of them. As will be shown below, Denjongke evinces too much pragmatic conditioning and too little syntactic control, such as obligatory ergative or absolutive marking, for the system to be meaningfully called paradigmatic EA.

Coupe (2017), on the other hand, argues that instead of syntactically oriented NA and EA patterns for argument marking, many Tibeto-Burman languages evince a pragmatically oriented agentive vs. anti-agentive pattern, a system which Coupe considers a precursor to NA and EA patterns. Coupe (2017) further notes that the agentive vs. anti-agentive pattern has previously been falsely considered a type of EA system (similarly LaPolla 1995: 214) and lists the following Tibeto-Burman languages as evidence for the prevalence of pragmatic rather than syntactic patterning of clausal arguments: Qiang (LaPolla 2003), Darma (Willis 2011), Meithei (Chelliah 1997, 2009), Mongsen Ao (Coupe 2007, 2011), Kurtöp (Hyslop 2010), Yongning Na (Lidz 2011), Singpho (Morey 2012), Sumi (Teo 2012), (Lhasa) Tibetan (Tournadre 1991). Partly inspired by Coupe (2017), the analysis here adopts the term "agentive" rather than "ergative" for overt marking of argument A. Altough Denjongke is probably one of those Tibeto-Burman languages which in

Coupe's (2017) analysis are in a transitional stage from pragmatic core argument marking to purely syntactic alignent, Denjongke has not yet arrived in the destination.²⁰⁶

Analyzing transitivity in Denjongke is complicated because Denjongke, like probably all other Tibetic languages (e.g. Denwood 1999: 191; Bartee 2007: 117), excels in zero anaphora where core and peripheral arguments are elided when the speaker deems them sufficiently deducible from the previous context. Therefore it is difficult to make transitivity judgments on verbs based on purely syntactic, distributional criteria at the sentence level. Rather, transitivity of any verb and clause is determined by the potential number of core arguments. Moreover, zero anaphora should be distinguished from argument suppression (Andersen 1987: 285ff; Andvik 2010: 115). In zero anaphora, an argument is omitted because it is so central. The argument can be recovered from the context and therefore does not need to be mentioned. In argument suppression, on the other hand, the argument is suppressed because it is not important or identifiable, or in words of Payne (1997: 233), the identity of the argument "has not been established and need not be established in order for the speaker to achieve his/her communicative goal." Therefore, suppressing argument A will cause an effect that can be called a "functional passive" (Givon 1984: 164), see §5.5.1.

Making a morphosyntactic distinction between core and peripheral arguments in Denjongke is difficult at least for three reasons: First, arguments P and R (the latter corresponding to "indirect object") and other, more peripheral arguments such as directionals (e.g. 'to Gangtok') may all be marked by dative-locative. Second, arguments typically considered peripheral, such as locationals and directionals may, similarly to P argument, occur zero marked. For instance, if the S argument is dropped, as in (5.5), disambiguation is purely semantic ("a toponym cannot be the goer") and pragmatic ("[s]he is answering my question concerning person X").

(5.5) $\operatorname{Het} \tilde{f}^{\text{T}} \operatorname{Agg} \tilde{f}^{\text{T}} \operatorname{He}^{\text{T}}$ $g\tilde{a}:to? gju-do b\epsilon?.$ TPN go-IPFV EQU.NE '(He's) going to Gangtok.' (KN e)

Thirdly, because of zero anaphora, core arguments may be elided from a clause, whereas peripheral arguments may be retained. The definition for core and peripheral arguments used here is adopted from Andvik's (2010: 119) description of Tshangla, another Tibeto-Burman language with zero anaphora. A core argument is an argument "which can be omitted only if its reference is recoverable from the discourse context, i.e. under zero anaphora" ("omitting" here excludes argument suppression), whereas a peripheral argument is an argument "which can be omitted even when not recoverable from the discourse context" (Andvik 2010:119).

5.2 Intransitive clauses

The S argument in intransitive clauses is typically non-marked but may also be agentive marked to emphasize identity or agentivity, as will be shown below. For a non-marked S argument, consider (5.6) and (5.7). The S argument is given in bold:

²⁰⁶ Similarly, Watters (2018: 217) characterizes argument marking in Dzongkha, a closely related language, as a "pragmatic marking system that does not distinguish consistently between grammatical and semantic roles."

(5.6) क्षॅंग्दर्ने कुराय्य रदी गी वर्त र्या गें केवा

ódi gjaple $p\dot{e}$:=gi $n\dot{e}$? t'ak-o \dot{t}' . that after 1SG.GEN=GEN disease get.healed-2INF EQU.PER 'After that my disease got healed.' (TB life story)

tc^hi-lo tciktõ: lò=i gubgja gupteu k'oná nànca one.thousand nine.hundred ninety ninety.five year=GEN inside outside-year ĩ:. cé:da số:-bo ŋà gã:to? nànca Buddhist.institute inside go.PFV-2INF EQU.PER 1SG TPN 'In 1995 (of Gregorian calender), I went to the Higher Institute of Nyingmapa Studies in Gangtok.' (RB life story)

In (5.7), the directional/destination is marked as an overtly peripheral argument with the postposition *nàŋca* 'inside'. However, destination and location may also be non-marked, as shown in (5.8) and (5.9) respectively (the latter is a copula clause). In both cases, also the dative-locative =lo can optionally be used. When the destination argument is non-marked, disambiguation of the clause is based on word order and semantics of the argument lexemes.

(5.8) ตัวสร้า สูราคพารา สูราร์ๆ (ค้า) สราร์า ติลุ

ódi gjable $\eta \dot{a}$ gã:to?(=lo) *òm-bo* \dot{t} :. that after 1SG TPN(=DAT) come-2INF EQU.PER 'After that I came to Gangtok.' (TB life story, KN e)

(5.9) المَّا عَلَّةُ الْمَا (مَنَّ) المَّا

 $k^{h}u$ $g\tilde{a}:to?(=lo) j\dot{\phi}?$. 3SGM TPN(=DAT) EX.PER 'He is in Gangtok.' (KN e)

Argument S is often elided if the context makes it obvious, as exemplified in the following S-less pair of an intransitive subordinate and main clause:

(5.10)જો વર્તર ઇજા જેન્ઝાન્ટ જો વર્તર અરુ ગવા વર્તિવા આવત વત્વા ગા વર્તર અંતુ વાયર ગ્રેવર્ષ જે જો જો જો જો જો જો જો ámba? tok^hẽ: ódi p'ja-somdã: ódi daku=di dvnk^ha that do-SIM that guava pluck.NMLZ owner=DEMPH in.front.of số:-bo bɛ?. lep-ti ónale arrive-NF then go.pfv-2inf EOU.NE 'As (they) were doing that, (they) arrived by the guava-picking owner (of the guava tree) and then went away.' (TB pear story)

The S argument of an intransitive clause, which is typically zero marked, may receive agentive marking for the purpose of emphasizing the identity of the actor, as in (5.11), where the S argument is also shifted to clause-final position. Similar pragmatic use of agentive with intransitive verbs, especially for marking contrastive agency, has been attested in Lhasa Tibetan (Tournadre 1991).

(5.11) هَمْ هَا هَ عَلَيْهُ عَلَيْ عَلَيْهُ عَلَيْ عَلَيْ عَلَيْهُ عَلَيْ عَلَيْهُ عَلَيْهُ عَلَيْهُ عَلَيْ عَلَيْهُ عَلَيْهُ عَلَيْهُ عَلَيْهُ عَلَيْهُ عَلَيْهُ عَلَيْهُ عَلَيْ عَ المَا عَلَيْهُ عَلَيْ المَا عَلَيْهُ عَلَيْهُ عَلَيْهُ عَلَيْهُ عَلَيْهُ عَلَيْهُ عَلَيْ عَلَيْ عَلَيْهُ عَلَيْ عَلَيْ عَلَيْ عَلَيْ المَا عَلَيْهُ عَلَيْ عَالَيْ عَلَيْ الْعَلَيْ عَلَيْهُ عَلَيْهُ عَلَيْ الْعَلَيْ عَلَيْ عَلَيْ

5.3 Transitive clauses

In this section, I first present some general comments on the morphosyntax of argument structure in monotransitive clauses and then describe the semantic/pragmatic factors in the split marking of A argument (§5.3.1) and P argument (§5.3.2). This is followed by a discussion on ditransitive clauses with T and R arguments (§5.3.3). Due to zero anaphora it is difficult to find enough natural examples of transitive clauses with all arguments in place to make trustworthy generalizations. Therefore this section includes quite a few elicited clauses.

Argument marking in monotransitive clauses is very flexible if the arguments are proper names. Consultant KN asserted that all the four different combinations of A and P marking in (5.12) and (5.13) can be used in spoken language. Example (5.12) represents a past/perfective clause, and (5.13) an imperfective one. The A argument may be either agentive or zero marked while the P argument may be either dative-locative or zero marked. When marking on both arguments is dropped, see (5.12d) and (5.13d), disambiguation of arguments is done through word order, similarly to English.

(5.12) a) สูญนาลส์สาฏิลา กาสานั่า นลนารั่า ลนา

gjalts ^h en=gi	karma=lo	sé-po	bɛ?.
PN=AGT	PN=DAT	kill-2INF	EQU.NE
'Gyaltshen kil	led Karma.'	(KN e)	

b)	ਗੁੁੁੁੁਪਾ ਲਕੱਰ '	শার্ষ'র্ন্য	বঝ্দ্'র্থি' শ্বদ্
	gjalts ¹ 8n	karma=lo	sé-po be?.
c)	ਸ਼ੂਆਕਾਂ ਕਰ ਸ਼ੂਕਾ	শারু'	বঝ্ব'র্য়, শ্বিধী
	gjalts ^h ɛn=gi	karma	sé-po be?.
d)	ਗ਼ੁੁੁੁੑੑੑੑੑੑੑੑੑਗ਼੶ਸ਼ਜ਼ਗ਼	শারু'	বঝ্ব'র্য়, শ্বিধী
	gjalts ¹ En	karma	sé-po (be?)

(5.13) a) ଶ୍ରୁଦ୍ୟ'କ୍ଷର୍ଚ୍ଚ ଶ୍ରିଷା ମୁଣ୍ଟ ଦିମ କରିବ ଦମ୍ମା

gjalts ^h en=gi	karma=lo	tip-tsen	du?.
PN=AGT	PN=DAT	beat-PROG	EX.SEN
'Gyaltshen is	hitting Karn	na.' (KN e)	

b)	ਗ਼ੁਕਾ ਅਰੱਕ	শ্ব্য'র্ন্	नियायविषा पर्युय	1
	gjalts ^h en	karma=lo	tip-tsen	du?.
c)	ক্ৰুণ্ম'অৰ্চ্চৰ্'দ্ৰীম'	শক্ষ	नियायविषा पर्युय	1
	gjalts ^h en=ki	karma	tip-tsen	du?.

d)	ਗ਼ੂਧਾ ਕਛੱਕ	শার্কা	हिम'मलेव' ५८ग		
	gjalts ^h en	karma	tip-tsen	du?.	

It is my impression that some speakers who have received formal education in Denjongke writing rules are prone to implement the writing rules, which are to some degree affected by Classical Tibetan, to their grammaticality judgments of spoken language in a prescriptive rather than a descriptive way. Consequently, some of the above forms, especially (5.11d) and (5.12d), may be considered "ungrammatical" (i.e. against the rules of writing) by some speakers. A similar observation was made by Sandberg (1895: 25) about the use of the agentive case:

Where both subject and object occur in any sentence, the subject is put in the agentive case, except where the verb of the sentence is part of the verb 'to be'. Rule though this is, it is generally not observed by the uneducated; and therefore we shall not keep to it in our conversational examples to be given hereafter, the nominative being usually heard as in English.

Sandberg's word "usually" with reference to the nominative (=zero-marking) leaves open the option that even "the uneducated" occasionally use agentive, suggesting that the use of the agentive was pragmatically conditioned even in Sandberg's data. Finding natural examples of all the combinations of A and P argument marking would require an extensive corpus which is not available at the moment. This chapter offers some preliminary remarks.

If arguments are marked by pronouns or common nouns, rules for assigning the A argument are somewhat stricter than with proper nouns. In perfective past, as in (5.14a), agentive marking is usually mandatory, although see the natural example (5.17) below where a past tense verb with a highly affected P has a zero-marked A argument.²⁰⁷ In the future tense, as shown in (5.14b), agentive marking is also preferred. The A argument in imperfective clauses is typically zero marked, whereas agentive marking is reserved mainly for contrastive agency (5.14c). The P argument is either zero marked or dative-locative marked. Semantic/pragmatic differences between the two markings of P are discussed in §5.3.2

(5 1 4)	~)		\sim	< ~	\sim
(5.14)	- A I		י מי(פוי)	הקיקי	MAI
	<i>u</i>)	21-1-1	NG(')	· / · ·	0,01

dã:	ŋá:	k ^h u(=lo)	tip-o	í.		
yesterday	I.AGT	3sgm(=dat)	beat-2INF	EQU.PER		
'I beat him yesterday.' (KN e)						

- c) ८'र्सें ८'/८४' ख़'/ख़'र्ये' हेरा हें' क्षेत्रा

t'ato \eta \dot{a}/\eta \dot{a}: k^h u/k^h u = lo tip-to \dot{\tilde{i}}:. now 1SG/1SG.AGT 3SGM/3SGM=DAT beat-IPFV EQU.PER 'I'm beating him now.' (KN e)

²⁰⁷ Cf. DeLancey's (1990: 306) observation on Lhasa Tibetan that ergative (here agentive) marking is required in the perfective aspect but is otherwise optional.

5.3.1 Argument A

As a system evolving from pragmatic argument marking towards syntactic alignment (see Coupe 2017), it is not surprising that Denjongke attests a combination of both syntactic and pragmatic control of clausal argument marking. As a sign of syntactic control, the A argument of some past verbs is obligatorily agentive marked (see Table 5.2 below). One sign that pragmatics has an effect on argument marking is that in elicitation both agentive and zero-marking are often offered for argument A, see (5.15).²⁰⁸ Argument A is given in bold.

(5.15) $\mathfrak{g}^{\mathsf{A}^{*},\mathsf{A},\mathfrak{a},\mathfrak{a},\mathfrak{a}^{*},\mathfrak{g}}(\mathfrak{g}^{\mathsf{A}^{*}}) \rightarrow \mathfrak{f}^{\mathsf{A}^{*}},\mathfrak{g}^{\mathsf{A}^{*},\mathfrak{g}}(\mathfrak{g}^{\mathsf{A}^{*}}) \rightarrow \mathfrak{g}^{\mathsf{A}^{*},\mathfrak{g}}(\mathfrak{g}^{\mathsf{A}^{*}}) \rightarrow \mathfrak{g}^{\mathsf{A}^{*}}) \rightarrow \mathfrak{g}^{\mathsf{A}^{*},\mathfrak{g}}(\mathfrak{g}^{\mathsf{A}^{*}}) \rightarrow \mathfrak{g}^{\mathsf{A}^{*}}) \rightarrow \mathfrak{g}^{\mathsf{A}^{*}},\mathfrak{g}^{\mathsf{A}^{*}}) \rightarrow \mathfrak{g}^{\mathsf{A}^{*}},\mathfrak{g}^{\mathsf{A}^{*}}) \rightarrow \mathfrak{g}^{\mathsf{A}^{*}},\mathfrak{g}^{\mathsf{A}^{*}}) \rightarrow \mathfrak{g}^{\mathsf{A}^{*$

Another sign of pragmatic control is that the syntactic tendency to have an agentive A argument with a past tense verb (with highly affected P) may be broken, as shown in (5.16) and (5.17). In the question-answer pair (5.16), the A argument is zero marked in the question and agentive marked in the answer.

(5.16) Q: เพาฐา พรรา เดณาณฑา พารเดิพารัง ณรา

 \acute{agja} $d\widetilde{a}$: $\epsilon \widetilde{\epsilon} : la?$ man-ze:-po da. elder.brother yesterday food.HON NEG-eat.HON-2INF be.similar 'It seems the brother (=you) didn't eat food yesterday.'

A: ८षा अ८८. तथा झवा	মন্-াব-মন্-	ર્ગ મેં એમ		
ŋá ː dãː	sàm	méŋkʰãː=na=rã:	sò-u	í.
I.AGT yesterday	food	hospital=LOC=AEMPH	eat-2INF	EQU.PER
'I ate yesterday	in the l	nospital.' (Richhi 23)		

The use of agentive in the answer in (5.16) is probably pragmatically motivated. The speaker's sister has suspected that the speaker did not eat dinner the previous night. The speaker responses by (5.16), emphasizing his agency in eating the previous night. The fact that the speaker is actually lying may also trigger an increased claim of agency. The lack of agentive marking in the question in (5.16), on the other hand, may be affected by negation and descreased assertiveness marked by the apparentive construction ending in -da 'be similar'.

In contrast to the agentive marking in the answer in (5.16), example (5.17) has a zero-marked A argument in an otherwise analogous clause, suggesting that the use of the ergative in the answer in (5.16) is indeed pragmatically conditioned.

(5.17) ราชีญา ตั้งเครา สีมา สาวัง ติสา

yàtea? ódi sàm sò-u \tilde{t} . 1PL that food eat-2INF EQU.PER 'We ate that food.' (DB trip story)

²⁰⁸ Similar observation was made by Huber (2002: 75) on Kyirong Tibetan, in which the argument marking system bears close similarity to Denjongke.

Coupe (2017) finds the following contexts for the use of pragmatic agentive in his survey of Tibeto-Burman languages: 1) to disambiguate semantics roles of NPs when they cannot be determined from the semantics of the NPs themselves, 2) to contrast one referent to another, 3) to express increased agency, 4) to encode the atypical or unexpected behavior of a referent. Similar pragmatic factors play a role also in Denjongke. Increased agency is illustrated by (5.18) where the use of the agentive, according to consultant KTL, would imply an intentional meeting and the use of zero-marking an incidental meeting.

(5.18) $\mathfrak{g}^{A''}\mathfrak{A}\mathfrak{E}\mathfrak{F}^{(\mathfrak{g}A'')}\mathfrak{g}_{\mathfrak{F}}^{(\mathfrak{A}'')}\mathfrak$

Similarly to intransitive clauses, agentive in transitive clauses is often used to mark contrastive agency, see (5.19). Zero-marking would be used in a non-contrastive context.

(5.19) بَعْ بَعْمَارَ بَعْرَامَ عَرَيْقُ بَامَّا عَرَيْقُ بَامَّا عَرَيْقُ بَامَّا عَرَيْقُ بَعْرَامًا عَرَيْقُ بَعْرَامًا عَرَيْقُ بَعْرَامًا عَرَيْقُ الْمَا عَلَيْهُ الْمَا عَلَيْهُ الْمَا عَلَيْهُ الْمَا عَلَيْهُ اللَّهُ عَلَيْهُ الْمَا عَلَيْهُ الْمَا عَلَيْكُ الْمَا عَلَيْكُمَ الْمَا عَلَيْهُ اللَّهُ عَلَيْهُ الْمَا عَلَيْ الْمَا عَلَيْهُ الْمَا عَلَيْهُ اللَّهُ الْمَا عَلَيْهُ اللَّهُ اللَّهُ عَلَيْنُ اللَّالِ الْمَا عَلَيْهُ الْمَا عَلَيْ الْمَا عَلَيْكَ الْمَا عَلَيْكَ الْمَا عَلَيْكُمُ الْمَا عَلَيْكُمُ الْمَا عَلَيْكُمُ الْحَالَ الْمَا عَلَيْكَ الْحَالَةُ الْحَلْمَا الْحَلْقُ الْحَالَةُ الْحَلْقُ الْحَلْقُ الْحَلْقُ الْحَالَ الْحَلْقُ الْحَالِي الْحَلْقُ الْحَلْقُ الْحَالَةُ عَلَيْكُمُ الْحَالَةُ الْحَلْقُ الْحَلْقُ الْحَلْقُ الْحَلْقُ الْحَالَةُ الْحَلْقُ الْحَالَةُ الْحَلْقُ الْحَلْقُ الْحَلْقُ الْحَلْقُ الْحَلْقُ الْحَلْقُ الْحَلْحَالَ الْحَلْقُ الْ الْعَالَيْ الْحَلَيْلَةُ الْحَلْقُ الْحَلْقُ الْحَلْقُ الْحَلَيْ الْحَلْقُ الْحَلْقُ الْحَلْقُ الْحَلْقُ الْحَلْ الْعَالَيْ الْحَلَيْلَةُ اللَّالِ الْحَلْعَالِي الْحَلْحَالَةُ اللَّا عَلَيْكَ الْحَلْعَ الْحَلْعَالِ الْحَلْقُ الْحَالِي الْحَالَةُ الْحَلْحَاتِ الْحَلْحَالَةُ الْحَلْحَالِ الْحَلْحَالِي الْحَلَيْلُ الْحَلْقُ الْحَلْحَالَةُ الْحَلْقُ الْحَلْحَالَةُ الْحَلْقُ الْحَلْحَالَةُ الْحَلْحَالَيْ الْحَلْحَالَ الْحَلْحَالَ الْحَلْحَالَيْلُ الْحَلْحَالَ الْحَلْحَالْ الْحَلْحَالُ الْحَلْحَا الْحَلْحَالَةُ الْحَلْحَالَةُ الْحَلْحَالَيْلُ الْحَلْحَالَةُ الْحَلْحَالَ الْحَلْحَالَةُ الْحَلْحَالَ الْحَلْحُ الْحَلْحَالَ الْحَلْحَالَ الْحَلْحَالَ الْحَلْحَالَ الْحَلْحَالَ الْحَلْحَالَ الْحَلْحَالَ الْحَلْحَا الْحَلْحَالَ الْحَ

The pragmatic nature of agentive marking is further illustrated by comparing (5.20) and (5.21) which employ the same verbal expression *néndzop tã:* 'oppress (lit. send oppression)' used by the same person in two instances within the same piece of discourse.

- (5.20)*ódi pε* zak-tiki khõ: néndzop tã:-bo $\tilde{i} = s$, oppression send-2INF EQU.PER=QUO that example set-NF 3SG.HON $\tilde{i} = s$ ódi ge:pu tã:-bo du-ke?. tε then that king send-2INF EQU.PER=QUO EX.SEN-IN 'Setting that as an example, there was (the rumour) that he brought oppression (on the people), the king brought that.' (CY interview)
- (5.21) הישּיון אָז' שָאידָאָליי) אָדָאָי יוזאָד אָדָאָז' יוזאָד אָדָאָז' אָזאָן $j atca = lo \ ge: py:=gi \ dem \ néndzop \ nấ:-bo \ tilde{i}:$ $1PL=DAT \ king.GEN=AGT \ such \ oppression \ do.HON-2INF \ EQU.PER$ 'It was the king who brought such oppression on us.' (CY interview)

In (5.20), A argument is zero marked and P argument is elided (it is recoverable from the context). In (5.21), on the other hand, argument A is agentive marked. Agentive marking in (5.21) is probably motivated both by the overt presence of two arguments (contra 5.20, which has only one overt argument and another one tracable by zero anaphora) and the atypical word order where argument P is topicalized as the first argument.

DeLancey (1990: 306) reports ergative (here agentive) as mandatory in Lhasa Tibetan transitive perfective clauses. Similarly, Takeuchi & Takahashi (1995: 284) state that with perfective verbs "the transitive subject [in Central Tibetan] is almost obligatorily in the ergative case." In some other Tibetic languages (e.g. Häsler 1999: 98), on the other hand, aspect is reported as insignificant for analyzing ergativity/agentivity. In Denjongke, (mono)transitive verbs can be divided into two groups based on the occurrance of the agent marker in past (perfective) clauses, see Table 5.2. Consultant KN reported that the verbs in the first group, when occurring in the periphrastic past construction VERB-2INF EQU have an obligatorily agentive-marked A argument. The past tense A argument of the second group of verbs, on the other hand, may be either zero or agentive marked. A glance at the list reveals that affectedness of the P argument is not a clear conditioning factor as verbs such as *sà* 'eat' and *cik* 'destroy' fall within the second group. Further research is needed to fully unravel the intricacies of agentive marking.

Agentive obligatory (AGT)		Agentive pragmatically conditioned (AGT/ZERO)			
tip	দীন.	'hit'	bak	নেন্যা.	'carry'
re:	হ্যন্য'	'tear'	sà	т.	'eat'
dɛk	क्षेया.	'put inside'	be?	ন্যন্থ'	'weed (out)'
dzip	र्यह्म.	'suck'	cé?	٩٢.	'comb'
hako	স্'র্শ	'understand'	cík	নিযাম'	'destroy'
сé:	नेख.	'(come to)	da:	नन्तर.	'sharpen'
		know'			
sé:	ন্দ্রন্থ'	'rinse'	zi:	শ্রিযাঝ'	'look, see (hon.)'
dze:	नहेनुः	'forget'	$dz \varepsilon:/p^h \varepsilon$	অহন্য'/বর্দ্রন (ସ্রিন্')	'meet'
сú	म्मु'	'peel'			

Table 5.2. Marking of A argument in the past tense of some verbs (consultant KN)

The A argument of verbs which depict involuntary actions may receive dative-locative marking. This point is here illustrated with three verbs, $t^{h}op$ 'receive', go? 'need to' and ga 'love, like'. The A argument with the verb $t^{h}op$ 'receive' can receive either zero, agentive or dative-locative marking. The reason for the possibility of dative marking is probably the semantics that run counter to the syntax: the A argument is actually the most patient like argument in the clause. The following clauses illustrate agentive, zero and dative-locative marking with $t^{h}op$, respectively.

(5.22) هَٰ
יָקָרָיָחָאַי צֿמַקאַי צֿמַיאַר װאָרי אָרָיזין?

 $m \partial by = ki$ $p^h o^2$ $t^h o p - \epsilon \epsilon$ $j \partial z - ka?$ wife=AGT salary receive-INF EX.PER 'Does the wife receive salary?' (BP BB discussion)

 $t\varepsilon$ $k^h \tilde{o}$: $c\dot{e}$:-ruŋ $k^h \tilde{o}$: $p\dot{y}$: $t^h op-o$ then3PLknow-CONCNEG-know-CCS3PLmoneyreceive-2INF $b\varepsilon$?.EQU.NE'Now whether they knew (it) or not, they got money.' (CY interview)

(5.24) אָרָיאִישְרֵי קַקרָישׁי, מָקָאִי אֲבָרָאִיחֵטָ קַקרָישׁי מָלָי אַקיזי אָרָן **khō:=lo**=jã: ố:tɛʰa, dɛndzoŋ=gi ố:tɛʰa=di tʰop-o 3PL=DAT=too authority Sikkim=GEN authority=DEMPH receive-2INF bɛ?. EQU.NE 'They too received authority, authority of Sikkim (subjects).' (CY interview)

In elicitation, consultant KN accepted either agentive or dative-locative marking in (5.25) but judged the use of zero-marking infelicitous:

(5.25) $\Box_{a}/\Box_{a}''' = a \Box_{a} \Box_$

When the verb $t^{h}op$ 'receive' occurs with a dative-locative-marked adverbial, the clause may have two dative-locative-marked arguments, which are disambiguated semantically and also perhaps by word order (the agentive $\eta \dot{a}$: could replace $\eta \dot{a}$ =lo in 5.26).

(5.26) riar range for a fright and the formula for the formula for the formula for the formula formula for the formula formula for the formula formula formula for the formula formu

The second verb that occurs with dative-locative marking of A argument is go^2 'be needed', see (5.27). Similar to $t^h op$ 'receive', go^2 allows for both dative-locative and agentive marking of A argument, whereas zero-marking is infelicitous:²⁰⁹

(5.27) (5.27) (5.27)

 $y\dot{a}:/y\dot{a}=lo/*y\dot{a}$ $tc^{h}u$ go?. 1SG.AGT/1SG=DAT/*1SG water be.needed 'I need hot water.' (KN e)

However, with the negated *mingo?* 'be not needed' all three forms, dative-locative, agentive and zero-marking were deemed acceptable.

(5.28) ८४/८ र्नेर/८ कुर्क्त क्षेत्र मेत्र र्नेषा

 $y\dot{a}:/y\dot{a}=lo/y\dot{a}$ $t\varepsilon^h uts^h \tilde{\varepsilon}:$ min-go?.1SG.AGT/1SG=DAT/1SG hot.waterNEG-be.needed'I don't need hot water.' (KT e)

²⁰⁹ Based on feedback from two additional consultants, only the dative-locative marked form $\eta \dot{a} = lo$ appears to conform to the standards of the written language.

The reason why the non-marked A can occur in a negated clause but not in the affirmed clause may be that negated clauses are lower on the semantic transitivity scale than affirmed clauses.²¹⁰

The third verb that allows dative-locative marking of the A argument is ga 'love, like'. For an example consider the question-answer pair (5.29-31). Note that in the forms volunteered by the consultant, the A argument in the affirmative form is dative-locative marked but the negated form remains non-marked.

- (5.30) $\Box \dot{\Delta} \Box \dot{\Delta}$
- (5.31) 5' क्रेंक्'त्यात| **yà** miŋ-ga. 1SG NEG-like 'I don't like (them).' (PED e)

In conclusion, the marking of argument A is conditioned by the nature of the verb and pragmatics.

5.3.2 Argument P

As shown in examples (5.12-13) above, argument P may be either zero marked or dative-locative marked. This optionality was noticed already by Sandberg (1895: 22), who comments that the accusative form of the word k^{him} 'house' may be either k^{him} (zero-marked) or $k^{him}=lo$ (dative-locative-marked). For the existence of two "accusative" forms, Sandberg (1895: 24) offers a functional motivation based on disambiguation: "The accus[ative] case may be expressed by the simple word without the affix lo, where no ambiguity would result as to which were the nominative, especially in imperative sentences."

In elicitation, consultant KN could not distinguish any difference in meaning between the zeromarking and dative-locative marking in (5.32). Argument P is given in bold.

²¹⁰ I am grateful for Bertil Tikkanen for suggesting this hypothesis. In Finnish, for instance, an affirmed clause may receive either a genitive object (totally affected) or partitive object (partially affected). In a negated clause, however, only partitive marking is allowed. Thus, less affectedness, which is a feature of lower semantic transitivity (Hopper and Thompson 1980), is associated with negation in Finnish.

²¹¹ Loan from Nepali.

²¹² According to consultant KUN, also zero-marked $y\dot{a}$ and agentive-marked $y\dot{a}$: could occur here instead of dative-marked $y\dot{a}=lo$. KUN further noted that dative-marking is not felicitous with the negated clause (5.31).

(5.32) $g^{\alpha_i \circ a_i \delta_i \delta_i}(\hat{g}^{\alpha_i}) \eta_{\delta_i}(\hat{\alpha}^{\circ}) \circ \alpha_i \delta_i^{\circ} \delta_$

In other instances, however, consultants were able to identify a pragmatic difference between zero-marked and dative-locative-marked P arguments. For an example, consider (5.33) and (5.34).

- (5.33) $\mathfrak{g}^{\mathcal{A}',\mathcal{A},\mathcal{B},\mathcal{B}',\mathcal{A}'$
- (5.34) العَلَى العَ العَلَى ال العَلَى العَلَى العَالَى العَلَى ال العَلَى اعَلَى العَلَى العَلَى العَلَى العَلَى العَلَى العَلَى العَلَى اع

Consultant KTL commented that in (5.33) the P argument (pig) is identifiable in the context. It is a specific pig which most likely is in the speakers presence. In (5.34), on the other hand, the P argument is unidentifiable, unspecific. It should be noted that the clauses differ not only with respect to P marking; (5.33) also has a demonstrative-emphatic =di which can mark definiteness. Consultant KTL tended to identify a pragmatic difference in P marking mainly in imperative clauses.

For another case of reported pragmatic difference, consider (5.35) and (5.36).

- (5.35) $\mathbb{R}^{r}(\widehat{\eta}\mathbb{N}^{r}) \stackrel{*}{\leftarrow}$ अर्घन् र्'र' क्षेत्र] $k^{h}u(=gi)$ do $t^{h}\widetilde{o}:-po$ $\widetilde{t}:$. 3SGM(=AGT) stone see-2INF EQU.PER 'He saw a stone.' (KN e)
- (5.36) $\mathbb{R}^{(\hat{\eta} \mathbb{N}^{*})} \check{\xi}^{\hat{\mathcal{A}}^{*}} a\check{\mathfrak{A}}\check{\xi}^{\hat{\mathcal{A}}^{*}} \check{\mathfrak{A}}_{\hat{\mathfrak{A}}_{1}}^{\hat{\mathcal{A}}^{*}} \check{\mathfrak{A}}_{\hat{\mathfrak{A}}_{1}}^{\hat{\mathcal{A}}^{*}} i a\check{\mathfrak{A}}_{\hat{\mathfrak{A}}_{1}}^{\hat{\mathcal{A}}^{*}} i a\check{\mathfrak{A}}_{1}^{\hat{\mathcal{A}}^{*}} i a\check{\mathfrak{A}}_{1}^{\hat{\mathcal{A}}} i a\check{\mathfrak{A}}_{1}^{\hat{\mathcal{A}}} i$

Consultant KN reported that whereas (5.35) would be a fairly neutral statement, the use of the dative-locative in (5.36) implies that the speaker saw a stone and not the other things that were available for seeing. Consultant KUN, on the other hand, considered the dative form with an inanimate object in (5.36) infelicitous. He suggested replacing do=lo [stone=DAT] with $k^h \tilde{o}:=lo$ [3SG.HON=DAT]. Based on comments from these two consultants, it may be initially summarized that P marking is conditioned by specificity/identifiability and animacy.

The role of animacy in differential P marking is confirmed by data from the novel Richhi, where specific human referents as P arguments of the verb $t^{h} \tilde{o}$: 'see' are always marked with dative-

locative, see (5.37) and (5.38a).²¹³ Negation does not appear to affect dative-locative marking of animate P-arguments, see negated (5.38b) which bears dative-locative marking.

- (5.37)mik jàte lóptø: ritc^hi. no:=lo karma=lo t^hõ:-pø: school.GEN direction=DAT eve up PN=DAT see-2INF.GEN hope 'Eyes up towards the school, in hopes of seeing Karma.' (Richhi 69) a) सु'गविषाख्त प्रयार्ख के मुगार्गे में रापविता यें रा गविषार्या के के राष्ट्र (5.38)*p*^h*am*=*tsu mitako* p'uzi-tcẽ: có:-zẽ: jò? khõ: children-AZR parents=PL impermanent feel-PROG EX.PER 3PL pí:-po=lo t^hõ∶-ti. two-COL=DAT see-NF 'Parents with children have a feeling of impermanence when seeing the two of them.' (Richhi 82)
 - b) $\tilde{\mathfrak{s}}_{\neg}$ $\tilde{\mathfrak{y}}$ \mathfrak{w}_{\neg} $\tilde{\mathfrak{v}}$ \mathfrak{w}_{\neg} \mathfrak{w}_{\neg} $\tilde{\mathfrak{w}}_{\neg}$ $\tilde{\mathfrak{v}}$ \neg \neg $\tilde{\mathfrak{w}}_{\neg}$ $\tilde{\mathfrak{w}}_{\neg}$ $\tilde{\mathfrak{v}}_{\neg}$ $\tilde{\mathfrak{w}}_{\neg}$ $\tilde{\mathfrak{w}}_{\sigma$

However, when the P argument is the pronominal ka- $j\tilde{a}$: '(not) anyone' (always followed by a negated verb), no dative-locative marking is needed, because the referent is unidentifiable: ²¹⁴

क्षें प्रदेन केंत् प्रमा गांधा स्टा ह्या गांधा गांधीता स्टा ही स्वार केंद्र से केंद्र मा से से से से से से से म (5.39)*ódɛtsʰika* karma hã:to:lo pí:sé:-ti $p^{h}a$:- $ts^{h}u$: thither-hither at.that.moment PN suddenly wake.up-NF ta-wø: gã: ka-jã: $ma-t^h \tilde{o}$. look-2inf.gen time who-even NEG-see 'At that time Karma suddenly wakes up and as he looks around he doesn't see anyone' (Richhi 115)

While all human P arguments of the verb $t^{h}\tilde{o}$: 'see' in the novel Richhi are marked as dativelocative, non-human (or inanimate) P referents of $t^{h}\tilde{o}$: 'see' are zero marked:²¹⁵

²¹³ Consultant KN, however, commented that =lo could be omitted in (5.37) and (5.38a).

²¹⁴ According to KN =lo could not be added in (5.39).

²¹⁵ However, consultant KN accepted using a dative-locative P argument instead of non-marking in (5.40).

(5.40)वन् रेंदि केन नन कें कन कोन केंन केन के कि के के के के के के nè:pø: mìŋ t'ã: lòts^he? mèː-po lo: patient.GEN name and age have.time.to NEG.EX.PER-2INF see k^hu ĩ. t^hetsom sà-ce 3SGM doubt eat-INF EQU.PER 'As soon as he sees the patient's name and age, he has doubts.' (Richhi 170)

The difference between dative-locative and zero-marking of P, however cannot be reduced to animacy, because human P referents may also be zero marked if unidentifiable/unspecific (although the P argument in [5.41], according to KN, could also be marked with =lo):

(5.41) $k^h u = gi$

खि.ग्रीख. थ. गड़िया. राखर. घूरी mí=tci? sép-o bɛ?. 3SGM=AGT (hu)man=INDF kill-2INF EQU.NE 'He killed a (hu)man.' (KTL e)

Affectedness of the P argument also appears to be a factor in dative-locative vs. zero-marking: P arguments which are totally affected and change a state are zero marked whereas less affected P arguments may be locative marked. Dative-locative marking with a totally affected P argument in (5.42b) was considered infelicitous by consultant TB.

(5.42)	a) હ્યુ'મીચ' દ્વરે' મેંચિ'ર્દ્ધ'	comput) ، بحج	er) বান্তবা'র্গা' শ্রদ্বা		
	k ^h u=gi	рè:	kompjutər	t£ak-o	bɛ?.
	3sgm=agt	1sg.gen	computer(Eng.)	break-2INF	EQU.NE
	'He broke m	y computer.	' (TB e)		
	b) * איזאי בא׳ זאשי:	ידהיצ (compi	iter) র্নি' বান্তবা'র্শা' শ্রদ্য		
	$*k^h u = gi$	nè: k	ompjutər=lo	tɛak-o	bɛ?.
	3sgm=agt	1SG.GEN c	omputer=DAT(Eng	.) break-2INF	EQU.NE
				,	-

Affectedness, however, appears to interact with humanness vs. non-humanness of the actor, as suggested by (5.43) and (5.44) which are again from consultant TB. If the direction of the action was from an animal to a human, both dative-locative and zero-marking were accepted for marking the P argument, see (5.43). With a human A argument and animal P argument, on the other hand, dative-locative-marked P argument was not considered felicitous, see (5.44).

(5.43)a) ଶ୍ମୁମ୍ଟ ସ୍ୱିମ୍ବା ରି' ସବ୍ୟମସଂଖ୍ୟମ୍ବ

> sé-b-be?. lán=gi тí bull=AGT human kill-2INF-EQU.NE 'A/the bull killed a man.' (TB e)

b) କ୍ଷୁଦ୍ଦ ସ୍ୱିନ୍ଦ ଭିଂନ୍ଦି ସ୍ୟୁଦ୍ଦ ଅନ୍ତ୍ର

sé-b-be?. láŋ=gi mí=lo bull=AGT human=DAT kill-2INF-EQU.NE 'A/the bull killed a man.' (TB e)

b) ??שָּׁמִימּשׁׁמִ שֵׁמִי אֵרִימֹי הַמּרְיגֿי אַרן
 ??gjaltshen=gi lấ:=lo sé-po be?.
 Gyalsthen=AGT bull=DAT kill-2INF EQU.NE

Analyzing transitive clauses is complicated by complex predicates, where the predicate already contains an element that looks like a P argument. For an example, consider the complex predicate *ka:gjur tã:* 'instruct' (lit. 'instruction send'), which seems a semantically unified concept (as suggested by the English translation 'instruct') but syntactically looks like a PV sequence. Based on clauses (5.45) and (5.46), it indeed looks like the language system considers *ka:gjur* a P argument. In (5.45), the bare verb *tã:* occurs with the zero-marked argument $\delta ni=di:=tsu$ 'the children', suggesting that in (5.46) *ka:gjur* is the P argument and the dative-locative $k^hu=lo=di$ a peripheral argument.

- (5.45) \tilde{k} (\tilde{k} : \tilde{g} : \tilde{k} : \tilde{g} : \tilde{k} : \tilde{g} : \tilde{k} : \tilde{g} : \tilde{k} : \tilde{g} : \tilde{k} : \tilde{g} : \tilde{k} : \tilde{g} : \tilde{k} : \tilde{g} : \tilde{k} : \tilde{g} : \tilde{k} : \tilde{g} : \tilde{k} : \tilde{g} : \tilde{k} : \tilde{g} : \tilde{k} : \tilde{g} : \tilde{g} : \tilde{k} : \tilde{g} : \tilde{g} : \tilde{k} : \tilde{g} : \tilde{g}
- (5.46) $\vec{\beta}$, \vec{a}_{1} , \vec{c}_{1} , \vec{a}_{1} , \vec{c}_{1} , \vec{a}_{1} , \vec{c}_{2} , \vec{c}_{2} , \vec{c}_{2} , \vec{c}_{2} , \vec{c}_{1} , \vec{c}_{2} , \vec{c}_{2} , \vec{c}_{1} , \vec{c}_{1} , \vec{c}_{2} , \vec{c}_{1} , \vec{c}_{2} , \vec{c}_{1} , \vec{c}_{2} , \vec{c}_{2} , \vec{c}_{1} , \vec{c}_{2} , \vec{c}_{2} , \vec{c}_{2} , \vec{c}_{1} , \vec{c}_{2} , \vec

However, example (5.47) shows that the P argument of the verb $t\tilde{a}$: may also be marked as dative-locative, leaving it undecided whether *ka*:*gjur* in *ka*:*gjur* $t\tilde{a}$: 'instruct' should be considered a P argument or part of the predicate.

वत्र र्यतेः गवत् स्थ्रूटयाः कर्यके सुया येत् मत् रत्या गाम्नागीया वत्र ये पत्ते र्ये देखेः यहत्यत् द्वियायवेवः यत् (5.47)né:tã: ts^haptcecy: j∂:cɛ=di: né:pø: karma=gi patient.GEN condition serious EX.PER=DEMPH.AGT PN = AGT né:po=di=lo dilli **tã:-**ce $p' i a - z \tilde{\epsilon}$ jờ?. patient=DEMPH=DAT TPN send-INF do-PROG EX.PER 'Because patient's condition is serious, Karma is preparing to send the patient to Delhi.' (Richhi 169)

Evidence for considering the P-like element in the verb complex a part of the verb and not the actual P argument comes from (5.48), where the P argument of the complex verb $j arg \varepsilon t \tilde{a}$: 'develop (lit. send progress)' is zero marked.

(5.48)dep p'ja-ti **nàtci ỳ:=di** ŋà jàrge? tã:-cε=di like.that do-NF our place=DEMPH progress 1SG send-INF=DEMPH nó:sam=di tã:-do bε?. thought=DEMPH send-IPFV EQU.NE 'Like that I'm thinking about developing our place.' (PED life story)

The same verb may also occur with a dative-locative-marked P argument:

(5.49) הישה' אך אל איז שישש אי אדר לעאישאי אדר אלישי אדן *nàtci* $k\epsilon:=di=lo$ *jàrge? tã:* $go=p\epsilon?$. 1PL.GEN language=DEMPH=DAT progress send be.needed=EQU.NE 'Our language needs to be developed.' (KL BLA 12)

In conclusion, P marking is sensitive to animacy, identifiability/specificity and affectedness of the referent. The first two factors may be connected in that human referents are probably more likely to be identifiable than non-human referents.

5.3.3 Ditransitive clauses (T and R argument)

Ditransitive clauses have three arguments A(gent), T(heme) and R(ecipient)/S(ource) (Haspelmath 2005). Haspelmath (2005: 2) introduces three types of alignment in ditransitive (or ditransitive) clause: 1) indirective alignment where T argument of the ditransitive clause aligns with the P argument of the monotransitive clause, leaving R separately marked, 2) neutral alignment where P, T and R are all marked the same way, and 3) secundative alignment where P and R are aligned in opposition to T. As suggested by the discussion at the beginning of the chapter, Denjongke does not fit nicely in any of the three patterns because the marking of P is split between zero-marking, which is also used for argument T, and dative-locative marking, which is also used for argument R.

The order of arguments is ATRV, as in the main clause in (5.50), or ARTV, as in (5.51), because either T or R may occur after A in the more focal position (see Givon 2001: 270). Occasionally, a topical R or T may also occur before A, see (5.57) further below. In (5.50), the R argument of the main clause, Choki, is already topical information whereas the T argument, letter, is new information and hence focal.

(5.50)	র্কুঝান্ধুনি. দ্রী	र्वेगायें दर्वे हें देंद	ননি শ্লবন্দান্য নাস্থা	<u>ৰ্বা</u> ন্তবা-বীৰ্ষা আঁন্বী' ব্যন্তিবা' ই	ग्रॅंग्यें' ईंब्र'नवेव'	
	t&^øki?	$p^{h}ilo=lo$	t ^h øn-di	òm-bø:	kap=lo	p'otso
	PN	outside=DAT	come.out-NF	come-2INF.GEN	time=DAt	child
	t£i=ki	jìgi=tɕi?	mù=lo	p`in-zɛ̃ː		
	one=AG	T letter=INDF	3sgf=dat	give-PROG		
	'As Cho	oki is coming ou	utside, a boy gi	ives her a letter (s	aying)' (I	Richhi 8)

In (5.51), on the other hand, argument R, being more focal, comes first.

(5.51) ה׳ בָּוֹל מִימִוּמִימִי בָּזָאָ דָאָז פֿאָ ידע מָדָי רָזָוֹמִיגורן nà bhaila=lo mén p'im-ba gju go:=pe?. SG PN=DAT medicine give-PUR go be.needed-EQU.NE 'I have to go to give medicine to Bhaila.' (Richhi 34)

In (5.51), the speaker has just requested the addressee to stay in her place while she goes to the hospital to attend the patient Bhaila. Because meeting Bhaila implies a contrast in location (here vs. at the hospital), argument R (Bhaila) is in this context more focal information than argument T (medicine).

In ditransitive clauses A argument is either agentive or zero marked, T argument is zero marked and R argument is typically dative-locative marked but may occasionally also be zero marked or be marked with the postpositional clitic =tsa: 'at, by'. It should be remembered that because of zero anaphora and argument suppression none of the arguments is mandatorily overtly present in a clause with a ditransitive verb such as *p'in* 'give'. The following two examples illustrate ditransitive clauses with agentive-marked argument A. Arguments T and R are zero marked and dative-locative marked, respectively.

- (5.52) $\neg_{G_{n}} \neg_{T_{n}} \neg_{T_{n$
- हे' क्ष'बे'गेक'र' झ्वा' ईक्ष' हे' दाख्या'र्थे' शुक्ष'शुक' ह' ∃काई' तत्रवा' देंह' ईक्र'हें' भ्रहा (5.53)ázi=ki=ra duk p'ja-ti nàtca=lo cý:cv: tε t'a then elder.sister=AGT=AEMPH pain do-NF 1PL=DAT a.little now ba? ồ: p'in-do sàm=tsu $b\epsilon$?. food.HON=PL carry come give-IPFV EQU.NE 'It was the elder sister who, toiling, used to bring a little food for us.' (PED life story)

The agentive in (5.52) is probably motivated by the inanimacy of the argument because of which its causal relation to the other arguments needs to be stressed. In (5.53), the agentive underlines the elder sister's agency, which is further highlighted by the emphatic marker =*ra*.

Examples (5.54) and (5.55) illustrate zero-marked argument A. In (5.55), argument R is elided because it is recoverable from the context.

(5.54) युः द्विंग्सं क्षें च्वें न्यत्वेन तर्गा

k ^h u	$k^{h}i=lo$	to	p'in-zen	du?.
3sgm	dog=DAT	cooked.rice	give-PROG	EX.SEN
'He's gi	ving the do	g rice.' (TB e)	C	

(5.55) $\vec{p} \cdot \vec{u} \vec{u} \cdot \vec{q} \cdot \vec{p} \cdot \vec{z} \vec{u}^{216} \vec{z} \vec{q}$ $t\varepsilon \quad \dot{am} = di \qquad t\varepsilon \quad t'om \qquad p'in.$ then mother=DEMPH then butter.dough²¹⁷ give 'Then the mother gives butter-dough (to the child). ' (LA birth in Lachung)

In (5.56), the R argument is zero marked, similarly to directives and locatives (see §5.1).

(5.56) $\vec{p} \cdot \vec{a} \times \vec{E} \times \vec{p} \cdot \vec{\Delta} \cdot \vec{E} \times \vec{p} \cdot \vec{D} \cdot$

The reason why dative-locative marking can be elided in (5.56) is probably that the R argument is a location/destination rather than a person who can possess the item in an abstract sense. Argument A is here suppressed (not recoverable from the context), the clause becoming a "functional passive" (see §5.5.1).

Example (5.57) illustrates the use of the postposition =tsa: 'at, by' used in place of the dativelocative as a marker of the R argument. The T argument is the topic established by the previous context so it occurs here even before A argument.

In elicitation, consultant KN provided past and future forms of p in 'give' with an agentive A argument (5.58a-b) and the imperfective with a zero-marked A (5.58c).

(5.58)	a) מקדי דמי ופימ	ग देय. बहिब	ા ફોંન 'ર્ચે' ક્ષેન્		
	dã:	ŋá:	$k^h u = lo$	t'ep=tei? p'im-bo	$\tilde{\iota}$
	yesterday	I.AGT	3sgm=dat	book=INDF give-2IN	NF EQU.PER
	'Yesterday	I gave l	nim a book.' (K	(N e)	
	b) র্ন্ন্য নম দ্ব	ुर्वे देय गई	^{કે} ગા ફેંવ વત્ જેવા		
	t ^h orã:	ŋá:	$k^h u = lo$	t'ep=tci? p'in-ce	ź.
	tomorrow	I.AGT	3sgm=dat	book=INDF give-INI	F EQU.PER

^{&#}x27;Tomorrow I'll give him a book.' (KN e)

²¹⁶ Consultant KT would write this word as $\breve{a}_{\neg \neg \neg}$ and pronounce it as $t' \varnothing m$, exemplifying a dialectal difference between LA (Lachung, North Sikkim) and KT (Bermeok, South Sikkim).

²¹⁷ Cooked tsampo-dough mixed with butter.

²¹⁸ Mixture of butter, tea and flour, has harder texture than *t* 'om 'butter-dough'.

c) $\neg \check{\mathfrak{F}} : \neg \check{\mathfrak{R}} : \check{\mathfrak{A}} : \check{\eta} \check{\mathfrak{S}} \check{\eta} : \check{\mathfrak{G}} \check{\mathfrak{A}} : \check{\mathfrak{T}} : \check{\mathfrak{R}} \check{\mathfrak{A}} : \check{\mathfrak{R}} \check{\mathfrak{A}} : \check{\mathfrak{R}} \check{\mathfrak{A}} : \check{\mathfrak{R}} \check{\mathfrak{A}} : \check{\mathfrak{R}} : \check{\mathfrak{R}} \check{\mathfrak{A}} : \check{\mathfrak{R}} : : \check{\mathfrak{R}} : \check{\mathfrak{R}}$

For T arguments marked by complement clauses, see §14.1.1.

5.4 Clauses with verbs of being and becoming

5.4.1 Equative clause

In a prototypical equative clause two non-marked noun phrases occur as copular arguments. The noun phrase may also be substituted by a nonfinite clause, see (5.60b). In equative clauses, the equative copulas i: and $b\epsilon$? either equate two arguments (5.59) or identify/classify the first argument as a member of a category (5.60). Dryer (2007: 233) calls similar comparative categories "referential" and "non-referential" respectively, whereas Pustet (2003: 29) calls them "identificational" and "ascriptive" respectively.

nílam làp-kh $\tilde{\epsilon}$: *nílam=rã*: *b* ϵ ?. dream say-NMLZ dream=AEMPH EQU.NE 'A dream is (just) a dream.' Lit. '(The thing) called dream is indeed a dream.' (Richhi 116)

- b) مَن حَمْ: مَن حَمْ: هُمَا mù nè: mòby? í:.
 3SGF 1SG.GEN wife EQU.PER
 'She is my wife.' (KN e)
- (5.60) a) 成二 अवर 是· 图5

 $k^{h}\tilde{O}$: $\acute{a}mdzi$ $b\varepsilon$?.3SG.HONdoctorEQU.NE'He is a doctor'

²¹⁹ This word may also be spelled and gran gnid-lam, which reflects the pronunciation better.

The second argument may be a genitive-marked nominal which identifies the second argument as the possessor of the first argument, see (5.61). This construction is best seen as a case of elision where the possessed item is elided from the second argument.

(5.61) a) पर्ने प्रेंग ĩ. di nè: this 1SG.GEN EOU.PER 'This is mine.' (KN e) b) गुवरकेंगीया धेर्यादे । यामुना क्षेत्रे कर केंग केंग केंग केंग केंग के या या प्रवित *kantc*^{*h*}*i*=g*i* jìgi: k^hadzã: ta-ti younger.sister(NEP)=AGT letter.GEN address look-NF t¢^høki=ki làp-zẽ: ái $b\varepsilon = s\varepsilon$ elder.sister PN=GEN EQU.NE= QUO say-PROG 'Kanchi looks at the address in the letter and says (it) is elder sister Choki's.' (Richhi 139)

In spoken language, the equative copula is sometimes elided, hence the (round) brackets in the translation of (5.62).

८.२०२५. पर्येश किंट्रा, थेता. क्षेत्र, त्रांच, त्रांच, ग्रांच, श्रांच, श्रांच, श्रंच, संया. क्षेत्र, यांक्र्या, पट्ट, चेंट्र, स्रेंग्रां, स्रंग्र, स्रंग्र, त्रं, त्रं, स्रंग्र, संग्र, स्रंग्र, संग्र, स संग्र, संग संग्र, स संग्र, संग संग्र, स (5.62) $d \in ndz \tilde{o}$: $n a \eta lo p^h u k - t \varepsilon^h \tilde{e}$: z iiờ:-kʰɛn bɛ?. nàtci $p^huk-tc^h\tilde{e}$ 1PL.GEN Sikkim inside cave-great four EX-NMLZ EQU.NE cave-great tci:=di cár-tc^ho: $b\varepsilon:p^hu?$. one=DEMPH east-direction Bephuk 'In our Sikkim there are four great caves. One great cave (is) eastern Bephuk.' (SGD cave story)

Equative copulas are also used in possessive constructions (§5.4.3), with predicate adjectives (§5.4.4) and, marginally, with locative arguments (§5.4.2).

5.4.2 Existential and locative clauses

Existential clauses have a single non-marked argument which occurs with an existential copula. The three basic existential copulas are the personal $j\partial^2$ (neg. $m\partial^2$), neutral $j\partial bb\epsilon^2$ (neg. $m\partial bb\epsilon^2$)²²⁰ and sensorial du^2 (neg. $mndu^2$). For the semantic differences between them and for examples of more marginal existential copulas, refer to §7.

(5.63) R. Bara. ũ

 $k^{h}u$ $k^{h}im=na$ $j\partial 2$. 3SGM house=LOC EX.PER 'He is at home.' (KT e)

²²⁰ Reduced from nominalized construction $j\dot{a}$:-po be? (neg. $m\dot{e}$:-po be?), see §7.3.2.1.2.

(5.64) يَقْدَرُسَانِهُ هُمَاتَكُ بَجَ عَمَامَعُ سَمَرَ مَعَالَمُ عَمَاتُكُ فَمَامَعُ سَمَرَ مَعَالَمُ مُعَالًا مُعَالً

(5.65) Q: র্কু' র্ইন্ব'শ্।?

ts^ha t^hop-ka? salt find-PQ 'Is there salt?' (KN e)

A: אָאָ־רְקָשְׁאָרְן mindu?/mè?. NEG.EX.SEN/NEG.EX.PER 'There is not.' (KN e)

As shown in the answer in (5.65), even the single argument may be elided if the context allows, thus leaving the existential copula the only obligatory word in an existential clause. Locative clauses are an extension of existential clauses in that they add an adverbial argument which specifies the location of existence. The more topical argument comes first. In (5.66), the locative adverb is topical and therefore comes first. In (5.67), on the other hand, the argument that marks the located object is topical and hence comes first (note that [5.67] also has a temporal adverbial).

(5.67) a) कॅर्बा क्रिन गुन केन ते ही आ न थेन

 $tc^h \phi ki?$ k'uŋsiŋ=na $k^h im=na$ $j \dot{\phi}?$. Choki holiday=LOC house=LOC EX.PER 'Choki is at home during the holiday.' (Richih 157)

'Is there anyone (lit. human) at home?' (rna-gsung 5)

b) ק'פּ' בּּרֹיטעקאי צַאַק'נעריק' עַק'יק?
 t'ato bhaila méŋkhã:=na jà:-ka?
 now PN hospital=LOC EX.PER-PQ
 'Is Bhaila now in hospital?' (Richhi 24)

The locative argument may also occur in the antitopical right dislocated position.

With frequently used toponyms elision of locative marking is quite frequent:

(5.69) $\mathbb{R}^{n} \approx \mathbb{R}^{n} \mathbb{R}^{n}$ $\mathbb{R}^{n} \mathbb{R}^{n}$ \mathbb{R}^{n} $\mathbb{R}^{$

Similar to equative copulas, existential copula may be elided in spoken language.

(5.70) अप्त झुरुप पात?

áca lamo k'ana? PN PN where 'Where (is) Asha Lhamo?' (SGD cave story)

Occasionally locative arguments may co-occur with an equative copula. While the evidentially neutral $b\varepsilon_i^2$ is fairly frequent in this use (5.71a), the only context in my data where a locative argument is accompanied by personal \hat{t} is telling one's place of origin (5.71b).

(5.71) a) هَرَ، هَرَ عَرَيْ اللهِ (5.71)

έ:, óna bε?.oh there EQU.NE'Oh, it is there!' (SM kitchen discussion)

b) $\Box \hat{\alpha}$ $\forall \forall \theta$ $\dot{\tilde{\alpha}}$ $\hat{\delta} \hat{\alpha}$ $\hat{\delta} \hat{\alpha}$ $\hat{\delta} \hat{\alpha}$ $n\hat{e}: p^{h}ay: sine=lo \tilde{t}:.$ 1SG.GEN fatherland Sinye=DAT EQU.PER 'My native place is (in) Sinye.' (DB life story)

5.4.3 Predicative possession

In predicative possession, the possessor, which accompanies an existential copula, is expressed either by a locational (5.72) or a genitive-marked argument (5.73) (for these and other types of predicate possession, see Stassen 2009). The locational argument is marked either by dative-locative (5.72a) or the cliticized postposition =tsa: 'at, beside' (5.72b). Using the postposition =tsa: implies that the possessed item is with the speaker at the time of speaking.

(5.72) a) הַיהֹי שּׁיחִי שְׂשׁקִי שְׁהִישִּתְקִי שֵּק t'a ŋà=lo tc^ha:ka=tci? jà:-k^hɛn bɛ?. now 1SG=DAT item=INDF EX-NMLZ EQU.NE 'Now, (let's imagine) I have a certain item.' (YR canteen video)

- b) $a\hat{\gamma} \cdot c\hat{\alpha} \cdot \underbrace{\mathfrak{F}}_{i} \underbrace{\mathfrak{W}}_{j}$ $di \quad n\hat{e}:=tsa: \quad j\hat{\phi}?.$ this 1SG.GEN=at EX.PER 'I have it (with me).'/'I have it (on me).' (KN e)
- (5.73) a) สังสาคงสา นณิ พาสูง ผันๆ

ónalɛ pè: átɛu jǿ?. then 1SG.GEN elder.brother EX.PER 'Then I have an elder brother.' (PED life story)

- b) $[g\hat{\alpha}, \tilde{\alpha}, \tilde{\alpha}, \tilde{\alpha}, \eta]$ $k^{h}u=i$ p'otso súm du?. 3SGM=GEN child three EX.SEN 'He has three children (I see/saw).' (KN e)

Although genitive-marked constructions have traditionally been termed "genitive possessives" (e.g. Heine 1997), Stassen (2009: 107) proposes the term "adnominal possessive." The reason is that Stassen (2009) analyzes the adnominal possessive construction as having only one nominal constituent, where the genitive-marked possessor is a modifier of the following possessee argument (in English analogy, "[my house] exists" instead of "[my] [house] exists"). Stassen (2009: 113-115), however, acknowledges that if there is evidence against the status of the possessor and possessee as forming one argument (for instance, the possessor and possessee can be separated by another constituent), the genitive-marked possessive may in fact be a "variant of the Locational Possessive" (Stassen 2009: 113). Example (5.73c) indeed provides such evidence. The locative argument $g\tilde{a}:to:=lo$ 'in Gangtok' follows the genitive-marked argument and so separates the purported modifier/adnominal (*karma=gi*) from its head (*t'ep*). Because I have no evidence from elsewhere that the constituents of a noun phrase could in prose²²¹ be separated in this way, I prefer to analyze the genitive-marked possessor as a separate constituent, which functions analogously with cases of locational possession.

If the possessor is marked locationally (i.e. with dative-locative or postposition =tsa), it can occur as the second argument in constructions where the possesse is topicalized and hence occurs first, see (5.74a) and (5.74b). The genitive-marked possessor, however, cannot occur second in a possessive construction, see (5.74c).

²²¹ In poetic language, however, the usual ordering may be broken, as shown in the example below, where the typical prose ordering $n\dot{e}$:=gi sém 'my mind' is changed and an adverbial occurs in between.

रोक्ररा भेयाङ्घे दर्भगी वर्षा झुन्

sém l'épti n e:=gi n a-u b e?.

mind very.much 1SG.GEN=GEN be.ill-2INF EQU.NE

^{&#}x27;My heart was in great pain.' Lit. 'My mind fell very ill.' (Richhi 85)

(5.74)a) क्रेंग्दर्ने उल्लगा दर्ने विंत्र कें येंत/दर्गा i*à?/du*?. ódi $k^h \tilde{o} := lo$ tcala=di that thing=DEMPH 3SG.HON=DAT EX.PER/EX.SEN 'He has that thing.'/'That thing is with him.' (KUN e) b) क्रेंग् देने उल्लया दने विंत रूर दन्या $k^h \tilde{o}$:=tsa: j*à?/du*?. ódi tcala=di that thing=DEMPH 3SG.HON=with EX.PER/EX.SEN 'That thing is with him.' (KUN e) c) *र्से तर्न र जन्मा तर्न वित्त्वी सेन/तर्मा *ódi tcala=di $k^h o \eta = gi$ j*à?/du*?. that thing=DEMPH 3SG.HON=GEN EX.PER/EX.SEN

The possessive construction can, perhaps surprisingly, also occur with an agentive argument. A simple possessive construction has the possessor argument marked as dative-locative (5.75), but when a P-like argument such as lenge:=lo in (5.76) or $te^h \emptyset:=lo$ (5.77) is added, the possessor argument switches to agentive.²²²

- (5.75) $\neg \forall \hat{\tau} \cdot \hat{\tau} \cdot \hat{\vartheta} \cdot \check{u}_{\gamma}$ $\eta \dot{a} = lo \quad rite^{h}i \; j\dot{\phi}?.$ 1SG=DAT hope EX.PER 'I have hope.' (KN e)
- (5.76) $\sum_{n \neq n} \sum_{j \neq n} \sum_{j \neq n} \sum_{i \neq j} \sum_{i \neq n} \sum_{j \neq n} \sum_{i \neq j} \sum_{j \neq$

'That thing is his.' (KUN, KN e)

5.4.4 Predicate adjectives

Both the existentials (personal $j\partial 2$, neutral $j\partial b\epsilon^2$, sensorial du^2) and the equatives (personal \tilde{t} ; neutral $b\epsilon^2$) are used in adjectival predication, analogously to the functionally similar forms in Lhasa Tibetan (Chang & Chang 1984: 608, 614-616; Tournadre & Dorje 2003: 119-122). For examples, see (5.78) for existentials and (5.79) for equatives. The semantic differences between the various copulas are addressed in more detail especially in §7.2.3.

²²² $\eta \dot{a}$: and $\eta \dot{a} = gi$ are two alternative ways to mark 1SG agentive.
(5.78)	a)	ાલું તે ગો. ચ.છેત્રો. (pike) લદ્દી. લુન. લાહતાં. જ્ઞાનાં વાટી					
		$k^{h}u=i=gi$ bai $k=di$ lèp màla? j ∂ ?.					
		3SGM=GEN=GEN bike(Eng.)=DEMPH very.much fast EX.PER					
		'His (motor)bike is very fast.' (NB e)					
	b)	क्षें न प्रेय के में त्र के में के में में में में में में में में में मे					
		ónale hindi=di lèp dzikţa? jèbbe?.					
	then Hindi=DEMPH very excellent EX.NE						
	'Then, (their) Hindi is most excellent.' (DR discussion with KL)						
	``	murer in Article de Article					
	c)	and the age of the age					
		ιοrpar=κι <i>μιηρο</i> ιερτι ιε <i>m</i> αυγ.					
		X-ray=GEN essence very good EX.SEN					
		The results of the X-ray look very good. (Richni 29)					
(5 79)	a)	क्रॅंग् दर्रेग प्रयोग प्रयोग याया केवा क्रिया					
(5.77)	u)	$\dot{o}di$ $\dot{h}\dot{a}ko$ go:-po k' :: $te^{h}\dot{\tilde{t}}$: $\dot{\tilde{t}}$:					
		that understand need.to-2INF important EQU.PER					
	'It is important to understand that.' (Richhi 7)						
	b)	हतः चन्छः बग्गयान्छन् भयाः तह्यायाः हयायः अवी जन्मन्नः अवी					
		<i>ne:</i> $p'otso$ $t^{n}amtee = le$ $dzikta?$ <i>i:</i> , <i>lem i:</i> .					
		'My shild is the best of all a good one '(PPM story of my son)					
		My child is the best of an, a good one. (KDW story of my son)					
	c)	क्षें पर्दर चेत्र ह्वे चेत्र ह्वत्					
	-)	ódi lèpti lèm be?.					
		that very good EQU.NE					
		'That is very good.'					
	-	2 222					
	d)	लह्लाःग्रेह्र पर्देवः गविषाःश्रह्ला अवे.ख.रह्ता वा.यह्यू त्यायाः ह्यायाः ह्यायाः ह्यायाः व्य					
		$dzamlin$ di : $n\dot{\varepsilon}$: $t\tilde{a}$: $n\dot{\varepsilon}$: $mu=r\tilde{a}$: $j\dot{a}mts^{h}ita$? $b\varepsilon$?.					
		world this.GEN condition really=AEMPH amazing EQU.NE					

'The condition of this world is really amazing.' (Ricchi 136)

As already shown in §3.4.1, property concepts can be expressed, in addition to an adjective accompanied by a copula, also by property concept verbs which inflect like other verbs. The availability of these two options places Denjongke among languages which use "mixed" encoding type for property concepts (Stassen 2013a). For comparative constructions, see §5.6.1.3.2 below.

5.4.5 Clauses of becoming and happening

²²³ This is an experimental phonetic spelling of the word. The form usually seen in writing is $a_i = a_i = a_i$, pronounced *jàmts^henţa*?

is both in my spoken and written data much more frequent than the latter one, which may be a loan word from Tibetan literature. The verbs of becoming may occur with a single argument with the meaning 'happen', 'arise' or even 'begin (to exist)':

(5.80) a) गुरु वेंतु र्चे क्षु र्?

k'ar t^h*om*-bo bε?? what happen-PST EQU.NE 'What happened?' (TB bull story)

b) & אי שיפָריזרי

δ ma-tc 'uŋ-ruŋ that NEG-become-CONC 'although that did not happen...'(BLA 6)

c) बेम्रबार्ये पहुबान्चून जुनारुन

sém=lo dzu:na: tc'uŋ-ruŋ mind=DAT sensation.of.fear arise-CONC 'despite sensations of fear arising in the mind...' (Rna-gsung 3)

d) ך־פִּי אֲִׂק־ײַזְל יַחָר־שָּׁר־ שָׁק־דִק־ שָׁק *t'ato lóbdø: k'uŋsiŋ=jã: tʰon-rap jò?.* now school.GEN holiday=too become-IMF EX.PER 'Now also the school's holiday is about to begin.' (Richhi 62)

Predicate nominals can be nouns (5.81) or nominalized clauses (5.82). The noun phrases linked with $t^{h} on$ in (5.81) and (5.82) are marked with square brackets.

(5.81)

(5.81)

أربانيا المجافظ المجاف المجافظ الم

(5.82) a) สังสา สาวสา สิวารัา สัสารัา ลูกุ

[$m\dot{u}$:] [t^hap $m\dot{e}$:-po] $t^h\theta m$ -bo $b\varepsilon$?. 3SGF.AGT means NEG.EX-2INF become-2INF EQU.NE 'She became one without means (to do something).' (Rna-gsung 6)

b) קאי אַר קדאי אָר אַשאַר אָשאָר אַ אָאָר אַדאָ אָד אַדאָ אָד אַדאָר אַדאי אַדא אַדי אָדא ג' אַדאיז אַדאיז אַדאין אַדאי ג' אַדאיז אַדאין אַדאי ג' אַדאיז אַדאין אַדאָן אַראָאָא אַראָאַן אַדאין אַדאָן אַדאין אַדאָן אַדאין אַדאָן אַדאָן אַראָן אַראָדאָן אַדאַראַן אַדאָן אַדאַן אַדאָן אַדאָן אַדאָן אַדאָען אַדאָן אַדאַן אַדאַן אַדאָען אַדאָן אַדאַן אַדאַן אַדאַן אַדאָן אַראַן אַראַן אַראַן אַראַדעע געראַאַראַראָאָאָאן אַדאאַראַין אַדאָאַראַין אַדאָן אַדאָן אַדאָען אַראַאַן אַדאַן אַדאַן אַדאַן אַדאָען אַדאַןע For $t^h ø n$ and $te' u\eta$ with predicate adjectives, consider the following examples (note that [5.85] has a phrasal adjective with a nominal element):

- (5.83) مَنْ تَقْتَابِحْ مَعْلَمَ عَلَمْ عَلَمْ عَلَمْ عَلَمْ عَلَمْ عَلَمْ مَعْلَمْ عَلَمْ مَعْلَمْ عَلَمْ مَعْلَمُ عَلَمْ الْعَلَى الْمُعْلَمُ الْعَلَى الْمُعْلَمُ الْعَلَى عَلَى الْعَلَى الْعَلَى الْحَلَى الْعَلَى الْعَلَ الْعَلَى الْعَلَى الْعَلَى الْعَلَى الْعَلَى الْعَلَى الْعَلَى عَلَى الْعَلَى الْعَ الْعَلَى الْعَلَى
- (5.84) جَ نَعْرَمَجُ طَعَمِعَ تَنْ طَعَ عَجَرَهُمَا *te ódi lèpo lèm te uŋ-ze.* so that very good become-PST 'So that turned out very good.' (RD BLA)
- (5.85) দান্ডবা ঝিললা দ্বানাদ্র বানাদ্র বানাদের বাদের বানাদের বাদের বাদের

With an ablative adverbial, the verb $t^h \emptyset n$ can also have the more concrete meaning 'exit, go/come out':

(5.86) a) सु. ट्रॅ'या द्विंगलया वेंब्र'यते क्यूटा

 $k^{h}u$ *t*'o:pa $k^{h}im = l\varepsilon$ *t*^hom-bo: $g\tilde{a}$: 3SGM morning house=ABL EXIT-2INF.GEN time 'When he left home in the morning...' (RS driver joke)

b) $\vec{\beta}$ ה' \vec{w}_{a} (school) אומי ההיומדי \vec{a} \vec{a} , \vec{a} , \vec{a} , \vec{a} , \vec{a} $t\varepsilon \ \eta \dot{a} \ iskul = l\varepsilon \ payk^ha \ t^h \varepsilon m$ -bo $b\varepsilon 2$ t'a. so 1SG school(Eng.)=ABL outside exit-2INF EQU.NE now 'So at that point I left school (permanently).' (PED life story)

5.5 Valency modification

This section describes how valency, i.e. the number of verbal arguments, may be modified in Denjongke. The lexicalized phonological distinction between controllable and non-controllable verbs was already introduced in §3.3.3. The two strategies for valency change are argument suppression²²⁴, which decreases the valency of the clause, and causative construction, which increases clausal valency.

5.5.1 Argument suppression

Denjongke lacks a morphosyntactic passive construction marked on the verb which would elide an A argument and raise P in its place. Instead, Denjongke can form a "functional passive" (Givon 1984: 164) by eliding argument A. In (5.87), argument A is suppressed and a peripheral argument 'from hospital' is fronted to the initial position typical of the A argument. In this context, it is not

²²⁴ Argument suppression is to be distinguished from zero anaphora, see the introduction to this chapter.

important, who within the hospital is the actual giver of the leave of absence. Because the actual giver of the permission cannot be deduced from the previous context and therefore does not fall under zero anaphora, (5.87) exemplifies argument suppression.

(5.87) $\arg (act a d t) \stackrel{\text{def}}{\text{def}} = le \quad n \hat{e}: po = lo \quad gompo \quad t \tilde{a}: -do.$ hospital=ABL patient=DAT leave.of.absence send-IPFV 'The patient is being discharged from the hospital.' (Richhi 172)

In (5.88), the main clause following the concessive clause consists simply of the verb. Here the missing A argument is unidentifiable and therefore (5.88) exemplifies a functional passive construction.

(5.88) $\operatorname{Aur}(\operatorname{Aur})$ $\operatorname{Aur}(\operatorname{Aur}$

In example (5.89), the A argument of the verb $t^{h} \tilde{o}$: 'see' is suppressed and the P argument of the equivalent transitive clause occurs zero marked (in a monotransitive clause a dative-locative would likely occur with this type of a referent, see 5.3.2), suggesting that the verb has truly become intransitive in this clause (hence the intransitive translation 'be visible').²²⁵

(5.89) ສເອລາ ອີຖ ອັງ ເຊິ່ງ ເຊິ່

5.5.2 Causative

The valency of an intransitive verb may be increased through a causative construction formed with the help of the secondary verb *tcu?* 'cause, put into'.

- (5.90) ইর্বামার্শ্ব স্থ্রব্যামার্শ্ব হ্রব্যামার্শ্ব হ্রব্যা pempa gjuk-o be?. PN run-2INF EQU.NE 'Pempa ran.' (KN e)

²²⁵ KN, however, commented that the dative-locative =lo could be added to the P argument here.

In elicitation, it was possible to increase the valency of a monotransitive verb. Example (5.92) illustrates a monotransitive clause, which is in (5.93) and (5.94) changed into a ditransitive one through a causative construction.

- (5.93) $\eta_{\mathfrak{A}'}(\eta_{\mathfrak{A}'}) \mathfrak{g}_{\mathfrak{A}''\mathfrak{A}\mathfrak{A}\mathfrak{A}\mathfrak{A}\mathfrak{A}}$ $\mathfrak{A}' \mathfrak{A}'' \mathfrak{A}_{\mathfrak{A}'} \mathfrak{A}'} \mathfrak{A}_{\mathfrak{A}'} \mathfrak{A}_{\mathfrak{A}'} \mathfrak{A}_{\mathfrak{A}'} \mathfrak{A}'} \mathfrak{A}_{\mathfrak{A}'} \mathfrak{A}'} \mathfrak{A}_{\mathfrak{A}'} \mathfrak{A}' \mathfrak{A}'} \mathfrak{A}_{\mathfrak{A}'} \mathfrak{A}' \mathfrak{A}'} \mathfrak{A}' \mathfrak{A}' \mathfrak{A}'} \mathfrak{A}' \mathfrak{A}' \mathfrak{A}' \mathfrak{A}' \mathfrak{A}'} \mathfrak{A}' \mathfrak{A}' \mathfrak{A}'} \mathfrak{A}' \mathfrak{A}' \mathfrak{A}'} \mathfrak{A}' \mathfrak{A}' \mathfrak{A}'} \mathfrak{A}'$
- (5.94) गाम्ग (गीवा) कुवा सम्बद श्वेव पार्वा नेपा पहुवा गी श्वन्। karma(=gi) gjaltshen pempa=lo tip teuk-o be?. PN(=AGT) PN (=DAT) PN=DAT hit cause-2INF EQU.NE 'Karma made Pema hit Gyaltshen.' (KN e)

The A argument of the original bivalent clause of (5.92) is in the causative clause (5.93) marked as dative-locative, whereas the optionally dative-locative-marked P argument of (5.92) occurs in (5.93) as obligatorily zero marked. Example (5.94) shows that word order does not play a role in deciding who hit whom; the argument marked with dative-locative is the actor. Despite (5.93) and (5.94) were accepted in elicitation, I suspect that in actual language use such formulations are almost non-existent. Clauses such as (5.95), which elide the original P argument, are probably more common.

The causative secondary verb tcu? is also used in optative constructions, see §11.5.

5.6 Adverbial modification

Adverbials in a clause may be divided into non-clausal adverbials and clausal adverbials (adverbials that are dependent clauses in a sentence). Whereas clausal adverbials are discussed elsewhere (see §15), this section describes non-clausal adverbials, i.e. case-marked nouns/noun phrases, postposition phrases and adverbs.

5.6.1 Case-marked nouns as adverbials

The grammatical cases agentive and genitive do not participate in forming non-clausal adverbials²²⁶ but spatial cases dative-locative²²⁷ (\$5.6.1.1), locative (\$5.6.1.2) and ablative (\$5.6.1.3) do.

5.6.1.1 Dative-locative =*lo*

In addition to the patient, recipient and possessor functions, which were decribed above, the dativelocative also marks adverbials of location and time. For stative, adessive type of location, a casemarker is typically used (5.96), but especially with frequent toponyms case-marking may be dropped (5.97). In the examples, discussed items are given in bold, and adverbials longer than one word are given in square brackets.

(5.96) جَرَةَ بِيَحَ^{رَ}مَ المَّامَ مِوْمَامِينَةُ? *dorjiliy=lo k'ana zu:-to?* TPN=DAT where live-IPFV 'Where do you live in Darjeeling?' (Richhi 13)

(5.97) **षु**' क्षून'र्नेग' थेंन्।

 $k^{h}u$ **gã:to?** $j \partial$?. 3SGM TPN EX.PER 'He is in Gangtok.' (KN e)

Allative type of goal-oriented directional meanings can be expressed with =lo (5.98), but non-cased-marked directionals seem to be more frequent with toponyms (5.99).

(5.98) a) क्रें प्रहें पान व्याया र्वा कें कें कें कें

to $ts^{h} \emptyset$:-wa **nà:tsẽ:=lo** $s \delta$:-zɛ. food²²⁸ search-PUR forest=DAT go.PFV-PST '(He) went to forest to look for food.' (KT animal story)

b) ฉัลเว้าผ์ ลิจง ลิเซ็ญลาผ์

bombai=lo $l \in p$ mi- $ts^h u$:=lo. TPN=DAT arrive NEG-be.able.to=REP 'He cannot arrive in Bombai, I hear.' (Richhi 138)

(5.99) a) هَٰ مَحْرَّ عَلَى اللَّهُ اللَّهُ عَلَى اللَّهُ (5.99) a) هُ عَلَى اللَّهُ عَلَى اللَّهُ عَلَى اللَّهُ مُنْ اللَّهُ عَلَى اللَّهُ a) and a structure of the story of the story of the story of the story and the story of the story of the story of the story and the story of th

²²⁶ An exception to this rule is the adverb(ial) $t^{halamgi}$ (B) "NATE of the clearly, which seems to have an agentive/genitive ending.

²²⁷ The dative-locative case is both a grammatical and a locative case, see §3.7.1.1.

²²⁸ Here the word *to* 'cooked rice' obtains an extended meaning 'food'. The clause refers to a tiger, a carnivorous predator.

It is my impression that case-marking in directionals is more frequently dropped in spoken language (5.100) than in written language (5.101).

(5.100) ך שִאי אָד תַשָּיאָד קַשָּיאָבאיקָד t'a y:kor gju-somda now tour go-SIM 'Now, when going on a tour...' (DB day trip)

(5.101) यार्थे सुया क्रेंग त्यु अपवा

làlo y:kor=lo gju- $k^{h}\tilde{\epsilon}$: some tour=DAT go-NMLZ 'some (of them are) tour-goers' (Richhi 120)

Example (5.102) illustrates the use of =lo for expressing time. The dative-locative is used exclusively with $g\tilde{a}$: 'time' (5.102a, b). The locative =na is more common with $t'yts^h @?$ 'time', but =lo is also used (5.102c). Note that in (5.102c) the noun has a preceding genitive-marked complement clause.

- b) $\hat{a}_{i} [n: q \approx i \hat{j} \approx \omega_{n} \cdot \hat{g} \approx \xi \hat{a}_{i} \cdot \xi \approx \tilde{a}_{n} \cdot \tilde{a$

The dative-locative also occurs as an optional element in temporal adverbs referring to times of the day, such as t'o:pa(lo) 'in the morning', t'arin(lo) 'today' and $p^{hi:tsham}(lo)$ (see §3.5.1).

The dative-locative functions as an additional (non-obligatory) locative-marker in some relator noun constructions:

(5.103) ८ सु. रु.चुं. गुरुषा क्रेटाय में रुर्ड्र क्रेंगी.

t'a k^hu [*rubi*?=*tci*? *tɛŋk^ha=lo*] *pjã:-tiki* now 3SGM climber=INDF on=DAT hang-NF 'Now, hanging [on a climber plant]...' (KT animal story)

 $tc^h \phi g \epsilon l \quad tc^h impu = lo$ nàtci míwã: ge:pu=gi [tende: $p' \varepsilon y$ ódi 1PL sovereign king great.one=DAT Nepal king=GEN ceremony that nàŋca=lo] dencu nấ:-m bɛ?. inside=DAT invitation do.HON-2INF EOU.NE "...our ruler, His Sovereign Majesty the King, was invited [in that ceremony] by the king of Nepal.' (CY interview)

Furthermore, =lo, along with the ablative =l ε (e.g. gjabl ε 'after'), occurs in relator nouns such as gjablo 'after', dz ε :lo 'after', dynlo 'before', t ε nlo 'above, on', etc. (see §3.6.8). The dative-locative may also attach to circumstantial clause marking progressive -t $\varepsilon \tilde{\varepsilon}$:/ $z\tilde{\varepsilon}$:/zin (§15.8.3) and circumstantial-purposive -pa/ba (§15.5.1).

The dative-locative-marker also occurs in case-stacking constructions (see §3.7.1.3) following the locative case marker =na, see (5.105). In spoken data, the construction occurs especially with the frequently used word k^{him} 'house', see (5.105a)

(5.105) a) ८. भुषा भुषा के हेंगा द्विया व र्या पर्छया हें र हेंश sý:sy: minto? khim=na=lo ŋà tsuk-to t'ato. 1sg a.bit flower house=LOC=DAT plant-IPFV now 'I'm planting a bit flowers at home now.' (PED life story) b) દ્વિશ્વ સ્વાયત્ર છે. જે. અને આ છે તો ત્યાં આ પ્રાયત્ય છે. આ પ્રાયત્ય છે છે. છે *k*^{*h*}*im*=*na*=*lo*=*jã*: mí=tsu námtei? ódem=sã: k^ha -lap house=LOC=DAT=even human=PL with that.much=until mouth-speak mè?. NEG.EX.PER 'Even at home, there is not that much talking with people.' (Richhi 164-165)

Other examples of =na=lo in Richhi are given in (5.106). Note that in (5.106) =na=lo attaches to a genitive marked noun, presenting an instance of three stacked cases (for more on case stacking, see §3.7.1.3).

(5.106) a) हेनु हेंदे नेपानु कें

 $[nint^h o = i$ $t' \varepsilon p = na = lo$] day-list.GEN book=LOC=DAT '[in the calender]' (Richhi 7)

- b) ক্ল্র্বান্থ্বি'ক্'র্ন্ *lóbdø:=na=lo* school.GEN=LOC=DAT 'inside the school' (Richhi 31)
- c) आजादी बोबाबा ने र्येग

[*ámø:* sém=**na=lo**] mother.GEN mind=LOC=DAT '[in the mother's mind]' (Richhi 83)

5.6.1.2 Locative =na

The locative =na typically marks spatial (5.107-108) or temporal adverbials (5.109). Analogously to the dative-locative, the semantics of =na cover both inessive type of stative locatives (5.107) and illative type of goal-oriented directionals (5.108).

t'ato bhaila mɛ́ŋkʰã:=na jò̃:-ka? now PN hospital=LOC EX.PER-PQ 'Is Bhaila now in the hospital?' (Richhi 24)

- b) أَكَمَّ الحَرَّ حَوْا مَحَرَّ حَوْا مَعَامَ مِنْ عَلَى اللَّهُ اللَّهُ عَلَى الْحَالَ عَلَى اللَّهُ عَلَى الْحَالِ اللَّهُ عَلَى الْحَالَ عَلَى الْحَالِ اللَّهُ عَلَى الْحَالِ اللَّهُ عَلَى الْحَالِ عَلَى الْحَالَ عَلَى الْحَالَ عَلَى الْحَالَ عَلَى الْحَالِ عَلَى الْحَالِ عَلَى الْحَالِ عَلَى الْحَالِ عَلَى الْحَالَ عَلَى الْحَالَ عَلَى الْحَالِ عَلَى الْحَالِ عَلَى الْحَالِ عَلَى الْحَالِ عَلَى الْحَالِ عَلَى الْحَامَانِ عَلَى الْحَالَ عَلَى الْحَالِ عَلَى الْحَالِ عَلَى الْحَالِ عَلَى الْحَالَ عَلَى الْحَالِ عَلَى الْحَالِ عَلَى الْحَالِ عَلَى اللَّهُ عَلَى اللَّهُ عَلَى الْحَالَ عَلَى الْحَالَ عَلَى الْحَالَ عَلَى الْحَلَى عَلَى اللَّهُ عَلَى اللَّهُ عَلَى اللَّهُ عَلَى الْحَالِ عَلَى الْحَالَ عَلَى الْحَالِ عَلَى الْحَالِ عَلَى الْحَالِ عَلَى الْحَالِ عَلَى الْحَالَ عَلَى الْحَالَ عَلَى الْحَالِ عَلَى الْحَالَ عَلَى الْحَالِ عَلَى الْحَالَ عَلَى الْحَالِ عَلَى الْحَالِ عَلَى الْحَالَ عَلَى الْحَالِ عَلَى الْحَالَ عَلَى الْحَالَ عَلَى الْحَالِ عَلَى الْحَالَ عَلَى الْحَالِ عَلَى الْحَالَ عَلَى الْحَالِ عَلَى الْحَالِ ع المَالِحَالِي عَلَى الْحَالَ عَلَى الْحَالَ عَلَى الْحَالَ عَلَى الْحَالَ عَلَى الْحَالَ عَلَى الْحَالَ عَلَى عَلَى الْحَالَ عَلَى الْحَالَ عَلَى الْحَالَ عَلَى الْحَالَ عَلَى عَلَى عَلَى الْحَالَ عَلَى عَلَى الْحَالَ عَلَى عَلَى عَلَ
- (5.108) ८ में आवा प्युर्भे कीवा

nà t^hom=na gju-do í:.
1SG town=LOC go-IPFV EQU.PER
'I'm going to town/market.' (KN e)

(5.109) a) ନ୍'୍ଲେକ୍' ଝ୍ଟାସନ୍' କ୍ଷୁସଷ'ଶ' । ସଂସୟ' କ୍ରୁସଷ'ନ୍' ଅଟ୍' ସଂସ୍'୍ରି ମୁସ୍' ସ୍'୍ରା ସ୍'୍ରା ସ୍'୍ର୍ୟୁସ୍' ଅନ୍'

t'izã: [ódi kap=na] k^ha=lε kjap-εε? mèm-bo t'i-εε=ki lùksø:
but that time=LOC mouth=ABL do-INF except write-INF=GEN tradition mèbbε?.
NEG.EX.NE
'But [at that time], in addition to spoken language, there was no tradition of writing.' (KL BLA 12)

b) אֹימה הַקַאישה הַשִּי
 b) אֹימה הַשָּי
 b) אֹימה הַשָּי
 c) למו ל'yts^ho=na
 c) that time=LOC
 f(at that time)' (PED life story)

5.6.1.3 Ablative =*lε*

The ablative $=l\varepsilon$ marks spatio-temporal adverbials (§5.6.1.3.1) and also the standard of comparison in comparative constructions (§5.6.1.3.2). Because the standard of comparison in comparative

constructions is, similarly to locative adverbials, a peripheral NP, comparison is here treated under adverbial modification.

5.6.1.3.1 Spatio-temporal uses

The ablative $=l\varepsilon$ expresses a locative or temporal starting point. With locatives, $=l\varepsilon$ typically expresses direction from a source (5.110), but especially with relator nouns, it is also used for expressing stative location (5.111). The ablative with relator nouns is not separated from the root with the clitic marker = (i.e. *gjable* instead of *gjab*=*l* ε), because the case ending has essentially merged into one word with the relator noun.

(5.110) सु' र्ट्र'य' द्विमालमा वेंत्र'यते' क्षट'

 $k^{h}u$ t'o:pa $k^{h}im=l\varepsilon$ $t^{h}om-bo$: $g\tilde{a}$: 3SGM morning house=ABL exit-2INF.GEN time 'In the morning when he was getting out [from the house...]' (RS driver joke)

(5.111) हिं कुन लाग में हैं गठिया झ झेंन लत्या

[*t'i* **gjable**] *p'otso=tei?* ba dø: du?. chair behind child=INDF hide stay EX.SEN 'A boy is hiding [behind the chair].' (PD spatial topography interview)

Examples in (5.112) illustrate the use of $=l\varepsilon$ as a temporal starting point.

- (5.112) a) هَرَ (مَحَ) عَلَيْهُ اللَّهُ اللَّهُ عَلَيْهُ اللَّهُ الللَّهُ اللَّهُ اللَّالَةُ اللَّهُ اللَّ
 - b) קיאקרימעי

t'ariŋ=lɛ today=ABL 'from today' (SGD wedding customs)

Similarly to the dative-locative =lo, $=l\varepsilon$ can attach to the locative case marker =na, see (5.113). The meaning is comparable to a relator noun construction such as the one in (5.114).

 $[t^{hinl\varepsilon:=ki} \quad k^{him=na=l\varepsilon}] \quad t'ato=s\tilde{a}: \quad ka-j\tilde{a}: \quad ma-l\varepsilon p-\varepsilon \varepsilon^{2}=di:$ Thrinley=GEN house=LOC=ABL now=until who-even NEG-arrive-INF=DEMPH.AGT 'Because no one has so far arrived [from Thrinley's house]...' (Richhi 43)

(5.114) هم هن مجرع المعن من من المجاب بي تكر المحص الحي المحركة المحتركة المحتركة

5.6.1.3.2 Comparative uses

In comparative constructions, the standard of comparison is marked by the ablative case $(=l\varepsilon)$, while the quality compared is expressed by an adjective (5.115), a stative verb (5.116) or an adverb (5.117). In Stassen's (2013b) classification, this type of comparative construction is termed "locational" and, further, "from-comparative." Denjongke has no separate comparative adjectival form. The standard of comparison may be presented, depending on topicalization, either before the comparee (5.115) or after the comparee (5.116). The quantifier *lako* 'more (than), (in) excess' may be used in addition to the ablative, see (5.117).

(5.115) ฉริ เจเลา ซัเลริ ซัลเร เลรา

 $di=l\epsilon$ ódi bompu du?. this=ABL that big EX.SEN 'That is bigger than this.' (TB e)

(5.116) $a\hat{\tau}$ שַאַין אָן מָלָ׳ מָלָ׳ מָלָ׳ מָלָ׳ אָרָי אָרָי אָרָן $di \quad puku=di \qquad di=l\epsilon \quad rin \qquad b\epsilon?.$ this pencil=DEMPH this=ABL be.long EQU.NE 'This pencil is longer than this.' (TB e)

(5.117) वहे ज्या क्षा में कु के द्वाया र्ड्या क्षा र्यु के द्वाये हा के र्या के र्यु क

 $di=l\varepsilon$ **lako** gjats^hita?-p'ja súŋ t^hop-o-dã: this=ABL more great-ADVZR observe receive-2INF-CONJ 'when we get to hold (the celebration) in a more grandiose way than this...' (Richhi 87)

In addition to being a noun, the standard of comparison may be a a subordinate clause, see §15.11.

5.6.2 Postposition phrases as adverbials

Postpositions are listed in Table 5.3. The first four rows list items which do not typically occur with a genetive complement (i.e. nantar, na

Postposition		Gloss		
nàŋtar, nàŋzin	वट्रा झर, वट्र पविवा	'according to, similar to'		
t'ønzin(gi)	र्ने न न न न न न न न न न न न न न न न न न	'according to,		
		in accordance with, in view of'		
(t'ãː) námtei?,	(८८) अनुवा छेगा,	'with'		
(ť ʿãː) námpu	(૬૬.) અનુઅ:સુ.			
$s\tilde{a}$:te, =s \tilde{a} :	<u> </u>	'until'		
mèmbo ²²⁹	ਕਰ,'ਧੋਂ'	'except'		
mèntã:	अव्र'नन्द'	'except'		
ts ^h aplo, ts ^h ama(lo)	ર્ळવર્ગ્સે, ર્ઠવર્જ્સર્યે	'instead of'		
$tsak^ha, tsa:lo, =tsa:$	₹'ष(म)', इ'र्ये', इ(म)'	'at, by, with'		
t'ønlo, t'ønlɛ,	र्नेवार्थें', नेवालक्ष',	'for (the purpose of)'		
t'øndalo, t'øndale	<u> </u>			
t ^h o:lɛ	র্হুবা.লপ্র.	'through, via, on the basis of, by'		
$(=l\varepsilon)$ gjy:ti	(બાજા.) ર્શુટ્.ક્ષે.	'through, via, by'		
go:le	নর্যা এম	'from, through'		
korlo, korlɛ	र्झेन्यें, झून्य्य	'about'		
yo:lo/hõ:lo, yo:tɛ/hõ:tɛ	ર્દેજાયોં, દેંજાસ્ટ્રે	'toward, in the direction of'		
ทู้oːlɛ/hõːlɛ	Karaar	'from the direction/side of, through'		
gjablo, gjablɛ	ক্রুন'র্ন্ম', ক্রুন'এম'	'behind, after'		
$dz \varepsilon: lo^{230}$	हेरा'र्ये'	'after'		
(kum)dynk ^h a, (kum)dỹ:lo,	(झु')अनुव'ाय(म)', (झु')अनुव'र्थे',	'in front of, in the presence of (kum- is		
kumdỹ:	भु'यट्व'	honorific prefix)'		
p'aːna	মন'ৰ'	'between'		
pénlo, pénle, pǿma,	ই্ইব'র্নি', ই্ইব'এম', ই্ইব'ঝ',	'before'		
hɛnlɛ, hɛma	দ্বে এম্বা, দ্ব ঝা			
tɛŋkʰa, tɛŋlo	छेमाम(म)', छेमार्थे'	'above, on'		
=gu, gulo	सगु (તેં)	'above, on'		
ò:lo, ò:lε	र्तेया में, तेया भषा	'below'		
buːna, buːlo	क्षुंत, क्षुंत्रं	'in the middle of'		
bolo(-k ^h a), bololo	त्र्यें र्भें (षर्), त्र्यें र्भें भें	'next to'		
nàŋɛa(lo), nàŋlo, nàŋna	व८ म. (यू.), व८ यू.	'inside'		
	वृत्त'व			
p ^h ilo?	र्द्धु र्त्यग	'outside'		
paŋk ^h a	ちち、(な(ナ)、	'outside'		

Table 5.3. Postpositions

The following examples illustrate the uses of postposition headed phrases as adverbials in the same order as they occur in the table. The example sentences illustrate simple uses with noun phrases. Examples (a) below have the complement noun in citation form, whereas examples (b) have a genitive-marked noun complement, if such forms are used. Some examples have (c), which

²²⁹ Occurs in a negated or interrogated clause.

²³⁰ This word from Tibetan is mainly used by literate people with monastic training.

illustrates an alternative form of the postposition. Most postpositions which are relator nouns are also used as independent adverb(ial)s, see the next section §5.6.3.

- - b) هَٰ مَرْجَرَ مَا مِحْرَجَ حَرَّحَمَّ مَالَمَمَا عَلَيْمَ مَحْرَمَ مَعْنَ اللَّهِ اللَّهِ عَلَيْمَ عَلَيْمَ عَلَيْمَ مَعْنَ مَعْنَ مَعْنَ اللَّهُ اللَّهُ وَلَقَعْ مَعْنَ مَعْنَ مَعْنَ اللَّهُ مَا يَعْنَ مَعْنَ مَعْنَ مَعْنَ اللَّهُ مَعْنَ اللَّهُ مَعْنَ اللَّهُ مَعْنَ مَعْنَ مَعْنَ اللَّهُ مُعْنَ اللَّهُ مَعْنَ اللَّهُ مَعْنَ مَعْنَ مَعْنَ مَعْنَ مَعْنَ اللَّهُ مُعْنَ مَعْنَ مَعْنَ مَعْنَ مَعْنَ مَعْنَ مَعْنَ مَعْنَ مَعْنَ مَعْنَ اللَّهُ مَعْنَ الْمُعْنَ الْعُنْقُ مُعْنَ مَعْنَ مَعْنَ مَعْنَ الْعُنْ مُعْنَ مَعْنَ اللَّهُ مَعْنَ اللَّهُ مَعْنَ الْعُنْنُ مَعْنَ مَعْنَ مَعْنَ مُعْنَ مُعْنَ الْمُعْنَ مُعْنَ مُعْنَ مُعْنَ مَعْنَ مَعْنَ مَعْنَ مُعْنَ مُعْنَا الْعُنْ مُنْ مُعْنَا مُعْنَ مُعْنَا مُعْنَا مُنْ مُعْنُ مُعْنَ مُعْنَا مُعْنَا مُعْنَا مُعْنَا مُنْكَنَا مُعْنَا مُعْنَا مُعْنَا مُعْنَا مُعْنَا مُعْنَا مُعْنَا مُعْنَا مُعْنَ مُعْنَا مُعْنَ مُعْنَا مُعْنَ مُ
 - b) רְיִשִּיִשְּרִיחִן בְּרִי אַשְׁיָשָּרִיחָן בְרָי אַשָּׁיָשָּׁי אָז בְּרָי אַשּׁיָשָּׁי אָרִיבוּשָּׁשָ אַיִרְשָּׁשִי בְּרָי אַשָּׁי אָרִיבוּשָּׁי בָּרָי אַשּׁי אָרָיבוּשָּׁי בָּרָי אַשּׁי אָרָיבוּשָּׁי בָּרָי אַישּישָ ג' געשַ בָּרָי אַ געשַ בּיּשָּׁי בָרָי אַשּׁישָ בַּרָע בַּרָאַ געשַ בּיּשָּׁי בָרָי אַשּׁישָּׁי בָרָי אַשּׁישָּׁי בָרָי אַ געשַ בּיּשָּׁי בָרָי ג'שּׁישָּׁי בָרָי ג'שּׁישָּׁי בָרָג' געשַ בּיּשַיּ ג' געשי בענייש ביע בּרָאַריישָּי בָרָג' געשַ ביע בערייש ביע בערייש ביע בערייש ביע געשיי בערישָּיי בערייש איז געשיי בעריש ביעריש ביעריש ביעריש ביעריש ביעריש ביעריש ביעריש ביעריש ג' ג' געשיי ביעריש ג' ג' ג' געשיי ביעריש ג' ג' ג' ג' ג' ג' ג' געשיי ביעריש ביערי
- (5.120) a) र्रेगुषार्द्धं व्यनुव्रासुः लेगवार्य्युषाः क्वेनुः क्षेतुः त्ववा

[*ro:=tsu* **pàmpu**] *lèm-p'ja dø: ín-(n)am.* friend=PL with good-ADVZR stay EQU.PER-ATTQ 'Live well [with (your) friends], eh.' (TB phone call)

b) \check{n} \check{n}

(5.121) a) ད་མོ་ མོ་སོ་ མོགམ་ར་ མོད། [t'ato sã:tɛ=to] lɛ̀m=ra jà?. now until=CEMPH good=AEMPH EX.PER '[Until now], (I) am well.' (PED life story)
b) མུན་ དྲག་གོ་མོ་ སོྲོཔ་གྲ་ ཆོགས། [tʰyn t'uko=sã:] lóbda tsʰo?. session sixth=until school gather

The postposition $m \dot{e}mbo$ 'except' (nominalized form a negated existential copula) occurs either in a negated (5.122a) or an interrogative clause (5.122b).

'The school is on [until the sixth session].' (Richhi 10)

nấ:-bo náŋ-k^hẽ: nà: [lópɛn=la:=gi mèmbo] zømø dem like.that do.HON-2INF do.HON-NMLZ here teacher=HON=AGT except other $ma-t^h \tilde{o}$.²³¹ ke:po ηά: ηά: NEG-see I.AGT many I.AGT 'I have not seen many, [except the teacher] here (=you), doing like that.' (KL discussion with DR) b) आर्याते कॅंपायें कॅंपार्या अवर्यं गाववा गा घेंप? *ápø:* ts^haplo [$te^h\phi$:= $r\tilde{a}$: **mèmbo**] zen ka $j\dot{\phi}$? father.GEN in.place 2SG.L=REFL except other who EX.PER 'Who is there in place of (your) father [except yourself]?'(Richhi 84) (5.123) a) เหาะเลิ เฮาาลัา ฮัการกา ลเสาร์า ลเลสา กา นัก? ts^haplo] $t \varepsilon^{h} \phi := r \tilde{a}$: [ápø: mèmbo zen ka j*à?*? father.GEN in.place 2SG.L=REFL except other who EX.PER 'Who is there [in place of (your) father] except yourself?' (Richhi 84) b) जगा नेषा ग्रीषा कंपा आया प्राप्त गांधेया मुप्तषाया पा केंप्रा केंद्र [taci=ki tshamalo] t'arin jó? kjap-a nà òm-bo ĩ. work do-PUR 1SG come-2INF EQU.PER PN=GEN in.place today 'I came today to work [in place of Tashi].' (KN e) The only example of $m \epsilon n t \tilde{a}$: 'except' (literally 'NEG-send') occurs in an interrogative: (5.124) रूट वर्ष यहट गार्वे कें र्ह गीय यह में रा रे ?

[$r\tilde{a}$: $m \tilde{e}nt\tilde{a}$:] $z\phi m\phi = tsu = gi = j\tilde{a}$: $t^{h}op - o$ $p\dot{a}$? you except other=PL=AGT=too receive-2INF EQU.PER.Q '[Except you], did others also receive (it)?' (KN e)

²³¹ The repetition of $\eta \dot{a}$:[I.AGT] here is unnecessary from the perspective of written and polished spoken language.

(5.125) a) यामया गाया तेवा (captain) रायाया गावमाया गाविया घेंया गाव? [jàːp kæpten tsa:=lɛ] nấ:wa=tci? t^hop-kam? nobleman.HON Captain(Eng.) at=ABL permission=INDF receive-ATTQ '(I wonder) whether (we) will get a permission [from Captain Sir].' (KN, CY interview) b) ८. रुपायी. इ. यात्रा पट्रे. या. हे. क्रेंग्रा जेट. रा. होट. होव. या. यात्रा t^hets^hom mè:-po be? $[\eta \hat{a}t \varepsilon a = ki \quad tsa = l\varepsilon] \quad di = lo$ by=ABL this=DAT doubt EX-2INF EQU.NE 1PL=GEN íŋ-ga=la. EQU.PER-PQ=HON '[From our side], there is no doubt about that, is there?' (KLT Bumchu video) c) गाहा, केंद्र' व' लेंग्ले' यर इर बद क्रेय देव यें। karma, tc^hø? nà: ópi=lo $[j\dot{a}] = tsa] = s\tilde{a}$ ket p'in lo. 2sg.l here small.child=DAT up=by=until bring give TAG.Q PN 'Karma, you take the child here [all the way up], okay?' (Richhi 40) (5.126) हे' मरकते' मॅबर्गा विषा मम का रही में के प्राय के कि कि कि कि कि कि te [nàtei t'øndale] $d\tilde{a}$: $p^{h}am = tsy$: t'øndale] lóp<u>t</u>a=ra so 1PL.GEN for and parents=PL.GEN for school=AEMPH mè:-k^hɛn *bε*?. EX-NMLZ EQU.NE 'So, [for us and for (our) parents] there wasn't a school.' (CY interview) (5.127) a) क्रॅंग्वर्ने न्दीगी नहान बिन नन न्दीगी न क्रिंगे कुराय क्रेंन के क्राय के का के ना के के के के के क tee:zip t'ã: nè:=gi [nè:=gi t'ato=i nàmjõ: t^{ho} : $l\varepsilon$] ódi that 1SG.GEN=GEN research and 1SG.GEN=GEN now=GEN experience through sé-patsene=di ηà 1SG say-COND=DEMPH 'If I tell that [by my research and my present experience]...' (YR boys' and girls' clothing) k^hokø: [jìgiː tho:lo] thalamgi sé:ta? nàŋ=gi tam=tsu p'ja-ti insides.GEN inside=GEN word=PL letter.GEN through clearly clear do-NF jờ?. hako-z \tilde{e} : understand-PROG EX.PER 'Inmost thoughts are [through letter(s)] clearly and unambiguously being understood.' (Richhi 152)

(5.128) a) สังรรรที่ เด็ญาที่ดิงสาวสงทสงาลิ่า จดกา พีรราชดิง ทุกสงสูง พิทาลัญา สูราลิ่ง ทุสงสิ่ง สุลงสัญาสา สูกสง ลักาลกา คริง mù=ran=gi $k^h o k \sigma = n a$ jà-pø∶ sák-ti za: 3SGF=REFL=GEN innards.GEN=LOC accumulate-NF set EX-2INF.GEN *tam=tsu* [jikt^ho? **giv:ti**] karmø: námtco=na lùk writing through PN=GEN ear=LOC word=PL pour $t^{h}op$ -ce=direceive-INF=DEMPH.AGT "...because she has gotten a chance to pour into Karma's ear [through writing] the words that have been accumulated and stored in her own inmost being.' (Richhi 148) b) ਘੇਂਧੀਕੇ ਕਾਸ਼ੁਸ਼ ਕਾਤਸ ਰੱਕਾਰੇ ਕਾਲ ਗੁਰਾ ਵੇ ਕਾਨੀ ਕਾਲ ਤਾ ਕਿ ਕਾਲ ਗੁਰਾ ਵਿੱਚ ਕਿ ਕਾ ਗੁਰਾ ਦੇ ਕੋਣ ਦੀ ਦੇ ਦੀ ਦੇ ਦੇ ਦੇ ਦੇ ਦ $k^hat \in \tilde{a}$: $ta-s \tilde{a}$: [bombai=**l**E gjy:ti] gã:to?, $[g\tilde{a}:to=l\varepsilon$ jìgi: letter.GEN address look-TERM TPN=ABL via PN **TPN=ABL** giv:ti] sòmbare t'i-ti=u be? write-RDP-2INF EQU.NE via TPN 'As (she) looks at the letter's address, (it) is written Gangtok [via Bombay], Sombare [via Gangtok]...' (Richhi 162) (5.129) ह्वेद्र वया राते तया भाषा ह्या रहे के लु. मुद्र के जु. भाषा [*nint^hakpø*: **go:le**] $t^h u dz i t e^h e$ cù-cĩ:=la. bottom.of.heart.GEN through thank.you say.HUM-NPST.PER=HON 'I thank (you) [from the bottom of (my/our) heart].' (CY interview) (5.130) a) קיאי ועדיארי מקאידי פֿמייזי איזי אַגיישאריטן אַאַקישאלי קאיאַאי אַאַדיאי איזאַאי אַריאי איזאַאי איז t'ato $k^h u = r\tilde{a}$: *ní:-po teiku* [*k*^hasaŋ=gi lóbdø: t'y:tøn two-COL only yesterday=GEN school.GEN celebration now 3sgm=aemph korlo] lógju? *cέ-z*ε̃:=lo tell-PROG=DAT about story 'Now only the two of them telling stories [about yesterday's school celebration]' (Richhi 99) b) ราวีรา รา รุฑราสพา อารามเกลาญี่) สมา รุรา รุมราสพา อารามเกลาญี่) สมาญี่) ที่ราอพา ธัญา ญี่อยาญสิหาอยา ลาร์ เพิ่ม ηά [karzε: $lap-k^h \in \eta = gi$ t'ariŋ sàm dã: márze: todav I vegetarian.food say-NMLZ=GEN food and nonvegetarian.food $i^{\prime} \cdot 232$ làp-k^hɛŋ-gi korle] ts^hik tciní:=tci? cù-do sàm=**gi** say-NMLZ=GEN food=GEN about word a.few=INDF say.HON-IPFV EQU.PER

'Today I will say a few words [about the food called karze and about the food called marze].' (DL about food)

²³² According to consultant KUN, the words *karze*: (lit. white-food) and *márze*: (lit. red-food) refer to the colour of milk/eggs-whites and meat/blood respectively.

(5.131) a) क्षु क्लिन नन र्रेगुका रुति केवा क्रन्का तार्वर रेका की *laki? t'ã: ro:=tsu=i* mìk [lankor **no:lo**]. and friend=PL=GEN eye car toward PN 'Lhaki's and friends' eye(s are) [toward (the) car].' (Richhi 68) b) ਤਿਹਾ ਸੱਚਾ ਦੇ ਕਹਾ ਸੀ [k^him ŋoːtɛ] gju-do. toward go-IPFV house '(I'm) walking [towards the house].' (TB e) c) हा सहार्मगा यया केंट्रायदे देवायवा केंटार्च क्षेत्रा ĩ. số:-bo gã:to? [làm tømø: no:le] nà road upper.GEN from.the direction go.IPFV-PST 1SG TPN EQU.PER 'I went to Gangtok [through (=from the direction of) the upper road].' (KN e) Note that g_{jable} 'behind, after' can be used both spatially (5.132) and temporally (5.133). (5.132) a) हिं कुन यथ में रें के परिवा हा हेंन वर्त्तवा $[t^{hi} giable] p'otso=tci?ba$ dø: du?. chair behind child=INDF hide stay EX.SEN 'A boy is hiding [behind the chair].' (PD spatial topography interview) র্দ্রদম্ম দ্ব স্থি দুর্দ্র মাদব। làp-ti làlɛ [ge:go=i]dynlo] te'ømbolekso t'ã: [gjablo] t^hudzite^he gate=GEN in.front.of welcome and behind thank.you say-NF some jì:du thinkha kjap-ti f'i-ti re: ka:pø: pjan-k^h~~?: cloth white.GEN letter azure do-NF write-NF hang-NMLZ 'Some (are those who) write and hang [in front of the gate] "welcome" and [behind (it)] "thank you" in blue letters on white cloth.' (Richhi 71) (5.133) a) हे' लेबा गठेगा कुपायण मा मा के के पार्ट के के के के के लाग क *ถั:-ธĩ*: te [nim tei? gjable] nà сá kimba da:-wa пá. so day one after 1sg meat loan take-PUR come-NPST.PER TAG.ASR 'So [after one day] I will come to take (back) the meat loan, eh.' (KT animal story) b) ନ୍ୟ ସାହିସାଂସ୍ୱା ଶ୍ରୁସଂବ୍ୟୁ ନି' ସାନ୍ତିୟ' ଶ୍ରୁସ୍ୟ'ନ୍ସ୍'ମ୍ବି' ସ'ହିସି'ମ୍ବି' ମୁକ୍ୟାର୍ଚ୍ଛିମ୍' ସାହିସ୍' ଧିମ୍ବା

[lo tci:=ki gjable] te nén kjap-ce=ki nàtci=giyear one=GEN after so marriage do-INF=GEN 1PL.GEN=GENt'yts^hø:=tci? jô?.time=INDF EX.PER

'[One year later] we have then a time for having the wedding.' (SGD wedding customs)

(5.134) केंबर मलिया हेयायेंग तरी स्वायदेंगर आश क्षांबेंग तरी येंगा क्षेत्र द्वारा गांग के कि केंगे $dz \epsilon: lo = di$] [nìm níː ná:dzor áca lamo=di lók-ti p'jaku=na day two after=DEMPH yogini PN PN=DEMPH return-NF cave=LOC $t^h \phi n$ - $z \varepsilon = lo$. appear-PST=REP '[After two days] yogini Asha Lhamo appeared back in the cave, it is said.' (SGD Sikkim caves) (5.135) a) જૉંગ્સર્તર અર્કેંગ વર્તર અતુરાય બાં જૉંગ્સ મારુવેર પાવિત્રથી સુત્ર પાવા વર્ત્તયાય વર્ત્તયાયો એ $ts^{h}o=di$ dynk^ha=lo] óna nàtsi pacin=gi $p^h u \eta$ [ómø: lake=DEMPh in.front.of=DAT there 1PL.GEN bamboo=GEN heap za: milk.GEN du- $k\varepsilon$ =s. set EX.SEN-IN=QUO "[In front of the lake of milk], there was placed a heap of our bamboos (it is said)." (SGD cave story) b) राष्ट्रव रहें वर्गी अन्व में हें र्श्वेयर प्रिया रहेगा du?. [tenzin=gi dyn=lo] do *bompu=tci?* PN=GEN in.front.of stone big=INDF EX.SEN 'There is a big stone [in front of Tenzing].' (KN e) c) विंदा मुनार्ये र्हु भुग्यत्व प्यया दागीया कें प्रदेव की की का t^ho:-po ĩ. [$k^h \tilde{o}$: gempo=tsu kumd \tilde{v} :=le] ηά:=gi ódepti 3PL elder=PL in.front.of=ABL 1SG=AGT like.that hear-2INF EQU.PER '[In the presence of them elders] I heard like that.' (CY interview) d) ८ गु.र. रेव.र्. क. भे. भरे थ. य राष्ट्र, ηà [guru rimputc^he kumdỹ:] tcaː-di 1SG guru precious.one in.front.of.HON come.HUM-NF 'I came [to the presence of Guru Rimpoche] and...' (CY interview) (5.136) a) अत्रानुवा पूरा क्षूरार्नेषा पत्रावा रुवानेषा थेंना t'ã: gã:to? **p'a:na**] rumte? j*à*?. [màrtam between TPN TPN and TPN EX.PER '[Between Martam and Gangtok] there is Rumtek.' (KN e) [átaŋ=gi tc^høki? t'ã: t'atø: tc^høki=ki **p'a:na**] nám t'ã: always=GEN PN and now.GEN PN=GEN between sky and sá=i k^hepar. earth=GEN difference '(There is) a difference of heaven and earth [between the usual Choki and the Choki of today].' (Richhi 157)

(5.137) a) ८ र् र्धेर याक्षा ८ वरी याक्षा में द्या वरी रेवा र्धेर याका केया पठरा देयाया t'a pénle ηa [di=le lò t'u? de:tei? nènle=tci?] now before 1SG this=ABL year six that.much before=INDF $tca:-z\varepsilon=la.$ come.HUM-PST=HON 'Now earlier, I came (here) [some six years earlier than now].' (unknown man on Bumchu video, see KLT) b) देव. गोषा रॉवे. हॅव या आक्तु या हैवा यहन ही क्रेंट करा [nim ke:pø: *pénlo*] ágja=lo tsi: te:-ti day many.GEN before elder.brother=DAT count entrust-NF mjõz-ts^haz. complete-CMPL '[Many days ago], I have completed entrusting (my body and mind) to the brother (=you).' (Richhi 147) c) ८ राख्या क्रेंप्टरे जुवाया nàtsa? [ódi hema] that before 1PL '[Before that] we...' (DR discussion with KL) (5.138) a) सुम्र' दर्न' क्रु' झेन्दायर' जुन्य' झेन्द' पर्यागो। p'um=di [tc^hu $t \in \eta k^{h} a$] $p \in d \sigma$ du-ke. above sleep stay EX.SEN-IN girl=DEMPH water 'The girl is floating (lit. lying) [on the water].' (TB e) b) विंदा गविषाक दाक्ष वयाविते क्रेट्रा क्रेंट्र येंट्र $k^h \tilde{o}$: ni: $t \epsilon^h a$ t'ato [nè:t^hi: tenlo] j*à*?. dø: 3PL two.of.them now bed.GEN above sit EX.PER 'The two of them are now sitting [on the bed].' (Richhi 18)

The relator noun gu(lo) occurs only in data from Martam (East-Sikkim), where the form occurs alongside $tegk^ha/teglo$. Note in (5.139c) that the cliticized form =gu may also attach to the relator noun teg 'up'.

(5.139) a) $\vec{\tau}$ ' $\vec{\tau}$ ' $\vec{\tau}$ ' $\vec{\tau}$

²³³ This may perhaps be considered a lexicalized loan word from English, because it is so frequently used and refers to higher working desks and eating-tables than the traditional Lhopo table *tcente*.

b) ୟ୍ର' ନରି'ଷଣ୍ଡ' ରଟ୍ଟସ୍'ଥିଣ୍ଡ୍'(?) ଛୁଁନ'ସି' ଛ୍ରମ୍ବା k^hu bε?. $[n\hat{e}] = gu$ zuktca? lõ:-bo 3sgm 1SG.GEN=on finger.pointing arouse-2INF EQU.NE 'He put blame [on me].' (KN e) c) यालन होन अगु $[pala\eta^{234}]$ tengu] bed on 'on the bed' (KN e) (5.140) a) ୮.୭୦୩ ସ୍ଥିୟ ଧାରୁ ଉଦ୍ୟୁ କିମ୍ବ ଅନ୍ୟୁ କିମ୍ବ ସ୍ଥାନ ବ୍ୟୁ କିମ୍ବ ସ୍ଥାନ କ୍ଷ୍ୟ ହିସ୍ଥା କ୍ଷ୍ୟ କିମ୍ବ କ୍ଷ୍ୟ କ୍ଷ୍ୟ କ୍ଷ୍ୟ କ୍ଷ୍ୟ କିମ୍ବ କ୍ଷ୍ୟ କ $nàtea?^{235}$ k^him jèbbe? ódi тòи gãi, [gari làm **ò:l**ɛ]. 1pl EX.NE that time car road below house down 'Our house was down (there) at that time, [below the car-road].' (DB life story) b) אוֹזייון־דָדִי (Thikadar) בָי הַמיאָה־אַריאֹי אוֹיק׳ דימימריףי (Ramasanhkha) אוֹ הֿמויממי אוֹמיבֿי אַר׳ אוֹיבּן אוי אוֹמיבֿי אַר׳ אוֹיבּן איי óna [ramasaŋk^ha=gi $t^{h}ikadar=tsu$ $t'yts^{h}o?$ $g\tilde{a}:=lo$ ò:lɛ] ke:p thikadar=PL time time=DAT there PN=GEN under many $h\epsilon^2$ mi=tsu. EQU.NE human=PL 'At the time of the thikadar-rulers, there were a lot of people there [under Ramasangkha].' (TB discussion with KT) (5.141) a) के कर्त्र हु हु क रामें के पा रामुनका mànpu **bu:na**] nà=lo mìk-tsum ma-kjap. [mí in.the.middle 1SG=DAT human many eye-close NEG-do 'Do not wink at me [in the midst of many people].' (song lyrics) b) ସିଁଦ୍ୟୁତ୍ତିର୍ମି ଅଂଗ୍ର ୩୫ ଅନ୍ୟ କାଶ୍ୱା ଅନ୍ୟ ଅଂଶ୍ଳର ଜ୍ୟାମ୍ପର୍ନି ଅଂନ୍ୟୁତ୍ୟୁ କାର୍ଯ୍ୟ କାର୍ଯ୍ୟ ହିନ୍ଦ୍ରା karma=jã: làka: $[k^h \tilde{o}_{i} = tsu = i \quad bu:na]$ dampe: bak-ti 3PL=PL=GEN in.the.middle PN=too in.hand Tibetan.guitar carry-NF lújã: tã:-zẽ: jò?. melody send-PROG EX.PER '[Among them] Karma too, carrying a Tibetan guitar, is singing a song.' (Richhi 120)

²³⁴ This word is so widely used that it should probably be considered a Nepali loan rather than an instance of codemixing. The more Tibetic word for bed is $n \hat{\epsilon} : t^h i$ (WD gauge *nyal-khri*).

²³⁵ I am not sure whether this non-genitive form is a mistake or intentional.

- lankor k'alv: deorali tshõ: thom k'aly: p'ja-z $\tilde{\epsilon}$: [g \tilde{a} :to? t^{h} om, car slow slow do-INF TPN town TPN market bu:na=le] t^høn-di in.the.middle=ABL come.out-NF 'Going slowly, the car comes out [from the midst of Gangtok town and Deorali market] (and)...' (Richhi 120)
- d) هَ مَ حَرَّ عَلَى مَ حَرَّ عَلَى اللَّهُ مَ حَرَّ عَلَى اللَّهُ عَلَى اللَّالَةُ عَلَى اللَّهُ عَلَى اللَّا عَلَى اللَّا عَلَى اللَّا عَلَى اللَّالِ عَلَى اللَّهُ عَلَى اللَّهُ عَلَى اللَّهُ عَلَى اللَّا عَلَى اللَّا عَلَى اللَّا عَلَى الْ المَالِكُلُوا عَلَى اللَّهُ عَلَى اللَّهُ عَلَى اللَّهُ عَلَى عَلَى اللَّالِكُلَى اللَّالِكُلَى اللَّالَةُ عَلَى اللَّا عَلَى اللَّا عَلَى اللَّا عَلَى اللَّالَةُ عَلَى اللَا عَلَى اللَّالَالَا عَلَى الللَّهُ عَلَى الللَّهُ عَلَى اللَ
- (5.142) a) ह्येगा दर्च लेगा जुलाह्ये

[gɛ:ka **bolo?**] nɛ̀:t^hi. window beside bed '(There is) a bed [beside a window].' (nga'i 'gan 1)

b) विंते तर्रे लेगा में जुः वर्षा पति आहे कार्य गठिग थेंन

 $[k^h u=i$ **bololo**] $gjat^h \tilde{a}: k^h \phi:$ $\dot{a} dzo=t \epsilon i \partial$ $j \dot{\phi} \partial$ 3SGM=GENbesideIndian.plains.GENgrandfather=INDFEX.PER'[Next to him]is a grandfather from the plains of India.' (Richhi 120)

 $[d \in ndz \tilde{o}: nan \in a = lo]$ dku [opo=tsu punnun=tsu $me:-k^{h} \in n$ $b \in = eo = la.$ Sikkim inside=DAT uncle Lhopo=PL few=PL EX-NMLZ EQU.NE=AT=HON '[Within Sikkim], Lhopos are not few.' (KT discussion with TB)

b) येंग्येंते वृत्त्ये सुते भूत हा क्षेक के हा सुन ता

[$j \delta l \delta :$ $n \delta \eta l \delta$] $l \dot{u} = i$ $k \varepsilon : da$ $\eta \varepsilon n t \varepsilon^h i t \varepsilon^h i$ $dam n \tilde{\varepsilon} := na$ curtain inside song=GEN sound pleasant Tibetan.guitar=LOC '[Within the curtain] (there is) a pleasant sound from a Tibetan guitar.' (Richhi 81)

c) क्षें वृद्त यो द्वीं वृत्ते वृत्त व

[*ónaŋ=gi gjømpo nàŋna*] there=GEN monastery inside '[within the monastery there]' (DB trip story)

The forms p^{hilo2} and $payk^{ha}$, both meaning 'outside', were in my natural data used independently as adverbs. Consultant KN, however, affirmed that they are also used as postpositions, as shown by elicited examples (5.144) and (5.145).

(5.144) यु. युम्र (गु.) ही केंग केंद्र हेंद्र तत्या

 $k^{h}u$ $k^{h}im(=gi)$ $p^{h}ilo?$ $l\tilde{o}$: $d\emptyset$: du?. 3SGM house(=GEN) outside stand stay EX.SEN 'He is standing outside the house.' (KN e)

(5.145) यु. ख्रिय (क्रु) यहायर येंहा झूंहा वहुवा

 $k^{h}u$ $k^{h}im(=gi)$ $p^{h}a\eta k^{h}a \ l\tilde{o}$: $d\omega$: du?. 3SGM house(=GEN) outside stand stay EX.SEN 'He is standing outside the house.' (KN e)

5.6.3 Adverbs as adverbials

This section exemplifies how adverbs are used for expressing manner (§5.6.3.1), location (§5.6.3.2), time (§5.6.3.3), quantification (§5.6.3.4) and other meanings (§5.6.3.5).

5.6.3.1 Adverbs of manner

Adverbs of manner, which were introduced and listed in §3.5.2.1, are exemplified in (5.146-149).

- (5.146) a) הַיִּשִּ וֹעָהִ יַשְׁשָּׁמִיגִי שִּשְּשִי שִׁיִשְׁ שִּׁהַ שִּׁהָ שִׁהָן
 t'ato k^hõ: ní:-po pámteilo to sà-zɛ̃: jò?.
 now 3PL two-COL together food eat-PROG EX.PER
 'Now the two of them are eating food together.' (Richhi 20)
 - b) होते. हार्ट्य रावर. हार्वेश हवा. पत्री. घूटा. रा

mí: dyŋk^ha pámtɛiʔ gju t^hop-o-dã: human.GEN in.front.of together go receive-2INF-CONJ 'When getting (a chance) to go together in front of people...' (NAB BLA 7)

(5.147) هَ مَجْ الْمَعْمَةُ مَجْ اللَّ عَالَمَةُ مَحْ اللَّ عَالَمَةُ اللَّ عَالَمَةُ اللَّ عَالَمَةُ اللَّ عَالَمَةُ اللَّ عَالَيْ اللَّ عَالَيْ اللَّ عَالَيْ اللَّ عَالَيْ اللَّ عَالَى اللَّهُ مَعْنَى اللَّ عَالَيْ اللَّهُ عَالَى اللَّهُ مَعْنَى اللَّهُ عَالَى اللَّهُ عَالَى اللَّهُ اللَّهُ عَالَى اللَّهُ اللَّهُ عَالَى اللَّهُ عَلَى الْحَالِي عَالَى اللَّهُ عَالَى اللَّهُ عَالَى اللَّهُ عَالَى الْحَالَى الْحَالَى الْحَالَى الْحَالِي عَالَيْ عَالَى الْحَالَى الْحَالَى الْحَالَى الْحَالِي مَالِكَ عَالَى الْحَالِي عَالَى الْحَالِ الْحَالَ عَالَيْ عَالَةُ عَالَيْ عَالَيْ عَالَى الْحَالَةُ عَالَى اللَّهُ عَالَيْكَ عَالَيْكَامِ عَالَيْ عَالَى الْحَالِي عَالِي عَالَيْ عَالَةُ عَالَى الْحَالَى الْحَالَى الْحَالَى الْحَالِي عَالِي عَالِي عَالَةُ عَالَيْكَ عَالَى الْحَالَةُ عَالَى الْحَالَى الْحَالَى الْحَالِي عَالَةُ عَالَيْ عَال المَالِي عَالَةُ عَالَيْ عَالَيْ عَالَيْكَ عَالِي عَالَيْ عَالِي عَالِي عَالِي عَالِي عَالِي عَالِي عَالَةُ عَالُ عَالَ عَالَ عَالَيْ عَالَيْ عَالَيْ عَالَيْ عَالَ عَالَيْ عَالَةُ عَالَيْ عَالَةُ عَالَيْ عَالَةُ عَالَيْ عَاحَالَيْ عَ

(5.148) المَّانِ مَحْرَ عَلَيْ اللَّهُ عَلَيْهُ اللَّهُ اللَّهُ عَلَيْهُ اللَّهُ عَلَيْ اللَّهُ عَلَيْ اللَّهُ عَلَيْهُ اللَّهُ عَلَيْهُ عَلَيْ اللَّهُ عَلَيْ اللَّالِي الْحَالَيْ اللَّهُ عَلَيْ اللَّا عَلَيْ الْحَالَةُ اللَّهُ عَلَيْ الْحَالَيْ الْحَالَةُ اللَّالِيلَةُ اللَّهُ عَلَيْ عَلَيْ الْحَالِي الْحَالَةُ اللَّا عَلَيْ الْحَلَيْ عَلَيْ الْحَالَةُ اللَّالِي الْحَلْقُولَةُ اللَّالِي الْحَلَيْ عَلَيْ عَلَيْ الْحَالَةُ عَلَيْ الْحَالَةُ اللَّالِي الْحَلَيْ عَلَيْ الْحَلْ المَا عَلَيْ اللَّالَةُ عَلَيْ اللَّالَةُ عَلَيْ اللَّا عَلَيْ اللَّالِي الْحَلَيْ عَلَيْ الْحَالَةُ الْ الْعَلَيْ اللَّالَةُ اللَّالَةُ اللَّالِي الْحَالَةُ عَلَيْ اللَّالِي الْحَالِي اللَّالِي اللَّا عَلَيْ الْحَالِي الْحَالَةُ اللَّا عَلَيْ الْحَالَةُ اللَّالِي الْحَالَيْ الْحَالَةُ عَلَيْ الْحَالُ الْحَالَةُ الْحَالِي الْحَالَةُ عَلَيْ الْحَالِي الْحَالِي الْحَالَةُ اللَّالِي الْحَالَةُ عَلَيْ الْحَالَةُ عَلَيْ الْحَالَةُ عَلَيْ الْحَالَةُ عَلَيْ الْحَالَةُ عَلَيْ الْحَالَةُ عَلَيْ الْحَالِي الْحَالَةُ عَلَيْ الْحَالَةُ عَلَيْ الْحَالَةُ عَلَيْ الْحَالَةُ الْحَالَةُ الْحَالَةُ الْحَالَةُ عَ المَا اللَّا عَلَيْ اللَّا عَلَيْ الْحَالَةُ عَلَيْ الللَّا عَلَيْ الْحَالُ الْحَالَةُ عَلَيْ الْحَالَةُ عَلَيْ الْحَالَةُ عَلَيْ الْحَالَةُ الْحَالَةُ الْحَالَةُ الْحَالَةُ عَالَ عَلَيْ الَحَالَةُ عَلَيْ الْحَالَةُ عَلَيْ الْحَالَة

Three adverbs are used for manners of sleeping:

- (5.149) हु गॅन् भूभ / यास्त / युरा हे जया हे र यह गा
 - $k^h u$ $k^* \tilde{o}: k\epsilon/k^h a b u p/s \tilde{u}: t\epsilon$ $p \tilde{\epsilon}:$ $d \phi:$ d u 2.3SGMon.back/on.tummy/on.sidesleep stayEX.SEN'He's is sleeping on (his) back/on (his) stomach/on (his) side.' (KN e)

As suggested by examples (5.146-149), adverbs of manner generally tend to occur close to the final verb, see (5.146b), (5.147) and (5.149). Temporal adverbs, on the other hand occur more frequently in clause-initial position or otherwise well before the verb. The manner adverb *hatok*^ha 'suddenly', which has a temporal nuance, is in (5.148) positioned quite far-away from the verb.

5.6.3.2 Locative adverbs

Locative adverbs typically occur in preverbal position (5.150) but may also be fronted if topical (5.151-152).

- (5.150) التي التي المعربي المعربي المعربي المعربي المعربي المعربي المعربي (5.150) (المعربي المعربي المعربي المعربي المعربي المعربي المعربي المعربي (المعربي المعربي المعربي المعربي المعربي ((KN e) المعربي المعربي (S.150) (S.150)
- (5.151) $\exists a_1 : \exists c_1 : a_1 : \exists c_1 : a_1 : a_2 : a_1 : a_2 : a_1 : a_2 : a_$
- (5.152) $rac{1}{7}$, $rac{1}{7}$, $rac{1}{7}$, $rac{1}{8}$, $rac{1}{8$

Most postpositions which are relator nouns (for definition, see §3.6.8) can be independently used as locative adverbials, see (5.153), which shows that postpositions used as locative adverbs typically occur just before the verb.

sém lèp màla? mù=i ga-ti màla? *ò:=lo* p'ap. 3SGF=GEN mind very.much rejoice-NF fast fast down=DAT descend 'Rejoicing in her mind very much, (she) came down very quickly.' (rna-gsung 4) b) अर्. छूटा ट्रटा लटला कुला कुटा लेंग छेंट्र छे latcun t'ã: sánge? gjablo dø:-ti and PN at.the.back sit-NF PN 'Lhachung and Sange, sitting in the back and...' (Richhi 51)

²³⁶ In written language and polished spoken language, the repetition of *óna* is considered infelicitous.

c) हा. पट्ट. जूरे. क्रूट. वर. जूट. हू. jòu **tɛŋkʰa** số̃ː-di $k^h u = di$ 3SGM=DEMPH up above go.PFV-NF 'He went up above and...' (KT animal story) d) ८. प्र्यालक खे. र. श्रुयलयामा $\tilde{i} = la.$ à:=l₽ cù-do ηà 1SG down=ABL speak.HUM-IPFV EQU.PER=HON 'I speak from down (here).' (KT phone call) e) गोव र्रेंट पर्ट श्चुव प्रवया kintsõ:=di za?. bu:na maize=DEMPH in.the.middle set 'The maize is put in the middle.' (PL interview) g) הי קריקי מקי מקן te nànca dzy: làp. so inside enter say

'So enter inside, (he) says.' (KT animal story)

For demonstrative pro-adverbs, see §3.6.5 and §6.4.

5.6.3.3 Temporal adverbs

(5.154) ਉੱਤਾ ਬਕਥਾਰਨਾ ਤਕਾ ਤਾਂ ਦੇ ਹੋਨਾ

Temporal adverbs tend to occur in clause initial position (5.154) or after the agent (5.155-156), depending on topicality considerations. The examples below illustrate adverbs referring to time of day (5.154), day (5.155) and year (5.156).

	$p^{h}iru?$ $t^{h}amtee?$ sàm sà-ti $mj\tilde{O}$:. at.night all food eat-NF finish
	'At night everyone has finished eating.' (Richhi 4)
(5.155)	\tilde{a} ' ך'דר' תַלָּא' אָד'אָד' עָר', אי דר' דר' אַק'תַר' דָר' אָזי תַר' אָז' אַד' גער', אי דר' דר' אַז' גער' גער' גער' גער' גער' גער' גער' גער
	3SGF today smile-IDEO EX.PER always and NEG-similar 'She is smily today, unlike usually.' (Richhi 148)
(5.156)	దా. న్రాంకింగా ఈ సినినాలు. ఇార్తేదా. ఇద్దా కారా ఇద్దా. ఇద్దా ఇద్దా. ఇద్దా ఇద్దా ఇద్దా ఇద్దా ఇద్దా ఇద్దా ఇద్దా
	yà t'utci? t'a ma-lɛp, nàpĩ: =di tsʰaː píː, nàniŋ
	1SG this.year now NEG-arrive last.year=DEMPH turn two last.year
	mébbe, zø:nin da.
	NEG.EX.NE two.years.ago AP
	'This year I didn't go (there), last year two times, it wasn't last year, it's two years ago, I think.' (KT discussion with TB)

The different placing of the temporal adverbials in (5.157) and (5.158) is conditioned by topicality.

bhaila=ki go tok-ti dã: nàtci di dø: j*à:-kap* yesterday 1PL.GEN PN=GEN head hit-NF fall stay EX-SIM k^hu *bε*?. p'usim=la: mè:-patsene óna=rã: *сі-с*г there=AEMPH die-INF EQU.NE younger.sister=HON NEG.EX-COND 3SGM 'Yesterday when our Bhaila was lying fallen after hitting his head, if it wasn't for the sister, he would have died on the spot.' (Richhi 12)

In (5.158), the topic ($k^h u$ 'he') that has arisen from the previous context is fronted and the time adverbial $d\tilde{a}$: occurs within the comment/focus part which provides new information about the topic. In (5.157), on the other hand, the topic established by the previous context is *p'usim* 'younger sister' and the proposition (5.157) draws attention to what happened the previous day in relation to the younger sister.

The postpositions $p\acute{e}nlo/p\acute{e}nlo$ 'before' and $gjablo/gjabl\epsilon$ 'behind; after' are also used as temporal adverbs, see (5.159). Whereas $p\acute{e}nlo/p\acute{e}nl\epsilon$ is temporal, $gjablo/gjabl\epsilon$ can also mark location, see (5.153b) above.

(5.159) a) वैता वर्ता कुतालया आहिं मेतावृता क्षेत्र

 $\epsilon i \eta = di$ **gjable** $\dot{a} tsi$ $ri\eta - \epsilon \epsilon$ $\dot{\tilde{t}}$. tree=DEMPH afterwards a.bit be.long-INF EQU.PER 'Later this tree will grow a bit.' (KN e)

b) אָד צָּק מעשי הא שָֿק שָׁקשייד אָד אָדן *kʰu píɛ́nlɛ nɛ̀: t 'oku ím-bo bɛ?.* 3SGM earlier 1SG.GEN friend EQU-2INF EQU.NE 'Earlier he was my friend.' (KT e)

Temporal adverbs *t'arun* 555 'again, yet, still' and *te:ra* 55 'x' 'again' express frequency:

(5.160) ה' נאֹיק ימושי ה' ה' הישה' משוי מה ימושק צורן t'a ónale t'a t'arun doly=tei? jò:-khen be?. now then now still custom=INDF EX-NMLZ EQU.NE 'Now then there is still another custom.' (SGD wedding customs) (5.161) हेन्वा हे रा हेन् ??

tsim tɛ:ra tsi:-ca? game again play-INF.PQ 'Shall (we) play the game again?' (PT kitchen discussion)

Example (5.162) illustrates an indefinite temporal adverb, which occurs in a repetitious bisyndetic construction.

(5.162) रे. तयात' अभाव' जुभ, रे. तयात' सॅट क्षे' धर र्खर' तया

rega: $m\hat{\epsilon}$:=na $n\hat{\epsilon}$:, **rega:** $l\hat{o}$:-ti $p^{h}a$: $ts^{h}u$: gju. sometimes bed=LOC sleep sometimes rise-NF thither hither go 'At times (he) lies on the bed, at times (he) rises up and walks to and fro.' (Richhi 117)

5.6.3.4 Verb-modifying quantitative adverbs

Quantifying adverbs, which are summarized in Table 5.4 (see also §3.5.2.4), modify the verb or the whole clause. Each of the forms are exemplified after the table.

	mouny mg que	
lèp(ti)	लेपा(हूँ))	'very much'
ke:p, ke:po	गोषाया, गोषार्था	'much, a lot'
та̀ури, та̀уро	ષ્રદ્ર', ષ્રદ્ર'ર્યે'	'much, a lot'
ts ^h ɛdɛ̃:	र्बन:स्रव:	'considerably'
màŋts ^h ø?	यट. कूर.	'to great degree, more (than)'
pùŋtsʰø? ²³⁷	લુન્-`ર્સન્'	'little, less (than)'
сў:су:	প্রম:প্রম	'a bit'
сý:tcy:	প্রুষা তিযা	'a bit'
cýry	. मुह्य में	'a bit'
átsi(m)	জান্ট(অ')	'a bit'
átem	क्षाः ह्येव्याः (?)	'a bit' (rare)
րնդրսդ	લુ⊏.લુ⊏.	'little, few'
tee:/dze:	रें'/तहें	'at all' (+negation)
tsa(:)lɛ	5. UN	'at all, never' (+negation)
bekki	न्दर्गे.	'at all, anyhow' (+negation) (Lachung)

(5.163) यदेव यात्रा केव यदेव यात्रा हे लेय कर्वे वे पर्ट

 $d\epsilon\eta$ -gam min- $d\epsilon\eta$ -gam te $l\epsilon p$ nó:- $n\epsilon$ =di be.true-ATTQ NEG-be.true-ATTQ so very.much think-COND=DEMPH 'So if one thinks a lot about whether it is true or not...' (CY interview)

²³⁷ Also pronounced *pùŋts^hi?*

(5.165) أَنْ المِعْمَ عَلَيْهِ عَلَيْ عَلَيْهِ عَلَيْهِ عَلَيْهِ عَلَيْ عَلَيْهِ عَلَيْ عَلَيْ عَلَيْ عَلَيْ عَلَيْهِ عَلَيْ عَلَيْ عَلَيْ عَلَيْ عَلَيْ عَلَيْهِ عَلَيْ عَلَ

(5.166) مَنْ مَعَرَبُ مِعْمَانُ هَرَ عَجَمَ مَالَمَ عَرَبُ عَمَ مَالَ مَعْرَبُ عَمَ مَالَمُ عَرَبُ عَمْ مَالًا عَلَى مَالًا مَعْرَبُ عَمْ مَالًا عَلَى مَالًا عَلَى مَالًا عَلَى مَالًا عَلَى مَالًا عَلَى مَالًا مَالًا عَلَى مَالًا مَالًا عَلَى مَالًا مُعْلَى مُعْمَا مَالًا مَالًا مَالًا مَالًا مَالًا مَالًا مَالًا مَالًا مَالَا مَالَا مَالَا مَالًا مَالًا مُعْلَى مُعْلَى مَاللَا مَالًا مَالَا مَالَا مَالَا مَالَا مَالًا مَالًا مَالًا مَالًا مَالًا مَالًا مَاللَّ مَالًا مَالًا مَالًا مَالًا مَالًا مُواللًا مُواللًا مُواللهُ مَاللًا مُنْ مَالًا مَاللَا مَالًا مُواللًا مُواللًا مُوالل مَالًا مُوالل مُوالل مَالًا مُعْلَى مُوالل مَاللَا مُعْلَى مُوالل مَالًا مُوالًا مُوالل مَالًا مُوالل مَالَا مُوالل مَاللَا مُوالل مُوالل مُوالل مُوالل مُوالًا مُوالل مُواللمُوالل مُوالل موالل موال

The quantitative adverb $m a \eta t s^h \emptyset ?$ 'to great degree, more (than)' expresses a great degree or majority of cases, see (5.167). Often an idea of comparison is involved and, hence, 'more' may be used in translation, see (b) and (c).

- (5.167) a) व्यन्तर्केन मेन रावते मी वर्क हेंबर मन क्षेत्र कें प्रनि क्षन cé:mø:=gi ĩ mànts^hø? nàts^ha t^høn-se? ódi gã:. great.degree measles=GEN illness become-INF EOU.PER that time 'There was a great deal of measles at that time.' (PED life story) b) र्विन अनुवासु गा र्हेन्से वन केन nàmpu ka $k^h \tilde{o}$: de:-po màŋtsø? with who stay-2INF more 3rd.hon "...who stays with him/her most/more." (YR boys' and girls' clothing) c) नृत्तःसुग्गुः नुषार्क्वनायाषा नेताषाता के केंग्राषा अत्तार्क्वना यानार्या भ्रुना t'ã:pu=gi $t'yts^h \emptyset := l\varepsilon t'ins \tilde{a}$: míts^ho? mànts^hø? jờ-po long.ago=GEN time=ABL nowadays crowd(s) great.degree ex-2inf *bε*?. EOU.NE 'Nowadays there are more people than in the earlier times.' (KUN e) (5.168) डुगुराय'ग' पर्ने' अन्दर्कत' नृत्त' न्द्र' राज्या' क्वें'ब्रेंब'ग' पर्ने' ज़न्द्र केंत्र' अप्यायाया क्वें'पर्ने' केंब्र' कुय'ग्रीय' गवन्यावन' कें केंद्र' केंब्र' कुय'ग्रीय' गवन्यावन' कें केंद्र' केंब्र' क्वें' केंब्र' केंद्र' केंब्र' केंद्र' के

(5.169) नुषः नुषः ११ भनः ढुंगुषार्धेः दर्धेदः र्धेनः ५८ गा cý:cy: k^ha làp ts^hu-po *t^høn* dø: du? a.bit mouth speak be.able.to-2INF become stay EX.SEN 'He has achieved a state where he is able to talk a bit.' (Richhi 23) (5.170) हे दारुवा वीषा सुषा हेवा हुव र्झेट थेंटा nàtea=ki cy:tey: pèn dø: j@?. tε then 1PL=AGT a.bit listen stay EX.PER 'So we have listened to (these songs) a bit.' (RS song intro) (5.171) नाक्ष वनायां सुषामे द्वायां हुवाई ťato nè:po cýry ťempo sýn-di now patient a.bit memory awaken-NF 'Now the patient has regained consciousness a bit (and)...' (Richhi 168) (5.172) ฉริ ยญทั ฉริ พาซิ ริราร์ ซุรา t^hako=di átsi rim-bo bɛ?. di this rope=DEMPH a.bit be.long-2INF EQU.NE 'This rope is a bit (too) long.' (KN e) (5.173) ૬'ર્સેં' દારુવા લાસેયા દોવુ ઘેંદ્રા t'ato yàtea? átem p'jeu jò?. a.bit hurry EX.PER now 1PL 'Now we are in a bit of a hurry' (DB day trip) (5.174) દાજ્યા હેંગ્લર્ન સુદ રાગવા બેવ સુદ સુદ દેવા મુદ્દ હેવા મુદ્દ છે. $\eta \dot{a} t \epsilon a^2 \circ di g \tilde{a}$: $s \dot{a} - \epsilon \epsilon^2 = l \dot{e} p \cdot \mu \eta \mu \eta t^h o p - \epsilon \dot{t}$. receive-NPST.PER 1pl that time eat-INF very little 'At that time we got very little to eat.' (PED life story)

Three adverbs tee, tsa:le and bekki occur with a negated verb and emphasize the fact that the action denoted by the verb was not done 'at all'.

(5.175) ब्राह्म गहुवा में रही हो के अब वार केंट्र रहा वर्षा में

 $l\tilde{a}$: $t \varepsilon u k u = di$ p'i $t \varepsilon \varepsilon$: man-za-wa $d \sigma$: $d u - k \varepsilon$. bull other=DEMPH fodder at.all NEG-eat-CIRC stay EX.SEN-IN 'The other bull lived without eating any fodder.' (TB bull story)

(5.176) हे' वर्ने' मारुषा साथकाम वर्षेत के खेना

 $t\varepsilon$ di $\eta \dot{a}t\varepsilon a$? $tsa: l\varepsilon = ra$ $t^h \omega n$ mi-si?. well this 1PL at.all=AEMPH happen NEG-be.possible 'Well, it's not at all possible for us to make it happen.' (NAB BLA 7) *ódi ts^hø: ta-wo ť:-mɛ. bɛkki ma-ne?*.²³⁸ that search look-2INF EQU.PER-?? at.all NEG-find 'I searched it (but) did not find it anyhow.' (KUN e)

The initial syllable of $te^{ha}:l\epsilon$ 'all' also occurs in a converb-looking construction, see (5.178). The verb-looking $te^{ha}:$ is tentatively glossed as 'be all'

(5.178) جٚ^א' (קה') कू⁻ଛୁ⁻ t'o? (**t'ã**:) tɛʰaː-ti

load (and) be.all-NF 'all the loads' (KN e)

The construction $(d\tilde{a}:)$ te^ha:-ti can co-occur with additional quantification:

(5.179) באי (ברי) שיציי אממיסרי

t'ep (*t'ã:*) *te^ha:-ti t^hamtee*? book (and) be.all-NF all 'all the books entirely' (KN e)

5.6.3.5 Other adverbs

Other adverbs include the epistemic adverbs né:mu(ra) कात्रे ख़ु'(रू') 'really' (5.180), mèntcene/mèntceno कात्र छे त्रे '/जात् 'छे'त्रें' 'perhaps, maybe' (5.181) and mènne/mènni²³⁹ कात्र त्रे ' perhaps, maybe' (5.182).

- (5.180) العربي المعربي (5.180) المعربي المعربي المعربي المعربي (180) المعربي (180) المعربي (180) المعربي المعربي المعربي (180) المعربي (18
- (5.182) هم تم تم تم تح معت معن معن تمن تمين تم شي تم تم تم تم تم تم شي المعن مي تم تم تم تم تم تم mènni t^horã: ŋà màrtam gju أُ:-to. maybe tomorrow ISG TPN go come-PROB 'Maybe I'll go to Martam tomorrow.' (KN e)

²³⁸ This clause from Lachung has two peculiarities. The first is the verbal ending $-m\epsilon$, which does not occur in my data elsewhere (hence the gloss ??) and which was reported by KUN to convey that the sentence is not complete. The form resembles Dzongkha "exophoric copula" *imme* (Watters 2018: 338, 442). The second peculiarity is the word *ne?* 'find' (WT $\frac{2}{8}5$ ' *rnyed* 'gain, find') instead of which *t^hop* 'find' is used elsewhere in my data.

²³⁹ The form *menni* is from consultant KN (Martam).

The postposition $t' onlo \neq \bar{\gamma} q$ for the purpose of can be amplified by the reflexive/anaphoric emphatic $=r\tilde{a}$: to form the epistemic adverb $t' onlor\tilde{a}$: 'really, truly', see (5.183).

(5.183) דְׁקִיאֹיִדְהִ׳ הִישּׁתִ׳ יוּקָאִיאָּן עֹקִיעִדִישּׁיקִ *t'onlo=rã: nàtsi lɛ̀:námko jò:-patsene* real=AEMPH 1PL.GEN fate EX-COND 'Really, if it is our fate...' (Richhi 113)

The restrictive adverb *teiku/teuku* (η) $\exists \eta \neg \eta'$ only' is postposed to the element it modifies. It can modify the whole dependent clause (5.184), another adverb (5.185), a numeral (5.186) or a noun (5.187).

(5.184) a) विं वेंगा क्वेनका क्वे रुपा गा फुर्गा

 $k^h u$ $l \partial k$ $l \varepsilon p - ti$ $t \varepsilon i k u$ hako.3SGMreturnarrive-NFonly understand'He understands (it) only upon returning.' (Richhi 26)

b) दे. र्या प्रया झुला दा हेया गा दें दा र्या क्षेत्र

t'ɛ:lu k^ha : $k\varepsilon$:-wa **tciku** ∂m -bo \tilde{t} : just.like.that message bring-PUR only come-2INF EQU.PER 'I have just come like that to bring you a message.' (nga'i 'gan 11)

(5.185) हर्ग्वते क्रुग्मर्क्व र्ग्निगीया मुझ ठेयागा जार्यार्भ क्षेत्र

 $\eta \dot{u}$ -wø: gjumts^h \tilde{e} : $k^h u = gi$ t'ato teiku hako-wo \dot{t} :. weep-2INF.GEN reason 3SGM=AGT now only understand-2INF EQU.PER 'He understood the reason of (her) weeping only now.' (Richhi 170)

- (5.186) ج: عِنْ جَعْابَيْ *p'u ná t'u? teiku* boy five six only 'only five-six boys' (PL interview on farming)
- (5.187) ज्लूब्र कॅलायरेवा हेवा गा *lám ts^høpsl tsiku*

lama PN only 'only lama Chopel' (LT kitchen discussion)

5.7 Summary remarks

This chapter showed that simple categorizing of Denjongke as either nominative-accusative or absolutive-ergative is not feasible, because argument marking of A and P is to a considerable degree conditioned by pragmatics and lexical choices. Therefore case-marking for the A argument was called agentive, a semantically-oriented term, rather than ergative. It was shown that the sole

argument of intransitive clauses may be agentive marked for emphatic purposes, while the marking of A argument in transitive clauses shows signs of both syntactic control (some verbs require agentive-marking in the past tense) and pragmatic control (e.g. A arguments with and without agentive-marking are offered in elicitation). The marking of P argument was seen to be sensitive to animacy, identifiability/specificity and affectedness. Moreover, this chapter showed that alignment of ditransitive clauses does not clearly fit any of Haspelmath's (2005) alignment types (indirective alignment, neutral alignment and secundative alignment), because the marking of P argument is split between zero-marking (also used for T[heme] argument in ditransitive clauses) and dative-locative-marking (also used for marking R[ecipient] argument).

It was shown that valency modification can be accomplished through valency decreasing argument suppression (resulting in "functional passive," see Givon [1984: 164]), and valency increasing causative constructions. Adverbial modification (not including adverbial clauses) is accomplished through case-marked noun phrases, postposition phrases and adverbs. Comparative clauses accomplished by the ablative can be categorized as "locational" and further as "from-comparative" (Stassen 2013b).

6 Deixis and reference

This chapter addresses issues related to deixis and reference in Denjongke. Deixis refers to the way in which context helps to determine the referent of a linguistic expression (Levinson 1983: 54). For instance, the referent of the personal pronouns *I*, *you*, and *she* is determined by the context. Similarly, the referent of spatial terms such as *here* and *there* is revealed by the context. The term "reference" covers "definite" (e.g. *he*) and "indefinite" (e.g. *someone*) ways of referring to people and objects. The treatment is divided into personal pronouns (or personal deixis) (§6.1), reflexives and reciprocals (§6.2), indefinite reference (§6.3) and demonstratives (§6.4)

6.1 Personal pronouns

The use of the personal pronouns is described here. For ease of reference, the forms already introduced in §3.6.1 are reproduced in Table 6.1.

1 uon	uble 0.1.1 ersonal pronouns						
			Singular	Plural			
1p			ŋà ⊏	<i>ŋàtɛa? ८</i> :उग			
2p	low-level		$tc^h \phi$? \breve{a}_{5} .	(<i>tc^hø:=tsu</i>	k'utca? শ্যুতব্য		
	mid-level		rã: ᠵᠵ	rã:=tsu रूप:र्डु:			
	honorific		lenge? झुव-मुल-	k'utca lɛŋgɛʔ નુ·રુગ સુર્વ क्रुष			
				<i>lɛŋgɛ:(=tsu) স্থ্র</i> সন্ট্রম'(র্ <u>র</u> ')		
3p	ordinary masc.		$k^h u$ \mathbb{R}^{-1}	$k^h ilde{o}$: $ ilde{o}$			
		fem.	mù/mò zy				
	honorific		khõ: 丙二·	$k^h \tilde{o}$: $lenge$: (= tsu) \check{n} =: \dot{e}	ईव.मैथ.(र्थ.)		
			<i>kʰõː lɛŋgɛ? ་</i> མོང་ ཕྱན་གྱས་				

Table 6.1. Personal pronouns

The first person plural form does not take stance as to whether the addressee is included in the "we" or excluded from it (see §3.6.1). The use of $\eta \dot{a}$ is illustrated in (6.1). For inclusive and exclusive uses of $\eta \dot{a}tea?$, refer to (6.2) and (6.3) respectively.

(6.1) क्षॅ.वर्द्दे.र्च्य. 240 ट. झॅ्या. घ.क्यांग

ódi-p'ja **yà** *dok ma-ts^hu?*. that-ADVZR 1SG read NEG-be.able.to 'Therefore I could not study.' (PED life story)

(6.2) ८. राज्या व के गोषार्य वर्हकाय छे वे

yàtea? nà: mí kɛ:po dzom-batɛɛnɛ 1PL here people a.lot gather-COND 'If we (incl.) gather here as many people...' (NT BLA 6)

²⁴⁰ Some Denjongke authors prefer the written form <u>wink</u>, which suggests that the demonstrative is in agentive form [that.AGT-ADVZR]. In spoken pronunciation, however, I have found no evidence for a longer vowel or other signs of agentive marking. Therefore, the form is here written without agentive marking.

dã:	$k^{h}a:nu:=lo$	ŋàtɛa?	р ^ь ои	jó?	kjap	de:-pø:
yesterday	the.day.after.yesterday=DAT	1pl	over.there	word	do	sit-2INF
gã:						
time						
'A few day	ys ago when we (excl.) were w	orking o	ver there?	(PL i	ntervi	ew)

Second person singular may be referred to on three levels. The familiar level $te^h \phi ?$ is used with close friends, social inferiors and those one despises. The mid-level $r\tilde{a}$, which literally means 'self' (see §6.2 for reflexive pronouns) can be used with one's equals or inferiors with whom a relationship already exists. The honorific lenge? is typically used with social superiors, strangers and anyone to whom the speaker wants to show respect. The same form lenge? is also used for referring to second person plural and as an honorific in conjunction with third person plural referents (it is not used for 3^{rd} singular referents).²⁴¹ Because lenge? is used for both 2^{nd} and 3^{rd} person referents, it is here glossed just as an honorific pronoun (PRN.HON) whose exact reference has to be understood from the context.²⁴² The three levels of address in second person singular are illustrated in (6.4-6). Note that in examples (6.5) and (6.6), taken from a Denjongke vocabulary cum phrase book (the data there conforms to spoken data), lenge? 'you (hon.)/honorific pronoun' occurs with the honorific noun ts^hen 'name (hon.)' whereas $r\tilde{a}$: 'you' is accompanied by the ordinary noun mn 'name'.

(6.4) केंद्र' केंद्र' यार'/यात' र्दे?

tchø?mìŋk'ar/k'anbo?2SG.Lname whatEQU.NE.Q'What's your name?'

(6.5) रूट मी' केट मुत र्य?

ray=gimìŋk'anbo?2SG.M=GENname what EQU.NE.Q'What is your name?' (JWD book p. 115)

(6.6) क्षेत्र क्रुवा ग्री अर्ह्य र्या पत्र लुया पत्र राष्ट्र या मार्ट रे जायाय?

Ordinary level second person plural is *k'utea?*, as in (6.7). The honorific pronoun *lenge?* is the default choice in formal situations, see (6.8). The mid-level $r\hat{a}$: can also be pluralized, although

²⁴¹ People from the villages of Lachen and Lachung in North Sikkim are reported to have a more direct way of speaking called $k^{h}ari? k^{h}atu? retained to have a more direct way of speaking of the lack of honorifies. Lachenpas and Lachungpas may address even strangers by <math>tc^{h}\phi?$, a practice which is considered vulgar by more southern speakers.

²⁴² An indication that lenge2 has wider semantics than 2SG is that once a person whom I addressed with lenge2 did not immediately understand I was referring to him personally.

this use is rare in my data. Note that (6.9) combines the use of $r\tilde{a}:=tsu$ and lenge:=tsu. The use of lenge=tsu in (6.9) seems to be a general reference to a group of individuals.

- (6.7) ญ องญาณ์ จงงงาร์ จาสาราราชางา พีราราอิเสิ **k'utca**=lo sámpo zã:ta? j*à-patse(ne)* 2PL=DAT intention good EX-COND 'If you have good intentions...' (SGD wedding customs) (6.8) ८. रुतिः अत् पर्दा पर्दा र देवा है यहे में र में ke:=di dep ŋàtci tsom-diki t'ato lópøn 1PL.GEN language=DEMPH like.that compose-NF now teacher lenge:=gi t&ha:le? k'adzø? ze: jøː-kʰɛn bε?, PRN.HON=AGT work.HON how.much eat/obtain.HON EX-NMLZ EQU.NE íŋ-ga=la. EQU.PER-PQ=HON 'Making compositions in our language like that, you teachers have done so much work, haven't you?' (RD BLA 9)
- (6.9) $\mathfrak{W}: \mathfrak{Y}: \mathsf{x} \in \mathfrak{F} \circ \mathfrak{F$

The plurality of lenge2 can be made explicit by adding the plural marker =*tsu*, lenge2=*tsu* 'you (pl.)':

(6.10) ขุลสามเฉสา สูสาฐสาสู่ " วุภา คิลาวารี "ผิขาสา" *séŋ-k^hɛ̃: lɛŋgɛ:=tsu=lo tacidile?*. listen.HON-NMLZ PRN.HON=PL=DAT greeting 'Greetings to you, listeners.' (KT animal story)

Plurality may also be made explicit through modifying words, e.g. $lenge? t^hamtee?$ 'all of you', $lenge? t^hamtee? k'ompu/k'\tilde{a}:pu$ 'all of you', lenge? pimpu 'the two of you'. The ordinary level k'utea? may be combined with lenge? into the honorific second person plural k'utea? lenge?:

(6.11) สูาธสา สุสาฐสา สมสาธรา ทัศารา มญิสารณิ รารฤาสา สุรา
 k'utca? [enge? thamtce? k'õ:pu khem-bø: daku be?.
 you PRN.HON all entire know.HON-2INF.GEN owner EQU.NE
 'All of you are possessors of (this) knowledge.'(NT BLA7)

The third person singular ordinary pronouns are $k^h u$ for males and $m \dot{u}/m \dot{o}$ for females:

(6.13) بي يَجْ المَرامَةُ المَرَامَةُ المَرَامَةُ المَرْجَاحِ اللهُ المُحْلَمَةُ اللهُ المُحْلَمَةُ مُحْلَمَةُ المُحْلَمَةُ مُحْلَمَةُ المُحْلَمَةُ المُحْلَمَةُ المُحْلَمَةُ مُحْلَمُ مُحْلَمَةُ مَا مُحْلَمُ المُحْلَمَةُ مَا مُحْلَمَةُ مُحْلَمَةُ مَا مُحْلَمَةُ مُعْمَاتُ مُحْلَمَةُ مُحْلَمَةُ المُحْلَمُ الْحُمَلَ مُحْلَمَةُ مَا مُحْلَمُ مُعْتَلَمَةُ مُعْتَاحُ مُحْلَمَةُ مُحْلَمَةُ مُحْلَمُ مُعْلَمَةُ مُعْتَاحُ مُحْل

The honorific for both of them is the gender-neutral $k^h \tilde{o}$: (see 6.14), which also functions as a third person plural pronoun (see 6.15). Often the plural $k^h \tilde{o}$: is supplemented by the plural =tsu for disambiguation, as in (6.16).

(6.14)beroteana gjagar=le p' g = lot'ã:pu ín-(n)o tc'øm-bø: gã: long.agoEQU-COND PN India=ABL Tibet=DAT come.HON-2INF time $k^h \tilde{o} = to$ lúŋ dup-ti tc'øm-bo=lo. 3sg.hon=cemph air achieve-NF come.HON-2INF=REP 'If it's the ancient times (we consider), when Berocana came from India to Tibet, he came through air, it is said.' (KL BB discussion)

(6.15) جَرَجٌ جَرَحَّسَ آمَجَ مَنْ عَلَيْ عَلَى عَلَى اللَّهِ عَلَى اللَّهُ عَلَى الْحَاسَ عَلَى اللَّهُ عَلَى الْحَلَى اللَّهُ عَلَى اللَّالَ عَلَى الْحَلْ الْعَلَى اللَّهُ عَلَى اللَّهُ عَلَى اللَّهُ عَلَى اللَّهُ عَلَى اللَّهُ عَلَى اللَّهُ عَلَى الْحَلَى الْحَلَى عَلَى الْحَلَى الْحَلَى الْحَلَى الْحَلَى الْحَلَى الْحَلَى الْحَلَى الْحَلَى الْحَ الْعَلَى اللَّهُ عَلَى اللَّهُ عَلَى اللَّهُ عَلَى اللَّهُ عَلَى اللَّهُ عَلَى اللَّهُ عَلَى الْحَلَى عَلَى اللَّالَةُ عَلَى الْحَلَى الْحَلَى الْحَلَى الْحَلَى الْحَلَى اللَّهُ عَلَى الْحَلَى الْحَلْ الْعَلَى الْحَلَى اللَّالَةُ عَلَى اللَّالِي عَلَى الْحَلَى عَلَى اللَّهُ عَلَى الْحَلَى الْ

(6.16) אָרָיאָיאָי אָיזיאי אָדָאַיאָן אָדָאַיאָן אָדָן **k'ô**:=tsu=lo nepali kjap-ce? dzampu be?. 3PL=PL=DAT Nepali speak-INF easy EQU.NE 'It is easier (lit. easy) for them to speak Nepali.' (RL interview)

In addition to the plural marker =tsu, another way to make plurality of $k^h \tilde{o}$: explicit is to add a modifier to the pronoun, e.g. $k^h \tilde{o}$: t^hamtee 'they all', $k^h \tilde{o}$: $k^h \tilde{e}$: le 'they all', $k^h \tilde{o}$: nim(pu) 'the two of them', $k^h \tilde{o}$: simpo 'the three of them'.

As noted above, the pronoun lenge2 may have either second person singular or plural meaning. In addition, lenge2 is used in honorific references to the third person plural, following a noun or a pronoun with third person referent:

- (6.17) היחאי פּיפּיד החתי שֿאי הר האָזי דָי אָזיאַי דָי פָזאָשיישֿ הישאָרי דָי פָזאָשיישֿ שּוּאָאיבי דּי פֿיז $\eta \dot{a}:=gi$ $c\dot{u}\cdot zo-u$ $kam\phi=di$ $k^{h}\tilde{o}:$ lenge:=lo t'a 1SG=AGT say.HUM-RDP-2INF discussion=DEMPH 3PL PRN.HON=DAT now $t^{h}up\tilde{o}:$ $s\dot{o}:-patsene$ benefit.HON offer-COND 'Now if what I say offers some benefit to them...' (CY interview)
- (6.18) দান্তরি' দ্লামা স্কুর' ক্রুমা স্কু' র্মাবা' স্ট্রী' দ্বদ্ভীবা দ্রী' ব্রিজা ঝাযবা' শ্রদ্ব' থেযাব্যা nàtei **lám lɛŋgɛ**:=tsu lòkti índzi mi-cé:-kʰɛn bɛ=la. 1PL.GEN lama PRN.HUM=PL again English NEG-know-NMLZ EQU.NE=HON 'Our lamas again do not know English.' (YR canteen video)

In addition to independent uses, personal pronouns may be used as appositional modifiers of nouns, see §4.1.2.4.1.

6.2 Reflexives and reciprocals

It was shown above that $r\tilde{a}$: functions as a mid-level second person personal pronoun. Literally $r\tilde{a}$: means 'self' and it is used as an independent reflexive pronoun, as in (6.19), and as a reflexive enclitic $=r\tilde{a}$:/=ra on personal pronouns, as in (6.20-23). In spoken language the dependent reflexive form $=r\tilde{a}$: tends to be shortened and denasalized to =ra.

(6.19)गूभाके द्वारा दर्दा रत्यों झूट दर्दा दर्यों हेंद्र भाषा देषा दर्या खायत छटा $k'\varepsilon:t\varepsilon^{h}ita=di$ $k\epsilon = di$ nénle cé: **ran**=gi *g0* important=DEMPH own=GEN language=DEMPH beginning first know go:-k^hen bε?. be.needed-NMLZ EQU.NE 'The important (thing is) that one has to know one's own language at first.' (KL BLA 12) רמי קשי מקריאן מצי מקי היאריאמי שטישדיעי אאן (6.20) $dzo^{243} = di$ thaktee:-po ĩ. t'ep di=gi nà=**ran**=gi ná: book this=GEN price=DEMPH 1SG=REFL=AGT decide-2INF I.AGT EOU.PER 'I decided the price for that book by myself.' (KL BLA 12) (6.21) שיתקי איקאראי דרי איאארישקו ódi mímã: $k^h \tilde{o}$:= $r\tilde{a}$: ma-cé:-b-be?. 3PL=REFL NEG-know-2INF-EQU.NE that mass(es) 'The masses themselves didn't know that.' (KL BLA 12) (6.22)केंश क्रुन कें रू रेगा राग राय $tc^h \phi ki? m \hat{u} = r \tilde{a}$ tciku mè:=na.

PN 3SGF=REFL only bed=LOC 'Only Choki herself (is) in bed.' (Richhi 4)

²⁴³ Also $g\tilde{o}$: (WD $\tilde{\eta}$) and $rin/r\tilde{i}$: (WD $\tilde{\tau}$) are used for referring to 'price'. All speakers do not recognize the word *dzo*.
(6.23) $\neg \check{a} \check{a} \neg \check{a} \check{a} \neg \check{g} \check{a}$ now 2SG.L=REFL=AGT=AEMPH good do-NF thought send 'Now think through it yourself carefully.' (Nga'i 'gan 14)

Note that in (6.23), the reflexive is followed by the homophonous anaphoric emphatic $=r\tilde{a}$:/*ra* (which is a further grammaticalization of the reflexive).

The form $r\tilde{a}:m\tilde{\epsilon}:/r\tilde{o}:m\tilde{\epsilon}: T_{a}$ 'oneself' is also used, at least in Martam (East Sikkim):

- (6.24) $\pi \pi$ and \mathfrak{g} at \mathfrak{g} a

Another reflexive form, which may be characterized as distributive because it encompasses all members in a group, is *rã:rã: so:so:* 'each one themselves':

(6.26)	८. रुगा. २८. २८. ब्र. ब्र. जू. मेंब. ट्रीब. वट. ट्र.							
	ŋàtca?	rã:rã:soso=lo	køn	tøn	g0:-68	bo?		
	1pl	each.oneself=DAT	blame	show	be.needed-INF	F EQU.NE.Q		
	'Are we	e to blame each one our	selves?' ((KN, C	CY interview)			

The reflexive $=r\tilde{a}$:/=ra can also attach to other forms than personal pronouns. Then it functions as an anaphoric emphatic, see §16.1.1.

Three reciprocal pronominals occur in my data, see Table 6.2 (see Nedyalkov [2007: 12] for the basic difference between pronominal and verbal reciprocals).

 Table 6.2. Reciprocal pronouns

Form		Gloss
tci:=ki tci:(=lo)	यछिया'यीद्य' यछिया'(व्यॅ')	'one to another' (lit. 'one to one')
tei:=ki zen(=lo)	যন্তিযা'যীম' যান্বব'(র্নি')	'one to another' (used in Richhi-novel) ²⁴⁴
p ^h ɛntsỹː	यव र्ह्व	'each other'

²⁴⁴ The novel Richhi is the only source where I have come across the construction $t\varepsilon i:=ki \ z\varepsilon n(=lo)$ instead of $t\varepsilon i:=ki \ t\varepsilon i:(=lo)$. The reason may be either that the deviant construction is used in the novel's author's dialect area or that the construction is influenced by Nepali and/or English which both have reciprocal constructions analogous to $t\varepsilon i:=ki \ z\varepsilon n(=lo)$, i.e. Nepali *ek arkaa-lai* [one another=DAT] 'to one another', English *to one another*.

The first is tei:=ki tei:(=lo) [one=AGT one(=DAT)] 'one to another, (to) each other' (lit. 'one to one'), see (6.27)

(6.27)विंद गविकार्ये न क्रिंपरी के के के न न न के के के न न न के के न न महिया मेरा के के न न महिया मेर के के के के क ím-batsene t'a $k^h \tilde{o}$: n(m-po)*ódi* p*imts*^h ε t'a tε ťa, nàtci now that date 3PL two-COL then now 1PL.GEN EQU-COND now tci:=ki zats^hã: t^hem-bo he? tci? ťa. one married.couple become-2INF EQU.NE now one=AGT 'The two of them, on that day, now when it comes to our (tradition), become each other's spouses.' (SGD wedding customs)

The second one is $t \in iki \ z \in n = lo$ [one=AGT other=DAT] 'one (to) another', which occurs in the novel Richhi instead of the first construction. I have not come across $t \in iki \ z \in n = lo$ elsewhere in my data.

(6.28) מוּצּׁמוּ מוּמָק'^{גֿע} שִׁמְיֹשׁי שַׁמְיָשָׁאַ מוּמָדָי שַׁמָיָשָׁאַ מוּמָדָי שַׁמָי שַׁמָי שַּׁמָי **נוּוּ:=ki zɛn=lo** p^hɛmbo p'ja-ɛɛ? giwø jó? גָ:. one=AGT other=DAT help do-INF merit.GEN word EQU.PER 'Helping one another is a meritorious act.' (Richhi 5)

6.3 Indefinite reference

The discussion on indefinite reference is divided into indefinite pronouns (§6.3.1) and indefinite expressions formed by question words (§6.3.2).

6.3.1 Indefinite pronouns

Indefinite pronouns are words that refer to people, objects or places without exactly specifying the referent, see Table 6.3. Indefinite pronouns also function as quantifiers which modify nouns, see §4.1.3.3.

 Table 6.3. Indefinite pronouns

tʰamtɕɛʔ (kˈãːpu)	म्रियया रुद्र (यद्र सु.)	'all, everyone'
kʰɛːlɛ (kˈãːpu)	यिंग्वें (यद्राः	'all, everyone'
tɕʰaːlɛ (kˈãːpu)	ক্র'ঝের্ম' (বার্ব'র্ন্ডু')	'all, everyone'
dzaŋki	<u>بر</u> الم	'all, everyone' (Lachung)
k'oːm	यॉन्ट्य.	'all, everyone' (Lachung)
màŋtɕʰiɕo?	यम्.कृ.चूय.	'most' (includes the adjectival superlative ending -co?)
màŋtɕʰiţa?	ञन्दः क्वे:न्द्रयाह्य:	'most' (includes the adjectival ending - <i>ta?</i>)
khace?	দ্রে.রিশ্র.	'some(one)'
làri?	artar.	'some(one)'
rere	रेंग्रें	'each one'
ka:kut£i?	गा'गु'र्च्चिया'	'a few, some'
tci:pi:	যান্তিযা'যান্তিম'	'a couple (of), a few' (lit. 'one-two')
ripi (ripi)	रे'यानिह्य (रे'यानिह्य)	'a couple (of), a few' (lit. 'one-two')
làla…làla	વાવા વાવા	'someothers'
<i>ri, -ri</i> (also <i>rε</i>)	<i>रे</i>	'one, each'

As shown by Table 6.3, several forms correspond to meaning 'all, everyone' and 'a bit'. The variants represent some dialectal variation. The most frequent items for 'all, everyone' are *t*^hamtee? and $k^h \varepsilon : l\varepsilon$, the first of which seems to be more frequent in West Sikkim and the latter in East and North Sikkim, although both are readily understood all over Sikkim. The form *dzayki* and *k'o:m* are from Lachung.

The independent uses of indefinite pronouns, except for $te^{ha:le}$, k'o:m and $mante^{hita?}$ (of which I have only noun-modifying examples), are below illustrated in the same order that they occur in Table 6.3.

- (6.30) สมเขาอกา เด้า ณริสา องส์ เกมเมา กรกา รักษ์ จำสา *thamtce=ki ódem=tci? nó:sam tõ:-botce sé-na* all=AGT like.that=INDF thought send-COND say-COND 'If all think like that...' (DR discussion with KL)

Note that in (6.32) *màŋte^hieo?* is followed by an appositional, explanatory noun. As a modifier *màŋte^hieo?* would typically follow the noun.

(6.32) مَحَرَّفَ عَلَى مَحَرَّمَ مَحَرَّمَ مَحَرَّمَ مَحَرَّمَ مَحَرَّمَ مَحَرَّفَ مَحَرَّفًى مَعْرَبَةً مَعْرَبَ مَحْرَبًا مَعْنَقَعَ مَعْرَبًا مَعْنَقَ مَعْنَعَ مَعْنَقَ مَعْنَقُ مَعْنَا مَعْنَق مُعْنَقَ مَعْنَقَ مَعْنَقَ مَعْنَا مَعْنَقَ مَعْنَا مَعْنَقُ مَعْنَق مَعْنَقُ مَعْنَقُ مَعْنَقُ مَعْنَا مَعْنَقُ مَعْنَا مَعْنَا مَعْنَا مَعْنَقَ مَعْنَا مَعْنَقَ مَعْنَا مَعْ مَعْنَا مَعْنَا مَعْنَا مَعْنَقَ مَعْنَا مَعْنَا مَعْنَقَ مَعْنَا مَعْنَا مَعْنَا مَعْنَا مَعْنَا مُعْنَ مَعْنَا مُعْنَا مَعْنَا مُعْنَا مُعْنَا مَعْنَا مَعْنَا مُعْنَا مَعْنَا مَعْنَ مَعْنَا مَنَ مَعْنَ مَعْنَا مَعْنَ مَعْنَا مَعْنَا

 $^{^{245}}$ This written form given by consultant KUN is surprising in that it suggests pronunciation as *ts'aŋki* rather than *dzaŋki*.

- (6.33) שֿׁיַמּלְאּישׁר וּשִיּשְּליישָאי פּיַשּאָזיי פּיַשאָר אַישּאי *odem=tci=jã: גּױּמבּ=ki cù-zouke=la.* like.that=INDF=even some=AGT say.HUM-PROG.APH=HON 'Some (people) were even saying like that.' (CY interview)
- (6.34) مح : عمر ، فا حرج : عمر الله المعادية المرجم : (thikadar) والمعارفة المرجم ا t'a **làri?** tciku=di gjømpa $kat \varepsilon^h \phi = na$ zu:-ce=di now some only=DEMPH monastery recitation=LOC sit.HON-INF=DEMPH nấ:-bo kjap-ti t'a ódi=lo=di átsi=tɕi? gõ:jã: dа do-NF now that=DAT=DEMPH a.bit=INDF excuse do.HON-2INF be.similar *bε*?, ódi gã:=lo *t*^{*h*}*ikadar=tsu=gi*. EQU.NE that time=DAT thikadar=PL=AGT 'At that time, only some, because they were living in monasteries doing recitation, now it seems exemption (from labour duty) was given to them by the thikadar-rulers.' (CY interview)
- (6.35) حج جے تھ مَع تھ مَع تھ مَع تھ ج حَبَ مَع ج مِعْلَمَ اللہ مَعْلَمَ تَعْلَمُ مَعْلَمَ تَعْلَمُ مَعْلَمُ مُعْلَمُ مُعْلًا مُعْلَمُ مُعْلًا مُ وعالم معالم الما مُعالم معالم محالهم مع ما مع محالم معالم محالم معالم معالم محالم معالم محالهم معالم معالم معال
- (6.36) ﷺ جِمْعَ مَانَعْ عَلَمَ عَلَي عَلَى عَلَي عَل (If (you) pluck five-six, a few (oranges), it will suffice.' (KN e)
- (6.37) = المعارية: تَعَامَ اللَّهُ: المَّحَامَ اللَّهُ: المَحَامَ اللَّهُ: المُحَامَ المُحَامَعَ المُحَامَ المُحَام المُحَامِ المُحَامِ المُحَامِ المُحَامِ المُحَامِ المُحَامِ المُحَامِ المُحَامِ مُحَامَعَ مَحَامَ المُحَامِعَامِ المُحَامِ المُحَامِ مُحَامَ المُحَامِ مُحَامَعَ مَحَامَ المُحَامِعَامِ المُحَامِ مُحَامَ المُحَامِ مُحَامَ مُحَامَ مُحَامَ مُحَامَعَ مُحَامَ مُحَامَ مُحَامَعَ مَحَامَ مُحَامَعَ مُحَامَ مُحَامَ مُحَامَ مُحَامَ مُحَامِ مُحَامَعَ مُحَامَ مُحَامَ مُحَامًا مُحَامًا مُحَامَ مُحَامِ مُحَامَ مُحَامَ مُحَامًا مُحَامَ مُحَامَ مُحَامَ مُحَامَ مُحَامًا مُحَامَ مُحَامًا مُحَامَ مُحَامَ مُحَامَ مُحَامًا مُحَامًا مُحَامًا مُحَامَ مُحَامِ مُحَامَ مُحَامَ مُحَامَ مُحَامَ مُحَامِ مُحَامَ مُحَامَ مُحَامًا مُحَامَ مُحَامَ مُحَامَ مُحَامَ مُحَامَ مُحَامَ مُحَامَ مُحَامِ مُحَامَ مُحَامَ مُحَامًا مُحَامًا مُحَامًا مُحَامَ مُحَامَ مُحَامَ مُحَامًا مُحَامًا مُحَامًا مُحَامَ مُحَامَ مُحَامَ مُحَامَ مُحَامًا مُحَامًا مُحَامًا مُحَامَ مُحَامَ مُحَامُ مَحَامُ مُحَامَ مُحَامَ مُحَامَ مُحَامُ مُحَامُ مُحَامَ مُ
- (6.39) מימי מקאי מקי, מימי מקי מקן *làla rɛ: làp, làla bja làp* some rice.kernel say some rice.kernel say
 'Some call it (=rice kernel) [rɛ], others call it [bja].' (PL interview)

(6.40)gõ:=di=lo sàm ódi=gi tiru? gja-**ri** gja-**ri**, food that=GEN price=DEMPH=DAT rupee hundred-each hundred-each sàm **ri**=lo tiru? gja-**ri** *bε*?. food each=DAT rupee hundred-each EQU.NE 'The price of that food was hundred hundred each, a hundred rupees per one (portion of) food.' (DB trip story)

A phonologically reduced form of the question k'an inam (reading-style pronunciation)/k'an paim (spoken pronunciation) 'What is it?' is used as a frequent conversation filler, which signals that the speaker does not remember a word, see (6.41). Therefore the form can be considered an indefinite pronoun. Different stages of reduction are attested in spoken language: k'an inam > k'an $paim > k'aj \epsilon m > k'a \epsilon m > k' \epsilon m$.

b) हे' तुंगीका क्रेंग्त गताक्षुका ईकाक्षेगी क्रेंग्दे गात ख्रुका मेंचाक क्राका पदी क्षारे मेंचाक जाक

te $k^{h}u=gi$ óna k'ajem p'ja-tiki, ódi k'ajemso 3SGM=AGT there what.is.it do-NF that what.is.it $t^{h}op-sa-le=di$, $ta:ri t^{h}op-sa=le$ receive-CMPL=ABL=DEMPH axe receive-CMPL=ABL 'When he had done that whatever, received that whatever, received the axe...' (JDF axe story)

6.3.2 Indefinite reference with question words

Meanings equivalent to such English indefinite expressions as whoever, anyone, wherever, anywhere, whatever and anything are formed with the help of question words. In affirmative clauses, conveying meanings of the type 'whoever', 'wherever' and 'whatever', the questions word is supplemented with a concessive equative form \tilde{i} -run 'EQU-CONC', see Table 6.4. Meanings which are semantically the polar opposites of meanings such as 'someone', 'somewhere', somehow' (i.e. not anyone/no one, not anywhere/nowhere, not anyhow) are formed by a combination of a question word, an optional clitic $=j\tilde{a}$: 'even' and an obligatory negated verb, see Table 6.5.

²⁴⁶ According to consultant KT, this word should be sàjo ज्ञरूपायॉ.

		lee with question words
k'an ấ:ruŋ ²⁴⁷	ਧੀਕ੍ਰ' ਲੀਕ੍ਰ'ਤ੍ਰ≍'	'whatever'
k'ar(ɛ) î̈́ːruŋ	ग'(रे') क्षेव'रुप्'	'whatever'
k'adi î:ruŋ	ਧਾਕਟੀ ਐਰਾਤਸ	'whichever, whoever'
ka îːruŋ	गा' क्षेत्र'उ्र-	'whoever'
nàm(lo) ĩ̃:ruŋ	ववा [.] (यॅं') क्षेत्र'रुप्त'	'whenever'
k`ana î:ruŋ	गान अनिउट	'wherever'
k`atem ĩ̃:ruŋ	ग'हेवा क्षेत्र'उट्-'	'what ever kind'
k'ate p'ja(ti ī́ː)ruŋ	ગ' કે' 5ાંચ'(ફે' બેવ') રુદ્દ'	'however'
k'ambja í̇́ːruŋ	ਧਕਿ'ਉੱਕਾ ਐਕ'ਤੁਸ	'for whatever reason ("whyever")'
k'adzø? î:ruŋ	गार्केन्' क्षेत्र'रुन्'	'however many'

Table 6.4. Affirmative indefinite reference with question words

Table 6.5. Negated indefinite reference with question words

$k'an(=j\tilde{a})$	শ্বর'(অন্ন')	'anything (+neg)'
k'amo(=jã:)	गव'र्बे'(धट्र')	'anything (+neg)'
k'adi(=jã:)	٩	'any(one) (+neg)'
$ka(=j\tilde{a})$	শা.(লেশ.)	'anyone (+neg)'
nàm(lo)(=jã:)	বৃষা(র্মি')(৸৸')	'ever (+neg)'
k'ana(=jã:)	শা'বৃ'(ਘদ্ম')	'anywhere (+neg)'
$k'ate(m/p)(=j\tilde{a})$	या'झेवा'(धट्र')	'any kind (+neg)'
k'ate p'jati=jã:	য়ৼ৾৾ঀৢ৾৾৾য়ৢ৾৾য়৾ৼ৾৾ঀৢ৾৽৸ৼ৾৾৽	'anyhow (+neg)'
k'an p'jati=jã: ²⁴⁸	য়ঀৢ৾৾৽ঀ৾ৢয়৾৾৽ৼ৾৾ৢ	'for any reason (+neg)'
k'adzø=(jã:)	শ'র্ক্রন্'(অন্-')	'any number of (+neg)'
k'andɛː(=jãː)	षावु'तर्द्रे'(धट्र')	'anything (+neg)'
$k'ar\varepsilon(=j\tilde{a}:)$	<u>য</u> া' र े'(ঋ <u>र</u> ,)	'anything (+neg)'

As suggested by Table 6.4 and Table 6.5, the concessive form \hat{i} :ruŋ is obligatory in the affirmative constructions, whereas the formative $=j\hat{a}$: is not obligatory in the negated constructions. The last two words in Table 6.5 do not have independent interrogative uses in affirmative clauses but only occur in negated clauses, although *k'are* also participates in the affirmative construction *k'are* \hat{i} :ruŋ 'whatever'. Some affirmative forms are illustrated in (6.42-45). Note that morphemes may intervene between the question word and the concessive equative, as exemplified by the anaphoric emphatic in (6.43) and (6.44).

(6.42) מאַן אָז אָז אָז אָז אָז אָז געש: ג'מח ז:ruy=tei?

k'an i̇́:ruŋ=tci? t^hɛm-batcɛnɛ what EQU-CONC=INDF happen-COND 'whatever (might) happen(s)' (PED life story)

²⁴⁷ This form also occurs as $\neg \neg \neg \neg \neg \neg \neg$ with pronunciation [k' \tilde{a} :ru η].

²⁴⁸ The shorter form k'ambja=j \tilde{a} : was deemed infelicitous by consultant KN. The clitic =j \tilde{a} : is obligatory.

- (6.44) جِسَّانَ هُمَ تَنْ سَابَعُمَّاتِمَ نَمَا عَجَدَ *t'ako cèmpo k'atem=ra tٌ:-ruŋ* noble fool/poor how=AEMPH EQU-CONC 'noble or fool/poor, whatever (you) are like' (KT life story)
- (6.45) $\exists \forall x \in \mathfrak{F}^{249} q \in \mathfrak{F}^{249} q \in \mathfrak{F}^{249} \mathfrak{F}^{24$

The equative may be dropped from the construction, making the result more lexeme-like than the full form:

(6.46)nénts^hẽ: k'oːm, átcu, ái. ádzo, p^hamiŋ, elder.brother elder.sister grandfather father's.relatives relatives all k'an-run jà-patee p'ja-run khe:le $k' \tilde{o} : pu = gi ta : -c \varepsilon$ $b\epsilon$?. what-CONC EX-COND do-CONC all full=AGT append-INF EOU.NE 'All relatives, elder brother, elder sister, grandfather, middle-man whatever (relative) is there, each and every one offer (a silk scarf).' (LA intro to Lachung)

The negated clauses, both with or without the clitic $=j\tilde{a}$: are illustrated in (6.47-54).

(6.47) र्जुमान्नमा गुत्रायमा मेरायमा

p'ja-tap $k'an=j\tilde{a}$: $m\dot{e}$:- pa^{250} do-means what=even NEG.EX-CIRC 'there being no chance of doing anything' (Richhi 159)

²⁴⁹ WD for this word is adopted from Dokhangba (2001). Other WD forms I have been suggested are $\Re \Re mi-ci$, $\Re \Re mi-tsus$ and $\Re \Re mi-tsus$ and $\Re \Re mi-tsus$. The form from Dokhangba (2001) is chosen for use here because it corresponds most clearly with the spoken output $m \chi tsy$?.

²⁵⁰ *k*'and ε : or kar ε could also be used here instead of *k*'an=j \tilde{a} : (KUN).

- (6.48) অব্যান্থ দুদাই আব nám=lo huŋpo k'ana mè?.²⁵¹ sky=DAT cloud where NEG.EX.PER 'In the sky, there aren't clouds anywhere.' (Richhi 151)
- (6.49) לישה ישר שא ימלי שישה שא ימלי שיש אימלי שישה ליונג *k'adi=jã: k'andɛ: làp mi-tsʰu?.* but which.one=even anything say NEG-be.able.to 'But anyone was not able to say anything.' (Richhi 53)
- (6.51) a) ฉุรุสา ข้าหาง ราชญา สุสานุรา สำษัญางรา]
 dɛm k'okap ŋàtɛa? nàm-jã: mi-t^ho(p)=pɛ. such opportunity 1PL when-even NEG-receive=EQU.NE
 'We will never get such an opportunity.' (KN e)
 - b) רְישׁמי שִׁמּישִׁ שִׁמּישִׁ שִׁמּישִׁ שִׁמּישִׁ בְּישׁמי מָמימִישִר מָמימִישָר מַמּיַמִישָר מַמּיַמַישָר מַמּיַמַישָר מַמּיַמַישָר מַמּיַמַישָר מַמּיַקרין *nàm=lo=jã*: *p*ham=tsu p'otso=lo màlɛp t'ã: kjota? nàm=lo=jã: *p*L parent=PL child=DAT bad and crooked when=DAT=even nó:sam mi-tã:.
 thought NEG-send
 'We parent do not ever plan bad and crooked (things) for (their) children.' (Richhi 154)

mi **k'atem=jã**: nim tei:=na siliguri lep mi-ts^hu?. human what.kind=even day one=LOC TPN arrive NEG-be.able.to 'A human, whatever kind (s)he is, cannot arrive in Siliguri (by walking) in one day.' (KN e)

(6.53) בה׳ דר ש׳ ש׳ ש׳ ש׳ לק׳ שר׳ ארן

pè:=tsa: ts^halum k'adzø=jã: mè?. 1SG.GEN=at orange how.many=even NEG.EX.PER 'I do not not have any number of oranges.' (KN e)

²⁵¹ The mere question word in conjunction with a negated verb here expresses the meaning 'anywhere'. However, as suggested by consultant KUN's desire to correct this clause by adding the clitic $=j\tilde{a}$; the fuller form $k'ana=j\tilde{a}$: is probably more frequent.

²⁵² k'an=j \tilde{a} : and k'an \tilde{i} :ruŋ could here replace k'an $d\varepsilon$:=j \tilde{a} : (KUN).

(6.54) جَرَّ מَجَ عَلَى اللَّهُ عَلَى اللَّالَ عَلَى اللَّهُ عَلَى اللَّهُ عَلَى اللَّهُ عَلَى اللَّهُ عَلَى اللَّهُ عَلَى الْحَلَى اللَّهُ عَلَى اللَّالَةُ عَلَى اللَّهُ عَلَى اللَّهُ عَلَى اللَّهُ عَلَى اللَّهُ عَلَى اللَّهُ عَلَى اللَّهُ عَلَى اللَّ

The general interrogative k'a: 'what, where, why' is used as a component of the following expressions which appear to have lexicalized:

(6.55)	k`aː-jø?	۳٠٤٠٦	'whatever (there is)' (lit. what-EX.PER)
	k'a:-t ^h op	শ'র্শ্রন'	'wherever' (lit. where-find)
	k'a-sa-k'a=la) গাঁমাগার্শি	'wherever' (lit. what-ground-where=DAT)

These three words are illustrated below. Note that in the written Denjongke sources these expressions are written as one word.

- (6.56) حَقْقَا بَنَ اللَّهُ عَلَيْهُ اللَّهُ اللَّهُ عَلَيْهُ اللَّهُ عَلَيْ اللَّهُ عَلَيْهُ اللَّهُ اللَّهُ عَلَيْعَتَى اللَّهُ اللَّعَلَيْنَ اللَّالِي اللَّهُ عَلَيْنَا اللَّهُ عَلَيْنَا اللَّهُ عَلَيْهُ اللَّهُ عَلَيْ اللَّهُ عَلَيْنَا اللَّهُ عَلَيْنَا اللَّهُ عَلَيْنَا اللَّهُ عَلَيْهُ اللَّهُ عَلَيْهُ اللَّهُ عَلَيْ اللَّهُ اللَّهُ اللَّهُ عَلَيْنَا اللَّهُ عَلَيْنَا اللَّهُ عَلَيْنَا اللَّهُ عَلَيْنَا اللَّالِي اللَّهُ عَلَيْ
- (6.57) بِهْمَا الْأَلَّمَةُ: गَا لَقْتَمَ مَا لَعْتَى الْحَالَةُ لَمْ الْحَالَةُ لَحَالَةًا لَحَالَةُ مَا لَحَالَةًا لَحَالَةًا لَحَالَةً مَا لَحَالَةً مَا لَحَالَةًا لَحَالَةًا لَحَالَةً مَا لَحَالَةً مَا لَحَالَةًا لَحَالَةً مَا الْحَالَةُ مَا لَحَالًا لَحَالَةً مَا إِلَيْحَالَةًا لَحَالًا لَحَالًا لَحَالَةًا لَحَالَةً مَا لَحَالَةً مَا لَحَالًا لَحَالَةً مَا مَا لَكُلُكُمَا لَحَالًا لَحَالَةً مَا لَحَالَةً مَا لَحَالًا لَحَالَةً مَا لَحَالَةً مَا لَحَالًا لَحَالَةً مَا لَحَالَةً مَا لَحَالَةً مَا لَحَالًا لَحَالَةً مَا لَحَالَةً مَا لَحَالَةً مَا لَحَالًا لَحَالَةً مَا لَحَالَةً مَا لَحَالَةً مَا لَحَالَةًا لَحَالًا لَحَالَةً مَا لَحَالَةًا مَا لَحَالَةًا لَحَالًا مَا لَحَالَةً مَا لَحَالًا لَحَالَةً مَا لَحَالًا لَحَالَةً مَا حَالَةً مَا لَحَالَةً مَا لَحَالَةً مَا حَالَةًا مَا لَحَالًا مَا لَحَالًا مَا لَحَالَةً مَا حَالَةً مَا حَالَةً مَا حَالًا لَحَالَةً مَا لَحَالَةً مَا لَحَالَةً مَا مَا حَالَةًا مَا مَا لَحَالَةً مَا حَالَيْ مَا مَا حَالَةً مَا مَا حَالَةً مَا مَا مَا حَالَةًا مَا حَالَةًا مَا مَا حَالَةًا مَا حَالًا مَا حَالَةًا مَا حَالَةً مَا حَالَةًا مَا حَالَةً مَا حَالَةً مَا حَالَةًا مَا حَالَةً مَا حَالًا مَا مَا حَالَةً مَا مَا مَا حَالًا مَا حَالَةً مَا حَالًا مَاحَالَةًا مَالْحَاحَاتُ مَا حَالًا مَالْحَالَةً مَا
- (6.58) শ'শ'শ'শ নেয়ু'র্দ্র'ন্ন' k'asak'alo gju-ruŋ=ra wherever go-CONC=AEMPH 'Wherever (we) go...' (song lyrics)

6.4 Demonstratives

Demonstratives are deictic words which define a person, object or location in terms of its spatial relationship to the speaker. Demonstratives may be pronouns, pro-adjectives and pro-adverbs. Denjongke demonstratives occur both independently and as noun modifiers. The roots from which demonstrative expressions are formed are listed in Table 6.6. The roots that usually do not occur independently (except as homonyms having a different meaning) are marked with a hyphen. For instance, the distal marker \dot{o} - does not usually occur independently but has to be complemented by another element, e.g. $\dot{o}di$ 'that', $\dot{o}na$ 'there', $\dot{o}k^ha$ 'there'. Occasionally, and exclusively in spoken language, \dot{o} - occurs by itself or with the plural marker =tsu as $\dot{o}=tsu$ 'they', which is an alternative for the more frequent $\dot{o}di=tsu$ 'they'.

1 abic 0.0.	Demonsu	
di	مثر.	proximal, 'this'
do-	पर्ने.	emphatic proximal, 'this right here'
<i>ó</i> -	র্জা	distal, 'that'
nà:	व.	'here'
p ^h ou, p ^h i-	र्चतुः	'over there'
jòu, jì-	ર્વેાતુ, ધે	'up (there)'
mòu, mì-	ર્વેતુ', ત્રે'	'down (there)'
p ^h a(:)	শ(শ)	'further, thither, on the other side'
$ts^h u(:)$	र्ख्(रू)'	'closer, hither, on this side'
zen	गवतु.	'other'

 Table 6.6. Demonstrative roots

The difference between the proximal di and the emphatic proximal do- is that whereas both can be accompanied by pointing to an object in the speaker's proximity, the deictic force is stronger in the emphatic do- (i.e. the speaker is more likely to actually point at something). The more general proximal di has further grammaticalized into an emphatic particle that has lost its referential function, see §16.1.3. For a comment on the use of proximal =di as a definiteness marker, refer to §4.1.6.

More demonstratives may be derived from the roots of Table 6.6. Demonstrative pronouns are formed by combining a demonstrative root with the proximal *di*, e.g. *ódi* 'that'. Demonstrative proadverbs of location are formed by supplementing the roots by the locative case marker =*na* (probably deriving from *nà*: 'here'), dative-locative case marker =*lo* or the less productive locational suffix -*k*^h*a*, see Table 6.7. Reduplication (e.g. *p*^h*o*:*p*^h*ouna* 'way over there') functions as an ideophonic strategy to imply further distance. The list of locative expressions in Table 6.7 is not exhaustive but only presents the forms which I have come across in my present data.

Root		Derived object	Derived location
do-	emphatic proximal	dodi 'this right here'	<i>dodik^ha, dona</i> 'right here'
di	proximal	<i>di</i> 'this'	<i>dikha</i> 'here'
<i>ó</i> -	distal	<i>ódi</i> 'that'	<i>óna</i> , <i>ók^ha</i> 'there'
nà:	'here'	$n\dot{a}=di$ 'the one here'	<i>nàk^ha(lo)</i> 'here'
<i>р</i> ^{<i>h</i>} ои,	'over there'	$p^{h}ou=di, p^{h}idi$	p ^h ou=na, p ^h ouk ^h a, p ^h ok ^h a, p ^h ina, p ^h ik ^h a
p^hi -		'that over there'	'over there'
			<i>p</i> ^h o: <i>p</i> ^h ina, <i>p</i> ^h o: <i>p</i> ^h ou=na, <i>p</i> ^h o: <i>p</i> ^h ouk ^h a
			'way over there
jòu,	'up (there)'	jòdi, jìdi	<i>jìna, jòu=k^ha</i> 'up (there)'
jì-		'that up there'	<i>jòːjina, jòina</i> 'way up (there)'
mòu,	'down (there)'	mòdi, mìdi	<i>mòuna, mòuk^halo, mòu=lo, mìna</i> 'down
mì-		'that down there'	(there)'
			<i>mò:mouk^halo</i> 'way down (there)'
zen	'other'	$z \in n = di$ 'the other'	$z \epsilon n k^h a (= lo)$ 'in another place'
$p^{h}a(x)^{253}$	'further, thither,	$p^{h}a = di$ 'the one	$p^{h}a:k^{h}a$ 'on the other side, further'
	on the other side'	thither/further'	
$ts^h u(:)$	'closer, hither, on	$ts^h u = di$ 'the one	<i>ts^hu</i> : <i>k^ha</i> 'on this side, closer'
	this side'	hither/closer'	

Table 6.7. Derived demonstratives

Demonstratives occur as prenominal (6.59) and postnominal (6.60) noun modifiers but are also used independently (6.61). Whereas this section focuses on independent uses, prenominal and postnominal uses as noun modifiers are more fully discussed in §4.1.2.1 and §4.1.3.4 respectively.

- (6.59) $\hat{\beta}$ '= κ ' q' q= κ ' \hat{a} ' $\hat{\alpha}$ '
- (6.61) র্জানেন্ট ব্যান্য উদ্যোগ্য শ্বন্য
 ódi k'ɛ:tɛʰita? bɛ?.
 that important EQU.NE
 'That is important.' (KL BLA 12)

Some of the deictic forms from Table 6.7 are exemplified in (6.62-6.71).

²⁵³ According to consultant KN, forms derived from $p^{h}a$: and $ts^{h}u$: are marginal, not used by all Denjongke speaking communities.

- (6.62) 여국 여국 학풍·중· 영직·제 **dodi**=di potso=tsu pè:-sa. this.right.here=DEMPH child=PL sleep-place 'This right here (is) the children's sleeping place.' (PD altar room video)
- (6.63) أَن تَوْرَ اللَّهُ عَلَى اللَّهُ اللَّهُ عَلَى اللَّعَامَ عَلَى اللَّهُ عَلَى الْعَلَى الْحَالَى الْحَالَ عَلَى الْحَالَى الْحَالَى الْحَالَةُ عَلَى الْحَالَى الْحَالَى الْحَالَى الْحَالَى الْحَالَى الْحَالَى الْحَالَةُ عَلَى الْحَالَى الْحَالَةُ عَلَى الْحَالَى الْحَالَى الْحَالَ حَالًا عَلَى الْحَالَى الْحَالَى الْحَالَى الْحَالَى الْحَالَةُ عَلَى الْحَالَى الْحَالَى الْحَالَى الْحَالَى الْحَالَى الْحَالَى الْحَالَى الْحَالَى الْحَالَ عَلَى الْحَالَى الْحَالَى اللْحَالَ عَلَى الْحَالَى الْحَالَةُ عَلَى الْحَالَى ا
- (6.64) דֿידֿי אֿיןמימאזי, שָּלּשן מדְיוֹמימאזי משָׁאַיקדי rè-re **ókʰa=le**, tei? **dikʰa=le** gjom-da one-one there=ABL one here=ABL gather-CONJ 'when gathering one from there and one from here' (KT e)

Example (6.65) illustrates the locative demonstrative $n\dot{a}(k^halo)$ 'here'. Example (a) has bare $n\dot{a}$:, whereas in (b) and (c) the form is complemented by two additional locative elements to form $n\dot{a}k^halo$. Example (c) is interesting in that the deictic form $n\dot{a}k^halo$ allows a possessor as a genitive modifier, a construction that in English requires a noun such as *place* instead of the deictic *here* (*'in the old lady's here' > 'in the old lady's place').

(6.65) a) व. प्रयाग मेंगा जे

nà: bak có: ná here carry come TAG.ASR 'Bring (it) here, eh.' (PT kitchen discussion)

c) ה׳ ה׳ҳ̄ъ׳ आ‑ན་གི་ ན་ལ་མོ་ རང་ སོད་འི།
 ŋà t'ariŋ ána²⁵⁴=gi nàkʰa=lo=rã: døː-ni.
 1sG today old.lady=GEN here=DAT=AEMPH stay-3INF
 'Today I will indeed stay in the grandmother's (=your) place.' (rna-gsung 7)

In addition to taking a modifier, $n\dot{a}$: itself can function as a modifier meaning 'the one here', as shown in (6.66):

²⁵⁴ In kinship terms, *ána* refers to maternal grandparent's brother's wife, see §17.2.

(6.66) ગાંચ, ર્કેન્' વ' લેં' છે' તેં' બન જ રાન્ય સુભા દીવ' તેં karma, tɛʰøʔ [nà: ópi=lo] ja:=tsa:=sa: ke: p'in lo. here small.child=DAT up=by=until bring give TAG.Q PN 2sg.l 'Karma, you take this child here all the way up, okay?' (Richhi 40)

In (6.67), $p^{h}ou=di$ is used independently, whereas the shorter form $p^{h}idi$ occurs as a noun modifier. Because the demonstrative-emphatic can attach to both locative adverbs and nouns, it is not clear whether $p^{h}ou=di$ should be interpreted as referring to location ('over there') or the item located ('the one over there').

ર્ષેલુ વર્તુ વૈત્ત સંવર્તુ જો જારુ phou=di (6.67) είη **p^hidi=lo** áru one.over.there=DEMPH tree one.over.there=DAT peach(Nep.) 'That over there, the tree over there (is called) [aru].' (PD surroundings video) धे'व' तन्त्र' रोग्रराख्व' यहिया' तन्यागी (6.68)jìna dem símt $c\tilde{e}$:=tci? du-ke. like.that animal=INDF EX.SEN-IN up.there 'Up there, there is some type of an animal.' (UU Deer story) व'वेर' गॉर्व'वेर' हे' पॅरि'व' र'रुग' क्षॅ'र्स्रे' झर'... nàpĩ: zø:pĩ: te jòi=na (6.69) nàtea? ódi gã:... two.years.ago so up.there=LOC 1PL last.year that time 'Last year or the year before that, up there, at that time we...' (KNA kitchen discussion) (6.70)कें केंदेंग्यें न्वीव देंग् नूनः झावनः केंवाका गोकार्येः दनुवा वी gjompo t'ã: lạk^hã: só? ke:po *mò:mou=lo* $du = \epsilon o$. down.there=DAT monastery and shrine etc many EX.SEN=AT

'Down there, there are many monasteries, shrines and such things, you know' (rna-gsung 25)

(6.71)

ଞ୍ଥି:ଶ' ସ\ତିସ|'I¹'ର୍ଘ, ନ୍ତି'ଶ' ସାବଟ 'I¹'ର୍ଘ| kisa t₆i-k^ha=lo, nèsa zenkha=lo. birthplace one-at=DAT staying.place in.another.place=DAT 'Birth-place in a place, staying place elsewhere.' (Richhi 160)

Denjongke also has forms with the double function of proadverb of manner and proadjective, see Table 6.8.

Table 6.8. Proadverbs of manner and proadjectives

dodem	वर्ने वनेग	'like this right here'
de: dem, dep	תבלי, תלמי, תקתי	proximal, cataphoric, 'like this', 'like that', 'such'
ódɛː, ódɛm, ódɛp	איֹתלי, איֹתלאי, איתלקי	distal, anaphoric, 'like this', 'like that', 'such'

The spatial orientation of the 'proximal' and 'distal' forms in Table 6.8 with reference to the speaker is less clear than with other demonstratives, hence the exactly same glosses. In discourse, the proximal forms are typically cataphoric (6.72) and distal forms anaphoric (6.73). Example (6.72) is an announcement followed by a quotation of the prayer in question. The distal in example (6.73), on the other hand, refers to a topic discussed earlier.

- (6.72) $\tilde{k} = \tilde{k} =$

A further indication that cataphoric reference is accomplished through proximal demonstratives is given by the cataphoric use of the proximal di in (6.74).

र्षितः तत्ते यात् राष्ट्रात्वे तत्रा र्खेत्या वत्ता के स्वाया के स्वाया के प्राय के स्वाय के यात्र योग क्रा के स्वाया के स्वाय (6.74)র্শিশ শ্বনা $k^h \tilde{o}$: dendzõ: nànca di=jã: súŋ-ze. lopo *m*í*r*i=*t*s*u*=*l*o Sikkim within Lhopo **3SG.HON** this=even say.HON-PST people=PL=DAT j*à:-k^h*ẽ: kaltcə $k\epsilon = di$ $t^{h}amt \varepsilon \varepsilon = l \varepsilon dzik - \varepsilon o$? *bε*?. EX.PER-NMLZ culture(Eng.) language=DEPMH all=ABL excellent-SUP EOU.NE 'He also said this: In Sikkim the culture and culture that the Lhopo people have is the most excellent.' (NAB BLA 7)

6.5 Summary remarks

This chapter discussed deixis and indirect reference in Denjongke. It was shown that second person pronouns exhibit a three-way distinction in politeness (ordinary vs. mid-level vs. honorific), while third person pronouns have a two-way distinction (ordinary vs. honorific). The difference between masculine and feminine ordinary 3rd person singular forms is neutralized in the honorific register. A typologically interesting pronominal feature was seen to be the fact that Denjongke, unlike many Tibetic languages, does not make a clusivity (exclusive vs. inclusive) distinction in first person plural pronouns.

The section on indefinite reference showed that Denjongke uses question words to express affirmative concepts such as 'whatever' and 'whoever' and negated concepts such as 'nothing/not anything', no one/not anyone'. In the first case (affirmative), the question word is accompanied by

a concessive form of the equative copula. In the second case (negated), the question word is accompanied by the additive clitic $=j\tilde{a}$: 'even' and a negated verb.

This chapter also introduced an array of demonstratives which refer to objects, places, directions, amounts, manners and qualities. An interesting feature was shown to be the existence of two proximal forms, the emphatically deictic *dodi* 'this right here' and the less emphatically deictic *di* 'this'.

7 Copulas and evidentiality in copulas

Typically of Tibetic languages, evidentiality in Denjongke largely derives from copulas, which, in addition to copular uses, also function as auxiliaries with other verbs. The current chapter discusses copulas and evidential phenomena associated with them. The auxiliary uses of copulas with other verbs are discussed later in §9 after the chapter on tense, aspect and mood (§8), because it is easier to discuss and understand evidentiality in periphrastic constructions only after those constructions have been introduced.

This chapter starts with the general discussion on evidentiality (§7.1) and then describes simple copulas (§7.2). The next section discusses complex copulas, which consist of more than one morpheme (§7.3). Lastly, simple copulas are compared with cognates in some other Tibetic languages, providing evidence of significant differences and suggesting a direction of diachronic change (§7.4). The discussion on copulas is largely based on Yliniemi (2017) but also improves on it.

7.1 Definition of evidentiality

Evidentiality is usually understood as being concerned with "information source" (Aikhenvald 2004). This definition, however, has proved problematic in Tibetic languages for describing the category that has been variously termed "egophoric" (Tournadre 2008), "ego" (Garrett 2001, Gawne 2013), "self" (Bartee 2007: 137), "personal" (Hill 2012: 391), "old knowledge" (Huber 2000), "assimilated knowledge" (van Driem 1998: 127) and "strong empathy" (Häsler 1999: 151). Following Hill (2012: 391), I use the term "personal" for reasons that will be given later in the chapter.²⁵⁵ This "typologically unusual" category (DeLancey 2018: 9), which forms a system with other more typically evidential categories such as the sensorial evidential, has received differing responses from linguists. Lapolla and Tournadre (2014: 241) broaden the definition of evidentiality in order to subsume the Lhasa Tibetan category egophoric within the redefined definition of evidentiality. DeLancey (2018), on the other hand, specifically states that "[t]he Tibetic Egophoric category is not part of the evidential system." Gawne (2013: 152) prefers the term "modality" to "evidentiality" as a cover term for copula distinctions in Yolmo in order to accommodate ego copulas within the same general descriptive category with other copulas. All of the abovementioned scholars seem to agree that the definition of evidentiality as being simply concerned with information source is not applicable to the category ego(phoric)/personal in Tibetic languages.

Because the copulas function as a system and therefore receive part of their meaning in relation to other copulas, I find it useful to refer to all the copula categories with the same general term. For this pragmatic reason, I here adopt Lapolla and Tournadre's (2014: 240) definition of evidentiality as "the representation of source and access to information according to the speaker's perspective and strategy." This definition subsumes within evidentiality the category ego(phoric)/personal.

²⁵⁵ Although I find the term "personal" helpful for describing Denjongke, it needs to be kept in mind that Hill (2012: 391) applies the term "personal" to Lhasa Tibetan, in which the category functions, as will be shown in this chapter, somewhat differently from Denjongke.

7.2 Simple copulas

Simple copulas consist of the basic copula forms, see Table 7.1, and two additional forms, which have copular uses. The two additional forms are the verb $\dot{\delta}$: 'come', which has existential functions (see §7.2.5.1) and the reportative =lo, which may function as a reportative equative by substituting the typical equative copula \tilde{i} : or $b\epsilon 2$ (see §7.2.5.2). In Table 7.1, affirmative and negated forms are separated by a slash. Table 7.1 lists only unanalyzable interrogative copulas. Copulas may also be interrogated with the regular polar question marker -ka/ga (ig-ga, $m\epsilon g-ga$, $b\epsilon-ka$, $m\epsilon mb\epsilon-ka$, $j\delta$:-ka, $m\epsilon ka$, $d\epsilon$:-ka, $d\epsilon$:-ka, see §10.1.3.2).

		Personal		Sensorial	Neutral		
					Ordinary		Apparentive
		decl.	interr.		decl.	interr.	
EQ	PRS	î:/mề̃: क्षेत्र:/व्यत्र:	ná/mèna न ^{:/} मन्-न	(índu?) (छोत्र'२८२ग')	<i>be?/mèmbe?</i> ఇగ్ ^{:/} శుశ్రాష్	<i>bo/mèmbo</i> র্ন ^{./} ঝবৃ'র্ন ^{.256}	de:/re: 255 ^{,257}
	PST						
EX		j <i>ò?/m</i> è?		du?/mìndu?	(jèbbe?/mèbbe?)		
(pos./	neg.)	شر المكر		तनुगा/वांब.पनुगा.	(थॅन्न-भ्रन्/येन्न-भ्रन्)		

Table 7.1.Basic copulas

As shown in Table 7.1., the types of evidentiality marked by Denjongke copulas are personal, sensorial and neutral. The neutral forms can further be divided into ordinary neutrals and the apparentive '(it)seems to be'. The equative copulas $(\tilde{t}:/m\tilde{\varepsilon}: \text{ and } b\epsilon?/membe?)$ have separate interrogative forms. The existential copulas, on the other hand, are interrogated by the regular polar question marker -ka/ga, e.g. duka/minduka.

The six basic declarative copulas are 1) the personal equative \hat{i} : $\hat{k}_{n_{T}}$ (neg. $m\hat{e}$: $\bar{k}_{n_{T}}$), 2) the personal existential $j\hat{\partial}^{2}\tilde{k}_{n_{T}}$ (neg. $m\hat{e}^{2}\tilde{a}_{n_{T}}$), 3) the sensorial existential $du^{2}a_{n_{T}}$ (neg. $m\hat{n}du^{2}\tilde{a}_{n_{T}}$, $m\hat{n}du^{2}\tilde{a}_{n_{T}}$, which can also be used, perhaps surprisingly, for past equation/identification, 4) the evidentially neutral $b\epsilon^{2}\tilde{a}_{n_{T}}$ (neg. $m\hat{e}mb\epsilon^{2}\tilde{a}_{n_{T}}$), which is basically equative but also has some existential type of uses, 5) the apparentive $d\epsilon$: $/r\epsilon$:, which is a merger of $da b\epsilon^{2}$ 'be like', and 6) neutral existential $j\hat{e}bb\epsilon^{2}$, which is given in brackets, because it is an abbreviation of the complex copula $j\hat{a}po b\epsilon^{2}$, which is introduced later in §7.3.2.1.2. The sensorial equative $indu^{2}$ is so rare that it is given in brackets in Table 7.1 and introduced under combinatory copulas in §7.3.1. Whereas the copulas du^{2} and $b\epsilon^{2}$ are clearly distinct from ordinary verbs in that they do not inflect for tense, aspect and mood (e.g. present habitual $*du - k^{h}\epsilon n b\epsilon^{2}$, $*b\epsilon - k^{h}\epsilon n b\epsilon^{2}$), the personal copulas \hat{i} : and $j\hat{a}^{2}$ can form many of the same constructions as the ordinary verbs (e.g. present habitual $i\eta - k^{h}\epsilon n b\epsilon^{2}$, $j\hat{a}$: $-k^{h}\epsilon n b\epsilon^{2}$).

This chapter focuses on declarative forms, with an emphasis on evidentiality. Interrogative forms are covered in the general discussion on interrogation in §11.1. Rather than describing each evidential category within the copulas by comparing them to some purported typological category established on the basis of other languages, it is useful to describe the evidentiality of each copula with reference to the other copulas within the system. The meaning of the copulas are defined as a system, with reference to each other. The personal copulas \hat{t} and $j\partial^2$ express the speaker's personal

²⁵⁶ This form is homophonic and homographic (in WD) with the nominalized form $m \grave{e}m$ -bo used in such expression as $m \grave{e}m$ -bo $b \varepsilon$? 'is/was not'. Some writers prefer to write the affirmed form $a \breve{s}'$ 'bo instead of \breve{s}' bo.

²⁵⁷ This is an innovative WD form deriving from the full disyllabic 35' 35'.

knowledge. The knowledge is considered personal either because the speaker already possesses it (in contrast to recently acquired knowledge marked by sensorial copulas) or because the referent of the proposition is present at the time of speaking (in contrast to neutral copulas, which are used for spatiotemporal backgrounding). Moreover, in nominalized expressions ending in \tilde{t} , "personalness" may be realized as the speaker's emotional involvement (see §7.3.2.2). In addition, \tilde{t} is associated with performing a type of speech act of identification, whereas *be*? focuses on the consequences of identification (see §7.2.3).

The reason for using the term "personal" rather than "egophoric" for describing Denjongke is that Tournadre (2008: 296) defines the egophoric category in Standard Tibetan in a way that is not applicable to Denjongke: "Egophoric auxiliaries are used with the first person occurring overtly, covertly or by anticipation, regardless of its function in a given clause (subject, object, indirect object, locative complement, etc.)" The more semantically oriented personal category in Denjongke is syntactically less restricted by the first person than its counterpart in Lhasa/Standard Tibetan.²⁵⁸ Tournadre (2017: 111) also specifically comments that "egophoric markers do not generally occur in the southern Himalayas," where Denjongke speakers are situated.

In contrast to the personal copulas \hat{i} and $j\hat{o}$?, which are based on the speaker's already existing knowledge, the basically existential copula du? refers to a specific event where the knowledge was sensorially acquired (similarly Gawne's [2013: 164] perceptual for Yolmo). When used for present occurrences, du? has overtones of newness (contra oldness implied by $j\hat{o}$?). When used as an auxiliary, du? has overtones of momentariness (contra continuation implied by $j\hat{o}$?). The term "sensorial," earlier used by Tournadre & Jiatso (2001: 78), was chosen as a category name, because it is the shortest way to refer to sensory experiences. Alternative terms are "sensory evidential" (Hill 2012: 389), "testimonial" (Tournadre & Dorje 2003: 110) and "perceptual" (Gawne 2013: 163).

The neutral, basically equative copula $b\epsilon^2$, on the other hand, does not refer to a sensory experience as du^2 , and lacks the cognitive assimilation and spatiotemporal proximity implied by $t^2/j\partial^2$.²⁵⁹ Even when having either old personal or recent sensorial knowledge about an event, the speaker may for contextual reasons background these sources of knowledge and instead use the neutral $b\epsilon^2^{260}$. When $b\epsilon^2$ syntactically overlaps with the sensorial du^2 , the use of $b\epsilon^2$ signifies that the proposition is generally asserted without reference to a specific sensory experience. It can be used, for instance, when the speaker and the addressee share the same visual experience at the moment of speech, and, therefore, it would be redundant for the speaker to use an evidential to make explicit how the information was received. According to DeLancey (2018: 17), the basic meaning of Lhasa Tibetan "factual" (analogous to Denjongke "neutral") "is simply the absence of any specification of source of knowledge." The same can be said of Denjongke, and thus the term "neutral" is adopted. The term "neutral" should not and cannot be understood as a typological category that could be applied as such to other languages. Its meaning derives from the Denjongle system where neutrality is defined as absence of sensorialness and personalness.

In the following subsections, the copulas marking the three basic evidential distinctions, personal (§7.2.1), sensorial (§7.2.2) and neutral (§7.2.3) are discussed separately. The section on neutral copulas provides summarizing, comparative examples. This is followed by a brief

²⁵⁸ Unfortunately, the difference of Denjongke and Lhasa Tibetan categories is hidden by the fact that the term "personal" is also used of Lhasa Tibetan (Hill 2012).

²⁵⁹ Hein's (2001: 43) defines Tabo/Spiti Tibetan category "speaker's unspecified knowledge" very similarly to Denjonke "neutral."

²⁶⁰ See Hill (2013) for contextual semantics of Lhasa Tibetan.

description of the apparentive equative (§7.2.4). The reportative =lo and the verb \hat{o} : 'come', which both have copular uses, are addressed last (§7.2.5).

7.2.1 Personal copulas

The personal knowledge expressed by the personal copulas may mean that 1) the proposition in question is evidentially based on their old, existing knowledge, 2) that the referent of the proposition is spatiotemporally proximate to the speaker or 3) the speaker is emotionally involved in the event. The last sense has been attested only with \hat{i} as the final copula of nominalized constructions. The first two senses are expressed by both personal copulas \hat{i} and $j\partial 2$. There is, however, a difference in that \hat{i} seems to primarily convey spatiotemporal proximity, whereas $j\partial 2$ expresses more clearly both old knowledge and spatiotemporal proximity. The reason for this difference between \hat{i} and $j\partial 2$ seems to be that the semantics of the personal copulas are affected by the other copulas they evidentially contrast with.

Because $j\partial^2$ in its ELPA-functions contrasts with both the sensorial du^2 and the neutral nominalized copula construction $j\partial -po b\epsilon^2/j\dot{\epsilon}bb\epsilon$, it has developed semantics in opposition to both of these contrastive copulas. The focus on the speaker's old, already existing (and hence personal) knowledge arises from the opposition to du^2 , which makes reference to a specific, usually recent knowledge-acquiring event. The sense of spatiotemporal proximity ("here and now"), on the other hand, arises from the contrast with the neutral nominalized construction $j\partial -po b\epsilon^2/j\dot{\epsilon}bb\epsilon$, which is used for spatiotemporal backgrounding ("there and then," similarly to mere $b\epsilon^2$).

The equative personal copula \hat{t} , in contrast, lacks a contrastive sensorial equative (the marginal combinatory sensorial equative *indu?*, see §7.3.1, does not contrast with \hat{t} in most contexts) and therefore the semantics of \hat{t} , focusing on spatiotemporal proximity, are mainly affected by its contrast with the neutral, spatiotemporally backgrounding *b* ϵ ?. Nevertheless, as shown in §7.2.1.1, a case can be made for \hat{t} also making reference to the speaker's already existing knowledge. In addition to the above three senses, \hat{t} is associated with a type of speech act of identification, as shown in §7.2.3.

7.2.1.1 Personal equative \tilde{t} :

In equation, the personal \hat{t} : contrasts frequently with the neutral $b\epsilon^2$ (§7.2.3) and marginally with the sensorial *indu*? (§7.2.2). In attributive sentences, \hat{t} : contrasts with the neutral $b\epsilon^2$, the personal $j\partial^2$ (§7.2.1.2), the sensorial du^2 and the neutral $j\dot{\epsilon}bb\epsilon^2$ (from $j\dot{\rho}$ -po $b\epsilon^2$) (§7.3.2.1.2).

In equative sentences such as (7.1) and (7.2), it is usually not obvious that \tilde{t} would mark older knowledge than $b\epsilon^2$, because both sentences could be used as soon as the knowledge is gained. The difference is rather characterized in terms of the presence or absence of the referent, the referent being present in (7.1) and absent in (7.2) (see also §7.2.3).

(7.1)	k ^h oŋ=gi	mìŋ	ts ^h eriŋ	Ĩ.	विंद यी. क्षेट कें रेट केंबी
(7.2)	k ^h oŋ=gi	mìŋ	ts ^h eriŋ	bE?.	विंट यों केट कें रेट भ्रह
	3sg.hon=gen	name	PN	EQU	
	'His name is Ts	hering.'			

In attributive sentences such as (7.3) and (7.4), however, the difference of \hat{i} and $b\epsilon^2$ with reference to integration of knowledge becomes clearer.

- (7.3) $\mathbb{R}^{r} \oplus \mathbb{P}^{r} \mathbb{R}^{r} \mathbb{R}^{r}$ $\mathbb{R}^{r} \oplus \mathbb{R}^{r}$ \mathbb{R}^{r} \mathbb{R}^{r
- (7.4) $\mathbb{R}^{\circ} \mathfrak{F}^{\mathfrak{P}} \mathbb{R}^{\mathfrak{N}^{\circ}} \mathfrak{R}^{\mathfrak{N}^{\circ}} \mathfrak{R}^{\mathfrak{N}^{\mathfrak{N}^{\circ}} \mathfrak{R}^{\mathfrak{N}^{\circ}} \mathfrak{R}^{\mathfrak{N}^{\mathfrak{N}^{\circ}}} \mathfrak{R}^{\mathfrak{N}^{\mathfrak{N}^{\circ}}} \mathfrak{R}^{\mathfrak{N}^{\mathfrak$

Consultant KN commented that in order to say (7.3) of a person who is present, the referent has to be the speaker's earlier acquaintance, whereas (7.4) could be said when seeing the referent for the first time.²⁶¹

The semantic difference between personal \tilde{t} and neutral $b\epsilon P$ is also seen when the copula is followed by the (clausal) attention marker $=\epsilon o$, which may mark a proposition as attention-worthy either to the speaker or to the addressee (see §16.2.2). When used with the personal copula \tilde{t} , which marks integrated knowledge, $=\epsilon o$ marks the information in the proposition as attention-worthy to the addressee, not to the speaker. For an example, see (7.5).

(7.5) $\neg \tilde{n} \tilde{n} \tilde{n} \tilde{n}$ $\tilde{n} \tilde{n} \tilde{n}$ $\eta \hat{a} \quad k^{h} o \eta = g i \qquad p' u \quad \tilde{i} = c o.$ 1PL 3SG.HON=GEN son EQU.PER = AT 'I'm actually his son (which you don't seem to know).' (PT e)

In (7.5), Person A and B are talking about a certain man. The man who is the topic of the discussion is actually A's father. In the course of the conversation, A has reason to believe that B is not aware of this fact. To counter this false assumption, A uses the attention marker to communicate to the addressee that he (the speaker) knows that what he is saying is probably unexpected and newsworthy, and hence attention-worthy, to the addressee.

With $b\epsilon$?, on the other hand, $=\epsilon o$ may mark the proposition attention-worthy either to the speaker (7.6) or to the addressee (7.7).

(7.6) $[a]! \in \overline{a}:=tsu$ $b\varepsilon=co.$ Oh child=HON=PL EQU.NE=AT 'Oh, it's the children.' (Richhi 25)

The proposition in (7.6) is accompanied by an exclamation to underline the noteworthy character of the information about the comers' identity to the speaker. In (7.7), in contrast, the

²⁶¹ The difference in choosing $b\epsilon$? rather than du? is addressed in §7.2.3.

speaker draws, by the use of = εo , the addressee's attention to the fact that he is going to tell the main teaching of his pedagogical story. The fact that with $\tilde{t} = \varepsilon o$ attention-worthiness is addressee-oriented but with $b\varepsilon = \varepsilon o$ either speaker or addressee-oriented suggests that \tilde{t} is a marker of old, already existing knowledge, whereas $b\varepsilon 2$ is neutral with respect to when and how the information was acquired.

The copula \hat{i} : (as also $b\epsilon$?) co-occurs with any of the first, second or third person pronouns, see (7.8), showing that the "personal" semantics of \hat{i} : have not been grammaticalized into a syntactic requirement for the first person to appear with \hat{i} : or into a semantic requirement for the referent to be closely related to the speaker (contra description of "Standard Tibetan" by Garrett 2001: 141-142). The semantic difference of using \hat{i} : and $b\epsilon$? is discussed in §7.2.3.

Prototypically equative copulas describe situations that exist in the present, but in appropriate contexts, they may refer to past events. This is exemplified in (7.9) where the adverbial $\sum_{n=1}^{\infty} n \epsilon n \epsilon$ 'earlier' enforces a past interpretation of the sentence with \hat{i} :

(7.9)	म. पर्टे. क्रुंबे.जब. द	દ્ર. લદ્ર. ફ્રેંવ.બલા વાંવ.લટ્ર. રે.આ.વ્રા.આતવ. લુ. વાંકવા. લુલી						
	ŋà=di	nénle	k'ande:	ha-maŋ-go-kʰɛ̃ː	mí=t£i?			
	1sg=demph í :.	earlier	anything	understand-NEG-understand-NMLZ	person=INDF			
	EQU.PER							
	'I was earlier a man who didn't understand anything.' (KT life-story)							

When used with an adjectival argument, as in (7.10) and (7.11), the use of the equative copula \dot{i} implies that the adjective expresses a defining or identifying characteristic of the nominal it is linked with.

(7.10) $\tilde{w}'(\alpha \hat{\gamma}', \tilde{\gamma}'\tilde{\eta}', \tilde{\gamma}\tilde{\eta}', \tilde{v}'\tilde{v}', \tilde{w}_{\alpha}, \tilde{w}_{\alpha})$ $\dot{\delta di} \quad h\dot{a}ko \quad go:-po \quad k'\varepsilon:t\varepsilon^{h}\tilde{t}: \tilde{t}'.$ that understand need.to-2INF important EQU.PER 'It is important to understand that.' (Richhi 7)

दि' र्दे'र्हे' वग्रवा ठट्र'यवा' दर्हिगवा द्वावा केंद्र येगवा' केंद्र विवाय केंद्र विव केंद्र विवाय केंद्र केंद्र विवाय केंद्र व केंद्र विवाय केंद्र केंद्र विवाय केंद्र व विवाय केंद्र विवाय के (7.11)í. $t^hamt \epsilon \epsilon = l\epsilon$ p'otso dzikta? Ĩ. nè: lèm 1SG.GEN child all=ABL excellent EOU.PER good EOU.PER 'My child is the best of all (lit. excellent from all), a good one.' (RBM story of my son)

The negated form of \tilde{t} is $m\tilde{\tilde{c}}$, except in the circumstantial construction (see §15.8.1), where \tilde{t} is negated by the prefix *ma*-. For examples, consider (7.12) and (7.13), which present two alternatives ways to negate the circumstantial-purposive converb.

(7.12) קאדיאַי איאָקיאַי איאָלי אַאָקיאַי má:pu ma-ím-ba átsi hømpu red NEG-EQU-CIRC a.bit blue-green 'not being red, a bit blue-green' (KN e)

(7.13) न्यरासु याक्षेवार्ये र्ज्याक्षे कार्रे र्ध्वासु

máːpu	ma-ím-bo	pʻja-ti	átsi	hømpu		
red	NEG-EQU-2INF	do-NF	a.bit	blue-green		
'not being red, a bit blue-green' (KN e)						

More examples of \tilde{i} are found in §7.2.3, where \tilde{i} is contrasted with $b\varepsilon^2$ and the other copulas. The use of \tilde{i} as the final copula of nominalized constructions is addressed in §7.3.2.2.

7.2.1.2 Personal existential *j* ∂ ?

Similar to i, the personal existential copula $j\partial i^{262}$ codes the speaker's already existing knowledge (contra sensorially acquired knowledge marked by du?) and spatiotemporal proximity (contra spatiotemporally backrounding nominalized copulas, e.g. $j\partial -po b\epsilon^2/j\dot{\epsilon}bb\epsilon^2$). The use of $j\partial i$ usually also entails that the situation depicted in the sentence continues to exist at the moment of speech (contra du^2 which reports an observation at a particular moment). The personal $j\partial i^2$ can only mark those experiences about which it is possible to acquire personal knowledge over time (e.g. what a friend's character is like), whereas the other existential copula du^2 will be used for coding momentary experiences (e.g. what a friend is wearing today). It seems impossible to gain personal knowledge and present actuality of the proposition implied by the use of $j\partial i$, they use the nominalized constructions $j\partial -po b\epsilon^2/j\dot{\epsilon}bb\epsilon^2$ and $j\partial i \cdot k^h\epsilon n b\epsilon^2$, which are discussed in §7.3.2.1.

The type of knowledge coded by $j\partial^2$ is illustrated by (7.14).

(7.14) $\mathbb{R}^{\hat{a}}$ $\overline{\mathfrak{r}}^{\hat{c}}$ $\overline{\mathfrak{r}}^{\hat{c}}$

The condition of the motorbike in (7.14) is part of the already existing knowledge of the speaker, who knows the bike and its owner. When commenting on an unknown biker who just passes by fast, the immediate sensory evidential du^2 would be chosen. In Kyirong Tibetan, a sentence equivalent to (7.14) and a cognate of $j\partial^2$ as copula implies that the speaker has had a "personal experience" of the speed of the bike by riding it (Huber 2002: 138).²⁶³ In Denjongke, however, riding the bike oneself is not required for a sentence such as (7.14). It is enough just to know the condition of the bike, for one reason or another, very well. In other words, $j\partial^2$ expresses the knowledge state of the speaker but does not reveal how the knowledge was gained.

²⁶² The copula $j\partial^2$ is pronounced by most children and young adults as $j\partial^2$, without rounding in the vowel. Rounding in front vowels, in general, seems to be disappearing.

²⁶³ khā: paħkpà:-de tsħē:mè: jo:bā jø:
he.GEN motorbike-DEF very fast EXPER
'His motorbike is very fast.' (Kyirong, Huber 2002: 138)

Example (7.15), taken from Bhaichung Tsichudarpo's novel *Ricchi*, shows how the author of a novel may use personal forms by virtue of having personal knowledge because he has created the characters and the storyline.²⁶⁴

(7.15)	ଶୁସ [.]	યાર્ફ્યેન્ન્યાગ્રી ર્સેયા સુપરે ક	अवायमावा अवार्च	ฃฐา ๛ัฐา			
	nùp	dɛndzoŋ=gi	sòmbare	méŋkʰãː=na	mémpo	karma	j ò ?.
	west	Sikkim=GEN	TPN	hospital=LOC	doctor	PN	EX.PER
	'It is in	n West Sikkim's	Sombare ho	ospital that doctor	r Karma is.	' (Richhi	161)

The exact semantic interpretation of $j\partial^2$ is dependent on the context. This is illustrated in (7.16), in which $j\partial^2$ may convey either personal knowledge gained through metaphorical proximity to the referent (friendship) or personal knowledge gained by literal proximity (being in the referent's presence).

(7.16) $\frac{1}{2} a^{-1} \hat{q}^{-1} \hat{$

Example (7.16) implies either that the speaker is Bill Gate's friend and so personally knows about his wealth (contra sensorial du?, which would imply recent discovery) or that Bill Gates is present at the time of speaking (contra neutral and spatiotemporally backgrounding *j*ebbe? with no such implication).

The copula $j\partial \hat{\rho}$ is not a typical choice for a simple, second person attributive sentence, perhaps because it would seem arrogant to claim ingrained personal knowledge about another person's qualities to their face, see (7.17).

(7.17) $\mathbb{R}^{n/2} \cdot \mathbb{E}^{n} \oplus \mathbb{R}^{n} \oplus$

According to van Driem (1998: 136), second person attributive sentences with the Dzongkha copula $\Im_{n} j\phi$ (cognate of Denjongke $j\phi$?) are not allowed. Instead, $\Im_{n} j^{-1} du$: (cognate of Denjongke du?) has to be used. Van Driem (1998: 136) states that in attributive sentences "knowledge about the second person referent is by definition objective" (and hence not personal). In Denjongke, however, the second person version of (7.17) is acceptable at least in the special case when the speaker tries to convince the addressee who is reluctant to believe the proposition. In these cases, the copula $j\phi$? may be followed by the attention marker =co to emphasize the addressee's counterexpectation and, hence, the newsworthiness of the claim for the addressee. All the other

²⁶⁴ The novel *Richhi* also quite systematically uses the personal auxiliary construction VERB-*po* i: rather than the neutral VERB-*po* $b\epsilon$? for third person referents' past actions within the author's omniscient narration. Using the cognate form VERB-*pa*-yin is infelicitous in Standard Tibetan (Tournadre & Dorje 2003: 206).

copulas, in different contexts, can more freely link the second person with an adjectival attribute. This is shown in (7.18). For semantic differences between the copulas in (7.18), see the discussion under example (7.52) below.

(7.18) $\widetilde{\mathfrak{s}}_{5}^{-} \mathfrak{g}_{\eta}\mathfrak{q}_{\lambda}\mathfrak{q}_{\lambda}$ $\widetilde{\mathfrak{k}}_{3}^{-/}\mathfrak{q}_{5}\mathfrak{q}_{1}^{-/}\mathfrak{g}_{5}$ $te^{h}\phi^{2}$ gja:nam \widetilde{t} :/du?/be?. 2SG.L fat EQU.PER/EX.SEN/EQU.NE You are a fat one./ (I see) you are fat./ You are fat.

It is a well-known phenomenon in Tibetic languages that when forming questions speakers do not evidentially base their copula choice on their own knowledge but on the anticipated knowledge of the addressee (cf. Tournadre's [2008: 296, 300] "rule of anticipation" in Standard Tibetan, see also Hyslop [2014] for the same in non-Tibetic Kurtöp). For Denjongke, this is illustrated in the question and answer pairs (7.19) and (7.20), where the use of the personal copula in the question does not reflect the speaker's own knowledge state but their estimation of the addressee's knowledge state.

(7.19) a) הדמי מֹק־יזן? *nám jo*:-*ka*? sugar EX.PER-PQ 'Is there sugar?'

> b) ^شم *jò?*. EX.PER 'Yes, there is.'

(7.20) a) æ

a) الله تَعْرَبُونَهُ عَلَيْهُ اللَّهُ عَلَيْهُ اللَّلِ عَلَيْهُ اللَّهُ عَلَيْهُ عَلَيْهُ عَلَيْهُ عَلَيْهُ اللَّهُ عَلَيْهُ اللَّهُ عَلَيْهُ اللَّهُ عَلَيْهُ اللَّهُ عَلَيْهُ اللَّهُ عَلَيْهُ عَلَيْ اللَّهُ عَلَيْهُ اللَّهُ عَلَيْهُ اللَّهُ عَلَيْهُ اللَّهُ عَلَيْهُ اللَّالَ اللَّالَةُ عَلَيْهُ عَلَيْ اللَّعَامُ اللَّالَ اللَّهُ عَلَيْ اللَّهُ عَلَيْ اللَّالَةُ عَلَيْهُ اللَّهُ عَلَيْهُ اللَّهُ عَلَيْ اللَّالَةُ اللَّالَةُ اللَّهُ عَلَيْنَا عَلَيْنَا اللَّالَةُ اللَّالَةُ اللَّالَةُ اللَّهُ عَلَيْ اللَّا عَلَيْ اللَّا عَلَيْ اللَّا عَلَيْ اللَّا عَلَيْ اللَّالَةُ اللَّا عَلَيْ اللَّالَةُ اللَّالَةُ اللَّا عَلَيْ اللَّالَةُ اللَّالَةُ اللَّالَةُ اللَّالَةُ اللَّالَةُ اللَّالَةُ اللللَّالَةُ اللَّالَةُ الْحُلَيْلَةُ اللَّالَةُ اللَالَاللَالَةُ الللَّالَةُ اللْحُلُلْلَةُ اللللَّ الللَّا الللللْحُل

b)	দরি (স্ট্রি·)/দের্শ্বে দ্রুঝা ব্যম্জুঝা র্ঘ্মদা			
	nè:(=gi)/ŋà=lo	p'um	súm	jờ?.
	1sg.gen(=gen)/1sg=dat	daughter	three	EX.PER
	'I have three daughters.'			

In (7.19a) and (7.20a), the use of the personal copula $j\partial^2$ suggests that the speaker deems the addressee as someone who has personal knowledge of the questioned fact. It would be an interesting line of research to find out what copula is used in the answer if the copula in the question does not correctly reflect the addressee's knowledge state. De Villiers et al (2009: 44) observed that "Tibetan children are not led by the evidential posed in a question, but base their answers on their own judgment of the scenario."

More examples of $j\partial \hat{\rho}$ are provided in next sections §7.2.2 and §7.2.3, where its use is contrasted with $du\hat{\rho}$ and $b\hat{e}\hat{\rho}$ respectively.

7.2.2 Sensorial copula du?

The discussion here is divided into existential uses of du^2 (§7.2.2.1), which form the great majority of cases, and past equative uses (§7.2.2.2). The last sections describes the intensifier suffix $-k\varepsilon$, which may attach to du^2 (§7.2.2.3).

7.2.2.1 Existential uses of *du?*

The use of the sensorial existential du^2 indicates that the proposition is evidentially based on a specific, most often recent or current event that the speaker has sensorially (not necessarily visually) attested. Information expressed through the personal evidentials is also first acquired sensorially, but later with time and/or repeated exposure the knowledge becomes so assimilated that no reference to a specific event needs to be made (similarly Gawne [2013: 203] on Yolmo). Whereas $j\partial^2$ conveys that the speaker's knowledge state has existed before ("I already know"), du^2 implies that the knowledge was recently acquired ("I came to know"). The neutral $b\epsilon^2$, on the other hand, marks a proposition non-committed as to the type of knowledge. Whereas $b\epsilon^2$ is used when the speaker and the addressee share a sensorial experience, du^2 is primarily used when the addressee does not share the sensorial experience with the speaker.

Because du^2 often refers to a recent event where knowledge was acquired, it can gain overtones of "newness" or "mirativity" (DeLancey 1997). The overtones of newness in the cognates of this copula in other Tibetic languages have been reported, among others, by Bielmeier (2000: 104), Denwood (1999: 123), Hongladarom (2007: 29) and Huber (2002: 139). It should be noted, however, that "newness" does not necessarily entail "unexpectedness/surprise" (Zeisler 2000: 40). Hill (2012) argues for the basic meaning of du^2 in Standard Tibetan being sensorial rather than mirative. Although the use of du^2 in Denjongke often implies recently acquired knowledge, Denjongke has a separate attention marker =co that can be attached even to the sensory evidential du^2 (du:=co) to emphasize the attention-worthiness (caused by surprise, counterexpectation, sudden realization etc.) of the information either to the speaker or to the addressee (see examples [7.22] and [7.27]). Although du^2 may have some undercurrents of newness, the Denjongke language system does not appear to grammaticalize any "surprise" value with du^2 .

The implied momentariness of du^2 , in contrast to the permanence suggested by $j\partial^2$, is especially seen when the two copulas are used as auxiliaries. In auxiliary uses with the progressive $-z\tilde{\epsilon}$, both $k^hu j\partial^2 p'ja-z\epsilon n du^2$ 'He was working' and $k^hu j\partial^2 p'ja-z\tilde{\epsilon}: j\partial^2$ 'He is working' could be said in a situation where the speaker does not see the man working at the moment of speech. Choosing the option with du^2 implies that the speaker recently saw the referent working, but is agnostic as to whether the referent is still working at the moment of speech (hence the past translation). The option with $j\partial^2$, however, implies the speaker's personal knowledge that the action still continues at the time of speech (hence the present translation).

Examples (7.21-34) illustrate the evidential semantics of du?. First, consider (7.21), a question where the speaker has to make an estimate of the addressees' state of knowledge.

(7.21) הجمع مرجم (۲.21) *nám du-ka?* sugar EX.SEN-PQ 'Is there (any) sugar?'

In (7.21), the speaker assumes that the addressee is not in personal possession of the knowledge asked for, i.e. that the addressee may have to look around right then to find out whether there is sugar. In (7.19) above, on the other hand, where the copula $j\partial 2$ is used instead of du^2 in the

otherwise identical sentence, the speaker assumes that the addressee already has assimilated knowledge on the availability of sugar and can answer the question without searching.

In light of what was said above, example (7.22) seems at first sight anomalous.

(7.22)a) ອ້ຽ. ຢູ່. ສິ້ ন্দ্র্যাশা? tc^hø? nò: bjã:-bo=lo. du-ka? 2SG.L disappear-2INF=REP EX.SEN-PQ cow 'Your cow is said to have disappeared. Is it (here)?' b) ५५,गामो. मा du-ke=co. EX.SEN-IN=AT 'Why, it is indeed.' (TB e)

In (7.22), the first speaker has found a cow that he brings to the second speaker. When making an estimate of the addressee's state of knowledge, speaker A in (7.22) would perhaps be expected to use the personal copula, because the addressee is supposed to have personal, integrated knowledge about his cows. The focus here, however, seems to fall on the specific sensory experience of identifying the cow, not on the existing knowledge state. The attention marker =*co* in B's answer expresses the speaker's surprise, indicated by the old-fashioned exclamation 'why' in the translation.²⁶⁵

The contrast of du^2 and $j\partial^2$ is further illustrated in (7.23-25). The question in (7.23) is formulated in a way that eliminates the possibility of echoing in the answer the same copula that was used in the question.

(7.23) \vec{a} \vec{a} \vec{a} \vec{n} ? $ts^ha t^hop-ka?$ salt find-PQ 'Is there (any) salt?'

(7.24) and

mè?. NEG.EX.PER 'No, there isn't.'

(7.25) ^{ঝे}বৃ'৭5্শ] *mindu?*. NEG.EX.SEN 'No, there isn't.'

To a customer's question (7.23) the shopkeeper may answer (7.24) if he knows from before that there is no salt ($m\dot{e}$? is the negated form of $j\dot{\varrho}$?), or (7.25), if he is not sure from the outset but finds out whether there is salt by looking around (mindu? is the negated form of du?).

²⁶⁵ Consultants KT and KUN commented that the question du-ka in (7.22a) has the meaning "Did you find it?". KT and KUN considered the context of (7.22) strange or surprising.

In the above examples, du^2 refers to the speaker's sensory experience at the time of speaking or just prior to the speech act. Examples (7.26-27), on the other hand, illustrate the uses of du^2 in which the sensory experience happened in more distant past.

(7.26) विंदा यानियां केंदि हियाव क्षेत्रयायते क्षद कें

 $k^{h}\tilde{o}$: $n\tilde{i}$:-po $m\tilde{u}$ =i $k^{h}im$ =na $l \varepsilon p$ - ϕ : $g\tilde{a}$: $m\tilde{u}$ 3PL two-COL 3SGF=GEN house=LOC reach-2INF.GEN time 3SGF 'When the two of them reached the house, she wasn't

 $\hat{\mathbb{Q}}^{a,a}$, $\hat{\mathbb{Q}^{a,a}$, $\hat{\mathbb{Q}^{a,a}}$, $\hat{\mathbb{Q}^{a,a}}$, $\hat{\mathbb{Q$

At the time of arriving at their friend's house, the protagonists in (7.26) sensorially attested that she was not at home. This use of *mindu?* can either be seen as case of the author of this literary work taking the viewpoint of the characters or, as Zeisler (2000: 50) suggests, as the author looking at the scene as if from a window as an observer.

Now consider (7.27), another example of a past use of du?, and a rare instance of du? being used of the first person.²⁶⁶

(7.27) $a = 5^{-1} = 5^{-1} = 10^{$

 \mathfrak{A}^{m} \mathfrak{A}^{m}

Usually information about oneself is by definition personal, and hence marked by \hat{t} and $j\partial 2$, but here the speaker has observed himself in a dream. When waking up from a dream, the dreamer gets an outsider's perspective into their own life. Therefore, the sensorial evidential du^2 can be used when talking about oneself. The copula is here followed by the attention marker =co, which indicates that the information was, and perhaps still is at the moment of speaking, surprising to the speaker.

In (7.28), the speaker is helping another person sit inside a car. The choice of du^2 as copula indicates either that the speaker does not expect his addressees to have definite knowledge about the whereabouts of the pillow or that he is speaking to himself.

²⁶⁶ See, for instance, Denwood (1999: 123) for similar examples from Lhasa Tibetan.

(7.28) ما براجام ما براجام ما براجان ما براحان ما براز

In (7.29), a doctor is examining a patient's X-ray pictures and comments on them:

(7.29) عَيْصَاحِمَتُ عَلَى اللَّهُ عَلَى اللَّهُ عَلَى اللَّهُ عَلَى اللَّهُ عَلَى اللَّهُ عَلَى اللَّهُ عَلَى الْحَافَةُ اللَّهُ عَلَى الْحَافةُ الْحَافةُ الْحَافةُ الْحَافةُ الْحَافةُ الْحَافةُ الْحَافةُ (7.29) المُحافةُ (7.29) (7.29) المُحافةُ (7.29) (7.29

In (7.29) the doctor who looks at the X-ray pictures uses du^2 probably either because the addressee(s) cannot see what he sees or cannot intepret what they see as he can. The sensorial du^2 is mainly used when the addressee does not share the same sensorial experience as the speaker. If the speaker and the addressee both see the same thing $b\varepsilon^2$ is more likely used.

Although information coded by du^2 is most often visual, it can also mark knowledge as deriving from the other senses, hearing (7.30), tasting (7.31), smelling (7.32) or touching (7.33).

- (7.30) 5577457 (WRN' 2579] $payk^ha: dly? du?.$ outside cat EX.SEN 'There's a cat outside (as I heard it meowing).'
- (7.31) $\begin{array}{c} \alpha \widehat{\uparrow}' \end{array} \stackrel{\text{diver}}{=} \eta \widehat{\forall} \alpha' \cdot \widehat{\eta} \widehat{\forall} \alpha' \cdot \widehat{\gamma}' \cdot \alpha \widehat{\neg} \eta \\ di s \acute{o}: tym cimpu du?. \\ this curry delicious EX.SEN$ `This curry is delicious (as I can taste).'
- (7.33) (7,3)

Example (7.34) presents a problem for anchoring du^2 to the speaker's specific sensory experience. The information has been heard from other people or read from books.

(7.34) अग्गुगोषा वायाषवार्य? मुमार्या कवार्या र्डुगीषा

 $\dot{a}ku=gi$ ma-sém-bo? t'appø: gempo=tsu=gifather's.younger.brother =AGT NEG-hear-2INF long.ago.GEN old.man=PL=AGT 'Hasn't the uncle heard? The elders of old (used to say):

"र्चें र्ड न' कें सुन भेव मन छोव वो गरुन पहुन" p'otso mòby? ťa lèn-ce? ín-ne k'atcun dun take-INF child wife metal.ornament hit now EQU.PER-COND "If (your) child is to take a wife, make a kachung-plate."

हे. या.इट. व्यय्ः हे. यानेव.मी. स्वाया स. केंग्रीय.

 $t\varepsilon$ k'atcunlàp-ti $p\acute{e}n=gi$ ta2p'umu=gisometal.ornamentsay-NFwedding=GENsigngirl=AGTSo there is indeed an old tradition stating that "kachung" is the sign of

Although the speaker of (7.34) probably has known the information for a long time, he cannot use the personal $j\partial^2$ here because that would imply that he was present himself at the time when the tradition was formed. Because it is not possible to gain personal knowledge of such a historically oriented word as "tradition," the speaker uses sensorial *du*?, which makes reference to the event(s) in which he has gained the information. An alternative for using *du*? would be to background the handing down of information by using one of the evidentially neutral nominalized construction $j\partial - po b\epsilon/j\epsilon bb\epsilon^2$ or $j\partial :-k^h\epsilon n b\epsilon^2$, the first of which is used analogously to (7.34) in (7.87).

7.2.2.2 Past equative uses of *du*?

In addition to existential uses, du^2 can be used for equative clauses that refer to situations that held in the past, see (7.35) for a declarative and (7.36) for an interrogative example.

(7.35) تَ يَجْمَعَ عَلَى اللَّهُ عَلَى اللَّالَةُ اللَّالَ اللَّهُ عَلَى الْحَلَى اللَّالَةُ عَلَى اللَّهُ عَلَى اللَّهُ عَلَى اللَّهُ عَلَى اللَّهُ عَلَى اللَّهُ عَلَى اللَّهُ عَلَ اللَّهُ اللَّالَةُ اللَّالَةُ اللَّالَةُ عَلَى اللَّهُ عَلَى اللَّهُ عَلَى اللَّالَةُ اللَّالَ اللَّالَةُ عَلَى اللَّالَ اللَّا عَلَى اللَّالَةُ عَلَى اللَّالَةُ عَلَى اللَّالَ اللَّالَةُ عَلَى اللَّالَ عَلَى الْ اللَّاللَّهُ عَلَى اللَّالَةُ عَلَى اللَّالَةُ عَلَى اللَّا عَلَى اللَّالَةُ عَلَى اللَّالَةُ عَلَى اللَّالَةُ عَلَى اللَّالَ عَلَى اللَّا عَلَى اللَّالَةُ عَلَى اللَّالَةُ عَلَى الْحَالَةُ عَلَى اللَّالَةُ عَلَى اللَّالَةُ عَلَى اللَّالَ اللَّ اللَّ اللَّالَةُ اللَّالَةُ اللَّالِ اللَّالَةُ اللَّهُ عَلَى اللَّهُ عَلَى اللَّهُ عَلَى اللَّهُ عَلَى اللَّهُ عَلَى اللَّالَ اللَّالَةُ عَلَى الْحَلَى اللَّالَةُ اللَّهُ عَلَى اللَّهُ عَلَى اللَّالَةُ عَلَى اللَّا عَلَى الْحَالَةُ عَلَى اللَّهُ عَلَى الْ عَلَى اللَّهُ

(7.36)เปราสา, สีรา สีรารุญา ผิสารณ์ พูงสาล้า ณรา ราชณ์ ๆชี ซู้า สีสารา ณรา ๆ รั? tc^hø? lópt^hu? ím-bø: kap=lo=di рета, nàtci tsotci PN 2sg.l student EQU-2INF time=DAT=DEMPH 1PL.GEN chief lémpu=di ka du-ko? minister=DEMPH who EX.SEN-2INF 'Pema, when you were student, who was our Chief Minister?'(YR e)

According to consultant YR, using the sensorial du^2 in (7.36) implies that the same chief minister is no longer in power, whereas using the personal $j\phi^2 (j\dot{\phi}-po/j\dot{e}-po)$ would leave open the possibility that the same chief minister is still in power. Example (7.37), however, suggests that du^2 , at least in the declarative, may be used for past equation even when the situation still holds in the present. In (7.37), a person who has been outside Sikkim for some ten years reacts to news about the name of the current Chief Minister:

מיד׳, ר׳ צַאָראי צַיאֹי חקאיצָר׳ האיד מי עריער׳ אָר׳ אָר׳ אַר׳ אַרי אַיערי איאַר איאַר איאַרי איז אַרי אַיאַרי א (7.37)néma tc^hilo pí:tõ: zi=na nà: jà-pø: ŋà gã: are, 1SG earlier year 2000 four=LOC here EX-2INF.GEN EXCLAM(Nep.) time ódeteika=jã: pawan tsamlin=ra $du:=\varepsilon o.$ that.time=too PN **PN=AEMPH** EX.SEN=AT 'Wow, when I was earlier here in 2004, at that time too (it) was (the same) Pawan Chamling (as Chief Minister), you know.' (KN e)

Similar to present uses, past equative du^2 marks recently acquired sensory information. It may be used when expressing information about things and other persons than oneself (7.38a) but not when the speaker conveys information about oneself (7.38b).

(7.38) a) هَرْجَرَعَرَ حَرَّ مَحَرَ طَحَرَ طَحَ طَعَ مَعْرَ عَلَيْهُ مَحَرًا عَلَيْهُمْ مَعْرَبُ عَلَيْهُ مَعْرَبُ عَلَيْهُمْ عَلَيْهُمْ مَعْرَبُ عَلَيْهُمْ مَعْرَبُ عَلَيْهُمْ عَلَيْهُمْ مَعْرَبُ عَلَيْهُمْ عَلَيْ عَلَيْهُمْ عَلَيْهُمْ عَلَيْهُمْ عَلَيْهُمْ عَلَيْهُمْ عَلَيْكُمْ عَلَيْهُمْ عَلَيْهُمْ عَلَيْكُمْ عَلَيْهُمْ عَلَيْهُمْ عَلَيْهُمْ عَلَيْهُمْ عَلَيْهُمْ عَلَيْهُمْ عَلَيْ عَلَيْهُمْ عَلَيْ عَلَيْهُمْ عَلَيْ عَلَيْكُمْ عَلَيْهُمْ عَلَيْكُمْ عَلَيْ عَلَيْ عَلَيْ عَلَيْ عَلَيْ عَلَيْ عَلَيْكُمْ عَلَيْكُمْ عَلَيْ عَلَيْكُمْ عَلَيْكُمْ عَلَيْكُمْ عَلَيْكُمْ عَلَيْكُمْ عَلَيْكُمْ عَلَيْكُمْ عَلَيْكُمْ عَلَيْكُمْ عَلَيْ عَلَيْكُمْ عَلَيْ عَلَيْكُمْ عَلَيْكُ عَلَيْكُمْ عَلَيْكُمْ عَلَيْ عَلَيْكُمْ عَلَيْكُ عَلَيْكُمْ عَلَيْكُمْ عَلَيْكُمْ عَلَيْكُمْ عَلَيْكُمْ عَلَيْكُمْ عَلَيْكُمْ عَلَيْكُمْ عَلَيْكُ عَلَيْكُمْ عَلَيْكُ عَلَيْكُمْ عَلَيْكُ مُعْتَعْتُ عَلَيْكُ مُعْلَيْكُمْ عَلَيْكُمْ عَلَيْكُمْ عَلَيْكُمْ عَلَيْكُ عَلَيْكُمْ عَلَيْكُمْ عَلَيْ عَلَيْ عَلَيْكُ عَلَيْ عَلَيْ عَلَيْ عَلَيْ عَلَيْكُكُ عْ

b) * \check{z} a' ($\neg v$ a' $\neg \tau$ ' $\neg c$ â' \check{a} $\neg \tau$ ' \not{e} a' \check{a} \check{a} $\neg r$ \check{e} a' \check{a} $\neg r$ \check{e} a' \check{a} $\neg r$ \check{e} a' \check{a} $\neg r$ \check{e} a' \check{a} \check{e} a' \check{a} $\neg r$ \check{e} a' \check{a} \check{e} a' \check{a} a' \check{e} a' \check{a} a' \check{e} a' $\check{e$

Note that du^2 cannot be used for present identification/equation, as shown by (7.39), a faulty attempt to communicate 'who is the man (now over there)?'.

(7.39) *\$וֹ הָלִי יוֹ הְקַיוֹיוֹ? *mi=di ka du-ko? human=DEMPH who EX.SEN-2INF

7.2.2.3 Intensifier - $k\varepsilon$

The sensorial du^2 is often accompanied by the suffix $-k\varepsilon$, which is called here an intensifier. For instance, as an answer to the question *Is there salt*? by using the intensified $-k\varepsilon$ form *mindu-k* ε instead of mere *mindu*?, the speaker can emphasize his/her engagement or involvement in the situation. Whereas *mindu*? could be said after just looking around, *mindu-k* ε would be appropriate after spending some time moving objects while searching. In addition to personal involvement, the intensifier $-k\varepsilon$ may imply certainty. For instance, according to some of my consultants $du-k\varepsilon$ is considered to carry more certainty than mere du^2 when reporting sensory experience. In this respect, it is similar to $-k\tilde{\varepsilon}/-g\tilde{\varepsilon}$: in Kyirong Tibetan, which is reported to mark increased assertiveness (Huber 2002: 136).

When being prompted to comment on the difference between (7.40) and (7.41), consultant PTB commented that (7.40) would be more appropriate when the referent of $k^{h}u$ 'he' is no longer present, although du^{2} may also be used in the referent's presence.

(7.40) R' \mathfrak{H}^{a} \mathfrak{H}^{a}

7.2.3 Equative neutral $b\epsilon$? in comparison with other copulas

The copula $b\epsilon_i$ is basically equative but it also syntactically overlaps with existential copulas $j\partial_i^2$ and du_i^2 not only in adjectival predication but also in quantified existentials and quantified locatives. The neutral $b\epsilon_i^2$ is evidentially non-committed unlike the sensorial du_i^2 and the personal copulas $j\partial_i^2$ and \dot{t} . Therefore $b\epsilon_i^2$ can be used in many contexts as a matter-of-fact generally asserting variant of the other copulas.

Probably the most difficult task in analyzing Denjongke copulas is to identify exactly what is the difference between equative sentences which differ only in the choice of copula \hat{t} : vs. $b\epsilon$?. Two things, however, can be said. First, \hat{t} : seems to perform a type of speech act of identifying, whereas $b\epsilon$? takes the identification for granted and leaves room for the implications of this identification. For a very similar characterization of difference of *yin* and *ree* in Lhasa Tibetan, see Yukawa (2017: 193-194). For an example, consider the two questions-answer pairs in (7.42) and (7.43), which were volunteered by one of my consultants, when I was trying to tease out the difference between \hat{t} : and $b\epsilon$?.

- (7.42) a) $\operatorname{sgr}(\mathfrak{g}^{\mathrm{sgr}},\mathfrak{g}^{\mathrm{sgr}})$ $\operatorname{sgr}(\mathfrak{g}^{\mathrm{sgr}})$ $\operatorname{sgr}(\mathfrak{g}^{\mathrm{sgr}})$ ka bo? PRN.HON what EQU.NE.Q 'Who are you?'
- (7.43) a) স্থ্ৰ'ক্ৰম' গ্ৰ' র্জা'আম্ব ন্ই? lenge? k'an p'ja-khen bo? PRN.HON what do-NMLZ EQU.NE.Q 'What do you do (for living)?'

b) $5^{\circ} (MAT) \stackrel{?}{\in} N \stackrel{}{\to} 1$ $\eta \dot{a} \qquad \dot{a}mdzi \qquad b\epsilon^{2}.$ $1SG \qquad doctor \qquad EQU.NE$ 'I'm a doctor.' (TB)

In the above examples, \hat{t} is used in the answer to the question concerning identity (7.42), and $b\epsilon 2$ is used when the question relates to doing (7.43). This implies that \hat{t} is more concerned with the act of identifying itself, as if performing a type of speech act of identifying, whereas $b\epsilon 2$ takes some distance from identifying and so suggests focusing on the implications of this identification (e.g. activities of a doctor). These are, however, not fixed rules; in another instance, the same consultant gave the sentence $\eta \hat{a} \hat{a}mdzi b\epsilon^2$ as an answer to the question in (7.42).

The possibility of choosing between \tilde{i} and $b\epsilon i$ to convey different evidential nuances about the same situation shows, similarly to Lhasa Tibetan (Hill 2013: 50), that there is no strict epistemological hierarchy among the copulas within which the speaker would have to choose the one considered to carry the highest degree of certainty.

When bringing up this same topic of \tilde{t} : vs. $b\epsilon^2$ with two other consultants, they volunteered comparative sentence pairs (7.44-45) and (7.46-47) respectively ($m\tilde{\epsilon}$: and $m\tilde{\epsilon}mb\epsilon^2$ are the negations of \tilde{t} : and $b\epsilon^2$ respectively).

We mustn't do like that.' (NB e)

(7.44)	দ্রান্থনা	ষ্ধ্রন'র্জ্বগ	জিব্য
	ŋàtco	ı?	lòpţ

	1 _{PL} 'We are	students.'	EQU.PER (NB e)					
(7.45)	দ'ন্তব্য' র্ন্নিদ'	শ্বেশা' শ্বদাদ'তথা	. पट्रेश. टींश. श्र.ज	নিশ				
	ŋàtɛa?	lòpt ^h u?	bE?.	ŋàtɛa?	dɛm	p'ja	mì-lɛʔ.	
	1pl	student	EQU.NE	1pl	such	do	NEG-be.good	ł

ĩ.

(7.46) $\overset{\scale{a}}{\neg} \overset{\scale{a}}{\Re} \overset{\scale{}}{\Re} \overset{\scale{a}}{\Re} \overset{\scale{a}}{\Re} \overset{\scale{a}}{\Re} \overset{\sca$

'We are students.

lòpt^hu?

- (7.47) a) $\tilde{\mathfrak{s}}_{\overline{1}} \circ \tilde{\mathfrak{s}}_{\overline{1}} \circ \tilde{\mathfrak{s}}_{\overline{1}}$ $t\varepsilon^{h}\phi^{2} \quad di \quad dok-t\varepsilon^{h}(i).$ 2SG.L this read-IMP.FRN 'You, read this!'
 - में क्रिंग खंग अब ख़ना में न्दीब हैं। क्रिंग के नेबा b) ηà lópt^hu? mèm-be?. ηà índzi dok mì-ceː. student English NEG-know 1SG NEG-EQU.NE Ι read 'I'm not a student. I can't read English.' (YR e)

Again, in both (7.44) and (7.46) \hat{t} is used for simple identification of people, whereas in the use of $b\epsilon$? in both (7.45) and (7.47) it is the implications of identification that are in focus. Example (7.45) is concerned with responsibilities of students (they should behave in a certain way) and in

(7.47) the central question is abilities of a student (they can read English). Whereas \tilde{t} in (7.44) and (7.46) identifies certain people by their occupational status (or lack of it), the use of $b\epsilon$? in (7.45) and (7.47) focuses on responsibilities and abilities of students in general.²⁶⁷

The above analysis based on elicited examples is corroborated by the following example from the novel Richhi:

(7.48)ग्विर-गी' गर्थिग' मुनस'आवत हें' झर' र'क्षे' त' थेर' में ररस ग'रमु zun=gi *jó? kjap-k^hen=to* bE?. t'ato nà: jò?. government=GEN work do-NMLZ=CEMPH EQU.NE now here EX.PER t^horã: k'a: gju. tomorrow where go 'I'm a government employee. Now I'm here, tomorrow (who knows) where (I) go.' (Richhi 95)

In (7.48), the speaker, rather than telling the addressee new information about his identity (in that case \tilde{t} would be used), focuses on the undesirable consequences of being a government employee. The act of identifying is backgrounded and its consequences are foregrounded.

The second thing that can be said about the difference between \hat{t} : and $b\epsilon^2$ is that \hat{t} : is associated with spatiotemporal proximity, with the "here and now," whereas $b\epsilon^2$ is associated with spatiotemporal distancing, "there and then." A conditioning factor in choosing between \hat{t} : and $b\epsilon^2$ is the presence or absence of the referent in the clause. Consultant PT (Tashiding, West Sikkim) preferred the identifying, equative copula \hat{t} : when the person referred to was present, whereas $b\epsilon^2$ was preferred when the referent was absent. This observation is illustrated in examples (7.49-51) below:

(7.49)	a)	ম্রে. ক্রুযাঝ'রঝ'	জিব্য	
· · ·	,	$k^h u$	gja:nam	Ĩ.
		3sgm	fat	EQU.PER
		'He's (a)	fat (one).' (referent present)

- b) $(\mathbb{R}^{\circ} \oplus \mathbb{P}^{\mathbb{N}^{\circ}} \mathbb{P}^{\mathbb{N}^{\times}} \mathbb{P}^{\mathbb{N}^{\circ}} \mathbb{P}^{\mathbb{N}^{\circ}} \mathbb{P}^{\mathbb{N}^{\circ}} \mathbb{P}^{\mathbb{N}^{\circ}} \mathbb{P}^{\mathbb{N}^{\circ}} \mathbb{P}^{\mathbb{N}^{\circ}} \mathbb{P}^{\mathbb{N}^{\times}} \mathbb{P}^{\mathbb{N}^{\mathbb$
- (7.50) a) $\tilde{k} \subset \tilde{g} \tilde{g} \sim \tilde{k} \tilde{g}$ $k^{h} \tilde{o}: t \varepsilon^{h} i g \varepsilon: b o \tilde{i}:.$ 3SG.HON foreigner EQU.PER 'He's a foreigner.' (referent present)

²⁶⁷ Yukawa (2017: 193-194) provides a very similar analysis of the difference between Lhasa Tibetan *yin* and *red*. In Yukawa analysis of the clauses *khong slob-phrug yin* and *khong slob-phrug red*, both meaning 'He is a student', *yin* "is used simply to report that the speaker is a student (a fact she is imminently familiar with)" whereas the semantics of using *red* subsume "a nuance of obligation associated with being a student."

- (7.51) a) $a_{\Gamma}^{c} \tilde{k}_{\Gamma} a_{\Gamma}^{c} \tilde{k}_{\Gamma}^{c} \tilde{k$

Consultant YR, when given the task of describing the difference between the sentences $k^{h}\delta$: ámdzi \hat{t} : and $k^{h}\delta$: ámdzi be? 'he is a doctor', first commented that in the first sentence the person is alive and in the second one dead, thus just bringing the presence vs. absence distinction to another level and adding temporal distance to spatial distance. Similarly, Chang & Chang (1984: 609) provide an example from Lhasa Tibetan where a boy says about his dead father $t^{h}a$ tì $\eta \epsilon$: pápá rè: 'Now, this is my father'. As the copula $j\hat{i}$: (or yin) would be usually used if the father were alive, Chang & Chang see the choice of $r\hat{e}$: as copula to indicate "emotional distance." Their analysis appears similar to Häsler's (1999: 151) description of Derge Tibetan jin as marking "strong empathy" and $r\hat{e}$: marking "weak empathy" and Kretschmar's (1986: 65) "die innere Regung des Sprechers" (the speaker's inner emotion).

It was already shown in (7.19) and (7.23-25) above that in questions Denjongke speakers make estimates about their addressee's state of knowledge. In questions relating to identity, however, copula choice may also be conditioned by whether the questioner wants to present themselves as someone who already knows or at least has a hypothesis of the answer (\hat{t}), or as someone who does not know the answer ($b\epsilon$?). For an example, consider (7.52).²⁶⁸

(7.52)	a)	ર્ક્રેન્ સ્નેંગલ્	र्षुग' झट्र'ग?		b)	र्केन्' र्श्वेन'ह	क्षुयाः स्नेत्रःया?	
		tc ^h ø?	lópť ^h u?	b <i>ɛ</i> -ka?		t€ ^h ø?	lòpt ^h u?	íŋ -ga?
		2sg.l	student	EQU.NE-Q		2sg.l	student	EQU.PER-Q
		'Are yo	ou a student	t?'		'You a	re a student,	aren't you?'

In swiftly transitory attributive situations, as shown in (7.53), $j\partial^2$ cannot be used because it suggests that the information in the sentence is old and ingrained. Then, the choice of copulas is narrowed down to du^2 and $b\varepsilon^2$.

(7.53) a) $a\hat{\gamma}^{*} \tilde{\eta}^{*} a^{*} a^{*} \tilde{\varsigma}^{*} \tilde{\gamma}^{*} \tilde{\gamma}^{*} \tilde{\varsigma}^{*} \tilde{\gamma}^{*} \tilde{\gamma}^{*} \tilde{\varsigma}^{*} \tilde{\gamma}^{*} \tilde{\gamma}^{*} \tilde{\varsigma}^{*} \tilde{\varsigma}^{*} \tilde{\gamma}^{*} \tilde{\varsigma}^{*} \tilde{\varsigma}^{*} \tilde{\varsigma}^{*} \tilde{\gamma}^{*} \tilde{\gamma}^{*} \tilde{\varsigma}^{*} \tilde{\varsigma}^{*} \tilde{\gamma}^{*} \tilde{\gamma}^{*} \tilde{\varsigma}^{*} \tilde{\varsigma}^{*} \tilde{\varsigma}^{*} \tilde{\gamma}^{*} \tilde{\varsigma}^{*} \tilde{\varsigma}^{*} \tilde{\gamma}^{*} \tilde{\varsigma}^{*} \tilde{\varsigma}^$

²⁶⁸ There are also other ways to form questions which are not treated here but in §11.1.

b) A_{5}^{c} $\tilde{\P}^{a_{1}a_{2}a_{1}}$ \tilde{S}^{c} $\tilde{S}^{a_{1}a_{1}}$ $A_{5}^{a_{1}}$ $di \ k'ola=tsu \ t'ika \ du?.$ this clothing=PL dirty EX.SEN 'These clothes are dirty (I see).'

One context for saying (7.53a) rather than (7.53b) is when the sensory experience where the knowledge acquired is shared by the speaker and the addressee. In these cases, there is no need to base one's assertion with an evidential.

In clock-times, $b\varepsilon^2$ is used in expressions where the minutes have gone past the hour (7.54), whereas du^2 is used when minutes have not yet reached the full hour (7.55).

(7.54)	कुःर्केन्- गर्छ्या- य	हुर्द्र भें ग	a. 28. 321			
	t& ^h uts ^h ø?	tci?	duŋ-di	karma	tcu	bɛ?.
	clock.time	one	strike-NF	minute	ten	EQU.NE
	'It's ten pa	st one.	.' / 'One how	ur having	struc	k, (it) is ten minutes.' (DB e)

The last two examples (7.56) and (7.57) summarize the evidential differences between the basic declarative copulas by contrasting $b\epsilon^2$ with other copulas in locative and attributive use respectively.

(7.56)	a)	विनः वनीवा हैं गोबार्ये थेंना	€ìŋ=di=na	do	kɛːp(o)	j <i>ò</i> ?.
	b)	विन्नः यन्तिःवः हैं गोषार्येः यनुगा	sìŋ=di=na	do	kɛːp(o)	du?.
			'There are a lot of sto	ones in the	field.'	
	c)	विनः वनीवा हैं गोबार्ये झन्	sìŋ=di=na	do	kɛːp(o)	bE ?.
	d)	विन्नः यनीवा हैं गोषार्ये क्षेवा	sìŋ=di=na	do	kɛːp(o)	ť.
			field=DEMPH=LOC	stone	much	COP
			'The stones in the fie	ld are man	ıy.'	

Whereas (7.56a) could be said by the owner of a field, who has old, personal knowledge about his field, (7.56b) would be said by someone who has just seen the field for the first time (or after a very long time) as a comment to someone else who does/did not share the same experience. Example (7.56c), in contrast, featuring the general neutral copula $b\epsilon^2$, can be said by someone who has never seen the field before to an accompanying friend who also sees the field. In this case, the sensory evidential du^2 is not needed, because the knowledge is mutual (they both see the field). Furthermore, (7.56c) could also be said in a situation where the speaker has knowledge about the field from before (old knowledge) but wants to, for some reason, distance himself from the epistemically more committed copula $j\partial^2$, which would imply personalness of knowledge. Example (7.56d), using the personal equative, is somewhat marginal in that it seems rarer than options (a-c). Moreover, consultants' felicity judgments diverged with regard to (7.56d). It was rejected as infelicitous by consultant KT and KUN but readily accepted by DB and YR. The semantic difference between (7.56c) and (7.56d) is probably similar to the difference described for examples (7.44-47).

The attributive use of $b\epsilon$? in contrast with the other copulas is illustrated with the adjective *gjanam* 'fat' in (7.57) below.

(7.57)	a)	ম্রি. ক্রুদামাবন্স, জিবা	$k^h u$	gjaːnam	Ĩ.	'He is a fat one (as I know).'
	b)	હ્ય. શુવાય વસ પેંની	k ^h u	gjaːnam	j <i>ò</i> ?.	'He is fat (as I know).'
	c)	শ্বি. শ্রীবাধ্য.প্রস্রা. ওর্টিরামি	k ^h u	gjaːnam	du?.	'He is fat (as I just saw).'
	d)	র্দ্রি. শ্রীবাধ্য প্রধ্য স্লিন্	k ^h u	gjaːnam	bɛ? .	'He is fat (as I generally assert).'

The first sentence with i: (7.57a) identifies the referent as a member in the class of "fat ones." The copula du? in (7.57c) is used when (or shortly after) meeting the described person for the first time (or after a long time). Whereas du? codes knowledge acquired by momentary recent observation, the use of $j\partial$? in (7.57b) suggests that the statement is based on the speaker's already existing knowledge. The copula $b\varepsilon$? in (7.57d), on the other hand, is neutral in these respects, implying neither the personalness of $j\partial$? nor the immediacy and sensorialness of du?. With $b\varepsilon$?, the emphasis falls on the information expressed in the sentence rather than on the type of knowledge the speaker purports to have.

7.2.4 Apparentive equative *de:/re:*

The appentive equative $d\varepsilon:/r\varepsilon$: merges the apparentive marker da '(be) similar' and the neutral equative $b\varepsilon$? to express the meaning '(it) seems to be'. Some informants were not aware that $d\varepsilon:/r\varepsilon$: originates with $da b\varepsilon$? 'be like'.

- (7.58) هَ¹ ج: تَحْمَّا عَلَيْ عَلَى مَعَرَبَ مَعَرَبَ مَعَرَبَ مَعَرَبَ مَعَرَبَ مَعَرَبَ مَعَرَبَ مَعَرَبَ مَعَرَبَ مَعْرَبَ مَعْرَبَ مَعْرَبَ مَعْرَبَ عَلَى مَعْرَبُ مُعْرَبُ مَعْرَبُ مُعْرَبُ مُ
- (7.59) $\tilde{\mathfrak{s}}'\mathfrak{a} \tilde{\mathfrak{s}}'$ $\tilde{\mathfrak{s}}'\mathfrak{a} \tilde{\mathfrak{s}}'$ $\tilde{\mathfrak{s}}'\mathfrak{a} \tilde{\mathfrak{s}}'\mathfrak{a} \tilde{\mathfrak{s}}'\mathfrak{a} \tilde{\mathfrak{s}}'\mathfrak{a}'$ $(\mathfrak{corrist})^{269} \mathfrak{a} \tilde{\mathfrak{s}}'\mathfrak{a} \mathfrak{s}'\mathfrak{a} \tilde{\mathfrak{s}}'\mathfrak{a} \tilde{\mathfrak{s}'}\mathfrak{a} \tilde{\mathfrak{s}'}\mathfrak{a$

For auxiliary uses of the apparentive copula, see §8.5.2.

7.2.5 Other forms used as copulas

In addition to the exclusively copular words describes above, the verb \hat{o} : 'come' and the reportative marker = lo may function as copulas.

 $^{^{269}}$ The speaker seems to use the loan word "tourist" as an adjective. The Denjongke word for "tourist" is *dy:korwo* מקמי אָדָיקֿי.
7.2.5.1 The verb $\dot{\tilde{o}}$: 'come' as existential

The verb \tilde{o} : 'come' has in some Denjongke varieties developed into an existential copula which, as shown in (7.60) and (7.61), is typically negated but which sometimes may also occur in the affirmative, as in (7.63). Consultant KT noted that he does not use this construction in his speech.

(7.60)क्षेंग्दर्ने' हे' द्वि' खु' वृत्त' वा हेटा *ódi tε* nè: сù-се ma-hõ:. that then 1SG.GEN say-INF NEG-come 'I cannot say that.' Lit. 'That's not mine to say.' (PAD Tashiding story) $\neg \check{a}$ ' $\neg \check{a}$ ' તર્દ ત્ર ' ગ્રે ' તે ' \check{a} ' તે ' \check{a} ' ' \check{a} (7.61)t'ato t'a dzamlinna mi-o:-to, dainoso:s. now now world NEG-come-PROB dinosaurs(Eng.) 'Nowadays (they) aren't there in the world, dinosaurs.' (KN e) (7.62) $ts^ha du - ka/j\dot{a}$:-ka? salt EX.SEN -Q/EX.PER.-Q 'Is there salt?' (KN e) (7.63) a) A1: هۡ a' هَ' ts^ha ma-hõ:.

salt NEG-come 'There isn't (any) salt.' (KN e)

As seen in (7.60) and (7.61), both the perfective negator ma- and the imperfective negator mi- may negate the existential use of \tilde{o} . When \tilde{o} is negated by ma- in its ordinary verbal uses, the result is phonetically /ma- \tilde{o} :/> [mõ:]. In the existential use, on the other hand, the pronunciation is [mahõ:], probably to underline the difference to the regular verbal use. With the negator mi- no laryngeal occurs between the negator and the verb \tilde{o} :, /mi- \tilde{o} :/> [miõ:]. The pronunciation difference between [mahõ:] and [miõ:] is reflected in WD here as $\arg f = ma-hong$ and $\arg f = mi-ong$ respectively.

I have come across one example of an interrogative existential \hat{o} , which in (7.64) occurs alongside the proper existential (personal) copula form $j\hat{o}$?.

(7.64) a) هِ نَشْمَ عَلَيْهُ اللَّهُ عَلَيْهُ عَلَيْهُ مَنْهُمْ عَلَيْهُ (7.64) a) مَعْ نَشْمَ اللَّهُ عَلَيْهُ اللَّهُ عَلَيْهُ اللَّهُ عَلَيْهُ اللَّهُ عَلَيْهُ اللَّهُ عَلَيْهُ اللَّهُ (7.64) salt EX.PER-ATTQ 'Is there salt, I wonder.' (KN e)

b) क्रु कॅट पाठा? *ts^ha òŋ-gam*? salt come-ATTQ 'Is there salt, I wonder?' (KN e)

In addition to the typical copula uses, *mahõ*: occurs as an alternative negated auxiliary to the personal *mè*? and sensorial *mindu*?, as shown in (7.66), which presents three alternative answers to question (7.65). In contrast to $j\hat{\rho}$? and *du*?, \hat{o} : appears to be evidentially neutral, although more research is needed to established that fact.

- (7.65) Q: פִי שְׁשָׁשִ' בַאַיִישָּׁפָז' בּרָשִ'שַן? ג^hu jó? p'ja-zen du-ka? 3SGM work do-PROGEX.SEN-PQ 'Is he working?'
- (7.66) a) A1: אָי קעֿעָק׳ לָאַיאָ׳ אָרן $k^h u$ jó? p'ja-u mè?. 3SGM work do-2INF NEG.EX.PER He's not working (I know).'
 - b) A2: [3] η \check{u} $\check{\eta}$ $\check{\eta}$ $\check{\tau}$ $\check{\tau$
 - c) A3: Ry $\eta \tilde{u} \tilde{\eta}' \tilde{\eta}'$

In addition to the uses as an ordinary verb and a copula, \hat{o} also occurs as a future auxiliary, see §8.2.6.

7.2.5.2 Reportative *=lo* as equative substitute

The reportative marker $=lo^{270}$ may replace an equative copula and thus function as a reportative copula, see (7.67). In existential reportative clauses, on the other hand, the copula is obligatorily present, as shown in (7.68) and (7.69).

(7.67) ^[3] (30.87) (7.67)

 $k^{h}u$ $\acute{a}mdzi=lo.$ 3SGM doctor=REP 'He's reportedly a doctor. / He's a doctor, I hear.' (YR e)

 $^{^{270}}$ Some of the other Tibetic languages that employ =*lo* as a reportative/hearsay marker are Classical Tibetan (Jäschke 1881: 551-552), Dzongkha (van Driem 1998: 405-406), Lamjung Yolmo (Gawne 2013:323), Lhomi (Vesalainen 2016:189) and Kyirong Tibetan (Huber 2002: 107).

(7.68) $\tilde{\mathfrak{G}}' \tilde{\mathfrak{G}}' \tilde{\mathfrak$

(7.69) * $\check{\mathfrak{G}}$ ' $\check{\mathfrak{G}}$ ' $\check{\mathfrak{G}}$ ' $\check{\mathfrak{G}}$ ' $\check{\mathfrak{G}}$ ' * \acute{ona} $ts^{h}a = lo$ there salt= REP

The use of the reportative marker =lo with the existential copulas $j\partial^2$ and du^2 shifts the evidential anchoring of the copula from the speaker to the person who is the source of information. In other words, "evidential information is retained from the original utterance" (Gawne 2013: 135, see also Tournadre 2008: 295-296). The shifting of evidential anchoring is illustrated in (7.70-71).

- a) रषु भूम मेंग व. पर्या (7.70)gã:to:=lo $k^h u$ du?**TPN=DAT** 3SGM EX.SEN 'He's in Gangtok (I saw him).' b) सुः झन् र्हेगातः तर्ह्यालें। k^hu gã:to:=lo du:=lo. 3SGM **TPN=DAT** EX.SEN=REP 'He's reported to be in Gangtok (they told they saw him).'
- (7.71) a) $\mathbb{R}^{t} \mathfrak{A}^{t} \widetilde{\nabla} \mathfrak{A}^{t} \mathfrak{A}^{t} \widetilde{\nabla} \mathfrak{A}^{t}$ $k^{h}u$ $g \widetilde{a}: to:= lo$ $j \partial 2$. 3SGM TPN=DAT EX.PER 'He's in Gangtok (I know it well).'
 - b) $\mathbb{R}^{t} \cong \mathbb{T}^{t} \mathbb{T}^{t}$ $\mathbb{T}^{t} \mathbb{T}^{t} \mathbb{T}^{t}$ $k^{h}u$ $g\tilde{a}:to:=lo$ $j\tilde{\theta}:=lo.$ 3SGM TPN=DAT EX.PER=REP 'He's reported to be in Gangtok (they know it well).'

Whereas in (7.70a) it is the speaker himself who saw the person under discussion, in (7.70b) the copula du^2 reports someone else's sensory experience. Similarly, in (7.71a) the speaker bases their statement on their own personal, already existing knowledge, whereas in (7.71b) the speaker presents that information as personal knowledge to someone else. The reason why the speaker consider their source to have personal knowledge in (7.71b) could be, for instance, that the source of information has personally ordered the person in question to go to Gangtok and has seen them leave in the morning. The use of du^2 (7.70a-b) implies that the speaker just claims that at a past point the person in question was reported to have been seen in Gangtok but that there is no guarantee of the person still being there. The use of $j\partial^2$ in (7.71a-b), on the other hand, suggests

²⁷¹ For the difference between $j\dot{\phi}:lo$ and du:lo see examples (7.70-71).

not only that the information is based on personal knowledge but also that the person in question is still in Gangtok.

The discussion so far has focused on the basic copulas $(\hat{i}, j\hat{\partial}, du, b\epsilon)$ and $d\epsilon$) and two other monosyllabic forms with copular functions $(\hat{o}, and = lo)$. The following section addresses the complex copula constructions.

7.3 Complex copulas

In addition to the basic copulas, Denjongke employs a number of complex copulas in which two basic copulas are combined together, either directly (combinatory copulas) or with the help of nominalization (nominalized copulas). These complex forms fill communicative gaps in the copula system, i.e. they help Denjongke speakers express evidential nuances that cannot be expressed by mere basic copulas, and by using them the speaker can avoid unwanted meanings that are implied by the basic copulas. I first describe the two combinatory copulas (§7.3.1) and then the several nominalized copula constructions (§7.3.2).

7.3.1 Combinatory copulas *imbe?* and *indu?*

The basic copulas may be directly combined to form the emphatic equative *imb* ϵ ? and the infrequent sensorial equative *indu*?. The emphatic equative *imb* ϵ ?, which resembles in form the Dzongkha $\Re_{\overline{a}_1,\overline{a}_1,\overline{a}_1}$ *immä*, often marks the speaker's agreement with what the addressee has just said. In the same vein, Dzongkha $\Re_{\overline{a}_1,\overline{a}_1}$ *immä* can be used to "politely punctuate someone else's narrative" (1998: 127) and "is found primarily in clauses of agreement" (Watters 2018: 342). In examples (7.72) and (7.73), the speaker concurs with somebody else's statement.

(7.72) a) अामुाधाम नुवयाकिते वामम ववार्य?

ágja=jã:	máːmiː=na=rãː	mèmbo?
elder.brother=also	army.GEN=LOC=AEMPH	NEG.EQU.NE.Q
'Isn't the brother also i	n the army?'	

(7.73) अंत्र : अत्र रू: येया जन्में जन्में स्टा

ímbe?.	ám	raŋ=gi	làp-o	dɛn	bɛ?.
EQU.EMPH	mother	2sg.m=agt	say-2INF	true	EQU.NE
'It is indeed	so. Moth	her, what you	say is true.'	(rna-gs	ung 39)

It is noteworthy that in (7.72) both the negated question $m \epsilon m bo$ and the concurring emphatic *imbe*?, although both basically equative copulas, receive a locative argument (for a similar use of $b\epsilon$?, see §5.4.2).

In addition to the concurring uses, $imb\epsilon^2$ can be used for emphatic effect without an explicit previous statement with which to concur, see (7.74) and (7.75). The speaker of (7.74) is a smart farmer who flatters an over-confident peddler into beginning a story-telling competition:

Example (7.75) contrasts the emphatic equative *imbe*? with non-emphatic *be*?. In (7.75a), the speaker presumes that the pencil is indeed the addressee's and seeks confirmation for the claim from the addressee whereas in (7.75b) such a grammatically coded preconception is lacking.²⁷²

(7.75)a) वर्ते कॅंदरगोु रोवर् केवर (pencil) क्षेत्र ख्वद गा? ímbe-ka? di te^hø:=ki pensil this 2SG.L=GEN pencil(Eng.) EQU.EMPH-PQ 'Is this indeed your pencil?' (TB e) b) दर्ने कॅन्गे: रोक सेना (pencil) भ्रन्ग? di tc^hø:=ki be-ka? pensil this 2SG.L=GEN pencil(Eng.) EQU.NE-PQ

'Is this your pencil?' (TB e)

Now consider (7.76-78), which exemplify the sensorial equative indu?.

याञ्चतः र्चत्राग्नीमः यमेः मारुदिः क्वेंत्रार्येः गर्देष्पं रेवाणाः व्यत्पमः (7.78)mèm-ba:²⁷³ pawan *tsamlin=di* nàtci lǿmpu tsou tciku Pawan Chamling=DEMPH 1SG.GEN minister main only NEG.EQU-CIRC 'Pawan Chamling is not only our Prime Minister,

बे' धेषा'रेषाह्र' आवर्ष'र्द्राह्र' पश्चिष' क्षेत्र'त्र्द्रुषा mi jigri? $k^h \varepsilon: ta? = t \varepsilon i?$ indu?. person literature skillful=INDF EQU.SEN but he is also a skillfull literary figure.' (KT e)

 $^{^{272}}$ Pragmatically, however, the very fact that the speaker is asking the question in (7.75b) may be seen as an indication that they presume the questioned fact to be the case.

²⁷³ Consultant PTB (Tashiding) would use here instead of *mèm-ba*: a more complex construction involving an adverbializing nonfinal converb, *ma-ím-bo p'jati* [NEG-EQU-2INF do-NF].

The copula *indu*? combines some of the meanings of both copulas i: and *du*?. Whereas i: marks the equative function, *du*? implies that there was a past personal sensory experience where this knowledge was gained. In (7.78), for instance, the speaker both identifies Mr. PC as a skilful writer and implies that he has had the sensorial experience of reading Mr. PC's writings.²⁷⁴ The difference between *indu*? and the equative use of *du*? (see §7.2.2.2) is that *indu*? is used for present identification (based on past sensorial experience) and *du*? for past identification.

Apart from *imbe*? and *indu*?, no other combinations of basic copulas (e.g. **bedu*?, **dube*?, **bei*?) were acceptable to my consultants.

7.3.2 Nominalized copula constructions

The dichotomy between \hat{t} : and $b\epsilon^2$ within equative copulas, and $j\dot{\partial}^2$ and du^2 within existential copulas, is neutralized in nominalized copula forms so that only \hat{t} : and $j\dot{\partial}^2$ may be nominalized by the markers $-\epsilon\epsilon^2$ (I infinitive) -po/bo (II infinitive) and $-k^h\tilde{\epsilon}$: (nominalizer) (reasons for distinguishing "infinitive" and "nominalizer" are given in §3.2.3).²⁷⁵ Therefore, the morpheme glosses of nominalized copulas below do not have information on evidentiality, e.g. *im-bo* is glossed as EQU-2INF (not as EQU.PER-2INF). The evidential value of a nominalized construction is based on the last copula, e.g. $b\epsilon^2$ in the construction *im-bo* $b\epsilon^2$ and \hat{t} : in the construction $j\dot{\partial}:-po$ \hat{t} . The nominalized part of the construction only marks the equative vs. existential dichotomy, e.g. *im-bo* in *im-bo* $b\epsilon^2$ marks the construction as equative and $j\dot{\partial}:-po$ \hat{t} : marks the construction as existential.

The nominalized equative expressions are *im-bo* $b\varepsilon$? (neg. $m\grave{m}$ -bo $b\varepsilon$?), *im-bo* \dot{t} : (neg. $m\grave{m}$ -bo \dot{t} :), *in-k*^h ε n $b\varepsilon$? (neg. $m\grave{m}$)-k^h ε n $b\varepsilon$?), *in-k*^h ε : \dot{t} : (neg. $m\grave{m}$)-k^h ε : \dot{t} :), \dot{t} :- $\varepsilon\varepsilon$ $b\varepsilon$? (neg. $m\grave{e}$:- $\varepsilon\varepsilon$ $b\varepsilon$? [?]) and \dot{t} :- $\varepsilon\varepsilon$ \dot{t} : (neg. $m\grave{e}$:- $\varepsilon\varepsilon$ \dot{t} : [?]). The existential expressions are $j\grave{p}$:-po $b\varepsilon$? (neg. $m\grave{e}$:-po $b\varepsilon$?), $j\grave{p}$:-po \dot{t} : (neg. $m\grave{e}$:-po \dot{t} :), $j\grave{p}$:-k^h ε n $b\varepsilon$? (neg. $m\grave{e}$:-k^h ε : \dot{t} : (neg. $m\grave{e}$:- $b\varepsilon$?), $j\grave{p}$:- $\varepsilon\varepsilon$ $b\varepsilon$? (neg. $m\grave{e}$:-po \dot{t} :), $j\grave{p}$:- $c\varepsilon$ $b\varepsilon$? (neg. $m\grave{e}$:- $c\varepsilon$ $b\varepsilon$? [?]) and $j\grave{p}$:- $c\varepsilon$ \dot{t} : (neg. $m\grave{e}$:- $c\varepsilon$ \dot{t} : [?]). Table 7.2 gives a summary of the different nominalized forms. Hypothetical (negated) forms of which I do not currently have any examples are marked with a question mark in brackets.

²⁷⁴ My consultant's attempt to translate indu? in (7.78) into Nepali was हो रहेछ ho rahecha.

²⁷⁵ Garrett (2001: 105) considers these type of contexts, where only ego evidentials (the equivalents of \hat{i} : and $j\hat{\partial}\hat{\rho}$) can appear, as evidence for his view that ego evidentiality is not coded lexically in the copulas, but is a "pragmatic property" caused by the absence of other, overt evidentials such as a_{557} 'dug.

Equ./	Evid./epist.	Nominalized co	onstruction		
Ex.	_	Affirmative		Negated	
Equat	Personal	ím-bo í:	क्षेव 'र्चे' क्षेव'	mềm-bo ĩ̃: (?)	बव र्च क्षेव (?)
•		íŋ-k ^h ẽ: ĩ:	દ્યેવ સાયવ દેવ	mèŋ-kʰẽː ĩ́ː	यव यागव खेव
		<i>ี่</i> เ:-ce i:	क्षेत्र.वट्र. क्षेत्र.	mề:-se í: (?)	ਕਕਾ.ਜਨ. ਲੇਕ. (?)
	Neutral or	ím-bo bɛ?	क्षेत्र'र्च' श्वन्'	mèm-bo be?	यव 'र्ये' श्चन
	assertive	íŋ-kʰɛn bɛ?	क्षेत्र स्थन	mèŋ-kʰɛn bɛ?	यव यायव : भ्रद
		í:-se be?	क्षेन् . भन. झन.	mề:-se be? (?)	ਕਰਾਜ਼ਨਾ ਝ਼ਨਾ (?)
Exist.	Personal	j <i>à:-po ĩ</i> :	थॅन्'र्ये' क्षेत्र'	mèː-po í́ː	बेट्र'र्ये' खेव'
		j <i>à:-k^hẽ: ĩ</i> :	र्धेन् :आयव् : क्षेव्	mèː-kʰɛ̃ː í̃ː	र्वत्र आयवः खेवः
		jø:-se í:	थॅन् नन् क्षेत्र	mèː-sɛ í́ː (?)	बेट्र'न्ट्र' क्षेत्र' (?)
	Neutral	jàː-po bɛʔ	थॅन्'र्ये' झन्'	mèː-po bε?	बेन्'र्ये' ञ्चन
		jèbbe?/jøbbe?		mèbbe?	बेन्न'ञ्चन
		j <i>à:-kʰɛn bɛ</i> ?	र्षेत् : श्रायत् : श्रत्	mèː-kʰɛn bɛ?	बेट्र'बायवु' झ्ट्र'
		jø:-se be?	बॅन्'नन्' भन्'	mè:-ɛɛ bɛ? (?)	बेन्:नन्: भ्रन्: (?)

Table 7.2. Nominalized copulas

Most frequently, the nominalized constructions end in the neutral copula $b\varepsilon$?. Constructions ending in the neutral copula $b\varepsilon$? are used by Denjongke speakers to dissociate themselves from the evidential values of \tilde{t} ; $j\partial$? and du?. I first give examples of neutral constructions ending in $b\varepsilon$? (§7.3.2.1) and after that personal constructions ending in \tilde{t} : (§7.3.2.2). More frequent forms with -po/po and $-k^h\tilde{\varepsilon}$: are given first and the less frequent forms with $-\varepsilon\varepsilon$? last.

7.3.2.1 Evidentially neutral constructions (ending in *bε?*)

Nominalized copula constructions ending in $b\epsilon^2$ are evidentially neutral. The following two subsections discuss first equative (§7.3.2.1.1) and then existential constructions (§7.3.2.1.2).

7.3.2.1.1 Equative constructions

The neutral equative constructions are *im-bo be*?, *in-khen be*? and *i*:-*ee be*?. The first two may refer to both present and past states of being, whereas the last one is used for futute and present uncertain states of being. In (7.79), the use of *im-bo be*? conveys assertive force of coming to a certain conclusion. The speaker realizes in her mind that her father's advice had, after all (or indeed), been correct. Assertive force is further added by the adverbial $n\dot{\epsilon}:mu=r\tilde{a}$: 'surely, really, certainly'.

बेंदि' बेग्रबा'त' दर्ने' वर्ते मन् क्षेत्र' वर्त्र' क्षेत्र' क्षे (7.79)ĩ mù=i sém=na de: nόː-εε né:mu=rã: ím-bo 3SGF=GEN mind=LOC like think-INF EOU.PER real=AEMPH EOU-2INF bE?. EOU.NE 'She thinks in her mind like this: (it) really is (like that).' (Richhi 119)

For, the negated form, consider (7.80).

(7.80) สังลริง ขุลงสิงรุขุลง ผลงจัง พรา *ódi k'ɛ:tɕʰit̥a? mèm-bo bɛ?.* that important NEG.EQU-2INF EQU.NE 'That is not important.' (KL BLA 12)

In (7.81), *iŋ-kħen be*? occurs as part of an even more complex assertive copula expression \hat{i} : *iŋ-kħen be*? ("it is indeed the case"). Similar to *im-bo be*? in (7.79), *iŋ-kħen be*? in (7.81) signifies extra assertive force of coming to a conclusion. In (7.81), the omniscient narrator has taken the perspective of the novel's characters that are admiringly looking at a young couple leading a ceremony.

(7.81) $\hat{a}^{i}\check{a}^{j}\eta^{a_{1}}\check{c}^{i}\check{a}^{j}$, $\hat{a}^{j}\check{c}^{a_{1}}\check{c}^{a_{1}}\check{c}^{a_{1}}\check{c}^{i}\check{c}^{j}\check{c}^{i}$ $mits^{h}o?$ $k\varepsilon:po=lo$ $t\varepsilon m$ $t\phi n-di$ mi=tsu=i $d\phi:ji?$ crowd a.lot=DAT show show-NF human=PL=GEN desires '(They were) able to capture people's deepest desires while acting in front of a

र्द्यग्राय्यायता क्षेत्र क्षेत्र यायता छन्, नाक्ष वर्णा নেই্র্রিঁযা' p^hjok ĩ ts^hu:-k^hẽ:. ín-k^hen be?. t'ato ã:=gi snatch can-NMLZ EQU.PER EQU-NMLZ intil=GEN EQU.NE now great crowd. Yes indeed, all the arrangement so far

र्वे। क्षेगुषा वय्यान्न, पिट, गुनुषार्यते, पर्वा पविटाव.

k'odi? $t^{h}amtee?$ $k^{h}\tilde{o}:$ ni:-pø: $gok^{h}i:=na$ arrangement all 3PL two-2INF.GEN leading=LOC had been (successfully) fulfilled under their

The negation of *iŋ-k^hɛn bɛ*? is *mɛ̀ŋ-k^hɛn bɛ*?:

(7.82) दे'र्ये' अव आयव भ्रदा

t'ε:lu mèŋ-k^hεn bε?. just.like.that NEG.EQU-NMLZ EQU.NE 'It is not (i.e. it does not happen) just like that.' (DR discussion with KL)

In line with the above description of $i\eta - k^h \varepsilon n b\varepsilon^2$ and $im - bo b\varepsilon^2$ as somewhat assertive in meaning, consultant YR commented that the constructions $i\eta - k^h \varepsilon n b\varepsilon^2$ and $im - bo b\varepsilon^2$ are used in debates to make assertions that are true contemporaneously with the speech act.

In addition to present assertive uses, *im-bo* $b\varepsilon^2$ and *in-khen* $b\varepsilon^2$ are compatible with past events/states. In (7.83-84), *im-bo* $b\varepsilon^2$ and *in-khen* $b\varepsilon^2$ appear to be used quite interchangeably.

(7.83)	ષ્ટ્ર. ફ્રેંવ.ખથ	१ मदि र्षेषयागु	क्षेत्र'र्चे' झन्			
	k ^h u	pénle	рè:	ť oku	ím-bo	bɛ? .
	3sgm	before	1sg.gen	friend	EQU-2INF	EQU.NE
	'He was	s my friend	before.' (k	KT e)		
(7.84)	ख्र. ८्रत. म्	વાષાગ્યા છે. આવા	र. झन्।			
	$k^h u$	nè:	ť oku	íŋ-kʰɛn	bE?.	
	3sgm	1sg.gen	friend	EQU-NML2	Z EQU.NE	
	'He wa	s my friend	.' (PT e)			

Formally $\tilde{t} = \epsilon \epsilon b \epsilon^2$, which occurs only once in my data, in (7.85), is a nonpast construction (see §8.2.5), which is used for referring to present and future states and events.

(7.85)- ร्रियाग्रिया के र्हेया वर्त्त वेया क्षे र्ड्र्याया राज्यायीया हे र्र्यायया कु वर्ळ्य क्षें वर्त्त क्षेत्र म्ह - र्यायया के र्हेया वर्ष्त क्षेत्र म्ह - र्यायया के र्र्यायया का क्षेत्र क्षेत्र महा - र्यायया के र्र्यायया का क्षेत्र क्षेत्र क्षेत्र महा - र्यायया के र्र्यायया का क्षेत्र क्षेत्र क्षेत्र महा - र्यायया के र्यायया का क्षेत्र क्षेत्र क्षेत्र महा - र्यायया के र्यायया का क्षेत्र - राज्य क्षेत्र - राज्य क्षेत्र - क्षेत्र - क्षेत्र क्षे क्षे क्षेत्र paksam minto=di lèpte tsã:ta? nàtca=gi tsi go:-p balsam flower=DEMPH very.much clean 1PL=AGT reckon be.needed-2INF gjumts^hen ódi **í:-ce** bE?. that EOU.PER-INF EOU.NE reason 'That may be the reason why we have to consider balsam flower very clean.' (RS bee story)

Yeshe Rinzing Bhutia's *Bhutia language learning course book* reports that the form \tilde{t} :- $\epsilon\epsilon b\epsilon$? "indicates statements about which the speaker is not certain" (Bhutia 2008: 53). Uncertainty certainly is a natural corollary of future. Consultant KUN commented that the use of \tilde{t} :- $\epsilon\epsilon b\epsilon$? in (7.85) should rather be translated as "maybe is" than as simply "is." The form \tilde{t} :- $\epsilon\epsilon b\epsilon$? thus presents an interesting case of interplay between tense-aspect and epistemic modality.

7.3.2.1.2 Existential constructions

Examples (7.86-93) illustrate the neutral existential constructions $j\partial po b\epsilon^2$, $j\partial po b\epsilon^2$ and $j\partial po b\epsilon^2$. First consider the semantically similar forms $j\partial po b\epsilon^2$ and $j\partial po b\epsilon^2$ and $j\partial po b\epsilon^2$ illustrated in (7.86) and (7.87) respectively. Examples (7.86a) and (7.87a) are taken from two folk-stories where the speaker does not want to give the impression, by using the lone copula $j\partial^2$, that he was personally involved in the events of the story, or by using the sensorial du^2 , that the event was recently sensorially attested by someone. Therefore, the neutral nominalized copula construction is chosen. Negated examples are given in (7.86b-c) and (7.87b).

(7.86)	a)	विंट्रार्स्डुतिः र्वेरादर्धवानिः से	व्ययाख्तुः यः म्राूटः रः	र्नूम: सुषा					
		$k^h \tilde{o}$:=tsy:	nò:do?=na	símtcẽ:	p'a	lấ:	rà	ťã:	lù?
		3PL=PL.GEN	cattle=LOC	animal	cow	bull	goat	and	sheep
		'In their cattle, t	hey had a lot o	of animals	such as	cows,	bulls, g	oats ar	nd

भार्षेषाका गोषार्दा धॅन् $र्दा ध्वन् <math>b\epsilon$? $laso: k\epsilon:p jo:-po b\epsilon$? etc a.lot EX-2INF EQU.NE sheep etc.' (rna-gsung 1) b) वे केंबर बेट् रें भ्रहा

t^hεts^hom mè:-po bε?. doubt NEG.EX-2INF EQU.NE 'There is no doubt (about that).' (KLT Bumchu video)

(7.87) a) नून्यें हेवा गा गुन् गुन् गहेवा व. के गहेवा. येन आवव. झन

t'ã:pu-teika $k'j\tilde{o}:=tei=na$ mi=tei? $j\tilde{o}:-k^{h}en$ be?. long.ago-APPR village=INDF=LOC man= INDF EX-NMLZ EQU.NE 'In a bygone time, there was a man in one village.'

b) לְקִי הַלְּ הַתְּ אָדִ אֹק־אַקּז אַקן *t'ɛp=di nè:=tsa:* mè:-kʰɛn bɛ?. book=DEMPH 1SG.GEN=at NEG.EX-NMLZ EQU.NE 'I do not have that book.' (KL BLA 12)

The construction $j\partial$:-po be?, which allows the existential meaning to be taken from $j\partial$? and the evidential meaning taken from be?, is so common that in spoken language this evidentially neutral existential form (vs. existentials $j\partial$? and du? which are evidentially loaded) has merged into $j\partial bbe?/jebbe?$ (neg. mebbe?). The Standard/Lhasa Tibetan (close to) pragmatic equivalent to $j\partial bbe?/jebbe?$ is $j\partial$:re:, which is etymologically a nominalized construction as suggested by one of the alternative written forms yod.pa.red (Denwood 1999: 119, Hill 2010).²⁷⁶ In Denwood's (1999: 122) analysis of Lhasa Tibetan, $j\partial$:re: "implies no such first hand knowledge [as $j\partial$: and du:], though it does not specifically rule it out." The same can be said of $j\partial$ -po be? (or $j\partial bbe?/jebbe?$) in Denjongke. The neutral evidential value of the construction is derived from the last copula be?.

In addition to being used for past events $j\partial po b\epsilon^2$ and $j\partial k h\epsilon n b\epsilon^2$ are also used for present events.²⁷⁷ In these cases, using the simple copula $j\partial^2$ is not desirable, because the information in the sentence is presented as uncontested, general knowledge. The following three examples exemplify the present uses of $j\partial k po b\epsilon^2$ (7.88), $j\partial k h\epsilon n b\epsilon^2$ (7.89) and $j\partial k b\epsilon^2$ (7.90), the colloquial equivalent of $j\partial k po b\epsilon^2$.

²⁷⁶ In Lhasa Tibetan, however, there is a current distinction between the historically nominalized form $j\hat{o}:re$: and the synchronically nominalized construction $j\hat{o}-bo-re$: (Denwood 1999: 119).

तुषा न मेन पान न रखा के रुति मन पति वियालें (7.88)t'ari $\eta = j\tilde{a}$: nàt£a? mí=tsv: thi:=lo t'ykã:pø: period today=too 1_{PL} human=PL.GEN foot.GEN heel=DAT 'Even at present time there is a tradition saying that the fact that demø:²⁷⁸ nèkjõ: j*à-po=di* ódi: kap t'vnotch EX-2INF = DEMPH period that.GEN time demoness.AGT there is a notch in the heel of the human foot is a mark of the demoness र्येषाः ह्वे चार्चा स्ट्रिया क्षेत्र के भयास्ति í̃:=sε pok-ti làp-ø: sà-sa-wø: бÝ. pluck-NF eat-RDP-2INF.GEN EQU.PER=QUO say-2INF.GEN trace at that time having plucked and खुगुबार्श्वेवा केंद्रार्थे झुदा lùksø: jò-po bE?. tradition EX-2INF EOU.NE eaten (that place).' (rna-gsung 19-20) କର୍ଟରିନ ଭିନ୍ତୁ । ଭିଂନ୍ଦର୍ନରା ମିକାର୍ଯ୍ୟ ଭିକ୍ରାସନ କ୍ଷମ (7.89)lìu ódɛm jøː-kʰɛn bE?. nàtci ke:po **1PL.GEN** phase like.that many **EX-NMLZ** EOU.NE 'We have many such phases.' (SG wedding customs)

Example (7.88) is the last line of a folkstory which purports to give the origin of a current fact, and (7.89) is a summary statement at the end of an exposition on the various phases involved in getting married among the Denjongpo. The information in both examples is presented as uncontested, general knowledge.

The example of $j\dot{\epsilon}b$ - $b\epsilon$? (7.90) comes from my discussion with one consultant.

(7.90) $\vec{w} \cdot \vec{q} \cdot \vec{q} \cdot \vec{v} \cdot \vec{q} \cdot \vec{v} \cdot \vec{v}$

In the context of (7.90), the consultant KN is telling about his father, who is an overseer of a small monastery. When I asked where the monastery is located, KN continued with a description of the location and ended in (7.90). The copula $j \epsilon b b \epsilon^2$ here marks generally known, uncontested knowledge. It is the location of the monastery that is the topic of the discussion, not whether or not there is a monastery somewhere. Had the original question been whether or not there is a monastery

 $^{^{278}}$ This reading-style pronunciation by consultant KT is likely influenced by the spelling, i.e. the final -*s* fronts the vowel.

somewhere (potentially contested knowledge), the speaker would have more likely used the personal evidential $j\partial^2$ rather than the neutral $j\partial bb\epsilon^2$.

Examples (7.91) and (7.92) illustrate the difference between $j\partial^2$ and $j\partial^2$.

(7.91) تَعْطَّ عَلَى اللَّهُ (Bill Gates) عَنَّ مَحَيَّ مَالِمَةُ عَلَى اللَّهُ عَلَى اللَّالَ عَلَى الْحَالَ عَلَى الْحَالَ عَلَى اللَّهُ عَلَى اللَّهُ عَلَى اللَّهُ عَلَى اللَّهُ عَلَى اللَّ Bill Gates has a lot of money (as I have come to know personally).' (KT e)

(7.92) $\frac{1}{2}a_{1}\cdot \hat{\eta}\cdot \hat{a}_{2}\cdot \hat{\eta}\cdot \hat{u}\cdot (\text{Bill Gates}) \stackrel{\text{```}}{=} \tau_{5}\cdot a_{1}\cdot \hat{\eta}\cdot \hat{u}\cdot \hat{u}\cdot \hat{\eta}\cdot \hat{u}\cdot \hat{u}\cdot \hat{\eta}\cdot \hat{u}\cdot \hat{u}\cdot \hat{\eta}\cdot \hat{u}\cdot \hat{$

Example (7.91) expresses the speaker's personal knowledge by implying either that the speaker is Bill Gate's friend or that Bill Gates is present at the time of speaking. In the latter case, the difference between $j\partial^2$ and $j\partial^2 b\epsilon^2$ is similar to the difference between i and $b\epsilon^2$ in (7.49-51). Example (7.92), on the other hand, is a general statement where the connection to a specific sensory event (contra du^2) and personal knowledge (contra $j\partial^2$) are backgrounded by using the neutral $j\partial^2 b\epsilon^2$.

The existential construction $j\partial :-\varepsilon \varepsilon b\varepsilon^2$, occurring nine times, is more frequent in my data than the equative construction $i:-\varepsilon \varepsilon b\varepsilon^2$. The construction $j\partial :-\varepsilon \varepsilon b\varepsilon^2$ resembles in meaning the more frequent present habitual construction $j\partial :-k^h\varepsilon n b\varepsilon^2$. Both constructions allow the speaker to express generally holding facts, as in (7.93).

(7.93)*di*=*lo*=*di* isto:rum, kot^ha tc^huntcun bo: storeroom (Eng.) room(Nep.) this=DAT=DEMPH small nàksu **j**ờː-ce *bɛ*? di=lo di=lo. this=DAT dark EX-INF EQU.NE DEMPH=DAT 'This (is called) [bo], a stroreroom, this small room, it's dark there, that one.' (PD storeroom video)

b) वर्ते जेंग वर्ते रासु अपाक्त ध्रत, गोव कॅंट पर्ह्या वर्त्ती रासु वें क्षावर ग्री पार्वव कें रा केंट वर्त ध्र di = lo = di $rap^{h}u$ làp- $\varepsilon\varepsilon$ b ε ?, kintsõ: maize.sowing.stick say-INF EQU.NE maize this=DAT=DEMPH tsuk-ce=ki rap^hu. nò: ta-ce=gi zømø=ra maize.sowing.stick cow look-INF=GEN other=AEMPH plant-INF=GEN j*à:-ce be*?. EX=INF EQU.NE 'This is called [rap^hu], maize sowing [rap^hu]. There are (or: will be) other (sticks) for herding cows.' (PL interview)

Based on the decreased certainty implied by the equative nonpast construction \hat{i} - $\epsilon\epsilon$ b ϵ ? (see [7.85]), future research should remain open to the option that in some contexts $j\hat{\varrho}$:- $\epsilon\epsilon$ b ϵ ? may mark uncertainty, although such an interpretation does not seem to fit the sentences in (7.93).

7.3.2.2 Personal constructions (ending in \tilde{t} :)

The following two subsections discuss personal constructions ending in the personal copula \hat{t} ; again first describing equatives (§7.3.2.2.1) and then existentials (§7.3.2.2.2). Thus far, I have found no examples of nominalized constructions ending in \hat{t} : in naturally occurring texts, either spoken or written. Therefore, all of the examples below are elicited. A fuller description of personal nominalized copulas would require natural examples from an extensive corpus.

7.3.2.2.1 Equative constructions

The equative personal nominalized copulas illustrated here are $i\eta - k^h \tilde{\epsilon}$: \tilde{t} :, im-bo \tilde{t} : and \tilde{t} :- $\epsilon \epsilon \tilde{t}$:. First consider the use of $i\eta - k^h \tilde{\epsilon}$: \tilde{t} : in (7.94).

(7.94)यु' मंदे' गलेव'यळव' क्षेव'यापव' क्षेत्रा क्षेत्र' कु' म'ये' $k^h u$ nénts^hẽ: ín-khĩ: ĨĽ. *î*:run k^hu nè: nà=lo relative EQU-NMLZ EQU.PER **1SG.GEN** still 3SGM 1SG=DAT 3SGM 'He is (supposed to be) my relative. Still, he doesn't look

 $\tilde{A}^{\alpha}\eta^{\alpha}\tilde{R}^{\alpha}$ $\tilde{R}^{\alpha}\tilde{R}^{\alpha}\eta^{\alpha}\tilde{R}^{\alpha}$ \tilde{R}^{α} lokta mi-kja(p) $b\varepsilon^{2}$ care NEG-do EQU.NE after me.' (KT e)

In (7.94), by using the emphatic nominalized construction $i\eta \cdot k^h \tilde{\epsilon}$: \tilde{t} : rather than just \tilde{t} :, the speaker appears to underline the fact that the referent is the speaker's relative, thus giving rise to the idea of a relative's obligations that have been neglected ("he is supposed to be"). Using mere \tilde{t} : would simply identify the referent as a relative. In (7.94), the speaker also shows his personal emotional involvement (he is disappointed) by using the personal auxiliary \tilde{t} : rather than neutral $b\epsilon$?. The latter clause of (7.94), which uses the neutral copula $b\epsilon$? as auxiliary, does not imply the speaker's emotional involvement but rather just explains the reason for the speaker's disappointment. In expressing present feelings of the speaker, the Denjongke \tilde{t} : appears to bear resemblance to its cognate in Drokpa Tibetan, which is described as marking personal engagement ("personliche Engagement") and inner (e)motion ("innere Regung") (Kretschmar 1986: 65).

Now consider (7.95) and (7.96) where personal and neutral equatives are contrasted. Nominalized constructions are used because the sentences refer to the past.

(7.95)	7.95) રહું. જ્રેંચ ગ્યુર્થ મંદ્રી સંગુર્થાં છે.					
	k ^h u	nénle	рè:	ť oku	ím-bo	Ĩ.
	3sgm	before	1sg.gen	friend	EQU-2INF	EQU.PER
	'He wa	as my frie	nd before.'	(KT e)		
(7.96)	দ্বি. ষ্ট্রব.জ	ষা দনি শ্র্রিযাম	.ગાુ. લેવ.ર્સ જ્ઞની			
	k ^h u	nénle	nè:	ť oku	ím-bo	bɛ?.
	3sgm	before	1sg.gen	friend	EQU-2INF	EQU.NE
	'He was my friend before.' (KT e)					

Consultant KT commented that the difference between (7.95) and (7.96) is that in (7.95) the speaker expresses that (s)he is presently experiencing sadness about a broken relationship whereas (7.96) is a purely factual statement with no emotional overtones. Another consultant YR (from

Kewsing), according to whom $i\eta \cdot k^{h}\tilde{\epsilon}$: \tilde{i} : and $i\eta \cdot k^{h}\epsilon n b\epsilon^{2}$ could also be used in (7.95) and (7.96) instead of $im \cdot bo \tilde{i}$: and $im \cdot bo b\epsilon^{2}$ respectively, commented that (7.96) is a neutral statement that does not presuppose any continuation of the discourse. The addressee of (7.95), on the other hand, is expecting the speaker to continue by giving the reason for his emotional involvement implied by the personal \tilde{i} : at the end. YR also noted that (7.95) could be said on the basis of the referent being present at the time of speech. The justification for using $im \cdot bo \tilde{i}$: or $i\eta \cdot k^{h}\tilde{\epsilon}$: \tilde{i} : in (7.95) could thus be either emotional involvement of the speaker or the presence of the referent.

Consultant PT, commenting on sentences (7.97) and (7.98),

- (7.97) אַ הֹא זֿעזעזיזין אָאַן אַעסק אָז אַן $k^{h}u$ $n\dot{e}$: t'oku $i\eta$ - $k^{h}\tilde{e}$: \tilde{t} :.²⁷⁹ 3SGM 1SG.GEN friend EQU-NMLZ EQU.PER 'He was my friend.' (PT e)
- (7.98) ^{মৃ.} দেন ম্র্যায়ায় জীব আমব শ্রদ্য *k^hu pè: t'oku iŋ-k^hɛn bɛ?. 3SGM 1SG.GEN friend EQU-NMLZ EQU.NE 'He was my friend.' (PT e)*

said that whereas in (7.98) the relationship is totally over, the personal evidential in (7.97) suggests that there is some continuation of the relationship in the form of perhaps seeing now and then. Thus, the use of the personal evidential seems to suggest some type of present personal relevance, or spatiotemporal foregrounding, for the speaker.

The speaker's current emotion is again the driving force in the use of the personal copula in (7.99). This time the emotion is confusion. The speaker's established belief is challenged by some new information. The use of the personal construction *im-bo* \hat{i} : (according to some consultants also $i\eta - k^{h}\tilde{\epsilon}$: \hat{i} : could be used here), implying emotional involvement, calls for an explanation that is given in the following sentence.

(7.99)	àr afr ़ेर्हेवrar अवाह	र क्षेत्र'र्चे' क्षेत्रा			_
	mí=di	пéтa	ámdzi	ím-bo	Í.
	man= DEMPH	before	doctor	EQU-2INF	EQU.PER
	'Earlier this ma	n was a doo	ctor,		
	न'र्झें में' प्रोंग में' भ्रुपा	यायवः ईंवःळंगोः	র্ণা		
	t'ato=to	k ^h orlo	ky:-kʰɛ̃ː	t ^h øn-ts ^h ak	e=60.
	now=CEMPH	wheel	drive- NMLZ	become-P	FV.APH=AT
	but now he has	become a d	river! (I'm con	nfused)' (KN	le)

In the speech of PT from Tashiding (West Sikkim), both \tilde{t} and *im-bo* \tilde{t} can be used in the present meaning, as shown in (7.100) and (7.101).

²⁷⁹ PT said that using *im-bo* \dot{t} : and *im-bo* $b\varepsilon$? in (11.95) and (11.96) would have about the same meaning as $i\eta$ - $k^{h}\varepsilon$: \dot{t} : and $i\eta$ - $k^{h}\varepsilon$ n b ε ? respectively but that the former constructions are not actively used in his speech variety.

(7.100)	אדי בתי שח	' क्षेम्			
	$k^h \tilde{o}$:	nè:	jàːp	Ĩ.	
	3sg.hon	1sg.gen	father.HON	EQU.PER	
	'He is my	father.' (P	Ге)		
(7.101)	אַבי באַ שק	· क्षेत्र'र्चे· क्षेत्रा			
	$k^h \widetilde{o}$:	nè:	jàːp	ím-bo	Í.
	3sg.hon	1SG.GEN	father.HON	EQU-2INF	EQU.PER
	'He is my	father.' (P	Ге)		

When inquired about the difference between (7.100) and (7.101), PT answered that the latter clause (with *im-bo* \tilde{t}) was "more calm," "more polite," "nicer" and "making the listener feel good." Another consultant (YR), on the other hand, claimed that only (7.100) and not (7.101) could be used when the father is present.

As pointed above, existential copulas have a clear motivation for forming nominalized evidentially neutral constructions, because both of the existentials $j\partial^2$ and du^2 are by themselves evidentially loaded. Among the equative copulas \tilde{i} and b ϵ ?, however, it is more challenging to describe the exact difference between the sentences in (7.100).

ámdzi –	bɛ?.
ámdzi	ím-bo bɛ?.
ámdzi	íŋ-kʰɛn bɛ?.
ámdzi	ím-bo í̃:.
ámdzi	íŋ-kʰɛ̃ː ĩ̃ː.
EMPH doctor	EQU
as a doctor.'	
	ámdzi ámdzi ámdzi ámdzi ámdzi ÉMPH doctor as a doctor.'

Some things, however, can be said. All the forms in (7.102) can be used for both past and present events/states, although with \tilde{t} , and perhaps also with $b\epsilon$?, a past interpretation usually requires a past adverbial. The difference between the personal forms ending in \tilde{t} : (a, e, f) and the neutral forms ending in $b\epsilon$? (b, c, d), as already discussed in §7.2.3, is that the personal forms are concerned with the act of identification whereas the neutral forms leave more room for the consequences of the identification. In existential constructions the nominalizers *-po* and $-k^h\tilde{\epsilon}$: appear to be used quite interchangably, but with equatives, *-po* and $-k^h\tilde{\epsilon}$: seem to have more specialized uses, at least for some speakers. For instance, PT from Tashiding can use (e) for a living person, whereas (f) would be preferred when speaking about a dead person. PT's characterization of the difference between sentences analogous to (a) and (e) was already given with example (7.100) and (7.101). Furthermore, it has been shown above that (7.102c) and (7.102d) may add assertive force to a statement and that (7.102e) and (7.102f) may express the speaker's emotional involvement.

At present, my hypothesis is that the speaker of sentences such as (7.102) will choose *im-bo b* ϵ ? (c) instead of $b\epsilon$? (b) when they want to emphasize the equative function of \tilde{t} : (which is backrounded by $b\epsilon$?) in contexts where the lone \tilde{t} : is undesirable either because of its personal evidentiality or because of its preference for deictical anchoring in the here and now. The nominalized copula constructions have a reduced anchoring to the present compared to lone copulas, lending themselves both to present and past uses, analogously to stative verbs.²⁸⁰

Nominalized personal equatives can also be formed by the infinitive marker $-\epsilon\epsilon^2$. The only two examples in my data are presented in (7.103a) and (7.103b), which are both emphatic or assertive in meaning (hence the gloss *indeed*). Note that in (7.103b) the contrastive emphatic =*to* occurs between the nominalizer and the final auxiliary.

(7.103) a) इन् इन हाम क्षेत्र ते क्षेत्र भन के

 $m \dot{y} t s y^2 l \dot{a} m$ in- $n \varepsilon$ $\dot{t} :- \varepsilon \varepsilon$ $\dot{t} :.$ other lama EQU-COND EQU-INF EQU.PER 'If the other (one) is a lama, (he) is (or: will be) indeed a lama.' (KN kitchen discussion)

b) אֹק'קר'דָ' אֹק'פ' איזן' לען קור'דָ' אָלן
 t:-ce=to f: ná áku rigzã:
 EQU-INF=CEMPH EQU.PER TAG.ASR father's.younger.brother PN
 súm-bo=di.
 say.HON-2INF=DEMPH
 'It is indeed, uncle Rigzang, as (you) said.' (sbar-phung 88)

7.3.2.2.2 Existential constructions

In personal existential constructions, existentiality is expressed by the nominalized copula $j\dot{\partial}$ - $po/j\dot{\partial}$ - $k^{h}\tilde{\epsilon}:/j\dot{\partial}:-\epsilon\epsilon(?)$ and the personal evidential value by final \tilde{t} . For an example on $j\dot{\partial}$ -po \tilde{t} , consider (7.104).

(7.104) $\neg \tilde{w} \bar{\gamma} \tilde{w} \bar{\gamma} \tilde{w} \bar{\gamma} \tilde{w} \bar{\tau} \tilde{\kappa} \bar{\gamma} \tilde{w} \tilde{\pi} \tilde{\tau} \tilde{w} \bar{\tau} \tilde{w} \bar{\tau} \tilde{w} \bar{\tau} \tilde{w} \bar{\eta}$ $\eta \dot{a} \dot{o} na j \dot{\theta} - p \dot{\theta} \tilde{z} t 'y ts^h \dot{\theta} \tilde{z} = lo k^h u = j \tilde{a} \tilde{z} j \dot{\theta} - p o$ $1 \text{ sG there EX-2INF.GEN time=DAT 3 \text{ sGM=too} EX-2 \text{ INF} \tilde{t} \tilde{z}$. EQU.PER 'At the time I was there, he was (there) too.' (YR e)

In (7.104) the speaker uses the nominalized construction $j\partial po$ rather than the mere copula $j\partial \hat{\rho}$ because $j\partial \hat{\rho}$ typically implies that the described situation persists at the moment of speech. The event referred to in (7.104), however, happened in the past and the speaker does not want to imply its present actuality. On the other hand, the personal copula \hat{t} rather than the neutral $b\epsilon \hat{\rho}$ is chosen as the final auxiliary because the speaker was personally present at the referred time.

(7.105) is another example of a personal existential construction. Here $j\hat{\phi}:-k^{h}\tilde{\epsilon}:\tilde{t}:=\epsilon o$ could be used in place of $j\hat{\phi}-po$ $\tilde{t}:=\epsilon o$.

²⁸⁰ For eventive/dynamic verbs, the infinitivizer *-po/bo* has in effect become a past tense marker, e.g. $s\dot{a}$ -bo \dot{t} : $> s\dot{a}$ -u \dot{t} : '*ate*', but for stative verbs the nominalized form can be used in the present meaning, e.g. ga-bo \dot{t} : > ga-u \dot{t} : '*love*'. Therefore the copulas side with stative verbs in letting the context be the final arbiter with reference to present vs. past interpretation.

(7.105)के, देय वहे दत इर थेंद रां केवरों $t' \varepsilon p = di = to$ $\mathbf{\tilde{t}} = \mathbf{\varepsilon}o.^{281}$ jờ-po έĽ *pè:=tsa:* book= DEMPH=CEMPH 1sg.gen-at EX-2INF EQU.PER=AT 0 'O, I would have had the book (all along), you know.' (KN e)

In (7.105), person A has been trying to get hold of a certain book by asking from his various friends. After finally managing to obtain the book, he meets person B who has not heard about A's need for the book. After A tells B about his search and finding the book, B answers (7.105). The nominalized $j\partial po$ is used instead of mere $j\partial p$ because the speaker makes reference to a past point of time. He had the book when his friend was looking for it. Using mere $i\hat{\phi}^2$ (or $i\hat{\phi}^2 = \epsilon o$) would put the emphasis on having the book presently ("I have the book"), whereas the nominalized form enables to convey the past-oriented meaning equivalent to English "I would have had the book (if you had asked me)." The personal final copula \hat{i} in (7.105) most likely signifies the fact that the speaker had the personal experience (and thus personal knowledge) of possessing the book at the time when the addressee was looking for it.

For the only two examples of existential $j\partial :-\varepsilon \varepsilon \tilde{i}$ in my data, consider (7.106). In (7.106a), the glide /j/ in the existential is reduced to /h/.

(7.106) a) خَرَسَة: (beer) مَتَرَجَحَ شَعَرَةًا biər **he:-eiŋ**=eo.²⁸² beer(Eng.) EX-NPST.PER=AT 'There's beer (inside), you know.'(oh, Tashiding)

> $tc^{h}uts^{h}g^{2}$ gu:=loŋà t^horã: t^hom=na **jø:-ce** ĩ. 1SG tomorrow clock.time nine=DAT town=LOC EX-INF EQU.PER 'I'll be at town tomorrow nine o'clock.'(KN e)

As shown by (7.106), $j\hat{a}$: $\epsilon \epsilon \hat{i}$: can refer to both currently holding (7.106a) and future states (7.106b). The latter use distinguishes $j\dot{\partial}$: $c\varepsilon$ \dot{i} : from $j\dot{\partial}$: $k^{h}\tilde{\varepsilon}$: \dot{i} ; which can refer to present but not future states. Although the exact semantics of $j\partial z - \varepsilon \varepsilon \tilde{i}$ are difficult to pinpoint based on the scarce current data, my hypothesis is that in the use such as the one in (7.106a), which refers to a state that holds at the time of speaking, the meaning is, analogously to the equative \tilde{i} -ce \tilde{i} . emphatic/assertive compared to mere $j\partial^2$. That is, whereas mere $j\partial^2$ would convey the meaning 'there is (I know)', the nominalized formulation in (7.106a) carries the meaning 'there certainly is (you will find out if you check)'. Analyzing (7.106a) is made complex, however, by the attention marker = co which by itself may convey the idea of insisting. In (7.106b), on the other hand, the nominalized (nonpast) construction seems to simply mark future.

²⁸¹ As already shown in (7.5), with the personal copula \dot{i} : the attention-worthiness marked by = ϵo is addressee-oriented. ²⁸² Some speakers pronounce [h] for initial /j/ when it is followed by a front vowel. Rounded vowels also tend to get unrounded, especially with younger speakers. As a result of these two changes, /j@?/ may be pronounced as /he?~he?/. The form $-\epsilon \tilde{i}$ is a reduction of $-\epsilon \epsilon \tilde{i}$.

7.4 Simple copulas compared with some other Tibetic languages

This section briefly compares Denjongke basic copulas to copulas in the better known related languages Dzongkha and Standard/Lhasa Tibetan. A notable difference between the Denjongke copula system and that of Dzongkha (see Table 7.3 below), a closely related language, is the nature of contrast between equative copulas. Dzongkha makes a central contrast between old information (marked by $\Re_{\overline{A}}$ '*ing*, a cognate of Written Tibetan $\Re_{\overline{A}}$ '*yin*, similarly to Denjongke, on the other hand, makes a central contrast among equatives between \hat{t} , which marks old information and spatiotemporal proximity, and *b* ϵ ?, which marks evidential neutrality and spatiotemporal distance. Although the Denjongke sensorial equative *indu*? (see §7.3.1) bears some functional similarity to Dzongkha '*immä*, the Denjongke marker is too marginal to be considered to correspond to the frequently used '*immä*.²⁸³

	Assimilated (old)	Aqcuired (new)
Equative	क्षेत्र [.] 'ing	ક્ષેન પાય 'immä
Existential	ũĩ <u>¬</u> ' <i>jö</i>	त्रन्त्या [.] du:

Table 7.3. Dzongkha copulas (adapted from van Driem 1998)

The difference in the nature of contrast between equatives appears to cause a slight semantic difference in the reflexes of WT $\log_{\nabla} yin$ between Dzongkha and Denjongke. Dzongkha 'ing essentially marks assimilated/old knowledge because it is contrasted with 'immä, which marks newly acquired knowledge. The meaning of Denjongke \tilde{i} ; on the other hand, focuses on spatiotemporal proximity (rather than oldness of information) because it is paradigmatically contrasted, not with a copula expressing newly acquired knowledge, but with the neutral copula $b\epsilon 2$, which implies spatiotemporal backgrounding.

An important fact about Denjongke "personal evidential" is that it is not as much restricted by the concept of grammatical person as the related category "ego(phoric)" in "Standard Tibetan" (Garrett 2001, Tournadre & Dorje 2003). According to Garrett (2001: 103), ego(phoric) copula constructions are "rather free, allowing the overt or implied first-person to be a grammatical subject, object, possessor of a subject or object, or even a possessor of a possessor. Nevertheless, *all ego sentences share a first-person restriction of some kind*" [italics added]. Garrett (2001: 141-142) further notes that in some uses of *yin [jîn*], such as (7.107), the 1st person may be syntactically absent. In these cases, however, the referent has to be "closely related to the speaker, e.g. his son." (The example is edited from the original.)

(7.107) Standard Tibetan (Garrett 2001: 142) ? $\check{\uparrow}$ $\check{\uparrow}$ $\check{\uparrow}$ $\check{\uparrow}$ $\check{\uparrow}$ $\check{\uparrow}$ $\check{\uparrow}$ $\check{\downarrow}$ $\check{\uparrow}$ $\check{\downarrow}$ $\check{\downarrow}$ $\check{\uparrow}$ $\check{\downarrow}$ \check{I} $\check{$

²⁸³I came across the first instance of *indu?* after several years of Denjongke studies. The form does not occur even once in my digitized data, which includes, among other things, the whole novel Richhi.

(7.108) Denjongke $\tilde{\Lambda} \subset \tilde{\lambda} \subset \tilde{\lambda}$ $k^{h} \tilde{o}: l \acute{o} p \tilde{o}: \tilde{\lambda}:$ 3SG.HON teacher EQU.PER 'He is a teacher.'

The difference between Standard Tibetan (7.107) and Denjongke (7.108) is that in Denjongke the personal copula i (cognate of $\log y$ yin) is freely used without any requirement for the referent to be closely related to the speaker. In an interesting contrast to Garrett's (2001: 141-142) description, Yukawa's (2017: 192) discussion of Lhasa Tibetan provides example (7.109) with the following comment "the person denoted by *koy* is often a family member or a close friend, but the essential meaning here is that the speaker feels familiar with the fact that the person is a student."

(7.109) Lhasa Tibetan (Yukawa 2017: 192) آمج تيم المجابة المحافظة محافظة المحافظة المحاضي م

Yukawa's gloss of (7.109) does not suggest a semantic restriction resembling that posed by Garrett (2001: 141-142). Instead, Yukawa (2017: 194) defines *yin* and *jod* (cognate of Denjongke *jò?*) as denoting "a state with which the speaker (or the listerner in interrogative sentances) feels familiar." Thus, Yukawa's description of *yin*, like my description on Denjongke, implies less syntactic restriction (of the first person) than Garrett's description of "Standard Tibetan." Yukawa's language data, which seems to come from the end of the 1960s or the beginning of the 1970s²⁸⁴, appears to have been collected roughly 30 years before Garrett (2001). This begs the question whether Yukawa's and Garrett's descriptions document a diachronic change from semantically oriented meaning towards more syntactic restriction.²⁸⁵

Such diachronic change is reported by Hongladarom (2007) for Rgyalthang Tibetan (a variety of Kham Tibetan). According to Hongladarom (2007: 22), Rgyalthang Tibetan folkstories and songs use "egophoric/self" forms in contexts where in everyday speech one expects a non-egophoric form. This suggests that Rgyalthang folkstories preserve an earlier form of the language, in which the current "egophoric/self" forms (somewhat corresponding to "personal" in Denjongke) are less restricted by the syntactic category of person than in the present spoken Rgyalthang.²⁸⁶

Moreover, Widmer (2017: 7) notes a similar diachronic change from semantic marking to more syntactic restriction in Bunan (Tibeto-Burman, non-Tibetic):

In the genealect of the oldest speaker generation, which roughly comprises speakers that were born before 1950, set A endings can express *epistemic involvement regardless of the semantic role that the speaker assumes*. In the genealect of younger speaker generations, set A endings have a narrower range of application and can only express

²⁸⁴ Yukawa (2017) is Nathan W. Hill's translation on an original Japanese article of (1975), which in turn is a revision of the same author's article of (1971).

²⁸⁵ Other factors influencing the issue are the age of the consultants, dialectal differences and the researchers ways of describing.

²⁸⁶ For another example where a difference between spoken and written language gives evidence to diachronic change in evidentiality, see the comment on Denwood (1999) under example (9.18).

epistemic involvement in contexts in which the speaker is co-referent with the most agent-like participant in the clause." (emphasis added)

If diachronic change towards more syntactic restriction has happened and perhaps is happening within Tibetic languages, Denjongke, along with Lhasa Tibetan described by Yukawa (2017[1975]), can be characterized as more "archaic" than Garrett's and Tournadre & Dorje's (2003) descriptions of "Standard Tibetan."

Another Tibetic language, in which the cognate of WT $\log vin$ behaves more semantically than its Lhasa/Standard Tibetan counterpart, is Lamjung Yolmo. Gawne (2013: 192) comments that Yolmo "ego copulas do not relate to the subject of the sentence, or the relationship of the speaker to the subject, but instead express the speaker's knowledge." Gawne (2017: 79), furthermore notes on the Classical Tibetan used in the biography of Milarepa by Gtsang smyon (1452-1507) (described by Oisel 2013: 81) that "*yin* was used in contexts that capture the personal knowledge of the speaker" and that "the distribution of the egophoric at this time was more like what we find in modern varieties such as Kyirong and Yolmo." In other words, Yolmo and Kyirong (and Denjongke) preserve an earlier, semantically oriented use of the WT $\log vin$, whereas Lhasa Tibetan has progressed towards more syntactic control (i.e. the requirement for the first person to occur with egophorics). Gawne (2017: 80) suggests that the split between Yolmo and Kyirong (and Denjongke) from Central dialects is likely to have taken place before increased syntactic control developed in Central Tibetan into "egophoric" in the sense of Tournadre (2008, 2017).

Similarly to Yolmo, Denjongke personal copulas refer to the speaker's personal knowledge rather than the speaker's involvement in the event²⁸⁷ or relationship to the subject. A possible difference between Denjongke and Yolmo, however, is that in a sentence such as (7.108) ('He is a teacher') above the "personal" evidentiality of \tilde{i} , owing to the contrast with the spatiotemporally backgrounding equative $b\epsilon 2$, appears to focus more on the spatiotemporal closeness of the referent (i.e. the person introduced is present) than on the speaker's already existing knowledge. Some other Tibetic languages, which do not share the 1st person restriction of Standard Tibetan with reference to the cognate of the "egophoric" *yin*, are Balti, Purik and Lower Ladakhi/Nurla (Bielmeier 2000).

In summary, ego(phoric)/personal copulas in Tibetic languages appear to occur on a grammaticalization scale from more semantically oriented marking to more syntactic restriction by the first person. The most grammaticalized end seems to be occupied by Standard Tibetan, which has developed a syntactic requirement for the presence of the first person in association with the egophorics (corresponding to "personal" here) (Tournadre 2008: 296). Exceptions are only allowed if the referent is closely related to the speaker, see (7.107). Shigatse and Themchen Tibetan (Haller 2000: 187), on the other hand, appear not to have a syntactic restriction but have instead a semantic restriction: the speaker has to be involved in the event. Denjongke (together with Yolmo, see Gawne 2013: 191-193) represents a yet less grammaticalized stage. The use of i as copula is not syntactically restricted to the first person, the referent in the clause does not need to have an especially close relationship to the speaker, and the speaker's involvement may be non-existent or very weak.

The development from semantically oriented marking of speaker's personal knowledge (e.g. Denjongke, Lamjung Yolmo) towards more syntactic control (Lhasa Tibetan) can be seen to arise quite naturally through speaker's involvement, a notion which has been described as central, for instance, for Shigatse and Themchen Tibetan (Haller 2000:187). First, personal knowledge, which

 $^{^{287}}$ However, spatiotemporal proximity of the referent (see §7.2.3) and the speaker's emotional involvement (see §7.3.2.2) may be viewed as a weak type of speaker-involvement in Denjongke.

typically coincides with personal involvement, is reinterpreted as personal involvement. Then, personal involvement, which frequently coincides first person syntax (on agent, patient or other constituent), is reinterpreted as a need for the presence of first person syntax. This hypothesis for grammaticalization of WT $\arg vin$ is schematized in Figure 7.1, where Yolmo and Denjongke take place towards the left of the continuum, Lhasa Tibetan towards the right, and Shigatse and Themchen Tibetan (based on Haller's [2000:187] brief characterization) in the middle.

Figure 7.1. Hypothesis of grammaticalization of WT yin	
Speaker's knowledge \longrightarrow Speaker's involvement \longrightarrow	Required first person syntax

Note that the three concepts in Figure 7.1 are present in Tournadre's (2017: 110) latest exposition of the category egophoric as he applies it to Tibetic languages (emphasis mine):

The use of an 'egophoric' auxiliary expresses the speaker's **personal knowledge**. The speaker is often directly implied **involved** in the event that is being described (see Tournadre and Dorje 2003: 93), "Egophoric auxiliaries are used **with first person** occurring overtly, covertly [...] regardless of its function in a given clause (subject, object, indirect object, locative complement)" (Tournadre 2008: 296).

While Tournadre's (2017: 110) definition describes the synchronic situation of those Tibetic languages which have arrived at the rightmost end of Figure 7.1²⁸⁸, the other terms in Figure 7.1 hypothesize the route through which the required first person syntax has arisen. Moreover, Figure 7.1 also suggests that the same grammaticalization cline is synchronously represented by various Tibetic dialects.

Sometimes the syntactically motivated terms "disjunct" (equivalents of $b\epsilon^2/du^2$) and "conjunct" (equivalents of \tilde{t} :/j ∂ ?), originating from Hale (1971, 1980), have been used in describing Tibetic copulas (e.g. DeLancey 1990, 1992). However, if applied to Denjongke, these syntactic terms referring to co-reference fail to facilitate an insightful analysis, because the real factors behind copula choice are semantic and pragmatic rather than syntactic. For a thorough criticism of using the concepts of "disjunct" and "conjunct" in describing Standard Tibetan, see Tournadre (2008).

7.5 Summary remarks

In this chapter on copulas and evidentiality, it was shown that Denjongke has a particularly wide array of copula forms, which mark three evidential values: personal, sensorial and neutral. The personal evidential is associated with well-integrated knowledge, spatiotemporal proximity of the referent and emotional involvement. The sensorial evidential refers to a sensory experience. Neutral evidentiality refers to the lack of personal and sensorial evidential values. It was shown that these evidential values are expressed through simple copulas and several complex constructions consisting of combinatory copulas and nominalized copulas. An interesting discovery was that the sensorial du^2 , which typically functions as an existential, can be used as an equative if the proposition describes something that held in the past. The last section of the chapter showed that the category "personal" in Denjongke differs from "egophoric" in Standard Tibetan (Tournadre & Dorje 2003) in that the Denjongke category is more semantically-oriented than the similar category in Standard Tibetan. I also outlined a hypothesis on how the more semantic type

²⁸⁸ According to Tournadre (2017: 111), "[e]gophoric markers are found in Tibet (Ü-Tsang, Tö-Ngari, Kham and Amdo, etc.) but do not generally appear in the Tibetic languages in the southern and Western Himalayas."

of marking, as exemplified by Denjongke, may have grammaticalized into the "egophoric" category that evinces more syntactic control, as exemplified by Standard Tibetan.

8 Tense, aspect and modality

This section describes verbal constructions which are related to tense, aspect and modality. Tense refers to how the action depicted by the verb relates to the time of speaking (e.g. past, present, future) (Timberlake 2007: 304). Aspect refers to the internal structure of the event described by the verb (e.g. progressive, imperfective, perfective) (Comrie 1976: 3). Modality is concerned with the speaker's judgments about a proposition, for instance certainty, obligation, permissibility and ability (cf. Palmer 2001: 8-10). While this chapter includes cursory remarks on evidentiality, a more detailed discussion on evidentiality in presented in §7 (copulas) and §9 (auxiliaries).

The following discussion is divided into five parts. The first part discusses those forms which describe past events from various aspectual standpoints (§8.1). The second part introduces present habitual and future forms (§8.2). The third part describes forms which mark ongoing action at a past or present time (§8.3). The various TAM-related uses of the possessive-like construction VERB-INF EX are addressed in (§8.4). The final section (§8.5), describes various modal forms expressing the speaker's assessments on certainty, permissibility, ability and obligation.

8.1 Past, completive and perfect forms

The different past constructions are summarized in Table 8.1. For simplicity, in the table *-tee* stands for *-tee/ze* (past marker) and *-po* for *-po/bo/u* (infinitivizer). The auxiliary copulas referred to in Table 8.1 are equatives \tilde{i} : and be^2 (EQU) and existentials $j\partial^2$, du^2 and (less frequently) $j\dot{e}bbe^2$ (EX).

Name	Form	Function
past	VERB- <i>tce</i>	past action
periphrastic past	VERB-po EQU	
	(dynamic verb)	
	VERB- <i>po</i> EQU	past or present state
	(stative verb)	
completive	VERB $-ts^ha$:	completed action
secondary verb	VERB mjồ:	having completed/finished or experienced
'finish'	(inflects like an ordinary verb)	the action marked by the primary verb
perfect	VERB(-RDP)- <i>po</i> EX	past action/state with present relevance
resultative	verb jø?	continuity of the results of an action
		(dynamic verbs), continuity of state
		(stative verbs)
sensorial	VERB du?	sensorially attested (action or its results)
past/present		present or past action/state
secondary	VERB za: EX	emphasizes the lasting effect of a past
resultative verb	VERB <i>zak-o</i> EQU	action
iterative past	verb- <i>po</i> verb-kjã: equ	iterativity
	VERB- $po(=l\varepsilon)$ VERB- $t\varepsilon im$ EQU	

Table 8.1. Past constructions

8.1.1 Past forms

Denjongke has two ways of expressing past tense, through the past verbal suffix $-t\varepsilon\varepsilon/z\varepsilon$ and through a periphrastic construction VERB-*po* EQU, where the verb root is infinitivized with -po/bo/u and then

followed by an equative copula. The equative copula is either \hat{i} : or $b\epsilon^2$, depending on how the speaker presents the situation evidentially (see §9). As shown in Table 8.1, the meaning of the periphrastic past construction is dependent on the nature of the verb. With a stative verb (e.g. ga 'love', $j\partial^2$ 'exist'), the construction may refer either to present or past states. Therefore the nominalizer/infinitivizer -po/bo/u is always glossed as an infinitive (2INF) and not as a past marker.

The suffix $-t\varepsilon\varepsilon/z\varepsilon$, on the other hand, is evidentially non-committed and does not need any additional morphology for finishing a sentence. For an example on $-t\varepsilon\varepsilon/z\varepsilon$, consider (8.1) and (8.2):

(8.1) দ' অদদ' ইদ'ন্ট' জি'ন।

yà dã: t^hõ:-tɛɛ óna. 1SG yesterday see-PST there 'I saw (it) yesterday, there.' (DB trip story)

(8.2) ซารา รัฐสาราวิ่า สาราวัง วิสามาร ภูราลารายา

 $p^{h}a$: $t\epsilon'\phi m - b\phi$: $g\tilde{a}$:=lo gagdza $kjap-t\epsilon\epsilon$ =la. over.there come.HON-2INF.GEN time=DAT obstacle do-PST=HON 'When (he/we) came over there, (they) obstructed him.' (CY interview)

In (8.3), the use of -tee imposes an eventive reading on the usually stative verb eé: 'know'.

(8.3) أَمَّعَ مِلَّى عَرَبَ عَرَبَ عَرَبَ عَلَيْ مَا عَلَى مَا عَمَرَ عَلَيْ مَا عَرَبَ عَرَبَ عَرَبَ عَرَبَ عَ nim súm nàŋca lɛngɛ:=ki k'adzø? cé:-tɛɛ, lɛkɛ? k'adzø? day three within PRN.HON=AGT how.much know-PST Lhoke how.much cé:-tɛɛ? know-PST 'In (these) three days, how much did you come to know (=learn)? How much Lhoke did (you) come to know? (YB restaurant discussion)

For the periphrastic past construction, consider (8.4) and (8.5).

(8.4) क्षें प्रदेश परी प्रा हे का रेंग के मा

 $\delta d\varepsilon m = di$ $\eta \dot{a}$: $t^h o$:-po \dot{t} :. like.that=DEMPH I.AGT hear-2INF EQU.PER. 'I heard (a thing) like that.' (KN e)

 p^ha : $t\epsilon'\phi m b\phi$: $g\tilde{a}$:=logagdza $manpu=t\epsilon i$?odeptiover.therecome.HON-2INF.GENtime=DATobstaclemany=INDFlike.thatkjap-o $b\epsilon$?. $b\epsilon$?.do-2INFEQU.NEequation be come over there, (they) obstructed him in many ways.' (CY interview)

Note that example (8.5) comes from the same speaker and piece of discourse as (8.2) and refers to the same situation. I am not aware of any other semantic difference between *-tce* and *-po be*? in (8.2) and (8.5) respectively, except the fact that *-tce* remains evidentially neutral by definition and

that the periphrastic construction -po be? is evidentially neutral by choice, i.e. because the neutral copula $b\epsilon$? is chosen instead of the personal copula \tilde{t} , which may also occur in this construction. In elicitation, speakers have not been able to describe any semantic difference between the verbal expressions in sentences such as (8.2) and (8.5). An extensive corpus study of the two forms would undoubtedly bring forth some results, but that type of undertaking is beyond the scope of this grammar.

The periphrastic past construction is also used in an idiomatic way to refer to imminent future:²⁸⁹

(8.6) مستهم منظر المربع (10% (10% - 10

Furthermore, it can also be used for irrealis reference in the apodosis of a conditional sentence:

(8.7) अ'र्तेट्र्या'र्यते'त' क्षट्ट' वर्ट्र' वर्ट्र' यार्क्र्य' अ'र्वेत'त' वर्ट्ट' विट्र'र्छ् गति' नर्ख्या'र्गे' गत्रट' आपत्र' वर्ट्ट' क्र्ट' केट्ट' वेत'र्दे' क्षट्ट màonpø:=na ke:=di jàrge? ma-t^høn-na=di $k^h \tilde{o}$:=tsufuture.GEN=LOC language=DEMPH proress NEG-happen-COND=DEMPH 3PL=PL zi tsuk-o $n\hat{a}$:- $k^{h}\tilde{\epsilon}$:=dinấ:mè? thøm-bo hE?. do.HON-NMLZ=DEMPH neglected become-2INF EQU.NE foundation plant-2INF 'If the language will not develop in the future, the foundation layers, they will have become neglected.' (KL BLA 12)

The verb in the periphrastic past construction may be reduplicated, see (8.8) and (8.9), although reduplication in this construction is rather infrequent. Reduplication emphasizes the resultativity of the action.

(8.8) गु'रु' देव'र्र'केंगीय' कॅंप्रदेन' नगात' गवटागवटया क्षेत्रा

 $guru rimputc^{h} \varepsilon = gi$ $\acute{o} d\varepsilon p$ ka $n \tilde{a}:-n \tilde{a}:-m$ $\tilde{t}:$. Guru Rimpoche=AGT like.that order do.HON-RDP-2INF EQU.PER 'Guru Rimpoche has said so.' (CY interview)

(8.9) दति' तुवा कें'ले' (Nep.) तदी' गलेत' मुनक' मुनक' रा क्षेत्र

pè: $n \dot{u}m$ *saili*²⁹¹=*di pén kjap-kjap-o* 1SG.GEN sister.of.a.woman third.daughter=DEMPH wedding do-RDP-2INF *ž:* EQU.PER 'My younger sister (who is the third daughter of my parents) is married.' (PED life story)

²⁸⁹ This use is analogous to Nepali clauses *mo gaẽ* 'I went' and *mo gaeko* 'I have gone', by which the speaker may signal her departure.

²⁹⁰ Or *gju*: [go.NMLZ]

²⁹¹ This is a loan from Nepali. The equivalent Denjongke expression is *nùm súmpo* 'third sister (of female)'.

Reduplication is more characteristic of the perfect construction VERB-RDP-2INF EX, which is introduced in §8.1.4. It is difficult to say what the exact semantic difference is between periphrastic (reduplicated) past $ka \ n\tilde{a}:-n\tilde{a}:-m \ \tilde{i}:$ and perfect $ka \ n\tilde{a}:-n\tilde{a}:-m \ j\tilde{\phi}$? forms.

The full nominalized form may also be reduplicated with the first instance in genitive case to add emphatic force to the statement, as shown in (8.10), where the emphatic nature of the clause is underlined by the presence of the contrastive emphatic =to and the conjunction $t'iz\tilde{a}$: 'but' in the following clause.

The periphrastic past form can be negated in three ways, by the prefix *ma*- (e.g. *ma*-*làp[-o bɛ?]*), by the prefix *mi*- (e.g. *mi*-*lap[-o bɛ?]*) or by negating the final copula (e.g. *làp-o mɛmbɛ?*). The prefix *ma*- is the most frequent, neutral way of negating a past action, see (8.11).

- (8.11)a) มหาสุราชา สัญสังสา สา สังขุázaŋ=gilóu=dimú:ma-tho:.maternal.uncle=GENspeech=DEMPH3SGF.AGTNEG-hear'She didn't hear the uncle's words.'(SN kitchen discussion)
 - b) אָזְקִישִי תּשָּי איצֿקיילי אָקן *lópţa gju ma-thop-o bɛ?.* school go NEG-receive-2INF EQU.NE '(She) did not get to go to school.' (PED life story)

Using the imperfective negator seems to force a past or present imperfective/continuous meaning on the clause, see (8.12) and (8.13). This construction is rare in my data,

(8.12)	क्षें पदी ह्यून न पदी रायम नहुवा वाया आववा वेयावा के हेंबा में छन्।								
	ódi	gã:	t'a	dikʰa ný:	làp-kʰɛ̃ː	lèm	mi-tʰɛm-bo	bE?.	
	that	time	now	here money	say-NMLZ	good	NEG-come.out-2INF	EQU.NE	
	'Now at that time there was not much money around.' (PED life story)								

(8.13) มิ'สุรารั มีรา

mi-nùp-o bε?. NEG-sink-2INF EQU.NE '(She) does/did not sink (under water).' (TB comment on a video)

Negating the final copula results in an emphatic type of negation, which is frequently used in the context of persuading. Of the six examples of this construction in my written data, in three the

²⁹² An innovative Denjongke spelling inspired by Dzongkha $\frac{1}{20}$ 'mind; conversation, speech, talk, word'.

speaker first expresses disagreement with the addressee by the negated interjection $m\tilde{\varepsilon}$; $m\tilde{\varepsilon}$: 'no, no' and then uses the emphatic negation at the end of the clause, see (8.14). Also in (8.15), the speaker counters what the addressee has said previously.

- (8.14)अत, अत, मंदी सेमागहसामु मेंत्रायां मागीया पार्थेण वर्म र्झ्यायां अत्र। mềː, mề:. mìntam=gi t'ønlo nè: ná:=gi NEG.EQU.PER NEG.EQU.PER 1SG.GEN reputation=GEN for.purpose.of 1SG=AGT p'ja-u mÊ. ió:=di work=DEMPH do-2INF NEG.EQU.PER 'No, No, I didn't do that thing for my reputation.' (nga'i 'gan 4)
- (8.15) สูง เพลส์ ราทิพ สีราจ์ รัสามีราวรุขามิราทิ ขุรมาวงศราร์ มสุ]
 gjalts^hen ná:=gi te^hø:=lo t'ønme? da:me:=ki tam eé-po
 PN 1SG=AGT 2SG.L=DAT vain void=GEN word tell-2INF
 mê:.
 NEG.EQU.PER
 'Gyaltshen, I didn't tell you vain and void words.' (nga'i 'gan 10)

Another context for emphatic negation is contrast with an adjacent affirmative clause, as shown in (8.16) and (8.17), where the negated clause occurs first, followed by the affirmative clause.

- (8.16) t'ato nàtea? k'are mềː. cù-wa tca:-bo t'ɛːlo now 1PL anything ask-PUR come.HUM-2INF NEG.EQU.NE just.like.that Ĩ. dze:-wa tca:-bo nàtea? t'arin nà: sər=tsa: today here sir(Eng.)=at meet.HON-PUR come.HUM-2INF EQU.PER 1PL 'Now we didn't come to ask for anything. We just came here today to meet (you) Sir.' (NAB BLA 7)
- (8.17) אָרָיאָר אָרָ בּיבָר אָלָי בּיבָר אָלָי אָאָן איינאָיאָזאָזי אָדָי אָאָן אייער אָזיאַזאָזישר אָזיאַזער אָזיגער אַזער אָזיגער אָזער אָזיגער אַזיגער אָזיגער אָזיגער אָזיגער אָזיגער אַזער אַזיגער אַזער אַזיגער אָזיגער אַזען אָזיגער אָזיגער אָזיגער אָזיגער אָזיגער אָזיגער אַזער אַזער אַזער אַזען אָזיגער אַזעער אַזער אַזער אַזער אַזען אָזער אַזעער אַזען אָזיגער אַזען אַזעגער אַזען אַזער אַזען אַזעגער אַזען אַזעגעער אַזעער אַזעער אַזעער אַזעער אַזעער אַזען אַזעגער אַזעער אַזעגעען איזעגעעןעען אַזעגעעןעען אַגעען אַגעען אַעגעעןעען אַגעעעןעען אַזעען אַזעגעעןעען אַגעעעעעןעען אַגעעעןעעןעען אַגעעעןעען אַגעעעןעען אַגעעעןעען אַגעעעןעען אַעען אַאָעען אַאָען אַאַען אַאַען אַיען אַאָען אַאָען אַאָען אַאָען אַאָען אַאָען אַאָען אַאַען אַאָען אַאָען אַאען אַאָען אַאָען אַאָען אַאַען אַאַען אַאַען אַאַען אַאַען אַאען אַאַען אַאען אַאען אַאַען אַאען אַאען אַאַען אַאָען אַאַען אַאַען אַאַען אַאַען אַאַען אַאַען אַאַעען אַאַען אַאַאַען אַאַען אַאַאַען אַאַען אַאַעען אַאַען אַאַען אַאַעןען אַאַען

The two negation strategies of using a prefix and negating the final copulas can also be combined to form a twice negated construction. Note that negated construction is followed, typically of emphatic negation, by an affirmative clause about the issue in question. (8.18)พรัพ สุราชิ่า ซาซี ซิ่า สัสาร์ เพชิยาที่พา ราชยา ผสุลาษัรลาที่) ซู้าร้า สิ่าริยาลาทั้ง สายสุราชา พลาซุรา t'ato sã:te tsotei nàtca? dendzon=gi lèmpu=tc^ho:=ki lopo now until chief minster=honorable=AGT 1 PL Sikkim=GEN Lhopo ma-nã:-bo jờ?. míri=lo mèmbe?. ke:po nã: za: a.lot give.HON people=DAT NEG-give.HON-2INF NEG.EQU.NE put EX.PER 'It is not that until now the Chief Minister has not given to us Lhopo people, (he) has given a lot.' (NAB BLA 7)

In addition to the aforementioned past constructions, past meaning can also be conveyed, if the context allows, by bare verb roots. This practice is usual in story-telling, as exemplified in (8.19) and (8.20).

- (8.19) جَ تَعْرَ عَرَصَعَ حَرَعَمَ حَمَّا حَمَّا مَعْ عَرَصَةً بِعَرَمَهُ اللَّهُ عَمَّا لَحَ مُعْ عَمَّا حَمَّا مَعْ عَمَا مُحْمَا مُحْمَا مُحْمَا مُحْمَا مُحْمَا مُحْمَا مُعْ عَمَا مَعْ عَمَا مَعْ عَمَا مُعْ عَمَا مُحْمَا مُحْمَ
- (8.20) ন্যু'পেযু' ঘাইন t'aca?

t'aca? zo. house.at.monastery build '(We) built a house (which is owned by the monastery).' (DB life story)

In addition to past, the bare verb root may also mark imperative (see §11.3) and steady state present (see §8.2.1).

8.1.2 Completive

The completive form $-ts^ha(:)$ derives from the Classical Tibetan verb sex tshar 'complete' and denotes a completed action. The completion of action is illustrated in (8.21) where $-ts^ha$: is contrasted with the past marker -tse. In (8.21a), the speaker completed reading the whole book, while in (8.22b) the speaker finished an act of book-reading but did not necessarily read the book to the end.

(8.21) a) अन्न न्या नेया यहेवा र्झेवा करा

 $d\tilde{a}$: $\eta \dot{a}$: $t' \varepsilon p = t \varepsilon i 2$ **dok-ts^ha**:. yesterday I.AGT book=INDF read-CMPL 'Yesterday I finished reading a book.' (NB e)

b) aff friction fr

Although $-ts^ha(:)$ has grammaticalized into an independent marker of completion that can end a sentence (8.22), it may be followed by an equative copula (8.23) or the existential copula du^2 (8.24). Moreover, $ts^ha(:)$ resembles an ordinary secondary verb in that it occurs in nominalized periphrastic constructions as $-ts^ho-u$ (8.25).

(8.22)nàtci jà:p †'õ:-ti lò ní:-ts^hø? lep-tshaz. 1PL.GEN father.HON die.HON-NF year two-about reach-CMPL 'Some two years have passed since our father passed away.' (Richhi 35) 5. र्हे. ट. ट्रे. पर. (driver)293 ईव. र्कर. भुना (8.23) t'ato t'a draivər thøn-tsha: be?.²⁹⁴ now then driver(Eng.) become-CMPL EQU.NE 'Now he's become a driver.' (KN e) (8.24)षुः तेंन्ज्रं तर्ग्यागो k^hu òn-ts^ha du-k_E. 3SGM come-CMPL EX.SEN-IN 'He's arrived.' (KN e) (8.25)a) ମ'ର୍ହ୍ମ ସ୍ୟସ୍ୟ ତମ ସା'ନି' ଛ୍ଲିମ୍ୟ'ହିଁ୍ୟ କ୍ରମ୍ *bε*?. t'ato t^hamtce? gari lep-tsho-u now all car reach-CMPL-2INF EOU.NE 'Now everything has arrived by car.' (RBM discussion on the roof) b) ନି:=୮ ୮.୭ନି: ଏନି: ଖିଁ'ଏନି: ଖ୍ମ୮. ୮ ଶି'ସନ୍ତମ୍ବା ଏସଂଞ୍ଚି: ସନ୍ଦମାଞ୍ଚି:ମି' ଏନି' ଦିଁ' ସନ୍ତ: ସଶ୍ୟ ସ' ଏ୮ ଛିଁପ୍ର' ଖ୍ରମ୍ବ t'izã: nàtci=di ódi gã: t'a mi-teu? làp-ti but 1PL.GEN=DEMPH that time now NEG-insert say-NF *zak-tiki=di* lò teu-thamba lã:-tsho-u leave.aside-NF=DEMPH year ten-NUM come.up.to-COMPL-2INF $b\epsilon$?. EQU.NE 'But at that time our (parents) said (we) do not put (you to school) and leaving (me outside school) I reached (the age of) ten years.' (PED life story) c) यियाता प्राणकरांचा खेतातवा? ра́т? ²⁹⁵ *k*^{*h*}*im*=*na lòk-ts*^{*h*}*o-u* house=LOC return-CMPL-2INF EQU.ATTQ 'Has he returned home, I wonder?' (Richhi 24)

²⁹³ The equivalent word in Denjongke given by consultant YR is אָאָא'קעָלָד. קאָאַיאַקעָל *num-khor ky:-kh*ɛ̃: [oil-wheel drive-NMLZ].

²⁹⁴ The original utterance has the verb ending $t^h \omega n - ts^h a - k\varepsilon = \varepsilon o$ [become-CMPL-IN=AT] but the consultant also re-uttered the clause with $t^h \omega n - ts^h a b\varepsilon$?

²⁹⁵ This pronunciation represents typical spoken language. Reading-style pronunciation would be *lok-ts^ha:-bo*.

t'o? bak-ti ódɛtɕika di=tsu tε t'a námo then that.time now camel these=PL load carry-NF t'vts^hø? [ep-ts^ha-k^hen gju-wø:=gi, gju-ce=ki bE?. going-2INF.GEN=GEN go-INF=GEN time arrive-CMPL-NMLZ EQU.NE 'Now then at that time, the time had come for these camels to go, to go carrying loads.' (PD bet story)

In (8.26), the completive occurs with the past suffix $-z\varepsilon$. The form $-ts^{h}ou$ is tentatively glossed as a secondary verb 'finish' without nominalization, because adding the past marker $-t\varepsilon\varepsilon/z\varepsilon$ to a nominalized form would be the only such example in my data.

(8.26) ອຼັສຸ ເສັດເອົາ *tc 'øn tshou-ze*. go.HON finish-PST '(They) already went.' (KT e)

In one instance in my data, the completive is followed by a morpheme which looks like the imperfective marker *-to* but which probably is the contrastive emphatic =to (see §16.1.2).

र्च्या अल त्या छे. ट.जू. ट्र. प्र्राज. यहट छे. टपु. क्रैय जिया र् याद्र्या. र् याद्र्या. र् याद्र्या. र् (8.27) $p'ja:=l\varepsilon$ tã:-di nè: òtε nà=lo do di: gjapk^ha precipice=ABL down 1SG=DAT stone fall send-NF 1SG.GEN back do=tci? phok-tsha=to=lo. stone=INDF hit-CMPL=CEMPH=REP '(You) sent a stone-fall from the precipice towards me and a stone fell on my back, (he said).' (UU deer story)

For evidential distinctions of the various constructions with $-ts^{h}a(:)$, consider §9.1.3.

8.1.3 Secondary verb mjo2 'finish, experience'

The verb $mj\tilde{o}$: $mj\tilde{o}$: $mj\tilde{o}$: finish' at the end of a SVC expresses that the action depicted by the SVC has ended or, less frequently, that the actor has experience of the action in question (also implying that the action has been completed). The former use has probably developed from the latter, as suggested by the meaning of the cognate WT $mj\tilde{o}$: myong 'experience'. It is likely that after having developed the meaning 'finish, complete' $mj\tilde{o}$: has pushed the earlier verb ts^ha : with a similar meaning 'complete, finish' from ordinary verbhood towards becoming a grammatical completive marker. The secondary verb $mj\tilde{o}$: 'finish' differs from the completive morpheme ts^ha : in that in addition to the serialized construction $mj\tilde{o}$: can occur in a nonfinal converbal construction (with little difference in meaning) whereas ts^ha : cannot, e.g. $sa-ti mj\tilde{o}$: 'finished eating', * $sa-ti ts^ha$:.

The use of $mj\tilde{o}$: referring to finished action is illustrated by (8.28) and (8.29).

(8.28) वयया रुट्र र्युया र् हेंट श्रदा

t^hamtcε? **p'ja mj**δ̂: bε?. all do finish EQU.NE '(We) finished doing all.' (DB life story) (8.29) هَرْعَ بِيَرْسَعَ تَوَلَّمَ عَلَى اللَّهُ عَلَى اللَّهُ عَلَى اللَّهُ عَلَى اللَّهُ عَلَى اللَّهُ عَلَى اللَّهُ اللَّهُ (He) finished taking education there.' (CY interview)

Examples (8.30) and (8.31) illustrate the use referring to experience, which in my impression are more frequent in negated clauses such as (8.31).

- (8.30) 就方可 되知知'る「 「「「「「「「「「「」」」」」」」
 (8.30) 就方可 「 日本知()る」「「」」」」
 (8.30) 就方可 「 日本知()る」「 「」」」
 (8.30) 就了 「 日本知()る」「 「」」」
 (8.30) 就了 「 日本知()る」「 「」」」
 (8.30) 就了 「 日本知()る」「 「」」
 (8.30) 就了 「 日本知()る」「 「 日本和()る」「 」」
 (8.30) 就了 「 日本和()る」「 」」
 (8.30) 就可 「 日本和()」
 (8.30) 就可 「 」」
 (8.30) 就可 「 」」
 (8.30) 就可 「 」」
 (9.30) 就可 「 」
 (9.30) 就可 「 」」
 (9.30) 就可 「 」」
 (9.30) 就可 「 」
 (9.30) 就可 」
 <
- (8.31)t'atø: t'yts^hø? nànca=lo kjap-k^hẽ: tho: ty:lu ódem ná: now.GEN season inside=DAT ode like.that strike-NMLZ 1SG.AGT hear ma-mjőz. **NEG-experience** 'At the present time, I have not heard (people) who sing odes like that.' (KT intro to an ode)

8.1.4 Perfect

In the perfect construction VERB(-RDP)-po/bo EX the verb is usually reduplicated (8.32) but occasionally non-reduplicated (8.33). By using the perfect construction, Denjongke speakers suggest that the action/state or its results continue until the time of speaking and have present relevance.²⁹⁶ In (8.32), the statement has present relevance because the speaker is going to reclaim an old loan.

nè: ápo=di ódep cìn-le: kjap-tiki ódep p^hjuku p^hjuku like.that field-work do-NF like.that rich rich 1SG.GEN father=DEMPH p^hjuku jà-pø: $tsa=l\varepsilon$ gã: ran=gi ápa=gi ápø: nè: EX-2INF.GEN time 2SG.M=GEN father=AGT 1SG.GEN father.GEN at=ABL rich tiru? dvntcu-t^hamba bak-sõ:-sõ:-bo i*ø*:=s làp-o=lo. seventy-NUM carry-go-RDP-2INF EX.PER=QUO say-2INF=REP rupee 'At that time when my father got rich, very rich doing work in the fields, your father took (in loan) seventy rupees from my father, (he) said, so the story goes.' (PD betting story)

²⁹⁶ In Jespersen's (1924: 269) definition, perfect "represents the present state as the outcome of past events, and may therefore be called a retrospective variety of present."

- (8.33)مكَرُ المعر الأرامة المحتمد ا सु' केंग भ्रयायवेव र्घुग र्थेट र्ये थेंटा dile ódi gjable nà te t'ato nà: jurop po:len làp-sa then that after 1SG so now here Europe Poland say-NMLZ.SPAT pámpu te^ho? t&hunteun=tei? nànca ŋà ť y:ku inside 1sg reincarnated.teacher small=INDF with doctrine lap-zin-p'ja dø:-po j*ò*?. teach-PROG-ADVZR stay-2INF EX.PER 'Then after that I have been staying here in Europe, Poland, teaching a small tulku (reincarnated Buddhist teacher).' (RB life story)
- (8.34) جريح بيج بيج بيج بيج بيج بيج المراجع المراحي المراجع المراجع المراجع المراجع المراجع
- (8.35)
 (8.3

didom=nat'i-ti-ujèbbe?.thisbox=LOCwrite-RDP-2INFEX.NE'It's written (something) in this box.' (TB e)

The perfect construction may be negated in two ways, by simply negating the final copula (8.36-37) or by the special negated construction VERB-*Ey*: NEG.EX (8.38-39).

- (8.36) a) 두 두 흙 국자 한 국자 한 국자 한 국자 한 국자 한 국자 (basketball) 홍두 홍두 节 බ도) ŋà t'ato sā:tɛ baskɛtbol tsi-tsi-u mè?. ISG now until basketball play-RDP-2INF NEG EX.PER 'I haven't played basketball so far (in my life).' (KN e)

 - c) קיזרי שריאי הישיו יויזי מיתי משאילי ארן

t'arin sà:te nàtea? k'are cù-wa tea:-bo mè?. today until 1PL anything ask-PUR come.HUM-2INF NEG.EX.PER 'Until today we haven't come to ask for anything.' (KN e)

d) अत्रे.सु. र्रेग्स.र्ह्स ८ र्ग्स गत्र.र्ट् गहार र्ग सेर्ग ग?

 $n\dot{\epsilon}:mu$ ro:=tsu $n\dot{a}=lo$ $k'and\epsilon:$ $s\dot{u}m-bo$ $m\dot{\epsilon}:-ka?$ really friend=PL 1SG=DAT anything say.HON-2INF NEG.EX.PER-PQ 'Haven't the friends really told anything to me?' (Ricchi 69) As shown by (8.36c) the non-reduplicated negated perfect construction can occur with a perfect meaning. The non-reduplicated construction, however, is also used in a future-oriented sense expressing lack of permission, intention or ability, see (8.37). Example (8.37b) leaves unclear whether the speaker expresses lack of ability or just lack of intention.

(8.37) a) ८ हारी र र प्युर्भे केन्

 $y\dot{a}$ $k^{h}u=i=tsa$: **gju-wo mè?**. 1SG 3SGM=GEN=by go-2INF NEG.EX.PER 'I can't go to his place (e.g. because we are not in good terms).' Lit. 'I have no going to his place.' (KUN e)

b) ה׳ אֹ׳אִ׳אָ׳אָ׳ אַקָּאָ׳ אַקן *ŋà siliguru gju-wo mè?*. 1SG TPN go-2INF NEG.EX.PER 'I have no going to Siliguri' (KUN e)

The negated form illustrated in (8.37) is also used for negating progressive, continuous and imperfective constructions, see §8.3.

The second way of negating the perfect construction is to use the morpheme *-cy:*, which etymologically likely derives from WT $\operatorname{gree} shul$ 'trace, remains'. Therefore *-cy:* is here tentatively glossed as 'trace', which fits the clausal meaning. In the novel Richhi, both the forms $\operatorname{gree} shul$ and $\operatorname{gree} shus$ occur in writing.

- (8.38) ה׳ ה׳ אָדָ אָר אָדָ בְּאַי אָקוּי אָדָ יאָרי אָדָ יאָרי אָדָן *nà t'ato sã:tɛ baskɛtbol tsi:-ɛy: mè?.* 1SG now until basketball(Eng.) play-trace NEG EX.PER 'I haven't played basketball so far (in my life).' (KN e)
- (8.39) للَّا تَوَادُ عَرَضَ المَّامَ حِتَامَ²⁹⁷ عَلَّمَ بَجُ العَرَمَجُ عَلَى العَرَمَجُ عَلَى العَرَمَةُ العَرَيمَةُ العَرَمَةُ العَرَضَةُ العَرَمَةُ العَرَضَةُ العَرَمَةُ العَرَمَةُ العَرَضَةُ العَرَض (Richhi 164)

The construction VERB-*cy*: NEG.EX is further illustrated in the question-answer-pair (8.40):

²⁹⁷ There is most likely a spelling-mistake here, the right spelling being upor super-

²⁹⁸ Morpheme analysis here follows the written form in the novel Richhi, which often has a genitive preceding the locative case. Following this analysis, here three cases are stacked together (GEN=LOC=DAT). An alternative would be to analyze *nalo* as a postposition meaning 'inside'. For case-stacking, see §3.7.1.3.

b) אָיזָר אָאי לַמי לַמי לַמי לַמי אָרן *t'aruŋ lèm-p'ja t'ak-ey: mè?.* yet good-ADVZR be.cured-trace NEG.EX.PER 'He hasn't recovered well yet.' (Richhi 26)

The positive construction VERB- ϵy : EX is rare in my data. The only instances in my data are the question answer pair (8.41) and clause (8.42) from the novel Richhi.

> $r\tilde{a}$: $n\dot{a}m=lo=j\tilde{a}$: $siliguri \ lep-tee-ga$? 2SG.M when=DAT=even TPN arrive-PST-PQ 'Did you ever go to Siliguri?' (NAB e)

b) ה' אָלאזי לָקא' עָרָן *ז`מ lep-cy: jò?*. 1SG arrive-trace EX.PER '(Yes) I have gone (there).' (NAB e)

वर्ना र्यते गविषा झेर्रे क्वा के मुबा लेंदा भर्ता नर्ने पर्य गाँग गोषा वर्ता र्या पर्ने क्वा देखी पहर मुबा पर्य (8.42)tshaptce-cyr299 **iò:-**cε=di: nè:pø: né:tã: karma=gi patient.GEN condition become.severe-trace EX-INF=DEMPH.AGT PN=AGT nè:po=di=lo dilli tã:-ce? p'ja-z $\tilde{\epsilon}$: iờ?. patient=DEMPH=DAT TPN send-INF do-prog EX.PER 'Because the patient's condition had signs of severeness, Karma is attempting to send the patient to Delhi.' (Richhi 169)

8.1.5 Resultative

In harmony with its copular function, $j\partial^2$ following a verb root marks the present continuity of a state. With dynamic verbs, see (8.43-45), this implies that the state is a result of an action, hence the name resultative for this construction. Resultative forms are in meaning very close to the perfect construction described above. Examples (8.43) and (8.44) include both affirmed forms (a) and negated forms (b).

(8.43) a) $\operatorname{sgr}_{\mathfrak{sgr}}^{\mathfrak{sgr}} \operatorname{sgr}_{\mathfrak{sgr}}^{\mathfrak{sgr}} \mathfrak{sgr}_{\mathfrak{sgr}}^{\mathfrak{sgr}} \mathfrak{sgr}_{\mathfrak$

²⁹⁹ Consultant KN (Martam) suspected that there might be a mistake here because in his language variety *-sy:* would be used here instead of *-cy:* as a diminutiave 'a bit serious'.

³⁰⁰ Some spell this word बे'र्ड्ते' instead of हान्'हुन्'.

(8.44) a) ร สิรง นัร

yà lɛp jò?. 1SG arrive EX.PER 'I have arrived.' (KN e)

b) ८ सेन् होन् राज्य के	/	<u>୮</u> . ୩.ଞ୍ଚିଅନ୍ୟା
ŋà Įɛp mè? .		ŋà ma-ļɛp.
1SG arriveEX.PER		1sg NEG-arrive
'I haven't arrived.'		'I did not arrive.' (KN e)

Note that $j\partial i$ may occur in a complex construction, as exemplified by $j\partial i \varepsilon \varepsilon b\varepsilon i$ in the irrealis/future in (8.45).

With stative verbs, the meaning focuses on continuity:

(8.46) هَرْجَمْتِ هُمَرْخَرْ جَمْمَرْخَرْ جَمْمَرْخَرْ مَعْرَبْ عَلَى اللَّهُ اللَّهُ الْمَرْجَمْتُ عَلَى الْمُحْرَفْ الْمَرْجَمْتُ الْمُحْرَفْ الْمُحْتَى الْحَدَى الْمُحْتَى الْحُمْتَى الْحَدَى الْحُدَةَ الْمُحْتَى الْمُحْتَى الْحُمْتَ الْحَدَى الْحُدَى الْحُدَى الْحُدَى الْحُدَى الْحُمْتَ الْحَدَى الْحُدَى الْحَدَى الْحُدَى الْحَدى الْحُدَى الْحُدَ مُعْلَى الْحُدَى الْحَدَى الْحُدَى الْحُدَى الْحُدَى الْحُدَى الْحَدَى الْحُدَى الْحَدَى الْحَدَى الْحَدَى الْ

According to consultant KN, (8.46) can be negated by replacing a negated copula for the affirmed one, i.e. $\epsilon \dot{a}$: $m \dot{e}$?

When used in the resultative construction, the verb $d\phi^2$ 'sit' is ambiguous as to dynamic ('have taken a seat') or stative reading ('are sitting') but nevertheless marks the continuing state of sitting:

(8.47) المحت الم المحت الم المحت (Richhi 18) (Richhi 18) المحت (Richi 18) المحت ال

³⁰¹ The colour words in this song occur in disyllabic Tibetan-style forms instead of the typical monosyllabic Denjongke form (*sé:p, má:p, ka:p*) probably for poetic and rhythmic reasons.

8.1.6 Sensorial resultative/past

The construction VERB du^2 , which is rather infrequent in my data, expresses sensorial resultative and sensorial past meanings. As shown in §7.2.2 and §9.1.2, sensoriality refers to the fact the speaker bases a proposition on a sensorial experience, typically visual. The difference to a similar construction with $j\partial^2$ is that whereas $j\partial^2$ implies that the resulted state continues at the time of speech, du^2 only makes reference to an event where knowledge was gained and remains uncommitted as to whether the state is still ongoing. A construction with du^2 only implies that the state-of-affairs held at the time of observing. In examples (8.48-52), where the time of observing coincides with the time of speaking, the construction is resultative, i.e. marking a state achieved by the verbal action.

(8.48) দ্বিদ রী এবা ঘারি এই রক্ষা এন্যা t'ariŋ mila? zi dzom du?. today person four gather EX.SEN 'Four people have/are gathered today, I see.' (PTB e)

According to consultant KN, (8.48) cannot be negated by just replacing the affirmed existential with a negated one. Negation strategy is adopted from the perfect construction (see §8.1.4):

(8.49) ק־דֹּבִי אֵׁישִׁקִי הַפְּ תַצָּׁאַשִּידִׁי אָק־תַקַיַקן *t'ariŋ míla? zi dzom-bo mìndu?.* today person four gather-2INF NEG.EX.SEN 'The four people haven't gathered today, I see.' (KN e)

The copula may, however, be negated in a construction with $(p^h a)g\varepsilon$ 'except', see (8.50). The meaning corresponds to English *more than* accompanied by a negated verb or the English *only* followed by an affirmative verb.

(8.50) ק־דֹבִ מَישִׁשִן שִוֹם (שִי)שִוֹ בּוֹשָׁמִשִיבָׂ מּמִיבָּ שָׁמָשִיבָּ מֹז מִקָרָשָן t'arin mila? zi (pha)ge dzom mindu?. today person four except gather NEG.EX.SEN 'Today there aren't (any) more than four people gathered, I see.'/'Today only four people have gathered, I see.'/ (KN e)

For another example of resultative use, consider (8.51) with an affirmed (a) and a negated (b) clause.

(8.51) a) अर्ळें जु. पहत. जुया. स्रंट. पर्या

 $ts^{h}o = na$ dza suk $s\delta$: du?. lake=LOC rainbow pierce go.PFV EX.SEN 'A rainbow is touching (or: has come and touched) the lake.' (DB, describing a picture)
b) addition defined a definition of the set of the se

Note that in the negated version (8.51b) the secondary verb $s\hat{\sigma}$ is elided and the main verb may occur with the nominalizer or without.

The knowledge on which the statement with the construction VERB du^2 is based can be gained either through direct observation of the verbal event, as in (8.48-51) above³⁰², or through observing the results of past action, as in (8.52).

अन्न विम्युत्तार्ये मारुवा सेंदि वर्णियो भीतेका हूर्ये स्था वर्ष्या वर्ष्वती पर्धि उत्तवी जूर्य क्रियो क्रियो क्र (8.52) $k^{h}a:nu:=lo$ nàtea? p^hou dã: jó? yesterday the.day.before.yesterday=DAT 1pl over.there work $g\tilde{a}$: \dot{a} :m=tci2³⁰³ kjap de:-pø: t^hu bak lő: số: du-ke do stay-2INF.GEN time jackal=INDF pick carry rise go.PFV EX.SEN-IN 'The other day, when we were working over there, a jackal came and carried (the hen) away.' (PL interview)

The use of the sensorial du- $k\varepsilon$ in (8.52) is based on visible evidence of the event's results (a dead, half-eaten hen in the forest), not the event itself. That is, the evidence against the jackal is only circumstantial, not direct. Here it is worth noting that although some linguists (e.g. Hengeveld & Olberz 2012: 495, DeLancey 2012: 540) underline the fundamental difference between direct perception and indirect perception (or inference from the results of an action), Denjongke uses the sensorial du? for reporting both direct evidence of seeing an action and indirect evidence of seeing the results of an action. In both cases, something is sensorially perceived, and thus both instances can be marked with the sensorial du? For the close connection of sensorial and inferential in Lhasa Tibetan and several other languages, see Hill (2017).

In contrast to (8.48-52), where the sensory experince coincides with the time of speaking or is a recent one, example (8.53) illustrates a reference to a past sensory event. The speaker describes an act by a historical figure about whom he has gained knowledge from a written document or by word of mouth. As in the previous example, in (8.53) the use of du^2 is not based on direct evidence of the depicted action but on written or spoken secondhand reports.

(0 = 2)	\sim	\sim		~		\sim
(0.33)	বেদ্র'যাঝ'	হার'এম'	25'	বদবাধা	শারন'	2241.MI
()	1.1.1.1.1	21-12			· / · / ·	. // .

$k^{h}o\eta = gi$	mélam	$d\varepsilon$:	tap-o	nấ:	du-kɛ.
3SG.HON=AGT	prayer	like.this	sow-2INF	do.HON	EX.SEN-IN
'He prayed like	this.' (KI	T Bume	hu video)		

 $^{^{302}}$ Strictly speaking, in (8.51a) the speaker does not claim to have seen the movement of the rainbow onto the lake (as suggested by the verb 'go'). The speaker claims to see or have seen the state resulting from the movement of the rainbow onto the lake.

³⁰³ According to consultant KT, the lack of agentive and overt patient argument in this clause makes the jackal appear to be, on the clausal level, the patient and not the doer of the action. The context, however, makes clear that the jackal is the agent and a hen the patient.

According to consultant KN, (8.53) can be negated by replacing the affirmative existential with a negated one:

(8.54) $[t = 1] = 10^{-10} \text{ (8.54)}$ $[t = 10^{-10} \text{ (100)}]$ $k^{h}o\eta = gi$ $m \ell lam d\epsilon$: tap-o $n \tilde{a}$: $m ln du-k\epsilon$. 3SG.HON=AGT prayer such sow-2INF do.HON NEG.EX.SEN-IN 'He did not pray such a prayer.' (KN e)

The auxiliary du^2 may also be used as a story-telling technique, where the speaker invites the addressee(s) to become part of the scene by observing events in real time, see (8.55).

(8.55) طع محد عَمْلَ مَعْرَبُ عَلَيْمَ مَعْرَبُ عَلَيْمَ اللَّهُ عَلَيْ عَلَيْ عَلَيْ اللَّهُ اللَّهُ عَلَيْ عَلَيْ عَلَيْ اللَّهُ اللَّهُ عَلَيْ الْعَلَيْ اللَّهُ عَلَيْ الْعَلَيْ الْعَلَيْ الْعَلَيْ اللَّالِي الْعَلَيْ اللَّهُ عَلَيْ اللَّهُ عَلَيْ اللَّا اللَّالَةُ اللَّهُ عَلَيْ الْعَلَيْ اللَّاللَّهُ اللَّالِي الْعَلَيْ اللَّهُ عَلَيْ اللَّهُ عَلَيْ اللَّا عَلَيْ اللَّهُ عَلَيْ الْعَلَيْ الْعَلَيْ اللَّا اللَّالَةُ اللَّا اللَّا اللَّا اللَّهُ عَلَيْ اللَّا اللَّا اللَّالِي اللَّا اللَّا اللَّا اللَّا اللَّ

As suggested by uses in (8.51), (8.52) and (8.55), the construction VERB du^2 is particularly common with the suppletive verb $s\delta^2$ 'went'.

8.1.7 Resultative secondary verb *zak*

The secondary verb *zak/za?* (often *za*.) 'set, place, put' (WD (WD (WD (WD))), which also occurs as a primary verb, may follow a primary verb to emphasize the resulting state and lasting effect caused by an action, as illustrated by the imperative construction in (8.56).³⁰⁴ Note that *zak* does not have the meaning "improperly" or "to deleterious effect" as its cognate serial verb in Lhasa Tibetan (DeLancey 1991: 9).

(8.56) $\tilde{\mathfrak{H}}^{a'} \tilde{\mathfrak{G}}^{c'} \operatorname{dom} p^{h} i za?.$ door open put 'Leave/keep the door open.' (PT e)

In the indicative mood, the secondary verb *zak* can occur both in the resultative construction VERB EX, see (8.57) and (8.58), and the past periphrastic construction VERB-*po* EQU, see (8.59) and (8.60).

³⁰⁴ This frequent construction resembles the Nepali verbal forms supplemented by $h\bar{a}lnu$ or $r\bar{a}khnu$ 'put', e.g. *bhan-i-rākh-nu* [say-LNK-put-INF] 'to say'.

- (8.59) $\forall x \in [n] \forall x \in [n] \forall x \in [n] \forall y \in [$
- (8.60) מוֹשָּׁמוֹ מִרְ' הַוּקר' הַוּמְזוּ וּאָמִ הַבְיוֹמן $t_{ci:}=di$ $t\tilde{a}:$ zak-o $\tilde{t}:$, $paŋk^ha$. one=DEMPH send put-2INF EQU.PER outside 'One (hen) I sent out (free), outside.' (PL interview)

The construction may be negated by prefixing the negator ma- to the secondary verb.

(8.61) $[nderivan] \leftarrow nderivan] \leftarrow mathematical and and an equivalent and the mathematical an$

The resultativity may be stressed by reduplicating *zak*.

(8.62) $\int \vec{x} \cdot \vec{x}$

Like many other complex verbal expressions, the construction VERB *zak* EX likely derives from converbal construction from which the converb morpheme has been dropped:

(8.63) ह्युत्र राषा यस्त्र हे रावया थेंना

mémpa: súg-di za: jô?. doctor.AGT say.HON-NF out EX.PER 'The doctor has said...' Lit. 'The doctor has by saying placed' (Richhi 167)

8.1.8 Iterative past

Iterativity and intensity can be marked with two constructions which are both sound symbolic in that iterativity in meaning corresponds to iterativity of form.

8.1.8.1 Iterative with *-kjã*:

The first construction, VERB-*po* VERB-*kjã*: EQU/*p'ja*, is formed with the help of $=kj\tilde{a}$; which is an alternative form of the more frequent additive clitic $=j\tilde{a}$: 'too, even, again'. The construction may

³⁰⁵ This word refers to a sticking glue-like sap from a certain tree.

occur in a finite clause (followed by an equative auxiliary), see (8.64-65), or in an adverbial clause followed by the verb/adverbializer p'ja 'do', see (8.66). In (8.64), the meaning is clearly iterative. In (8.65) the action is not iterative but extends over a long period. In (8.66), it is not clear whether the action is continuous or consists of intermittent bursts.

- (8.64)สู'มิ่นางขุลา ขุดิลารันา มีหาลา ลางมาที่ลา นาลา รู้าร้า ขุลนาร์า ขุลนามีนา ผิลา p'usim=la: korlo áma=gi $\eta \dot{a} = lo$ ní:pø: f'iwa younger.sister=HON two-ORD.GEN about mother=AGT 1sg=dat question nã:-bo náŋ=kjã: ĨĽ. give.HON-again EQU.PER give.HON-2INF 'The mother has been keeping on asking me about the second sister.' (Richhi 27)
- (8.66) $\xi: \overline{\chi}: \xi: \overline{\chi}: \overline{\chi$

In (8.67), the iterativity concerns several different undergoers, i.e. several different people have died.

(8.67) बे' मे' मे' मे' मु'गुर' भ्रता

mí εί-u εί-kjõ: bε?. human die die-again EQU.NE 'People died and died.' (KN e)

8.1.8.2 Iterative with -tcim

The second construction with which iterativity and intensity may be marked is VERB- $po(=l\varepsilon)$ VERBteim EQU. The formative -teim is of unknown origin and is in (8.68-69) preliminarily glossed simply as a nominalizer.³⁰⁶ In the emphatic construction (8.68), the same verb root occurs thrice. The iteration refers to different undegoers (i.e. several people died), not to one person undergoing the experience again and again (i.e. one person died many times).

(8.68)	תבי פאאישבי	ने-चे-गो	প্রি'র্য'শব্দ	ਸ਼੍ਰੇ ਤਿਸ	<u>95</u> 1
(0.00)	1	T R ''	1 N N	4 001	コン

	1 1 2 1	1 1	- 11						
kħõ:	t ^h amtse?	cí-tiki	¢í−u=le	cí-tcim	bE?.				
3pl	all	die-NF	die-2INF = ABL	die-NMLZ	EQU.NE				
'They all died and died.' (PB discussion with TB)									

³⁰⁶ A possible origin of *-tcim* is nominalization of the progressive form *-zen/tcen* (*tcem-bo*) where the nominalizer has reduced to *-m*.

Whereas in (8.68) -*tcim* is followed by an equative copula, (8.69) shows that the nominalized construction ending in -*tcim* may also be followed by the verbalizer p'ja.

(8.69) ଷ୍ଟ ଷ୍ଟ୍ର ଜ୍ୟା ମୁମଷାଧିଁ ମୁମଷାତିଷା ପ୍ରଷାଧିଁ କ୍ଷମ

mù úpa? kjap-o kjap-teim p'ja-u bɛ?. 3SGF force strike-2INF strike-NMLZ.FEM do-2INF EQU.NE 'She kept on forcing (one to do something).' (KT e)

8.1.9 Note on the Sandberg's (1895) past forms

The old variety of Denjongke recorded in Sandberg (1895) employs forms I have not come across in my data. One of these forms is the "past indefinite" tense formed with the help of the secondary verb số: 'went'. One of Sandberg's (1895: 42) examples of the past indefinite is "He has written a letter: *Kho yige chi p'i song du'*," corresponding to $k^h u$ *jigi tçi? p'i³⁰⁷ sõ: du?* [he letter=INDF write went EX.SEN]. According to Sandberg (1895: 42), the verb ci 'die' "always forms the past tense with song." He gives the examples shi song 'he died', shi song zhe 'has (quite) died, is dead' and shi song du' 'did die (emphatic)'. In my data, however, these forms are supplanted by the completive $\epsilon i ts^h a$: 'has died' and periphrastic past $\epsilon i u \tilde{i} / b\epsilon 2$. The form $\epsilon i s \tilde{o} - z\epsilon$ (presumably corresponding to Sandberg's shi song zhe) was reported by consultant KN to have a purposive meaning equivalent to $\epsilon i \cdot wa s \delta z \epsilon$ 'went to die'. The form $\epsilon i s \delta du$, on the other hand, was reported by consultant TB to be a curse-like wish, probably something in the effect 'let him die'. For past tense of p'ja 'do', Sandberg (1895: 49) lists Zhe song, a form which seems to combine the Central Tibetan *tee* 'do' with the secondary verb $s\delta$ ' 'went'. In my data, p'ja employs the same past forms as other verbs, e.g. the perfective past $p' ja - z\varepsilon$, periphrastic past form $p' ja - u \tilde{i} / b\varepsilon^2$, and the completive p'ja-ts^ha:. It is noteworthy that Sandberg does not record the completive form $-ts^ha(:)$ at all, suggesting that this form may be a later development.

8.2 Present habitual and future forms

As shown in Table 8.2, there are five constructions that express general facts holding in the present and three constructions that express future. The nonpast construction is used for both.

Name	Form	Function
steady-state present	VERB	present habitual
simple present	verb be?	
present habitual I	VERB- $k^{h} \widetilde{e}$: EQU	
present habitual II	STATIVE.VERB-po EQU	
nonpast	VERB- <i>ce</i> EQU	
		future
future	VERB <i>ò</i> :	-
imminent future	VERB-rap EQU/EX	'be about to'

Table 8.2. Present habitual and future constructions

³⁰⁷ The language variety recorded by Sandberg (1895) has $/p^{\circ}i/$ 'write' for what most speakers nowadays have $/t^{\circ}i/$. I have heard that the form $/p^{\circ}i/$, which is more faithful than $/t^{\circ}i/$ to the general tendency of Denjongke to correspond WT /br/ with /py/, is still used in East Sikkim around Rhenock.

The present and future forms are here discussed in the same order as they occur in Table 8.2. In addition to the productive forms presented in Table 8.2, some infinitival constructions are formed with $-ni/n\epsilon$, which resembles the Dzongkha infinitive -ni (van Driem 1998: 338). These infinitival constructions are discussed in §8.2.8. The imperfective -to/do, which is introduced in §8.3.1 below, may also express immediate future.

8.2.1 Steady state present

In the steady state present tense, a bare verb root expresses an ongoing state.³⁰⁸ The verb is usually stative, as in (8.70) and (8.71), but can also be a dynamic/eventive one which expresses habituality, as in (8.72) and (8.73).

(8.70) حمر أحمر المراجع (8.70) من أحمر المراجع (8.70) من أحمر المراجع (1.70) من أحمد (1.70) من أ

1SG.AGT know 'I know (it).'

- (8.71) $\neg \widetilde{a} \neg \widetilde{a} \neg \widetilde{a} \neg \gamma \eta \gamma$ $\eta a \quad t \varepsilon^h \phi := lo \quad ga.$ 1SG 2SG.L=DAT like 'I like you.' (KN e)
- (8.73) a) यह्नुव रहें वर्गीका लगाने च.गे.

tenzin=gi p^hak-ca $s\hat{a}$ -ga?Tenzing=AGTpig-meateat-PQ'DoesTenzingeat pork?' (PT e)

b) *¬*| *sà.* eat 'Yes, he does.' (lit. 'eats') (PT e)

The steady state present is negated by the prefix *mi*-, see (8.74).

(8.74) $a_{T}^{2} a_{T}^{2} a_{T}^{$

³⁰⁸ The name "steady state present" for this category is adopted from Van Driem's (1998: 195) description of the analogous category in Dzongkha.

Alternative ways to say approximately the same thing as (8.71) are (8.75) and (8.76).

- (8.76) $r \in \tilde{a}r \cap \tilde{a}r \cap \tilde{a}r \cap \tilde{a}r$ $\eta a t c^{h} \phi := lo ga-u \tilde{t}$. 1SG 2SG.L=DAT like-2INF EQU.PER 'I like you.' (KN e)

8.2.2 Simple present

The verb root may be followed by the evidentially neutral equative $b\varepsilon^2$ (or the cliticized variant $=p\varepsilon^2$) to form a construction which is mainly used for present habitual meanings (8.77-79) but which, with an appropriate adverbial, may also express future events (8.80-81). In interrogatives, the bare verb root may also be followed by the interrogative personal equative copula $p\dot{a}$ (historically *in-na*), see later example (11.12) and also Table 10.6.

(8.77) গ্রন্থ অর্ধন মন্

 $k'\tilde{a}$: $t'\delta$:= $p\epsilon$?. snow-mountain see=EQU.NE 'The mountain is visible.' (TB e)

- (8.78) a) אָי קעֿעָק' לָאַי אָרָיַק? $k^h u \quad j ó? \quad p'ja \quad b\epsilon - ka?$ 3SGM work do EQU.NE-PQ 'Does he work?' (KN e)
 - b) אַ קַי מְעָשִׁי אָק־לָאַי אָרן *k^hu jó? mim-bja bɛ?.* 3SGM work NEG-do EQU.NE 'He does not work.' (KN e)

(8.79) שֿוֹתְלִידִי שִּדִי תְקַשִי שִי תְשֵי צִּדן *ódɛra tʰa: ba? ja: gju bɛ?.* like that realease carry go go EQU.NE '(They) go free just like that.' (CY interview) (8.80)गवरमा कें र्रे पर कु केंदा पकु में म्रांग पर मुपय है क्यों। दर्म मेम्यत के का दा राष्ट्र मेया का मार्ग की ही क náŋts^hi t'o:pa tc^huts^hø? tcu=lo ló?var kjap-ti the.day.after.tomorrow morning clock.time ten=DAT X-ray do-NF ta-ge. nè: sém=na t'a t'ak jà: gju bE?. look-hort 1sg.gen mind=LOC now get.well go EQU.NE go 'Let's take an X-ray at ten o'clock in the morning of the day after tomorrow and look. In my opinion, he'll get better now.' (Richhi 27)

(8.81) $\mathbb{R}^{t} \check{a}^{t} \pi \pi \pi^{t} \pi^{t} \check{a}^{t} (\text{Nep.}) \operatorname{Agg} \check{a}^{t} \check{a}^{t}$ $k^{h}u$ $t^{h}or\tilde{a}^{t}$ badzar **gju** be?. 3SGM tomorrow market go EQU.NE 'He will go to the market tomorrow.'

Example (8.82) contrasts simple present and steady state present forms respectively. According to consultant PT, (8.82a) could be said if the speaker has just seen Tenzing eat pork, whereas (8.82b) implies old knowledge about Tenzing's pork-eating habit.

(8.82) a) यञ्चर पहेंद्र ग्रेश प्रण म झ हा

ter	nziŋ=gi	p ^h ak-sa	sà	bE?.	
Te	nzing=AGT	pig-meat	eat	EQU.NE	
ΎT	enzing eats	pork.' (PT	e)		
1 \	~ ~				

b) $\neg \Re \neg \alpha \overleftarrow{e} \overrightarrow{a} \cdot \cancel{y} \aleph' \lor \neg \neg \neg \neg$ $tenzin = gi \quad p^h ak \cdot ca \quad sa.$ Tenzing=AGT pig-meat eat 'Tenzing eats pork.' (PT e)

The simple present construction is negated by the prefix *mi*-.

- (8.83) מוֹק : جَنْهُ مَّا جَ تَمَّا جَ تَمَا يَعْتَ تَعْتَ عَلَيْهُ مَا يَحْ عَلَيْهُ مَا يَحْ عَلَيْ الْحَافَ مَنْ الْعَلَى عَلَيْهُ اللَّهُ عَلَيْهُ عَلَيْ عَلَيْهُ عَلَيْهُ عَلَيْهُ عَلَيْهُ عَلَيْهُ عَلَيْهُ عَلَيْهُ اللَّهُ عَلَيْهُ عَلَيْهُ عَلَيْهُ عَلَيْهُ عَلَيْهُ عَلَيْهُ عَلَيْهُ عَلَيْهُ عَلَيْ عَلَيْهُ عَلَيْهُ عَلَيْهُ عَلَيْهُ عَلَيْهُ عَلَيْهُ عَلَيْهُ عَلَيْهُ عَلَيْ عَلَيْهُ عَلَيْ عَلَيْكَ عَلَيْ عَلَيْكَ عَلَيْ عَلَ المَعْلَيْ عَلَيْ المَعْلَيْ عَلَيْ عَلَ
- (8.84)ราสิราชี" สตา "ขอรา เคริ" รับส์ รับส์ "สังส์ชุพุพ" พราt'arin=githaktee:=dit'atarethenmi-tshube?.today=GENdecision=DEMPHnowhappenNEG-be.able.toEQU.NE'Today'sdecision cannot be made now.'(BB BB discussion)

8.2.3 Present habitual I

The present habitual construction, similar to steady state present, is used in contexts which are reports of ongoing, stable state-of-affairs, see (8.85)

a) क्रॅंग् दर्ने के क्रॅंग् दर्ने के क्रॅंग् दर्ने यात हुआ के भाषा आपता हुन (8.85)bE?. *ódi=lo=di* te ódi k'aiem mi-làp-k^hen that=DAT=DEMPH so that what.is.it NEG-say-NMLZ EQU.NE 'It's not called that, whatever.' (PL interview) b) ज्ञूम् लुप्ता छेन्द्रे पर्वे दिप्त प्राय प्रत्याय मार्ग्य स्ट्रा zu:-k^hen lám sù-watsene gjømpa=na bE?. lama say.HUM-COND monastery=LOC live.HON-NMLZ EQU.NE 'If (we) talk about lamas, (they) live at monastery.' (YR interview) c) गाया के द्वाषा वहीं रहा थीं भूत वहीं वर्षा हेंद्र यात्र के द्वीं का वर्षा का क k'ɛːt c^h ita=di ran=gi ke:=di nénle cé: *g0* beginning first know important=DEMPH own=GEN language=DEMPH go:-k^hen hE2. be.needed-NMLZ EQU.NE

'The important thing is that one has to know one's own language at first.' (KL BLA 12)

The habitual present form lends itself to historical present uses, as shown in (8.86).

(8.86) $\widehat{\uparrow}:=: \widehat{\mathfrak{g}}^{\alpha_1} \underbrace{\mathfrak{g}}^{\gamma_1} \underbrace{\mathfrak{k}}^{\gamma_1} \underbrace{\mathfrak{k}}^{\gamma_2} \underbrace{\mathfrak{k}}^{\gamma_1} \underbrace{\mathfrak{k}}^{\gamma_2} \underbrace{\mathfrak{k}}^{\gamma_1} \underbrace{\mathfrak{k}}^{\gamma_2} \underbrace{\mathfrak{k}}^{\gamma_1} \underbrace{\mathfrak{k}}^{\gamma_2} \underbrace{\mathfrak{k}}^{\gamma_2}$

The construction $lap-k^{h}\tilde{\epsilon}$: \tilde{i} :/ $lap-k^{h}\epsilon n b\epsilon 2$ can be negated either with the perfective negator maor the imperfective mi-. The instances in my data suggest that ma- is used for referring to events which have a natural ending point and which are viewed as a whole, see (8.87). The negator mi-, in contrast, is used for events/states which are ongoing or prospective, see (8.88).

³⁰⁹ The English loan word here occurs here in a surprising form (in genitive case), which is on the first line "corrected" to *literature*.

³¹⁰ According to consultant YR, the Denjongke words for literature and and university are সমূল্যন্জ্য *bstan-bcos* and ব্যর্ব্যাথ্য র্দ্রান্যন্দ্র' *gtsug-lag slob-khang* respectively.

b) สุลาลเกลา (การ เมา m(a)- $\partial \eta$ - $k^h \varepsilon n$ $n amk^h a = l \varepsilon p' a p - t i$ bE?. sky=ABL descend-NF NEG-come-NMLZ EQU.NE '(They) do not come descending from the sky.' (NAB BLA 7) (8.88)a) ଦିଁକ୍ର କୁଷ୍ୟ ଭିଂନ୍କର୍ନି ଭାସ୍ତିର ଆସର ଛୁକ୍ରା $k^h \tilde{o}$: lenge? ódi mi-khen-khen hE?. 3PL PRN.HON that NEG-know.HON-NMLZ EQU.NE 'They did not know that.' (CY interview) b) ตั้งคริงศัก ตั้งคริง गुरु ख़ुरु के भाषा आपता खुरा लें रही में रही के साम का मुरा खार का का का का का का का क *ódi=lo=di* te ódi k'ajem mi-làp-k^hen be?. that=DAT=DEMPH so that what.is.it NEG-say-NMLZ EQU.NE ódi=lo biko làp-se $b\epsilon^2$. that=DAT stick say-INF EQU.NE 'It's not called that, whatever. It is called 'biko'.' (PL interview) c) אימרלי ארי לי אירי יומרי מקרי מלי אי מבאים ארי מרימי מרימי ארן למיארישרי שמיטילי מרי שמיטיל מאמרי אי מארי אי מאמרי אי מאמי מאמרי אי מאמי यायवः स्रदा ódi gã: t'a *p*^hilin zun=di dzambulin nànca ļo that time now foreign government=DEMPH south continent inside *bε?*. p'jasonzã: ge:pu=lo=jã: *cúktc*ẽ: ge:pu sún powerful EQU.NE therefore king=DAT=even king say.HON mi-ts^hu-k^hen bE?. NEG-be.able.to-NMLZ EQU.NE 'At that time the foreign (=British) government was strong outside Sikkim. Therefore the king couldn't even be called a king.' (CY interview)

A counter example to this basic pattern is (8.89), where the perfective negator ma- is used for an action that generally holds in the present.

Example (8.89), however, was spoken by a lady from Lachung, an area in North Sikkim which is dialectically somewhat different from more southern and western varieties. One consultant commented that the negator *mi*- should be used in the context of (8.89).

8.2.4 Present habitual II

As already discussed in §5.1, stative verbs (to which copulas are included), when nominalized by -po/bo/u and followed by an equative auxiliary, may refer, depending on the context, to present habitual state (8.90-92) or past state (8.93).

- (8.90) \mathbb{R}^{r} $\subset \mathbb{R}^{r}$ $\subset \mathbb{R}^{r}$ \mathcal{R}^{r} \mathcal{R}^{r}
- (8.91) $a_{1}^{2} = a_{1}^{2} i_{1}^{2} a_{2}^{2} i_{2}^{2} i_{3}^{2} i_{3$
- (8.92) $\frac{1}{2}$ מי מי $\frac{1}{2}$ מי מי (Bill Gates) מֹ הַבָּמִי חָמִיזי מֹרְזיִצּרְן $bil \ geits = lo \ n\acute{y}: \ ke:p \ jeb-be?.$ Bill Gates = DAT money much EX.NE 'Bill Gates has a lot of money.' (YR e)
- (8.93)a) ମ୍ ଡିମ୍ ଅଁ କ୍ରମ୍ମି: କ୍ଥୁ ଭିଂ ମୁଣ୍ଡମ ଭିମ୍ ଆର୍ସ ମମ୍ମାନ୍ୟ ସମ୍ବି ମାନ୍ୟା ନ୍ମାନ୍ୟ ମ୍ବା ସିଂହିଁ ଭିଷ୍ୟ ସିଂ କ୍ଷମ୍ବ tc^hø:=ki kankara=di t'atcĩ: na=lo pjõ: j*à:-k*^hẽ: kankara=gi crab=DEMPH recently 2sg.l=gen nose=dat hang ex-NMLZ crab=GEN p'otso ím-bo *bɛ*?. child equ-2inf EOU.NE 'The crab that was recently hanging from your nose is/was a baby crab.' (rna-gsung 33)
 - b) ג'לי מיצ'י מ'ל מאזי'צ'י אַרן *אָרָי מאזי'צ'י אַרן אָרָי אַרָּי אַרָּי אָרָי אַרָּי אַרָּי*

However, consultant KN commented that a clause like (8.93b) would, taken out of any further context, suggest for him that the referent has passed away, thus making the past interpretation the default case.

8.2.5 Nonpast

The construction VERB- $\epsilon\epsilon$ EQU can refer both to habitually true present facts and future events, hence the name nonpast (glossed NPST). The personal form $-\epsilon\epsilon$ \hat{i} is often abbreviated to $-\epsilon\tilde{i}$./- $\epsiloni\eta$ (see 8.101). Present habitual uses, which resemble in meaning the present habitual form VERB- $k^{h}\tilde{\epsilon}$. EQU, are illustrated in (8.94-96).

> *ódi=lo biko làp-cε bε?.* that=DAT stick say-INF EQU.NE 'It's called "biko".' (PL interview)

- (8.95) দ' দেশ্য' দ্বিশ' স্থ্ৰদ' ব' স্থ্ৰদ' দেশ' জীবা ŋà tasidiŋ=na do:-ce t:.³¹¹ ISG TPN=LOC live-INF EQU.PER 'I live in Tashiding.' (JD life story)
- (8.96) होगा मन के हे दे दा

 $s\acute{e}k-c\widetilde{i}r=s$ $p^{h}iru?$. burn-NPST.PER-QUO at.night '(My eyes) have a burning sensation at night.' (TB discussion)

Similar to the present habitual construction VERB- $k^{h}\tilde{\epsilon}$: EQU, the nonpast construction may be used for historical present, see (8.97) and (8.98). The nonpast construction in (8.97) ends in the reportative =lo, which can replace equative copulas in both pure copula and auxiliary uses (see §7.2.5.2) and which frequently occurs in storytelling. In (8.98), the speaker tells about his childhood.

 $t\varepsilon$ t'aypy:=gi $d\varepsilon ndzoy=gi$ kor $t'\tilde{a}:$ guru $b\varepsilon n\varepsilon = di = gi$ korzithenlong.ago=GENSikkim=GENabout andguruhidden.land=DEMPH=GENaboutn ayca=loodepti $kam\phi$ $dt\tilde{a}:-ra$ $n\tilde{a}:=c\varepsilon$ $\tilde{t}:$.inside=DATlike.that discussionalways-DEMPHdo.HON=INFEQU.PER'Then(we)would like that alwayshold discussionsabout ancientGuru's hiddenland.'(CY interview)

The nonpast form can express what the speaker just did or is doing:

³¹¹ This clause comes from a consultant from Tashiding, who was at the time fifteen years old. In the same piece of discourse, he also used other nonpast forms to refer to habitual actions. However, two other consultants, who hail from the villages of Lachung and Yangang, claimed that (8.95) is infelicitous as a habitual statement. In their opinion, the nonpast construction in (8.95) could only refer to future intention to stay in some place, whereas habitual residing in a place would be expressed through the imperfective $d\emptyset$:-to \hat{t} : (for the imperfective, see §8.3.1).

³¹² These type of double genitives, which are frequent in spoken language, do not occur in written Denjongke, where mere $5553 \cdot t'\tilde{a}$:py: is used instead.

For future uses, consider (8.100-102).

(8.100) अं ने न का में राम का मा के में मा के की

έ: t'εnε t^horã: ŋà nà: ba dø:-εε ΐ:.
o then tomorrow 1SG here hide sit-INF EQU.PER
'O, in that case tomorrow I'll sit hiding here.' (KT animal story)

- (8.101) $\int \tilde{a} = \tilde{a} = \pi \delta \tilde{a} = \tilde{a} =$
- (8.102) क्रेंग्व, मुनलाम्ह, क्षेव, मा

*óna kjap-ciŋ la*³¹³. there strike-NPST.PER okay 'I'll throw (the ball) there, okay?' (oh, Tashiding)

The nonpast construction can be negated in three ways. In the first, the negator prefix *mi*attaches to the verb root and no TAME-markers follow (8.103). In the second construction, the same prefix is used but is supplemented with infinitive and equative morphemes (8.104). The third construction negates the final auxiliary (8.105). Analogously with negation in periphrastic past constructions, the last construction (with negated auxiliary) is less frequent and here preliminarily considered an emphatic negating construction.

(8.103) हे' ठवा वागीया अपाक्षे' ठेवा'गु' दर्म' क्वेव'क्षेवया अवाया $k^h a = gi$ làp-ti tciku=di min-dik=la. tε tcam then just.like.that mouth=AGT say-NF only=DEMPH NEG-be.alright=HON 'It will be not (or: it is not) alright to just do it just like that orally.' (CY interview) (8.104) র্গমর্মান্বন্ মান্যবন্ ক্রমার্ব্রা নার্তনি স্পন্ এন্টা আব্যার্যা রাজ্য রাজ্য জিল্লিন্ প্রদা sémne? ma-nấ:-tsʰa-no yàtei k ke:=di jàrge? mind.pain NEG-do.HON-CMPL-COND 1PL.GEN language=DEMPH development mi-t^høn-ce bE?. NEG-happen-INF EQU.NE 'If we do not have pangs of conscience, our language will not develop.' (KL BLA 12)

³¹³ The typical Denjongke pronunciation is *lo* (the pronunciation here follows Nepali from which the morpheme is borrowed).

 $d\epsilon = r\tilde{a}$ $(n) \in to$ nàtea=tsu ran=gi p'a:na like.that=AEMPH EOU-COND=CEMPH 1PL=PL self=GEN in.between tci:=ki zen=lo *ló=rã*: te:-6e mèmbe? ín-(n)am one=AG other=DAT mind=AEMPH entrust-INF NEG.EQU.NE EQU.PER-ATTQ *t*'oku=tsu. friend=PL 'In that case, there is no trust among us between each other, is there, friends.' (mthunsgril 4)

b) דָ׳ מִדְ׳ הַיִּאִי מָדְ׳ אָדָ׳ אָי, דָ׳ כָּשִי הָשָּׁ שָּאַיִגִי מִדְ׳ אָשָּ שָּׁדָן t'a=di ŋà=lo=di, k'arsa, tiru? gja-t^hamba=di now=DEMPH 1SG=DAT=DEMPH what.to.say rupee hundred-NUM=DEMPH **p'in-ce mèmbe?**. give-INF NEG.EQU.NE 'Now he won't give me, what's that, a hundred rupees.' (PD bet story)

The three negating constructions are summarized in (8.106).

(8.106) a) हु' ईं'र्र्न्स' प'ईर' (Nep.) बेतु' दशु'

k^hu t^horã: badzar miŋ-gju. 3SGM tomorrow market NEG-go 'He will not go to the market tomorrow.' (KN e)

- b) אָי אָדיגדאי קידָאַי (Nep.) אָזי רְאָשָ' אָדָי אַדן *k^hu t^horã: badzar miŋ-gju-cɛ bɛ?. 3SGM tomorrow market NEG-GO-INF EQU.NE 'He will not go to the market tomorrow.' (KN e)*
- c) אָד צֿיאָרָאי קיבָּאַ (Nep.) מָשָּיַרְך אַק־אָק' אַק־אָרָן *k^hu t^horã: badzar gju-cɛ mɛ̀mbɛ?*. 3SGM tomorrow market go-INF NEG.EQU.NE 'He will not go to the market tomorrow.' (KN)

Consultant KN commented that the forms $gju\varepsilon\varepsilon \ m\dot{\epsilon}mb\varepsilon^2$ (negating the final copula) and $mi\eta$ gju\varepsilon\varepsilon $i/b\varepsilon^2$ in (8.106) (using negator mi- but adding the infinitive followed by positive copula) implied less certainty than the mere $mi\eta$ -gju. Future research is needed to fully understand the semantic differences between the formally different nagated nonpast forms.

8.2.6 Future with secondary verb \vec{o} : 'come'

In the future construction, the main verb is followed by the secondary verb \hat{o} : 'come', which functions as an auxiliary. It can refer to quite unlikely events, such as (8.107), or to very probable events, such as (8.108). When invited to comment on the difference between the nonpast construction (see §8.2.5) and the future construction, the consultants said that the event referred to

by the nonpast form is more fixed, whereas the future construction leaves more room for contingencies.³¹⁴

- (8.107) ਕਰ੍ਹਾਜ਼ੂ ਗੋਕਾ ਕੋਨਾ। *maŋ-gju?. di: ੈ.* NEG-run fall come 'Don't run. (You)'ll fall.' (NB e)
- (8.108) $\underset{k'an}{\Re} \cdot \eta_{\mathfrak{A}'} \cdot \eta_{\mathfrak{A}$
- (8.109) gratch^{i} , gratch^{i} , $\hat{\text{gratch}}^{i}$, $\hat{$

The future construction may be made explicitly uncertain by adding the probabilitative *-to* to form the construction VERB \hat{o} :-*to*, see §8.5.1.

8.2.7 Imminent future

The imminent future suffix *-rap* is appended to the verb root. It codes something that, in the speaker's opinion, is going to happen in the imminent future (glossed IMF). This form may be followed by either an equative or an existential copula, as shown in (8.111-113), or even by the verb $t^{h} on \not \equiv_{\overline{A}}$ 'come/go out, happen, become', see (8.114).

 $k^h u$ gju-rap $\tilde{i}:/b\epsilon^2/j\delta^2/du^2$. 3SGM go-IMF EQU.PER/EQU.NE/EX.PER/EX.SEN 'He's about to go.' (KN e)

³¹⁴ In my original PhD thesis, I used the term "uncertain future" for this construction. The qualification "uncertain" reflected the less fixed nature of the future event compared to more certain future events expressed by the nonpast construction. However, because there are no other constructions termed simply "future," it is superfluous to define the only "future" form as "uncertain." Therefore the description here adopts the simple term "future" for this construction. The events described by the future construction vary in degree of certainty/fixedness but are generally perceived as less fixed than future uses of the nonpast construction.

 $\eta \dot{a} = to$ $l \dot{e} p t i$ $t c^h \sigma :-rap$ $t^h \sigma m - bo$ $b \varepsilon ?$ 1SG = CEMPHvery.muchbecome.mad-IMFbecome-2INFEQU.NE'I was (lit. became) about to go very crazy.' (nga'i 'gan 22)

The imminent future marker may also be followed by other elements than a copula, for instance a case marker, as in (8.115a), or the secondary verb $d\phi^2$ 'sit, live', as in (8.115b).

(8.115) a) $\mathfrak{F}(\mathfrak{F}) = \mathfrak{F}(\mathfrak{F}) = \mathfrak{F}($

b) คิรารา อัราราลิ สุราสา สุราสา สุราสา
 cí-rap do:-pø: kap nàŋɛa=lo die-IMF stay-2INF.GEN time inside=DAT
 'At the moment when he was about to die...' (KT animal story)

The imminence of the action may be stressed by reduplication:

(8.116) 5' ผสูสารจ' ผสูรารจ' พัญ

yà t^h*uŋ-rap t*^h*uŋ-rap jò*?. 1SG drink-IMF drink-IMF EX.PER 'I'm just about to drink.' (DB e)

Imminent future construction does not occur negated in my natural data. When asking about the possibility of negation, consultant KN was at first reluctant to provide a negated example but then volunteered the following example (the translation is preliminary):

8.2.8 Tense, aspect and modality with the infinitive *-ni*

Sandberg (1895: 40) reports two infinitive forms *-she* (- $\varepsilon \varepsilon$?) and *-nyi* (-ni) for Denjongke and comments that the former is used in Denjongke spoken in Sikkim and the Tibetan variety spoken in the Tsang region of Tibet³¹⁵, whereas the latter is used in Denjongke spoken in the Darjeeling district. In my data, the infinitive *-ni* is used in a variety of idiomatic constructions some of which also occur with the infinitive *-\varepsilon \varepsilon \v*

The uses of *-ni* in my data are associated with such concepts as future, irrealis mood and uncertainty and are divided into following categories: uncertain future, inability, unrealized planned activity, future-oriented question, request/suggestion and future conditional. Common to all these categories is that the actions denoted by the verbs are not known to have happened as the speaker is talking (hence the description "irrealis").

8.2.8.1 Uncertain future

Using -ni in future constructions such as (8.118) implies more uncertainty than the use of the regular nonpast construction VERB- $\epsilon\epsilon$ EQU.

Consultant UTR commented that the construction in (8.118) is not much used in Tashiding and involves uncertainty (hence 'may' in gloss).

Example (8.119) shows that -ni cannot function as a replacement of the infinitive $-\varepsilon \varepsilon(2)$ in the nonpast construction $gju - \varepsilon \varepsilon \hat{i}$: 'will go' (b).

(8.119) *ך ר׳ ସ୍ଥୁ ନି ଜିମ୍ବ **t'a ŋà gju-ni ấ:*. now 1SG go-3INF EQU.PER

The infinitive -ni may also express uncertain future, or resemblance, in conjunction with the demonstrative proadverb *dem* 'like (it)' (the infinitive $-ee^2$ also occurs in this construction).

(8.120) ह्येंत्र जे. पर्ट्या गी

p'in-pi dem du-ke.
give-3INF like.that EX.SEN-IN
'It looks like (we) are to give (our daughter in marriage).' (SGD wedding customs)

8.2.8.2 Inability

Another irrealis use of -ni is the possessive-type-of construction which expresses inability. The infinitive $-c\epsilon^2$ also occurs in an analogous construction (see §8.4).

³¹⁵ Sandberg (1895: 12) reports Sikkimese Bhutias (=Denjongpos/Lhopos) to have originally come from the Tsang region in Tibet.

(8.121) हे' वही' यहेन गांवा केन यहेन गांवा वहीं हा हा केना

tε di dεŋ-gam min-dɛŋ-gam di t'a ŋà εù-ŋi then this be.true-ATTQ NEG-be.true-ATTQ this now 1SG say.HUM-3INF *mè?*. NEG.EX.PER 'Now whether this story is true or not, I cannot tell.' (RS bee story)

(8.122) म'र्से' वर्ने' बि'ने' बेन

 $y\dot{a}=lo$ di $c\dot{u}$ -pi mè?. 1SG=DAT this say.HUM-3INF NEG.EX.PER 'It's not mine to tell (=I do not know).' (PD interview)

8.2.8.3 Unrealized planned activity

Followed by the verb p'ja 'do', the infinitive -ni forms a construction which expresses what the speaker attempts/attempted to do or is/was hoping to do but has not been able to realize thus far. The -cc-infinitive could replace -ni in this construction without any or much change in meaning.

(8.123) E' ABE' ''' (13)

tc 'a *t*^huŋ-ni p'ja-u [†]. tea drink-3INF do-2INF EQU.PER 'I was about to drink tea./I attempted to drink tea/I would like to drink tea.' (TB e)

(8.124) ८ मार्थेग र्मुबा के र्मुबा के रा

nà jó? p'ja-ni p'ja-u í:.
1SG work do-3INF do-2INF EQU.PER
'I was going to work (but...).' (KN e)

Unrealized planned activity may also be expressed with the verb nó: 'think':

(8.125) הַ خَرْجَ مَلَّمَ الْعَـ بَعَـ جَ مَلَّ عَلَى مَعْ جَ جَ الْمَكَ (Nep.) جَ مَلَّمَ جَمِعَ جَرَعَ مَلْعَ اللَّهُ اللَّهُ اللَّهُ عَلَى اللَّهُ اللَّهُ عَلَى اللَّهُ اللَّ اللَّهُ اللَّالَ اللَّهُ اللَّهُ اللَّهُ اللَّهُ اللَّهُ اللَّهُ اللَّالَةُ اللَّهُ اللَّهُ اللَّهُ اللَّهُ اللَّ اللَّهُ اللَّهُ الللللَّ الللَّالَةُ اللللَّا اللَّهُ اللَّالَةُ اللَّالَةُ اللَّالَةُ اللَّالَةُ اللَّالَةُ اللَّالَةُ اللَّا لَقُلْحُالَةُ اللَّالَةُ اللَّالَةُ اللَّالَةُ اللَّالَةُ اللللَّالَةُ اللَّالَةُ اللللَّالَةُ اللَّالَةُ الللللَّالَ اللَّالَةُ اللَّالَةُ اللَّالَةُ اللَّالَةُ اللَّالَةُ اللَّالَةُ اللَّالُ اللَّاللَّاللَّالَةُ اللَّالَةُ اللَّا لَا اللَّالَةُ اللَّالَةُ اللللللَّ

8.2.8.4 Future-oriented questions

Another mode of presentation where the verbal action has not taken place are future-oriented questions.

(8.126) ך (مَجَ حَبَ الْعَامَة) حَامَاتِي (المَعَ عَلَيْ اللَّهُ عَلَيْ اللَّا عَلَيْ اللَّهُ عَلَيْ اللَّالَةُ عَلَيْ اللَّهُ عَلَيْ اللَّهُ عَلَيْ اللَّهُ عَلَيْ اللَّهُ عَلَيْ اللَّهُ عَ اللَّهُ اللَّالِي اللَّهُ عَلَيْ اللَّهُ عَلَيْ اللَّهُ عَلَيْ اللَّهُ عَلَيْ اللَّهُ عَلَيْ اللَّهُ عَلَيْ الْ المَالَا اللَّهُ اللَّا اللَّا اللَّهُ عَلَيْ اللَّهُ عَلَيْ اللَّهُ عَلَيْ اللَّهُ عَلَيْ اللَّهُ عَلَيْ اللَّ In example (8.128), the question functions as complement of dau 'like' and is therefore not a true question. The syntagm given in bold is an idiom used three times in the novel Richhi.

(8.128) الله: جَرَبُحَتَ الله: جَرَبُحَتَ الله: اللله: الله: الل

In the interrogative construction in (8.129), the uses of -pi (a) and $-c\epsilon^2$ (b) overlap syntactically.

(8.129) a) নন্ত্ৰান্টান্ম? gju-pi-ga? go-3INF-PQ 'Are you going?' (UTR e)

b)	<u> ଏ</u> କ୍ସି.ସ	(from:	دىق. تىك ئەركە.
	gju-ca?	(from:	gju-ce-ga?)
	go-INF.PQ		go-INF-PQ
	'Are you g	oing?' (U	TR e)

8.2.8.5 Request and suggestion

Another context for -ni are urgent requests (8.130) and suggestions (8.131). The request construction with the urgentive $-m\phi$?, according to consultant KN, is used in Tashiding (West Sikkim) but not, for instance, in Martam (East Sikkim).

(8.130) جوج كَرَّ مَعْمَ *t^huŋ-pi=mø?.* drink-3INF=URG 'Drink, by all means!' (KN e)

In (8.131), the construction VERB-*ni* EQU functions as a suggestion rather than a statement about future. The speaker has first inquired whether the addressee has any plans for the coming holiday, and upon hearing that there are no definite plans, he continues with:

³¹⁶ This kinship term also has other meanings, see §17.2.1.

bombai=lotɛ 'øn-µibɛ?. p^ha :ts^hu:ta-ɛɛlèptilèmBombay=DAT come.HON-3INFEQU.NE thitherhitherwatch-INFverygood $j\partial$?.EX.PER'There is (this option of) coming to Bombay. Doing sightseeing here and there is verygood.' (Richhi 101)

Example (8.131) presents a clear point of difference with infinitive $-\varepsilon \varepsilon 2$. Using $-\varepsilon \varepsilon 2$ instead of $-\mu i$ in (8.131) would result in a typical and frequent future construction meaning '(s)he will go to Bombay'.

8.2.8.6 Future conditional

The infinitive -*pi* also occurs in the future conditional construction VERB-*pi* EX-COND.

(8.132) दायापाय, वर्त्ता र्या त्या के येंत्ता क्षवायया यावा द्वियाया के र्यता यहार रहा द्वियाया क्षेत द्या मता केवा nè:po da = lajø:-ne mémpa: k^hik-ti f'aː-pi k'ana doctor=HON physician.AGT where patient get.well-3INF EX-COND lead-NF số: súŋ-ruŋ k^hik-ti gju-se ĩ:. go.IMP say.HON-CONC lead-NF go-INF EQU.PER 'Doctor, if the patient is to get well, (I) will take (him) wherever the doctor tells (me) to.' (Richhi 169)

(8.133) (8.133) يَّقْ مَجْمَّتِي عَلَى حَجْنَى عَلَى عَ مَا عَلَى عَلَى

Conditionality can be combined with the use of tem/dem 'like (it)' to form a construction which is, in the realis-irrealis continuum, even further away from a realis assertion of a happened fact than a bare conditional.

(8.134) هم حرك بخم عن علي علي علي علي علي علي علي المحالي المحالي المحالي (8.134) علي المحالي المحال

8.2.8.7 Other uses of *-ni*

Iterativity and excessiveness can be expressed by postposing to VERB-*ni* the same verb again as a type of reduplication:

শ্লবিষা বদান্তার্থা te $k^h u = lo$ *pim* $te^{h}ame?$ $\delta d\varepsilon = ra$ ka òm-bo $k^h \tilde{o}$: boto like.that=AEMPH who come-2INF so 3sgm=dat day every 3PL beating *pìm te^hame:=lo tei:=ki* rɛk-ti õĽ dun-ni dun dø:-pø: kap feel-NF day every=DAT one=AGT come hit-3INF hit stay-2INF.GEN time nànca=lo inside=DAT 'So when daily like that anyone who came beat him up, and as he received every day beating after beating from anyone who came...' (KT animal story)

The infinitive -ni also occured in the following construction, where -ni appears to refer to a presently holding fact. Together with the discourse particle $t\varepsilon$ 'so', the reading is causal.

(8.136) معرابَ وَعَبَّرَ مَعَرَابَةُ مَعَرَابَةُ وَعَرَابَةُ وَعَامَا مَعَنَّ جَابَةُ وَعَرَابَةُ وَعَرَابَةُ وَعَرَابَةُ وَعَرَابَةً وَحَرَابَةً وَعَرَابَةً وَعَرَابًا وَعَرَابًا وَعَرَابَةً وَعَرَابَةً وَعَرَابَةً وَعَرَابَةً وَعَرَابَةً وَعَرَابًا وَعَرَابَةً وَعَرَابًا وَعَرَابَةً وَعَرَابَةً وَعَرَابَةً وَعَرَابَةً وَعَرَابَةً وَعَرَابَةً وَعَرَابَةً وَعَرَابَةًا وَعَرَابَةً وَعَرَابَةًا وَعَرَابَةً وَعَرَابَةً وَعَرَابَةً وَعَرَابَةًا وَعَرَابَةًا وَعَرَابَةًا وَعَرَابَةًا وَعَرَابَةًا وَعَرَابَةً وَعَرَابَا وَعَرَابَةًا وَعَرَابَةً وَعَرَابَةً وَعَرَابَةً وَعَرَابَةً وَعَرَابَا وَعَرَابَةً وَعَرَابَةًا وَعَرَابَةً وَعَرَابَةً وَعَرَابَةً وَعَرَابَةً وَعَرَابَةً وَعَائَعَا وَعَابَاتًا وَعَا وَعَرَابَةً وَعَرَابَةً وَعَرَ

8.3 Imperfective, progressive and continuous forms

Denjongke has several partly overlapping ways of expressing that the action denoted by the verb is ongoing at the time of speech or, in the case of the imperfective, was ongoing at a past time, see Table 8.3. The terms "progressive" and "continuous" are very similar in meaning. The difference of progressive and continuous categories here, however, is based on the properties described for these categories in Denjongke, not on the semantics of the English terms. Different names for semantically similar construction are needed in order to refer to the constructions unambiguously.

Name	Form	Function
imperfective	VERB-to/do (EQU)	past habitual or ongoing action/state,
		present habitual or ongoing action/state,
		immediate future
continuous	VERB dø: ex	past, present, or future ongoing action/state
	VERB <i>dø</i> : (+normal inflection)	(with atelic verbs, e.g. 'stand'),
		past, present or future resultative (with telic
		verbs, e.g. 'arrive')
progressive	VERB- $tc\tilde{\epsilon}$:/ $z\tilde{\epsilon}$: EX	past or present ongoing action
alterphoric	VERB- <i>tsunge/zunge</i> (Tashiding)	perceived ongoing action
progressive	VERB- <i>tcouge/zuŋge</i> (Martam)	
durative	VERB <i>bak(-ti)</i>	emphasizes durativity of the action

 Table 8.3. Imperfective, progressive and continuous constructions

8.3.1 Imperfective

The imperfective -to/do may refer to past habitual, past ongoing, present habitual, present ongoing or immediate future actions/states.³¹⁷ Habitual uses seem more frequent than others in my data. Past habitual uses are illustrated by examples (8.137) and (8.138).

(8.137) العملام من من من المحالية المعلمان من المحالية المحالية المحالية (Nep.) مكم تعلم المحالية ال محالية المحالية المحالي المحالية ال

In (8.139), the imperfective is used, perhaps surprisingly, for a past telic action (telling a lie). With this strategy of vivid storytelling, the speaker appears to bring the addressees in the middle of the action within the story.

(8.139) क्वेंग्रेग्गुन्द क्वेंग वायर के रायर क ló-mi-tε tøn ma-teuk-ee=ki t'ønda=le de: mind-NEG-entrust show NEG-CAUS-INF=GEN purpose=ABL like.this â: dem=tei? kjap-to ín-k^hen bE?. p'ja-tiki like=INDF strike-IPFV EQU-NMLZ EQU.NE do-NF lie 'In order that it wouldn't be shown that he didn't believe (the story), he's telling as a lie like this:' (PD story)

For the evidential implications of using the nominalized copula construction $i\eta$ - $k^{h}\epsilon n b\epsilon^{2}$ as the auxiliary in (8.139), refer to example (9.22) in §9.1.1.5.

Examples (8.140) and (8.141) illustrate the present habitual use of the imperfective. Note that both (8.141) and (8.137) above occur in the same piece of discourse where an elderly lady compares the old style of living in her village to the present one. Whereas in (8.137) the imperfective construction is used for a past habitual action, in (8.141) the same construction (negated) describes present reality.

(8.140) नुम्र कुट र्च कॅरा क्रुट लय र्ह भुन

 $p'um te^{h}umbo$ $te^{h}\phi ki$? **làp-to b** ϵ ?. girl small.one PN say-IPFV EQU.NE '(My) youngest daughter is called Choki.' (DB life story)

³¹⁷ The use of *-to/do* differs from the related language Dzongkha in that the homophonous morpheme in Dzongkha cannot refer to past time and cannot be used with past adverbials, such as 'yesterday' (van Driem 1998: 202). For past uses, the related form *dowä/deä* is used in Dzongkha.

(8.141) र्रे विश्व विश्व मार्ट विर्म के क्या के क bɛ?. t'o? $t^{h}amtee$? gari bak òn-do тí thamtee? ka:m load all car carry come-IPFV EQU.NE people all foot tsuk-ti man-go:-to bE?. gju plant-NF go NEG-be.needed-IPFV EQU.NE 'All loads are carried by cars. People don't have to go by foot.' (RBM discussion on the roof)

Examples (8.142) and (8.143) illustrate a context where the imperfective marks a present ongoing action/state. The equative copula may be dropped when presenting present ongoing and immediate future events.

- (8.142) $\overleftarrow{b} \overleftarrow{\gamma} \cdot \overrightarrow{\eta} \cdot \overrightarrow{\alpha}$? $tc^{h} \phi$? k'a: gju-do=s? 2SG.L where go-IPFV=QUO 'Where are you going (he said)?' (KT animal story)

 $t'ato te^huts^h \otimes 2$ tei:=to **duy-do.** now clock-time one=CEMPH hit-IPFV 'It's one o'clock now (lit. it's striking one o'clock now).' (Richhi 124)

For an immediate future use of -to/do, consider (8.144).

(8.144) गुत्र के केंन प्रा नित्र र्यु र र तेंन में

kante^{hi318} te^hø? gju-zẽ: p'ja, ŋà òn-do. younger.sister 2SG.L go-PROG do 1SG come-IPFV 'You go on, sister, I'm coming.' (Richhi 53)

The affirmative imperfective form has several corresponding negated forms. The first, negation by the negator prefix *ma*-, was already ilustrated in (8.141). Initial data from consultants KN and KUN suggests, secondly, that the prefix *mi*- could be used instead of *ma*- without change in meaning. The third negation strategy is borrowed from the non-reduplicated perfect construction, as shown in (8.145b), which negates (8.145a). The fourth strategy, which according to consultant KN is used at least in Tashiding (West Sikkim), is to negate the final existential of the affirmed imperefective construction (8.145c).

(8.145) a) यु. गर्थेग र्जुश र्र भ्रु

 $k^{h}u'$ jó? **p'ja-do bɛ?**. 3SGM work do-IPFV EQU.NE 'He is working.' (KN e)

³¹⁸ A loan word from Nepali.

b) אַ יַשְׂאַשְ לָאיד אָרְקישָרן *k^hu jó? p'ja-u mèbbe?.* 3SGM work do-2INF NEG.EX.NE 'He is not working.' (KN e)

For negating the affirmative question in (8.146), several functionally roughly equivalent options are possible, see (8.147). In (147a), the negated form derives formally from the affirmed imperfective form. The negated forms (147b) and (147c) build on the non-reduplicated perfect form. Construction (147d) uses the typical past periprastic question construction VERB-*po pá* put replaces the perfective negator *ma*-, which would occur in a past construction, with the imperfective negator *mi*-.

(8.146) 逝 5' 5 3 7 · デ · 新 5' 初 5 · 7 ?

t¢ ^h ø?	índzi	ke?	kjap-to	pá?
2sg.l	English	language	strike-IPFV	EQU.PER.Q
'Do you	speak E1	nglish?' (NA	AB e)	

(8.147) a) هَرَ ٢ جَعَمَ ٦ جَعَمَ ٦ اللهُ عَمَى اللهُ عَمَى اللهُ عَمَى اللهُ عَمَى اللهُ عَمَى اللهُ المُ

 $t \varepsilon^h \phi$?indzi $k \varepsilon$?mi-kjap-to $p \acute{a}$?2SG.LEnglish languageNEG-strike-IPFVEQU.PER.Q'Don't you speak English?.'(NAB e)

8.3.2 Continuous

The continuity of an action or its results can be expressed by the verb $d\phi^2$ 'sit, stay' in two type of constructions. In the first, the secondary verb $d\phi^2$ is followed by an existential auxiliary (personal

 $j\partial^2$, sensorial du^2 or neutral $j\partial po b\varepsilon^2$). In the second, less grammaticalized use, $d\phi^2$ 'sit, stay' postposed to a verb inflects like an ordinary verb, allowing nominalized/infinitivized forms. The more grammaticalized and probably more frequent uses with an existential auxiliary are first illustrated in (8.148-161). Uses with typical verb inflection are exemplified in (8.162-163).

With atelic expressions, which have no natural end-point, the continuous construction marks actions and states which are ongoing. In (8.148) and (8.149), the action/state is ongoing at the time of speech, whereas in (8.150) the action was ongoing at a(n imaginery) past time.

(8.148) দ্রে' ব্যরিম' র্ইন্' র্ন্বা

k^hu zim dø: du2. 3SGM sleep.HON stay EX.SEN 'He's sleeping (I see/saw).' (TB e)

(8.149) देवा लेवा लेवाया मर र्हेट तर्वा

nìm lèp lèm cá: dø: du?. sun very.much good shine stay EX.SEN 'The sun is shining very nicely (I see/feel).' (TB e)

ge:pø:	p^h	odã:	ódi	tẽ:=lo	lep-ø.	•	gã:	óna	ge:pø:
king.GEN	r pa	lace	that	top=DAT	arrive	e-2INF.GEN	time	there	king.GEN
p ^h oːdãː	nàŋce	a t ^h a	:kor=	lo	óna	ge:pø:	p'um	tci?	kja
palace	inside	e sui	rround	ding=dat	there	king.GEN	girl	one	hair
sé-tse:		sé-tsê	Ĭ.	ZUS	dø:	j <i>ð:-kʰɛn</i>	bE?.		
comb-PR	OG	comb	-PROC	G sit.HON	stay	EX-NMLZ	EQU.N	νE	
'When h	e arri	ved up	p at tl	hat royal pa	lace, i	nside the ki	ing's p	balace.	, in the surroundings,
there was	s a da	ughter	of th	e king comb	oing ar	nd combing	(her) l	nair.' ((PD bet story)

The fact that $d\emptyset$, the ordinary verb meaning 'sit, stay', in (8.150) occurs following the honorific zu: 'sit, stay (hon.)' shows that the use of $d\emptyset$: is grammatical rather than lexical. Lexically, one honorific form collocates with other honorific forms and thus the use of the honorific zu: would evoke the use of other lexical honorifics.

With telic expressions, which have a natural end point, the meaning is resultative, i.e. highlightning the ongoing state accomplished through the verbal action, see (8.151-154).

- (8.151) $\mathfrak{F}_{\mathcal{F}}$ $\mathfrak{F}_{\mathcal{F}}$ $\mathfrak{F}_{\mathcal{F}}$ $\mathfrak{K}_{\mathcal{F}}$ $\mathfrak{K}_{\mathcal{F}}$ $\mathfrak{K}_{\mathcal{F}}$ $m \acute{y} tsy ri$ $k \varepsilon 2$ $b j \tilde{a}$:d o: $j \dot{o}$: $-k^{h} \varepsilon n$ $b \varepsilon 2$.other-EMPH.GENlanguagedisappearstayEX-NMLZEQU.NE'The language of others has disappeared' (RS interview)
- (8.152) مَحَدَ مَحَدَ بِحَجَ سَمَرَ بَنَ مَعَ ثَنَ جَنَةً وَجَ

$k^h \widetilde{o}$:	ŐĽ	dø:	j <i>ð:-p(o)</i>	mèmbo,	p'otso=tsu?
3pl	come	stay	ex-2inf	NEG.EQU.NE.Q	child=PL
They	y have	e retur	med, haven'	t they, the children	n?' (LT, KN kitchen)

pe=na kolidz levəl nànca nàtea=lo nàtci example=LOC college(Eng.) level(Eng.) inside 1PL=DAT **1PL.GEN** keri? lep dø: jò?. language arrive stay EX.PER 'For instance, our language has reached college level.' (NAB BLA 7)

(8.154) विं भ्रुवायत्व भ्रेययायते भ्रत इते ययाया द्वारे मेव भ्रि स्वर स्व के स्वर स्व के स्वर स्व के स्वर के स k^hu $m \epsilon \eta k^h \tilde{a}$:=na [ep-ø: gã: bhaila t'embo sìn-di hospital=LOC arrive=2INF.GEN time PN 3sgm consciousness awaken-NF сý:сv: k^ha -làp $ts^h u$ -po *t*^hØn du? dø: a.bit mouth-speak be.able.to-2INF become stay EX.SEN 'When he arrives at the hospital, Bhaila has regained consciousness and has become able to talk a bit.' (Richhi 23)

The continuous construction may also be used for future actions/states if the final copula is in the nonpast construction $j\dot{\theta}$:- $\epsilon\epsilon$ \dot{t} , see (8.155), contrasting with analogous past (8.156) and present expressions (8.157).³²⁰

- (8.156) جَ مَعْرَجَ حَيَّ هَيْ جَ مَعْنَ مَعْنَ مَعْنَ مَعْنَ مَعْنَ عَيْنَ مَعْنَ عَنْنَ مَعْنَ عَنْنَ مَعْنَ عَنْنَ مَعْنَ عَنْنَ مَعْنَ عَنْنَا مَعْنَ عَنْنَا مَعْنَ مُعْنَ مَعْنَ مُعْنَ مُعْنَا مَعْنَ مَعْنَ مَعْنَ مَعْنَ مَعْنَ مُعْنَ مُعْنَ مُعْنَ مُعْنَ مُعْنَ مُعْنَ مُعْنَ مَعْنَ مَعْنَ مَعْنَ مُعْنَ مَعْنَ مُعْنَ مُعْنَ مُعْنَ مُعْنَ مُعْنَ مُعْنَ مُعْنَ مُعْ

nà t'ato gju: dø: jô?.
1SG now go stay EQU.PER
'I keep on walking now.' (TB e)

The continuous construction has most probably developed through the also existing converbal construction VERB-NF $d\emptyset$: EX by eliding the converbial marker -ti/di, see (8.158) and (8.159).

(8.158) न पुत्रायम पर्द्या हे न तनेय हेन थेन मज्या वा

 $t'a \ k^h \varepsilon : g\tilde{a}: tsuk-ti$ $t'a \ d\varepsilon p \ d\sigma: j\tilde{\sigma}^2$, $\eta \tilde{a}t\varepsilon a^2 \ n\tilde{a}:$ now family plant-NF now like that stay EX.PER 1PL here 'Now here we live like that, having established families.' (DB life story)

³¹⁹ The WD word is ঝর্ষাইন স্ক্রিন্যা *mtho-rim slob-grwa* (YR).

³²⁰ It is not possible to form a future construction of the other existential copula, sensorial du?.

At least some verbs may occur in both a converbal (8.160) and continuous constructions (8.161).

(8.160) यायाबा हेंट्राबा बाहेरायमा येंटाही हेंट्रा येंट्रा

làla døsa ma-t^hop-(p)a lồ:-ti dø: jồ?. some.AGT sitting-place NEG-find-CIRC stand-NF stay EQU.PER 'Some, not finding a place to sit, are standing.' (Richhi 75)

(8.161) $\operatorname{Aux}(\operatorname{Aux}, \operatorname{Aux})$ $\operatorname{Aux}(\operatorname{Au$

The converbal construction in (8.160) places emphasis on the manner of staying, i.e. standing, which is contrasted with the possibility of sitting, whereas the continuous construction in (8.161) is a simple statement about what the speaker sees people doing.

In addition to the auxiliary construction where $d\phi^2$ is followed by an existential, $d\phi^2$ may be inflected like a typical verb. In some of the uses, the verb $d\phi^2$ has a more grammatical sense where it underlines continuity of the action, see (8.162). In other uses, the secondary verb $d\phi^2$ is used in a more concrete way with the meaning 'sit, stay', see (8.163).

(8.162) a) सुत्रा वर्दन कवाकेंवा मुंबा वर्युः झून्य झन्

p'um=di $tc^haltc^hol-p'ja$ gju dc-b=bc?. girl=DEMPH IDEO.NN-ADVZR go stay-2INF=EQU.NE 'The girl keeps on going (around) acting like a deranged person.' (KN e)

b) ਕੇ ਕਨ ਦੁੱਖ ਤੁੱਕ ਤੁੱਕ ਕਹੁ ਵੱਨ ਦੱ ਤੁੱਕ

mi=di $p^hjarp^hjor-p'ja$ gju do:-po $b\epsilon 2$. human=DEMPH IDEO.NN-ADVZR go stay-2INF EQU.NE 'That person keeps on going (around) not wearing clothes properly.' (KN e)

c)	5'5'5	ं कुन-कुन-लबा वा देव	।'झे' पॅंट-'झे' चलग'झे' झे	<u> </u>	7.5451			
	t'a	ŋà=to	t&huyteuy=le	nà:	òtε	òn-di	jàk-ti ³²¹	dø:
	now	1sg=cemph	small=ABL	here	down	come-NF	set-NF	stay

³²¹ This speaker from Lachung frequently used the verb zak/jak 'set, put' together with other verbs. Consultant YR noted that the use of zak/jak here "doesn't sound good."

dE-b=bE?.

sit-2INF=EQU.NE

'Now, I have come and settled down here from an early age.' (LA intro to Lachung)

(8.163) a) ज़ूर स्वाः झेंगी झ र्झेन वर्न केवा

gjun $p^huk-tiki$ **ba** $d\theta:-c\varepsilon$ **í**:. basket pierce-NF hide stay-INF EQU.PER 'Piercing the basket I will stay hiding (there).' (KTL animal story)

b) $a \widehat{\uparrow} (a_{N}, \forall i_{\mathcal{B}}, \underbrace{\mathfrak{g}}_{\mathcal{T}}, \underbrace{\mathfrak{g}}, \underbrace{\mathfrak{g}}_{\mathcal{T}}, \underbrace{\mathfrak{g}}, \underbrace{\mathfrak{g}}$

lóbda $n an = la^{322}$ $te^{h}o^{2}$ sà:te tacidin gjompø: ná lò tcu monastery.GEN school inside=DAT I year ten until TPN doctrine p'ja dø:-po Ĩ. stay-2INF EQU.PER do 'I stayed as long as ten years studying at the Tashiding monastery school.' (RB life story)

Note that in (8.162c) the concrete use of $d\phi$? is followed by a grammatical use.

8.3.3 Progressive VERB-tcɛ̃:/zɛ̃:/zin EX and VERB-tcuŋgɛ/zuŋgɛ

The progressive constructions VERB- $te\tilde{e}:/z\tilde{e}:/zin$ EX and VERB-teunge/zunge (in Martam: VERB-teouke) mark the verbal action as ongoing at a specific time determined by the existential copula and the context. The form $-te\tilde{e}:/z\tilde{e}:$ occurs in writing as WD/WT $\operatorname{deg}: bzhin$. The reading-style pronunciation is zin, a form which also occurs in the spoken language of literate speakers. The historic origin of the form teunge/zunge is more difficult to determine because I have not come accross it in written Denjongke.³²³

The semantics of the progressive $t_{\tilde{e}\tilde{e}}:/z\tilde{e}:/zin$ seem more limited to a certain specific time than the semantics of continuous secondary verb $d\vartheta$; which may include habituality in addition to continuity at a specific point of time. For instance, consultant NAB commented that *gju-zin du*? [go-PROG EX.SEN] refers to an event happening at the time of speaking but *gju d\varnotheta*: du? [go stay EX.SEN] could also refer to habitual action.

8.3.3.1 Progressive VERB-tce?:/ze?:/zin EX

The construction ending in the existential $j\partial^2$ implies that the speaker is personally well-acquainted with the situation and that the action is ongoing at a specific reference time, which typically is the time of speaking. In (8.164), the author of the novel Richhi uses the structure with $j\partial^2$ as if to invite the readers to observe a real-time event.

³²² Dative-locative form =la instead of =lo here is Tibetan influence.

³²³ Consultant KT specifically stated that *-tcuŋgɛ/zuŋgɛ* is only used in oral, not written language.

(8.164) यहे राह र्जुया हर्षेत्र र्योत्र र्ये अहला आपत्र के रेरे पानिया पनिया जीवा र्याहर प्रिया वितर यहा $d\varepsilon = r\tilde{a} = p'ja$ $dz\varepsilon$:- $k^{h}\tilde{\varepsilon}$: ní:-ní: gømpo тí re-re like.that=AEMPH=ADVZR monastery meet-NMLZ person each-each two-two Ő:-zĩ: $p'ja-z\tilde{\varepsilon}$ j*ò*?. do-prog come-PROG EX.PER 'Like that, people visiting the monastery are coming each two by two.'(Richhi 2)

The first instance of $-z\tilde{\varepsilon}$: in (8.164), p'jaz $\tilde{\varepsilon}$; illustrates an adverbial use without a following auxiliary. In this respect, $-t\epsilon\tilde{\epsilon}:/z\tilde{\epsilon}:/zin$ resembles the English progressive form -ing, which occurs both as an adverbial without an auxiliary and as an element in a finite construction followed by an auxiliary.

In (8.165), the use of *-tcen* with *làp* 'say' marks the continuing factuality of a proposition heard earlier (he disappeared) rather than the fact that a third person is speaking at the same time as the speaker and the addressee of (8.165) are speaking.

(8.165) בַבָּרִידִי אוֹשְ מים היש אוֹש (bjõ:-bo ť: làp-teen du?disappear-2INF EQU.PER say-PROG EX.SEN '(He) disappeared, (they) are saying.' (TB phone call)

The aspect marker $-t_{\varepsilon}\tilde{\varepsilon}/z\tilde{\varepsilon}/z\tilde{\epsilon}/zin$ is the only verbal suffix which may be supplemented by the infinitive marker -po/bo. The nominalized construction may be used in identical contexts with the non-nominalized construction, as shown by the two possible answers (8.167a) and (8.167b) to the question (8.166).

- (8.166) मुभायळंत्र मार्नें गता र्घ्यापवित्र पर्या? gjalts^hen t'ato k'an **p'ja-zen** *du?*? Gyalsthen now what do-PROG EX.SEN 'What is Gyaltshen doing now?' (KN e)
- (8.167) a) विंदा दार्क्ष देया क्रेंगायविवार्य वर्त्तगा t'ato t'ep **dok-zim-bo**³²⁴ $k^h \tilde{o}$: du?3SG.HON now book read-PROG-2INF EX.SEN 'He is now reading a book (I see).' (KN e)

b) विन न कें नेय क्रेंग नवेवा वर्ग t'ato t'ep dok-zin kħõ: du?. 3sg.HON now book read-PROG EX.SEN 'He is now reading a book (I see).' (KN e)

In addition, the nominalized form, however, can be used for what in English are called present perfect continuous meanings:

³²⁴ For some reason, the consultant used here and in the following example the literary pronunciation instead of the typical oral -tcen.

(8.168) تربيحما التي تعالى المراجع التي المراجع المراحي المراجع المراجع المراجع المراجع المراجع المراجع المم

The nominalized progressive also occurs with an equative as auxiliary. The construction has a past (hence -po EQU) progressive (hence $-tc\tilde{\varepsilon}:/z\tilde{\varepsilon}:/zin$) meaning, as shown in (8.169).

(8.169) अन्न क्षत्र कुषा न में में में पर (call) कुन्रषा प्रदे क्षन न के प्रायमें प्रायमें की प्रायम की प dã: lenge? nà=lo kol kjap-ø: ηà t^hom=lo gã: yesterday PRN.HON call(Eng.) do-2INF.GEN time I marker=DAT 1sg=dat Ĩ. gju-zim-bo go-prog-2inf EOU.PER 'When you phoned me yesterday, I was going to town.' (KN e)

Finally, the progressive occurs in an idiomatic construction followed by the verb p'ja 'do':

- (8.170) খন্তব্যথ্যখন্ট্ৰির' দ্রার্থান্ট' ব্যব্দ'র্য' ব্যব্দ'] zu:-tɛɛ̃: p'ja-u nấ:. sit-PROG do-2INF do.HON 'Please sit (and wait here)' Lit. 'Please do sitting'. (oh)
- (8.172) รถ ขุรร สูง พิขุ ธิรจิสุร ฮูง รุร ติรุ *ná: denzu jìgi t'i-zẽ: p'ja-ce í*:. I.AGT invitation letter write-PROG do-INF EQU.PER 'I will be writing an invitation letter.' (Richhi 42)

The use of the progressive in (8.170) underlines the durative nature of the action. A simple request $zu:-po n\hat{a}$: [sit.HON-2INF do.HON] 'Please sit down' would be used when a standing guest is advised to sit down. Example (8.170), on the other hand, was used when the guest was already sitting and the host needed to go away for a while. In (8.171), the progressive construction underlines the iterativity and duration of the action. The alternative shorter construction *tok-renkha*: 'when plucking' without the progressive could be interpreted as being about one fruit, whereas (8.171) presupposes an iterative process of plucking. In (8.172), the speaker announces her immediate future action, focusing on the durativity of that action.

According to Jäschke (1881: 483), the etymon of the progressive marker $-tc\tilde{\epsilon}:/z\tilde{\epsilon}:$, WT $\neg a_{\overline{d}}$. *bzhin*, has the meanings 'face, countenance', 'agreeably, in conformity, according to' and 'like, as'. These WT meanings are reflected when *-zin* occurs as a component of the postpositions t' onzin(=gi) 'in accordance with, in view of' (t' on 'purpose') and pakozin(=gi) 'similarly, in accordance with' (the origin of the form *pako* is unknown to me at present).

(8.173) جَ هَ مَحْمَ بِعَمَ مَ عَلَى جَ حَصَّا مَرَى مَ مَحَالًا مَ مَحَالًا مَ مَعْلَى اللَّهُ اللَّهُ اللَّهُ عَلَيْهُمْ عَلَيْكُمْ عَلَيْهُمْ عَلَيْكُمْ عَلَيْكُمْ مَا عَلَيْكُمْ عَلَيْكُمُ عَلَ مَعْنَا عَلَيْكُمْ عَلَيْكُمْ عَلَيْكُمْ عَلَيْكُمْ عَلَيْكُمْ عَلَيْ عَلَيْ عَلَيْكُمْ عَلَيْكُمْ عَلَيْ مَعْنَا عَلَيْكُمْ عَلَيْكُمْ عَلَيْكُمْ عَلَيْ عَلَيْ عَلَيْ عَلَيْ عَلَيْكُمْ عَلَ مَعْلَيْ عَلَيْكُمْ عَلَيْكُمْ عَلَيْكُمْ عَلَيْ عَلَيْ عَلَيْ عَلَيْ عَلَيْ عَلَيْكُمْ عَلَيْ عَلَيْ عَلَيْ عَلَيْ عَلَيْكُمْ عَلَيْ عَلَيْكُمُ عَلَيْ عَلَيْ عَلَيْ عَلَيْكُمْ عَلَيْ عَلَ مَعْلَيْ عَلَيْ مَعْلَيْ عَلَيْ عَلَ

(8.174) تَى تَعْدَ مَرْجَا مَالَى اللَّهُ اللَّهُ اللَّهُ عَلَى اللَّهُ الللَّهُ اللَّهُ الللَّهُ اللَّهُ اللَ اللَّهُ اللَّهُ اللَحُلْقُولُ اللَّهُ اللَّهُ اللَّهُ اللَّهُ اللَّالَةُ اللَّهُ اللَّهُ اللَّهُ اللَّهُ اللَّهُ اللَّهُ اللَّهُ اللَّهُ اللَّالَةُ اللَّهُ اللَّالَةُ اللَّالَ اللَّالَةُ اللَّالَةُ اللَّالَ اللَّالَةُ اللَّالَ اللَّهُ اللَّالَ اللَّ اللَّهُ اللَّهُ اللَّالَ اللَّالَ اللَّالَ اللَّا اللَّاللَّالَةُ اللَّالَةُ اللَّالَةُ اللَّا اللَّا اللَّالَةُ اللَّالَةُ اللَّالَةُ اللَّالَةُ اللَّالَةُ اللَّ

For more on the postpositions t' on zin(=gi) and pakozin(=gi), see §3.6.8, §5.6.2 and §15.8.5.3. Finally, (8.175) records a unique (interrogative) example of a local non-standard construction where an equative auxiliary accompanies the progressive:

(8.175) अन्त मुरा सन्त मन्त मन्त में रायवेव भन्ये?

lenge? dã: gãto:=na kor-zin be-po? PRN.HON yesterday TPN=LOC go.around-PROG EQU.NE-2INF 'Were you roaming in Gangtok yesterday?' (PL e)

Attaching *-po* to the neutral equative $b\varepsilon$? is a marginal phenomenon, see §11.1.2.2. One consultant wanted to replace $b\varepsilon$ -*po* here with $j\dot{\rho}$ -*po*.

8.3.3.2 Alterphoric progressive VERB-tcuŋgɛ/zuŋgɛ

The progressive construction VERB-*tcuŋgɛ/zuŋgɛ* (also *-tcyŋgɛ/zyŋgɛ*, in Martam *tcougɛ*) does not occur in written Denjongke. This construction is probably an abbreviation of the fuller form VERB-*tcɛn dukɛ*, which also occurs in writing. Because in my data VERB-*tcuŋgɛ/zuŋgɛ* does not occur with 1SG actors, I have tentatively and analogously to the completive construction, which most likely uses the same marker (see §9.1.3), glossed *-tcuŋgɛ/zuŋgɛ* as alterphoric progressive (PROG.APH). The term "alterphoric" here simply means that the form is incompatible with first person actors.³²⁵ The alterphoric progressive is here illustrated in (8.176-178).

³²⁵ The examples in my data have third person actors. Uses with second person actors are left open for future research.

(8.176) ह्वा पा स्वयूषा कुषा की स्वर प्रवित वी खेंदी 326

dawa $námg\varepsilon:=gi \ k\varepsilon^2 \ t^h on-zy\eta g\varepsilon$ jòu.PNPN=GENvoice become-PROG.APHup'Dawa Namgyal's voice is calling out from up (there).' (PT kitchen duscussion)

- (8.177) \hat{a}_{i} $\alpha \hat{\beta}_{i}$ $\alpha \hat{\beta}_{i$

8.3.4 Durative secondary verb *bak*

The secondary verb *bak* 'carry' can be used either quite literally referring to carrying something on oneself, as in (8.179), or more metaphorically referring to carrying on doing an action, see (8.180) and (8.181). In the latter case, *bak* has overtones of durativity or continuity, as suggested by the fact that carrying something along is an event of some duration. In (8.181), durativity/iteration is further signaled by reduplication of $ts^h \emptyset$: *bak* 'search carry'. As suggested by all the three examples, the secondary verb *bak* typically occurs in a nonfinal construction (i.e. followed by *-ti/di*).

(8.179) गाम्रा भ्रुत रेलामु र्हेषा तनगा है लेगा भ्रेनण

karma mén ri:bu pò: bak-ti lòk lɛp. PN medicine tablet buy carry-NF return arrive 'Karma, having bought and brought the medicine, arrives back. (Richhi 11)

karma t'ã: nòrbu p^hi:ts^ham tc^huts^hø? zi=lo tc^høki? døsa dusk clock.time four=DAT PN and PN dwelling PN số: rit^{ch}i bompu tap bak-ti. went hope big sow carry-NF 'Around dusk at four o'clock Karma and Norbu go to Choki's dwelling, having great hopes.' (Richhi 96)

³²⁶ One consultant wanted to replace ਛੱਡ 'ਸਕਿਤ' ਪਾ *thøn-zyŋ-gɛ* with ਛੱਡ 'ਸਕਿਤ' ਪਨ੍ਤਾ 'ਸੇ' *thøn-zɛn du-kɛ*, underlining the fact that the former is probably a reduction of the latter.

lấ: k^hu ódi-p'ja k'jõ:=na ts^hØ: bak ts^hø: bak that-ADVZR 3SGM village=LOC bull search carry search carry sốː-di $k'j\tilde{o}:=t\epsilon i=na$ lấ: $tc^h a = tci?$ t^hop-ti go.PFV-NF village=INDF=LOC bull pair=INDF find-NF 'Therefore going to village(s) keeping on searching and searching for bull(s) and finding a pair of bulls in one village he...' (TB bull story)

8.4 Tense, aspect and modality with the construction VERB-INF EX

The construction VERB-INF EX resembles an existential clause (8.182) or a locative/possessive clause (8.183) where a verbal action in infinitive occurs in place of a located/possessed noun.

(8.183) $\mathbb{R}^{i\tilde{A}'/\mathbb{R}'}$ $\mathfrak{q}^{\tilde{M}}\mathfrak{q}' \mathfrak{g}^{\tilde{A}'/\mathbb{R}'}$ $\mathfrak{q}^{\tilde{M}}\mathfrak{q}' \mathfrak{g}^{\tilde{A}'/\mathbb{R}'}$ $\mathfrak{q}^{\tilde{M}}\mathfrak{q}' \mathfrak{g}^{\tilde{A}'/\mathbb{R}'}$ $\mathfrak{g}^{\tilde{A}'/\mathbb{R}'}$ $\mathfrak{g}^{\tilde{A}$

As a sign of grammaticalization, the genetive or locative marking of the possessor/location in (8.183) has became optional.

The construction is negated by replacing an affirmative copula by a negated one. For negated declaratives, consider (8.184) and for negated interrogatives, see (8.185).

(8.184) a) הَ عَ تَ حَدَمَةُ عَلَى اللَّهُ اللَّهُ عَلَى اللَّهُ اللَّهُ اللَّهُ اللَّهُ عَلَى اللَّهُ (KN e)
b) הَ أَحْدَ مَعْ عَرَجَعَ اللَّهُ (KN e)
b) הَ أَحْدَ مَعْ عَرَجَعَ اللَّهُ اللَّالِ اللَّ اللَّالَ اللَّالَةُ اللَّهُ اللَّهُ اللَّهُ اللَّهُ اللَّهُ اللَّالِ اللَّالَةُ اللَّالَةُ اللَّالَةُ اللَّالَةُ اللَّالِي اللَّهُ اللَّهُ اللَّهُ اللَّهُ اللَّهُ اللَّالَةُ اللَّالِ اللَّالِي اللَّالَةُ اللَّالِ الللَّالِي اللَّالَةُ اللَّالَةُ اللَّا عَلَيْعَالِي اللَّالَةُ الللَّالَةُ اللَّالَةُ اللَّالَةُ اللَّالَةُ اللَّالَةُ الللَّالَةُ اللَّالِي اللَّالَةُ اللَّالَةُ الللَّالَةُ اللَّالَةُ اللَّالَةُ اللَّالَةُ اللَّالَةُ اللَّا اللَّا اللَّا اللَّالَةُ اللَّالَةُ اللَّالَةُ اللَّا اللَّا اللَّالَةُ اللَّالَةُ اللَّالَةُ اللَّالَةُ اللَّالَةُ اللَّالَةُ اللَّالَةُ اللَّالَةُ الللَّالَةُ اللَّالَةُ اللَّالَةُ اللَّا اللَّا اللَّا اللَّا اللَّالَةُ الللَّالَةُ اللللَّالَةُ الللَّا اللَّا الللَّا الللَّا الللَّا ال

(KT discussion with TB)

(8.185) a) यु. गर्थेग. र्जुरा. मून. केव. २८ ग. ग?

k^hu jό? p'ja-cε? mìndu-ka?
3SGM work do-INF NEG.EX.SEN-PQ
'Isn't he working?'/'Isn't he going to work?' (KN e)

b) גַי קעֿק׳ לָאִיָּקְ׳ אָק׳יַק? *k^hu jó? p'ja-cɛ? mè:-ka*?³²⁷ 3SGM work do-INF NEG.EX.PER-PQ 'Isn't he working?'/'Isn't he going to work?' (KN e)

The construction VERB-INF EX can obtain various shades of meaning. Whereas (8.183) above marks prospective action in the future, (8.186) and (8.187) below express abstract possession of a tradition (in the past) and something to say (in the present/future), respectively.

- (8.186) הָדְיָלָי צָּשָׁקְיֹמִי הִישּׁמְי בּוּרָ עָרְי עָרְישׁוּמִקָי צָּרְ הן t'ã:pu nénlo nàtea? rìn ze:-ee? jò:-kʰɛn bɛ? t'a. long.ago before 1PL price obtain.HON-INF EX-NMLZ EQU.NE now 'Long ago earlier we had (the custom of) receiving money (for the bride).' (SGD marriage customs)
- (8.187) $\neg ; \check{\mathfrak{F}}; \check{\mathfrak{N}}; \check{\mathfrak{S}}$ $\check{\mathfrak{N}}; \check{\mathfrak{S}}$ $\check{\mathfrak{N}}; \check{\mathfrak{S}}$ $\check{\mathfrak{N}}; \check{\mathfrak{N}}; \check{\mathfrak$

In example (8.188), the meaning is present habitual.

(8.188) कें.नु८्'ग्रीक' र्ववाक' र्वत्र भूट' केंट्र'ग?

 $m \delta by = gi \quad p^h o? \quad t^h op - \varepsilon \epsilon? \quad j \delta$:-ka? wife=AGT salary receive-INF EX.PER-PQ 'Does the wife get a salary?' (BP BB discussion)

The same construction can also express what the speaker can or cannot do, referring either to ability, as in (8.189-191), or willingness, as in (8.192).

(8.189)	र्षेक्षषाःच्वनः अन्यःक्षायत्रः यन्तिः यत्यः भन्तः कोन्दिन्दः कार्त्रे नषाकाः ननन्दः भन्तः यनुषाःगो								
	símtcẽ:	làp-k ^h ɛn=di=jãː	$k^h a$	làp-се?	mè:-ruŋ	nó:sam			
	animal	say-NMLZ=DEMPH=even	mouth	say-INF	NEG.EX-CONC	thought			
	tã∶ -cɛ?	du -kɛ.							
	send-INF EX.SEN-IN								
	'The animal, even though it cannot talk, can think (I see).' (RB butcher story)								

³²⁷ The difference between (8.185a) and (8.185b) lies in what the speaker expects the addressee's level of knowledge to be. If the addressee is supposed to have personal knowledge, $m\dot{e}$:-*ka* is used. On the other hand, if the speaker expects that the addressee needs to check what the case is, *minduka* would be used. For further information on evidentiality in questions, see §7.2.1.2 and §7.2.2.1.

(8.190) ตาร์าสู่าที่ลา เลาสุกุลา ณินา สิยุลาร์สุลา ณนา.ครา ณีราสเตสา มหา

 $\dot{a}dzo=tsu=gi$ $k^{h}atam$ $l\dot{e}p$ dzikta? $l\dot{a}p-ce?$ $j\dot{o}:-k^{h}en$ be?. grandfather=PL=AGT proverb very excellent say-INF EX-NMLZ EQU.NE 'The grandfathers have most excellent proverbs to tell.' / 'The grandfathers are able to tell most excellent proverbs.' (KN field notes)

(8.191) ନି' ୍ଟ କ୍ସୁ କ୍ଟିକ୍ଟ କ୍ରମ୍ୟ କ୍ଟ ରିମ୍ବ

te mù loke kjap-ce? mè?. well 3SGF Lhoke speak-INF NEG.EX.PER 'Well, she cannot speak Lhoke.' (SN kitchen discussion)

 $y\dot{a}$ $k^{h}u=i=tsa$: **gju-ce? mè?**. 1SG 3SGM=GEN=at go-INF NEG.EX.PER 'I cannot go to his place (because of our bad relationship).' Lit. 'I have no going to his place.' (KUN e)

The negated form of the verb \hat{o} : 'come', *ma-h* \hat{o} : [mah \tilde{o} :]³²⁸, can replace *mè?/mindu?* as negated existential (see §7.2.5.1):

(8.193) \tilde{w} $a\hat{f}$, $a\hat{f}$, \hat{f} , $\bar{c}\hat{a}$, g, g, π , \tilde{a} , \tilde{a} , $\tilde{c}\hat{u}$ - $\varepsilon \epsilon^2$ ma- $h\hat{o}$. di=di te $p\hat{e}$: $\epsilon\hat{u}$ - $\varepsilon \epsilon^2$ ma- $h\hat{o}$. that=DEMPH so 1SG.GEN say.HUM-INF NEG-come 'I cannot tell that.' Lit. 'That is not mine to tell.' (PAD Tashiding story)

8.5 Modality (probability, possibility and necessity)

The term "modality" here refers to the speaker's judgments about a proposition in a very general sense.³²⁹ Judgments about the factuality/certainty of a proposition are epictemic modals. Other types of modal judgments which may be made explicit in Denjongke grammar are judgments about obligation, possibility, permissiveness, temporality, morality, ability and trustworthiness/genuineness (pretensive construction). As shown by Table 8.4, the grammatical means for expressing modality distinctions can be divided into one suffix, three complex constructions and eight secondary verbs. Among secondary verbs, the category "modality" is assigned to those markers which are in linguistics typically treated under the category modality (ability, obligation, permission, possibility). For other secondary verbs and introduction to secondary verbs in general, see §4.2.3.

³²⁸ The pronunciation differs from the non-copular negation $ma \cdot \dot{o} :> [m\tilde{o} :]$.

³²⁹ This definition is inspired by Palmer's (2001: 8) characterization of epistemic and evidential modalities as expressing the speaker's jugdments about a proposition.

Function	·	Form		Meaning	Morpheme
					type
Epistemic/	probabilitative	-to	Ťٜ; Ť; š;	'maybe, probably'	suffix
certainty	apparentive	(<i>-po</i>) <i>da</i> (EQU)	<u>جج</u> .	'seem'	complex
	approximative	raːgju, tʰɛːgju	?	'seem'	constr.
Obligation/		go?	न्यॉर्थः	'need to, must'	secondary verb
deontic					
Permission	objective	tc ^h o?	र्केया'	'be allowed'	
	subjective	tup	বদ্যুব'	'deem fit'	
Evaluation	temporal	ren	न्तु:	'be time to'	
	practical/moral	(mi-)lɛ? ³³⁰	(মৃ.)ন্যবাধ্য	'be good to'	
Ability	general	ts ^h u?	র্ন্তুবাৎ্য	'be able to'	
		сé:	.विह्य.	'know (how to)'	
	physical	k ^h ø?	ĨĨŢ	'have strength to'	
	temporal	lồ:	а́г.	'have time to'	
Possibility		sí?	£15.	'be possible'	
Tentativity		ta	ar.	'look'	
Pretension		-dzin t ^h ap	हुँब. घय.	'pretend to'	complex
					constr.

Table 8.4. Modality markers

8.5.1 Probabilitative -to

The probabilitative *-to* attaches directly to the verb root (8.194) or the completive marker *-ts^ha*: (8.195) and is not followed by any tense, aspect or evidentiality markers. The certainty implied by *-to* varies between 'maybe' and 'probably'.

(8.194) र्रेलाकें लुपाकें ते अग्नेतरों

rolmo εù-watεεnε k^hεn-ţo. cymbal say.HON-COND know-PROB 'If (I) say [rolmo] (you) probably know (what it is).'/'(You) probably know (what is a) [rolmo].' (YR canteen video)

(8.195) ยๆ ริ่นางสา อีสายนางสาง ลุณา ลัสมสารทับ

 $t^ha:rin=l\varepsilon$ $tc'\phi n-c\varepsilon = di:$ $c\varepsilon:$ $kom-ts^ha:-to.$ far.away=ABLcome.HON-INF=DEMPH.AGTmouth.HONthirst-CMPL-PROB'Since (you)have come from far away, (you)have perhaps become thirsty.'(rnam-rtog 5)

The probabilitative is negated by the prefix *mi*-.

³³⁰ This form mainly occurs as negated.
(8.196) हे' याया अग्नेत्र उद्ग दर्दी' ख्याया या के जलया गाँ।

 $t\varepsilon$ làla k^h εn -ru η di t^hu:=lo **mi-za:-to**. then some know.HON-CONC this mind.HON=DAT NEG-set-PROB 'Then although some (may) know (the story), (they) may not have committed it to memory.' (PAD bet story)

(8.197) केंद्र द्र' कर्के वर्द्र' मुवा के कुंगुरा र्यो

 $tc^h \phi$?t'a $ts^h o = di$ $g \varepsilon$: $mi - ts^h u: -to$.2SG.Lnowlake=DEMPHcrossNEG-be.able.to-PROB'Now you probably won't be able to cross the lake.' (KT animal story)

The probabilitative co-occurs with personal copulas, see (8.198) and (8.199), but not with neutral (8.200) or sensorial copulas (8.201).

(8.198) र्क्तेस क्षेत्र ग्राँ।

ts^hom ín-ţo. mortar EQU-PROB 'It's maybe a mortar.' (TB e)

- (8.199) স্থ্রবাহ্মার্ বাধ্যনা বাথান্ডরান্ট অবির্গা ? *lóbdø: jó? k'ɛ:tɛɛ̃:=to mè:-to.* school.GEN work important=CEMPH NEG.EX-PROB 'There's probably no important school work.' (Richhi 55)
- (8.200) *逝河 習行河 **ts^hom bε-to*. mortar EQU.PER-PROB
- (8.201) *สังลา ขุณิชา ขุณา ชองาร์า สิสา (45 ขา) ขัง
*lóbdø: jó? k'ɛːtɕɛ̃:=to
school.GEN work important=CEMPHmìndu:-to.

The probabilitative also attaches to auxiliary copulas:

(8.202) הַמישּקיקֿידִי 1,50000 מְקִיצִּין אָדִידִי אָּקיאָן אָדָייָ שָּׁאָי שִאָּדיין t'y:mene=ra de:d lak de:tei? số:-bo ín-to, k^him about=AEMPH 1,5(Nep.) lakh(Nep.) that.much go.PFV-2INF EQU-PROB house teo:-pø: gã:. repair-2INF.GEN time 'Maybe about some 1,5 lakh (rupees) went when repairing the house.' (KT discussion with TB)

The verb \hat{o} : 'come' functions as an existential, which often co-occurs with the probabilitative marker:

(8.203) 委 萬二河 ts^ha **o:-to**. salt come-PROB 'There's probably salt.' (KN e)

The form $\hat{o}:-to$ may be postposed as an auxiliary to verb roots instead of mere -to (compare with 8.194):

(8.204) אָאָקיקצֿואָ אופאַ אָרָיקַיאַשָּאַן *ווויאס אופא אינער אופא אינער אינער אופא געריין און אופא אינער אופא געריין אופא געריין אופא געריין אופא געריין אינער אופא געריין אינער אופא געריין אינער אופא געריין אינער געריין אינער געריין גערין געריין גערין גערין געריין געריען געריין געריין געריין געריין געריין געריין געריעעען געריין געריין געריין געריען געריען געריין געריין געריען געריען געריען געריין גען געריין געריין גען געריין געריין געריין געע*

In the novel Ricchi, -to is six times out of total seven followed by the morpheme ni (spelled both $\hat{\mathfrak{g}} \cdot ni$ and $\hat{\mathfrak{g}} \cdot ne$), which is probably a loan of the Nepali discourse-oriented attention marker ni (see Yliniemi 2016b), rather than a cognate of the Classical Tibetan topicalizer -ni (see Beyer 1992: 275-278) This collocation may be typical of Denjongke spoken in Yanggang, the native place of the author. The rest of my data has dozens of examples of -to but only one followed by ni. If ni in (8.205) is indeed an attention marker, it probably signals a slight change in the topic of the discussion.

(8.205) 5'या में आ में में अब मुंब दी की का गा?

hapa bombai=lo **man-dzøn-to ni** *iŋ-ga.* now Bombay=DAT NEG-come.HON-PROB AT(Nep.) EQU.PER-PQ 'You perhaps haven't so far been to Bombay, have you?' (Richhi 101)

8.5.2 Apparentive *da*

The Classical Tibetan verb α_{5} da 'be similar' (see Beyer 1992: 254) also occurs in Denjongke as a marker which expresses what, in the speaker's opinion, apparently is the case, hence the term "apparentive" and the gloss "AP." The apparentive occurs independently without other verbal elements and also postposed to the verbal suffixes $-\epsilon\epsilon(2)$ (1. infinitive), -po/bo (2. infinitive) and -to/do (imperfective). It would be problematic to name da a apparentive copula, because it is very frequently followed by an equative copula.³³¹ For the independent, copular type of uses, consider (8.206) and (8.207):

³³¹ The combination $da b\epsilon 2$, however, merges into an apparentive equative copula $d\epsilon z/r\epsilon$; which is discussed in §7.2.4.

(8.207) $\check{\alpha} \prec \check{\gamma} \cdot \check{q} \prec (gang) \land \varsigma \cdot \check{q}$ $k^{h} \tilde{o} := to \qquad g \varepsilon \eta \qquad da = s.$ $3PL = CEMPH \qquad gang(Eng.) AP = QUO$ 'They seem (to belong to) one group.' (KN kitchen discussion)

Uses of da postposed to a combination of verb and verbal suffix are more frequent than independent copular type of uses. The following examples illustrate the co-occurrence of da with past action marked by the second infinitive -po/bo/u (8.208-212), future action marked by the first infinitive $-ce^{2}$ (8.213-215) and ongoing action marked by the imperfective -to/do (8.216-220).

Past with -po-infinitive

When preceded by the *-po*-infinitive form of the verb, the main verbal action in the apparentive construction is typically understood as having taken place in the past, see (8.208-210). Example (8.208b) illustrates the negated construction.

(8.208) a) איז אקדי מדיאי מבי אַרן

 $k^h u$ $d\tilde{a}$: ∂m -bo $da b \epsilon ?$.3SGMyesterdaycome-2INFAP EQU.NE'It seems he came yesterday.'(KUN e)

b) תי אקדי אימדיאי מקי ארן

 $k^h u$ $d\tilde{a}$:ma-om-bo $da b \epsilon$?.3SGMyesterdayNEG-come-2INFAP EQU.NE'It seems he didn't come yesterday.' (KUN e)

(8.209) ผลุลผา ครา พรรา เกาสุรา พัญาร์สาร์า ครา

 $n\acute{a}m=di$ $d\widetilde{a}$: $k^{h}a$:nup $l\acute{o}k$ te'øm-bo daughter-in-law=DEMPH yesterday the.day.before.yesterday return come-2INF da.

nnorontlu

'Apparently the daughter-in-law returned yesterday or the day before yesterday.' (SN kitchen discussion)

(8.210) \check{w} ah \check{c} \check{w} \check{s} \check{s} \check{q} \check{q} \check{q} \check{s} \check{v} \check{q} \check{s} \check{s} \check{q} \check{s} \check{s} \check{s} \check{q} \check{s} \check{s} \check{s} \check{q} \check{s} \check{s} \check{s} \check{q} \check{s} \check{s} \check{q} \check{s} \check{s} \check{q} \check{s} \check{s} \check{q} \check{s} \check{q} \check{s} \check{s} \check{q} \check{s} \check{s} \check{q} \check{s} \check{q} \check{s} \check{s} \check{q} \check{s} \check{s} \check{q} \check{s} \check{s} \check{s} \check{s} \check{q} \check{s} \check{s} \check{s} \check{q} \check{s} \check{s} \check{s} \check{q} \check{s} \check{s} \check{s} \check{q} \check{s} \check

Exceptions to the generalization that the apparentive constructions with *-po*-infinitive refer to past actions are stative verbs, see (8.211), and forms negated with the imperfective negator mi-, see (8.212). The copula (which is a stative verb) in (8.211) expresses a state holding at the time of speaking. In (8.212), on the other hand, the *-po*-infinitive is negated by mi-, which implies that the

reported state holds at the time of speaking and in the projected future. The negation in (8.212) contrasts with the negation with *ma*- in (8.208b), where one past act is negated.

(8.211) รารีรา ราชส์ สิาส์ สรีส์ สรีลา พรณาตา สรสาส์ พิสาร์ สรา พิสาๆ พาสู di: $dz \varepsilon: k^h a t^h a m o$ t'arin nàtci mí-ts^he ím-bo dа 1PL.GEN human-life this.GEN meeting last today EOU-NMLZ AP íŋ-ga? EQU.PER-PQ 'It seems like today is our last meeting in this human life, doesn't it?' (Richhi 173)

(8.212) यु. गर्धेग दर्न के र्चुक र्य दर्ज

 $k^{h}u$ $j \phi := di$ **mi-j \overline{-}u qa**. 3SGM work=DEMPH NEG-do-2INF AP 'Apparently he's not doing the work.' (KN e)

Future with *-ce*-infinitive

With $-\varepsilon\varepsilon$ -infinitive, the apparentive construction refers to apparent future action. In (8.215), the sequence $da \ b\varepsilon$? merges into the apparentive copula $d\varepsilon$? (also $d\varepsilon$:/ $r\varepsilon$:/ $r\varepsilon$?).

(8.213) ד' אָז' פּיָרָ בד' ד' בי ד' אָד' בּד' שִׁמִי אָד' בּד' בּבּ שָׁמ vear twenty that.much now 1PL.GEN language=DEMPH disappear-INF AP t:=la. EQU.PER=HON 'Now it seems that (in) some twenty years our language will disappear.' (RL interview)

(8.214) جَرَّمِعَنَّ جَرَّعَنَّ مَعَرَّجَمَّ مَعَرَّجَمَّ مَعَرَّ حَيَّ مَعَرَّ حَيَّ مَعَرَّ مَعَرَّ مَعْرَ مَعْرَ مَعْرًا مَعْرَ مَعْرًا مَعْرَا مَعْرَ مَعْرًا مَعْرَا مَعْرًا مُعْرَا مُعْرَا مُعْرًا مُعْرَا مُعْرًا مُعْرَا مُعْرَا مُعْرًا مُوالْعُا مُعْرًا مُعْرًا مُعْرًا مُعْرًا مُعْرًا مُعْرًا مُعْرًا مُعْرًا مُعْرًا مُوالْحُولُ مُعْرًا مُعْرًا مُعْرًا مُعْرًا مُعْرًا مُوالْعُولُ مُعْرًا مُعْ مُعْرًا مُ فَعْرًا مُعْرَا مُعْرَا مُعْرَا مُعْرَا مُعْرًا مُعْرَا مُعْرًا مُعْرًا مُعْرًا مُعْرًا مُعْرًا مُعْرًا مُعْرًا فَعْلَمُ مُعْرَا مُعْرَا مُعْرَا مُعْرَا مُعْرَا مُعْرَا مُعْرَا مُعْرًا مُعْرًا مُعْرًا مُعْرًا مُعْرًا مُعْ

(8.215) क्षे मॅबर्फे रखे सुवार्थे र्हेन वन वहना (=वर भना)

 ε :, t' = to ranipul=lo $d = \varepsilon \varepsilon$ d = 2. oh meaning=CEMPH TPN=DAT stay-INF AP.EQU.NE 'Oh, (that) means it looks like (she) will stay (the night) in Ranipool.' (KNM kitchen discussion)

Ongoing with imperfective

In conjunction with progressive marker, the apparentive construction, unsurprisingly, refers to apparent ongoing action. It is noteworthy that the equative copula \tilde{t} (neg. $m\tilde{\tilde{\varepsilon}}$) may occur in the progressive apparentive construction, as shown in (8.217) and (8.218).

- (8.217) क्रॅंग् तद्रेया क्षेत्र में तद्

ódɛm=tɛi? ín-do da. like.that=INDF EQU-IPFV AP 'It seems to be like that.' (YR boys' and girls' clothing)

(8.218) arg. 2. 22

mèn-doda.NEG.EQU-PROGAP'It seems not to be (so).' (KNM kitchen discussion)

t'ato $k^h \tilde{o}$: pi:-po pén t^ha:riŋ=lo dze:=ki kiː-kiː ý: place far.away=DAT marriage now 3PL two-col after=GEN enjoyment-RDP né:kor (hanimun=lo) gju-do da-u j*à*?. sighseeing.tour honeymoon(Eng.)=DAT go-IPFV AP-2INF EX.PER 'Now the two of them are as if going on a post-wedding enjoyment-tour (honeymoon).' (Richhi 129)

In the majority of instances in my data, da is followed by an equative copula, most often $b\varepsilon$?, see (8.208), (8.210), (8.216), (8.219), (8.221) and (8.222), but also with \tilde{t} , see (8.213).

Through frequent use, the combination $da b\varepsilon^2$ has merged into an apparentive equative copula $d\varepsilon:/d\varepsilon^2/r\varepsilon:/r\varepsilon^2$, see also §7.2.4. Some of my consultants were not aware that $d\varepsilon:/r\varepsilon$: is related to $da b\varepsilon^2$. The apparentive copula is often accompanied by the attention marker $=\varepsilon o$, which marks attention-worthy, often new information (see §16.2.2 and Yliniemi 2016a). The co-occurrance of the apparentive copula with the attention marker is quite natural, because both markers are often used in the context of finding out something new, as illustrated in (8.221) and (8.222).

(8.221) 減点子 新方 新方 新方 新方 新方 新方 新方 新一 (= 高方 新方 新)
 ódi gã: nò:-nò: rɛ:=co. that time buy-buy AP.EQU.NE=AT
 '(They) had apparently bought (those) at that time (I found out).' (TB discussion with KT)

(8.222) \tilde{w} (\tilde{a} \tilde{f} , ηq , \tilde{a} ? \tilde{w} , \tilde{c} , \tilde{c} , \tilde{w} , \tilde{c}

When -da is negated by the negator-prefix mi-, the meaning of the form is 'be different':

(8.223) $\int (3\pi)^{n} a^{n} da^{n} da^$

(8.224) לישה קראי מאסימה אסימה אסימה אסימה אסימה (1.224) *t'iza: t'ã:pu dem min-da be?.* but long.ago like.that NEG-AP EQU.NE 'But (it) was different from how (things were in) in earlier times.' (DB day trip)

Because the negation of da ($b\epsilon$?), min-da ($b\epsilon$?), has the meaning 'be different', da itself cannot be negated and retain the meaning 'does not seem'. Rather, the associated verb needs to be negated, quite analogously to the English expression 'seems not to'.³³² Example (8.225) is negated in (8.226).

- (8.225) \mathbb{R}^{n} $anggrives \tilde{a} \in \tilde{A}^{n}$ $\tilde{a} \in \tilde{A}^{n}$ $\tilde{a} \in \tilde{A}^{n}$ $\tilde{a} = \tilde{A}^{n}$
- (8.226) B. arggreen arggreen

In addition to the apparentive uses, *da* is used as the base of the adjectives *dau/dou* 'similar, like' (neg. *man-dau/man-dou* 'dissimilar, unlike'), see (8.227), and *da:mandau/da:minda* 'different kinds of', see (8.228).

(8.227) אָזי מָלָ זָדאָ מְקָמָ שִׁק שִׁן p'um=di nòrbu dou i̇́:=s. girl=DEMPH gem like EQU.PER=QUO 'The girl is like a gem (it is said).' (SGD wedding customs)

³³² English has two options for negating the apparentive *seem*, 'He doesn't seem to have come' and 'He seems not to have come', whereas in Denjongke the negation of the apparentive is possible only analogously to the latter English construction.

(8.228) यानुमः मुया पदा मेवा पदा गोषा दा धाँदा आपवा भ्रदा

tamgy? daminda $k\epsilon$: po $j\dot{\phi}$:- $k^{h}\epsilon n$ $b\epsilon$?. legend different.kinds.of a.lot EX-NMLZ EQU.NE 'There are a lot different types of legends.' (SGD cave story)

8.5.3 Approximatives *da:gju* and *t*^h*ɛ:gju*

There are further two constructions that resemble in meaning the probabilitative and apparentive forms, da:gju/da:kju (may also be pronounced ra:gju/ra:kju, written tentatively as $a:g_{i}(a:g)$) and $t^{h}\varepsilon:gju$ (written tentatively as $a:g_{i}(a:g)$), which occur in the same positions as the apparentive da ($b\varepsilon$?). The form da:gju/da:kju, which may include the apparentive da, is used at least in Ralang and Tashiding in West Sikkim, whereas $t^{h}\varepsilon:gju$ is used in Martam, East Sikkim. The two forms are here, inspired by the use in (8.229), tentatively glossed as 'approximative' (APPR). However, the meaning is in many instances indistinguishable from the probabilitative and apparentive forms. Similarly to copulas, the approximative forms may occur independently without other verbal marking:³³³

- (8.229) สูาฉ้าล้าร์ พิลงาฏลา (school) ฏิ ชิ (fee) สาวสูา กรากสูาลฤญาลฤพ dou=lo=to iskul=gi p^hi: ŋábgja ra:kju=la.³³⁴ month=DAT=CEMPH school(Eng.)=GEN fee(Eng) 500 APPR=HON 'The monthly school fees are around 500 (rupes).' (TB discussion with KT)
- (8.230) জাঁব নাই ন্যাঁম পন্ পন্ পন্ পন্ পন্ প্রা *óna dzo go:=ce ra:kju*. there build be.needed=INF APPR 'It appears (we) have to build (it) there.' (TB discussion with KL)

The following examples illustrate the use of the approximative in auxiliary position. Examples (8.231-233) illustrate uses postposed to a nominalized verb, and (8.234) a use with the progressive.

(8.231) בל שאי קראיצי ד' אי קד אי קי חצי אראי מביתש

pè: ám t'õ:-ti t'a lò teu teeŋa **làm** 1SG.GEN mother die.HON-NF now year ten fifteen reach.2INF **da:gju.** APPR 'It's probably ten to fifteen years since my father died.' (DB life story)

(8.232) जन्मार्च तन् त्यु, म अर्धेनार्च भन्

 $t\tilde{a}$:-bo da:(k)ju, $\eta \dot{a}$ $ma-t^hob=b\epsilon$?. send-2INF APPR 1SG NEG-receive.2INF=EQU.NE 'Probably (they) sent (it), (but) I didn't receive (it).'(PED life story)

³³³ My data has no examples of the independent use of $t^{h}\varepsilon : gju$, but because $t^{h}\varepsilon : gju$ seems the Martam equivalent of ra: gju, independent uses probably do exist.

³³⁴ This clause is a typical case of code-mixing. The official language in schools is English and therefore the speakers are more used to the English expression "school fee" than its Denjongke equivalent $lopta=gi la \tilde{a}_{TT}$ ".

- (8.233) $a^{av} = \pi a^{av} a^{b} a^{av} a^{av} a^{b} a^{av} a^{av$
- (8.234) हे केंप्रदे केवर्ट वेप्रणु

tε ódε: ín-do t^hε:gju. so like.that EQU.PER-PROG APPR 'So it seems to be like that.' (KN kitchen discussion)

8.5.4 Permissives *tcho?* and *tup*

The two permissive secondary verbs $tc^h o?$ 'be allowed' and tup 'deem fitting, deem proper' have partly overlapping semantics. By using the secondary verb $tc^h o?$ 'be allowed, be right' the speaker evaluates whether something is permissible or not with reference to some moral or other norm. Hence $tc^h o?$ may be termed an objective permissive in contrast to subjective permissive tup, which expresses that evaluation happens with reference to individual person's tastes and practical purposes. The permissives may be negated by either ma- or mi-, used with past and present respectively. According to informant KN, the past form of the affirmative $tc^h o?$ is $tc^h ok$ -o $\hat{t}:/bc?$, although this form does not occur in my natural examples. See (8.235) for an affirmative example and (8.236-237) for negated examples. Note that in (8.235) the affirmative use has the meaning 'should'.

(8.235) न्युले'र्न् तनेया र्जुया केंगा

 $c\dot{c}:-to$ dcm p'ja tcho?. seize-PROG like.that do be.allowed 'it's like (they) should be caught' (CY interview)

(8.236) तहा गॅावा के केंगा

t^hu ko: mi-tc^ho?. pick throw NEG-be.allowed 'It's not right to throw (it) away (like that).' (KNA kitchen discussion)

(8.237) ८. ज्या योषा पट्ने मुन्य मुन्य के केंया के

 $y \dot{a}tca = ki = di$ kum kjap mi-tc^ho:=s. 1PL=AGT=DEMPH stealing do NEG=be.allowed=QUO 'We are not allowed to steal.' (YR canteen video)

The affirmative form may be followed by an equative copula (in my data, the negated form is not followed by an equative):

(8.238) $x_{\nabla} = x_{\nabla} = x_{\nabla$

(8.239) אָז מי מקי אָקן *láma làp-ruŋ làp teʰo bɛ?.* lama say-CONC say be.allowed EQU.NE 'It's alright to call (it) [lama].' Lit. 'Even if you call (it) a lama, it's alright.' (KNA kitchen discussion)

Whereas te^{ho2} 'be allowed' makes reference to some objective norm against which an action is evaluated, the secondary verb tup 'deem fitting' expresses a more subjective evaluation of the appropriateness of an action, see (8.240-242). In (8.240), an M.C. is asking a festival crowd whether they enjoy the program.

(8.240) 직명자계? *tup-ka*? deem.fitting-PQ 'Is (it) fitting/alright?' (NAB, oh)

- (8.241) هُمَّ عَلَيْهُ مَعْنَ عَلَيْهُ عَلَيْ عَلَيْ عَلَيْهُ عَلَيْهُ عَلَيْهُ عَلَيْهُ عَلَيْ عَلَيْ عَلَيْ عَلَيْ عَلَيْهُ عَلَيْهُ عَلَيْ عَلَيْهُ عَلَيْهُ عَلَيْهُ عَلَيْ عَلَيْهُ عَلَيْهُ عَلَيْ عَلَيْ عَلَيْ عَلَيْ عَلَيْ عَلَيْهُ عَلَيْ عَ المَا عَلَيْهُ عَلَيْهُ عَلَيْهُ عَلَيْهُ عَلَيْهُ عَلَيْهُ عَلَيْ عَلَيْ عَلَيْ عَلَيْ عَلَيْ عَلَيْ عَ المُعْلَيْ عَلَيْهُ عَلَيْهُ عَلَيْهُ عَلَيْهُ عَلَيْ عَلَيْهُ عَلَيْ المَعْلَيْ عَلَيْ عَ المَا عَلَيْ عَ
- (8.242) $\eta_{a'}\hat{a}_{5}\cdots \hat{a}_{5} \hat{a}_{5}\cdots \hat{a}_{5} \hat{a}_{5}\cdots \hat{a}_{5}\cdots \hat{a}_{5}\cdots \hat{a}_{5}\cdots \hat{a}_{5} \hat{a}_{5}\cdots \hat{a$

Although in (8.240-242) a human agent makes the evaluation, *tup* may also occur in contexts without a human agent, as if fate is personified:

(8.243) المعربة تقامير المعربة عنهم تقامير المحافي المحافية المح

The permissive *tup* is also used in contexts analogous with te^{ho2} , as shown by comparison of (8.244) with (8.237) above. Based on other uses, it may be surmised that *tup* in (8.244) leans towards expediency and te^{ho2} in (8.237) towards moral obligation. The generally valid statement in (8.244) is negated by *mi*- and the past statement in (8.243) by *ma*-.

(8.244) দান্তব্যাধীমা ইন্যা প্রতিদা রীদ্যদুদ্য *nàtea=ki só: teé? mi-tup*. 1PL=AGT life cut NEG-be.fitting 'We shouldn't take life.' (YR canteen video)

8.5.5 Temporal evaluative *rɛn* 'be time to'

By using the secondary verb *ren* 'be time to' the speaker expresses that in their evaluation a time for the verbal action (denoted by the primary verb) has come. The verb *ren* differs from other secondary verbs in that it does not occur as a primary verb, and even as a secondary verb it has a very limited distribution. It either occurs as the final marker in the clause, see (8.245) or is followed by the imperfective marker *-to* (8.246). Other constructions, such as the perihrastic past *VERB *renbo be*? and completive *VERB *ren-ts^ha*: do not occur. A nominalized form, however, occurs in the idiom given in (8.247).

(8.245) ८ दें र्येट क्रेनिय रवा

t'ato $k^h \tilde{o}$:lepren.now3SG.HONarrivebe.time.to'Now it's time for him to come.' (KUN e)

(8.246) सिंबा रिंबा रज्हें सामया क्षाके

 \acute{om} $k^h \theta$: ren-to. màla? ta-te^hi. milk boil be.time.to-IPFV quick look-IMP.FRN 'It's time for milk to boil. Look at (it) quickly.' (KUN e)

(8.247) त्युः रता र्हेन रता राग्या न तनेवा याता र्हायार्थ क्षेत्रा

gju ren dø: ren-po=le t'a dem k'an p'ja-u \tilde{t} ? go be.time.to stay be.time.to-2INF=ABL now such what do-2INF EQU.PER 'Just as everything was going smoothly, why did (you) do like that?' (KUN e)

Note that in (8.246) the person has not seen the milk boil but evaluates that sufficient time has passed for the action to take or have taken place.

Sandberg (1895: 74) reports a similar use of $r \in n$. His imperfective marker, however, has a voiced initial (-*do*) in contrast to -*to* in my data.

8.5.6 Moral and practical evaluative (*mi-)le*? 'be (not) good'

The secondary verb (mi-) $l\epsilon$?, which typically occurs negated, expresses the speaker's evaluation about the goodness or practicality of a course of action.

(8.249) र्खे तर्ने र र्जुष से लेगा

ódi **p'ja** *mi-lɛ***?**. that do NEG-be.good 'It is not good to do that.' (KN, CY interview)

(8.250) ज्ञ अप्रदे कें र गीषा ज्ञ के लेगाषा

s a m = di $t c^h \emptyset$: = ki s a $mi-l \epsilon 2$. food=DEMPH 2SG.L=AGT eat NEG-be.good 'It is not good for you to eat the food.' (Rna-gsung 9)

(8.251) שאי איש, האמידי אדריארן שריאי הישלי אישריאי אישאי אישאי

 $t \epsilon^h \epsilon$ àgja, jã:ne nàtei làka=lɛ nãː-mø?. ze:-po oh.no elder.bother eat.HON-2INF do.HON-URG 1PL.GEN hand=ABL or ze: mi-lè:-po? eat.HON NEG-be.good-2INF 'Oh no, brother, please eat, by all means. Or is it not good to eat from our hands?' (Richhi 20)

In many situations, *mi-le*? is functionally very close to negated permissives *mi-te^{ho}*? 'be not allowed' and *mi-tup* 'deem not fitting'.

8.5.7 General abilitatives *ts^hu?* 'be able to' and *cè*: 'know'

The secondary verbs $ts^h u$? 'be able to' (WT \mathfrak{serv} tshugs) and $\mathfrak{c}\acute{e}$: 'know' (WT \mathfrak{serv} shes) express the speaker's evaluation about the proposition's actor's ability to do something. The form $ts^h u$? is described first, then $\mathfrak{c}\acute{e}$: The abilitative $ts^h u$? may occur as the sentence-final morpheme or be followed by other verbal suffixes. The negated forms are present/future mi- $ts^h u$? (EQU) and past ma- $ts^h u$? (EQU). The affirmative past form is $ts^h u$ -po EQU. In (8.252) and (8.253) $ts^h u$? occurs as sentence-final morpheme.

(8.252) ディエンデ (P'-9' オロネンズオ オン・9' ジロ' みよう 愛行 オン・9' やっ 愛ロン 愛切 マ グロン ジロン ボー te rã:=to khasa nà:tsẽ: nàŋsa p'ja? di=tsu=i nàŋsa lèp then 2SG.L=CEMPH deer forest inside cliff this=PL=GEN inside very.much gjuk tshu?. run be.able.to 'But you, a deer, are able very well to run in the forest and cliffs.' (UU deer story) (8.253) क्षॅंग्द्रि-र्द्युबा मा क्रेंगा वार्द्धगुबा

ódi-p'ja ŋà qok ma-ts^hu?. that-ADVZR 1SG study NEG-be.able.to 'That's why I could not study.' (PED life story)

Both the affirmative (8.254-255) and negated forms (8.256) can also be followed by an equative copula.

- (8.254) $a\hat{\gamma}_{i} \hat{\gamma}_{i} \hat$

(8.256) ק־דָּבִישָּׁן שַמַּוְיַשְּבָּזְ הַיָּשֶׁיבֹי שָׁמָן שָּבָן *t'ariŋ=gi tʰaktɛɛ̃:=di t'atarɛ tʰɛn mi-tsʰu bɛ*?. today=GEN decision=DEMPH now happen NEG-be.able.toEQU.NE 'Today's decision cannot be made right now.' (BB BB discussion)

The following examples illustrate the nonpast, past and present habitual uses of $ts^h u^2$ respectively:

(8.257) या'त' त्यु: र्ख्यायाः मृत्त' आदा?

k'ana gju ts^hu:-cɛ? ápa? where go be.able.to-INF father 'Where will (we) be able to go, father?' (AB kitchen discussion)

³³⁵ Here the contrastive emphatic is pronounced in the Nepali way as *ta* and not as typical Denjongke *to*. The written Denjongke, however, has $\frac{1}{2}$ *to*. The same happens, with the same speaker, in (8.255).

(8.259) යකුහ මූහ ගති හි a_{n}^{2} $a_{n}^$

In addition to uses as primary verb, *cé:* 'know' can be used as a secondary verb marking ability:

(8.260) \check{h} \check{h} : \check{h} : \check{a}

(8.261) र्ड्सग: नेषाग?

dok cé:-ka? read know-PQ 'Can (you) read?' (UT e)

(8.262) म्राप्तः ह्रां नेषागावा?

lấ: mó cé:-kam? bull plough know-ATTQ 'Can (he) plough with a bull, I wonder?' (SGD weddin customs)

Abilitative modality may also be expressed by the construction VERB-INF EX, see §8.4.

8.5.8 Mental abilitative *nùm* 'dare, have courage to'

The secondary verb n um 'dare, have courage' may be characterized as one type of mental ability (alongside, for instance, intellectual ability). The following examples illustrate the use of n um in a negated statement (8.263), a question (8.264a) and an affirmative answer to a question (8.264b). Note that in (8.264b), where the primary verb in undertood from the context (the question), n um can be used independently.

(8.263) ਨਾ ਕੁਝੂ' ਕੇ'ਰੁਗ੍ਰ *ŋà gju mi-nùm*. 1SG go NEG-dare 'I don't dare to go.' (KTL e)

(8.264) a) אָאָק' קאָ' קאָייק? *biu=di* sé? nùm-ka? snake=DEMPH kill dare-PQ 'Do (you) dare to kill the snake?' (KTL e)

 $^{^{336}}$ \approx here represents the Nepali emphatic *po*.

b) המי جرها
 nàm.
 1SG.AGT dare
 'I (do) dare.' (KUN e)

8.5.9 Physical abilitative $k^h \theta^2$ 'to have the strength to'

The secondary verb $k^h \emptyset$? (WD $\mathfrak{g}_{\uparrow} khyod$) expresses a special case of abilitative, 'to have the strength to'. This verb collocates especially with the verb *bak* 'carry':

(8.265) हि. क्रे. पट्रे.व. ह्येगार्गे वालुव. पट्रेवा र्घ्रया क्रे. सवाव. हि. ह्यार्गे क्रे.पट्रे. पर्या हिंट.या k^hu $ts^{h}\varepsilon = di = na$ diko ma-nun dem-p'ja $ts^{h}\varepsilon p^{h}am = na$ 3SGM life=DEMPH=LOC sin NEG-few such-ADVZR life other.side=LOC $k^h u$ diko ódi **bak k^hø**:-ka? 3SGM sin that carry have.strength-PQ 'Because of so many sins in this life, will he be able to carry that sin in the coming life?' (KN e)

8.5.10 Temporal abilitative *lõ*: 'have time to'

The secondary verb $l\tilde{o}$, which does not occur as an independent verb (and thus cannot be termed "versatile," see Matisoff 1969, 1973 and Delancey 1991), is homophonous with the verb $l\tilde{o}$. 'stand', but differs from it syntactically in that whereas $l\tilde{o}$. 'stand' functions like a typical verb, $l\tilde{o}$. 'have time to' is always followed by an existential auxiliary. Typically the construction is negated, as in (8.266) and (8.267), but it also occurs as affirmed, see (8.268) and (8.269).

(8.266) गर्रेग् मार् लुर्गेन केन

só:za cù lô: mè?. tea.HON have.HUM have.time.to NEG.EX.PER. '(We) do not have time to have tea.' (DB trip story)

(8.267) न हेंग न आहु हर त्यु मेंन केन हिंग न मेंग येंग येंन

t'ato ŋà átɛu=tsa: gju lồ: mè?. k^him=na jó? now 1SG elder.brother=at go have.time.to NEG.EX.PER house=LOC work j*à*?. EX.PER '(I) have now no time to come to the brother's (=your) place. (I) have work at home.' (YR e)

(8.268) ८. गाँधना सेना दशुः सेना

ŋà	jò?	mè?.	gju	lồ:	jờ?.
1SG	work	NEG.EX.PER	go	have.time.to	EX.PER
'I do	not ha	ave work. (I) h	ave tii	ne to go (there).' (YR e)

(8.269) ה׳ ק׳צּׁישִׁשִּ׳ מָה׳בִּ׳ קַשְׁישָׁהֵ׳ מְשִישָׁהִ׳ מְשָישׁתִּ׳ מָּהִ׳בִּי מָשָּישׁתִּ׳ מָּהִ׳בִּי מָשָּישׁתִּ׳ מָּהִ׳בִּי מָשָּישׁתִי nà t'atei=le מֿ:-ni p'ja=sã: k'amø: מֿ: lồ: SG a.bit.earlier=ABL come-3INF do-TERM what.GEN come have.time.to duk-o? EX.SEN-2INF 'Although I tried to come a bit earlier but whence the time for coming? (rnam-rtog 1)

A particularly frequent nominalized adverbial use of $l\hat{o}$: 'have time to', expressing the semantic equivalent of English 'as soon as', is exemplified in (8.270):

(8.270) วัมเวิเช้ สิวพาณัร มิรุรั ผิวา สิงคร ผิส bombai=lo lep lo: mè:-po jìgi t'i-ce î:. TPN=DAT arrive have.time.to NEG.EX-2INF letter write-INF EQU.PER 'I'll write a letter as soon as I arrive in Bombay.' (Richhi 138)

The reason why $lo^{:}$ 'have time' is not used as primary verb is probably that Denjongke has a another verb $t'\varepsilon$: 'have time to', which can be used independently:

(8.271) هُمّ ترمانهٔ ا min-de:=lo. NEG-have.time=REP '(He) does not have time, (they) heard (him say).' (KNA kitchen discussion)

8.5.11 Deontic modal go? 'be needed, must'

The affirmative form of the deontic modal secondary verb go2 'be needed, must' indicates that something, in the speaker's opinion, needs to be done (in affirmative), see (8.272).

(8.272) $\widetilde{\mathfrak{s}}_{\gamma} \cdot \widetilde{\mathfrak{g}}_{\gamma} \cdot \widetilde{\mathfrak{g}}_{\gamma} \cdot \widetilde{\mathfrak{g}}_{\gamma} \cdot \widetilde{\mathfrak{g}}_{\gamma} \cdot \widetilde{\mathfrak{g}}_{\gamma} \cdot \widetilde{\mathfrak{g}}_{\gamma} = ki \quad t' \varepsilon p \quad t'i \quad go?.$ 2SG.L=AGT book write be.needed 'You have to write a book/books.' (YR canteen video)

The construction can be negated in two ways, which bear a semantic difference. Preposing the negator prefix to the secondary verb go^2 has the meaning 'does not need to' (8.273), whereas preposing the negator to the primary verb has the meaning 'must not' (8.274).

(8.273) ตัลงละ รางลงสูารา มีรา ผิสารที่สารัๆ

ómdze? sámdup dø: miŋ-go:=co.
second.lama PN sit NEG-be.needed=AT
'Omze Samdrup does not need to participate (in the ceremony), you know.' (LT kitchen discussion)

(8.274) ८. राज्या यीया होया या याया र देवीया

yàtca=ki sók ma-sé go?. 1PL=AGT life NEG-kill be.needed 'We mustn't take life.' (YR canteen video) The form can also be used as an independent primary verb (i.e. without a preceding verb) with the meaning 'need to have' (the various forms of the S argument in these clauses are commented in §5.3.1):

- (8.275) $\sum_{n' \in A} \frac{1}{n} \sum_{n' \in A} \frac{1}{n} \sum_$
- (8.276) $\Box_{N'}/\Box_{\tilde{A}'} / \Box_{\tilde{A}'} = \delta_{\tilde{A}'} \delta_{\tilde{A}'} = \delta_{\tilde{A}'} \delta_{\tilde{A}'} = \delta_{\tilde{A}'} \delta_{\tilde{A}'} \delta_{\tilde{A}'} = \delta_{\tilde{A}'} \delta_{\tilde{A}'} \delta_{\tilde{A}'} \delta_{\tilde{A}'} \delta_{\tilde{A}'} = \delta_{\tilde{A}'} \delta_{\tilde{A}'} \delta_{\tilde{A}'} \delta_{\tilde{A}'} \delta_{\tilde{A}'} = \delta_{\tilde{A}'} \delta_{\tilde{$

The equative copula $b\epsilon \partial = p\epsilon^2$ may follow go^2 (the equative \tilde{i} : does not occur in this position in my data).

(8.277) अन्तु प्रॉन्यरां लु प्रॉययना

ágya gom(po) εù go:=pε?. elder.brother permission request be.needed-EQU.NE 'Brother, I need to request permission (to leave).' (Richhi 70)

(8.278) क्षॅंग्वद्वेयः तुः वेवर्त्यायः भ्रतः में

 $\delta d\varepsilon m$ εu min-go? $b\varepsilon = \varepsilon o$. like.that say.HUM NEG-be.needed EQU.NE=AT '(They) needn't say like that, you know.' (KN kitchen)

Typically *go*? is postposed to the verb root, but in some instances it is appended to an infinitival form of the verb:

In addition to the nonpast form exemplified in (8.279), *go*² may occur in the periphrastic past (8.280) and present habitual construction (8.281):

- (8.281) ה' שְׁשָׁשִׁי הַשְׁשִׁשִּעָשִ אַרָן *t'a jó? p'ja go:-kʰɛn bɛ?.* now work do be.needed-NMLZ EQU.NE 'Now work needs to be done.' (NAB BLA 7)

Denjongke does not have an exact equivalent of the English verb 'want'. Resembling semantics, however, may be expressed through a combination of the deontic modal and the verb *no* 'think':

(8.282) केंबर माह्य कानुका रेंग या लाग मुर्गेषा कर्ते उत्त

mú: karma pàmpu k^ha-lap go? nó-ruŋ 3SGF.AGT PN with mouth-speak be.needed think-CONC 'Although she wanted to talk with Karma...' Lit. 'Although she thought she must talk with Karma...' (Richhi 126)

(8.283) द्युः दर्गेषः अर्वे र्घे क्षेतु

gju go? nó-u \tilde{t} . go be.needed think-2INF EQU.PER '(I) wanted to go.' Lit. '(I) though (I) must go.' (TB e)

8.5.12 Possibility with sí? 'be possible'

The rather infrequent secondary verb sí? expresses possibility and, as negated, impossibility.

(8.284) ਗੁ'ਕਰੇਂਨੇ ਦਾ ਕਿ ਸ਼ੇਰ' ਨੇ ਦਾ (8.284)

gjamtshø:tshukhomsí? \tilde{o} :.sea.GENwaterbecome.drybe.possible come'It is possible for the water of the ocean(s) to dry up.' (song lyrics)

Consultant KN commented that si^2 does not occur in past constructions, while example (8.285) shows that si^2 does occur in the nonpast construction:

(8.285) वायावि त्यां येवा का क्रेंग्रवा हुट खेट वट क्रेंग

màzidoli? te^ha nómpo $te^u\eta$ si:-eebe?.actually traditionpairlevelbecome be.possible-INFEQU.NE'Actually, it will be possible for the tradition to become equal (for the rich and the poor).'(sbar-phung 91)

The negated examples below suggest a frequent collocation of negated si^2 with $t^h on$ 'happen, become' and tsale 'at all':

(8.286) हे' दर्न' माख्या साथयाम दर्वे के होना

te di yàtea? tsale=ra t^hon mi-si?. so this 1PL at.all=AEMPH happen NEG-be.possible 'It is not at all possible for us to make it happen.' (NAB BLA 7)

(8.287) अतुः दर्ने कें स्थाया दर्वतुः के श्वेन

 $m\tilde{\varepsilon}$: $d\varepsilon$:=to tsale then mi-si?.

no like.that=CEMPH at.all happen NEG-be.possible

'No, it is not at all possible that that happens.' (Richhi 116)

8.5.13 Tentativity with ta 'look'

The secondary verb *ta* 'look' denotes tentative and experimental action whose outcome is uncertain. Similar use of the verb 'see' to mark "tentative aspect" has been reported, for instance, in Tibeto-Burman Lahu (Matisoff 1973) and Galo (Post 2007: 497), the Tai-Kadai language Maonan (Lu 2008: 310) and Indo-Aryan Assamese (Post 2008: 65). I consider, however, tentative modality a better term, because the category is not essentially about time-perspective. The secondary verb *ta* frequently collocates with t'i 'ask', which denotes an action whose result is by definition uncertain, see (8.288) and (8.289). Because experimental action is frequently expressed through a suggestion, the secondary verb *ta* often occurs in the hortative mood, see (8.289), (8.290) and (8.291) or another type of suggestive construction, see (8.292).

(8.288) केंद्र गुत्र तर्च्र वया है दे क्षां राजे

 $te^{h\phi^2}$ kan $bj\tilde{o}$: làp-ti **t**'i ta-u=lo. 2SG.L what disappear say-NF ask look-2INF=REP 'What of yours disappeared (he) asked (so the story goes).' (JDF axe story)

(8.289) **લાર્ચા લાઆર્થા લાઈ** છેવા દાજવા દી ક્ષેત્રો

ápo áma=lo átsi=tei? ŋàtɛa? t'i ta-gɛ. father mother a.bit=INDF 1PL ask look-HORT 'Let's see a bit and ask father and mother.' (SGD wedding customs)

(8.290) हा पर्वे द्वित क्षणो के दति केंसुन कें।

 $t\varepsilon$ 'a zo **p**'in ta- $g\varepsilon$ =s $p\dot{e}$: $m\dot{o}by$ =lo. tea make give look-HORT=QUO 1SG.GEN wife=DAT 'Let's look and make tea to give to my wife.' (KT e)

 $g\varepsilon:po=di=lo$ $l\partial kt\varepsilon\varepsilon=r\tilde{a}$: k^ha-lap $ta-g\varepsilon-tc^hi$.king=DEMPH=DATagain=AEMPHmouth-speaklook-HORT-IMP.FRN'Let's again try and speak (in incantation) to this (spirit) king.' (mam-rtog 8)

The honorific equivalent zi: 'look, see (hon.)' can also be used in similar function to ta.

(8.293) $\hat{\Xi}^{\cdot}$ שִׁשִּׁשִּׁידִוֹי שִקְּהֵישָּׁן t^{i} zi:-po $n\hat{a}^{:}$ -tc^hi. ask look.HON-2INF do.HON-IMP.FRN 'Try and ask (him/her).' (KN e)

8.5.14 Pretensive - dzin thap

The pretensive is formed by postposing to the verb *-dzin t^hap*, consisting of WT $\underline{\xi}\underline{\gamma}$ *rdzun* 'lie' and the verbalizer *t^hap* of uncertain etymological origin (possibly WT $\underline{\alpha}\underline{\beta}\underline{\gamma}$ '*thab* 'fight, quarrel').

(8.294) \mathbb{R}° קשׁמִי לָא בָא בּא ילי באן $k^{h}u$ jó? **p'ja-dzin** t^hap-o bɛ?. 3SGM work do-PRET VBLZ-2INF EQU.NE 'He pretended he was working.' (KN e)

When the pretensive is used with a constituent complement clause, the verb occurs twice, and the complement is formed from a genitivized infinitive or nominalized form, see (8.295), where gju- $c\varepsilon$ =ki [go-INF=GEN] could be replaced by gju- $b\emptyset$: [go-2INF.GEN].

(8.295) यु. व. नाम प्रा. मेरे. मेरे.

 $k^{h}u$ [$na:=l\varepsilon$ gju- $\varepsilon\varepsilon=ki$] gju-dzin $t^{h}ap$ -o b ε ?. 3SGM here=ABL go-INF=GEN go-PRET VBLZ-2INF EQU.NE 'He pretended he was going from here.' (KN e)

The pretensive form, which attaches directly to the verb root, is a grammaticalization of the complex verb $dzin t^h ap$ 'pretend', which occurs with a nominalized complement (complement given in square brackets):

(8.296) לישרי משקריארמי הלמי שמיאי אמי ריאי ד׳ אמישרי ד׳ מיאמיעלי באן בריד׳ הבן ť izã: né:tã:=di: nà=lo cé:-run p'ja-ti mú: nò but situation=DEMPH.AGT do-NF 3SGF.AGT 1SG=DAT face know-CON $[\eta \hat{\rho} ma - \epsilon \hat{e} \hat{z} - p \hat{\sigma} \hat{z}]$ dzin thap-o da. face NEG-know-2INF.GEN lie VBLZ-2INF be.similar 'But because of this situation, although she knows me, it is as if (she) pretends (she) does not know.' (Richhi 171)

8.6 Summary remarks

This lengthy chapter described constructions which express tense, aspect and modality. The discussion on tense and aspect was divided into past oriented constructions, present/future constructions, and progressive/durative constructions. A lot of semantic overlap was shown to exist between forms expressing completion (completive, perfect, resultative, secondary verb 'finish') and durativity (imperfective, continuous, progressive, durative). Nuances of difference between the semantically similar constructions call for further research.

The present analysis was shown to differ from Sandberg (1895), especially with reference to past and completive forms: Sandberg reports a 'past' form (- $s\tilde{o}$.) that does not occur in the present description. Second, Sandberg does not report the completive form which does occur in the present description ($-ts^ha$.). It was also noted that the infinitive forms - $c\epsilon a$ and =ni syntactically overlap in some constructions but not in others. Furthermore, I described 16 modality-expressing markers, most of them secondary verbs. As an example of modal functions, secondary verbs can express ability from several perspectives: general, mental, physical and temporal.

9 Evidentiality marked by auxiliaries and clitics

This chapter continues the discussion on evidentiality which began in chapter 7 on copulas. Chapter 8 on tense, aspect and modality-marking construction was placed between the two chapters on evidentiality because it is easier to understand the discussion on the evidential uses of copulas and clitics in various constructions after those construction have been introduced. The current chapter widens the discussion on evidentiality into two directions. First, here the discussion on evidential functions of copulas is extended to their auxiliary uses at the end of the verb complex (§9.1). Second, this chapter addresses the evidential clitics, reportative =lo and quotative = $s\varepsilon$ (§9.2). For a definition of evidentiality, see §7.1.

The description in this chapter concerns only finite main clauses. Reportative and quotative markers do not occur in dependent clauses,³³⁷ and in dependent clauses evidential contrasts in auxiliaries are neutralized so that only the personal copulas \hat{t} and $j\partial 2$ (and not neutral $b\epsilon 2$ and sensorial du?) occur.³³⁸ The concessive construction in (9.1) exemplifies an evidentially neutral, dependent construction. In (9.1), \hat{t} cannot be replaced by $b\epsilon 2$.

(9.1) हे' क्षेत्र कुषार्दु' या' पत्वयाषा आयता आत्र आत्र रा

te lenge:=tsu *k'a:* $zu:-k^h\tilde{e}:$ $\tilde{t}:-rug$ so PRN.HON=PL where live.HON-NMLZ EQU-CONC 'So wherever you live...' (KT life story)

9.1 Evidentiality marked by auxiliaries

As shown by the ensuing discussion, the copulas \hat{i} , $b\epsilon^2$, $j\partial^2$ and du^2 retain their basic meaning when used as auxiliaries. The use of the personal equative \hat{i} and personal existential $j\partial^2$ is associated with 1) well-integrated old knowledge, 2) spatiotemporal proximity and 3) emotional involvement, whereas the neutral auxiliary $b\epsilon^2$ is associated with the lack of these three characterizations. The sensorial existential du^2 is associated with sensoriality and momentariness³³⁹.

The discussion below first addresses equative auxiliaries (§9.1.1) and then existential auxiliaries (§9.1.2). The final section (§9.1.3) discusses evidentiality of the various completive constructions which involve the completive $ts^ha(:)$.

9.1.1 Equatives \tilde{i} : and *b* ϵ ? as auxiliaries

The equatives \tilde{i} : and $b\epsilon^2$ occur in past (VERB-po EQU)³⁴⁰, present habitual (VERB- $k^h\tilde{\epsilon}$: EQU), imperfective (VERB-do [EQU]), future (VERB- $\epsilon\epsilon$ EQU) and nominalized progressive (VERB-zim-bo EQU) constructions. Table 9.1 summarizes these constructions and exemplifies each form with the verb(alizer) kjap 'strike, do'.

³³⁷ The quotative $=s\varepsilon$ can mark an embedded clause but this embedded clause (a quotation) is not dependent in that it can also occur independently.

³³⁸ However, causal clauses with *k'ambjasene* (see §15.4.1) and causal/purposive clauses with *làpti* (see §15.4.4 and §15.5.3) allow finite constructions and thus also occur with *be*? and *du*?.

³³⁹ Momentariness refers to the fact that by using the auxiliary du^2 the speaker claims only to have had a sensory experience but does not necessarily claim that the reported state-of-affairs continues at the time of speaking. In contrast, the personal auxiliary $j\partial^2$ typically suggests continuation of the action or its results at the time of speaking.

³⁴⁰ With stative verbs, this structure may also have a present habitual meaning.

Name	Form	Example of personal	Example of neutral
past (periphrastic)	VERB- <i>po</i> EQU	kjap-o íĽ	kjap-o bε?
present habitual	VERB- $k^{h} \tilde{\epsilon}$: EQU	kjap-kʰɛ̃ː í̇́ː	kjap-kʰɛn bɛʔ
imperfective	VERB-do (EQU)	kjap-to íĽ	kjap-to bε?
nonpast	VERB- <i>ce</i> EQU	kjap-ce î๊:	kjap-se be?
nom. prog.	VERB- <i>zim-bo</i> EQU	kjap-zim-bo íĽ	kjap-zim-bo bε?

Table 9.1. Constructions with equatives as auxiliaries

The evidential meaning of the personal auxiliary \tilde{t} : is essentially the same as the meaning of the copula \tilde{t} ? By using the auxiliary \tilde{t} ? the speaker claims familiarity with the reported situation. Similarly to the copular use of *b* ϵ ?, by using the auxiliary *b* ϵ ? speakers imply that they distance themselves from the claim of familiarity with the situation. Typically, when talking about oneself, one feels familiar with one's situation and hence the personal copula is used. The use of the neutral auxiliary *b* ϵ ? with 1st person actor ³⁴¹ implies that the speaker distances herself from the proposition. When talking about non-1SG subjects, the neutral auxiliary is frequently used, but in certain contexts, which are exemplified below, the personal auxiliary may be used for situations which have a non-1st person actor.

The following examples will illustrate personal forms with 1st person actor (§9.1.1.1), personal forms with non-1st person actor (§9.1.1.2), neutral forms with non-1st person actor (§9.1.1.3) and neutral forms with 1st person actor (§9.1.1.4). Section §9.1.1.5 describes evidentiality with complex equative auxiliaries. More space is given to the less frequent combinations (personal forms with non-1st person actor and neutral forms with 1st person actor) than to the more frequent combinations (personal forms with 1st person actor). The exceptions to the default collocations of the personal forms with 1st person actor and neutral forms with 1st person actor and neutral forms with 1st person actor and neutral forms with 1st person actor.

9.1.1.1 Personal forms with 1st person actor

Clauses with 1st person actor are typically marked with personal copula \hat{i} , which implies familiarity. People are by default well familiar with their own actions. The examples below illustrate the periphrastic past (9.2), present habitual (9.3), imperfective (9.4) and nonpast contexts (9.5). The nonpast form $-\epsilon\epsilon \hat{i}$ is often, as in (9.6), reduced to $-\epsilon\tilde{i}$ in spoken language.

(9.2) देन दर्न र्याट केव कार्मुका यो भार रेंग किवा

 $t'\varepsilon p = di$ $k'\tilde{o}:t\varepsilon^{h}\tilde{\varepsilon}:$ $ma-j\hat{a}-g\varepsilon$ $l\hat{a}p-o$ $\tilde{t}:$. book=DEMPH expensive NEG-do-HORT say-2INF EQU.PER 'Let's not make (the price of) the book expensive, I said.' (KL BLA 12)

(9.3) ८ ८ ५ में प्रगद कें द सुर कें र भाषत के सा

nà t'ato kalimpon do:-k^hẽ: í:.
1SG now TPN live-NMLZ EQU.PER
'I stay now in Kalimpong.' (TB e)

³⁴¹ The word actor is here defined in a very loose sense so that it includes experiencers of events such as seeing and hearing.

(9.4) तरे लुप्तविद प्यमा दते केंग तर्म तर्म तर्म तर्म का प्रविगार्म केंदरो

 $d\varepsilon$: $\varepsilon \hat{u}$:- $zin=l\varepsilon$ $n\hat{e}$: $ts^{h}ik$ dumdodi $n\hat{a}$:like.that say.HUM-PROG=ABL1SG.GENword short this.right.here herezak-to \tilde{t} :=s.put-IPFVEQU.PER=QUO'Having said that, I hereby end these few words of mine.' (KTL life story)

(9.5) यात्र र्मुया को ते सुर्ये दया नया नया भून किता

k'amjasine $k^h u = lo$ $\eta \dot{a}$: $s \dot{\epsilon} - \epsilon \tilde{\iota}$:because3SGM=DAT1SG.AGTkill-NPST.PER'Because I will kill him.'(KTL animal story)

In (9.3), the present habitual construction with the personal equative (VERB- $k^{h}\tilde{\epsilon}$: \tilde{t} :) functions similarly to the equative copula \tilde{t} : in that it identifies the person as being someone who has the quality which is described by the nominalized verb. Using the neutral auxiliary $b\epsilon^{2}$ in (9.3) would imply that the identifying function is backgrounded, i.e. VERB- $k^{h}\epsilon n$ $b\epsilon^{2}$ puts more emphasis on verbal action than on identifying (see also the discussion in §7.2.3).

9.1.1.2 Personal forms with non-1st person actor

The use of personal forms with non-1st person actors in Denjongke is more frequent and semantically and syntactically less limited than the use of the equivalent egophoric forms in Standard Tibetan (see §7.4). The three examples (9.6-8) illustrate the past construction. In (9.6), the speaker reminds the addressees of something that they have just said. By using the personal form in (9.6) the speaker probably emphasizes the fact that he himself just recently heard his conversation partners utter the words referred to.

t'atci?	k'utɛa?	làp- o	í́:=co	íŋ-ga.
a.bit.earlier	2pl	say-2INF	EQU.PER=AT	EQU.PER-PQ
'You (pl.) just	t said (so)), you know	, didn't you.' (KL BLA 12)

Example (9.7) shows that the omniscient narrator of a novel is entitled to use the personal form in third person narration.

(9.7) दु'र्चेते' क्रु'अर्ळव' गिंगोष' द'क्ष' ठेपागा' द'र्यो'र्चे' क्षेवा

ŋù-wø:	gjumts ^h ẽ:	$k^h u = gi$	t'ato	tciku	hako-wo	Ĩ.
weep-2INF.GEN	reason	3sgm=agt	now	only	understand-2INF	EQU.PER
'He understood t	he reason of	(her) weeping	only	now.'	(Richhi 170)	

In (9.8), a boy delivers a letter and reports to the addressee from whom the letter is. Similarly to (9.6), the personal knowledge implied by the use of the personal auxiliary in (9.8) probably derives from the speaker's personal experience and involvement in the process.

(9.8) अ.म. गमा गामा गामा रामे हा भाषा रामे भाषा र

ágja karma=gi tã:-bo ť:. elder.brother PN=AGT send-2INF EQU.PER 'Brother Karma sent (this).' (Richhi 8)

Examples (9.6-8) show that Denjongke personal forms have wider distribution than the cognate intentional egophoric *pa-yin* in Standard Tibetan. The Standard Tibetan form can only be used with a 1st person subject (Tournadre 2008: 296).³⁴²

The following examples illustrate the use of personal forms with non-1st person actor in present habitual (9.9), imperfective (9.10) and nonpast construction (9.11-12). The use of the personal auxiliary in (9.9) suggests that the speaker knows very well the person he refers to.

(9.9) क्वेंन'न्रॅंक' यमना मार्थे', यमना मा कनामा केंन'र्सेक' मकना केंक' में *cέ:da=lo*, lópøn lóp*õ*: cé:da nàŋca Buddhist.institute teacher Buddhist.institute=DAT inside teacher *i*r=60. náŋ-k^hẽ: do.HON-NMLZ EOU.PER.=AT 'The teacher (=he) has a teacher's tenure at the Shedra-institute, within Shedra-institute, you know.' (KL phone call)

The speaker of (9.10) quotes the words he heard the previous day from the guard of a sacred lake, who gave the speaker instruction on why not to feed the fish with puffed rice. The use of the personal auxiliary reflects the guard's familiarity with anything concerning the lake.

(9.10) अर्कें जुमया हें क्षेत्र यो

 ts^{ho} pàm-to $\tilde{t}r=s$. lake weaken-IPFV EQU.PER=QUO 'The lake is getting weaker, (he) said' (DB day trip)

In (9.11), the doctor who has just examined an unconscious patient has the authority to claim such familiarity with the situation as to use the personal nonpast form:

(9.11) مِحَرَّنَ سَابِهِمَ عَلَيْهُ عَلَيْهُ جَمَانَةُ جَمَانَةُ جَمَانَةُ عَلَى مَعْرَبُ هُمَا nè:po k'aly? k'aly? p'ja-ti t'embo sìn-ce t. patient slow slow do-NF consciousness catch-INF EQU.PER 'The patient will slowly, slowly regain consciousness.' (Richhi 14)

Similarly to (9.7) above, the omniscient narrator of (9.12) uses a personal construction, because he is familiar with the characters he has written in the story. I did not find personal forms in the narrative parts of folkstories, which are (typically) not of the storyteller's own design.

³⁴² A second difference is that, unlike for Standard Tibetan VERB-*pa-yin*, the action does not have to be intentional in Denjongke for the construction VERB-*po* \hat{i} : to be used.

(9.12) यामुका पर्देतेः झ्रॅर'र्थे गुम्र'गीष भेराङ्गे कर्ते जषक राम्र-'भून' क्षेत्रा

tam di: korlo karma=gi lèpti nó:sam tã:- $c\epsilon$ i:. talk this.GEN about PN=AGT very.much though send-INF EQU.PER 'Karma thinks a lot about that thing.' (Richhi 36)

9.1.1.3 Neutral forms with non-1st person actor

Clauses with non-1st person actors are typically marked with neutral forms, because speakers are less likely to feel familiar with other people's actions. The past, present habitual, progressive and nonpast constructions are presented in (9.13-16) respectively.

(9.13) न्दे' अरु केंग आहे' क्वेंग वेंच'र्य भुन

pè: $\dot{a}tcu$ $tc^{h}o^{2}$ $\dot{a}tsi$ dok $t^{h}op-o$ **b** ϵ **?**. 1SG.GEN elder.brother doctrine a.bit read receive-2INF EQU.NE 'My elder brother got (an opportunity) to study a bit.' (PED life story)

- (9.14) ন' পার্মিশ' র্নাম' নামিন শ্রন্থন্ শ্রন্থন্য শ্রন্থন্য ক্রাজ্য ব্যামান্য শ্রন্থন্য প্রদা t'a jó? p'ja go:-kʰɛn bɛ?. now work do be.needed-NMLZ EQU.NE 'Now work needs to be done.' (NAB BLA 7)
- (9.15) ₅. रेट. केंगुरा हॅ. भुना

t'ariŋ ts^ho:-do bε?. today gather-IPFV EQU.NE '(The ceremony) is taking place today.' (LT kitchen discussion)

(9.16)พन्म मार्ग्य के स्वर्थ के स्वर् dã: bhaila=ki go nàtci tok-ti di: dø: iờː-kap yesterday 1PL.GEN PN=GEN head hit-NF sit EX-SIM fall *p'usim=la*: mè:-patcene $k^h u$ óna=rã: *cí-c*e bE?. there=EMPH die-INF EQU.NE vounger.sister=HON NEG.EX-COND 3sgm 'Yesterday when our Bhaila was lying fallen after hitting his head, if it wasn't for the sister, he would have died on the spot.' (Richhi 12)

It should be kept in mind, however, that there are also other factors than the actor's person that affect the choice of auxiliary. Temporal distance in (9.13) and imaginary action (irrealis mood) in (9.16) can result in even the speaker using $b\epsilon^2$ of their own actions, as shown in the next section.

9.1.1.4 Neutral forms with 1st person actor:

By using an evidentially neutral form, speakers distance themselves from the situation depicted by the proposition. Some reasons for distancing are the following: focusing on the consequences of the action rather than the action itself (9.17), historical distance and lack of control (9.18), speaker's imagined (irrealis) action (9.19-20) and distant, theoretically understood future rather than immediate future (9.21),

According to consultant KN, (9.17) would be appropriate as an answer to the question "How did you get that wound?". Similar to the copular use of $b\epsilon^2$ (see §7.2.3), the use of the auxiliary $b\epsilon^2$ backgrounds the event itself and focuses the addressee's attention on the consequences of the event.

(9.17) $\sqsubset \alpha \widehat{\mathfrak{A}} \widehat$

Spatiotemporal backgrounding may also be used about distant events that happened to oneself, as in (9.18), where the speaker is telling the birth order of her siblings. Lack of control may also be a factor in using the evidentially neutral auxiliary.

(9.18) $\tilde{\mathfrak{M}}$ ' $\mathfrak{A}''\mathfrak{A}'' \mathfrak{H}' \mathfrak{H}'$ $\hat{\mathfrak{M}}$ ' $\mathfrak{A}''\mathfrak{A}'' \mathfrak{H}'$ $\hat{\mathfrak{M}}$ ' $\mathfrak{A}''\mathfrak{A}'' \mathfrak{H}'$ $\hat{\mathfrak{M}}$ ' $\hat{\mathfrak{A}}'\mathfrak{A}'' \mathfrak{H}'$ $\hat{\mathfrak{M}}$ ' $\hat{\mathfrak{A}}'$ ' $\hat{\mathfrak{H}}$ '

Denwood (1999: 143) comments that the "other-centred" (contra "self-centred") Lhasa Tibetan past construction VERB-*pa-red*, which is somewhat equivalent to the Denjongke construction used in (9.17) and (9.18), "is said to be possible with a first-person subject and is certainly found in writing..., but I have not heard it in spontaneous speech." This suggests that Denjongke and written language in Lhasa may embody more "archaic" grammar than spoken Lhasa Tibetan.

The following two examples illustrate the use of neutral evidentiality in imagined, irrealis situations. The speaker of (9.19) asked in a telephone conversation, jokingly, the addressee to put a hefty sum of money on his account. After having been pried as to what he would do with the money, the speaker spontaneously answered:

(9.19) ה׳ جَعَنَ عَلَى جَ عَلَى عَلَى اللَّهُ (9.19) ad **gju-do bɛ?**, t^ha:riŋ, kor bak-ti. 1SG go-IPFV EQU.NE far.away tour carry-NF 'I'm going, far away, roaming around.' (KN phone call 2)

Example (9.20), on the other hand, is a build-up sentence for a linguistic example, sketching an irrealis situation.

(9.20)	લવ જીયાવા ત્યા છે. ટી.તા તા ટી. હતી. જ્ઞા						
	lenge:=lo	$p^{h}at\varepsilon$	t'iwa	ŋà	1'i-ce	bɛ ʔ.	
	PRN.HON=DAT	thither	questio	n 1sG	ask-INF	EQU.NE	
	'(Let's imagine) I'll ask you a question.' (KN e)						

Lastly, (9.21) reports the words of Siddhartha Gautama, who after seeing a dead person realizes that the same fate awaits him, not necessarily in the near future but at some indefinite point.

(9.21) جَ جَ جَ جَ (Nep.) جَ جَ عَلَ t'a yà puni³⁴³ **εί-εε bε?**. now 1SG too(Nep.) die-INF EQU.NE 'Now, I too will die (one day).' (TB story of Buddha)

A clause analogous to (9.21) but said by a very sick person, on the other hand, would likely have the personal auxiliary \tilde{t} . Proposition (9.21) refers to a theoretical understanding about the reality of death, not to one's immediate projected fate.

9.1.1.5 Evidentiality with complex equative auxiliaries

Occasionally, the final auxiliary is a complex nominalized or infinitival copula. The nominalized copula construction $(n-k^{h}\epsilon n \ b\epsilon)$ in (9.22) underlines the nonhabitual (hence $(n-k^{h}\epsilon n \ b\epsilon)$ instead of mere $b\epsilon$?) and irrealis/past meaning (hence spatiotemporally backgounding nominalized $(n-k^{h}\epsilon n \ b\epsilon)$ rather than mere personal \tilde{i} .) of the clause.

(9.22)କ୍ଷିଁ କିଂଦ୍ୱାନ୍ୟ କ୍ଷିଣ ଅପରୁଦ୍ୟା ମୁମ୍ବର୍ମ୍ବା ନିର୍ବ ମହା ଏହି ସମ୍ଭାରଣ କରି କ୍ଷିତ୍ର କ୍ଷିର୍ବ କୁମ୍ବର୍କ୍ତ କ୍ଷିଣ ଆହର କ୍ଷମ tøn ma-tsuk-se=ki ló mi-te t'ønda=le de: mind NEG-entrust show NEG-CAUS-INF=GEN purpose=ABL like.this p'ja-tiki \hat{a} : $d\epsilon m = t\epsilon i$? kjap-to íŋ-k^hɛn *bɛ*?. do-NF lie like=INDF strike-IPFV EQU-NMLZ EQU.NE 'In order that it wouldn't be shown that he didn't believe (the story), he was telling (it) as a lie like this.' (PAD bet story)

In (9.23) the nominalized auxiliary *im-bo* \hat{i} : makes the past interpretation of verbal action more explicit than if the auxiliary were mere \hat{i} ; although the temporal interpretation of the verb form is also revealed by the adverb $p\dot{\epsilon}nl\epsilon$ 'earlier'. By choosing a personal copula (*im-bo* \hat{i} :) instead of a neutral one (*im-bo* $b\epsilon$?) the speaker underlines his own involvement in seeing the flowers himself.

Example (9.24) combines a typical past tense nominalization with a neutral nonpast copular construction \hat{t} - $\epsilon\epsilon$ be?

(9.24) ດີ່ ເຈົ້າຫຼືຈາ ດີ່ ເວົ້າ ເຈັ້າ ເຈລັງ ເຊິ່ງ ອູລາ ເວັງເວົ້າ ທີ່ສຳຊາ ສິງ di = tsu = gi = di $k^{h} \tilde{o}: \eta \tilde{a} = lo$ $s \tilde{e}: -\epsilon \varepsilon = ki$ $t \varepsilon' y$ $t \tilde{a}: -bo$ this = PL = AGT = DEMPH 3PL 1SG = DAT kill-INF = GEN means send-2INF $\tilde{t}: -\epsilon \varepsilon$ $b \varepsilon 2$. EQU-INF EQU.NE 'These guys, they are probably plotting to kill me.' (KT Animal story)

³⁴³ The equivalent Denjongke morpheme is $=j\tilde{a}$.

In the context of (9.24), the main protagonist, a marten, has seen a tiger's tail peeking out from under a basket, making him suspect that the tiger-couple, whom he is visiting, are plotting to kill him. As the simple auxiliaries $b\epsilon^2$ and \tilde{t} : would typically mark an event which happened prior to the time of speaking, the use of the complex copula \tilde{t} - $\epsilon\epsilon$ b ϵ^2 rather than $b\epsilon^2$ or \tilde{t} : shows that the action is ongoing at the time of speaking. Based on the copular use of \tilde{t} - $\epsilon\epsilon$ b ϵ^2 in example (7.85) above and its context here, the auxiliary use in (9.24) probably also indicates uncertainty.

9.1.2 Existentials $j\partial^2$ and du^2 as auxiliaries

The existentials $j\partial^2$ and du^2 are used as auxiliaries in the constructions given in Table 9.2. The table also includes the non-copula form $-k\epsilon/g\epsilon$, which was earlier shown to occur as an intensifier attached to sensorial du^2 (see §7.2.2.3).

Name	Form	Example with j@?	Example with <i>du?</i>
completive	VERB- $ts^h a(:)$ EX		$k jap-ts^ha du(k\varepsilon)$
	VERB-ts ^h ake		kjap-ts ^h ake
	$(VERB-ts^{h}ou EX)^{344}$		$(kjap-ts^{h}ou du[k\varepsilon])$
	VERB- <i>ts^houk</i> ε		kjap-ts ^h ouke
perfect	VERB(VERB)-po/bo EX	kjap(kjap)-o j <i>à</i> ?	$kjap(kjap)-o du(k\varepsilon)$
resultative	VERB EX	kjap j <i>à</i> ?	$kjap du(k\varepsilon)$
progressive	VERB- <i>tGẼ:/zɛ̃:/zin</i> EX	kjap-tɛɛ̃ː jờ?	kjap-teen du(ke)
	VERB- <i>tsuŋgɛ/zuŋgɛ³⁴⁵</i>		kjap-teuŋge
	(Martam VERB- <i>tcouke</i>)		(Martam <i>kjap-tεoukε</i>)
continuous	VERB- <i>dø:</i> EX	kjap-dø: j <i>à</i> ?	kjap-døː du(kɛ)
possessive-	VERB- <i>ce</i> EX	kjap-ce jờ?	kjap-ce du(ke)
like			

Table 9.2. Constructions with existentials as auxiliaries

The use of personal auxiliary $j\partial^2$ implies the speaker's personal acquaintance/familiarity with the proposition. It also stresses the present relevance of a past action or suggests that the action is still ongoing at a reference time, which is usually the time of speaking. By using *du*?, in constrast, the speaker bases her proposition on sensory evidence. The interpretation of what the speaker has seen depends on the nature of the event (as coded by tense and aspect markers). If the action is presented as ongoing with progressive or continuous forms, the sensory evidence implied by *du*? most likely refers to seeing the action itself. If the action has been completed earlier, as suggested by the completive and perfect forms, the sensory evidence implied by *du*? refers to some result of the action rather than the action itself.³⁴⁶

In Table 9.2, $-k\epsilon/g\epsilon$ occurs as an element that may be appended to du?, but in the completive and progressive forms $-k\epsilon/g\epsilon$ also occurs independently of du?. The uses without du? (VERB- $ts^hak\epsilon$, VERB- $ts^houk\epsilon$, VERB- $t\epsilon\epsilon\eta g\epsilon/z\epsilon\eta g\epsilon$) are hypothesized to be abbreviated forms, which may have originally retained the sensorial meaning but are currently, because of the absence of du?, becoming to be used in contexts where there is no sensory evidence. The forms VERB- $ts^hak\epsilon$, VERB- $ts^houk\epsilon$

³⁴⁴ This form was reported as marginal by consultant KN.

³⁴⁵ This form is probably an abbreviation from VERB-*tcen duke*.

³⁴⁶See Hill (2017) for an illuminating study on how sensorial/experiential forms may have inferential semantics.

and VERB-*tsenge/zenge* are preliminarily termed alterphoric, because in my data they do not occur with the first person.

The discussion below is divided into personal forms with 1st person actor (§9.1.2.1), personal forms with non-1st person actor (§9.1.2.2), sensorial and neutral forms with non-1st person actor (§9.1.2.3) and sensorial and neutral forms with 1st person actor (§9.1.2.4). Complex evidential auxiliaries are briefly addressed in §9.1.2.5.

9.1.2.1 Personal forms with 1st person actor

The speaker is typically well familiar with propositions which have a first person actor. Therefore clauses with a first person actor usually opt for a personal rather than a sensorial or neutral auxiliary (counter-examples are given later). The examples below exemplify perfect (9.25), resultative (9.26), progressive (9.27) and continuous forms (9.28).

- (9.25) rest direction di

nà lɛp jò?. 1SG arrive EX.PER 'I have arrived.' (KN e)

A complex auxiliary with $j\hat{a}$:- $c\epsilon \hat{i}$ is used in (9.28) to refer to the future.

 $\eta \dot{a}$ t^horã: tɛ^huts^hø? gu=lo t^hom=na gju dø: j \dot{o} :-ɛɛ t́:. 1SG tomorrow clock.time nine=DAT town=LOC go stay EX-INF EQU.PER 'Tomorrow nine o'clock I will have gone to town' (BT grammar exposition)

In (9.28), evidentiality is marked as personal with \tilde{i} , whereas $j\partial \hat{i}$, which occurs in an evidentially neutralized position (*du*? does not occur with infinitive), only marks the construction as an existential.

9.1.2.2 Personal forms with non-1st person actor

Personal auxiliary $j\partial^2$ also occurs with non-1st person actors, if the speaker wants to claim familiarity with the situation. Note that these uses fall outside the purview of Standard Tibetan egophoric, a category similar to Denjongke personal.

- (9.29) قرح تَوَّامَّا مَ مَعْمَا هَمْمَا مَعْمَا مَعْمَا مَعْمَا مُعْمَا مُحْمَا مُحْمَا مُعْمَا مُعْمَا مُحْمَا مُعْمَا مُحْمَا مُعْمَا مُحْمَا مُحْمَا مُعْمَا مُحْمَا مُحْمَا مُعْمَا مُع

רְצִוֹים אוֹימִי בּישׁמִי (college level)347 מְרֵים בישּמוימֹי בישׁמוי אָרֶיצֿמוּמי אֵביר מוֹרן (9.32) $p\varepsilon = na$ kolidz levəl nàŋca nàtea=lo nàtei keri? example=LOC college(Eng.) level(Eng.) inside 1PL=DAT 1pl.gen language lep dø: jò?. stay EX.PER arrive 'For instance, our language has reached college level.' (NAB BLA 7)

In the novel Richhi, the omniscient narrator is entitled to use the personal forms in third person narration:

- (9.33) $\neg \beta i \overline{k} \neg i \overline{k} i \overline{k} \neg$
- (9.34) $\check{n} \check{r} \check{r} \check{r} \check{g} \check{r} \check{r} \check{g} \check{r} \check{g} \check{r} \check{g} \check{r} \check{r} \check{g} \check{r} \check{r} \check{g} \check{r} \check{r} \check{r} \check{r} \check{r}$ $k^{h} \check{o}: ni:-tea t'ato n \check{r}:t^{h}: tenlo do: j \check{o}?.$ 3PL two-pair now bed.GEN on sit EX.PER 'The two of them are now sitting on the bed.' (Richhi 18)

Example (9.35) illustrates the use of the personal form with the reportative =lo. The speaker reports the words of a guard at a sacred lake. Similar to (9.10) above, by using the personal auxiliary, the speaker reports the guard's (and not his own) familiarity with the proposition.

³⁴⁷ WD equivalent ਕਬੱ'ਤੇ ਕਾ ਡੂੱਸਾ ਗੁ mtho-rim slob-grwa 'high-level school'.

(9.35) दुर्ग्स इस साम्चेंद्र से पासुन पालग येंन में

 $p\dot{a}=lo$ sàm mam-bi:=s sún za: $j\dot{o}$:=lo. fish=DAT food NEG-give=QUO say.HON set EX.PER=REP 'It has been said, do not give food to the fish, I heard.' (DB trip story)

9.1.2.3 Sensorial and neutral forms with non-1st person actor

Sensorial and neutral forms are quite typically used with non-1st person actors because it is rarer to claim familiarity (marked by the personal forms) with propositions concerning other people. I first describe the use of sensorial forms and then neutral forms. Using the evidential auxiliary du^2 usually implies a recent or current sensorial experience by the speaker, see (9.36), where a doctor comments on the x-ray pictures in front of him.

(9.36) جَ مَا عَلَى مَعْمَ مُعْمَ م

The sensory event, however, may also have taken place a long time ago, as in (9.37), where an elderly speaker reports what he saw decades ago.

हे. क्रॅ'र्ट्स, श्रैट.जू. लेला.पापु.ग्री. पहूर्य.तटेंची. पट्र. क्रॅ.ट्र. क्रॅ.पटेंच.ह्र. क्रॅ.पुंच्या. क्र्य.जूब. क्रूट.जूब. ह्या. ह्या. ह्या. क्रं.याय. (9.37)तर्ने र्हु आर कें तर्ने न हु नकर नवीत पर्या te ódi gã:=lo $\dot{v}:k^h \phi:=gi$ dzinda? $di=tsu=j\tilde{a}$: gjompa ódepti so that time=DAT village.GEN=GEN sponsor this=PL=too monastery like.that tsõ:ri? p'ja-tiki t c^h uri? di=tsu=jã: lori? mèri? ódepti **tea:-zen** Lhopos Lepchas Limbus do-NF Nepali this=PL=too like.that come.HUM-PROG *du?*. EX.SEN 'At that time also sponsors from the village, Lhopos, Lepchas, Limbus and also Nepalis were coming like that to the monastery.' (CY interview)

The sensory progressive form is frequently used with verbs of saying. In (9.38), where the speaker explains why he thinks the absent referent is in the toilet, the reference to a specific sensory event is clear. In examples (9.39) and (9.40), on the other hand, du^2 do not seem to refer to a specific sensory experience but generally to what people can be heard to say.

(9.38)	૪જ. બ્રુવ રૉ' કર્ને ક્ષેક બવર્વક લડ્યા						
	t'ateile súpo	nà-u	í	làp-tɛɛn	du?.		
	recently stomach	fall.ill-2INF	EQU.PE	R say-prog	EX.SEN		
	'He is saying his stomach has become ill lately.' (Richhi 98)						

(9.39) $a_{T}^{A_{T}} f_{T}^{A_{T}} f_{T}^{A_{T}} (bowl) a_{T}^{A_{T}} f_{T}^{A_{T}} f_{T}^{A_{T}}$

(9.40)में कुषा या गावा द्या हुंबा हुंबा हुं के लिने प्राह्म रावया पर्या du?lògju? k^ha ódε: súŋ sé:ta? p'ja-ti za: story mouth clear do-NF like.that say.HON set EX.SEN 'It has been so said in clear words.' (KLT Bumchu video)

The sensoriality marked by du^2 may refer to indirect evidence (inference) rather than direct evidence. Example (9.41) is a comment by a farmer, who has lost a hen the previous night. Half of the eaten hen has been found in the forest. The likely culprit is a jackal. The sensory experience implied by du^2 is not that of the action itself depicted by the verb but of a result of that action (half of a ripped hen in the forest). Example (9.41) shows that the sensorial marker can with a past telic action express inferential semantics, i.e. the sensorial implies seeing the results of an action, not the action itself.

(9.41) = العند المحرمة المالي المحرمة المحرمة المحرمة المحرمة المحرمة المحرمة المحرمة المحرمة (9.41) (9.41) =

As with the copula du?, the sensoriality implied by the auxiliary du? does not have to be visual but may be gained through other senses, as shown illustrated by (9.42), which is based on gustatory evidence.

(9.42) אק׳ קאֹןאיגוק, אק׳ קאֿןאיגוק, אלק׳ קאָן קא׳ געק, אלק׳ קאָן פּס:=pɛ?, tsho:-tsha:
 take.down be.needed=EQU.NE take.down be.needed=EQU.NE be.cooked-CMPL du?.
 EX.SEN

 (It) needs to be taken down, (it) needs to be taken down, (it)'s cooked.' (Richhi 89)

It was shown above that an omniscient narrator can use personal forms when talking about characters (s)he has designed. The omniscient narrator may also invite the reader/hearer to take the viewpoint of one of the story's characters by using the sensorial form. In the continuous(-resultative) construction in (9.43), the reader of the novel Richhi is invited to view "in real time" the scene from the perspective of the protagonist who enters a hospital room:

(9.43)	ॉॅंग भ्रुकायटावा श्लेपवारादी श्लटा झेंदी व्ययावा दवार्चे चेवां झे. सुवां सुवा पा व्यता क्रुंयावारी दर्वेवा झॅंटा दर्र्या						
	k ^h u	méŋkʰã:=na	ļɛp-ø:	gã:	bhaila	ť 'embo	sìn-di
	3sgm	hospital=LOC	arrive-2INF.GEN	time	PN	consciousness	awaken-NF
	сý:су:	kʰa-làp	ts ^h u-po	<i>t^høn</i>	dø:	du?.	
	a.bit	mouth-speak	be.able.to-2INF	beco	me stay	EX.SEN	
	'When I	he arrives at the	e hospital, Bhaila	has re	gained co	onsciousness an	d has become able
	to talk a	bit.' (Richhi 2	3)				

The neutral existential auxiliaries $j\partial po b\epsilon/j\partial b\epsilon$ and $j\partial \cdot k^{h}\epsilon n b\epsilon^{2}$ are used when the speaker wants to background sensoriality and personality (or familiarity), i.e. put full emphasis on the content of the proposition without revealing how they received the information.

- (9.46) מרמי שַמי האָש פָּק הקמי מּיהָל מוּמָרִמּי הוּמִין מֹק־אַוימָד צָּקן sánge tsomdendẽ: ódi sún za: jò:-kʰen be?. PN PN that say.HON set EX.PER-NMLZ EQU.NE 'Sange Tsomdenden (=Shakyamuni Buddha) has said so.' (YR canteen video)
- (9.47) אָרָיאָרָי אָרָי mýtsy-ri kɛ? bjã: dø: jò-kʰɛn bɛ?. other-EMPH.GEN language disappear stay EX-NMLZ EQU.NE 'The language of other's has disappeared.' (RS language situation)

In (9.48), it would not be desirable to use the personal auxiliary $j\partial^2$ because it would imply that the action were still ongoing at the time of speaking. The neutral form allows the progressive action a reasonable ending point before the time of speaking.

(9.48)ร เญ รายิ่า ของาา พิศาลา์ เ รย์คาพิ่า ๆ สำหรัดสาร์ เ รย์คาล่าง เ รย์คาล่าง เ ราย์คาล่าง เ ราย์คาล่าง เ ราย์คาล่า rubi=tci? tenk^ha=lo pjaŋ-tiki t'a $k^h u$ k'alo ómtcun now 3SGM hang-NF creeper=INDF on=DAT what.is.it swinging pjaŋ-zin jè-po bɛ?. hang-PROG EX-2INF EOU.PER 'Now, hanging from creeper, he was, what's that, swinging.' (KT animal story)

9.1.2.4 Sensorial and neutral forms with 1st person actor

Sensorial and personal forms with first person actor are in my data less frequent with auxiliary copulas than with pure copulas. As already mentioned in §7.2.2, a person asking a question takes the repliers point of view and thus may use the sensorial form with a first person actor. In my data, the only natural examples of sensorial auxiliary with first person actor are real questions (9.49) or indirect questions for which a reply is not expected (9.50).

t'a p\dot{e}: lok\epsilon=di k'adzø? k'odatøpo y\dot{a}: kjap du?? now my Lhoke=DEMPH how.much understandable 1SG.AGT strike EX.SEN 'Now how understandable Lhoke did I speak?' (PAD bet story)

(9.50)নদন'ষ্ট্র' ই্র্নি' ৭5ুবা t'a ágja, *raŋ=gi=ra* t^hu:sam ze:-po nấ: now elder.brother 2SG.M=AGT=AEMPH though.HON have.HON-2INF do.HON ná:=gi $k'adz \phi = s\tilde{a}$: gato-ti dø: du?. k'adzø? sã:te nà 1sg how.much=until rejoice-NF live EX.SEN 1sg=AgT how.much until *ritc^hi tap-ti* du?àgja=lo dø: elder.brother=DAT hope sow-NF stay EX.SEN 'Now brother, you think for yourself how happy I am, how I've put my hope in the brother (=you).' (Richhi 146)

Like the sensorial copula (see §7.2.2), the sensorial auxiliary may also be used in special circumstances, such as seeing oneself in a dream or on a video-recording:

(9.51) מוּפָּרַיאָשּי בּבְּישָ (football) בּיקפֿיב באַ מוּן *nílam=di=na nà futbol tsi-zen du?*. dream=DEMPH=LOC 1SG football(Eng.) play-PROG EX.SEN 'In the dream I was playing football.' (KT e)

By using a neutral auxiliary with first person actor, the speaker creates distance towards his own action, see (9.52), where the actions is non-voluntary (mistake).³⁴⁸

(9.52) هَن مَرْجُرُمُ المَانِي عَلَمَ مَرْجَ حَامَامًا عَلَمَ مَرْجَ عَلَى عَلَى مَرْجَ عَلَى عَلَى مَرْجَ عَلَى *di: k^hate mèn-ne ná:=gi nòrt^hy:=to p'ja* that.GEN speaking.manner NEG.EQU-COND 1SG=AGT mistake=CEMPH do *jò:-ce be?.* EX-INF EQU.NE 'If he does not have speaking manners, I will have made a mistake.' (Nga'i 'gan 13)

Example (9.53) shows that the speaker may create distance to an action which happened a long time ago by using the neutral copula, see (9.53a). In (9.53b), on the other hand, the breaking of the cup has taken place just a while ago.

(9.53) a) خَّارِي جَارِي اللَّهُ عَالَي اللَّهُ (49.53) a) خَارَ اللَّهُ عَالَي اللَّهُ عَالَى اللَّهُ عَالَى اللَّهُ عَالَي اللَّهُ عَالَى اللَّهُ عَالَى اللَّهُ عَالَى اللَّهُ عَالَي عَالَي عَالَي عَالَي اللَّهُ عَالَي عَالَى اللَّهُ عَالَي عَالَى الْحَالِي عَالَي عَالَي عَالَي عَالَي عَالَي عَالَ عَالَي عَالِي عَالَ عَالَي عَالَي عَالَ عَالَي عَالَ عَالَي عَالَي عَالِي عَالِي عَالِي عَالَي عَالِي عَالَي عَالَي عَالَ عَالَي عَلَي عَلَي عَالَي عَالَ عَالَي عَالَي عَالَي عَالَ عَالَي عَالَي عَالَي عَلَي عَالَي عَالَي عَالَي عَالَي عَالَي عَالَ عَالَ عَالَ عَالَي عَالَ عَالَي عَالَ

tcoktsi $te\eta = gi$ $kaj \emptyset := di$ $\eta \dot{a} := gi$ teak-o $\dot{t} :$ table top=GEN cup=DEMPH 1SG=AGT break-2INF EQU.PER 'It was I who broke the cup that was on the table (just now).' (KT e)

³⁴⁸ Cf. Takeuchi's (2015: 412) characterization of Lhasa, Central and Kham Tibetan *yin* as "within the speakers will" and *red* as "outside the speakers will."

9.1.2.5 Evidentiality with complex existential auxiliaries

In my data, only one construction with an existential auxiliary, the existential/possessive-like construction VERB-INF EX (see §8.4), occurs with a complex copula. In the construction VERB-INF EX, the final existential auxiliary can occur in the complex evidentially neutral forms $j\partial -po b\epsilon^2$ (9.54) and $j\partial \cdot k^{h}\epsilon n b\epsilon^2$ (9.55). The first construction conveys a past meaning and the second one a present habitual meaning.

(9.54) هم بحر بج مع المعالم المعال المعالم ا

(9.55) هَرَ اللَّذِي اللَّهُ اللَّا المَا اللَّا اللَّا اللَّا اللَّهُ اللَّهُ اللَّهُ اللَّا اللَّهُ اللَّالَ اللَّهُ اللَّهُ اللَّعْلَيْلَةُ اللَّهُ اللَّهُ اللَّهُ اللَّهُ اللَّهُ اللَّهُ اللَّلَ اللَّالَ اللَّلُولَا اللَّالَ اللَّهُ اللَّهُ اللَّهُ اللَّالَةُ اللَّهُ اللَّهُ اللَّهُ اللَّالَ اللَّلَ اللَّالَةُ اللَّ وَاللَّاللَّالَةُ اللَّالَةُ اللَّهُ اللَّالَةُ عَلَيْ الللَّالِي اللَّالَ اللَّالَةُ اللَّالَةُ اللَّالِ (The grandfathers have most excellent proverbs of tell.' / 'The grandfathers are able to tell most excellent proverbs.' (KN field notes)

The investigation mentioned in (9.54) refers to inauspicious omens which have just taken place, an uneven number of orange pieces and a broken curd bowl.

9.1.3 Evidentiality with the completive *-tsha(:)*

The completive marker $-ts^ha(:)$ deserves a separate treatment, because it forms several constructions with other morphemes (see Table 9.3) and occurs in a construction to which the heretofore used evidential terms personal, sensorial and neutral seem inapplicable. The new category is tentatively termed "alterphoric" (glossed APH) and described below.

Construction	Evidentiality
VERB- <i>ts^ha</i> :	neutral
VERB- <i>ts^ha</i> EQU	personal/neutral
VERB- <i>ts^ho-u</i> EQU	personal/neutral
VERB- <i>ts^ha-k^hɛn</i> EQU	personal/neutral
VERB- $ts^h a \ du(k\varepsilon)$	sensorial
VERB- <i>ts^ho-u du?</i>	sensorial
VERB- ts^ha - $k\varepsilon$	alterphoric
VERB- <i>ts^ho-u-k</i> ε	alterphoric

Table 9.3. Constructions with the completive $-ts^{h}a(:)$

As suggested by Table 9.3, $-ts^ha(:)$ may occur alone as the final marker or be followed by one of the following formatives: equative copula $(\tilde{t}:/b\epsilon^2)$, existential copula du^2 or the morpheme $-k\epsilon$. Moreover, $-ts^ha$ forms constructions both without nominalization and with -po/bo/u and $-k^h\tilde{\epsilon}$. The evidentiality of the constructions ending in an equative auxiliary copula depends on the copula, as outlined above in this chapter (\tilde{t} : personal and $b\epsilon^2$ neutral). The discussion here focuses on the other

forms, which do not use equative copulas as auxiliaries. For examples on the equative auxiliaries in conjunction with the completive $-ts^{ha}(:)$, refer to §8.1.2.

With completive $-ts^ha(:)$ (henceforth just $-ts^ha$ within the text), the typical contrast with the personal $j\partial$? and sensorial du? is neutralized in that the personal auxiliary does not occur postposed to $-ts^ha$. In the completive construction, however, the sensorial evidential contrasts with neutral evidentiality (i.e. lack of evidential marking) implied by the lone $-ts^ha$. As evidentially neutral, the lone $-ts^ha$ occurs both with 1SG (9.56) and non-1st person actors (9.57). The sensorial form $-ts^ha$ $du(k\varepsilon)$ and the alterphoric form $-ts^hak\varepsilon$, on the other hand, typically occur with non-1st person actors, as shown by (9.58a) and (9.59-61). Exceptions are "endopathic"³⁴⁹ and sensorial clauses where the speaker may be conceived to observe themselves. Example (9.58b) with $-ts^hak\varepsilon$ illustrates an endopathic use, the first words of a boy who wakes up after having fallen asleep while tending a cow. Having observed his own state of consciousness, he speaks about himself to himself with the same verbal form that would typically be used when speaking about a third person to an interlocutor. I suspect that using the sensorial $-ts^ha du(k\varepsilon)$ is also possible with first person actor when the speaker sees themselves in a dream or on a video, although my data has no such examples.

(9.56) 5' مَحْ مَحْ مَعْ

nà òn-ts^ha:. 1SG come-CMPL 'I've arrived.' (KN e)

(9.57) אי אדי אין

k^hu òn-ts^ha:. 3SGM come-CMPL 'He's arrived.' (KN e)

(9.58) a) ?٢. ٤٢ ه. ٣)

?*ŋà òn-ts^hakε*. 1SG come-CMPL.APH ?'I've arrived.' (KN e)

b) מי, ה׳ בּוֹיהֵשִׁישֹׁיחִישׁיחֹי בּוֹן
 ja?, ŋà mik^hu:-ts^hakε=co
 oh 1SG fall.asleep-CMPL.APH=AT
 'Oh, I fell asleep (I realize).' (Short film Dream)

(9.59) ?בי מביעדי מקשייון

 $?\eta \dot{a} \quad \dot{o}n-ts^h a$ $du(-k\varepsilon)$.1SGcome-CMPLEX.SEN(-IN)?'I've arrived.' (KN e)

³⁴⁹ The term "endopathic" is adopted from LaPolla and Tournadre (2014: 242-244) and refers to a type of access to information that is based on the speaker's self-awareness and is only available to the speaker.
(9.60)المِّ: كَمْتِ هَنْ سَال k^hu on-tshake. 3sgm come-CMPL.APH 'He's arrived.' (KN e)

(9.61) اللا مُحر هج المالي

> khu du-k_E. òn-ts^ha 3sgm come-CMPL EX.SEN-IN 'He's arrived.' (KN e)

The difference between $-ts^h a$ and $-ts^h a du(k\varepsilon)$ is that $-ts^h a du(k\varepsilon)$, as in (9.62), makes reference to a specific sensory event, whereas $-ts^h a$ does not, see (9.63). Example (9.62) are words of a doctor, after checking a patient's pulse.

(9.62)

du?now good become-CMPL EX.SEN 'Now (the patient) has turned out well.' (Richhi 29)

(9.63)८.२०२५ लग मॅन्सान्ने. प्र वेश्रिक्स्ट्र म्रेनसक्त्र

> lò pí:-ts^hø? nàtci jà:p ť õː-ti lep-tshar. father.HON die.HON-NF year two-about reach-CMPL 1pl.gen 'Now about two years have passed since our father passed away.' (Richhi 35)

The completive marker occurs both as non-nominalized $-ts^h a$ and nominalized $-ts^h ou$. I am not aware of a semantic difference between these two forms. The nominalized form is illustrated in (9.64).

देष्भ्रेषास्य (facebook)350 दा र्ते रार्केतु गो केंग (9.64) \hat{o} :-ts^houkɛ=lo. feisbuk=na facebook(Eng.)=LOC come-CMPL.2INF.APH=REP '(It)'s been posted in Facebook, I heard' (LT kitchen discussion)

The completive may be supplemented by either $-du(k\varepsilon)$ as in (9.65) or $-k\varepsilon$ as in (9.66-9.67). In (9.65), the speaker, having found a child with a wound from hitting his head in a stone, hands over the child to his mother.

अम्रायम्ब, रेंग्र्हेते' मर्वे' येन' यहिंगवाळर ५८५ या (9.65) go lèp tok-ts^ha du?ám-la:, p'otsø: mother-HON child.GEN head very.much hit-CMPL EX.SEN 'Mum, the child's head has been severy hit.' (Richhi 3)

³⁵⁰ According to consultant YR, the WD equivalent is المجترجة gdong-deb [face-book].

- (9.66) $\mathfrak{F}_{\mathfrak{T}}$ (מימן ד מימן ד מימן ד מימן מימי (מימן ד מימן ד מימן מימי) און $\mathfrak{F}_{\mathfrak{T}}$ $\mathfrak{F}_{\mathfrak{T}$ $\mathfrak{F}_\mathfrak{T}$ $\mathfrak{F}_\mathfrak{T}$ $\mathfrak{F}_\mathfrak{T}$ $\mathfrak{F}_\mathfrak{T}$ $\mathfrak{F}_\mathfrak{T}$ $\mathfrak{F}_\mathfrak{T}$
- (9.67) $\mathfrak{F}_{\mathsf{RF}}$ מימך מך מקר מקר מקראי מעראי מע

Commenting on the difference between $du(-k\varepsilon)$ and $-k\varepsilon$, consultant KT said that whereas in (9.68a) the proposition is visually attested, in (9.68b) it does not have to be so.

(9.68) a) ख़ु दॅम कॅम वमुगगो

 $k^h u$ *òn-ts***^ha du(-k\varepsilon)**. 3SGM come-CMPL EX.SEN(-IN) 'He's come (I saw him).'

b) الله: مُحَمَّ مَعْمَى الله مُعْمَى مُعْمَى الله مُعْمَى الله مُعْمَى الله مُعْمَى مُحْمَا مُعْمَى مُعْمَى مُعْمَى مُعْمَا مُعْمَا مُعْمَى مُعْمَا مُعْما مُعْمَا مُعْما مُعْما مُعْما مُعْما مُعْمَا مُعْما مُعْمَا مُعْما مُعْما مُعْما مُعْما مُعْمَا مُعْما مُعْمَا مُعْمَا مُعْمَا مُعْمَا مُعْما مُعْمَا مُعْمَا مُعْمَا مُعْمَا مُعْمَا مُعْمَا مُعْمَا مُعْما مُعْمَا مُعْما مُعْما مُعْمَ مُعْما مُعْما مُعْما مُعْع

KT's observation is supported by the examples above in that the proposition with du^2 , (9.65), is based on a visual experience, while in examples with $-k\varepsilon$ (9.66) is based on the speaker's own visual evidence but (9.67) is likely based on a story the speaker has heard (furthermore, he could not see Tshothang Uncles thoughts).

Further evidence that action marked with $ts^ha-k\varepsilon$ does not have to be sensorially attested is provided in (9.69), where (9.69a) presents a question posed on the telephone and (9.69b) a comment after the phone call. The comment (9.69b) is evidentially based on a report heard on the telephone, not on sensorially witnessing the event or its results.

(9.69) a) ਖੱਕੁ ਸ਼੍ਰੇਹਬਾਰ ਸੰ ??

p^hou lɛp-ts^ho-u pá? over.there reach-CMPL-2INF EQU.PER.Q 'Has he arrived over there?' (KN phone call 3)

b) र्देतुः श्चेन्नर्थः कॅंग्गे

p^hou lep-ts^hake. over.there reach-CMPL.APH 'He's arrived over there.' (KN conversation) The form VERB- ts^ha - $k\varepsilon$ may have originated as an abbreviated version of VERB- ts^ha du- $k\varepsilon$. This is suggested by the fact that, in contrast to $-ts^ha$ $du(k\varepsilon)$, I have not come accross $-ts^ha$ - $k\varepsilon$ in written Denjongke. I hypothesize that while the abbreviated form $-ts^ha$ - $k\varepsilon$ originally retained the sensory meaning from the full form $-ts^ha$ $duk\varepsilon$ (and hence even synchronously $-ts^ha$ - $k\varepsilon$ does not occur with 1st person actors), the lack of du? in the construction is currently being reinterpreted as unnecessity of sensoriality (hence 9.69b). Because $-ts^ha$ - $k\varepsilon$ does not typically occur with first person actors (association with the first person could be called "egophoric") and because it does not seem to be necessitate sensoriality, I have tentatively named it "alterphoric," a term just suggesting that the category is typically disassociated with first person actors, although endopathic uses such as the one exemplified in (9.58b) are an exception. Further research is needed to clarify the semantics of $-duk\varepsilon$ vs. $-k\varepsilon$.

A distinction similar to that between completive $-du(k\varepsilon)$ and $-k\varepsilon$ can possibly be posited between the progressive forms $-t\varepsilon\varepsilon n du(k\varepsilon)$ and $t\varepsilon u\eta - g\varepsilon$, see §8.3.3.2. The latter form, which does not occur in written Denjongke may be a spoken reduction of the former. At present, however, I do not have adequate evidence to show an evidential distinction (sensorial vs. alterphoric) between the two progressive forms.

9.2 Evidentiality marked by clitics

Evidentiality concerns source and access to information (see the definition in §7.1). Therefore, the reportative (§9.2.1) and quotative markers (§9.2.2), which are both used for referring to information gained from another person than the speaker, fall under the purview of evidentiality.

9.2.1 Reportative =lo

The use of =lo in a clause implies that the speaker presents the proposition as originating with someone else. This section briefly illustrates the declarative (§9.2.1.1), interrogative (§9.2.1.2) and imperative uses of =lo (§9.2.1.3).

9.2.1.1 Declarative use

Examples (9.70-72) illustrate the typical declarative uses of the reportative.

(9.70) เตารับ ฏักรณาสราคับ

 $\dot{a}po$ $t'\tilde{o}$:- $ts^{h}a$:=lo. father die.HON-PRF=REP '(His) father has died, (he) says.' (BP BB discussion)

```
(9.71) \arg_{\neg \neg \neg \neg} \ \arg_{\neg \neg} \ \arg_{\neg} \ \operatorname{d}_{\neg} \
```

(9.72) הריבן אריבן יוצרים יי אלי ארי ארי ארי אין אריבן יאשי אריבן און ג'*ĩ:pu lấ:bu=tci? dɛp làm nàŋɛa gju-do=lo*. Long.ago elephant=INDF like.this road within go-IPFV=REP 'Long ago an elephant was walking like this on the road (so the story goes).' (UU deer story)

In (9.70), a group of people are talking about a person who has just left their company for a while. The addressees know who has uttered the reported proposition (9.70). In the context of (9.71), on the other hand, several people are trying to determine the date of a Buddhist ceremony. The addressees are not likely to know who told the speaker about the Facebook update. Example (9.72) is the beginning line of a folk-story. Here even the speaker does not know who originated the proposition. The use of the reportative in story-telling is frequent, see the folk-story in the appendix.

A more rare declarative use of =lo is illustrated in (9.73), taken from a story about a marten.

(9.73) क्षे., म्दि लगा में हेगा गा कें, म्दी मन रें केंन् में लग

		1	1 1 1				
έː,	<i>n</i> è:	làko	tciku= lo ,	рè:	kãːpo	jờ:=€0	làp.
eh,	1sg.gen	hand	only=REP	1SG.GEN	foot	EX.PER=AT	say
'(I th	ought) I had	l only ha	nds, (but) I l	have also fee	et, I realiz	ze.' (KT anii	mal story)

In the story, the marten's hands have one by one been stuck onto a wall smeared with glue-like sap from a tree. Then the marten realizes that his earlier thought that he could only use his hands to get off the glue was false; he can also use his feet. The interesting feature about the use of =lo here is that the speaker is not reporting someone else's proposition but rather reporting his own earlier thoughts that he is now questioning.

9.2.1.2 Interrogative use

In some works on related languages, interrogative uses of the cognates of the reportative =lo are not commented on (van Driem 1998), reported not to exist (Vesalainen 2016: 189) or limited to a narrow context with a question word (Huber 2002: 108, Gawne 2015). In Denjongke, on the other hand, the reportative is used in interrogatives (without question word), see (9.74).

(9.74) 适可可能? *tc 'øn-ca=lo*? come.HON-FUT.Q=REP 'Shall (we) go, (he) asks.'/ 'Will (you) go, (he) asks.' (oh, Tashiding)

The speaker of (9.74), taking cover from rain, has just heard a friend offer a car ride. He conveys the invitation to his friends in another room by shouting.

9.2.1.3 Imperative use

In addition to declaratives and interrogatives, the reportative is used in imperatives, a feature which is rarely exemplified in Tibetic languages (exception: Gawne 2015) and which places Denjongke within a typological minority. Aikhenvald (2004: 250) states that "[a]n overwhelming majority of languages with evidentials do not use them at all in imperative clauses." Example (9.75),

illustrating the imperative use, was spoken to me by my host TB, who had just visited the kitchen and was conveying an invitation from his wife.

The reportative =lo in Denjongke is pronounced as unstressed with a low pitch. However, Denjongke also has a segmentally identical tag question lo, a loan of the frequent Nepali tag la. The tag question, which is pronounced with rising intonation, may be used in syntactic contexts identical with the reportative, as shown by (9.76), which is syntactically analogous to (9.75)

(9.76) מוְפַמְשִיהִפֹּק: לַשִּיהִי מוֹק בִי מֹן $zu:-te\tilde{e}:$ p'ja-u $n\tilde{a}:$ **lo**. sit.HON-PROG do-2INF grant TAG.Q 'Please stay here, ok (while I go away for a while).' (TB, oh)

The pitch traces of (9.75) and (9.76) (recorded later with TB) in Figure 9.1 below show that whereas =lo is pronounced with a low pitch, the pitch rises with the tag question lo. The starting point of the reportative =lo (left) and tag question lo (right) are henceforth marked in the pitch trace with a vertical line line.

Figure 9.1. The pitch in reportative =lo (9.69) and tag question lo (9.70) contrasted (TB)



In some languages (see Aikhenvald 2015: 263), the reportative may be extended to such uses where the speaker is not reporting someone else's proposition. In these uses, the reportative helps the speaker to save face by presenting a request indirectly, and hence more politely, as if it came from someone else. This appears to happen also in Denjongke, as shown in (9.77). The following discussion, however, shows that interpreting the data is not straightforward.

(9.77) ฉริ จรุลสม อัสาน์

di $n \dot{a}m$ $t \epsilon' \phi n = lo$. this carry.HON go.HON=REP 'Please take this away.' (BBP BB discussion)

Example (9.77) occurs on an audio-recording which was recorded over a lunch. The speaker, DL, an elderly gentleman, hands over a dish to someone else, saying (9.77). When I played this recording to consultant KT (male, 60 years), he commented that the speaker is an old person who

speaks in a polite, humble way. KT contrasted this non-stressed, low pitch use of =lo with the tag question lo, which is pronounced with rising intonation. KT commented that whereas the man on the recording used =lo to make a polite request, using the tag question lo with rising intonation would suggest insisting.

I played (9.77) from the recording to two more consultants, TB (male, 40 years) and KL (male, 45-50 years). TB viewed (9.77) as an instance of reportative =lo instead of the tag question lo. He, however, was not aware of the extended indirect use of =lo but rather just commented that the speaker of (9.77) speaks as if transmitting some else's request onwards. Consultant KL, on the other hand, did not recognize (9.77) as an instance of the reportative =lo at all but considered it an instance of the tag question lo, which he specifically said comes from Nepali. I asked both TB and KL to repeat on recording what they thought DL had said on the original recording. Figure 9.2 presents the pitch traces from the original recording with TB's and KL's repetitions of it. A difference on the last syllable is obvious between TB and KL.

Figure 9.2. Pitch trace of original (9.71) and TB's and KL's repetitions of it respectively



Whereas DL's original recording and TB's repetition of it have a low intonation pattern typical of the reportative =lo, KL produces a rise in intonation typical of the tag question, albeit in an attenuated way, probably because on the original recording there is no rise at all. My hypothesis, based on the facts mentioned above, is that Denjongke reportative has developed an extended use marking indirectness/politeness. This extended use appears to be limited to certain conversational contexts (KT specifically mentioned handing over items to other people) and possibly to certain dialectal areas. However, because of bilinguality of most Denjongke speakers in Nepali and the ubiquity of the Nepali tag question la, even those uses of =lo which are extensions of the reportative marker, and hence a language-internal development, are currently being reinterpreted as cases of the borrowed tag question lo.

In conclusion, the Denjongke reportative =lo can be used in declarative, interrogative and imperative moods. I drew attention to two rare uses that, to my knowledge, have not been reported in related languages. One is the declarative use in which the speaker reports their own earlier proposition/thought rather than someone else's proposition. In the second extended function, =lo in (honorific) imperatives/requests is used as an indirectness/politeness marker. The request is presented *as if* it originated with someone else to attenuate its force. This indirect use appears to be currently reinterpreted as the tag question lo borrowed from Nepali.

9.2.2 Quotative $=s(\varepsilon)$

The quotative marker is an enclitic that is appended to the end of a clause. Its basic meaning is to indicate that someone is being quoted. The quotation may come from a person other than the speaker, or the speaker may quote something that they themselves said earlier. Typically, $=s\varepsilon$ functions as a complementizer for verbs of saying and writing, as shown in (9.78) and (9.79).

- (9.78) $3 = \frac{1}{2} = \frac{1}{2} \frac{1}{2} = \frac{1}$

Note that in (9.78) = s marks a quotation within the story/discourse, whereas the reportative =lo, which follows, is a type of metacomment on the story, meaning something like "so the story goes." The complementizer may occur in a different clause than the verb of saying/writing:

(9.80)ଡିଁଷା କ୍ଷିମ, ସ୍ଥୁ କୁମ୍ମା ଭିଂସାରି ଭାଷ ସିଂସନ୍ମାର୍କ୍ ସ୍ୱିର୍ମ୍ ସନ୍ମା ସ୍ଥିରେ ଅନ୍ମା ସ୍ଥା ବିଶ୍ୱା ଭାଷ କୁ କିଂହ୍ୟିମବ୍ୟ କା te^høki?, paːlaː=gi t'utci? nàtea? jìgiː lèn *t'i tã: lo.* father=GEN letter.GEN answer write send TAG.Q PN this.year 1pl námtei? ágia $mi-l\epsilon p=s(\epsilon)$. elder.brother NEG-arrive=OUO with 'Choki, write a letter to the father, okay, saying that the elder brother is not coming with us this year.' (Richhi 138)

Often, however, the verb *làp* 'say' or its converbal form *làpti* functions as an additional complementizer.³⁵² In these cases $=s(\varepsilon)$ is little more than a "closing quotation mark" (Tournadre & Dorje [2003: 214] for the cognate in Standard Tibetan).

(9.81)	ন্যু'তব্য'র্ন্ম' বর্ষম'	র্ষা বল্পনার্যাঝা র্মনান্যান্তা	(ते) त' अख' र्देया हे' त	দে:প্ৰ'অম্ব' ৰ্হ্টুৰ' ৰ্দ্তুগ	NRI-515. 3. 2	ম' অব'ল্পি' বাধ্যদ'র্ব'র্থা
	k'utca=lo	sámpo zaŋţa?	j <i>à-pats</i> ɛ(nɛ)	nà:=lɛ	òtε	nàŋ£a=lɛ
	2PL=DAT	thought good	EX-COND	here=ABL	down	inside=ABL
	te 'øn	$ts^h u = p\varepsilon$	ná=s	làp-ti	súm-bo	=lo.
	come.HON	be.able.to=EQU.M	NE TAG.ASR=0	QUO say-NF	say.HON	N-2INF=REP
	'Then (som	neone) said, "if yo	ou have good in	ntentions, yo	u can inc	leed go down from inside
	here" (so th	he story goes).' (S	SGD Wedding	customs)		

If the context reveals with whom a saying or opinion originated, the quotative can appear as the final marker without any words of saying. In (9.82), the speaker is explaining the rationale for an old marriage custom. The quotative refers the proposition back to ancestors.

³⁵¹ In Denjongke authors' works, the quotative can be seen either written separately or attached to the previous word (the same author may use both ways of writing). In the spoken examples here, I have written the quotative separately. Examples from written sources follow original conventions.

³⁵² *làp-ti* also functions as a complementizer by itself without = $s\varepsilon$, see §14.2.2.1.

(9.82) सुमा पर्ने कॅरामु पर्य कें के रो p'um=di nòrbu dou $\tilde{t} = s$. girl=DEMPH gem like EQU.PER=QUO 'The girl is like a gem (it is said).' (SGD wedding customs)

In some uses of $=s(\varepsilon)$, speakers appear to quote themselves, see (9.83-85). In (9.83), the speaker responds to the question "Where are the children?". Looking around, he is surprised to find out that the children who were there just a moment ago are nowhere to be seen. By using the quotative in (9.83), the speaker appears to repeat, and thus quote, his own earlier thought.

(9.83) ८ द में व र कें र के t'ato nà: j \dot{g} :=s. now here EX.PER=OUO '(But they) just were here.' (KT e)

Similarly to (9.83), the quotative is used in (9.84) in a situation where the speaker's earlier assumption is challenged: a pencil is no longer in a place where the speaker expected it to be.

nám?

(9.84) ดังสา นี่สา ลิ่ม ๆ ลี่สารี ผูล? $\delta na j \dot{a} := s(\varepsilon).$ $k'a: s \dot{\delta} :- bo$ there EX.PER=QUO where go.PFV-2INF EQU.ATTQ

'(But) it was there. Where did it go, I wonder?' (YR e) When the addressee does not hear or believe what was said, speaker may use $=s\varepsilon$ as a forceful restatement of their case with a peremptory tone and possibly showing irritation:

(9.85) थॅन् से (थॅन्) jờ:=s€ $(j\dot{\varrho})$. EX.PER=QUO (EX.PER) '(Yes) there is, I tell you.' (PTB e)

The quotative can occur postposed to the reportative, see (9.86) and (9.87).

ମ୍ର ଅଗ୍ୟାସଜ୍ୟାର୍ଯ୍ୟା (9.86)

t'a min-ze := lo = s. now NEG-have.HON=REP=QUO 'He's not having (drinks) now, I heard him say.' (PT kitchen discussion)

(9.87) हे विन्यीय दर्न अव दशु रत् दशु के लेगाय

> te $k^h o \eta = gi = di$ maŋ-gju=rõ: $g_{ju}=lo=s_{i}=la.$ so 3sg.hon=agt=demph NEG-go=DEMPH go=REP=QUO=HON 'He says (cf. $=s\varepsilon$) he heard (cf. =lo) he has no way of not going.' (AB kitchen discussion; all instances of 'he' in the translation have the same referent)

Whereas =lo refers more generally to the fact that the speaker has heard something from others, the quotative seems to make more clear reference to a specific speech act. In the preliminary translations in (9.86) and (9.87), =lo is translated as 'I heard' and $=s(\varepsilon)$ as 'him say'. The semantic nuances of $=s\varepsilon$ and its relationship to the reportative =lo derserve further study.

9.3 Summary remarks

This chapter described evidentiality marked by copula auxiliaries and two clitics (reportative and quotative). The focus was on describing the less typical combinations of evidential value and person marking, i.e. personal forms with non-1st person actor (context where sensorial and neutral forms are more frequent), and sensorial/neutral forms with the first person actor (context where personal forms are more frequent). It was shown that, similar to copulas, Denjongke personal auxiliaries occur more freely with non-1st person actors than "egohopric" forms in Standard Tibetan (Tournadre & Dorje 2003).

Inquiry into evidentiality in various completive constructions marked by $-ts^ha(:)$ gave preliminary evidence for the category "alterphoric," which refers to disassociation with the first person. The reportative marker was shown to occur in declarative, interrogative and imperative uses, the last one of which is typologically rare (Aikhenvald 2004: 250). Finally, it was also seen that, in addition to prototypical uses, the quotative can refer to speaker's own earlier thoughts or speech, possibly conveying counterexpectation or annoyance.

10 Negation

While negation strategies for individual constructions have been discussed under relevent headings, this chapter summarizes negation patterns in Denjongke. A few words on the terminology of the ensuing discussion are in order. I will use the terms "symmetric and asymmetric negation" deriving from Miestamo (2000, 2003, 2005). In a symmetrically negated construction, the negated clause differs from the corresponding affirmative clause by nothing else than adding a negator morpheme. An example of this can be taken from English: the affirmative clause *I'm going there* is negated by adding the negator morpheme *not*, as in *I'm not going there*. No other modifications take place.

Asymmetry in negation, on the other hand, may be viewed from two perspectives, as constructional asymmetry or paradigmatic asymmetry. In constructional asymmetry, a negated statement differs from the affirmative clause also by some other modification(s) than the adding of a negator. As an example Miestamo (2005: 3) gives the Finnish sentence *nuku-n* [sleep-1sG] 'I sleep.'/'I'm sleeping', which is negated as *e-n nuku* [NEG-1SG sleep.CNG] 'I do not sleep'/'I'm not sleeping.' Here negating is more complex than merely adding a negator morpheme: the negated auxiliary takes personal inflection and the inflected verb of the affirmative clause occurs in non-finite form.

Paradigmatic asymmetry in negation means that negated forms in the verbal paradigm do not have one-to-one correspondence to the affirmative forms. For instance, one negated form may correspond to more than one affirmed form, or the other way round.

The treatment is divided, taking inspiration from Miestamo's (2016) questionnaire for describing the negation system of a language, into clausal negation (\$10.1), non-clausal negation (\$10.2.) and, finally, notes on other aspects on negation (\$10.3).

10.1 Clausal negation

Denjongke accomplishes clausal negation through the negating prefixes *ma*- and *mi*-, which attach to the verb, and through negated forms of the copulas working either as pure copulas or as auxiliaries. The negators are summarized in Table 10.1.

Prefixes	म्रे [.] mi-		imperfective (present, future)
	रू' <i>ma-</i>		perfective (past, imperative)
Copulas	$m \widetilde{ec{e}}$ ਂ ਕਰਾ	(affirm. í क्षेत्र [.])	personal equative
	mèmbe? अतुःञ्चनुः	(affirm. $b\epsilon^2 \approx 5$)	neutral equative
	mè? ar	(affirm. <i>jà?</i> ຟັຽ.)	personal existential
	mindu? झेत्र तनुग	(affirm. du? مرجماً)	sensorial
	<i>mèmbo</i> ਕਰ _ੱ ਜੱ	(affirm. <i>bo</i> 국)	interrogated equative neutral
	mèna aara	(affirm. <i>ná</i> ₃ .)	interrogated equative personal

Table 10.1. Clausal negation formatives

With the perfective negator *ma*-, the verb has either past or imperative meaning, e.g. *man-dø?* $\operatorname{sug}^{\cdot}$ 'did not sit, do not sit!'. With the imperfective negator *mi*-, the meaning is present habitual or future oriented, e.g. *min-dø?* $\operatorname{sug}^{\cdot}$ 'does not sit, will not sit'.

Negation in declarative verbal main clauses, excluding copulas and existentials (which often have a differing negation strategy), is called standard negation (Miestamo 2013). The following subsections first describe standard negation (10.1.1) and then negation in copular clauses (10.1.2), non-declarative clauses (10.1.3) and subordinate clauses (10.1.4).

10.1.1 Standard negation

In Denjongke, standard negation is accomplished either symmetrically by attaching a negating prefix (mi- or ma-) to the verb root, as in (10.1), or asymmetrically by replacing the affirmed final auxiliary copula with a negated copula, as in (10.2). The asymmetrical use is emphatic and typically occurs in contrastive contexts, such as the one in (10.2)

- (10.1) a) $\tilde{a}(\tilde{a}; \mathfrak{H}\mathfrak{Y}\mathfrak{Y}, \mathfrak{H}\mathfrak{Y}\mathfrak{Y})$ $c \tilde{c}(\tilde{a}; \mathfrak{A}\mathfrak{Y}, \mathfrak{Y}\mathfrak{Y})$ $c \tilde{a}(\tilde{a}; \mathfrak{A}\mathfrak{Y}, \mathfrak{Y}\mathfrak{Y})$ $n \tilde{o}:-kja?, kjako=di=lo dzuga làp.$ cattle-excrement excrement=DEMPH=DAT cow-dung say 'Cow-dung, dung is called /dzuga/.' (PL interview)
- ריאי היסמי אילי פירי משאילי אמן ליאי הישמי ריאה מי אדי (Sir) איז מבמירי משאילי אאמן ליאי הישמי ריאה אילי אין ר (10.2)t'ato nàtea? k'are sù-wa tsa:-bo mềː. t'e:lo ask-pur come.hum-2inf NEG.EQU.NE just.like.that now 1PL anything nàtea? t'arin nà: sər=tsa: dze:-wa tca:-bo Ĩ. here sir=by meet.HON-PUR come.HUM-2INF EQU.PER today 1PL 'Now we didn't come to ask for anything. We just came here today to meet (you) Sir.' (NAB BLA 7)

In serial verbs, the negator occurs before the last $verb^{353}$, as shown by the affirmative and negated pair in (10.3).

(10.3) a) ณสูา ทัศงาร์า ติลุ

thuko:-boí:.pickthrow.away-2INFEQU.PER'(He)picked and threw (it) away'. (KN e)

b) ત્વુ અર્ગોષા

t^hu ma-ko.
pick NEG-throw.away
'Do not pick and throw (it) away.' (KN e)

The correspondance of affirmed and negated constructions in standard negation is summarized in Table 10.2, where EQU = $i/b\epsilon^2$, NEG.EQU = $m\tilde{\epsilon}:/m\epsilon mb\epsilon$, EX = $j\tilde{\phi}^2/du^2$ and NEG.EX = $m\epsilon^2/mindu^2$

³⁵³ Negated serial verbs do not have more than two verbs.

(for evidential distinctions marked by these distinctions, see §7). The parts in square brackets may be added to the shorter forms. The simplified glosses in Table 10.2 do not convey all the semantic nuances.

Constr.	Affirmative	Gloss	Negated	Gloss
state	làp	'is called, says'	mi-làp	'is not called'
simp. prs	làp bɛ?	'says'	mi-làp bɛ?	'does not say'
IPFV	làp-to (EQU)	'used to say,	ma-làp-to (EQU)	'used not to say, is not saying'
		is saying'	mi-làp-to (EQU) ³⁵⁴	
			lap-o NEG.EX ³⁵⁵	'is not saying'
CONT	<i>làp dø:</i> EX	'is saying'		
PROG	<i>làp-tɛɛn</i> EX	'is saying'		
periphr.	<i>làp-o</i> EQU	'said'	<i>làp-o</i> neg.equ	'did not say (emphatic)'
PST			mà-lap(-o EQU)	ʻdid not say'
PST	làp-tse	'said'		
CMPL	làp-ts ^h a:	'has said'		
PRF	<i>làp-làp-o</i> EX	'has said'	<i>làp-làp-o</i> NEG.EX	'has not said'
			<i>làp-ɕyː</i> NEG.EX	'has not said'
RES	làp jờ?	'has said'	làp mè?	'has not said'
SEN.PST/	làp du?	'said'	làp mìndu	'did not say'
SEN.RES			làp-o mìndu	'did not say'
NPST	<i>làp-εε</i> EQU	'will say'	<i>làp-εε</i> NEG.EQU	'will not say (emphatic)'
			<i>mi-lap(-се</i> EQU)	'will not say'
FUT	làn ồ'	'will say'		
101	<i>iup</i> 0.	will say		
poss.like	<i>làp-с</i> ε ех	'has/hadto say'	làp-ce neg.ex	'has/had not (anything) to say'
HAB. PRS	làp-k ^h ữ: EQU	'said, says'	ma-làp-k ^h ĩ: EQU	ʻdidn't say'
			<i>mi-làp-kʰɛ̃ː</i> EQU	'doesn't say'
			<i>mi-lap-o</i> EQU	'does not say, used not to say'

Table 10.2. Negation of declarative final forms

As seen in Table 10.5, there is a slightly smaller number of affirmative constructions (15) than negated construction (17), showing paradigmatic asymmetry. Further asymmetry is evident in that:

 $^{^{354}}$ I have have no evidence that the choice of negator (*ma*- vs. *mi*-) would mark a semantic distinction in this construction. However, I suspect that *ma*- can be used both for past and present (habitual) actions ("used not to say, is not saying"), whereas *mi*- is only used for present actions ("is [in the habit of] saying ").

³⁵⁵ In Tashiding, West-Sikkim, the construction *làpto mè?* is also used.

1) Negation of the imperfective is split into symmetric negation (*ma-làp-to* [EQU] or *mi-làp-to* [EQU]), and non-symmetric negation, which is formally borrowed from non-reduplicated perfect construction and which occurs in the present continuous use (*làp-o* NEG.EX).

2) In the progressive type of constructions (imperfective, progressive, continuous), there is mismatch of form and function in that the negated form (lap-o NEG.EX) for these constructions is not derived from any of the three progressive type of constructions but from the non-reduplicated perfect construction (lapo EX). In this respect, Denjongke works analogously to Indo-Aryan Nepali, which uses the same negation strategy for perfect and present continuous.³⁵⁶

3) The same negated construction ma-làp(-o EQU) corresponds to three affirmed forms, the periphrastic past (-*po* EQU), past (-*tc* ϵ ?) and the completive (-*ts*^{*h*}a:).

4) The periphrastic past and nonpast constructions both have one affirmed form corresponding to two negated forms, a neutral negation accomplished through a negator prefix and an emphatic negation accomplished through negating the final auxiliary copula.

5) The future form (VERB \hat{o}) is negated by the same construction (*mi-làp*[- $\varepsilon \varepsilon$ EQU]) as the nonpast construction (- $\varepsilon \varepsilon$ EQU).

6) The affirmed perfect construction has two corresponding negated constructions, one of them involving a morpheme not occurring anywhere else in the negation system (-*ey*:, etymologically probably state 'shul' trace').

7) The habitual present form $(-k^{\hbar}\tilde{\epsilon} \cdot EQU)$ is typically negated by symmetric constructions formed with *ma*- and *mi*-, but occasionally asymmetric negation of the periphrastic past form with *mi*- (*mi*-VERB-*po* EQU) functions analogously (see the last row in Table 10.2).

Table 10.3 summarizes the types of finite clause negation in terms of two types of symmetry. The first is constructional symmetry, which tells whether the negated form in question is formed simply by adding a negator morpheme to an affirmative form (symmetric) or by other means (asymmetric). The second type of symmetry value marked in Table 10.3 expresses the relationhip of the negated form to its functionally equivalent affirmative form. If the negated counterpart of a certain affirmative construction is formed from the affirmative construction itself, negation is symmetrically related to the affirmative form. If the negated form is based on another affirmative construction, the relationship is asymmetric. For instance, the relationship between the affirmative past form *làptee* and its negated form *ma-làp-o* be? is asymmetric, because the negated form is formed/borrowed from the affirmative periphrastic past form làp-o be?. As seen in Table 10.3, constructional symmetry and relational symmetry may have opposite values. For instance, the negated construction ma-làp-o $b\epsilon$? is constructionally symmetric because it is formed from the affirmative form $lap-o b\epsilon^2$ by simple adding a negator morpheme. However, ma-lap-o be2 is also the negated equivalent of the affirmative completive form $lap-ts^{h}a$, with which the relationship is asymmetric because the negated form is not formed from the completive but borrowed from another construction.

In Table 10.3, S refers to symmetry and AS to asymmetry. The use of brackets in "(S)" means symmetry if the final TAME-marking is present (i.e. $m\dot{a}$ -lap-o \tilde{i} : is symmetric negation of $l\dot{a}p$ -o \tilde{i} , while the shorter form $m\dot{a}$ -lap is not).

³⁵⁶ Nep. *paani par-dai cha*. [water fall-PROG EX.NPST.3SG] 'It's raining.', *paani par-e-ko chaina* [water fall-PFV-NMLZ NEG.EX.NPST.3SG] 'It's not raining./It has not rained.'

Constr. Affirmative Negated		Constructional	Relational symmetry of	
			symmetry	neg. and affirm. form
state	làp	mi-làp	S	S
simp. prs	làp bɛʔ	mi-làp bɛ?	S	S
IPFV	làp-to (EQU)	ma-làp-to (EQU)	S	S
		<i>mi-làp-to</i> (EQU)		
		lap-o NEG.EX ³⁵⁷	AS	AS
CONT	<i>làp dø:</i> EX			AS
PROG	làp-tɛɛn EX			AS
periphr.	<i>làp-o</i> EQU	<i>làp-o</i> NEG.EQU	AS	S
PST		mà-lap(-o EOU)	(S)	S
PST	làp-tse		< / <	AS
CMPL	làp-ts ^h a:			AS
PRF	(làp-)làp-o	(làp-)làp-o	AS	S
	EX	NEG.EX		
		làp-ɛyː NEG.EX	AS	AS
RES	làp jờ?	làp mè?	AS	S
SEN.PST/	làp du?	làp mìndu	AS	S
SEN.RES	-	làp-o mìndu	AS	AS
NPST	làp-se EQU	làp-εε NEG.EQU	AS	S
	_	mi-lap(-ce EQU)	(S)	(S)
FUT	làp ồ:			AS
poss.like	làp-се EX	lap - $\epsilon\epsilon$ NEG.EX	AS	S
HAB. PRS	làp-k ^h ĩ: EQU	ma-làp-k ^h ế: EQU	S	S
		mi-làp-k ^h ẽ: EQU		
		<i>mi-lap-o</i> EQU	S	AS

Table 10.3. (A)symmetry in finite negated constructions

A deeper discourse-oriented analysis of negation may reveal further complexities, perhaps a complex network of relations between affirmative and negated forms similar to Contini-Morava's (1989) description of Swahili negation.

10.1.2 Negation in copular clauses

This section summarizes negation in copulas, which may be divided into simple copulas, combined copulas and nominalized copulas, which were already introduced in §7. As shown in Table 10.1 above, negated forms of simple copulas seem to historically derive from symmetric constructions, where the negator prefix attaches to an affirmative copula. Thus, negation in copulas may be termed historically symmetric (because the negated copula was formed by adding a negator prefix) but synchoronically asymmetric (because the process is no longer transparent or productive).³⁵⁸ The combinatory copulas, emphatic *imbe*? and sensorial equative *indu*? (see §7.3.1), do not have distinct negated forms. The form *mèmbe*? negates both the neutral equative *be*? and the emphatic *imbe*?. No

³⁵⁷ In Tashiding, West-Sikkim, the construction *làpto mè?* is also used.

³⁵⁸ Croft (1991) has put forward a hypothetical cycle according to which standard negation evolves from negation in existentials (see also Veselinova 2014).

negated form for *indu?* has been attested in natural speech or elicitation, and the hypothetical form **mèndu?* was rejected in elicitation.

I am aware of only one construction, the circumstantial adverbial, where a negator prefix attached to a copula is used instead of a negated copula:

(10.4)a) हु के उत्र द्वारा हेवा रा के का रा रेवा ठवा था के का का के का का के का के का के का के का के के का के के के क í. $k^h u$ teiku ma-ím-ba riktcen=jã: тí tcã:ta? 3SGM human beautiful only NEG-EQU.PER-CIRC intelligent=also EQU.PER 'He is not only a handsome person but also intelligent.' (referent present) (PT e) b) अन्त दर्न के का अक्र न दर्न के दहेगा का उत्त के जाविया के ता ána=di тí ma-ím-ba demo dziksu runpo=tci? old.lady=DEMPH human NEG-EQU.PER-CIRC demoness PN **PN=INDF** ĩ. EOU.PER 'The old lady, not being a human, is one demoness Dziksu Rungpo.' (ma-gsung 10)

The negated forms of nominalized copulas are given in Table 10.4. The nominalizers are -po/bo (2. infinitive), $-k^{h}\tilde{e}$: (nominalizer) and $-\epsilon\epsilon^{2}$ (1. infinitive). Nominalized copula constructions have two copulas, the first making the equative vs. existential distinction and the second one making the evidential distinction personal vs. neutral. The first, nominalized copula is always one of the personal copulas \tilde{i} : (neg. $m\tilde{e}$:) or $j\partial^{2}$ (neg. $m\tilde{e}$?). The final, evidentiality marking copula is either personal \tilde{i} : or neutral $b\epsilon^{2}$. The forms given with a question mark do not occur in my data. However, consultant KUN commented that he has heard all of them used, with the exception of $m\tilde{e}m$ -bo \tilde{t} :, the felicity of which he doubted.

Equ./	Evid./epist.	Nominalized co	onstruction		
Ex.	_	Affirmative		Negated	
Equat	Personal	ím-bo í́:	क्षेव 'र्च' क्षेव'	mèm-bo ĩ̃: (?)	यव'र्चे' क्षेत्र' (?)
•		íŋ-k ^h ẽ: ĩ:	જીવ.જાયવ. જીવ.	mèŋ-k ^h ẽ: ĩ:	मव सायव सेव
		<i>ี่</i> เวีธะ นี้:	क्षेत्र.वट्न. क्षेत्र.	mề̃:-ɕɛ í̃: (?)	ಷವ વૃત્ત જોવ (?)
	Neutral or	ím-bo bɛ?	क्षेत्र'र्च' श्वन्'	mềm-bo be?	बन्दां भून
	assertive	íŋ-kʰɛn bɛ?	छोव :यायव : भ्रद	mèŋ-kʰɛn bɛ?	षात्र स्थन. स्थन.
		í:-se be?	क्षेव् . १८ . झर	mề̃:-ɕɛ bɛʔ (?)	ਕਾਰ. ਰਟ. ਬਟ. (?)
Exist.	Personal	jờː-po ĩ́:	थॅट्र'र्ये' क्षेत्र'	mèː-po ĩ́:	बेट्र'र्रे' क्षेत्र'
		j <i>àː-kʰɛ̃ː í̃:</i>	धॅन् आपवः क्षेव	mèː-kʰɛ̃ː í̃ː	बेट्र'बायव्' क्षेव्'
		jờː-se ĩ:	र्षेन् मृन् क्षेत्र	mèː-ɛɛ í́ː (?)	ચેન્ નન્ ક્ષેત્ર (?)
	Neutral	jờː-po bɛʔ	थन्रायं भ्रन	mèː-po bɛʔ	बेन्'र्रे' झून्'
		jèbbe?/jàbbe?	थॅन्न भ्र	mèbbe?	એન્ન'સ્ન
		j <i>à:-kʰɛn bɛ</i> ?	र्थेन् अपन भ्रन	mèː-kʰɛn bɛ?	बेट्र'बायव्. झ्ट्
		jờ:-ce be?	र्षेन् १९ - भून	mè:-ɛɛ bɛʔ (?)	बेन् वन् झन् (?)

Table 10.4. Nominalized copulas

The nominalized copula construction are cases of asymmetric negation where the first, nominalized copula is replaced by a negated one, while the last, evidentiality-marking copula stays the same. For examples, refer to §7.3.2.

Although I do not currently have examples, the nominalized copulas have additional negated forms in which the final copula, not the first one, is negated (e.g. \hat{t} - $\epsilon\epsilon$ mèmbe? instead of mè:- $\epsilon\epsilon$ be?, and $j\partial$ -po mèmbe? instead of mè:-po be?). For examples on analogous negated constructions with other verbs, see §8.1.1 (past construction) and §8.2.5 (nonpast construction). Consultant KUN commented that the affirmative forms in Table 10.4 could be negated by negating the final copula, i.e. \hat{t} :- $\epsilon\epsilon$ mɛ́:. The only form that he was not certain about was the form $j\partial$:- $\epsilon\epsilon$ mɛ́:, which he haid he had not heard used. Negation of complex copulas leaves much to be investigated.

10.1.3 Negation in non-declarative clauses

Negation in non-declarative clauses is here addressed in two parts, non-copular clauses (§10.1.3.1) and copular clauses (§10.1.3.2)

10.1.3.1 Negation in interrogative, imperative, hortative and optative

Negation in imperative, hortative and optative moods is accomplished by adding the prefix *ma*- to the affirmative form, i.e. non-declarative negation in symmetrical, see Table 10.5. In imperative and hortative moods, the negator is prefixed to the verb root, while in the optative mood the negator occurs between the verb root and optative marker. Negation is not applicable to the exclamative mood (see §11.2).

I dole 1	rubie roler regulion of non deelalative non eopalar elauses							
IMP	làp(-tɛʰi)	'say!'	ma-làp(-tɛʰi)	'do not say'	symmetric			
HORT	làp-kɛ	'let me/us say'	ma-làp-kɛ	'let us/me not say'				
OPT	làp-tsuk	'let (her) say'	làp ma-tsuk	'let (her) not say'				

Table 10.5. Negation of non-declarative non-copular clauses

Negation of interrogated clauses is summarized in Table 10.6, where the glosses are simplified. Analogously to Table 10.3, constructional symmetry (S) refers to the fact that negation is accomplished simply by adding a negator morpheme to the formally corresponding affirmative interrogative (in any tense-aspect category). Constructional asymmetry (AS) refers to all other cases. Relationally symmetric are those negated forms which are derived from the temporally and aspectually corresponding affirmative form. Relational asymmetry refers to negated forms which are formed on the basis of another (non-corresponding) affirmative form. For instance, the affirmed imperfective interrogative *kjap-to-ka* 'is he doing' has two semantically corresponding negated forms are constructionally symmetric in that they derive from an affirmed form by simply adding the negator prefix. In terms of relational symmetry, however, *mi-kjap-to-ka* is symmetric because the form is derived from the affirmed imperfective interrogative *kjap-to-ka*. The other negated form, *mi-kjap-ka*, is relationally asymmetric, because it is formally derived from another construction, the steady state present form *kjap-ka*.

For a more detailed description of tense, aspect and modality expressed by the forms in Table 10, refer to §8, and for question formation, see §11. The forms in Table 10.6 mostly reflect the language of consultant KN who judged the forms in brackets marginal. As a general rule, it seems that questions negated by the negated interrogative equative auxiliaries $m \grave{e}mbo$, $m \grave{e}n$ -a, $m \grave{e}g$ -ga and the affirmative interrogative *ig-ga* are in effect like tag questions, whereas questions negated by other means are normal non-tag questions.

Table 10.6. Negation of interrogatives

Constr	Affirm.	Gloss	Negated	Gloss	Constr.	Relation.
	1. 1				symm.	symm.
state	kjap-ka	'does he do'	mi-kjap-ka	•does he not do	S	S
simp. prs	kjap bɛ-ka	'does he do'	kjap mèmbe-ka	'does he not do'	AS	S
	kjap ná	'does he do'	mi-kjap ná	'does he not do'	S	S
			kjap mèna	'he does, doesn't	AS	S
				he'		
IPFV	kjap-to-ka	'is he doing'	mi-kjap-ka	'does he not do'	S	AS
			(mi-kjap-to-ka)	'is he not doing'		S
	kjap-to ná	'is he doing'	ma-kjap ná	'is he not doing'	AS	AS
			(ma-kjap-to		S	S
			pá/bo)			
			kjap-to mèna ³⁵⁹	'he is doing, isn't	AS	S
				he'		
			ma-kjap-to mèna	'he is not doing,	AS	S
			<i></i>	is he'		
	kjap-to ín-ga	'he is doing,	kjap-to mèn-ga,	'he is doing, isn'i	AS	S
	51 50	isn't he	51 507	he'		
	kjap-to bo	'is he doing	ma-kjap-to bo	'whether he is	S	S
	51	(I wonder)	(alternative	not doing or'		
			<i>question</i>)	8		
			kiap-to mèmbo	'he is doing.	AS	S
			July to memory	isn's he'		~
			ma-kiap-to	'he is not doing.	AS	S
			mèmbo	is he'		~
	kian-to he-ka	'is he doing'	kian-to mèmbe-ka	'is he not doing'	AS	S
	Ngup to be ku	is ne doing	kjap to memor ka	'is he not doing'	AS	AS
			kjap o me. po	is no not doing	110	110
			kjap o minau ko kjap-o mè'-ka	'is he not doing'	Δ <u>S</u>	Δ
			kjap-0 meka	is ne not doing	10	A 5
			ма-kian-o ná	'is he not doing	S	45
			ma-kjup-0 jiu ma kjap o ho	did he not do'	5	ло
PROG	kian taon iài	'is he doing'	kian o màrka	fis he not doing?	15	45
TROO	kjup-isen jø	is ne doing	kjup-0 me.ku	is he not doing	AS	AS
	κα, κjap-ιωεπ du ka		кјар-о тіпай-ка			
	lian teon ià	tig ha daing?	kian taon mài no	fig ha not doing?	15	C
	Kjup-isen jø-	is ne doing	kjap-isen mepo	is he not doing	AS	3
	ро, кјар-њен					
	ай-ко		kian o mài no	tig ha not dain -'	45	15
			кјар-о те:-ро	is ne not doing	AS	АЗ
CONT	1. 1.	(° 1 1 ° °	кјар-о тіпай-ко	· · · · · · ·	10	4.0
CONT	кјар ав: јв:-	is he doing	кјар-о те:-ка	is he not doing	AS	AS
	ка, кјар dø:		kjap-o mindu-ka	(* 1 · · · ·		4.0
	du-ka		kjap dø:-p mè:-ka	'is he not doing'	AS	AS

³⁵⁹ The Nepali equivalent offered by KN was gar-dai-cha, hoina [do-PROG-NPST.3SG EQU.NPST.NEG.3SG]

			kjap døː-p mìndu- ka			
	kjap dø: j <i>à-</i> po	'is he doing'	kjap-o mèː-po kjap-o mìndu-ko	'is he not doing'	AS	AS
	kjap dø: du- ko		kjap dø:-p mè:-po kjap dø:-p mindu- ko	'is he not doing'	AS	AS
periphr.	kjap-o ná	'did he do'	ma-kjap-o ná	'did he not do'	S	S
PST			ma-kjap ná	'did he not do'	AS	AS
			kjap-o mèna	'he did, didn't	AS	S
			kjap-o mèŋ-ga	he'	AS	AS
			ma-kjap(-o) mèna	'he did not do,	AS	S
			ma-kjap(-o) mèŋ-ga	did he'	AS	AS
	kjap-o bo	'did he do	ma-kjap-o bo	'did he not do'	S	S
		(I wonder)	kjap-o mèmbo	'he did, didn't he'	AS	S
			ma-kjap(-o) mèmbo	'he did not do, did he'	AS	S
	kjap-0 ³⁶⁰	'did he do'	ma-kjap-o	'did he not do'	S	S
PST	kjap-tɛɛ ɲá, *kjap-tɛɛ bo	'did he do'	ma-kjap ná	'did he not do'	AS	AS
			kjap-t£ɛ mèna	'he did, didn't he'	AS	S
			kjap-t£ɛ mèmbo	'he did, didn't he'	AS	AS
	kjap-t£ɛ-ka kjap-t£a	'did he do'	ma-kjap ná	'did he not do'	AS	AS
CMPL	kjap-ts ^h aː ná	'has he done'	ma-kjap-o ná	'did he not do', 'has he not done'	S	AS
	kjap-ts ^h o-u ná		kjap-ts ^h o-u mèna	'he has done, hasn't he'	AS	S
	kjap-ts ^h a: bo	'has he done'	ma-kjap-o bo	'did he not do', 'has he not done'	S	AS
	kjap-ts ^h o-u bo		kjap-ts ^h o-u mèmbo	'he has done, hasn't he'	AS	S
PRF	(kjap-)kjap-o jøː-ka, (kjap-) kjapo du-ka	'has he done'	(kjap-)kjap-o mè:- ka, (kjap-)kjap-o mìndu-ka	'has he not done'	AS	S

 $^{^{360}}$ Essential for this form and the corresponding negation is raised pitch at the end.

	(kjap)-kjap-o jø-po, (kjap-) kjap-o duko	'has he done'	(kjap)-kjap-o mèː- po, (kjap-)kjap-o mìndu-ko	'has he not done'	AS	S
	kjap-ɛyː jờː- ka, kjap-ɛyː du-ka	'has he done'	kjap-£y: mè:-ka, kjap-£y: mìndu-ka	'has he not done'	AS	S
	kjap-ɛyː jò̀- po, kjap-ɛyː du-ko	'has he done'	kjap-sy: mè:-po, kjap-sy: mìndu-ko	'has he not done'	AS	S
RES	kjap j <i>à:-ka</i>	'has he done'	kjap mèː-ka	'has he not done'	AS	S
SEN.PST SEN.RES	kjap du-ka	'did he do'	kjap mìndu-ka kjap mìndu-ko	'did he not do'	AS	S
NPST	kjap-se ná	'will he do'	mi-kjap ná ³⁶¹	'will he not do'	AS	AS
	(kjap-sa)		mi-kjap-ka	'will he not do'	S	AS
			kjap-se mèna	'he will do, won't he'	AS	S
	kjap-se bo	'will he do'	mi-kjap-ka	'will he not do'	S	AS
			kjap-se mèmbo	'he will do, won't he'	AS	S
FUT	kjap òŋ-ga	'will he do'	mi-kjap-ka(m) ³⁶²	'will he not do'	S	AS
poss.like	kjap-se jð:- ka, kjap-se du-ka	'has he to do'	kjap-se mè:-ka, kjap-se mìndu-ka	'he has not to do'	AS	S
	kjap-£ɛ jờ-po, kjap-£ɛ du-ko	'has he to do'	kjap-ɛɛ mèː-po, kjap-ɛɛ mìndu-ko	'he has not to do'	AS	S
HAB. PRS	kjap-kʰɛ̃ː ná	'is he the one doing'	ma-kjap-k ^h ẽː ná mi-kjap-k ^h ẽː ná	'is he (the one) not doing'	S	S
			kjap-kʰɛ̃ː mɛ̀na kjap-kʰɛ̃ː mɛ̀ŋ-ga	'he is (the one) doing, isn't he'	AS	S
			ma-kjap-k ^h ẽ: mèna,	'he is not (the one) doing, is he'	AS	S
			ma-kjap-kʰɛ̃ː mɛ̀ŋ-		AS	AS
			ga, mi-kjap-k ^h ẽ: mòna		AS	S
			minu, mi-kjap-kʰɛ̃ː mɛ̀ŋ- ga		AS	AS
	kjap-kʰɛn bo	'is he the one doing	ma-kjap-kʰɛn bo	'is he (the one) not doing'	S	S
		(I wonder)'	kjap-kʰɛ̃ː mɛ̀mbo	'he is (the one) doing, isn't he'	AS	S

³⁶¹ This may also be a statement. ³⁶² The addition of -m makes the form an attenuated question.

	ma-kjap-k ^h ẽ: mèmbo mi-kjap-k ^h ẽ: mèmbo	'he is not (the one) doing, is he'	AS	S
	mi-kjap-o bo	'does he not do',	S	AS
		'is he not doing'		

Table 10.6 underlines the richness and complexity of question formation in Denjongke. For clausal examples of some of the negated questions, refer to the relevant constructions in §11. The semantic nuances of question formation in Denjongke leave a lot to be covered by further research.

10.1.3.2 Negation of interrogative copulas

My data has no examples of imperative, hortative or optative forms of the copulas. Verbs of becoming $t^{h} \otimes n$ 'come out, happen, become' and $t \in 'u\eta$ 'become', on the other hand, are used in the optative form, $t^{h} \otimes n$ ma-t $\in u^{2}$, $t \in 'u\eta$ ma-t $\in u^{2}$ 'let not be/become'. The non-declarative forms of proper copulas consist of interrogatives, which are given in Tables 10.7 (equative copulas) and 10.8 (existential copulas).

	Q type	Affirmed	1	Negated	
Р	suffix	íŋ-ga	'is (it)' (tag)	mèŋ-ga	'isn't (it)' (tag)
Е		íŋ-gam	'I wonder whether (it) is'	mèŋ-gam	'I wonder whether (it) is not'
R	neg. cop.	ná	'is (it)'	mèna	'isn't (it)' (tag)
		nám	'I wonder whether (it) is'	mènam	'I wonder whether (it) is not'
Ν	suffix	bɛ-ka	'is it'	mèmbe-ka	'is (it) not'
Е	neg. cop.	bo	'is (it, I wonder)'	mèmbo	'isn't (it, I wonder)' (tag)

Table 10.7. Negation of interrogated equative copulas

Table 10.8. Negation of interrogated existential	copulas
--	---------

	Affirmative		Negated		
Per	r $j \phi(:)$ -ka 'is (there)'		mè(:)-ka	'is (there) not'	
	jø(:)-kam	'I wonder whether (there) is'	mè(:)-kam	'I wonder whether (there) is not'	
Sen	du-ka	'is (there)'	mindu-ka	'is (there) not'	

As seen in Table 10.7 and Table 10.8, both equative and existential copulas may be interrogated with the suffix -ka/ga and its attenuated counterpart -kam/gam, whereas only equative interrogatives have separate negated forms (*mèna*, *mènam*, *mèmbo*). The attenuated forms occur only in the personal copulas i and $j\partial 2$, which represent diachronically older copulas, tracing back to Classical Tibetan.

10.1.4 Negation in subordinate clauses

Negation of those adverbial clauses for which I have data on negation is summarized in Table 10.10. The negation of copulas in subordinate clauses differs from other verbs in that copulas are negated by replacing the affirmative copula by the negated copula.

Form	Function	Affirmative	Negative	Constructional	Relational
				symmetry	symmetry
-nɛ/n0 ³⁶³	conditional	kjap-ne	ma-kjap-ne	S	S
-ruŋ	concession	kjap-ruŋ	ma-kjap-ruŋ	S	S
-ti	nonfinal	kjap-ti	ma-kjap-a	S	AS
-pa(r)	circumstance	kjap-a			S
-podã:	simultaneous	kjap-o-dã:	ma-kjap-o-dã:	S	S
-tsubdã:	simultaneous	kjap-tsubdã:			AS
-sondã:	simultaneous	kjap-sondã:	ma-kjap-sondã:	S	S
-somdã:		kjap-somdã:	ma-kjap-somdã:		
-kap	simultaneous	kjap-kap	ma-kjap-kap	S	S
		kjapø: kap	ma-kjapø: kap		
-dy:	simultaneous	kjap-dy:	ma-kjap-dy:	S	S
-rɛŋkʰa	simultaneous	kjap-rɛŋkʰa	ma-kjap-rɛŋkʰa	S	S
gã:	simultaneous	kjap-øː gãː	ma-kjap gã:	AS	AS
р'агри	simultaneous	kjap-kjap	ma-kjap p'a:pu	AS	AS
		p'aːpu			
$-s\overline{\tilde{a}}$:,	terminative	kjap-sonzã:	ma-kjap-sonzã:	S	S
sonzã:					
-ce=di:	causal	kjap-ce=di:	ma-kjap-ce=di:	S	S

Table 10.9. Negation of nonfinal forms

As shown by Table 10.10, negation in non-finite clauses is constructionally mostly symmetric, showing asymmetry in only two constructions (simultaneous $g\tilde{a}$: and p'a:pu). Two constructions (nonfinal and simultaneous *-tsubdã*:) show paradigmatic asymmetry in deriving the negated form from another construction. Simultaneous constructions with $g\tilde{a}$: and p'a:pu evince a peculiar paradigmatic asymmetry in having a unique negated construction which is not derived from any affirmative construction. The simultaneous construction formed with $g\tilde{a}$: 'time' differs from the functionally equivalent affirmative construction by dropping the nominalizer *-po* and the simultaneous construction formed by p'a:pu 'in between' differs from its functionally equivalent affirmative form by eliding reduplication.

10.2 Non-clausal negation

10.2.1 Negated replies

While Denjongke has no word corresponding to English *no*, one word negated replies are possible with the negated copulas. Negated existentials *mindu?* and *mè?* can function as one-word replies to questions relating to existence and location.

(10.5) Q: র্কু' র্ইন্ব'শ্।?

ts^ha t^hop-ka? salt find-PQ 'Is there salt?' (KN e)

³⁶³ Speakers literate in Tibetan also use the form *-na*, a loan from Classical Tibetan.

A: सेव रद्ग/सेदा mìndu?/mè?. NEG.EX.SEN/ NEG.EX.PER 'There is not.' (KN e)

The negated equative $m\tilde{\varepsilon}$ is used in the meaning 'it is not as you are saying' for contending with what the speaker's interlocutor has said. Typically, however, the form is not used totally independently but is followed by a further negated verb form.

 $\mathfrak{m} \tilde{\mathfrak{e}}$:, $t' \mathfrak{arin}$ $z \mathfrak{e}$:-p(o) $m \tilde{\mathfrak{e}}$:=lo. (10.6)NEG.EQU.PER today eat.HON-2INF NEG.EX.PER=REP 'No, (he) is not eating today, I hear.' (PT kitchen discussion)

Affirmed and negated questions differ with respect to whether they allow a one-word negated answer. In an affirmative question (10.7a), a negated reply with a mere negated copula was deemed infelicitous (10.7c). A negated verb is needed in addition (10.7b).

(10.7)a) Q: अन्न केंन्रेन कें राष्ट्र रा कु?

> $ts^{h} \varepsilon ri\eta = lo p^{h} \varepsilon - po$ dã: pá? vesterday PN=DAT meet-2INF EQU.PER.Q 'Did you meet Tshering yesterday?' (KN e)

- b) A1: अत्र, अ'र्म्रेट्रा mĩ:, $ma-p^{h} \varepsilon ?$. NEG.EQU.PER NEG-meet 'No, (I) didn't meet.' (KN e)
- c) A2: *অব্য *mĩ:. NEG.EQU.PER 'No (it is not like that).' (KN e)

In a negated question, on the other hand, a reply with a mere negated existential copula was deemed felicitous:

(10.8)

Q: अन्न- केंन्द्रे-'गॅ' दर्धन-'रॅ' अव'ग? dã: ts^hɛriŋ=lo p^hɛ-po mèŋ-ga? yesterday PN=DAT meet-2INF NEG.EQU.PER-PQ 'Didn't you meet Tshering yesterday?' (KN e)

A1: अत्र, (अ'तर्झ्त्') mề: $(ma-p^{h} \varepsilon^{2}).$ NEG.EQU.PER NEG-meet 'No (I did not meet).' (KN e) A2: $\operatorname{ag}(\mathfrak{F})$ $m\tilde{\tilde{e}}:$ ($p\dot{a}$). NEG.EQU.PER TAG.ASR 'No (indeed). (KN e)

10.2.2 Negative indefinites and quantifiers

Negative indefinites with meanings such as 'nobody', 'never', and 'nothing' are formed by appending the additive clitic $=j\tilde{a}$: 'even, too' to an interrogative word such as 'who', 'when', and 'what' and negating the following verb, see (10.9) and (10.10). Two forms, $k'and\varepsilon$: and $k'ar\varepsilon^{364}$ '(not) anything', do not function as interrogatives at all but only occur in negated statements, see (10.10) and (10.13).

- (10.9) $\operatorname{grav} \check{\mathfrak{A}}' \check{\mathfrak{$
- (10.10) ने चन या पत्रियायन यात्र तर्दा वाया के संयाया

t'izã: k'adi:=jã: $k'and\epsilon:$ làp $mi-ts^hu?$. but which.one.AGT=even anything say NEG-be.able.to 'But anyone was not able to say anything.' (Richhi 53)

For more examples on negative indefinites, refer to §6.3.2.

In quantification, (surprisingly) big numbers/amounts are often expressed through negated constructions. The negated item is typically the verb (10.11-13) but at least in once instance a quantifying adjective, *ma-pun* 'not a few, many' (10.14).

- (10.12) आयाया मुखाखेषा येन्।

ámmamma, cý:tey? mè?. EXCLAM(Nep.) a.little NEG.EQU.PER 'O my goodness, it's not a few (stray dogs that we have here).' (PL interview)

(10.13) कॅन् ग्रीस वार्म्य या रे बेन्

 $tc^{h} \sigma := ki$ mam-bja-u k'ar ϵ mè?. 2SG.L=AGT NEG-do-2INF anything NEG.EQU.PER 'There is not anything you haven't done.' ('dras-ljongs gsung-gtam 45)

 $^{^{364}}$ k'are is probably a shortened form of k'ande:. The retroflex /d/ alternates with /r/ also elsewhere.

(10.14) 5' 5' लहेबा बालूर' 55 वा होंगे!

t'a ŋà dɛm ma-μuŋ ŋý: p'in-diki now 1SG such NEG-be.few money give-NF 'Now I, having given so much money...' (TB bull story)

10.2.3 Negation of adjectives

Adjectives are negated either through a negator prefix attached to the adjective, see Table 10.10, or through a nominalized negated copula following the property concept word, see Table 10.11.

Positive		Negative		
lèm	good	mà-lɛm, mà-lɛp ³⁶⁵	bad	
tɛõːtaʔ(TB), tɛãːtaʔ (KT)	beautiful	ma-tɛom (TB) ma-tɛãːpo (KT)	ugly, not beautiful	
tsã:ţa?	clean	ma-tsãːm	dirty, unclean	
<i>dau</i>	similar	man-dau	dissimilar	
dempo	true	man-dɛmpo	untrue	

Table 10.10. Adjectives negated by a prefix

As seen in Table 10.10, adjectives are mainly negated by the perfective negator prefix *ma*-. The imperfective *mi*- may be used when forming future-oriented ad-hoc adjectives from verbs through nominalization, as shown in (10.15).

(10.15)	Positive		Negative	
	ञ्चुन्तः र्खुयाह्यःदे	1	- ⁵ ્ ત્રે જ્વા	ম'র্ন
	tc`uŋ	ts ^h u-po	tc`uŋ	mi-ts ^h u-po
	happen	be.able.to-2INF	happen	NEG-be.able.to-2INF
	'possibl	e'	'imposs	ible'

When an adjective is negated through a nominalized negated existential copula ($m\dot{e}$:- $k^{h}\tilde{e}$: or $m\dot{e}$:po), the adjectival suffix (e.g. -ta?, $-te^{h}\tilde{e}$:), which occurs in the positive form, is dropped and the resulting form, which is negated, is rather a noun describing a quality (e.g. sharpness, strength) than an adjective. This way of negating adjectives seems more productive than prefixing a negator, which has become more lexicalized. A few examples are given in Table 10.11.

Positive	0	Negative	
<i>nøta?</i> র্র'দেযাঝ'	sharp	nø mèːkʰɛ̃ː/mèːpo ǯ̃ aləːənəəsi/aləːːšī	blunt, not sharp
sém-súkts ^h ễ:	courageous	sém-cúk mèːkʰɛ̃ː/mèːpo	coward,
बेग्रब' मुगुब'ळेव		જોઅષ્ય: ભુષાષા એનુ:અષ્યનુ:/એનુ:ર્દ્ય:	not courageous,
<i>ɛúk-tɕʰɛ̃</i> ː ঀुषाषा केव	strong (lit.	cúk mèːkʰɛ̃ː/mèːpo ભुषाषा એન્'આવન્'/એન્'ર્સં	weak, not strong
top-tɕʰɛ̃ː क्लॅनषा केव	strength big)	top mèːkʰɛ̃ː/mèːpo ૹ૾ૢૼવષા એન્'આવન્'/એન્'ર્સં	

Table 10.11. Adjectives negated by a negated copula

³⁶⁵ There is no other, non-derived form meaning 'bad'.

The adjectival meanings 'different' and 'different kinds' are expressed through negation by constructions listed in Table 10.12.

rubie 10:12: Rajeeuves meaning amerent (kinds or)				
min-da	मेव दर्	[NEG-be.similar]	'different'	
ma-t£ik-0	अ'यव्ठिया'र्गे'	[NEG-one-NMLZ]	'different, not one'	
da man-da	तर्न् : यत्र :	[be.similar NEG-be.similar]	'different kinds of'	
da min-da	तर्दु सेव रद्	[be.similar NEG-be.similar]	'different kinds of'	
man-da gun-da	अव रदा नगु रदा	[NEG-be.similar nine-be.similar] 'different kinds of'	

Table 10.12. Adjectives meaning 'different (kinds of)'

t'iza: t'ã:pu dem min-qa be?. but long.ago like.that NEG-be.similar EQU.NE 'But it is different from how (it was) long ago.' (DB trip story)

(10.17) के दर्न अग्वरेग में लया है

mí=di **ma-teik-o** *làp-ti* human=DEMPH NEG-one-NMLZ say-NF 'saying these people (are) different...' (LA intro to Lachung)

(10.18) ך נוסיד אָי פּיפּיפּים השׁישי השׁישי הבי אָשְרָי העֹק אַרָן t'a ŷ:tso ŷ:tso=na átsi=tei? doly? daminda jô:-kʰɛn bɛ?. now district district=LOC a.bit=INDF tradition various EX-NMLZ EQU.NE 'Now, in different districts, the traditions are a bit different.' (SGD wedding customs)

For an example of the negated idiom *man-da gun-da*, refer to §10.3.3 below.

10.2.4 Privatives

Privative (or abessive) meanings (cf. English *without* N, N-*less, un*-N) are expressed through negated existential copulas, see (10.19). This strategy is identical with negating certain adjectives, see Table 10.11 above.

(10.19) ८. रू. हेवागा तदेव खुवार्श्वेर तवार्श्व रेखुर केन्द्र तवार्श्व केन्द्र तवार्श्व तवार्श्व तवार्श्व केन्द्र खेन ý:kor bak-ti ri-lun nà=rã: teuku dep mèː-po k^hjam 1SG=AEMPH only like.this tour carry-NF hill.valley EX-2INF roam gju dø: bak-ti j*ò*?. carry-NF go sit EX.PER 'Like this I'm travelling alone, without a place (to stay) in the hills or valleys, roaming around.' (PAD bet story)

Clausal privatives (cf. 'without doing') are expressed through negated circumstantial converb or through the construction NEG-VERB-*po* p'ja(ti), see §15.8.1.

10.3 Other aspects of negation

This section provides a note on negation in complex clauses (\$10.3.1) and describes two idiomatic constructions involving negation, the negated restrictive *-ma* (\$10.3.2) and the variety marking idiom NEG-VERB *gu*-VERB (\$10.3.3).

10.3.1 A note on negation in complex clauses

The negated connector mi- $ts^{h}\varepsilon^{2}$ (NEG-stop) 'not only, in addition' functions both as a clause connector and a more loose discourse connector, see §12.2 and §15.9.1. There are no forms corresponding to English *neither...nor*, but the same function is covered by using either the independent connector $j\hat{a}$: 'and, again' or its cliticized form $=j\tilde{a}$: with a negated verb in both clauses, see §12.3.

A functional approximation of English subordinate clauses with *lest* can be formed by $m \hat{\epsilon} nn\epsilon/m \hat{\epsilon} nni$, which was in §3.5.2.5 introduced as an epistemic adverb meaning 'maybe, perhaps'. The form $m \hat{\epsilon} n(n)\epsilon/m \hat{\epsilon} n(n)i$ likely derives from the conditional form of the negated personal equative $m \hat{\epsilon}$: and thus seems to have an air of negation similar to English *lest*. In (10.20), the relatives of a bride insist on seeing the prospective groom, lest they be deceived.

(10.20) אַמִיּזֹי בְּרָ׳ מִיָּבִי עָבְיָצִי, זוֹיִבּאָיאון אַמִיזֹי אָבָי אָזי אַז אַמִיזוֹי אַבָּיבָ אַמִיאון אַמִיזוֹי אַבָּיבָ אַמִיאון אַמִיזוֹי אַבּיבָ אַמִיאון אַמיזיזי בּבּי máko=di k'ana jè-po, kolɛ=s. máko mèn(ɛ) groom=DEMPH where EX-2INF where=QUO groom perhaps mik-ca:zou=rõ: oُ:. eye-lame=AEMPH come 'Where is the groom? Where? Lest the groom be blind.' (SGD wedding customs)

In (10.21), *mènni*, in combination with the preceding conditional construction and following negated verb, forms a functional approximation of the English imperative followed by a *lest*-clause.

(10.21) جِسَّ مَعْرَ عَرْجَ مَعْرَعُ مَعْرَعْ مَعْرَعْ مَعْرَعْ مَعْرَعْ مَعْرَعْ مُعْرَعْ مُعْرَعْ مُعْرَعْ مَعْرَعْ مُعْرَعْ مَعْرَعْ مَعْرَعْ مَعْرَعْ مَعْرَعُ مَعْرَعْ مَعْرَعْ مَعْرَعْ مَعْرَعْ مَعْرَعْ مَعْرَعْ مُعْرَعْ مَعْرَعْ مُ مَعْرَعْ مَعْمَ مَعْرَعْ مَعْمَ مَعْرَعْ مَعْمَ مُعْمَ مَعْرَعْ مَعْمَ مَعْمَ مَعْمَا مَعْمَ مَعْ مُعْمَ والمَعْلَمُ مَعْذَعُ مَعْمَا مَعْمَا مَعْمَا مَعْمَ مَعْرَعْ مَعْمَا مَعْنَعْ مَعْمَ مَعْرَعْ مَعْمَ مَعْمَ مَعْ والمَعْنَا مَعْنَا مَعْمَا مَعْمَا مَعْمَ مَعْ م

Note that while English *lest* co-occurs with an affirmed verb, $m \hat{e} nni$ in (10.21) is followed by a negated verb.

10.3.2 Negated restrictive -ma 'more than (+NEG.EX), only'

The restrictive suffix *-ma*, which is distinct from the negator prefix *ma*-, attaches to quantifying words and together with a negated existential expresses about the same meaning as English '(there is) no more than'. It is functionally close to the adverb *teiku* 'only' used with an affirmative existential. Consultant KN commented that *=ma* in (10.22) could be replaced by the marker $p^{h}a:g\varepsilon$ or *ma:g* ε and retain the same meaning.

- (10.22) riseq à grigria a first riseq for a first firs

In the novel Richhi and the audio-play *rnam-rtog*, *-ma* is written together with the word it is postposed to, see (10.23-25). In the context of (10.24), a man and a woman are each counting pieces of an orange split in two to find out whether the comparative number of orange pieces bodes well for their prospective marriage.

(10.24) A: $\hat{a}_{\$} = c\hat{h} \cdot \vec{s} \cdot \vec{p} \cdot \nabla c \sigma \vec{q} \cdot \vec{s} \cdot \vec{s} \cdot \vec{q} \cdot \vec{q} \cdot \vec{s} \cdot \vec{p} \cdot \vec{q} \cdot \vec{q} \cdot \vec{s} \cdot \vec{q} \cdot \vec{q} \cdot \vec{s} \cdot \vec{q} \cdot \vec{q} \cdot \vec{s} \cdot \vec{p} \cdot \vec{q} \cdot \vec{s} \cdot \vec{q} \cdot \vec{q} \cdot \vec{q} \cdot \vec{s} \cdot \vec{q} \cdot$

B: ८ते.र्ड'र्ये'थ८' पर्ट्रुव'स' सेव'पर्ट्या |"

 $p\dot{e}$: $tsa=lo=j\tilde{a}$:dyn-mamindu?.1SG.GENat=DAT=evenseven-(no.)more.thanNEG.EX.SEN'I too haven't (any) more than seven.'

(10.25) A: झुत' वर्ने' यत्र' हॅग्रागय' अ'कु।

 $m \acute{e}n = di$ man-dzo:-kam ágja? medicine=DEMPH NEG-be.finished-ATTQ elder.brother 'Isn't the medicine finished, brother?'

B: ק׳ אַמיצעי אא אירקעי צעיאטי ארן

t'a	сý:tcy: -ma	mìndu?.	dzoː-rap	bɛ?.
now	a.bit-(no.)more.than	NEG.EX.SEN	be.finished-IMF	EQU.NE
'Now	there's no more than a	bit. It's about t	to be finished.' (rn	am-rtog 33)

A similar meaning may be expressed by *mèmbo* accompanied by a negated verb:

(10.26) कॅर्बा क्रिन: क्षुत रहुंग वियात मेंगा क्षेग तमुग्तरा मा केता यहेगा यता के मा

 $t \varepsilon^h \phi ki$?pyn=tsu $k^h im=na$ $l \partial k$ -tigju-wat'an im $t \varepsilon i$? $m \epsilon mbo$ PNsibling=PLhouse=LOCreturn-NFgo-PURnowdayoneNEG.EQU.NE $m \epsilon$?NEG.EX.PERNEG.EX.PER

'It is no more than one day and Choki will return to (her) sibling's house.' (Richhi 136)

10.3.3 Variety marking idiom NEG-VERB gu-VERB

The negated construction NEG-VERB gu-VERB marks undefined variety, an apt translation often being 'doing various kinds of'. The formative gu means nine, rendering a literal translation 'not doing nine doing'. In the novel Richhi, the construction always collocates with the verb $n\delta$ or $n\delta$:sam tã:, which both mean 'think', see (10.27)

(10.27) هَ (المَ اللَّٰ عَرَابَةُ اللَّهُ عَرَابَةُ اللَّهُ عَرَابَةُ اللَّهُ عَرَابَةُ اللَّهُ عَرَابَةُ اللَّهُ عَرَابَةُ عَرَابَةُ اللَّهُ عَرَابَةُ عَرَابَةُ عَرَابَةُ اللَّهُ عَرَابَةُ اللَّهُ عَرَابَةُ اللَّهُ عَرَابَةُ عَرَابَةُ اللَّهُ عَرَابَةُ عَرَابَةُ اللَّهُ عَرَابَةُ اللَّهُ عَرَابَةُ اللَّهُ عَرَابَةُ اللَّهُ عَرَابَةُ اللَّهُ عَرَابَةُ اللَّهُ عَرَابَةُ عَرَابَةُ اللَّهُ عَرَابَةُ اللَّهُ عَرَابَةُ اللَّهُ عَرَابَةُ اللَّهُ عَرَابَةُ اللَّهُ عَرَابَةُ اللَّالَةُ اللَّهُ عَرَابَةُ اللَّالَ اللَّهُ عَرَابَةُ اللَّالَ اللَّكَ عَرَابَةُ عَرَابَةُ اللَّعَالَ اللَّهُ عَرَابَةُ اللَّهُ عَرَابَةُ اللَّهُ عَرَابَةُ اللَّهُ عَرَابَةُ اللَّهُ عَرَابَةُ عَنَابَةُ اللَّالَ اللَّالَ اللَّهُ عَرَابَةُ عَرَابَةُ اللَّهُ عَرَابَةُ عَرَابَةُ اللَّ

'That night Karma thinks many various kinds of thoughts.' (Richhi 171)

(10.28) สสาคร รฐาคร

man-qa gun-qa NEG-be.similar nine-be.similar 'various kinds of' (TB, oh)

10.4 Summary remarks

This chapter provided a summary of features related to negation. Special emphasis was given to the concept of symmetry and asymmetry, which was addressed from two perspectives, constructional and relational. It was shown that some negated tense-aspect constructions do not have a distinct negated form but borrow the negation strategy from another construction, leading into less negated forms than affirmative forms. This tendency, however, is counterbalanced by that fact that some affirmative constructions can be negated in more than one way, leading into more negated forms than affirmative forms. For instance, past and nonpast constructions have, in addition to the ordinary negated construction (formed by the negator prefix), also an emphatic negated form (formed by negating the final auxiliary copula).

Negation of questions was found to be highly complex and deserving of further study. Adjectives are negated in two ways the first of which resembles verbal negation (negator prefix) and the other one nominal privative constructions (negated and nominalized existential).

11 Non-declarative clauses

While other chapters discuss mainly declarative sentences, this chapter focuses on non-declarative sentence moods: interrogative (§11.1), exclamative (§11.2), imperative (§11.3), hortative (§11.4) and optative (§11.5). The bulk of the discussion addresses interrogatives, which show a wide variety of constructions, while the other clause types receive less attention. The imperative is concerned with second person commands and requests, hortative first person suggestions and optative third person wishes.

11.1 Interrogatives

Interrogative clauses in Denjongke consist of polar questions, content questions, alternative questions and tag questions.³⁶⁶ Polar interrogatives, also known as yes/no-questions, question whether something is the case, or, when negated, whether something is not the case. Although ideally a polar question expects a "yes" or a "no" as an answer, there is a range of replying possibilities in between, for instance 'most likely', 'maybe' or 'hardly'. Content questions, also known as "constituent interrogatives" and "who', 'where' and 'why'. Through content questions, also known as "constituent interrogatives" and "information questions" (König & Siemund 2007: 291), the speaker seeks information whose type is revealed by a wh-word in the clause, for instance 'who' for identity, 'where' for location' and 'why' for reason. Alternative questions present the addressee with two alternatives and seek information as to which is the case, for instance 'Would you like to have apples or oranges?". Tag questions in Denjongke are interrogated affirmative and negated equative copulas (affirmed *iŋ-ga, bɛ-ka* and negated *mèmbo, mèna, mèŋ-ga*), which are appended to the end of the clause to make a declarative clause an interrogative. At the same time, tag questions raise expectations about the answer (König & Siemund 2007: 296).

Denjongke interrogative markers are briefly introduced in Table 11.1. Their use is exemplified in the following sections. Formally, interrogative markers can be divided into interrogative suffixes and interrogative equative copulas. Existential copulas do not have separate interrogative forms but they are interrogated by the same question suffixes as the other verbs (-ka/ga and -kam/gam).

³⁶⁶ It is possible to make a difference between the concept of question (pragmatic act) and interrogation (grammatical category) so that what functions as a question is not necessarily an instance of grammatical interrogation. This grammar, however, does not take into account such a distinction. Question and interrogation are used synonymously, referring to grammatical category.

Verb	Polarity	Direct	polar	Attenuated	polar and content
type		questions		questions	
All Vs	-	-ka/ga	गा'/ग	-kam/gam	गाठा/याठा
Personal	Affirm.	ná	3 ^{./} छोत्र'त' ³⁶⁷	па́т ³⁶⁸	ષ્ટ્રચ:/બ્રેવચ:/બ્રેવ'વચ:
equative	Neg.	mèna	ਕਰ-ਰ-	mènam	षाव 'ववा'
Neutral	Affirm.	bo	(ন)র্ন'		
equative	Neg.	mèmbo	মন্ব'র্ন'		

Table 11.1. Interrogative morphemes

As shown in Figure 11.1, the interrogative suffixes, which can attach to the verb root or verbal suffixes, are $-ka/ga^{369}$ and its attenuated counterpart -kam/gam. The attenuated question suffix -kam/gam (along with the attenuated interrogative copula pidm) tones down the directness of a question by posing it as if the speaker were wondering to themselves quite like in the English expression *I wonder* (*whether*).³⁷⁰ While -ka/ga is a polar question marker, -kam/gam occurs in both polar and content questions, and is the preferred choice in interrogative complement clauses. The pre-verbal polar interrogative \dot{a} , which is used in village of Lachung (North Sikkim), is not included in the table but is separately described in §11.1.1.5.

The interrogative equative copulas are personal $p\dot{a}$ (neg. $m\dot{e}n$ -a), its attenuated counterpart $p\dot{a}m$ (neg. $m\dot{e}nam$) and evidentially neutral bo (neg. $m\dot{e}mbo$). While the interrogative copula $p\dot{a}$ is used in polar questions, its attenuated counterpart $p\dot{a}m$ occurs in both polar and content questions, and is the preferred choice in interrogative complement clauses. The interrogative bo, similarly to declarative $b\epsilon^2$ (§5.4.2), may in addition to equation also be used for location, e.g. k 'ana bo? [where EQU.NE.Q] 'Where is (it)?'. The marker bo occurs in polar, content and alternative questions. Finally, there is a marginal alternative question marker $-l\phi^2$, which will illustrated in §11.1.3.4.

In addition to the separate interrogative forms, copulas can be interrogated, similar to other verbs, by the suffixes *-ka/ga* and *-kam/gam*, as shown in Table 11.2.

³⁶⁷ Both these forms occur in Denjongke writing. I am using the simpler form $n\dot{a}$, although it does not represent high register implied by the initial of the source form ($\Re \eta \cdot \eta$). Note that the form suggested to me for writing the attenuated form $n\dot{a}m$ gav has a superscript which implies high register.

³⁶⁸ The form $p \dot{a} m$ most likely derives from the interrogated personal copula in - (n)am (in WD both δa_{a} and δa_{a} are used), which is still productive in both polar and content questions

 $^{^{369}}$ The phonetic output tends to be /g/ when preceded by voiced sounds and /k/ when preceded by voiceless sounds, including the glottal stop.

³⁷⁰ Consultant KL translated the clause *k'ana số:-bo nám?* [where go.PFV-PST EQU.PER.Q] 'Where did (he) go, I wonder?' into Nepali as $kah\tilde{a}$ ga-yo holā [where go-PST.3SG COP.PROB.3SG] 'Where might he have gone?', using the Nepali dubitative copula holā to correspond to Denjongke *nám*.

Copula	Polarity	Direct polar questions		Attenuated polar and content	
type				questions	
Personal	Affirm.	íŋ-ga	क्षेत्र ग	íŋ-gam	জিব শাব্য
equative	Neg.	mèŋ-ga	ষব যা	mèŋ-gam	মব যাম'
Neutral	Affirm.	bɛ-ka	শ্বন্'শ্'	bɛ-kam	ञ्चन् गगवा
equative	Neg.	mèmbe-ka	য়ঀ৾৾৽য়ঀ৾৾৾৾৾৾৾৾য়	mèmbe-kam (?) ³⁷¹	बात्र : झुन् गावा (?)
Per. ex.	Affirm.	jờː-ka	थॅन्ग	jờː-kam	धॅट्-'गठा'
	Neg.	mèː-ka	बेम्'गा'	mèː-kam	बेद्त्'गव्य'
Sen. ex.	Affirm.	du-ka	दर्नुगागा.	du-kam ³⁷²	तनुयाःगवः
	Neg.	mindu-ka	बेव २५५ गा गा	mìndu-kam	बेव .पट्या गावा.

Table 11.2. Negating copulas with -ka/ga

As shown by Tables 11.1 and 11.2, the equative copulas have two types of interrogated forms, the separate interrogative copulas $p\dot{a}$ (neg. $m\dot{\epsilon}na$), $p\dot{a}m$ (neg. $m\dot{\epsilon}nam$) and bo, and the regularly formed corresponding $i\eta$ -ga (neg. $m\dot{\epsilon}\eta$ -ga), $i\eta$ -gam ($m\dot{\epsilon}\eta$ -gam), $b\epsilon$ -ka (neg. $m\dot{\epsilon}mb\epsilon$ -ka), and $b\epsilon$ -kam (neg. $m\dot{\epsilon}mb\epsilon$ -kam). The existential copulas have only the regular interrogated forms with -ka/ga and -kam/gam.

Sandberg (1895: 47) reports *-na* as an interrogative morpheme and gives the example *Chhö ām chi t'ong-che-na* 'Did you see a silver fox', which is given in edited version in (11.1).

(11.1) Sandberg (1895: 47) (edited transcription, WD and glossing mine)

 \tilde{a} ਨੂੰ ਕੂਬਾ ਗ੍ਰੇਗਾ ਕਬੇਂ ਨਾਲੇ ਕ? $tc^h \phi$? $\dot{a}:m=tci$? $t^h \tilde{o}:-tc \varepsilon-na$? 2SG.L fox=INDF see-PST-PQ 'Did you see the silver fox.'

In my data, however, -*na* does not occur as a question marker outside the copula forms $in-na(m) > p\dot{a}(m)$, $m\dot{\epsilon}n-na(m)$.

The following sections are divided into polar questions (\$11.1.1), question words and content questions (\$11.1.2), alternative questions (\$11.1.3), tag questions (\$11.1.4), questions with the reportative =*lo* (\$11.1.5), and exclamative questions with (*h*)*o*: (\$11.1.6). For the affirmed and negated polar questions in various tense and aspect constructions, refer to Table 10.5 in \$10.1.3.1.

11.1.1 Polar questions

Polar questions in Denjongke may be formed by rising intonation accompanied by *-po*-infinitive (\$11.1.1.1). More frequently, however, polar questions are formed by the polar interrogative suffix *-ka/ga* or the polar interrogative copula $p\dot{a}$ (often phonetically reduced to $j\dot{a}$) but also by other markers, which occur both in polar questions and content questions: the attenuated interrogative suffix *-kam/gam*, the attenuated interrogative copula $p\dot{a}m$ (often reduced to $j\dot{a}m$) and the interrogative copula/auxiliary *bo* (neg. $m\dot{e}mbo$). The polar uses of these markers are discussed in \$11.1.12 (*-ka/ga* and $p\dot{a}$), \$11.1.13 (*-kam/gam* and $p\dot{a}m$) and \$11.1.14 (*bo*). The last section under

³⁷¹ I do not currently have examples of *mèmbe-kam* but its existence can be hypothesized on the basis of the positive form *be-kam*.

³⁷² I have heard the forms *du-kam* and *mindu-kam* in use but I have no example sentences of them.

polar questions (\$11.1.1.5) describes a further interrogative marker \dot{a} , which is used in Lachung. Section \$11.1.1.6 provides a summary on polar questions.

11.1.1.1 Polar questions with rising intonation and -po-infinitive

Polar questions in Denjongke may be formed by rising intonation at the end of a clause which ends in a *-po*-infinitive, see (11.2) and (11.3), and their pitch traces in Figure 11.1 and Figure 11.2 respectively. With stative verbs (11.2), the question concerns a present state, whereas with eventive verbs (11.3) the question concerns a past event.

(11.2) गु'र्रा' इत थॅन र्रा?

gapo=tsa: jè-po? elder.man=at EQU.PER-2INF 'Does sir (=you) have (that)?' (DR discussion with KL)

Figure 11.1. Intonation in polar question (11.2)



(11.3) দ্বিক্ত অ'নেস্ত্র্যার্শী? t'itsi ma-tsuk-o? a.few.days.ago NEG-set-2INF '(You) haven't put (it there) lately?' (TB telephone call)

Figure 11.2. Intonation in polar question (11.3)



Sandberg (1895: 73) reports the polar question *chhö* ts'ongkhen hlam du' bo?, which is given in edited form as (11.4). As suggested by the glossing in (11.4), I assume that Sandberg's final morpheme is the nominalizer/infinitivizer -*po/bo*, which does not assimilate to the underlying velar in *duk/du*? (hence *du-bo*³⁷³). For comparison, see example (11.5) where the nominalizer reduces to -*o* (hence *duk-o*).³⁷⁴

³⁷³ I would have expected *-po* instead of *-bo* because final glottal (such as the one in du?) is in my data followed by voiceless sounds.

³⁷⁴ Sandberg reports also words with the sequence /kp/ which in current Denjongke have been reduced to /k/, e.g. Sandberg's (1895: 33) reports *nak-po* 'black' whereas my data has *nàku* 'black'.

(11.4) Sandberg (1895: 73) (WD, phonological transcription and glossing mine) $\tilde{\mathfrak{s}}_{\overline{1}} \sim \alpha \tilde{\mathfrak{s}}_{\overline{1}} \approx \alpha \tilde{\mathfrak{s}}_{\overline{1}} \approx \alpha \tilde{\mathfrak{s}}_{\overline{1}} \tilde{\mathfrak{s}}_{\overline{1}}^{2}$ $t \varepsilon^{h} \phi^{2} t s^{h} o g - k^{h} \tilde{\varepsilon}$: lam du-bo? 2SG.L sell-NMLZ shoe EX.SEN-2INF 'Have you any boots to sell?'

(11.5) দশ্যন:স্ক্রিস্ট্রন্রে এন্ব্রার্গ? kalimpoŋ=na dɛm duk-o? TPN=LOC such EX.SEN-2INF 'Are there such (things) in Kalimpong?' (KN photo discussion)

Because the nominalizer does not in my data attach to the sensorial du^2 in other contexts than questions,³⁷⁵ it is possible that the suffix -*po/bo/o* is in conjunction with du^2 becoming in effect a question marker.

11.1.1.2 Polar questions with -ka/ga and pá

The polar question markers -ka/ga and $p\dot{a}$ differ from each other in that $p\dot{a}$, being an interrogative copula, replaces the equivalent declarative equative copula (11.6), whereas -ka/ga may be appended to both equative (11.7) and existential copulas (11.8) to form interrogatives.

(11.6) a) र्क्षन् झुरे प्रदे खु?

te^hø? ta:ri di **pá**?
2SG.L axe this EQU.PER.Q
'Is your axe this (one)?' (JDF axe story)

- b) $\widetilde{\mathfrak{s}}_{\Gamma} \widetilde{\mathfrak{R}} \widetilde{\mathfrak{r}} \widetilde{\mathfrak{R}} \widetilde{\mathfrak{R}}_{\Gamma} \widetilde{\mathfrak{R}}_{\Gamma}$ $t \varepsilon^{h} \phi^{2} ta : ri di \widetilde{t}:$ 2 SG.L axe this EQU.PER 'Your axe is this (one).' (KN e)
- (11.7) $\widetilde{\mathfrak{s}}_{\mathsf{T}}^{\mathsf{T}} \widetilde{\mathfrak{A}}^{\mathsf{T}} \widetilde{\mathfrak{A}}^{\mathsf{T}} \widetilde{\mathfrak{A}}^{\mathsf{T}}} \widetilde{\mathfrak{A}}^{\mathsf{T}} \widetilde{\mathfrak{A}}^{\mathsf{T}}}?$ $te^{h}\phi^{2} \quad lopt^{h}u^{2} \quad be-ka?$ $2\text{SG.L} \quad \text{student EQU.NE-PQ}$ 'Are you a student?'
- (11.8) בי עוֹק ייזן?

tc'a jø:-ka? tea EX.PER-PQ 'Is there tea?' (Barapathing discussion)

However, when copulas function as auxiliaries, they are often elided in interrogatives. This results in $p\dot{a}$ and -ka/ga occurring in syntactically analogous environments, as shown in (11.9) and

³⁷⁵ In other contexts with nominalization, the evidential distinction between personal existential $j\partial^2$ and sensorial du^2 is neutralized so that only $j\partial^2$ occurs as nominalized with -po/bo ($j\partial -po$).

(11.10) respectively. Example (11.10b) shows that with the imperfective -to/do the auxiliary is optional even in the declarative form, hence $b\epsilon^2$ in brackets.

(11.9) अूत्र गुर्थाण मुत्राल के? *lenge? jó? kjap-to pá*? PRN.HON work do-IPFV EQU.PER.Q 'Are you working?' (KN e)

> b) אָר אַד אֹשְאַ אַשָּר אָשָאַ אָדָי (אָדָ)? *dzã:dar lêm tʰøn-do (bɛ?).* training good happen-IPFV (EQU.NE) 'The training is turning out good.' (KN e)

Based on the distribution of -ka/ga and $p\dot{a}$ with copulas ($p\dot{a}$ replaces equative copulas and -ka/ga is appended to both equatives and existentials), it can be said that in (11.9) the copula is replaced by $p\dot{a}$, whereas in (11.10a) the copula is elided. The syntactically overlapping and non-overlapping contexts of -ka/ga and $p\dot{a}$ are described in more detail later in this section.

Both -ka (11.11) and $p\dot{a}$ (11.12) may be postposed to a verb root:

(11.11) a) 도직 직기까?

ná: làp-ka? I.AGT say-PQ 'Shall I tell?' (KN e)

b) अप्रे, रूट के केंट ग?

ái, rã: mi-ts^hiŋ-ka? elder.sister 2SG.M NEG-get.angry-PQ 'Sister, won't you get angry?' (Richhi 41)

(11.12) রুণ্' স্ট্র্র্য নৃ?

sùk kjap µá? pain strike EQU.PER.Q 'Is it hurting? (TB e)

The intonation in polar interrogatives with -ka/ga rises at the end of the clause, however not on final -ka/ga but on the penultimate syllable, as seen in Figure 11.3, which gives the pitch trace from (11.13), and in Figure (11.4), which provides the pitch trace from (11.14).

(11.13) (אַיקרישָאַי צָאַקאַי אַאָקרי אָקרי אָרָישָ? moby:=ki pho? thop-ce jo:-ka? wife=AGT salary receive-INF EX.PER-PQ 'Does (your) wife get salary?' (Bp BB discussion)

Figure 11.3. Intonation in polar question (11.13) with *-ka/ga*



Figure 11.4. Intonation in polar question (11.14) with -ka/ga



Figure 11.5 presents the pitch trace from (11.15), which is a declarative clause with the same past suffix *-tee* as in Figure 11.4, illustrating that while the pitch on declarative *-ze* is lower than the previous word p'ja 'do', the pitch on the interrogative *-ze* is higher than on p'ja.

(11.15) हे. ८. ज्या क्रेंपरने केया महया पश्चन र्युमाली

 $t\varepsilon$ $\eta \dot{a}t\varepsilon a$? $\dot{o}d\varepsilon$:= $t\varepsilon i$? dzuk-dy: p'ja- $z\varepsilon$. then 1PL like.that=INDF ending do-PST 'So at that point we ended.' (NAB BLA 7)





Intonation in polar questions with the interrogative copula $p\dot{a}$ is exemplified in Figures 11.6 and 11.7, which give the pitch traces from (11.16) and (11.17) respectively. In Figure 11.5, the decrease in the pitch trace at the end of syllable $ts^{h}o$ is caused by background noise. The pitch in both clauses has a slight rise on the penultimate syllable, thus resembling interrogative intonation with -ka/ga.

(11.16) 「管 遊叭下 弓?
 t'ato ts^ho:-to pá?
 now gather-IPFV EQU.PQ
 'Are (they) gathering now.' (KN kitchen discussion)

Figure 11.6. Intonation in polar question (11.16) with pá



(11.17) \check{a}_{∇} \check{e}_{A} (phone) $\nabla_{\nabla} \nabla$ \check{a} ? $te^{h}\phi^{2}$ fon $t\tilde{a}$:-bo $p\dot{a}$? 2SG.L phone(Eng.) send-2INF EQU.PQ 'Did you call?' (KN e)

Figure 11.7. Intonation in polar question (11.17) with *pá*



The interrogative suffix -ka/ga does not have a negated form, but the interrogative copula pa has the specific negated form mena, which functions very similarly to the regularly formed mena. ga. In my data, mena is more frequent than menaber ga, the use of which is limited to consultant KN.

Example (11.18) illustrates an independent copular use of $m \epsilon na$, while (11.19) provides auxiliary uses, which correspond to English tag questions. For $m \epsilon \eta - ga$, consider (11.20).
(11.18) यासुत्रा यासुत्रा मुनया नर्याया भन्न झन्। यात्र, रावि रावे यात्र वा súm súm kjap go:-ce *bε*?. mẽː, zi zi four four three three do be.needed-INF EQU.NE NEG.EQU.PER mèn-a? NEG.EQU.PER-PQ 'Three of each needs to be made. No, isn't (it) four each?' (KNA kitchen discussion) rã: k^hõ:fa mèn-a làp-o=lo. sák-to accumulate-IPFV NEG.EQU.PER-PQ say-2INF=REP 2SG.L anger 'You are getting angry, aren't you, he said (so the story goes).' (PD bet story) b) ਘ, ८ जियामर अवाव, र्रायर र्रायया प्र्या र्योवामर अवार्य? mèn-a? pè:-se jà: t'a t'o:pa t'ole gju g0:-68 well now sleep-INF NEG.EQU.PER-PQ morning early go be.needed-INF mèmbo? NEG.EQU.NE.Q 'Well, now we shall go to sleep, shan't we? We need to go early in the morning, don't we?' (Richhi 67) t'a nàtea? pí:-po gompo cù-ce mèn-a. now 1PL two-COL leave ask-INF NEG.EOU.PER-PO 'The two of us are taking a leave now, aren't we?' (Richhi 28) (11.20) a) हा जाता है जाता ग? k^hu ámdzi mèŋ-ga? doctor NEG.EOU.PER-PO 3SGM 'Isn't he a doctor (assuming he is)?' (KN e) b) अन्म कें रेम प्रधुन में अव ग? mèŋ-ga? dã: ts^herin $p^{h}e$ -po meet-2INF NEG.EQU.PER-PQ vesterday PN 'You met Tshering yesterday, didn't you?' (KN e)

In addition to interrogative uses, $p\dot{a}$ also occurs as a frequently used tag in declarative and imperative clauses. This tag, which has likely developed from a tag question that has lost its interrogative force, adds assertive force to a statement or a request/command. Declarative instances of $p\dot{a}$ are postposed to the equative and existential copulas (11.21b), whereas interrogative $p\dot{a}$ can only replace an equative copula (11.21a). The clause given as A2 (11.21c) is infelicitous as an answer (i.e. a declarative clause), because this construction with $p\dot{a}$ is by definition a question. The declarative uses of $p\dot{a}$, which are here termed assertive tags, are treated in more detail in §16.3.1.

(11.21) a) Q: المجمع: المحمد المجمع: المحمد المحمد المحمد المحمد المحمد (11.21) a) Q: المحمد المح

- b) A1: ה مَنْسَا بِعَامَاتِهُ هُمَ جَا nà jó? kjap-to í: **pá**. 1SG work do-IPFV EQU.PER TAG.ASR 'I am indeed working.' (KN e)
- c) A2: *¬ എഫ്എ நூலார் ஒ? *ŋà jó? kjap-to **pá**? 1SG work do -IPFV EQU.PER.Q

Although $-ka/ga^{376}$ is a suffix and $n\dot{a}$ a copula, they have partly overlapping distributions. Distributions overlap when ka/ga and $n\dot{a}$ are postposed to a verb root, imperfective marker -to/do, future-marking infinitive $-c\varepsilon$ and marginally to the past marker $-tc\varepsilon$, of which I have only one interrogative example with $n\dot{a}$ and several with -ka/ga. Overlapping uses of -ka/ga and $n\dot{a}$ are here described first and distinctive uses after that.

Verb root

When -ka/ga attaches to a bare affirmative verb stem, the action refers to immediate future, as in (11.22a), or is a general fact, as in (11.22b).

(11.22) a) הַ מְשָּיחַ? אָמ **gju-ga**?³⁷⁷ ISG go-PQ 'Shall I go?' b) מָהְ' מוּשִי מָהָ' שִׁישִׁייִדָּ' אֵׁקִשִּיחַ? di làm=di siliguri lɛp-ka? this road=DEMPH TPN reach-PQ 'Does this road lead to Siliguri?' (KN e)

Negated verb stems in polar questions with -ka/ga are future-oriented (or general facts) when negated by the imperfective negator mi-, see (11.23), and past-oriented when negated by the perfective negator ma-, see (11.24).

(11.23) कॅन्'ग्रीस' रूप'क्षुत्र'र्थे' क्रे'भय'ग्?

tc^hø:=ki rabdɛn=lo mi-làp-ka? 2SG.L=AGT PN=DAT NEG-say-PQ 'Aren't you telling (it) to Rabden?' (KN e)

³⁷⁶ A cognate of -ka/ga occurs in Standard Tibetan as a future interrogative marker (Tournadre & Dorje 2003: 170).

³⁷⁷ Another consultant preferred the perfective form of the suppletive verb here, $\eta \dot{a} s \tilde{o}$:- $ga \subseteq \check{a} \subseteq \check{a} \subseteq \check{a}$?

The interrogative copula $p\dot{a}$ may sometimes attach directly to the verb root, as shown in (11.25). In these cases, the corresponding declarative also has a copula, see (11.25b). Consultant KN commented that -ka/ga instead of $p\dot{a}$ would not be acceptable in (11.25a), although -ka/ga was acceptable in (11.22) above. The difference may be caused by lexical semantics of the verbs.

(11.25) a) 弐可: 西ふね: う?
sùk kjap pá?
pain do EQU.PER.Q
'Does it hurt?' (KN e)
b) 弐可: 西ふね:ふつ
b) 弐可: 西ふね:ふつ
sùk kjap=(p)ɛ?
pain do=EQU.NE

'It hurts.' (KN e)

Both -ka/ga and pa may occur with deontic modality marker te^{ho2} 'be allowed', see (11.26) and (11.27). I am not aware of any semantic difference between the two clauses. The equative interrogative copula *bo* also occurs in analogous contexts, following te^{ho2} 'be allowed'.

- (11.26) $\neg \cdot \neg \cdot \neg \cdot \overleftarrow{e} \neg \cdot \overleftarrow{e} \neg \cdot \overrightarrow{g}$? $\eta \dot{a} \quad n \dot{a}: \quad d \sigma: \quad t c^h o: \qquad p \dot{a}$? 1SG here sit be.allowed EQU.PER.Q 'Can I sit here?' (GB e)

Similarly to -ka/ga in (11.23), pa may be postposed to a negated verb stem, see (11.28) and (11.29). Both examples feature the imperfective negator mi-, because ma- does not occur in my data in this context.

(11.28) هَتْحَ: جَرَقَ بَيَّاتَ مَعَرَمِقَ جَ؟ ts^hø? dordziliŋ miŋ-gju pá? 2SG.L TPN NEG-go EQU.PER.Q 'Are you not going to Darjeeling?' (KN e)

di làm=di siliguri mi-lɛp pá? this road=DEMPH TPN NEG-arrive EQU.PER.Q Doesn't this road lead to Siliguri? (KN e)

Imperfective

Both -ka/ga and ná occur postposed to the imperfective marker -to/do.

(11.30) भुष्त'यहँत' लेख' दर्धेत' तु' गु? dzã:dar lèm t^høn-do-ga? training good happen-IPFV-PQ 'Is the training turning out good.' (Richhi 69)

(11.31) Q: גַיאָי גֿעאידי פ?

t'ato ts^ho:-to pá? now gather-IPFV EQU.PER.Q 'Are (they) gathering now?' (KN kitchen discussion)

A: لَاسْمِعْنَى اللَّهُ اللَّانِ عَلَى اللَّهُ اللَّهُ عَلَى اللَّعْظَمَ عَلَى اللَّهُ عَلَى اللَّهُ عَلَى اللَّهُ عَلَى اللَّهُ عَلَى اللَّعَ عَلَى اللَّهُ عَلَى اللَّعَلَى اللَّهُ عَلَى اللَّعَلَى اللَّهُ عَلَى اللَّهُ عَلَى اللَّهُ عَلَى اللَّهُ عَلَى اللَّهُ عَلَى اللَّالَ عَلَى اللَّهُ عَلَى اللَّهُ عَلَى اللَّهُ عَلَى اللَّهُ عَلَى اللَّهُ عَلَى الْحَلَى اللَّهُ عَلَى الْحَالَى الْحَالَةُ عَلَى الْحَالَى الْحَالَةُ عَلَى الْحَالَى الْحَالَى الْحَالَى الْحَالَى الْحَالَى الْحَالَى الْحَالَى الْحَالَى الْحَالَى الْحَالِ عَلَى الْحَالَى الْحَالَةُ عَلَى الْحَالَةُ عَلَى الْحَالَى الْحَالَةُ عَلَى الْحَالَةُ عَلَى الْحَالَى الْحَالِ عَلَى اللَّهُ عَلَى الْحَالَةُ عَلَى الْحَالَةُ عَلَى الْحَالَةُ عَلَى الْحَالَةُ عَلَى الْحَالَةُ عَلَى الْحَالَةُ عَلَى الْحَالَى الْحَالَةُ عَلَى الْحَالَى الْحَالَ عَلَى الْحَالَةُ عَلَى الْحَالَ عَلَى الْحَالَةُ عَلَى الْحَالَةُ عَلَىهُ عَلَىهُ عَلَى الْحَالَ عَلَى الْحَالَى الْحَا

Example (11.32) below provides a further comparative example, showing that there is a semantic differences in the choice between $p\dot{a}$ and -ka/ga in some contexts, although this is the only contrastive example I have to show as evidence.

(11.32) a) المجتربة المحربة ا

b) אָקישָׁאי אָרִיאָשי (football) אָרִיאָ יוּמְרִיאָ'יוּ? *lɛŋgɛ? futbol tsi:-bo nắ:-do-ga?* PRN.HON football(Eng.) play-2INF do.HON-IPFV-PQ 'Do you (habitually) play football?' (TB e)

Past

Both -ka/ga and $p\dot{a}$ occur with the past marker $-tc\varepsilon$ in my data, although there are several examples of -ka/ga, two of which are given in (11.33-34), but only one elicited example of $p\dot{a}$ in this construction (11.35). In colloquial language, *VERB-tc\varepsilon-ka* often merges into *VERB-tc\varepsilon*, see (11.34).

 $^{^{378}}$ The Denjongke term for 'football' given by consultant YR is $\pi \subset \mathfrak{F}^{\mathfrak{C}}$ is $i \in \mathcal{K}^{\mathfrak{C}}$.

(11.34) ८ रेंगायों के तेन (meeting) त' पत्वायां ?

t'atci=ki mi:tin=na zu:-tca? recent=GEN meeting(Eng.)=LOC sit.HON-PST.PQ 'Did you sit in the recent meeting?' (oh, Barapathing)

(11.35) অস্ট ন্ত?

làp-tee ná? say-PST EQU.PER.Q 'Did you/(s)he/they say (it)? (KN e)

One of the reasons why the use of $p\dot{a}$ in (11.35) may be marginal is that the past marker *-tce* in the declarative cannot take a copular auxiliary, unlike the imperfective *-to/do* (\tilde{t} :/*be*?) and the periphrastic past *-po* \tilde{t} :/*be*?. Consequently, it is not clear whether $p\dot{a}$ in clauses such as (11.35) is interrogative or declarative (at least in the segmental level). In imperfective and periphrastic past clauses, on the other hand, the occurrence of the copula auxiliary disambiguates between interrogative ($p\dot{a}$ replaces copula) and declarative uses ($p\dot{a}$ occurs after copula).

Nonpast

Both -ka/ga and pa' may interrogate the periphrastic nonpast form VERB- $\epsilon\epsilon$ *i*:/ $b\epsilon$?, which marks both habitual present facts and future actions, see (11.36-38). Similarly to the past construction $-t\epsilon\epsilon$ -ka, which is abbreviated to $-t\epsilon a$, the future $-\epsilon\epsilon$ -ka is often reduced to $-\epsilon a$ in spoken language, see (11.37).

- (11.36) גֿיקק־יָּשָאַ צֿקעאַ צָּק־יַשְ? moby:=ki p^ho? t^hop-cɛ-ka?³⁷⁹ wife=AGT salary receive-INF-PQ 'Does the wife receive salary?' (BP BB discussion)
- (11.37) 豪云和· 劳· 天· 豪云· 丹?380

tsim tɛ:ra tsi:-ca? game again play-INF.PQ 'Shall we play a game again?' (PT kitchen discussion)

k^hu jó? **p'ja-εε μά**? 3SGM work do-INF EQU.PER.Q 'Will he work?' (TB e)

³⁷⁹One consultant wanted to correct this question into $m \partial by := ki p^h o^2 t^h o p - \varepsilon \varepsilon j \dot{\phi} : -ka^2$.

³⁸⁰ r is an innovative Denjongke spelling, which represents the merged pronunciation $-c\epsilon -ka > -ca$.

Distinctive uses

The examples above illustrated the use of -ka/ga and $p\dot{a}$ in identical environments. What follows describes the distinctive uses of the two markers, beginning with copular questions. The fact that the interrogative equative copula $p\dot{a}$ simply replaces the corresponding declarative copula was already illustrated in (11.6) above. Being an equative, $p\dot{a}$ is not used for interrogating the existentials copulas $j\dot{a}$? and du?. This leaves two copula forms interrogated with the suffix -ka/ga, $i\eta$ -ga (neg. $m\dot{e}\eta$ -ga) and $b\varepsilon$ -ka (neg. $m\dot{e}mb\varepsilon$ -ka), which roughly correspond to the equative interrogative copula $p\dot{a}$. The form $i\eta$ -ga, however, has developed uses that set it semantically apart from $p\dot{a}$. The interrogative $i\eta$ -ga is mainly used as an often-heard tag question, see (11.39), but it also occurs as an exclamation upon hearing new information, see (11.40), and as an ordinary (non-tag) question, see (11.41). By using $i\eta$ -ga as a non-tag question the speaker is predisposed to believe that the questioned claim is true.

- (11.40) אי אאָז ייז יאיזאָז? ג:, **iŋ-ga**=la. oh EQU.PER-PQ=HON 'Oh, is it so?' (KNA kitchen discussion)
- (11.41) कॅन् र्रेन सुग केव ग?

tc^hφ? *lópt^hu*? *iŋ-ga*? 2SG.L student EQU.PER-Q 'Are you a student (I think you are)?' (YR e)

By using the evidentially neutral equative $b\epsilon^2$ in a polar question, on the other hand, the speaker does not reveal their preconceptions about the answer, see (11.42), contrasting with (11.41).

(11.42) \widetilde{a}_{∇} \widetilde

The semantic difference of $i\eta$ -ga in (11.41) and $b\varepsilon$ -ka in (11.42) is particularly noteworthy, because it reflects a difference in the speaker's own epistemic stance towards the proposition, 'I think it is the case' for $i\eta$ -ga and 'I do not know' for $b\varepsilon$ -ka. In descriptions of other Tibetic languages, the speaker's choice of copula in questions is usually determined not by the speaker's own beliefs about the truth value of the statement but by what copula the speaker anticipates the addressee to use in their answer based on the addressee's own knowledge, see Tournadre's (2008: 296, 300) rule of anticipation. The rule of anticipation is more prominent with existential interrogatives $j\partial$:-ka and du-ka, as will be pointed out a few paragraphs below.

Example (11.43), where $p\dot{a}$ is used instead of $b\varepsilon$ -ka, is very close in meaning to (11.42). I hypothesize that the difference here can be understood in terms of the rule of anticipation: in (11.43)

the anticipated answer has the personal equative \tilde{i} : (focusing on the identification), whereas in (11.42) the anticipated answer has the neutral equative $b\epsilon^2$ (focusing on the consequences of identification). For the difference of \tilde{i} : and $b\epsilon^2$, refer to §7.2.3.

The interrogative $b\varepsilon$ -ka is also used as a tag question (but not as often as $i\eta$ -ga), as shown in (11.44).

(11.44) לישרי אלרי אריארי ריסאי איארי אלי אדי אלי איא אואנא איאנא t'iza: $k^h \tilde{o}$:=diŋàtci k^hon=gi ran-ke? loke=di 3PL=DEMPH own-language 1PL.GEN Lhoke=DEMPH but 3PL=AGT mi-k^hɛm-bo $b\varepsilon = la$, mi-khen-khen NEG-know.HON-2INF NEG-know.HON. EQU.NE=HON *bε?*. **b***ɛ***-***k***a**=*l*a? EQU.NE EQU.NE-PQ=HON 'But they don't know our own language Lhoke, don't know, isn't it (so)? (YR canteen video)

It also occurs with nominalized copulas in ordinary non-tag questions, as shown in (11.45), where $j\hat{e}bb\varepsilon$ -ka is a reduction of $j\hat{\phi}$ -po $b\varepsilon$ -ka:

The interrogator $b\varepsilon$ -ka also occurs as an auxiliary. Consequently, although -ka/ga cannot directly attach to nominalized verbs like the interrogative copula $p\dot{a}$ can, -ka/ga may interrogate the periphrastic past construction VERB-2INF EQU by attaching to the final copula, see (11.46). Consultant KN reported that (11.46) may either be a question or an exclamation (for more on the use of $b\varepsilon$? in exclamation, see §11.2.2), without difference in intonation. It appears that this constructions for interrogating the periphrastic past is rather infrequent, construction with $p\dot{a}$ being more frequent (11.47).

(11.46) 展 茶云茶 聖云河?

 $k^h u$ ∂m -bo $b \varepsilon$ -ka?3SGMcome-2INFEQU.NE-PQ'Did he come?'/'He came, eh. (KN e)

(11.47) $\mathbb{R}^{\cdot} \check{\alpha} \subset \check{\nabla}^{\cdot} \mathfrak{F}^{\circ}$? $k^{h}u \quad \partial m - bo \quad p\acute{a}$? 3SGM come-2INF EQU.PER.Q 'Did he come?' (KN e)

Only -ka/ga (and not $n\dot{a}^{381}$) occurs as an interrogator with existential copulas $j\dot{a}$? and du?, forming $j\dot{a}$:-ka and du-ka, as exemplified in (11.48) and (11.49). The choice of copula between $j\dot{a}$? and du? in existential questions depends on whether the speaker thinks the addressee has personal information of the questioned fact or not. When addressee's personal information is presupposed and hence anticipated in the answer, the copula in the question is $j\dot{a}$?, otherwise du?.

(11.48) $\in \check{u} \subseteq \check{\eta}$? $te'a j \check{\theta}:-ka$? tea EX.PER-PQ 'Is there tea?' (oh, Barapathing)

(11.49) $\forall g: \pi : ag: g \pi : \pi : ag:$

My data includes one elicited counter-example to considering -ka/ga as an exclusively polar question marker. In (11.50), -ka/ga occurs in a content question, although the interrogative can also be formed by omitting -ka/ga.

(11.50) ผู สุล สันร์ รู ๆ?

 $k^{h}u$ nàm òn-do-ga? 3SGM when come-IPVF-PQ 'When is he coming?' (KN e)

Two further constructions in which $p\dot{a}$ and -ka/ga have differing distributions are the periphrastic past construction $-po b\epsilon/\dot{t}$, which is negated in differing ways by $p\dot{a}$ and -ka/ga, as already shown in (11.46-47) above, and the perfect $-ts^ha(c)$, which in my data is interrogated with $p\dot{a}$ but not with -ka/ga. Example (11.51) was used in a telephone conversation not many minutes after I had initially concluded, upon elicitation, that such a form does not exist. While (11.51) presents the form that was actually used in the conversation, (11.52) and (11.53) were reported as variants which are semantically roughly equivalent (the same variation occurs in the declarative, see §8.1.2).

(11.51) के ले गु रे क्वेप्र के जु?

siliguri lɛp-tsʰa: µá? TPN arrive-CMPL EQU.PER.Q 'Have (you) arrived in Siliguri?' (KT phone call 3)

³⁸¹ However, declarative $p\dot{a}$ (see §16.3.1) may be postposed to existentials to form $j\dot{a}$: $p\dot{a}$ and du: $p\dot{a}$.

(11.52) สิ[.] ลิ. ซูาร์: ฐิวสุพ.ฮ์. ที่ 3?
 siliguri lep-ts^hake pá?
 TPN arrive-CMPL.APH EQU.PER.Q
 'Have (you) arrived in Siliguri?' (KT e)

(11.53) के' मे' मु' मे' क्रेनक' क्रेंदु' द?

siliguri lɛp-tsʰo-u pá? TPN arrive-CMPL-2INF EQU.PER.Q 'Have (you) arrived in Siliguri?' (KT e)

Finally, I have one elicited example where $p\dot{a}$ co-occurs with the nominalized existential *duk-o*, which could, with rising intonation, form a question by itself without $p\dot{a}$, see (11.54). It may be that here $p\dot{a}$ is used as an assertive tag rather than as a question marker (see §16.3.1 for uses of $p\dot{a}$ as an assertive tag).

(11.54) हुः क्षेंनुः दर्त्त्यार्गेः दु?

 $k^h u$ óna duk-o pá? 3SGM there EQU.SEN-2INF EQU.PER.Q 'Is he there (I think he is)?' / 'Is he indeed there?' (KT e)

In conclusion, $p\dot{a}$ is a polar equative interrogative copula which historically originated as an interrogated copula *in-na* and synchronically forms a question by replacing the corresponding declarative copula (which may function as an auxiliary). The polar question suffix *-ka/ga*, on the other hand, is a non-copular question marker which attaches to copulas but still has an overlapping distribution with $p\dot{a}$ when occurring with verb roots, the imperfective *-to/do*, nonpast *-ce* tbc? and (marginally) past form *-tce*.

11.1.1.3 Simple polar questions with the attenuated markers -kam/gam and pám

The attenuated interrogative markers -kam/gam and pam mainly occur in alternative questions but occasionally also in simple polar questions, as illustrated for <math>-kam/gam in (11.55) and (11.56). The only verbal suffix that -kam/gam attaches to in my data is the imperfective marker -to/do.

k^h*epar átsi jð:-to-gam*? difference a.bit EQU.PER-IPFV-ATTQ 'Is there perhaps a bit difference?' (KN e)

The use of $p\dot{a}m$ in a polar question is illustrated in (11.57).

(11.57) तन्त्रा तन्त्रा मेरावेषायाय भगषा पवेषा में भुग?

 $d\varepsilon m \ d\varepsilon m \ mi-ze:-k^{h}\widetilde{\varepsilon}:=la.$ ze:-to $p\acute{a}m?$ such such NEG-drink.HON-NMLZ=HON have.HON-IPFV EQU.ATTQ 'He doesn't drink such and such things. (Or) does he perhaps drink?' (SM kitchen discussion)

Because (11.57) is structurally not a typical alternative interrogative, it is here grouped among simple polar questions. Semantically, however, (11.57) resembles an alternative question in that the question is preceded by negated speculation about the state of affairs. For more typical uses where -kam/gam and pam form polar alternative questions, refer to §11.1.3.1.

The negated form of *nám*, *mènam*, similarly to negated interrogatives in general, forms a leading question in which the speaker presumes the affirmative proposition to be true, see (11.58) and (11.59). The gloss 'I ask' in brackets in (11.58) attempts to transfer the meaning of the final quotative.

(11.58) ८ केंग्रेंग मुंबर दर्वाया मन्तर अव वया पाववर गा दशु केंग्रे

t'ats'o: $ts'\phi n$ go:-sem in-am? $z \in n$ kanow son.in.lawgo.HONbe.needed-INFNEG.EQU.PER-ATTQanotherwho $gju-ni=s(\varepsilon)$?go-3INF =QUO'Now the son-in-law should go, shouldn't he? Who else is to go (I ask)?' (rnam-rtog 30)

(11.59) รา ผสุดาสั, กลารัสา ครัราฮิรารักษ์ ซิญาภูา ตักคริมา สราชิร สัา ๆธรายรายารา มาย์พาสา รุญายุรา ผสาสุม]
 t'a nám, p'embõ: boŋt^hiŋbo=lo³⁸² teiku ódem sấ:te lóte:-ee
 now daughter-in-law Bon.priest Lepcha.priest=DAT only such until trust-INF
 mam-bja-ne **['ak-ee mèn-am**?
 NEG-do-COND be.alright-INF NEG.EQU.PER-ATTQ
 'Now, daughter-in-law, it would be better not to put your trust only in Bon priests and bongthings, wouldn't it?' (rnam-rtog 32)

The negated question menam may be followed by the affirmative tag question *inga*, see (11.60). Interestingly, Bhaicung Tsichudarpo, the author of the play *rnam-rtog*, from which examples (11.58-60) are taken, uses the question mark only after examples such as (11.60), which have a tag question, but not with tagless (11.58) and (11.59).

(11.60)	5.g. a	षाःगॉ्'	तिः होटार्थे मु	781.95.	यव्य वया क्षेव	('শ্ব?		
	t'ato	làko	di:	teŋlo	кјар-с е	mèn-am		íŋ-ga?
	now	hand	this.GEN	on	do-INF	NEG.EQU	.PER-ATTQ	EQU.PER-PQ
	'Now	/ it's d	lone on th	is har	nd, isn't i	t, or wha	t?' (rnam-rtog	27)

³⁸² WD ກສຸກັສ[,] ban-bon refers to 'Buddhist and Bon priests' while WD ກັສ-ກິສສ-(ສັງ' 'bong-thing(-bo) denotes a 'male ritual specialist of the Lepchas' (Balikci 2008: 378)

11.1.1.4 Polar questions with the interrogative copula bo

The third interrogative copula, apart from $p\dot{a}$ (neg. $m\dot{e}$ -na) and $p\dot{a}m$ (neg. $m\dot{e}$ -nam), is bo (neg. $m\dot{e}mbo$), which, like $p\dot{a}m$, occurs both in polar and content questions. For examples in polar questions, consider (11.61) for affirmative and (11.62) for negated constructions:

(11.61) a) ぶ あち あち うす ざ のうち ざ ?
lò ts^hame? tc 'ɛm-bo nấ:-do bo?
year every come.HON-2INF do.HON-IPFV EQU.NE.Q
'Do you come (here) every year?' (Bumchu-video)
b) ਤੁਗ ਰੁਤਾ ਰੁਤਾ ਹੋ ਤਾ ਹੈ
b) ਤੁਗ do-2INF EQU.NE.Q

The negated form $m \epsilon m bo$ is in effect similar to English negated tag question following an affirmed clause, i.e. the speaker suspects that the affirmative is the case.

(11.62) a) สักา สักา ซักา พักกา พลาร์?

'Did it hurt?' (NB e)

 $k^{h}\tilde{o}: \quad \dot{\tilde{o}}: \quad d\sigma: \quad j\dot{\sigma}:-p \qquad m\dot{\epsilon}mbo, \qquad p'otso=tsu?$ 3PL come stay EX.PER-2INF NEG.EQU.NE.Q child=PL 'They have come, haven't they, the children?' (LT kitchen discussion)

b) รับรารับเลขา รับเลขา รับเลขารับ?
 t'o:pa t'ole gju go:-ce mèmbo?
 morning early go be.needed-INF NEG.EQU.NE.Q
 '(We) have to leave early in the morning, don't we?' (TB phone call 2)

While the negated equative $m \dot{\epsilon} \eta$ -ga is used alongside the more frequent specific negated equative $m \dot{\epsilon} n$ -a (at least by one of my consultants), I do not currently have examples of evidentially neutral $m \dot{\epsilon} m b \varepsilon$ -ka, which would be a form analogous to personal $m \dot{\epsilon} \eta$ -ga.

11.1.1.5 Polar interrogatives with preverbal *á* (Lachung)

The polar interrogative \dot{a} (or \dot{a} -), which does not occur in the southern and western varieties of Denjongke is used at least in the village of Lachung in North Sikkim. This formative is placed before the verb.

(11.63) कॅंग्वन्देयार्फे का क्षेत्र?

(11.64) R' & ara?

 $k^{h}u$ ód ε : **á** làp? 3SGM like.that PQ say 'Does he say like that?' (KUN e) Yukawa (2017: 191, 194) reports that a similar interrogative marker a (WD \mathfrak{sr}) in Lhasa Tibetan may precede yöö (cognate with $j\partial$?) and yin (cognate with \tilde{t} :) but not ree (functionally quite similar to $b\epsilon$?). Yukawa's translation for questions with a includes the frame 'I wonder whether'. A similar question marker a^{53} is also reported for Dongwang Tibetan (Bartee 2007: 412). Moreover, a preverbal vocalic element for polar questions (prefix *i*-) is found in the Tibetan variety spoken in Bragg.yab (Schwieger 1989: 50).

11.1.1.6 Summary on polar questions

In conclusion, polar questions can be formed by four interrogative morphemes (excluding \dot{a} , which is specific to Lachung) and also by mere intonation. These five options are illustrated in (11.65). Three of the interrogative morphemes, $p\dot{a}$, $p\dot{a}m$ and bo, are interrogative copulas, which may occur in a syntactically identical environment (see a-c). The interrogative marker -ka/ga, on the other hand, attaches to the copula (d). In the last example (e), the interrogative is formed by mere raised intonation on the last copula, which in a declarative would have a descending pitch.

(11.65) a) हु' गुलेब' मुनय'र्थ' ज़?

	k ^h u	pén	kjap-o	ná?	
	3sgm	wedding	do-2inf	EQU.PER	2.Q
	'Did he	get married	?' (KN e)		
b)	ातुः यन्त्रि न ् ह	મુગ્રષ'ર્યે' સ્નુસ?			
	$k^h u$	pén	kjap-o	pám?	
	3sgm	wedding	do-2inf	EQU.AT	ſQ
	'I wond	ler if he got	married?' (H	KN e)	
c)	ातुः यन्त्रि न ् ह	দ্রবন্ধ'র্য' র্ব?			
	$k^h u$	nén	kjap-o	bo?	
	3sgm	wedding	do-2INF	EQU.NE.	Q
	'Did he	get married	?' (KN e)		
d)	ম্রি' শবিষ' হু	দ্র্বিষার্থি স্থ্রদৃ'শ্।			
	k ^h u	pén	kjap-o	bɛ-ka?	
	3sgm	wedding	do-2inf	EQU.NE-	PQ
	'Did he	get married	?' (KN e)	-	-
		C			
e)	रषुः गनिनः ह	મુગ્રષ'ર્યે' સ્વત્?			
	k ^h u	pén	kjap-o	bɛ??	(rising intonation on final copula)
	3sgm	wedding	do-2inf	EQU.NE	
	'Did he	get married	?' (KN e)		

At this stage, the question whether there is an evidential difference between (11.65a) and (11.65c) and whether the speaker anticipates the use of a different copulas in the answers to these two questions (\tilde{i} : for $p\dot{a}$ and $b\epsilon^2$ for bo) has to be left open for further research.

11.1.2 Question words and content questions

Content questions are expressed by placing a question word in the clause. Therefore content questions may be formed without other interrogative markers than the question word, as illustrated by the declarative copula in the interrogative clause (11.66a), or with additional interrogative markers such as *nám* in (11.66b) and *bo* in (11.66c).

(11.66) a) तर्ने गव भून? di k'an b ϵ ?? this what EQU.NE 'What is this?' (KN e) b) तर्दे गव हुस? di k'an pám this what EQU.ATTQ 'I wonder what this is?' (KN e) c) तर्दे गुरु में? di k'an bo? this what EQU.NE.Q 'What is this?' (KN e)

The discussion on content questions is divided into question words (§11.1.2.1), content questions without final question markers (\$11.1.2.2), content questions with -kam/gam and pám (\$11.1.2.3) and content questions with bo (\$11.1.2.4).

11.1.2.1 Question words

Table 11.2 lists Denjonke question words, which are illustrated in sentential context after the table.

k'an, k'ar	गव, गर.	what
k'adi	ما رولي ا	which
k'ana, k'ak ^h a	যা'ব', যা'ম(ম)'	where
ka	या'	who
nàm	ववा.	when
k'ambja ³⁸³	শাব, র্হার্ম.	why
k'adzø?, k'ats ^h ø?	या केंट्र	how many
k'atem	या'झेव्र'	what kind of
k'ate	या झे	how

Table	11.2.	Question	words

(11.67) गुत्र गुत्र रॉट्? **k'an** nấ:-do? what do.HON-IPFV 'What (are you) doing?' (TB e)

³⁸³ Also occurs in longer form *kan p'ja(ti)* and in more reduced forms *k'amja* and *k'ame*.

- (11.68) $\mathbb{R}^{*} \P^{*} \mathfrak{A}^{+} \mathfrak{A}^{*}$? $k^{h}u$ $k^{*}adi$ bo? 3SGM which EQU.NE.Q 'Which one is he?' (TB e)
- (11.69) युवा दर्ने गाव हुव?

ý:=di k'ana nám? place=DEMPH where EQU.ATTQ 'Where is that place, I wonder?' (AB kitchen discussion)

- (11.71) גערק אָזי דֿיָשַקאַ? אָזי קאָזי קישאַ? *odi* **nàm** bo=la? lò nì: halɛ? that when EQU.NE.Q=HON year two before 'When was that? Two years ago?' (RS [in]auspicious days)
- (11.72) गव्र र्मुब दर्न में हे के दर्म ?

k'ambja $d\epsilon p$ $p^{h_i:-ti}$ ∂m -bo?whylike.that be.late-NFcome-2INF'Why did you come late like that?' (RS pupil joke)

(11.73) ५.३८ वेश गार्केट र्झेट मट?

t'aruŋ nìm k'adzo? dø:-ɛɛ?? still day how.many stay-INF 'How many days will (you) still stay (here)?' (oh, Tashiding)

The answer to $k'at\varepsilon m$ 'what kind' tends to be an adjective, whereas the answer to $k'at\varepsilon$ 'how' tends to be an adverb. The adverbial nature of $k'at\varepsilon$ 'how' is revealed by the indefinite form $k'at\varepsilon$ p'jati *î:ruŋ* 'however', which in opposition to $k'at\varepsilon m$ *î:ruŋ* 'whatever kind', includes the adverbializing converb p'jati (see §6.3.2).

(11.74) ริเลก สังกรร์ พิมาอีสา สงคลาร์ ขาลลาล์ ขาลิ สาร์สา สู่ขาลาลร์

t'izã: $m\hat{u}=r\hat{a}:=to$ *lèm-p'ja ma-cé:-po.* zen=lo *k'ate lap* but 3SGF=REFL=CEMPH good-ADVZR NEG-know-2INF other=DAT how teach *p'in* $ts^hu:-ce?$? give be.able.to-INF 'But she herself does not know (it) well. How to be able to teach other(s)?' (Richhi 65) (11.75) a) वट्रायार्थे वयायया याक्षेय प्रवेश प्रहेया हे?

cè:la? k'atem tcuk-to? nè:po=lo ze: patient=DAT food.HON what.kind.of eat cause-IPFV 'What kind of food is the patient being fed?' (rnam-rtog 23)

b) गाः क्षेत्राः गत्रित्राः गत्र नार्यः? nấ:-bo? k'atem zim-bo do.HON-2INF what.kind.of sleep.HON-2INF 'How did you sleep?' (TB e)

The uses of *k'atem* 'what kind' and *k'ate* 'how' overlap when asking how people are doing:

(11.76) a) ८ रु. इ. दि. यापाया पा होया थें ८ यापाय? t'ato bhaila k'atem *jà*:=*la*? now PN what.kind.of EX.PER=HON 'How is Bhaila now?' (Richhi 10)

> t'ato bhaila k'ate j*à?*? how EX.PER now PN 'How is Bhaila now?' (Richhi 25)

The word *k'ama* is used, at least in riddles, similarly to *k'an bo/mo* 'What is it?'. In addition to the above questions words, the form k'a' is used as a more general, contextually decipherable question word with a meaning covering 'what', 'where' and 'why', see (11.77) and (11.78).

(11.77) केंद्र ग द्यु र्द्र शे

tc^hø? k'a: gju-do=s. 2SG.L go-IPFV=OUO where 'Where are you going (he said)?' (KT animal story)

dep tc^ha:p kjap-ne gari k'a: like.that rain strike-COND why/what wash car

'If it's raining like that why wash a car?' (oh, Lachen)

11.1.2.2 Content questions without final question markers

As shown above, the question markers *pám*, *-kam/gam* and *bo* occur both in polar questions and in content questions. Content questions, however, can also be formed without these final question markers. With existential copulas $j\partial^2$ and du^2 , content questions usually have the copula nominalized with -po:

 $k^{h}ju?$

- (11.79) गान थॅन रॉ? k'ana jè-po? where **EX.PER-2INF** 'Where is (he)? (SG wedding customs)
- (11.80) र्ट्यट्रायेंग् या क्षेत्रा तट्या में?

ts^hepo k'atem duk-o? heat how EX.SEN-2INF 'How hot is it?' (TB phone call)

Bare copulas are also sometimes used:

- (11.81) झुः झुन् गवरा रहेना गवरा येन? laki? né:ts^hy: k'an j*à*?? PN news what EQU.PER 'What news are there, Lhaki?' (Richhi 69)
- (11.82) अपमुगी ट्रें पार्केंट पर्या? k'adzø? ágja=gi=co du??elder.brother=GEN=AT how.many EX.SEN 'And as for the big brother, how many do (you) have?' (Richhi 99)

Some Denjongke-speaking communities also allow *-po* to be appended to the neutral equative be?:

(11.83) देन वर्देगी वहें गर्छेंद श्वर्र् $t'\varepsilon p = di = gi$ dzo: k'adzø? $b\varepsilon$ -po? book=DEMPH=GEN price how.many EQU.NE-2INF 'What is the price of this book?' (KT e)

Other communities prefer an assimilated nominalized form, $b\varepsilon$ -go (note that the preferred word for 'price' also changes):

(11.84) جَمَ⁻ مَجَ⁻ بَّابَ ثَابَ⁻ गِ⁻ عَبْحَ⁻ عَبْحَ⁻ ثَابً? t'ɛp=di=gi gõ: k'adzø? be-go? book=DEMPH=GEN price how.many EQU.NE-2INF 'What is the price of this book?' (PT e)

Yet other communities, however, prefer the use of the interrogative copula bo in contexts such as (11.83-84), i.e. the forms $b\varepsilon$ -po and $b\varepsilon$ -go are replaced by mere bo. Interrogative clause is the only context where $b\epsilon^2$ occurs as nominalized with -po in my data. This nominalized use of $b\epsilon^2$ in interrogatives is probably triggered by analogy with the existential nominalized forms $j\dot{\rho}$ -po and du-ko. It is also possible that through constant association with interrogation, the nominalizer/infinitivizer attaching to copulas in questions has been or is being reinterpreted as a question marker.

The following three examples further illustrate content questions which lack a final question marker, see completive (11.85), past (11.86) and periphrastic past³⁸⁴ (11.87):

(11.85) द' र्ड्रेद र्छे गी' देवा या केंट्र यह कर? pìm k'adzø? nà: tɛ'øn-diki là:-ts^ha:? here come.HON-NF day how.many come.up.to-PRF 'How many days is it since you came here?' (oh, Tashiding) (11.86) कॅला या कॅंट्र झॅया डे? $tc^{h}o$? k'adzø? dok-tee? studies how.many read-PST 'How much did (s)he study?' (BP BB discussion) (11.87) हे. र्यून ट्रांब अनुवास पटी गाव वहना रा? te lópøn pámpu=di k'ana dze:-po?

so teacher with=DEMPH where meet.HON-2INF 'Where then did he meet the teacher (=you)?' (BB discussion)

Figure 11.8 produces the pitch trace from (11.87), showing that no clause-final rise in intonation is needed because of the presence of a wh-word that marks the clause as a question.

Figure 11.8. Intonation in content question (with a wh-word) (11.87)



Finally, my data contains one example of a question marker *-go* which occurred in an interrogated progressive sentence (11.88). The interrogative *-go* corresponds to declarative *-k* ϵ /g ϵ , see §8.3.3.2.

(11.88) ८षा गर र्जुषा रावित गॅ?

ná: k'ar p'ja-zuŋ-go? I.AGT what do-PROG-Q 'What am I doing?' (KN e)

I suspect that (11.88) is a spoken abbreviated version of the fuller form (11.89).

³⁸⁴ In the interrogative, however, the form is not periphrastic because the copula is dropped.

(11.89) दबा गर र्जुबा पर्या गें?

yá: k'ar p'ja-zɛn duk-o? I.AGT what do-PROG EX.SEN-2INF 'What am I doing?' (KN e)

Note that the auxiliary used with first person actor in (11.89) is a sensorial, because the speaker anticipates the addressee to answer using the sensorial auxiliary.

11.1.2.3 Content questions with -kam/gam and pám

In addition to polar alternative interrogatives (and marginally simply polar interrogatives), the attenuated markers *-kam/gam* and *pám* occur in content questions. Example (11.90) gives an example of a content interrogative with *pám* in a complement clause:

(11.90) दे चट क्रेंग्रे' धेया व्यत दर्दा वर क्रेंग्रेय में क्रेंग्रेय के क्रिक्य यह क्रिय क्रेंग्रेय क्रेय क्रेय क्रेय क्रेंग्रेय क्रे . ૧૬ જેવા t'izã: mù=i jìglĩ: t'arun=sã: ma-lep-ø: gjumts^he: k'an but 3SGF=GEN letter.answer still=until NEG-arrive-2INF.GEN what reason $ts^{h}o=na$ số:-di nó:-zẽ: nà=i sém=na nám tam think-PROG fish=GEN lake=LOC go.IPFV-NF mind=LOC saying EQU.ATTQ ĩ. ke:po t'en-ce a.lot remember-INF EQU.PER 'Thinking what might be the reason for the letter-answer having not yet arrived he goes to the fish pond and reminisces many things in his mind.' (Richhi 151)

The following two examples illustrate independent (non-complement) content questions with *nám:*

(11.91)	শ'র্ক্টন্'নম' র্ভ্রি'	न्वॉर्श्वन् हे	ोवःववा?		
	k'adzø?	p'a:	tc'øn	g0:-68	pám?
	how.many	interval	come.HON	be.needed-INF	EQU.ATTQ
	'What time shall I come, I wonder?' (Richhi 57)				

Example (11.93) exemplifies the use of -kam/gam in a content question with a question word:

(11.93) ماجها: قَارَحَ: مَا تَعْتَرَ عَلَمَ مَا يَعْتَرَ عَلَمَ مَا يَعْتَرَ عَلَمَ مَا يَعْتَرَ عَلَمَ مَا يَعْتَرُ عَلَيْهُ مَا يَعْتَرُ عَلَيْنَا عَلَيْهُ مَا يَعْتَرُ عَلَيْهُ مَا يَعْتَرُ عَلَيْنَا يَعْتَرُ عَلَيْنَا يَعْتَرُ عَلَيْنَا يَعْتَرُ عَلَيْنَا عَلَيْنَا يَعْتَرُ عَلَيْنَا يَعْتَرُ عَلَيْنَا يَعْتَرُ عَلَيْنَا يَعْتَلُونَا يَعْتَ عَلَيْنَا يَعْتَلُكُ مَا يَعْتَرُ عَلَيْنَا يَعْتَرُ عَلَيْنَا يَعْتَرُ عَلَيْنَا عَلَيْنَا عَلَيْنَا عَلَيْنَا يَعْتَلُكُ عَلَيْنَا يَعْتَلُكُ عَلَيْنَا يَعْتَلُكُ عَلَيْنَا يَعْتَلُكُ عَلَيْنَا يَعْتَلُكُمُ عَلَيْ عَلَيْنَا يَعْتَلُكُ مَا يَعْتَعُ عَلَيْنَا عَلَيْنَا يَعْتَعْتُ عَلَيْ عَلَيْ عَلَيْنَا عَنْتُكُمْ عَلَيْ عَلَيْنَا يَعْتَعْتُ عَلَيْ عَلَيْنَا يَعْتَعْنَا يَعْتَعْتُ عَلَيْ عَلَيْنَا يَعْتَعْتُ عَلَيْنَا يَعْتَعْتُ عَلَيْنَا عَلَيْنَا يَعْتَعْتُ عَلَيْنَا عَلَيْنَا عَلَيْ عَلَيْ عَلَيْ عَلَيْنَا عَلَيْ عَلَيْ عَلَيْ عَلَيْ عَلَيْ عَلَيْكُمْ عَلَيْ عَلَيْ عَلَيْكُمُ مَا يَعْتَ (To what degree is that word true, I wonder?' (KN, CY interview) Figure 11.9 below presents the intonation contour in content question (11.94) with the attenuated copular interrogative $p\dot{a}m$. The stress is on the wh-word, which has a raised pitch. No rise in pitch is observable at the end of the clause.

(11.94) गुत्र र्जुया के मुनया रेंग क्रुवर

k'amp'jati kjap-o pám=la? why do-2INF EQU.ATTQ=HON 'Why did (they) do that?' (KN, CY interview).

Figure 11.9. Intonation in content question (11.94) with pám



11.1.2.4 Content questions with *bo*

The evidentially neutral interrogative equative *bo* alternates with its declarative counterpart $b\epsilon^2$ in content questions. Examples (11.95) and (11.96) further illustrate the overlap of *bo* and $b\epsilon^2$ respectively in analogous complement clauses.

- (11.95) $\vec{r} \cdot \vec{r} \cdot$
- (11.96) مَامَرَ عَلَمَ مَرْدَ عَلَيْكُ مَرْدَ عَلَيْكُ مَرْدَ عَلَيْكُمْ عَلَيْكُمْ عَلَيْكُمْ عَلَيْكُمْ مَرْدَ عَلَيْكُمْ مَرْدَ عَلَيْكُمْ مَرْدَا عَلَيْكُمْ مَرْدَا عَلَيْكُمْ مَرْدَا عَلَيْكُمْ مَرْدَا عَلَيْكُمْ مَرْدَا عَلَيْكُمْ مَرْدَا عَلَيْكُمْ مَرْدَ عَلَيْكُمْ مَا عَلَيْكُمْ عَلَيْكُمْ مَا عَلَيْكُمْ مَا عَلَيْكُمْ مَا عَلَيْكُمْ عَلَيْكُمْ عَلَيْكُمْ مَا عَلَيْكُمْ عَلَيْكُمْ مَا عَلَيْكُمْ عَالْكُمْ مَا عَلَيْكُمْ مَا عَلَيْكُمْ مَا عَلَيْكُمْ مَا عَلَيْكُمْ مَا عَلَيْكُمْ عَلَيْكُمْ عَلَيْكُمْ مَا عَلَيْكُمْ عَلَيْكُمْ عَلَيْكُمْ عَلَيْكُمْ عَلَيْكُمْ عَلَيْكُمْ عَلَيْكُمْ عَلْكُمْ عَلَيْكُمْ عَلَيْكُمْ عَلَيْكُمْ عَلَيْكُمْ عُلَيْكُمْ عَلَيْكُمْ عَلَيْكُمْ عَلَيْكُمْ عَلَيْكُمْ مُ مُعْلَيْكُمْ عَلَيْكُمْ عَلَيْكُمْ عَلَيْكُمْ عَلَيْكُمْ عَلَيْكُمْ عَلَيْكُمْ عَلَيْكُمْ عَلَيْكُمْ عَلَيْكُمْ عُلَيْكُمْ عَلَيْكُمْ عَلَيْكُمْ عَلَيْكُمْ عَلَيْكُمْ عَلَيْكُمْ عَلَيْكُمْ عَلَيْ عَلَيْكُ مُعْتُ عَلَيْكُمْ مُ مَا عَالْكُ مُعْتُ عَلَيْكُمْ عَلَيْ عَلَيْكُ مُعْتُ عَلَيْكُ مَا عَلَيْكُ مُعْتُ عَلَيْ عَالَيْكُ مَا عَا عَا عَا عَا عَلَيْكُ مَا عَا عَلَيْكُ مُعْتُ عَلَيْ مُعْلَيْكُمْ عَلَيْكُمْ عَلَيْكُمْ عَلَيْكُمْ عَلَيْ عَلَيْكُمْ عَلَيْ عَلَيْكُ مُعْتُ عَلَيْكُ مَا عَلَيْكُ عُلْ عَلَيْكُ مَا عَلَيْ عَلَيْكُ عَلَيْكُ مُ عَلَيْكُ عَلَيْ عَاعْتُ عَلَيْ عَاعْتُ عَلَيْ عَا عَا عَا عَا عَا عَاعْتُ عَا

Examples (11.97) and (11.98) exemplify copular and auxiliary uses of bo respectively.

(11.97) ළ·ආ· ති?

 $k^{h}u$ ka **bo**? 3SGM who EQU.NE.Q 'Who is he?' (KN e) (11.98) デ· 末済、資本、 資方、4万、 石? *tɛ ranipul kɛ:=tsa: dø:-ɛɛ bo?*so TPN who.GEN=at sit-INF EQU.NE.Q
'So with whom will (she) stay in Ranipool?' (KN kitchen discussion)

In interrogative nonpast constructions (- $\epsilon\epsilon$ -infinitive followed by an equative copula), *bo* may be elided, as seen in (11.99), where the latter of two almost identical clauses has no copula.

(11.99) न' केंन' ग'हे' हैं' दग' वन 'र्य? केंब' होन' न' केंन' ग'हे' हैं' दग'वन? t€^hø? t'a tc^hø? k'ateti t'ak-ce **bo**? t¢^høki? k'ateti t'a get.well-INF EQU.NE.O PN now 2sg.l how now 2SG.L how *t'ak-ce??* get.well-INF 'Now how will you get well? Choki, now how will you get well?' (Richhi 171)

That *bo* is the interrogative equivalent of the declarative *b* ϵ ? is born out by the fact that *bo*, similarly to *b* ϵ ? (and unlike *n* \dot{a}), occurs in locative uses:

र्दे हिंगी के इव आपवा दरी गावा में अयायते दुषा केंन वटावा (11.100)mì: $s vn - k^h \varepsilon n = di$ òn-diki k'ana làp-ø: bo come-NF fire kindle-NMLZ=DEMPH where EQU.NE.Q say-2INF.GEN t'yts^hø? nàŋca time inside ...she came there and at the time when she said (to herself) where is the one who lighted the fire...' (PAD Tashiding story)

11.1.3 Alternative questions

Alternative questions are formed mainly by the attenuated markers -kam/gam, pam and the interrogative copula *bo* but also with the direct marker -ka/ga (with existential copulas) and the more marginal suffix $-l\phi$?. The interrogative copula pa does not occur in alternative questions in my data.

11.1.3.1 Alternative questions with -kam/gam and pám

Example (11.101) illustrates the use of the attenuated question markers $p\dot{a}m$ and -kam/gam in a context where, having been asked about the whereabouts of a person named Bhaila, the speaker comments:

(11.101) $\exists a_1 = i \quad t^{i} \quad a_2 = i \quad a_1 = i \quad a_2 = i \quad a_1 = i \quad a_2 = i \quad a_1 = i \quad a_2 = i \quad a_2 = i \quad a_1 = i \quad a_2 = i \quad a_2 = i \quad a_1 = i \quad a_2 = i \quad a_1 = i \quad a_2 = i \quad a_2 = i \quad a_2 = i \quad a_1 = i \quad a_2 =$

In (11.101) the connection between the clauses marked by $p\dot{a}m$ and -kam is somewhat looser than in typical alternative questions, as suggested by the use of two different interrogating morphemes. Example (11.102) exemplifies the use of $p\dot{a}m$ in a typical alternative question, where the same interrogating morpheme occurs in both clauses.

(11.102) المجترية المحترية المحترية

Interestingly, (11.102) combines a polar question (first) and a content question (second). The interrogative clauses occurs as a complement clause, i.e. as an argument of another clause ('I do not know...). With complement interrogatives, the whole clause does not function as a question unless the superordinate clause forms a question as well.

In independent interrogative clauses, the use of attenuated question markers instead of the more direct question markers -ka/ga and pa seems to imply more politeness. In complement clauses, on the other hand, -kam/gam and pam appear to completely displace -ka/ga and pa as question markers. Polar questions with the attenuated question markers are usually presented as alternative questions of the type 'is it or is it not' and 'did he or did he not', see (11.103) for an independent use and (11.104) for uses in a complement clause. The glosses in (11.104) do not include 'I wonder' because the distinction between -ka/ga vs. -kam/gam does not occur in complement clauses.

(11.103) a) ५२ ८ र परेवरणवा केवरपरेवरणवा?

di t'a dεη-gam min-dεη-gam? this now be.true-ATTQ NEG-be.true-ATTQ 'Now is that true or not true, I wonder?' (DR discussion with KL)

b) דָרינוֹ מר אָלָרי אָלרי אָאי איאָלרזי אַאי איאַלרזי אַאי

tsøpo=di	mjồ∶-ts¹o-u	рáт	ma-mjồː-p	pám?
debate=DEMPH	finish-CMPL-2INF	ATTQ	NEG-finish-2INF	ATTQ
'Has the debate	ended or not, I wond	ler?' (A	B kitchen discuss	sion)

(11.104) a) क्षेत्र याव्य कर याव्य म. म. के मेया *íŋ-gam mèŋ-ga*

iŋ-gam mèŋ-gam t'a ŋà mi-cé: EQU.PER-ATTQ NEG.EQU-ATTQ now 1SG NEG-know 'Now I don't know whether it is or is not (so).' (KNA kitchen discussion)

b) ८. ५८, र्स. क्षेत्रा थू. क्षेत्रा वीच्याया राख्या क्षे.

t'a $d\varepsilon$: p^{ho} **pám** $m \delta$ **pám** zi: zak-tinow like.that male EQU.ATT female EQU.ATT see.HON set-NF 'Now like that it is seen whether it is a male or a female (and)...' (SGD wedding customs) In alternative questions, the interrogative morpheme at the end of the first alternative has a raised pitch in anticipation of the second alternative. For a pitch trace of (11.103a), consider Figure 11.9.





As seen in the example above, the interrogative copula $p\dot{a}m$ always occurs two times in alternative questions. The interrogative suffix *-kam/gam* (along with the more direct *-ka/ga*), on the other hand, may be repeated, as shown in (11.105), or not repeated, as shown in (11.106).

- (11.105) דָי לְרִישִרִי שִׁיֹתְלְאִי לֵשְׁדִישְׁאִי אָדְיַשְאִי אָדִישְׁאִי אָיָשְׁאוּ t'a t'iŋzã: ódɛm p'iŋ-kʰɛ̃: jòː-kam mèː-kam now nowadays such give-NMLZ EX.PER-ATTQ NEG.EX.PER-ATTQ mi-cé:. NEG-know 'Now I don't know whether there are such givers nowadays.' (LA Intro to Lachung)
- (11.106)धारपा गापातेतु (captain) रावेत्य गात्रपा धार्मा गात्र केन वाया है। क्षें प्रमेश गात्र स्रुवा याहेगा tsa=le nấ:wa j*à:-kam* mè? jà:p kæpten nobleman captain(Eng.) at=ABL permission EQU.PER-ATTQ NEG.EX.PER làp-ti ódɛm k'ajem=tei? say-NF like.that what.is.it=INDF 'Whether (we) have a permission or not from Captain Sir, something like that...' (KN, CY interview)

11.1.3.2 Alternative questions with *bo*

The interrogative copula *bo* may be repeated in alternative questions, as in (11.107), or the second occurrence may be elided, as in (11.108). Note that in (11.108), the sentence-final *-bo* is a nominalizer and not the interrogative copula *bo*.

b) $a\widehat{\uparrow}$, \widehat{g} , $\overleftarrow{\uparrow}$, $w_{i}a_{i}w', \overleftarrow{\uparrow}$? $di \quad k^{h}i \quad bo \qquad diy? \quad bo?$ this dog EQU.NE.Q cat EQU.NE.Q 'Is it a dog or a cat?' (KN e)

The intonation contour of (11.108) is given in Figure 11.10. The stress (rising/higher pitch and intensity) is on information that is being questioned, in the first part on the word $n\dot{a}$: 'here' and in the second part on the syllable $l\ddot{o}$: from $l\ddot{o}$: $s\dot{o}$:-bo.

Figure 11.10. Intonation in alternative question (11.108) with bo



11.1.3.3 Alternative questions with -ka/ga

The polar question marker -ka/ga is used for asking more direct alternative questions than the attenuated marker -kam/gam. The question marker typically occurs only once after the affirmed verb and is not repeated after the negative:

- (11.110) ך جَدَ الْعَظَمَ عَرَبُومَ اللَّهُمَ عَمَرُ اللَّهُ عَلَيْمَ اللَّهُ عَلَيْمَ اللَّهُ عَلَيْمَ اللَّهُ عَلَيْهُ اللَّهُ (11.110) المَحْدَةُ عَلَيْهُ اللَّهُ اللَّهُ عَلَيْهُ اللَّالِي اللَّهُ عَلَيْهُ اللَّالِي اللَّهُ عَلَيْ اللَّالِي اللَّالَ اللَّهُ عَلَيْ اللَّالِي اللَّالِي اللَّهُ عَلَيْ اللَّالَ اللَّالَ اللَّالَ اللَّالَ اللَّالَةُ اللَّالَ اللَّالَةُ عَلَيْ اللَّالَةُ عَلَيْ اللَّالَةُ اللَّا عَلَيْ اللَّالَةُ اللَّالَةُ عَلَيْ اللَّالَةُ عَلَيْ عَلَيْ اللَّالِي اللَّالِي عَلَيْ اللَّالَةُ عَلَيْ اللَّالَةُ اللَّالَةُ عَلَيْ اللَّالَةُ اللَّا عَلَيْ اللَّالِي اللَّالَ اللَّالَ اللَّالَةُ اللَّالِي اللَّالَةُ اللَّالَ اللَّالَةُ اللَّالِي عَلَيْ اللَّا عَلَيْ الْحَالَيْ اللَّالِي الللَّالِي اللَّالِي عَلَيْ الْحَالَيْ عَلَيْ الْحَالَةُ عَلَيْ الْحَالِي عَلَيْ عَلَيْ اللَّهُ عَلَيْ اللَّهُ عَلَيْ الْحَالَيْ الْحَلَيْ اللَّالِي اللَّالَةُ عَلَيْ الْحَالَةُ عَلَيْ عَلَيْ اللَّالِي عَلَيْ الْحَالَةُ عَلَيْ الللللَّالِي اللَّا عَلَيْ اللَّالِي اللَّالِي عَلَيْ اللَّالِي عَلَيْ الللَّا عَلَيْ اللَّالِ اللَّالَةُ عَلَيْ اللَّهُ عَلَيْ الْحَالِي اللَّهُ عَلَيْ اللَّالِي الللَّهُ عَلَيْ اللَّهُ عَلَيْ اللَّ عَلَيْ عَلَيْ الللَّهُ عَلَيْ عَلَيْ اللَّ عَلَيْ عَلَيْ عَلَيْ اللَّ عَلَيْ عَ
- (11.111) רַיִמי שְּבּשִי הָשִישִי אָבּקיין אָבירָקין? ג'יב: tsi? du-ka mindu?? score one EX.SEN-PQ NEG.EX.SEN 'Is there twenty or (is there) not?' (KT discussion with TB)

11.1.3.4 Alternative questions with -*lø*?

The marginal bisyndetic alternative question marker $-l\phi^2$ only occurs in my data twice, both times as a complement clause of the verb *ta* 'look', see (11.112) and (11.113). The latter one is spoken by a balsam flower to an overly confident bumble bee in a folk story.

- (11.112) $\operatorname{Agr}_{\tilde{u}} \tilde{u}_{\Gamma} \operatorname{ar}_{R} \operatorname{agr}_{\tilde{u}} \tilde{u}_{\Gamma} \operatorname{gr}_{\tilde{u}} \widetilde{\mathfrak{s}}_{\Gamma}$ $t^{h}uy \cdot l \phi 2 \quad ma \cdot t^{h}uy \cdot l \phi 2 \quad ta \cdot t \varepsilon^{h} i.$ $drink \cdot Q \quad NEG \cdot drink \cdot Q \quad look \cdot IMP \cdot FRN$ 'Look whether (the cow) drank it or not.' (TB e) (11.113) $\widehat{\Gamma}_{,}^{r} \operatorname{an}_{,} \alpha \widehat{\in}_{,} \widetilde{u}_{,} \widetilde{\Gamma}_{,} \operatorname{gr}_{,} \widetilde{u}_{,}^{r} \widetilde{\mathfrak{s}}_{,} \widetilde{u}_{,}^{r} \widetilde{\mathfrak{s}}_{,} \widetilde{u}_{,}^{r} \widetilde{\mathfrak{s}}_{,} \widetilde{\mathfrak{s}}_{,}^{r} \widetilde{\mathfrak{s}}_{,} \widetilde{\mathfrak{s}}_{,}^{r} \widetilde$
 - t'ene dzip-lø? man-dzip-lø? ta-g ϵ =s làp-o=lo then suck-Q NEG-suck-Q look-HORT=QUO say-2INF=REP 'Then, let's see (whether you will) suck or not (nectar from me), it said.' (RS bee story)

A cognate of $-l\phi 2$ is found in Standard Tibetan, where it occurs as a question marker (Tournadre & Dorje 2003: 230). The morpheme $-l\phi 2$ is also used in exclamation, see §11.2.1.

11.1.4 Tag questions

Denjongke uses interrogated equative copulas as tag questions, which are separate utterances, often preceded by a pause. By tag questions the speaker invites the addressee to listen attentively. Often a tag question also manages to trigger some type of verbal response from addressee (e.g. la- \tilde{t} : [HON-EQU.PER] 'yes, it is so'). The tag questions described here are the interrogated copulas $i\eta$ -ga and $b\varepsilon$ -ka, and the Nepali loan lo. In addition, the negated copulas $m \tilde{e}mbo$, $m \tilde{e}na$ and $m \varepsilon \eta ga$ are used in questions pragmatically similarly to tag questions, see Table 10.6 above. Examples (11.114) and (11.115) exemplify $i\eta$ -ga:

(11.114)	<u>न्'र्ह्रे' न्' रून'</u> ब	नुत्रार्छ्या २ थे	र्षत्वःश्चन्द्रम् अवग्याया	JAN?	
	t'ato t'a	rã:	pámt£i=ra	jὲbbɛ=ɕo,	íŋ-ga =la?
	now now	2sg.l	with=AEMPH	EX.NE=AT	EQU.PER-PQ=HON
	'Now she'	s with yo	ou, isn't she?' (l	BB BB discu	ussion)

(11.115)	קֿקי הק בה צז	९ बेट्र'बाय्व' झट्र' क्षेव'	ম?					
	t'ɛp=di	nè:=tsa:	mè:-k ^h ɛn	bɛ?,	íŋ-ga ?			
	book=DEMP	H 1SG.GEN=at	NEG.EX-NM	LZ EQU.NE	EQU.PER-PQ			
	'I don't have the book with me, do I?' (KL BLA 12)							

In (11.114), speaker looks for a spoken confirmation of his preconceived idea, so the sentence is also pragmatically a request for information. In (11.115), on the other hand, the speaker, by using the tag question, is not trying to confirm the truth value of the clause but rather just aiming at keeping the addressee engaged in listening.

For the less frequent copular tag question $b\varepsilon$ -ka consider:

(11.116)לישרי וערימלי אריארי ריסהי איארי מלי וערייזאי איאוטאיאטאיאטאיי איאטארי איאטאיארי איאטאין אייזארי איזיארי איזאיז ran-ke? t'iza: $k^h \tilde{o}$:=di loke=di nàtci $k^{h}on=gi$ but 3PL=DEMPH own-language 1PL.GEN Lhoke=DEMPH 3PL=AGT mi-k^hɛm-bo $mi-k^h \in \eta - k^h \in n$ $b\varepsilon = la$, NEG-know.HON-2INF EQU.NE=HON NEG-know.HON-NMLZ bɛ?. $b\epsilon$ -ka=la? EQU.NE=HON EQU.NE-PQ=HON 'But they don't know their own language, our Lhoke, they don't know, isn't it (so)? (YR canteen video)

A tag question is typically pronounced with a raised pitch. Figure 11.11 illustrates the intonation rise on the tag question $i\eta$ -ga from (11.115).

Figure 11.11. Intonation with tag question *inga* in (11.115)

pè:- tsa: mèk^hen be? t'ep=di íŋ-ga

Another tag question, *lo*, is a loan of the frequent Nepali tag question *la*. Using *lo* in requests or orders is polite because the speaker requests for the addressee's compliance rather than takes it for granted.

(11.117)	र त्युः तॅ
	nà gju lo.
	ISG go TAG.Q
	'I'm going now, okay?' (oh, Martam)
(11.118)	न्त्वग्रयान्वितः ईग्रान्तेः गव्रदाः या
	$zu:te\tilde{e}: p'ja-u n\tilde{a}: lo.$
	sitting do-2INF do.HON TAG.Q
	'Please stay here, okay (while I go away for a while)?' (oh, TB)
(11.119)	દ્રતે વયાવરા (number) હુદ્દે છેવુ લેં બાળાયા
	pè: nambər di í: lo ː=la
	1SG.GEN number(Eng.) this EQU.PER TAG.Q=HON
	'This is my number, eh.' (KT phone call)
(11.120)	ब्रह्य:२८. योत्र यो. ज्ञालयाया
	dzε:-εε náŋ-gε lo :=la.
	meet.HON-INF do.HON-HORT TAG.Q=HON
	'So, let's keep on meeting, okay?' (KT phone call)

The tag question marker *lo* is pronounced with raised pitch, as shown in Figure 11.12, which presents the pitch trace from (11.120). When *lo* is followed by the honorific clitic =la, as in Figure 11.12, the vowel is lengthened to [lo:].



Figure 11.12. Intonation with tag question lo in (11.120)

11.1.5 Questions with the reportative *=lo*

Interrogatives may be marked by the reportative =lo, which can replace equative copulas (see §7.2.5.2). Examples (11.121) and (11.122), which were used during a phone call, exemplify a polar interrogative and a content interrogative respectively. While the exact context for the utterances is unclear, (11.121) appears to ascertain that the wedding mentioned in the phone call (hence the reportative) was indeed the addressee's own wedding. The use of =lo in (11.122) is more difficult to decipher. The use of the reportative perhaps indicates that the date of the gathering mentioned in the clause is announced by someone else than the addressee, ultimately presumably by an astrologer who determines an auspicious date.

(11.121) रूट यो यानेव या रे

ran=gi pén=lo? you=GEN wedding=REP '(Are you saying/Did you say) it's your wedding?' (KN oh, phone call)

(11.122)

قرم المعن مركز معن معن معن مركز $ts^{h}o:-c\varepsilon = di$ $n am = l\varepsilon = lo$? gather-INF = DEMPH when = ABL = REP 'When is the gathering together (according to them)?' (KN oh, phone call)

Intonation in (11.121) and (11.122) follows the pattern already established above. The polar interrogative (Figure 11.13) has a rising pitch and the content interrogative a low pitch at the end of the utterance (Figure 11.14).

Figure 11.13. Rising intonation with *lo* in polar question (11.121)





11.1.6 Exclamative questions with (h)or

The formative $(h)\delta$, which is most likely a borrowing of the Nepali equative copula ho, also occurs in equative polar questions where it replaces both the copula and the question marker and has the meaning 'is it true that, is it so that', see (11.123-125). Because $(h)\delta$: also occurs as a noninterrogatory exclamative tag (see §16.3.2), it has an air of exclamativity. This exclamativity is reflected also in the interrogative, hence the gloss as exclamative question (EXCLAM.Q). Because clauses with (h)o: here are treated as basically interrogative but having exclamative nuance, they are discussed here under interrogation (§11.1) rather than under exclamation (§11.2).

- (11.123) $ext{dar}$ $ext{ir}$ $ext{dar}$ $ext{campo}$ $ext{dug}$, $ext{dar}$, $ext{dif}$? $ext{tsampo}$ $ext{dug}$, $ext{len}$ = $ext{di}$ $ext{dir}$? flour beat-NMLZ=DEMPH EXCLAM.Q 'Is (it) the one who beat the flour.' (PT e)
- (11.124) A: جَنْعُسَ جُنَيْمَعَسَ (direct) جَنَةِ سَمَحَ عَلَى اللَّهُ مَنْ حَرَ عَلَى اللَّهُ مَنْ حَرَ عَلَى اللَّهُ مَنْ حَرَ عَلَى اللَّهُ مَنْ حَرَى اللَّهُ عَلَى اللَّهُ مَنْ حَرَى اللَّهُ مَنْ حَرَى اللَّهُ عَلَى اللَّهُ مَنْ حَرَى اللَّهُ عَلَى اللَّهُ مَنْ حَرَى الْحَرَى الْحَرَى اللَّهُ عَلَى الْحَرَى الْحَرَى الْحَرَى الْحَرَى الْحَرَى الْحَرَى الْحَرَى الْحَرَى الْحَرَى اللَّهُ مَنْ حَرَى اللَّهُ مَنْ حَرَى الْحَرَى الْحَرَ اللَّهُ اللَّا عَلَيْ حَرَى اللَّهُ مَنْ حَرَى اللَّهُ عَلَى الْحَرَى الْحَرَى الْحَرَى الْحَرَى الْحَالَى الْحَلَى الْحَالَ عَلَى الْحَالَ عَلَى الْحَالَ حَالَ الْحَالَ الْحَالَ عَلَى الْحَاكَ مَا الْحَالَ الْحَالَ عَلَى الْحَالَةُ اللَّا عَلَيْ حَرَى الْحَالَ حَالَى الْحَالَ حَالَ حَاجَةَ مَا الْحَالَةُ الْحَالَةُ الْحَالَةُ عَلَى الْحَالَةُ الْحَالَ عَلَى الْحَالَ الْحَالَ عَلَى الْحَالَ عَلَى الْحَالَ عَلَى الْحَالَ الْحَالَةُ الْحَالَى الْحَالَةُ الْحَالَ الْحَالَ لَكَ الْحَالَ الْحَال اللَّالَةُ اللَّاحَاكَ اللَّهُ عَلَيْ حَالَى الْحَالَةُ عَلَى الْحَالَةُ عَلَى الْحَالَةُ الْحَالَ الْحَالَ عَلَى الْحَالَةُ الْحَالَةُ الْحَالَةُ عَلَى الْحَالَةُ الْحَالَةُ الْحَاكَ مَالَ الْحَالَى الْحَالَةُ الْحَالَةُ الْحَالَةُ

B: ईंट्र'र्रे' दर्दे' क्रें?

tsøpo=di **ó**:? debate=DEMPH EXCLAM.Q 'You mean the debate?' (KN kitchen discussion)

(11.125) a) अत्रे'ग्र'र्र्र् क्षे?

né:mu=rã: ó:? really=AEMPH TAG.EXCLAM 'Really, is it?' (Richhi 99) b) बठ्'रॅं' ५८२ेंग' कॅप'के'गॅंद' झुब्'ग्पट' व' छि्ग' झे' ५ग' भू' गॅंद' भे' कॅं। nè:po dem ts^haptee-lø? méŋk^hã:=na k^hik-ti t'ak-ee patient such serious-EXCLAM hospital=LOC lead-NF get.well-INF jò:=se ó:? EX.PER=QUO EXCLAM.Q 'Is it so that taken to the hospital in such a serious condition the patient is to get well?' (rnam-rtog 17)

The quotative $=s\varepsilon$ in (11.125b) shows that the speaker refers to another person's words/idea. The tag δ : functions in (11.125b) as an interrogative predicate which has a declarative clause as an argument.

11.2 Exclamatives

Exclamative clauses not only inform but also "express an affective response to what is taken to be a fact" (König & Siemund 2007: 316). Exclamative clauses can be formed by using the suffix $-l\phi^2$, which typically collocates with specific other words described below (§11.2.1), by non-interrogatory use of the interrogated copula $b\varepsilon$ -ka (§11.2.2) or through interjections (§11.2.3). Interjections are words that comprise an utterance in themselves (Schachter & Shopen 2007: 57) and express the speaker's spontaneous emotions and reactions to something they have experienced or heard. Therefore interjections can be considered a special case of exclamation.

11.2.1 Exclamation with -lø?

The exclamative suffix $-l\phi^2$ can be added to stative verbs to form an exclamation, e.g. 'how tasty!'. Whereas in Lhasa Tibetan (Tournadre & Dorje 2003: 230) the cognate of $l\phi^2$ occurs in genuine questions, in Denjongke the formative is used in exclamative rhetorical questions. The use of $-l\phi^2$ is illustrated in (11.126-129) with examples of ϵim 'be tasty' (cf. $\epsilon impu/\epsilon imta^2$ 'tasty'), $l\epsilon^2$ 'be good' (cf. $l\epsilon m$ 'good'), dzik 'be excellent' (cf. $dzikta^2$ 'excellent) and ga 'rejoice' (cf. $gata^2$ 'happy'). In exclamative clauses $-l\phi^2$ collocates with a question word (11.126-127) or forms an idiomatic succession with the reportative =lo (11.128).

- (11.126) עון אָלאָל (פֿאי שָׁאָי דָ, אָישׁ! לָי אן) k'amø: בוּשּ-וּשָּׁאַ bo, $\delta j\varepsilon$, p'ja-ca!what.GEN be.tasty-EXCLAM EQU.NE.Q oh chicken-meat 'How tasty it is, oh, chicken-meat!' (Richhi 89)
- (11.127) गाञ्चेय गट्राट्र वेगयांग्यंया

k'atem k'e:da: lɛ̀:-løʔ! what.kind cleaning be.good-EXCLAM 'How well tidied up!' (Richhi 45)

In example (11.129), $-l\phi^2$ does not occur with a question word or the reportative =lo but is followed by the demonstrative $\delta d\epsilon m$ 'such, like that'. Here $-l\phi^2$, together with the demonstrative, functions rather as an intensifier of the property concept ('such happy') than as a marker of clausal level exclamation.

(11.129) בַּזְרְגִיאָלָא שׁׁיִמְבָאישׁרָאָ שָּׁאי שָׁאָיאַדָּי *ga:-lø?* ódɛmø: kʰimzi ko:-di rejoice-EXCLAM such.GEN home throw.away-NF 'forsaking such a happy home...' (nga'i 'gan 14)

The form $-l\phi^2$ is also used in alternative questions, see §11.1.3.4. For etymological information on $-l\phi^2$, see §3.3.6.12.

11.2.2 Exclamative use of the interrogated copula bɛ-ka

In addition to the interrogatory uses, the interrogated neutral copula can be used for exclamation, as in (11.130) and (11.131). The difference between glosses a) and b) in (11.131) has to be determined contextually.

(11.130) E' AS' ANK! E	(11.130)	हा तही वियास ह
------------------------	----------	----------------

tc'a=di cimpu bɛ-ka! tea=DEMPH delicious EQU.NE-PQ 'Isn't this tea good!' / 'How good this tea is!' (KT e)

(11.131)	र्गेवाऱ्यात्ते [.] (Co	mpany) ন'র্শ	र्द्धेतुः (phone) यह	८.र्ट्र. झट.ग/://			
	kompani	ŋà=lo	fon	tã:-do	bɛ-ka		
	company	1sg=dat	phone(Eng)	send-IPFV	EQU.NE-PQ		
	a) 'Does the company keep on phoning me?' (question)						
	b) 'How the company keeps on phoning me!' (exclamation) (KN e)						

11.2.3 Interjections

An illustrative list of Denjongke interjections, which occurred in §3.6.10, is reproduced in Table 11.3. Some of the interjections are exemplified below the table.

³⁸⁵ With consultant KT, $-l\phi$? typically collocates with =lo (or). The reportative =lo is segmentally homophonous with the tag *lo*. Further exploration is needed to find out for certain which one is used here.

Form		Gloss	Function
áme:	લ્ય'એ'	'wow'	expressing enthrallment, rapture
làso	শ্যবাধ্য র্মি	ʻokay'	approval
ák(ʰ)aː	মে'মে'	'gee, gosh'	surprise, amazement
ádzei	<i>अ</i> 'हें'	'gee, gosh'	surprise, amazement, quite similar to $\dot{a}k^ha$:
ójε, ói	ર્લેપ્પે	'oh, hey'	used for getting someone's attention
έː, jàː	ક્ષે', પ્પ'	ʻoh'	expresses engagement or surprise when listening,
			keeps the conversation going
á:	હ્ય	'hey',	1) informal address to get someone's attention
		'yes',	2) non-honorific response to being called, addressed to social
		'what?'	inferiors (corresponds to honorific <i>la</i> : below)
là:	অধ্যম	'yes'	1) polite response to being called
		'excuse	2) expressing that the speaker did not hear or understand
		me?'	what was said
kei	ਸ਼੍ਰੋ', ਸ਼੍ਰੋ'	'O (voc.)'	honorific address (e.g. 'O lama')
t€ ^h a	æ	'alas'	disappointment (e.g. after a bad shot in a game of carrom or
			<i>kerembot</i>), loan from Nepali
t€ ^h E∶	æ.	'ah, oh no'	disapproval, discomfort (e.g. when someone does not answer
			phone), the response to being tickled
úf	জ্বন্ব.	'phew'	expression of pain or discomfort
ádzi:	<i>खां हें</i> '	'oh no'	1) (unpleasant) surprise, 2) fear
áijo:	જો [.] ર્ਘે'	'ouch'	pain or fear of pain
á(i)ja:	જ્ઞે'પ્યૂ', જ્ઞ'પ્પ'		
óho:	ર્લે કેં	'oh no',	sadness, response to bad news (KT)
		'gosh'	(neutral) surprise at information (KUN)

Table 11.3. Some interjections

(11.132) เหาล่า ณิชาชาลังชาลั!

ámε: lɛ̀:-lø:=*lo!* wow be.good-EXCLAM=REP 'Wow, how good it is.' (KT e)

(11.133) (אַזיְאָיאָׁי קֿיקּאָי שָׁקָיאָ' זוּקָקָיין **làso**, t'ɛnɛ tɛ'øm-bo nấ:. alright then go.HON-2INF do.HON 'Alright, in that case please go.' (Richhi 17)

b)	ર્લે ખે! દિ	अ'त' बे'थेंट्'ग?		
	ójε:,	k ^h im=na	mí	j <i>à:-ka</i> ?
	hey	house=LOC	human	EX.PER-PQ
	'Hey	, is anyone a	it home?'	(rna-gsung 5)

- (11.135) $\hat{\alpha}' \hat{\alpha}' \hat{\gamma}' \hat{\gamma}' \hat{\gamma}' \hat{\gamma}' \hat{\alpha}' \hat{\gamma}' \hat{\alpha}_{\eta}$ $\acute{\epsilon}' \acute{ona} b\epsilon^2$. $\eta \dot{a} t'ato t^{ho:-po} \hat{t}'$. o there EQU.NE 1SG now hear-2INF EQU.PER 'O, it's there. I heard it (only) now.' (PT kitchen discussion)
- (11.137) a) गरा!

karma! PN 'Karma!'

b) व्ययाह्य। यात्र: याह्युद्र: रॅं? *la:*, *k'an súm-bo?* yes what say.HON-2INF 'Yes, what did you say?' (Richhi 15)

(11.138) a) अूर्'हे! ब्रिंश'त'र्ने' यत्र'त्यु'ये।

ádzi:, $k^{h}im=na=to$ maŋ-gju-kɛ! oh.no house=LOC=CEMPH NEG-go-HORT 'Oh no, let's not go to (his) home.' (reaction to suggestion) (Richhi 24)

b) $a_i \stackrel{\text{dist}}{=} a a_i \stackrel{\text{dist}}{=} a a$

(11.139) æj

tc^ha! darn 'Darn! (after an unsuccessful strike in a game of kerembot)' (oh, Tashiding)

- (11.141) هَرْ مَحْرَ عَلَى اللَّهُ اللَّ (11.141) مُعْرَبُهُ اللَّهُ اللَّ (11.141) مُعْرَبُهُ اللَّهُ اللَّ مُعَامُ اللَّهُ اللَّ
- (11.142) אינע ב, מי אַרא איז אַרא איני (מי גר) איני איני

àijo: $y\dot{a}$, $k^{h}ap=di$ mi-kjap *áijo:*! ouch 1SG needle=DEMPH NEG-strike ouch 'Woe is me, do not inject the needle, ouch.' (rnam-rtog 28)

11.3 Imperative

Sentences in imperative mood convey commands and requests. Imperatives in Denjongke can be expressed by using the bare verb root (§11.3.1) or by imperative suffixes $-tc^{h}i$, -da and -na (§11.3.2). Urgentive nuance to imperatives can be provided by $=m\phi^2$ (§11.3.3).

11.3.1 Verb root as imperative

The simplest imperative form consists of the bare verb root, see (11.143)

(11.143) এবামার্ম্ব, জাক্তু ব্রাইরমান্যমার্ র্রা

làso ágja, zim-pa tc'ð:. alright elder.brother sleep.HON-PUR go.HON 'Alright brother, please go to sleep.' (Richhi 57)

Example (11.144) further illustrates that the imperative mood is negated by the perfective prefix *ma*- and that the imperative may be followed by an assertive tag (see §16.3.1).

(11.144) เช จรุสา ชัวา สาจา สัา

 k^ha tsum $d\phi$?. ma-làp no. mouth closed sit NEG-speak TAG.ASR 'Be silent. Don't speak, I tell you.' (KNA kitchen discussion)

nấ:.

Imperative with the periphrastic honorific verbal form VERB-*po* $n\hat{a}$: is illustrated in (11.145) and (11.146). The negator prefix attaches to the honorific verb $n\hat{a}$: 'do (hon.), grant'.

(11.145) ק־זַּה־ מאָרִיאָ מַסְרִין t'a(r)uŋ s**úm-bo**

again say.HON-2INF do.HON 'Please say it again.' (TB e)

(11.146) র্ট্রবার্ষ আল্বনা

tc'em-boma-nã:.come/go-2INFNEG-do.HON'Please do not come/go.' (TB e)

Using the bare verb root for imperative causes ambiguity on the clausal level (which context usually disambiguates on the discourse level), because final suffixes and auxiliaries are often elided in declarative clauses, as shown in (11.147).

(11.147) চি ক্লবায়াস্থ বনাপ প্রশান প্রদা te lópta=tsu nàŋea pel-bo nấ:. so school=PL within spread-2INF do.HON 'So (they) spread (them) within schools.' (CY interview)

Example (11.147) is clearly a declarative on the discourse level but could be mistaken for an imperative on the clause level.

Sandberg (1895: 42) lists three imperative construction: bare verb root, the root appended with $t\tilde{a}$: (ordinary) and the root appended with $n\tilde{a}$: or $p\dot{a}$ (honorific). The secondary verb (or verbalizer) $t\tilde{a}$: 'send' also sometimes accompanies the verb root in imperative in my data:

(11.148) ন্যান্ডন্য ব্রাব্যস্কর্ম র্ভাব্যস্কর্চনা র্জার্স্কন ব্রাব্যস্কর্ম র্ভাব্যস্কর্চনা র্জার্স্কন ব্রাব্যস্কর্ম ব্রান্ডর্জন **p'ja tã**:. ótsõ: dã: benda **t'a: tã**:. 2PL what.is.that do send onion and tomato slice send 'You, do what's that, slice onions and tomatos.' (PT kitchen)

Sandberg's (1895: 42) more polite form, where the verb $n\hat{a}$: 'do (hon.), grant' is directly appended to the verb root, also occurs in my data, see (11.149), but not as frequently as the nominalized construction exemplified in (11.145) above:

(11.149) $at = c \cdot a \cdot c \cdot a \cdot a = c \cdot a$

11.3.2 Imperative suffixes -tc^hi, -da, -na

While the bare verb root can function as an imperative, the imperative mood may also be marked unambiguously by the suffixes $-te^{hi}$, -da and -na. Using $-te^{hi}$ and -da makes the request/command more friendly than using the bare verb root, hence the term "friendly imperative" (similarly Denwood 1999: 168). I have not, however, been able to find any semantic difference between $-te^{hi}$ and -da, which are illustrated in (11.150) and (11.151). Example (11.150), taken from the novel Richhi, presents a doctor'a polite instruction to a nurse:

Although the use of imperative suffixes in negated imperatives is rare in my data, at least $-tc^{h}i$ occurs in a negated imperative:

The suggestive *-na* (glossed SUG), on the other hand, softens down the tone of the imperative towards a suggestion and thus make a request/command more polite than using the bare verb root or one of the other suffixes. The use of *-na* is illustrated in (11.153) and (11.154) by sentences from the novel Richhi and the play *nga'i 'gan*, respectively. In the first example, *-na* is used in an honorific verbal expression and in the second with an ordinary verb root, showing that *-na* does not directly participate in the honorific system, where the use of one honorific (e.g. an honorific noun) in good style requires the use of other honorifics (e.g. an honorific verb).

रु'र्श्वेन्य्याय, अन्न सु'र्श्वेन्रन रेवा'गुया पर्झेत'र्ये' यात्रन्यायत' यासुन्यू: न'रेन्ये' कॅर' यार्चेया' पर्झेत'र्ये' यात्रन्या (11.153)p'usim=rã: p'usimla, dã: tciku: younger.sister=HON yesterday younger.sister=AEMPH only.AGT nấː-kʰĩː súnlu gøm-bo t'arin=lo $ts^{h}a$: tci? sing.HON-2INF do.HON-NMLZ song.HON today=DAT turn one nấː-na. gøm-bo sing.HON-2INF do.HON-SUG 'Sister, what if you'd please sing today once the song that you (lit. sister) sang alone yesterday.' (Ricchi 90)

(11.154) $\delta ce:kuk^{h}\tilde{e}: rindz\tilde{i}:, tc^{h}\phi$? $n \dot{a} \eta = l \varepsilon$ màla? lòkti **gju-na** t'ønda? 2SG.L inside=ABL quick again go-SUG please PN meaning mè:-pø: ke:po ma-cé-pa. tam NEG.EX-2INF speech many NEG-tell-CIRC 'Please Rinzing, what if you go quickly back from inside (this room) without speaking those meaningless words.' (Nga'i 'gan 9-10)

The suggestive *-na* differs from friendly imperatives $-te^{hi}$ and -da in that *-na* attaches to the imperfective root of the verb gju (gju-na 'go, what about going'), where as $-te^{hi}$ and -da: attach to the suppletive perfective form $s\delta$: of the same verb (e.g. $s\delta$: $-te^{hi}$ 'go!').

Example (11.155) summarizes the four different imperative constructions, all essentially meaning 'please come here'. Note that the bare verb root as imperative is in (11.155a) represented by the final verb $n\hat{a}$: of the honorific nominalized construction. The use of the honorific

construction results in all the forms in (11.155) being basically polite. Simpler, and less honorific forms for a bare verb imperative would be $\epsilon \delta 2$ 'come! (direct)' and $t\epsilon' \phi n$ 'please come! (hon.)'.

(11.155)a) ব' র্ট্রব'র্ন' গ্রদ্য nấː. nà: tc'em-bo here come.HON-2INF do.HON 'Please come here.' (polite) b) वः र्चुवःर्त्तं गवनःकी nấ:-tc^hi. nà: tc'em-bo here come.HON-2INF do.HON-IMP.FRN 'Please come here.' (polite, friendly) c) व' र्चुव'र्चे' गवराररा। nà: tc'ɛm-bo nấː-da. here come.HON-2INF do.HON-IMP.FRN 'Please come here.' (polite, friendly) d) तुः र्च्चेतुःर्त्तेः गुत्रृत्त्र तु nấː-na. nà: tc'em-bo here come.HON-2INF do.HON-SUG 'What if you would please come here.' (polite, suggestive)

Imperatives may be supplemented with the exclamative tags $p\dot{a}$ and $inam^{386}$, which add nuance to the command. Whereas $p\dot{a}$ is considered friendly, *inam* is usually spoken by a person of a greater social standing than the addressee and has an air of obligation. Example (11.156), illustrating $p\dot{a}$, was spoken by a younger person to an elder one, whereas (11.157), illustrating *inam*, was spoken on the telephone by an older brother to a younger brother who was working in the Indian capital Delhi.

(11.156)	গ্লব্র্ গ্রহ র sém-bo nấ: pá . listen.HON do.HON TAG.ASR 'Please listen (to me), will you.' (PB discussion with TB)
(11.157)	क्ष ईंग्लें' (Nep.) ईश्व केंब वज्य देगा पत्वा केंब वज्य átsi dzokai p'ja ín-(n)am! t'ɛk zaː ín-(n)am! a.bit saving(Nep.) do EQU.PER-ATTQ store set EQU.PER-ATTQ 'Save a bit (money), I tell you! Set (it) aside, I tell you!' (TB phone call)

The pitch trace from (11.157) is presented in Figure 11.15, showing the rising pitch on *inam*.

³⁸⁶ The form *inam* is segmentable as *in-(n)am* [EQU.PER-ATTQ], whereas the form *na* has merged into a unit from the historical segments *in-(n)a* [EQU.PER-PQ].





In some language varieties, the forms ino (Bermeok) and no (Martam) as used for pá/ina:

- (11.158) $\mathfrak{H}(??) \mathfrak{a} \mathfrak{h}(\mathfrak{H}) \mathfrak{K}(\mathfrak{H}) \mathfrak{K}(\mathfrak{H}) \mathfrak{K}(\mathfrak{H})$ $gjat^{h}a \quad di=tsu \quad súp \quad \textbf{ino.}$ hatch this=PL close TAG.ASR 'Close the hatches, will you.' (KT animal story)
- (11.159) $[\pi^{,} \neg \neg \overleftarrow{a}, \overleftarrow{a}, \neg \neg \overleftarrow{a}]$ $k^{h}a$ tsum $d\phi$?. ma-làp no. mouth closed sit NEG-speak TAG.ASR 'Be silent. Don't speak, eh.' (KNA kitchen discussion)

11.3.3 Urgentive =*mø*?

The enclitic $=m\phi^2$ (pronounced also as $=me^2$) can be added to imperatives, hortatives and optatives to make a plea more urgent. Within imperatives $=m\phi^2$ may attach to the verb root and suggestive -na but not to the friendly imperatives $-te^{hi}$ and -da. Rather than forming a sentence mood of its own, $=m\phi^2$ may be seen as a modifier of the imperative, hortative and optative moods. One consultant commented that $=m\phi^2$ may be used when all other verbal persuasion resources have been exhausted. Because $=m\phi^2$ may attach to several elements it is not analyzed as a suffix but as an enclitic. The following examples illustrate the use of $=m\phi^2$ appended to verb root (11.160-162), imperative -na (11.163), hortative $-k\varepsilon/g\varepsilon$ (11.164) and infinitive -ni (11.165-166). The use with optative $-t\varepsilon u^2$ is illustrated in §11.5 below. In example (11.160), a folk story on animals, a bear has just heard from a marten that the marten might be able to offer him (the bear) a good job by asking the king. The bear responds to the marten:

(11.160) क्षे हे लुर्च गुवर कॅन

έ:, tɛ cù: nấ:=mò?. oh then ask.2INF do.HON=URG 'Oh, then ask (him), by all means.' (KT animal story)

(11.161) $a_{i} [i \in \tilde{A}^{i}] = lo$ hako-ga làp t'i ta=**mø?.** here 3SG.HON=DAT understand-PQ say ask look=URG 'Try and ask him here, by all means, whether he knows (the story I am about to tell).' (JDF discussion on the roof)
(11.162) স্বিশ্যস্য স্বিশ্বর্শ্বর্দ্য ze:kar ze:=me?. chilli.HON eat.HON=URG 'Go ahead and take chilli.' (PTM kitchen discussion)

Of the imperative suffixes, the urgentive may attach to the suggestive -na:

(11.163) লন্স ল'র'র্শ্বন্য sàm sà-na=**mø?.** food eat-IMP.SUG=URG 'Would you eat (some) food, by all means!' (PT e)

In (11.164), a line from a contemporary Denjongke song, the urgency marker is appended to hortative $-g\varepsilon$:

Consultant KN also reported that $=m\phi^2$ is used, especially in Tashiding (not the consultant's native place), in the following idiomatic expressions that employ the infinitive form *-ni*, see (11.165-166).

- (11.165) $\exists \cdot \hat{\vartheta} \cdot \tilde{\aleph} \gamma$ $s \hat{a} - n i = m \emptyset ?$. eat-3INF=URG 'Eat, by all means.' (KN e)

11.4 Hortative -*kɛ/gɛ*

Whereas the imperative concerns second person commands and requests, the hortative is mainly used for first person suggestions either in the singular, as in (11.167), or plural, as in (11.168). The hortative marker $-k\epsilon/g\epsilon$ is appended directly to the verb root. Note that in (11.167) the velar is elided because the preceding verb root also ends in a velar.

(11.167) ८ ८ २ २ २ १ १ १ १

ηà bak-ε. 1SG carry-HORT 'Let me carry (it).' (GB oh) (11.168) $\neg e^{\pi q} e^{\pi q} \neg e^{\pi q} e^{\pi q} \neg e^{\pi q} e^{\pi q} \neg e^{\pi q} e^{$

The friendly imperative suffixes may be appended to the hortative, as exemplified by $-tc^{h}i$ in (11.169).

(11.169) $\mathfrak{H}^{a_1}\mathfrak{X}' \mathfrak{A} \mathfrak{L}' \mathfrak{X}' \mathfrak{A} \mathfrak{Y}' \mathfrak{Y}' \mathfrak{A} \mathfrak{Y}' \mathfrak{A} \mathfrak{Y}' \mathfrak{A} \mathfrak{Y}' \mathfrak{A} \mathfrak{Y}' \mathfrak{A}$

Although the hortative is mainly used in the first person, I heard the second person plural address (11.170) in Martam (East Sikkim). The overheard clause has two optional translations.

The hortative construction is negated by the perfective negator prefix ma-, see (11.171) and (11.172).

11.5 Optative with tcu?

Optative mood expresses hopes about desirable future. The optative is formed by the causative secondary verb *teu?* $\neg \neg \neg \neg \neg$ 'cause' (see §5.5.2.), which attaches to the verb root without tense, aspect and modality marking. For an example, consider (11.173), which also employs = $m\phi$? to mark urgency (see §11.3.3).

³⁸⁷ The verb p'ja 'do' is usually reduced to ja when a negator is prefixed.

(11.173) $\hat{\neg}$ 'त्रका क्षेंप्रदेश पॉर्थेण' \neg 'र्थेप्पूर' ईन' नहुष्य'कें \neg 'ले t'ene ódem jó? $\eta \dot{a} = lo = j\tilde{a}$: $t^{h}op$ $teu = m\dot{o}$:=s. then such work 1SG=DAT=even receive cause=URG=QUO 'Then, by all means, let me also get such work, he said.' (KTL animal story)

The optative construction may be used in purposive adverbial clauses, see affirmative (11.174) and negated (11.175), which is negated by prefixing *ma*- to the secondary verb.

(11.174)nè:=gi làp-ti ηà 1SG.GEN=GEN birthday.party(Eng.)=LOC 3SGM come cause say-NF 1SG fon p'ja-u ĩĽ. phone(Eng.) do-2INF EOU.PER 'I phoned (him) so that he would come to my birthday party.' (literally: 'Saying let him come to my birthday party, I phoned.') (KN e) विंद रदी र्डु गीया वयया ठट गीया कॅया कुया व वट मा द्वया हेंट्या येवा हेंव या पर्छवा ये यावर रदी कें परेटा कुपया हे विंद रवावा (11.175)ळ. भियक.त्. स्रेटी $k^h \tilde{o}$:=di=tsu=gi $t^{h}amtee = ki te^{h} \phi ge: na: nanea$ dendzõ lòk

here inside all=AGT Sikkim return 3PL=DEMPH=PL=AGT king tc'øn mà-tcu? sé-k^hen=di ódep kjap-ti k^hõ: come.HON NEG-cause say-NMLZ=DEMPH like.that do-NF 3pl gagdza kjab-be?. obstacle do.2INF-EQU.NE 'All of them like that made obstacles so the king would not come back here to Sikkim.' (CY interview)

Usually the optative and causative uses of teu?/teuk can be distinguished by the presence/absence of tense/aspect/modality/evidentiality (TAME) marking: the optative is not followed by TAME markers whereas the causative is. In causative (11.176), however, the periphrastic past construction is elided because it is retrievable from the context (dzy: teuk-o be? > dzy: teu?). Thus, in the absence of tense and aspect marking, the context is the arbiter between optative and causative reading of teu?.

11.6 Summary remarks

This chapter described non-declarative clauses, i.e. interrogatives, exclamatives, imperatives, hortatives and optatives. The main focus was on interrogatives, which form a complex system. Polar questions can be formed either by polar question markers or by intonation without segmental interrogation markers. Polar interrogatives include the suffix -ka/ga and several interrogative copulas (formed either with -ka/ga or by other means). Attenuated interrogating morphemes are used for softening down questions by making them resemble speaker's speech to themselves. Attenuated question markers are also used in content questions, which, however, do not necessarily require any other interrogatory marking than the question word. Some more marginal question marking morphemes were also described.

Interjections were shown to be a special case of exclamatives, which also include the formally interrogated copula $b\varepsilon$ -ka and the suffix $-l\phi$?. The various imperative constructions express such semantic nuances as directness, politeness, friendliness, suggestiveness and urgentness. The hortative marker occurs both with singular ('let me do') and plural first person ('let us do'). Lastly, it was shown that optative clauses are formed by postposing the causative secondary verb stem to the primary verb.

12 Connecting finite clauses

The discussion on clause combining is divided into four chapters. Connections between finite clauses are described in this chapter. The following three chapters address constituent-modifying clauses $(\$13)^{388}$, complement clauses (\$14) and adverbial clauses (\$15). The discussion in this chapter begins with an introduction to concepts and terminology (\$12.1) and continues with a description of the uses of monosyndetic (\$12.2) and bisyndetic (\$12.3) connectors.

12.1 Introduction

Crosslinguistically, connections between finite clauses may occur in a range from coordinationresembling connections to looser discourse connections. According to Haspelmath's definition (2007: 1), in coordination "two or more units of the same type are combined into a larger unit." The last part of the definition, "combined into a larger unit," is particularly challenging to apply to Denjongke because it is uncertain whether the connector words (see §3.6.9) combine the clauses, as Haspelmath's (2007:1) definition posits, "into a larger unit" called sentence, or whether the connectives provide looser logical connections between individual sentences. Consequently, the words "connect" and "connective" are used instead of "combine" and "conjunction" because the latter pair of words suggests coordination, in which two clauses "combined" with a "conjunction" form a larger entity, sentence. The words "connect" and "connective" are meant to include looser relationship between two clauses than the one implied by coordination.³⁸⁹

The reason why coordination in Haspelmath's (2007: 1) definition is not a prominent feature of Denjongke is that Denjongke is a clause-chaining language. Longacre (2007: 375) divides languages into co-ranking languages such as English, in which it is possible to have several verbs "of the same rank" within one sentence, and chaining languages such as many New Guinean languages, in which it is not possible to have more than one final verb form in a sentence. Denjongke is one of the chaining languages in which only the last verb in a sentence is finite and previous, dependent verbs within the same sentence are not "of the same type" as the final verb. This means that Denjongke uses a structurally different strategy for describing situations which in English are expressed through coordination. For instance, consider the Denjongke equivalent of the English clause 'Go today and stay home tomorrow':

t'arin gju-ti thorã: $k^{h}im = na$ $d\phi 2^{.390}$ today go-NF tomorrow house=LOC stay 'Go today and stay home tomorrow.' (Richhi 59)

Although the Denjongke and its English translation in (12.1) are semantically equivalent, they differ structurally in two important respects. First, English uses verb forms of the same type,

³⁸⁸ Constituent-modifying clauses represent clause combining in that the constituent which is modified is an argument in another clause.

³⁸⁹ Payne's (1997: 443) definition of coordination as "linking two clauses of equal grammatical status" leaves open whether the linking results in a new entity. Thus Payne's definition would perhaps subsume looser connections under coordination than Haspelmath's (2007: 1) definition does.

³⁹⁰ One consultant would have preferred the perfective form of the verb $s \delta$: to imperfective gju in the nonfinal clause.

whereas Denjongke uses two different verb forms, nonfinal converbal form marked with -ti/di and the verb root, which functions as an imperative. Second, English uses the conjunction 'and', whereas the Denjongke sentence lacks a conjunction because the nonfinal verb form can convey analogous semantics to the English conjunction 'and'. In summary, (12.1) is structurally not an instance of coordination but of subordination/dependency although it is functionally analogous to an English coordinated sentence.

However, while finite clauses in Denjongke cannot be coordinated by $t'\tilde{a}$: 'and', nominalized clauses can:

(12.2)ลผา ลา ผาสีราวัง รรา จุสูรา ผาสีราวกิง คังสูญ พราวัง จุรุญ [sàm sà *ma-thop-o*] ťã: [t^hun ma-t^hop-øː] lògju? màŋpu food eat NEG-find-2INF and drink NEG-find-2INF.GEN story a.lot du?. EX.SEN 'There are a lot of stories [that (people) haven't found food to eat] and [haven't found (anything) to drink].' (Class 9-10 grammar, 134)

There are other connectors than $t'\tilde{a}$: 'and' that connect finite clauses. My intention in this chapter is to show how finite clauses are linked through these connectives, without entering into a discussion on whether and by what criteria the connected elements should be considered units. The connectors vary in how amenable they are to coordinative interpretation. The connective $t'iz\tilde{a}$: 'but, however', for instance, is at times used like a coordinating conjunction. The connector t'ene 'then, in that case', on the other hand, is hardly a coordinating conjunction because it typically implies, in addition to a semantic/logical connection, that the speaker has changed.

12.2 Monosyndetic connectors

Table 12.1 lists monosyndetic connectors, which have a single connector.

t'izã:	<i>∽</i> `≡∽	'but'
<i>โ:ruŋ</i>	ਲੀਕ੍ਰ-ਤੁਸ-	'but, however', lit. 'although it is (so)' [EQU-CONC]
ínajã:	क्षेत्र'त्र'थम	'but, however', lit. 'even if it is (so)' [EQU-COND-even]
índã:jã:	क्षेत्र'नूम्'ਘम्'	'but, however' lit. 'it is (so) and yet' [EQU-and-even]
ĩ́:jãː	ਲੇਰ੍ਹ ਘ_	'but, however', lit. 'it is (so) yet' [EQU-even]
zenne/zene/zøne	याववाने'	'otherwise' lit. [other-COND] or [other-TOP]
zẽ:mene	गवित्र अत्र ते.	'otherwise', lit. 'if (it) is not elsehow' [other-NEG.EQU-COND]
jầː	とう.	'and, (then) again'
jầːnɛ	थम् वे	'or'
jầːmɛnɛ	ਘ⊏'ਕਰ'ਰੇ'	'or in other case, otherwise', lit. 'and if it is not (so)' [and-
		NEG.EQU-COND]
mi-tshe?	ۿڗۿڗ	'not only (but also); moreover; in addition ', lit. 'not-stop(ping)'
k'ambjasene ³⁹¹	यव द्विश्व श्वे.	'because, this is for the reason that', lit. 'if told why'
dile	तर्नु'लाखा	'then (temporal sequence)' (lit. 'this=ABL', but because of
		frequent use should probably be considered a lexeme)
t'ene	ट्रे'वर्ष', ट्रे'वर्ष'	'then, in that case (logical consequence)' (often collocates with
		te 'well, then', as in te t'ene 'well then')
ódi=lε=to	ૹ૾ૼૻૡઽૢ૾ૺૼ૾ઌૹૻૻૻ૽ૼૢૼ	'rather' (lit. that= ABL=CEMPH)

Table 12.1. Monosyndetic clause connectors

The examples below illustrate the use of the connectors from Table 12.1 in the same order they occur in the table. The two forms which are most often used as contrastive connectors are $t'iz\tilde{a}$: 'but' and \tilde{i} :ruŋ 'but, however'. The contrastive connector $t'iz\tilde{a}$: is of unknown etymological origin. In written Denjongke, $t'iz\tilde{a}$: occurs, depending on the author and possibly the context, either as a sentence-initial marker (following |, the closest equivalent to full stop in written Denjongke), as in (12.3), or as a sentence-medial marker (without 1), as in (12.4).

- (12.3) $\underbrace{\mathfrak{K}}_{\mathsf{N}} \underbrace{\mathfrak{K}}_{\mathsf{N}} \underbrace{\mathfrak{K}} \underbrace{\mathfrak$
- (12.4) המהמיני אַ אַמאיפה מַדָּאָמייס ה' אַיָּאָר מַזְיּבָּי אַזיבי מַזְיּבָּי אַזיבי מַזָיבָּי אַדי אַקישָאַ *mánpu thamtee? dzom-ze t'izã: godze khu=rã: man-dzøn.* multitude all gather-PST but leader 3SGM=REFL NEG-come 'All the people gathered but the leader himself did not come.' (Class 9-10 grammar, 135)

³⁹¹ The first part 'why' in this word may be pronounced k'amja, k'ambja or k'amp'ja, depending on the level of phonological reduction. The last pronunciation k'amp'ja suggests a succession of words rather than a single word, because p' typically only occurs word-initially. The verb of speaking $s\dot{\epsilon}$ (also si) can be replaced by lap 'say' or ϵa 'say (hum.)', e.g. $k'amjalapn\epsilon$, $k'amjacun\epsilon$. The last syllable, which is a conditional marker, may also take the forms *-no* and *-na*, the latter of which is probably affected by Tibetan spelling, e.g. k'amjaseno (eastern and northern pronunciation), k'amjasena (literary pronunciation).

In (12.3), the actor (Choki) is elided in the second clause, suggesting coordination, where the clauses belong to the same sentence. However, argument elision is frequent even in independent clauses, and therefore this argument for the presence of coordination is not fully persuasive. Of all the connectors, $t'iz\tilde{a}$: is the most amenable to coordinating interpretation (which is also suggested by the omission of | in Denjongke writing). Prosodic phenomena in the clause, however, deserve further study.

The contrastive connector $\hat{i}:ruy$ (or $\hat{i}:-ruy$) 'but, however' is the concessive form of the equative $\hat{i}:$, which through frequent use may be considered to have lexicalized. It is not clear, whether (12.5) and (12.6) should be considered to consist of one sentence or two sentences.

R. टपु. योजेव. अक्व. श्रेव. आपव. श्रेवी श्रुव.उट. पि. ट.जू. प्रा. झ. शु. यागे. श्र. शु. यागे. (12.5)โ้:-ruŋ k^hu nè: $p \in nts^h \tilde{e}$: $i \eta - k^h \tilde{e}$: ĩ. k^hu nà=lo relative EQU-NMLZ EQU.PER 3SGM 1SG.GEN EOU-CONC 3SGM 1SG=DAT lòkta mi-kjap be?. care NEG-do EQU.NE 'He is (supposed to be) my relative. However/but, he doesn't look after me. ' (KT e)

(12.6) العمرة حرب ألم ترجب حربي الأمر حرب عامر الأمر عرب حربي المحرب المحرب المحرب المحرب المحرب حربي الأمر المحرب حربي المحرب المحرب

Three additional, more complex forms *inajã:*, *indã:jã:*, *í:jã:* are used for contrastive cohesion. These forms begin with the equative copula *i̇:* and end in the clitic =jã: 'too, even, yet, still'. The difference is what, or whether anything, occurs in between. In *inajã:*, the copula occurs with the literary conditional form *in-(n)a* (which also occurs in Central Tibetan), see (12.7) and (12.8). In *indã:jã:*, the intervening element is the conjunctive t'ã: 'and', conveying the meaning 'it is so and yet', see (12.9). The last form *i̇:jã:* has no intervening element, conveying the equivalent of English 'it is so yet', see (12.10).

אי מקרי שלי שרי מלי ארי מרי אי מריאי ריצהי ארי מרי רי מישאי שליאי איזי איזי אי ארי מרי איי אי ארי מרי איי אי ארי מרי איי (12.7)क्षेंग् भूनि मेरे मेरे स्टर्भ मेरे. स्टर् क्षेंगीया सन् हियाया याववा के हियाया सन् की आधेव आपके सन् te ó ódep gã:=di nànca=lo nàtci p'ja-wø: like.that do-2INF.GEN time=DEMPH so that inside=DAT 1PL.GEN lèm p'ja-ti ke:=di ódɛm=to t'a lap-ce=to now good do-NF like.that=CEMPH language=DEMPH teach-INF =CEMPH mè:-k^hɛn *bε?*. *ínajã*: $g\tilde{a}$:=giдетро di=tsu=gi ódi t'a this=PL=AGT NEG.EX-NMLZ EQU.NE however that time=GEN now elder $mi-k^h \in \eta - k^h \in n$ kɛːdza? $t \epsilon^h u k \epsilon = di$ *bε*?. zømø, Nepali.language=DEMPH NEG-know.HON-NMLZ EQU.NE language.HON other 'When doing like that, within that, our language was not taught well. However, the elder people in those times did not know other languages, did not know Nepali.' (CY interview)

Example (12.8) illustrates that occasionally connecting words, here *t'izã*: and *inajã*:, co-occur.

- (12.8)मुनया ही गी' के पर्नेता गोया रें के क्वा प्रत्या पर्य या के रा ke? te ódi p'ja-ti nàtei cý:tcy? *ódepti pàmte^ha?* so that do-NF language a.bit like.that decline **1PL.GEN** ĩ. số:-bo t'izã: ínajã: t'ato lópța=di nànca but however now school=DEMPH inside go.PFV-2INF EQU.PER jðː-ɛɛ? kjap-tiki ódɛp lák bak da: ma-số:. ke:po=to EX-INF do-NF like.that much=CEMPH be.ruined carry chase NEG-go.PFV 'So for that reason our language went into decline like that. But however (the case), because (the language) is within schools it has not been that badly damaged.' (CY interview)
- איקידי יוקיתלי ארן אארידריערי מליועדי רישיוי אאיידרי מקיין אי מידי יוקרין (12.9)l*àn=to* k'ande: mè?. índã:jã: dik^ha ŋàtca? lèm=rã: message=CEMPH any NEG.EX.PER however here 1PL good=AEMPH làp $du:=s\varepsilon$ nã:. EX.SEN=QUO say do.HON (I) do not have any message. But tell (them) we are alright here.' (nga'i 'gan 15)

(12.10) – एश्वः पत्तः मः रुषा यत्र र्स्ततः यत्तः यत्तः यत्तः यत्या येनः प्राण्यः वयः वर्ष्ते यार्स्तवाया क्षेत्रः यत्ते र्त्तवायां श्वः श्चितः (12.10) t'atawa: $\eta at ca? p^{h} ent s^{h} \tilde{v}$: p^ha dze: ts^hu: dze: thither meet.HON hither nowadays 1PL mutual meet.HON cè: $ts^h o$ ma-ts^hu?. Ĩ:jã: mè:-pe: nè: NEG.EX-2INF.AGT face recover NEG-be.able.to however **1SG.GEN** ro:m laki female.friend PN 'These days, because we have not met each other here and there (I) could not recognize (him). However, my friend Lhaki...' (Richhi 46)

Still other contrastive connectors are built around the word *zen* 'other'. These forms are *zenne/zene/zøne* (12.11-12), *ze:men* (12.13), *zẽ:mene* (12.14). The ending *-ne* in *zenne* and *zene/zøne*, which probably represents a reduced pronunciation of *zenne*, may be etymologically a conditional form³⁹² or possibly the topicalizer *=ne*. The other two forms supplement *zen* with the negated equative $m\tilde{e}$; which may occur alone, as in *ze:men* (12.13) or in the conditional form, as in *zẽ:mene* (12.14).

³⁹² As the conditional marker attaches only to verbs, zen-ne may represent a reduction of the more complex zen menne [other NEG.EQU.PER-COND].

- (12.11) हॅन्-र्रेंग र्जुया भन्न प्रतीय र्जुया हे गुवया क्रेंन मुनया वेन भन्न गुवव हे ये वेन यावत भन्न tsøpo *bε*?. p'ja- $c\varepsilon$ =di=p'jate né:kor kjap t^hop ZENNE debate do-INF = DEMPH = ADVZR so sightseeing do receive EQU.NE otherwise mi-t^hop-k^hen bε no. NEG-find-NMLZ EQU.NE TAG.ASR 'Because of (participating) the debate (he) gets to do sightseeing. Otherwise it is not available, eh. (KNA kitchen discussion)
- (12.13) ત્રાંગ્રતો તર્યો દે સું સું જે બંદરયા દીય છે તે વાવય હવા બહાર થય ગયત સંગ પ્રાયમ સંગ પ્ય સંગ પ્રાયમ સંગ mù=ri godze kjokju ódi: p'ja-ti тí zen=tsu: 3SGF=AEMPH.GEN crooked that.GEN do-NF leader human other=PL.AGT mèmbo z**ɛ̃:mɛn**=to t^hõ∶-sa mù=rã: køntce: $t^h \tilde{o}$:-po 3SGF=AEMPH guilty see-2INF except otherwise=CEMPH see-NMLZ.SPAT mù=lo køn $m\dot{e}$:=co. 3sgf=dat blame NEG.EX.PER=AT 'She has become the culprit in the eyes of other people for no other reason than this crooked leader of hers. Otherwise, she is not to be blamed for anything, you know. (nga'i 'gan 4)
- (12.14) ขาง ซิรา ขารสาวร้า สานนานขัสง ขาสสา สู่ขาสาวั อูราสา สานนานขัสง สรสายขา รรรา อีสาร์ ขาสราซิ มีรณาสา มีรณาสา สู่ขาลาวัรา (गवित अत्र त रहे कें केंदि माया अहर आपत क्षा केंग राषा रहे के के केंदि k'ɛːsi? tam=di k^hando: sén $ts^h u$ -po *tc* '*uŋ-n*ɛ word=DEMPH dakini.AGT listen.HON be.able.to-2INF become-COND if nấː-di kjop-ne k^hando: sã:t^ha? tã: p'im-bo kiop dakini.AGT copper-rope send give-2INF do.HON-NF protect-COND protect ts^hu õĽ. zẽ:mene demo mù=i $k^h a = l \varepsilon$ be.able.to come otherwise she.demon 3SGF=GEN mouth=ABL t^ha :- $k^h\tilde{\epsilon}$: tca:so? $dup-o=tsu=l\varepsilon$ mi-õ: iron-life fulfil-2INF=PL=ABL be.released-NMLZ NEG-come 'If this word will become audible to the dakini and the dakini, sending a rope of copper, protects (you), (you) will be able to be protected. Otherwise, no one except those of ironbody will escape from the mouth of the she-demon.' (rna-gsung 12)

The following three connectors build on the word $j\hat{a}$: 'again, and', an independent form related to the clitic $=j\hat{a}$: 'too, even, yet, still'. The connector $j\hat{a}$: occasionally occurs alone to mark that the information presented in the clause is somehow added to the information in the previous clause (12.15).

(12.15) ଅନ୍ କୁନ୍ୟୁର୍ମ୍ ମିଷ୍ୟ ଶ୍ରା ତ୍ୟାର୍ଘ୍ ଅଂନ୍ୟିସାସ ଶ୍ରହ୍ୟାରି :

jãː	ge:py:=gi	k'utca=lo	màlɛp	p'ja-ne
again	king.GEN=AGT	2PL=DAT	bad	do-COND
'Again,	if the king did ba	d things to y	ou' (CY interview)

The connector $j\tilde{a}$ may be supplemented, similarly to the connectors presented above, by $-n\varepsilon$, which is likely a conditional marker or a topicalizer (12.16-17), or by $m\varepsilon n\varepsilon$, a conditional form of a negated equative (12.18-19). When used with negated verbs in both clauses, the connector $j\tilde{a}:n\varepsilon$ expresses meanings corresponding to English *nor* (which marks additional negated information), see (12.17).

(12.16) שאי איש, האמידי אמריאר אריאר שריא בישאר אישיאי איאיאיאי איאיאיזיי איאיאיאין איידי איאיאיאין איידי איאיאין איידי

 $te^{h}\varepsilon$:agjaze:-po $n\tilde{a}:=m\emptyset?$. $j\tilde{a}:n\varepsilon$ yàtci $làka=l\varepsilon$ oh.noelder.bothereat.HON-2INFdo.HON=URGor1PL.GEN hand=ABLze: $mi-l\tilde{\varepsilon}:-po?$ eat.HONNEG-be.good-2INF'Oh no, brother, please eat, by all means. Or is it not good to eat from our hands?' (Richhi20)

- nàtca=ne sátha? *rim-bo=l*ε nóme? $logiu? t^{h}o:-ce?$ 1PL=TOP distance be.long-2INF=ABL thoughtless story hear-INF mè?. mìndu?. *jã:nɛ p^ha:* ts^hu: mí: $dy = j\tilde{a}$ thither hither human.GEN travel=too NEG.EX.PER NEG.EX.SEN or 'We do not hear thoughtless stories from far away. Nor are there people traveling here and there.' (rnam-rtog 4)

nàtea? eí-ne тí ϵi -tsub=l ϵ làlo mí lèm sǿnam 1pl die-COND human die-CMPL.2INF=ABL some human good merit i*à∶-n*ε=di mí=lo mí-ly: $t^h op$ zanpo jãːmene good human=DAT human-body receive or.in.other.case EX-COND=DEMPH símter: j**ā:ne** bu? làp k^hor. animal or bug say rotate

'If we die, after a person's death, someone, if (s)he is a good person of good merit, that person receives a human body. Otherwise, (the person) transmigrates as an animal or a bug.' (KT discussion with TB)

The negated form of the verb $ts^{h}\varepsilon^{2} \neq \gamma$ 'stop, break off, discontinue', $mi-ts^{h}\varepsilon^{2}$, is used as a conjunction meaning 'not only (but also); morerover; in addition'. It can be used either as a looser connector, which does not combine two clauses into one sentence but provides logical cohesion (akin to English 'moreover, furthermore'), see (12.20), or it can be used with a nominalized and thus subordinated construction. The subordinated use is described later under adverbial clauses, see §15.9.1.

(12.20) वर्दे अत्र न देट अट मी सुम्र गवित मुनम में सम र्ये सुम्य में मान के सुम्य के स t'inzan=gi p'um pén kjap-kjap-o=tsu: di mèn-ne this NEG.EX-COND nowadays=GEN girl marriage do-RDP-2INF=PL.AGT $p\tilde{a}:d\tilde{\epsilon}:=di$ mi-tup. mits^hE? pã:dĩ: l*ù*:=ki kap NEG-deem.fitting moreover apron=DEMPH body=GEN apron cover teŋk^ha t'ã: cérpo=gi $k' \sigma t c^{h} \epsilon = t \epsilon i P (m - b \sigma)$ mànte^hico? p'øpo clothing=INDF EQU-2INF.GEN above most Tibetan and Sherpa=AGT lùksø: ĩ. k'øm-bø: tradition wear-2INF.GEN EOU.PER 'Otherwise, married girls nowadays do not want to wear pangden-apron. Moreover, in addition to pangden-apron being a clothing for the body, most Tibetans and Sherpas have a tradition of wearing (it).' (sbar-phung 93)

The connector *mits*^h ε ? may also occur sentence-initially following a demonstrative:

(12.21) دَجْ مَا عَجْ جِمَا عَلَى اللهُ عَلَى اللهُ اللهُ عَلَى اللهُ اللهُ اللهُ اللهُ اللهُ المُحْ

di **mits^h\varepsilon?** $p\dot{e}:=gi$ $mi-ts^{h}\varepsilon=di$ $n\dot{a}\eta\varepsilon a=lo$ this in.addition 1SG.GEN=GEN human-life=DEMPH inside=DAT 'In addition to this, in my life...' (KT life story)

A possible origin of $mits^{h}\epsilon^{2}$ as a nonfinal form $mi-ts^{h}\epsilon-ti(ki)$ is suggested by (12.22), which is the only recorded instance of a negated nonfinal converb in my data (the negated form is usually borrowed from the circumstantial construction ma-V-pa). A possible reason why negation is possible with a nonfinal form here is that through frequent use the form has lost the connotation of negatedness in the mind of the speakers and has instead developed semantics akin to 'in addition, adding to what was said'. (12.22) אַמישיי הקמימי אַבָּראיזי אָדיזיפירמיזי אָדיזיפיזאיזי אַזייזיפיזאיזי רי אָריאַרי אַרי

 $b\epsilon jyl$ $d\epsilon madzon=gi \ korzi=na \ mi-ts^h\epsilon-tiki$ $\eta a \ t\epsilon^hunt\epsilon^hun \ g\tilde{a}$: hidden.land Sikkim=GEN topic=LOC NEG-stop-NF 1SG small time 'Still continuing about the topic of the hidden land Sikkim, when I was a child...' (CY interview)

For the causality marking connector k'*ambjasene* (and its variants), refer to §15.4.1. The connector *dile* 'then' connects finite clauses. It expresses temporal sequence:

ĩ:. t^hartcim-po dile dendzõ: sé:da cé:da ηà dendzõ: 1sg Sikkim institute reach.end-2INF EQU.PER then Sikkim institute $tc^{h}o^{2}$ dok mjo:-pø: gjable doctrine read finish-2INF.GEN after 'I finished (studying at) the Higher Institute of Nyingmapa studies. Then, after finishing studies at the Higher institute of Nyingmapa studies...' (RB life story)

The connector $t' \epsilon n \epsilon$ 'then, in that case' has two main functions. It expresses a logical consequence of what was said before and it also typically indicates a change of speaker. In other words, when $t' \epsilon n \epsilon$ is used the speaker typically reacts to something that another person has said. By suggesting that the speaker has changed $t' \epsilon n \epsilon$ helps the listener to track who is speaking in a story. The connector $t' \epsilon n \epsilon$ typically occurs at the beginning of the clause (12.24) but after interjections (12.25) and discourse particles (12.26). It frequently co-occurs with the discourse particle $t\epsilon$ 'so' to form $t\epsilon$ $t' \epsilon n \epsilon$ or $t' \epsilon n \epsilon$ te 'so then, so in that case', see (12.26)

- (12.24) $\hat{r}_{,ax} \tilde{s}_{,ax} \tilde$
- (12.25) a) क्षे देवया वेंग्रा म का क्षे केंद्र का की

b) מעקאיאֿי לִיקאי שָׁקיאָי שַקרין
làso t'ene te'om-bo nấ:.
alright then go.HON-2INF do.HON
'Alright, in that case please (feel free to) go.' (Richhi 17)

(12.26) a) בَ' בَ'שָּמִי בֿ'יּצֿה' שַמִיצָּי בָּ' בָּרָיבָרִי שֵּׁשְוּבֵּוֹ שָּמִישִׁי שִּׁשִּׁשִ שִּׁשִ שִּׁשִ *tɛ t'ɛnɛ p'otsø: pʰam=tsu=lo t'ariŋ=rã: tɕʰadi pʰy:-gɛ,* so then child.GEN parent=PL=DAT today=AEMPH letter.HON offer-HORT *iŋ-ga?* EQU.PER-PQ 'So then let's today offer a letter to the children's parent, shan't we.' (Richhi 20)

Occasionally, *t'ɛnɛ* may precede the verb:

(12.27) ८ र्गेगा हे रा दे त्वया हा गो

t'a lokti=ra t'ɛnɛ ta-gɛ. now again=AEMPH then look-HORT 'Now in that case, let's look again.' (JDF axe story)

My written data has six instances of $t' \epsilon n \epsilon$, all marking a change in speaker within a dialogue. In spoken data, 16 out of 18 clauses with $t' \epsilon n \epsilon$ imply a speaker change. The two exceptions are given in (12.28) and (12.29) respectively. In the narrative example (12.28) $t' \epsilon n \epsilon$ seems to occur within the narrator's own meta-speech, not within the narrative's characters' dialogue. I am not certain whether the iteration of linguistic form denotes iteration of action or the speaker's hesitation.

(12.28) بَنَ مَسْتَابَجُ جَرَعَمَّ جُرَعَمَّ جُرَعَمَّ جُرَعَمَّ مَسَابَعُ جَرَعَمَّ مَسَابُعُ جَرَعَ مَتَ عَلَيْ اللَّهُ اللَّ واللَّاللَّهُ اللَّهُ اللَّهُ اللَّهُ اللَّهُ اللَّهُ اللَّهُ اللَّهُ اللَّ اللَّاللَّالَةُ اللَّهُ اللَّهُ اللَّهُ اللَّهُ اللَّهُ اللَّهُ اللَّهُ اللَّهُ اللَّالَةُ اللَّالِي اللَّالَةُ اللَّالِي اللَّالَةُ اللَّالِي الللَّهُ اللَّالِي اللَّالَةُ اللَّالَ اللَّالَةُ اللَّالَةُ اللَّالَ الللللَّالِللللَّالِي اللَّالَةُ الللَّالِي اللَّالِي اللَّالِي اللَّالِي اللَّالِي اللَّالَةُ اللَّالَةُ اللَّالِي اللَّالِي اللَّالِي اللَّالَةُ اللَّالُ اللَ اللَّاللَّاللَّالَةُ اللَّالَةُ اللَّالَةُ اللَّالَةُ اللَّالَةُ اللَّالَةُ اللَّالَةُ اللَّالَةُ اللَّالَ اللَّ

The other example, (12.29), is from a monologue, where $t' \epsilon n \epsilon$ introduces a new item in a lengthy list.

दे'त्रबा क्षेबार्गे'यर (scholar) र्हु' या क्वेप'क्वे' सेयागवा (welcome) र्जुबा दर्याबा मर्त् रे? intənæcənəl levl nàŋca k'atepti te ikspous p'ja international(Eng.) level(Eng.) inside how so expose(Eng.) do íŋ-ga? bo. t'ene skolar=tsu k'atepte **39**.-68 be.needed-INF EQU.NE.Q EQU.PER-PQ then scholar(Eng.)=PL how welkam p'ja go:-ce bo?welcome(Eng.) do be.needed-INF EQU.NE.Q '(We have to think) how to expose our language to the international level, isn't it? Then, how to welcome (international) scholars?' (DR discussion with KL)

The last monosyndetic connector is $\delta di = l\epsilon = to$ (that=ABL=CEMPH) 'rather', see (12.30) and (12.31).

- (12.30) & (a, c, b) and (a
- (12.31) هَتْجَ جَ مَعْنَ مَجَ مَعْنَ مَعْن مَعْن مَعْنَ مَعْن مَعْنَ مُعْنَ وَعْنَ مَعْنَا مُعْنَا مُعْنَا مَعْنَ مُ مُ مُعْنَ مُعْنَ مَعْنَ مَعْنَ مُعْنَ مُعْنَ مَعْنَ مَعْنَ مُنْ مُ مُعْنَ مَعْنَ مَع وَعْنَ مَعْنَا مُعْنَا مَعْنَا مَعْنَا مَعْنَا مَعْنَ مَعْنَ مَعْنَ مَعْنَ مَعْنَ مَعْنَ مَعْنَ مَعْنَ مَعْنَ مُ مَعْنَ مُعْنَ مَعْ وَعْنَ مَعْنَ مُ مَعْنَ مَعْنَ مَعْنَ مَعْنَ مَعْنَ مَعْنَ مَعْنَ مَعْنَ مَعْنَ مَعْ مَعْنَ مَعْنَ مَعْنَ مَعْنَ مَعْنَ مَعْنَ مَعْنَ م

12.3 Bisyndetic connectors

Bisyndetic constructions are formed by using the same conjunction twice, once in each clause. Table 12.2 lists bisyndetic connectors.

Table 12.2.	Bisyndetic clause connectors
-------------	------------------------------

$j\tilde{a}$: $j\tilde{a}$: or $=j\tilde{a}$: $=j\tilde{a}$: (=) $j\tilde{a}$: NEG.VERB(=) $j\tilde{a}$: NEG.VERB	ಬ ್ಬ್.	'bothand' (lit. alsoalso, eveneven) 'neithernor'
jäːnɛjäːnɛ ³⁹³	धन्देधन्दे.	'eitheror' (lit. oror)

When $(=)j\tilde{a}$: is followed by an affirmative verb in both clauses, the meaning is 'both...and', see (12.32).

³⁹³ The monosyndetic form $j\hat{a}:n\varepsilon$ up \hat{a} 'or' is used in the coordination of noun phrases, see §4.1.7.

jã: $k^{h}u = r\tilde{a}$: be? jã: mí=lo sà tcuk-o sà-u both 3SGM=REFL eat-2INF EQU.NE and human=DAT cause-2INF eat hE2. EQU.NE 'He both ate himself and fed (other) people.' (Class 9-10 grammar, 135)

When $(=)j\tilde{a}...(=)j\tilde{a}$ is followed by a negated verb in both clauses, the meaning becomes 'neither...nor', see (12.33), where the speaker explains the meaning of the word ' $di\eta$ ' mediocre':

(12.33) אקישהי אקי, מפתימישהי אקן

 $rap=j\tilde{a}$: $m e^{2}$, $t^{h}ama=j\tilde{a}$: me^{2} .supreme=evenNEG.EX.PERlast=evenNEG.EX.PER'(It) is neither first-class nor last (in quality).' (KN e)

Clause-initial use of $j\hat{a}:n\varepsilon$ 'or' in adjoining clauses expresses two options in a way similar to English 'either...or', see (12.34).

jã:ne jìk-len man-di-u-p'ja dik^ha òn-do mù=rã: or letter-answer NEG-write-2INF-ADVZR 3SGF=REFL here come-IPFV ηà=lo pám, jã:ne mú: gokor=to tã:-bo EQU.PER.ATTQ or 3SGF.AGT 1SG=DAT deception=CEMPH send-2INF mèn-gam? NEG.EOU.PER-ATTO 'I wonder whether she is coming here herself without answering (my) letter or whether she is perhaps not deceiving me.' (Richhi 149)

12.4 Summary remarks

This chapter described clausal connectors and how finite clauses are connected through them. It was shown that some clausal connectors may facilitate coordination-like linking (particularly *t'izã:* 'but, however') whereas others are looser cohesion-adding connectors (e.g. *t'ɛnɛ* 'then, in that case', which typically implies that the speaker has just changed). Denjongke was shown to have both monosyndetic and bisyndetic connectors.

13 Constituent-modifying clauses

This chapter continues the discussion on clause combining by dealing with constituent-modifying clauses. As detailed in the introduction (§13.1), constituent-modifying clauses can be divided into relative clauses (§13.2), correlative clauses (§13.3), noun complement clauses (§13.4) and postposition complement clauses (§13.5).

13.1 Introduction

Thompson et al (2007: 238) divide subordinate clauses into three categories: 1) complement clauses, which function like noun phrases, 2) relative clauses, which modify nouns, and 3) adverbial clauses, which modify the verb complex or the entire clause. The same division, with one modification, is followed here. Complement clauses and adverbial are discussed in §14 and §15 respectively, while Thompson et al's category "relative clauses" is extended into "constituent-modifying clauses," which covers relative clauses, noun complement clauses and postposition complement clauses. The reason for this modification is that these three types of clauses are morphologically identical in being genitive-marked nominalized clauses, see Table 13.1 (relative clauses formed with the nominalizer $-k^h \tilde{\epsilon}$, however, are not genitive marked). The nominalizing suffixes in Table 13.1 are the infinitive markers $-\epsilon\epsilon(2)$ and -po/bo and the nominalizers $-k^h \tilde{\epsilon}$ and -sa.

Constituent-	Noun-modifying	Relative clauses	$-k^{h}\tilde{e}$:	No genitive
modifying	clauses		-po/bo + GEN	Genitive-marked
clauses			-sa + GEN	
		Noun complement	-po/bo + GEN	
		clauses	$-\mathcal{E}\mathcal{E} + \text{GEN}$	
	Postposition-modif	ying clauses/	-po/bo + GEN	
	postposition complement clauses		$-\mathcal{E}\mathcal{E} + \text{GEN}$	

Table 13.1. Constituent-modifying clauses

Relative clauses and complement clauses are distinguished by a syntactic criterion: in relative clauses, the modified noun functions as an argument (whether core or peripheral) in the modifying clause whereas in complement clauses it does not. The modifying clause types are introduced in examples (13.1-5). A fuller discussion follows. The RCs and complement clauses are given in brackets. The head noun phrase of the RCs, noun complement clauses and postposition complement clauses is underlined.

Relative clause

 $\begin{bmatrix} t^{h}amtce = ki & n\delta: & ts^{h}u - p\emptyset: \end{bmatrix} \qquad \qquad \underbrace{t'ep} \\ all = AGT & buy & be.able.to-2INF.GEN & book \\ \underbrace{book} [that everybody can buy]' (KN e) \end{bmatrix}$

Clausal complement clause

Noun complement clause

(13.3) \vec{r}_{τ} (\vec{r}_{τ} (\vec{r}_{τ}) (\vec{r}_{τ})

Postposition complement clause with -po + genitive

(13.4) देया पर्दे वय्या ठू गीया हे क्याया परि देवाया

 $[t' \varepsilon p = di$ $t^h amt \varepsilon \varepsilon = ki$ $n \delta$: $ts^h u - p \phi$: $t' \phi n lo$ book=DEMPHall=AGTbuybe.able.to-2INF.GENfor.the purpose.of'For the purpose[that everybody can buy the book]' (KN e)

Postposition complement clause with $-\epsilon\epsilon$ + genitive

(13.5) วิจา คริ ยมพาธราฏิพา ด้า สูงาพางคา รัสาน้า

 $\begin{bmatrix} t' \varepsilon p = di & t^{h} amt \varepsilon \varepsilon = ki \ po : & ts^{h} u : -\varepsilon \varepsilon = ki \end{bmatrix} \qquad \frac{t' \omega n lo}{\text{for.the purpose.of}}$ book=DEMPH all=AGT buy be.able.to-INF=GEN for.the purpose.of '<u>For the purpose</u> [that everybody can buy the book]' (KN e)

In many languages, verb forms which modify nouns are termed participles. In Tibeto-Burman languages, however, participles and nominalizers tend to merge together, i.e. the same form may be used for both noun modification and argument nominalization (e.g. Chantyal nominalizer -wa in Noonan [1997: 375-377]). If the decision between an analysis as participle or nominalizer is made based on the primary function, which is the main criteria used by many typologists (Shagal 2016: 31-32), it may be argued that the markers -po/bo and -sa are nominalizers³⁹⁴, because their citation forms are used for nominalizing clausal arguments whereas their noun-modifying (more participial like) uses have to be further marked for genitive, i.e. the modifying uses are extensions of the uses as clausal arguments. The marker $-k^{h}\tilde{\epsilon}$, however, is not genitive marked when functioning as a noun modifier, and therefore it is not as clear whether noun-modifying or argument nominalizing uses are primary.³⁹⁵ However, the present productive use of $-k^{h}\tilde{\epsilon}$ in derivativation favours an interpretation as a nominalizer, see §3.2.4.3 (similarly DeLancey 2002 on the cognate of $-k^{h}\tilde{\epsilon}$: in Lhasa Tibetan). In summary, all the markers used in constituent modification are here analyzed as nominalizers, although when used in nominal modification they may be functionally termed participles (forms with $-k^{h}\tilde{\epsilon}$) or participal constructions (genitivized forms with -po, -sa and $-\epsilon\epsilon$?).

Constituent-modifying clauses are now discussed in the same order they occur in Table 13.1.

³⁹⁴ The suffix *-po* is further categorized as infinitive marker because it creates a nominalized form that refers to action itself, whereas *-sa* creates a form which refers to a place where the action is done.

³⁹⁵ Sandberg (1895: 38) describes "*k*^hen" as participle marker.

13.2 Relative clauses

Relative clause (henceforth RC) in Denjongke is here defined as an embedded clause which modifies a noun phrase in the main clause and which shares a common argument with the main clause (the shared argument need not be in the same syntactic role in both clauses).³⁹⁶ The modifying function distinguishes RCs from (clausal) complement clauses. A complement clause is similar to a RC in involving nominalization but it differs from an RC in that, instead of modifying an argument in the main clause, the complement clause *is* one of the arguments of the main clause. As a sign of modifying function, the pre-head relativizing nominalizer *-po* is genetivized as *-pø*:, whereas clausal complement clause marking *-po* is not genitivized. The second feature in the definition of an RC, the requirement for a common argument between the RC and the main clause, on the other hand, excludes from the definition noun complement clauses (§13.4) and postposition complement clauses (§13.5).

Denjongke employs two basic strategies for forming RCs. One is the typically Tibeto-Burman strategy of appending a nominalized clause to a noun. The other option is the typically Indo-Aryan strategy of having an RC with a relative pronoun followed by the main clause with a presumptive demonstrative.³⁹⁷ This latter use can be called a correlative (or co-relative) construction, because the two clauses can be argued to instantiate coordination rather than embedding (Dixon 2010b: 356). In Denjongke correlative clauses, interrogative pronouns are used in place of separate relative pronouns. As pointed out by Genetti (1992: 408), who found a similar (typically) Indo-Aryan relativizing strategy in Dolakha Newar, this latter strategy "is probably due to contact influence." Nominalized relative clauses are the topic of this section, while correlative clauses are described in §13.3.

Nominalized RCs in Denjongke are mainly externally headed or headless but one example of internally-headed clauses was also found. Relativization is achieved mainly through the second infinitive -po/bo and the nominalizer $-k^{h}\tilde{\epsilon}$, and more rarely through the spatial nominalizer -sa and the quantitative nominalizer $-ts^{h}\epsilon^{2}$ ("as much as is x-ed"). RCs usually precede the head noun but may occasionally also follow it. Pre-head RCs formed with -po/bo and -sa are marked as noun modifiers through genetivization. Post-head RCs are generally not genitivized and thus function syntactically as appositions. RCs with $-k^{h}\tilde{\epsilon}$: are not genitivized even in pre-head position.

Table 13.2 summarizes the various forms that pre-head RCs take with -po/bo, $-k^{h}\tilde{\epsilon}$: and -sa. The empty cells represent forms that do not occur in my data (but could, perhaps, be possible).

³⁹⁶ See Andvik (2010: 238) for a similar definition of RC in Tshangla.

³⁹⁷ This is a subcase of Payne's (1997: 437) "pronoun retention" strategy of forming RCs.

-khĩ:	-po	-sa	Temporal/aspectual function
VERB- <i>k^hẽ</i> :	VERB- <i>pø:</i>	VERB-SØ.	neutral (although unlikely future)
VERB jø:-k ^h ĉ:	VERB j <i>ò-pø:</i>		resultative (dynamic verbs), progressive (stative verbs)
	VERB-RDP- <i>pø:</i>		iterativity, habitualness (past or present)
	VERR(-ti) za: jờ-pơ:		resultative
VERB- <i>zin p'ja-k^hẽ</i> . ³⁹⁸	VERR-zin-pø:		progressive
	VERB-zin j <i>à-pø:</i>		progressive
	VERB <i>-zin dø: jø-pø:</i> (+time word)		progressive
	VERB- <i>(ti) dø: jø-pø:</i> (+time word)		progressive
	VERB-INF <i>mèː-pøː</i>		future ('which will not be x-ed')

Table 13.2. Temporal and aspectual values in pre-head RCs

As seen in Table 13.2, $-k^{h}\tilde{\epsilon}$:, -po and -sa all may attach directly to the verb, forming a construction whose temporal perspective is determined by the context. Temporal and aspectual values can be explicitly expressed by various constructions ending in the nominalized (genetivized) existential auxiliary $j\partial$?. The infinitivizer -po/bo occurs in the greatest number of constructions, e.g. progressive VERB-zin $(d\sigma)$: $j\partial$ - $p\sigma$:, perfect VERB-RDP- $p\sigma$:, and resultatives VERB $j\partial$ - $p\sigma$: (expressing, however, progressivity with stative verbs such as $d\phi$? 'sit, stay') and VERB za: $j\partial$ - $p\sigma$:.

The head word of an RC is usually a noun, but may also be a personal pronoun (13.6) or a demonstrative such as *di* 'this' (13.7), *ódi* 'that' (13.8) or *ódɛm=tɛiʔ* 'one like that' (13.9).³⁹⁹ Note that the personal pronoun in (13.6) refers to second person plural. In Kham (Tibeto-Burman, Nepal), for instance, "the referent of the subject relative clause is always 3RD person" (Watters 2002: 201). In (13.7), the nominalized form $làp-k^h\tilde{\varepsilon}$: does double duty as a complementizer to the clause on its left and as a nominalized modifier to the head pronoun on its right.

(13.7) वुप्यमा वेंत्तावावा वर्ता र्हु कें ना कें मना मुनमा तुमा

[nà=lɛ	$\dot{\tilde{o}}$:- $k^h \tilde{\epsilon}$:]	<u>di=tsu</u> =to	t'a	ļoke	kjap-dỹ:
here=ABL	come-NMLZ	this=PL=CEMPH	now	Lhoke	strike-SIM
'But now v	when <u>those</u> [wł	no come from here	e] spea	k Lhoke	' (RL e)

³⁹⁸ This construction employs a combination of the progressive marker *-zin* and the verb p'ja 'do', see (13.12).

³⁹⁹ The demonstrative adjective/adverbial $\delta d\epsilon m$ 'like that, such' is here nominalized with a strategy similar to the English strategy for the expression 'one like that', i.e. employing the word for 'one', *tei*?, which also functions as an indefinite marker.

- (13.8)୍ର ଧ୍ରସାବ୍ୟାଦ୍ୟି ଜନ୍ମି ଅନ୍ମାଇଟିମ ନ୍ମା ନ୍ମୋଇସ୍ କୁମୁୁୁ ଅନ୍ମାର୍ଯ୍ୟ କୁମ୍ମାର୍ଯ୍ୟ କୁମ୍ମାର୍ଯ୍ୟା ସ୍ଥାନ୍ ଅନ୍ମାର୍ଯ୍ୟା କୁମ୍ମାର୍ଯ୍ୟା କୁମ୍ମ $[tc^huk=lo=di]$ $d\tilde{a}$: $\eta at \epsilon a$? $l \phi - m \tilde{\epsilon}$: = lo = dimánts^hø? Nepali=DAT=DEMPH more and 1PL Lhopo-Lepcha=DAT=DEMPH $phints^{h}i$? $lap-k^{h}\tilde{e}$: ódi $tc^h \phi g \varepsilon = gi n \tilde{a} - n \tilde{a} - b o$ minduk= εo =la. say-NMLZ that king=AGT do-RDP-2INF NEG.EX.SEN=AT=HON less 'The king did not do that (thing) [which is to say more (was to be given) to the Nepali and less to the Lhopos and Lepchas].' (CY interview)
- (13.9)้สิส : ฮกาลักา สาขาวกับ สาขาวกา อู่นาทั่า นลการ์า รัฐกาสนอง เสียวสาร์สาขายกา นักา nấ:-di màlɛp [$pin tc^{h} \epsilon me^{2}$] màlep=ro tsuko $d\emptyset:-k^{h}\tilde{\epsilon}:$] day ceaseless bad=AEMPH do.HON-NF stay-NMLZ bad only *ódɛm=tɕi=jã*: jờ?. like.that=INDF=too EX.PER 'There are also those of such kind [who live committing only bad (deeds) upon bad (deeds) every day].' (SS Proverb explanation)

The following sections present a separate treatment for RCs formed by $k^{h}\tilde{\epsilon}$; -po/bo, and -sa respectively.

13.2.1 Relativization by $-k^{h}\tilde{\varepsilon}$:

The nominalizer $-k^{h}\tilde{\varepsilon}$: can be used in both headed (§13.2.1.1 and §13.2.1.2) and headless relative clauses (§13.2.1.3). The nominalizer $-k^{h}\tilde{\varepsilon}$: is the most usual nominalizer when the head noun is coreferential with the actor/agent in the RC. However, as shown below, $-k^{h}\tilde{\varepsilon}$: may also occur with patient and locative arguments. The construction with $-k^{h}\tilde{\varepsilon}$: is in itself neutral with respect to time-reference, which has to be deduced from the context.⁴⁰⁰ The RC with the nominalizer $-k^{h}\tilde{\varepsilon}$: may occur either before the head noun of the RC (§13.2.1.1) or after it (§13.2.1.2). With the indefinite expression $ka:(=ki)=j\tilde{a}$: 'whoever', it may also form an internally headed RC (§13.2.1.4). The term "pre-head RC" used here means that the RC precedes the head noun. Thus, "pre-head RC" corresponds in meaning to "post-headed RC" used by some authors.

13.2.1.1 Pre-head RCs

Typically the nominalizer $-k^{h}\tilde{\epsilon}$: occurs before its head noun. The head noun of the RC may be an actor, a patient or a locative.

Head noun as RC actor

The head noun functioning as the actor of an RC formed with $-k^{h}\tilde{\varepsilon}$: may be either animate (13.10) or inanimate (13.11).

(13.10) هَتْحَ بَيَّا عَيْ² مَتْ جَخَبَ مَتْجَ مَعْمَ الْمَتْ الْحَالَةُ عَلَى اللَّهُ عَلَى اللَّهُ عَلَى اللَّهُ عَلَى اللَّهُ عَلَى الْحَالَةُ عَلَى اللَّهُ عَلَى الْحَالَةُ عَلَى الْحَال وَعْلَى الْحَالَةُ عَلَى الْحَالَةُ عَلَى الْحَالَةُ عَلَى الْحَالَةُ عَلَى الْحَالَةُ عَلَى الْحَالَةُ عَلَى الْ وَعْلَى الْحَالَةُ عَلَى الْحَالَةُ عَلَيْكَ عَلَى الْحَالَةُ عَلَى الْ

⁴⁰⁰ Sandberg (1895: 38) refers to " $k^h en$ " as "the participle" and notes that "we find no difference in expression between the present and the past participle. The context must determine the time to the English speaker."

(13.11) कु' वृत्तान क्षुमामापवा तरी क्षारे तरी

 $[te^{h}u \ nane a \ lum - k^{h}en = di]$ water inside fall-NMLZ=DEMPH axe=DEMPH 'the <u>axe</u> [that fell into the water]' (JDF roof discussion)

Note that in (13.11) the demonstrative-emphatic = di occurs two times, both after the RC and after the head noun.

Example (13.12) illustrates a specific construction with the verb p'ja 'do', enabling explicit progressive marking:

(13.12) टु'पवित्र' र्ज्रुषायत्र' र्पे र्छ' वट्टे' व्य

[$\eta \hat{u}$ -zin p'ja- $k' \tilde{e}$:] $\underline{p'otso} = di = lo$ weep-PROG do-NMLZ child=DEMPH=DAT 'to the <u>child</u>, [who was weeping]' (RS pupil joke)

Head noun as RC patient

The head noun may also be a patient argument in the RC.

(13.13) नुम्राकु'त्र' ह्यूया'यायत्र' कु'

 $\begin{bmatrix} p'umte^{h}u = na & l\acute{u} - k^{h}\widetilde{e}: \end{bmatrix} \xrightarrow{te^{h}u}$ Bumchu.festival=LOC pour-NMLZ water '<u>Water</u> [that is poured at Bumchu-festival]' (KLT Bhumchu video)

Example (13.14) shows that a genitive modifier (here $c \partial = i$) occurs closer to the head noun than the RC, which modifies the same head noun.

(13.14) $\eta_{\overline{a}i'\overline{a}i'} \stackrel{\text{gen}}{\to} \eta_{\overline{a}i'} = \frac{1}{2} \frac$

A frequent context in which an RC with $-k^{h\tilde{\epsilon}}$ is used for a patient argument is with verbs of saying (*làp* 'say', *súŋ* 'say [hon.]', *sé* 'say, be called'), referring to what items are 'called':

(13.15) $\widehat{\pi}_{i} \widehat{\mathbb{B}}_{i} \operatorname{eqr}_{i} \operatorname{eqr}_{i} \widehat{\mathbb{A}}_{i} \widehat$

Head noun as RC locative

In (13.16), the head noun is a locative argument in the RC (board on which something is written). The RC, on the other hand, modifies the patient role in the main clause.

(13.16) विन्तं योया थाया दी आवत पन्तं योहेया यहना छा

 $k^{h}o\eta = gi$ [*jìgi t*'*i*-*k*^h*ɛ*:] <u>*pã*:lɛp=tɛi?</u> nấ́:-di 3SG.HON=AGT letter write-NMLZ board=INDF request.HON-NF 'He requested for a <u>board</u> [on which to write letters] and...' (KT e)

13.2.1.2 Post-head RCs

RCs marked by $-k^{h}\tilde{\epsilon}$: may also occur after the head noun. In the three examples (13.17-19) below, the head nouns take an actor role in the RC. Based on these three examples, therefore, it may be hypothesized that post-head RCs with $-k^{h}\tilde{\epsilon}$: only allow the relativization of the highest ranking member of the relativization accessibility hierarchy, the subject (Keenan & Comrie 1977). Note that the adjectival modifiers in (13.17) and (13.18) occur closer to the head than the RC.

- (13.17) هَن طعر بهرت طي تعامل مَن نَشَر عَالَي اللهُ عن الله من المعالي (13.17) المَن عن الله من المعالي (13.17) (13
- (13.18) ରିଂगାର୍ଟ୍ ନହାଁ ଭିଷା ନିଷା ସେନ୍ତିଆ ସମ୍ଭ ସିଷା ହାରି କାର୍ଯ୍ୟ କି

<u>mi</u> <u> $zen</u> [doli? ce:-k^h\tilde{e}:]=tei?$ p'a:mi zo-ti man another custom know-NMLZ=INDF mediator make-NF '<u>Another man</u> [who knows the (wedding) customs] is made a mediator and....' (sbarphung)</u>

(13.19) ริ. ซ.ส. นร. ซุเลร์. ซุเลร์. สุนามส. นร.ซ. นุร. สินาส. นร. ลินาส. นร. สินาส. นิยา.

 $t\varepsilon$ óna=diódi $[p^h o k \emptyset$:p'a:mi $\partial p \cdot k^h \tilde{\varepsilon}:=di$]so there=DEMPHthatgroom.side.GENmediatorcome-NMLZ=DEMPH $tc^ha:p^hy:$ $k^h \varepsilon: l \tilde{\varepsilon}:$ kjap-tikiprostrationincantationdo-NF'So there the one[who comes as the groom's mediator] does prostrations and incantation(and)...'(SGD wedding customs)

13.2.1.3 Headless RCs

The clause nominalized with $-k^{h}\tilde{\epsilon}$: typically describes an animate, usually human, referent ('the one who does x'). Because the animate referent is presumed, $-k^{h}\tilde{\epsilon}$: naturally forms headless relative clauses which "themselves refer to the noun that they modify" (Payne 1997: 433). Typically the implied head noun is the actor in the RC (13.20-23), but other semantic roles are also possible, as shown by the patient role in (13.24) and oblique in (13.25).

Implied head noun as RC actor

(13.20) ผสมสงาณ์ จาลุสาสงามเกลา พิสา ซาร์ เลาร์ เจาสา

[$ts^{h}am = lo$ $zu:-k^{h}\tilde{e}$:] $k\epsilon:p$ $b\epsilon$? áno=la:. retreat=DAT sit.HON-NMLZ many EQU.NE grandmother=HON '[Those who sit in (meditation) retreat] are many, grandmother.' (PTB SM kitchen)

- (13.21) $\hat{\neg}_{i=\pi}$ a $\alpha_{i=\pi}$ a $\hat{\alpha}_{i=\pi}$ a

Implied head noun as RC patient

(13.24) [adzo=la:=gi ka $n\tilde{a}:-k^{h}en=di$ $n\dot{a}:$ da:=di $n\dot{a}:$ $k^{h}e:len-ee$ grandfather=HON=AGT word give.HON-NMLZ=DEMPH I.AGT accept-INF $\tilde{i}:$. EQU.PER 'I accept [what the grandfather (=you) is saying].' (Richhi 49)

Implied head noun as RC oblique

(13.25) $\exists_{a_1} \leq a_2 \leq a_1$ $\exists_{a_2} \leq a_2 \leq a_1$ $\exists_{a_2} \leq a_2 \leq a_2$ $[t^hinle=ki \quad d\tilde{a}: \quad logju? \quad c\dot{e}\cdot k^h en = di] \quad b\varepsilon - ka?$ Thrinley=AGT yesterday story tell-NMLZ=DEMPH EQU.NE-PQ 'Is (she) [the one about whom Thrinley told the story yesterday]?' (Richhi 27)

The argument roles taken by the clauses with $-k^{h\tilde{\epsilon}}$: in the main clauses above are equative copula subject (13.20), dative-locative-marked possessor in a possessive/locative sentence with existential copula (13.21), A(ctor) argument (13.22), genitive modifier of a P(atient) argument (13.23), non-marked P(atient) argument (13.24), copula complement⁴⁰² (13.25). Complement marking $-k^{h\tilde{\epsilon}}$: can

 $^{^{401}}$ Note that the complement clause is here, similarly to correlative clauses (see §13.3), followed by a demonstrative. The difference of (13.21) to correlative clauses is that, unlike in correlative clauses, the modifying clause does not have a questions word which is coreferential with the demonstrative.

⁴⁰² The terms "copula subject" and "copula complement" are from Dixon (2010b). Copula subject is the first argument and complement the second argument of a copula.

occur in various cases, as shown by the dative-locative in (13.21), agentive in (13.22) and genitive in (13.23).

The nominalized forms of the verbs of saying $lap - k^h \tilde{\epsilon}$: $s\dot{\epsilon} - k^h \tilde{\epsilon}$: and $s\dot{u}\eta - k^h \tilde{\epsilon}$: frequently form headless RCs. The forms lap and $s\dot{u}\eta$ occur as productive verbs, whereas $s\dot{\epsilon}$ in my data is used only in the nominalized construction $s\dot{\epsilon} - k^h \tilde{\epsilon}$: and as the quotative $=s(\epsilon) \sim s(i)$. The non-honorific forms $lap - k^h \tilde{\epsilon}$: and $s\dot{\epsilon} - k^h \tilde{\epsilon}$: tend to follow a noun and express the meaning "the one which is called" (13.26-27),

(13.26) a) אַרָּקָיָקָאָק אָרָיאַרָאָ אָרָאָיאַרָאָ אָרָאָיאַרָן (*thapon làp-khɛ̃:=tsi?*) *jobbɛ?*. performer say-NMLZ=INDF EX.NE 'There's [someone who is called [thapõ:]].' (SGD wedding customs)

b) גַזן היה אומק קאימי אק אימי אק אק אומן
[t'u? làp-kʰɛ̃ː] nàmlo man-za go?.
poison say-NMLZ ever NEG-eat be.needed
'One should never eat [(the thing) called poison].' (KN e)

(13.27) הישהי אישומש (13.27) הישהי (13.27)

The headless relative use of $sin_{j}k^{h}\tilde{\epsilon}$: may refer to the content of what is being said, as in (13.28), or to the person who is saying something, as in (13.29).

(13.29) अप्याया भ्रुवायमावा परिया देवा परिया होवा वर्षा या साम्रा के स

[$m \dot{a} la^2 m \dot{e} y k^h \ddot{a} := na k^h ik t \epsilon' \phi n s \dot{u} y - k^h \ddot{\epsilon} :]$ $n \dot{a} : p' u sim$ quickly hospital=LOC lead go-HON say.HON-NMLZ here younger.sister $b \epsilon = \epsilon o$. EQU.NE=AT '[(The one) who told (us) "take (him) quickly to hospital"] is the sister here.' (Richhi 12)

13.2.1.4 Internally-headed RCs

In an internally-headed RC, the head word occurs within the RC, not outside of it (Dryer 2013). Denjongke can use an internally headed RC-construction at least with the indefinite expression $ka:(=ki)=j\tilde{a}:$ 'whoever' (see §6.3.2), as shown in (13.30).

 $[\underline{mi} \quad \underline{ka:=ki=ja:} \quad n\dot{a}: \quad \dot{o}\eta \cdot k^h \tilde{\epsilon}:] \qquad k^h u = lo = j\tilde{a}:^{403} \qquad du\eta \quad j\dot{a}.$ human who=AGT=even here come-NMLZ 3SGM=DAT=even beat TAG.ASR '[<u>Whosoever</u> person comes here] even beat him, eh.' (KT animal story)

13.2.2 Relativization by -po/bo

The II infinitive marker -po/bo can form both pre-head and post-head RCs. Pre-head clauses are typically marked as noun modifiers by genitivization, although with reduplicated roots genetivization seems optional, see (13.32) and (13.35) below. Post-head clauses are generally not genetivized but their end is marked by the demonstrative-emphatic =di or the plural marker =tsu. Similarly to Lhasa Tibetan, -po/bo is generally used when the "head noun is coreferential with a non-actor NP in the RC" (DeLancey 1999: 234). The actor role is typically marked by the nominalizer $-k^{h}\tilde{\epsilon}$; although it also occurs in other roles. The ensuing discussion addresses pre-head and post-head RCs marked with -po/bo.

13.2.2.1 Pre-head RCs

The majority of RCs occur before the head noun. The examples here are categorized according to the semantic role that the head word takes in the RC.

Head noun as RC actor

Although RC actor role is typically marked with $-k^{h}\tilde{\epsilon}$; the nominalizer -po/bo may also suffix to RC verbs where the modified noun is in the actor role. In (13.31), the head noun ($\underline{mi=tci2}$) is coreferent with the actor of the intransitive verb 'come' in the RC. The RC modifies the copula complement of the main clause.

(13.31) קריבי האי איצר ליצ קמי לרימתי הי בי בי בי אי אי (Buddhist) מריאי אידיאלי ארימי ארימי אי משמין אאן מריאי אי משמין אין $t'yts^h = l\epsilon \eta \dot{a}$ t'ã:pu nè: ádzo podzo [budis jã:ne long.ago grandfather forefather time=ABL 1SG Buddhist(Eng.) my or nầːpøː $tc^h \phi := ki$ nànca=lo òm-**bø**?] ĩ. mí=tci? insider.GEN teaching=GEN inside=DAT come-2INF.GEN human=INDF EQU.PER 'From bygone times of my grandfathers and forefathers, I am a man [who came within the Buddhist or insiders' religion].' (KT life story)

In (13.32), the head noun functions again as the actor of the RC. Note that the reduplicated verb stem is not genitivized.

(13.32) हे' मरहते' सुमार्थे इंनुईनुर्ये र्वनुर्ये वनुर्धे

te	[ŋàtɛi	<i>ỳ</i> :=lo	dø:-dø:- po]	<u>pʻøbo</u>	$di=tso^{404}$
then	1pl.gen	place=DAT	stay-RDP-2INF	Tibetan	DEMPH=PL
'Ther	n the <u>Tibeta</u>	<u>ns</u> [who live	in our place]' (LA Intro	to Lachung)

⁴⁰³ The use of $=j\tilde{a}$: may be a mistake, because there is nothing in the context suggesting that there is an additional object for beating (which would be the natural reading of $=j\tilde{a}$: here).

⁴⁰⁴ The plural form =*tso* in the northern village of Lachung resembles the Central Tibetan plural form =*ts*^{*h*}o.

Example (13.33) presents another example of an actor argument marked with *-po/bo* in an intransitive clause. The example is a proverb and may thus embody a form (perhaps influenced by Classical Tibetan), which is not preferred in current spoken language.

(13.33) ज्ञ' अ'र्ह्येन्'पति' के' ज्ञ', ब्रेतु' गठिंग' र्द्धेग'ग' जठा तया कर्या कर्र्येन्'पति' के' तया, हेकप्प' भ' पषका ma-mjõː-**pø**ː] [sà тí sà, NEG-experience-2INF.GEN human eat eat $p \in u = t \in i^2$ $p^h j o ka s a m$, snot=INDF taste food [gju ma-mjõ:-**pø**:] тí gju, human go NEG-experience-2INF.GEN go t^hempa là: sám threshold pass think 'A man [who is not acquainted with eating] eats and finds the taste of snot (like) food. A man [who is not acquainted with walking] walks and thinks a threshold is a mountain pass.' (UT proverb)

Head noun as RC patient

In the following two examples, the head noun functions as the P(atient) argument of the RC. Note that the reduplicated stem in (13.35) is not genitivized.

(13.34) אוֹזיגֿי האָרְיִיגוֹ אוֹשָאַי אָרָירָי גּאָקייזיאַזען [kɛ:p ɛɛ́-pø:] <u>lògju</u> bɛ? t'a iŋ-ga=la. a.lot tell-2INF.GEN story EQU.NE now EQU.PER-PQ=HON 'Now (it) is a <u>story</u> [that is much told], isn't it.'(PT kitchen)

(13.35) هَٰ حَرَّ خَرَخْخُ عَرَّ عَرَيْخُ حَرَّ عَرَيْ مَرْجُ عَرَيْ عَرَيْ (medal) خَرْ عَمَرْ *ódi* [p'otsa=tsu t^hop-t^hop-o] <u>medal=tsu</u> be?. that child=PL receive-RDP-2INF medal=PL EQU.NE 'Those are <u>medals</u> [that the children have received].' (PD altar room video)

Head noun as RC locative argument

In the three examples below, the head noun is semantically the locative argument of the RC, expressing where the action denoted by the nominalized verb took or takes place. In (13.36) and (13.37), the RC functions as the copula complement in a copular clause where both the copula subject and copula itself are elided. In (13.38), the RC modifies a noun which functions as a complement of a locative postposition.

(13.36) มส์มสา พลุญพาพลุญพาพลิ พลุพา ชี้ เช่

 $[ts^{h}am zu:-zu-po:]$ <u>né:</u> $ts\tilde{a}:ts\tilde{a}:$ seclusion sit.HON-RDP-2INF.GEN site purely '(it's) a <u>site</u> [that is purely for sitting in solitary meditation]' (SM kitchen) (13.37) สารา ริสารัาซิ ๆ ๆ สม พอสม านุลุขุมานุลขุมานุลา

[guru rimputc^h ε t' \tilde{a} :pu ts^ham zu:-zu-po:] <u>né:</u> guru Rimpoche long.ago seclusion sit.HON-RDP-2INF.GEN site '(it's) a <u>site</u> [where Guru Rimpoche used to sit in solitary meditation long ago]' (SGD cave story)

(13.38) वर्दते ज्यामें येन्यते दे हुझ देया ज्या क्षें क सुवा हेया हे में में में में के की के की की की की की की

di: [làko jề-**p** θ :] <u>p'otɛum</u> $\partial:l\varepsilon$ óna ɛý: hok-tiki t'õ: this.GEN hand EX-2INF.GEN lower.stomach below there a.bit scratch-NF hole t ϕ n.

tear

'Scrathing a bit, tear a hole below this <u>lower stomach</u>, [where the hand is].' (spoken by a marten from within a dead elephant's body to a tiger outside) (KT animal story)

Head noun as copula subject of an existential RC

In (13.39), the head noun is the copula subject of the existential RC, and the RC modifies the copula subject of the existential main clause (in which the existential is elided).

(13.39) $a_{\nabla \nabla} \cdot [a^{*}a_{\nabla} \cdot a^{*}a_{\nabla} \cdot a^{*}a_{\nabla}$

13.2.2.2 Post-head RCs

RCs that occur after the head noun are less frequent in my data than those preceding the head. Unlike pre-head RCs, post-head RCs with -po/bo do not require genetive marking. Instead, they are typically marked by a final demonstrative-emphatic =di, see (13.42). Case marking of the noun phrase occurs after the post-head RC, see (13.40) and (13.41). Note that the reduplicated verb *kjap-kjap-φ* in (13.41) occurs in the genitive not to mark relativization but to make the form amenable to agentive marking.

1 1 / 10 0		· • • •	/	
$p\varepsilon = na$	hindu=i	<u>p`um</u> [pén	ŀ	kjap-kjap- o]=lo
example=LOC	Hindu=GEN	girl wedding	g (do-RDP-2INF =DAT
maŋgalsutra		ta:-ce?	jờ:-	-ne
mangalsutra.n	ecklace(Nep.)	append-INF	EX-	-COND
'For instance,	if there is (the	tradition of) ty	ing t	the mangalsutra-necklace on a Hindu girl
[who has been	n married]' (sbar-phung 88))	-

(13.41) 🕅 🕷 🛪 ५८.५५ ५ ५००४ । पार्चन वायोंका मुनका मुनका मुनका रात्रि मीका आ हेका हैया झाया का प्रान्य का प्रा óna t'ã:pu k^hatsã: sáiv? $[kjap-kjap-\boldsymbol{\theta}] = gi$ hale there long.ago before some.time.ago earthquake strike-RDP-2INF.GEN=AGT átem=tci? tsawa ram du-ke. a.bit=INDF foundation destroy EX.SEN-IN 'There before some time ago an earthquake, [which struck], destroyed the foundation a bit (I saw).'(DB day trip)

Post-head RCs seem more appositive in nature than pre-head RCs and can convey meanings similar to English unrestricted RCs. In (13.42), the post-head relative clause presents information that the speaker already knows. There are no other camels to be contrasted with than the ones mentioned in the RC. Therefore, the post-head RC here is more descriptive of the content of the head noun than it is restrictive of its reference, hence the English translation with a comma. The head noun is the patient argument in the main clause. The RC occurs after the demonstrative modifier δdi .

(13.42) ביאֹרי אֹימר, שרושי (אַראי) איאָשואילי מר, באיאראי אַראי אַרישי אַרישי אַרישי אַרישי אַרישי אַרישי אַרישי אַרישי אַראי אַרישי אַרישי אַרישי אַראי אַרישי אַרישי אַרישי אַראי אַראי אַרישי אַראי אַראי אַרישי ódi [k`jaŋk^ha (kjap) $ma-ts^hu-po=di$] p'itcun=gi námu NEG-be.able.to-2INF = DEMPH bird=GEN camel that counting (do) ka:m tenkha kimi=di ki:-ti foot on thread=DEMPH tie-NF 'those camels, [which could not be counted], being tied by a thread to the bird's feet...' (PAD bet story)

Example (13.43) presents another post-head RC that is more descriptive/appositive than restrictive:

वर्देवे'व' चम्दर्फे क्षेवा [nìːza? di: $p'ja=s\tilde{a}$: pè: k^hokø: nàŋ=gi tam=tsu do=TERM 1SG.GEN innards.GEN day.and.night this.AGT inside=GEN word=PL za: jò-**po**=di] t'ariŋ t'ã: dau ke:po=sã: sák-ti and month many=until accumulate-NF set EX-2INF=DEMPH today ĩ. iìgi di:=na cé-to letter this.GEN=LOC tell-IPFV EOU.PER 'Therefore I'm telling (you) today (my) inmost words, [which have been stored accumulating day and night for many months].' (Richhi 143)

Example (13.44) illustrates a complex post-head RC with two clauses:

(13.44) कु' क्रॅन्ग' छे'गी' हेट'हेट'गें।

tc^hu [*kø:-tiki rɛŋ-rɛŋ-ko*]⁴⁰⁵ water boil-NF make.cool-RDP-2INF '<u>water</u> [that is boiled and made cool]' (TB e)

In (13.45), the post-head RC occurs in the genitive because the RC is part of a pre-head adjectival modifier of the noun $t'ubd\varepsilon$:

One motivation for placing the nominalized (and reduplicated) verb after the noun in (13.45) probably is that if the nominalized verb would be placed before the noun (*bhaila=ki kjap-kjap-ø:* <u>ló?par</u>), Bhaila would easily be interpreted as the actor who takes the X-ray, with the genitive =ki taken, when the text is read aloud, as the homophonous agentive marker.

Lastly, (13.46) illustrates a post-head RC without a final =di, ending in the construction $t'\tilde{a}$: dau '(be) similar, resemble'. Note that another, pre-head locative RC, modifying the noun $n\dot{\epsilon}$: 'site', is embedded within the RC which modifies p'jado 'feather'.

(13.46) المجارعة على المحافظة المحافة المحاضة المحافة المحافظة

Summarizing the examples above, RCs formed with *-po/bo* may occur preceding their head noun or following the head noun. Moreover, the RC head noun (or common argument) may occur at least in the following roles in the RC: actor, patient, locative and copula subject of an existential clause. In the main clause, the head noun may occur at least in the following roles: actor, patient, copula complement in equative clause, complement of a locative postposition and copula subject of an existential clause. RCs nominalized with *-po/bo* can express various temporal and aspectual distinctions listed in Table 13.2 above.

13.2.3 Spatial nominalizer -sa

The construction VERB-*sa* can roughly be translated as 'a place where x is or can be done', x representing the verb to which the nominalizer -sa is attached. In my data, constructions nominalized with -sa occur in pre-head and headless clauses.

⁴⁰⁵ Consultant KUN commented that in his language variety $re\eta$ -re η -ko would require the patient argument $t'\phi$? 'heat', which expresses what is being cooled, $[k\phi:-tiki t'\phi]$ $re\eta$ -re η -ko].

13.2.3.1 Pre-head RCs

All the headed constructions occur preceding the headword and are genitivized. Based on my current corpus, RCs with *-sa* are more frequent in writing than in speech. A rare example of a headed RC with *-sa* from spoken data is (13.47). In spoken language, nominalizations with *-po/bo* and *-k^h* $\tilde{\epsilon}$:, which can both be used when the head noun has a locative function in the RC, are used in place of written constructions with *-sa*.

(13.47) Ry $\check{\alpha}_{\Sigma}$ value (and the case) of the story) (13.47) Ry $\check{\alpha}_{\Sigma}$ value (13.47) Ry $\check{\alpha}_{\Sigma}$ value

The personal pronoun $k^h u$ in (13.47) could be taken either as belonging to the RC (as suggested by the square brackets) or to the main clause, in which case the relative clause would consist of merely the nominalized verb.

Most of the following examples in this section illustrate uses of *-sa* found in written sources. The head noun is always in locative function in the RC. The examples are headlined based on the role of the head noun in the main clause (MC).

Head noun as MC patient

In (13.48), the head noun zoim 'tub' functions as the patient of the main clause verb ton 'show'.

(13.48) $\widetilde{a}_{q_{N'}} \widetilde{g}_{q'} \widetilde{$

(13.49)	ର୍ଷ:ର୍ଷ୍ତ ସଣ୍ୟାତ୍ତର୍ମ	<u> २८ २८ कें के</u>	र्गतेः र्झेनुःश्वतेः सःगवस्यः दर्वे			
	mí=tsu	t ^h amtse?	rãːrãːsoːsøː	[dø:- sø:]	sáne?	ts ^h ø:-ti
	human=PL	all	each.oneself.GEN	sit-NMLZ.SPAT.GEN	place	seek-NF
	dø: j <i>à</i> ?.					
	sit EX.PE	R				
	'All people	are sittir	ig, each having so	ought their own place	where t	o sit].' (Richhi 75)

Head noun as MC locative adverbial

In the following two examples, the head noun functions as a locative argument in the main clause. The locative expression $g\tilde{a}:to? m\acute{e}\eta k^{h}\tilde{a}:=na$ 'in Gangtok hospital' in (13.50) can be seen either as part of the RC or the main clause (the brackets follow the latter interpretation).

(13.50)	শ্বদ-র্দৃশ শ্ব	<u>નાયન</u> નાસ્ટ્રાપ્તે.	र्येः चलगुःखदेः ष्ट्रि	,ब्र'केग'त्र'		
	gã:to?	méŋkʰãː=na	[p'um=ø:	nè:po	zak-sø:]	<u>k^himmik=na</u>
	TPN	hospital=LOC	girl=GEN	patient	set-NMLZ.SPAT.GEN	room=LOC
	'In Gang	gtok hospital, ii	n the <u>room</u> [where fe	male patient are plac	ed]' (Richhi 166)

(13.51) สัง สูงาจารติสา 5.345 วัยาจารัง รัฐงาจารัง สิมาราชา

 $m\dot{u}$ $gjuk-z\tilde{\epsilon}$: $[\eta\dot{u}-k\epsilon\hat{r}$ $\dot{\tilde{o}}$:- $s\theta$:] $\underline{t}\epsilon^{h}o:=lo$ $l\epsilon p-kap$ 3SGF run-PROG cry-sound come-NMLZ.SPAT.GEN direction=DAT arrive-SIM 'When she arrives running to the <u>direction</u> [from which the sound of crying comes]...' (Richhi 2)

For some reason, in (13.52), the RC is not genetivized. The form *làpsa* also occurs as a headless RC, see §13.2.3.2

(13.52) สิ.สิ.ส์ พราพ ผู้พา สุราค พราพั

 $[te^{hete^{heto}}e^{bo} lap-sa]$ $\underline{\dot{y}}$:naneabe:=lo.TPNsay-NMLZ.SPATplace insideEQU.NE=REP'It's in a place [(that is) called Chechebo], it was said.' (PD Tashiding story)

Head noun as a genitive attribute in MC

In (13.53), the RC head word *zimteun* 'bedroom' is a genitival modifier of another noun, *gom* 'door'.

In (13.54), the head noun $\epsilon \dot{a}:lo2$ 'bamboo wall' is a genitivized modifier of the postposition *teylo* 'on'.

(13.54) གན་ཀོག་ རྒོག་ག་ འགྲུབ་གོ་ འགྲུ་གོག་གོ་མོ་... ཐི་སྱི་ karma=gi [dø:gar tʰap-sø:] <u>cá:lo=ki</u> tɛŋ=lo... t'i-ti PN=AGT play act-NMLZ.SPAT.GEN bamboo.wall=GEN top=DAT write-NF 'Karma writes... on the <u>wall</u> [of the place where the play is going to be acted] (and)...' (Richhi 71)

Head noun as the only argument of an existential in the MC

The head noun of an RC formed with *-sa* may function as the only argument in an existential clause (although the existential copula is elided in the below example).

(13.55) สู้านนิ่า สูรา ซิญาณ์ ดู ญสังสนิ่า สสังผุนั่ว

 $gok^h \emptyset$: $s \hat{u}$: $t \varepsilon i$:=lo $[n \hat{a} \quad s \hat{o} \cdot s \vartheta$:] $\underline{ts^h okor}$ doorway.GENsideone=DATfishkeep.alive-NMLZ.SPAT.GENpond'on one side of the doorway, (there is) a pond[where fish are kep]t' (Richhi 32)

I have no examples of post-head RCs with -sa.

13.2.3.2 Headless RCs

Because of its nominal origin (sá 'ground') the nominalizer -sa can be used as a headless relative clause, which itself refers to the noun it modifies (definition from Payne 1997: 328). Therefore, when -sa is used in a headless RC, the English translations below express the implied head noun by the word *place*. Many headless uses of -sa may be considered already lexicalized or close to being lexicalized. The forms which should likely be considered lexicalized because of their frequency include words such as *zak-sa* 'place to put something, storage', $d\phi$ -sa 'place to stay, dwelling', *zu:-sa* 'place to stay, dwelling (hon.)', *dzim-sa* 'place to sleep, bedroom (hon.)', *ki-sa* 'place of birth'. For examples of headless RC with -sa, consider (13.56-58).

(13.56) ଗ୍ରକ୍ଷା ଶ୍ରହ୍ୟା କ୍ରିମ୍ବା ସୁସ୍ଥା ଅନ୍ମା କିର୍ସା କରୁ କରୁ ଅନ୍ମା କିର୍ବାଧ୍ୟ କରୁ କରୁ ଅନ୍ମା କରୁ ଅନ୍ମା କରୁ ଅନ୍ମା

 $p'um = tei? p'ja-ti p^{h}iru? [ka t^{h}op-sa] gju-ee? mi-lè?.$ girl=INDF do-NF at.night who find-NMLZ.SPAT go-INF NEG-be.good 'Being a girl, it is not good to go at night [to the place of anyone one finds].' (Richhi 119)

(13.57) वगारेट के त्याया हि त्याया के राषा हें राषा

 t^{ha} :rin $[mi-dy: k^{hi}-dy: m\dot{e}:-sa] d\sigma:-n\varepsilon$ far.away human-travel dog-travel EX-NMLZ.SPAT live-COND 'If one lives far away [in a place where there is no roaming of either people or dogs]...' (nga'i 'gan 5)

(13.58) ८ देग अन्न भर्भ देट हे

 $n\dot{a}$ [\dot{o} ? $l\dot{a}p$ -sa]=lo $\dot{o}n$ -di1SG TPN say-NMLZ.SPAT=DAT come-NF 'I came to [the place (that is) called Wok] (and)...' (KT life story)

13.2.4 Quantifying nominalizer -tshe?

The quantifying nominalizer $-ts^{h}\varepsilon^{2}$, which derives from WT $\equiv_{\nabla} tshad$ 'limit, degree' is rather rare in my data. It attaches directly to a verb root and forms a post-head RC with the meaning 'as much as is x-ed' or 'the (full) extent of x-ing'. The RC is postposed to a noun which it modifies. Note that in the English translations the RC is in the pre-head position, as in (13.59), or scattered on both sides of the head noun, as in (13.60) and (13.61).

(13.59) & (13.5

The RC formed with $-ts^{h}\varepsilon^{2}$ may be followed by additional quantifying modifiers, the most typical of which, based on examples (13.60) and (13.61), is $k'\tilde{a}:pu t^{h}amt\varepsilon\varepsilon^{2}$ 'totally all'.

(13.60) $\in \operatorname{cg} \subset \operatorname{cg} \operatorname{cg} \operatorname{cg} \operatorname{cg} \subset \operatorname{cg} \subset \operatorname{cg} \operatorname{$

(13.61) مَامَنَّمَ عَلَى اللَّهُ عَلَى اللَّهُ عَلَى اللَّهُ عَلَى اللَّهُ عَلَى اللَّهُ عَلَى اللَّهُ عَلَى الْعُلَى اللَّهُ عَلَى الْحَالَى الْحَلَى اللَّهُ عَلَى الْكَلَى اللَّهُ عَلَى الْحَالَةُ عَلَى الْحَالَةُ عَلَى الْحَالَةُ عَلَى الْحَلَى اللَّالَةُ عَلَى اللَّهُ عَلَى اللَّهُ عَلَى اللَّ الْحَالَيْ الْحَلَى اللَّا عَلَى اللَّهُ عَلَى الللَّا عَلَى اللَّهُ عَلَى اللَّا عَلَى اللَّهُ عَلَى اللَّهُ عَلَى الْحَلَى الْحَلَى اللَّالَ عَلَى الْحَلَى الْحَلَى الْحَلَى الْحَلَى الْحَلَى الْحَلَى الْحَلَى اللَّهُ عَلَى الْحَلَى الْحَلَى حَالَى اللَّهُ عَلَى الْحَلَى الْحَلَى

13.3 Correlative clauses

Correlative clauses consist of two clauses with a common argument marked in the first clause by a question word and in the second clause by a coreferential resumptive demonstrative. The interrogative pronoun occurs in a truly question-like construction, but the presence of the resumptive demonstrative in the following clause distinguishes correlative clauses from indirect question clauses. The term "correlative (or co-relative)" refers to the clauses being "essentially coordinated, rather than one being embedded within the other" (Dixon 2010b: 356). The first clause in each of the examples (13.62-65) could occur as an independent question. The question word in the following examples is given in bold and the presumptive pronoun is underlined.

(13.62) ५ र्ड्रेन या गर यें र रेंग कें परी र प्रवया र रेंग्या

[t'a	пє́та	k'ar j <i>à-po</i>]	[<u>ódε=ra</u>	zak	go?.]
now	earlier	what EX-2INF	like.that=AEMPH	set	be.needed
'Wha	atever wa	s before, has to be	e preserved like the	at.' (C	Y interview)

(13.63) या र्केंट्र पर्झे कें पट्रे साया साया हा हा

[k'adz o? zo] $[\underline{ode:}$ $ma-l\tilde{o}:-k^h \epsilon n$ $b\epsilon ?.]$ how.maymakelike.that NEG-be.enough-NMLZEQU.NE'However many are made, that's not enough.' (DR discussion with KL)

- [t'ene p'otsø: $p^{h}am = tsu$ ka-ka=lo dencu go:-po] be.needed-2INF then child.GEN parent=PL who-RDP=DAT invite $ts^{h} \epsilon n - t^{h} o = t \epsilon i ?$ [*ódi=tsu=i Z*,*O* lo].that=PL=GEN name.HON-list=INDF make TAG.Q 'Then make a name-list of all those parents of the children who need to be invited, okay.' (Richhi 42)

As suggested by the brackets, the demonstrative-emphatic =di in (13.65) behaves somewhat differently from the other demonstratives in that it belongs phonologically to the first clause but syntactically to the latter clause, where it functions as the resumptive demonstrative. If =di is dropped, (13.65) becomes formally an interrogative clause that functions as a clausal complement. Those clauses where the resumptive demonstrative is co-referential with the whole first clause and not just the question word are analyzed as complement clauses, see §14.2.3.

13.4 Noun complement clauses

Noun complement clauses are a special case of complementation. The clause complementing a noun is usually nominalized and genitivized. Typically nominalization is accomplished by the infinitive marker -po/bo (§13.4.1), although the infinitive marker $-c\epsilon^2$ is also used (§13.4.2.). However, the noun complement clause may also be a finite clause which is followed by a nominalized verb of saying, functioning as a complementizer (§13.4.3). Only rarely is a noun complement clause formed by attaching the genitive clitic directly to the finite clause (§13.4.4).

13.4.1 Nominalization with -po/bo

The nominalized and genitivized complement clauses resemble in form relative clauses. The only difference to relative clauses is that the noun to which a complement is appended is not an argument within the complement clause. The head noun of a relative clause, on the other hand, is an argument within the relative clause. For instance, the complementized noun t^him 'law' in (13.66) is not an argument in the complement clause látea=jã: p'ince? me`:-po: '(that) even wages are not given'. Rather, the complement clause explains what the law is about. The complemented noun is underlined and the complement clause is given in square brackets.

- (13.66) 黒(あいいて) (こうかいて) (こうかいで) (こうかいで) (こうかいで) (こうかいで) (こうかいで) (こうかいて) (こう
- (13.67) यु. व्रेंब्र नदी हुबा र्जुबा प्रेंच थें में हुबा येंग

 $k^h u$ $[t^h \varnothing m - b \varnothing:]$ $\underline{t \varepsilon' \gamma^2}$ $p' ja - zin j \vartheta - po$ $b \varepsilon^2$.3SGMexit-2INF.GENmeansdo-PROGEX-2INFEQU.NE'He was searching a way[of getting out].' (KT animal story)

(13.68) यायाया अवार्गी या बाखुया मुनयारायी या मुखा घेन् रा भ्रन

lala=lo [máko lò súm kjap-ø:] <u>lògju?</u> j \dot{p} -po b ϵ ?. some=DAT son-in-law year three do-2INF.GEN story EX-2INF EQU.NE 'Some have the <u>narrative</u> [that the son-in-law does three years (of work service with inlaws)].' (SGD wedding customs)

- k'ã:dotci? nà tsa=le tiru? lèm-bøː] ηà [di: t^hap р'ја-се 1SG this.GEN at=ABL rupee 1sg take-2INF.GEN do-INF some means $\tilde{t} = s$. EQU.PER=QUO 'I'll find a way [to take from this one a few rupees], he said.' (PAD bet story)
- (13.70) هَ مَ مَ عَلَى اللَّهُ عَلَى اللَّهُ عَلَى اللَّهُ عَلَى اللَّهُ عَلَى الْعَلَى الْ الْعَلَى الْعَلَ
- (13.71) هَٰٚ (مَרْ: مَا سَامَ مَّا حَلَّهُ: عَلَّى مَعْمَ مَرْكُ: [*ódi ma-né:-pø:*] <u>gjumts^hɛ̃:</u>=di that NEG-reside-2INF.GEN reason=DEMPH 'The <u>reason</u> [why that (condition) did not remain]' (CY interview)

In the novel Richhi, the author may use either a temporally neutral form (13.72) or a progressive form (13.73) to refer to events that are taking place at the time of speaking/writing/reading.

(13.72) B' A'B' A'B'A' 551 4A' MT

 $[k^hi \quad au \quad au=lo \quad hap-ø:]$ $\underline{k\varepsilon^2}$ dog woof woof=DAT bark-2INF.GEN sound'the sound [of a dog barking woof woof]' (Richhi 1)

(13.73) $\exists a_1 : \exists c_1 : a_2 : a_3 : a_4 : a_5 : \exists c_1 : a_3 : a_3 : \exists c_1 : a_3 : a_3$

Time adverbial words often receive a complement clause:

(13.74) ริ ขุลิขุญานิ รูญาชีราชา เชีรา รายราชรา

 $t\varepsilon$ [*zi:-pø*:] <u>*t*'*yts*^{*h*}*ø*:=*na*</u> *k*^{*h*}*õ*: *hatã:tc*^{*h*}*ø*?. then look.HON-2INF.GEN time=LOC 3SG.HON be.amazed 'Then at the <u>time</u> [(that she) looked (at it)], she was amazed.' (PAD Tashiding story)

(13.75) אקרישרי די קאי קאישאי קאישאי

 $\begin{bmatrix} n\tilde{a}:te^{h}\tilde{a}:=tsa: tea:-p\phi: \end{bmatrix} \qquad \underline{nimts^{h}i}=di$ bethrotal=at come.HON-2INF.GEN date=DEMPH '(on) the day [when coming to the bethrotal]' (SGD wedding customs)
(13.76) هَتْعَامَا بَحَرَّ هَتْعَامَا بَحَرَّ مَسْطَعَمَا هُ عَلَيْمَا مَتَنَا الْعَنْ الْمَعْمَانَةُ عَلَيْمَا مَتَنَا الْعَنْ الْمَعْمَانَةُ عَلَيْمَا مَتَنَا الْعَنْ الْمَعْمَانَةُ عَلَيْهُ مَعْمَانَةً عَلَيْهُ مَنْ عَلَيْهُ مِنْ عَلَيْهُ مَنْ عَلَيْهُ مِنْ عَلَيْهُ مَنْ عَلَيْ عَلَيْهُ مَنْ عَلَيْهُ مَنْ عَلَيْهُ مَنْ عَلَيْكُمُ مَنْ عَلَيْكَ مَنْ عَلَيْهُ مِنْ عَلَيْ عَلَيْ عَلَيْكُمْ مَنْ عَلَيْ عَلَيْ عَلَيْ عَلَيْ عَلَيْ عَلَيْ عَلَيْ عَلَيْكُ مَنْ عَلَيْ عَلَيْكُ مَنْ عَلَيْكُمُ مَنْ عَلَيْكَ مِنْ عَلَيْكُمُ مَنْ عَلَيْ عَلَيْكَ مَنْ عَلَيْكَ مَنْ عَلَيْ عَلَيْكَمْ مَنْ عَلَيْكَ مَنْ عَلَيْ عَلَيْكَمْ مَنْ عَلَيْكَ مَنْ عَلَيْكَمْ مَنْ عَلَيْكَمْ عَلَيْكَمْ مَنْ عَلَيْكَ مَنْ عَلَيْكَ مَنْ عَلَيْكَمْ مَنْ عَلَيْكَمْ عَلَيْكَ مَنْ عَلَيْكَ مَنْ عَلَيْكَ مَ الْعَلَيْكَ مَنْ عَلَيْكَمْ مَنْ عَلَيْكَمْ مَنْ عَلَيْكَمَا مَنْ عَلَيْكَمْ مَنْ عَلَيْكَمْ مَنْ عَلَيْ عَلَيْ

(13.77) ସମ୍ରଷ୍ୟ କୁଁକ୍ଷ୍ୟ ନ୍ୟା ନିର୍ବ୍ୟ କରି ଅନ୍ୟୁ କରି ଅନ୍ୟୁ ଅନ

 $\begin{array}{ll} [dendz\tilde{o}: lok & te'\phi m-b\phi:] & \underline{g\tilde{a}:=lo} \\ \text{Sikkim return come.HON-2INF.GEN time=DAT} \\ \text{`...at the time [when coming back to Sikkim]...' (CY interview)} \end{array}$

Example (13.78) presents an exception to the description presented above: a genitivized complement clause occurs in post-head position. Note that there is a relative clause $(k^{him}=na l ep l \delta : m e^{-po} karma tsa=le \delta : j e^{-po}$ within the complement clause.

The reason for the exceptional constituent order in (13.78) is likely that it is easier to process a long RC after its head noun rather than before it (for the effect of processing on grammar, see Hawkins 2004). Significantly, (13.78) occurs in a piece of writing, giving an air of standardization to this unorthodox-looking construction.

The complement clause may start with $m \hat{e} n \varepsilon$ 'not perhaps', a grammaticalized conditional form of a negated equative. By using $m \hat{e} n \varepsilon$, the content of the hope is expressed with an air of negation, i.e. 'a hope that Karma comes' becomes in form more like 'a hope that Karma would not perhaps come'.

PN.AGT return-2INF.GEN before this.year 3SGF=DAT meet-PUR *òm-bø*:]
come-2INF.GEN
'She still has a <u>hope</u> [that perhaps brother Karma will come to meet her this year before he returns].' (Richhi 119)

13.4.2 Nominalization with -ce?

Noun complement clauses may also be formed by genitive-marked - $c\epsilon$ -infinitive clauses, although these type of clauses are less frequent than those nominalized by -*po/bo*.

(13.80) क्षत्र कुष गत्र र्जुष में गत्र न्यू मुंग नगत केंग येत्र में?

lenge? [*k'an p'ja-u nấ:-ɛɛ=ki*] <u>*ga:tsho:*</u> *jè-po?* PRN.HON what do-2INF do.HON-INF=GEN liking EX-2INF 'What would you like to do?' Lit. 'You have <u>enjoyment</u> [of doing what]?' (TB e)

(13.81) ארי אָז דָי שְׁמְקיאָ שָּקָאָ אָיאָדָעי אָרָשָ אַי דָאַיאַרָאָי אָרָשָ אַי אָדָעי אָרָשָ אַי אָדָעי אָרָשָ אַי [rã:=lo mèmbo t'a zɛn=lo nén mi-kjap-ɛɛ=ki] <u>t'amtea</u> 2SG.M=DAT except now another=DAT marriage NEG-do-INF=GEN promise '<u>promise</u> [to now not marry anyone else except you]' (Richhi 143)

13.4.3 Finite clause with a complementizer

Although noun complement clauses are typically nominalized, also finite clauses may complement nouns if followed by a complementizer, see (13.84) and (13.85), where a nominalized form a verb of saying functions as a complementizer. The complementizer is nominalized either by $-k^{h}\tilde{\epsilon}$; which does not occur with genitive, or -po/bo, which is genitivized. Note that the English translation of (13.84) has an indirect question clause.

 $\begin{bmatrix} mi & t'\tilde{a}:pu & l\tilde{e}p & t'\tilde{a}:pu & go:-po & l\tilde{a}p-k^{h}\tilde{e}: \end{bmatrix} & \underline{t'gnda2} & odi \\ \text{human honest very.much honest be.needed-2INF say-NMLZ reason that} \\ b\varepsilon = \varepsilon o. \\ \text{EOU.NE=AT} \end{aligned}$

'The <u>reason</u> [why a man needs to be honest, very honest], is that, you know.' (JDF axe story)

- (13.85) مام بَعْن هَمَ بَعْن مَعْن مَن مَامَ بَعْنَ مَامَا بِعَنْ مَامَا بِعَنْ مَامَا بَعْنَ مَامَا بِعَنْ مَامَا بِعَامَ مَامَا مُ مامام مام مامام مام مامام مام مامام مام مامام مام

13.4.4 Genetive-marked finite clause

Lastly, a non-nominalized genitive-marked clause may also occasionally complement a noun:

(13.87) ८ विंदा देना वर्ती याक्षेत्रा पेंदा दत्ता केराग्री याहका कटाना वर्ती दाख्या कार्यात्रायाव छत्ता k'atem j*à*? t'a $[k^h \tilde{o}]$ t'ep=di t'õ: $m \dot{e} = ki$] now 3sg.hon book=DEMPH how EX.PER and NEG.EX.PER=GEN nàtea? ma-són-k^hen bɛ?. tam nàŋca=di speech inside=DEMPH 1pl NEG-go.PFV-NMLZ EQU.NE 'Now we didn't go into speaking (lit. word/speech) [what his book was like or not like].' (KL BLA 12)

13.5 Postposition complement clauses

Postpositions may take three types of complement clauses. The first is marked by a genitivized *-po*-infinitive (\$13.5.1). The second is marked by the bare non-genitivized *-cc*-infinitive (\$13.5.2). The third option is a sentence-like complement clause with a resumptive demonstrative in the following clause (\$13.5.3). The aspectual choices for these types of clauses are the same as those given for RCs in Table 13.2. Table 13.3 lists some postpositions that can take a nominalized and genitivized clause as a complement.

Table 15.5. Some postpositions w	men take complement clauses	
pénle, pénlo	ই্ইৰ্ব'অম্ব', ই্ইৰ্ব'ৰ্ম	'before'
gjablɛ, gjablo	ক্র্বান্থমা, ক্রুবার্থা	'after'
nàŋlɛ, nàŋlo, nàŋɕa	वृत्तः भूषः, वृत्तः वृत्तः वृत्तः वृत्	'inside'
t'ønlo, t'ønlɛ, t'øndalɛ, t'øndalo	੶੶੶੶੶੶੶੶੶੶੶੶੶੶੶੶੶੶੶੶੶੶੶੶੶੶੶੶੶੶੶	'for the purpose of'
kor, korlɛ, korlo	ङ्ग्रीतः, ङ्ग्रीतः व्ययः, ङ्ग्रीतः व्यः	'about'

Table 13.3. Some postpositions which take complement clauses

13.5.1 Genitivized -po-infinitive

The order of presentation here follows the order of postpositions in Table 13.3.

- (13.88) ດຽສາ'ອິຼັດສາລັງ ອັງລາວດີ' ອັງລາວດີ' ອັງລາວດີ' ອັງລາວດີ' ອັງລາວດີ' ອັງລາວດີ' ອີງລາວດີ' ອີງລາວດີ' ອີງລາວດີ [dɛndzo:=lo man-dzøm-bø:] <u>nɛ́nlɛ</u>=di Sikkim=DAT NEG-come.HON-2INF.GEN before=DEMPH '<u>before</u> [coming to Sikkim]...' (KLT Bhumchu video)
- (13.89) $\mathfrak{F}_{\mathfrak{A}^{N}}$ $\mathfrak{A}\mathfrak{E}_{\mathfrak{A}^{N}}$ $\mathfrak{A}\mathfrak{A}^{N}$ $\mathfrak{A}\mathfrak{A}^{N}$ \mathfrak{A}^{N} \mathfrak{A}^{N

(13.90) هَٰ مَرْجُمْ يَرْمَا عَرْجَمْ عَرْجَ عَرْجَ مَحْ مَرْعَا عَرْجَ عَنْ عَرْجَ عَ that.AGT chatting do-NF sit-RDP-2INF.GEN inside=ABL considerable p^hi:-ts^ha:.</sup> be.late-CMPL 'Thus, as [(they) sit (and sit) chatting] it becomes quite late.' (lit. '<u>within</u> sitting') (Richhi 110)

(13.91) a) म्राया रुप्रांगु रेके प्रमुद र्यते र्यु येंग

 $\begin{array}{ll} [t^hamtee=ki & rite^hi \ k\tilde{a}:-b \phi:] & \underline{t' \phi n lo} \\ all=GEN & hope \ fill-2INF.GEN & for.purpose.of \\ \underline{'In \ order \ [to \ fulfil \ the \ hope \ of \ all]...' (Richhi \ 75-76) \end{array}$

b) क्षेत्र उत्त्यात्र यानुवा दवाद्वयात्रा यानुवा, दवाद्वयात्रा यानुवा विद्यालया क्षेत्र क्या क्षेत्र क्रुया क्षेत्र क्रियायां यावायात्र स्थित्यां आवर्य क्रियायां क्षेत्र क्रियायां यावायात्र स्थित्यां आवर्य क्षेत्र क्रियायां क्षेत्र क्रियायां यावायात्र क्षेत्र क्रियायां यावायात्र क्षेत्र क्रियायां क्षेत्र क्रियायां यावायात्र क्षेत्र क्ष ุรเล่นางที่นิ่า รัสานี้" พาสู่นางาร์" นรุ่มเซินา ซีราล่า î:ruŋ=jã: [tam t'amfa? pí:, t'amta? pí:=tci? ηà k'utca lenge? however=still word tight two tight two=INDF 1SG 2PL PRN.HON kumdỹ:=lo ts^hokpo sá:py:=gi man-zak-ø:] t'øn=lo in.front.of=DAT NEG-set-2INF.GEN association new.GEN=GEN purpose=DAT ma-ts^hu-po $d\epsilon m = t\epsilon i^2 ts^h o^{-} z\epsilon$. NEG-be.able.to-2INF such=INDF feel-PST 'However, I felt like I can not refrain (myself) from presenting a few concise words in

your presence, (members of) the new association.]' (lit. 'However, I felt like someone who is not able <u>for the purpose of</u> [not presenting a few concise words in the presence of you, the new association.]') (NT BLA 7)

(13.92) a) เสข สู่สุดขางนิ พิ้าราง

 $\begin{array}{ll} [n\acute{a}: ts^hu - p\mathscr{O}:] & \underline{kor} \\ \text{I.AGT be.able.to-2INF.GEN} & about \\ `about [what I am able (to do)]' (Richhi 36) \end{array}$

⁴⁰⁶ The fact that $ts^ha(:)$ is nominalized shows that Denjongke grammar treats $ts^ha(:)$ in this construction like a secondary verb meaning 'finish' rather than as a fully grammaticalized completive marker.

নাল্লীক' আঁদ'র্শ্বা [t'ato sã:te $k^h o \eta = gi$ dzamlių nàų ca=lo $dzut^{h}ul \quad k^{h}o:jo:=ki$ now until 3sg.hon=gen world inside=DAT miracles servant=AGT nấː-zin lóːm=tsu=gi j*à-p*ø:] $t^hamt \in \varepsilon = ki$ korlo disciple=PL=AGT do.HON-PROG EX-2INF.GEN about all=AGT sén-zin j*à:-*to. hear.HON-PROG EX-PROB 'All are probably hearing about [(the fact) that his servants, disciples are doing miracles in the world].' (KT life story)

As shown by the progressive form in (13.92b), the nominalized verbal construction in the complement clause may be complex.

13.5.2 Bare -ce-infinitive

The postposition complement clause may also be a non-nominalized infinitive clause marked by $-c\epsilon^2$:

- (13.95) $\check{m}'af' af' \check{\pi}'af' \check{\pi}'af' \check{\pi}'af'' \check{\pi}'af'' \check{\pi}' \check{\pi}'$

13.5.3 Sentence-like complement

Finally, a postposition complement can be a sentence-like clause (i.e. the form could occur independently) followed by a resumptive demonstrative (in genitive), which is coreferential with the whole previous clause. In my data, this may happen with the postposition *korlo* 'about' (WD $\Im_{\nabla} \check{\Delta}$ ').

- (13.96) ฉริเพลา ลเนนารรับสารรา ลเนนารรับสมาทาทานี้ แล่งสารที่สายรับ ลริลา มีรานี้ รายการเป็นการเรา เป็นการเรา เ [dile thap *i*: t^hapøm ka-ka=lo ťã: sé: g0:-68 bolthen actor and actress who-RDP=DAT choose be.needed-INF EQU.NE.Q di: korlo nàtea? ní:-po: k'jotyn kjap-ke. this.GEN about 1pl two-COL.AGT discussion do-HORT 'Then let the two of us discuss about [who all should be chosen as actors and actresses.]' (Richhi 41)
- ĩ:-run [ts^hima? nànca=lo=jã: ran-ke:=di k'ate càpcù however future.life inside=DAT=too own.language=DEMPH how serve g0:-68 bo]di: korlo ηà sámlo tã:-zin jò?. 1SG thought send-PROG EX.PER be.needed EOU.NE.O this.GEN about 'However, I'm thinking about [how I should serve (the cause of) my mother tongue also in the future].' (KT life story)

The first clause in both (13.96) and (13.97) could function as an independent question but is here linked with the second clause with the help of the resumptive proximal demonstrative di, which is coreferential with the whole interrogative clause. The second clause could also occur independently. Example (13.96) comes from the novel Richhi, where the writer signals the linking relationship achieved by juxtaposition and resumptive demonstrative by leaving out the equivalent of the full stop (1), which would normally occur at sentence boundary.

13.6 Summary remarks

This chapter described "constituent-modifying clauses," a term which covers all clauses that modify a single word. The modified word can be a noun (relative clauses and noun complement clauses) or a postposition (postposition complement clauses). The modifying clause is nominalized and typically genetivized. Genetivization does not take place in post-head RCs or if the modifying clause is nominalized with $-k^h \tilde{\epsilon}$. Morever, genitivization is not mandatory when the verb root is reduplicated. All the relativizing nominalizers $-k^h \tilde{\epsilon}$. -po/bo, and -sa were seen to occur in headed RCs and two of them, $-k^h \tilde{\epsilon}$: and -sa, also in headless RCs. The nominalizer $-k^h \tilde{\epsilon}$: was seen to have a marginal internally-headed use. Moreover, this chapter showed that correlative clauses, which are probably Indo-Aryan influence (see Genetti1992: 408), are functionally similar but formally dissimilar to RCs. Correlative clauses, while not demanding nominalization and genitivization, require a question word in the first and a resumptive demonstrative in the second clause. Finally, I defined the difference between relative clauses and noun complement clauses in the following way: the noun to which a complement is appended is not an argument within the complement clause, but the head noun of a relative clause is an argument within the relative clause.

14 Complement clauses

This chapter continues the discussion on clause combining and subordinate clauses. Complement clauses are clauses that function as an argument of another clause (Noonan 2007: 52). They can be either non-finite or finite (sentence-like). The following subsections discuss non-finite complement clauses (§14.1) and finite complement clauses (§14.2).

14.1 Non-finite complement clauses

In my data, non-finite complement clauses may be marked by the elements listed in Table 14.1:

Table 14.1.	Elements	forming	complement	t clauses
10010 11.11	Liements	Torining	complement	ciuuses

1) II infinitive marker *-po/bo* (§14.1.1)

2) I infinitive marker $-c\epsilon^2$ (§14.1.2)

3) progressive marker -*tcɛ̃:/zɛ̃:/zin* (§14.1.3)

4) postposition kor 'about' heading a postposition complement clause (§14.1.4))

Headless relative clauses marked by $-k^{h}\tilde{\varepsilon}$: and -sa resemble complement clauses, but because headless clauses with $-k^{h}\tilde{\varepsilon}$: and -sa imply a referent that is modified (person who does with $-k^{h}\tilde{\varepsilon}$:, and place where something is done with -sa), they are categorized as relative clauses, see §13. The infinitives in Table 14.1 may occur with or without demonstrative-emphatic =di. Infinitival complement clauses typically occur as copula subjects. The four types of complement clause are described in the following subsections in the same order they occur in Table 14.1.

14.1.1 Complement clauses with -po-infinitive

In my data, complement clauses marked by *-po*-infinitive occur as the copula subject or the copula complement of change-of-state verbs and copulas, or as the P argument of other types of verbs listed in Table 14.2 (where "collocate" means "in my data typically co-occurs"). The verbs listed in Table 14.2 do not include the honorific equivalents of certain verbs (zi: $\eta \exists \eta \forall \forall$ 'see [hon.]', sen $\eta \forall \forall \forall$ 'hear [hon.]', dze: $\exists t \in \P$ 'meet [hon]'). Although I do not have examples of the honorific verbs, it is safe to assume that that they behave analogously to the ordinary verbs.

a) Being:	\tilde{t} : ເຫັງ', $b\epsilon^2$ ເຊ $_{3}$ ' (equative), $j\dot{\theta}^2$ ເພັງ', $du^2 c_{5}$ ' (existential); collocate
	with all types of verbs; structurally receive a complement clause, but
	functionally the combination marks tense/aspect/modality
b) Change of state:	thøn র্টর 'come out, become, happen', tc 'uŋ ভ্রুন 'become', zo দ্র্রা
	'make (into)', \tilde{i} way 'be (equ.)' $j\partial 2$ and 'be (ex.)'; collocate with go:-
	<i>po</i> [be.needed-2INF] and <i>ts^hu-po</i> [be.able.to-2INF]
c) Perception:	$t^h \tilde{o}$ ਕਸ਼ੱਸ 'see', $t^h o$ ਸ਼ੱਕ 'hear'; collocate with all types of verbs
d) Mental activity:	hako 5'र्गे' 'know, understand', t'Ensin इत्र चेत्र' 'remember', nó: अर्वे'
	'think'; <i>hako</i> and <i>t</i> ' <i>ensin</i> ; collocate with all types of verbs, except $n\dot{o}$; which collocates with go :- $po = 5\bar{\eta} \times \bar{x}$ [be.needed-2INF] ⁴⁰⁷
e) Meeting ⁴⁰⁸ :	$p^{h_{\mathcal{E}}^{2}} \cap (m_{\mathcal{E}}^{2})^{*}$ 'meet', $t^{h_{\mathcal{U}}} k_{\mathcal{E}}^{\eta}$ 'touch, meet, face'; $p^{h_{\mathcal{E}}^{2}}$ collocates with $j \partial$ -
	po वॅन्ट्र [EX-2INF], t ^h uk collocates with go.po न्यॅल्र रॅंग [be.needed-2INF]
f) Speaking and writing:	$t'i \exists \cdot \text{'write'}, \epsilon \epsilon^2 \neg_{\neg \neg \neg \neg} \text{'tell'; collocate with } go:-po [be.needed-$
	2INF]

Table 14.2. Verb types receiving a complement clause with -po/bo

As seen in the list above, the verb go? 'be needed' occurs frequently as the verb which is heading the complement clause. The complement clauses with the various types of verbs are now exemplified in the same order they occur in the list above.

As shown by (14.1), both the equative and existential copulas occur postposed to clauses marked by *-po*. Because these constructions look formally like the complement constructions described below, they are briefly mentioned here. Since the function of the combination, however, has become more grammatical (the construction in [14.1a] marks past tense and the one in [14.1b] perfect aspect), a fuller description is presented in §8.1.1. and §8.1.4 respectively.

Verbs of change of state

Note that the demonstrative pro-adverb $d\varepsilon m$ is used alongside nominalization to introduce comparison, see (14.2c).

⁴⁰⁷ Even more often *nó*: collocates with *go*? without the nominalizer, e.g. *p'ja go*? *nó* [do be.needed think] 'think that one has to do'.

⁴⁰⁸ Covers concrete ($p^h \varepsilon P$) and metaphorical ($t^h uk$) meeting.

(14.2)	a) وإمر، ومر، مرمر، وقرم، يَرْ، مرقم، يَوْلَم، مَرْمَى مَعْمَ، عَلَى مَعْمَ، مَرْمَاً مَعْمَ، مَرْمَاً مَعْمَاً مَعْمَ، وَمَرْمَاً مَعْمَاً مُعْمَاً مُعْمَاطًا مُعْمَاطًا مُعْمَاً مُعْمَاً مُعْمَاً مُعْمَاطًا مُعْمَاطًا مُعْمَاطًا مُعْمَاطًا مُعْمَاطًا مُعْمَاطًا وَعْمَالًا مُعْمَالًا مُعْمَاطًا مُعْمَاطًا وَحْمَالًا مُعْمَالًا مُعْمَاطًا مُعْمَالًا وَحْمَالًا مُعْمَا مُعْمَا مُعْمَا مُعْمَا مُعْمَالًا مُعْمَا مُعْمَا مُعْمَا مُعْمَا مُعْمَاطًا مُعْمَا مُعْمَالًا مُعْمَا مُحْمَا مُعْمَا مُعْمَ مُعْمَا مُعْمَا
	b) אָקישִהִי זְאָי זְאָרִיזָאַ אָקָי אָקָ אָקָי רְזָזְאִיזִי צָּקִי רְזָזְאִיזִי צָּקן זֹ:jã: kʰu [kʰõ:tʰo=di nɛ́n-di dø: go:-po] however 3SGM anger=DEMPH suppress-NF stay be.needed-2INF tʰom-bo bɛ?. become-2INF EQU.NE 'However, he became [obliged to keep his anger suppressed].' (Class 7 textbook 61)
	c) $\widehat{r}_{i} \equiv r_{i} e_{i} e_$
	d) \hat{p} , \hat{c} ' $;$ aav η aa $\neg \eta$ ' \hat{s} ', η $;$ aav \hat{s} ', $\neg \eta$, η ,
	e) યાત્ય ક્ષેત્ર યામ્રસ વર્ત્તે આયવ વર્ષે ચાય્ય યાયલ દ્વાયાર્પે દુત્ત્વ k'ɛːsi? tam=di [kʰandoː sɛ́n tsʰu-po] tɛʿuŋ-nɛ if speech=DEMPH dakini.AGT hear.HON be.able.to become-COND 'If this speech will become [such that the dakini can hear it]' (rna-gsung 12)
Verbs 0 (14.3)	f perception a) ਗੁਸ਼ ਸ਼੍ਰੇਸ਼ ਸ਼੍ਰਾਸ਼ਕ੍ਰੇਕਾ ਕਾਂਸ਼ ਕੋਸ਼ ਬੱਸ਼ਾ ਬੱਸ਼ਾ ਬੱਸ਼ਾ ਬੱਸ਼ਾ ਬੱਸ਼ਾ ਬੱਸ਼ਾ ਬੰਸ਼ਾ ਕੁੰਦਾ ਕ [k'ursin set and a set and a set a strand a sugarcanel ' (KTL animal story)

ĺ	~8a	cane	Peer	1100	eur moo		500 200	Lácura
((He)) saw ((him)	[peelin	ng and eatir	ng sugarcane].' (KTL a	animal story)

- b) גַמִּי בְּבִיאַבִּי מֵּי מְמִדְיָשָׁי יימִיחְרָי" שְּבִיפָ מְשָּיַצָּמִי מִיחְרָי בּוּקַמְשִּיצִי מְשִׁיצָי מְשִׁיצָי מְשָׁיצָי מָשָּׁיצָי מְשָׁיצָי מָשָּׁיצָי מְשָׁיצָי מָשָּׁיצָי מָשָּׁיצָי מוּשָׁיצָי מוּשָׁיצָי מוּשָׁיצָי מוּשָׁיצָי מוּשָּר גַי מוּשָׁיצָי מוּשַ מוּשַי גערי מוּשָׁיצָי מוּשָׁיצָי מוּשָׁיצָי מוּשָר מוּשָׁיצָי מוּשָׁיצָי מוּשָׁיצָי מוּשָׁיצָי מוּשָּיצָי מוּשָע מושעי שוּשוּש שוּיצָי מוּשַר מוּשַר מוּשַר מוּשָר מוּשָר מוּשָר מוּשָר מוּשָר מוּשָר מוּשָר מוּשָר מוּשַר מוּשּר מוּשַר מוּ מוּשַר מוּשר מוּשַר מוּשר מוּי געיד מושעי גענער מושעי שוּיד מושעי גענער גענער מושעי גענער מושער גענער מושער גענער מושער מושער גענער מושער מוער מושער מושער גענער מושער מועי גענער מושער גענער מושער גענער מושער מושער מושער מושער גענער גענער גענער מושער גענער מושער גענער גענער מושער גענער מושער גענער מוער גענער מועגענער גענער גענער גענער גענער גענען גענען גענעגענען גענער גענען גענען גענען גענען גענעגענען גענען גענען גענען גענען גענען גענעען גענען גענען גענען גענעגענען גענען גענען גענען גענען גענען גענענען גענען גענען גענען גענען גענען גענען גענען גענעגענען גענעגענען גענעגענען גענעגענען גענעגענען גענעגענען גענעגענען גענעגענען גענעגענען גענען גענעגעען
- c) ८ ८ ८ ४ ४ में भ्रु प्रमेश भ्रु प्रमेश के प्रदेश रागपर मुबर में झेंश रें क्षेत्र [ŋàtɛi pʰamo kude: di=tsu=gi ódɛp ka nấ:-bo] tʰo:-po 1PL.GEN parents elder this=PL=AGT like.that order do.HON-2INF hear-2INF i:. EQU.PER 'I heard [these parents and elders of ours say like that].' (CY interview)

Verbs of mental activity

(14.4) a) เพลา ณัญา ฐิเวลานี้ 5 ซ้าริ

[*ám* l*∂k lεp-o*] *hako-ti* mother return arrive-2INF know-NF 'finding out [that the mother has returned]' (Richhi 32)

- b) هَذَهُ: اللَّاحَ، مَاهَمَ: عَلَى اللَّامَةُ: اللَّقَامَةُ: الْقَامَةُ عَلَيْكَامَةُ: الْحَلْقُلْقُلْحَامَةُ: الْحَلْقُلْحَامَةُ: الْحَلْقُلْحَامةُ: الْحَلْقُلْحَامةُ عَلَيْحَامةُ اللَّعْلَيْحَامةُ عَلَيْنَاعَامَةُ الْحَلْقُلْحَامةُ عَلَيْكَامَةُ عَلَيْنَا الْحَلْقُلْحَامةُ لَقَامَةُ عَلَيْحَامةُ عَلَيْنَا الْحَلْقُلْحَامَةُ عَلَيْحَامةُ عَلَيْ حَامَةُ عَلَيْحَامةُ عَلَيْنَا الْ
- c) केंद्र ग्री. क्षार्रेग क्षत्रा श्चे वेंग श्चत्र का का ले. सु.श्चेट्र. र्हु तेत्र सुया केषा कु. व्या जु.गीषा श्चेंग्वतित श्चेंद्र ट्वेंषार्रेग केंद्र ग्रीषा कर्ते क्रुंगाषार्रेग गा। $[t \epsilon^h \phi] = ki \quad a p o$ áт pélopynts^ha ázi p'usim=tsu 2SG.L=GEN father mother male.relatives elder.sister younger.sister=PL $pint^{h}up mikte^{h}u la: pa=gi p^{h}i:-z\tilde{e}:$ nó: dø: go:-**po**] tc^hø=ki arm=AGT wipe-PROG stay be.needed-2INF 2SG.L=AGT think all.day tear ts^hu-po?

be.able.to-2INF

'Are you able to think [that your father, mother, brothers and sisters have to live daily wiping their tears to their arms]?' (nga'i 'gan 7)

Verbs of meeting

(14.5)a) क्रेंग्रिन क्रेंनग्रायन गाव के नृत्त क्षायहेंवा क्रेंनग्रायनग्रा के वे केंवा त्राया क्षेत्र केंवा प्रविव केंन्द्र प्रदीत ódi ts^høka [kantei t'ã: ladzi? $ts^{h}o\eta k^{h}\tilde{a}$:= $l\varepsilon$ tcini nò: bak-ti that time PN and PN market=ABL sugar(Nep.) buy carry-NF **p**^h**e?**.⁴⁰⁹ lòk-zẽ: i**ò-po**] return-prog ex-2inf meet 'At that time (they) met [Kanchi and Lhadzi returning from the market, carrying sugar (they) had bought].' (Richhi 40) b) रे.के. यहवां हैं! हॅन् हॅन् र्यटा राये. बटाये. प्राया. कट. ट्येंबाये. खुया [*rit^ha*: $te^{h}\epsilon$? rits^hi tap-ti dø:-dø:-pø: nàŋlo go:-po] hope sow-NF sit-RDP-2INF.GEN inside be.disappointed be.needed-2INF t^huk. meet 'In the midst of keeping on hoping, he is faced with [having to be disappointed].' (Richhi 116) Verbs of writing and speaking (14.6)f'i p'in ồ. [mén k'an k'an nò: go:-po] t^horã: medicine what what buy be.needed-2INF tomorrow write give come 'Tomorrow (I) will write for you [what medicines need to be bought].' (Richhi 29) [mí t'ønda? ódi ťã:pu lèp f'ã:pu go:-po] làp-k^hẽ: human honest very.much honest be.needed-2INF say-NMLZ reason that $b\varepsilon = \varepsilon o$. EOU.NE=AT 'The reason for saying [the person has to be honest, very honest] is that, you know.' (SGD wedding customs)

Clauses marked by *-po*-infinitive are also used as arguments in copular clauses of quantification (14.7), attribution (14.8) and existence (14.9). As suggested by the examples below, the nominalized verb is typically go? 'be needed'.

⁴⁰⁹ The complement clause here could alternatively be analyzed as an NP with a following post-head RC, that is [*kantei* $t'\tilde{a}$: [*adzi*? [*tshoykh* \tilde{a} :=*le teini pò: bak-ti lòk-z* \tilde{e} : *jò-po*]]. Two facts suggests that an analysis as a complement clause is better: 1) In the novel Richhi, where the example is from, the NP expressing the patient of the verb *phe*? 'meet' is typically, although not exclusively, marked with the dative-locative =*lo* or postposition *námtei*? 'with'. Dative-locative marking of post-head RCs occurs at the end of the RC, as shown in (13.40). The absence of case or postposition marking on the potential post-head RC (or the nouns), suggests that (14.5) is likely a complement clause. 2) Clear posthead RCs are typically not as complicated as the potential one in (14.5).

- (14.7) $\check{n}^{i}\check{n}^{i} \check{\neg}^{i}\check{\neg}^{i}\check{\neg}^{i}\check{\neg}^{i}\check{\neg}^{i}\check{\neg}^{i}\check{\neg}^{i}\check{\neg}^{i}\check{\neg}^{i}\check{\neg}^{i}\check{\neg}^{i}\check{\neg}^{i}\check{\neg}^{i}\check{\neg}^{i}\check{\neg}^{i}\check{\neg}^{i}\check{\neg}^{i}\check{\neg}^{i}$ $[k^{h}u=lo \quad go:-po=r\tilde{a}:] \quad \acute{o}di-ts^{h}\varnothing? \quad \acute{i}:.$ $3SGM=DAT \quad be.needed-2INF=AEMPH \quad that-as.much.as \quad EQU.PER$ $`[What he needs] \quad is that much.' (Richhi 102)$
- (14.8) নমজা এক্তম্ নদৃদ্য দ্বাঁখাম্য দ্বী পাথান্তব জিল্বা দৈ দেওলা দীমা দুর্গী শ্রদ্ব [[sámteʰa: tã: go:-po=di] k'ɛːtɛɛ̃: ím-bo] ŋàtɛa=gi plan send be.needed-2INF =DEMPH important EQU.PER-2INF 1PL=AGT hako bɛ?.
 understand EQU.NE
 'We understand [that it is important [that (we) have to do planning].' (sbar-phung 92-93)
- (14.9) المَّنْ عَنْ عَنْ مَعْنَا مَحَنْ حَنْ مَنْ عَنْ حَنْ مَنْ عَنْ حَنْ مَنْ عَنْ عَنْ مَنْ عَنْ الْمَالَةُ عَنْ مَنْ الْعَنْ عَلَى اللَّهُ الْعَنْ عَلَى اللَّهُ الْعَنْ اللَّهُ عَنْ اللَّعْنَ اللَّهُ عَنْ الْحَالَةُ عَنْ اللَّهُ عَلَى الْحَالَةُ عَنْ الْحَالَةُ عَنْ الْحَالَةُ عَلَى الْحَالَةُ عَنْ اللَّهُ عَلَى اللَّهُ عَلَيْ الْحَالَةُ عَنْ الْحَالْ الْحَالَةُ عَنْ الْحَالَةُ عَنْ الْحَالَةُ عَنْ الْحَالَةُ عَنْ الْحَالَةُ عَنْ الْحَالَةُ عَنْ الْحَالَ الْعَامَةُ عَنْ الْحَالَةُ عَنْ الْحَالَةُ عَنْ الْحَالَةُ عَنْ الْحَالَةُ عَنْ الْحَالَةُ عَنْ الْحَالَ الْحَالْحَالَةُ عَلَيْكَامِ اللَّا عَالَيْ الْحَالَةُ عَلَيْ عَالَةُ عَلَيْ الْحَالَةُ عَالْحَالَ الْحَالَةُ عَالَةُ عَالَيْعَالَةُ عَلَيْ عَالَيْ عَالَيْ الْحَالَةُ عَالَةُ عَالَةُ عَالَةُ عَلَيْ الْحَالَةُ عَلَيْ الْحَالَةُ عَلَيْ الْحَالَةُ عَلَيْ الْحَالَةُ عَالَيْ الْحَالَةُ عَالَيْ الْحَالَةُ عَالَةُ عَالَيْ الْحَالَةُ الْحَالَةُ عَالَيْ الْحَالَةُ عَالَ الْحَ الْحَالَيْ الْحَالَةُ عَالَيْ الْحَالَةُ عَالَةُ عَالَيْ الْحَالَةُ عَالَيْ الْحَالَةُ عَالَيْ الْحَالَةُ عَالَ لَ الْحَالَةُ عَلَيْ الْحَالَةُ عَالَيْ الْحَالَةُ الْحَالَةُ عَالَةُ عَالَيْ الْحَالَةُ عَلَيَا الَحَالَةُ عَلَيْ الْحَا

Finally, nominalized go? 'be needed' collocates with $k'\varepsilon:t\varepsilon^{h}i$ in a construction which may be characterized either as verbless attribution or a verbal use of the adjective $k'\varepsilon:t\varepsilon^{h}i$ 'important':

14.1.2 Complement clauses with -cc-infinitive

A complement clause may also be formed by an infinitive marked by $-c\epsilon^2$. An infinitive marked by $-c\epsilon^2$, which typically refers to an action in a more abstract way than -po/bo, occurs at least as the copula subject (14.11) and as a patient of the verb p'ja 'do' (14.12).

(14.11) a) रूप् भून तर्जुम वर्त सुर (Nep.) बालेवान भ्रम

 $[ran-k\epsilon^2 bj\tilde{a}:-\epsilon\epsilon=di]$ puramàlepbe?.own-language disappear=INF=DEMPHvery(Nep.)badEQU.NE'(It) is very bad [that the mother tongue disappears].' (YR canteen video)

b) गुहेना ने गुवहा के प्रदेश स्वर्भ स्वर्भ स्वर्भ स्वर्भ स्वर्भ गुयेंग के ने गुयेंग के ने गुयेंग के ने गुयेंग के ने गुये के गु 1000 न गुये के गुये गुये के गुये के गुये के गुये के गुय (14.12) a) याक्षे कुंक्षे चुँतामा जुराये।

 $[p^{h}at\varepsilon ts^{h}ut\varepsilon p'in-\varepsilon\varepsilon^{2}] p'ja-g\varepsilon.$ thither hither give-INF do-HORT 'Let us do [giving mutually (to each other)].' (PD bet story)

b) ๑
๑
๙
๙
๙
๙
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
๓
<

14.1.3 Complement clauses with progressive -tcee:/ze:/zin

The progressive form of the verb can act as a complement of sensory verbs.

(14.13) ซาลรา ลิ่า ฏิสา ภูกลากติสา ผลักานกุ

 $[p^{h}a:k^{h}a \quad mi \quad k'jo? \quad kjap-zin] \quad t^{h}\tilde{o}:=p\varepsilon?.$ over.there human chatting do-PROG see=EQU.NE 'It is visible (to me) [that there are people chatting over there].' (KN e)

14.1.4 Postposition clause with kor 'about' as a clausal complement

The postposition *kor* 'about' together with its complement clause may functions as a clausal complement:

गंगवान्यविताः येंदाः ह्यां [t'ato sã:te $k^h o \eta = gi$ dzamlin nànca=lo dzut^hul $k^{h}o:jo:=ki$ now until 3sg.hon=gen world inside=DAT miracles servant=AGT ló:m=tsu=gi nấ∶-zin j*à-pø*: **korlo**] $t^{h}amtee = ki sen-zin$ disciple=PL=AGT do.HON-PROG EX-2INF.GEN about all=agt hear.HONjờ-to. **EX-PROB** 'All are probably hearing [about (the fact) that his servants, disciples are doing miracles in the world until now].' (KT life story)

14.2 Finite (clause-like) complement clauses

Finite, or clause-like complement clauses are such clauses which could occur independently. They may occur without a complementizer (§14.2.1) or with a complementizer (§14.2.2). Moreover, the finite complement clause may occur with a resumptive demonstrative (§14.2.3).

14.2.1 Finite complement clauses without complementizer

Finite complement clauses may be either declarative (§14.2.1.1) or interrogative (§14.2.1.2)

14.2.1.1 Declarative complement

Declarative complement clauses without a complementizer are frequent with the verb $n\dot{o}$: 'think', see (14.15) and (14.16), but also occur with other verbs, see (14.17).

- (14.15) $rm{r}$ $rm{s}$ $rm{r}$ $rm{s}$ $rm{s}$ r $te^{h} \phi = lo$ [rấː-mi ĩ:] nó:-ti t'ɛːlu k^ha nà=to 1SG=CEMPH 2SG.L=DAT own-human EQU.PER think-NF just.like.that message ĩ tciku òm-bo ke:-wa bring-PUR only come-2INF EQU.PER 'Thinking [that (you) are (my) own people] I have just come like that to bring you a message...' (nga'i 'gan 11)
- (14.16) a_{ij} $\eta \in \eta$ $\hat{a}_{ij} \in \eta$ $\hat{a}_{ij} \in \eta$ $[l\dot{u} = tci? \quad t'i \quad go?] \quad n\dot{o}:-ti$ song=INDF write be needed think-NF 'Thinking [(that I) have to write a song]...' (nga'i 'gan 11)
- (14.17) गासुषा दुषा प्रविग यामेषा ये?

[*k'aly:-p'ja za?*] *ma-cé:-po*? slow-ADVZR set NEG-know-2INF 'Did you not know (enough) [to place it (there) slowly].' (Richhi 106)

14.2.1.2 Interrogative complement

Interrogative clauses functioning as indirect questions occur as complements in the same form in which they would occur as independent questions.

(14.18) สัราร์าสูา สาสสัรา

 $[\partial m - bo \quad n\dot{a}]$ $ma - t^h \tilde{o}$. come-2INF EQU.PER.PQ NEG-see 'I did not see [whether he came].' (DB' wife, oh)

- (14.19) הקלא יחאי אלא היהלא יחאי ה' האהי אלא ילי [*deŋ-gam min-deŋ-gam*] *te lèp nó:-ne=di* be.true-ATTQ NEG-be.true-ATTQ so much think-COND=DEMPH 'so if (I) carefully (lit. a lot) think [whether it is true or not]...' (CY interview)
- (14.20) होते केवर्षान यान यान केंट गांगीय क्या क्याय में

[*mí*: $s \neq m = na$ *k* '*an k* '*an j* \neq ?] ka = gi *làp* $ts^{h}u - po$? human.GEN mind=LOC what what EX.PER who=AGT say be.able.to-2INF 'Who can tell [what all is within the human mind]?' (Richhi 2)

(14.21) द्वेष राया या वा राष्ट्र राष्ट्र गा गीया मार्गे ?

[t ^h inlɛ	k'ana	sốː-bo]	ka=gi	hako?
PN	where	go.pfv-2inf	who=AGT	know
'Who k	nows [wl	here Thrinley h	as gone]?' (Richhi 6)

(14.22) ସିଁସା'ମାନି'ଟ' ସାଟ' ଧିଁକ୍ 'ହାଁ ସାହରା ହି' ସାନ୍ସା' ରି'ର୍ଦ୍ୟସାର୍ଷ'ନ୍

[k ^h okø:=na	k'an jờ-po]	dam-di zak	mi-lèː=lo.
innerds.GEN=DAT	what EX-2INF	bind-NF set	NEG-be.good=REP
'It's not good to bin	d and store [what	t is inside], it's	said.' (Richhi 143)

Note that (14.20-22) resemble correlative clauses in that they, like correlative clauses, have a question word in the first clause (see §13.3). The difference, however, is that the complement clauses are not followed by a resumptive demonstrative in the main clause.

Both the declarative and interrogative examples of complement clauses without a complementizer are rather short, suggesting that complement clauses without a complementizer may on average be shorter than those with a complementizer. The presence of a complementizer naturally aids in processing the clause and thus could allow longer complements.

14.2.2 Finite complement clauses with complementizer

The main complementizers are $=s\varepsilon$ and *làpti*, which both derive from verbs of saying (see §14.2.2.1). The more marginal complementizer *ki* is loan from Nepali (see §14.2.2.2).

14.2.2.1 Complementizers = $s\varepsilon$ and lap(ti)

The complementizers $=s(\varepsilon)/s(i)$ and $lap(ti)^{410}$ can occur independently or as a combination $=s\varepsilon$ lap(ti). Finite complement clauses with a complementizer occur especially with verbs of speaking, writing, thinking and knowing.

- (14.23) $\check{a}_{i} \check{q}_{j} \check{s}_{j} \check{s}_{j} \check{s}_{i} \check{$
- (14.24) हे केंद्र वदी गवर्ग्वरा हीं यें लग हैं दें केंद्र बद

t ε [*t* $\varepsilon^h \emptyset$:=*di k'amja p^hi:-po làp-ti*] *t'i-sonzã:* then 2SG.L=DEMPH why be.late-2INF say-NF ask-SIM 'Then when asked [why you were late]...' (RS pupil joke)

(14.25) $\exists r : \mathfrak{g} :$

⁴¹⁰ The nonfinal converbal form *làp-ti* functions analogously with the Nepali complementizer *bhan-era* 'say-NF'.

(14.26) $\underset{i}{\text{g}}$ $\overset{\text{g}}{\text{f}}$ $\overset{\text{g}}{\text{f}}$ $\overset{\text{g}}{\text{f}}$ $\overset{\text{g}}{\text{f}}$ $\overset{\text{g}}{\text{f}}$ $\overset{\text{g}}{\text{f}}$ $\overset{\text{f}}{\text{f}}$ $\overset{\text{g}}{\text{f}}$ $\overset{\text{f}}{\text{f}}$ $\overset{f}{\text{f}}$ $\overset{f}{\text{f}}$ $\overset{f}{\text{f}}$ $\overset{f}{\text{f}}$ $\overset{f}{\text{f}}$ $\overset{f}{\text{f}}$ $\overset{f}{\text{f}}$ $\overset{f}{\text{f}}$ $\overset{f}{\text{f}}$ $\overset{f}}{\text{f}}$ $\overset{f}{\text{f}}$ $\overset{f}{\text{f}}$ $\overset{f}{\text{f}}$ $\overset{f}{\text{f}}$ $\overset{f}{\text{f}}$ $\overset{f}}{\text{f}}$ $\overset{f}{\text{f}}$ $\overset{f}{\text{f}}$ $\overset{f}{\text{f}}$ $\overset{f}{\text{f}}$ $\overset{f}}{\text{f}}$ $\overset{f}}{\text{f}}$ $\overset{f}{\text{f}}$ $\overset{f}}{\text{f}}$ $\overset{f}{\text{f}}$ $\overset{f}{\text{f}}$ $\overset{f}{\text{f}}$ $\overset{f}}{\text{f}}$ $\overset{f}}{\text{f}}$ $\overset{f}}{\text{f}}$ $\overset{f}}{\text{f}}$ $\overset{f}}{\text{f}}$ $\overset{f}}{\text{f}}$

The fact that the honorific form is used in the complement clause shows that (14.23) does not exemplify direct quotation. If the complement were a direct quotation, a non-honorific word choice $(l\partial k-ti \partial n-do \hat{t})$ would be expected, i.e. speakers are not expected to use honorifics when referring to themselves. As a result, the use of honorifics can be used as a test for determining directness vs. indirectness of speech.

According to consultant KN, a complementizer is not needed when the actor of the main clause and the quoted person is the speaker (14.27). If the quoted person is someone else than the speaker (14.28) or if the actor of the main clause is someone else than the speaker (14.29), a complementizer is used. Note that the second clause in (14.27) is functionally a complement clause although the two clauses are, in the absence of the complementizer, formally juxtaposed finite clauses.

(14.27) ה׳ הַאֵּיִקָּהָ׳ אָאָזְ הֵ אָ אָזָ הֵ אָ אָזָי הַ אָדָי אָדָן הָאָזָאָרָ אָאָן nà t'amtea? p'ja-ce t: [ŋà ló gjabma lòk ồ:-ce 1SG promise do-INF EQU.PER 1SG year following return come-INF t:]. EQU.PER 'I promise [I come back the following year].' (KN e)

- ĩ k^hu t'amtsa? р'ја-и [nà ló gjabma lòk promise do-2INF 3SGM 1sg year following return EOU.PER *õ*:-*c*ɛ *î*: làp-ti]. come-INF EQU.PER say-NF 'He promised [that he would come back the following year].' Lit. 'He made a promise [saying I come back the following year].'(KN e)
- (14.29) $\overleftarrow{b} = \widehat{b} \cdot \widehat{b} = \widehat{b} \cdot \widehat{b} \cdot$

Examples (14.27-29) above already exemplify sentences where the complement follows the main clause. Sentences (14.30-31) below are analogous in the order of clauses in the sentence but differ in that the main clause has another filler-word taking the place where complement clause would occur if it were embedded. The filler-word makes the main clause syntactically complete.⁴¹¹ Therefore what follows in the next clause is not in the same way required by the syntax as with

⁴¹¹ The filler-word is somewhat analogous to *that* in English *that*-complement clauses, although in English the filler-word has grammaticalized into a complementizer. In present English, a better analogy is the clause *I know the fact* [*that x*], where *the fact* functions are a filler that makes the main clause syntactically complete even without the complement clause.

embedded complement clauses in (14.27-29). It is noteworthy that this construction occurs several times in the novel Richhi, suggesting some degree of acceptability/standardization of ending a sentence with a complementizer if a coreferential filler-word occurs in the main clause. Now consider (14.30), where the coreferential filler-word is represented by *it* in the English translation. The word in the main clause that replaces the moved complement clause is underlined.

(14.30) कॅंग्લॅंग् याव्र (बद्दे: अन्त: वर्त्त): केंद्र' दे केंद्र' थो यो क्षेत्रका थेंद्र' अन्त क्षेत्र' यात्र' क्षेत्र' थो यो क्षेत्रका थेंद्र' अन्त क्षेत्र' थाये व्रे क्षेत्र' थाये व्रे क्षेत्र' थाये क्षेत्र' थाये क्षेत्र' थाये क्षेत्र'

In (14.30), the main clause $m\dot{u}=lo\ k'and\varepsilon$: $l\dot{a}p$ - $c\varepsilon\ m\dot{e}^2$ could occur independently, because the filler word $k'and\varepsilon$: fills the position where an embedded complement clause would occur. For another example, consider (14.31).

t'izã: tam=di $t\epsilon^h \phi ki = lo \ lap - \epsilon\epsilon \ me^2$ [bhaila t'arin $m \epsilon \eta k^h \tilde{a} = l \epsilon$ but word=DEMPH PN=DAT say-INF EX.PER PN hospital=ABL today ĩ $t^{h}op-to$ làp-ti]. gompo permission receive-IPFV EQU.PER say-NF 'But Choki does not know this fact [that Bhaila is getting a permission to leave the hospital today].' (Richhi 30)

The motivation for placing the complement clause after the main clause in (14.30) and (14.31) is probably that this order is easier to process when the complement is long.

The last two examples below illustrate two alternative strategies for making a complement clause for the noun $rite^{hi}$ 'hope'. The first, (14.32), involves a finite clause with the complementizer lap-ti whereas the second, (14.33), uses a non-finite strategy, genitivized -po-infinitive.

- (14.32) هَمْ بَهْمَ جُوَمَ مَعْرَبُكُمَ عَلَى مَعْرَبُكُمْ مُعْرَبُكُمْ مَعْرَبُكُمْ مُعْرَبُكُمْ مُعْرَبُكُمْ مُعْرَبُكُمْ مُعْرَبُكُمْ مُعْرَبُكُمْ مُعْرَبُكُمْ مُعْرَبُكُمْ مُعْرَبُكُمْ مُعْرَبُكُمْ مُعْمَا مُعْمَا مُعْرَبُكُمْ مُعْرَبُكُمْ مُعْرَبُكُمْ مُعْرَبُكُمْ مُعْرَبُكُمْ مُعْرَبُكُمْ مُعْرَبُكُمْ مُعْذَعُ مُعْرَبُكُمُ مُعْرَبُكُمُ مُعْرَبُكُمُ مُعْرَبُكُمُ مُعْرَبُكُمُ مُعْتُكُمُ مُعْذَعُ مُعْتُكُمْ مُعْتُكُمْ مُعْتُكُمُ مُعْتُكُمُ مُعْتُكُمُ مُعْتُكُمُ مُعْتُكُمُ مُعْتُ مُعْتُكُمُ مُعْذَعُ مُعْتُكُمُ مُعْذًا مُعْتُ مُعْذَعُ مُعْذَعُ مُعْذًا مُعْنَا مُعْذَعُ مُعْرَبُكُمْ مُعْذَعُ مُعْتُكُمُ مُعْذَعُ مُعْذًا مُعْتُكُمُ مُعْذًا مُعْذَعُ مُعْذَعُ مُعْتُكُمُ مُعْتُكُمُ مُعْذَعُ مُعْذَعُ مُعْتُكُمُ مُعْذًا مُعْذَعُ مُعْتُكُمُ مُعْتُكُمُ مُعْتُكُمُ مُعْذَعُ مُعْتُكُمُ مُعْتُكُمُ مُعْتُكُمُ مُعْتُ مُعْتُ مُعْتُكُمُ مُعْتُ مُعْتُ مُعْتُ مُعْتُ مُعْتُكُمُ مُعْتُ مُعْتُ مُعْتُ مُعْتُعُمْ مُعْتُعُمْ مُعْتُ مُعْتُعُمُ مُعْتُعُمْ مُعْتُعُمْ مُعْتُعُمْ مُعْتُ مُعْتُ مُعْتُ مُعْتُعُمْ مُعْتُ مُعْتُ مُعْتُ مُعْتُ مُعْتُ مُعْتُ مُعْتُ مُعْتُ مُعْتُ مُعْتُعُمْ مُعْتُ مُعْتُ مُعْتُعُمْ مُعْتُعُمُ مُعْتُعُ مُعْتُ مُعْتُكُمُ مُعْتُ مُعْتُ مُعْتُعُمْ مُعْتُعُمُ مُعْتُعُمُ مُعْتُعُمُ مُعْتُ مُعْتُكُمُ مُعْتُكُمُ مُعْتُ مُ مُعْتُعُ

14.2.2.2 Complementizer *ki* (loan from Nepali)

The Nepali complementizer ki is occasionally used also in spoken Denjongke. As a sign of its approval as a loan among some speakers, it is used in an example sentence of class 9-10 Denjongle grammar and spelling textbook:

(14.34) ธัริการ์มีพา พราร์า ซุกาทีา โดพา สุมาพ์า ครา มาภูลพ

ts ^h eriŋ=gi	làp-o	be?	[ki	k ^h u:	nàmlo	à:	mi-kjap].
PN=AGT	say-2INF	EQU.NE	COMP	3sgm.agt	ever	lie	NEG-strike
'Tshering s	aid [that he	never lie	s].' (Clas	s 9-10 gram	mar, 136	5)	

14.2.3 Complement clauses with a resumptive demonstrative

These clauses differ from correlative clauses (see §13.3) in that the resumptive demonstrative is not coreferent with the question word in the first clause (as is the case in a correlative clause) but with the first clause as a whole. The complement clause is given in brackets and the resumptive demonstrative is underlined.

(14.36) مَتْ جَحَّتْ حَصَّاتُمَا عَلَى يَعْصَعَاتُ مَعْتَ مَحَتَّا عَلَى يَعْتَقَصَاتُ مَعْتَ beginning at.first 1PL=AGT community say-NMLZ=DEMPH what EQU.NE.Q <u>ódi</u> hako go:-po k'ɛ:tɕʰī: í:. that understand be.needed-2INF important EQU.PER 'First, it is of importance to need to understand [what (the thing) called society is].' (Richhi 7)

Examples (14.35) and (14.36) resemble clauses (14.30-31) in having a filler-word in the main clause which is coreferent with the whole complement clause. There are, however, three differences. First, the order of the clauses is different, in (14.30-31) [main clause + complement clause] and (14.35-36) [complement clause + main clause]. Second, in (14.35-36) the filler-word is a demonstrative whereas (14.30-31) use other filler-words. Third, the complement clauses in (14.30-31) have a complementizer whereas the ones in (14.35-36) do not.

14.3 Summary remarks

longer on average than those without the complementizer, suggesting that the presence of the complementizer aids procressing and enables longer clauses. Denjongke was seen to have two main complementizers, which can be used indendently or together, and a third form borrowed from Nepali. The last section introduced complement clauses with a resumptive pronoun, which differ from correlative clauses in that the resumptive demonstrative is coreferent with the whole complement clause, not one word in the complement clause (as in correlative clauses).

15 Adverbial clauses

This chapter discusses adverbial clauses. The other types of subordinate clauses are treated in §13 (constituent complement clauses, including relative clauses) and §14 (clausal complement clauses). Adverbial clauses modify the verb complex or the entire clause (Thompson et al 2007: 238). The treatment begins with an introduction in which the various constructions used in adverbial clauses are categorized according to form into four types (form-to-function ordering) (§15.1). In the actual discussion after that, the various constructions are described under functionally motivated headings (function-to-form ordering). The functional headings are the following: nonfinal clauses (§15.2), temporal clauses (§15.3), causal clauses (§15.4), purposive clauses (§15.5), conditional clauses (§15.6), concessive clauses (§15.7), circumstance and manner clauses (§15.8), additive clauses (§15.9), substitutive clauses (§15.10), comparative clauses (§15.11) and various uses of the terminative converb (§15.12).

15.1 Introduction to forms

The types of formal marking used in adverbial clauses are converb, postposition with accompanying modifying clause, noun with accompanying modifying clause and other types of marking respectively, see Tables 15.1-4. Note that the form kap (WD MPN skabs) occurs both as a converb (*-kap*) attached directly to the verb root and as a noun (*kap* 'time') with a nominalized and genetivized complement clause.

Table 15.1. Converb	al endings used in adv	erdial clauses
-ti/di		nonfinal
-pa/ba	<u>ち(</u> れ)/ち(れ)	circumstantial-purposive
-(patee)ne	(યમજે)ઠે	conditional
-ruŋ	र्डूट.	concessive
-sondãː, -somdãː,	אדיקדי, אדמיקדי	simultaneous
-tsubdã:	ૡૢૼ ऺਙ ॱ ॸੑੑ <u></u> ॸॱ	simultaneous
-kap	শ্বনিধ'	simultaneous
-dy:	<u>5</u> ar.	simultaneous
-renk ^h a	रतु'यर्'	simultaneous
-sãː, -sonzãː	<u> </u>	terminative

Table 15.1. Converbal endings used in adverbial clauses

nàŋɛa, nàŋlo	बन्दाः, बन्दांसं	'inside', circumstantial
t'ønlo	र्नेन'र्यें'	'for the purpose of', purposive
gjablɛ	র্য্যুন্ম'এম	'after', anterior
pénle	ই্ইন্ব'এম্ব'	'before', posterior
nàŋtar, nàŋzin	वर्तः क्षेत्रः, वर्त्तः विवः	'according to', comparative manner
t'ønzin(gi)	र्देव'नविव'(ग्रीष')	'in accordance with, in view of', comparative manner
p`aːpu	ন্দ'র্য	'in between', simultaneous

Table 15.3. Nouns heading adverbial clauses

ken	र्गुव.	'cause, condition', causal	
gã:	到上、	'time', simultaneous	
kap	শ্ধবন্ধ'	'time', simultaneous	

1 auto 13.4. Other constructions used in autorbial clause	Τa	able	15.4.	Other	constructions	used in	adverbial	clauses
---	----	------	-------	-------	---------------	---------	-----------	---------

<i>=ki/gi</i> (agentive)	-(শ্বী)শ্ব	causal
$=l\varepsilon$ (ablative)	-910	anterior (temporal)
-po-dã: ⁴¹²	<u> </u>	2INF-CONJ, simultaneous
k'amjasene	শ্বব'র্দ্রুম্ব'ম্বি'	connector 'because', causal
mits ^h e?	ঈ'র্ক্র'	connector 'in addition' (lit. 'not stopping'), additive
mèmbo	মন্ব'র্ন'	connector 'except (+negation in the following main clause)',
		negated additive
lồ: mè:po	र्वेट केट र्धे	'no soonerthan', 'as soon as', immediate anteriority
-po tsamgi	র্থা র্তুঝায়্রীঝা	'as soon as', immediate anteriority
làpti	વ્યત્ર ક્ષે.	nonfinal converb of <i>làp</i> 'say', purpose and causality
tcẽ 1/zẽ 1/zin	नवित्र'	progressive marker, circumstance/manner

The analysis here applies the term "converb" to several Denjongke verbal forms. Therefore a brief discussion on converbs is in order. Haspelmath (1995: 3-8) defines a converb as "a non-finite verb form whose main function is to mark adverbial subordination". Converbs are subordinate forms, which modify verbs but not nouns. They are characterized as verbal adverbs which do not function as either clausal or nominal complements (Haspelmath [1995: 3-8]). This definition fits Denjongke with two caveats. The first caveat is that, similar to many other Tibeto-Burman languages such as Mongsen Ao (Coupe 2006: 146), one of Denjongke converbal suffixes, nonfinal *-ti/di*, can be used, in Coupe's (2006: 146) words, "in a more coordinative manner" to form clause chains, which describe sequences of events. The second caveat is that the conditional converbal suffix may attach either to a verb root, in which case the construction is clearly non-finite, or it may attach to the final auxiliary of a periphrastic construction with temporal and aspectual values, both features of finiteness. Only those verbal suffixes that attach directly to the verb root and involve no clear nominalization are here considered converbal suffixes.

Other verb-modifying adverbial clauses involve nominalization and the use of cases and postpositions. The ten converbs are the nonfinal converb marked by -ti/di, the circumstantial-purposive converb marked by -pa/ba, the conditional converb marked by -(patce)ne/-(batce)ne, the concessive converb marked by -ruy, the simultaneous converbs marked by $-pod\tilde{a}$:, $-sond\tilde{a}$:/ $somd\tilde{a}$:, $-tsubd\tilde{a}$:, -kap, -dy: and $-reyk^ha$ respectively, and the terminative converb marked by $-sonz\tilde{a}$:, see Table 15.1. These converbs occur at various levels of specialization. For instance, the nonfinal one is, in Nedjalkov's (1995: 106-110) terms, "contextual" in that its meaning is context-dependent, whereas the conditional and concessive converbs are highly "specialized" in that their basic meaning is invariable.

⁴¹² This form, along with *-sondã:*, *-somdã:* and *-tsubdã:* often occurs in a reduced pronunciation without nasalization on the final vowel, *-poda(:)*, *sonda(:)*, *-somda(:)* and *-tsubda(:)*. The forms given here reflect a more conservative pronunciation

15.2 Nonfinal converb marker -ti/di

The most basic function of the nonfinal converb marker -ti/di is to indicate that another verb is following, i.e. that the sentence is not complete. Therefore -ti/di is here glossed merely as nonfinal (NF).⁴¹³ In practice, most of the instances are conceived as anterior to the main verb but in many cases, as shown below, the verbal action marked by -ti/di is temporally simultaneous to the main verb. This is especially the case when the nonfinal converb modifies a nonspecific verb of motion, for instance gju (WD rgg) 'go'. In a clause-chaining language such as Denjongke, the nonfinal converb is often used where non-clause-chaining languages such as English would use coordination. The most natural English translation is often, as in (15.1), mere 'and'.

(15.1) $\vec{r} \cdot \vec{r} \cdot$

The clause-chaining construction in (15.1) may be described as dependent in that it is followed by another verb form.⁴¹⁵ On the other hand, the clause-chaining uses of -ti/di are not clearly subordinate in the sense that they would modify the matrix clause. Hence the clause-chaining uses have been termed "cosubordinate," see Van Valin and LaPolla (1997: 453-454) for a general discussion and Watters (2018) for application to the related language Dzongkha. More complex clause-chaining with -ti/di is illustrated in (15.2) with four instances of the nonfinal converb, each given in bold.

(15.2) วังซัง กรากราชิญ รังยุรา อิรางสา อูสาอิ สมัย สมัยสายสายมา อิสารคิสา มีราวัง สมัยราชิง เสมาร์รา สายงาน รังสา เนยูาอิง กังสายอิสา युगा र्चेषा चेंत नवित प्रा गेत है हैंन येंन प्रा भीत भाष p'otso tã:tã:=tci? $do-p^h u\eta$ k^hja? $t \in \eta = lo$ **lum-di** go=le fall-NF head=ABL blood child middle.size=INDF stone-pile top=DAT $t^h \phi n - z \tilde{\epsilon}$: jờ-po t^hõː-ti *ámp^hi* nàn=le tora come.out-PROG EX-2INF see-NF breast.pocket inside=ABL handkerchief t^hu-ti tora=gi $k^h ja?$ $p^{h}i$: $p'in-z\tilde{\epsilon}$: p^haːpaː **lèn-di** dø: wipe give-PROG lap.LOC take-NF stay pick-NF handkerchief=AGT blood jø-pø: kap EX-2INF.GEN time 'When she saw that a middle-size child had fallen off a stone-pile and blood was coming out of his head, she picked a handkerchief from her breast-pocket and, wiping out blood with the handkerchief, took (him) into (her) lap...' (Richhi 2)

The first converb (*lum-di* 'having fallen') in (15.2) forms a chain with another verb ($k^h ja^2 t^h øn$ 'bleed'). Both verbs in the chain express what happened to the same referent, a child in the story.

⁴¹³ See Watters (2018) for the same glossing for the cognate category in Dzongkha and DeLancey (1991: 3) for the same glossing of a functionally equivalent category in Lhasa Tibetan.

⁴¹⁴ The voicing/voicelessness of *-ti/di* following a velar nasal (usually realized as a nasalized vowel) has to be learnt on a case by case basis, for instance t^huy -di 'drinking', $s\delta$:-di 'going', δn -di 'coming', riy-di 'becoming long', but $mj\delta y$ -ti/mj\delta:-ti 'finishing', $tc^h\delta$:-ti 'rushing' (see also §2.8.1).

⁴¹⁵ For a defence for allowing converbs to mark clause-chaining (contra Haspelmath 1995), see Genetti (2005).

This clause-chain, then, becomes the complement of the first of three nonfinal verbs ($t^{h}\tilde{o}:-ti$ 'having seen', $t^{h}u-ti$ 'having picked up', $l\tilde{e}n-di$ 'having taken'), which describe three consecutive actions of another person, a certain lady in the story. Note that in addition to the nonfinal clauses there are also other dependent clauses in (15.1). The whole example forms a simultaneous clause ending in *kap* 'time' (see §15.3.3.4), and there is also a manner clause marked by the progressive $z\tilde{e}$: (see §15.8.3) and a complement clause ending in *-po*-infinitive (see §14.1.1).

In addition to anterior sequence, the verbal action marked by -ti/di may be simultaneous to the verbal action that follows. In these cases, the converb may be seen as an adverbial modifier to the accompanying action, a more typically converbal function than clause chaining illustrated in (15.2). The converbal construction itself does not suggest a difference between sequential and modifying relationship (similarly Genetti 2005: 50 for Dolakhā Newar). The context, which to a high degree consists of verb choice, determines the interpretation. While example (15.3) is ambiguous between anterior/consequtive ("having sat down") and simultaneous reading ("sitting"), in the examples in (15.4-6) the simultaneous reading is preferable. The nonfinal converbs in (15.4-6) modify the general verbs gju 'go' and ta 'look' by specifying manners of going and looking.

- (15.3) ﷺ الحالية الحالية المحالية ال
- (15.4) यत्र र्स्ट्र केषा देग्रे हे क्षेत्र के

 $p^{h}a:ts^{h}u:$ mìk **gir-di** ta-sã: thither-hither eye stare-NF look=TERM 'When/as (she) looks intently here and there...' (Richhi 98)⁴¹⁶

(15.5) ८.२४.२५ अर्घर रे. या या वर्षे रे. या वर्षे या वर्षे रे. या वर्षे य

 $p\dot{a}=lo=to$ $t^{h}\tilde{o}$:-ruŋ ma- $t^{h}\tilde{o}$:-po $d\varepsilon m$ **zu:-di gju**: 1SG=DAT=CEMPH see-CONC NEG-see-2INF like.it avoid-NF go.2INF $b\varepsilon$?. EQU.NE 'Although (he) saw me (he) walked avoiding me, as if not seeing (me).' (nga'i 'gan 16)

(15.6) a) झुत्रायम् तु छिया हि देशु त

 $m \epsilon \eta k^h \tilde{a}$:=na $k^h i k$ -ti gju-n ϵ hospital=LOC lead-NF go-COND 'If (he) is taken to the hospital...' (rnam-rtog 7)

⁴¹⁶ The construction $ta = s\tilde{a}$: is analogous to Nepali *her-e-samma* [look-PFV-until] 'when looking, as far as one can see'.

b) ନି' ଝି'ଦେନି'ଡିସା'ग' ମ' ମ' ଝାଁଛିଳ' ଦନି'ର୍ଟ୍ ମିଷା ଦରସା ହ୍ୱି' ଦଣ୍ଡ'ସହି'ମି', ଦଣ୍ଡ'ମମ'ମ୍ରି' ମୁଷା ଝିଁମ' ହ୍ରାସଷା ଅମ

t'o? bak-ti ódɛtɕika námo di=tsu tε t'a gju-wøz=gi, then that.time now camel these=PL load carry-NF go-2INF.GEN=GEN t'ytshø? [ep-tsha-khen he?. gju-ce=ki go-INF=GEN time arrive-CMPL-NMLZ EQU.NE 'Now then at that time, the time had come for these camels to go [carrying loads].' (PD bet story)

Whereas in (15.3), the converbal form expresses an attendant circumstance to the finite verb, in (15.4) and (15.5) the converb, rather than joining two clauses, functionally forms a complex predicate with the main verb. These types of constructions, in which the latter verb is typically a verb of motion, have given rise to serial verb constructions by dropping the converbal marker. The clauses in (15.6) exemplify such forms where the converbal ending could be dropped, $k^{h}ik$ -ti gju > $k^{h}ik$ gju 'bring (a human or a large animal)', bak-ti gju > bak gju 'take away (a thing)'. For more details on serial verbs, see §4.2.3.

When the converb marked by -ti/di is used, the actor of the converbal clause and the main clause are usually the same. Indeed, by using the nonfinal converb, the speaker typically indicates that the following verb has the same actor as the converbal clause. In contexts where the actor switches, one of the simultaneous constructions is typically used (see §15.3.3). These facts suggest that Denjongke is developing a switch-reference system, which has already been described for the Tibetic languages Shigatse Tibetan (Haller 2009) and Dzongkha (Watters forthcoming). However, example (15.7) provides an exception to the rule that -ti/di presumes an unchanged actor. This example, I suspect, may be somewhat confusing to the hearer, because the changed actor of the last clause is not explicitly mentioned.

क्षेंग्वदीर्द्युया यु. ग्रींटरवर यूटर वर्ळ्या वर्य्या वर्ळ्या वर्य्या केंटर्ष्ट्र ग्रींटर यहेवगुवर यूटर का यहेवा घेंचर्ष्ट्र यूटर वहेर वहेर केंटर्स्य (15.7)เนณ ปอยา พยาร์า มียา k^hu lấ: ódi-p'ja k'iõ:=na ts^hø: bak ts^hø: bak số:-di bull search carry search carry go.PFV-NF that-ADVZR 3SGM village=LOC $tc^{h}a$ tci? $t^{h}op-ti$ $l\tilde{a}$:=dik'jõ: lấ: dzo:=ditci:=na village one=LOC bull pair one find-NF bull=DEMPH price=DEMPH tõ:t^ha? khe: làp-o bɛ?. tci? thousand score one say-2INF EQU.PER 'Therefore he went into villages searching and searching and in one village he found a pair of bulls and (the seller) said the price was 20,000 rupees.' (TB bull story)

The nonfinal converb construction is typically used for expressing how long the results of a previous action have existed:

(15.8) مَجْ العَمَّةُ جَ مَعَمَّةً عَنْ مَعَاجًا مُعَاجًا مَعَاجًا مَعَاجًا مُعَاجًا مُعَ معاجمع م المحاجمع معاجمع مع

In spoken language, -ti/di is often accompanied by the marker -ki/gi (or =ki/gi), which looks like a genitive or an agentive, see (15.9). This form almost never occurs in written language, although

the novel Richhi has one instance written $\hat{\eta} \cdot -ki$ ($\neg \eta \neg \dot{\vartheta} \cdot \dot{\eta} \cdot bshad-sti-ki$ [cétiki] 'saying'), suggesting that the author considers the marker a genitive ($\hat{\eta} \cdot -ki$) rather than an agentive ($\hat{\eta} \cdot -kis$).

(15.9)Rundan Wart महा पर रहे. में स्टार्थना प्रया गहिना पहिना हीत होता ही सार पर रहे. ही ही की प्रया प्रता की सार की $k^h u = gi$ *ódi lấ:=di=lo* tõ:t^ha? k^he: tei? pí: p'in-diki $k^h u$ 3SGM=AGT that bull=DEMPH=DAT thousand score one two give-NF 3SGM *lấ:=di* $k^{h}i?$ ∂m -bo nòː-ti *bε*?. bull=demph buy-nf lead come-2inf equ.ne 'Giving 22,000 for the bull, he bought the bull and brought it (home).' (TB bull story)

The nonfinal converb may be followed by the dative-locative =lo when the clause expresses manner of doing:

(15.10) रुपाक्षे'र्ये' रुपारुपा र्घुषाणे।

rup-ti=lo join.together-NF=DAT together do-HORT 'Let's join and do it together.' (LT e)

The nonfinal marker does not co-occur with the negator prefixes, *ma-VERB-ti, *mi-VERB-ti⁴¹⁷. Negation is accomplished by a strategy identical with the circumstantial converb, ma-VERB-pa. This is quite natural, because when an anterior action is negated, it in effect becomes a manner or circumstance for the following action. For instance, when the *eating* in *I ate and came* is negated, it becomes a manner or circumstance of coming, *I did not eat and came* > *I came without eating*. For examples on negating the circumstantial converb -pa/ba, refer to §15.8.1. There is, however, a circumlocution of the unacceptable form *ma-VERB-ti which combines nominalization and the nonfinal form of the verbalizer p'ja 'do'. In (15.11), in order to avoid the infelicitous *ma-VERB-ti, the speaker nominalizes the expression as man-zak-o and assigns the nonfinal marking to the following verbalizer. A functionally equivalent alternative would be to use the negated circumstantial converb man-VERB-(p)a.

(15.11) a) केर-र्झुंग्र, केंट्र' के फ़ेंग' रे यत्र पालगार्गे र्झुक्ष क्षेत्र किता केंद्र

sé:bjam, $tc^h \phi$? minto? tcc: **man-zak-o p'ja-ti** $dzip-k^h \tilde{c}$: \tilde{t} : bumble.bee 2SG.L flower at.all NEG-leave-2INF do-NF suck-NMLZ EQU.PER 'Bumble bee, you are one who sucks flowers without leaving any aside.' (RS bee story)

⁴¹⁷ I have one recorded exception, see *mi-ts^he-tiki* in example (12.22). Moreover, consultant KN commented that he has heard some people use forms *mi-kjap-ti(ki)* or *ma-kjap-ti(ki)* but he considers them incorrect forms.

In my written data, the nonfinal marker does not occur with copulas. Consultant KN, however, commented that while using -ti/di is perfectly acceptable with the existential $j\partial^2$, see (15.12), its use with the equative \tilde{i} : is limited to uses in ritual language of incantations, see (15.13), where the translation is tentative.⁴¹⁸

- ke:=di teeta? jø-ti raŋ=gi raŋ-gi ke? kjap-ce self=GEN strike-INF language=DEMPH lovable EX-NF self=GEN language *bε*?. EOU.NE '(One's) own language being lovable, (he) speaks (his) own language.' (YR canteen video)

15.3 Temporal clauses

Temporal clauses are divided into anterior, posterior and simultaneous clauses, based on the temporal relationship between the subordinate and main clause. Anterior clauses mark verbal action that happens before the following main verb. It was already shown in §15.2 that nonfinal clauses are typically interpreted as anterior. Anterior constructions described here, however, mark anteriority more explicitly than the nonfinal converb, which may also have simultaneous uses expressing manner, as described above. Posterior constructions, or 'before'-clauses, mark action that happened after the action presented by the verb in the following main clause. Simultaneous constructions, on the other hand, mark actions that are at least partly simultaneous with the following verb.

15.3.1 Anterior clauses

General anteriority can be marked by two constructions. The first construction is formed by the ablative marker $=l\varepsilon$, which can attach to *-po*-infinitive or the completive marker $-ts^ha$: (§15.3.1.1). The second construction expressing general anteriority employs the postposition *gjable* 'after' or $dz\varepsilon$:(*lo*) 'after' (§15.3.1.2). In addition, two constructions (VERB-*lõ*: *mè*:-*po* and VERB-*po tsamgi*) express immediate anteriority, which corresponds to the expressions "as soon as" and "no sooner than" (§15.3.1.3).

15.3.1.1 Anteriority with ablative $=l\varepsilon$

The ablative marker = $l\varepsilon$ may be combined to -*po*-infinitive (15.14), the completive - $ts^{h}a(:)$ (15.15) or its infinitivized form - $ts^{h}a$ - $u/ts^{h}o$ -u (15.16) to mark an action that precedes the action coded by

⁴¹⁸ However, I have some initial evidence that some speakers may accept a wider use of *in-di*.

the following main clause. Of these forms, the first and the last occur both in the spoken language and the novel Richhi, whereas the form $ts^{h}a = l\varepsilon$ has been only attested in the spoken language.

- (15.14) เหมา ฏักลารับเขพา การขา เมิมาสา มีกายกา อุกกาชกา áт *t'õ:-po=lɛ* $nàtea? k^{h}im=na$ $d\emptyset$:- $c\epsilon$? $t\tilde{a}$:- ts^ha :. mother die-2INF=ABL 1PL house=LOC live-INF leave-CMPL 'After mum died, we stopped living in (our) house.' (Richhi 133) (15.15) 55 र्ज थेंतु याक र्रेग सुवा मुनक के रायक हागा करी हुंग याक के रायक हा हुंग के रायक के राय के रायक के राय के रायक के रायक के राय jõu=le dokt^hum kjap-ts^ha=le hãːtoma ta:=di $p'ja:=l\varepsilon$ òtε up=ABL kick do-CMPL=ABL tiger=DEMPH cliff=ABL suddenly down lum-di cí-u ĩ. fall-NF die-2INF EOU.PER 'After/since (he) had given a kick from above, the tiger fell off the cliff and died.' (KT animal story)

As shown by the options in the English translation of (15.15) and (15.16) ("after/since"), in some contexts the use of the ablative may suggest a causal rather than simply anterior reading. Causal interpretation of ablative marked clauses is particularly typical of stative verbs, see §15.4.5.

The construction with the nominalized completive form is realized in spoken language in various reduced forms, depending on the language community. The following forms occur in my data: $-ts^{h}ou=l\varepsilon$, $-tsub=l\varepsilon$, $-ts^{h}o:=l\varepsilon$ and $-so:=l\varepsilon$. The last two are phonologically conditioned variants given by the same speaker. The form $-tsub=l\varepsilon$ was said by a consultant to be a reduced form of $ts^{h}a-u=l\varepsilon$ ($< ts^{h}a-wo=l\varepsilon$). For examples, consider (15.17-19). The form $-tsub=l\varepsilon$ is here illustrated by two examples (15.18-19) by two different consultants from different locations (Bermeok and Lingdum) to show that the form is not restricted to one locational variety of Denjongke:

(15.17) नूरागुायका धारा दहेंगार्केंदुायका यमाञ्चर केंवा हिमा केंवा धेंदा

 $tarku=l\varepsilon \ ja$: $dz \varepsilon k-ts^h o-u=l\varepsilon$ $lam-m\varepsilon^2$ $ona \ k^h im \ ona \ ja^2$. TPN=ABL up climb-CMPL-2INF=ABL road-below there house there EX.PER 'Having climbed up from Tarku, there's a house below the road there.' (KT discussion with TB) nàtea? eí-ne cí-tsub=le làlo mí lèm sǿnam тí 1pl die-COND human die-CMPL.2INF = ABL some human good merit zanpo jø:-nε=di EX-COND=DEMPH good 'If we die, after a human dies, if someone is a person of good fortune...' (KT discussion with TB)

ki:-tsub=le p'a:=dibɛ?. ťa ki-u átsi gjaple interval=DEMPH grow-2INF EQU.NE now grow-CMPL.2INF=ABL a.bit after hapta pí: súm gjaple t'a di=lo ko: **30:-6**8 *bε*?. tε, then week two three after now this=DAT dig be.needed-INF EQU.NE 'Meanwhile it grows. Then a little after having grown, after two or three weeks it then has to be dug.' (PL interview)

The forms $-ts^{h}o:l\varepsilon$, see (15.20), and $-so:l\varepsilon$, see (15.21), were conditioned in the speech of consultant SGD so that the reduced form $-so:l\varepsilon$ occurred when the verb had an initial affricate⁴¹⁹ (e.g. $ts^{h}\tilde{a}:-so:=l\varepsilon$ 'after completing', $t\varepsilon a:-so:=l\varepsilon$ 'after coming'), whereas the fuller form $-ts^{h}o:l\varepsilon$ occurred otherwise (e.g. $mj\tilde{o}:-ts^{h}o:=l\varepsilon$ 'after finishing', $dik-ts^{h}o:=l\varepsilon$ 'after arranging'). The variant pronunciations $-ts^{h}o:l\varepsilon$ and $-so:l\varepsilon$ illustrate the tendency of Denjongke to reduce or elide the second aspirated sound in a word, especially if the second aspiration occurs in a verbal suffix.

- (15.20) المعرَّمْ عَلَى اللَّهُ اللَّهُ عَلَى اللَّهُ اللَّهُ عَلَى اللَّهُ اللَّهُ اللَّهُ عَلَى اللَّهُ اللَّلُلَ اللَّهُ اللَّلُ
- (15.21) ਸ਼ੁ'ਕਸ਼ੇ' ਕ੍ਰੇਕਾਰ' ਧਣਸ'ਛੱਸ਼ੇ' ਕਾਲਾ ਕਸ਼ੇ' ਸ਼ੁਕੇ'ਸੀ' ਘਾਧਾ ਸ਼ਸ਼' ਕਸ਼ੇ'ਸੀਆ $p'umy: k^{h}im=na$ tca:-so:= $l\epsilon=di$ p'u-i-gigirl.GEN house=LOC come.HUM-CMPL.2INF=ABL=DEMPH boy=GEN=GEN ja:p $t'\tilde{a}: juma=di=gi$ father.HON and mother.HON=DEMPH=AGT 'Having come to the girl's house, the boy's father and mother...' (SGD wedding customs)

15.3.1.2 Anteriority with the postpositions gjable 'after' and dze:(lo) 'after'

Anterior constructions are also formed by attaching the postposition gjablo/gjable 'after' or dze:(lo) 'after' to the genitive form of a nominalized verb. As suggested by the three examples below, the construction is particularly common with the secondary verb $mj\tilde{o}$: 'finish'. Note that in (15.23) the nominalizer attaches to the completive marker $-ts^ha$:

⁴¹⁹ Consultant RS has *-so:le* also with initial aspirated lateral /l/, e.g. lep-so:=le (RS) vs. $lep-ts^ho:=le$ 'after arriving' (SGD).

(15.22) מַלְיָמִשׁי מַשִּׁשִי אַשָּריש הארי אַן מוש אַ אַריאַמי בי אריאַמי קטי דעצרי ארי אריאר איז אַריאַמי בי אריאַמי קעיביאר אריאי אוי אַריאַמי אַריאַנעי אַרי אַריאַ אַיעאי אַזיין אַריאַגע אַריאַריאַן אַיעאַ אַרי र्ये क्षेम् dile [dendzon $\epsilon \dot{\epsilon}$:da $t \epsilon^{h} o$? dok mjőː-**pø**: gjable] nà gã:to? institute doctrine read finish-2INF.GEN after then Sikkim 1SG TPN ĩ:. perdzø? lakhã: nànca jó? p'ja-zin-po-p'ja dø:-po library shrine inside word do-PROG-2INF-ADVZR live-2INF EOU.PER 'Then, [after finishing studies at the (Buddhist) institute,] I lived working at a Buddhist

library in Gangtok.' (RB life story)

(15.23) هَامَ عَامَ مَحَانَ هَامَ عَانَ مَعَانَ مُعَانَ مَعَانَ مَ معامَعَانَ مَعانَ مَعَانَ مَ معامَعَانَ مَعانَ مَعانَ مَعانَ مَعَانَ مَ مَعْنَا مَعَانَا مَعَانَ مَعَانَا مَعَانَ مَعَانَ مَعَانَ مَعَانَ مَعَانَ مَعَانَ مَعَانَا مَعَان

(15.24) שאי הקריאי אלריאראי איזייאי

 $[tc^{h}o^{2} c\dot{\varepsilon} ti mj\dot{o}:-sum gjable]$ doctrine tell-NF finish-go.PFV.NMLZ(?) after '[After finishing (his) teaching]...' or 'He finished teaching. Afterwards...' (TB boat story)

Note that the verb in (15.24) is not genitivized. The formative *-sum*, which typically occurs in the simultaneous construction *-sumdã*: (see §15.3.3.2) is here used as the sole verbal marker. Consultant KN commented that *gjablɛ* here is probably rather a clause initial adverb (see the second translation option) than an anteriority marking relator noun.

The postposition $dz\epsilon$: (lo) 'after' is used by some speakers alongside $gjabl\epsilon$. According to some of my consultants, $dz\epsilon$: (lo) is more Central Tibetan than Denjongke.

- (15.26) $a\overline{f}$ ' $\overline{z}\overline{z}$ ' $\overline{f}\overline{g}\overline{q}$ ' $u\overline{g}\overline{q}$ ' $\overline{z}\overline{z}'$ ' $\overline{z}u\overline{g}\overline{q}\overline{q}\overline{n}'\overline{z}u\overline{c}'$ ' $\overline{f}\overline{g}\overline{q}'$ ' $u\overline{g}\overline{q}\overline{n}'\overline{z}u\overline{c}'$ ' $\overline{f}\overline{g}\overline{q}'$ ' $u\overline{g}\overline{q}\overline{n}'\overline{z}u\overline{c}'$ ' $\underline{f}\overline{g}\overline{n}'$ ' $\underline{f}\overline{g}\overline{q}'$ ' $\underline{f}\overline{g}\overline{n}'$ ' $\underline{f}\overline{g}\overline{q}\overline{n}'\overline{z}u\overline{c}'$ ' $\underline{f}\overline{g}\overline{n}'$ ' $\underline{f}\overline{n}'$ ' $\underline{f}\overline{n}'$ ' $\underline{f}\overline{g}\overline{n}'$ ' $\underline{f}\overline{n}'$ ' $\underline{f}\overline{n}''$ ' $\underline{f}\overline{n}''$ ' $\underline{f}\overline{n}''$ $\underline{f}\overline{n}''$ ' $\underline{f}\overline{n}'''$ ' $\underline{f}\overline{n}''$ '

15.3.1.3 Immediate anteriority

Two constructions mark immediate anteriority, corresponding to the English expression 'as soon as' and 'no sooner than'. These constructions emphasize that the action in the main verb follows immediately after the action depicted by the subordinate verb. The first construction, which is more frequent, is VERB $l\tilde{o}$: $m\dot{e}$:po, consisting of the modal secondary verb $l\tilde{o}$: ⁴²⁰ 'have time to' (see §8.5.10) and the negated existential in -po-infinitive $m\dot{e}$:-po.

- (15.27) تَن جَنَحُمَّ: حَنَحَمَّ: حَنَحَمَّ: حَنَّحَمَّةُ عَنْ جَنَحَمَّةُ عَنْ جَنَحَمَّةُ عَنْ جَنَحَمَّةُ عَنْ حَنْعَانُ مَعْنَا مَا مَعْنَا مَا مَعْنَا مَعْنَا مَنْ وَعْنَا مُعْنَا مُنْكَمَا مَعْنَا مَنْ عَنْنَا مَعْنَا مَعْنَا مَا مَعْنَا مَا مَا مَعْنَا مَا مَا مَعْنَا مَعْنَا مَا مَا مَا مَا مَعْنَا مَا مُ
- [ŋà dɛndzõː=lɛ lố ágja ágja lòk-ti lep elder.brother elder.brother 1SG Sikkim=ABL return-NF arrive have.time.to mèː-po] jìgi gu: gu:-pa. EX-2INF letter wait wait-CIRC 'Brother, brother, [as soon as (or: since) I arrived back from Sikkim], I have been waiting and waiting a letter.' (Richhi 146)

The second and less frequent construction is VERB-po tsamgi, which is based on WD star tsam 'barely'.

(15.29) झ्रयाका वर्त्ते यान्त्र वान्ह्रेया वान्ह्रया वान्ह्रयां का यान्यां का स्वायां का स्वायां का स्वायां का स्वायां का स्वायां का स्वायां की स्वायां की स्वायां की स्वायां की स्वाय स्वय स्वाय स्व ta:=di=jã: k'atep [àːnja=gi làp-**o** tsamgi] súpø: tiger=DEMPH=too how marten=AGT say-2INF as.soon.as belly.GEN sù: $k^h a k^h a = gi$ làko símo=gi t'õ: p^huk-tiki corner mouth=AGT hand nail=AGT hole pierce-NF 'Somehow the tiger too, [as soon as the marten had said (so)], pierced a hole in the corner of the (elephant's) side with (his) mouth and hand nails, (and)...' (KT animal story)

(15.30) अर्वेद्तर्भें र्ड्यांग्रेस रत्तर्पत्वित्रंग्रें सेव्रसार्थें दद्ते द्वें कें रत्त क्षेत्रं से व्रकें में कें केंद्र

 $\begin{bmatrix} t^{h}\tilde{o}:-po & tsamgi \end{bmatrix} \quad r\tilde{a}:zingi & s\epsilon m=lo=di & d\epsilon mo=r\tilde{a}: \\ see-2INF & as.soon.as & by.itself & mind=DAT=DEMPH & she.demon=AEMPH \\ b\epsilon:=s\epsilon & n\delta:-bo & mi-ts^{h}\epsilon? \\ EQU.NE=QUO & think-2INF & NEG-stop \\ `[As soon as (she) saw (her)], (she) automatically thought that (this) is indeed a she-demon and moreover...' (rna-gsung 5)$

15.3.2 Posterior clauses with postposition *pénle* 'before'

Posterior constructions with the postposition $p\acute{e}nl\epsilon$ 'before' (also $p\acute{e}nlo$ and $h\epsilon nl\epsilon$) mark action that happened after the action presented by the verb in the following main clause. This means that the order of the actions in the sentence is contrary to the real life temporal sequence. Similarly to

⁴²⁰ homophonous with $l\tilde{o}$: 'stand'

⁴²¹ The use of the genitive here is surprising, perhaps a mistake. The speaker is referring to a group of five people one of whom he is himself.

constituent-modifying clauses (relative clause, noun complement clause and postposition complement clause), the complement of *pénle* is a genitivized *-po*-infinitive, see (15.31-33). Note that the posterior clause in (15.33) has an embedded nonfinal clause (*mú: jìgi cík-ti*).

- (15.31) المحتجة: مكتر مرفقة معن المحققة المحتجة: مكتر المحققة المحتجة: مكتر المحققة المحتجة: مكتر المحققة المحتجة: مكتر المحتجة (Istaining beginning plant-2INF.GEN before PN=AGT friend=PL=DAT '[Before beginning the training], Lhaki (says) to the friends: '(Richhi 64)
- [t'izã: lú pèm-**bø**: $n \epsilon n lo$] $t \epsilon^{h} \phi$:=lo tam=tci? làp-a song listen-2INF.GEN before 2SG.L=DAT word=INDF but say-PUR ĩ òm-bo gjalts^hen. come-2INF EQU.PER PN "[But before listening to (your) song], I have come to tell you a message, Gyaltshen." (nga'i 'gan 24)
- (15.33) هَلَّمَّ سَابَ مَوْمَا جَجَ عَلَمْ مَالَمُ عَلَمْ مَنْ عَلَى مَنْ عَلَى مَنْ عَلَى مَنْ عَلَى مَنْ عَلَى مَنْ عَلَى مَالًا عَلَى مَا مَالًا عَلَى مَالَى مَالًا عَلَى مَالًا عَلَى مَالَي مَالَى مَالَكَ مَالًا مَالَا عَلَى مَالَا مَالَكَ مَالَى مَالَى مَالَى مَالًا عَلَى مَالًا عَلَى مَالَى مَالَا عَلَى مَالَا عَلَى مَالًا عَلَى مَالًا عَلَى مَالًا عَلَى مَالَى مَالَى مَالَى مَالَى مَالًا عَلَى مَالًا عَلَى مَالًا عَلَى مَالَى مَالَى مَ مَالُولُولُكُلُولُ عَلَى مَالَى مَاللَا عَلَى مَالَى مَالَى مَالَى مَالَى مَالَى مَالَى مَالًا عَلَى مَالًا مَالًا عَلَى مَالًا مَالًا عَلَى مَالُولُ عَلَى مَالَعَالَى مَالَى مَالَا عَلَى مَالًا مَالًا عَلَى مَالًا مَالَا مَالًا عَلَى مَالًا مَالًا عَلَى مَالًا مَالًا مَالًا عَلَى مَالًا مَا عَلَى مَالًا عَلَ مَاللَا مَالِحَالًا عَلَى مَالَا مَالًا مَالًا مَالًا مَالًا عَلَى مَالًا عَلَى مَالًا عَلَى مَالًا عَلَى مَالًا مَالًا مَالًا عَلَى مَالًا مَالًا عَلَى مَالًا عَلَى مَالًا عَلَى مَا مَالًا عَلَى مَالًا عَلَى مَالًا عَلَى مَالًا عَلَ

When the event in the main clause has already taken place, the posterior clause is negated, see (15.34-36). The negated verb may occur non-nominalized (15.34) or nominalized (15.35-36).

- (15.34) خَيْطَ (Paul) שוּקָשָ אישָראי צָשָׁשִי הָה' שָּאַ הָרָ' שָׁרָ־שָּׁדן [pol nén ma-kjap nénle] אולי מאשי הָרָ' שָׁרָ־שָּׁדן PN marriage NEG-do before my mother=DEMPH die.HON-CMPL '[Before Paul married], my mother died.' (DB life story)
- (15.35) ເດີ ເພາເນາກົາ ເພດ ຫຼັດເລັ້າ ເຊິ່ງ ເລັ້າ ເລີ້າ ເລີ້າ ເລັ້າ ເລີ້າ ເລີ້າເລີ້າ ເລີ້າ ເລີ້າ ເລີ້າ ເລີ້າ ເລີ
- (15.36) ผสสาลัยาสาร์สานนิ มีสานสา ผสาร์

 $[d \in ndzo:=lo$ man-dzøm-bø: $p \in nl \in =di$ Sikkim=DATNEG-come.HON-2INF.GENbefore=DEMPH'[before coming to Sikkim]...' (KLT Bumchu story)

As an alternative to -po-infinitive, the complement of the postposition may be a genetivized - $\epsilon\epsilon$ infinitive:

(15.37) מקימידי מביקרייון קמימימי

 $[dik^ha=ra]$ δ̃:-εε=ki henle] here=AEMPH come-INF=GEN before '[Before coming here]...' (PTA kitchen discussion)

15.3.3 Simultaneous constructions

Denjongke has several constructions which imply at least partial temporal overlap between the verbs in the subordinate clause and the main clause (these are comparable to English whenclauses). The various constructions are listed in Table 15.5 and described in the following subsections.

Construction	Origin
VERB-po-dã:	nominalizer <i>-po/bo</i> and temporal extension of coordinator $t'\tilde{a}$: 'and'
VERB-sondã:	$s \dot{o}$:-bo [go.PFV-2INF] + t' \ddot{a} : 'and' (?)
VERB-somdã:	$s \delta$:-bo [go.PFV-2INF] + t' \tilde{a} : 'and' (?)
VERB-sumdã:	$s\tilde{o}$:-bo [go.PFV-2INF] + t'\tilde{a}: 'and' (?)
VERB-tsubdã: (CY)	$ts^{h}o$ - u [CMPL-2INF] + $t'\tilde{a}$: 'and' (?)
VERB-pø: gã:	WT (ק) کار sgang 'time'
VERB-kap	WT Skabs 'time'
VERB-dỹ:	WT 5 ^{sy} dus 'time, season'
VERB- <i>rɛŋ-kʰa</i>	WT πa 'be time to' + spatial suffix $k^h a$
VERB lõː mèː-po	secondary verb WT Tor 'have time to' + NEG.EX-2INF
VERB-RDP p'a:pu	WD ਹਨ ਦੱ bar-po 'in between, while (negated 'while not')' (from WT
NEG-VERB <i>p'a:pu</i>	קج: bar 'between, middle')

Table 15.5. Simultaneous dependent verbal constructions

The great number of options for expressing simultaneity in Table 15.5 naturally raises the question, what, if any, the functional differences are between the different constructions. The following subsections present only the beginning of the enquiry into the simultaneous constructions. More understanding remains to be gained through further study.

15.3.3.1 Simultaneous -po-dã⁴²²

Similar to other simultaneous constructions, the use of VERB-*po-dã*: (may be reduced to VERB-*m* $d\tilde{a}$:) implies at least partial temporal overlap between two actions/states:

⁴²² Sandberg (1895: 49) reports the similar form -pa tang, as in Ngá minda kyap-pa tang, mi sum hlum song zhe 'On my firing the gun, three men fell.'

(15.38) ราลัร ราวรา ตัลเญา พาซี ซิตา ลัญา ลักา ลิ่าสา รู้ อัลาวีารา ลรายรา ครายรา ครามการเกาสา พัวรา กริสา พัวรา t'ato [rawã: t^hom=gi átsi=tci? òtε tc'em-bo-dã:] mòu barfun now TPN town=GEN a.bit=INDF down down go.hon-2inf -conj tpn iờː-kʰɛn $b\epsilon^2$. lìndam làp-k^hế: óna k'jo:=tci? TPN say-NMLZ there village=INDF EX-NMLZ EQU.NE 'Now when (one) comes a bit down of the town of Rabang, there is a village there called Barphung Lingdam.' (SGD wedding customs)

Usually constructions with $-d\tilde{a}$: signify a switch in the actor referent, as seen in (15.39), where each instance of $-po-d\tilde{a}$: is followed by a switched actor (clock-Indians-bear-people). The two instances of -ti/di, on the other hand, signify actor continuity.

tc^huts^hø? teu:ni: [ep-o-dã:] [*t*^hik lòk gjate^hu=tsu ke:po te:ta? exactly clock.time twelve arrive-2INF-CONJ many stick again Indian=PL [*lòk* t'om=di=lo tip-tip-o-dã: t'om ba? òn-**di** t^henki hit-RDP-2INF-CONJ bear carry come-NF again bear=DEMPH=DAT suddenly mòu=le dzu:-ti $[k\epsilon^2]$ có:-bo-dã:] тí t^hamtce? p'jo: down=ABL be.afraid-NF sound call-2INF-CONJ human all escape jà:-bo. go-2INF

'When it was exactly 12 o'clock, the Indians came back bringing sticks and started hitting (him) suddenly. Then when the bear was afraid and let out a cry from under (the sack-cloth), all the people escaped.' (KT Animal story)

Example (15.40) is the only instance of VERB-*po*- $d\tilde{a}$: in the novel Richhi. The clauses are functional passives, so actor switching or retaining is not applicable in this sentence.

(15.40) スティディ خاصم حجر خاصم حجر خاصم حقاق المعالي المحالي المحالي (15.40) حجر خاصم حجر خاصم حجر خاصم حجر خاصم حجر خاص حجر المحالي المحالي المحالي المحالي المحالي المحالي المحالي المحالي المحالي (15.40)

The construction is negated by the prefix *ma*-. The form $m - \dot{\tilde{o}}: -b - d\tilde{a}:$ in (15.41) is an abbreviation of *ma-om-bo-d\tilde{a}:* [NEG-come-2INF-CONJ].

 $[k^{h}u \quad m-\ddot{o}:-b-d\tilde{a}:]$ $\eta\dot{a} \quad j\dot{o}:=di \qquad p'ja \quad mj\ddot{o}:-ts^{h}a:.$ 3SGM NEG-come-2INF-CONJ 1SG work=DEMPH do finish-CMPL 'I finished the work [before he came]. ' (KN e) (15.42) ผูโฉ้ญาทาทิญ มาสู์ฤญาส์ สุดาญารี

 $[k^{h}\varepsilon:ga=gi$ ma-ts^hu:-ts^hu-p-da?] strength=AGT NEG-be.able.to-RDP-2INF-CONJ '[when I wasn't able to do (those things) with (my) strength]...' (CY interview)

Note that the negated simultaneous construction in (15.41) becomes, in effect, a temporal 'before'-clause. Negated simultaneous constructions are used in many languages, which do not have a morpheme meaning 'before', to express 'before'-clauses (Thompson et al 2007: 248).

In addition to the grammaticalized temporal use, -po $t'\tilde{a}$: also occurs in the non-temporal conjunctive meaning:

(15.43) ข้าง ภูกงาร์ 55 คร. 65 ยา 3

k'jo? kjap-**o** t'ã: àrùr=gi da and hurly-burly=GEN sound chatting do-2INF 'the sound of chatting and hurly-burly' (mam-tog 24)

15.3.3.2 Simultaneous converb markers -sondã:/somdã:/sumdã:/tsubdã:

The use of the converbal markers -sondã:/somdã:/sumdã:/tsubdã:, which I have only come accross in spoken language, suggests that the converbal action is at least partly simultaneous with the action marked by the verb in the following clause. For hypotheses of the origin of these forms, which do not occur as such in written Denjongke, see §3.3.6.18. The four forms are illustrated in (15.44-47). The construction is negated by the prefix ma(n)-, see (15.45).

jà:-sondãː] te k^hu 3SGM so '[When (his) friend goes away weeping], he...' (RS pupil joke)

This simultaneous construction may, depending on the context, also express reason:

(15.45) हुन अव ह्वेग्या रॉन्स न न न के ही गी खुया रें हुन [lúŋ man-dik-somdã:] t'a ŋà $l\hat{u}$ -b=b ϵ ?. nàː-tiki air NEG-be.fitting-SIM now 1SG be.ill-NF remain-2INF=EQU.NE '[Since the air was not fitting (for me)], I was left ailing.' (PED life story)

Sometimes both a simultaneous and causative interpretations are possible:

(15.46) विमा मर र्येनमानन कें परने में मामन हा [*pìm cá:-sumdã:*] óde: εí-kʰεn bɛ?. like.that die-NMLZ EQU.NE sun shine-SIM '[When/because the sun shines], (the seeds) die like that.' (KT discussion with TB)

In (15.47), the simultaneous construction is followed by an explicitly causal construction δdi *p'ja-ti* 'because of that' (lit. 'that doing').

(15.47) รา สังกรุราชิ เกราวสิขาลา จารา สูงาลารังรารา รา สังกรา อุสาชา ราชกา ซิลา สาวาริกา สิงาริกา สาวารา สาวาร ťa *ódɛpti* [k^hjapdaː lèp $t\tilde{a}$:-tsubd \tilde{a} :]⁴²³te ódi p'ja-ti nàtei send-SIM so that do-NF now like.that spreading much 1PL.GEN cý:tcy? ódepti pàmtɕʰa? số:-bo ĩ. ke? language a.bit like.that decline go.IPFV-2INF EQU.PER 'Now, [when (the Nepali language) was much promoted], for that reason our language went a bit into a decline.' (CY interview)

The fact that *-tsubda:* is in (15.47) followed by a causal construction suggest that although a causative interpretation is at times possible, the basic meaning of *-tsubdã:* and the related forms *-sondã:/somdã:/somdã:* is simultaneous.

15.3.3.3 Simultaneity with gã: 'time'

The next four simultaneous constructions use an explicit word referring to time. The word $g\tilde{a}:(=lo):^{424}$ (at the) time' is postposed to genitivized *-po*-infinitive to mark simultaneity, a typical noun complement construction (see §13.4). Phonetically $g\tilde{a}$: is frequently reduced to $[\eta\tilde{a}:]$.

(15.48) الله: قَحْرَ ثَلَّهُ: عَمْرَ مَحْمَّ عَمْرَ مَحْمَ عَمْرَ مَعْرَ مَعْنَ عَمْرَ مَعْنَ عَمْرَ مَعْنَ عَلَى اللَّهُ عَمْرَ عَلَى اللَّهُ عَمْرَ اللَّهُ عَمْرَةُ اللَّهُ عَمْرَ اللَّهُ عَمْرَةُ اللَّهُ عَمْرَةُ اللَّهُ عَمْرَ اللَّهُ عَمْرَ اللَّهُ عَمْرَةُ اللَّهُ عَمْرَةُ عَمْرَةُ اللَّهُ عَمْرَ اللَّهُ عَمْرَةُ عَمْرَةُ اللَّهُ عَمْرَةُ عَمْرَةُ عَمْرَ اللَّهُ عَمْرَى اللَّهُ عَمْرَ اللَّهُ عَمْرَى الْحُكْمَالِ عَمْرَةُ عَمْرَى الْحُمْنَ عَمْرَةُ الْحُلْقُلْ عَمْرَى الْحُمْنَ الْحُمْنَ عَمْ الْحُمْعَ عَمْنَ الْحُمْنَا عَمْنَ اللَّعْمَا عَمْ الْحُمْعَ الْحُمْعَالَ الْحُمْعَالِ اللَّهُ عَمْنَ الْحُمْعَالِي الْحُمْعَالِي عَمْ الْحُمْ عَمْنَ الْحُمْ عَمْ عَلَيْ الْحُمْعَ الْحُمْعَالَ الْحُمْعَالُ اللَّا عَمْدَةُ الْحُمْعَالَ اللَّالَةُ عَمْعَالَ اللَّهُ عَمْ عَمْ الْحُمْعَالِ اللَّالِحُمْعَالِ اللَّالِ الْ اللَّالْحُمْعَالَ اللَّالِ عَمْدَ عَمْعَالِ اللَّالِحُمْعَالِ اللَّالِ عَمْ الْحُالِي الْحُمْعَالِ الْحُمْعَ عَ اللَّالْحُمْعَالَ الْحُمْعَالَ عَمْدَ عَمْعَالَ عَمْعَالَ اللَّالِ عَلَيْ عَلَيْ عَالَيْ عَالَيْ عَالَيْ الْحُمْ عَلَيْ الْحُلْعَالَ الْحُلْعَالَ الْحُلْعَالِي عَلَيْعَالِ عَلَيْ الْحُالِي عَلَيْ الْحُالِ الْحُلْعَالِي الْحُالِ الْحُلْعَالُ الْحُالْحُالِ الْحُمْعَا الْحُمْعَا الْحُمَالِي الْحُلْحُالِي عَلَيْ الْحُمْعَالِي عَلَيْ عَالَيْ عَالَيْ عَالَ الْحُلْعَالَ عَلَيْ الْحُلْعَالُ عَلْ الْحُمَالِي الْحُلْحُلُكُلُ الْحُلْحُلْحُالِ اللْحُلْحُلُكُ عَلَيْ عَالَيْعَالِ عَالَا اللْحُعْلُ الْحُلْعَا عَلَيْ ال

In (15.48), the construction with VERB-pø: $g\tilde{a}$: and the following clause have the same actor. Example (15.49), on the other hand, shows that the construction can also occur when the actor switches.

ső:-bø: daku=di [ónale ódi t^horã:=tci? $g\tilde{a}$] ó then that tomorrow=INDF owner=DEMPH go.PFV-2INF.GEN time that lấ: tsuko=di p'i t^hamtce? sà-ti lèp dã:-ti dø: bull other=DEMPH fodder all eat-NF very.much be.satisfied-NF sit du?EX.SEN 'Then, the following day, at a time when the owner had gone away, the other bull ate all the fodder and appeared very satisfied.' (TB bull story)

The expression may be supplemented by the dative-locative (15.50) or ablative marker (15.51):

⁴²³ The form $t\tilde{a}$:- $tsubd\tilde{a}$: could either be a reduced version of $t\tilde{a}$:- ts^har -bo t' \tilde{a} : (cf. VERB- ts^har - $bo = l\varepsilon > VERB$ - $ts^hub = l\varepsilon$) or have the abilitative secondary verb ts^hu^2 'be able to', $t\tilde{a}$: ts^hu -po- $d\tilde{a}$: > $t\tilde{a}$: ts^hub - $d\tilde{a}$. In the latter case, example (15.47) would be an instance of simultaneous construction -po- $d\tilde{a}$:, see §15.3.3.1.

⁴²⁴ This word may also be directly postposed to demonstratives, nouns and adjectives, e.g. *ódi* $g\tilde{a}:(=lo)$ 'that time', *p'jaby:* $g\tilde{a}$: 'when being child', $te^hu\eta te^hu\eta g\tilde{a}$: 'when being small'.

(15.50) ซา้า อัสารณ์ สู่กาล์ เล่าเกิน เล่าสาย เล่าจาย

[$p^{h}a$: tc'øm-bø: $g\tilde{a}$:=lo] gakdza kjap-tc ε =la. over.there come.HON-2INF.GEN time=DAT obstruction do-PST=HON '[When (he) came over there], (they) obstructed (him).' (CY interview)

(15.51) ଦିଁକ୍ର ୩ନିଷାର୍ଯ୍ୟା କା ପର୍ଯ୍ୟାର୍ଥ୍ଯ ଅନ୍ଥିକା ଅନ୍ଥିକା ଅନୁକାରଣ ଅନ୍ତର୍ଥ କା ଅନୁକାରଣ କା ସ୍ଥାନ୍ତ୍ର ଅନ୍ତର୍ଯ୍ୟ ଅନ୍ [khõ: ní:-po tc'a t^huŋ-di ma-mjõ: $g\tilde{a} = l\varepsilon$ kantc^hi 3PL two-COL tea drink-NF NEG-finish time=ABL youngest.daughter(Nep.) t'ã: ladzi=ki $k^h \tilde{o}$: pi:-po=locè:la? bak-ti lep. and PN=AGT 3PL two-COL=DAT food.HON carry-NF arrive '[When the two of them haven't (yet) finished drinking tea], Kanchi and Lhadze arrive, bringing food to the two of them.' (Richhi 19)

Example (15.51) above illustrates a negated form, which occurs without nominalization and genitivization. The nominalizer may also be dropped if the verb root is reduplicated:

(15.52) ぎきょうや あち あち なち

[dordziliŋ $s\ddot{o}:-s\bar{o}:$ $g\ddot{a}:=di$]TPNgo.PFV-RDPtime=DEMPH'[When I was going to Darjeeling]...' (UTR plains story)

The reduplication in (15.52) probably emphasizes that the verbal action in the following clause happened on the way to a location ("when I was going") and not after the speaker had reached the location ("when I went").

15.3.3.4 Simultaneity with kap 'time'

The second simultaneity marker deriving from a WT word for 'time' is kap (WT MVW skabs), which can be postposed to a genitivized -po-infinitive (15.53-54) and progressive forms (15.55) of the verb, or to their combination (15.56). The word kap may head the time adverbial by itself or be followed by case or postpositional marking. Unlike $g\tilde{a}$:, which collocates with the dative-locative =lo, kap frequently collocates with the postposition nàŋea 'inside' (kap=na and kap=lo also occur). One consultant, see (15.54), used the Central Tibetan equivalent kapsu 'when, while' accompanied by the dative-locative. The actor of the clause following a construction with kap may be either the same as or different from the clause with kap.

(15.53) อัสารณิ มูรง สุร.ศ.ศ.

[*tc'øm-bø: kap nàŋca=lo*] come.HON-2INF.GEN time inside=DAT '[When coming]...' (KT phone call)
- (15.54) די גוימדי לקידופקיים איינטיים איינטי ביינט ברי די מאדים ארגיים ארגיים ארגיים אייני ביסרי איר איינטי ביי केव रेंग्लेंग नवासुवा कुवासुगीका हेव रहोवा क्रेंगर्दी वट मुर्वे गत्व खु गवटका छत्त te ódi t'ønzingi [p'ey:=gi ge:pu=di t'a sé:t^hinásol $p^{h}ou$ so that according Nepal=GEN king=DEMPH now enthronement over.there kapsu=lo] nàtci míwã: $t \epsilon^h \phi g \epsilon l \quad t \epsilon^h impu = lo$ zu:-pø: p'ey: sit.HON-2INF.GEN time=DAT 1PL.GEN ruler king great.one=DAT Nepal $g \varepsilon p u = g i t \varepsilon n d \tilde{\varepsilon}$ ódi nànca=lo dencu nấː-m *bε*?. king=AGT ceremony that inside=DAT invitation give.HON-2INF EOU.NE 'So in accordance to that, [now when the King of Nepal was having coronation over there], our ruler, the great king, was invited in that ceremony by the king of Nepal.' (CY interview)

 $[k^h u]$ pankha thøn-ce? nó:sam tã:-zin kap nànca=lo] k'ate? te 3SGM how outside come.out-INF thought send-PROG time inside=DAT then *t^høm-bø*: tc'y? p'ja-zin ta? òm-bo hako-tiki khu 3SGM plot do-PROG tiger come-2INF know-NF come.out-2INF.GEN j*à-po* be?. EX-2INF EQU.NE 'When he was thinking how to get out, he, having found out that a tiger had come, was scheming to get out.' (KT animal story)

tea.HON snacks eat-PROG-NMNZ.GEN time '[as (they) are/were enjoying tea and snacks]...' (Richhi 12)

In written language, however, *kap* may be directly postposed to the verb root (in Richhi *kap* occurs both with non-nominalized and nominalized verbs). The form directly attached to the verb root is here considered a converbal ending and glossed as simultaneous (SIM).

- (15.57) هَرَ اللَّهُ عَلَى اللَّهُ (اللَّهُ اللَّهُ اللَّالَ اللَّهُ اللَّهُ عَلَيْ اللَّهُ عَلَيْ اللَّهُ عَلَيْ اللَّالِ اللَّالَ اللَّهُ عَلَيْ اللَّالِي اللَّهُ عَلَيْ اللَّالَ اللَّهُ عَلَيْ اللَّالَ اللَّهُ عَلَيْ اللَّالَ اللَّالَ اللَّالَ اللَّهُ عَلَيْ اللَّالَ اللَّالَ اللَّالَ اللَّالِ اللَّالَ عَلَيْلُولِ الللَّالَ اللَّ
- (15.58) المجمع محرية مع محرية المحلي المعالية المحلي المعالية المحلية (15.58) المجلي المحلية المحل محلية المحلية ال

The construction is negated by ma-.

15.3.3.5 Simultaneous converb marker -dy:

The simultaneous converb marker -dy: (derives from $\sqrt[5]{N}$ t'y: 'time, period') is attached directly to the verb root. This use is less frequent in my data than the other simultaneous constructions.

(15.60) $a_{i} = k_{i} + k_{i$

(15.61) מַאָן קריע די מָריקאי

[go t'anpo nà: **òn-dy**:] start first here some-SIM '[When (I) at first came here]...' (KT life story)

In the novel Richhi, $-d\tilde{y}$: occurs only once, in a poem:

(15.62) สิ.เมสา ฮัสารสา

[*mi-ly: t^hop-dy:*] human-body receive-SIM 'When receiving a human body...' (Richhi 74)

The converb -dy: occurs once in my data in a type of double simultaneity marking construction with *kapso*, which also occurs in (15.54) above as the variant *kapsu*.

(15.63) येंग प्रहुन्धः श्चेपयानुया भ्रपयायां मा यत्रयार्थ्वेम्या प्रमनाया वमाम केंया र्यया र्थनाया केंवा teø:na lep-dy: kapso] nà dendzõ: cé:da [lò nànca 1sg Sikkim Buddhist.institute inside year fifteen reach-SIM time $tc^{h}o?$ p'ja dø:-po ĩ. doctrine do sit-2INF EQU.PER '[When I reached fifteen years (of age)], I stayed at Gangtok Shedra studying.' (RB life story)

Negation is accomplished by the prefix *ma*- and genetivization is optional.

(15.64) مرجز جن بھرت بھرت جن میں من تقرین (درامَن) جرمن جن مکمن بھی تقرین (درامَن) جرمن درامَن (درامَن) جرمن درامَن (درامَن) جرمن درامَن محمد بھی تقرین (درامَن) جرمن درامَن محمد بھی تقرین (درامَن) جرمن درامَن محمد بھی تقرین (درامَن) جرمن درامَن درامَال درامَن درامَان درامَن درامَان درامان دراما

15.3.3.6 Simultaneous converb marker -reykha

The fourth construction based on a WT word for 'time' (WT $\pi \pi$ ' *ran* '[it is] time') is the converb formed with the suffix *-renkha*, which combines the secondary verb *ren* 'be time to' with the locational/temporal derivative suffix *-kha* (which some speakers reduce to *-ka*). For uses of *ren* as a secondary verb, see §8.5.5.

- (15.66) اَجَ مَنْسَاتِمَ اللَّهُ عَلَى اللَّا عَلَى اللَّهُ عَلَى الْعَلَى الْحَلَى اللَّهُ عَلَى اللَّا عَلَى اللَّهُ عَلَى اللَّهُ عَلَى اللَّهُ عَلَى اللَّهُ عَلَى اللَّ اللَّهُ اللَّهُ اللَّا اللَّهُ اللَّا عَلَى اللَّا عَلَى اللَّا عَلَى اللَّالَةُ عَلَى اللَّا عَلَى اللَّ المَا عَلَيْ اللَّا عَلَى اللَّالَةُ اللَّا عَلَى اللَّا عَلَى اللَّالِ اللَّالِي اللَّالِي اللَّالِي اللَّالِ عَلَى اللَّالِي اللَّالِي الْحَالِي الْحَالِي اللَّالِي الْحَالَ اللَّا عَلَى الْحَلُ اللَّا عَلَى اللَّا عَلَى اللَّا عَلَى اللَّالَةُ عَلَى اللَّا عَلَى اللَّالَةُ عَلَى اللَّالَةُ عَلَى اللَّالِ اللَّا عَلَى اللَّالَةُ عَلَى اللَّالَةُ عَلَى اللَّالِي عَلَى اللَّالَةُ عَلَى اللَّالِي اللَّالَةُ عَلَى الْحَلَى اللَّالِ اللَّالَةُ عَلَى اللَّالَ اللَّالَةُ عَلَى اللَّالَةُ اللَّال المَالَةُ عَلَى اللَّالَةُ عَلَى اللَّالَةُ عَلَى اللَّا عَلَى اللَّالَةُ عَلَى اللَّالَةُ عَلَى الْحَالَةُ عَلَى المَالَةُ عَلَى اللَّالَةُ اللَّا عَلَى اللَّهُ عَلَى اللَّاللَالَةُ عَلَى اللَّهُ عَلَى اللَّهُ عَلَى اللَّالِ عَلَى

The construction VERB- $re\eta k^h a$ is negated by the negator prefix ma-, ma-VERB- $re\eta k^h a$.

nà: nawa⁴²⁵ nànca $[k^h u]$ k^hu-ri-gi cínto? tok-teã: 3SGM=AEMPH.GEN=GEN here thought inside 3sgm fruit pick-PROG p'ja-renkha] [òte ma-tho:-renkha] down do-SIM NEG-see-SIM '[When he was here in his thoughts picking fruit] (and) [when he did not see down]...' (RB pear story)

In the novel Richhi, the construction VERB- $r\epsilon\eta k^h a$ occurs only once, see (15.68). In addition, $\pi \overline{q}$. ran [ren] is once in Richhi used without the suffix $-k^h a$, see (15.69).

⁴²⁵ According to consultant KN, this word is code-switching from Tibetan.

(15.68) שאי איצר אי אין איז אי איצר אי איצראי איצראי

 $tc^{h}\varepsilon$: [$\dot{a}dzo$ $\epsilon i - r \epsilon \eta k^{h}a$] t'a $k'am \omega$: $m \dot{a}ni$ $g j \tilde{a}$: $-\epsilon \epsilon^{2}$? EXCLAM grandfather die-SIM now what.GEN prayers extend-INF 'Oh heck, [when the grandpa (=you) is dying], what's (the use of) saying prayers?' (Richhi 79)

(15.69) 5' P'TA')

t'a ci-ren=to. now die-be.time=IMPFV 'Now it's (my) time to die.' (Richhi 79)

15.3.3.7 Simultaneity with p'a:pu 'in between'

The formative ${}^{426}p'a:pu$ is used in two constructions to express about the same idea in an affirmed ('while, as long as') and a negated way ('while not, as long as not'). In the affirmed construction, the verb root is reduplicated:

(15.70) قِحَ قَحَ مَحَ بَخَرَ مَحَ مَعَ عَلَيْ عَمَرَ مَعَ عَلَيْ عَجَ مَعَ عَلَيْ عَمَرَ مَعَ عَلَيْ عَلَيْ مَعْ مَعْ عَلَيْ عَلَيْ عَلَيْ اللَّهُ عَلَيْ عَ الْعَلَيْ عَلَيْ الْعَلَيْ عَلَيْ الْعَلَيْ عَلَيْ عَلَيْ

In the negated construction, the non-reduplicated verb root is preceded by the negator prefix ma. In its original context, example (15.71) directly follows (15.70), adding a negated perspective ('while we are not dead') to the affirmed perspective ('while we are alive').

[ma-cí p'a:pu] t^hamtce? t^hyn-di tci:=ki zen=lo ro:ramda: p'ja-ti NEG-die between agree-NF do-NF all one=AGT other=DAT help p^hembø: jó:=tsu p'ja-ge. work=pl do-hort benefit.GEN '[While not dead,] let us all agree, help each other and do beneficial works.' (mthunsgril 5)

15.4 Causal clauses

This section describes eight reason-marking constructions. The terminative $=s\tilde{a}$: 'until', which may also express reason, is not discussed here but in §15.12. The constructions described here are the following:

1) finite clause marked by the connector k'amjasene 'because' (§15.4.1)

2) clause employing the noun gjumtshe? 'reason, cause' (WD gratar rgyu-mtshan) (§15.4.2)

⁴²⁶ I am hesitant to call *p'a:pu* a postposition because it does not occur with nouns in my data.

3) construction based on the word ken 'cause, condition' (WD 演示 rkyen) (§15.4.3)

4) converbal form *làp-ti* [say-NF] 'saying, having said' (§15.4.4)

5) ablative construction with a copula (which with other verbs marks anteriority) (§15.4.5)

6) agentive marking (§15.4.6)

7) the converbal form *p'ja-ti(ki)* [do-NF] 'doing, having done' (§15.4.7)

8) circumstantial-purposive converb which is used causally (§15.4.8)

15.4.1 Causality with connector k'amjasene 'because'

The connector k'amjasene 'because' has a clausal origin as k'amja sé-ne [why say-COND] 'if said why'.⁴²⁷ Three facts suggest that the speakers are still aware of the clausal origin of k'amjasene. First, the speakers may change the verb of saying used in the connector (k'amjasene/k'amjalapne/k'amjacune, using sé/si 'say', làp 'say' and eu 'say.HUM' respectively), Second, Denjongke authors may write the form jointly as k'amjasene or separately as k'amja sene. Third, k'amjasine may be interrupted by other elements:

(15.72) สู้ ฯลายิ गุสาปูล ฯลาร์ ลาสิา

 $\begin{array}{cccccccc} l \acute{o} & p^h am-di & k`amja & p^h am-bo & si-n\varepsilon \\ \mbox{mind be.defeated -NF why be.defeated-2INF say-COND} \\ `(Our) \mbox{mind being downcast, if (we) talk about why (our mind) is downcast...' (RS language situation)} \end{array}$

Typically k'amjasene is used as a word-like connector/conjunction meaning 'because', see (15.73-75). Causal clauses with k'amjasene are finite.

 $k^h u = lo = di$ tε *lópøn=di=gi átsi=tei?* ka:gjur tã:-bo then teacher=DEMPH=AGT 3SGM=DAT=DEMPH a.bit=indf instruction send-2INF *ódi gjammø=di=lo* [k'amjasine ný: t^hu: $b\epsilon = lo$. EQU.NE=REP that latter=DEMPH=DAT because money pick.2INF $b\varepsilon = \varepsilon o.$] EQU.NE=AT 'Then the teacher instructed him a bit, the latter one, [because (he) had taken the money, you know].' (RS pupil joke)

(15.74) รู ซิสา เต ซิสาลิ เมลาย์สาลิ เมลาย์สาลิ เกิราย์สาลา เกิราย์สาราย์ เมลา เกิราย์ เ

t'utei? $\dot{a}tsi\ l\dot{e}m\ t^{h} on-z\epsilon$ [k'amjasene $k^{h} on=gi$ $n\dot{a}=lo$ this.year a.bit good become-PST because 3SG.HON=AGT 1SG=DAT $s\dot{o}:ra\ n\ddot{a}:-bo$ in=gi=la.] gift give.HON-2INF EQU.PER=NC=HON 'This year was a bit good, [because he gave me a gift].' (KT discussion with TB)

⁴²⁷ Nepali has analogous *kinabhane* 'because', which consists of *kina* 'why' and *bhan-e* 'say-PFV'.

(15.75) कें ट्रंडेया र्वेअप्टेंग्यं क्षेत्रका केंर्स्र्यकार्यों यात्र र्डाकाकें त्र आप्तें प्रयाद र्वेत्र केंस्रुम्रायं दशु दर्याका mù t'utci? *bombai=lo lɛp* mi- $ts^h u$:=lo. [k'amjasene ápø: NEG-be.able.to=REP because 3SGF this.year TPN=DAT arrive father.GEN t'ønzin màisur=lo gju go?]. ka be.needed order according TPN=DAT go 'She cannot come to Bombay this year, we hear, [because (she) has to, according to the father's word, go to Mysore].' (Richhi 150)

Longer form of the conditional (see §15.6) may also be used:

(15.76) בישרי צילי מאד מידיליאו מדילא איליפיא איליפיאי איליפיאי איליפיאי איליפיאי איליאי מרי אינאי אילי אינא איליפיאי ĩ nà=jã: p^hi :-po làp-o=lo. [k'amja sí-betsene k^hu 1sg=too be.late-2INF EQU.PER say-2INF = REP why say-cond 3sgm ódi tiru? ná-po=di [um-run] paisa=di money(Nep.)=DEMPH that rupee five-COL=DEMPH fall-CONC 'I too am late, (he) says, [because although his money, those five rupees fell]...' (RS pupil joke)

In addition to occurring independently as a causal marker, k'amjasene may co-occur with the causal construction VERB-INF=DEMPH.AGT, which also occurs as the sole reason marker (see §15.4.6):

(15.77) कॅका अहेका में कं पवित्र येंन गत र्झका के तर हे का में का स्वर्ग के अनुका हैया न या में के जिन मान हैं जिन मन के जिन के कि का मान के कि का मान के जिन के कि का मान के जिन के कि का मान के कि क $t \epsilon^h \phi dz i \eta \delta t s^h a - z \tilde{\epsilon}$ jờ? k'amjasene nò ma-seː-pøː PN be.ashamed-PROG EX.PER because face NEG-know-2INF.GEN pámtei? nagacel(?) tei? kap-ti nè:-ce=di:. тí human with blanket one cover-NF sleep-INF=DEMPH.AGT 'Chodzi is ashamed, because she has slept covering (herself) in the same blanket with an unknown person.' (Richhi 131)

15.4.2 Causality with gjumtshe?: 'reason'

Causality/reason may also be expressed by a simple juxtaposition of a complement clause headed by the word *gjumts*^h $\tilde{\varepsilon}$: 'reason' and a clause that explains what the reason is. Both examples (15.78) and (15.79) have the demonstrative-emphatic =*di* following *gjumts*^h $\tilde{\varepsilon}$:, suggesting that the emphatic may be obligatory or preferred in this construction.

(15.79) هَمْ تِرَيْمُ اللَّهُ عَنْ الللَّهُ عَنْ اللَّهُ عَنْ الْحَالَةُ عَنْ اللَّهُ عَنْ اللَّالَ اللَّالَ اللَّالَ اللَّعْنَالَ اللَّالَ اللَّهُ عَنْ الْحَالَةُ عَنْ الْحَالَةُ عَنْ الْحَالِ اللَّالَ اللَّالَ اللَّالَةُ عَنْ الْحَالَةُ عَنْ الْحَالَةُ عَنْ الْحَالَةُ عَنْ الْحَالَةُ عَنْ الْحَالَةُ عَنْ الْ اللَّا عَنْ الْحَالَةُ عَنْ اللَّالَةُ عَنْ اللَّالَةُ عَنْ الْحَالَةُ عَنْ الْحَالَةُ عَنْ الْحَالِ اللَ اللَّالْ اللَّالَةُ عَنْ اللَّالَةُ عَنْ الْحَالَةُ عَنْ الْحَالَةُ عَنْ الْحَالَةُ عَنْ اللَّالِي اللَّالَةُ عَا اللَّا عَالَةُ عَنْ الْحَالَةُ عَنْ الْحَالَةُ عَالَيْ عَا الْحَالَةُ عَالَيْكَ عَلَيْ الْحَالِ الْ الْحَالَةُ عَنْ اللَّالَةُ عَنْ اللَّا عَالَيْ عَالَةُ عَنْ اللَّالَةُ عَالَيْ عَالَيْ عَالَيْ عَالَيْ عَالَيْ اللْحَالِ اللَّالَةُ عَالَى الْحَالَةُ عَالَى الْحَالَةُ عَالَيْ عَالَيْ عَالَيْعَالَيْ عَالَيْ عَالَ الْحَالَةُ عَالَيْ الْحَالَةُ عَالَيْ عَالَى اللَّ عَالَيْ الْحَالْ الْحَالَ اللَّالَةُ عَالَيْ عَالَيْ عَالَيْ عَالَيْ عَالَيْ عَالَيْعَالَيْ عَالَيْ عَالَيْ عَالَيْكَ عَالَ الْحَالَ الْحَالَةُ عَالَى الْحَالَةُ عَالَيْ عَالَيْ عَالَ اللَّعَالَ عَالَ عَالَيَ لَا عَالَ لَكَالِ عَالَ عَالَ عَالَيْ عَا

The second clause, which defines the reason, may also be introduced with k'amjasene 'because' (note that =di is not obligatory here).

(15.80) มิี่สารเลิ่า สูมามส์สา ๆสายีา์ห้าสา ยามนิ าๆดิสาส์สาร์า้น้ำ ยู่ๆามคนา มิๆา สที่หารนิว สัสารๆ ๆดิหา พัศาร์า มีกา gjumts^hɛ̃ː] [k'amjasinɛ p'umø: [*mè*:**-***pø*: $n\acute{\epsilon}nts^h\widetilde{\epsilon}$:=tsu=loNEG.EX-2INF.GEN reason because girl.GEN relative=PL=DAT t'ønda? pí: tc^handze: dik go:-pø: j*à-po* $b\epsilon$?]. arrange be.needed-2INF.GEN purpose two EX-2INF EQU.NE present.HON '[The reason why it is not there (is)] [because there are two purposes for having to arrange a present to the girls' relatives].' (sbar-phung 15)

Spoken language often resorts to longer expressions than mere $k'amjasin\varepsilon$ to accompany $gjumts^{h}\tilde{\varepsilon}$: 'reason'.

(15.81) ସନ୍ ଦ ଦିଂଶୁଷା ସମ୍ମ ମୁସ୍ସିଷା ସମ୍ମ ମୁସ୍ପିଷା ସନ୍ ସୁଅଟେ ସନ୍ ସମ୍ପ ସୁଷା ଗ୍ରାସିଷା ଗ୍ରାସିଷା ସ୍ଥାନ୍ ସିହାର ସ୍ଥାନ ସ go:-**pø**: gjumts^h *ẽ*ː=di] [di ηà lògju? cé? [k'amja this 1SG story tell be.needed-2INF.GEN reason=DEMPH why cù: ĩ súŋ-bɛtɕɛna nàtci loke=di di=na this=LOC tell.hum.2inf equ.per say.HON-COND 1PL.GEN Lhoke=DEMPH $b\varepsilon = \varepsilon o = la$]. EQU.NE=AT=HON 'If I am to tell the reason why I had to tell this story, why I told it, (it's because) in it is our Lhoke (=Denjongke language), you know.' (PAD bet story)

In (15.81), gjumts^h $\tilde{\epsilon}$: is followed by a second reference to telling the story (ϵu 'say.HUM' restates what was expressed by $\epsilon \epsilon 2$ 'tell' in the previous clause). In addition, k'amjasin ϵ , which has grammaticalized from a conditional clause (meaning 'if (I) tell why') into a connector word, is replaced by a more explicitly conditional clause k'amja...súŋ-bɛtɛɛna.

In yet another construction, $gjumts^{h}\tilde{e}$: 'reason' co-occurs with the dummy verb p'ja 'do' in a simultaneous construction which is interpreted as expressing reason.

(15.82) די אַ אשלא אירד באיארידרי ארי

[*t*ɛ *gjumts^hɛ̃: ódi p'ja-sondã:=di*] so reason that do-SIM=DEMPH '[So for that reason]...' Lit. '[Because of the doing of that reason]...' (PAD Tashiding story)

15.4.3 Causality with ken 'cause, condition'

Another way to express reason/causation through a dependent clause is to append the word *ken* 'cause, condition' (WT $\frac{1}{2}$, *rkyen*) followed by agentive/instrumental or ablative marking to the nominalized and genitivized form of the verb. Agentive marking is exemplified in (15.83) and ablative marking in (15.84).

(15.83) a) ८ गीया है। यार्क्स्यायायती मुनियुंग्रिया ५८ हेवा वर हे ये में में दिरा (recording) वर्ता मां केंगा यायर्क्तयाया क्षेत्र केंग ma-ts^hu-**pø**: ken=gi] hato? [náː=gi ťi sã:te nà: 1SG=AGT write NEG-be.able.to-2INF.GEN cause=AGT suddenly until here reko:diŋ nànca=lo ts^hik ma-ts^hun-po ín-(n)o recording (Eng.) inside=DAT word NEG-fit-2INF EQU-COND 'If there are, suddenly, unfit words on this recording [because I was not able to write (the story down)]...' (KT animal story)

 b) הֹישִה׳ וֹשָׁשִׁ שִׁשִׁי שִׁשִׁי שִׁקִישִי וֹשָׁשִי שִׁהְישִׁל שֵׁק׳ שִׁק׳ שִׁקִי שִׁק׳ שִׁקִי שִׁקּי שִׁקִי שִׁקִי שִׁקּי שִׁקּי שִׁקּי שִׁקִי שִׁקּי שִׁקִי שִׁקּי שִׁקִי שִׁקִי שִׁקִי שִׁקּי שִּיקּי שִׁקּי שִׁקּי שִׁקּי שִׁקּי שִׁקּי שִׁקּי שִּיקּי שִיקּי שִּיקּישִּיקּישִּיקּי שִּיקּי שִּיקּישִּיקּיקּיקּישִּיקּישִּיקּישִּישִּיקּישִּיקּיקּישִיקּישִּיקּישִּיקּישִּיקּיקּישִּיקּישִּיקּיקּישִיקּיקּישִּיקּיקּישִּיקּישִּיקּיקּישָּיקּישִּיקּישִּיקּישִּיקּיקּיישִּיקּישִּיקּישָּישִּיקּישָּיקּיקּייקּ בעוּא שִּישִּישִּיקּישִּיקּישִּיקּי שִּיקּי שִּיעוּ שִּישִּיקּישִּיקּישִי שִּיעוּישִי שִּישִיעוּישִיעָּישִיעוּישִיעוּישִיעוּישִיעוּישִיעוּישִיעוּישִיעוּישִיעוּישִיעוּישִיעוּישִיעוּ שִישִּישִּיעוּיעוּישִיעו שִּיעוּישִיעוּיעו שִּיעוּי

(15.84) दर्ने ते' कर' के' रेगुल' गुरुष येर' गुरुष' येत्र के के रेगुल' ये के रेगुल' ये के रेगुल' के रेगुल' ये के रेगुल' के रेगुल' ये रेगुल' ये ते रेगुल' ये ते रेगुल' ये रेगुल' ये रेगुल' ये रेगुल' ये रेगुल' ये रेगुल' ये ते रेगुल' ये रेगे रेगुल' ये रेगे रेगुल' ये रेगुल' ये रेगुल' ये रेगुल' ये रेगे रेगुल' ये रेगुल' ये रेगे रेगुल' ये रेगुल' ये रेगुल' ये रेगे रेगे रेगे रेगुल' ये रेगुल' ये रेगे रेगे रेगे रेगे रेग

 $\begin{bmatrix} di: & n \partial \eta^{428} & m iri? & k'j \tilde{a}:me? & n \dot{\epsilon}: -p \sigma: & k \epsilon n = l \epsilon \end{bmatrix}$ this.GEN inside people countless reside-2INF.GEN cause=ABL '[Because countless people reside(d) in it (=Sikkim)]...' (da-lto'i 'bras-ljongs 1)

In spoken language, the construction with ken occurred mainly in the speech of consultant KT.

15.4.4 Causality with nonfinal converb làp-ti 'saying'

Reason can also be expressed by the nonfinal converb *làp-ti* 'saying', which refers to the mental process of rationalization by the actor of the clause.

(15.85) قَحْتَ مَامَحَمَّ هُنْ مَانَ أَجْ مَانَ مَعْتَ مُعْتَا مَعْتَ مُعْتَا مَعْتَ مُعْتَا مَعْتَ مُعْتَا مَعْتَ مُعْتَا مَعْتَ مُعْتَا الْعَامَةُ مُعْتَا الْعَامَةُ مَعْتَا الْعَامَةُ مَعْتَا الْعَامَةُ مَعْتَا الْعَامَةُ مَعْتَا الْعَامَةُ مُعْتَا الْعَامَةُ مُعْتَا الْعَامَةُ مَعْتَا الْعَامَةُ مَعْتَا الْعَامَةُ مُعْتَا الْعَامَةُ الْعَامَةُ مُعْتَا الْعَاحَاتِ الْعَامَةُ مُعْتَا الْعَامَةُ مُعْتَا الْعَامَةُ مُعْتَا الْعَامَةُ مُعْتَامَةُ مُعْتَا الْعَامَةُ مُعْتَا الْعَامَةُ مُعْتَا الْعَامَةُ مُعْتَا الْعَامَةُ مَا الْعَامَةُ مُعَامَةً الْعَامَةُ مُعَامًا الْعَامَةُ مُعْتَا الْعَامَةُ مَا الْعَامَةُ مُعَامَةً مَا الْعَامَةُ مَا الْعَامَةُ مَعْتَ الْعَامَةُ مُعَامًا الْعَامَةُ مُعْتَا الْعَامَةُ مُعْتَا الْعَامَةُ مُعْتَا الْعَامَةُ مَا الْعَامَةُ مُعْتَعَامَةُ مُعْتَاعَا الْعَامَةُ مُعْتَا الْعَامَةُ مَا الْعَامَةُ مُعْتَا الْعَامَةُ مُعْتَا الْعَامَةُ مُعَامَةُ مُعْتَا الْعَامَةُ مُعْتَا الْعَامَةُ مُعْتَا الْعَامَةُ مُعَامَةُ مَعْتَا الْعَامَةُ مَا لَحْتَيْنَا الْعَامَةُ مُعَامَةُ مُعْتَا الْعَامَةُ مُعْتَامَةُ مَا الْعَامَةُ مُعَامًا الْعَامَةُ مُعَامَةُ مُعْتَا الْحَامَةُ مُعْتَا الْعَامَةُ مُعْتَعَا الْحَامَةُ مُعَامَةُ مُعْتَعَا الْحَامَةُ مُعَامَةً مُعَامًا مُعَامَةُ مَالَحَا الْحَامَةُ مَا لَحَامَةُ مُعَامَةً مُعَامَةُ مُعَامَةً مُعَامَةُ مُعَامًا مُعْتَعَامَةُ مُعَامَةً مُعَامَةُ مُعَامًا الْعَامَةُ الْحَامَةُ مُعَامَةُ مُعَامَةُ مُعَامَةُ مُعَامَةُ

The same form can also be used as a purposive, see §15.5.3.

15.4.5 Causality with ablative $=l\varepsilon$

In addition to anteriority (\$15.3.1.1), the ablative marker = $l\varepsilon$ may mark causality. The causal uses seem to be more frequent with stative verbs, see (15.86) and (15.87), although they also occur with

⁴²⁸ In spoken language, the case marker =*na* (WD \mathfrak{q}) or a disyllabic postposition *nàŋɛa/nàŋlo* ($\mathfrak{q} \neg \mathfrak{q}'/\mathfrak{q} \neg \mathfrak{q}''$) is typically used instead of this written short postpositional form $\mathfrak{q} \neg \mathfrak{q}'$.

dynamic verbs, especially when they are negated, see (15.88). When negated, even eventive/dynamic verbs cannot obtain a sequential meaning because there is no action to place into sequence with other actions. Therefore negated action marked with $=l\epsilon$ naturally obtains causative instead of anterior meaning (cf. 15.3.1.1). In my data, $=l\epsilon$ attaches either to the completive marker $-ts^{h}a(:)$, as in (15.86) or -po-infinitive, as in (15.87-15.88). With copulas, due to their stative nature, the usually anterior construction $-ts^{h}a=l\epsilon$ is reinterpreted to express reason, see (15.86), where the full completive form $-ts^{h}a$: is reduced to -sa.

 $[mi \quad t'\tilde{a}:pu \quad in-sa=l\varepsilon]$ $t\varepsilon \quad t^hamtc\varepsilon^2 c\varepsilon -po=lo.$ human honest EQU-CMPL=ABL then all tell-2INF=REP '[Because he was an honest man], he then told everything.' (JDF axe story)

(15.87) אָר־אי ואָרי קאמי קאמי קאמי מקראי אַראידימאי

[$lú\eta ma \ k^h \emptyset:si:si:=di: \ k'j\tilde{a}:-bo=l\varepsilon$] wind chilly=DEMPH.AGT be.cold-2INF=ABL '[Because the chilly wind made them feel cold]...' (Richhi 118)

(15.88) विंदायी तथा केंदा दाउदा वा खेतवा रेंग वाया

 $\begin{bmatrix} k^{h}o\eta = gi & t'yts^{h}\phi^{2} t'aru\eta & ma-\[ep-o=le] \\ 3SG.HON=GEN time & yet & NEG-arrive-2INF=ABL \\ `[Because his time had not yet come]...' (KT e)$

15.4.6 Causality with agentive

In written language, reason may also be expressed by two agentive-marked constructions. In nominal uses, the agentive marks the argument which causes action or is the intrument of the verbal action. In clausal use, it is the verbal action, or the whole clause, that is seen as the causer of another verbal action. In the first construction, the agentive-marked demonstrative-emphatic =di: (WD rachter N '*dis*) attaches to $-c\varepsilon$ -infinitive form of the verb.

(15.89) באָק ימימי דָיאָ בּר ייזישר מיאַקמי בָר בָּבָּ [thinle=gi khim=na=le t'ato=sã: ka=jã: ma-lep-ce=di:] Thrinley=GEN house=LOC=ABL now=until who=even NEG-arrive-INF=DEMPH.AGT '[Because nobody has so far arrived from Thrinley's house]...' (Richhi 43)

sà̃:tɛ $zimk^h\tilde{a}$:= $l\varepsilon$ [dã: nàtea=lo lòk nè: ágja vesterday 1SG.GEN elder.brother house.HON=ABL direct 1pl=dat return $\eta at ca? k^h im = na$ tc'øn-ce=di:] lèm-ba t'arin lók-ti take-PUR come.HON-INF=DEMPH.AGT today 1pl house=LOC return-NF gju-do í. go-PROGEQU.PER '[Because my brother came yesterday directly from home to take us back], we are going back home today.' (Richhi 59)

In the second construction, which is mainly used in literary language, the formative $-p\varepsilon:/b\varepsilon$: (WD \sqrt{N}/\sqrt{N} pas/bas), which looks like an nominalizer in agentive case, attaches to the verb root (15.91) or the completive suffix (15.92).

(15.91) A' a'ra' m'r A'r M'r M'r

[cá sà-wø: lồ:cu jờ:-pɛ:] meat eat-2INF.GEN habit EX-2INF.AGT '[Because (she) had the habit of eating meat]...' (rna-gsung 8)

[di: gjable $k^h \tilde{o}$:=tsu lèpti t^hoţa? sá=lo this.GEN after 3PL=PL very.much high ground=DAT $[ep-ts^ha-be]$ nó? ma-ts^hu-po-dã: demo: arrive-CMPL-2INF.AGT she.demon.AGT reach NEG-be.able.to-2INF-CONJ '[Since they after that arrived in a very high place], the she-demon could not reach (them) and...' (rna-gsung 17)

With verbs of saying, the use of $-p\varepsilon$:/b ε : is rather sequential than causal (although a line in a conversation can be seen as causing the next line by the other speaker).

(15.93) هَا مَن حمر مَن مَنْ مَرْجَ عَلَى وَاحمر مَن مَنْ عَرَبْ عَلَى عَلَى مَنْ مَنْ مَنْ مَنْ مَنْ مَنْ عَلَى اللَّهُ عَلَى اللَّهُ عَلَى اللَّهُ الْعَلَى اللَّهُ الْعَلَى الْ الْعَلَى ال الْعَلَى الْعَلَى

Spoken Denjongke prefers phonologically more complex forms than mere vowel modification for marking causation. Vowel length, such as the one that is suggested by the reading pronunciation of written a_{n}^{2} dis [di:], is not such a prominent feature of spoken Denjongke as to be the only phonological clue of causation on sentence level. In spoken language, the agentive form of the deictic emphatic is expressed more explicitly with =*ki*, see (15.94-95). Moreover, agentive marking is more central in written language than in spoken language, which typically uses other constructions, such as *p'jati(ki)* described in the next section, for expressing causation.

(15.94)

(15.94)

أرباني المجارع المجامع المج

(15.95) היסּמֹידִי אָרָי אָרָאָרָי מָלָיאָשִי [ŋàtɕi=ri kɛ? kjap-ɛɛ=di=gi] וPL.GEN=AEMPH.GEN language strike-INF=DEMPH=AGT '...[because (they) spoke our language]...' (RL interview)

Example (15.96) illustrates a rarer agentive form of which I have only one example

(15.96) মন্বাই'ম'ডৰ শ্ৰীৰ্ম [denri-wa-teen=gi] believe-CIRC-PROG/ADJZR=AGT '[because you believed]' or '[you being a believing one]' (TB discussion with KT)

Here the agentive is attached to the form *-tcen*, which occurs as a progressive marker but is also a derivative suffix, which attaches to nouns to form adjectives, see §3.4.2. In (15.96), the verb is accompanied by the circumstantial marker *-ba*, which makes the verb more amenable to nominal operations, such as the use of the adjectivizer *-tcen*.

15.4.7 Causality through nonfinal converb p'jati(ki) 'doing'

In spoken causal clauses, it is more usual to use the nonfinal converb p'ja-ti(ki) 'doing' than it is to use the agentive (see §15.4.6). The converb typically occurs following - $\epsilon\epsilon$ -infinitive accompanied by =di, see (15.97), but occasionally directly following the infinitive, see (15.98).

(15.97) $\mathfrak{F}_{q}^{\mathsf{eq}} \mathfrak{a}_{\mathsf{e}}^{\mathsf{e}} \mathfrak{a}_{\mathsf{e}}^{\mathsf{e}}} \mathfrak{a}_{\mathsf{e}}^{\mathsf{e}}} \mathfrak{a}_{\mathsf{e}}^{\mathsf{e}}} \mathfrak{a}_{\mathsf{e}}^{\mathsf{e}} \mathfrak{a}_{\mathsf{e}}^{\mathsf{e}}} \mathfrak{a}_{\mathsf{e}}^{\mathsf{e}} \mathfrak{a}_{\mathsf{e}}^{\mathsf{e}}} \mathfrak{a}_{\mathsf{e}}^{\mathsf{e}}} \mathfrak{a}_{\mathsf{e}}^{\mathsf{e}}} \mathfrak{a}_{\mathsf{e}}^{\mathsf{e}}} \mathfrak{a}_{\mathsf{e}}^{\mathsf{e}} \mathfrak{a}_{\mathsf{e}}^{\mathsf{e}}} \mathfrak{a}_{\mathsf{e}}^{\mathsf{e}} \mathfrak{a}_{\mathsf{e}}} \mathfrak{a}_{\mathsf{e}}^{\mathsf{e}}} \mathfrak{a}_{\mathsf{e}}^{\mathsf{e}}} \mathfrak{a}_{\mathsf{e}}^{\mathsf{e}}} \mathfrak{a}_{\mathsf{e}}^{\mathsf{e}}} \mathfrak{a}_{\mathsf{e}}^{\mathsf{e}}} \mathfrak{a}_{\mathsf{e}} \mathfrak{a}_{\mathsf{e}}^{\mathsf{e}}} \mathfrak{a}_{\mathsf{e}}^{\mathsf{e}} \mathfrak{a}_{\mathsf{e}}^{\mathsf{e}}} \mathfrak{a}_{\mathsf{e}}^{\mathsf{e}}} \mathfrak{a}_{\mathsf{e}}^{\mathsf{e}}} \mathfrak{a}_{\mathsf{e}}^{\mathsf{e}}} \mathfrak{a}_{\mathsf{e}}^{\mathsf{e}}} \mathfrak{a}_{\mathsf{e}}^{\mathsf{e}} \mathfrak{a}_{\mathsf{e}}^{\mathsf{e}} \mathfrak{a}_{\mathsf{e}}^{\mathsf{e}} \mathfrak{a}_{\mathsf{e}}^{\mathsf{e}} \mathfrak{a}_{\mathsf{e}}^{\mathsf{e}} \mathfrak{a}_{\mathsf{e}}^{\mathsf{e}} \mathfrak{a}_{\mathsf{e}}^{\mathsf{e}} \mathfrak{a}_{\mathsf{e}}^{\mathsf{e}} \mathfrak{a}_{\mathsf{e}}^{\mathsf{e}} \mathfrak{a}_{\mathsf{e}} \mathfrak{a}_{\mathsf{e}}^{\mathsf{e}} \mathfrak{a}_{\mathsf{e}}^{\mathsf{e}} \mathfrak{a}_{\mathsf{e}}^{\mathsf{e}} \mathfrak{a}_{\mathsf{e}}^{\mathsf{e}} \mathfrak{a}_{\mathsf{e}}^{\mathsf{e}} \mathfrak{a}_{\mathsf{e}}^{\mathsf{e}} \mathfrak{a}_{\mathsf{e}}^{\mathsf{e}} \mathfrak{a}_{\mathsf{e}}^{\mathsf{$

(15.98) $\eta_{e}\hat{R}^{c} = \pi^{c} \cdot q^{n} \tilde{A}^{c} \cdot \tilde{A} - \pi^{c} \cdot q^{n} \tilde{A}^{c} \cdot \tilde{R}^{c} \cdot \tilde{R}^{c} \cdot \tilde{R}^{c} \cdot \tilde{R}^{c} \cdot \tilde{R}^{c} \cdot p^{c} \cdot p^{c} \cdot p^{c} \cdot p^{c} \cdot p^{c} \cdot p^{c} \cdot q^{c} \cdot q^{c$

The nonfinal form p'ja-ti(ki) and its reduced form p'ja also attach to nouns (15.99) and pronouns (15.100):

 $\begin{bmatrix} t'iz\tilde{a}: & n\dot{\varepsilon}:t\tilde{a}:=di: & p'ja-ti \end{bmatrix}$ but condition=DEMPH.AGT do-NF '[Because of the condition (of hers)]...' (Richhi 171)

(15.100) $\breve{w}(a\beta')(\beta')$) a) $\acute{odi} p'ja$ b) $\acute{odi} p'ja-ti$ c) $\acute{odi} p'ja-tiki$ that do-NF 'Because of that...'

15.4.8 Causal uses of the circumstantial-purposive converb

In certain contexts, such as (15.101) and (15.102), the circumstancial converb can be interpreted as expressing reason/causality:

(15.101) אוֹיקָקָ׳ מִקְ׳ מִישָאַ׳ אַישָאַי אַישָאַי אַישָאַי אַישָאַי אַישָאַי אַישָאַי אַישָאַי אַישָאַי אַישָאַי [moby=di lèm ma-jà-wa] cí-u bc?. wife=DEMPH good NEG-do-CIRC die-2INF EQU.NE '[(His) wife not doing well (in caring for him)], he died'/'He died [because (his) wife did not do well (in caring for him)].' (KN e)

(15.102) הַ הַשָּקון שִישִישִי אֹקי שָּׁקָקִי אָשָׁקָקִי אָשָׁקָשִישָּׁדן [ŋà gjuk jà:-ja-wa] lèp k'jop to:-ts^ha:. 1SG run go-RDP-CIRC very.much stomach hunger-CMPL '[Having run and run], I became very hungry.' (KN e)

For circumstantial uses, refer to §15.8.1.

15.5 Purposive clauses

Denjongke uses mainly three constructions for expressing purpose: 1) circumstantial-purposive converb marker -pa/ba used with a verb of motion and a few other verbs, 2) postposition complement clause with t'ønlo 'for the purpose of', and 3) the nonfinal converb lap-ti 'saying'. Purpose and reason are naturally related concepts, as supported by the fact that the converb lap-ti is used for expressing both.

15.5.1 Purpose with circumstantial-purposive converb marker -pa/ba

The uses of the converb -pa/ba can be roughly divided into affirmed purposive uses and negated circumstantial uses.⁴²⁹ A similar connection between manner (similar to circumstantial) and purposive converbs is noted by Vanhove (2016: 330), who comments that in Beja (Cushitic) "the Manner converb of action verbs may encode a purposive interclausal relation if the following verb is a verb of motion." The purposive uses of -pa/ba occur with verbs of motion (15.103), and some other verbs illustrated by go? be 'needed' (15.1104) and $t^{h}op$ 'receive' (15.105). Note that the purposive may be followed by dative-locative case-marking, see (15.104) and (15.105).

- (15.103) הישקי מאיצוי באלימקאי איז איז דעבדי לי שאקן *nàtea? ní:-po* [bhaila ta-wa] tea:-bo í: איז 1PL two-COL PN watch-PUR come.HUM-2INF EQU.PER 'The two of us came [to see Bhaila].' (Richhi 11)
- (15.104) هَم: تَمَمَّ: مَالَمَ عَمَّاتِ مَالَ عَمْرَةُ اللَّهُ عَمَّاتِ مَالَحَ عَمَّاتُ عَمْرَ عَمَّاتُ عَمْرَ عَمَّاتُ عَمْرَ عَمَّاتُ عَمْرَ عَمَّاتُ عَمْرَ عَمَالًا وَ [sé:wø: damŋa? **dup-a=lo**] cà? dyn=lɛ lako maŋ-go?. hail.GEN instruction achieve-PUR=DAT night seven=ABL more NEG-be.needed '(He) did not need more than seven nights [in order to achieve (mastery of) the hailmantra]. (mi-la ras-po 4)

⁴²⁹ However, see exceptions in §15.8.1.

In spoken language, the converbal marker may be dropped if context allows. According to consultant KN, all the forms (15.106) are possible for a purposive construction.

(15.106)	a) <i>nà</i>	sé?			số∵-zε	छ' नष्यमु' र्सेम्'बे।
	b) <i>nà</i>	sé	-pa		số∶-zε	9. 202 . 20 . 20 . 20 . 20 . 20 . 20 . 2
	c) <i>nà</i>	sé		=lo	số∶-zε	छ. ययर् सें सेंट ले
	d) <i>nà</i>	sé	-pa	=lo	số∶-zε	छ' यह्र राखें र्सेट ले
	fish	kill	-PUR	=DA'	Г go.PFV-	PST
	'(I) v	vent f	ishing	.' Lit.	'(I) went	[to kill fish].' (KN e)

Another example of a mere verb root as purposive is (15.107).

(15.107) $\check{\mathfrak{M}}$ 'AFA' $\check{\mathfrak{T}}$ 'AFA' $\check{\mathfrak{T}}$ 'AFA' $\check{\mathfrak{T}}$ 'AFA' $\check{\mathfrak{M}}$ 'A

Similarly, Sandberg (1895: 66) gives the example clause *Ngá chhö tá ong che* 'I have come to see you', where the mere verb root is used purposively. The clause is given in edited form in (15.108).

(15.108) Sandberg (1895: 66) (WD, transcription and glosses mine) $\overrightarrow{b} \overrightarrow{b} \overrightarrow{b} \overrightarrow{c} \overrightarrow{b} \overrightarrow{b}$ $y a t c^h \phi P t a \overrightarrow{o} \cdot t c \varepsilon.$ 1SG 2SG.L look come-PST 'I have come to see you'

The actor of the converbal clause with -pa/ba is always the same as that of the main clause. If the actor is different, a construction with the complementizer *làpti* is used, see §15.5.3.

15.5.2 Purpose with the postposition t'onle/t'onlo

Purpose clauses can also be formed by adding a complement clause to the relator noun *t'ønle/t'ønlo* (also *t'ønda?/t'øndale/t'øndalo*) 'for the purpose of, in order to; because', deriving from the noun t'øn(da?) 'meaning, sense; purpose; reason'. The complement clause may either be a genitivized *-po*-infinitive-clause (15.109) or a clause with *-ce*-infinitive, either with genitive marking (15.110) or without further marking (15.111).

- (15.109) العقدة: حريق: تَحَدَّ مَعَبَّرَ تَحَمَّ عَبَّرَ عَرَبَهُ عَبَّرَ عَرَبَعَ الْمَعَ الْمَعَ الْمُعَالِي اللَّهُ الْمُعَالِي الْمُعَالُ الْمُعَالِي الْمُعَالِي الْمُعَالِي الْمُعَالِي الْمُعَالِي الْمُعَالِي الْمُعَالِي الْمُعَالِي الْمُعَالِي الْحَامِ الْمُعَالِي الْمُعَالِي الْمُعَالِي الْمُعَالِي الْمُعَالِي عَلَيْكَامِ عَلَ الْعُلْمُعَالِي الْمُعَالِي الْمُعَالِي الْمُعَالِي عَلَيْكَ عَلَيْكَ مَعَالَي عَلَيْكَامِ عَلَيْكَامِ عَلَيْ مُعَالِي عَلَي عَلَيْ عَالَةُ عَالَي عَلَيْكَ مَعَالِي الْمُعَالِي عَلَيْ عَالِي الْمُعَالِي الْمُعَالِي عَلَيْكَ مَعَالِي الْمُعَالِي الْمُعَالِي الْمُعَالِي الْمُعَالِي الْمُعَالِي الْمُعَ الْعُلْمُعَالِي الْمُعَالِي الْمُعَالِي عَلَيْكَ عَلَيْ عَلَيْ عَلَيْ عَالِي الْمُعَالِي الْمُعَالِي الْمُعَالِي عَلَيْ عَالَي الْحَالَةُ عَلَيْ عَالِي الْحَالَةُ الْحَالَةُ عَلَيْ عَالَي عَلَيْ عَالَي عَلَيْ عَالَةُ عَالَيْ عَالِي
- (15.110) $\sqrt[4]{a} \sqrt[4]{a} \sqrt[4$
- (15.111)k^hu [paksam cíŋ=gi t'øndale] minto? dzip-ce? sá balsam tree=GEN flower suck-INF for.purpose.of 3SGM ground $\partial = l\varepsilon$ ts^hã: zo-tiki gu: $d\emptyset$:- $\varepsilon\varepsilon$ =lo=s. nest make-NF wait sit-INF=REP=QUO below=ABL '[In order to suck (nectar from) balsam tree's flower], he makes a nest underground and sits waiting (so the story goes).' (RS bee story)

In (15.112), the purpose clause is postposed to the main clause, resulting in a clause that ends in a postposition rather than a verb.

The reason why the purpose clause in (15.112) is postposed to the main clause is probably that it would be difficult to process such a long clause if it were placed between the topical actor *ápo* 'father' and the patient *gompo* 'leave-of-absence' in the main clause. The dependence of the purpose clause is signalled by the lack of the sentence boudary marker in Denjongke writing.

15.5.3 Purpose with nonfinal converb làpti 'saying'

In addition to forming causal clauses (see \$15.4.4) and functioning as a complementizer (see \$14.2.2.1), the nonfinal form of the verb *làp* 'say' may also form purpose clauses by attaching to declarative (15.113), optative (15.114) and hortative verb forms (15.115).⁴³⁰ Example (15.113a) is a comment by a house-owner about the purpose of the 40-centimeter-high threshold. Example (15.113b), on the other hand, was an answer to the question, what is the purpose of heaping earth at the feet of maize stalks.

⁴³⁰ Similar to Nepali purposives with *bhanera*-constructions (bhan-era [say-CPTCP]).

a) ५२ २ ही अप्राय रहें वर्त्त रहें के राम के रहें के राम के राम के राम के के राम के (15.113)di t'a $[k^{h}i \quad \dot{a}ly:=tsu \ n\dot{a}\eta \epsilon a \quad dzy:-\epsilon \tilde{i}:$ làp-ti] t'ã:py: this now dog cat=PL inside enter-NPST.PER say-NF long.ago.GEN $\tilde{i} = s.$ mí=gi dem dzo-u human=AGT like.that build-2INF EQU.PER=QUO 'Now people long ago built this like this [so that dogs and cats would not enter inside]. (Lit. 'People long ago built this like this [saying that dogs and cats will enter inside].') (PD living room video)

b) هَٰ (مَרْ) أَمَ عَٰ الْحَادِ مَامَاً عَٰ عَامَ الْحَادِ مَامَاً عَامَاً عَامَ الْحَادِ مَامَاً عَامَانِ الْحَادِ مَامَاً عَامَاً مَامَاً عَامَاً عَامَاً مَامَاً عَامَاً مَامَاً عَامَاً مَامَاً مَامَاً عَامَاً مَامَاً مَامَاً مَامَاً عَامَانِ مَامَاً مَامَا مَامَاً مَامَا مَامَاً مَامَا مَامَا مَامَاً مَامَاً مَامَا مَامَاً مَامَاً مَامَاً مَامَاً مَامَاً مَامَالِي مَامَا مَامَا مَامَالِي مَامَا مَامَاً مَامَا مَ مَامَاً مَامَاً مَامَاً مَامَاً مَامَاً مَامَاً مَامَاً مَامَالَا مَامَا مَامَا مَامَالَا مَامَالَ مَامَاً مَامَاً مَامَاً مَامَاً مَامَاً مَامَا مَامَاً مَامَاً مَامَاً مَامَا مَامَا مَامَاً مَامَا مَامَا مَامَا مَالْمَا مَامَا مَامَا مَامَا مَامَا مَامَاتِ مَامَا مَامَ مَامَا مَامَامَا مَامَا مَامَا مَامَامَا مَامَا مَامَا مَامَا مَامَامَام

Negated purpose clauses are formed by negating the optative construction by prefixing *-ma* to the secondary verb. For affirmative and negated counterparts, see (15.114a-b).

a) दते गी नूत देते रा ते (birthday party) व हा केंद्र नरुवा वर हे द देवा के केंद्र (15.114)k^hu õ bəððɛipa:ti=na [nèː-gi tcu? làp-ti] ηà birthday.party(Eng)=LOC 3SGM my=GEN come cause say-NF 1SG fon p'ja-u \tilde{l} phone(Eng.) do-2INF EQU.PER 'I phoned (him) [so that he would come to my birthday party].' (literally: '[Saying let him come to my birthday party], I phoned (him).') (KN e) b) दतिगी पूर्टरेत पार्ते (birthday party) दा हा पॅटापरुवा वपाक्षे रा देवा देवा के का $k^h u \hat{\tilde{o}}$: ma-tcu? [nè:-gi bəððɛipa:ti=na làp-ti] ηà birthday.party(Eng)=LOC 3SGM come NEG-cause say-NF 1SG my=GEN p'ja-u î. fon phone(Eng.) do-2INF EQU.PER 'I phoned (him) [so that he would not come to my birthday party].' (literally: '[Saying let him not come to my birthday party], I phoned (him).') (KN e)

In (15.115), *làp-ti* forms a purpose clause in conjunction with the hortative form $-k\varepsilon/g\varepsilon$.

(15.115) تَ عَرْجَ مِعْنَ مَعْنَ عَرْبَ مَعْنَ عَرْبَ اللَّهُ عَرْبَة اللَّهُ عَرْبَة اللَّهُ عَنْ اللَّالَ عَنْ اللَّهُ عَنْ اللَّهُ عَنْ اللَّهُ عَنْ اللَّهُ عَنْ اللَّهُ عَنْ اللَّهُ عَنْ الْحَالَةُ عَنْ الْحَالَةُ عَنْ اللَّالَ اللَّالَةُ عَنْ الْحَالَيْ عَنْ الْحَالَةُ عَنْ الْحَالَةُ عَنْ الْحَالَةُ عَنْ الْحَالَةُ عَنْ الْحَالُ اللَّالَ اللَّالَةُ عَنْ الْحَالَةُ عَ الْعَامَا اللَّالَةُ عَنْ الْحَالَةُ عَنْ اللَّا عَالَيْ اللَّالَةُ عَنْ الْحَالَةُ عَنْ الْحَالَةُ عَنْ الْحَالُ اللَّالَالِي اللَّالَةُ عَالَيْ عَالَةُ عَالَةُ عَالَةُ عَالَى الْحَالَةُ عَالَةُ عَالَيْ الْحَالَةُ عَالَيْ عَا الْحَالَةُ عَالَيْ عَالَةُ عَالَيْ عَالَةُ عَالَيْ عَالَةُ عَالَيْ عَالَةُ عَالَيْ عَالَةُ عَالَةُ عَالَى الْحَالَةُ عَالَةُ عَالَيْ الْحَالَةُ عَالَةُ عَالَيْكَ عَالَى الْحَالَةُ عَالَيْ عَالَيْ عَالَةُ عَالَيْ عَالَةُ الْحَالَةُ عَالَى الْحَالَةُ عَالَيْ عَالَيْ عَالَةُ عَالَيْ عَالَيْ عَالَةُ عَالَيْ عَالَ عَالَيْ عَالَيْ عَالَيْكَالَ الْحَالَةُ عَالَيْ عَالَيْ عَالَيْ عَالَيْ عَالَ الْحَالَةُ عَالَيْ عَالَةُ عَالَةُ عَالَيْعَالَ عَالَةُ عَالْ

15.6 Conditional clauses

A conditional clause is formed by an obligatory final converb $-(pate\epsilon)n\epsilon$ (also $-bate\epsilon n\epsilon$), which may be accompanied by the optional initial word $k'\epsilon si^2$ 'if'.⁴³¹ I am not aware of any meaning difference between the shorter form $-n\epsilon$ and the longer form $-pate\epsilon n\epsilon$. The relator noun tenle '(from) upon' may also be used in a conditional sense, see (15.133). The conditional marker attaches to the verb root (which may be a final auxiliary copula) or the completive $-ts^ha(\epsilon)$. In (15.116), the conditional attaches to the simple verb root and, typically of this construction, obtains a present/future reading. In (15.117), on the other hand, the conditional is postposed to the final auxiliary copula, a construction which allows referring to the past. As a general rule, those TAME⁴³²-constructions which end in an auxiliary copula, either equative or existential, can occur in the conditional form (for the various constructions, see §8).

3SGF=DAT 2SG.L=GEN address give

'[If (you) get an address from her], you also give her your address.' (Richhi 95)

गायां श्वेर राज्या गीया वर्त के रावाया में राये के राज्या के राज्या ही याया के का का के के के के के के के के के (15.117)[k'e:si? nàtca=ki di=na nórť^hy: có:-bo ím-batsene] nàtca? this=LOC mistake slip-2INF if 1PL=AGT EOU-COND 1PL ĩ:. t^hudẽ:gõ:da? cù-ce request-INF EQU.PER apology '[If in this we made mistake(s)], we apologize.' (NAB BLA 7)

The conditional only attaches to the personal copulas \hat{i} and $j\hat{\partial}^2$ and not to neutral $b\epsilon^2$ and sensorial du^2 , e.g. $*b\epsilon \cdot (pat\epsilon\epsilon)n\epsilon$, $*duk \cdot (at\epsilon\epsilon)n\epsilon$. Thus evidential distinctions are neutralized with the conditional.

The optional $k' \varepsilon: si?$ 'if' is not strictly initial in that other elements may precede, i.e. a term of address and a temporal adverbial in (15.118).

⁴³¹ k'ɛ:si? functions quite analogously to the optional Nepali *jedi* 'if'.

⁴³² Tense, aspect, modality, evidentiality

अरमु गुन्स्सेन भ्रानेषात्र गया में राषा देया है के के के नाम के का के का के क (15.118)ágja [k'uŋsiŋ kap=na k'ɛːsi? bombai=lo *òm-batgene*] elder.bother holiday time=LOC if TPN=DAT come-COND õ:. par=tsu nà=rã: bak lep picture=PL 1SG=AEMPH carry arrive come 'Brother, [if I come to Bombay during the holiday], I will arrive bringing the pictures myself.' (Richhi 108)

The conditional marker may attach to the completive marker $-ts^ha$, allowing an aspectual value to be expressed within a conditional clause. In (15.119), $-ts^ha$ is preposed to the longer form -patcene and in (15.120) to the short form -ne.

(15.120) ך: هَنْ جَ جَسَّ مَعْ: جَسَّ مَعْ: جَ مَعْ: جَ مَعْ: مَعْ: جَ مَعْ: مَعْ: عَلَى مَعْنَ مَعْنَ عَلَى مَ مَعْنَ عَلَى مَعْنَ مَعْنَ مَعْنَ عَلَى مَعْنَ عَلَى مَعْنَ عَلَى مَعْنَ عَلَى مَعْنَ مَعْنَ عَلَى مَعْنَ عَلَى مَعْنَ عَلَى مَعْ

The conditional construction is negated by ma-, which may reduce to m- with vowel intial roots:

 (15.121) a) תקקי אימָר קֿ אָר אָר אָר אַימִקן
 [ba? m(a)-õ:-nɛ] mýtsy=lo ma-làp. carry NEG-come-COND other=DAT NEG-say '[If (it) is not brought], don't tell (it) to others.'(KNA kitchen)
 b) ך אָר אָר אַיאָק אָר איאַק קיקי פֿיק׳.

[*t'a loke ma-k^hem-bateene*] now Lhoke NEG-know.HON-COND '[Now if (they) do not know Lhoke]...' (YR canteen video)

Note, however, that the imperfective negator mi- may be used with a verb if the conditional form of the verb $s\acute{e}/si$ 'say' follows:

⁴³³ The translation 'either with me or alone' derives from the apposed forms $s\delta$: and gju. The first, $s\delta$:, is a command to the second person, whereas the second form, gju, is used as a hortative 'let's go', which includes both the speaker and the addressee.

(15.122)nấːtɕʰãː nìmts^hi [t'a máko lò ní: *mi-kjap sí-ne*] nàtea? now groom year two NEG-do say-COND 1PL engagement date mákø:=gi ke? ťi g0:-68 $b\epsilon$?. groom.GEN=GEN gain ask be.needed-INF EOU.NE 'On the day of the engagement, [now if the groom says (he) is not going to work for two years (in the bride's home)], we have to ask for the groom's gift.' (SGD wedding customs)

In the following two clauses the completive conditional form *bom-ts^ha-n* ε 'if (one) grows' is used for prospective future.

(15.123) جَرَعَت عَلَى اللَّهُ اللَّاللَّالَ اللَّالَ اللَّالَ اللَّلَ اللَّهُ اللَّهُ اللَّهُ اللَّا لَيَحَامُ اللَّالَ لَقُلْقُلْقُلْ اللللَّالَ اللَّ اللَّاللَّالَةُ اللَّالَةُ اللَّالَةُ اللَّالَةُ اللَّالَةُ الللَّالِ الللَّ اللَّا اللَّالَةُ اللَّالَةُ اللَّالِ الللَّالَةُ الللَّالِي اللَّالِي الللَّالِي اللَّالِي اللَّالَةُ اللَّالِ لَقُلْقُلْقُلْقُلْلِ الللَّالَ اللَّالَ اللَّالَ اللَّالَ اللَّ اللَّالَ الللَّالُ اللَّا اللَّالَةُ اللَّالَةُ اللَّالَةُ اللَّالَةُ عَلَيْ اللَّالَةُ عَلَيْ الللَّالَةُ اللَّالَةُ اللَّالَةُ لَ اللَّاللَّالَةُ اللَّالَةُ اللَّالِي الللَّالِي الللَّالِي اللللَّالِ اللللَّالِ اللَّالِي اللللَّ لَاللَّالَ لَ اللَّاللَّاللَّالَةُ اللَّالَةُ اللَّالَةُ لَاللَّاللَّ لَاللَّالَ لَالَ اللَّاللَّالَةُ اللَّالَةُ اللَّالَةُ اللَّاللَّالَةُ الللَّالَ الللَّالَ لَاللَّالَالَةُ اللَّالَةُ اللَّ

I suggest two hypothetical motivations for using the completive conditional form to convey an essentially anterior future meaning in (15.123) and (15.124). The first is that there may be a division of function between the completive anterior construction $ts^ha=l\varepsilon$, which is restricted to past and habitual uses, and the completive conditional construction $ts^ha-n\varepsilon$, which covers future anterior uses. The second is that the past uncertainty of a child to survive into adulthood has been encoded in the grammar as uncertainty implied by the conditional. If the latter is true, *bom-ts^ha-n*\varepsilon in (15.124) could have the implied meaning 'if he will have managed to grow up'

The conditional construction can also be used counterfactually to describe a past alternative reality, i.e. what would have happened if something else had not taken place, see (15.125) and (15.126)

⁴³⁴ In English, it is more natural to talk about such an everyday occurrence as the growing of a child with the temporal expression 'when'.

(15.125)พन्म मार्ग्य के राज्य के राज dã: nàtci bhaila=ki go tok-ti di: dø: j*à:-kap* yesterday 1PL.GEN PN=GEN head hit-NF fall stay EX-SIM [p'usim=la: *mè:-patcene*] khu óna=rã: *cí-c*ε $b\epsilon$?. vounger.sister=HON NEG.EX-COND 3SGM there=EMPH die-INF EQU.NE 'Yesterday when our Bhaila was lying fallen after hitting his head, [if it wasn't for the sister], he would have died on the spot.' (Richhi 12)

(15.126)พहें र्रोट हेट यो लय में जब हे टर्ट येया राज्य राज्य यात हेत हेत या गीय जार हेंगा है के लय है जा है के लय है जी [ádzo *bont*^{*h*}*in*=*gi* cè:=lo nèn-di t'arun Lepcha.priest=GEN mouth.HON=DAT listen-NF grandfather still k^him=na zak-ne] k'an t^høn *t^høn* ka=gi làp house=LOC leave-COND what happen happen who=AGT say ts^hu-po? be.able.to-2INF '[If we had listened to grandfather bongthing's words and left (the patient) still at home], who can tell, what all would have happened.' (rnam-rtog 31)

The conditional marker may be followed by the demonstrative-emphatic = di.

(15.127)	ને. ખુત્ર, જાશ્રુ, પુર્ટુ, તટ્ટુ, તટ્ટે સંગ્રે સંગ્રે, તટ્રે ત્રારુવા, પર્ટવા, પ્ર							
	[tɛ	lèp	nóː-nɛ=di]	dența?	dɛm=tɕi?	du-kɛ.		
	then	very.much	think-COND=DEPMH	true	such=INDF	EX.SEN-IN		
	' [If (]) think hard	(about it)], (it) looks	s like it's	true.' (CY i	nterview)		

While (15.128) exemplifies a frequent use where the conditional form of a verb of speaking introduces a new topic for discussion, (15.129) provides a metacomment on the discourse.

- त्र को सुं, खे. य. कु. थ. थ. य. भू. प्रांग, पट्ट. द्विता. क्रे. त्व प्रा. क्रे. क्रे. क्रे. क्रंथ. क्रेंथ. प्रा. क्रंथ. त्या. क्रं. ज्या. क्रं. ज्या. क्रं. ज्या. क्रं. ज्या. क्रं. (15.129)[t'a né:mu cù-watsene] te k^hik-ti máko=di now real(ly) say-COND then son-in-law=DEMPH lead-NF $k^h a t \epsilon^h \tilde{a}$: nìmts^hi tøn go:=s làp-ti jøbbe?. Khachang-ceremony date show be.needed=QUO say-NF EX.NE '[If (we) speak the truth], then there is (the custom) that the son in law has to be brought and shown on the date of the Khachang-ceremony (of marrying).' (SGD wedding customs)

In fast speech, the full conditional form -patcene may be reduced to -patce (15.130) or even -pa, represented by the allophonic varient -a in (15.131).

- (15.130) מוקאי מויא: הערך קמא מרויגויציאי די בישמי אוי [tam=tsu k'ate cé? go? làp-atce] ŋàtea=ki word=PL how tell be.needed say-COND 1PL=AGT '[If (we) speak about how the words should be told], we...' (YR canteen video)
- (15.131)t'a tsi:pø:=tsa: gju-tiki [tsi? dik-atsene] t'a nấ:tc^hã ke: now astrologer.GEN=at go-NF now bethrotal astrology fit-COND bring [k'e:si? tsi? *bε*?. *man-dik-a*] $(t) \mathcal{L}(h) O$ be.allowed EQU.NE if astrology NEG-fit-COND 'Now, having gone to the astrologer, [if the astrological calculations (of the prospective bride and groom) are fitting], bethrotal can be arranged. [If the astrological calculations do not fit]...' (SGD wedding customs)

In (15.131), the context helps the addressee interpret the abbreviated form *man-dik-a* as a conditional rather than the homophonous circumstantial-purposive converbial form. First, *man-dik-a* forms a logical pair with the full conditional form dik-atsene in the previous clause. Second, the short form is accompanied by k'e:si? 'if', which helps to disambiguate the clause as conditional.

In addition to the typical converbal uses illustrated above, the conditional form $-n\varepsilon$ is used in an idiomatic construction with the secondary verb *ta* 'look' and the attention marker $=\varepsilon o$ to form questions/suggestions presenting an alternative course of action.

The relator noun *teyle* '(from) upon' is once in my data used in a conditional sense:

(15.133) $\underset{k}{\Re q} \cdot \alpha - \gamma \cdot \alpha - \eta \alpha \cdot \beta \cdot \beta - \gamma - \alpha - \eta \alpha \cdot \beta \cdot \delta - \gamma \cdot \delta \cdot \eta = [m \epsilon n = di l \epsilon m p' j a - ti s a - w \phi : teyle] m \epsilon nni t \epsilon^h \phi ? m i - \epsilon i. medicine=DEMPH good do-NF eat-2INF.GEN upon perhaps 2SG.L NEG-die '[If (you) eat the medicine well], perhaps you will not die.' (KN e)$

15.7 Concessive clauses

The concessive converb marker *-ruŋ* is suffixed to the verb root and functions similarly to the English concessive 'although' and "concessive conditional" 'even if' (Thompson et al 2007: 261). Unlike the conditional -(patce)ne, my data has no examples of *-ruŋ* attached to the completive marker *-ts^ha*. The time reference in concessive clauses of the form VERB-*ruŋ* has to be deduced from the context. In (15.134), the verbal action marked by *-ruŋ* has already taken place.

In (15.135), the time reference is present and in (15.136) hypothetical future.

(15.135)ลิดิ ทุลุทุพ รัม วิ ณ์ พัว รุรา ลิมพ ดรุพ ดรุพ ลิรา จุมีราย ภูมางาลิรา พัว [mi]zu? bombai=lo j*àː-ruŋ*] sém=di: dzamlin korwa human.GEN body TPN=DAT mind=DEMPH.AGT EX-CONC world roaming kjap-z \tilde{e} : j*à*?. do-prog EX.PER '[Although (his) human body is in Bombay], his mind is travelling around the world.' (Richhi 135)

तनेवा क्षेत्र'त'र्ने' वीं'ठवा'रूम यात्र' त्युं'रुम केंम म वीं'ठवा' व्यववा वेत्र'त्युं (15.136)k'utca=rã: [k`ana gju-run] số:. dem $(n) \in to$ ηà like.that EQU-COND=CEMPH 2PL=AEMPH where go-CONC go.PFV 1sg k'utca? pámtci? miŋ-gju. 2pl with NEG-go 'In that case, you (pl.) go [wherever you go]. I'll not go with you (pl.).' (Richhi 80)

As seen in (15.136), a concessive verb postposed to an interrogative word functions similarly to English '-ever' suffixed to a question word. Other examples are *k*'an *p*'ja-run [what do-CONC] 'whatever (he) do(es)', *ka* \tilde{i} :-run [who EQU-CONC] 'whoever (she/he is)', see §6.3.2.

The concessive verb is negated by the prefix ma-. A construction which juxtaposes the affirmative and negated concessive uses functions similarly to English expression 'whether...or not':

- (15.137) \vec{p} , $\vec{n} \in \vec{q}$ and $\vec{q} \in \vec{n}$ and $\vec{q} \in \vec{n}$ and $\vec{q} \in \vec{n}$ and $\vec{q} \in \vec{n}$. $t \in [k^h \vec{o}: c e :-r u \eta]$ [ma-c e :r u \eta] $k^h \vec{o}: p \dot{y}: t^h o p - o$ be?. so 3PL know-CONC NEG-know-CONC 3PL money receive-2INF EQU.NE 'So [whether they knew it] (or) [did not know], they received money.' (CY interview)
- (15.138) = (isà-khế: jô:-ruŋ] [mè:-ruŋ] t'ytshø:=na sàm zo za?. eat-NMLZ EX-CONC NEG.EX-CONC time=LOC food make put '[Whether there are eaters] (or) [not], have the food prepared in time.' (Class 9-10 grammar, 135)

The initial $k' \varepsilon si?$, which occurs in conditional clauses, can also be used with *-ruŋ* with the meaning 'even if'. The uses with *-ruŋ*, however, seem rare. The only example in my data is (15.139).

The verb root to which the concessive marker *-ruŋ* attaches may be the final copula of a complex predicate. In these cases, tense values are explicitly marked by the verb complex, in contrast to the earlier examples, where the temporality was contextually interpreted. For instance, (15.140) exemplifies a concessive present habitual construction.

(15.140)	हे. क्षेत्र क्रुतार्खः वा. टालुवायाः आवत्र अत्र उद्र त्यायाः ह्यांव	ଷ' ସନ୍ଧି' ଷଞ୍ଚଷ୍ୟ' ସକ୍ତୁମ	''ব'', ই্র্রিযাঝ' যা' দ্বন্ত্রেযাঝ'র	१[यव: क्षेव:उट: ट:ठग]					
	या र्ख्याया रूप्या झूर्ये धर क्रुया यहरायी खे	_							
	[tɛ lɛŋgɛː=tsu kʿaː zuː-kʰɛː	Ĩː-ruŋ]	[dɛndzõː tɕʰoː	zi so					
	PRN.HON=PL where live.HON-NMLZ	EQU-CONC Si	ikkim direction	four					
	$ts^{h}am$ ge:=na $te^{h}o$: $k^{\prime}a$:	zu:-k ^h ẽ:	Ĩ:-ruŋ]	yàtɛa?					
	border eight=LOC direction where	live.HON-NML	Z EQU-CONC	1 PL					
	$k'a: ts^{h}u^{2}$ $rag=gi$ $k\varepsilon:=lo$	jàrgɛ	taŋ-gɛ=s.						
	what be.able.to own=GEN language=	DAT develop	ment send-HOR	T=QUO					
	'[So wherever you live], [wherever you live in the four directions and eight borders of								
	Sikkim], let's develop, however we can, our own language.' (KT life story)								

The copula concessive form \tilde{i} -ruy has further developed into a contrastive connective 'but, however', see §12.2.

15.8 Clauses of circumstance and manner

Denjongke uses several constructions for forming adverbial clauses which express an attendant circumstance or manner associated with the main verb. The constructions described here are formed around the circumstantial converb marker -pa/ba (§15.8.1), the postposition nayea/naylo 'inside' (§15.8.2), progressive marker -teen/zen (§15.8.3), the word $k^ha=lo$ 'mouth=DAT' (§15.8.4), comparative manner markers dem 'like (it)', naytar(gi) 'according to', nayzin 'according to, similar to' and t'ønzin(gi) 'in accordance with' (§15.8.5), and genitivized -po-infinitive (§15.8.6). These markers and the constructions associated with them are addressed in the following subsections.

15.8.1 Circumstantial-purposive converb marker -pa/ba

The circumstantial-purposive converb marker -pa/ba can mark both attendant circumstance to the action in the following main verb (mainly in negated clauses) and purpose (in affirmative clauses and mainly in conjunction with verbs of motion). The purposive uses are decribed in §15.5.1 above. While affirmative circumstantial uses do exist in my data, the majority of affirmative uses of -pa/ba are purposive. All negated uses, whether occurring with verbs of motion or not, are by necessity

circumstantial (e.g. *I came here without eating*), because purposive uses are highly unlikely (e.g. *I came here in order not to eat*).

Examples (15.141-142) illustrate the rare affirmative circumstantial uses of -pa/ba. Note that in (15.141) the converb, which typically occurs before the main clause, is postposed to the main clause. In (15.142), the reduplication of the verb root favours the circumstantial rather than purposive interpretation.

The negated circumstantial use of -pa/ba, which also functions as the negation of the nonfinal converb -ti/di, is illustrated in (15.143). Note that (15.143d) has a clause chain where the negated verb occurs with -pa/ba and the affirmed one with -ti/di.

- (15.143) a) אַשִּׁי מִקָר ה׳שִּׁשִׁאִ׳ מּישְׁשָּׁה׳ מּישָשָּׁה׳ מּישָשָּׁה׳ מּישָשָּׁה׳ מּישָשָּׁה׳ מּישָשָּׁה׳ מּישָשָּׁה׳ מּישָשָּׁה׳ מּישָשָּׁה׳ מּישָשָּׁה׳ מּישַשָּׁה׳ מּישַשָּׁה׳ מּישַשָּׁה׳ מּישַשָּהָרָגי דָּי וֹמָה׳ מּי שָּרָ בּאַן [só:=di ŋàtea=ki ma-tee:-pa] te k^hõ:=lo ts^he t^ha:. life=DEMPH 1PL=AGT NEG-cut-CIRC so 3PL=DAT life release '[Not killing life], we (are to) release them (=sentient beings) alive.' (YR canteen video)
 - b) ה׳ מקה׳ מוֹשָׁרְיאִי מוֹשָׁרְיאִי הַקַּמְאִיק׳
 t'a bunt^hɛ̃: tɕi? [ma-pʰɛ-pa] zu: ná. now moment one NEG-budge-CIRC sit.HON TAG.ASR
 'Now sit a while [without budging], will you.' (rnam-rtog 28)
 - c) हे. ट.क्यां गीया पट्टी केंगा पट्टी जेया पहेंबारी गीया कार्युयात्रा खेटाझेंट वा झूँट रा ट.क्यु. क्रुया पट्टी क्रेया पट्टी क्रेया पट्टी क्रेया पट्टी क्रेया पट्टी क्रेया पट्टी क्रेया पट्टी क्र . लु. नृत्त गर्वेत र्थे थे र्र्डु ग्र्या lèp dzukta? te nàtea=ki=di $[ts^hik=di$ ma-jà-wa] [cènã: so 1PL=AGT=DEMPH word=DEMPH very frightening NEG-do-CIRC anger $ts^{h}ik=di$ penta? penta?-p'ja сù-се? ma-lóː-pa] nàtci NEG-make.rise-CIRC 1PL.GEN word=DEMPH soft soft-ADVZR say.HUM-INF $z\phi m\phi m i = tsu = lo.$ other human=PL=DAT

'So we, [not making our words very frightening] (and) [not arousing anger], are to speak in soft, soft words to other people.' (YR canteen video)

d) לִישִּה׳ מִלְיָשָּׁ הִשְּׁשִׁ בּרִיבְמִי מִימִשְׁ מִימִשָּׁ מִימִשָּׁ מִימִשְׁ מִימִשָּׁ מִימִשָּׁ מִימִשָּׁ מִשּׁיצֿי שּׁשּן t'izã: tam=di=tsu t'ato=sã: [nê: k^ha=lɛ ma-t^hom-ba]⁴³⁵ but word=DEMPH=PL now=until 1SG.GEN mouth=ABL NEG-come.out-CIRC sém=na=rã: ry:-di lù:-po í:. mind=LOC=AEMPH rot-NF remain-2INF EQU.PER 'But [without getting out of (my) mouth] these words have until now remained rotting in my mind.' (Richhi 143)

As already pointed out in §15.2, an alternative to the simple negation (15.144) is the periphrastic nonfinal construction (15.145).

- (15.144) $ax^{i}a'' ag a' a' ag a'' a'''$ [mà:=lo te^hu**ma-lúk-a**]butter=DAT water NEG-pour-CIRC'[without pouring water into the butter]' (KN e)
- (15.145) $ax^{i} \overline{a}; \overline{a}; \overline{a}; \overline{a}; \overline{a}; \overline{a}; \overline{b};$ $[m\dot{a}:=lo tc^{h}u ma-l\acute{u}k-o p'ja-ti]$ butter=DAT water NEG-pour-2INF do-NF '[without pouring water into the butter]' (LA birth in Lachung)

Converb marked with *-pa/ba* may also express how much time is left until something happens, see (15.146). The gloss purposive fits this future-oriented context better than circumstantial.

The negated -pa/ba may express how much time has passed since something happened, see (15.147). This form also functions as the negated functional equivalent of the affirmative nonfinal construction (VERB-*ti*), see (15.8) above.

(15.147) مَلَّمَّ سَلَمْ عَلَيْ عَلَيْ عَلَيْ مَعَ عَلَيْ مَعْ عَلَيْ عَلَى عَلَيْ عَلَيْ عَلَى عَلَيْ عَلَى عَلَيْ عَلَيْ عَلَى عَلَيْ عَلَيْ عَلَيْ عَلَى عَلَيْ عَلَى عَلَى عَلَيْ عَلَيْ عَلَى عَلَيْ عَلَيْ عَلَيْ عَلَى عَلَيْ عَلَى عَلَى عَلَيْ عَلَى عَلَيْ عَلَى عَلَيْ عَلَى عَلَيْ عَلَيْ عَلَى عَلَيْ عَلَيْ عَلَيْ عَلَى عَلَيْ عَلَى الْكَلَيْ عَلَى عَلَيْ عَلَى عَلَيْ عَلَيْ عَلَى عَلَيْ عَلَى عَلَيْ عَلَى عَلَيْ عَ [mit's today (being) more than four months [since she wrote a letter to Karma].' (Richhi 161)

⁴³⁵ Here and elsewhere it is not always clear whether certain words, like $t'ato=s\tilde{a}$: here, belong to the subordinate clause or are part of the main clause.

In clock-times, -pa/ba expresses how many minutes are left before the full hour, see (15.148), whereas the nonfinal form -ti/di, faithful to its tendency to mark anteriority, expresses how many minutes have gone since the last full hour, see (15.149).

- (15.149) العَنْقَرْبَ مِنْ عَرْبَعُ اللَّهُ: مَنْ عَرْبَعُ عَامَةً مَنْ عَرْبَعُ عَامَةً اللَّهُ اللَّهُ عَلَيْهُ عَلَيْهُ عَلَيْهُ عَلَيْهُ اللَّهُ عَلَيْهُ اللَّهُ عَلَيْهُ عَلَيْهُ عَلَيْهُ عَلَيْهُ عَلَيْهُ عَلَيْهُ عَلَيْهُ عَلَيْهُ عَلَيْهُ اللَّهُ عَلَيْهُ عَلَيْهُ عَلَيْهُ عَلَيْهُ عَلَيْهُ عَلَيْهُ عَلَيْهُ عَلَيْهُ عَلَيْهُ عَلَيْ عَلَيْهُ عَلَيْهُ عَلَيْهُ عَلَيْهُ عَلَيْهُ عَلَيْهُ عَلَيْهُ عَلَيْهُ عَلَيْهُ عَلَيْ عَلَيْهُ عَلَيْهُ عَلَيْهُ اللَّهُ عَلَيْ عَ اللَّهُ اللَّهُ اللَّهُ عَلَيْ عَلَ اللَّهُ اللَّهُ اللَّا عَلَيْ عَلَيْ

The circumstantial marker occurs both with the equative \hat{i} and the existential $j\partial \hat{i}$. For examples, refer to (15.150) and (15.151) respectively. The use of of *-pa* especially in (15.150a) deserves further research.

 a) מיפרי יוסידי תברידייי אי אסיר, דאידי? (15.150)ázã: kantca bjõ:-bo=lo=s maternal.uncle youngest.male(Nep.) disappear-2INF=REP=QUO $t^{h}o$:-tca?. *ím-pa*, EOU-CIRC hear-PST.PO 'Uncle is said to have disappeared, did you hear?' (translation tentative) (TB phone call) NEG.EQU.PER recently grandfather=AGT king=DEMPH=DAT a.lot incantation mènni ódi: p^hen-do-gam. nã:-ze. [*mèm-ba*] perhaps that.AGT help-IPFV-ATTQ NEG.EQU-CIRC do.HON-PST p^hen-do-gam? mέn di: medicine this.AGT help-IPFV-ATTQ 'No, the grandfather just made many incantations to the (spirit) king. I wonder if that would perhaps help. [In case not], I wonder whether this medicine will help.' (rnam-rtog 31)

(15.151) a) $_{a}$ קר; $_{a}$ ና ຊ ເຊັ່ງເງິງ ເລືອດ ເລສັດ ເລຊັ ເຊັ່ງເຊີ ເພັດ ເລສັດ ເລຊັ ເຊັ່ງ ເຊິ່ງ ເຊິ່

b) אַ אָראיאומק אָרְיזי באי אַ אָראיזי אָק [*là: kjap-kʰɛ̃: mè:-pa*] *yá: là: kjap-o bɛ?.* field do-NMLZ NEG.EX-CIRC I.AGT field do-2INF EQU.NE '[As there was no plougher], I ploughed (the fields).' (DB life story)

In a 12th class text book called $\operatorname{asym} \operatorname{psym} \operatorname{psym} \operatorname{bras-ljongs} gsung-gtam$, the circumstantial converb is used somewhat differently from the description given above and attested by my oral and written data elsewhere. Whereas data from elsewhere has a rough distinction between negated circumstantial uses and affirmed purposive uses (with motion verbs), *'bras-ljongs gsung-gtam* has plenty of affirmative circumstantial uses and a different strategy for purposive uses. Instead of the converbal form, the purposive uses in *'bras-ljongs gsung-gtam* drop the converb marker *-pa/ba* and attach the dative-locative =lo directly to the verb root, e.g. $\operatorname{sigm} \operatorname{sigm} \operatorname{sigm$

(15.152)้ ตั้าคนๆ บลก ณ แก้ง เป็นเมือง เป็นเมือง เมือง เมือ *ǿ:la? za:mi* ní:=gi [k^himts^hE biu=tci? [ep-a] kipu arrive-CIRC enjoyment crow householder two=AGT neighbor snake=INDF man-dzun-run p'ja-tap k'an=jã: ma-t^hop-o bɛ?. NEG-become-CONC do-means what=even NEG-find-2INF EOU.NE '[As a snake arrived (to live) as a neigbour], although the crow couple was not happy, (they) could not find means to interfere anyhow.' (Dras-ljongs gsung-gtam 13)

In addition to dependent uses listed above, -pa/ba sometimes occurs as the final verb form, see (15.153-158). The glosses are tentative (and hence accompanied by a question mark), because the semantics of these constructions need further study. In the context of (15.153), six people are going to be divided into three pairs for working. One of the six says to the one who is responsible for dividing the pairs:

(15.153)	कुव्य'अर्ळव्' ८८' gjalts ^h en PN 'Gyaltshen	∽ અલ઼અ જેવા વાયેવ t'ã: ŋà n and 1SG t and I (are to)	' ক্রন্থার্থা aámtei? ogether work tog	<i>jó? kjap- work do-Cl ether.' (?) (k</i>	<i>a.</i> IRC XN e)	
(15.154)	ాషే జుక్ సాష్ $j\dot{a}=ni$ 1SG=TOP $n\dot{o}:-ti$ think-NF 'As for me (=you) wor	^{5[.]} ¶' ^æ f' ఇ∹ ^ష ' ఫ్రే ádzo grandfather ta-ta-wa . watch-RDP-C e, I have been uld come.' (?)	র শ্বম অর্ই স্ট্র' <i>t'ariŋ</i> today PIRC looking (rnam-rto	ଙ୍ଗ বিশ্ব <i>k'adzø?</i> how.much and looking og 1)	<i>p'a:=lo</i> interval=DAT , thinking at w	<i>tc'øŋ-gam</i> come.HON-ATTQ hat time the grandfather

- (15.156)में मिंग्रे कि रायरांग मरुर हो के के राय के कि कि के राय क का राय के राय क का राय के राय का राय के राय का राय के ताय के राय k^hoi nà p^hou simk^harka tca:-ni nóː-wa tε 1SG over.there TPN think-CIRC then where(Nep.) come.HUM-INF $dz\varepsilon$: min- $d\varepsilon$: hou at.all NEG-have.time EXCLAM 'I've been thinking to come to Simkharka, but how, I don't have time at all, eh.' (?) (KT discussion)
- (15.157)dɛndzõ:=lɛ ágja lòk-ti ágja ηà lep elder.brother elder.brother 1sg Sikkim=ABL return-NF arrive lõ mè:-po jìgi gu: gu:-pa. have.time.to EX-2INF letter wait wait-CIRC 'Brother, brother, as soon as (or: since) I arrived back from Sikkim, I have been waiting and waiting a letter.' (?) (Richhi 146)

Consultant KN commented that the form $j\partial pa$, as used in (15.159b), cannot be found in books, suggesting that clause-final *-pa* is mainly an oral construction.

(15.159) a) স্কুব'ক্তুম' দ'র্দ্র' দশ্যন'র্দ্রব্'স্তুদ'ব' অঁদ'র্দ্ব? *lɛŋgɛ? t'ato kalimpoŋ=na jò-po?* PRN.HON now TPN=LOC EX.PER-2INF 'Are you now is Kalimpong?'

> b) ה׳ ק׳אָּי אַזאָג׳אָאָד׳אָד׳ אַק׳אַן yà t'ato kalimpoŋ=na=ra jò:-pa. ISG now TPN=LOC=AEMPH EX-CIRC 'I'm now indeed in Kalimpong.' (?) (KN e)

The finite-looking uses of -pa need more investigation.

⁴³⁶ KN's Nepali translation was *maile euta sentence bana-e-ko chu ta* [1SG.ERG one sentence build-PFV-NMLZ EX.NPST.1SG CEMPH].

15.8.2 Circumstantial construction with postposition *nàyca/nàylo* 'inside'

The postposition *nàŋca/nàŋlo/nàŋlɛ* preceded by a genitivized infinitive can be used for forming circumstantial adverbial clauses roughly corresponding to the English clauses with 'as', (e.g. *As they were drinking tea...*). The novel Richhi has twelve examples of this construction and in all of them the nominalized verb is reduplicated.

- (15.161) مَجْ حَدَّ فَلَّا سَمَّا مَعْ عَلَى اللَّٰذِي اللَّٰذِي اللَّٰذِي اللَّٰذِي اللَّٰذِي اللَّٰذِي اللَّٰذِي الْحَدَةُ الْحَ الْحَدَافَةُ الْحَدَةُ الَحَاجُاءُ الْحَدَةُ الْحَدَةُ الْحَدَةُ الْحَاجُ الْحَاجُ

15.8.3 Circumstantial/manner use of progressive -tcɛ̃:/zɛ̃:/zin

The progressive marker $-tc\tilde{\epsilon}:/z\tilde{\epsilon}:/zin$, which forms finite constructions with existential auxiliaries, may also be used without the final auxiliary as a dependent adverbial clause marker. These uses mark a manner or attendant circumstance for the following main verb. In this use, the progressive may occur alone (15.162) or with dative-locative (15.163) or ablative case-marking (15.164):

- (15.163) $\exists a_1 = 0$ (16.164) $\exists a_2 = 0$ (16.164) (

The progressive frequently co-occurs with verbs of motion (especially gju 'go'), providing a manner or attendant circumstance of literal going⁴³⁸, as in (15.165), or metaphorical going, that is, changing, as in (15.166).

- (15.165) هام العربية المحافظ المحافظ

Repetition of the progressive emphasizes duration, as shown by (15.164) above and (15.167) below.

(15.167) ยังส สูงรุง ราติส สูง คยูงรัติ พุรุง
 [k jo? kjap-zɛ̃: kjap-zɛ̃:] gju-wø: kap chatting do-PROG do-PROG go-2INF.GEN time 'When (they) go on chatting and chatting...' (Richhi 130)

The circumstantial/manner use of the progressive is negated by the prefix ma-.

15.8.4 Circumstantial clauses with *k*^{*h*}*a*=*lo*

Circumstantial clauses can also be formed using the dative-locative form of the word k^ha 'mouth', preceded by genitivized *-po*-infinitive. The meaning is close to both simultaneous and causal clauses, as suggested by the gloss 'as/when/since'.

⁴³⁸ For a similar use of the nonfinal -ti/di, consider examples (15.3-6) above.

⁴³⁹ ³³⁷ *bas* [bɛ:] is a Central Tibetan clause connector which does not occur in my spoken data.

(15.169)[*nà*-ri=tsa: mèː-pøː $k^{h}a=lo$] ηά rã:=lo k'ate 1SG-AEMPH.GEN=at NEG.EX-2INF.GEN mouth=DAT I 2SG.M=DAT how p'in-ce bo? give-INF EQU.NE.Q '[As/when/since I do not have myself], how could I give to you?' (KN e) (15.170)८ २ में कु र्वेर पारे केन प्रति पार्ये के र्रे पानेव क्व श्वन्या [nà=lo gjunor k'are mè:-pø: $k^{h}a=lo$] pù:pents^hẽ: 1SG=DAT riches any NEG.EX-2INF.GEN mouth=DAT relatives pã:. abandon '[As/when/since I didn't have any riches], (my) relatives abandoned (me). (Class 7

Further research is needed in order to find out whether this use of $k^h a = lo$ is limited to collocating with the negated existential $m\dot{e}^2$, which occurs in both (15.169) and (15.170).

The dative-locative form $k^h a = lo$ also marks additive meaning (§15.9.3), and the suffix $-k^h a$ is used in locative adverbs (§3.5.2.2).

15.8.5 Comparative manner

textbook 10)

Comparative manner can be expressed with a nominalized construction involving one of the following words: demonstrative pro-adverb $d\epsilon m$ 'like (it)' (§15.8.5.1) or one of the postpositions $n a \eta tar(gi)$ 'according to', $n a \eta z in$ 'according to, similar to' (§15.8.5.2) or $t' \theta n z in(gi)$ 'in accordance with' (§15.8.5.3).

15.8.5.1 Comparative manner with *dɛm*

The verb preceding $d\varepsilon m$ 'like (it)' is typically a *-po*-infinitive (without genetivization), see (15.171) and (15.172), but the imperfective form also occurs in this position, see (15.173).

(15.171)	ર્લે લેનું અવે સું સ્ટ- ર્રેન્ટ ગ્રીયા બનાર્યા બને સામ બેન કે કે કે સું દ્વાયા જ્ઞેની									
	ódi né:mu=ra	[tc ^h ø:=ki	làp- o	dɛm]	lèpti	kjota?	bɛ?.			
	that really=AEMPH	2SG.L=AGT	say-2INF	like.it	very.much	sad	EQU.NE			
	'That really is, [like	you said], v	very sad.' (ng	ga'i 'gan	25)		-			
(15.172)	८.जू.मू. बह्रू. र. व.बह्र. र. पर्यं त्वें . तंवें. ही. पंकी. हा.									
	ŋà=lo=to	t ^h õ:-ruŋ	[ma-tho:-po	dem]	zu:-di	gju:				
	1SG=DAT=CEMPH	see-CONC	NEG-see-2IN	NF like.it	avoid-NI	go.21	NF			
	bɛ?.									
	EQU.NE									
	'Although (he) saw me, (he) walked avoiding me[, as if not seeing (me)]' (nga'i 'gan									
	Richhi 23)									

(15.173) $\check{\alpha}_{\Sigma'} : \check{\pi}_{\Sigma'} : \check{\pi}_{\Sigma'} : \check{\alpha}_{\Sigma'} : \check{\alpha}_{\check'} : \check{\alpha} : \check{\alpha}_{\check'} : \check{\alpha}_{\check'} : \check{\alpha}_{\check'} : \check{\alpha}_{\check'} :$

In line with its function as a proadverb of manner, $d\varepsilon m$ may be followed by manner adverbializer p'ja(ti):

(15.174)ราวิกา พาา ผริมา ลาวา ผริมา อู่พาลา พาา พลารทัพาวา ลา сù-**bo** [t'arin làp dɛm, dem p'ja-ti] làp today say like.it say.HUM-2INF like.it do-NF say man-go:-b-be?. NEG-be.needed-2INF-EQU.NE 'It was not necessary to mention (it) [as we speak (it), as we say (it) today].' (NT BLA 6)

15.8.5.2 Comparative manner with nàŋtar(gi)/nàŋzin

- (15.176) אַזיאי קאָר־דֿי קר־יקאָק־ [*láma: súm-bo nàŋzin*] lama.AGT say.HON-2INF according.to '[According to what the lama had said]...' (mi-la ras-po 6)
- क्षेत्र क्रुया ग्रीया मेंया ये. वेटा क्षेत्र टा रेटा वा मठता में खेता (15.177)[lenge:=ki p'o:-po nàŋtar] t'ariŋ nà: tca:-bo PRN.HON=AGT invite-2INF according.to today here come.HUM-2INF ĩ. EOU.PER 'I came here today [on the basis of you having invited (me)].' (KN e)

15.8.5.3 Comparative manner with *t'ønzin(gi)*

A clause headed by the postposition t' onzin(gi) 'in accordance with' can be postposed to the verb root (15.178), bare *-po*-infinitive (15.179), genitivized *-po*-infinitive (15.180) or a correlative clause (15.181).

קיארייםי שליואייתקמיק ישיוריאיי בייוארי איי אריארי אייאאיי בייארי איאאייארי איאאייארי איאאייארי איאאייארי אייא (15.178)न्वित्रा गवन्तान्त्रा सुगत्राहे के विमन्त क्षेत्रा tha:tee? nấ: $[t'arin=gi ts^{h}o:dy:=na]$ t'ønzin] kamø today=GEN meeting.GEN=LOC discussion decision do.HON in.accordance.with $t^{h}amtee = ki r \tilde{a}: r \tilde{a}: s \delta: s \sigma:$ t^hu:gen 6È. ze: all=AGT each.oneself.GEN responsibility.HON mouth.HON eat.HON nấː-bɛː $t^h u dz i t c^h \varepsilon$ $c \dot{u}$ - $c \varepsilon$ ĩ. do-2INF.AGT thank.you say.HUM-INF EQU.PER 'I offer thanks because all (of you) have taken each (your) own responsibility [in accordance with the decision we took in today's meeting.]' (Richhi 51)

- (15.179) اَجَ هَٰ مَحْتُ جَمَ عَرَامَمَ عَنَّامَ جَ حَرَّصَ الْحَرَّاتِ حَرَّاتٍ حَرَّاتٍ حَرَّاتٍ حَرَّاتٍ حَرَّاتٍ حَرَّاتٍ حَرَّاتٍ عَرَّاتٍ عَرَّاتٍ عَرَّاتٍ عَلَيْهِ عَرَّاتٍ عَلَيْهِ اللَّهُ عَلَيْهُ عَلَيْهُ اللَّهُ عَلَيْهُ عَلَيْهُ عَلَيْهُ اللَّهُ عَلَيْهُ عَلَيْهُ عَلَيْهُ عَلَيْهُ عَلَيْهُ عَلَيْهُ عَلَيْهُ عَلَيْ اللَّهُ عَلَيْ الْحَ اللَّاللَّهُ اللَّالَةُ اللَّهُ عَلَيْهُ عَلَيْهُ عَلَيْ اللَّالَةُ اللَّهُ عَلَيْ اللَّالَةُ اللَّالَةُ اللَّهُ عَلَيْ اللَّا عَلَيْكَ اللَّا عَلَيْ اللَّهُ عَلَيْكَ اللَّهُ عَلَيْ اللَّهُ عَلَيْ الْحَالَةُ عَلَيْ الْحَالَةُ عَلَيْ اللَّا عَلَيْ عَلَيْ عَلَيْ اللَّالِ اللَّهُ عَلَيْ اللَّا عَلَيْ اللَّهُ عَلَيْ اللَّهُ عَلَيْ اللَّا عَلَيْ اللَّا عَلَيْ اللَّا عَلَيْ الللَّا عَلَيْ الللَّا عَلَيْ اللَ اللَّا عَلَيْ اللَّعَالَيْ اللَّا عَلَيْ اللَّهُ عَلَيْ اللَّالِ اللَّا عَلَيْ اللَّا عَلَيْ الْحَالِي الْحَالِ عَلَيْ الْحَالِي اللَّا عَلَيْ اللَّا عَلَيْ عَلَيْ الللَّالَةُ عَلَيْ اللَّا عَلَيْ عَلَيْ الْعَلَيْ الْعَالِي اللَّ اللَّا عَلَيْ اللَّا عَلَيْ اللَّا عَلَيْ اللَّهُ عَلَيْ اللَّهُ عَلَيْ اللَّا عَلَيْ اللَّا عَلَيْ الْعَلَيْ الْ اللَّا عَلَيْ اللَّالَةُ عَلَيْ اللَّهُ عَلَيْ اللَّالَ اللَّ عَلَيْ اللَّا عَلَيْ اللَّالَةُ اللَّا عَلَيْ الْعَلَيْ ا اللَّا عَلَيْ اللَا عَلَيْ اللَّالَةُ عَلَيْ اللَّالَةُ عَلَيْ اللَح
- (15.180) [tirestimatesidestima
- रा'रा' रेक'र्ये'के'गीषा यात्र' यासुन्यासुन्ये' सन्य' संयदने' मेंत्र'यवित्र'ग्रीषा सुयाषा याषेत्र' होगी' अया'याय यात्रत' हे हा (15.181)[guru rimput $c^h \varepsilon = gi$ k'an sún-sum-bo j*à-po* ódi guru Rimpoche=AGT what say.HON-RDP-2INF EX-2INF that tcha:le? nấ:-do t'ønzingi] t^hu:sam ze:-tiki *bε*?. in.accordance.with thought.HON take.HON work.HON do.HON-IPFV EQU.NE '(The king) used to work thinking [in accordance with what Guru Rimpoche had said].' (CY interview)

A postpositional phrase headed by *t'ønzin* may be supplemented by the adverbializer -p'ja(ti), which forms adverbials of manner (see §3.5.2.1):

(15.182) مَجْ صِمَّ عِنْ مَعْمَّ مِعْتَ مَسَمَّ مَعْتَ مَعْت then girl=too mother.AGT what order give.HON-2INF.GEN
t 'onzin]-p'ja dø:-po bɛ?.
according.to-ADVZR stay-2INF EQU.NE
'Then the daughter lived [in a manner that accorded with what the mother had
ordered].' ('dres-ljongs gsung-gtam, class 12, 39)

15.8.6 Genitivized -po-infinitive

A genitivized *-po*-infinitive may, although infrequently, form an adverbial clause of circumstance/manner:

15.9 Additive clauses

Affirmative additive clauses can be formed by postposing to a *-po*-infinitive form one of the forms $mits^{h}\varepsilon^{2}$ 'in addition, not stopping' (§15.9.1), $t\varepsilon\eta lo$ 'on, above' (§15.9.2) or $k^{h}a=lo$ which literally means 'at the mouth' (§15.9.3). The first one is the most frequent, while I am aware of only one instance of each of the latter two. In addition to the three affirmative constructions, negated additive clauses may be formed by $m \varepsilon mbo$, which is followed by a negated or an interrogated clause (§15.9.4).

15.9.1 Additive with *mits^hε*?

Additive clauses can be formed by postposing $mits^{h}\varepsilon^{2}$ 'in addition, not stopping' to the bare *-po*-infinitive form of the verb. The additive clause precedes the main clause:

(15.184)dau tsyndy: mam-bja-ne $[t' on=j\tilde{a}]$: di min-dup-o *mi-ts^h*e?] this like effort NEG-do-COND purpose=too NEG-fulfil-2INF NEG-stop bompu mjoz-ce duŋŋal ĩ. experience-INF EQU.PER difficulty big 'If (we) do not make effort like that [it is not only that (our) purpose will not be fulfilled] (but) that (we) shall go through big trouble.' / 'If (we) do not make effort like that (our) purpose will not be fulfilled and, moreover, (we) shall also go through big trouble.' (Class 7 textbook 5)

The form mi-ts^h ε ? also occurs as a looser cohesive marker 'moreover, furthermore' following a finite clause, see §12.2.

15.9.2 Additive with *teylo*

The relator noun *teylo* 'on, upon, above' can mark additive clauses:

(15.185)พात मन्यता पर्ने येपन्ने वेंग्म्यान्ने सुन्ध्यान्याया वेंत येंन पेंन रों न हेंग्रम मेय पर्वता येंन परि क्रेन्ये क्रायने मेन्या ána ge:m=di [*l*èpti lò ge:-ti ùdu:ta? old.lady old.lady=DEMPH year become.old-NF out.of.breath very.much *t*^høn j*à-po* t'ato=rã: сí-rap dem j**ò-pø**: tenlo] now=AEMPH die-IMF like.it EX-2INF.GEN become EX-2INF upon na=di rinku nose=DEMPH long 'The old lady, [in addition to having become out of breath with age (and) being as if about to die right at that moment], (had) a long nose...' (rna-gsung 5)

15.9.3 Additive with $k^{h}a=lo$

In addition to circumstance (see §15.8.4), $k^ha=lo$ 'mouth=DAT' may express additive meanings:

(15.186)רמי קיארי איפי מרי ביקאי לי מיאי לי שאי (beer) לי הדי לי תפרילי שאן [lè:de: biər ná: t'arin=to k'ã: sà-wø: k^ha=lo] I.AGT today=CEMPH plate full eat-2INF.GEN mouth=DAT beer(Eng) t^huŋ-bo ĩ. botr do two drink-2INF EQU.PER bottle 'Today, [in addition to eating a full plate], I drank two bottles of beer.' (KN e)

15.9.4 Negated additive with *mèmbo*

The form m & mbo, which is a somewhat grammaticalized form of the negated interrogative equative m & m & mbo [NEG.EQU.NE.Q], can form an additive construction, if followed by a negated or (rhetorical) interrogated main clause. When followed by a negated declarative clause, as in (15.187), m & mbo forms a type of negated additive, telling that the verbal action of the main clause did/does/will not happen in addition to the action in the subordinate first clause.

(15.187)ने'=
के'
के'</p $[k^h a = l\varepsilon]$ ť izã: ódi kap=na mèmbo] t'i-ce=ki kjap-se? but that time=LOC mouth=ABL except write-INF=GEN do-INF mèbbe?. lùksø: tradition NEG.EX.NE 'But at that time, [in addition to spoken language], there was no tradition of writing.' (KL BLA 12)

When the main clause is a rhetorical question, as in (15.188), the assumed answer is negative, i.e. the action in the main interrogated clause will not happen in addition to the action in the dependent clause marked with *mèmbo*.

(15.188)ขุโรม รสารัง รรง กรุ่มาก สัญาญา พลารัง ขุดสา ซีราชิม พิเมิร์ เพิ่ม เป็นเป็น เป็นเป็น เป็นเป็น เป็นเป็น เป็นเป็น t'izã: dolma dɛmø: p'um=di=lo sønkor t'ã: gjapkor p'ja-ti but PN like.GEN girl=DEMPH=DAT care and support do-NF $t^{h}a:p^{h}u=na [te^{h}o:=ri$ rangi za:da zo-ti tenk^ha mìntam own=GEN spouse make-NF end=LOC 2sg.l=aemph.gen above fame pèmpo t'ã: p^hjark^ha $t^{h}o:-\varepsilon \epsilon^{2}$ **mèmbo**] $z \epsilon n$ $t \epsilon^{h} \sigma:=ki$ mìsisikø: hear-INF except other 2SG.L=GEN ridiculous.GEN bad and disgrace bompu k'an t^hop? mìntam reputation big what receive 'But having cared for and supported a girl like Dolma and having made her your spouse, what other grandly ridiculous reputation can you get, [except for (or: in addition to) hearing at the end disrepute and disgrace (directed) at yourself]?' (nga'i 'gan 4)

15.10 Substitutive clauses with *tshaplo*

Substitutive clauses are formed with the help of the relator noun *tshaplo/tshamalo* 'in place of', which is appended to a genetivized infinitive form:

(15.189) التابعة المحافظ الم

15.11 Comparative clauses with $=l\varepsilon$ (*lako*=EMPH)

Comparative clauses can be formed by attaching the ablative $=l\varepsilon$, optionally followed by *lako* 'more, (in) excess, rather', to a nominalized verb form. In (15.190) and (15.191), nominalization is accomplished by *-po*-infinitive and the demonstrative-emphatic =di respectively. The word *lako* may be followed by an emphatic, see =to in (15.190b) and =di in (15.191).

(15.190) a) المجترَّبَ عَبَرَ تَنْ سَعَنَ عَلَى اللَّهُ عَلَى اللَّعْنَا اللَّهُ عَلَى اللَّالَ اللَّهُ عَلَى اللَّعَلَى اللَّهُ عَلَى اللَّهُ عَلَى اللَّهُ عَلَى الْعُلَى الْحَلَى اللَّ

b) ਗ਼ੁਕਾ ਕੁਛੱਕ 'ਕੇਂ' ਤੁੱਕ 'ਬੇਂ' ਕੁषा 'ਸੱ' 'ਸੋ ਗੋ ਕਾ ਕੇਂ' ਤੁੱਕ 'ਬੇਂ' ਤੁੱਕ 'ਬੇਂ' ਤੁੱਕ 'ਬੇਂ' ਤੁੱਕ

 $[gjalts^{h} \varepsilon n = lo \ p'im-bo = l\varepsilon]$ lako=to $karma=lo \ p'im-bo$ t'ak.PN=DATgive-2INF=ABLmore=CEMPHPN=DATgive-2INFbe.better'It is better rather to give to Karma [than to give to Gyaltshen].' (KN e)

(15.191)nàt⊊a=di $[k^h on = gi$ ť'i-u $n\tilde{a}$:-sonz \tilde{a} :=di=l ε] 1PL=DEMPH 3SG.HON=AGT write-2INF do.HON-TERM=DEMPH=ABL *lako=di* min-go? $b\epsilon$?. be.needed EOU.NE excess=DEMPH 'We do not need (anything) more [than the extent that he has written].' (KL discussion with DR)

For uses of the comparative construction $=l\varepsilon$ (*lako*) with nouns, see §5.6.1.3.2.

15.12 Various functions of the terminative converbs -sã: and -sonzã:

The terminative converb has two markers, $-s\tilde{a}$: and $-sonz\tilde{a}$:, which both probably derive form the postposition $=s\tilde{a}:(t\varepsilon) \equiv (\overline{s})$ 'until'. The longer form $-sonz\tilde{a}$: is most probably a historical combination of the secondary verb $s\tilde{a}$: (WD \overline{s} song) and the postposition $=s\tilde{a}$. The general meaning of both converbs may be approximated by the English expression *as far as*, although, as will be seen, the forms have considerable interpretive freedom, expressing meanings such as terminative, simultaneous and causal.

The terminative use, which most resembles the postpositional use of $=s\tilde{a}$: with nominals, (i.e. clearly expresses the semantics of 'until') is illustrated as negated in (15.192) and affirmed in (15.193).

- (15.192) ה׳ הְיאָה׳ אָשְׁי אִימָה־שִּה׳ אָשָׁה׳ אָשָּה׳ הַישָּה׳ אָשָׁה׳ (15.192) ה׳ הַיאָרָי אָשָׁר אַימָרִישָּה׳ אָשָּהָי אָשָּה׳ (15.192) ה׳ הַיאָר אַימָרישָר׳ אָשָּר אָשָּהי (100 אושר 100 אושר 1000 אושר 100 אושר

In (15.192), the speaker is waiting for something that has not yet happened. Therefore the verb with $=s\tilde{a}$: occurs as negated. In (15.193), on the other hand, the speaker is waiting for an ongoing activity to stop from happening and therefore an affirmed, reduplicated verb form is used.

Frequently, $=s\tilde{a}$: expresses simultaneity:

(15.194) rightarrow interest in the matrix <math>interest in the matrix interest in the matrix <math>interest in the matrix interest in the matrix <math>interest in the matrix interest in the matrix interest in the matrix <math>interest in the matrix interest interest interest interest in the matrix interest intere
- (15.197)ਸੇ' मारुषा' गुन्ध र्छन' द्वीतु याँक' (culture program) पर्मेक 'यो' जन्म नाम मारेका के के सिंह (stage) के रायता [nàtea? kaltea prougram làp-sã:] tε tøn-ge culture(Eng.) program(Eng.) take.out-HORT say=TERM then 1PL nàt£a=lo steidz mìndu?. 1PL=DAT stage(Eng.) NEG.EX.SEN 'Then [when we say let's hold a cultural program], we do not have a stage.' (NAB BLA 7)
- (15.198)यरर्त्त्र हेगा द्यीर हे सुवार केंर मुक्ते वाद द्वार ही कार्य राष्ट्र ही कार्य राष्ट्र के केंद्र हो के केंद्र हो $[p^ha]$ ts^hu: mìk gir-di ta-sã:] nòrbu=to làm cùk-ti eye stare-NF thither hither look=TERM PN=CEMPH road enter-NF k^him no:=lo thõ:-ti gju: direction=DAT go. 2INF see-NF house '[When/as (she) looks intently here and there], (she) sees Norbu going along the road toward (his) home, and...' (Richhi 98)⁴⁴⁰

The terminative may be followed by the conditional marker:

A verbal construction with $=s\tilde{a}$: can also express reason:

⁴⁴⁰ The construction $ta = s\tilde{a}$: is analogous to Nepali *her-e-samma* [look-PFV-until] 'when looking, as far as one can see'.

⁴⁴¹ The speaker KN himself volunteered a translation into Nepali with the simultaneous construction *her-da-k^heri*.

(15.201)รพา ริจา ผริา สู้ขา ครา ผข้า จรูขาสรา รณิฑา รัขพา ผจาลิา สู้ขา พาล์จา $[\eta \dot{a}: t \in p = di$ dok-ce? tsuk-sã:] nè:=gi go I.AGT book=DEMPH read-INF beginning insert-TERM 1SG.GEN=GEN ro? lep-ti dok ma-t^hop. friend arrive-NF read NEG-receive '[Because I had started to read the book], my friend, having arrived, could not read (it).' (KN e)

With verbs of speaking, the meaning may approximate 'according to, as far as (someone) says':

(15.202) $\mathbb{R}^{n} \mathbb{A} \mathbb{A}^{n} \mathbb{A}^$

In some contexts, the meaning may be concessive, as illustrated by (15.203), where the speaker offers an excuse for being late.

The longer form of the terminative is $-sonz\tilde{a}$:. Its use in examples (15.204) and (15.205) is akin to the English expression as far as (or 'according to'), which is clearly semantically related to the semantic content of the postposition $s\tilde{a}:(t\varepsilon)$ until. Example (15.204) occurs in the book sbar-phung ling-dam 'sgro-lis, where the form is written $\sin \sin s$ song tsang, suggesting an origin in WT $\sin s$ tsang 'because', although the clause does not express reason.

(15.204)	ລິ ພາພາກີສາ ພຣາສັ້ຮາອີຮາ ຍາຍີ່ສາຍິງ หຼາຍສາ ແລ້ ເພາຍາຊາ ຜີສາພັງ								
	[mí	làla=gi	làp-sonzã:]	pén=gi	ta:=di	pãːdɛ̃ː			
	human	some=AGT	say-TERM	marriage=GEN	sign=DEMPH	apron			
	$\tilde{\iota}=lo.$								
	EQU.PER	=REP							
	'[As far as some people say], the sign of marriage is the pangden apron, I hear.'/								
	'[Accore	ding to what	some people s	say], I've heard, th	ne sign of marr	iage is the pangden			
	apron.' ((sbar-phung	89)						

ने'चट क्रेंप्टी' प्रमायांक पटी' टया कुपाया अर्थेट केंटाचट पटेवायाया केवापटेवायाया हे' येप अर्थे वे' पटे पटेवाट्याया पटेवा (15.205)यांडिया' तत्या गो gjapma tho:-sonza:] den-gam *t'izã: ódi gagdza=di* [ŋáː but that obstacle=DEMPH I.AGT see-TERM be.true-ATTQ after dɛnta? min-den-gam te lèp nό:-nε=di dem=tei? NEG-be.true-ATTQ so much think-COND=DEMPH true like=INDF duke. EX.SEN-IN 'But that obstacle, [as far as I see it (now) afterwards], if considering carefully whether it was true or not, it seemed to be true.' (CY interview)

For simultaneous/anterior use, consider (15.206), and for a clearer simultaneous use, see (15.207), where the differing pronunciation *-sumzã*: raises the question of different etymology from the typical form *-sonzã*:.

(15.206) र्दु'र्सेट ज्ञट हे' खुर्य अयर्थ गेंग

[t"i-sonz"a:] te $k^h \dot{u}:$ $l \dot{a} p - o = lo$ ask-TERMthen3SGM.AGTsay-2INF=REP'[As (the teacher) asked], he then said (so the story goes):' (RS pupil joke)

(15.207) מֹה־אָּהָשִישָּה מָשִּׁיחָן מָלְ הֵיהַפּׁמִיחָן מָלָ הֵיהַפּּמָ מָּהִיהַפּּמָיחָן [*òn-sumzã*:] tɛuku=di ŋù-zin òn-tɛuŋgɛ=lo. come-TERM other=DEMPH weep-PROG come-PROG.APH=REP '[When (they) come], the other one comes crying (so the story goes).' (RS pupil joke)

In (15.208) and (15.209), the construction expresses reason:

Both the short form $-s\tilde{a}$: and the long form $-sonz\tilde{a}$: occur in spoken language, although the long form is more frequent. The novel Richhi does not have any examples of the typical spoken

form -*sonzã*:, see (15.210), but instead uses the shorter form -*sã*:, see (15.211). The negation of the copular terminative *in-sonzã*: is $m\tilde{\tilde{\varepsilon}}$:-*sonzã*:.

- (15.210) สังละวัน สิ่ง สังนายน [ódi: **ín-sonzã:**] that.AGT EQU-TERM '[Because of that]...' (CY interview)
- (15.211) ਕਟ੍ਰੇਆ ਐਕ੍'ੜਨ' [*di: ín-zã:*] this.AGT EQU-TERM '[Because of this]... ' (Richhi 60)

In written language, the demonstrative in (15.211) is marked as an agentive/instrumental. I suspect that in spoken language, the contrast with the non-marked and agentive-marked demonstrative is non-existent, especially because the difference is only in vowel length, a feature which is contrastive but does not bear much functional load. For the time being, however, even the spoken form in (15.210) is here marked as agentive.

For an example of the terminative marker in a comparative clause, refer to example (15.191).

15.13 Summary remarks

This lengthy chapter described the variety of constructions used in forming adverbial clauses, which were organized under eleven main functional headings: nonfinal, temporal, causal, purposive, conditional, concessive, circumstantial, additive, substitutive, comparative and terminative. The constructions used in forming adverbial clauses were seen to include ten converbs, seven postpositional clauses, three noun-headed clauses and ten other constructions. Some of the markers, such as the terminative converb, were shown to occur in a variety of uses and thus to be highly context sensitive, whereas other markers, such as the concessive and conditional markers, were seen to have specialized uses. The circumstantial-purposive marker showed an interesting polarity-related division: affirmative uses tend to be purposive and negated uses circumstantial, although in one literary source these forms were seen to function differently. Simultaneity (expressed with seven constructions) and causality (expressed with eight constructions) were seen to be the functions which showed the greatest constructional variety, both leaving room for further research.

16 Discourse phenomena

This chapter addresses discourse and context related markers and phenomena, which typically make reference beyond the clause. For instance, the contrastive clitic =*to* suggests that a contrast is being made with another proposition/clause in discourse and the honorific =*la* suggests that the speaker has made an evaluation of their own social status in relation to the addressee in the non-linguistic context. The bulk of the discussion will focus on phrase-level emphatic clitics (\$16.1), clause-final clitics (\$16.2.), assertive and exclamative tags (\$16.3) and discourse particles (\$16.4). Briefer treatment is given to recapitulation (\$16.5), right-dislocation (\$16.6) and intonation in discourse (\$16.7).

16.1 Emphatic clitics

Denjongke has four emphatic clitics, which in various ways invite the addressee to pay attention to a certain constituent in the clause. Being aware of Payne's (1997: 345) claim that "focus" and "emphatic" "are probably the most overused and misused terms in linguistics," I aim to, at least, specify the distinguishing characteristics of the emphatics.

Table 16.1. Emphatic clitics

	r	
$=r\tilde{a}$:/ $=ra$	<u> २८</u> ./२.	anaphoric emphatic
=to	র্ট	contrastive emphatic
=di	جمح.	demonstrative-emphatic
=ni/nɛ	वे'	topicalizer-emphatic
=jã:	೮ಗ.	additive emphatic

As examplified in (16.1), emphatic clitics in a noun phrase occur after case-marking.

In the following sections, the emphatic clitics are discussed in the same order they occur in Table 16.1.

16.1.1 Anaphoric emphatic =*rã*:

The emphatic clitic $=r\tilde{a}$: (or $=ra^{442}$) can add general emphasis to almost any clausal element. The marker $=r\tilde{a}$: also has similar anaphoric potential as its etymon πr : rang '-self, same' in Classical Tibetan. Beyer (1992: 218) offers the following description of nominals followed by rang in Classical Tibetan: "These reflexives specify the scope of the nominal as limited to an entity or set of entities already stated or implied in the preceding text." In line with that description, Denjongke $=r\tilde{a}$: is often, although not exclusively, used in contexts where the referent to which $=r\tilde{a}$:

⁴⁴² In spoken language, the emphatic $=r\tilde{a}$: tends to become shortened and denasalized =ra, although consultant KN commented that both nasalized and non-nasalized forms are heard.

appended has already been mentioned in the discourse and has thus been activated in the speaker's mind. Topical continuity in the English translations below is often conveyed by 'indeed'. Because of its anaphoric potential, $=r\tilde{a}$: sits naturally together with anaphoric demonstratives describing referents that are already given, e.g. $\dot{o}di=ra$ 'that indeed', $\dot{o}d\epsilon m=ra$ 'indeed like that'. Example (16.2) shows a use of the emphatic, where the referent to which the emphatic is attached has already been mentioned. A group of people are discussing where a certain ceremony is going to be held. One person has suggested Varanasi but person B has objected. Then, person A in (16.2) suggests the right location, Dorjeden, to which B concurs and marks the location, now the new topic of discussion, with the emphatic =ra.

(16.2) A: جَ'تَحْسَ' جَ'تَجَ' سَاحَ مَ'سَا

t'utci? dordzid \tilde{e} :=lo. this.year TPN=DAT 'This year (it's) in Dorjeden.'

B: $\frac{1}{5} \cdot \frac{1}{5} \cdot$ this.year direct(Eng.) TPN=DAT=AEMPH AP.EQU.NE 'This year (it) indeed seems to be directly in Dorjeden.' (KN kitchen discussion)

In (16.3), the emphatic demonstrative adverb $\delta d\varepsilon = ra$ 'just like that' makes anaphoric reference to the previous clause in the same sentence:

(16.3)

za: go?. now earlier what EQU-2INF like.that=AEMPH set be.needed 'What was before, needs to be preserved just as it is/was.' (CY interview)

As shown in §6, the form $\pi \sigma$ is also used as a second person singular pronoun and a reflexive marker. Example (16.4) shows that the emphatic $=r\tilde{a}$ is distinct from the reflexive 'self' in that both forms may occur in the same clause. The reflexive attaches to the pronoun before casemarking, whereas the emphatic is postposed to the case-marker. This time the meaning is hardly anaphoric, as in the two examples above, but generally emphatic. In the translation emphasis is suggested by 'all' in 'all by myself'.

त्र गव र्घ्र त्र्ग् - त्र त्र p'ja go? pa-ray=gi=rã: (16.4) t^hak - $\epsilon\epsilon$? I.AGT what do be.needed 1SG-REFL=AGT=AEMPH decide mi-ts^hu? NEG-be.able.to 'I cannot decide all by myself what I should do.' (Richhi 112)

⁴⁴³ The *a-chung* subscript *q* is in this word used to suggest that the pronunciation of the coda consonants breaks the conventional Tibetan spelling rules.

The emphatic $=r\tilde{a}$: attaches to a variety of word classes and constructions. The examples above already illustrated uses with a noun (16.2), demonstrative adverb (16.3) and pronoun (16.4). In the following three examples, $=r\tilde{a}$: attaches to an adjective (16.5), postposition (16.6) and a question word (16.7).

- (16.5) $\neg \tilde{\mathfrak{R}} = \neg \tilde{\mathfrak{R}} \tilde{\mathfrak{R}} = \tau \tilde{\mathfrak{R}} \tilde{\mathfrak{R}} \tilde{\mathfrak{R}} \tilde{\mathfrak{R}} = \tau \tilde{\mathfrak{R}} \tilde{\mathfrak{R}} \tilde{\mathfrak{R}} = \tau \tilde{\mathfrak{R}} \tilde{\mathfrak{R}} = \tau \tilde{\mathfrak{R}} \tilde{\mathfrak{R}} = \tau \tilde{\mathfrak{R}} \tilde{\mathfrak{R}} = \tau \tilde{\mathfrak{R}$
- (16.6) $\vec{r}_{i} = \vec{r}_{i} \cdot \vec{$
- (16.7) באיאַבי (level) קדידי גאַק־די געו ג'ar=**ra**

levl k'ar = ra $\tilde{t}:-run$ level(*Eng.*) what=AEMPH EQU-CONC 'whatever level (it is)' (KL BLA 12)

The emphatic $=r\tilde{a}$:/*ra* also attaches to numerals, as shown in (16.8-10). In (16.8), the numeral functions as a noun modifier. Example (16.9) illustrates an independent use of *tei2* 'one' with =ra, obtaining the meaning '(one and) the same'. In (16.10), the emphatic attaches to =tei2, which functions as an indefinite marker.

- (16.8) هَرَ مَحْ اللَّهُ مَعْ اللَّهُ مَعْ اللَّهُ اللَّ

The following examples exemplify the use of $=r\tilde{a}:/ra$ in conjunction with verbs. The emphatic occurs attached to $-c\varepsilon$ -infinitive marking a clausal complement (16.11), to the first part of the

phrasal verb *ha-ko* 'know' (16.12), to *-po*-infinitive in the periphrastic past construction (16.13) and directly to the verb root (16.14-15).

(16.11) בתי משיקדיאבי מלי שאמן ĩ. nè: làp-se=**rã**: di my say-INF=AEMPH this EQU.PER 'What I have to say is this:' (Nga'i 'gan 10) (16.12) كَانج سَ عَلَى مَعْدَر مَعْلَمَ مَعْدَر مَعْلَمَ مَعْدَ مَعْدَ مَعْدَ مَعْدَ مَعْدَ مَعْدَ مُ p^hiru=ki $t \epsilon^{h} u t s^{h} \theta^{2} g \epsilon^{2} d u \eta - p o$ ha=**rã**: man-go. night=GEN clock.time eight strike-2INF know⁴⁴⁴=AEMPH NEG-know '(They) did not even know that it was already eight o'clock..' (Richhi 14) (16.13) କ୍ଷୁଣ କ୍ରୁଷ କ୍ଷ୍ୟୁମ୍ପିଷ ଆସ୍ତିଣ ସାହି ଦେବା ମ୍ବା କ୍ଷୁମ, ଆସ୍ତିଣ ଆସ୍ତିଣ ସିଂନ୍ୟ କ୍ଷମ୍ବା lenge:=tsu=gi k^hem-bø: daku bɛ?. PRN.HON=PL=AGT know.HON-2INF.GEN owner EQU.NE $k^{h} \varepsilon n - k^{h} \varepsilon m - bo - ra$ $b\epsilon$?. know.hon-rdp-2inf-aemph EQU.NE 'You are possessors of (this) knowledge, (you) indeed know.' (NAB BLA 7) (16.14) מב אילי מבידי אשידי מקו $\tilde{i} = ra$ di ta:ri di=**ra** làp. this axe this=AEMPH EQU=AEMPH say 'This axe is indeed the one, he said.' (JDF axe story) (16.15) वर्त्तार्या प्रदेवा वर्त्ता के सुरा के राषे के राष k'ande: nè:po sã:te lừ: *có:-có:* man-bja-wa dem body lose.control-RDP anything NEG-do-CIRC patient like.that until zak-o *bε=ra*. tcam just.like.that set-2INF EQU.NE=AEMPH 'Patient who has such a weak body is placed here just like that without anything done.' (rnam-rtog 6)

Finally, $=r\tilde{a}$: is also used in "copy verb constructions" (see Ozerov & Daudey [2017] and Konnerth [2014: 586])⁴⁴⁵, where the emphatic occurs between reduplicated forms of the same verb in a construction that underlines inevitability. Examples (16.16-18) illustrate affirmative constructions. The last one is a complicated construction with three emphatics.

(16.16) ট্রিশ্'ষ্ট্' নেদ্রু'ন' নেদ্রু'ন' জিন্ব $k^{h}ik-ti$ gju=ra gju-cĩ: lead-NF go-AEMPH go-NPST.PER 'I will surely take (you as my wife)' (song lyrics)

⁴⁴⁴ The meaning of the first part of the compound *ha-ko* 'know' is unknown.

⁴⁴⁵ Ozerov & Daudey (2017: 53) define a copy verb construction as [VERBAL_STEM=DISCOURSE_CLITIC FINAL_VERB]

(16.17) 5. रा यें दा रा दे रा रा के रा क inteimentei? mí: k^him t'uwa jø:-sø: no:=losmoke EX-NMLZ.SPAT.GEN direction=DAT necessarily human.GEN house ið:=rã: jø-po $da=s\epsilon$ nó:- $z\tilde{\epsilon}$: EX=AEMPH EX-2INF AP=QUO think-PROG 'Thinking that it seems that in the direction where there is smoke there necessarily has to be a house of a human...' (rna-gsung 4)

(16.18) جَ بَعْ عَلَى مَالَعَ عَرَبَ مَالَعَ مَا مَالَعَ مَا مَالَعَ مَا so Lhoke beginning know.HON-2INF = AEMPH EQU.PER 'So (they) certainly knew Lhoke in earlier times.' (CY interview)

A negator prefix may be preposed either to the first or the second verb in the copy verb construction. When the negator is prefixed to the first verb, the construction underlines obligation to do the action perhaps even against one's own will, see (16.19). In my data, the negator in this position is always ma-.

(16.19) a) สาสาญารรา สุรญา พรา

ma-bak=rã:bak $b\epsilon$?.NEG-carry=AEMPHcarry EQU.NE'There is no way of not carrying/It has to be carried.' (KN e)

When the negator is prefixed to the second verb, the main verbal action is negated and focus is on the absoluteness of inability of the verbal action to happen, see (16.20). In my data, the negator in this position is always mi.

 $^{^{446} =} r\tilde{o}$: is a dialectal variant of $= r\tilde{a}$:

(16.20) เกรา สลัการกา สิเพลัการ์ เมตุ

di $t^h \tilde{o}:=r\tilde{a}:$ $mi-t^h \tilde{o}:-bo$ $b\varepsilon^2$. this see=AEMPH NEG-see-2INF EQU.NE 'There's absolutely no seeing this (unclear photocopy).' (RL oh)

(16.21) क्षें प्रदेन वर्त्ता में वर्त्तर के में वर्षाय का मार्ग

ódi	nàŋɛa	t ^h øn-ra	mi-t^høy -k ^h ɛn	$b\varepsilon = s.$
that	within	happen-AEMPH	NEG-happen-NMLZ	EQU.NE=QUO
'Witl	hin that,	there is no way of	(that) happening.' (C	CY interview)

 (16.22) ฤลัสาสัง มดุสารา มิงมดุสามุกสา
 *zømø námpu nám-ra mi-nám-k^h*ẽ: other with match-AEMPH NEG-match-NMLZ
 '(it is of the kind that) absolutely does not match with others' (KT discussion with TB)

The negated copy verb construction with the existential copula, illustrated in (16.23) and (16.24), can be either once (16.24a) or twice negated (16.24b):

(16.24) a) यॅन्'रूट' केन्'अपव' भ्रन

 $j\dot{\theta}$:= $r\tilde{a}$: $m\dot{e}$:(- $k^{h}\varepsilon n$ $b\varepsilon$?). EX=AEMPH NEG.EX(-NMLZ EQU.NE) 'there absolutely is not' (TB e)

b) $\widehat{a}_{\nabla} \cdot \overline{x}_{\nabla} \cdot \widehat{a}_{\nabla} \cdot \widehat{a}_{\nabla} \cdot \widehat{a}_{\nabla} \cdot \widehat{a}_{\nabla}$ $m \widehat{e}:=r \widehat{a}: m \widehat{e}:(-k^{h} \varepsilon n b \varepsilon^{2}).$ NEG.EX=AEMPH NEG.EX(-NMLZ EQU.NE) 'there absolutely is not' (TB e)

16.1.2 Contrastive emphatic =*to*

The contrastive emphatic =*to* differs from the general and potentially anaphoric emphatic = $r\tilde{a}$: in that =*to* introduces an air of contrast to emphasis. In (16.25), for instance, the use of =*to* with the adverbial 'until now' suggests that the speaker makes a contrast between the past and the unknown future: thus far life has been good, but she does not know about the future.

(16.25) - - المج المج المج المج المحافظة محافظة المحافظة محافظة محا

t'ato $s\tilde{a}:t\varepsilon = to$ $l\tilde{c}m = ra$ $j\tilde{\phi}?$. now until=CEMPH good=AEMPH EX.PER '(We) are well until now (but I do not know about the future).' (PED life story) In (16.26), speaker A presents an assumption (in the form of a question) which contrasts with what speaker B knows: lama Kaching cannot be the proctor because he is abroad.

lám kats^hiŋ làp *tc^høtimbo=di* ka=tci? lám тò proctor=DEMPH lama PN say EQU.NE.O who=INDF lama $m\hat{o}=t\epsilon i??^{447}$ EOU.NE.O=INDF 'Is the proctor lama Kaching? (Or) which lama is it?' lám kats^hin=**to** mòu bɛ?. $t \epsilon^{h} i g \epsilon l = b \epsilon^{2}$. lam PN=CEMPH down EQU.NE abroad EQU.NE '(But) lama Khaching is down. (He) is abroad.' (KN kitchen discussion)

In (16.27), the speaker contrasts a man with his wife. The contrast is reflected by "but" in the English translation.

(16.27) $\overline{\eta} \otimes \overline{\eta} \otimes$

In (16.28), those who do not know the language of their ethnic community are contrasted with those who do.

(16.28) $\pi r^{i} \eta \eta r^{j} \eta r^{j} \eta \eta r^{j}$

The use of the contrastive emphatic may also convey disapproval, as in (16.29):

(16.29) यदेवा वराष्ट्रेक वायायुराकेंदा

 $d\epsilon m$ $s\tilde{a}:t\epsilon=to$ $ma-su\eta=m\emptyset^2$. like.that until=CEMPH NEG-say.HON=URG 'Please do not by all means go that far in speaking.' (Richhi 20)

Now consider (16.30), which has two instances of =to.

⁴⁴⁷ The use of the indefinite marker/numeral =tei? after the interrogative copula here is surprising.

⁴⁴⁸ The words $g \in m$ and g a p can refer to people of advanced years or to younger people who are shown respect.

- (16.30) A: ष्ट्र'हे! विधाय में अय पशुणी भय मन पत पत्वा मन केन केने विधाय पशुमन में के के कि विधाय के के के के क làp-ce? zak-ce? ádzi:. $k^{h}im = na = to$ man-gju-ge. home=LOC=CEMPH NEG-go-HOR say-INF put-INF oh.no mè:-pø: mí: k^him=na gju-ce? nò ts^ha-wa: NEG.EX-2INF.GEN person.GEN home=LOC go-INF face be.hot-CIRC 'Darn! Let's not go to (his) home. It is embarrassing to go to a man's home without anything to say or give.'
 - $B: \neg \dot{\tau} \dot{\tau}$, છુ \neg એ \neg . તેનું : તેનું : તેનું : તેનું : છુ $\neg \dot{\tau}$: છ $\neg \dot{\tau}$: $\dot{\tau}$: $\dot{\tau$ $k^h \epsilon$: $t'ar \varepsilon = to$ $m\dot{e}$:= εo . nàtea? now=CEMPH difference NEG.EQU.PER=AT 1_{PL} bhaila=lo ta-wa òm-bo $\tilde{\iota} = s\varepsilon$ làp tcho? **PN=DAT** see-pur come-2inf Equ.per=quo say be.allowed í́:=co. EQU.PER=AT 'But now it doesn't matter, you know. We can say that we have come to see Bhaila.' (Richhi 24)

In (16.30), person B has suggested that she and person A should go to see a boy named Bhaila in his home. Person A expresses her disapproval/disagreement in (16.30) by using the contrastive marker =to, followed by the reason for disagreement. Person B, in turn, uses the contrastive emphatic with the adverbial 'now' ($t'ar\varepsilon = to$) in order to claim that the circumstances at the time are, contrary to what person A thinks, conducive to the action she has suggested. The latter instance of =to is accompanied by the clause-final attention marker =co, which underlines the attention-worthiness of the proposition caused by the contrast.

The contrastive emphatic quite frequently collocates with the conditional form of the verb:

The contrastive emphatic may occur in the middle of a complex copula construction to convey the meaning 'although it is the case that':

(16.32) \breve{w} $(a_{1}^{-}, \breve{w}a_{1}^{-}, \breve{f}^{-}, \breve{w}a_{1}^{-}, \breve{w}a_{1}^{-}, \breve{f}^{-}, \breve{w}a_{1}^{-}, \breve{w}a_{$

Finally, =*to* occurs in an negated emphatic copy verb construction VERB=*to* NEG-VERB (for copy verbs, see Ozerov & Daudey 2017):

(16.33) মিনার্ন্ত ক্রুনমার্ট মাক্রুনমান।

 $k^{h}\tilde{o}:=tsu$ $gj\tilde{a}:=to$ $mi\eta$ - $gj\tilde{a}:-wa$ 3PL=PL fill.up=CEMPH NEG.fill.up-CIRC 'They do not at all fill up (mutterings of prayer)' (TB discussion with PB)

16.1.3 Demonstrative-emphatic =di

In addition to the demonstrative uses discussed in §6.4, the proximal di 'this' has been grammaticalized into demonstrative-emphatic⁴⁴⁹ =di, which does not have referential function but brings emphasis to the element it is postposed to. The use of the demonstrative-emphatic is prevalent in spoken language, sometimes occurring several times in one clause. It co-occurs with the prenominal modifying demonstratives *ódi* (16.34) and *di* (16.35).

(16.34) שֿוֹ מָלָיחִן דָי מָלָ זוּ מְלָי זוּז מָז מָלָי מוּזימַל שָּרָן *odi* mí=di=gi gju=di k'andɛ: mè:-kɛn that man=DEMPH=GEN possessions=DEMPH anything NEG.EX-NMLZ bɛ?. EQU.NE 'That man didn't have any possessions.' (JDF axe story)

di $t' \varepsilon p = di$ $dzo: k'adz \emptyset po?$ this book=DEMPH price how.much EQU.NE.Q 'How much is the price of this book?' (KT e)

In (16.36a), =di functions as a specifier which signifies that one referent is chosen from among others. In (16.36b) without the emphatic, on the other hand, the (topic) selection had already been established.

(16.36) a) $\mathbb{R}^{n} \mathbb{R}^{n} \mathbb{R}^{n}$

'He is a doctor.' (YR e)

Sandberg (1895: 21) notes that "[o]rdinarily *di* follows its noun; but where any singling out of the noun is desired we have one *di* placed before and another *di* after the word." Similarly, Zeisler (2011: 278) observes that in the Kenhat dialect of Ladakhi the "definiteness marker" -*de* co-occurs with prenominal demonstratives and can even be postposed directly to to the proximal /i/ *hdi* 'this'

⁴⁴⁹ The term "demonstrative" refers both to the origin of the emphatic marker and to its present, more demonstrativelike uses. The term "emphatic" underlines the fact that the marker has developed discourse-oriented functions which do not naturally fall under the umbrella of categories such as "demonstrative" and "definite."

and distal /a/ ha 'that' to form /ire/ hdi-de and /are/ ha-de respectively. A similar use of the "definite article" -de occurs in Kyirong Tibetan (Huber 2002: 70).

Although the semantic range of =di in Denjongke subsumes functions which in other languages have been characterized as "definiteness," its uses go beyond definiteness. Such categories as proper nouns, personal pronouns and demonstrative pronouns are by definition definite, as suggested by the fact that the English definite article does not co-occur with them. The Denjongke =di, however, may be postposed to all the three above-mentioned categories and also to others.⁴⁵⁰ Example (16.37) illustrates the use of the demonstrative-emphatic with a proper noun and a demonstrative pronoun, and example (16.38) exemplifies a use with a personal pronoun.

dzambuliŋ=na ý: dɛndzõː=di ļo k'adi t'ã: mìnda, Sikkim=DEMPH south continent=LOC region which and unlike tsa:t c^{h} impo \tilde{t} :=s. né: di=di lèpti this=DEMPH very.much place invaluable EQU.PER=QUO 'Sikkim (is) different from any region in the rest of the world, this is a very precious place.' (CY Interview)

Examples (16.39-41) further illustrate that =di may attach to postnominal demonstrative modifiers *ódi*, *di* and *dodi* respectively.

(16.39) দ্' দ্বঁম' ঈ'এদি' এদি'ম' ঈুঁম্মান্ট্ৰ ঈ'ৰ' স্থ্ৰিদম'ষ্ট্ৰ' ঈদ্ দ্বঁ হ'ৰ্মা to:-ti óna lep-ti now bear that=DEMPH very.much=AEMPH be.hungry-NF there arrive-NF ôm-bo.
come-2INF
'Now that bear arrived there very hungry.' (KT animal story)

(16.40) בקי מבי מבי

 $t' \varepsilon p \quad di = di$ book this=DEMPH 'this book (holding the book in hand)' (SGD wedding customs)

⁴⁵⁰ Similar extended, non-referential, emphatic use of the demonstrative *se* 'it' occurs in Finnic languages (Grünthal 2015: 277, Kittilä & Yurayong (forthcoming). Grünthal (2015: 280), for instance, describes the non-referential use of *se* in Veps (Finnic) as an "unspecific focus particle." Moreover, in Classical Greek (e.g. Xenophon's Anabasis) and Koine Greek (e.g. New Testament), the "definite" article, which has a demonstrative origin, may be preposed to personal names, a use which seems to fall outside the purview of "definiteness."

(16.41) מְלָימָלִי מְלָי זוֹצֿיצָי פְּאַישון *dodi=di* potso=tsu nɛ̀:-sa. this.right.here=DEMPH child=PL sleep-place 'This right here (is) the children's sleeping place.' (PD altar room video)

A demonstrative-based marker di or =di may occur either before the plural marker (16.42) or after it (16.43).

রূরম'র্য স্থ্রন'এগবা্ $k^h \tilde{o}$: di = tsu = gite^høge: nà: nàŋea t^hamtce=ki dendzõ lòk 3PL this=PL=AGT all=AGT king here inside Sikkim return tc'øn mà-tɕu? sé-k^hen=di ódep kjap-ti k^hõ: come.HON NEG-cause say-NMLZ=DEMPH like.that do-NF 3PL gagdza kjab-be?. obstacle do.2INF-EQU.NE 'All of them like that made obstacles so the king would not come back here to Sikkim.' (CY interview)

(16.43) हे र्यट्र र्डु दर्न कें क्ष आपत हुन

t ε *k*^h δ *:=tsu=di m* δ *ta-k*^h ε *n b* ε *?*. so 3PL=PL=DEMPH divination look-NMLZ EQU.NE 'They were (lit. are) diviners.' (UT plains story)

Apart from positioning, a further formal difference between di in (16.42) and =di in (16.43) is that the vowel in pre-plural di tends to be somewhat longer than in the emphatic post-plural =di, suggesting that the pre-plural di begins a new phonological word whereas post-plural =di is a clitic attached to the previous word. Differing distribution and phonology suggest a possible difference of meaning. I am, however, not aware of such a difference. Nevertheless, to reflect the different positioning and phonology, the pre-plural variant di is in this grammar written as a separate word and glossed as a proximal demonstrative 'this', whereas the post-plural variant is written as a clitic glossed as an emphatic =DEMPH.

Furthermore, di/=di may occur on both sides of the plural marker, as in (16.44), or two times with an intervening case-marker, see (16.45).

(16.44) \overline{a} של של הי של ה

 The first instance of =di in (16.44) and (16.45) functionally covers the fields of demonstrativity and definiteness. The second use (following the plural marker) adds further emphasis on the noun phrase. However, with a singular instance of di/=di in a noun phrase, the position of the morpheme in relation to the plural marker is not necessarily a trustworthy indication of its function as either demonstrative/definiteness marker or an emphatic. This is illustrated by the following examples, in which both the post-plural =di (16.46) and pre-plural di (16.47-48) occur with an already definite nominal, a personal pronoun (16.46-47) or a demonstrative (16.48).⁴⁵¹

(16.46) אָר:אָי אבן *k^hõ:=tsu=di*

3PL=PL=DEMPH 'They...' (UTR plains story)

(16.47) xx xx xx

 $k^{h}\tilde{o}$: di=tsu 3PL this=PL 'They...' (CY interview)

(16.48) \breve{m} at at $d\tau$ at $d\tau$ $\delta(d)i \ di = tsu$ that this=PL 'They...' (PED life story)

The demonstrative-emphatic can intervene between a noun and its genitive modifier, as shown in (16.49), where =di seems to bring emphasis on the possessor.

(16.49) ເຈັດເຈົ້າງ ດີເງ ດັ່ງ ອູເຈາ yàtci=gi=di lògju? 1PL.GEN=GEN=DEMPH story 'our story' (YR canteen video)

It also attaches to postpositions:

(16.50) בי מקמיצי הלי

nà pámpu=di
1SG with=DEMPH
'With me' (RS animal song intro)

In (16.51), the first instance of =di occurs as a type of substantivizer/nominalizer attached to an adjective:

⁴⁵¹ Somewhat analogously, the Finnish plural demonstrative ne 'they' has grammaticalized into a plural emphatic which may attach to the homophonous demonstrative, ne=ne 'they-DEMPH' (personal knowledge).

Moreover, =di attaches to verb forms. Co-occurrance with an infinitive is exemplified in (16.52).

The emphatic = *di* can also attach to the anterior construction shown in (16.53) and the nonfinal construction exemplified in (16.54).

(16.53) สัญาทุกิรที่ พรุสาหระ สิจพระสัญาพร กรีร $l \delta b d \phi := g i \qquad dy \eta k^h a : \ l \varepsilon p - so := l \varepsilon = d i$ school.GEN=GEN in.front.of arrive-CMPL.2INF=ABL=DEMPH

'After arriving in front of the school...' (RS pupil joke)

(16.54) 載雲 如象如 道氣 整泊 요子 하 黃丁 如氣 요널드 如氣 요널드 따라 繋泊 午 菜 花 ബ bjõ:,
tc^hu-la=tci? tc'øn-diki=di "tɛ tc^hø? k'an bjõ:,
water-god=INDF come.HON-NF=DEMPH so 2SG.L what disappear k'an bjõ:" làp-tiki t'i ta-u=lo.
what disappear say-NF ask look-2INF=REP
'A water-god came and asked (him): "So what of yours disappeared? What disappeared?"" (JDF Axe story)

The demonstrative-emphatic can attach only to the longer, nonfinal form -tiki/diki, not the short form -ti/di, presumably because of the phonetic similarity of =di and -ti/di. The longer converbal form occurs almost exclusively in spoken language. In writing, mere $t_{\mathcal{G}} \circ pn-di$ would be used in place of $t_{\mathcal{G}} \circ pn-di=ki=di$.

Moreover, =*di* can be added to -*po*-infinitive forms:

(16.55) मारुवा या प्रेन वाहमाया द्विराया प्रेन या होन या होन वा राये प्रवाश्य होने के लिने हो के जिने क्रियाया

 $\eta \dot{a}t\varepsilon a = lo = di$ $n \dot{\tilde{a}}:wa$ p'im-bo = di = lo $pint^h akp \sigma$:1PL=DAT=DEMPHpermission give-2INF=DEMPH=DATbottom.of.heart.GEN $go: l\varepsilon$ $t^h udzitc^h \varepsilon$ $c \dot{u} - c \tilde{i} := la$.through thank.yousay.HUM-NPST.PER=HON'I thank you from the bottom of (my/our) heart for giving us the permission.' (CYinterview)

The emphatic =di makes nominalized verbal constructions more conducive to nominal operations such as case-marking. In (16.55), for instance, adding the dative-locative case-marker directly to the nominalized form (*p'im-bo=lo*) would result in a form that is homophonous with the typical past reportative construction often heard in stories (both dative-locative and reportative are marked with =lo). An intervening =di thus functions as a disambiguator. Especially in written language, the agentive form of =di attached to $-c\epsilon$ -infinitive marks causation, see §15.4.6.

16.1.4 Topicalizer-emphatic =*ni/nɛ*

The topicalizer-emphatic $=ni/n\varepsilon$ is typically used for activating new referents in discourse. In this way, it is dissimilar to the emphatic $=r\tilde{a}$, which typically emphasizes already activated referents but similar to =di and =to both of with which it semantically overlaps. In some of its uses $=n\varepsilon$ can be characterized as a topicalizer in that it draws the addressee's attention to new topics. In other uses, it is safer to just say that $=n\varepsilon$ simply emphasizes a certain constituent without necessarily making it the topic under discussion in the next clause. The term "topic" is here defined in a non-technical sense as something that the sentence is about.

The topicalizing function of $=n\varepsilon$ is illustrated in (16.56) and (16.57).

(16.56) هَتَرَ مَنْ مَتَرَ مَعْرَ مَعْنَ مَعْرَ مَعْرَ مَعْرَ مَعْنَ مَعْرَ مَعْنَ مَعْرَ مَعْنَ مَعْرَ مَعْنَ مَعْن مُنْ مَعْنَ مَعْنَ مَعْنَ مَعْنَ مَعْنَ مَعْنَ مَنْ مَنْ 28. 28. (L=DAT father=TOP NEG.EX.PER father.GEN instead 28.3.) other who EX.PER 'You don't have a father. Who is there in place of your father except yourself?' (Richhi 83)

(16.57) गात्र, केंद्र' गा'त' र्शेद'र्? त्रग्रु'र्दे' क्षेत्र' बा'अप्र'यम्। दारुषा'ते' येप'ङ्गे' खेबबाञ्चग' झेंद' थेंदा gju-do í karma tc^hø? k'ana số:-bo? ma-làp-a. $\eta at \epsilon a = n\epsilon$ Karma 2sg.l where go.PFV-PST go-IPFV EQU.PER NEG-say-CIRC 1PL=TOP sémdu? dø: j*à*?. lèpti very.much worry stay EX.PER 'Karma, where did you go, without saying you were going? As for us, we were very worried.' (Richhi 110)

In (16.56) the word $\dot{a}po$ 'father' is topicalized by $=n\varepsilon$ and occurs as the topic of the next sentence. In (16.57), the use of $=n\varepsilon$ signifies a topic-switch from the addressee to the speaker (and her reference group).

Example (16.58), presents a longer stretch of text in order to illustrate three instances of $=n\varepsilon$ in context. In the example, a girl threatened by an undesirable marriage is talking in her head to her hoped-for groom (sadly, in vain).

(16.58) गाहा, केंट्र' ट्र'र्ये' येवर्रात्र' व्याया' र्वया ट्र' ट्र'वे' nà=lo lèm-ba màla? *сó?*. karma. tc^hø? t'a ηà=nε PN 2SG.L 1SG=DAT take-PUR quick now 1SG=TOP come.IMP क्रें'व' ज्जुव'र्र्र्य' दर्धेव'र्ळर्रा क्रें'धण' पर्छन्'शण्व' न' केंन्' श्रव'र्ये' ne=na sỳn-rap $t^h on - ts^h a$. ne-tha? tee:-khe: t'a te^hø? mèmbo trap=LOC enter-IMF become-CMPL trap-rope break-NMLZ now 2sG.Lexcept പരു गागोषाया के क्रिंगुषा गाहा, यातुव गागोषाया zen ka=gi=jã: karma, $z \in n$ ka=gi=jã: mi-ts^hu?. other who=AGT=even NEG-be.able.to other who=AGT=even PN มิ สู์ขุณุ ๆ ๆ ๆ ๆ ๆ ๆ ยู่ณ คุๆ, กลิ่าอ์กา คุณ ขุลุ วิ mi-ts^hu?. t'a $n\dot{a}$: k'an p'ja- $\epsilon\epsilon$?, $n\dot{e}$:=tsa: t^haplam zen=**n**e NEG-be.able.to now I.AGT what do-INF 1SG.GEN=at means other=TOP गुत्र रद्दे : २४ में गुर्भ र् युवा के दा यो में गुरू र यो के रा यो k'ɛːsiʔ k'and ε :=j \tilde{a} : cókts^hip jà:-ne=to me?. nám=le any=even if NEG.EX.PER wing EX.PER-COND=CEMPH sky=ABL तस्र हेः तें र केंग केंता कें लरे के र र केंग केंग केंग $p^h u$:-di \tilde{o} : ĩ. $t \epsilon^h o$ ódi=**n**ɛ nà=lo cókts^hip mè?. fly-NF come be.allowed EQU.PER that=TOP wing 1SG=DAT NEG.EX.PER 'Karma, come quickly to take me. I $[=n\varepsilon]$ have become one about to be entrapped. There is now no one except you who can break the trap-rope. Now what should I do? I have no other means $[=n\varepsilon]$. If I had wings, I could come flying through the sky. But that's the issue $[=n\varepsilon]$ (lit. that= $n\varepsilon$): I do not have wings.' (Richhi 155)

The first use $=n\varepsilon$ ($\eta \dot{a}=n\varepsilon$) in (16.58), topicalizes the speaker's own situation. The second use $(t^h a plam z\varepsilon n = n\varepsilon)$ introduces a new topic, i.e. ways of escaping the present situation. The third use $(\dot{o}di=n\varepsilon)$, which resembles a verbless clause, directs the addressee's attention to a relevant facet in the previous clause, which, again, is a new topic.

In harmony with its name as topicalizer-emphatic, some of the uses of $=n\varepsilon$ are better described as simply emphatics rather than as topicalizers. For an examples, consider (16.59), with two instances of =ni.

karma $te^{h}\phi^{2}$ $t'a=n\varepsilon$ bom- $ts^{h}a$:. $n\dot{\varepsilon}:pa=lo$ $tat\phi^{2} p'ja-k^{h}\tilde{\varepsilon}:$ Karma 2SG.L now=TOP become.big=CMPL patient=DAt care do-NMLZ $m\dot{\varepsilon}mpo=n\varepsilon$ $t^{h}\phi n-ts^{h}a$:. doctor=TOP become-CMPL 'Karma, you have now become important. You have become a doctor who takes care of patients.' (Richhi 114)

The topicalizer-emphatic $=n\varepsilon$ can be used contrastively very similarly to =to, see (16.60) and (16.61). Note that in (16.60) $=n\varepsilon$ attaches to a pronominal expression which signifies topic

continuation rather than new topic. The function of $=n\varepsilon$ is rather to bring into focus the contrast between the past and the present. In (16.61), the contrast is reflected as *but* in the translation.

(16.60) हे कॅंप्रिट् र्ड् वे देन्या केव पर्या गो लुम्ह केवा ĩ:. te ódi=tsu=**ne** t'iŋzã: mìndu-kɛ сù-се so that=PL=TOP nowadays NEG.EX.SEN-IN say.HUM-INF EQU.PER 'Those are not there nowadays, I submit.' (CY interview) भूमा भाषा के सेव भीषा dzāːdar mjoː-mjoː-po dɛːtsʰika k'ɛːsi? pè: mémpø: ín-ne at.that.time if my doctor.GEN training finish-RDP-2INF EQU-COND kjap te^ho? ĩ dzā:dar ma-mjõ: $m\hat{u}=lo$ t'ato=rã: nén 3SGF=DAT now=AEMPH marriage be.allowed EQU.PER training NEG-finish do $g\tilde{a}$:= $l\varepsilon$ = $n\varepsilon$ min-dik. time=ABL=TOP NEG-be.fitting 'At that time, if my training is finished, (it) would be acceptable to marry her immediately. (But) while the training is not finished, (it) is not fitting.' (Richhi 112)

The topicalizer-emphatic can also attach to quantifiers:

Lastly, $=n\varepsilon$ is used in clauses which define and identify, see (16.63) and (16.64). The emphatic functions as the drawer of the addressee's attention before the definition/identification is given in the next word. In these uses, =di could be used instead of $=n\varepsilon$. In an attempt to reflect the Denjongke structure, the English translations are somewhat cumbersome.

(16.64) मुंग इत्यायाया के र्यत्य थेंता

k'jo $[i\tilde{a}:-k^{h}\tilde{\epsilon}:=n\epsilon$ $p'\tilde{o}:bu$ $j\tilde{\phi}?$. wheat carry-NMLZ=TOP donkey EX.PER 'As for carriers of wheat, donkeys were there.' (PD bet story)

16.1.5 Additive emphatic $=j\tilde{a}$:

The additive emphatic $=j\tilde{a}$: expresses both additive ('also, too; either [neg.]') and scalar meanings ('even'). In affirmed clauses, $=j\tilde{a}$: is typically additive, but in negated clauses it is typically scalar. The definitions for "additive" and "scalar" are adopted from Krifka (1999: 111), according to whom addititive uses express "that the predication holds for at least one alternative of the

expression in focus". For instance, in (16.65) the use of $=j\tilde{a}$: implies that someone else, in addition to the brother, is in the army. In (16.66), the additive use of $=j\tilde{a}$: implies that in addition to intelligence there is another positive quality about the referent.

(16.65) अपकुष्पत्त नुवग्विते व्यत्त व्यत् थे?

(16.66) हि से रू र्यास रेवागा संक्षेत्र म रेवा रव स्था केता

 $k^{h}u$ mí $t\epsilon \tilde{a}:ta$? $t\epsilon iku$ ma-ím-ba $rikt\epsilon \epsilon n=j\tilde{a}:$ $\tilde{i}:$. 3SGM human beautiful only NEG-EQU.PER-CIRC intelligent=also EQU.PER 'He is not only a handsome person but also intelligent.' (PT e)

Scalar uses, on the other hand, imply "that the predication holds for the expression in focus, and presuppose that this predication is prima facie less likely than the alternative predications" (Krifka 1999: 111). For two examples, both of them negated, consider (16.67) and (16.68). In (16.67) the condition of not talking to people at home is depicted as less likely than not talking to people in other places.

(16.67) ผู้สาสานักพรา สิ่าส์ พฤสาชิญา สำคริสาสรา เขามา สิรา

k^him=na=lo=jã: mí=tsu námtei? ódɛm=sã: k^ha-lap house=LOC=DAT=even human=PL with that.much=until mouth-speak mè?.
NEG.EX.PER
'Even at home, there is not that much talking with people.' (Richhi 164-165)

(16.68) هرج حتحما في قَلْع حج محتمادًا والمعامة والمحتمد والمحتفية في الحد والمعنية في المحتفية الحتفية

Although scalar uses of $=j\tilde{a}$: (glossed as 'even') in my data occur in negated clauses, not all negated clauses with $=j\tilde{a}$: can be characterized as scalar. The use of $=j\tilde{a}$: in (16.69) is best described as negated additive and translated into English as 'either'.

 Example (16.69) does not suggest that Choki's not eating is less likely than Karma's not eating. It simply expresses that in addition to Choki there is someone else who does not eat.

Although the uses of $=j\tilde{a}$: in affirmative clauses in my data are typically additive, comparative clauses are an exception, because comparison is by nature scalar. The use of $=j\tilde{a}$: in (16.70) implies that Gyaltshen is so tall that the option that someone else would be taller is less likely.

(16.70) און אי פר אַשיאשל יאנאי ארי אריזן פראן karma=di gjalts^hen=le=**jã**: riŋku du?. PN=DEMPH PN=ABL=even long EX.SEN 'Karma is even taller than Gyaltshen (I see).' (KN e)

In addition to the additive and scalar uses described in this section, $=j\tilde{a}$: occurs in two additional uses, which are described elsewhere in the grammar. First, when attached to a question word and followed by a negated verb, $=j\tilde{a}$: is used for forming negated indefinite expression such as '(not) anything', '(not) anyone' and '(not) anywhere', see §6.3.2. Second, the enclitic $=j\tilde{a}$: and independent $j\tilde{a}$: are used for connecting clauses in both monosyndetic and bisyndetic manner, see §12.2 and §12.3 respectively.

16.2 Clause-final clitics

Clause-final clitics are listed in Table 16.2.

=lo	র্ম	reportative
=sɛ/si/s	र्ह्स'	quotative
=la	এযাম'	honorific
=60	Ĩ.	attention marker
=ki/gi	गो	non-commitment marker

Reportative =lo and quotative = $s\varepsilon$ were discussed under treatment of evidentiality in §9. This section addresses the remaining clausal clitics =la (§16.2.1), =co (§16.2.2) and =ki/gi (§16.2.3),

16.2.1 Honorific =la

By placing the honorific enclitic at the very end of the clause, the speaker shows respect to the addressee. The honorific occurs after other clausal clitics (16.71) and tags (16.72):

(16.71) a) خَرَقْتُمْ مَرْجُرُ خَمْتُ مَنْتُوْمَمْ اللَّهُ مَنْ الللَّهُ مَنْ اللَّهُ مَنْ اللَّهُ مَنْ اللَّهُ مَنْ اللَّهُ مَنْ اللَّهُ مَنْ اللَّهُ اللَّهُ مَنْ اللَّالِي اللَّهُ مَنْ اللَّالِي اللَّهُ اللَّهُ مَنْ اللَّالِي اللَّهُ مَنْ اللَّالِي اللَّهُ مَنْ اللَّالِي اللَّهُ مَنْ اللَّالِي اللَّالِي اللَّالِي اللَّالِي اللَّالِي اللَّالِي اللَّالِي اللَّالِي اللَّالَةُ اللَّالِي اللَّالَةُ اللَّالِي اللَّالَةُ اللَّالَةُ الْحَالَةُ مَنْ الْحَلْمَا اللَّالِي الللَّالِي اللَّالَةُ اللَّالَةُ الْحَلْقُلْمَا اللَّالَةُ اللَّالَةُ الْحَالَي مَا اللَّالَةُ الْحَالَةُ مَنْ اللَّالَةُ اللَّالَةُ اللَّالِي اللَّالَةُ الْحَلْقُلْعُالِي اللَّالَةُ اللَّالِي الْحَالَةُ اللَّهُ اللَّالِي اللَّالَةُ الْحَلْقُلْعَالِي اللَّالَةُ الْحَلْقَالَةُ الْحَلْقَالَةُ اللَّالِي اللَّالَةُ الْحَلْقَالَةُ اللَّالِي اللَّالَةُ اللَّالَةُ اللَّالِي اللَّالِي اللَّالِي اللَّالَةُ الْحَلْقَالَةُ الْحَالَةُ اللَّالِي اللَّالِي اللَّهُ اللَّالِي اللَّالِي اللَّعَالَةُ اللَّالِي اللَّهُ اللَّهُ اللَّالَةُ اللَّالِي اللَّالَةُ اللَّالَةُ الْحَلْحَالَةُ اللْحَالَةُ الْحَلْحَالِي مَا اللللْحَالِي اللَّالِي اللَّالْحَالَةُ اللَّالِي اللَّالَةُ اللَّالَةُ الْحَلْحَالَةُ اللَّالَةُ اللَّذَا الْحَالَةُ اللَّ الْحَالَي اللَّذَا اللَّالَةُ اللَّ لَقَالَ اللَّالَةُ لَقَالَ الللَّهُ اللَّالَةُ الْحَالَةُ لَقَالَ الْحَالَةُ الْحَالَةُ اللَّالَةُ الْحَالَةُ الْحَالَةُ اللَّ الْحَالَةُ اللَّالَةُ اللَّالَةُ اللَّالَةُ الللَّ اللَّالْحَالَةُ الْحَالَةُ اللْحَالَةُ الْحَالَةُ اللَ

c) वेव प्रवेष में भगषा

min=ze:=co=la. NEG=have.HON=AT=HON '(he) doesn't drink (it), you know.'(PT kitchen discussion)

- d) $a = a^{-i} a^{-i}$
- (16.72) a) הַיאָּי הַ דה מקמי צּׁקיד עוֹקה אָקיקימיקא? *t'ato t'a rã: námtsi=ra jèbbɛ=so, iŋ-ga=la?* now now 2SG.L with=AEMPH EX.NE=AT EQU.PER-PQ=HON 'Now she's with you, you know, isn't she?' (BB BB discussion)
 - b) אבמי אך יומד יוֹי מֹימיזאן *dzɛ:-cɛ? náŋ-gɛ lo=la.* meet.HON-INF do.HON-HORT TAG.Q=HON 'Let's meet, okay.' (KTL phone call)

The clitic = la: (typically pronounced with a longer vowel) is also used in forming honoric nouns referring to persons, see §3.7.4.3.

16.2.2 Attention marker = co

The uses of the attention marker =co are more fully described in Yliniemi (2016). This section provides a brief summary. The presence of the attention marker, which may be either speaker or addressee-oriented, indicates that something is brought to the forefront of the speaker's or the addressee's attention. The attention marker =co occurs in declarative uses postposed to a verb and in interrogative uses postposed to other parts of speech (verbless uses). Verbal uses, which resemble the notion "mirativity," are either speaker or addressee-oriented, whereas verbless uses, which resemble the notion "contrastive focus," are always addressee-oriented. When occurring with copulas, the function of =co as either speaker or addressee-oriented is partly dependent on the evidentiality of the copulas. With other verbs, the orientation of =co is dependent on other contextual factors.

A typologically interesting feature is that the same marker =co has both speaker-oriented uses (marking the proposition as newsworthy to the speaker) and addressee-oriented uses (marking the proposition as newsworthy to the addressee). The following two examples illustrate a speaker-oriented use (16.73) and an addressee-oriented use (16.74) of =co attached to the equative copula bc?. Speaker vs. addressee-orientation is contextually determined.

(16.74) and $\operatorname{$

In (16.73), the speaker expresses the newsworthiness of the information for herself (as also suggested by the use of the interjection $j\dot{a}$.). In (16.74), on the other hand, the speaker is carrying out a pedagogical monologue and wants, by using =co, to directs his addressees' attention to the significance of the story he has just told.

A frequent context for using =co is when the speakers provide information that they think is contrary to what the addressees believe:

- - B: $\tilde{\mathfrak{m}}$ 'ב $\tilde{\mathfrak{m}}$ ' השקילו השלילו ה

Although many uses of =co resemble clauses which in other languages have been termed "mirative"⁴⁵², example (16.76) shows that uses of =co are not determined in terms of newness to "overall knowledge structure" (DeLancey's [1997: 33] description of mirativity) but in terms of attention-worthiness of the proposition.

(16.76)	જો', દ્વિ	र व्ययार्गे रुवारगु	ર્લેં, દ્વિ	ਗ੍ਰਨਾਵਾਂ ਘੋਨ੍ਰਾਸ਼੍ਰਾ '	พฤ			
	έː,	nè:	làko	t£iku=lo,	лè:	kã:po	jò:= co	làp.
	eh,	1sg.gen	hand	only=REP	1sg.gen	foot	EX.PER=AT	say
	'(I the	ought) I had	only	hands, (but)	I have also	feet (I re	alize).' (KT	animal story)

In (16.76), the information that he has feet is not new knowledge to the speaker. Rather, this old piece of information is particularly useful, and thus attention-worthy, in the situation the speaker is in (hence 'I realize' in the translation). The speaker, a marten, is stuck in glue and has tried to get out of the trouble by using his feet. When his feet are stuck in glue, he comes up with a new realization that he has further limbs. Counter-arguments for subsuming the uses of =co under "mirativity" are presented in Yliniemi (2016a).

Verbless uses of =co are topic-switching questions that direct the addressee's attention to a new item. The topic-switching question may be translated '(and/but) what about...?'. When occurring

⁴⁵² Especially when mirativity, following Hengeveld & Olbertz (2012), is defined as consisting of both speaker and addressee-oriented uses (p. 487) and being "a linguistic category that characterizes a proposition as newsworthy, unexpected, or surprising" (p. 488).

within a clause, as in (16.77), the attention marker is followed by a pause (in harmony with its use as a clause-final clitic).

(16.77) عَلَّمَ عَلَى اللَّهُ اللَّلِ اللَّهُ اللَّهُ اللَّهُ اللَّهُ اللَّهُ اللَّهُ اللَّهُ اللَّلَ اللَّلَ اللَّالِي اللَّالَ اللَّالَ اللَّالَ اللَّالَ اللَّالَ اللَّلَالَ اللَّالَ اللَّالَ اللَّالَ اللَّالَ اللَّالَ اللَّ المَا اللَّاللَّةُ اللَّا اللَّالَةُ الللَّالِي اللَّالِي مَا اللَّالِي اللَّالَةُ الللَّالِي اللَّالِ لَقَلْمُ الللَّالَ اللَّالَ اللَّالَ اللَّالَ اللَّالَ اللَّالَ اللَّالِ اللَّالَ اللَّا لَقَلْمُ الللللَّ اللَّا اللَّالَةُ اللَّاللَّا اللَّالَةُ اللَّاللَّةُ عَلَيْعَالِي مَالَةُ اللَّاللَّاللَّالَ اللَّالَةُ اللَّ

A noun phrase with =co can also form an independent topic-switching question, as illustrated by (16.78), where the speaker shifts the topic from himself to the addressees.

(16.78) جَ هَمَ عَلَيْهُ عَلَيْهُ عَلَيْهُ مَعْلَى اَبْ اَلَّهُ عَلَيْهُ اَبْ اَلْحَ الْحَافَةُ الْحَافَةُ عَلَيْهُ الْمُ الْمُعْلَى الْمُعْلَى الْمُحْتَى الْحَافَةُ الْحَ المَا الْحَافَةُ الْحَ الاحَافَالَا الْحَافَةُ الْحَافَةُ الْحَافَةُ الْحَافَةُ الْحَافَةُ الْحَافَةُ الْحَافَةُ الْحَافَةُ الْحَافَةُ الْحَافَ الْحَافَةُ الْحَافَةُ الْحَافَةُ الْحَافَةُ الْحَافَةُ الْحَافَةُ الْحَافَةُ الْحَافَ الْحَاقَالْ

The attention marker =co with copulas can be used in a way that is functionally similar to verbless uses in that they also introduce new topics:

(16.80) أَنَّ عَامَةُ حَاجَةُ هَا اللَّهُ عَلَيْ المَّحَانِ عَلَيْ اللَّهُ اللَّهُ عَلَيْ اللَّهُ اللَّهُ عَلَيْ اللَّهُ عَلَيْ اللَّهُ عَلَيْ اللَّهُ عَلَيْ اللَّهُ عَلَيْ اللَّهُ عَلَيْ اللَّهُ عَلَيْهُ اللَّهُ عَلَيْ عَلَيْهُ عَلَيْهُ عَلَيْهُ عَلَيْهُ عَلَيْهُ عَلَيْهُ عَلَيْهُ اللَّهُ عَلَيْهُ عَلَيْهُ عَلَيْهُ عَلَيْهُ عَلَيْهُ عَلَيْهُ عَلَيْهُ عَلَيْهُ عَلَيْهُ عَلَيْ عَلَيْهُ عَلَيْهُ عَلَيْهُ عَلَيْهُ عَلَيْ اللَّا عَلَيْ اللَّا عَلَيْ عَلَ المَا عَلَيْ عَ

It is typologically interesting that =co has both clausal "mirative" like uses and uses which direct the addressee's attention to a clausal constituent, highlighting the similarity (i.e. directing attention) between the concepts mirative and contrastive focus.

16.2.3 Non-commitment marker =ki/gi

The final marker =ki/gi is a loan from Nepali, where ki is used as a question particle, conjunction 'or' and also as a very frequent clause-final marker. I have not yet fully understood the meaning of this marker in Nepali or, by extension, in Denjongke. Preliminarily, I name it, in harmony to its other uses in Nepali, a non-commitment marker (glossed NC).

- (16.81) המי מיקי קריאָל דָקיאשָלָק מָלי אָרָי אָזי מָלָי אָרָיאָן $\eta \dot{a}: \ l \dot{a} p - c \varepsilon = k i \quad t' onts^h \tilde{\varepsilon}:= d i = ra \quad odi \quad b \varepsilon:= k i.$ I.AGT say-INF=GEN subject=DEMPH=AEMPH that EQU.NE=NC 'The subject of my talking is just that.' (KL BLA 12)
- (16.82) هَرَ جَمْرَ مَرْ جَرَامَ جَمَارَ جَرَامُ اللَّهُ اللَّهُ عَلَيْ اللَّهُ عَلَيْهُ عَلَيْهُ اللَّهُ عَلَيْهُ عَلَيْهُ اللَّهُ عَلَيْ اللَّهُ عَلَيْنُ اللَّهُ عَلَيْنُ اللَّهُ عَلَيْنُ اللَّهُ عَلَيْنُ اللَّهُ عَلَيْنُ الْحُمَالَةُ عَلَيْ الْحُمَالَةُ عَلَيْنُ اللَّهُ عَلَيْ الْحُمَالِي الْحُمَالَةُ اللَّالَةُ اللَّا عَلَيْنُ الْحُمَالَةُ اللَّهُ عَلَيْنُ الْحُمَالَةُ اللَّالَ الْحُمَالَةُ عَلَيْ الْحُمَالَةُ اللَّا عَلَيْ الْحُمَالَةُ اللَّهُ اللَّا عَلَيْ الْحُمَالَةُ عَلَيْ الْحُمَالَةُ عَلَيْ الْحُمَالَةُ اللَّالَ الْحُمَالَةُ عَلَيْ الْحُمَا الْحُمَالَةُ اللَّالِ الْحُمَالَةُ عَلَيْ الْحُمَالَةُ اللَّالَةُ عَلَيْ الْحُلُولَةُ اللَّالَةُ اللَّا عَلَيْ الْحُمَالَةُ عَلَيْ الْحُمَالَةُ الْحُمَالُ اللَّا عَلَيْ الْحُمَالِ اللَّالَةُ عَلَيْ الْحُمَا الْحُمَا الْحُمَا الْحُمَالِي الْحُمَالَةُ اللَّا عَلَيْ الْحُمَا الْحُمَا عَلَيْ الْحُمَا عَلَيْ الْحُمَالِي عَلَيْ عَلَيْ عَلَيْ عَلَيْ عَلَيْ عَلَيْ الْحُمَا عَلَيْ الْحُمَالِ اللَّهُ عَلَيْ الْحُمَا الْحُمَا عَلَيْ الْحُمَا الْحُمَا الْحُمَالِ اللَّالِي عَلَيْعَالَيْعَالَيْ عَلَيْعَا اللَّالِ الْحُمَا الْحُمَا الْحُمَا الْحُمَا عَلَيْ عَلَيْ الْحُمَا الْحُمَا لَكُلُكُ الْحُمَالَةُ عَلَيْ الْحُمَا الْحُمَا الْحُمَا الْحُمَا الْحُمَا عَلَيْ الْحُمَا الْحُمَالَ الْحُعَالَةُ عَلَيْ الْحُعَالَيْ عَلَيْ الْحُالَةُ الْحُعَالَةُ اللَّا عَالَيْ عَالَةُ عَلَيْعَا الْحُعَا عَلَيْ عَلَيْ عَلَيْ حُمَا

Note that the homophonous *ki*, also a loan from Nepali, functions as an occasional complementizer, see §14.2.2.2.

16.3 Assertive and exclamative tags

The assertive tag $p\dot{a}$ and exclamative tag (h)o: are morphemes which are more loosely connected to the rest of the sentence than suffixes and clitics and are also used for interrogation. For interrogative uses of $p\dot{a}$ and (h)o:, see §11.1.1.2 and §11.1.6 respectively. The tag $p\dot{a}$ adds assertive nuance (§16.3.1) and (h)o: exclamative nuance to the proposition (§16.3.2).

16.3.1 Assertive tag pá

The form $p\dot{a}$ was in §11.1 shown to be an interrogated form of the personal copula \dot{t} . This form has, however, also grammaticalized into a non-interrogatory assertive tag which may attach even to copulas. The tag $p\dot{a}$ (along with its dialectal variants *ino* and *no*) occurs in declarative and imperative clauses. The phonological status of $p\dot{a}$ (both the copulas and tag) is open for more detailed research. For now, I have retained high pitch marking on the vowel to preserve connection to the source form *in-na* \Re_{a} . By using the assertive tag the speaker, rather than seeking a response

from the addressee, adds exclamative force or nuance to the statement or command, as if adding the equivalent of English 'I (certainly) tell you' or 'indeed'.

Example (16.84) illustrates the use of an interrogative $p\dot{a}$ in a question (a) and the exclamative $p\dot{a}$ in the answer to the same question (b):

b) ה׳ קעֿק׳ אָקראי אָ־ אּזי אָן קמ jó? kjap-to í: pá. ISG work do-IPFV EQU.PER TAG.ASR 'I am indeed working.' (KN e)

For two further examples of $p\dot{a}$ and *ino* appended to copulas, consider (16.85) and (16.86) respectively (the assertive force is suggested in the translation by *I tell you*).

(16.85) ব্ৰ'ৰ্জ ইল্বাম' জিব্'ন্য p'otso dzikţa? i: **pá**. child excellent EQU.PER TAG.ASR '(The) child is excellent, I tell you. (SN kitchen discussion)

(16.86) هَ[·] مَامِعَ، مرحماً، هَا عَلَى اللَّهُ مَا اللَّهُ مَا عَلَى اللَّهُ مَا عَلَى اللَّهُ مَا يَ عَلَى اللَّهُ عَلَى اللَّعَانَ مَا يَ عَلَى اللَّهُ مَا يَ عَلَى اللَّهُ مَا يَ عَلَى اللَّهُ مَا يَ عَلَى اللَّهُ عَلَى اللَّهُ عَلَى اللَّهُ مَا يَ عَلَى الْحَامَةُ مَا يَ عَلَى مَا يَ عَلَى الْحَامَةُ مَا يَ عَلَى الْحَامَةُ مَا يَ عَلَى الْحَامَةُ مَا يَعْنَى مَا يَ مَا يَ عَلَى الْحَامَةُ مَا يَ عَلَى مَا يَ مَا يَ عَلَى مَا يَ مَا يَ عَلَى مَا يَ مَا يَ عَلَى مَا يَ يَ يَعْلَى مَا يَعْلَى مَا يَ يَعْلَى مَا يَعْلَى مَا يَ يَعْلَى مَا يَ يَعْلَى مَا يَ مَا يَعْلَى مَا يَ يَعْلَى مَا يَعْلَى مَا يَ يَعْلَى مَا يَ يَعْلَى مَا يَعْلَى مَا يَ يَعْلَى مَا يَعْلَى مَا يَعْلَى مَا يَعْلَى مَا يَ يَعْلَى مَا يَعْلَى مَا يَعْلَى مَا يَعْلَى مَا يَ يَ مَا يَعْلَى مَا يَعْلَى مَا يَعْلَى مَا يَعْلَى مَا يَعْلَى مَا يَ مَا يَ مَا يَ يَعْلَى مَا يَ يَعْلَى مَا يَ مَ مَا يَ يَعْلَى مَا يَ مَا يَ يَعْلَى مَا يَ يَعْلَى مَا يَ مَا يَ يَعْلَى مَا يَ يَعْلَى مَا يَ عَلَى مَايْعَايَ مَايْ

In (16.87), the assertive tag is appended to a reportative form.

(16.87) هَٰ مَجُ تَابَعَتَهُ عَامَةُ مَا مَعَ مَحُ ثَامَ عَنْ عَامَ حَنْ عَامَ تَعْتَقَا عَامَةً مَنْ مَا مُعْتَقَقَ عَامَ مَا مَعْتَقَقَ عَامَ مَا مَعْتَقَقَ عَامَ مَا مَعْتَقَقَ عَامَ مَا مَعْتَقَعَ مَا مَا مَعْتَقَعَ مَا مُعْتَقَعَ مُعْتَعَ مَا مُعْتَقَعَ مَا مُعْتَقَعَ مَا مُعْتَقَعَ مَا مُعْتَقَعَ مُعْتَعَ مَا مُعْتَعَ مَا مُعْتَعَا مُعْتَعَا مُعْتَعَامَ مُعْتَعَامَ مُعْتَعَا مُعْتَعَ مَا مُعْتَعَ مَا مُعْتَعَا مُعْتَعَامُ مُعْتَعَامُ مُعْتَعَامُ مُعْتَعَامُ مُعْتَعَامُ مُعْتَعَامُ مُعْتَعَا مُعْتَعَامُ مُعْتَعَامُ مُعْتَعَامَ مَا مُعْتَعَامَ مَا مُعْتَعَامَ مَا مُعْتَعَامَ مُعْتَعَامَ مُعْتَعَا مُعْتَعَامُ مَا مُعْتَعَا مَا مُعْتَعَا مَا مُعْتَعَا مُعْتَعَ مَعْتَ مُعْتَعَ مَعْتَ مُعْتَعَا مُعْتَعَ مَا مُعْتَعَ مُعْتَ مُعْتَعَ مُعْتَعَ مُعْتَعَ مُعْتَعَ مُعْتَعَا مُعْتَعَ مُعْتَعَ مُعْتَعَ مُعْتَعَا مَا مُعْتَعَ مُعْتَعَ مُعْتَعَ مُعْتَعَا مُعْتَعَ مُعْتَعَ مُعْتَعَ مُعْتَعَ مُعْتَعَ مُعْتَعَ مُعَتَعَ مُعْتَعَ مُعْتَعَ مُعْتَعَ مُعْتَ مُعْتَعَ م

For an example of $p\dot{a}$ attached to an imperative, consider (16.88), which contains two imperatives. The speaker attaches $p\dot{a}$ to his second, emphatic appeal.

ର' ହିଁଷାତିଆରା ମସ୍ୟାର୍ଥ୍ୟ mjőː-mjőː-pøː t'a lògju? daːmè? ma-cɛ?. tam=gi now finish-RDP-2INF.GEN word=GEN story void NEG-tell ĩ. f'ak-f'ak-ø: тá kjo:-ne sù: lòtce=rã: kjap-ce heal-RDP. 2INF.EN wound repeat-COND pain again=AEMPH strike-INF EQU.PER ma-cé? **ná**. t'a lògju? tho:-tee-gam ónmu? now story NEG-tell TAG.ASR hear-PST-ATTQ PN 'Now do not in vain speak of things that have been dealt with. If you reopen old, healed wounds, it hurts again. Now do not tell (this) story, I tell you, did you hear me, Wangmu?' (nga'i 'gan 19)

In instructions, the tag $p\dot{a}$ may emphasize the obligation of the addressees:

(16.89) לישה מיז מיז איז אָר שער לעשי שן *t'izã: בּב: lèpti sún teuk go: pá.* but mouth.HON very.much guard cause be.needed TAG.ASR 'But the (patient) should be made to guard his mouth (from wrong food), I tell you.' (rnamrtog 24)

Consultant YR commented on (16.89) that if the tag $p\dot{a}$ were to be replaced by the equative copula $=p\epsilon^2$, the sentence would turn from an instruction given to others to a declaration which the speaker herself would be obliged to observe. That is, $p\dot{a}$ marks what "you have to do" whereas $=p\epsilon^2$ would mark what "we have to do." As a further illustration of the point, YR volunteered the following two clauses the first of which concerns a common obligation of the speaker and the addressee and the latter one the addressee's obligation:

(16.90) वेंग्रेन्स्य मारुषा कें क्षेन क्षेय नगेंवायना

t^horã: η *àtɛa? lokɛ? lap go:=pɛ?.* tomorrow 1PL Lhoke learn be.needed=EQU.NE 'Tomorrow we have to learn Lhoke.' (YR e)

(16.91) (أن عَن حَدَى الله المَع المُع المُ

A further example of the addressee's obligation is provided by (16.92).

pè: dun=di=lo jì: ma-t $c^h \varepsilon$ -n ε ran=gi ηà=lo story=DEMPH=DAT believe NEG-believe-COND you=AGT 1SG.GEN 1SG=DAT gja-t^hamba p'in go: tiruk ná. hundred-NUM give be.needed TAG.ASR rupee 'If you do not believe this story of mine, you have to pay me, I tell you, a hundred rupees.' (PD bet story)

Declarative $p\dot{a}$ is often pronounced with a rise in intonation, as in Figure 16.1, which provides the pitch trace from (16.93), and in Figure 16.2, which provides the pitch trace of (16.94). In (16.94), the pitch on $p\dot{a}$ is particularly high, and the clause triggered a response from the interlocutor, who said *lai* 'yes' (comes from WD angle dgs-*lags-in*).

(16.93) $\neg \hat{\alpha} \cdot \hat{\eta} \cdot \neg \cdot \hat{\alpha} \cdot \hat{x} \cdot \hat{\alpha} \cdot \hat{\alpha} \cdot \hat{\gamma} \cdot \hat{\alpha} \cdot \hat{\gamma} \cdot \hat{\alpha} \cdot \hat{\gamma} \cdot \hat{\gamma} = ji$ $p\hat{e}:=gi$ t'a $\hat{a}po$ $\hat{a}m$ $\hat{c}\hat{u}\cdot do$ $\hat{b}\hat{e}:$ $p\hat{a}$. 1SG.GEN=GEN now father mother say.HUM-IPFV EQU.NE TAG.ASR 'My father and mother used to say (like that), I tell you.' (CY interview)

Figure 16.1. Rising intonation on *pá* in (16.93)



(16.94) הישר הישר אילי (university) א שר אילי (*university*) א שר אילי אילי (*university*) א שר אילי (*t'aruŋ ŋàtci=gi junivə:siti lɛp ma-tsʰu-kʰɛn bɛ:* yet 1PL.GEN=GEN university(Eng.) arrive NEG-be.able.to-NMLZ EQU.NE *pá*. TAG.ASR '(It) hasn't yet been able to reach our university (level), I tell you.' (DR discussion with KL)

Figure 16.2. Rising intonation on *pá* in (16.94)



However, pitch is not always raised with declarative $p\dot{a}$. Figures 16.3 and 16.4, providing the pitch trace from (16.95) and (16.96) respectively illustrate a falling pitch pattern on $p\dot{a}$.

(16.95) र्चे र्डेंग दर्ने हेयायान्याया खेवा ना

p'o-tso=di dzikta? i: pta. child=DEMPh excellent EQU.PER TAG.ASR 'That child is excellent, I tell you.' (KL phone call)

Figure 16.3. Falling intonation on *pá* in (16.95)



(16.96) ད་མོལྲོག ŋà mi-cé: pá. ISG NEG-know TAG.ASR 'I don't know, I tell you.' (PTW kitchen discussion)

Figure 16.4. Falling intonation on *pá* in (16.96)



In the imperative, tags tend to have a pitch level which is raised from the previous context, see Figure 16.5 presenting the pitch trace from (16.97).

(16.97) গ্রম্বর র্মা গ্রদ রা

sém-bo nấ: **pá**. listen.HON-2INF give.HON TAG.ASR 'Please listen, eh.' (PB discussion with TB)



Further research is needed in order to determine whether raised pitch is more likely to evoke a response from the addressee than low pitch.

16.3.2 Exclamative tag (h)o:

Another commonly used tag is ho:/o:, an exclamative tag which adds exclamative force or nuance to both declarative and imperative clauses. The marker is most likely a borrowing of the Nepali equative copula ho, which functions as a similar clause-final exclamative also in Nepali. The same form also occurs as an interrogation marker, see §11.1.6. In (16.98) (h)o: is used in a declarative and in (16.99) in an imperative clause. (16.98) العائلة: المحتلم المحتلم المحتلم المحتلفة: الح

(16.99) দ ঝান্দা দ্র্যা

t'a ma-lap ho:. now NEG-say TAG.EXCLAM 'Now don't speak, eh. (KNM kitchen discussion)

Example (16.100) provides a corresponding use from Nepali (Hutt & Subedi [1999: 249], Romanization mine):

(16.100) Nepali (Hutt & Subedi [1999: 249]) *hijo maile tapāĩ aspatāl jā-na lāg-e-ko dekh-e-ko*yesterday I.AGT you.HON hospital go.INF begin-PFV-NMLZ see-PFV-NMLZ *thiẽ*, *ho*.
COP.PST.1SG COP.EQ.NPST.3SG
'I saw you going to the hospital, eh.'

In (16.101) the motivation for using $(h) \dot{o}$: was reported to be irritation of not being, at first, heard by the addressee.

(16.101) محترب مَعْرَبَ عَلَى مَعْرَبَ عَلَى اللَّهُ اللَّهُ عَلَى الْحَالَى الْحَلَى الْحَالَ عَلَى اللَّهُ عَلَى الْحَالَةُ عَلَى اللَّهُ عَلَى اللَّهُ عَلَى اللَّهُ عَلَى اللَّهُ عَلَى اللَّهُ عَلَى اللَّهُ عَلَى الْحَلَى اللَّهُ عَلَى اللَّا عَلَى اللَّهُ عَلَى الْحُ المَا عَلَيْ عَلَى اللَّهُ عَلَى اللَّهُ عَلَى اللَّهُ عَلَى اللَّهُ عَلَى اللَّالَةُ عَلَى اللَّهُ عَلَى اللَّ

Another context is emphasizing disagreement:

(16.102) 动页: 密方: 浙 *mèmbe? ó:*. NEG.EQU.NE TAG.EXCLAM '(It) is not (that), eh.' (PT, oh)

With the tag (h)o: the pitch is usually raised from the previous context, but the rise may be marginal, as shown in Figure 16.6, which presents the pitch trace of (16.103).

(16.103) $\check{h} \check{h} \check{h} \check{h}$: $\check{g} \check{h} \check{h} \check{h} \check{h} \check{h}$: $i \check{h}^{h} u k \varepsilon = ra$ $k j a p - t \varepsilon \varepsilon$ \acute{o} :. 3PL Nepali=AEMPH speak-PST TAG.EXCLAM 'It was Nepali they spoke, eh.' (CY interview) Figure 16.6. Intonation with the tag (*h*)o:



16.4 Discourse particles *t*'*a* and *t\varepsilon*

The particle t'a 'now' is a reduction of the fuller form t'ato 'now'. Whereas the full form has timereferential lexical content, t'a is used somewhat similarly to the English *now* in clauses such as *Now, I've got something to tell you*, where *now* has a discourse function rather than a timereferential function. In line with their distinctive functions, t'a and t'ato often co-occur in the same clause, as shown in (16.104) and (16.105). The latter example also illustrates the use of the other discourse particle $t\epsilon$, which is discussed below.

(16.104)	ન' યવિત્ર' મુનષ' સેંગો' ન'ર્સે બાં નજી લયાન' બન હતા								
	t'a	pén	kjap-tiki	t'ato	lò	tsu-t ^h amba	lầ:-tsʰaː.		
	now	wedding	do-NF	now	year	ten-NUM	arrive-CPML		
	'Now it's ten years since (they/he) got married.' (KT life story)								

(16.105) $\vec{p} \cdot \vec{\gamma} \cdot \vec{g} \cdot \vec{p} \cdot \vec{\gamma} \cdot \vec{q} \cdot \vec{q} \cdot \vec{r} \cdot \vec{q} \cdot \vec{q}$

The particle t'a can occur clause-initially (16.106), medially (16.107) and finally (16.108). In spoken language, the particle may also occur more than once in a clause, as shown in (16.108).

(16.106) جَ مَنْ طَعَمَّرَ جَ مَامَّ المَعْرَمَةِ مَنْ المُعْمَرَةُ المُعْمَرَةُ مَنْ المُعْمَرَةُ مُعْمَرًا المُعْمَرَةُ المُعْمَرَةُ المُعْمَدَةُ مُعْمَدَةُ المُعْمَدَةُ المُعْمَدِينَةُ المُعْمَدَةُ المُعْمَدَةُ المُعْمَدَةُ المُعْمَدَةُ المُعْمَدَةُ المُعْمَدَةُ المُعْمَدَةُ المُعْمَدَةُ مُعْمَدَةُ مُعْمَدَةُ المُعْمَدِينَةُ المُعْمَدَةُ المُعْمَدَةُ المُعْمَدَةُ المُعْمَدَةُ مُعْمَدَةُ المُعْمَدِينَةُ المُعْمَدِينَةُ المُعْمَدَةُ المُعْمَدِينَةُ المُعْمَاتُ المُعْمَدِينَةُ المُعْمَدِينَةُ المُعْمَدِينَةُ المُعْمَاتُ المُعْمَاتُ المُعْمَاتُ المُعْمَاتُ المُعْمَاتُ المُعْمَعْمَةُ المُعْمَعُ مُعْمَعُمَةُ المُعْمَاتُ المُعْمَاتُ المُعْمَاتُ المُعْمَعُمَةُ المُعْمَعُمَةُ المُعْمَاتُ المُعْمَاتُ المُعْمَاتُ المُعْمَعُمَةُ المُعْمَاتُ المُعْتَقَاتُ مُعْمَاتُ المُعْمَاتُ المُعْتَقُعْتُ المُعْمَاتُ المُعْتَقَاتُ المُعْتَقَاتِ المُعْمَ

(16.107) $\tilde{a} \subset \bar{b} \circ \tilde{c}^{*} \tilde$

(16.108)- २' ऑर्'ने' क्रिंग केंद्र गवित मर्कत मार्गों पर्ने' मर्मा क्रां में केंद्र न्यों के स्वार्थ केंद्र न्यों के स्वार्थ स्वार्थ के स स्वार्थ के स t'a ódi nimts^hi nù:nents^h \tilde{e} : máko=di t^hamtee=lo nò tøn now that date relative son-in-law=DEMPH all=DAT face show t'a.453 $b\epsilon^2$ **30:-6** be.needed-INF EQU.NE now 'Now on that day the groom has to be shown to all relatives.' (SGD wedding customs)

The particle $t\varepsilon$, which was already preliminarily illustrated in (16.105), signifies little more than that the speaker continues to speak. Its meaning can be approximated by some uses of the English words *well, then, so*⁴⁵⁴, *moreover*. Just like t'a, the particle $t\varepsilon$ occurs clause initially, medially and finally, and may occur more than once. For an initial use, see (16.109), and for medial and final uses, consider (16.110), which has two instances of $t\varepsilon$ in one clause.

- (16.109) \vec{p} \vec{w} $(\vec{a}\vec{r})$ \vec{g} $(\vec{u} \cdot \vec{a} \cdot \vec{r})$ $\vec{a}\vec{q}$ $(\vec{v} \cdot \vec{n})$ $\vec{a}\vec{n}$ \vec
- (16.110) $\overleftarrow{ar}(a_{7}^{-}, \overleftarrow{e}^{n}, a_{7}^{-}, \overleftarrow{r}, \overleftarrow{ar}, a_{7}^{-}, \overrightarrow{r}, \overrightarrow{r},$

The use of $t\varepsilon$ may also indicate topic change:

(16.111)- टाठगा के भगा नहुः र्यते कट भगा मिं नट टा भेन है तक्या में का के ता हे मुं होट हुते हैं न हो न का भाग तहे का मां में न्या के मां के मा मां के म के मां के मा मां के मां का मां के मां क मां के मां के मां के मां के मां का मां का मां का मां के मां के मां के मां के मां के मां के मां का मां के मां का म मां के मां के मां के मां के मां का मां का मां के मां का मां के मां क मां के मां के मां के मां का मां मां का मां का मा मा मां क मा मां का मां मां मां मां का मां का मां मा मा मा nàtea? míla? teu-pø: $n a \eta = l \varepsilon$ k^hu ť`ã: ηà lèpti 1pl person ten-2INF.GEN inside=ABL 3SGM and I very.much $tc^{h}am-co?$ ĩ tE *p*'*usim*=*tsu*=*i* lóbdø: agreeable-SUP younger.sister=PL=GEN school.GEN EQU.PER so $t^h \emptyset n$ - $\varepsilon \varepsilon = po^{455}$? k'unsin nàm=l ε holiday when=ABL happen-INF=EQU.NE.Q 'Among us ten, he and I are the closest. So/then, when does the sisters' (=your) school's holiday start?' (Richhi 100)

16.5 Recapitulation

Recapitulation (see Genetti 2005: 49) refers to a technique in story-telling, where the speaker repeats what was said in the previous (finite) clause. The repeated form is typically presented in

⁴⁵³ There is probably a mistake in word order here, the right order being: $t'a \ \delta di \ n \ m \ bind b = di \ n \ m \ bind b = lo$ $y \ \delta ton \ go: \varepsilon \varepsilon \ b \varepsilon^2 \ t'a.$

⁴⁵⁴ In such uses as *So, what do you think of my new jacket?*, which are not closely connected to what was said before.

⁴⁵⁵ Denjongke script has \breve{t} [po], which is probably a cliticized form of the interrogative equative *bo* (cf. declarative equative *be*?/=*p* ε ?).

the nonfinal converbal construction. The following examples come from a story of a trip the speaker had made on the previous day. The recapitulated parts are given in bold.

- (16.112). १९४१. १९४२ में में स्थान के में स्थान के साथ क сý:сv: pòptca kjap-o be?. nòptca kjap-ti óna a.bit shopping do-2INF EQU.NE shopping do-NF there óni=lo lam=tei? pò:. lam pò:. shoe=INDF buy shoe buy small.child=DAT '(They) did a bit shopping. Having done shopping, (they) bought shoes to a child. (They) bought shoes.' (DB trip story)
- (16.113) أَتَ تَعْنَ عَلَيْ عَلَى عَ مَعْلَى عَلَى عَ مَعْلَى عَلَى عَ مَعْلَى عَلَى عَ مَعْلَى عَلَى عَلَ

16.6 Right-dislocation

Typically Denjongke clauses end in a verb but occasionally clausal arguments occur postposed to the verb. In spontaneous spoken language, right-dislocation is a frequently used way for speakers to correct themselves and add information that helps the addressee to disambiguate the clause. The right-dislocated element may be a core argument, as in (16.114) or a peripheral argument, as in (16.115). The dislocated element may be but is not always preceded by a pause. The dislocated elements are given in bold.

- (16.115) বিনির্দের্য রিব দেল্যু আমর শ্রন্ দ্বমান্ট বৃদ প্রা khorlo miŋ-gju-khen be?, basti nàŋea. car NEG-go-NMLZ EQU.NE village(Nep.) inside 'Cars are not going, in the village.' (BP BB discussion)

Right-dislocation finds even written expression in the novel Richhi. Example (16.116) presents three consecutive instances.

(16.116)sà-ne sà. pa? làp-o ťã: dau man-za-ne say-2INF similar eat-COND eat NEG-eat-COND eat.AO.HUM and t^høn-z $\tilde{\epsilon}$: jò? karma=lo. become-PROG EX.PER PN=DAT 'It is happening to Karma as it says (in the proverb): "If you are about to eat, eat. If you are not about to eat, eat (anyway because you have to)."" (Richhi 65)

- b) ๑དོ་ ཞོན་ གསུམ་པོ་ ག་ལྲི་ གདོགས་གད་པོ, ལོ་ གསུམ་ལས་ ལྲག་ཆོད་ ཆོར་བཞིན་ གམ་མོ་ ད་རྒ། di nìm súm-po k'ate to:-ee bo? lò súm=le this day three-COL how contain-INF EQU.NE.Q year three=ABL laktsʰø? tsʰo:-zɛ̃: karma=lo t'ato. more.than feel-PROG PN=DAT now 'How to endure these three days, feeling now like more than three years for Karma?' (Richhi)

16.7 A note on intonation and discourse

The pitch at the end of an affirmative statement may rise as a sign that the speaker is about to continue the discourse, as shown Figure 16.7 representing the pitch trace from (16.117), the first sentence of a story.

Figure 16.7. Sentence-final rise in intonation in anticipation of continuation (16.117)



Figure 16.7 can be contrasted with Figure 16.8, which presents the non-rising intonation of (16.118), the second sentence in the same story:



16.8 Summary remarks

This chapter described discourse phenomena, with an emphasis on emphatic clitics, clause-final clitics and non-interrogative tags. It was shown that Denjongke has five emphatic clitics which bring differing but partly overlapping semantic nuance to emphasis. For instance, the anaphoric emphatic $=r\tilde{a}$: typically highlights a referent which has already been mentioned (hence the term "anaphoric") and thus activated in the addressee's mind. The demonstrative-emphatic =di has developed from a proximal demonstrative into a non-referential marker whose meaning subsumes definiteness but goes beyond it (demonstratives and personal pronouns can be marked with =di). Therefore =di was not named a definiteness marker, although that term is used for cognate morphemes in related languages.

The attention marker =co, which does not seem to have reported cognates in other Tibetic languages, was shown to have uses resembling both the categories "mirative" and "contrastive focus," underlining the similarity of the two categories (for details, see Yliniemi 2020). It was also shown that the tags $p\dot{a}$ and (h)o, which are also used for interrogation, have developed non-interrogative uses: $p\dot{a}$ can bring assertive and (h)o: exclamative nuance to a declarative proposition. The uses of the non-commitment marker ki borrowed from Nepali remain a fertile ground for future research (in both Nepali and Denjongke).
17 Notes on lexicon

This chapter describes vocabulary from five semantic domains which show particularly rich variety, such as idephones (§17.1), or are otherwise crosslinguistically or culturally interesting, such as kinship terms (§17.2), names (§17.3), colours (§17.4) and vocabulary used with small children (§17.5).

17.1 Ideophones

In this section, I first introduce the concept of idephones (§17.1.1). This is followed by a description of the various types of ideophones in Denjongke. Semantically ideophones are grouped into nonnormative ideophones (§17.1.2) and those ideophones for which normativity is not an issue (§17.1.3-5). Those ideophones for which normativity is not an issue are morphologically divided into reduplicating (§17.1.3) and near reduplicating words. The near reduplicating words are further divided into those which change vowel quality (§17.1.4) and those which change the initial consonant (§17.1.5). A somewhat distinct group are onomatopoeic ideophones (§17.1.6). There are also ideophonic, reduplicated suffixes, which can vividify adjectives (§17.1.6).

17.1.1 Introduction

Ideophones (or ideophonic adjectives and adverbs) in Denjongke are words which behave syntactically as adjectives and/or adverbs but are morphologically, phonologically and semantically distinct from other adjectives and adverbs. The morphologically distinct characteristics are reduplication, near reduplication and rhyming. The phonological distinctness consists of consonantal phenomena not attested in non-ideophones. Semantically, ideophones are more vivid representations of sensory experiences than non-ideophonic descriptions. Onomatopoeia is considered here to fall within the description "vivid," and thus onomatopoeic words are considered a subclass of ideophones (see §17.1.6). Other ideophones, however, are not as clearly connected with onomatopoeia, i.e. mimicking sounds.

The term "ideophone" was first introduced from within Bantu linguistics by Doke (1935: 118), who defined an ideophone as "a vivid representation of an idea in sound. A word, often onomatopoeic, which describes a predicate, qualificative or adverb in respect to manner, color, smell, action, state, or intensity." A more recent definition is provided by Dīngemanse (2011:25): "Ideophones are marked words that depict sensory imagery." Dīngemanse adds to Doke's definition the idea of "markedness" of ideophones. This means that ideophones stand out from the rest of the language by their phonology, morphology, syntax and semantics. Ideophones are theoretically interesting because they fall under sound symbolism, a concept that challenges one of the basic tenets of structuralist linguistics, the arbitrariness of the linguistic sign (Nuckolls 1999: 226). In Denjongke, ideophones are syntactically adjectives and/or adverbs but they have morphological, phonological and semantic characteristics which set them apart from other word classes.⁴⁵⁶

Ideophones employ reduplication, as in $ts^humts^hum \notin s_{a} \notin s_{a}$ 'opening and closing or being on and off intermittantly', or near-reduplication by vowel change, as in *barbur* $a_{a} \times a_{a} \times a_{a}$

⁴⁵⁶ For the analogous category "expressives" in the closely related language Dzongkha, see Watters (2018: 297).

Reduplicated adjectives occupy a gray area in terms of ideophonic status. Reduplication and near reduplication are the central strategies for forming ideophones. Reduplication in Denjongke (and probably in most other languages) evokes the idea of iteration and thus provides perhaps a more "vivid representation" (description used in Doke's [1935: 118] definition of an ideophone) than arbitrary sound combinations without reduplication. That would amount to saying that the reduplicated adjective *dumdum* $a_{5}a_{7}a_{5}a_{7}$ is a more vivid representation of 'short' than the non-reduplicated counterpart *dumta2* $a_{5}a_{7}a_{5}a_{7}$.

Many or most ideophones can be used both adjectivally and adverbially, as shown by (17.1), where (a) is adjectival and (b) adverbial. In the examples below, the ideophones expressing nonnormative qualities and behaviour are glossed simply as IDEO.NN and approximately translated in the free translation.

(17.1) a) ลิ ลริ อาชัก ซา

mi=di te^hapte^hop $b\epsilon 2.$ human=DEMPHIDEO.NNEQU.NE'That man is reprehensible.'(KN e)

b) אַסאי ראָלָי דּאַדיאָדאָי אָאַקיאָן sàm=di tɛʰaptɛʰop man-za. food=DEMPH IDEO.NN NEG-eat 'Do not eat the food inconsiderately.' (KN e)

The ambiguity of ideophones with respect to adjective vs. adverb distinction is further illustrated by (17.2), where sabzap = ar = ar' = ar' 'careful(ly)' is in (a) used alone adverbially and in (b) with the adverbializer p'jati = fati = ar', which turn adjectives into adverbs.

(17.2) a) สุราสุรา ซู้รา

sàbzap dø?. careful sit 'Remain careful(ly).' (KN e)

b) בְּקִיבְּק לָאִיצָּי צְּק sàbzap p'ja-ti dø?. careful do-NF sit 'Remain careful(ly).' (KN e)

⁴⁵⁷ Except for some foreign loan words.

Ideophones can even be used like nouns, as shown by (17.3) and (17.4). In both examples, (a) presents an adjectival/adverbial use of an ideophone and (b) a noun-like use of the same word.

(17.3)a) ଭିଂନ୍ତ୍ରୀ କ୍ରଣ୍ୟାନ୍ଦ୍ୟୁ ସ୍ଥାନ୍ ସ୍ଥାନ୍ ସ୍ଥାନ୍ ସ୍ଥାନ୍ ଅନ୍ତ୍ରୀ k^husimsim j*à*?. mí=tsu t^hamtce? silent(ly) EX.PER human=PL all 'All the people are silent(ly).' (Richhi 84) b) 5' क्ष' दहें रागुया का खिनी का लेगी t'ato dzinge:=na k^husimsim j*à*?. now class=LOC silence EX.PER 'Now there is silence in the classroom.' (Richhi 6) (17.4)a) ८ क्ष कर ख्वा केंबरों केंगे कुन केन पर्वे केंबर t'ato $ts^h \epsilon d\tilde{\epsilon}$: pômpo sòle **raprip** t^høn-ts^ha:. become-CMPL now considerably evening dusk dim 'It has become considerably dim with evening dusk.' (Richhi 40) b) ८.ज्या रुप रुप रुप सेपल में लेवा ĩ:. nàtea? **raprip**=na lep-o dim(ness)=LOC arrive-2INF EQU.PER 1pl

'We arrived at dusk.' (KT e)

In (17.3b), k^h usimsim \mathbb{R}^n \mathbb{R}^n \mathbb{R}^n is used like an abstract noun 'silence' as a copula subject in a locative clause, and in (17.4b), on the other hand, *raprip* 5^{n} , 7^n receives locative marking, which is typical of nouns, and the meaning becomes 'in the dimness, at dusk'.

Consider, furthermore, the riddle in (17.5), which uses three ideophones. Each ideophone is used like an abstract noun in a copulaless locative clause. The Denjongke writing and translation are preliminary.

(17.5)	ક્ષેન્ટ બેંચાયેંચા, વર્ત્ત ઘ્રુવા'ઘ્રુવા', તેંવા' ર્ઢન્ ર્ટ્તન					
	tẽ:	jòmjom,	p'ar	k ^h juk ^h juk,	ò?	ts ^h etse?.
	above	full.to.brim	middle	straight.movement	below	limit-limit(?)
	'Above	, full to the brin	n. Betwe	en, movement. Belov	w, limit(?)' (JDG fieldnotes)

The riddle in (17.5) describes water/river. The first ideophone refers to water which fills the river, the second ideophone describes the movement of the fish in the water, and the last ideophone refers to the sand at the bottom of the river.

17.1.2 Ideophones expressing nonnormativity

The phonological sequence $C1aC2.C1oC2^{458}$ (where C is a consonant or a consonant cluster and the numbers indicate that the same consonant is repeated in the second syllable) is in Denjongke associated with a quality or state of affairs that deviates from some type of norm. For instance, the

⁴⁵⁸ With reduplicated disyllabic words such as *k'abzi k'obzi*, *C1aC2.C3V3 C1oC2.C3V3*.

word *rakrok* 5π 5π 5π refers to tree trunks and roads which are not smooth/level (the norm) but bulgy/bumby. Some nonnormative ideophones are listed in Table 17.1 and exemplified after the table. The words in Table 17.1 are from consultants KL (Barapathing, East Sikkim) and KN from Martam (East Sikkim). The expressions have some local variation, as shown by the alternatives *k'abzi k'obzi* $\pi \pi$ 5π π (Barapathing) and *k'abzi k'oduŋ/hapzi hoduŋ* 5π 5π (Martam). The order of presentation is phonetic, beginning with bilabial initials and moving backwards through alveolars, palatals and velars to laryngeal fricative.

	siles expressii	gnonnormativity
раррор	শন্যর্থন	'having mixed big and small inappropriately'
pjarpjor	ર્ઘુત્ર'ર્ધુંત્ર'	'onomatopoeic for a disturbing sound of pouring liquid'
		(e.g. diarrhea or pouring water from a container to another)
praprop	র্ধ্রব:র্ধ্রব:	'in only slight resemblance of how something should be
		done, badly done'
p ^h jarp ^h jor	પર્શ્વ માલ્દ્વે મ	'clothes not appropriately worn'
tc ^h artc ^h or	ધુર્ત્ર.હે્ર.	
dzardzor	ਖਬੈਂਟ.ਖਬ੍ਰੈਂਟ.	
babbop	র্বন'র্ম্বন'	'with bulges of fat (of a fat person)'
bjarbjor	ਖਈ੍ ਮ.ਖਸ਼੍ਰੇ ਮ.	'inappropriately prepared (of food)' (e.g. mixing items not
		to be mixed or cooking unhygienically)
mjàkmjok	র্ন্ডাব্য'র্ন্ট্রিবা'	'tasteless (of tea or millet-beer)'
t ^h amt ^h om	ਬਕਾਬੱਕਾ	'not being able to function properly (of people)'
dakdok	न्गार्नेगा	'occurring as an assortment of small items of various sizes'
ts ^h apts ^h op	ಹా'హే'	'placed in an unorganized way' (e.g. pictures on the wall)
ts ^h arts ^h or	<u> </u>	'unknown sound (in the forest)' (KL)
sàrsor	র্ন:র্নুন:	'unpleasant, unclear sound (also human voice can have this
		quality during a cold)' (KL, KN)
làklok	लगा र्लेग	'unimportant enough to be able to be given away'
làplop	অন্য:র্মন্য	'lukewarm (not hot, not cold)'
rayroy	रूप:र्रेट'	'here and there (of big stones/rocks in the soil)'
rakrok, raprop	হ্যশ'ৰ্হ্যশ',	'with bulges, not smooth (of a road, floor plank, of a stone-
	হ্রন'র্জন'	wall in which some stones are loose), also <i>raja ruja p'ja</i> 5°
		עי היעי לאי 'make uneven with bulges'
tchaltchol	สญาสัญา	'doing this and that without completing anything of
		foolish behavior ("like a mad person" [KN])'
tchantchon		'in a careless and inconsiderate way (e.g. eating of a
is upis op		ni a carefess and meonsiderate way (e.g. cating, of a person)'
cáncon (KT)	สรา สีรา	'softness of leaves and hav in the forest'
sápsop (KN)	শ্বনার্থনা	solutions of leaves and hay in the lotest
nàrnor	สราส์รา	'sneak around the real subject' (also used of dog's sound)
jiaijioi		(KL)
iàrior	<u></u> ચત્ર'ર્ધેત્ર'	'uncontrolled walk of a drunk person'
kiakkiok	য্যুবা'র্য্যুবা'	'hulgy (of a tree)'
krakrok	হাঁবা হাঁবা	'onomatopoeic for sounds coming from a wrecked body'
khankhon	<u>ค</u> า คา	'with bulges (of a hilly horizon figure of land)'
n upn op		man cargos (or a miny nonzon, ngure or iana)

Table 17.1. Ideophones expressing nonnormativity

kʰjakkʰjok,	छिया छिंया.	'abnormal (of walking of a drunk or sick person)' (KN)	
gjagjok (KN)			
k ^h japk ^h jop	দ্রেন'র্ট্রিন'	'walking here and there'	
k ^h jark ^h jor	ড্রি-'র্ট্রি'-দ	'weak (of walking style or sick animal)' (e.g. the walk of a	
		drunken man) (KL)	
gapgop	নগ্রদান দের্গান,	'out of usual shape (of a tree trunk with bulges, of	
gaptagopta?	র্যান:শ্ব্যুম্বা	something drawn or written inappropriately)'	
gaptagopta?	র্য্যান:শ্বিশ্বাম্বা		
gargor	নগম নগ্	'mistaken forms in writing, bad hand-writing' (KL)	
kʻabzi kʻobzi (KL)	गनावि गॅनावि	'do something fast and inattentive of surroundings' (e.g.	
kʻabzi kʻoduŋ (KN)	यायाले यें। हुम्सा	with <i>gju</i> 'walk' > <i>k</i> ' <i>abzi k</i> ' <i>obzi gju</i> 'rush' and with <i>sà</i> 'eat'	
hapzi hoduŋ (LT) 5ናና ዓ ኝ ፟ታርጥ		> hapzi hoduŋ(lo) sà 'gulp')	
harhor	न्रर'र्न्रर'	'unplanned, fast, haphazard action' (e.g. entering	
		someone's unoccupied house uninvited)	

As shown in Table 17.1, several ideophones express bulginess. Clausal examples are given in (17.6). Note that the English translations are less vivid than the originals.

(17.6)	làm=di	<i>rakrok</i>	bɛ?	ન્યશ્ર લદ્દે. છેત્રી.છૂત્રી. જ્ઞેટી	'The road is rough.'
	¢íŋ=di	gapgop	bɛ?	विन्तः यहीः बायः ब्रेनिः झनी	'The tree is bulgy.'
	mí=di	babbop	bɛ?	थे. ५८. स्व.र्य्य. झरी	'The man is bulgy.'
	ri=di	k ^h apk ^h op	bɛ?	לי מלי ומקיואקי ארן	'The mountain-range is curvy.'459
	N=DEMPH	IDEO.NN	EQU.NE		

(17.7) รัฐานสารักา พลานคญ

do pappop man-za.stone IDEO.NN NEG-place'Do not place the stones in a disorderly manner.' (DB e)

- - b) હું ઇંત્ર ર્દીત્ર ચાર્ટ્યા

tc^hupjarpjor ma-jà.waterIDEO.NN NEG-do'Do not make disturbing sound with water.' (KN e)

⁴⁵⁹ This example is the only one which may be argued not to instantiate nonnormativity, because Sikkim is enveloped in a hilly landscape. A possible reason for this exception is that k^hapk^hop originates as a nonnormative description of something else and is then extended to hills. Alternatively, the ideophone may refer to the difficulty of travelling in steep hills.

- (17.9) after aft
- (17.10) جَعَرَ مَرْجَ ثَالَمَ عَلَمَ مَعْرَ عَلَمَ عَلَي عَلَي عَلَي عَلَي عَلَي عَلَي عَلَي عَلَى عَلَي عَلَى عَلَي عَلَ (KN e)
- (17.11) สูง คริ่าทิง สง คร์ร คร์ร จรั ซา p'um=di=gi sàm bjarbjor zo-u bɛ?. girl=DEMPH=AGT food IDEO.NN make-2INF EQU.NE 'The girl prepared the food in an inappropriate (i.e. unhygienic) way.' (KN e)
- (17.12) हा वर्तर र्ख्या र्ख्या श्चन

tc'a:=di $mj \lambda kmj o k$ $b \varepsilon ?$ millet.beer=DEMPHIDEO.NNEQU.NE'This tea is tasteless.' (KN e)

(17.13) a) मुका मेंका मेंका केंग

thamthom thøn-tsha:. IDEO.NN become-CPML '(He) has become unable to work.' (KT e)

b) लेन' वय'र्वेय' त्यु'र्धे'

lèp thamthom gju-di very.much IDEO.NN walk-NF 'walking very difficultly' (KN e)

(17.14) दर्ने रेजें कंन केंन दन्ग

di rimo ts^hapts^hop du?.⁴⁶⁰ this picture IDEO.NN EX.SEN 'These pictures are placed haphazardly.' (KN e)

(17.15) - (17.15) - (17.15) (17.15) (17.15) (17.15) (17.15)

 $\eta \dot{a}$ ts^harts^hor t^ho:-bo \tilde{t} :.⁴⁶¹ 1SG IDEO.NN hear-2INF EQU.PER 'I heard an unrecognizable sound (and was afraid).' (KL e)

⁴⁶⁰ According to consultant KUN *tharethore* used in place of *tshaptshop* could convey the same meaning.

⁴⁶¹ The nonnormativity here probably consists of the fact that the speaker does not recognize what or who made the sound.

(17.16) รุระริระ สุระสัระ ฮู่ สะจัสะ สูง (17.16) รุระริระ สุระส์ระ ฮู่ สะจัสะ สูง (17.16) รุระริระ สุระส์ระ ฮู่ สะจัสะ สูง (17.16) รุระริระ สระส์ระ สูง (17.16) รุระริระ สระส์ระ สูง (17.16) รุระริระ สระส์ระ สูง (17.16) รุระริระ สระส์ระ สูง (17.16) รุระริระ สุระริระ สุระริระ สุระริระ สูง (17.16) รุระริระ สุระริระ สุระริระ สุระริระ สุระริระ สุระริระ สุระริระ สุระริระ สุระริระ (17.16) รุระริระ สุระริระ สุระริระ (17.16) รุระริระ (17.16) รุระ (17.16 t'ant'in sàrsor p'ja-bø: da clang-clang IDEO.NN do-2INF.GEN sound 'the sound of noise-making, clang clang.' (Richhi 30) (17.17) गु. उते हर गाँधना लगा लेंगा थेंद के टार्थ लेंक नहरा nà=lo k'utci=tsa: jó? **làklok** jờː-ne lờn tãː. work IDEO.NN EX-COND 1sg=dat 2PL PL=at message send 'If you have (some) temporary work (to offer), send me a message.' (KN e) làplop be?. $t \epsilon^h u = di$ water=DEMPH IDEO.NN EQU.PER 'The water is lukewarm.' (KT e) (17.19) हॅं रूट र्रेट पर्याणी do rayroy du-ke. stone IDEO.NN EX.SEN-IN 'There are (big) stones here and there (and everywhere).' (KN e) (17.20) a) เปิลา กลารักา ทิลารัก กรุญาทิ k^him raprop ke:p du-ke. IDEO.NN house a.lot EX.SEN-IN 'There are a lot of (inappropriately) small houses, I see/saw.' (KN e) b) במי אסיאסי באי לי שאן *raprop-p*'ja ť i-u ĩ. ηά: IDEO.NN-ADVZR write-2INF EX.SEN I.AGT 'I wrote it haphazardly.' (KN e) (17.21) नुम्र' दन् कवाकेंवा र्युष द्यु र्थनार्य भना tchaltchol-p'ja $d\varepsilon$ - $b=b\varepsilon$?⁴⁶². p'um=digju girl=DEMPH IDEO.NN-ADVZR go stay.2INF=EQU.NE 'The girl keeps going (around) out of her mind.' (KN e) (17.22) a) สิ่งสุรี ฮสาชัส ซุร mí=di tchaptchop be?. human=DEMPH IDEO-NN EOU.NE 'That man is reprehensible.' (KN e)

⁴⁶² This form comes from either gju dø: -po bε? ດຫຼູ ອັງ ະັງ ອັງ dø: jø-po bε? ດຫຼູ ອັງ ເພັງ ເຫັງ ຫຼື

- b) אָאָי מָלְ׳ אָאָיאָדי אָאָיאן sàm=di tchaptchop man-za. food=DEMPH IDEO-NN NEG-eat 'Do not eat food inconsiderately.' (KN e)
- (17.23) $\frac{3}{9}$ 'g' ch^2 ' $aqqweat equivient quartity <math>\frac{3}{2}$ 'g' ch^2 ' $aqweat equivient quartity <math>\frac{3}{2}$ 'g' ch^2 ' ch^2 'g' ch^2 ' ch^2
- (17.24) مَتَرْبَعَ: مَاعَ: مَعْنَاتُمَ: *pàrpor k'an làp-o?* IDEO.NN what say-2INF 'What did you say in unclear mutter?' (KL e)

mi=di $t\epsilon^{h}\tilde{a}$: $t^{h}u\eta$ -di jarjor=lo gju- $zouk\epsilon$. human=DEMPH millet.beer drink-NF IDEO.NN =DAT go-PROG.SEN 'The man has drunk beer and walks in halting steps, I see/saw.' (KN e)

(17.26) אָר מר שַקיאַקי אָרן ciŋ=di kjakkjok bɛ?. tree=DEMPH IDEO.NN EQU.NE 'The tree is bulgy (and not straight).' (KN e)

(17.27) ماعمامه عمد محر بأما بأما فع في بار zu? thamtee? krakrok thon-diki body all IDEO.NN become-NF 'All (my) body has become wrecked...' (KN e)

(17.28) a) אואי קלי שקיאין ארן làm=di k^hjakk^hjok bɛ?. road=DEMPH IDEO.NN EQU.NE 'The road is curvy.' (KT e)

> b) דישָבּוֹיבֿוֹי מּשּוּימֿי מַשָּישּׂמָ מָשָּישׁמָ מָדָשָּ *ra:zipo làm=lo k^hjakk^hjok=lo gju-zɛn du?.* drunkard road=DAT IDEO.NN=DAT go-IMPF EX.SEN 'A drunkard is walking here and there on the road.' (KUN e)

⁴⁶³ It is not obvious how this ideophone referring to the undergrowth in the forest is "non-normative." Perhaps soft undergrowth is not considered safe or stable walking-ground.

(17.29) के दर्न पुरु हिंन र्चु र द्यु रवेव दर्ग

mi=di $k^hjapk^hjop-p'ja$ gju-zen du?. human=DEMPH IDEO.NN-AVDZR go-PROG EX.SEN 'The man is walking feebly (like a sick person).' (KT e)

(17.30) a) र्बे' वर्ने' छर'छेर' भ्रन

 $n \partial z = di$ $k^{h} jar k^{h} jor$ $b \varepsilon^{2}$. cow=DEMPH IDEO.NN EQU.NE 'The cow is weak/sick'. (KT e)

ádzo t'ariŋ sǿ:tc^hõ: dɛ?. *ze:-p* grandfather today millet.beer.HON drink.HON-2INF APP.EQU.NE átsi p^ha: ts^hu: k^hjark^hjor kjap-ti tc'øn-do. càp foot.HON a.bit thither hither IDEO.NN strike-NF go.HON-IPFV 'Grandfather seems to have drunk beer today, walking a bit unsteadily stepping here and there.' (KL e)

(17.31) धेर्गा दग्राय दर्गाय को मात्र देर्स?

jìgi **gapgop=lo** *k'an t'i-u?* letter IDEO.NN=DAT what write-2INF 'What did you write in such shapeless strokes?' (KN e)

(17.32) \widetilde{as}_{τ} η_{τ} \widetilde{e}_{τ} $\widetilde{\eta}_{\tau}$ \widetilde{e}_{τ} $\widetilde{\eta}_{\tau}$ \widetilde{e}_{τ} η_{τ} \widetilde{e}_{τ} η_{τ} \widetilde{e}_{τ} η_{τ} \widetilde{e}_{τ} \widetilde

(17.33) a) क्षेंग जरार्जरायें छेक्षे वटाम पॅटा पहटाबे

gom **harhor=lo** $p^{h}i$:-ti $\eta \dot{a}\eta \varepsilon a$ \tilde{o} : $t\tilde{a}$:- $z\varepsilon$. door IDEO.NN=DAT open-NF inside come send-PST '(He) opened the door and came in uninvited.' (KN e)

b) हु. छु.क्षुग. ५२.५ूर. च्र्य. ५८. यवु. ५८. यी

 $k^{h}u$ $p^{h}itu?$ **harhor**-p'ja hup-teen du?. 3SGM gruel IDEO.NN-ADVZR slurp-PROG EX.SEN 'He is slurping his gruel inconsiderately fast.' (KN e)

The ideophone *dakdok* 'occurring as an assortment of small items of various sizes' is exceptional in that it is not inherently negatively evaluated. In (17.34), (a) and (b) seems to have a negative context, but (c) and (d) present positive or neutral evaluations.

⁴⁶⁴ Consultant KL gave the Nepali equivalent *laddai pardai*.

(17.34)	a)	क्रे. पट्ट. वि.जल. विवा.क्रूवा.	न्त्या हेवा केंत्र केंग					
		mí=di	$k^ha=l\varepsilon$	kʰjaktsʰi?	dakdok	t ^h øn-ts ^h a:.		
		human=DEMPH	mouth=ABL	blood.in.spit	IDEO.NN	become.COMP	L	
		'Blood-clots in s	pit have come	out of that ma	in's mouth	.' (KN e)		
	c)	ई'9' द्वा देवा थें' षावर'	ৰ্শাঝ মৰ্যা নেতৃযা					
		p'ja-ca de	akdok=lo sákho	a ko:	za: dı	ı?.		
		chicken-meat ID	EO.NN grou	nd throw	set EX	K.SEN		
	'Chicken meat has been thrown on the ground in small pieces, I see/saw.' (KN							
	c) યા લેંત્ર. દ્વા સ. સ્વા દ્વા વેંગ ગળા જ્ઞેંટ. લર્ગા							
		k'apø: dome	a dakdok=lo	ta: c	dø: du?.			
		cucumber fruit	IDEO.NN=D	AT attach s	stay EX.SI	EN		
		'Cucumbers are	growing abund	antly.'	-			
	d) क्रॅ'न्ने' स्वार्स्या गोषार्यः ५८ वा							
		ópi dakdok ke	r(o) du?					
	child IDEO.NN a.lot EX.SEN							

'There are a lot of children (of various sizes), I see/saw.' (KN e)

17.1.3 Fully reduplicated ideophones

This section describes fully reduplicated ideophones (e.g. $ts^hapts^hap \, degree deg$

		· · · · · · · · · · · · · · · · · · ·		
sàbzap	ತರ.ತರ.	'careful(ly)'		
ļaplap	শ্বন'শ্বন'	'flicker (n.), show unsteady light' (of a light bulb and fire)', also		
		<i>laplap p'ja</i> 'flicker (v.)' (see also <i>pja laplap</i> in Table 17.3)		
ts ^h umts ^h um	ર્દ્યું અ'ર્દ્યુઅ'	'open and close or be on and off intermittently'		
t ^h akt ^h ak	ষণা'ষণা'	'clearly (of reading)'		
t\$c^hopt\$c^hop	ર્જેન'ર્સેન'	'right texture for chewing (of meat and some vegetables)'		
gjopgjop	মর্ট্র্যুন'মর্ট্র্যুন'	'hastily'		
dapdap	מקשימקשי	'texture of (a bit too wet) rice that sticks to ladle'		
ts ^h apts ^h ap	রর্জন'রর্জন'	'restless, (overly) quick in action' <i>tshaptshap p'ja</i> 'be hurried,		
		restless' (TB e)		
leplep	ક્ષેન'ક્ષેન'	'of scanty meat of a lean animal'		
ло̀рпор	র্নৃন'র্নৃন'	'soft, such that can be depressed with a finger (e.g. meat, skin)'		
		(opposite of <i>takta2</i> 'hard') (KT)		
burbur	ਕਰੁੱਤ.ਕਰੁੱਤ.	'bulging' (TB)' from <i>bur</i> 'spring up', <i>tc^hu bur-ce?</i> 'springing up		
		of water'		
ŗiprip	হ্যিন'হ্যিন'	'dim(ly)'		
k`ukk`uk	सुया सुया.	'moving up and down (e.g. nodding head or moving finger)'		
t ^h akt ^h ak	ষ্যা'ধ্যা'	'clear (at least of reading from memory)'		

Table 17.2. Fully reduplicated ideophones

(17.35) ผู สู้รุณ สาเลา ยุณ คาบ มิ ส์เลา

takjõ: sàpzap $p'ja=c\varepsilon=ki$ lót^ha? looking.after carefully do=INF=GEN decision 'decision to look after carefully' (Dras-ljongs gsung-gtam, for class 12, 13)

- (17.37) ລຫຼັກ ລຫຼັກ ລັດ ຊາງ ເສີ **gjopgjop** o: -ເເ hastily come-INF EQU.PER '(I)'ll come hastily.'(DB e)

(17.38) क्वेंग कर्न झ्व झ्व झ्व

 $l\partial^2 = di$ *laplap*⁴⁶⁶ $b\varepsilon^2$. light=DEMPH flickering EQU.NE 'The light is flickering.' (KN e)

⁴⁶⁵ This word may refer both to the front side (genitals) and the back side (buttock) of the anatomical bottom.

⁴⁶⁶ This ideophone is also used for expressing paleness of colour, see §17.4.2.

(17.39) דَ بَعَ: عَلَى تَعْ عَلَى اللَّهُ عَلَى اللَّالَ اللَّهُ عَلَى الْ (17.39) اللَّهُ اللَّهُ عَلَى اللَّهُ عَلَى اللَّهُ عَلَى اللَّهُ عَلَى اللَّهُ عَلَى اللَّهُ عَلَى اللَّ (17.39) اللَّهُ اللَّهُ عَلَى اللَّا عَلَى اللَّالَ عَلَى اللَّالِ عَلَى اللَّا عَلَى اللَّا عَلَى اللَّا عَلَى اللَّهُ عَلَى اللَّهُ عَلَى اللَّهُ عَلَى اللَّهُ عَلَى الْ اللَّالَكُلُولُكُلُولُكُولُكُولُولُكُولُكُولُولُ اللَّهُ عَلَى الللَّهُ عَلَى اللَّالَّالِي اللَّالِ اللَّا عَلَى اللَّا عَلَى اللَّالَةُ عَلَى اللَّا عَلَى اللَّا عَلَى اللَّالَالَ عَلَى اللَّالَةُ عَلَى الَحَالَةُ عَلَى اللَّالَال

(17.40) अर्वे गुया गुया र्मुया र्म्या र्म्या सेवा र

go k'ukk'uk p'ja-zẽ: head up.and.down do-PROG 'nodding (his) head' (mthun-sgril 17)

(17.41) हु. द्रोक. रही. जुन. हवा. हवा. वार्डिक. रहेवा.जी

 $k^{h}u$ $petc^{h}a=di$ lep $t^{h}akt^{h}ak$ $zu\eta$ du-ke. 3SGM book=DEMPH very.much clearly read.from.memory EX.SEN-IN 'He reads the scripture very clearly from memory.' (KN e)

Fully reduplicated ideophones also occur in phrasal constructions with a preceding monosyllabic element which may be either a noun or a verb, see Table 17.3. Some of the prefixed elements such as pja (meaning unclear) in pja-laplap 'eating greedily' seem tightly connected to the ideophonic suffix, whereas other elements, such as re:po 'hair' in re:po singing 'entangled hair' form a looser ideophonic syntagm which may be intervened by other words, see (17.46).

jèː-laplap	୴୶୳ୄୢଌ୕ୣୣୠୣୠ	'seeing unclearly as if alternately disappearing and
(also jè:-dapdap)		appearing' (KN), combining $j\hat{\epsilon}$: 'disappear' with <i>laplap</i>
		'flicker (n.), show unsteady light' (of a light bulb and
		fire)'
pja-ļaplap (KN)	ର୍ଗ୍ରଷଂଝ୍ଲସଂଝ୍ଲସ ^{.467}	'eating greedily, impatient behavior in front of food
p'ja(?)-lɛplɛp	र्चुषा'लेन'लेन'	consisting of stretching hands here and there to taste
	- 1	various dishes (typical of children)' (latter written form
		from Dras-ljongs gsung-gtam 45)
pu-sópsop	শ্নু'র্শ্রব'র্শ্রব'	'fluffy with hair (of skin), also associated with leanness',
		pu 'skin hair' (Dras-ljongs gsung-gtam 40)
núm-riːriː	ञ्चुम्रा रेला रेला	'smooth (of skin), also associated with fatness', num 'oil'
num-tã:tã: (KNA)	ञ्चुरुअ'यनम् 'यनम्'	(Dras-ljongs gsung-gtam 40)
k ^h øː-síːsiː	ସ୍ତ୍ରିଲ'ମବ୍ଧିଲ'ମବ୍ଧିଲ'	'chilly, cold', <i>k^hø</i> : 'chill'
hampo cờ:cø	न्त्रयार्थेः दियार्वेया	'loads of mold', hampo 'mold'
rɛːpo sìŋziŋ	শ্বমার্থা রিদারিদা	'tangled hair', <i>rɛ:po</i> 'hair'
go dzokdzok	वर्षे तह्वा तह्वा	'bowing, nodding, bending (of human, of trees in wind)',
		go 'head'

Table 17.3. Complex ideophones

⁴⁶⁷ Here, there is a mismatch with pronunciation and suggested Denjongke spelling: KN pronounced *pja* istead of *p'ja*.

(17.42) ସିଁ ହିଁ ପ୍ରଶ୍ୱାକ୍ଷମ କ୍ଷମ କାର୍ଯ୍ୟ

p'otso	pjalaplap	ma-jà.
child	impatient.eating.manner	NEG-do
'Child,	do not eat greedily.' (KN e	2)

(17.43) และสุราสุรา สสัรสา สุรา

jèː-laplap	t ^h õ∶-m	bɛ?.
disappear-intermittent	see-2INF	EQU.PER
'(I) saw (it) unclearly.	' (KN e)	

(17.44) مَعْرَا بَيْ مَجْرَ عَيْقَمَ عَمْرَ مَعْقَمَ مَعْمَ اللَّهُ عَلَى الْمُعْرَانِ الْمُعْرَانِ الْمُعْرَانِ الْمُعْرَانِ الْمُعْرَانِ اللَّهُ عَلَى اللَّهُ عَلَى اللَّهُ مَعْرَانَ اللَّهُ عَلَى اللَّهُ عَلَى اللَّهُ مَعْرَانَ اللَّهُ عَلَى اللَّهُ عَلَيْنَ اللَّهُ عَلَى اللَّهُ عَلَى اللَّهُ عَلَى اللَّهُ عَلَيْنَا اللَّهُ عَلَيْنَا اللَّهُ عَلَيْنَا اللَّهُ عَلَى اللَّهُ عَلَى اللَّهُ عَلَى اللَّهُ عَلَيْنَا اللَّهُ عَلَى اللَّهُ عَلَى اللَّهُ عَلَى اللَّهُ عَلَيْنَا اللَّهُ عَلَيْنَا اللَّهُ عَلَى اللَّهُ عَلَيْنَا اللَّهُ عَلَيْنَا اللَّهُ عَلَيْنَا اللَّهُ عَلَيْنَا اللَّهُ عَلَيْنَا اللَّهُ عَلَى اللَّهُ عَلَى اللَّهُ عَلَيْنَا الْحَالَى الْحَالَ الْحَالَى الْحَالَ الْحَالَى الْحَالَ الْحَالِ اللَّعْلَى الْحَالَ الْحَالَ الْحَالَى اللَّهُ عَلَيْ الْحَالَى الْحَالَ الْحَالَةُ عَلَيْنَا الْحَالَ الْحَالَةُ عَلَيْكَ عَلَيْنَا الْحَالَ الْحَالَةُ عَلَيْ الْحَالَةُ عَلَيْ الْحَالَةُ عَلَيْ الْحَالَيْنَا الْحَالَةُ عَلَيْ الْحَالَيْنَا الْحَالَ الْحَالَ عَلَيْ الْحَالَ عَلَيْ الْحَالَ الْحَالِ الْحَالَ الْحَالَ عَلَيْكَ عَلَيْكَ عَلَيْكَ عَلَيْ الْحَالَ الْحَالِ الْحَالَةُ عَلَيْكَ عَلَيْ الْحَالَةُ عَلَيْ الْحَلَيْ الْحَالَةُ عَلَيْكَا الْحَالَةُ عَلَيْكَا عَلَيْ الْحَالَةُ عَلَيْ الْحَالِي عَلَيْكَ عَلَيْكَ عَلَيْ الْحَالِي عَلَيْكَ عَلَيْ الْحَالِي عَلَيْ الْحَالِي عَلَيْكَانِ الْحَالِي عَلَيْ الْحَالِي عَلَيْكَ عَلَيْكَ عَلَيْ عَلَيْ الْحَالَةُ عَلَيْ عَلَيْكَامِ عَلَيْ الْحَالِ عَلَيْ الْحَالِي عَلَيْ عَلَيْ الْحَالِي عَلَيْ عَلَيْ الْحَالِي عَلَيْكَ عَلَيْ الْحَالَ الْحَالِ عَلَيْ الْحَالَةُ عَلَيْكَ عَلَيْ الْحَالِي عَلَيْ الْحَالَيْ عَلَيْكَ عَلَيْكَ عَلَيْكَ عَلَيْكَ عَلَيْ عَالَيْ عَلَيْ عَلَيْ عَالَيْ عَلَيْ عَا عَلَيْ عَا عَلَيْ عَلَيْ

- (17.45) \underline{a}_{j} רישי \underline{b}_{j} בישי \underline{b}_{j} (17.45) בישי \underline{b}_{j} בי
- (17.46) مَشَرَّهُ: مَعْنَجُهُ مَعْنَدُ أَمْ الْمَعْنَ مَحْنَ حَمَّا خَتْنَ الْمَعْنَا عَلَى الْحَالَةُ عَلَى الْحَالَةُ حَالَةُ الْحَالَةُ الْحَالَةُ الْحَالَةُ الْحَالَةُ الْحَالَةُ الْحَالَةُ الْحَالَةُ الْحَالَةُ عَلَيْحَالَةُ عَلَيْتُ الْحَالَةُ الْحَالَةُ الْحَالَةُ الْحَالَةُ الْحَالَةُ حَالَةُ الْحَالَةُ الْحَالَةُ عَلَيْحَالَةُ الْحَالَةُ عَلَيْحَالَةُ عَلَيْكَةُ الْحَالَةُ الْحَالَةُ الْحَالَةُ الْحَالَةُ الْحَالَةُ عَلَيْ عَلَيْحَالَةُ عَلَيْ الْحَالَةُ عَلَيْ الْحَالَةُ لَالْحَالَةُ عَلَيْنَالَ الْحَا الْحَالَةُ الْحَالَةُ الْحَالَةُ عَلَيْحَالَةُ عَلَيْتَ الْحَالَةُ عَلَيْحَالَةُ الْحَالَةُ عَلَيْحَالَةُ الْحَالُ الْحَالَةُ عَلَيْ الْحَالَةُ عَلَيْ الْحَالَةُ عَلَيْ الْحَالَةُ عَلَيْ عَالَةُ عَالَ الْحَالَةُ عَالَةُ عَالَيْحَالَةُ عَالَةُ عَالَةُ عَالَةُ عَالَةُ عَالَةُ عَالَةُ عَالَةُ عَالَةُ عَالَةُ عَالُ الْحَالَةُ عَالَةُ عَالَةُ عَالَةُ حَالَةُ عَالَةُ عَالَةُ حَالَةُ عَالَةُ عَالَةُ عَالَةُ عَالَةُ عَالَةُ عَالَةُ عَالَةُ عَالَةُ حَالَةُ عَالَةُ حَالَةُ عَالَةُ عَالَةُ عَالَةُ عَلَيْ حَ الْحَالَةُ عَالَةُ حَالَةُ عَالَةُ حَالَةُ عَالَةُ حَالَةُ عَالَةُ حَالَةُ عَالَةُ حَالَةُ عَالَيْكَ مَالَةُ عَلَيْ عَالَةُ عَالَةُ عَالَةُ حَالَةُ عَالَةُ حَالَةُ عَالَةُ عَا الْحَالَةُ عَالَةُ عَالَةُ عَالَةُ عَالَةُ عَالَةُ عَالَ

 $\dot{a}dzo$ $t \in ndzi\eta = gi$ go dzok-dzok $p'ja-z\tilde{e}$: grandfather PN=AGT head nod-nod do-PROG 'grandfather Tenzing, nodding his head...' (KT e)

Many reduplicated adjectives could be added to this category on the basis of the formal criteria of reduplication.

17.1.4 Near-reduplicated ideophones with vowel change

Some near-reduplicated ideophones which change the vowel are listed in Table 17.4 and illustrated below. In the examples in my data, the second vowel is typically a high vowel (/i/ or /u/) and always higher than the first vowel. As seen in Table 17.4, many of the near duplicated ideophones of the form C1aC2.C1uC2 resemble nonnormative ideophones in that the expressed quality seems to be negatively evaluated. As the vowels *o* and *u* are closely connected⁴⁶⁸, future research should find out whether the ideophonic sequence C1aC2.C1uC2 should also be described as nonnormative.

⁴⁶⁸ For instance, some disyllabic nouns ending in -po (\breve{x}) in writing may be pronounced either as -po or -pu.

<i>raprip</i>	হ্যন'হ্যিন'	'dim, dimness' <i>rapripna</i>
barbur	ਕਰਸਾਕਰੁਸਾ	'knotty, buckled, with bulges, uneven when it should be even' (KN, KL), collocates at least with $\frac{1}{2}$ <i>lúk</i> 'cast (of metal), pour', <i>barbur lúk-tsha-ke</i> 'it has been cast bulgy (not even)'
damdum	तन्त्रातन्त्रा	'short'
làblɛp	শব-শব-	'in small pieces (of wood)'
gragrik	तर्च्या'तर्च्या'	'uneven (of unpleasant texture of meat in the mouth, of the tactile
		feeling of non-smooth, rough stone surface)' (KT)
ļakl <i>ɛ</i> k	क्षया क्षेया.	'mud-like, too wet texture of boiled rice or dough' (may also refer
ļakta ļekta		to a thick texture of paint) (YR, KN)
rakruk	र्त्या:उया	'(small) stones (the size of goat poo) here and there' (cf. <i>do rayroy</i>
		'big stones here and there')
dzardzir	מדימדי	'texture of ground that has been watered enough to resemble dough (but not wet enough to become mud)'
sámsum	শব্য-ম্যৃব্য'	'disorganized, unfolded (of clothes)'
tc'aktc'ik	ह्या ह्या.	'having sucked in water (e.g. marshy land)'
màrmur	ષત્ર.શેન.	'indistinctive muttering of words'
ts ^h amts ^h um	చ్ న'చ్రేన'	'distressed mental state, mental state after hearing or experiencing something unpleasant'
ts ^h apts ^h up	ಹೆन'ಹ್ರೆन'	'nervous'
t ^h amt ^h um, t ^h ameit ^h umei	ਬਕਾਸ਼ੁਕਾ, ਬਕਾਸ਼੍ਹੇਸ਼ੁਕਾਸ਼ੇ	'of a drowsy person's eyes'

Table 17.4. Near-reduplicated ideophones

The examples below first exemplify those uses which occur independently and then those which strongly collocate with a noun to which they are postposed.

(17.48) ८. रुग रुग रुग के सेन कर हो के बा

yàtea? raprip=na lep-o \hat{i} : 1PL dim(ness)=LOC arrive-2INF EQU.PER 'We arrived at dusk.' (KT e)

(17.49) אַקאָדי אָקיאָן אָקיאָן אָקיאָ? **barbur** lúk-luk-o mèmbo? dented pour-RDP-2INF NEG.EQU.NE.Q 'Hasn't (this vessel) been dented?' (KL e)

(17.50) a) ארי מקשימקשי קאָי קאָזי קאָזי קאָזין *בוח damdum tok-ti za?.* tree short cut-NF put 'Leave the tree cut in small pieces' (KN e) b) קקמוילקי הקמיתקמי למייאקי
 cápten damdum p'ja-ce? ritual short do-INF
 'make a short (religious) ritual' (KN phone call)

- (17.51) هم، معرام محرج بهرا بهرا بالمرح معرف معرف معرف المرحم (17.51) هم، معرف المرحم المرحم (17.51) هم، معرف (17.51) معرف (17.51

- (17.54) ८ २ रेट ८ थेव कॅंस कॅंस हेंब है

t'ariŋ ŋà lèp tshamtshum thøn-di today 1SG very.much afflicted become-NF 'Today I'm very distressed.' (KL e)

In the following examples, the ideophone strongly collocates with a noun which makes the meaning of the ideophone more explicit.

(17.55) यॅक्षि भया सम्राह्य मन्द्रमा

k'ola? sámsum man-za?. clothes disorganized NEG-put 'Do not leave the clothes disorganized (not folded)' (KN e)

(17.56) याक कु ह्या हेया थेंग्र यते घया हे याठेया येंग

sátc ^h a	t¢ ^h u	tc`agdzik	jờ-sơ:	t ^h a:pi=tci=lo
land	water	having.sucked.in.water	EX-NMLZ.SPAT.GEN	near=INDF=DAT
'In the n	eighbour	rhood of a piece of marshy	land' (rna-sung 30)	

(17.57) রিমর্ম র্চ্রস্র্র্র

sém ts^hapts^hup mind nervous 'in a nervous state of mind' (KT e) (17.58) ลิรา พันสุวิมา คนาคนา สญัญา สนายา สุขา

cíŋ ódɛm làblɛp gjom ba? có?. tree like.that small.pieces gather carry come 'Collect and bring such small pieces of wood.' (KN e)

(17.59) ק־דֹרִי הַי אַמייאָקי אַמיאָק־אָקַמיאָק־ אָקַרמי אָ*ץ*י t'arin nà mì:do? thamsithumsi? kjap-ti

today 1SG eye drowsy do-NF 'Today my eyes are drowsy (and)...' (KN e)

The partly reduplicated ideophonic construction k'jotip k'jorun / k'jorum k'jotip, whose etymology and literal meaning are uknown to me, is an intensified quantifier approximating the meaning 'so many/much, crammed with'.

(17.60) a) ך حَد بَعْ عَ عَن بَعْ جَ عَ عَ عَن مَن الله عَن مَن الله عَن مَن الله عَن مَن الله عَن مَ مَ الله عَن مَ مَ الله عَن مُ

t'ariŋ óna mí **k'jotip k'jorum** $\tilde{\delta}$:-*ts^hak* ε . today there human very.many very.many come-CMPL.APH 'Today so many people came there.' (KUN e, imitating speech of Tashiding)

t'ariŋ sèu k'jorum k'jotip tã:-ts^ha:. today hail very.much very.much send-CMPL 'Today it hailed heavily.' (KUN e, imitating speech of Tashiding)

17.1.5 Rhyming ideophones

In rhyming ideophones, reduplication is accompanied by a change in the initial consonant of the reduplicated syllable, while the rest of the syllable is fully reduplicated. The change of consonant in the second syllable results in an independently meaningless syllable. For instance, the word $k' atep = \sqrt{2}\pi'$, 'how' functions as base for the ideophone $k' atepmatep = \sqrt{2}\pi'$, which has a rhyming but independently nonsensical second part *-matep*.

Table 17.5. Rhyming ideophones

k'atepmatep	गु'झेन् अ'झेन'	'(able to do) one way or another, somehow'
ínt¢imint¢i?	ਲੇਕ ਠੇਧਾ ਕਕ ਠੇਧਾ	'at any cost, necessarily'

Rhyming ideophones are exemplified in (17.61-63). Example (17.61) was an apology for mistakes in telling a folk-story.

(17.61) तरीया द्वांया क्षे मा तरी कया केंपा या क्षेपाया क्षेपा क्षेत्र उत्त ता क्षेत्र कुया रहा येंग यायता क्षेत्र क्षेत्र

tchaptchop k'atematep *î*:-ruŋ p'ja-ti ŋà=di di: do-NF 1SG=DEMPH IDEO.NN in.whatever.way EQU.PER-CONC this.AGT lenge:=tsu=lo súŋ=di $p^h y$:-do ĩ. nà: here PRN.HON=PL=DAT story=DEMPH offer-IPFV EQU.PER 'Therefore I here offer to you this story, although it's inadequate and haphazardly made.' (KT animal story)

(17.62) कुन अंतरे में म गांसेन कासेन हुंका म जेवा नरुम भेता

gjamø: lò ŋà k'atɛpmatɛp-p'ja ŋà lòk tɛa:-ɛĩ:. latter year 1SG in.whatever.way-ADVZR I return come.HUM-NPAST.PER 'A year after, I will return, whatever it takes.' (DB e)

17.1.6 Onomatopoeic ideophones

Onomatopoeic ideophones differ from other ideophones in that they more clearly mimic real sounds and are often appositional. Like with other ideophones, reduplication is typical of onomatopoeic words. Whereas other ideophones are syntactically adjectives or adverbs, onomatopoeic words often fall outside the basic clause structure. For instance, in (17.64-70) the onomatopoeic words given in bold are appositional eloborations of the underlined nouns. Note that example (17.68) evokes not necessarily only the sound but also the feeling of the wind.

(17.64) वगवार्ग्वत मुंखूद र्हुते क्मूट ठार्र्य हेरेया

nà:tsẽ:	p'itcuŋ=tsu=i	<u>ke?</u>	tcarap-tcirip
forest	small.bird=PL=GEN	sound	tweet-tweet
'sound o	of the forest birds, tw	eet tweet	t' (Richhi 1)

- (17.66) چَاخَدَ مِعْمَرَ مَعْمَرَ مُعْمَرَ مُعْمَرَ مُعْمَرَ مُعْمَرًا مُعْمَرَ مُعْمَرًا مُنْ مُعْمَرًا مُعْمَرًا مُنْ مُعْمَرًا مُعْمَلًا مُعْمَا مُعْمَلًا مُعْمَا مُعْمَلًا مُعْمَلًا مُعْمَلًا مُعْمَلًا مُعْمَلًا مُعْمَا مُعْمَا مُعْمَا مُعْمَا مُعْمَا مُعْمَعُ مُعْمَا مُعْمَ مُعْمَا مُعْمَا مُعْما مُعْمَا مُعْمَ مُعْمَا مُعْمَا

ting ting dun-pø: <u>kɛ:da</u> clang clang hit-2INF.GEN sound 'the sound of clanging, clang clang' (Richhi 1)

(17.68) कुट वा केर केर केर कुरका रेंग्टर

<u>lúŋma</u> sír sír sír kjap-o-dã: wind whoosh whoosh whoosh do-2INF-CONJ 'When the wind goes whoosh whoosh whoosh...' (KT animal story) (17.69) B' A'B' A'B' A'B' 57'4A' MT

 $k^{h}i$ **àu àu**=lo hap-ø: <u>ke2</u> dog woof woof=DAT bark-2INF.GEN sound 'the sound of a dog barking, woof woof' (Richhi 1)

 $k^{h}u$ <u>lósa?</u> $k^{h}\varepsilon kk^{h}\varepsilon k(=lo)$ kjap bak-ti gju dø: j ∂ ?. 3SGM cough cough.cough(=DAT) do carry-NF go stay EX.PER 'He keeps on walking around coughing, cough-cough.' (KN e)

Note that in (17.65) and (17.66) the onomatopoeic word is marked with dative-locative case, suggesting that the syntax treats the onomatopoeic word as an adverb.

An onomatopoeic word may co-occur with another ideophone, as in (17.71), where onomatopoeic *t'ant'in* elaborates on the nonnormative ideophone *sàrsor*, which is used like a nominal.

(17.71) รุธารุธา สุราร์รา ฮูสารณิ สู

t'ant'insàrsorp'ja-wø:daclang-clangIDEO.NNdo-2INF.GENsound'the sound of noise-making, clang clang.' (Richhi 30)

For an example of onomatopoeia in quoted speech, consider (17.72).

(17.72) हे' कें' वर्दा' हे'हे' अन्न' क्वे'गे' हे'हे' अन्य क्वे' द्याव र्य कें

t ε *m* \dot{o} =*di hehe l* $\dot{a}p$ -*tiki hihi l* $\dot{a}p$ -*ti ga*-*u*=*lo*. so 3SGF=DEMPH haha say-NF hehe say-NF laugh-2INF=REP 'So she laughed saying haha, saying hehe, so the story goes.' (RS driver joke)

In (17.73), the onomatopoeic expressions are used like adjectivals in a verbless clause.

(17.73) طِعلام عَرْبَ عَلَى اللَّهُ عَرْبَ عَلَى عَلَى عَلَى عَلَى عَلَى عَلَى عَلَى عَلَى اللَّهُ (17.73) على المُعام عَلى اللَّهُ عَلَى اللَّالَةُ عَلَى اللَّهُ عَلَى الْعُلَى اللَّهُ عَلَى الْ الْعَلَى اللَّهُ عَلَى اللَّ اللَّهُ عَلَى الْحَالَةُ عَلَى اللَّهُ عَلَى الْعَلَى اللَّهُ عَلَى اللَّهُ عَلَى اللَّهُ عَلَى الْحَلَى اللَّهُ عَلَى الْحَلَى اللَّهُ عَلَى اللَّهُ عَلَى اللَّهُ عَلَى الْحَلَى اللَّالَةُ عَلَى الْحَلَى اللَّهُ عَلَى اللَّالَةُ عَلَى اللَّهُ عَلَى اللَّهُ عَلَى الْحَالَةُ عَلَى اللَّهُ عَلَى الْ المَالِي اللَّالِي اللَّهُ عَلَى اللَّالِي عَلَى اللَّهُ عَلَى اللَّهُ عَلَى اللَّهُ عَلَى اللَّهُ عَلَى الْحَلْ عَلَى الْحَلَى الْحَلَى الْحَالَةُ عَلَى اللَّهُ عَلَى اللَّهُ عَلَى عَلَى الْحَلَى الْحَلَى الَقَلْ عَلَى الْحَالَةُ ع

17.1.7 Ideophonic suffixes

ন্দ্ৰ্ইন্ tc^høki? ŋò ts^ha-ti nénlo mì ťã: pìmø: t'ø:=di: dõ: face be.hot-NF before fire and sun.GEN PN heat=DEMPH.AGT face má:-ri:ri: tenlo t'ato p'jaca=i *tam=di*: above now chicken.meat=GEN word=DEMPH.AGT red-IDEO màx-rixrix t^høn. *lèpti=ra* very.much=AEMPH red-IDEO become 'Being ashamed, Choki's face which was earlier made red by the heat of the fire and the sun gets now very red because of the word about chicken-meat.' (Richhi 90)

Reduplicated ideophonic suffixes, which are frequently used with colour terms, are also used in some other contexts. Example (17.75) illustrates three ideophonic suffixes, which are used with the noun $dzum \operatorname{agar}$ 'smile'. One of them is $-ri:ri: \widehat{\pi}:\widehat{\pi}$, the other two, $-m\hat{e}:m\hat{e}$ (or $-m\epsilon rm\epsilon r$) $\widehat{\operatorname{agr}}:\widehat{\pi}:\widehat{\pi}$ and $-\epsilon\epsilon k\epsilon\epsilon k \operatorname{agr}:\widehat{\operatorname{agr}}:\widehat{\pi}$. I have not come across other words than dzum which use the latter two suffixes.

(17.75) a) dzum-riːriː দেইজাইন্টে 'smilingly' b) dzum-meːmeː দেইজারিম্জিম্ 'smiling secretly' c) dzum-cekcek⁴⁶⁹ দেইজান্মিয়া 'smiling very widely'

For -mè:mè: in clausal content, consider (17.76).

(17.76) هَٰי ק'זָּה' ﻣִבָּאַ אִדיאָד' שְׁק' מִידָה' קָה' אַק'מִק'אָן mù t'ariŋ dzum-mè:me: jŵ?, átã: t'ã: man-dou 3SGF today smile-IDEO EX.PER always and NEG-similar 'She is smily today, unlike usually.' (Richhi 148)

(17.77) عَتْرَ اللَّامَاتِ اللَّذِي اللَّامَاتِ اللَّذِي اللَّامَ اللَّاتِ اللَّامَ اللَّاتِ اللَّامَ اللَّاتِ اللَّامَ اللَّاتِ اللَّٰ اللَّاتِ اللَّاتِ اللَّٰ لَالَةُ اللَّٰ لَالَةُ اللَّٰ اللَّٰ لَقَاتِ اللَّٰ لَالَةُ اللَّٰ اللَّٰ اللَّٰ لَقَلْقَاتِ اللَّٰ لَقَلْقَاتِ اللَّٰ لَقَلْقَلْقَاتِ اللَّٰ لَقَلْقَاتِ اللَّٰ لَقَلْقَاتِ اللَّقَاتِ اللَّٰ لَقَلْقَاتِ اللَّٰ لَقَلْقَاتِ اللَّقَلْقَلْقَاتِ اللَّقَلْقَلْقَاتِ الْحَاتِ الْحَاتِ اللَّقَلْقَلْقَلْقَلْقَاتِ اللَّقَلْقَلْقَاتِ اللَّقَلْقَاتِ الْحَاتِ لَقَلْقَاتِ لَالَةُ لَالَةُ لَقَلْقَاتِ الْحَاتِ لَقَلْقَاتِ لَقَلْقَاتِ لَقَلْقَاتِ لَقَلْقَاتِ الْحَاتِ لَيَّ لَالَةُ لَقَلَاتِ لَقَلْقَلْقَاتِ لَ

The suffix -tõ:tõ: occurs in several other adjectives/adverbs than the colour terms:

(17.78)	a) <i>hã:</i>	55.	'feel loneliness'	>	hãːtõːtõː	'feeling of loneliness'
	b) dzam	त्ह्रस.	'be easy'	>	dzamtõ:tõ:	'easy, easily' (cf. dzamta?)
	c) <i>nɛn</i>	ञ्चेत.	'be sweet to hear)'	>	nentõ:tõ:	'soft(ly) (of speaking)' (cf. nenta?)

⁴⁶⁹ This expression used in a song made by a speaker from Tashiding, but consultant KN (Martam) does not recognize the form.

For clausal uses consider the following:

- (17.80) (תי) אַבָּאידָה־אָדָר אַרן (*kha) dzam-tõ:tõ:* mouth easy-IDEO

mouth easy-IDEO speak 'Speak easily.' (KN e)

làp.

(17.81) $(\arg \tau) \arg \tau \tau \tau$ $(k\varepsilon)$ $pen-t\tilde{o}:t\tilde{o}: làp.$ language easy-IDEO speak 'Speak softly.' (KN e)

17.2 Kinship terminology

Denjongke kinship terminology is presented here under generational headings with comments in between. Many kinship terms begin with \dot{a} , reflecting the historical prefix a-, which occurs in many Tibeto-Burman languages (Matisoff 2003:105). The description is divided between consanguineal relatives (one's own blood-relatives and their spouses), see §17.2.1, and relatives through marriage (in-laws), see §17.2.2.

17.2.1 Consanguineal relatives and their spouses

Kinship terms for the ego's grandparents' generation and beyond are given in Table 17.6. Terms for consanguineal (related by blood) relatives are given first, followed by terms for relevant affinal relatives (consanguineal relatives' spouses). Note that the paternal vs. maternal distinction in Table 17.6 is based on a difference in affinal terms, while the consanguineal terms are invariant.

Tuble 17.0. Leo s grandparents gen	neration and beyond	
grandfather	ádzo wiĘ:	
grandmother	áno wiž	
great-grandfather	tc'odzo Ĕ·Ĕ·	
great-grandmother	pòpo šŕš·	
forefather	p^hadzo दर्) ह	
paternal grandparent's elder brother	ádzo bompu জ'Ĕ' ૹ૾ૣૼ૱'ਚ;	wife: áno bompu জ'র্নৃ' খ্র্ররান্ড্র'
maternal grandparent's elder brother	ádzo bompu ຜາ້ະ ສິສາຍູາ	wife: ána bompu ຜາສາ ສິສາຮູາ
paternal grandparent's younger brother	ádzo teʰuŋtɕʰuŋ জ་རོ་ ཆུང་ཆུང་	wife: áno tc ^h uŋtc ^h uŋ জ་ནོ་ ཆུང་ཆུང་
maternal grandparent's younger brother	ádzo te ^h uŋte ^h uŋ ಜ [.] ಕ್. ಹ್ರದ.ಹ್ರದ.	wife: ána teʰuŋteʰuŋ জ་ན་ ཆུང་ཆུང་
grandparent's elder sister	áno bompu હ્ય'ર્ને' ફ્રેંચ'સ'	husband: ádzo अंहें
grandparent's younger sister	áno teʰuŋtɕʰuŋ ๙ลัฺ ฮฺ๛ฮฺ๛	husband: ádzo 欧芒

Table 17.6 Ego's grandparents' generation and beyond

As shown in Table 17.6, no distinction is made between maternal and paternal grandparents. Neither is there a difference in terms between one's grandfather's and grandmother's siblings (hence "grandparent" in Table 17.6). Among grandparents' siblings' spouses, however, a distinction is made based on whether the grandparent is patrilineal or matrilineal, hence *ápo* and ána respectively.

Whereas matrilineal and patrilineal terms are identical in the ego's grandparents' generation, ego's parent's siblings have different matrilineal and patrilineal terms, see Table 17.7 and 17.8. On the mother's side, an age distinction (elder or younger) occurs with sisters but not with brothers. On the father's side, on the other hand, an age distinction is made between brothers but not sisters.

Table 17.7. Matrilineal par	ents' generation			
mother	<i>ám</i> கலா, <i>áma</i> கான, <i>jùm</i> யூகா (hon.)			
mother's brother	$\dot{a}z ilde{a}$:, $\dot{a}z ilde{o}$: where	wife: <i>ápi</i> डा'द्रे', डा'द्रे'		
mother's elder sister	<i>ámbom</i> জন্মস্থ্রন husband: <i>ába bombu</i> জ			
		<i>ápo bompu</i> జాచా జ్ఞేశాన్రా		
mother's younger sister	ámtsuŋ warg<	husband: áku बाहु		
Table 17.8 Patrilineal pare	nts' generation			
Table 17.0. Faithinear parents generation				

Table 177 Matuili +~ [,]

Table 17.8. Patrilineal parents' generation				
father	<i>ápo</i> జాహా, <i>jàːp</i> ఆనా (hon.)			
father's elder brother	ápo bompu क्षर्रेंग क्वेंकासु	wife: <i>ámbom</i> জন্মস্থ্রন		
father's younger brother	áku w _{IG} .	wife: <i>ámtɛuŋ</i> জন্সস্তৃদ		
father's sister	ápi w?ð	husband: tsʰoː హాస్ (áku జ్యూ) ⁴⁷⁰		

⁴⁷⁰ Consultant KT (Bermeok) accepts *áku*, which is used for mother's younger sister's husband, as an alternative term for *tsho*:. Consultant TB (Ralang), however, accepts only *tsho*: here.

When a kinship term does not express birth order, e.g. dpi 'father's sister', it may be supplemented by the attributes gempo 'old one', $te^huygo/te^humbo/te^huyte^huy$ 'small one', p 'amo 'middle-one' and the Nepali terms maili (fem.) / maila (masc.) 'second in birth', saili (fem.) / saila (masc.) 'third in birth', kaili (fem.) / kaila (masc.) 'fourth in birth'. For instance, father's eldest sister may be called áni gempo, father's youngest sister áni te^huyte^huy and father's second sister áni maili. According to a consultant from Tashiding, in families where two or more brothers have sexual relations with the same woman, the children may address their legal father as aku 'father's younger brother' instead of ápo 'father'.

Table 17.9 presents terms for (typically) one's own generation.

elder brother, elder male cousin	áteu জऱ्हु [.] , ágja জ [.] मु [.]	wife: màtei? ज्ञ'डेग्'
younger brother, younger male cousin	pynlo ह्युत्र'र्ये'	wife: <i>nám</i> अवृत्रा
elder sister, elder female cousin	ázi w ते, ái अते, जे	husband: <i>ts^ho:</i> హానా
younger sister of a man, younger female cousin of a man	p'usim हु' होम्ब	husband: álo अर्थे
younger sister of a woman, younger cousin of a woman	nùm तुञ	husband: álo ल'र्वेंग
mother's brother's daughter	ápi wizr	husband: tsho: র্স্রন্ট
husband	p ^h ogja? र्दे' कुष्'	
wife	moby? कॅ.सून	
maternal uncle's son	átcu জন্ত্, ágja জন্ত্,	wife: màtci? ब्र'हेग्',
	ázãː tɛʰuŋtɛʰuŋ ʁvːནང་ ཆུང་ཆུང་,	ápi ङान्ने
	eā̀ːteʰuŋ འང་ཆུང་	
guardian brother	phami चासे.	

Table 17.9. Ego's own generation

As can be seen in Table 17.9, maternal uncle's son has, in addition to the typical $\arg disting distance distanc$

Classical Tibetan makes several distinctions in kinship terminology based on the gender of the ego, resulting in such forms as *phubo* 'older brother of a man', *nubo* 'younger brother of a man', *phumo* 'older sister of a woman' and *numo* 'younger sister of a woman' (Beyer 1992: 158). In Denjongke these distinctions based on ego's gender are reduced to one pair: *nùm* 'youger sister of a woman' and *p'usim* 'younger sister of a man'. Consequently, a man can address a woman younger than himself by saying

(17.82) جَاهَ مَعْنَى جَعْنَ جَعْ

Relatives (typically) younger than the ego are given in Table 17.10.

Tuble 17.10. Generations younger	thun 050	
son	p'u 5;, sé? 5/5/ (hon.)	wife: <i>nám</i> अत्रत्व
daughter	p'um zar, sé:m zywar (hon.)	husband: <i>màko</i> लग्ग [.] र्गे
child	p'otso ર્ને.*. ópi સૅન્ફે	
grandson, cousin's son, nephew	<i>ts^hap^hju?</i> र्ळ:सुग	wife: <i>nám</i> अत्रत्वा
granddaughter	ts ^h am ⁴⁷¹ र्ळञ'	husband: <i>màko</i> लग्ग [.] र्गे
niece, cousin's daughter	<i>ts^ham</i>	husband: màko अपा'र्गे', álo आ'र्थे'
male descendant	<i>jàpʰju?</i> ⁴⁷² ur:र्स्रवा [.]	wife: nám ลเส เส ลง
female descendant	jàm wa	husband: màko अया गॅ

Table 17.10. Generations younger than ego

Parents call their children either by name or by endearing terms such as *bhaitcuŋ* (combining the Nepali word for younger brother *bhāi* and the Denjongke word *tc^huŋ* 'small'), *sémla:* (honorific word for daughter followed by the honorific marker =*la:*), *p'otcuŋ* 'small child' and *bahinila:* (Nepali for 'younger sister' followed by the honorific marker). The word *bhaitcuŋ* has become a personal name of several Denjongpo men, who presumably are youngest sons in their families.

Consanguineal kinship terms are also used to refer to and address strangers, see Table 17.11. A few more general terms are also included in the list.

⁴⁷¹ Consultant KT (from Bermeok) commented that in his speech variety $ts^hap^hju^2$ is used both for grandson and granddaughter and that ts^ham , the word here given for 'granddaughter', only has the meaning 'niece'.

⁴⁷² The terms $jap^h ju^2$ and jam come from consultant TB (Ralang). Consultant KT (Bermeok), on the other hand, used the term $ts^h ap ju^2$ for male and female descendant beyond the grandchildren's generation.

Table 17.11. Terms for addressing strangers of various	ages
young child (appr. 0-6 years, male or female)	óni ší g.
child (0-15 years, male or female)	p'jaby: র্র'ন্রন্থ
child, underaged (appr. 0-18 years, male or female)	p'otso ন'ৰ্স্ত
female younger than oneself (said by man)	p'usim नु'ञ्चेम./नु'ञ्चेम्ब
female younger than oneself (said by woman)	nùm सुन
male younger than oneself	pynlo ह्युत्र'र्ये
female slightly older than oneself	ázi urð, ái uð
male slightly older than oneself	áteu urzy, ágja urzy
female approximately one's mother's age	<i>áni</i> લખ્તે, <i>ámtcun</i> લગ્ન <i>જ્⊏, ámla:</i> લગ્ન ભગ
male approximately one's father's age	áku wrg
female approximately one's grandmother's age	áno wiz.
male approximately one's grandfather's age	ádzo wě
nun	ánila: अन्ते व्यम्
monk	<i>lám(a)la:</i> ลูສ'ณฑุส', <i>jà:pla:</i> ພາສຊາ (مارم)

Table 17.11. Terms for addressing strangers of various ages

The term $u_{T,T'} u_{T,T'} j \dot{a}$: *j* \dot{a} : *j* $\dot{a}:$ *j* $\dot{a}:$ *j* $\dot{a}:$ *j* $\dot{$

17.2.2 One's spouses relatives

The closest of ego's in-laws are listed in Table 17.12.

Table 17.12. Spouses relatives		
father-in-law	ápo k 'joːp జూష్' స్పేశునా	
mother-in-law	<i>ám gjum</i> ક્ષર્ય સુયાસ	
grandfather-in-law	ádzo k'jo:p జుక్ గ్రేశానా	
grandmother-in-law	<i>áno gjum</i> జుశ్ర _{ప్రై} లానా	
wife's older sister	ázi gjum ख'ति' झुगठा	husband: tsho: र्झन्त
wife's younger sister	nim ฐิรุส	husband: pynlo झुन कॅ, áteu अन्हु
wife's older brother	áteu k'joːp ตาธูา ฏัญญา	wife: ázi (gjum) wig. (भूगम.)
wife's younger brother	kipu myy, myy ži	wife: nùm ajar, nám aja car

Table 17.12. Spouses relatives

As can be seen in Table 17.12, the terms k'jo:p 'male in-law' and gjum 'female in-law' are used in forming affinal relational terms from consanguineal ones. There are, however, two notable exceptions, pim 'wife's younger sister' and kipu 'wife's younger brother', the two exclusively affinal relational terms which are not used for consanguineal relatives. Other terms referring to the relatives of one's spouse are generally formed in the same way as one's own relatives and supplemented by k'jo:p for male and gjum for female relatives, e.g. $g_{V:T'} \notin g_{V:T'} \notin g_{V:T'} dba bompu k'jo:p$ 'wife's mother's elder sister's husband, wife's father's elder brother'. However, there are a few exceptions. The word ts^{ho} cannot be supplemented by k'jo:p (* ts^{ho} : k'jo:p), thus making its meaning by definition ambiguous between one's own (parent's) elder sister's husband and one's wife's (parent's) elder sister's husband.

17.3 Names

Denjongpo personal names typically consist of two parts, both usually disyllabic, e.g. p^hurba ts^herin^{473} , nim(a) ts^herin , dawa námge:, karma kunna, pasan lamu, sónam doma. The first name is often, for instance in p^hurba ts^herin and nim(a) ts^herin , derived from the day of the week the child was born. The names derived from week days/planets are given in Table 17.13.

14010 1711011	ames associate	a while days of the week			
za: dou	गत्र, भ्र.मू.	'Monday' (Moon)	>	dawa	
za: mìŋmar	ग्वतः वेग'न्वर'	'Tuesday' (Mars)	>	тіŋта	र्वया'न्वर'
za: ļako	ग्≒त: झैगार्गो	'Wednesday' (Mercury)	>	ļakpa	स्रैण'र्'
za: pʰurbo	গ্বন ধ্রমর্য	'Thursday' (Jupiter)	>	p ^h urba	સુત્ર'ન'
za: pasã:	শ্বর' র'শ্বন্থা	'Friday' (Venus)	>	pasaŋ	ち"を入去
za: ретро	ग्वतः झेवःर्येः	'Saturday' (Saturn)	>	рєтра	धेव'य'
za: nìm	ग्रन्तर 'तेवर'	'Sunday' (Sun)	>	pìm(a)	'ત્રેસ'/'ત્રે'સ'

Table 17.13. Names associated with days of the week

Some names are exclusively feminine and others exclusively masculine, while still others are used as names for both genders, see Table 17.14.

⁴⁷³ I have heard this name being reduced to $p^{h}urts^{h}i\eta$ by people who use it often to refer to their relative.

Male names		Female names		Neutral names	
paljor	न्यन्य पर्चेन	diki?	न्दे:भ्रेद	dawa	lar Tar
	'wealth'		'bliss and delight'		'moon'
dadul	र्मा पर्नुला.	doma	र्श्वन्य स	<i>ìima</i>	ন্টি'ঝ'
	'conquerer of enemy'	-	'Tara, saviouress'	-	'sun'
tobdzor	ষ্ট্রবন্ধ'মেট্র'ম'	kipa	<u>گ</u> ار.ي.	ts ^h ɛõː	<i>ळें</i> 'न् <u>न</u> न्'
	'strength and wealth'		'delight'		'life empowerment'
dzikdal	पहेबा.र्चल.	t¢ ^h øki?	ૡ૾ૼૹૻૻ૾ૢૢૢૢૢ૽ૺૢૼ	рета	<u> </u>
	'imperishable'		'delight in dharma'	-	'lotus'
lòbzaŋ	Ří dat.	lamu	સુ ⁻ સુ-	ts ^h eriŋ	<u>ಹ</u> ೆ.ಕ್ಲ [.]
_	'good mind'		'goddess'	_	'long life'
dordzi	ξ'È.	kidøn	গ্রিন'র্শ্রীন'	karma	गह्य'
	'thunderbolt sceptre'		'delight-builder'		'action'
tobge:	ষ্ট্রনথ.ফ্রীঅ.	kilamu	શ્चેન્. સ. જ.	taci	নশ্ৰা'শ্বিশ্ব'
_	'king of strength'		'goddess of delight'	-	'good luck'
óŋgɛl	<u> </u>	lá(d)ze:∕	झ. <i>च्</i> र्य.	rinziŋ	रेव पहेंव
	'conquest'	ļándze:	'divine substance'		'precious'
p ^h ynts ^h o?,	सुव केंगरा	jáŋki?	षाधार झेंत.	all names	
pynts ^h o?	'abundance'		'prosperous delight'	based on	
				week days	
námge:	ईर्याया.मिला.	jáŋtcẽ:	षाधार ठव.		
	'all-victorious'		'prosperous one'		

Table 17.14. Names according to gender

As a last name in official documents, Denjongpos may use the pan-Tibetan term for "Sikkimdwellers," กุรลาลัการา/กุรลาลักลาร์ Denjongpa/Denjongpo (also written as Denzongpa), the generic word $\tilde{\mathfrak{h}}^{c}$ w Bhutia, which is an exonym used for all Tibetan-related groups in the Southern Himalayas, or a clan name such as an action of a second action or a clan name such as an action of the second acti Inhabitants of the northern villages of Lachen and Lachung use the place-related last names Lachenpa and Lachungpa. Clan names are discussed in some detail in Mullard (2011).

17.4 Colours

Denjongke has five commonly used basic colour terms, see Table 17.15.

Table 1/.15. Basic colour terms			
máːp(u)	నరిశారా, నరిశారా	red	
sé:p(u)	र्बरूच, बेर-धें	yellow	
ka:p(u)	দশাম্ব', দশাম'র্ম'	white	
nàku	वया'गु'	black	
nompu/nompu/hompu	జ్ ॅंब्र'र्रे'	blue/green	

Table 17 15 D a a 1 a

⁴⁷⁴ Romanizations of clan names used here are those used by some Denjongpo authors themselves.

A cause of considerable confusion for Denjonke speakers is what to make of the English terms green and blue, which may both be referred to as $\underbrace{\mathbb{E}}_{\mathbf{n}} \underbrace{\mathbb{E}}_{\mathbf{n}} nompu/nompu$. For instance, in common parlance $\underbrace{\mathbb{E}}_{\mathbf{n}} \underbrace{\mathbb{E}}_{\mathbf{n}} nompu/nompu$ may refer both to the colour the tree-leaves and the colour of the clear sky. More specific terms for 'green' and 'azure blue', deriving from Classical Tibetan, are $\underbrace{\mathbb{E}}_{\mathbf{n}} \underbrace{\mathbb{E}}_{\mathbf{n}} \underbrace{\mathbb{E}}_{\mathbf$

There are also compound expressions, which often employ one of the generic terms for colour, মর্দ্রা do?, র্স্কর্বা ts^hendo? and দ্যুনা kja?, see Table 17.16.

námdo?	यानुम्राम् या	sky-blue (lit. sky-colour)	
ts ^h odo?	अर्ळे अर्देगा	lake-blue (lit. lake-colour)	
dolom kja?	र्दे'र्थेअ' गुग्'	violet (lit. eggplant-colour)	
k`otʰɛu kja?	र्वे। चेतुः गुग	grey (lit. ash-colour)	
ts ^h alum kja?	র্ক্ত:মুঝ' শ্র্যুযা'	orange (lit.orange-coulour)	

Table 17.16. Some compound colour terms

Genitive constructions may also be used: námkhe: do? गुत्र आयति कर्त्ता 'colour of the sky' (KN e), dolomki kja? इं.सॅ.स.चु. गुया 'colour of the eggplant, violet' (KN e).

17.4.1 Clear colours

Table 17.17. Colours terms with ideophonic suffixes

sé:tõ:tõ:, sé:tã:tã:, sé:tiŋtiŋ, sé:ri:ri:, sé:si:si:, sé:hu:hu:	bright yellow, very yellow
nà:tõːtõː, nà:tã:tãː, nà:tiŋtiŋ, nà:riːriː, nà:siːsiː, nà:huːhuː	totally black
máːtõːtõː, máːtãːtãː, máːtiŋtiŋ, máːriːriː, máːsiːsiː, máːhuːhuː	bright red
kjatõ:tõ:, kjatã:tã:, kjatiŋtiŋ, kjari:ri:, kjasi:si:	clean white
notoːtoː, notaːtaː, notintin, noriːriː, nosiːsiː, nohuːhuː	bright blue/green

Note that *kjahu:hu:* 'clean white' is absent from Table 17.17. The reason is that its meaning was described as 'the (pale almost colourless) colour of the desert'. Although *nà:hu:hu:* basically describes a totally black colour, it may extend to metaphorical uses which should not be taken literally:

(17.83) न् रेन् केंब करे वाद र्डा वया कु कु के र

t'arin nìm=di k'amja nà:hu:hu: bɛ=ɛo? today day=DEMPH why totally.black EQU.NE=AT 'Why is the sky (lit. day/sun) pitch-black today?' (KN e) (17.84) གན་གྲོས་ ད་རོང་ ཆོད་ཀྱི་ གརོད་ འདི་ ནག་རྒྱ་རྒྱ་ སད་ནོ? $k'amja t'arin tc^h \phi:=ki d\tilde{o}:=di nà:hu:hu: b\varepsilon=co?$ why today 2SG.L=GEN face=DEMPH totally.black EQU.NE=AT 'Why is your face today black (of anger)?' (KN e)

In addition to the terms in Table 17.17, clear specimens of colours may be emphasized by the use of $t'angi = \sum \hat{\eta}$ and $t'ingi = \sum \hat{\eta}$, which appear to be genitivized forms of two of the forms from Table 17.13, e.g. $m\dot{a}$: $t'angi = \sum \hat{\eta}$. 'clear red'. For a clausal example, consider (17.85).

(17.85) المَاتَى المَاتَى المَاتَى المَاتَى المَاتَى المَّاتِ المُحَدَّةِ المَّاتِ المُحَدَّةُ المَّاتِ المُحَدَّةُ المَّاتِ المُحَدَّةُ المَّاتِ المُحَدَّةُ المَّاتِ المُحَدَّةُ المُحَدَّةُ المُحَدَّةُ المُحَدَّةُ المُحَدَّةُ المُحَدَّةُ المُحَدَّةُ المُحَدَّةُ المُحَدَّةُ المَّاتِ المَّاتِ المَّاتِ المَّاتِ المُحَدَّةُ مَاتِهُ المُحَدَّةُ المُحَدَّةُ المُحَدَّةُ مَاتِي مُحَدَّةً مُحَدَّةً المُحَدَّةُ المُحَدَّةُ المُحَدَّةُ مَاتِي مُعَاتِي مُحَدَّةُ مُحَدَّةُ مَاتِي مُحَدَّةُ مَاتِي مُحَدَّةُ مُحَدَّةُ مَاتِي مَا مُحَد المَاتِ المَاتِحَدَةُ المَاتِ المَحَدَّةُ مَاتِ المَحَدَّةُ مَاتَةُ مَاتَةُ مَاتَةُ مَاتَةُ مَاتَةُ مَاتَعَةُ مُ

Of the ideophonic suffixes listed above, at least $-tintin_{\beta} = \frac{1}{2} \int e^{-\frac{1}{2}} e^{-\frac{1$

17.4.2 Pale colours

The forms *-lop* or *-sa loksy:* are used for referring to pale colour terms. In Tashiding (consultant DB), the suffix *-lop* is added to the colour term, see Table 17.18.

Table 17.18. Pale colours with <i>-lop</i> (Tashiding)			
kjalop	শ্রুষা'র্ন্মন'	pale colour, colourless	
má:lop	न्यर येंग	pale red, pink	
sé:lop	ঝন'র্ন্মন'	pale yellow	
nolop	র্ছু-র্মন-	pale blue/green	

Table 17.18. Pale colours with -lop (Tashiding)

In Martam (consultant KN), the colour term is followed by a more complex construction -sa loksy: ar anguar, see Table 17.19.

Table 17.19. Pale colours with *-sa loksy:* (Martam)

10010 17717710			
máːsa lòksy:	नुस्रमः संग्रासुलः	pale red, pink	
sé:sa lòksy:	र्षेत्र'ख' सेंगु'क्षुत्य'	pale yellow	
"nosa lòksy:	ર્ક્ષે.ય. પૂર્યા.શેળ.	pale blue/green	

In Martam mere *lòksy:* can be added to complex colour terms which end in the word *kja?* (pale) colour', e.g. *dolom=gi kja loksy:* 'pale/light purple/violet' (lit. eggplant's colour *lòksy:*). The construction COLOUR *laplap* may also be used for referring to pale colours, quite similarly to construction COLOUR-sa lòksy:, e.g. má: *laplap* קאדיפָּקיפָקי 'pale/light red', sé: *laplap* אדיפָּקיפָקי 'pale/light yellow'. Moreover, *kjalɛmlɛm* תַּקִיאָשׁל אַ refers to pale/light grey.

17.4.3 Dark colours

In Martam (consultant KN) dark colours are referred to with the construction -na? loksy: ব্যা ন্যান্ত্র্বা , which suggests that black $na^2 = \sqrt{\eta}$ is mixed to the colour in question, see Table 17.20.

	.20. Daik	colours with -na.	r loksy. (Waltalli)
máːna?	lòksy:	नुव्यत्र वया सेवा खुला	dark red
sé:na? l	lòksy:	र्षेत्र:वया र्येया सुला	dark yellow
pona? l	òksy:	र्द्े ज्या र्यया खुला	dark blue/green
0			

Table 17.20 Dark colours with -na2 loksy (Martam)

An alternative longer construction to má:na? loksy: is má: tɛŋkʰa nàksa loksy: त्वर केंत्र' क्र्या केंग May (lit. 'above red black'-sa loksy:).

17.4.4 Other colour terms

Mixed colours can be expressed with the formative $-t^h a \boxtimes$ 'variegated (colour)'. In Tashiding (consultant DB), the expression ka:p tashi? 57757 Bigging refers to basically white but mixed with other colours. In Martam (consultant KN), on the other hand, ka: tha sing or ka: that a sing of ka: that a sing of the single sin to white colour occurring on a backround of other colour(s). For instance, the expression $ka: p tenyk^h a$ paper.

Finally, Table 17.21 presents some additional colours terms from Tashiding.

Fable 17.21. Other coulour terms from Tashiding (consultant DB)			
nàːmu?	नगः झुग	dark (but not black)	
máːmu?	न्यर झ्या	brown (?)	
kjazā:	শ্রুমা-আন্দর্মা	bright-coloured, clear colour	
kjami?	गुगुगकेन्	colourless, pale	

17.5 Vocabulary used with small children

Some words are specifically used when talking to small children. The words I am aware are listed in Table 17.22 (with equivalent ordinary words given, if known) and some of them exemplified below. The verbs used in Table 17.22 are $t\tilde{a}$: 5π 'send', $l\hat{u}k_{a}$ bour', $p'ja_{a}$ is 'do', $kjap_{a}$ 'strike', sà \exists 'eat' and ze' $\exists a a a$ (eat, have (hon.)', although all of them tend to get semantically bleached when combined with other elements. As seen in Table 17.22, reduplication, probably accompanied by onomatopoeia, is prevalent when talking to small children (*cýcy*, *ã2ã2*, *màmam*, bunbun, tea:tea, bulubulu). The words referring to animals ∂mba 'cow (child talk)', $l\dot{\varepsilon}:l\varepsilon:/l\varepsilon:la$ 'goat (child talk)' and te 'idzi 'pig (child talk)' are the same as the words addressed to these animals when prompting them to move.

Ordinary	Child talk	Gloss
séːp tãː	сý:∼сýсу tã:, íс tã:, cýl tã: (KN), tсʰulu	urinate
	lúk (TB)	
kjako tã:	ế: tã:, ấ?ã? tã:	defecate
to sà, cèla? ze:	màmam ze:	eat food, rice
teʻa ze:	È: ze:	have tea
nèː, zim	nèːɲeː p'ja, zimzim p'ja	sleep
di:	buŋ(buŋ) p'ja	fall
gju	teaːteaː p'ja (PT), pʰaː (KN)	walk
tsʰuɛɛ kjap	bulubulu pʻja	wash
ákar	áka:	chilli
	paija:	spank
p ^h ɛn tãː	puk tã:	fart (verb)
ts ^h ata?	átsʰaː, átaː, hṟːhụː	hot
t'ika	haka, teitee, áte ^h i(te ^h i), te ^h ite ^h i (KN)	dirty
tcã:tc ^h itc ^h i	ásisi	beautiful
nò	<i>òmba</i> (cf. <i>óm</i> 'milk')	COW
ra	lè:la, lè:le:	goat
p ^h ako	te 'idzi	pig
óm	bubu (KN)	milk
	È:pa (KN)	sweet (of taste)

Table 17.22. Child talk vocabulary

Some words in Table 17.22 code adult's empathy towards children. One sign of empathy is phonetic assimilation to what the child would likely produce in attempting to say the "grown-up" version of the word: *ákar* 'chilli' is simplified to *aka:*, *tcã:tc^hitc^hi* 'beautiful' to *ásisi* and *ts^hata?* 'hot' to *áta:*. The word for cow, *ómba*, likely derives from *óm bak* 'carry milk', thus coding empathy in that the word is linked with the child's existing experience of drinking milk. The word *buŋbuŋ* presents onomatopoeically and emphathetically what will happen to a child if (s)he falls.

Some of the words from Table 17.22 are illustrated in (17.86-94). In the glosses, CHT refers to "child talk."

- (17.87) कॅं'ने' केन' जहना
 - *óni ἑ̃: p'ja.* child poo do 'Child, go poo-poo.' (PTB 5, 39)

(17.88) ঝ'ঝঝ' দাল্লিমা

màmam ze:. food.CHT have.hon 'Please eat some food/rice.' (17.89) $\hat{\alpha}^{,} \forall \forall \vec{\alpha} \neq \vec{\beta} \neq \vec{k}$ $\hat{\epsilon}^{,} ze^{,}$ tea.CHT have.hon 'Please have some tea.'

(17.90) สุสุราสุสุรา ฐ์พาศุรา ษิรา bunbun p'ja-ce be?. falling.CHT do-INF EQU.NE '(You) will fall (and hurt yourself).'

(17.91) कॅं'ने' नरुत'नरुत' र्मुषा

óni tɛa:tɛa: p'ja. child walking.CHT do 'Child, walk.' (PTB 5, 39)

- (17.92) \tilde{q} (מימים מילי של של של האי של ג'יפן ג'ola? $\dot{a}sisi$ של $b\varepsilon$?. clothes beautiful.CHT EQU.NE. ' (Your) clothes are beautiful.' (KT)
- (17.94) הקיתקי תפהין

bubu t^huŋ. milk.CHT drink 'Drink (some) milk!' (KN e)

Lastly, (17.95) records an expression which an adult may say to a child who has done something wrong and should show remorse.

(17.95) $\operatorname{High}(2\pi) \operatorname{High}(2\pi)$ $kuk^{h} \widetilde{e}: p'ja$ apologise.CHT(?) do 'Be sorry/Apologise!' (TB, PT)

17.6 Summary remarks

This chapter presented Denjongke vocabulary from five different domains: ideophones, kinship terms, names, colour terms and vocabulary used when talking to children. A major part of the discussion addressed ideophones, which occur in wide array and which have not been, to my knowledge, addressed in detail in other Tibetic languages than Dzongkha (see "expressives" in Watters [2018: 271ff]) and in Purik Tibetan (see "dramatizers" in Zemp [2018: 735ff]). Ideophones were categorized both by formal criteria (e.g. fully reduplicated, near-reduplicated) and functional criteria (e.g. nonnormative, onomatopoeic). A particularly interesting group of idephones was seen to be nonnormative ideophones, which associate a certain sequence of phonemes (*ClaC2.CloC2*) with nonnormativity (e.g. walking like a drunkard instead of walking normally).

Within kinship terms, it was shown that terms for ego's spouse's relatives are formed from the same terms that are used for ego's own relatives by adding in the end k'jo:p 'male in-law' and gjum 'female in-law'. Two exceptions were the terms for ego's wife's younger sister (*pim*) and wife's younger brother (*kipu*), which are expressed by dedicated terms not occuring among incosanguineal relatives. This, naturally, raises the question on the significance of these in-laws in Bhutia/Denjongpo culture.

This chapter further showed that Denjongke first names are strongly associated with astrology (names based on planets/week days) and Buddhism. Moreover, it was shown that Denjongke has five commonly used basic colour terms, with the same word being used for blue and green (although there are literary words for sky-blue/azure and green). Clear colours, pale colours and dark colours were shown to be expressed with specific constructions. Terms expressing clear colours were seen to excell in the use of reduplicated ideophonic suffixes. The last section introduced some vocabulary used by adults when they talk to children. It was shown that the vocabulary used when talking to children reflects adults' empathy toward children in terms of children's enunciatory ability and life-experiences.

Appendix 1: Text excerpts

This section provides some interlinearized and translated examples of various language genres with notes on the specific characteristics of each genre. The genres represented are proverbs, riddles, folkstory, novel and conversation.

Proverbs

Proverbs excel in scanty expression where interpretation may be highly dependent on collocation and word order rather than grammatical marking. For instance, the verb *ta* in (1) below would in ordinary prose in both clauses occur in nonfinal converbal form *ta-ti*. Proverbs typically consist of two lines which say analogous or somehow opposite things in parellel constructions, using partly the same words.

(1) के क्ष गयिंग र्ड्स र्ड क्ष कर मुह

mi ta jó? p'ja, p'ja ta tc^hu ma-t^huŋ. human look work do hen look water NEG-drink 'Looking (for an example) at people, do your work. Looking at the hen (for an example), do not drink water.' (KN)

(2) बेंग्क्यायां यानुवा म्राप्ता का वायायां ना

mi $g\varepsilon = lo$ tam, $l\tilde{a}$ $g\varepsilon = lo$ $\varepsilon \dot{a}$. human (be.)old=DAT word bull (be.)old=DAT meat 'Old (hu)man has sayings, old bull has meat.' (KN)

ra- ϵa sa-wø: $g\tilde{a}$: $t\epsilon\epsilon$ laplap, ra- $r\tilde{i}$: ta-wø: $g\tilde{a}$ goat-meat eat-2INF.GEN time tongue lick.lick goat-price look-2INF.GEN time mik lukluk. eyes wide.open

'When eating goat-meat, tongue goes lick-lick.When looking at the goat-price the eyes go wide open (in surprise).' (KN)

(4) कें कर्ते रेगका के गठिया मुं झुरेयका के गठिया

mí nóri? mi-tci?, *p'ja pu-ri? mi-tci?*. human thought-kind NEG-one bird feather-kind NEG-one 'People's ways of thinking are various. Bird's feathers are various.' (SS)

(5) अग्गूर केट्रे न स्व केव्र न कें सुर केट्रे सा के क्याया

 $\dot{a}kar \ m\dot{e}:-n\varepsilon$ $s\dot{a}m \ min-za. \ m\dot{o}by?$ $m\dot{e}:-n\varepsilon$ $l\dot{a}$ $mi-tc^ha?.$ chilly NEG.EX-COND food NEG-eat wife NEG.EX-COND enjoy NEG-enjoy 'Without chilli, there is no eating. Without a wife, there is no enjoyment.' (KN) (6) नेम लाया उद्य में प्रहेवा पहें ही के यापा उद्य में भया पहें ही dzek тí cín càga-tcɛ̃:=lo dzo-ti. tree branch-ADJZR=DAT climb be.comfortable-NF human k'o-wa-t $\epsilon \tilde{\epsilon}$:=lo làp dzo-ti. speak be.comfortable-NF understand-CIRC-ADJZR=DAT 'It is easy to climb to a tree with many branches. It is easy to speak to a man of understanding.' (UT) (7) אשי הדמי איזי איזי אין מי הדמיאי איזי אין איזי אין מי אדמיאי אין lòkso jờ?. $k^h a$ dzø:-na lòkso mè?. làm dzøː-na road err-COND returning EX.PER mouth err-COND returning NEG.EX.PER 'If one errs in road, there is returning. If one errs in speech, there is no returning.' (UT) (8) बे' दाँर्घ्र मा र्येका, रा दाँर्घ्र केंसु' दाँर्घ्रा $k^h a$ тí k^hjom k^hjom, ra k^hjom ts^hilu k^hjom. man difficulty mouth difficulty goat difficulty fat difficulty 'Man's bane is the mouth, goat's bane is the fat.' (explanation: A human being runs into trouble because of his/her mouth, and a goat runs into trouble [of being eaten] because of its fat) (UT) (9) କ୍ଟି ନ୍ମୋଂନ୍ଟି ଅର୍ମ କ୍ଟିର୍ମ ଭିଷ୍ୟ ମୁହ୍ୟିରା ଏକ ସିନ୍ତି ନ୍ୟୁର୍ମ ନିଦ୍ଧାରୁ ଭିଷ୍ୟ ମୁହ୍ୟିରା làm p^hiːpeː sà-wa topa? min-go?. to bja:to gju-wa to:ma? rice countryside.rice eat-PUR curry NEG-needed road level go-PUR stick miŋ-go?. **NEG-needed** 'In order to eat countryside rice for rice, no curry is needed. In order to walk on a level road, no (walking) stick is needed.' (UT) (10) ยิ่าสิขา จพราะา หาร่า ผิสารทัพ พังามา จพราะา สีว้า ผิสารทิพ sé-pa ta:ri miŋ-go?. kicik sóma sé-pa $t^h o$: black.flea kill-PUR axe NEG-be.needed flea.egg kill-PUR hammer min-go?. NEG-be.needed 'To kill fleas, no axe is needed. To kill flea-eggs, no hammer is needed.' (UT) (11) यन्त्र स्नान के राजे के राज làp-tã: jờ:-nɛ $k^h a = i$ sờn. gju-tã: j*à*∶-nɛ speak-manner EX-COND mouth=GEN caution walk-manner EX-COND kã:pø: sờn. foot.GEN caution 'If there is speaking manner, mouth's caution. If there is walking manner, feet's caution.' (UT)

t'ako p'in go? nó. sèmpo ín-ne ín-ne kv: nobleman EQU-COND give beneeded think fool EQU-COND cause.to.move g0? nó. be.needed think 'If (someone) is a nobleman, think that (you) have to give (him). If (someone) is a fool, think that (you) have to use (him).' (UT) (13) विवयां में द्विया मन क्षेत्र या भाषा ईन्या में ईवरमन क्षेत्र या भाषा kʰik-cĩ: cè:m=lo ma-làp. pjanka=lo p'in-cĩ: ma-làp. lead-NPAST.PER NEG-say poor=DAT give-NPAST.PER NEG-say foolish.woman=DAT 'Don't say to a foolish woman "I will take (you as my wife)," don't say to the poor "I will give."" (14) विंतु लि. म. वा पहुर र्ये र पहुर sòu k^haca dun-bo sà, rà duŋ food deer eat beat-2INF goat beat 'Deer ate the food, goat bore the beating.' (UT) (15) विवर्ग्स हिंग्वे गर्छिया भ्रत्य छत्र द्यार्गे हिंग्वे यविषा भ्रत्य वेत्र cèmpo k^hjo:-ne tei? ke: pã:. f'ako k^hjo:-ne ní: ke: get.angry-COND one portion forsake nobleman get.angry two portion fool $t^h op.$ find 'If a fool gets angry, give up giving (even) one gift. If a nobleman gets angry, (he) shall receive two gifts.' (UT) (16) હું'ગા, મુંલ. જ્રાર્થ્વાયા શેટ. ત્રા. ઇટે જ્યાં p^hjuku ki-u mi-ts^hu?. mi-ts^hu?. pjanka cí-u rich be.born-2INF NEG-be.able.to die-2INF NEG-be.able.to poor 'The rich are not able to be born. The poor are not able to die (properly).' (UT) (17) สังสัง จรูายา พาลา ฏิสา ลารา ชังสัง จรูายา พาจา ฤกลา ลารา $p^{h}o=lo$ mò=lo áma k'øn ma-ri. tce:ŋa ápa tce:ŋa tam

female=DAT fifteen mother wear NEG-hope male=DAT fifteen father word *ma-ri*.

NEG-hope

'A girl of fifteen (years), do not count on mummy clothing (you). A boy of fifteen (years), do not count on father's advice (but make your own decisions).' (UT)

- (18) न्वांत्र पति हिं हें वांत्र मुनया ते झायति खुवया के पंपा भन खन dogom kjap-ne k^hi lámø: $t^h u = lo$ phok-ce gjømpø: monastery.GEN dog stone strike-COND lama.GEN mind.HON=DAT hit-INF *bε*?. EQU.NE 'If you stone the monastery's dog, it will hit the lama's mind.' (UT) (19) ארי דָא מַזָקי אַ מּיאַאי⁴⁷⁵ באי זַר ווארי ארן
 - rã:dã: dup-nε tc^haly? pèn-ruŋ k^hεpar mè?.
 own.goal fulfil-COND style be.bad-CONC difference NEG.EX.PER
 'If one's own goal is achieved, it does not matter even though the means is bad.' / 'The end justifies the means.' (KL)
- (20) สา ลา ผู้ชี้หารน์น้ำ ลิ่า สา สิ่ญา ที่อี้ตา ซู้ซี่ตากา สลา | กุญา ลา ผู้ชี้หารน์น้ำ ลิ่า กุญา สลารา กา รลลา ma-mjõ:-pø: sà тí sà, neu=tei? *p^hjoka* sàm NEG-experience-NMLZ.GEN human eat snot=INDF taste food eat ma-mjõː-pøː тí gju gju, t^hempa là sám pass think walk NEG-experience-NMLZ.GEN human walk threshold 'An inexperienced eater eats and thinks a taste of snot is food. An inexperienced walker walks and thinks a threshold is a mountain pass.'

⁴⁷⁵ This WD form given by consultant KL may correspond to dictionary form argunar 'style, fashion, custom'.
Riddles

(1)	້ຢູ່ ຕັ້ຫຼາດຫຼາຍ ທີ່ຖາດຫຼາຍ ທີ່ຖາດຫຼາຍ ທີ່ ເຊິ່ງ ເຊ
	answer: ξ_{\parallel} <i>do</i> stone 'stone' (JDG)
(2)	$\begin{array}{cccccccccccccccccccccccccccccccccccc$
(3)	รั ^{·ພຸស} [·] બૅંદ [·] ਝ਼ੇ [·] 5ູເ ⁻ நॅंग [·] ब्रा ⁻ व्या [·] गॅं [·] कुत्रब [·] રૉં। गव्र [·] भ्रात्? t'o:lɛ lồ:-ti t'uŋto? nàŋɛa làko kjap-o. k'an bɛ?? early rise-NF tea.churn inside hand strike-2INF what EQU.NE 'Rising early, putting hand in a tea-churn. What is (it)?' (JDG) answer: ลักว ⁻ गॅล [·] มัล.วั1
	<i>tø:gu? k'øm-bo</i> or p^hylun/p^hydun short wear-2INF sleeve 'putting on a shirt' 'sleeve'
(4)	बे दन्ते बे वग्वा गहाका, झ्रे रगाया गहिवा। दन्ते गव छन्? mi=di $mila?$ súm, kera tei?. di k'an be?? human=DEMPH person three belt one this what EQU.NE 'Three people, one belt. What is it?'
	Answer: เตาสญา จุธัสาจุธัสาจัןเตาสญา จุธสางคุรๆára?ten-tem-boorára?teak-ce?liquorextract-RDP-2INFliquorcut-INF'extracting liquor''making liquor' (JDG)

Explanation (KN): The liquor-maker binds three vessels with one belt, which prevents the liquor from breaking the vessels.

(5) मु अबा र्येन वरा हे विर्म र यहवाबा यहरा र्ये। p'ø? sã:te k^hada gja=le ta: tã:-bo India=ABL Tibet until scarf append send-2INF 'It extends a khada-scarf from India to Tibet.' answer: कु tc^hu water 'water' (JDG) ସଣ୍ଟମ୍ମ ନିମ୍ନାପ୍ରୁଷ, ସର୍ବଷ୍ଟର୍ଶ୍ମ ବିଷ୍ୟା (6) t^hoko riŋkʰjam, sè:go? cìmpu long rope food tasty 'Long rope, tasty food.' answer: 3. ปสาว. อีป sé-ce? рà fish kill-INF 'to fish' (JDG) (7) बराषुः र्वेरुर्धेः गट्रंगेषः वर्ह्रबान्नीटः युत्रर्धे। mà:k^hu $p^{h}o:po \quad k'\tilde{a}:=gi$ dzamli η k^hjap-o. liquid.butter bowl full=AGT world spread.far-2INF 'A full bowl of liquid butter extends over the world'. answer: तुम् pìm sun 'sun' (JDG) (8) गार्केन् इंग्रें क्रेंकेंन् भुगार्गे पहनायावता k'adzø? ódzø? tã:-khĩ:. sò-u, kjako how.much eat-2INF that.much faeces send-NMLZ 'As much as it eats, that much it deficates.' (JDG) answer: মৃদ্র বিষয়া rã:t^ha? mill 'mill' (JDG)

(9) हे हवा थे छे. बह्वा झेटावर हैं छट द्वा वीष क्ष के किया है te:ta?p^hi:pe:=tci? tenk^ha p'iteun ťu:=ki lú kjap-o stick horizontal=INDF on small.bird six=AGT song do-2INF 'On a horizontal stick six birds are singing.' answer: मु'न्नेत्र सु শ্বীন্দ'ন্ত্ৰ' ধ্ৰা $p^h u$ límbu $p^h u$ gjaliŋ or blow gyaling.trumpet blow flute 'blowing the gyaling-trumpet' 'blowing the flute' (10) अग्रा होत्र जेंग, सुम्रा मायत तर्षे, यात्र तर्मा? áт k^{h} andu, k'an du?? símo, p'um mother cannibal.demoness daughter dakini what EX.SEN 'The mother a demon, the daughter an angel, what is (it)?' answer: ર્જ્રે'ભુષા ts^hɛːlum tshelum-plant 'tshelum plant' (a plant with thorns but sweet fruit) gju-reŋk^ha ta dem gjuk. $d\phi$:-reŋk^ha kynt $\epsilon \tilde{\epsilon}$: dem $d\phi$?. $\tilde{\delta}$:-reŋk^ha ge:p go-SIM horse like run sit-SIM thief like sit come-SIM king dem õ. òη-do di k'an bo? this what EQU.NE.Q come-proglike come 'When going, it runs like a horse. When sitting, it sits like a thief. When coming (back), it comes like a king. What is it?' answer: अया गी नहर महा kjako tã:-ce? faeces send-INF 'defecating' (UT) (12) मॅक्रायते के कन्तर क्रूट थी हरा का के मुनक रका पर झुरा के जिनक रकी पर से बाद र दे या रहे या रहे हरा हरा हर mìnda lú $\eta = gi dz \epsilon$? kompø: sá=lo kjap-renk^ha air=GEN bullet ground=DAT strike-SIM leather.GEN gun $p^{h}ok$. nagu tenkha di k'an bo? nose on hit this what EQU.NE.Q "When the gunpowder of a leather gun is shot toward the ground, it hits the nose. What is it?" answer: यत्।

p^hen fart 'fart' (UT)

Folkstory

Folstories are often pedagogic discourses about animals. Frequent use of the reportative =lo, typical of folk tales, reveals that the speaker has heard the story from others. The use of the reportative is in the story below typically translated as '(so the story goes)' (the brackets indicate that Denjongke is not as explicit as the English equivalent). The quotative $=s(\varepsilon)$, as shown by the first line of the story, can also mark that the speaker is repeating what has been said by others.

- भेग्में लेत अर्घेट अप्तमग्रम से (1)mí=lo cènt^hõ: ma-sá:=s. human=DAT contempt NEG-gather=QUO 'Do not heap contempt on people, it is said.' (2) มे'गें' देव'यहेंद प्रयायगां ठें दे' पर्म र देव' हे के के के के के के mílo cènt^hõ: $k^h a \epsilon a = ra$ t^høn-ce sák-etsene $(d\varepsilon)m$ human=DAT contempt gather-COND deer=AEMPH like become-INF $\tilde{i} = s$. EQU.PER=QUO 'If (you) heap contempt on people, (you) will end up like the deer.' ાત્ર તર્ને હિવુ. ત્રી તર્ને સે સે ત્રી તર્ને સે સે ત્રે તે સે ત્રે તે સે $k^{h}aca=di$ $k^{h}u=i=gi$ dem(3) naku dep súm-sum-bo jø:=lo. deer=DEMPH 3SGM=GEN=GEN like.that nose like.that shrink-RDP-2INF EX.PER=REP 'The deer, its nose is shrunken like that, it is said.' क्रेंग् दर्भे गाव सुग्र में भेपा छे वे मार मु ग्राम मार गारेगा परिय भग्न वरा मा द्यु में में , ईवा वरा मा भग्न (4) ódi k'an súm-bo sí-betsene t'anpu lấ:bu=tci? dev làm that what shrink-2INF say-COND long.ago elephant=INDF like.it road gju-do=lo, p'ja? nànca=le.nànca go-IPFV=REP cliff inside=ABL inside 'If (I) tell what shrunk it, long ago an elephant was walking on a road like this, on a cliff.'
- (5) gã:=di lấ:bu=di=gi gjap k^ha jòu p'ja:= $di=l\varepsilon$ gju-wø: go-2INF.GEN time=DEMPH elephant=DEMPH=GEN in.back up cliff=DEMPH=ABL ba? òn-di òtε dogom=tci? di: t'a:glo dogom=di stone=INDF down fall carry come-NF EXCLAM stone=DEMPH $p^{h}ok-o=lo=la$. hit -2INF=REP=HON 'When going, a stone came down falling from the cliff and, slam, the stone hit the elephant's back (so the story goes).'
- (6) ξ \check{q} \check{q}

(7)	$\vec{\beta}$ (מי פן רק אלי אד אד אד אלי רק אלי רק אי דאפין מי מי דעי אלי $te \ k^h a \epsilon a = di = lo$ $l \tilde{a}: bu = di$ dep ta $za:$ $l a p - o = lo.$ so deer = DEMPH = DAT elephant = DEMPH like.this look put say-2INF = REP 'So the elephant looked at the deer and said like this:'
(8)	
(9)	ອູ້ຫຼາ ເຈລັ, ເຊິ່ງ, ເວລັ, ເຊັ່ງ,
(10)	$[a^{i},q^{i},a^$
(11)	\vec{p} : $\tau \sim \vec{p}$: $[n:q]$: $a \neq n$ and $n \neq \vec{s}$: $q \neq \vec{s}$: $q \rightarrow \vec{s}$: $q \neq \vec{s}$:
(12)	جر: قرح: عَقَّرَامًا، قَمْرَ يَجْ مَحْرَ عَقَلَ rã: jòu gjuk số:-di ta số:-tc ^h i. 2SG.M up run go.PFV-NF look go.PFV-IMP.FRN 'You go running up to look.'
(13)	ग'ल्लेस' ग्रह्मेग' तर्न, र्न. में' हैं' ल्लेंग' त्योभ'आयत्र' तर्न, भन्न'आयत्र? $k'atem=tei?$ $du-ke$ di $\eta a=lo$ do lok $di:-k^hen=di$ how=INDF EX.SEN-IN this 1SG=DAT stone remove fall-NMLZ=DEMPH $lap-k^h\tilde{e}$:? ⁴⁷⁷ say-NMLZ

'What kind is the one who displaced and made the stone fall on me?'

⁴⁷⁶ This form is typically pronounced *na*. ⁴⁷⁷ The last syllable is unclear on recording. The nominalizer $-k^{h}\tilde{\epsilon}$: was an educated guess by a consultant.

(14) दे चट हे हूँग दर्गवा आपत दर्र हें र्ट्र का गठिया दर्या केंग स्रा गरिया के लाग के लाग के लाग के लाग के लाग di:-k^hen=di p'idim=tci? t'izã: do lok du = lo = la, but stone remove fall-NMLZ=DEMPH porcupine=INDF EX.SEN=REP=HON po:kjupain. porcupine(Eng.). 'But the one who displaced the stone and made it fall turn out to be a porcupine (so the story goes).' (15) हे सु मिन दर्न अय रॉग्से $t\epsilon k^h u$ k^haca=di làp-o=lo so 3sgm deer=DEMPH say-2INF=REP 'So he, the deer, says (so the story goes):' (16) धे'व' तर्ने अ' र्शे अर्था उत्र' यहिया' तर्नया गी $j\hat{i}=na$ dem símte \tilde{e} :=tei? du-ke. up=LOC such animal=INDF EX.SEN-IN 'There is such an animal up there.' (17) यह्याया रूट रॅटर्रेट पर्टे पर्वेया पर्याणी zu? tsã: dodem=tci? du-ke. ronron body thorn here.and.there right.like.this=INDF EX.SEN-IN 'It is one with body covered with thorns like this.' ódi=gi dep t'õ: kok-sa=l ε =di *do=di* di: ba? like.that hole dig-CMPL=ABL=DEMPH that=AGT stone=DEMPH fall carry nè:⁴⁷⁸ gjapk^ha p^hok-o òη-diki, k'atelo. $b\varepsilon = s$. how=DAT 1SG.GEN in.back hit-2INF come-NF EQU.NE=QUO 'That one like that dug a hole and (as a result) a stone fell and came and, how, hit my (probably should be: your) back (so the story goes).' (19) हे मुरानु दर्न अयार्थ में te lấ:bu=di làp-o=lo so elephant=DEMPH say-2INF=REP 'So the elephants said (so the story goes):' (20) ואיי הליועדי ההי מקקיועדי ואין $k^h u = lo$ dik^ha nè: dynk^ha kuk=s. 3SGM=DAT here 1SG.GEN in.front.of call=QUO

'Call him here in front of me, (he) said'

 $^{^{478}}$ This should be $\breve{a}_{\overline{T}}(\widehat{\mathfrak{Y}})$ 'your' as the speaker is the deer.

- (21) \mathbb{R}^{n} \mathbb{N}^{n} \mathbb{N}
- (22) हे मुरामु पर्ने अन्न रें के

te lấ:bu=di làp-o=lo. so elephant=DEMPH say-2INF=REP 'So the elephant said (so the story goes):'

rã: jàte p'ja? nànea dzek-ti jòu $k^{h}u=i=tsa$ gju g0? cliff inside 2SG.M up climb-NF up 3SGM=GEN=at go be.needed *jòk^hø:=di=ki* có? $lap-dzyng\varepsilon=s$ làp-sa=le. say-PROG.APH=QUO say-CMPL=ABL up.GEN=DEMPH=AGT come 'You have to climb up the cliff and go to him, because the one up (there) is saying "come".'

- (25) $r \cdot [g\hat{a} \cdot argar[ar \cdot ag] \cdot \hat{a} \cdot \tilde{g} \cdot$
- (26) $[\mathbb{Q}^{a}]^{a} (a\mathbb{Q}^{c}) (a\mathbb{Q}^{$
- (27) हैं', रत्र में' वतेश महमाश र्धेश्ररूत', रत्र वतेश महमाश' र्धेश्ररूत', ख़ु' वतेश र्धे'श्रमा' महिमा' अतुब'गर्र वर्गु श्रेर्ख्याय आपव' यत्त' है' क्षेत्र र्ख्या (insult) द्वार र्या या व्या मा

tsei *rã:=to* dem zu? bomteun, rã: dem zu? bomtcun, k^hu EXCLAM 2SG.M=CEMPH such body size 2SG.M suck body size 3SGM dem p^hima=tei? mi- $ts^h u$ - $k^h \tilde{e}$:= $j\tilde{a}$: dynk^ha gju làp-ti such small=INDF in.front.of go NEG-be.able.to-NMLZ=even say-NF cènt^hõ: insol $p'_{ja-u=lo=la,}$ sák-o=lo. do-2INF = REP = HON contempt gather-2INF = REP insult(Eng.) "Wow, you (are) of such a body-size, you (are) of such a body-size, but still you are not able to go in front of such a small one as he," insulted, heaped contempt (on him) (so the story goes).'

- cènt^hõ: sák-sa=le $k^h u = gi = di$ ódi elifənt tenkha, gather-CMPL=ABL 3SGM=GEN=DEMPH that elephant(Eng.) contempt on lấ:bu tenk^ha sènt^hõ: jì? ma-te^hi-pa tc^hi: sák-sa=le, contempt gather-CMPL=ABL believe NEG-believe-CIRC elephant on EXCLAM $g\tilde{a}$: te $k^h u = i = gi$ naku=di súm-di làpø: dep say-2INF.GEN time so 3SGM=GEN=GEN nose=DEMPH like.that shrink-NF $d \emptyset$:-po = lo = la. stay-2INF=REP=HON 'Having heaped contempt, he, having heaped contempt on that elephant, when (he) did not
 - believe and said "darn", his nose shrunk and stayed like that (so the story goes)."
- (29) مَحَ مَعْمَ هَذَي مَعْمَ مَعْمَ عَلَي مَعْمَ مَعْمَ مَعْمَ مَعْمَ مَعْمَ مَعْمَ مَعْمَ مَعْمَ مُعْمَ مُعْ

Excerpt from the Novel Richhi

The novel Richhi by Bhaichung Tsichudarpo, published in 1996, remains the only existing Denjongke novel. Below is an excert from the novel (p. 12-13). The excerpt is slightly modified from Yliniemi (2016a).

ສັ້ສາສິັສາຫຼີສາ ຫລີ້ທາຮາ ດລີ່ ທີ່ ຮອສສາອຽາລ້າ ຽຽສາດລິສາລ້າ ວ່າທູຽາ $t \varepsilon^h \phi d \tilde{\varepsilon} = g i s \phi z a$ zo:-ti thamt $\varepsilon \varepsilon = lo \ t`\tilde{a} - z\tilde{\varepsilon} := lo$ pe:lun tea.HON make-NF all=DAT carry-PROG=DAT bowl PN=AGT Choden makes tea, brings it to everyone and puts snacks in a basket in their वर्त्ताणे गर्सेन्य वित्र ठवाने झेत्यमें झुवा प्रविग দান্ধা বিদার্ন্ত অনস্যান্ডিয়া t'ato khõ:=tsu námtei? nàn=gi tenlo bu=na za?. sớ:cip tcente snacks middle=LOC inside=GEN table on put now 3PL=PL with midst on the table. Now as they sit together हेंद्र ह्वे' गर्वेग्य हा गर्वेग्य वेत्र ज्ञात्रवेत्र प्रति भ्रान्य यवे खुज्द द्विय k^him kap né:mu=rã: dø:-ti sø:za sà-zẽ:-pø: sø:cip time really=AEMPH home sit-NF tea.HON snacks eat-PROG-NMNZ.GEN enjoying tea and snacks they are really like members ขั้ง พิวสาวดิส พิวสาวดิส ทาง ग्रेगगे' प्रचतर्क्त ५५.२१ tci:=ki za:tshã: dau. k'jo? kjap-z \tilde{e} : kjap-z \tilde{e} : karma=gi one=GEN family.members like talk do-prog do-PROG PN=AGT of one household. As (they) keep on talking, Karma says *laki=lo, nà: p'usim=la:* k'ana zu:-k^hen bo? PN=DAT here younger.sister=HON where live.hon-nmnz EQU.NE.Q to Lhaki, "Where does the sister here live?" 'क्षुं भुनि' - "पिन' न' नन' अनुवार्छपानन' र्ख्वेन'यु'व' धुणायवा गवन'र्न' nấ:-do laki? – $k^h \tilde{o}$: ηà t'ã: pámtei=rã: lóbda=na te^ha:le? **3SG.HON** 1sg and with=AEMPH school=LOC PN work.HON do.HON-IPFV Lhaki: "She works together with me at the school. জিব'র্মা हें हे. म्रेट कें राखगणगणव केंवी तर्ने ।यर श्चेनका हे जाय ĩ ĩ:=co. dik^ha lep-ti dordzili $\eta = lo zu:-k^{h}\tilde{e}:$ hálam

EQU.PER=AT TPN=DAT live-NMLZ EQU.PER here arrive-NF about (She) lives in Darjeeling. (She) came here about some त्र में मजुन केंन लम के खाजा" ma; dou ge:-ts^hø? lã:-to ágja. Karma – month eight-some reach-IPFV elder.brother PN that.AGT=CEMPH eight months ago, brother." Karma: "That's it. dã: nàtci bhaila-ki tok-ti di: dø: j*à:-kap g0* vesterday 1PL.GEN PN=GEN head hit-NF fall sit EX-SIM Yesterday when our Bhaila was lying fallen after hitting his head, cí-ce? p'usim=la: *mè*:-*pateene* $k^h \tilde{o}$: óna=rã: bɛ?. dã: little.sister=HON NEG.EX-COND 3SG.HON there=AEMPH die-INF EQU.NE yesterday if it wasn't for the sister, he would have died on the spot. Yesterday ८.२०११ स्रि. मेरे स्ट. स्ट. स्ट. स्वर्गे स्वर्गे संग्रीत संग्री स संग्री स संग्री स nàtea=tsu p'jeu t'ã: hấ:pø: ¢úk=na p'usim=la:=lo hurry and shock.GEN strength=LOC little.sister=HON=DAT 1PL=PL in the heat of the hurry and shock we didn't even manage "अम्रायम् ! स्ट्राट ट र र होते स्रिते यम् श्चियाबाह्ते.क.लट. ख.ज.क्याब्री, $t^{h}udzite^{h}\varepsilon = i\tilde{a}$: ma-ts^hu?. dã: nàtci Bhaila сù ám=la: thank.you=even say.HUM NEG-can mother=HON yesterday we.GEN PN to say thank you to the sister. "Mother! Yesterday when our Bhaila had fallen it त्रगुभा येंद्र रादे ज्ञूदा यदायर भेव हैं युगा हेंबा रावेव ge: jà-pø: gã: p^hã:pa: lèn-di k^hja? $p^{h}i$:- $z\tilde{\varepsilon}$: fall EX-2INF.GEN time lap.DAT take-NF blood wipe-PROG was the sister here who took him to her lap, wiped the blood, $p^{h}\tilde{a}$:pa: Bhaila te:-z \tilde{e} : ámø: ám=la: màla? ménkhã:=na mother.GEN lap.dat pn entrust-PROG mother=HON quick hospital=LOC handed him to the mother's lap and and said: 'Mother, take her quickly রষ্ট্রিয়া র্দ্রবা যার্য্যনাথারা বা ন্যায়ীনা স্লান্য হা ਆਰਾਨਾ ਦੇ. k^hik tc'øn súŋ-k^hẽ: nà: p'usim $b\varepsilon = \varepsilon o$. áma: ηò take go.HON say-NMLZ here little.sister EQU.NE=AT mother.AGT face

to the hospital!'

Did the mother

⁴⁷⁹ The expression $\delta di = to$ is a challenging to understand and translate in this context. It can be said, for instance, with hindsight to a child, who without heeding a warning, experiences the unpleasant outcome.

અદ્યેવ લે વાય ?'' લા અ ''ક્ષારુંદ્વા દ્વ અવેય દ્વી	अन्न:
kʰɛn-zɛ-gam? ama – ãhã: ŋồ ma-ɛéː-ɛo.	dã:
know.HON-PST-ATTQ mother aha face NEG-know-AT	yesterday
recognise her?" Mother: "No, I didn't recognise her.	Yesterday in the
	1 ~ 2
र्डीतः कॅपार्स्तरावाः क्षेरियनेवाः क्षेर्ण्यात्तरत्तः व्येत्	ષ્યુ! જેંપ્લે
$p' j \in u$ $ts^h ap ts^h up = na$ $od \in m$ ta $lo: = r\tilde{a}$: $m \tilde{e}^2$.	já: óni
nurry nervous=LOC like.that look nave.time.to=AEMPH NEG.	EX.PER EXCLAM Child
must of hervous hurry there wasn't even time for as much as having a	IOOK. Ana, the child
ર્શ્વેન'શ' લદ્દેવે' ન'રદ્દ' છુવા'બષ' વાનદદ્દ?" જાગ્રાવે' બન'બે'	
$l \acute{o} b da di = na = r \tilde{a}$: $t c^h a : l \epsilon^2 n \acute{a} : - do?$ $\acute{a} m \sigma$:	lèn=lo
school this.GEN =LOC=AEMPH work.HON do.HON-IPFV mother.GEN	answer=LOC
works here at the school?" As an answ	er to the mother,
ळॅब्राक्नेुन्'ग्रीुब - ''व्य्याब्य' क्षेत्र। ८' क्वॅन्य्या' वर्न्टदे'त्र'र्र्न्र्' यॉर्थेया' क्रुनव्य'र्न्र' क्षेत्रा	
$t \epsilon^{h} \phi ki - ki - l \dot{a} - \tilde{t}$: $\eta \dot{a} = l \dot{o} p t a = r \tilde{a}$:	jó? kjap-to í:.
PN=AGT HON-EQU.PER ISG school this.GEN=LOC=AEMPH	work do-IPFV EQU.PER
Choki (says:) "Yes I do. I work at the school.	
ਕੋਨੀਯੁਨਾ ਕੋਰਨਾਲਾਂ ਨੂੰ ਨੂੰ ਕਾਰਾ ਰਸ਼ੂਨ ਕੋਨਨਾ ਕਾਨਨਾ ਸਿੱੰ $dik^ha:$ $lep-ti$ $t'a$ dau $ge:-ts^ha$? $l\ddot{a}:-to$. here arrive-NF now month eight-about reach-IPFV It is now some eight months since I arrived here."	
"લન્ને ત્વર યો. ૨૮. જાલેષા. હતો. તાંચ. રાલેતોજા.પ્રે.ડુ., જા.જાણ. ટુ.ટો.	
di-k ^h a ka t'ã: pámtci? k'ana zu:-to? ámø: t'i	wa.
here who and with where stay. HON-IPFV mother. GEN qu "With whom (and) where do you live here?" (is) the mother's question	estion
with whom (and) where do you live here? (is) the mother's question	1.
ଛଁଷଂଞ୍ଚିନ - "ଶ୍ୱିସଂମଧିୟ' ସଣସାଷ ମସ୍ତ ସ ହିଁମନି ଭିଷା ମାନ ଅନ୍ତର୍ଭ କାର୍ଯ୍ୟ ତିସ୍ତ ମଣ୍ଡ ସଣ	
$te^h\phi ki^2 - l\phi p\phi n = zu; ea; = na d\phi; -to \tilde{t}: nà námtei? ně$: nùm
PN teacher room=LOC stay-IPFV EQU.PER 1SG with 1S	G.GEN younger.sister
Choke: "I am staying in the teachers' quarters. And I have my tw	vo little sisters
ມດີສາເພ⊏ ພັ⊏ເຈມສາ" ຜາລສາ - "≭ະ∋ີ ມີ⊏ເຈັ້າ ມາລາ ສດມສາຮົ?	
$n(z=i\tilde{a}; i\tilde{a}; = la; \acute{a}ma = dordzilin=lo k'ana zu$:-to?
two=also EX.PER=HON mother.AGT TPN=DAT where sta	IV.HON-IPFV
with me too." Mother: "Where do you live in Darjed	eling?
with me too. Mother: Where do you live in Darjed	eling?
with me too." Mother: "where do you live in Darjed שרוי קר: שאוי אָן רופאי אָריזן?" jà:p t'ã: jùm=co zu: jà:-ka?	eling?
with me too. Mother: where do you live in Darjed (uni n n n n n n n n n n n n n n n n n n	eling?

هُمَّرَ بَهُمَّ - ''جَرَةَ بَهُمَّرَ بَالْمَا مَنْ مَنْمَ نَوْمَ نَوْمَ نَوْمَ نَوْمَ نَوْمَ نَوْمَ نَوْمَ عَلَ tc^høki? – dordziliŋ=lo gum gjømpø:=tsa: PN TPN=DAT TPN monastery=by Choki: "In Darjeeling by the Ghoom monastery.

भाषा पुरुष पुत्रि पुत्रि पुत्रि पुत्रि का $j\dot{a}:p$ $t'\tilde{a}: j\dot{u}m$ $p\dot{t}:tc^ha$ father.HON and mother.HON both Father and mother are both

مرومام[.] شرم" zu: jø?. live.HON EX.PER alive."

Excerpt from discussion

The following text is an excerpt from an unstructured discussion between six family members four of whom (marked K, B, T and F) are involved in the discussion excerpt. The recording was done in my absence by consultant KN, one of the family members. The representation of the discussion is simplified in that at times several people spoke at the same time and more than one line of discussion was ongoing.

- - B ମୁକ୍ତ ଇଁନ୍ତ ଅମ୍ବା ଅଂନ୍ୟକ୍ ରିଂ ସମ୍ବାଦ ଅଁକବାର୍କ୍ଷ ଭିକ୍ସ cánu mòu bɛ?. waranasi $k^hatc^hø:=l\epsilon$ $\tilde{t}:.^{480}$ PN down EQU.NE TPN teaching=ABL EQU.PER 'Shanu is down. (He) is at (Buddhist) teaching in Varanasi.'
- (11) B \check{r} \check{r} , \check{r} ,
- (12) T कात दाय्या कुया कर्क र् याद्वा कर् क्रा क्रिं? $m\tilde{e}$:, $p^{h}a:=le$ $gjalts^{h}en=tsu$ $j\tilde{e}$:-p $m\tilde{e}mbo$? NEG.EQU.PER over.there=ABL PN=PL EX-2INF NEG.EQU.NE.Q 'No, from over there Gyaltshen and his associates are there, arent't they?'
- (13) B ตั้ง สูงางสัส นั่น
 - *έ:* gjalts^hɛn j ∂ ? oh PN EX.PER Oh, Gyaltshen is there.
- (14) T $\overrightarrow{\beta}$ $\overleftarrow{\gamma}$ $\overrightarrow{\eta}$ $\overrightarrow{\eta}$ $\overleftarrow{\beta}$ $\overleftarrow{\kappa}$ \overrightarrow{k} $\overrightarrow{\eta}$ $\overleftarrow{\kappa}$ \overleftarrow{k} $\overrightarrow{\eta}$ $\overleftarrow{\kappa}$ \overleftarrow{k} $\overrightarrow{\eta}$ $\overleftarrow{\kappa}$ \overleftarrow{k} $\overrightarrow{\eta}$ $\overleftarrow{\kappa}$ \overleftarrow{k} \overrightarrow{k} \overrightarrow{k}

⁴⁸⁰ The use of the personal equative \tilde{i} : in a locative clause here is surprising and worthy of deeper investigation.

(15) B देर्श्व वर क्षु अर्केट्र कें केंट्र बर केंट्र केंटर केंट्र केंट्

t'itsina: $late^{h} = lo$ $s \delta: -s a: -ni$ a.few.days.agohereworship=DATgo.PFV-TERM-TOP $\eta \delta ts^{h} - u = lo = ki.$ be.ashamed=2INF=REP=NC'A few days ago when (I told him) go to the worship ceremony, he said he was ashamed.'

(16)ťa nà ó te gutor số: gju=to làp-zin nà tε now 1SG DISTAL so gutor-offering 1SG go.PFV go=CEMPH say-PROG so mà-nen-ts^ha-ne k'an p'ja-ce?? NEG-listen-CMPL-COND what do-INF 'Now, so when I tell him to go to the gutor-offering either with me or alone, so if (he) does not listen, what to do?' ⁴⁸¹

- (17) F 위·袞·方· 尓· 養尓·챡· 费져져져가 여필· 주책적·위方· 習行
 cánu=to t'a tsøpo kjap-a gju go:-ɛɛ bɛ?.
 PN=CEMPH now debate do-PUR go be.needed EQU.NE
 'But Shanu now has to go to participate in the debate.'
- (18) B Ry and an an equation of the set of
- (20) מוּפְקּ'שָׂ' אֵּ'אַלִיאוּמִק' אַרְ'אָן *בנחחנ mi-t^hop-k^hen be no.* otherwise NEG-find-NMLZ EQU.NE TAG.ASR 'Otherwise (it) is not available, I tell you.'

(21) B $\operatorname{Aut}(X') \cong f$ $la=b\varepsilon^2$ HON=EQU.NE 'Yes, it is (so).'

⁴⁸¹ For clarification of meaning, see the footnote in example (15.120).

- (22) $F = \eta_{\alpha} \eta_{\beta} + \tilde{f} + \tilde{f} + \tilde{f} + \eta_{\beta} +$
- (23) B ८. उत्ते । पर र्ड गीषा आदा पशु पतित पराते गात पशु र्खगाषा भूत आदा?

gju-zin⁴⁸² nàtci làp-ne k^hartsa=gi ápa k`ana gju 1PL.GEN expenses(Nep.)=AGT father go-PROG say-COND where go ts^hu:-ee? ápa. be.able.to-INF father 'Because of the expenses for us, father, even if (one) says (he's) going, where could (one) go, father.'

- (25) ٦' نَعْ مَحْمَ عَنْ عَامَ عَنْ اللَّهُ عَامَ عَنْ اللَّهُ عَامَ اللَّهُ عَامَةُ عَامَ اللَّهُ عَامَةُ عَامَ اللَّهُ عَامَ الْحَامَ مَا اللَّهُ عَامَ الْحَامِ مَا عَامَةُ عَامَ اللَّهُ عَامَةُ عَامَةُ عَامَ اللَّهُ عَامَةُ عَامَةً مَا عَامَةُ عَامَةً مَا عَامَةً مَا عَلَيْ عَامَةُ عَامَةً مَا عَامَةً مَا عَامَةُ عَامَةً عَامَةً مَا عَامَ مَا عَامَةُ عَامَا اللَّهُ عَلَيْ عَامَ عَلَيْ عَامَ عَلَيْ عَامَ مَا عَامَ مَا عَامَ اللَّهُ عَلَيْ عَامَ مَا عَامَ الْحَامَ عَلَيْ عَامَ مَا مَا عَامَ عَلَيْ عَامَ عَلَيْ عَامَةُ عَامَةُ عَامَ عَلَيْ عَامَ عَامَ عَلَيْ عَامَ عَلَيْ عَامَ عَلَيْ عَامَ عَلَيْ عَامَ عَلَيْ عَامِ عَامَ عَلَيْ عَلَ مَا عَامَ عَلَيْ عَامَ عَامَ عَلَيْ عَامَ عَلَيْ عَامَ عَلَيْ عَامِ عَامَ عَلَيْ عَالَ
- (26) B ANDRY BY $la=b\epsilon^2$ HON=EQU.NE 'Yes, it is (so).'

(27) בְּאָד' (phone) אָדאדי דא יארי אין phon kjap-tee làp-tee. phone(Eng) do-PST say-PST

'(Somebody) told (me) that (he) phoned.' (i.e. Shanu, who is the son of B, had phoned to a member of B's household)

(28) हेंन्'र्रे' वर्ने' हेंन्'र्केंवु' क्षुम्र' म'हेंन्न' क्षुम?

tsøpo=di	mjõː-tsʰou	nám	ma-mjồː-p	pám?				
debate=DEMPH	finish-CMPL.2INF	EQU.ATTQ	NEG-finish-2INF	EQU.ATTQ				
'I wonder whethe	er the debate has ende	ed or not.' (i	e. having heard so	ome of the contents of				
the phone call, the speaker still did not know whether the debate had ended)								

⁴⁸² Although this excerpt was transcribed with KN, I consider it possible that *gju-zin* here should, in fact, be transcribed as $gju-c\tilde{\iota}$: [go-NPST.PER].

- (29) F ω^{*} *jà:* aha 'Aha.'
- (30) K צַּק ימשי זַבּרָ דָ מיד קישׁ רַשָּ קר צַרן *חנהוב k^h*õ:=to waranasi gju-ce be?. before 3SG=CEMPH TPN go-INF EQU.NE 'Before (going to Dorjeden) he will go to Varanasi.'
- (31) B المَاتِمَةُ اللَّهُ اللَّالَ اللَّهُ اللَّالَ اللَّالَةُ اللَّذَا اللَّذَا اللَّهُ اللَّهُ اللَّهُ اللَّالَ اللَّقُولُ اللَّالَ الْحُلْقُلُولُ اللَّالُقُلُولُ اللَّالَ اللَّالَ اللَّالَ اللَّالَ اللَّالَةُ اللَّالَ اللَّالَ اللَّالَ اللَّلْحُالَةُ اللَّالَ لَقُلُولُ اللَّالَ لَقُلْقُلُولُ اللَّلْحُالِ اللَّقُلُقُلُولُ الللَّلُقُلُولُ الللَّالَ اللَّالَ اللَّالَ لَقُلُولُ اللَّالَ اللَّالَ اللَّالَ اللَّالَ اللَّالَ اللَّالَ اللَّ
- (33) T جَنْحُعَّا جَنَعَ عَامَةً عَنْمَا t'utci? dordzideelo. this.year TPN=DAT 'This year in Dorjeden.'
- (34) K קָיָצָּמִי דְּיִבְּמִחְדָ (direct) בָּוֹצְ מִיקִי מִיִּדִי מְקָדָן (=מְקָי צֵּקן) אָּיִמְלִי מִיִקִיקּמְרָיזּון *t'utci? dairekt dordzidɛ̃:=lo=ra dɛ?. ódɛ: làp-tɛyŋgɛ.* this.year direct(Eng.) TPN=DAT=AEMPH EQU.AP like.that say-PROG.APH 'This year (it) seems to be directly in Dorjeden. So (they) are saying.'
- (35) $F \notin_{T} \tilde{x}' q \hat{T} \tilde{x}'$? $ts \phi po = di$ \acute{o} debate=DEMPH EXCLAM.Q 'The debate?'
- (36) T איזקאיאאָקיאיזאָ *la-i*:=*la* HON-EQU=HON 'Yes.'
- (37) F هَن هُم مَ مَ اللهِ

 é:, iŋ-ga.
 oh EQU.PER-PQ
 'Oh, is (it) really (so)?'

- (38) B $5^{-i}\delta q^{-i} A^{i} T^{i} T^{i} A^{i} T^{i} A^{i} T^{i} A^{i} A^{i}$
- (39) $\underbrace{\mathbb{E}}_{a} \operatorname{A}_{a} \operatorname{A}_{a} \operatorname{A}_{a} \operatorname{B}_{a} \operatorname{E}_{a} \operatorname{B}_{a} \operatorname{E}_{a} \operatorname{B}_{a} \operatorname{E}_{a} \operatorname{B}_{a} \operatorname{E}_{a} \operatorname{B}_{a} \operatorname{E}_{a} \operatorname{$
- (40) F ॲ도· ॲ도·

õ: õ: yes yes 'Yes, yes.'

(41) B अन्न: सु. वर्ने (प्रम: क्रेंब. (phone) मुनया के जन के

dõ:	$k^h u$	dik ^h a	fon	kjap-tee	làp-tɛɛ.
yesterday	3sgm	here	phone	do-PST	say-PST
'(Somebod	y) told (n	ne) tha	at he pho	ned here	yesterday.'

Appendix 2: Differences between spoken and written language

Differences between spoken and written language were discussed throughout the grammar. Table 1 below provides a summary of the differences and refers to sections in the grammar where the differences are more fully illustrated and discussed. The headings in Table 1 describe the ways in which spoken language differs from written language. Spoken language is associated with phonological reduction and modification, morphosyntactic reduction (and possibly accompanying phonological reduction), semantic reduction (one example), morphosyntactic expansion, morphosyntatic flexibility, frequency of discourse-related constructions, and influence from Nepali and English (as opposed to influence from Literary Tibetan in written language).

Phonological reduction and modification	
VERB- po may be abbreviated to VERB- b (with verbs ending in the glottal stop or p	§1.2.7.3
and followed by the equative $b\epsilon$?) or to VERB- <i>m</i> (with verbs ending in the velar	
nasal and followed by the equative $b\epsilon^2$ or $\tilde{\iota}$).	
	80.0.1.4
In spoken language WD -al and -ar, which are occasionally pronounced in careful	§2.2.1.4
and spennig-style speech as /ai/ and /ai/ respectively, are pronounced as $\frac{1}{2}$ / and $\frac{1}{2}$	
Written progressive marker sate bzhin, which is pronounced /zin/ in spelling-style	\$3.3.6.5
pronunciation becomes $/z\tilde{\epsilon}'/\sim/te\tilde{\epsilon}''$ in typical spoken language (but /teou/ in	0
Martam).	
The written conditional form \overline{q} <i>na</i> corresponds to spoken forms $/n\epsilon/~/no/~/nu/$	§3.3.6.15
(depending on locality)	
The frequent anaphoric emphatic clitic $\pi rang$ tends to occur as short and	§6.2
denasalized $=ra$ in spoken language.	
The frequently-used complex copula $\tilde{u}_{\neg} \tilde{v} \approx j \tilde{\rho} - po b \epsilon^2$ (neg. $m \tilde{e} :-po b \epsilon^2$) is in	§7.3.2.1
spoken language reduced to jøbbe?/jebbe? (neg. mebbe?).	
The nominalized completive form র্জন ন্টা একা tshar-bo-las, suggesting the spelling-	§15.3.1.1
style pronunciation ts^har - bo - $l\varepsilon$, is in spoken language, depending on the consultant	
and phonological environment, reduced to $-ts^{h}o - u = l\varepsilon$, $-tsu - b = l\varepsilon$, $-ts^{h}o := l\varepsilon$	
or $-so:=l\varepsilon$.	
Most probably because of influence from Written Tibetan, many nouns that in	
pronunciation end in -pu are written as $\frac{1}{2}$ po by many writers (e.g. pompu $\frac{1}{2}\sqrt{3}$	
dpon-po).	
Written $\breve{\neg}$ bo, when following short vowels, reduces to $-u$ in pronunciation (e.g. \overline{a})	
ਸੱ $zla-bo > dau/dou$ 'month', ਜ਼੍ਹਾਸੱ ਯੋਗ $lta-bo$ in $> ta-u$ \tilde{t} 'I looked').	
Morphosyntactic reduction (possibly also phonological reduction)	P
The often used written form of the interrogative copula क्षेत्र ज्वा in-nam, suggesting	§3.3.7
a bimorphemic interpretation, merges in spoken language to monosyllabic and	
monomorphemic <i>nam</i> .	1

Table 1. How spoken language differs from written language

Relator nouns, which are in spoken language always followed by a case-marker,	§3.6.8
are occasionally in spoken language used without a case-marker, e.g. $t \in \eta = lo > t \in \eta$	
'on, above'.	
In spoken language, genitive-marking may be dropped from noun modifiers.	§4.1.2.2
In equative and locational clauses in spoken language, the copula may be elided.	§5.4.1 and §5.4.2
Case-marking in directionals seems more frequently dropped in spoken language	§5.6.1.1
than in written language.	
The frequent spoken filler-question /k'apam/~/k'ajɛm/~/k'aɛm/~/k'ɛm/ is an	§6.3.1
abbreviation of the written गुत्र [.] क्षेत्र जुत्रा <i>gan in-nam</i> .	
The apparentive construction $da \ b\epsilon$?, which corresponds to written $rac{5}{35}$ ' dra	§8.5.2
sbad and occurs as such in slower speech, is often in faster speech reduced to $d\varepsilon:/r\varepsilon$.	
The non-past construction $-\varepsilon \varepsilon i$ (WD \mathfrak{AS} \mathfrak{RA} shad in) often reduces to $-\varepsilon \tilde{\iota} : /\varepsilon i \eta$,	§9.1.1.1
which does not have a strictly corresponding written form.	
Past interrogative construction $-t\varepsilon\varepsilon$ -ka and future interrogative construction $-\varepsilon\varepsilon$ -ka	§11.1.1.2
in spoken language often reduce to -tea and -ea respectively. To my knowledge,	0
the reduced forms do not have standardized written equivalents.	
In spoken language purpose-marking converbal marker $-pa/ba$ (WD $\neg [\neg] \cdot \neg [\neg] \cdot \neg$	§15.5.1
may be dropped.	
The simultaneous converbal endings -sondã:/somdã:/sumdã: used in spoken	§3.3.6.18,
language probably derive from the longer construction $s\dot{\delta}$:-bo t' \ddot{a} : [go.PFV-2INF	§15.3.3.2
and] and the simultaneous converbal ending <i>tsubda</i> : possibly derives from <i>ts^ha</i> :-	
<i>bo t'ã:</i> [finish-2INF and].	
Semantic reduction	
The distinction of meaning between dative-locative and ablative case-marking,	§5.6.2
which is carefully retained in written language, largely collapses in spoken	
language (ablative is used for expressing non-directional stable location).	
Morphosyntatic expansion	
In spoken language, unlike in writing, the nonfinal converbal ending $-ti/di$ is often	§15.2
accompanied by the marker $-ki/gi$, which looks like a genitive or an agentive.	81546
whereas in written language causality may be expressed through agentive marking	§15.4.0, \$15.4.7
by the grapheme $-w$ -s, which suggests a lengthened vowel in spelling-style	§13.4.7
pronunciation, spoken language uses additional morphosyntax for the same	
purpose, i.e. the emphatic $= di$ and the nonfinal converb $p'jati(ki)$.	
Double/marking of genitive is frequent in spoken language and almost non-	§3.7.1.3
existent in written language.	82.2 < 17
Both the terminative endings $-sa$ and $-sonza$ occur in spoken language but only the short $a\tilde{a}$ in written language	§5.5.6.1/
Morphosyntostia flovibility	
In spoken language, the head noun of a noun phrase may acous between the	8/21
adjective modifier and adjective.	84.3.1
Agentive-marking in spoken language seems more pragmatically conditioned than	§5.3
in written language, which tends to be affected by prescriptive rules inherited from	
Classical Tibetan.	

The morphologically dependent distal demonstrative form o- 'that' is occasionally	§6.4		
in spoken language used independently			
Frequency of discourse-related morphemes and phenomena			
Unlike in written language, the use of the demonstrative-emphatic $=$ <i>di</i> is prevalent	§16.1.3		
in spoken language, sometimes occurring several times in one clause.			
Discourse particles $t'a$ and $t\varepsilon$ are more frequent in spoken than in written language	§16.4		
because they are used as fillers when the speaker does not know what to say.			
In spontaneous spoken language right-dislocation is a frequently used way for	§16.6		
speakers to correct themselves and add information that helps the addressee to			
disambiguate the clause.			
Influence of Nepali and English rather than Tibetan			
Code-mixing with Nepali/English and Nepali/English loan words are more			
frequent in spoken than in written language.			
Because Denjongke does not have a long literary history, the written language			
lacks standardized words for many concepts. Therefore authors, especially when			
translating, are likely to use Tibetan loan words which are not readily understood			
by non-literate speakers.			

Appendix 3: Vowel plot measurements

Table 2 below presents the F1 and F2 measurements for short and long vowels pronounced by TB, see Figure 2.7 in chapter 2 on phonology. It would have been ideal to record formant values in monosyllabic words but, because the audio data was recorded for other purposes than this particular study, disyllabic words are used in some cases to increase the number of examples.

Short	word	F1	F2	Long	word	F1	F2
vowel				vowel			
i	<i>cí</i> 'die'	388	2175	i:	phi:ru 'night'	287	2337
	$k^{h}i$ 'dog'	386	2309		ki: 'bind'	268	2565
	mí 'man'	256	2419		sì:p 'dew'	291	2327
	$p^{h}i$ 'open'	368	2292		dzi:ce? 'to split'	238	2423
	Average:	350	2299		Average:	271	2413
				e:	geː 'fall'	391	2441
					ge: 'weigh'	363	2529
					<i>dze:</i> 'place up'	373	2297
					<i>sé:p</i> 'yellow'	353	2218
					Average:	370	2371
3	ke 'neck'	471	2322	£ :	ge: win	471	2303
	gep 'bag'	501	2018		bɛːp frog	626	1914
	$d\varepsilon$ 'ghost'	493	2119		gɛːp king	455	2146
	<i>έmmo</i> 'left'	506	2147		she: rinse	461	2271
	pepo 'leech'	524	1912		Average:	503	2159
	Average:	499	2104				
a	tsa 'grass'	676	1640	aː	t'aːtɕiʔ 'again'	713	1626
	t'atci? 'recently'	701	1627		<i>te^ha:ka</i> 'thing'	855	1576
	k^ha 'mouth'	746	1450		ma:p 'red'	686	1314
	k^ha^2 'soup'	793	1629		ka:p 'whit'e	769	1498
	ŋà 'I'	747	1480		<i>tc^ha:p</i> 'rain'	832	1401
	<i>ápo</i> 'father'	711	1292		Average:	771	1434
	Average:	729	1587				
0	<i>ko</i> 'dig'	644	1093	O.	<i>ko:</i> 'throw'	386	791
	<i>p´jo</i> 'pour'	645	1245		<i>p'jo:</i> 'escape'	411	1072
	<i>tsho</i> 'lake'	701	1136		<i>tsho:</i> 'father'	s414	892
	do 'stone'	598	1156		sister's husband'		
	Average:	647	1158		do: 'touch'	394	957
					Average:	401	928
u	súpo 'stomach'	376	1221	u:	<i>phu:</i> 'fly'	317	960
	khu 'he'	333	894		<i>dzu:</i> 'be afraid'	343	1128
	chu 'water'	377	948		Average:	330	1044
	<i>tup</i> 'be alright'	397	1137				
	Average:	371	1050				
				y:	ky:ce? 'drive'	310	2319
					<i>doly?</i> 'custom'	311	2326
				1	dvn 'seven'	323	2165

Table 2. Vowel plot measurements

ý: 'place'	242	2303
average:	297	2278
ø: cø:m 'cockroach'	448	2125
søːna 'soot'	401	1946
<i>ø:la?</i> 'raven'	373	1962
sớ:za 'tea (hon.)'	343	2066
average:	391	2025

Appendix 4: Letter of informed consent

Informed consent from the consultants was received through the following Denjongke letter, which is translated into English on the next page. As suggested at the end of the written form, the contents of the letter were explained to the signers in spoken language.

यावत-'च' र्सेप'र्थर' भे'यो। તેંવા'ર્শ' हेंगर्भ, मुप्तस' झें' प्रस' University of Helsinki'બ્યસ' ફેંद्र, અપવર, 'ફ્રેં'झेट्'व, 'क़्सस' देप' डॉस' આવર,' Juha Yliniemi'ર્થ' प्र'गीस' બप' थॅट्'प्यते' 'ફ્रें'झेट्'ग्रे' recording'र्ड्र' ट्रट' प्रस' प्रमट,' थॅट्'प्यते' ढेंवा' ट्रट' ढेंवा'श्चुप्तर्स' हिं' तंप्र'' केंवा' डेंब'' ट्रट' ढेंवा'श्चुप्तर' हें' केंवा त्वा'र्थ' हेगर्भ, मुप्तस' झे' प्र'गेस' भप' थॅट्'प्यते' हें'झट्'ग्रे' संग'श्रस' हें' प्रहि' प्रात्न' हें' केंव'' पविट'प्त' हेगर्भ, मुप्तस' हैं' प्र'प्त' थें'' केंव'' पविट'प्त' हेगर्भ, मुप्तस' हैं' प्र'प्त' थें'' सेंच'' पविट' क्रम्स' मुप्तस' हैं' प्र'प्त' थें'' सेंच'' पविट' क्रम्स' केंप्र' पविट'यते' हें'झट थें''आप' यावट'प्त' हें' प्रात्न'' य' पविट' क्रम्स' देप''र्ट' केंग्र' पविट'यते' हें'झट थें''' यावट'प्त'' यावट'प्त'' क्रेंगोस' भय' थें'' हेंग'र्ड' पविट' विप'र्म' केंग' खेंद' प्त'' केंप' संग' हेंग'र्ड' केंग्र' पविट'प्त' हैं' क्रम्स' यावट'प्त'' क्रम्' विद' केंप' संग'र्ड' केंग्र' यावट'प्त'' क्रम' खेंप'र्स' केंग'र्ड' र्ट' केंग्र' यावट'प्त''र्ड' क्रम्स' केंप'र्स' यावट'प्त'' क्रम' क्रम' संग'र्स' केंग'र्ड' र्ट' केंग'र्ड' प्र'न'र्स' केंस' क्रि'स्ट'' क्रम'र्स' संग'र्स' केंग' पत्'' केंप'र्स' केंग'र्ड' र्ट' केंग'र्ड' प्र'न' केंप' क्रम'र्स' केंग'र्ड' केंग'र्ड' केंग'र्ड' केंग'र्ड' प्र'न'र्स' केंग'र्ड' क्रम'र्ड' केंस' देस' केंट' स्वस'र्ड' केंग'र्ड' केंग'र्ड' केंग'र्ड' प्र'न' प्र'र्ड' केंग'र्ड' देर' केंट' स्वस'र्ट' केंग'र्ड' केंग'र्ड' केंग'र्ड' प्र'र्ड' केंग'र्ड' केंग'र्ड' केंग'र्ड' केंग'र्ड' केंग'र्ड' देर' केंट' स्वस'र्ड' केंग'र्ड' केंग'र्ड' केंग'र्ड' केंग'र्ड' देर' केंट' स्वस'र्ड' केंग'र्ड' केंग'र्ड' केंग'र्ड' केंग'र्ड' केंग'र्ड' केंग'र्ड' केंग'र्ड' देर' केंट' स्वस'र्ड' केंग'र्ड' देर' केंर' स्वस'र्ड' केंग'र्ड' केंग'र्ड' दर' केंग'र्ड' केंग'र्ड' केंग'र्ड' केंग'र्ड' केंग'र्ड'' दर्त' केंट' केंग'र्ड' कें

र्वेषागाः तन्ते में र्थेन स्थापत्र केंगार्खते मेंत्र न्या तन्ते न्यां याययाद्याय र्घ्याक्षे याक्षन्त यानन थेना

ळें ग्नू ८२०:______

ਗਲੱਕ:		
<u> त्रा</u> र्न्स:		
্রান্ড ক্লান্য:		

Letter of permission

I_____ by ticking (the box) below grant permission to Juha Yliniemi from the University of Helsinki, who carries out research on Lhoke, to make known the words and sentences recorded from me to all people in the world through his research writing.

By ticking (the box) below, I grant permission to make available the recordings of my speech for the purpose of listening, reading and watching (video-recordings) not only to Juha Yliniemi but also to other researchers and anyone who is interested.

When words and sentences spoken by me occur in research writings

I give the permission to use my full name.

I give the permission to use my initials only.

my name or initials should not be used.

The contents of this letter have been clearly explained to me in spoken language.

Date:		
Name:		

Age:		
Birth	place:	

References

Aikhenvald, Alexandra. 2004. Evidentiality. Oxford: Oxford University Press.

- Aikhenvald, Alexandra. 2007. "Typological distinctions in word-formation". In Shopen, Timothy (ed.), *Language typology and syntactic description 3: Grammatical categories and the lexicon*, 1-65. Cambridge: Cambridge University Press. <u>https://doi.org/10.1017/CBO978051161843</u> 7.001
- Aikhenvald, Alexandra. 2015. "Evidentials: Their links with other grammatical categories". *Linguistic Typology* 19(2): 239-277. <u>https://doi.org/10.1515/lingty-2015-0008</u>
- Andersen, Paul Kent. 1989. "Zero-anaphora and related phenomena in Classical Tibetan". *Studies in Language* 11, 279-312. <u>https://doi.org/10.1075/sl.11.2.02and</u>
- Andvik, Erik. 2010. A grammar of Tshangla. Leiden: Brill. <u>https://doi.org/10.1163/ej.9789004178</u> 274.i-490
- Balikci-Denjongpa, Anna. 2002. "Kangchendzönga: Secular and Buddhist perceptions of the mountain deity of Sikkim among the Lhopos". In *Bulletin of Tibetology* 38(2), 5-37. Gangtok: Namgyal Institute of Tibetology. <u>http://himalaya.socanth.cam.ac.uk/collections/journals/bot/ pdf/bot_2002_02_01.pdf</u> Last access 21 Jan 2021.
- Balikci, Anna. 2008. *Lamas, shamans and ancestors: Village religion in Sikkim*. Leiden: Brill. <u>ht</u> tps://doi.org/10.1111/j.1748-0922.2012.01614_2.x
- Balikci-Denjongpa, Anna. 2011. "Introduction part II: Sikkim studies and the Namgyal Institute of Tibetology". In Balikci-Denjongpa, Anna ; and McKay, Alex (eds.), *Buddhist Himalaya: studies in religion, history and culture, volume II: The Sikkim papers* (Proceedings of the golden jubilee conference of the Namgyal Institute of Tibetology, Gangtok 2008), 11-28. Gangtok: Namgyal Institute of Tibetology.
- Bartee, Ellen. 2007. *A Grammar of Dongwang Tibetan*. Ph.D. diss, University of California, Santa Barbara. <u>http://hdl.handle.net/11707/1022</u> Last access 21 Jan 2021.
- Beyer, Stephan V. 1992. *The classical Tibetan language*. New York: State University of New York Press.
- Bhutia, Karma Lobsang. 2013. אָיקאָר קר קאָאיקאָר (Sikkimese Bhutia oral stories and moral dialects). Gangtok: Bhutia Kayrab Yargay Tsogpo.
- Bhutia, Norden Tshering and Takchungdarpo, Pema Ringzing. 2001. *Bhutia-English Dictionary*. Gangtok: published by authors.
- Bhutia, Pintso. 2005. Bhutia-English Dictionary. 2nd edn. Gangtok: published by author.
- Bhutia, Tshering Thendup and Bhutia, Thupten Palzang. 2012. འརོམ་གྲིར་ གྲོད་ਘིག་གི་ སྐྱོ་ནི་ མུན་གསོལ་ མུག
- Bhutia, Yeshe Rinzing. Unpublished manuscript 2008. Bhutia language learning course book (גָּי אָק־ אָק־אָק־ אָק־אַק־ אַק־אָק־ אַק־אָק־ אָק־אַק־ אָק־אַק־ אָק־אַק־ אָק־אַק־ אָק־אַק־ אָק־אַק־ אָק־אַק־ אָק־אַק־ אָק־אַק־ אַק־אַק־ אַק־אַק־ אַק־אַק־ אַק־אַק־ אָק־אַק־ אַק־אַק־ אַק־אַק־ אַק־אַק־ אַק־אַק־ אָק־אַק־ אַק־אַק־ אַקי
- Bielmeier, Roland. 2000. "Syntactic, semantic, and pragmatic-epistemic functions of auxiliaries in Western Tibetan". *Linguistics of the Tibeto-Burman Area* 23:2, 79-125.
- http://sealang.net/sala/archives/pdf8/bielmeier2000syntactic.pdf Last access 21 Jan 2021.
- Bielmeier, Roland. 1985. Das Märchen vom Prinzen Cobzan: eine tibetische Erzählung aus Baltistan, text, Übersetzung, Grammatik und westtibetisch vergleichendes Glossar. Sankt Augustin: VGH.

- Blake, B. J. 1994. *Case*. Cambridge: Cambridge University Press [Cambridge textbooks in linguistics 32]. <u>https://doi.org/10.1017/CBO9781139164894</u>
- Bradley, David. 1997. "Tibeto-Burman languages and classification". In Bradley, David (ed.), *Tibeto-Burman languages of the Himalayas* [Pacific linguistics series A-86, Papers in Southeast Asian linguistics 14], 1-72. Department of Linguistics, Research School of Pacific and Asian studies, the Australian National University. <u>http://sealang.net/sala/archives/pdf4/bradley1997</u> <u>tibeto-burman.pdf</u> Last access 21 Jan 2021.
- Bradley, David. 2002. "The subgrouping of Tibeto-Burman". In Beckwith, Christopher I. (ed.), *Medieval Tibeto-Burman languages* (PIATS 2000: Proceedings of the ninth seminar of the International Association for Tibetan Studies, Leiden 2000), 73-112. Leiden: Brill.
- Caplow, Nancy. 2016. "Stress patterns and acoustic correlates of stress in Balti Tibetan". *Himalayan Linguistics* 15(2): 1-49. <u>https://doi.org/10.5070/H915232601</u>
- Causemann, Margret. 1989. *Dialekt und Erzählungen der Nangchenpas*. Beiträge zur tibetischen Erzählforschung herausgegeben von Dieter Schuh, Band 11. Bonn: VGH Wissenschaftsverlag.
- Chafe, Wallace (ed.). 1980. *The Pear Stories: Cognitive, Cultural, and Linguistic Aspects of Narrative Production*. Norwood, New Jersey: Ablex. <u>https://doi.org/10.1017/S0047404500008</u> <u>897</u>
- Chang, Kun ; and Chang, Betty Shefts. 1984. "The certainty hierarchy among Spoken Tibetan verbs of being". *Bulletin of the Institute of History and Philology (Academia Sinica)* 55:4, 603-635. <u>https://doi.org/10.6355/BIHPAS.198412.0603</u>
- Chelliah, Shobhana L. 1997. *A grammar of Meithei*. Berlin: Mouton de Gruyter. <u>https://doi.org/</u> <u>10.1515/9783110801118</u>
- Chelliah, Shobhana L. 2009. "Semantic role to new information in Meithei". In Barðdal, Jóhanna and Chelliah, Shobhana L. (eds.), *The role of semantic, pragmatic and discourse factors in the development of case*, 377-400. Amsterdam: John Benjamins. <u>https://doi.org/10.1075/slcs.108.</u> 20che
- Comrie 1976. Aspect. Cambridge: Cambridge University Press.
- Contini-Morava, Ellen. 1989. Discourse pragmatics and semantic categorization: The case of negation and tense-aspect with special reference to Swahili. Berlin: Mouton de Gruyter [Discourse Perspectives on Grammar 1]. https://doi.org/10.1017/S0041977X00005292
- Coupe. 2006. "Converbs". In: Brown, Keith (ed.), *Encyclopedia of Languages and Linguistics* (2nd edn.), 145-152. Oxford: Elsevier. <u>https://www.academia.edu/616075/Converbs</u> Last access 21 Jan 2021.
- Coupe, Alexander. 2007. *A grammar of Mongsen Ao*. Berlin: Mouton de Gruyter[Mouton Grammar Library 39]. <u>https://doi.org/10.1515/9783110198522</u>
- Coupe, Alexander. 2011. "On core case markin patterns in two Tibeto-Burman languages of Nagaland". *Linguistics of the Tibeto-Burman Area* 34(2), 21-47. <u>http://sealang.net/archives/ltba/pdf/LTBA-34.2.21.pdf</u> Last access 21 Jan 2021.
- Coupe, Alexander. 2017. The typology of pragmatic case-marking in Tibeto-Burman. Presentation at South Asian Lingustic Analysis Roundtable (SALA33), Adam Mickiewicz University, Poznań, 15-17 May 2019. <u>https://www.academia.edu/33155200/The_typology_of_pragmatic_case_marking_in_Tibeto_Burman</u> Last access 21 Jan 2021.
- Croft, William. 1991. "The evolution of negation". *Journal of Linguistics* 27, 1-39. <u>https://doi.org/</u>10.1017/S0022226700012391
- Daudey, Henriëtta. 2014. *A grammar of Wadu Pumi*. Ph.D. diss, La Trobe University. <u>https://www.jcu.edu.au/data/assets/pdf_file/0006/1099590/A_Grammar_of_Wadu_Pumi.pdf</u> Last access 22 jan 2021.

- DeLancey, Scott. 1990. "Ergativity and the cognitive model of event structure". *Cognitive Linguistics* 1, 289-321. <u>https://doi.org/10.1515/cogl.1990.1.3.289</u>
- DeLancey, Scott. 1991. "The origins of verb serialization in modern Tibetan". *Studies in Language* 15(1), 1-23. <u>https://doi.org/10.1075/sl.15.1.02del</u>
- DeLancey, Scott. 1992. "The historical status of the conjunct/disjunct pattern in Tibeto-Burman". *Acta linguistica Hafniensia* 25, 39-62. <u>https://doi.org/10.1080/03740463.1992.10412277</u>
- DeLancey, Scott. 1997a. "Mirativity: The grammatical marking of unexpected information". *Linguistic Typology* 1, 33-52. <u>https://doi.org/10.1515/lity.1997.1.1.33</u>
- DeLancey, Scott. 1997b. "Grammaticalization and the gradience of categories: Relator nouns and postpositions in Tibetan and Burmese". In Bybee, Joan; Haiman, John; and Thompson, Sandra A. (eds.), *Essays on language function and language type* (Dedicated to T. Givón), 51-69. Amsterdam: John Benjamins. <u>https://doi.org/10.1075/z.82.07del</u>
- DeLancey, Scott. 1999. "Relativization in Tibetan". In Yadava, Yogendra and Glover, Warren (eds.), *Studies in Nepalese Linguistics*, 231-49. Kathmandu: Royal Nepal Academy. <u>https://www.academia.edu/3930401/Relativization_in_Tibetan</u> Last access 23 Jan 2021.
- DeLancey, Scott. 2018. "Evidentiality in Tibetic". In Aikhenvald, Alexandre (ed.), *The Oxford handbook of evidentiality*. Oxford University Press. <u>https://doi.org/10.1093/oxfordhb/97801987</u> 59515.013.27
- Denzongpo, Tashi; Tsichudarpo, Bhaichung; and Takchungdarpo, Pema Rinzing. 2009. אָישׁשִי אָשִי אַשִי אַשִי אַשִי אַשִי אַשִי אַשִי אַשִי אַשִי אַשיי אַשִי אַשיי אַשיי
- Denzongpo, Tashi; Tsichudarpo, Bhaichung; and Takchungdarpo, Pema Rinzing. 2011. אֹיעָאן' אָאַז' בָּקָאָ אָי (Class 7 Denjongke textbook). Gangtok: Human Resource Development, Government of Sikkim.
- Denwood, Philip. 1999. Tibetan. Amsterdam: John Benjamins. https://doi.org/10.1075/loall.3
- Dewan, Dick B. 2012. *Education in Sikkim: an historical retrospect, pre-merger and post-merger period.* Pedong, Kalimpong: Tender Buds' Society.
- Ding, Picus Sizhi. 2014. A grammar of Prinmi: based on the central dialect of Northwest Yunnan, China. Leiden: Brill [Brill's Tibetan Studies Library, Languages of the Greater Himalayan Region 5/14].
- Dingemanse, Mark. 2011. *The meaning and use of ideophones in Siwu*. Ph.D. diss, Radboud University. <u>http://thesis.ideophone.org/</u> Last access 22 Jan 2020.
- Dixon, R. M. W. 1997. *The Rise and Fall of Languages*. Cambridge: Cambridge University Press. https://doi.org/10.1017/CBO9780511612060
- Dixon, R. M. W. 2010a. *Basic linguistic theory*, Volume 1, Methodology. Oxford: Oxford University Press.
- Dixon, R. M. W. 2010b. *Basic linguistic theory*, Volume 2, Grammatical topics. Oxford: Oxford University Press.
- Dixon, R. M. W. and Aikhenvald, Alexandra. 2003. *Word: a cross-linguistic typology*. Cambridge: Cambridge University Press. <u>https://doi.org/10.1017/S0952675704210119</u>
- Doke, C. M. 1935. *Bantu linguistic terminology*. London: Longmans, Green. <u>https://doi.org/10.</u> 2307/1155306
- Dokhangba, Sonam Gyatso. 2014. Video-recorded oral exposition on Lhopo marriage customs (in Lhoke/Denjongke), January 2014.

- Driem, George van. 1998. *Dzongkha*. Leiden: Research School CNWS [Languages of the Greater Himalayan Region 1].
- Driem, George van 2007. "Endangered languages of South Asia". In Brenzinger, Matthias (ed.), *Language diversity endangered*, 303-341. Berlin: Mouton de Gruyter [Trends in Linguistics 181]. <u>https://www.himalayanlanguages.org/files/driem/pdfs/2007EndangeredLggsSouthAsia.</u> <u>pdf</u> Last access 22 Jan 2021.
- Dryer, Matthew. 2007. "Clause types". In Shopen, Timothy (ed.), *Language typology and syntactic description 1: Clause Structure*, 224-275. Cambridge: Cambridge University Press. <u>https://doi.org/10.1017/CBO9780511619427.004</u>
- Dryer, Matthew. 2013. "Order of Relative Clause and Noun". In Dryer, Matthew S. and Haspelmath, Martin (eds.) *The World Atlas of Language Structures Online*. Leipzig: Max Planck Institute for Evolutionary Anthropology. <u>http://wals.info/chapter/90</u> Last access 22 Jan 2021.
- Ebihara, Shiho. "The inclusive-exclusive distinction in spoken and written Tibetan". Undated manuscript.<u>https://kobe-cufs.repo.nii.ac.jp/index.php?action=pages_view_main&active_action = repository_action_common_download&item_id=1943&item_no=1&attribute_id=22&file_n o=1&page_id=13&block_id=17_Last access 22 Jan 2021.</u>
- Everett, Daniel. 2001. "Monolingual field research". In Newman, Paul; and Ratliff, Martha (eds.), *Linguistic Fieldwork*, 166-188. Cambridge: Cambridge University Press. <u>https://doi.org/10.10</u> <u>17/CBO9780511810206.009</u>
- Garrett, Edward. 2001. *Evidentiality and Assertion in Tibetan*. Ph.D. diss, University of California, Los Angeles. <u>https://linguistics.ucla.edu/general/dissertations/GarrettEdwardDissertationUCL A2001.pdf</u> Last access 22 Jan 20121.
- Gawne, Lauren. 2013. Lamjung Yolmo Copulas in Use: Evidentiality, Reported Speech and Questions. Ph.D. diss, University of Melbourne. <u>http://hdl.handle.net/11343/38104</u> Last access 22 Jan 2021.
- Gawne, Lauren. 2015. "The reported speech evidential particle in Lamjung Yolmo". *Linguistics of the Tibeto-Burman Area* 38(2), 292-318. <u>https://doi.org/10.1075/ltba.38.2.09gaw</u>
- Gawne 2017. "Egophoric evidentiality in Bodish languages". In Hill, Nathan. and Gawne, Lauren (eds.), *Evidential systems of Tibetan languages*, 61-94. Berlin: Mouton de Gruyter [Trends in Linguistics. Studies and Monographs [TiLSM] 302]. <u>https://doi.org/10.1515/9783110473</u> 742-003
- Genetti, Carol. 1986. "The development of subordinators from postpositions in Bodic languages". *Proceedings of the twelfth annual meeting of the Berkeley Linguistics Society*, 387-400. Berkeley: Berkeley Linguistics Society. <u>https://doi.org/10.3765/bls.v12i0.1873</u>
- Genetti, Carol. 1992. "Semantic and grammatical categories of relative-clause morphology in the languages of Nepal". *Studies in Language* 16(2), 405-427. <u>https://doi.org/10.1075/sl.16.2.07gen</u>
- Genetti, Carol. 2005. "The participial construction in Dolakha Newar: Syntactic implications of an Asian converb". *Studies in Language* 29(1), 35-87. https://doi.org/10.1075/sl.29.1.03gen
- Genetti, Carol. 2011. "Nominalization in Tibeto-Burman Languages of the Himalayan Area: A typological perspective". In Yap, Foong Ha. and Wrona, Janick (eds.), *Nominalization in Asian Languages: Diachronic and typological perspectives*, 163-193. Amsterdam: John Benjamins [Typological Studies in Language 96]. <u>https://doi.org/10.1075/tsl.96.06gen</u>
- Givon, Talmy. 1984. *Syntax: a functional typological introduction*. Vol. I. Amsterdam: John Benjamins. <u>https://doi.org/10.1017/S0022226700010434</u>

- Givon, Talmy. 2001. Syntax: an introduction. Amsterdam: John Benjamins.Goldstein, https://doi.org/10.1075/z.syn1
- Melvyn C.; Rimpoche, Gelek; and Phuntshog, Lobsang. 1993 [1991]. *Essentials of Modern Literary Tibetan: A reading course and reference grammar*. New Delhi: Munshiram Manoharlal Publishers.
- Grierson, G. A. 1909. *Linguistic Survey of India. Tibeto-Burman Family*. Vol 3, part 1. Calcutta: Superintendent of government printing. <u>https://dsal.uchicago.edu/books/lsi/</u> Last access 22 Jan 2021.
- Grimes, Barbara F. (ed.). 1996. *Ethnologue: languages of the world*, 13th edition. Dallas: Summer Institute of Linguistics. <u>https://www.ethnologue.com/13/</u> Last access 22 Jan 2021.
- Grünthal, Riho. 2019. *Vepsän kielioppi* (A grammar of Veps).. Helsinki: Suomalais-Ugrilainen Seura [Apuneuvoja suomalais-ugrilaisten kielten opintoja varten XVII].

https://www.sgr.fi/uudetsivut/items/show/5 Last access 22 Jan 2021.

- Hale, Austin. 1971. *Person Markers: Conjunct and Disjunct Forms*. SIL mimeograph [Topics in Newari Grammar 1].
- Hale, Austin. 1980, "Person markers: finite conjunct and disjunct forms in Newari". In Trail, R. et al (eds.), *Papers in Southeast Asian Linguistics* 7: 95-106.
- Haller, Felix. 2000. "Verbal categories of Shigatse Tibetan and Themchen Tibetan". *Linguistics of the Tibeto-Burman Area* 23(2), 175-191. <u>http://sealang.net/sala/archives/pdf8/haller2000verbal.</u> pdf Last access 22 Jan 2021.
- Haller, Felix. 2009. "Switch reference in Tibetan". *Linguistics of the Tibeto-Burman Area* 32(2), 45-70. <u>http://sealang.net/archives/ltba/pdf/LTBA-32.2.45.pdf</u> Last access 22 Jan 2021.
- Haller, Felix. 2012. "Vowel harmony in Shigatse Tibetan". *Linguistics of the Tibeto-Burman Area* 35(2), 33-47. <u>http://sealang.net/archives/ltba/pdf/LTBA-35.2.33.pdf</u> Last access 22 Jan 2021.

Hari, Anna Maria. 1979. An Investigation of the Tones of Lhasa Tibetan. Bern: University of Bern.

- Haspelmath, Martin. 1993. *A grammar of Lezgian*. Berlin: Mouton de Gruyter [Mouton Grammar Library 9]. <u>https://doi.org/10.1515/9783110884210</u>
- Haspelmath, Martin. 1995. "The converb as a cross-linguistically valid category". In Haspelmath, Martin and König, Ekkehard (eds.), *Converbs in cross-linguistic perspective*, 1-55. Berlin: Mouton de Gruyter. <u>https://zenodo.org/record/227108#.YArGvxbkvIU</u> Last access 22 Jan 2021.
- Haspelmath, Martin. 2001. "Word classes and parts-of-speech". In Smelser, N. J. and Baltes B. (eds.), *International Encyclopedia of the Social and Behavioral Sciences*, 16538-16545. Elsevier. <u>https://doi.org/10.1016/B0-08-043076-7/02959-4</u>
- Haspelmath, Martin. 2005. "Argument marking in ditransitive alignment types". *Linguistic Discovery* 3(1), 1-21. <u>https://doi.org/10.1349/PS1.1537-0852.A.280</u>
- Haspelmath, Martin. 2007. "Coordination". In Shopen, Timothy (ed.), *Language typology and syntactic description 2: Complex constructions*, 1-51. Cambridge: Cambridge University Press. https://doi.org/10.1017/CBO9780511619434.001
- Haspelmath, Martin. 2009. "Terminology of case". In: Malchukov, Andrej and Spencer, Andrew (eds.), *The Oxford handbook of case*, 505-517. Oxford: Oxford University Press. https://doi.org/10.1093/oxfordhb/9780199206476.013.0034
- Haspelmath, Martin. 2010a. "Framework-free grammatical analysis". In Heine, Bernd and Narrog, Heiko (eds.), *The Oxford Handbook of Grammatical Analysis*, 341-365. Oxford: Oxford University Press. <u>https://doi.org/10.1093/oxfordhb/9780199544004.013.0014</u>
- Haspelmath, Martin. 2010b. "Comparative concepts and descriptive categories in crosslinguistic studies". *Language* 86(3), 663-687. <u>https://doi.org/10.1353/lan.2010.0021</u>

- Haspelmath, Martin. 2016. "The serial verb construction: Comparative concept and cross-linguistic generalizations". *Language and Linguistics* 17(3). 291-319. <u>https://doi.org/10.1177/239700221</u> 5626895
- Hawkins, John, A. 2004. *Efficiency and complexity in grammars*. Oxford: Oxford University Press. https://doi.org/10.1093/acprof:oso/9780199252695.001.0001
- Hein, Veronika. 2001. "The role of the speaker in the verbal system of the Tibetan dialect of Tabo/Spiti". *Linguistics of the Tibeto-Burman Area* 24(1), 35-48. <u>http://sealang.net/sala/ar chives/pdf8/hein2001role.pdf</u> Last access 22 Jan 2021.
- Heine, Bernd. 1997. *Possession: Cognitive Sources, Forces and Grammaticalization*. Cambridge: Cambridge University Press. <u>https://doi.org/10.1017/CBO9780511581908</u>
- Hengeveld, Kees and Olbertz, Hella. 2012. "Didn't you know? Mirativity does exist!". *Linguistic Typology* 16(3): 487-503. <u>https://doi.org/10.1515/lity-2012-0018</u>
- Herrmann, Silke. 1989. *Erzählungen und Dialekt von Dinri*. Beiträge zur tibetischen Erzählforschung herausgegeben von Dieter Schuh, Band 9. Bonn: VGH Wissenschaftsverlag.
- Hildebrandt, Kristine. 2005. "A phonetic analysis of Manange segmental and suprasegmental properties". *Linguistics of the Tibeto-Burman Area* 28, 1-36. <u>http://sealang.net/sala/archives/pdf8/hildebrandt2005phonetic.pdf</u> Last access 22 Jan 2021.
- Hill, Nathan W. 2010. "Personal pronouns in Old Tibetan". Journal Asiatique 298(2), 549-571. https://doi.org/10.2143/JA.298.14.2062444
- Hill, Nathan W. 2012. "Mirativity' does not exist: hdug in 'Lhasa' Tibetan and other suspects". *Linguistic Typology* 16(3), 389-433. <u>https://doi.org/10.1515/lity-2012-0016</u>
- Hill, Nathan W. 2013. "Contextual semantics of 'Lhasa' Tibetan evidentials". SKASE Journal of Theoretical Linguistics 10(3). <u>https://eprints.soas.ac.uk/id/eprint/17855</u> Last access 22 Jan 2021.
- Hill, Nathan W. 2017. "Perfect experiential constructions: the inferential semantics of direct evidence". In Hill, Nathan and Gawne, Lauren (eds.), *Evidential systems of Tibetan languages*, 131-159.. Berlin: Mouton de Gruyter [Trends in Linguistics. Studies and Monographs [TiLSM] 302]. <u>https://doi.org/10.1515/9783110473742-005</u>
- Hongladarom, Krisadawan. 2007. "Evidentiality in Rgyalthang Tibetan". *Linguistics of the Tibeto-Burman Area* 30(2), 17-44. <u>http://sealang.net/archives/ltba/pdf/LTBA-30.2.17.pdf</u> Last access 22 Jan 2021.
- Hopper, Paul; and Thompson, Sandra. 1980. "Transitivity in grammar and discourse". *Language* 56(2), 251-299. <u>https://doi.org/10.1353/lan.1980.0017</u>
- Huber, Brigitte. 2000. "Preliminary report on evidential categories in Lende Tibetan (Kyirong)". *Linguistics of the Tibeto-Burman Area* 23(2), 155-174. <u>http://sealang.net/sala/archives/pdf8/</u> <u>huber2000preliminary.pdf</u> Last access 22 Jan 2021.
- Huber, Brigitte. 2002. *The Lende subdialect of Kyirong Tibetan: a grammatical description with historical annotations*. Ph.D. diss, University of Bern
- Hutt, Michael; and Subedi, Abhi. 1999. Teach yourself Nepali. London: Hodder and Stoughton.
- Hyslop, Gwendolyn. 2010. "Kurtöp case: the pragmatic ergative and beyond". *Linguistics of the Tibeto-Burman Area* 33(1), 1-40. <u>http://sealang.net/archives/ltba/pdf/LTBA-33.1.1.pdf</u> Last access 22 Jan 2021.
- Hyslop, Gwendolyn. 2011. A grammar of Kurtöp. Eugene: PhD. Diss, University of Oregon. https://my.b-ok.as/book/2664953/7bb589 Last access 22 Jan 2021.
- Hyslop, Gwendolyn. 2014. "On the category of speaker expectation of interlocutor knowledge in Kurtöp". In Leung, Herman, Zachary O'Hagan, Sarah Bakst, Auburn Lutzross, Jonathan Manker, Nicholas, Rolle; and Sardinha, Katie (eds.), *Proceedings of the fortieth annual meeting*

of the Berkeley linguistics society (February 7-9, 2014). Berkeley: Berkeley Linguistics Society. <u>https://doi.org/10.3765/bls.v40i0.3140</u>

- Hyslop, Gwendolyn; and Tshering, Karma. 2010. "Preliminary notes on Dakpa (Tawang Monpa)".
 In Morey, Stephen; and Post, Mark (eds.), *North East Indian Linguistics* 2, 3-21. New Delhi: Foundation/Cambridge University Press. <u>https://doi.org/10.1017/9789382993285</u>
- Jäschke, Heinrich August. 1881. A Tibetan-English Dictionary: with special reference to the prevailing dialects, to which is added an English-Tibetan vocabulary. Reprint 2017. Delhi: Motilal Banarsidass.
- Jespersen, Otto. 1924. Philosophy of grammar. London: Allen and Unwin.
- Jespersen, Otto. 1965. *A Modern English Grammar on Historical Principles*, Part VI, Morphology. London: George Allen and Unwin Ltd.
- Keenan, Edward. L. and Bernard. Comrie. 1977. Noun Phrase Accessibility and Universal Grammar. *Linguistic Inquiry* 8(1), 63-99. <u>https://www.jstor.org/stable/4177973</u> Last access 22 Jan 2021.
- Khenpo Lha Tshering. 2016. Speech at the Namgyal Institute of Tibetology, Gangtok, July 21, 2016.
- Kittilä, Seppo; and Yurayong, Chingduang. In press. "Demonstrative and pronoun as enclitic discourse markers in Finnish and Finnic". In: Yap, Foong Ha (ed.), *Deixis in discourse*.
- König, Ekkehard; and Siemund, Peter. 2007. "Speech act distinctions in grammar". In: Shopen, Timothy (ed.), *Language typology and syntactic description 1: Clause structure*, 276-324. Cambridge: Cambridge University Press. <u>https://doi.org/10.1017/CB09780511619427.005</u>
- Konnerth, Linda. 2014. A grammar of Karbi. Ph.D. diss, Eugene: University of Oregon. https://doi.org/10.7264/N3DN43BG
- Kretschmar, Monika. 1986. *Erzählungen und Dialekt der Drokpas aus Südwest-Tibet*. Sankt Augustin: VGH Wissenschaftsverlag [Beiträge zur tibetischen Erzählforschung, Bd. 8].
- Krifka, Manfred. 1999. Additive Particles under Stress. In: Strolovitch Devon; and Lawson, Aaron. (eds.), *Proceedings of SALT 8*, 111-28, Ithaca, NY: CLC Publications. <u>https://doi.org/10.3765/salt.v8i0.2799</u>
- Lama, Dorjee Rinchen. 2013. English-Bhutia-Hindi-Nepali dictionary. Gangtok: Kwality stores.
- LaPolla, Randy. 1995. "Ergative marking in Tibeto-Burman". In: Nishi, Yoshio; Matisoff, James A.;and Nagano, Yasuhiko (eds.), *New horizons in Tibeto-Burman morphosyntax*, 189-228. Osaka: National Museum of Ethnology [Senri Ethnological Studies 41]. <u>https://core.ac.uk/ download/pdf/14514807.pdf</u> Last Accessed Jan 22, 2021.
- LaPolla, Randy. 2003. *A grammar of Qiang with annotated texts and glossary*. Berlin: Mouton de Gruyter [Mouton Grammar Library 31]. <u>https://doi.org/10.1515/9783110197273</u>
- LaPolla, Randy; and Tournadre, Nicolas. 2014. "Towards a new approach to evidentiality: issues and directions for research". *Linguistics of the Tibeto-Burman Area* 37(2): 240-263. <u>https://doi.org/10.1075/ltba.37.2.04tou</u>

Levinson, Stephen. 1983. Pragmatics. Cambridge: Cambridge University Press.

- Lidz, Liberty A. 2011. "Agentive marking in Yongning Na (Mosuo)". *Linguistics of the Tibeto Burman Area* 34(2), 49-72. <u>http://sealang.net/archives/ltba/pdf/LTBA-34.2.49.pdf</u> Last accessed Jan 22, 2021.
- Longacre, Robert E. 2007. "Sentences as combinations of clauses". In: Shopen, Timothy (ed.), Language typology and syntactic description Vol. 2, 372-420. Cambridge: Cambridge University Press. <u>https://doi.org/10.1017/CBO9780511619434.007</u>
- Lu, Tinan Qiao. 2008. A grammar of Maonan. Florida: Universal Publishers. https://www.academia.edu/36095089/A_Grammar_of_Maonan_Last accessed Jan 22, 2021.

Matisoff, James. 1969. "Verb concatenation in Lahu: the syntax and semantics of 'simple' juxtaposition". *Acta Linguistica Hafniensia* 12(1): 69-120. <u>https://doi.org/10.1080/03740463.</u> 1969.10415426

Matisoff, James. 1973. The grammar of Lahu. California: University of California Publications.

- Matisoff, James. 2003. Handbook of Proto-Tibeto-Burman: System and philosophy of Sino-Tibeto-Burman reconstruction. Berkeley: University of California Press. <u>https://escholarship.org/uc/item/19d79619</u> Last accessed Jan 22, 2021.
- Mazaudon, Martine. 2003. "From discourse to grammar in Tamang: topic, focus, intensifiers and subordination". In: Bradley, David; Lapolla, Randy; Michailovsky, Boyd; and Thurgood, Graham (eds.), Language variation: papers on variation and change in the Sinosphere and in the Indosphere in honour of James A. Matisoff, 145-158. Canberra: Australian National University. <u>https://doi.org/10.15144/PL-555.145</u>
- Mazaudon, Martine; and Michailovsky, Boyd. 1988. "Lost Syllables and Tone Contour in Dzongkha". In Bradley, David; Henderson, Eugenie J.A.; and Mazaudon, Martine (eds.), *Prosodic analysis and Asian linguistics: to honour R.K. Sprigg*, 115-136. Canberra: Dept. of Linguistics, Research School of Pacific Studies, Australian National University. <u>https://www. academia.edu/2090741/Lost syllables and tone contour in Dzongkha Bhutan</u> Last accessed Jan 22, 2021.
- Miestamo, Matti. 2000. "Towards a typology of standard negation". *Nordic Journal of Linguistics* 23(1): 65-88. <u>https://doi.org/10.1080/033258600750045787</u>
- Miestamo, Matti. 2003. *Clausal negation: atypological study*. Ph.D. diss, Helsinki: University of Helsinki.
- Miestamo, Matti. 2005. *Standard negation: the negation of declarative verbal main clauses in a typological perspective*. Berlin: De Gruyter Mouton [Empirical Approaches to Language Typology 31]. <u>https://doi.org/10.1515/9783110197631</u>
- Miestamo, Matti. 2013. "Symmetric and asymmetric standard negation". In: Dryer, Matthew S.; and Haspelmath, Martin (eds.), *The World Atlas of Language Structures Online*. Leipzig: Max Planck Institute for Evolutionary Anthropology. <u>http://wals.info/chapter/113</u> Last accessed Jan 22, 2021.
- Miestamo, Matti. 2016. Questionnaire for describing the negation systems of a language. <u>http://tulquest.huma-num.fr/fr/node/134</u> Last accessed Jan 22, 2021.
- Miller, Roy Andrew. 1966. "Early evidence for vowel harmony in Tibetan". *Language* 42(2): 252-277. <u>https://doi.org/10.2307/411688</u>
- Morey, Stephen. 2012. "The Singpho agentive functions and meanings". *Linguistics of the Tibeto-Burman Area* 35(1): 1-14. <u>https://www.academia.edu/29378318/The_Singpho_agentive_functions_and_meanings</u> Last accessed Jan 22, 2021.
- Mullard, Saul. 2011. *Opening the hidden land: state formation and the construction of Sikkimese history*. Leiden: Brill.
- Nedjalkov, Vladimir P. 1995. "Some typological parameters of converbs". In: Haspelmath, Martin; and König, Ekkehard (eds.), *Converbs in crosslinguistic perspective: structure and meaning of adverbial verb forms adverbial participles, gerunds*. Berlin: Mouton de Gruyter [Empirical Approaches to Language Typology 13].
- Nedyalkov, Vladimir P. 2007. "Overview of the research: definitions of terms, framework, and related issues". In: Nedyalkov, Vladimir P. (ed.), *Reciprocal constructions* Vol. 1, 3-114. Amsterdam: John Benjamins [Typological Studies in Language 71]. <u>https://doi.org/10.1075/tsl.71.06ned</u>

- Newman, Paul; and Ratliff, Martha (eds.). 2001. *Linguistic fieldwork*. Cambridge: Cambridge University Press. <u>https://doi.org/10.1017/CBO9780511810206</u>
- Noonan, Michael. 1997. "Versatile nominalizations". In Bybee, Joan; Haiman, John; and Thompson, Sandra (eds.), *Essays on language function and language type: Dedicated to T. Givón*, 372-395. Amsterdam: John Benjamins. <u>https://doi.org/10.1075/z.82.21noo</u>
- Noonan, Michael. 2006. "Grammar Writing for a Grammar-reading Audience". In Payne, T.; and Weber T. (eds.), *Perspectives on grammar writing*, 113-126. Amsterdam: John Benjamins. <u>https://doi.org/10.1075/bct.11.07noo</u>
- Noonan, Michael. 2007. "Complementation". In Shopen, Timothy (ed.), *Language typology and syntactic description* Vol. 2, 52-150. Cambridge: Cambridge University Press. <u>https://doi.org/10.1017/CBO9780511619434.002</u>
- Nuckolls, Janis B. 1999. "The case for sound symbolism". *Annual Review of Anthropology* 28: 225-252. <u>https://doi.org/10.1146/annurev.anthro.28.1.225</u>
- Oisel, Guillaume. 2013. Morphosyntaxe et semantique des auxiliaires et des connecteurs du *Tibetain Litteraire*. Ph.D. diss, Paris: Universite Sorbonne Nouvelle. <u>https://tel.archives-ouvertes.fr/tel-00914393/document</u> Last accessed Jan 22, 2021.
- Ozerov, Pavel; and Daudey, Henriëtta. 2017. "Copy-verb constructions in Tibeto-Burman and beyond". *Linguistic Typology* 21(1): 53-99. <u>https://doi.org/10.1515/lingty-2017-0002</u>
- Palmer, Frank. 2001. *Mood and modality*. Cambridge: Cambridge University Press. <u>https://doi.org/</u> <u>10.1017/CBO9781139167178</u>
- Payne, Tom. 1997. *Describing morphosyntax: a guide for field linguists*. Cambridge: Cambridge University Press. <u>https://doi.org/10.1017/CBO9780511805066</u>
- Payne, Thomas E. 2006. *Exploring language structure: A student's guide*. Cambridge: Cambridge University Press. <u>https://doi.org/10.1017/CBO9780511806483</u>
- Payne, Thomas. 2014. "Towards a balanced grammatical description". In Nakayama, Toshihide; and Rice, Keren (eds.), *The art and practice of grammar writing*, 91-108. Honolulu: University of Hawai'i Press. <u>http://hdl.handle.net/10125/4586</u> Last accessed Jan 22, 2021.
- Phenasa, Ugen Lopsang. 2013. *New light multilingual dictionary prajna* (English-Sherpa-Bhutia-Nepali-Tibetan dictionary). Gangtok: Author.
- Plaisier, Heleen. 2006. *A grammar of Lepcha*. Ph.D. diss, Leiden: Leiden University. <u>https://hdl.handle.net/1887/4379</u> Last accessed Jan 22, 2021.
- Post, Mark W. 2007. *A grammar of Galo*. Ph.D. diss, Melbourne: La Trobe University. <u>https://www.academia.edu/197280/A_Grammar_of_Galo</u> Last accessed Jan 22, 2021.
- Post, Mark W. 2008. Grammaticalization and the discourse distribution of serial verbs in Assamese. Paper presented at the 14th Annual Meeting of the Southeast Asian Linguistics Society 2004. Bangkok, Thailand.

https://www.academia.edu/197267/Grammaticalization_and_the_discourse_distribution_of_se_rial_verbs_in_Assamese_Last accessed Jan 22, 2021.

- Pustet, Regina. 2003. *Copulas: universals in the categorization of the lexicon*. Oxford: Oxford University Press. <u>https://doi.org/10.1075/sl.31.1.12mai</u>
- Richter, Eberhardt; and Mehnert, Dieter. 1978. "On the structure and function of aspiration in modern Tibetan". In: Richter, Eberhardt; and Taube, Manfred (eds.), *Asienwissenschaftliche Beiträge: Johannes Schubert in memoriam*, 123-130. Berlin: Akademie-Verlag.
- Sadler, Louisa; and Nordlinger, Rachel. 2006. "Case stacking in realizational morphology". *Linguistics* 44 (3): 459 - 487. <u>https://doi.org/10.1515/LING.2006.016</u>

- Sandberg, Graham. 1888. Manual of the Sikkim Bhutia Language or Denjong Ke. Calcutta: Oxford Mission Press. <u>https://archive.org/details/manualofsikkimbh00sanduoft</u> Last accessed Jan 22, 2021.
- Sandberg, Graham. 1894. *Handbook of colloquial Tibetan*. Calcutta: Thacker, Sprink and Co. <u>https://archive.org/details/HandbookOfColloquialTibetan/mode/2up</u> Last accessed Jan 22, 2021.
- Sandberg, Graham. 1895. revised edition of 1888. Manual of the Sikkim Bhutia language or Denjong Ke. Calcutta: Hare Press. <u>https://pahar.in/mountains/Books%20and%20Articles/</u> <u>Indian%20Subcontinent/1895%20Manual%20of%20the%20Sikkim%20Bhutia%20Language</u> <u>%20by%20Sandberg%20s.pdf</u> Last accessed Jan 23, 2021.
- Schachter, Paul; and Timothy Shopen. 2007. "Part-of-speech systems". In: Shopen, Timothy (ed.), Language typology and syntactic description 1: Clause structure, 1-60. Cambridge: Cambridge University Press. <u>https://doi.org/10.1017/CBO9780511619427.001</u>
- Schwieger, Peter. 1989. *Tibetisches Erzähltgut aus Brag-g.yab: Texte mit Übersetzungen, grammatischem Abriß und Glossar* (Beiträge zur tibetischen Erzählforschung herausgegeben von Dieter Schuh, Band 10). Bonn: VGH Wissenschaftsverlag.
- Shafer, Robert. 1974. Introduction to Sino-Tibetan. Wiesbaden: Otto Harrassowitz.
- Shagal, Ksenia. 2017. *Towards a typology of participles*. Ph.D. diss, Helsinki: University of Helsinki. <u>http://urn.fi/URN:ISBN:978-951-51-2958-1</u> Last accessed Jan 23, 2021.
- Shopen, Timothy. 2007. *Language typology and syntactic description* Vol. 1-3. Cambridge: Cambridge University Press.
- Shor, Thomas. 2017. A step away from paradise. USA: City Lion Press.
- Simons, Gary F. and Fennig, Charles D. (eds.). 2017. Ethnologue: Languages of the World. Twentieth edition. Dallas: SIL International. <u>http://www.ethnologue.com</u> Last accessed Aug 14, 2018.
- Sprigg, R. K. 1961. "Vowel harmony in Lhasa Tibetan: prosodic analysis applied to interrelated vocalic features of successive syllables". *Bulletin of the School of Oriental and African Studies* 24:1, 116-138. London: University of London. <u>https://doi.org/10.1017/S0041977X00140431</u>
- Sprigg, R. K. 1991. "The spelling-style pronunciation of written Tibetan, and the hazard of using citation forms in the phonological analysis of spoken Tibetan". *Linguistics of the Tibeto-Burman Area* 14(2): 93-131. <u>http://sealang.net/sala/archives/pdf8/sprigg1991spelling.pdf</u> Last accessed Jan 23, 2021.
- Starosta, Stanley. 1985. "Relator nouns as a source of case inflection". In: Acson, Venetta Z.; and Leed, Richard L. (eds.), For Gordon H. Fairbanks, 111-133. Honolulu: University of Hawai'i Press [Oceanic Linguistics Special Publication 20].
- Stassen, Leon. 2000. "AND-languages and WITH-languages". *Linguistic Typology* 4: 1-54. <u>https://doi.org/10.1515/lity.2000.4.1.1</u>
- Stassen, Leon. 2009. *Predicative possession*. Oxford: Oxford University Press [Oxford studies in typology and linguistic theory].
- Stassen, Leon. 2013. "Predicative adjectives". In: Dryer, Matthew S.; and Haspelmath, Martin (eds.), *The World Atlas of Language Structures Online*. Leipzig: Max Planck Institute for Evolutionary Anthropology. <u>http://wals.info/chapter/118</u> Last accessed Jan 23, 2021.
- Stassen, Leon. 2013b. "Comparative constructions". In: Dryer, Matthew S.; and Haspelmath, Martin (eds.), *The World Atlas of Language Structures Online*. Leipzig: Max Planck Institute for Evolutionary Anthropology. <u>http://wals.info/chapter/121</u> Last accessed Jan 23, 2021.

- Sun, Hongkai. 1999. "On the category of causative verbs in Tibeto-Burman languages". *Linguistics* of the Tibeto-Burman Area 22(1): 183-199. <u>http://sealang.net/sala/archives/pdf8/sun1999</u> category.pdf Last accessed Jan 23, 2021.
- Takchungdarpo, Pema Ringzing. 1987. বহ্রষার্বন্রা প্রান্ত্রান্ট্রা (Stories from Sikkim) (class 12 textbook). Department of Education, Government of Sikkim.
- Takeuchi, Tsuguhito; and Takahashi, Yoshiharu. 1995. "Split ergative patterns in transitive and intransitive sentences in Tibetan: a reconsideration". In: Nishi, Yoshio; Matisoff, James A.; and Nagano, Yasuhiko (eds.), *New horizons in Tibeto-Burman morphosyntax*, 277-288. Osaka: National Museum of Ethnology. <u>http://doi.org/10.15021/00002999</u> Last accessed Jan 23, 2021.
- Takeuchi, Tsuguhito. 2015. "The function of auxiliary verbs in Tibetan predicates and their historical development". *Revue d'Etudes Tibétaines* 31: 401-415. <u>http://himalaya.socanth.cam.</u> ac.uk/collections/journals/ret/pdf/ret_31_25.pdf Last accessed Jan 23, 2021.
- Teo, Amos. 2012. "Sumi agentive and topic markers: *no* and *ye*". *Linguistics of the Tibeto-Burman Area* 35(1): 49-74.
- Thompson, Sandra A.; Longacre, Robert E.; and Hwang, Shin Ja J. 2007. "Adverbial clauses". In: Timothy Shopen (ed.), *Language typology and syntactic description 2: Complex constructions*, 237-299. Cambridge: Cambridge University Press. <u>https://doi.org/10.1017/CBO97805116194</u> 34.005
- Tikkanen, Bertil. 1991. *Hindin kielioppi* (A grammar of Hindi). Finland: Finnish Oriental Society [Suomen Itämaisen Seuran Julkaisuja 23].
- Timberlake, Alan. 2007. "Aspect, tense, mood". In: Timothy Shopen (ed.), *Language typology and syntactic description 3: Grammatical categories and the lexicon*, 280-333. Cambridge: Cambridge University Press. <u>https://doi.org/10.1017/CBO9780511618437.005</u>
- Tournadre, Nicolas, 1991. "The rhetorical use of the Tibetan ergative". *Linguistics of the Tibeto-Burman Area* 14: 93-109. <u>http://sealang.net/sala/archives/pdf8/tournadre1991rhetorical.pdf</u> Last accessed Jan 23, 2021.
- Tournadre, Nicolas. 2008. "Arguments against the concept of 'conjunct'/'disjunct' in Tibetan". In: Huber, Brigitte; Volkart, Marianne; and Widmer, Paul (eds.), *Chomolangma, Demawend und Kasbek. Festschrift für Roland Bielmeier zu seinem 65. Geburtstag 1*, 281-308. Saale: International Institute for Tibetan and Buddhist Studies [Beiträge zur Zentralasienforschung 12].
- Tournadre, Nicolas. 2010. "The Classical Tibetan cases and their transcategoriality: from sacred grammar to modern linguistics". *Himalayan Linguistics* 9 (2): 87-125. <u>https://doi.org/10.5070/H99223480</u>
- Tournadre, Nicolas. 2014. "The Tibetic languages and their classification". In: Owen-Smith, Thomas; and Hill, Nathan W. (eds.), *Trans-Himalayan linguistics*, 105-130. Berlin/Boston: Walter de Gruyter [Trends in Linguistics, Studies and Monographs 266]. <u>https://doi.org/10. 1515/9783110310832.105</u>
- Tournadre, Nicolas. 2017. "A typological sketch of evidential/epistemic categories in the Tibetic languages". In: Gawne, Lauren; and Hill, Nathan (eds.), *Evidential systems of Tibetan languages*, 95-129. Berlin: Walter de Gruyter [Trends in Linguistics, Studies and Monographs 302]. <u>https://doi.org/10.1515/9783110473742-004</u>
- Tournadre, Nicolas; and Dorje, Sangda. 2003. *Manual of Standard Tibetan*. Ithaca, N.Y.: SnowLions Publications.

- Tournadre, Nicolas; and Jiatso, Konchok. 2001. "Final auxiliary verbs in literary Tibetan and in the dialects". *Linguistics of the Tibeto-Burman Area* 24(1): 49-111. <u>http://sealang.net/sala/archives/pdf8/tournadre2001final.pdf</u> Last accessed Jan 23, 2021.
- Tournadre, Nicolas; and LaPolla, Randy. 2014. "Towards a new approach to evidentiality: Issues and directions for research". *Linguistics of the Tibeto-Burman Area* 37(2): 240-263. <u>https://doi.org/10.1075/ltba.37.2.04tou</u>
- Tshering, Norden; and Takchungdarpo, Pema Rinzing. 2001. *Bhutia-English dictionary*. Gangtok: Kwality stores.
- Tsichudarpo, Bhaichung. 1996 [2003]. रे.चे. (Hope). Unpublished manuscript.
- Tsichudarpo, Bhaichung. 1997 [2003]. 🚓 🛪 🛪 (Superstition). Unpublished manuscript.
- Tsichudarpo, Bhaicung. 2008. ਕਗ੍ਰਾਂ ਸੁੱਤ੍ਰ' ਗ੍ਰਤਕਾ ਬੁੱਤ੍' ਕਲ਼ਾ ਬੁੱਤ੍' (Plays of three headings, i.e. three plays). Unpublished manuscript.
- Tsichudarpa, Bhaicung. 2018. "Bhutia". In: Pandey, Balaram (ed.), *The languages of Sikkim*, 46-58. Himayatnagar, Hyderabad: Orient Black Swan [People's Linguistic Survey of India 26, part 2].
- Turin, Mark. 2005. "Language endangerment and linguistic rights in the Himalayas: a case study from Nepal". *Mountain Research and Development*, 25(1): 4-9. <u>https://doi.org/10.1659/0276-4741(2005)025[0004:LEALRI]2.0.CO;2</u>
- Turin, Mark. 2011. "Results from the linguistic survey of Sikkim: mother tongues in education". In: Balikci-Denjongpa, Anna; and McKay, Alex (eds.), *Buddhist Himalaya: studies in religion, history and culture* Vol. 2 (Proceedings of the golden jubilee conference of the Namgyal Institute of Tibetology, Gangtok 2008), 127-142. Gangtok: Namgyal Institute of Tibetology. <u>http://digitalhimalaya.com/projectteam/turin/downloads/Sikkim_Tibetology.pdf</u> Last accessed Jan 23, 2021.
- Turin, Mark. 2014. "Mother tongues and language competence: The shifting politics of linguistic belonging in the Himalayas". In: Toffin, Gérard; and Pfaff-Czarnecka, Joanna (eds.), *Facing globalization in the Himalayas: Belonging and the politics of the self*, 372-396. New Delhi: Sage [Governance, Conflict and Civic Action 5]. <u>https://www.researchgate.net/publication/342788730 Mother tongues and language competence The shifting politics of linguistic belonging in the Himalayas</u> Last accessed Jan 23, 2021.
- Turner, Ralph Lilley. 1962-1966. *A comparative dictionary of Indo-Aryan languages* (Includes three supplements, published 1969-1985). London: Oxford University Press. <u>https://dsal.uchicago.edu/dictionaries/soas/</u> Last accessed Jan 23, 2021.
- Walsh E. H. C. 1905. A vocabulary of the Tromowa dialect of Tibetan spoken in the Chumbi valley (so far as it differs from standard Tibetan): together with a corresponding vocabulary of Sikhimese and of Central (standard) Tibetan (The corresponding Sikhimese words supplied by Sub-inspector S. W. Laden La, Bengal police). Calcutta: Bengal secretariat Book depot. https://digital.soas.ac.uk/AA00000534/00001 Last accessed Jan 23, 2021.
- Vanhove, Martine. 2016. "The manner converb in Beja (Cushitic) and its refinitization". In: Chamoreau, Claudine; and Estrada-Fernández, Zarina (eds.), *Finiteness and nominalization*, 323-344. Amsterdam/Philadelphia: John Benjamins [Typological Studies in Language 113]. <u>https://doi.org/10.1075/tsl.113.13van</u>
- Valin, Robert van; and LaPolla, Randy. 1997. *Syntax: structure, meaning and function*. Cambridge: Cambridge University Press. <u>https://doi.org/10.1017/CBO9781139166799</u>
- Watters, David. 2002. A grammar of Kham. Cambridge: Cambridge University Press. <u>https://doi.org/10.1017/CBO9780511486883</u>
- Watters, Stephen. 2002. "The sounds and tones of five Tibetan languages of the Himalayan region". *Linguistics of the Tibeto-Burman Area* 25(1): 1-65. <u>http://sealang.net/sala/archives/pdf8/</u> <u>watters2002sounds.pdf</u> Last accessed Jan 23, 2021.
- Watters, Stephen. 2003. "An acoustic look at pitch in Lhomi". In: Kansakar, Tej R.; and Turin, Mark (eds.), *Themes in Himalayan language and linguistics*, 249-264. Kathmandu: South Asia Institute (SAI) Heidelberg and Tribhuvan University. <u>https://www.academia.edu/14152460/</u> <u>An acoustic look at pitch in Lhomi</u> Last accessed Jan 23, 2021..
- Watters, Stephen. 2018. A grammar of Dzongkha (dzo): phonology, words, and simple clauses. Ph.D. diss, Houston: Rice University. <u>https://hdl.handle.net/1911/103233</u> Last accessed Jan 23, 2021.
- Watters, Stephen. 2018. "Referential tracking in languages of the Himalaya: emerging switch reference?". Unpublished manuscript. <u>https://www.academia.edu/37645709/Referential tracking in languages of the Himalaya emerging switch reference</u> Last accessed Jan 23, 2021.
- Widmer, Manuel. 2017. "The evolution of egophoricity and evidentiality in the Himalayas: the case of Bunan". *Journal of Historical Linguistics*, special issue 7(1-2): 246-275. https://doi.org/10.1075/jhl.7.1-2.10wid
- Wilde, Christopher. 2001. Preliminary phonological analysis of the Limi dialect of Humla Bhotia. MA thesis, Helsinki: University of Helsinki. <u>http://urn.fi/URN:NBN:fi-fe20071359</u> Last accessed Jan 23, 2021.
- Willis, Christina. 2011. "Optional case-marking in Darma (Tibeto-Burman)". *Linguistics of the Tibeto-Burman Area* 34(2): 101-132. <u>http://sealang.net/archives/ltba/pdf/LTBA-34.2.9.pdf</u> Last accessed Jan 23, 2021.
- Vesalainen, Olavi; and Vesalainen, Marja. 1976. *Lhomi phonemic summary*. Kathmandu: Tribhuvan University Press.
- Vesalainen, Olavi. 2016. A grammar sketch of Lhomi. SIL International [Language and Culture Documentation and Description 34]. <u>https://www.sil.org/resources/archives/68227</u> Last accessed Jan 23, 2021.
- Veselinova, Ljuba. 2014. "The negative existential cycle revisited". *Linguistics* 52(6): 1327-1389. <u>https://doi.org/10.1515/ling-2014-0021</u> Last accessed Jan 23, 2021.
- Villiers, Jill de; Garfield, Jay; Gernet-Girard, Harper; Roeper, Tom; and Speas, Margaret. 2009.
 "Evidentials in Tibetan: Acquisition, semantics, and cognitive development". In: Fitneva, Stanka; and Matsui, Tomoko (eds.), *Evidentiality: A window into language and cognitive development*, 29-47. San Francisco: Jossey-Bass [New Directions for Child and Adolescent Development 125]. <u>https://doi.org/10.1002/cd.248</u>
- Yliniemi, Juha. 2005. *Preliminary phonological analysis of denjongka of Sikkim*. MA thesis, Helsinki: University of Helsinki. <u>http://urn.fi/URN:NBN:fi-fe20071259</u> Last accessed Jan 23, 2021.
- Yliniemi, Juha. 2014. Front unrounded vowels in Denjongke (Sikkimese Bhutia): an acoustic study with a historical note. Presentation at the 8th International Conference of the North East Indian Linguistic Society, Guwahati, Jan 31-Feb 2, 2014. <u>http://www.academia.edu/23501747/Front unrounded vowels in Denjongke Sikkimese Bhutia an acoustic study with a historical n ote</u> Last accessed Jan 23, 2021.
- Yliniemi, Juha. 2016a. "Attention marker =*co* in Denjongke (Sikkimese Bhutia)". *Linguistics of the Tibeto-Burman Area* 39(1): 106-161. <u>https://doi.org/10.1075/ltba.39.1.05yli</u>
- Yliniemi, Juha. 2016b. "Nepali attention marker *ni*". *Himalayan Linguistics* 15(2): 50-87. <u>https://doi.org/10.5070/H915231077</u>

- Yliniemi, Juha. 2017. "Copulas in Denjongke or Sikkimese Bhutia". In: Gawne, Lauren; and Hill, Nathan W. (eds.), *Evidential systems of Tibetan languages*, 297-349. Berlin: Walter de Gruyter. https://doi.org/10.1515/9783110473742-010
- Yliniemi, Juha. 2018. "The life of a semi-urban Lhopo/Sikkimese Bhutia family". *Himalayan Discoveries* 2.1 (Publication of the Society of Himalayan Studies in Finland): 95-109. <u>https://shsfinland.files.wordpress.com/2018/11/yliniemi1.pdf</u> Last accessed Jan 23, 2021.
- Yliniemi, Juha. 2020. "Similarity of mirative and contrastive focus: three parameters for describing attention markers". Unpublished manuscript.
- Yongda, Sonam. 2016. Video-recorded interview in Denjongke (Interviewer: Karma Namgyal Denjongpo).
- Yukawa, Yasutoshi (湯川恭敏).1975. "チベット語の述語 Chibettogo no jutsugo" (The predicates of Tibetan). アジア・アフリカ文法研究 Ajia, Afurika Bunpō Kenkyū / Asian and African Linguistics 4: 1-14. Tokyo: ILCAA.
- Yukawa, Yasutoshi. 2017. "Lhasa Tibetan predicates (translation)". In: Gawne, Lauren; and Hill, Nathan W. (eds.), *Evidential systems of Tibetan languages* (Translation of Yukawa 1975 by Nathan W. Hill), 187-224. Berlin: Walter de Gruyter [Trends in Linguistics, Studies and Monographs Vol. 302]. <u>https://doi.org/10.1515/9783110473742-007</u>
- Zeisler, Bettina. 2000. "Narrative conventions in Tibetan languages: the issue of mirativity". *Linguistics of the Tibeto-Burman Area* 23(2): 39-77. <u>http://sealang.net/sala/archives/pdf8/</u> <u>zeisler2000narrative.pdf</u> Last accessed Jan 23, 2021.
- Zeisler, Bettina. 2011. "Kenhat, the dialects of Upper Ladakh and Zanskar". In: Turin, Mark; and Zeisler, Bettina (eds.), *Himalayan languages and linguistics* (Studies in phonology, semantics, morphology and syntax), 235-319. Leiden: Brill. <u>https://doi.org/10.1163/ej.9789004194489.i-322.69</u>
- Zemp, Marius. 2018. *A grammar of Purik Tibetan*. Leiden: Brill [Languages of the Greater Himalayan Region 21].